Musqueam Reference Grammar

FIRST NATIONS LANGUAGES

The First Nations languages of the world, many of which are renowned for the complexity and richness of their linguistic structure, embody the cumulative cultural knowledge of Aboriginal peoples. This vital linguistic heritage is currently under severe threat of extinction. This new series is dedicated to the linguistic study of these languages.

Patricia A. Shaw, a member of the Department of Linguistics at the University of British Columbia and coordinator of the First Nations Languages Program, is general editor of the series.

The first two volumes in the series are:

The Lillooet Language: Phonology, Morphology, Syntax Jan van Eijk Musqueam Reference Grammar Wayne Suttles



First Nations Languages

Musqueam Reference Grammar

Wayne Suttles



UBCPress Vancouver Toronto

© UBC Press 2004

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, without prior written permission of the publisher, or, in Canada, in the case of photocopying or other reprographic copying, a licence from Access Copyright (Canadian Copyright Licensing Agency), www.accesscopyright.ca.

 $10 \ 09 \ 08 \ 07 \ 06 \ 05 \ 04 \qquad 5 \ 4 \ 3 \ 2 \ 1$

Printed in Canada on acid-free paper

National Library of Canada Cataloguing in Publication

Suttles, Wayne, 1918-Musqueam reference grammar / Wayne Suttles

(First Nations languages, ISSN 1206-9531) Includes bibliographical references and index. ISBN 0-7748-1002-5

1. Stalo language - Dialects - Grammar. I. Title. II. Series.

PM2381.S81S97 2004 497'.94 C2004-900532-4

Canadä

UBC Press gratefully acknowledges the financial support for our publishing program of the Government of Canada through the Book Publishing Industry Development Program (BPIDP), and of the Canada Council for the Arts, and the British Columbia Arts Council.

This book has been published with the help of a grant from the Canadian Federation for the Humanities and Social Sciences, using funds provided by the Social Sciences and Humanities Research Council of Canada. Financial support has also been received from the UBC Killam General Fund.

UBC Press The University of British Columbia 2029 West Mall Vancouver, BC V6T 1Z2 604-822-5959 / Fax: 604-822-6083 E-mail: info@ubcpress.ca www.ubcpress.ca To the memory of

Andrew and Christine Charles James Point Della Kew Arnold Guerin

Contents

Preface and Acknowledgments xx

3

Introduction xxiii

1 Phonology

- 1.1 Consonants
 - 1.1.1 Obstruents 4
 - 1.1.2 Resonants 7
- 1.2 Vowels 9
 - 1.2.1 Vowel Length 13
- 1.3 Stress and Pitch 13
- 1.4 Phonotactics 14
- 1.5 Morphophonemics 15
 - 1.5.1 Losses in Rapid Speech 15
 - 1.5.2 Assimilation of Schwa 15
 - 1.5.3 Metathesis of Non-Labial Resonant and Schwa 16
 - 1.5.4 Glottalized Resonant Disengagement 17
 - 1.5.5 Coalescence of Glottal Stop and Resonant 17
 - 1.5.6 Loss of the First of Two Identical Resonants 17
 - 1.5.7 Emergent /θ/ 18
 - 1.5.8 Intrusive /h/ 18
 - 1.5.9 Alternation of /n/ and /l/ 19
 - 1.5.10 Alternation of /c/ and /k w/ and of /x/ and /x w/ 19
 - 1.5.11 Vowel Gradation 20
 - 1.5.12 Vowel Mutation 20
- 1.6 Special Forms of Speech 22

2 Synopsis of Morphology

- 2.1 Words 23
- 2.2 Roots and Their Internal Modification 23
- 2.3 Affixes 25

- 2.4 The Person System 27
- 2.5 The Ordering of Affixes 27
- 2.6 Ambivalent Affixes 28
- 2.7 Particles 29

3 Syntax 1: Simple Sentences

- 3.1 The Minimal Sentence 31
 - 3.1.1 Predicate Heads 33
 - 3.1.2 The Second-Position Predicate Particles 33
- 3.2 Expanded Verbal Predicates 34
 - 3.2.1 Auxiliary Verbs 34
 - 3.2.2 Adverbs 38
 - 3.2.3 Verb Complements 39
 - 3.2.4 Hortative Forms 40
- 3.3 Types of Verbal Predicates 41
 - 3.3.1 Intransitive Predicates 41
 - 3.3.2 Active Transitive Predicates 42
 - 3.3.3 Passive Predicates 43
- 3.4 Nominal Adjuncts 44
 - 3.4.1 With Intransitive Predicates 45
 - 3.4.2 With Active Transitive Predicates 48
 - 3.4.3 With Passive Predicates 51
- 3.5 Verbal Adjuncts 53
- 3.6 Adverbial Adjuncts 57
- 3.7 Adjectival and Nominal Predicates 58
 - 3.7.1 Adjectival Predicates 59
 - 3.7.2 Nominal Predicates 60
 - 3.7.3 Nominal Predicates with Adjectival Modifiers 61
- 3.8 Expanded Nominal Adjuncts in Simple Sentences 63
 - 3.8.1 Possession 63
 - 3.8.2 Particles 64
 - 3.8.3 Adjectival Modifiers 64
 - 3.8.3.1 Adjectives 64
 - 3.8.3.2 Numerals 66
 - 3.8.3.3 Adverbs 67
 - 3.8.3.4 Adjective-like Forms 68
 - 3.8.3.5 Verb Forms as Modifiers 69
 - 3.8.3.6 Noun Premodifiers 70
 - 3.8.4 Coordination 70
 - 3.8.5 Apposition 71

4 Syntax 2: Complex Sentences

- 4.1 Relative Clauses 74
 - 4.1.1 Forms of Verbal Relative Clauses 75

- 4.1.1.1 With Extracted Subjects and Objects 75
- 4.1.1.2 With Extracted Oblique Adjuncts 76
- 4.1.1.3 With Extracted Possessors 78
- 4.1.2 Uses of Verbal Relative Clauses 79
- 4.1.3 Nominal Relative Clauses 81
- 4.1.4 Adjectival Relative Clauses 83
- 4.1.5 Pseudo-Cleft and Cleft Sentences 84
 - 4.1.5.1 Pseudo-Cleft Sentences 84
 - 4.1.5.1.1 With Nominal Predicates 84
 - 4.1.5.1.2 With Pronominal Predicates 88
 - 4.1.5.1.3 With Interrogative Predicates 89
 - 4.1.5.2 Cleft Sentences 89
- 4.2 Subordinate Clauses 92
 - 4.2.1 Conditional Clauses 93
 - 4.2.2 Indirect Commands 95
 - 4.2.3 Indirect Questions 95
 - 4.2.4 Complements of Main Clauses Referring to the Future 96
 - 4.2.5 Interrogative Complements with ² ol- 97
- 4.3 Nominalized Clauses 98
 - 4.3.1 Within Predicates 99
 - 4.3.2 As Nominal Adjuncts 100
 - 4.3.3 Clauses Nominalized with δx^{w} 104
 - 4.3.4 Nominalized Narrative Sentences 105

5 Syntax 3: Compound Sentences

- 5.1 Coordinate Compound Sentences 113
- 5.2 Qualifying Compound Sentences 114
 - 5.2.1 With Quantifying First Clauses 115
 - 5.2.2 With Interrogative First Clauses 115
 - 5.2.3 With Conditional First Clauses 116
 - 5.2.4 With Modal First Clauses 116

6 Syntax 4: Negation

- 6.1 Negation with ⁹ówə 118
 - 6.1.1 Subordinate Clauses with Intransitive and Active Transitive Predicates 118
 - 6.1.2 Subordinate Clauses with Passive Predicates 121
 - 6.1.3 Subordinate Clauses with Non-Verbal Predicates 122
 - 6.1.4 Nominalized Clauses 124
 - 6.1.5 With Relative Clauses 125
 - 6.1.6 Ellipsis with the Negative 126
- 6.2 Negation with $x^w \partial w \dot{e}$ 126
- 6.3 Negation with ²ówəte² 127

- 6.4 Negated Predicates in Complex and Compound Sentences 130
 - 6.4.1 In Relative Clauses 131
 - 6.4.2 In Subordinate Clauses 131
 - 6.4.2.1 As Conditional Clauses 131
 - 6.4.2.2 Expressing Necessity 132
 - 6.4.2.3 In the Sense 'Or' 132
 - 6.4.3 In Nominalized Clauses 133

7 Morphology of the Root 1: The Verb

- 7.1 The Verb Root 135
 - 7.1.1 Canonical Shapes 135
 - 7.1.2 Inherent Voice 136
 - 7.1.3 Internal Modification of the Root 137
- 7.2 The Progressive Aspect 138
 - 7.2.1 Progressives of CAC Roots 139
 - 7.2.1.1 Progressives of Bare CAC Roots 139
 - 7.2.1.2 Progressives of CAC Roots with Affixes 141
 - 7.2.1.3 Progressives of CÁCVC Roots 146
 - 7.2.2 Progressives of HAC Roots 1477.2.2.1 Progressives of HAC(C) PC Roots 149
 - 7.2.3 Progressives of TAH Roots 149
 - 7.2.3.1 Progressives of TAHT Roots 152
 - 7.2.4 Progressives of RAH Roots 1527.2.4.1 Progressives of RA⁹CC and RA⁹C Roots 153
 - 7.2.5 Progressives of HAH Roots 153
 - 7.2.6 Progressives of TəT Roots 153
 7.2.6.1 Progressives of TəTəC Roots 157
 7.2.6.2 Progressives of TTVC Roots 157
 - 7.2.7 Progressives of TəR Roots 158 7.2.7.1 Progressives of TəRəC Roots 160
 - 7.2.8 Progressives of R₂C Roots 160
 - 7.2.9 Progressives of H₂C Roots 162 7.2.9.1 Progressives of H₂C(C)V(C) Roots 162
 - 7.2.10 Progressives of CaCT Roots 163
 - 7.2.11 Unique Progressives 165
- 7.3 Plurals 165
 - 7.3.1 CAC Plural Perfectives 166
 - 7.3.1.1 CAC Plural Progressives 166
 - 7.3.2 HAC and HACaC Plurals 168
 - 7.3.3 CAH Plurals 168
 - 7.3.4 CaC Plural Perfectives 169 7.3.4.1 CaC Plural Progressives 170

- 7.3.5 CaCT Plurals 171
- 7.3.6 HəC(C)V(C) Plurals 171
- 7.4 Diminutives 172
 - 7.4.1 Diminutive Plurals 173
- 7.5 The Durative Aspect 174
 - 7.5.1 Plural Duratives 177
 - 7.5.2 Diminutive Duratives 178
- 7.6 The Dispositional-Iterative Aspect 181
 - 7.6.1 Progressives of Dispositional-Iterative Forms 181
 - 7.6.2 Plurals of Dispositional-Iterative Forms 181
 - 7.6.3 Diminutives of Dispositional-Iterative Forms 181
- 7.7 Resultative Forms 182
 - 7.7.1 Resultatives of CAC and CaCC Roots 182
 - 7.7.2 Resultatives of CaC Roots 183
 - 7.7.3 Resultatives of Other Tri-Consonantal Roots 184
 - 7.7.4 Resultatives of Irregular Roots 185
 - 7.7.5 Resultatives with Suffixes 185
 - 7.7.6 Anomalous Resultative Forms 187
 - 7.7.7 Plurals of Resultatives 188
 - 7.7.8 Diminutives of Resultatives 189
 - 7.7.9 Diminutive Plurals of Resultatives 189
- 7.8 Paradigms 190
 - 7.8.1 Problematic Paradigms 198

8 Morphology of the Root 2: The Noun

- 8.1 Types 199
- 8.2 Shapes of Noun Roots 200
- 8.3 Plurals 204
 - 8.3.1 With Infixed -l- 205
 - 8.3.2 With CV Reduplication 207
 - 8.3.3 With CVC Reduplication 209
 - 8.3.4 With Vowel Change Only 210
 - 8.3.5 Suppletion 211
- 8.4 Diminutives 211
 - 8.4.1 With C₁ Reduplication and No Vowel Change 212
 - 8.4.2 With C₁ Reduplication and Vowel Change 213
 - 8.4.3 With CV Reduplication 214
- 8.5 Diminutive Plurals 214
 - 8.5.1 With Infixed -l- 214
 - 8.5.2 With Vowel Change 215
- 8.6 Problematic Diminutives 215
- 8.7 Resultative Forms 217

- 8.8 The Inflection of Verbs Derived from Nouns 218
- 8.9 Verbs that Act as Nouns 218

9 Morphology of the Root 3: Adjectives and Adjective-like Words

- 9.1 Shapes of Adjective Roots 219
- 9.2 Plurals 221
- 9.3 Diminutives 222
- 9.4 Diminutive Plurals 223
- 9.5 Verbs Formed from Adjective Roots 223
- 9.6 Complex Adjectives 223
- 9.7 Adjective-like Words 224

10 Non-Personal Affixes 1: Voice

10.1 Transitive Suffixes 226 10.1.1 -t 'transitive' (TR) 226 10.1.2 $-n \partial x^w$ 'limited control transitive' (LCTR) 227 10.1.3 -x 'transitive' (TR) 227 10.2 Intransitive Suffixes 229 10.2.1 -*am* 'intransitive' (INTR) 229 10.2.2 -éls 'activity' (ACT) 232 10.2.3 - $\delta n \partial q$ 'someone else, another person' 233 $10.2.4 - \partial t \partial n \sim -\partial t i n$ 'someone, another' 234 10.3 The Causative Suffix -stax^w, 234 10.3.1 In the sense 'cause to be' 234 10.3.2 In the sense 'cause to do' 236 10.4 Applicatives 237 10.4.1 -stax^{w_2} 'comitative' (COM) 238 10.4.2 -nəs 'goal' (GOAL) 239 10.4.3 -as 'recipient' (RECIP) 240 10.4.4 -tc- 'benefactive' (BEN) 241 10.4.5 -*mət* 'concern' (CON) 242 10.5 Reflexives 243 10.5.1 -θət 'oneself' 244 10.5.2 -námət 'oneself (limited control)' 245 10.5.3 -stanámat 'oneself (causative)' 246 10.6 The Reciprocal $-t\partial \vec{l}$ 'each other' 246 10.7 The Permissive -s 247 10.8 The Subordinate Passive -at 249

11 Non-Personal Affixes 2: Aspectual and Modal Affixes

- 11.1 Aspectual Prefixes 250
 - 11.1.1 *wa*-₂ 'established' (EST) 251
 - 11.1.2 wət- 'already' 254

- 11.1.3 tow- 'like, somewhat, a bit, sort of (like)' 255
- 11.1.4 *x*^w*ən* 'still, soon' 257
- 11.1.5 *yə-i* 'along' 258
- 11.1.6 $y \partial_{-2} (\sim i)$ 'first, before doing anything else, above all (first)' 260
- 11.1.7 $n \partial x^w s$ (~ $x^w s$ -) 'habitually' (HAB) 260
- 11.2 Modal Suffixes 261
 - 11.2.1 -*ilmən* 'want to, intend to, seem about to' 261
 - 11.2.2 $-i \cdot t \sim -i \partial t (//-\partial y \partial t//)$ 'arrange to, seek to (arrange)' 262

12 Non-Personal Affixes 3: Derivational Affixes

12.1 Purely Grammatical Derivational Affixes 264 1211 s_{-1} '(word-) nominalizer' (NOM) 264 12.1.2 s_{-2} 'resultative' (RES) 265 12.1.3 x^{w-1} 'oblique relater' (OBREL) 266 12.1.4 $\check{s}x^{w}$ - 'oblique (word) nominalizer' (OBLNOM) 266 12.1.5 $-a^{2}l$ 'attributive' (ATT) 268 12.1.6 -t 'stative' (STAT) 269 12.2 Verbalizing Prefixes 269 12.2.1 c- 'get, have, make, do' 269 12.2.2 $x^{w} \partial - x^{w} i$ - 'become' 273 12.2.3 tx^{w} -, 'buy' 274 12.2.4 *t*-, 'partake, experience' 274 12.2.5 $\vec{\lambda}$ - 'go to' 275 12.2.6 cl- 'die of' 275 12.2.7 x^{w-2} 'move toward' 275 12.3 Verbal Suffixes 275 12.3.1 $-\partial l \sim -il$ 'move toward, turn to, become' 275 12.3.2 -im 'die from' 276 12.3.3 $-\dot{a}\cdot t$ 'travel by' 277 12.3.4 $-\acute{a}m\partial x \sim -\acute{a}m\partial x$ 'look, appear' 277 12.4 Miscellaneous Lexical Prefixes 278 12.4.1 m_{∂} - 'come' 278 12.4.2 tom- 'time of, season for' 278 12.4.3 *tən-* 'from' 279 12.4.4 $tx^{w_{-2}}$ (something) belonging to' 279 12.4.5 tx^{w} -, 'remaining, only' 280 12.4.6 *t*-, 'portion' 280 12.4.7 *čt*- 'fellow, co-' 280 12.4.8 x^{w-3} 'inward' 280 12.4.9 $w_{\partial-3}$ 'suddenly, with a burst' 283 12.4.10 $h \partial n - \sim h \partial m - \sim \gamma \partial m - \sim h \partial n - \circ come'$ 283

13 Non-Personal Affixes 4: Lexical Suffixes

- 13.1 Forms 285
- 13.2 Meanings 286
- 13.3 Lexical Suffixes and Related Words 287
- 13.4 Relations between Suffix and Root 288
 - 13.4.1 With Verbal Roots 289
 - 13.4.1.1 Root and Suffix Related as Verb and Patient, Locus, or Instrument 289
 - 13.4.1.2 Root and Suffix Related as Modifier and Head 290
 - 13.4.2 With Adjective Roots 291
 - 13.4.3 With Noun Roots 292
 - 13.4.3.1 Related as Head and Modifier 293
 - 13.4.3.2 Related as Modifier and Head 293
- 13.5 Compounding of Suffixes 294
- 13.6 The Inventory 294
- 13.7 Possible Lexical Suffixes 315
- 13.8 Suffixes in Formal Personal Names 318

14 Person Markers

- 14.1 Person and Number 320
- 14.2 Forms 321
 - 14.2.1 The First- and Second-Person Subject Particles 322
 - 14.2.2 The Third-Person Transitive Subject Marker 323
 - 14.2.3 The Subordinate Subject Person Markers 323
 - 14.2.4 The Possessive Affixes 324
 - 14.2.5 The Object Person Markers 327
 - 14.2.6 The Passive Person Markers 329
 - 14.2.7 The Personal Words 331
 - 14.2.7.1 As Predicate Heads 331
 - 14.2.7.2 As Adjuncts 333
 - 14.2.8 The Possessive Words 336
 - 14.2.9 The Third-Person Plural Particle 336

15 The Demonstrative System

- 15.1 The Articles 339
 - 15.1.1 Gender 340
 - 15.1.2 Position 341
 - 15.1.3 The Contrast with English 347
 - 15.1.4 The Oblique Article 348
- 15.2 The Demonstratives 350
 - 15.2.1 The Simple Demonstratives 351

- 15.2.2 The Compound Demonstratives 351
 - 15.2.2.1 The Locative Demonstratives 351
 - 15.2.2.2 The Third-Person Demonstratives 353
- 15.2.3 Use as Determiners 353
- 15.2.4 Use as Adjuncts 358
 - 15.2.4.1 Pronominal Uses 358
 - 15.2.4.2 Locative Uses 362
- 15.3 The Demonstrative Auxiliaries 364

16 Predicate Particles and Tags

- 16.1 Clause-Initial Particles 367
- 16.2 Second-Position Predicate Particles 367
 - 16.2.1 -əł 'past' 368
 - 16.2.2 [?] 'interrogative' (ROG) 372
 - 16.2.3 $y \partial x^w$ 'inferential' (INF) 373
 - 16.2.4 *co* 'quotative' (QUOT) 373
 - 16.2.5 ⁹e[•] 'plural imperative' (PLPER) 374
 - 16.2.6 *ce*⁹ 'future' (FUT) 375
 - 16.2.7 *to ~ te* 'imperative-optative' (PER) 377
 - 16.2.8 [?]al 'just' 378
 - 16.2.9 'ewət 'reassuring' (REAS) 381
 - 16.2.10 *yet* 'expectable' (EXP) 381
 - 16.2.11 *m* (sometimes *m*) 'certain' (CERT) 382
 - 16.2.12 *ctwa*² 'speculative' (SPEC) 384
 - 16.2.13 $k^{w} \partial \sim k^{w} e$ 'then' 385
 - 16.2.14 *q*́ә 'emphatic' (ЕМРН) 387
 - 16.2.15 θəł 'adversative' (ADV) 387
 - 16.2.16 $k^{w}t$ 'by now, by then'(?) 389
 - 16.2.17 wa? 'presumptive' (PRES) 390
 - 16.2.18 ²á²a 'emphatic interrogative' (ROG!) 391
- 16.3 Sentence-Final Tags 393
 - 16.3.1 $^{9}a \sim ^{9}e$ 'question tag' 393
 - 16.3.2 yé 'taunting echo' 393

17 Interrogative Words

- 17.1 wét, təwét 'who?' wé²əlt 'who (plural)?' 396
- 17.2 tx^wwét, təwét 'whose?' 398
- 17.3 stém 'what?' stélam 'what (plural)?' 398
- 17.4 témc (pl. téləmc) 'what people' 402
- 17.5 təmtém 'when?' 402
- 17.6 ctétəm 'be doing what?' 404
- 17.7 $c \delta s t \partial x^w$ 'do what with it? do what about it?' 404
- 17.8 ctámət 'do what with oneself?' 405

- 17.9 $x^{w}c\ell i$ 'go where?' 406
- 17.10 $x^w c \delta l \theta \delta t$ 'go where?' 407
- 17.11 $x^w c \ell l st \partial x^w$ 'put where?' 407
- 17.12 $sc\acute{e}k v a l$ 'how? how much? who?' 407
- 17.13 $sc \partial k^{w} \partial l i m$ 'how? how constructed? how intended?' 409
- 17.14 $x^{w} \partial^{2} i \cdot nt$ 'say what?' 410
- 17.15 $sx^w \partial^2 i nt$ 'what meaning?' 410
- 17.16 $x^w \partial^2 i \cdot ntst \partial x^w$ 'say what to him? say what to him about something?' 411
- 17.17 ?ánaca, ?ánca 'where?' 411
- 17.18 $? \delta n \partial c \partial s t \partial x^w$ 'put it where?' 413
- 17.19 nəcím 'why?' 413
- 17.20 k^{win} 'how many?' 414

18 Adverbs and Adverbial and Modal Words and Phrases

- 18.1 Adverbial and Modal Functions 416
- 18.2 Adverbs 417
- 18.3 Adverbial and Modal Predicates 419
- 18.4 Catalogue 422
 - 18.4.1 $y \dot{a} \theta$ 'always, frequently' 424
 - 18.4.2 $t \neq \dot{q}$ 'usually, generally, habitually' 424
 - 18.4.3 $\dot{c} \partial x^{w} l \dot{e}^{\gamma}$ 'sometimes, there are times when' 425
 - 18.4.4 *néč* 'at other times, sometimes' 425
 - 18.4.5 $n\partial \dot{c} ex^w$ 'once, one time' and so on 426
 - 18.4.6 $q \check{x} \acute{e} t$ 'often, many times' 427
 - 18.4.7 $\gamma \delta w \delta sc \ell k^{w} \delta \delta s$ 'never' 427
 - 18.4.8 $m \delta \vec{k}^w s w \delta y \delta l$ 'every day' and so on 427
 - 18.4.9 $hi\theta$ 'last long, be a long time' 428
 - 18.4.10 *náča² swéyəl* 'one day' and so on 430
 - 18.4.11 $\dot{t}\dot{a}t$ 'long before, from long ago, earlier' 431
 - 18.4.12 $q e^{i}$ is 'recently, just a short time ago, formerly' 432
 - 18.4.13 yét 'only now, only then' 433
 - 18.4.14 *wéyəl* 'tomorrow' and so on 434
 - 18.4.15 λe , λe^{2} 'again, also, too' 435
 - 18.4.16 *qəlét* 'do again, again,' or with a negative '(no) more, (no) longer' 436
 - 18.4.17 *ha*? 'if, when' 436
 - 18.4.18 $m \delta \vec{k}^w$ 'whenever' 437
 - 18.4.19 $x^{w} \partial n \dot{a}^{\gamma}$ 'first, when first, as soon as' 437
 - 18.4.20 yawéń 'before' 439
 - 18.4.21 *ti²á²aq^wt* 'after' 440
 - 18.4.22 $t \dot{a} \dot{x}^{w}$, $t \dot{a} x^{w}$ 'occur later, later, follow (an example)' 441

- 18.4.23 $x^{w} \partial n \ddot{x} \partial t \dot{e}^{2} e t \theta e^{2}$ 'meanwhile, in the course of that, after some time, finally' 441
- 18.4.24 *nán* 'too, very, very much' 441
- 18.4.25 scécoň 'really, certainly' 442
- 18.4.26 *θ∂*²*it* 'truly' 443
- 18.4.27 $t \dot{a} \ddot{x}^{w} \sim t \dot{a} x^{w}$, $t \dot{a} \ddot{x}^{w}$, $t \dot{a} x^{w}$ 'adjust, exactly, just' 443
- 18.4.28 *cəlél* 'nearly, about to' 444
- 18.4.29 *cəlél* [?]*al* 'barely' 444
- 18.4.30 $x^{w} \acute{e} l \partial q$ 'nearly' 444
- 18.4.32 sní?anł 'enough, just enough, permitting' 445
- 18.4.33 $\vec{\lambda} a x^w l \hat{a}^2 a s ({}^2a \tilde{l})$ 'worthless, of no account, no matter that, even if' 446
- 18.4.34 [?]*iwáwa*, [?]*iwáwa*[?] ~ [?]*iwáwwa*, [?]*iwáwa*[?] 'maybe, perhaps' 447
- 18.4.35 wála, wála, 'probably, presumably, I guess' 447
- 18.4.36 háy 'specifically, uniquely, as for, more' 448
- 18.4.37 wənáy 'only' 450
- 18.4.38 tx^wáy 'only remaining' 452
- 18.4.39 *cəlím* 'even' 453
- 18.4.40 $m \delta \vec{k}^w$ 'all' 454
- 18.4.41 $\check{x}^{w} \delta m$ 'move fast, be quick, immediately, be able to' 456
- 18.4.42 $s \dot{\lambda} \dot{a} \dot{\pi}$ 'enough, ought, can' 457
- 18.4.43 *sk^wéy* 'impossible, unable' 458
- 18.4.44 scowét 'smart, capable, able' 459
- 18.4.45 *sqí?qəl, sqíqəl* 'unable, ignorant (of how to do something)' 459
- 18.4.46 $23\dot{y}$ 'better' 459
- 18.4.47 $s \hat{\lambda} i^{\gamma}$ 'want, like' 460

19 Numerals

- 19.1 Simple Forms 462
- 19.2 Compound Forms 464
- 19.3 Ordinals 466

20 Exclamations and Interjections

- 20.1 Exclamations 468
- 20.2 Exclamatory-Interrogative Adjectives 468
 - 20.2.1 $\vec{x} q \vec{e} \vec{y}$ 'how far! however far, for an unknown distance' 468
 - 20.2.2 *léq* 'how fast! however fast, at an unknown speed' 469
 - 20.2.3 *tə²éň* 'how wonderful! what a ___! how powerful!' 470
 - 20.2.4 wéwə 'how wonderful!' 470
 - 20.2.5 *x^wawém* 'how big!' 470

		-			
	20.2.6	$\lambda i \dot{c}$ 'how long!' 470			
	20.2.7	$\dot{c}i\dot{\lambda}\partial m\partial \dot{t}^{\theta}$ 'how tall!' 470			
20.3	Interjections 470				
	20.3.1	<i>hé</i> ² <i>e</i> 'yes' 471			
	20.3.2	<i>?áwə</i> 'no' 471			
	20.3.3	<i>ná</i> 'what, yes' 471			
	20.3.4	$n \dot{a} \dot{w} \sim n \dot{a} \cdot \dot{w} \sim n \dot{a} w$ 'yes,' 'hey' 471			
	20.3.5	$ni^{?} \sim ni$ 'hey there, excuse me' 471			
	20.3.6	<i>?i</i> 'here' 471			
	20.3.7	$y e \sim y e^{\cdot}$ 'well, all right' 472			
	20.3.8	$^{2}\acute{a}$ 'ah, oh, aw, well' 472			
	20.3.9	? <i>o</i> ·, ? <i>s</i> · 'oh' 472			
	20.3.10	?áà?, ?á?à 'ahhhh! aha!' 473			
	20.3.11	² õ 'hmm' 473			
	20.3.12	<i>?óš</i> 'well!' 473			
	20.3.13	<i>šé</i> ² 'well!' 474			
	20.3.14	$\check{s}\acute{a}$, $\check{s}\acute{a}$? 'pshaw! oh, heck, etc.' 474			
	20.3.15	?écənè 'oh my!' 474			
	20.3.16	<i>⁹énene</i> 'oh my!' 474			
	20.3.17	<i>yá</i> . [,] 'my!' 474			
	20.3.18	<i>yu</i> [•] 'wow!' 474			

21 Kinship Terms

- 21.1 Consanguineal Relatives 476
 - 21.1.1 Parents and Children 476
 - 21.1.2 Siblings and Cousins 477
 - 21.1.3 Uncles and Aunts, Nephews and Nieces 478
 - 21.1.4 Grandparents, Grandchildren, and Beyond 479
 - 21.1.5 Relatives and Kin Groups 480
- 21.2 Affinal Terms 481
 - 21.2.1 Spouses 481
 - 21.2.2 Primary Affines 482
 - 21.2.3 Secondary Affines 483
 - 21.2.4 Step, Half, and Adopted Relatives 484
- 21.3 Terms of Impersonal Reference 484

22 Space and Time

- 22.1 Space 487
 - 22.1.1 The Expression of Movement and Location in Relation to Water 488
 - 22.1.1.1 Inland versus Shoreward 489
 - 22.1.1.2 Offshore versus Shoreward 490

- 22.1.1.3 Upstream versus Downstream 491
- 22.1.1.4 The Regional Axis 491
- 22.1.2 In Relation to the House and Fire 492
- 22.1.3 The Ceremonial Circuit 495
- 22.1.4 In Relation to the Canoe 496
- 22.1.5 Directions and Winds 498
- 22.1.6 Measurement of Space 499
- 22.1.7 Place Names 499
- 22.1.8 General Terms 500
 - 22.1.8.1 The Concepts "Here" and "There" and "To" and "From" 500
 - 22.1.8.2 The Concepts "Near" and "Far" 501
 - 22.1.8.3 The Concepts "Leaving," "Staying Away," and "Returning" 502
 - 22.1.8.4 The Concepts "Before," "Behind," and "Pass" 503
 - 22.1.8.5 The Concepts "In," "Out," Middle," "Between," and "Among" 504
 - 22.1.8.6 The Concepts "Along," "By Way of," "Across," and "Around" 505
 - 22.1.8.7 The Concepts "Turn," "Left," and "Right" 506
 - 22.1.8.8 The Concepts "Up" and "Down," and "Above" and "Below" 506
 - 22.1.8.9 The Concept "Everywhere" 507
- 22.2 Time 508
 - 22.2.1 Day and Night 509
 - 22.2.2 Time by the Clock 511
 - 22.2.3 The Week 512
 - 22.2.4 The Month 512
 - 22.2.5 The Tide 513
 - 22.2.6 The Year 514
 - 22.2.7 The Seasons 515
 - 22.2.8 The Calendar 516

23 Sample Texts

- 23.1 Getting Winter Supplies (CC 1) 519
- 23.2 The Deer Drive (CC 2) 524
- 23.3 Night Hunting in the Gulf Islands (CC 3) 528
- 23.4 Mink, Beaver, and Scouring Rush (JP 9) 532
- 23.5 The Origin of the Name "Musqueam" (JP 13) 539

Appendix 1 Index of Grammatical Elements and Lexical Affixes 549
Appendix 2 Names of Places and Peoples 566
Appendix 3 A History of Work on Halkomelem 577
Bibliography 585
Index 592

Tables

Table 11.1Aspectual prefixes (§11.1)251

- Table 14.1The person markers (§14.2)321
- Table 14.2 Object-subject paradigm (§14.2.5) 328
- Table 15.1 The articles (§15.1) 340

Table 15.2The simple demonstratives (§15.2.1)351

- Table 15.3The locative demonstratives (§15.2.2.1)352
- Table 15.4The third-person demonstratives (§15.2.2.2)353
- Table 15.5The demonstrative auxiliaries (§15.3)364
- Table 16.1 Order of second-position predicate particles (§16.2) 368
- Table 17.1Interrogative words (§17)395
- Table 18.1 Adverbs and adverbial and modal words and phrases (§18.4) 422

Maps

- Map 1 Halkomelem Territory (Introduction) xxvi
- Map 2 Musqueam Territory (Appendix 2) 567

Preface and Acknowledgments

The Musqueam people are now largely the residents of the Musqueam Indian Reserve on the North Arm of the Fraser River, adjacent to the city of Vancouver, British Columbia. Their language, which they call həndəmínəm, is one of the Downriver dialects of the language known to anthropologists and linguists as Halkomelem. This latter term is based on the Upriver form of the name, and the Musqueam see it as inappropriate for the Downriver and Island forms of the language. I will have to use it, however, when referring to the language as a whole.

A "reference grammar" is said to be one written for laymen rather than for linguists, describing the language with as little technical jargon as possible, illustrating its various features with abundant examples, and cross-referencing and indexing so that readers with some knowledge of the language can indeed look something up and find an answer. I have tried to follow these principles in the hope that this grammar will be useful not only to linguists specializing in Salishan languages but also to interested non-linguists, students of Salish literature, and above all the Musqueam and speakers of other dialects of the language.

In writing this grammar I have not been able to dispense with all technical jargon. As much as possible, I have tried to use traditional terms, which reflect a rather traditional sort of analysis. In organization, however, I have not been traditional, placing morphology largely after syntax. For placing syntax before morphology, I have the Thompsons' sketch of Clallam (Thompson and Thompson 1971) as a precedent. I was probably also influenced by my early exposure to the teaching method of Professor Henry S. Tatsumi, who began his course in Japanese with an outline of syntax, allowing the students to create Japanese sentences with a largely English vocabulary, or, in the case of nisei who were already doing this at home, making them aware of the principles they were using. But perhaps my choice was made largely on the basis of the feeling that tables and lists of affixes and particles are pretty dull anyway, especially if one does not yet know where and how these things are likely to be encountered.

Once the choice was made, however, it turned out to have a practical consequence. In the chapters on syntax, the sentences used as examples are graded according to what has already been introduced; I have tried to use no sentence with a construction that has not been previously accounted for. In the chapters on morphology, however, coming afterwards as they do, I have laboured under no such constraint and have been able to illustrate an affix, particle, or word with a special function in a construction of any sort.

This grammar is based on work I have engaged in from time to time over a period of many years. (For a history of the work, see Appendix 3.) Primarily it reflects the speech of Christine Charles (1894-1968) and James Point (1881-1979), as exemplified in elicited sentences and dictated texts. Some of the texts dictated by Mrs. Charles originated with her husband, Andrew Charles (1893-1961).

I reviewed all the texts dictated by Mrs. Charles with Mr. and Mrs. Charles's daughter, Della Kew (1929-1982), getting some additional material from her. Later I worked with Arnold Guerin (1910-1987), checking on many features of the language and going over many of the texts dictated by Mr. Point. Mr. Guerin's speech was, as he readily acknowledged, influenced by long residence in the Cowichan dialect area and so may be less representative of Musqueam. But working with him was doubly rewarding because he was a linguist in his own right and a colleague in the enterprise of exploring the language. I am profoundly grateful for all of the help these speakers of the language have given me and for the great pleasure it has been to work with them.

I should also express my gratitude to Professor Melville Jacobs for introducing me to the Native languages of our region and to Professor Harry Hawthorn for making my work at Musqueam possible. There are many others who have also given me help and encouragement in my work. In its early stages, Aert Kuipers generously gave me many hours of much-needed guidance. W.W. Elmendorf shared a summer's work with the Charleses and provided the concept and motivating force behind our joint paper on Halkomelem dialect differences (Elmendorf and Suttles 1960). Over the years, William H. Jacobsen, Laurence and Terry Thompson, Thomas Hess, and M. Dale Kinkade, among others, have shared insights and given advice. I have learned much about Halkomelem from the work of Brent Galloway on the Chilliwack dialect and of Thomas Hukari, Adrian Leslie, and Donna Gerdts on Island dialects. I must thank Barbara Efrat and Robert Levine, then of the Linguistics Division of the (then) British Columbia Provincial Museum, for inviting me to write this grammar for a projected series. Through no fault of theirs, the series did not materialize, but without their invitation and encouragement, I might not have completed the work. I thank Randy Bouchard and David Rozen for sharing information and references, Brent Galloway for computer help, and Mercedes Hinkson for suggestions on formatting. To Donna Gerdts I am especially indebted for her time and patience in reading and criticizing drafts of this work and for her efforts to keep me from blundering in logic and usage. I thank Dale Kinkade for carefully reading and correcting the version prepared for the museum. I thank Patricia Shaw for catching a number of errors in the present version, and Francis Chow for catching still more errors and inconsistencies in the course of his meticulous copy-editing. Remaining errors and inconsistencies are, of course, mine.

I am grateful to the University of British Columbia Press for agreeing to publish this work, to Holly Keller-Brohman for her guidance, and to Charles Ulrich for the use of his Straight font and his long work in preparing the manuscript for publication. I thank my son Cameron for his frequent advice and help with my computer and for the maps. And finally to Shirley, who has cheerfully lived with this for an unconscionably long time and unfailingly given good advice on style and sense, my heartfelt thanks.

I must also express gratitude for financial support for my work with Halkomelem from the Leon and Thea Koerner Foundation in Vancouver, the President's Committee on Research at the University of British Columbia, the American Council of Learned Societies, the Phillips Fund of the American Philosophical Society, the Desert Research Institute in Reno, the Faculty Committee on Research and Publications at Portland State University, the Royal British Columbia Museum, and the Melville and Elizabeth Jacobs Research Fund.

Introduction

Halkomelem

Halkomelem is one of the twenty-three languages that belong to the Salish Family of Northwestern North America. It is the language of the Native peoples of southeastern Vancouver Island from the west shore of Saanich Inlet northward to somewhere beyond Nanoose Bay, and of the mainland from the Fraser Delta eastward upriver as far as Harrison Lake and the lower end of the Fraser Canyon. These peoples are known by a number of names – in their English forms, Malahat, Quamichan, Penelekuts, Nanaimo, Musqueam, Kwantlen, Katzie, Chilliwack, Tait, and many more. These are names that once designated single villages or clusters of villages. Many have come to designate "bands," the administrative units established by the colonial and later federal authorities in the middle of the nineteenth century.

Within the Halkomelem area, the language is spoken with some marked differences in phonology, lexicon, and even grammar. Three main dialect groups are distinguishable: (1) an Island group, spoken by people whose old winter villages were on Vancouver Island and in the Gulf Islands, including the Cowichan, Chemainus, and Nanaimo dialects, with differences especially between Nanaimo and the rest; (2) a Downriver group, spoken by people whose old winter villages were around the mouth of the Fraser and upstream to the Stave River, including Musqueam, Tsawwassen, Kwantlen, and Katzie; and (3) an Upriver group, spoken from Matsqui on upstream, including Matsqui-Sumas (which shares some features with Downriver), Chilliwack, Chehalis, and Tait (Elmendorf and Suttles 1960; Galloway 1977, xii; Gerdts 1977). The Musqueam dialect, described in the present work, belongs in the Downriver group.

The language is identified in Island dialects as *həlqəmínəm*, in Downriver dialects as *hənqəmínəm* or *hənqəmínəm*, and in Upriver dialects as $h \ell l \dot{q} am \ell y l am$.¹ James Point identified $h a \dot{n} \dot{q} am (\dot{n} am)$ as the language as "all as far as understand one another, on the Fraser River and the Island." The term is thus the Musqueam equivalent of "Halkomelem." This name (pronounced hælkəméyləm in English) is an anglicization, first used by Charles Hill-Tout (1903), of the Native term for the language in its Upriver form. About the same time, the Downriver form was anglicized as "An-ko-me-num" by Crosby (1907). Wilson Duff used Hill-Tout's version of the term in his ethnography of the Upper Stalo Indians (Duff 1952, 11), and the term has been in general use among anthropologists and linguists since then.²

The Halkomelem language and its speakers were formerly often identified as "Cowichan" (spelled variously), as by Boas (1890, 806; 1897, 320), Hodge (1910, 1:355), and the Department of Indian Affairs (Canada 1970, 28-35), but this usage runs counter to others. On the one hand, in Native usage the name "Cowichan" (or its native form) is restricted to the people of the Cowichan River and environs on Vancouver Island; on the other hand, some writers (Tolmie and Dawson 1884; Newcombe 1909; Goddard 1934) at one time extended the name "Cowichan" to all of the contiguous Coast Salish north of Puget Sound, including speakers of as many as seven Salishan languages. All in all, it seems best to restrict "Cowichan" as in Native usage.

Halkomelem within Salish

In the most recent classification of the Salish languages (Thompson 1979, 693; Thompson and Kinkade 1990, 34-35), Halkomelem is one of the ten members of the Central Salish branch, which includes all of the Salish languages of the Strait of Georgia–Puget Sound Basin. There are four other branches of the family: Tsamosan, consisting of four languages in southwestern Washington; Interior Salish, consisting of seven languages in the Fraser and upper Columbia river drainages east of the mountains; Bella Coola, a single language on the central coast of British Columbia; and Tillamook, a single language on the northern Oregon coast. Bella Coola and Tillamook are geographically isolated from the rest; the other three branches are contiguous. The speakers of the Central and Tsamosan languages are identified in the ethnographic literature as "Coast Salish."

¹ My phonetic recording of the word in the Tait area was [hældjamélam] ~ [hældjabélab]. The Island form has been written "Hul'qumi"mum"" by Hukari and Peter (1995). The name is said, perhaps only in the Upriver area, to be derived from *ladjemal* or *nadjeman*, a village on Nicomen Slough, below Chilliwack (Duff 1952, 22; Galloway 1977, xviii). It is a progressive form with the suffix *-am* 'intransitive.' My guess is that if indeed it is derived from 'Nicomen,' its literal meaning is 'be Nicomening,' i.e., 'be speaking in the Nicomen manner.' There is a close parallel in the Northern Straits term for that language, given to me in the late 1940s by Samish and Lummi people as lak"əŋínəŋ, from lak"éŋən, the name of the people around Victoria.

² The term *Stalo* is from *stálo* (Downriver) or *stálo* (Upriver) 'river.' James Point believed that as a term for the peoples of the Fraser Valley it was first used by the early priests. Duff (1952) used the term "Upper Stalo" for the Halkomelem-speaking peoples from Chilliwack upstream. The spelling *Stó:lo* was adopted by the Coquileetlza Education Training Centre in the late 1970s.

In spite of the linguistic differences, the Coast Salish region appears to have been, before European settlement, a social and biological continuum within which speakers of each language had ties of marriage with their neighbours and participated with them in joint social and ceremonial activities (Elmendorf 1960; Suttles 1987b; Kennedy 1993, 2000). Nevertheless, there were cultural differences, and on the basis of these differences, four segments of the continuum have been distinguished: Northern Coast Salish, consisting of the speakers of Comox, Pentlatch, and Sechelt; Central Coast Salish, consisting of the speakers of Squamish, Halkomelem, Nooksack, Northern Straits, and Clallam; Southern Coast Salish, consisting of the speakers of Lushootseed and Twana; and Southwestern Coast Salish, consisting of the speakers of the Tsamosan languages, Quinault, Lower Chehalis, Upper Chehalis, and Cowlitz (Suttles 1990). Thus, the Halkomelem language belongs to the *Central Branch* of the Salish Family, while the traditional cultures of the Halkomelem-speaking peoples belong to the *Central Coast Salish* group of Coast Salish cultures.³

Diversity within Halkomelem

As we might expect, the differences are greatest between Island and Upriver dialects, with Downriver providing a link between the two other groups. An initial study (Elmendorf and Suttles 1960) suggested that Downriver Halkomelem is closer to Island phonologically and closer to Upriver lexically.

Phonologically, the most obvious differences are: Island has \check{s} where Mainland dialects have x, and Island and Downriver have both n and l, while Upriver has merged these as l. Upriver speakers often use \check{s} , \check{c} , and \check{c} , where Island and Downriver speakers have s, c, and \dot{c} . Upriver lacks the glottalized resonants and post-vocalic glottal stops of Island and Downriver, vowel length usually but not consistently appearing where Downriver and Island have post-vocalic glottal stops. Upriver dialects also have greater pitch differences, with some words being distinguished by pitch alone. All of these differences, together with an overall laxness of the voice, make Upriver speech sound remarkably different from Downriver and Island. In this last respect, Downriver falls between the laxness of Upriver and the tenseness of Island speech. Downriver speakers often drop final glottal stops and glottalize resonants more lightly, making them harder to detect.

³ Thus "Central Salish" as a taxon in a linguistic taxonomy should not be confused with "Central Coast Salish" as a taxon in a cultural taxonomy. The linguistic taxonomy, I should add, is based on much solid research, while the cultural taxonomy is somewhat impressionistic and was adopted for Volume 7 of the *Handbook of North American Indians* in part as a means of coping with the task of describing the cultures of the whole Coast Salish region. The term "Central Coast Salish" has been used for some time, however (Suttles 1968, 1977, 1990; Kew 1980; Kennedy 1993), to designate a culturally distinguishable segment of the Coast Salish continuum. The possible confusion of the taxa is unfortunate.



Map 1 Halkomelem Territory

Lexically, the most immediately recognizable difference among the major divisions of the language is in the third-person personal word 'he/she/it is the one,' nit on the Island and $\dot{x}a$ on the Mainland, which occurs with great frequency and forms the basis for a set of frequently used demonstratives. But there are many other lexical differences (see Elmendorf and Suttles 1960; Gerdts 1977).

The grammatical differences remain to be worked out, but the detailed grammar by Galloway (1993) suggests to me that at least some of the grammatical differences are the result of phonological changes like the loss of the oblique particle, the merging of the interrogative particle with a preceding auxiliary, and the loss of the locative meaning of the auxiliaries (for examples, see Suttles 1994).

In earlier times, there was probably greater diversity within each of these dialect areas than there has been in the last generation or two. It is said that once every village had some unique features of speech, and even segments of a village might differ. Some of the differences between my two principal sources at Musqueam were said to reflect such local variation, but even greater differences reportedly once existed there.

The diversity of Halkomelem dialects and the relationships among them (yet to be worked out) must be the products of complex social and economic forces as well as processes of linguistic change. The diversity certainly cannot be seen as simply the result of geographical barriers separating regions or of social and economic self-sufficiency isolating villages as "little communities." On the contrary, during the period for which we have historical and ethnographic information, there were neither geographical barriers nor isolated villages.

The Regional Social System

In particular, the Strait of Georgia was not a barrier. For the summer runs of salmon, many Island people crossed the strait and camped along the Fraser River, mainly in the Downriver area, while both Downriver people and Island people went on up the river to fish in the Upriver area. Northern Straits and Squamish speakers were also involved in this activity (Duff 1952, 25-26; Suttles 1998, 172-74). Thus, the Lower Fraser seasonally drew people together from all over the Halkomelem area and may have helped preserve the unity of Halkomelem even while promoting external contacts.

Both internal and external contacts were also promoted by the marriage system. As elsewhere in the Strait of Georgia–Puget Sound Basin, families with the means in each village arranged marriages for their children with families in other villages, often in other dialect or even language areas. Marriage ties established economic ties and led to participation in inter-village ceremonies and games. Without anything resembling European political institutions, the Halkomelem-speaking peoples were participants in a regional social network that had no discernible limits (Suttles 1960, 1963, 1987b; Elmendorf 1960, 277-305; Kennedy 1993). This regional social network probably extended, in spite of occasional conflict, throughout the Georgia-Puget basin and beyond.

Within this region, Halkomelem occupied an important position. Its territory lies close to the geographical centre of the Georgia-Puget basin and also close to the linguistic centre of the Central branch. To the north are (or were) four languages – Squamish, Sechelt, Pentlatch (now extinct), and Comox-Sliammon; to the south are another four (or five) – Nooksack (perhaps now extinct), Straits (perhaps divisible into Northern Straits and Clallam), Lushootseed, and Twana. These nine or ten languages constitute the Central Salish branch of the family. Traditions and genealogies indicate that the speakers of Halkomelem were aware of and had at least indirect ties with speakers of these other Central Salish languages. They were also aware of and occasionally intermarried with the nearest Interior Salish to the east and with the nearest Wakashans to the west and north.

Before the great epidemics that first came in the late eighteenth century, Halkomelem may also have been a demographic centre. In 1928 James Mooney estimated the pre-epidemic population of the Halkomelem-speaking peoples as 12,600, greater than that of any other Salishan language. A recent review of the data by Robert Boyd reduces the Halkomelem figure to 10,534, still much greater than the immediately adjacent languages but less than the 11,835 he gives for Lushootseed. For a comparison of Mooney's and Boyd's estimates for the whole Coast Salish, see Boyd 1999, 264-65.

It seems likely that the environmental richness of the Lower Fraser could account for a relatively larger population in the Halkomelem area. If this was the case, it may have had some influence on the course of linguistic history in the region. Hess (1979b) has shown that Halkomelem may be the centre of innovations in the terms for 'deer,' 'rabbit,' and 'lake,' and Kinkade's analysis (1986) of terms for 'blackcap' suggests a similar history.

Musqueam

"Musqueam" (pronounced móskwiyəm in English) is from $x^wm \delta \theta k^w \partial y \partial m$ 'place of $m\delta \theta k^w \partial y \partial r m \delta t^{\theta} k^w \partial y'$ (a plant no longer identifiable, but see Appendix 2 for further discussion), the name of a village at the mouth of the North Arm of the Fraser, just southeast of Point Grey, on the present Musqueam Indian Reserve. According to tradition, the Musqueam people once had villages at Jericho, the mouth of the Capilano River, and perhaps elsewhere on Burrard Inlet, at Marpole up the North Arm, and elsewhere in the delta (see Appendix 2 for place names). The people of Indian Arm are said to have spoken the same as the Musqueam. How homogeneous the speech of this area was and how much it resembled recent Musqueam are questions we cannot now answer. By the latter half of the nineteenth century, the Native people of Burrard Inlet were becoming mainly Squamish in speech, and Musqueam was the only Halkomelem village on the North Arm.

At that time, Musqueam itself consisted of two principal groups of houses, $m \acute{a} l \partial y$, toward Point Grey, and $s \acute{c} \partial l \acute{e} x^w$, at the head of a now vanished slough. There are said to have been some differences in speech between these two settlements, and some even within $s \acute{c} \partial l \acute{e} x^w$, where the people of the row of houses called $s c \partial l q^w \acute{a} \theta \partial n$ used n for l, making theirs an all-n dialect, maximally contrasting in that feature with the all-l dialects of the Upriver area (Elmendorf and Suttles 1960, 7). By the late 1940s, there was said to be only one old person left who used n for l, Old William at Coquitlam. However, for several words I have found individuals differing in which sound they used.

In recent times, there have been other differences among families at Musqueam. My two principal sources, Christine Charles and James Point, differed in their treatment of the second-person possessive, in a few lexical items, and in some preferences in syntax. I cannot say to what extent these differences reflect older differences within Musqueam or differences in their life histories. Andrew and Christine Charles and James Point were all native speakers of Musqueam and had lived as children at $s\dot{c}al\dot{e}x^w$, the upper part of the village, but each had some familiarity with another dialect of Halkomelem. Mrs. Charles was raised at Musqueam by her Musqueam mother and stepfather. She went to school for three years near Chilliwack and was familiar with the Upriver dialect. Mr. Charles was born at Musqueam of Musqueam parents, but after his father died, his mother remarried at Duncan in the Cowichan area, and he lived for a few years there and spoke Cowichan as well as Musqueam. Mr. Point also had Musqueam parents, but he went to school for several years on Kuper Island in Island Halkomelem country and learned that dialect. His first wife was Katzie, and he lived for a time with her family, who must have spoken a form of Downriver Halkomelem very close to Musqueam.

The differences among Musqueam speakers seem slight when their speech is compared with samples of Upriver or Island dialects, however, and my impression is that there were no great differences in speech within Downriver Halkomelem. James Point mentioned one or two words that were different at Tsawwassen, but I do not believe he mentioned any differences at Katzie. The people at Coquitlam were Musqueam families who had moved upriver a generation earlier. Mr. Point once asserted that what the Musqueam speak is really "Langley," and when we reconstructed one of Hill-Tout's Kwantlen texts, he found only two or three places where he would have chosen different words. This similarity between Musqueam and Kwantlen may simply reflect earlier proximity. According to tradition, the Kwantlen once lived nearby, at the site of New Westminster, and after Fort Langley was established in 1827, they moved further upstream to be near the fort. However, records of the Hudson's Bay Company's first contact with the Kwantlen show that they were already in the area where the fort was built (Suttles 1998, 170-71). Alternatively, the similarity between Musqueam and Kwantlen may reflect a later influence resulting from the prestige that the Kwantlen enjoyed after 1827 as the neighbours, allies, and affines of the Hudson's Bay Company men at the fort.

In recent years, however, the source of influence has clearly been from the other direction. In the 1950s and '60s, younger people at Musqueam who spoke Halkomelem were said to speak more in the Cowichan fashion. This probably had several causes. One may be that a number of Musqueam, beginning before the turn of the twentieth century, attended the residential school on Kuper Island, where they were exposed to Island forms. Another must be that the Island people have for a long time been more numerous and more active in maintaining the Native ceremonial system. This has given Cowichan a preferred status for some people in some situations. Speeches at the big dances are more often in Cowichan than in anything else, and speakers of other dialects may switch to Cowichan for formal speeches if they can do so. Referring to this, Mike Underwood of Duncan once remarked, "Cowichan is our national language, you might say." At present, however, there is a strong feeling among

the Musqueam that they ought to preserve Musqueam as a distinct form of speech.

Today, at the beginning of the twenty-first century, in all Halkomelem bands, Halkomelem speakers are a small minority. Most are middle-aged or older and very few are monolingual. Speakers are greatly outnumbered everywhere by children and young people who speak English only. It was probably inevitable that the Native people would learn English when, in the middle of the nineteenth century, English-speaking settlers and potential employers came into their country in overwhelming numbers, but their loss of their language must be largely the result of the policy of the government and of the church-operated residential schools to stamp out the Native languages (Levine and Cooper 1976). In several parts of the area, however, Native people are making efforts to teach the language to the young, and it has even been taught in public schools.

A Note on the Examples

Most examples of sentences are identified as to source by initials. The principal sources are Christine Charles (CC) and James Point (JP). A number appearing after these initials refers to the number of the text from which the example comes. I have tried whenever possible to use examples from dictated texts rather than from elicited sentences. I have used examples elicited from Della Kew (DK) and from Arnold Guerin (AG) to show differences among speakers and to supplement material from the principal sources. I have especially relied on Mr. Guerin in exploring problems that I was quite unaware of when working with Mrs. Charles and Mr. Point. Andrew Charles (AC) is the source of a few examples. (Actually he contributed a great deal as the source and inspiration of much of the material given by Mrs. Charles, but I have credited him only when the record shows that the example came directly from him.) A few forms came from Simon Pierre (SP) of Katzie. One sentence is from Herman Guerin (HG), Arnold's much older brother.

The examples often appear in four lines. The first is simply the Halkomelem as recorded, ignoring what may have been or ought to have been there, such as glottalization of resonants. The second line gives the Halkomelem segmented into morphemes, except that I have not tried to represent components in the internal morphology of roots (infixes, forms of reduplication, and so on) as separate elements. In this line, I have also indicated what the rules of grammar imply ought to be there, most often oblique particles. The third line identifies the segments of the second line. Many of these segments, especially the most commonly occurring grammatical elements, are identified by abbreviations, listed below. But I have not given labels to all grammatical elements; those easily translated are simply translated in the third line. The fourth line is a fairly close, but normal, English translation. If a sentence is such that a closer translation might help explicate it, I have added one, following "lit." (meaning "closer to a literal translation"). I have tried not to repeat an example but occasionally I have had to use the same sentence in two different contexts simply because it was the best example available.

Where the first and second lines in an example would be identical, I have given the example in only three lines.

Abbreviations

ADDICVIALIO	
ACT, act.	activity suffix -els, -əls
ADV	adversative particle $\theta \partial l$
ART	article $t\partial$, $\theta\partial$, $k^{w}\theta\partial$, $t\partial$, $k^{w}\partial$, $k^{w}s\partial$
ART(OBL)	oblique article $\dot{\lambda}$
ATT	attributive formant suffix -a·?ł
AUX	auxiliary verb ^{9}i , $ni \sim ni^{9}$
BEN	benefactive suffix - <i>lc</i> -
BE3P	'be third person' personal word $\vec{X}a$
BE3PL	'be third person plural' personal word <i>Xalam</i>
CAUS	causative suffix $-st \partial x^{w_{I}} \sim -st$ -
CERT	certainty particle <i>m</i> ²
CJ	Chinook Jargon
СОМ	comitative suffix $-st \partial x_{2}^{w} \sim -st$ -
CON	concern suffix -mət, -ámət, -mit, -me ⁹ t
DEM	demonstrative
DIM, dim.	diminutive
DISP	dispositional aspect
DUR	durative aspect
EMPH	emphatic particle \dot{q}_{∂}
EST	'established' aspectual prefix wa-
EXP	expectable particle <i>yeł</i>
FN	feminine nearby (article or demonstrative) to et al.
FP	feminine present (article or demonstrative) $\theta \partial$ et al.
FR	feminine remote (article or demonstrative) $k^w s \partial$ et al.
FUT	future particle <i>ce</i> ⁹
gen.	generic
GOAL	goal-causal suffix -nəs
INF	inferential particle $y \partial x^w$
INTR, intr.	intransitive suffix <i>-əm</i>
INSTR	instrument (lexical suffix) -tən
LCTR	limited control transitive suffix $-n \partial x^w \sim -n$, $-l \partial x^w \sim -l$ -
MN	non-feminine nearby (article or demonstrative) $k^{w}\theta \partial$ et al.
MP	non-feminine present (article or demonstrative) to et al.
MR	non-feminine remote (article or demonstrative) $\vec{k}^{w} \partial$ et al.
n.	noun
NOM	nominalizer s-

OBL	oblique particle ² ?
OBLNOM	oblique nominalizer $\check{s}x^w$ -
OBREL	oblique relater x^{w} -
p.n.	place name
PAS	passive
PER	imperative particle <i>le</i>
perf.	perfective
PERM	permissive suffix -s
PL, pl.	plural
PLPER	plural imperative particle ⁹ e·
PRES	presumptive particle <i>wa</i> ²
PROG, prog.	progressive aspect
QUOT	quotative particle <i>c</i> ²
REAS	reassuring particle ?ewəł
RECIP	recipient suffix - <i>əs</i>
RES, res.	resultative aspect <i>s</i> -
ROG	interrogative particle ?>
ROG!	emphatic interrogative particle ² a ² a
s.t.	something
sev.	several
SPEC	speculative particle <i>ctwa</i> ?
STAT	stative suffix -t
SUBPAS	subordinate passive suffix - <i>ət</i>
TR, tr.	transitive suffix $-t \sim -\partial t$, $-n\partial x^w \sim -n-$, $-l\partial x^w \sim -l-$, $-x$
V.	verb
2plpos	second person possessive suffix -?ələp
3PL	third-person plural particle ?e·ltan
3POS	third-person possessive suffix -s
3SUB	third-person subordinate subject -as
3SUBPAS	third-person subordinate passive $-\partial y \partial - \sim i - i$
3TR	third-person transitive subject -as

Symbols Used in Phonological Formulas

А	any full vowel (<i>i</i> , <i>e</i> , <i>a</i> , or <i>u</i>)	R	any resonant
С	any consonant	Т	any obstruent
Η	/h/ or /?/	V	any vowel (full vowel or schwa)
*	reconstructed phoneme or form	**	ungrammatical form

Musqueam Reference Grammar

1 Phonology

Halkomelem shares many of the areal features of phonology that made the Native languages of the North Pacific coast of North America so long impermeable to European understanding. It has a (seemingly) simple vowel system but a richly developed consonant system with several features unparalleled in the languages of Western Europe. These include the presence of lateral and uvular obstruents, the opposition of glottalized to plain plosives and resonants, and the opposition of labialized to plain velars and uvulars. Well over half of the consonants of Halkomelem have no counterpart in English.

1.1. CONSONANTS

The following consonants have been recorded in the Musqueam dialect of Halkomelem:

	Obstruents					
	Plosives			Spirants	Resonants	
	Plain	Glottalized	Voiced	Voiceless	Plain	Glottalized
Labial	р	ģ	{b}	{ f }	m	[ṁ]
Dental	(t^{θ})	ťθ	-	θ	_	_
Alveolar	t c	ỉ ċ	{d}	s	n	['n]
Lateral	Ż	_	-	ł	1	[İ]
Palatal	(č)	_	{j}	(š)	у	[ỷ]
Velar	(k)	(k)	_	х	{ r }	
Lab. velar	k w	k ^w	-	X ^w	W	[ŵ]
Uvular	q	ģ	-	ž	_	_
Lab. uvular	\mathbf{q}^{w}	ġ.	-	Х ^w	_	_
Laryngeal	_	?	-	h	-	_

Of these, twenty-eight (those not enclosed above) occur frequently and have unquestionable phonemic status. Five (enclosed in parentheses) occur less frequently or in limited environments and are of recent and/or peripheral phonemic status. Five other sounds, the glottalized resonants (enclosed in square brackets) occur frequently but pose a problem in phonemic analysis. And in addition five consonants (enclosed in braces) occur only in a few foreign words and their derivatives and in one sound-imitative word.

1.1.1. Obstruents

For descriptive economy, stops and affricates are grouped together as plosives. There are thus four series of obstruents: plain plosives, glottalized plosives, voiced plosives (of marginal status), and spirants. The plain plosives are generally less aspirate before vowels than in English but more aspirate finally. The glottalized plosives are ejectives but are usually not strongly released. Sequences of plosives are rearticulated; that is, the first is not released into the second. The obstruents are produced at as many as nine places of articulation. Adding accompanying labialization, we have thirteen sets of obstruents.

Labials. /p/ and /p/ are bilabial stops. /f/ is the labiodental spirant of English. It occurs only in recent loans from English and their derivatives, e.g., /káfi/ 'coffee,' /šx*kafiélə/ 'coffeepot,' /skíkf/ 'little skiff,' /číf/ 'chief.' In the past, older monolingual speakers reportedly used /p/ in such words. I have recorded /b/ only in /bá·s/ 'boss,' said to have been used for the first government-appointed chief.

Dentals. $|\theta|$ is usually a dental or interdental spirant somewhat like the English 'th' in *thin*. $|t^{\theta}|$ is a glottalized affricate $[t^{\theta}]$.¹ Some speakers, I believe, articulate these somewhat further back, reducing the difference between $|\theta|$ and |s| and between $|t^{\theta}|$ and |c'|.² Halkomelem $|\theta|$ and $|t^{\theta}|$ are the reflexes of Proto-Salish *c and *c and may have reached their present phonetic forms only recently.

I have recorded an unglottalized affricate, $[t^{\theta}]$, at Musqueam in only two instances. First, AC gave $[m \diamond t^{\theta} k^w \diamond y^{2}]$ as the name of the plant, now long gone, for which Musqueam itself was named, while CC and JP pronounced it $[m \diamond \theta k^w \diamond y^{2}]$. To explain AC's pronunciation, I can only suggest that he may have been using an old-fashioned realization of / θ / in this little-used and perhaps emotion-laden word. Second, JP occasionally used $[t^{\theta} \diamond]$ instead of the expectable $[t \diamond]$ for the non-feminine visible article. This is the Cowichan form, as JP acknowledged. In Cowichan, $[t^{\theta}]$ seems to be a fusion of the sequence //t- θ -// that

¹ Elmendorf and Suttles (1960) used the symbol 'ç for this phoneme, basing it on Boas's use of ç for θ. Galloway (1977) uses θ', noting that the phoneme has, in Upriver, a glottalized spirant allophone. Hukari et al. (1977) use ^{iθ}. Until the present work, I have used ż simply for convenience.

² Working with SP, who was missing some front teeth, I often mistakenly recorded "s" for /θ/ and "č" for /t^θ. With CC and JP I had little trouble with the spirants but occasionally had to recheck the glottalized affricates. Early to mid-nineteenth-century spellings of Native names using "s" for /θ/, as in "Musqueam" for /x m 5θk *sysm/, suggest that at the time /θ/ was more generally pronounced with the tongue more retracted.
occurs in the demonstrative system, and I have written the Cowichan article $/t^{\theta} a / when it$ has appeared. However, in Musqueam demonstratives, the sequence $/t\theta / is$ clearly a stop followed by a spirant; cf. Cowichan $[t^{\theta} \acute{e} \acute{y}]$ with Musqueam $[t\theta \acute{e} ?]$ 'he, that.' I do not believe we need to posit a phoneme $/t^{\theta} / for$ Musqueam, though it may be needed for Cowichan.

Alveolars. The stops /t/ and /t/ appear to me to be articulated as a point perhaps slightly forward of that of the usual English "t" and "d." I have recorded [d] in two items, [sk "dædæ] 'chickadee' (also a person's nickname), which may be imitative of the sound made by the bird, and [dædıs], a pet name for a little girl, probably Chinook Jargon for 'flower' (Jacobs 1936, 12). This does not seem to justify positing a /d/ phoneme.

It appears to me that the affricates /c/ and /ċ/, phonetically [t^s] and [t^s], and the spirant /s/ are somewhat more retracted than the /t/ and /t/, but this may be true of some speakers only, and the difference may not be great.³ Before a vowel, the affricate /c/, as in /cám/ 'go inland,'/cśsət/ 'be telling him,' is quite distinct from the sequence /ts/, as in /tsás/ 'poor,' /tsát/ 'approach it.' In final position, as in /yəpənéc/ 'going with the wind' versus //s-kwən-é-t-s// 'his holding it,' there may be little or no phonetic difference. However, in forms like the last, where the [t] and [s] represent different morphemes, it seems only sensible to write /ts/, and so I do. I have no contrasting examples with /ċ/ versus /ts/ but expect that they would be distinct.

In Musqueam, with rare exceptions, /c/ and /s/ do not occur before $/x^w/$. Where we might expect them for morphological or historical reasons, we find instead /č/ and $/š/.^4$ However, /c/ does occur freely before $/x^w/$, as in $/cx^w \acute{a}t/$ 'add to it.'

Halkomelem /c/ and /c/ are the reflexes of Proto-Salish *k and *k, while Halkomelem /s/ is the reflex of Proto-Salish *s. The source of /c/ in *k is reflected in one feature of the morphology (see §1.5.10 below).

Palatals. /č/ is a palatal affricate, phonetically $[t^{\delta}]$, like the English 'ch' in 'cheap.' /š/ is a spirant like the English 'sh' in 'sheep.' There is no glottalized palatal affricate. As indicated, /c/ and /s/ are usually replaced by /č/ and /š/, respectively, before /x^w/. If these were the only instances of [č] and [š], we could simply identify these two sounds as allophones of /c/ and /s/. However, there are a few words that have [č] and [š] in other environments, and these require, I believe, that we posit /č/ and /š/ phonemes. These include a very small number of words apparently of Native origin, as /čəčí?ġən/ 'mink' and

³ In Upriver dialects /c/, /c/, and /s/ can be as palatalized as [č], [č], and [š] (Galloway 1977, 5-9).

⁴ One exception is sx wə?int 'what meaning,' which was, I believe, consistently pronounced with an initial /s/. AG used an initial /s/ in /sx wəx wá/ 'lowered,' the resultative of /xwét/ 'lower it,' but JP used /š/. It may be that for some speakers, the s-2 'resultative' behaves differently from the s-1 'nominalizer.'

as /č= max/ 'pitch' (identified as Cowichan but evidently also used at Musqueam) and the interjections /š= /, /š= /, /š= /, /š= muspecified as on. They also include a small number of loan words from Chinook Jargon (CJ), French (Fr) via Chinook Jargon, and English (E), as /číkm=n/ 'metal' (CJ), /č= k (Fr) via Chinook Jargon, and English (E), as /číkm=n/ 'metal' (CJ), /č= k (Fr) via Chinook Jargon, and English (E), as /číkm=n/ 'metal' (CJ), /č= k (Fr) via Chinook Jargon, and English (E), as /číkm=n/ 'metal' (CJ), /č= k (Fr) via Chinook Jargon, and English (E), as /číkm=n/ 'metal' (CJ), /č= k (Fr) via Chinook Jargon, and English (E), as /číkm=n/ 'metal' (CJ), /č= k (Fr) via Chinook Jargon, and English (CJ (CJ (CJ), /č), /k=) (CJ), /č= k (CJ

Laterals. /ł/ is a lateral spirant. /Å/ is what is usually described as a glottalized lateral affricate, phonetically [\dot{t}^1], but the apex of the tongue at the onset is in the position for the lateral release rather than for a /t/ and there is little of the friction heard in the other affricates. /Å/ is quite distinct from the sequence /tł/, cf. /Åá/ 'be third person' and /tłád̥wt/ 'pry it off.' As in most other Salishan languages, there is no unglottalized lateral affricate.

Velars. /x/ is a strongly palatalized front velar spirant, phonetically $[x^y]$.⁵ It is the reflex of Proto-Salish *x, which has become /š/ in Island dialects of Halkomelem and in neighbouring Coast Salish Languages.

/k/ is a plain, and /k/ a glottalized, unrounded front velar stop. These are less strongly palatalized than the spirant. They are rare and, because Proto-Salish *k and *k have become /c/ and /c/ in Halkomelem, they may be of recent origin. I have recorded /k/ in two presumably Native words, /x "ankét/ 'hold on, wait a bit' and /skáti/ 'crazy.' It occurs in a small number of loan words from Chinook Jargon and English, as /kapú/ 'coat' (CJ < Fr *capote*), /lasék/ 'bag' (CJ < Fr *le sac*), /laklí/ 'key, lock' (CJ < Fr *le clef*), /lá·k/ 'log' (E), /ká·/ 'car' (E). It occurs in "baby talk" as a substitute for /q/, as in /ká?/ for /qá?/ 'water' when speaking to a child. I have recorded /k/ in only one word, where it is the babytalk substitute for /q²/ (see §1.6).

Labialized velars. $/k^{w}/$, $/k^{w}/$, and $/x^{w}/$ are, respectively, a plain stop, glottalized stop, and a spirant produced at about the point of articulation of English /k/ in *cool* and accompanied throughout by the lip-rounding of /w/.

In the sequence $/\delta x^w/$, especially in the common compound prefix $//s-x^w-//$ 'oblique nominalizer,' the rounding and velar friction of the $/x^w/$ can occur, it seems, simultaneously with the palatal friction of the $/\delta/$. In the Cowichan dialect, in most environments the $/x^w/$ of this prefix is lost, and Cowichan has $/\delta/$ where Musqueam has $/\delta x^w-/$. This pronunciation is also occasionally heard from Musqueam speakers, probably from Cowichan influence.

⁵ Elmendorf and Suttles (1960) and Galloway (1977) have used the symbol x^{y} for this phoneme, but because there is no [x] contrasting with [x^{y}], it seems more convenient to write /x/.

Uvulars. /q/ and /d/ are a plain and glottalized stop produced further back than English /k/, probably varying in place of articulation from uvular to back velar. The limited occurrence of /k/ and /k/ may permit a wider range of articulation than might be permitted otherwise. Nevertheless, velars and uvulars are distinct, in the adult and baby-talk forms cited above and in the only near-minimal pair discovered, /ká·/ 'car' and /qá?/ 'water.'é /X/ is a uvular fricative. It is produced with a good deal of friction or uvular vibration and contrasts strongly with /x/.

Labialized uvulars. $/q^w/$, $/\dot{q}^w/$, and $/\check{x}^w/$ have a uvular or near-uvular place of articulation and are accompanied throughout by lip rounding. $/q^w/$ and $/\dot{q}^w/$ are not always easily distinguished from $/k^w/$ and $/\dot{k}^w/$, but the uvulars have a different resonance produced by a different shape of the oral chamber and there may be some uvular friction. $/\check{x}^w/$ is more easily distinguished from $/x^w/$ because of its more audible friction or uvular vibration.

Glottals. /?/ is a glottal stop and /h/ a glottal spirant. They are listed here with the obstruents but possibly belong in a class by themselves. They are limited in their co-occurrence with other obstruents (obstruents proper), and they play unique roles in the morphology of the language.

1.1.2. Resonants

There are five plain resonants: two nasals /m/ and /n/, a lateral liquid /l/, and two semi-vowels /y/ and /w/. Perhaps /h/ too should be classed as a resonant, as Kuipers (1967, 21) has done for Squamish and Hukari (1976b) has suggested for Cowichan. I have also recorded [r] in [santusprí], also given as [santusplí] 'Holy Spirit' (< Fr *Saint-Esprit*, possibly through Chinook Jargon), and in *čéris* 'cherry.'

There may also be five glottalized resonant phonemes. Phonetically, there are glottalized resonants (i.e., resonants with accompanying glottalization), symbolized [\mathring{R}], and resonants preceded and followed by glottal stops, [?R] and [R?]. There seems to be some free variation between [?R] and [\mathring{R}], and there is certainly a good deal between [R?] and [\mathring{R}], but never between [?R] and [R?]. I have found no instances of contrastive distribution among any of the three. These observations allow for two possibilities: that there are two sequences of phonemes, /?R/ and /R?/, with overlapping [\mathring{R}] allophones, or that there is a single phoneme / \mathring{R} / that is realized in three ways. I have come to prefer the latter.⁷

Positing $/\mathring{R}/$, we can say that its allophones seem to be determined by the stress and quality of adjacent vowels. In all environments where $/\mathring{R}/$ occurs, $[\mathring{R}]$ seems to be a possible realization, but following a stressed full vowel (one other than schwa) and before an unstressed vowel, it is usually realized as [?R], e.g.,

⁶ Early in my work, I often recorded "k" for what turned out to be /q/, but much less often confused /k"/ and /q"/ or / \vec{k} "/ and / \dot{q} "/.

⁷ In his own transcription, AG treated [?R] as /?R/ and what I hear as [Å] and [R?] as /R?/.

/lámðt/ [lá?mət]	'throw it at him'
/píwํət/ [pí⁰wət]	'freeze it'
/sť ^θ áṁaqʷ/ [sť ^θ á ⁹ maqʷ ~ sť ^θ áṁaqʷ]	'skull'

Following an unstressed vowel and before a stressed vowel, it is usually [R?], e.g.,

/pəẅ́éls/ [pəwˀéls]	'freeze'
/ləṁéls/ [ləm?éls]	'throw something'
/šxʷṫ ^θ əṁíwən/ [šxʷṫ ^θ əmʔíwən]	'pit, seed'
/həṁá/ [həṁá ~ həm?á]	'pigeon'

Between a stressed schwa and an unstressed vowel, it varies between $[\hat{R}]$ and $[R^{9}]$, e.g.,

/mə́ǹə/ [mə́ǹə ~ mə́n?ə]	'child, offspring
/lə́məθət/ [lə́məθət ~ lə́m?əθət]	'bend over'

Elsewhere, between unstressed vowels and between a vowel and a consonant or zero, it is realized as $[\vec{R}]$, e.g.,

/sť ^θ əṁəléxૅən/ [sť ^θ əṁəléxૅən]	'arm bone'
/sť ^θ áṁcəs/ [sť ^θ áṁcəs]	'hand bones'
/sť ⁰ ám/[sť ⁰ ám]	'bone'
/spíwႆ/ [spíwႆ]	'ice'
/láṁ/ [láṁ]	'get hit by something thrown'

As may be seen from the examples, a root such as $/p\hat{iw}/\hat{r}$ freeze' or $/s\hat{t}^{\theta}\hat{am}/\hat{r}$ bone' can appear in several forms with the $/\hat{R}/alternately$ [?R], [R?], and [\hat{R}]. Assuming $/\hat{R}/\hat{r}$, we can identify this variation as purely phonetic. But one could of course posit /?R/ and /R?/ and consider the variation to be one of phonologically conditioned allomorphs.

Glottalized resonants do not occur as freely as the plain resonants. They do not occur in morpheme-initial position, with three exceptions: the particles $/\dot{m}_{2}/$ 'certain' and $/\dot{y}_{2}$ 'as usual,' and $/\dot{m}_{1}/$, a reduction of $/^{9}a\dot{m}_{1}/$ 'come.'

Although glottalized resonants can occur in roots, I have not noted any pairs of roots contrasting for glottalization of resonant only. Nevertheless, minimal pairs do exist. The glottalization of resonants plays a role in the morphology. For example, in progressive forms of the verb, resonants occurring after the stress (some, but not all, following rules not yet worked out) are glottalized; cf. /tí1əm/ 'sing,' /títələm' 'be singing.' In most progressive forms, this aspect is signalled by reduplication (as in the example) or a different vowel, and so the glottalization of any resonants is redundant and need not be very audible. However, for a few verbs the progressive differs from the perfective form by the presence of a glottalized resonant only; cf. /t̂θŕm/ 'pick berries' and /t̂θŕm/ 'be picking berries,' /tí.m/ 'pick' and /tł·m'/ 'be picking,' /xé.m/ 'cry' and /xé.m'/ 'be crying.'

There is some variation in the strength with which Musqueam speakers glottalize resonants. Transcribing texts dictated by JP, I often wrote plain resonants and only later discovered that they were glottalized. The glottal constriction of $[\mathring{R}]$ was especially gentle, the glottal stop of ['R] more apparent. CC's glottalized resonants were more easily heard but I missed some. In most Upriver dialects, glottalized resonants do not exist, while on the Island, they are, I believe, more sharply articulated. Downriver Halkomelem stands between the other dialect areas, linguistically as well as geographically, and possibly its speakers vary in this and other features depending on Island or Upriver influence.

1.2. VOWELS

Musqueam has five vowel phonemes:

	Front	Central	Back rounded
High	i	_	u
Mid	e	ə	_
Low	_	а	_

Two distinctions can be made among these vowel phonemes. First, four of them, /i/, /e/, /a/, and /ə/, are common, play important roles in the morphology, and are old in the language (being reflexes of earlier *i, *a, *u, and *ə, respectively), while /u/ occurs mainly in loan words, has no morphological role, and is very likely of recent origin.

Second, the full vowels, /i/, /e/, /a/, and /u/, can be distinguished from schwa /ə/. All five vowels occur both with and without stress, but schwa is by far the most common unstressed vowel. The full vowels occur with phonemic length, as /i·/, /e·/, /a·/, and /u·/, while schwa does not. The full vowels can be followed by glottal stops without restrictions, but schwa can be followed by a glottal stop only when it is unstressed and with a stressed vowel following the glottal stop. In the morphology, there is an alternation between schwa and the full vowels /i/, /e/, and /a/.

Unlike the consonant phonemes, which show little phonetic variation, the vowel phonemes vary considerably phonetically. The phoneme /i/ has three fairly distinct allophones. Following unrounded uvulars, it is realized as [e]. Preceding both unrounded and rounded uvulars, it is realized as [i] with a central off-glide. Elsewhere, it is realized as a low [i] or high [e]. It can be lower before a glottal stop or glottalized resonant, but speakers seem to vary in this. The /e/ is realized as a low to mid-front vowel, [ε] or high [ω]. The /a/ is low and central to back, usually close to [a]. The /u/ is fairly high, back, and rounded, a low [u] or high [o].

The schwa phoneme /a/ when stressed appears in most environments as a mid-central [a], but before /x/ it is fronted and higher, approaching [1]; before

/y/ it is also fronted, approaching [ϵ]; before /w/ it is lower and back, approaching [a]; and before rounded velars it seems mid-back, toward [o].

Unstressed /a/ is by far the most common unstressed vowel, and it seems even more chameleon-like than stressed /a/. Before /x/ and /y/ it can be as high as [1] or even [i], and before labialized velars and /w/ it can be realized as [0] or [v]. Moreover, an unstressed /a/ can be assimilated, by a kind of vowel harmony, to a stressed /e/ or stressed /a/ in an adjacent syllable.

Vowels before semi-vowels are especially troublesome, the schwa coming close to one or another of the full vowels. To complicate things further, there are some instances of forms with $/-3\dot{y}/$ and $/-1^{\prime}/$ alternating, such as $/s\theta \partial \theta \dot{y}/$ ~ $/s\theta \partial \theta \dot{1}^{\prime}/$ 'right,' 'fixed,' $/\dot{s}x w \partial x w \dot{y}/ ~ /\dot{s}x w \partial x w \dot{1}^{\prime}/$ 'awake.' In these cases, the morphology should yield the $/-3\dot{y}/$ form.⁸

The following sequences of stressed vowel and (plain or glottalized) semivowel can be distinguished:

- /śy/ in /x^wśy/ 'wake up,' /θśyt/ 'fix it,' /cśyx^wt/ 'dry it,' /?śỷ/ 'good,' /swśyqe?/ 'man, male,' /smśyəθ/ 'deer.'
- /éy/ in /skwéy/ 'impossible,' /θéỷt/ 'be fixing it,' /céỷxwt/ 'be drying it,'
 /sqwəméỷ/ 'dog,' /θéỷəm/ 'be baking,' /héỷ/ 'go ahead.'
- /áy/ in /háy/ 'stop,' /dáy/ 'die,' /cqwáy/ 'yellow,' /?áyəm/ 'slow.'
- /św/ in /sśwd/ 'seek,' /?śwkw/ 'be used up,' /šxw śwdan/ 'swan,' /?śwa/ 'no, not,' /nśwa/ 'you.'

/éw/ in /héwt/ 'rat,' /sqéwθ/ 'potato,' /séwd/ 'be seeking,' /?éwkw/ 'possessions.' /áw/ in /yəs?aŵθ/ 'in a rush,' /?áwkw/ 'being used up' (DK), /?áwə/ 'hour.'

⁸ With both CC and JP, I often wrote "éy" for what turned out to be /áy/ and "áw" for what turned out to be /áw/. (A few of these errors appear in Elmendorf and Suttles 1960.) I had less difficulty hearing the differences in the speech of DK and AG, because, I believe, in their Cowichan-accented speech there is more vocal tension differentiating full vowels from schwa. But they were also able to state explicitly, "That's a schwa."

No wonder Hill-Tout (1903, 370-71) complained bitterly that in the mouth of an informant, "-i, ai, and -e, as -o and au, were constantly interchanging in the most bewildering fashion." More recently, Thompson (1979, 697-98), discussing Salishan in general, has written:

Vowel systems are usually small, although vowels often exhibit wide variation. The central lax vowel ϑ , in particular, adapts strongly to its consonantal environments, and in many cases the foreign ear has great difficulty recognizing whether a variant of; or one of a tense vowels is being heard.

I would add that there may be difficulties even for Native ears. A younger but fluent speaker once assured me that 'be burning' is $/h\dot{g}\dot{q}w/$ and that 'fire' is $/h\dot{g}\dot{q}w/$. For older speakers, however, there is only one word, $/h\dot{g}\dot{q}w/$ 'be burning,' which if preceded by an article is 'that which is burning,' that is, 'fire.' One must suppose that for this younger speaker, and perhaps others, the two words in English and the closeness of $/\dot{g}\dot{y}/$ and $/\dot{e}\dot{y}/$ promoted a reanalysis of the Native word, splitting it into two.

/íw/ in /spíŵ/ 'ice,' /qíŵx/ 'steelhead,' /sníŵ/ 'advice,' /x^wíwəl/ 'go upstream,' /níŵət/ 'advise him.'

/úy/ in one presumably borrowed word, /šək^wlúy/ 'turnip.'

I have recorded no instances of /íy/, /íỷ/, /úw/, or /úŵ/.

The unstressed vowels differ from the stressed vowels in several respects. Unstressed /i/, /e/, and /a/ are relatively uncommon. In the interpretation presented here, unstressed /u/ does not occur, /ə/ is the only unstressed vowel that occurs with semi-vowels, and many instances of [i] and [u] are realizations of /əy/ and /əw/, respectively.

When stressed full vowels of roots lose their stress, as when they take stressed suffixes, they usually appear as schwa or zero. In affixes, too, unstressed vowels are usually schwa, and a stressed vowel, with loss of stress, usually becomes schwa.

Unstressed /i/ does appear as an unreduced full vowel in some words with stressed suffixes, as in / θ i θ áləs/ 'big-eyed' (from θ í 'big'), and in the unstressed auxiliaries / 9 i/ and /ni?/ ~ /ni/. It also appears as an infix in some verb forms, as in diminutives and duratives (for unstressed /i/ in progressives of duratives, see §7.5). Unstressed /i/ also appears as a rapid speech variant of the prefix /yə-/ 'along,' as an individual variant of the initial of /yəsélə/ ~ /isélə/ ~ / 9 isélə/ 'two,'⁹ and as a schwa assimilated to /í/, as in /ti⁹í/ for /tə⁹í/ 'this.'

Unstressed /e/ and /a/ occur in a small set of noun roots of the form CVCV?, e.g., /méqe?/ 'snow,' /téqe?/ 'salal berry,' /sk **áye?/ 'squirrel,' /náča?/ 'one,' /sqáma?/ 'breast,' /žáca?/ 'lake,' /sčáła?/ 'leaf,' or CVRCV?, as in /swáýqe?/ 'man, male.' Historically these unstressed vowels may be assimilated schwas; /e/ occurs most commonly following /é/ and /a/ most commonly following /á/, and there seems to be no schwa before a glottal stop in such an environment. But the occurrence of /e/ or /a/ here is not (at this point) entirely predictable. There are also individual differences. A reduplicated form of $náca^2$ 'one' meaning 'one person' was pronounced /nánàce?/ [ná?nəce?] by CC but /ná?ənċa?/ by JP. From both CC and JP I recorded $syalax*a^2$ 'old person,' but from AG I recorded $syalax*e^2$. Unstressed /e/ and /a/ also occur in a few suffixes. And they occur as ex-schwas that have become assimilated to nearby full vowels (of which more below).

Unstressed /əy/ and /əỷ/ occur in such words as /yəθéləỷ/ 'those,' /dáləy/ 'die (pl.),' /sθéθəỷən/ 'blood,' /cəỷx wéls/ 'dry something,' /'əỷθát/ 'become better,' /słénəỷ/ 'woman,' 'female,' /déməỷ/ 'young woman,' /sláwəỷ/ 'shredded cedarbark,' and /'əỷ/ 'and.' The unstressed /əy/ is phonetically [1y] or [i]; unstressed /əỳ/ can vary from [ɨỷ] to [1ỷ] to [1²]. I have three reasons for

⁹ Thinking I heard it from someone this way, I asked CC if 'two' could be pronounced [yəsélə]. "No," she said, "that's wrong. But of course a lot of Indians say it that way."

interpreting these as /ay/ and /ay' rather than as /i/ and /i?/, as others (Gerdts 1981; Hukari and Peter 1995; and Galloway 1977) have done for other dialects.

First, in CC's and JP's speech (though not in AG's), there seemed to be a phonetic difference between unstressed /əy/ and especially unstressed /əy' on the one hand and unstressed /i/ and /i[?] on the other. The word 'woman,' for example, can appear as [słéni?] or, especially in slower speech, with the final nearly as diphthongized and low, as [əy]. The conjunction 'and' I have recorded as [i], [əy], and even [?ey].¹⁰ On the other hand, there are unstressed [i] vowels that I have never heard as diphthongs, e.g., those in the auxiliaries /ni?/ and /?i/; /ni?/ is usually /ni/ and can be reduced to /nə/, but it is never [n+y], nor is the auxiliary /?i/ ever [?+y]. Phonemicizing these variable elements as /əy/ and /əŷ/, even though they may overlap allophonically with /i/ and /i?/, thus preserves what I believe is a real distinction.

A second reason for this choice is that it simplifies the description of the morphology. For example, / \dot{q} aləy/ can be described as / \dot{q} ay/ 'die' with the infix /-lə-/ 'plural,' paralleling /ném/ 'go,' /néləm/ 'go (pl).' And it preserves the basic form of the root; cf. / \dot{c} syx *t/ 'dry it' and / \dot{c} əyx *éls/ 'dry something,' /?ə́y/ 'good' and /?əỷ@at/ 'get better.' (In these last two and in parallel forms, however, there may be an alternation of /əŷ/ and /i?/ such as mentioned in note 8 for /ə̂y' and /i?/: 'dry something' may be / \dot{c} əŷx *éls/.)

Third, I am sure I have been influenced by the fact that for some of these words there is comparative evidence for /y/. For example, cf. Halkomelem [słénəỷ] ~ [słéni?], Squamish *słánay*², Lushootseed *słádəy*² 'woman.'

Unstressed / ϑ w/, phonetically [ϑ w] ~ [u], occurs quite often as an allomorph of the aspectual prefix / $w\vartheta$ -₂/ 'established.' It appears prefixed to roots as / $w\vartheta$ -/ but can also follow other elements as /- ϑ w/, e.g., following /s-/ 'nominalizer' as / ϑ w/ ([ϑ w] ~ [ϑ w] ~ [ϑ u]), generally translated 'then.' (In AG's speech, / $w\vartheta$ -/ 'established' was never [$w\vartheta$] but invariably [?u] or [?u?].)

Unstressed $\partial w/also appears in one set of demonstratives, /təwネá/ ([təwネá] ~ [tuネá]) 'he, that non-feminine visible one,' /θəwネá/ 'she, that feminine visible one,' etc. And it probably occurs in the prefix /təw-/ 'somewhat.'$

Unstressed /əŵ/, phonetically [əŵ] ~ [o[?]], appears in a number of words like /stáləŵ/ 'river,' / éləŵ/ 'arm, wing,' /k̃*áləŵ/ 'skin,' /łéłəŵ/ 'be escaping,' /łəẁélqł/ 'miss a shot, lose a fish, have one's prey escape.' In this case too, phonemicizing /əŵ/ simplifies description of the morphology. For example, /łéłəŵ/, the progressive of /łéŵ/ 'escape,' is a case of CV reduplication with the second vowel reduced to schwa, paralleling /k̃*áqʷ/ 'get hit,' /k̃*ák̃*əqʷ/ 'be getting hit.' Moreover, for some of these words there is comparative evidence

¹⁰ Transcribing texts dictated by CC, I usually wrote 'and' as [i], occasionally as [?i], and a few times as [?əỷ]. CC's daughter DK went over these texts with me later and expressed the opinion that [i] was correct. Perhaps the form was actually /yi/. In the example sentences, I have tried to leave the first line as I heard it.

for /w/; cf. Halkomelem /stálow/ 'river,' Lushootseed $stúlok^w \sim stúlog^{w}$ - 'river' (cf. Halkomelem $n\partial w$ -, Lushootseed $dik^w \sim dig^{w}$ - 'be/go inside,' Halkomelem /nówo/, Lushootseed $d\partial g^{w}i$ 'you,' Halkomelem /wét/, Lushootseed $g^{w}at$ 'who').

1.2.1. Vowel Length

Vowel length, written /·/, is phonemic in Musqueam. Long and short full vowels (but not schwa) contrast, as in /wí·l/ 'tule' and /wíl/ 'appear,' /té·m/ 'shout,' and /stém/ 'what,' /smé·nt/ 'rock, mountain' and /mén/ 'father,' /pá·t/ 'blow on it' and /ləpát/ 'cup,' /dá·yt/ 'kill him' and /dáy/ 'die,' /tú·x^w/ 'nine' and /pús/ 'cat.'

Phonetically some of the long vowels followed by resonants differ from the others in that the lengthening is shared by the resonant, but these two varieties of long vowels have not been observed to contrast and so they are marked the same way. Probably all of these long vowels are the products of either recent phonological change or ongoing morphophonemic processes. These sources will be discussed in \$1.5.3. and \$1.5.6.

Short vowels can vary somewhat in length. Stressed full vowels before final resonants may appear longer than normal short vowels in deliberate speech, as in $[n \Rightarrow h \acute{a} \cdot y]$ 'It's finished,' but re-eliciting shows these vowels to be short.¹¹

Rhetorical lengthening also occurs in conversation and narrative. A speaker may prolong a vowel in a key word for several times its normal length for emphasis. I mark this /../, for example, /cá··k^w/ 'It's faaar!' (cf. /cák^w/ 'far'), /səw háye..'s təwkaləm x^wíwəl/ 'Then awaaay they went upstream' (cf. /háye?/ 'go away').

1.3. STRESS AND PITCH

Stress consists of an increase in intensity and a rise in pitch. There are three levels of stress: primary (marked /'/), secondary (marked /'/), and weak (not marked). In every full word (as opposed to particles), there is one vowel with primary stress. Its occurrence is not wholly predictable at this time and so it is best marked.

In uninflected words with more than one vowel, the primary stress is generally on the first vowel, as in /léləṁ/ 'house,' /céləx/ 'hand,' /lśwəǎ/ 'rib,' /mɔ́ǹə/ 'child,' /ṫθ́íxʷṫθ̂əxʷ/ 'fishhawk,' /Å ɔ́xʷÅ ɔǎxʷ/ 'oyster.' But there are exceptions, such as /wəq́əq/ 'snail,' /ǎwə?ít/ 'wedge,' /xəmén/ 'enemy.' If the word contains a full vowel and one or more schwas, the full vowel usually bears the stress, as in the last two examples. But again there are exceptions, such as words with final glottal stop, which cannot be preceded by schwa, e.g.,

¹¹ Kava (1969, 39), writing on Cowichan, identifies these as "half-long allophones" of the short vowels.

/nóća?/ 'one.' As will be seen, various morphological processes involve shifts in stress.

Minimal pairs contrasting stress are rare but do exist. When a verb root consisting of a resonant, a schwa, and an obstruent is followed by -t 'transitive,' the primary stress can be on either the root or the suffix, giving us the minimal pair /mớk^wət/ 'salvage it' and /mək^w>t/ 'finish it all.' Compare also /mớd>ot/ 'swallow it' and /mətit/ 'bend it.' For roots with schwa, dispositional-iterative and plural forms may differ in stress only; cf. /sôdsəd/ 'likely to split' and /sədsədd>ot /sədsədd/ 'split it up, split several.' (There may be an intransitive /sədsədd/ 'several split,' but I have not recorded it.) Also, the assimilation of schwa to an adjacent full vowel (discussed in the next section) results in the minimal pair /sté?e/ (~ sté?ə) 'on the way' and /ste?é/ (~ stə?é) 'like, resembling.'

The secondary stress appears often in words composed of a root that has retained the stress and a stressed suffix. I find that in recording such words I have been inconsistent in marking which stressed vowel had the primary stress and which the secondary. For example, I recorded 'help me' as $\dot{c}ew\partial\theta amx$ and as $\dot{c}ew\partial\theta amx$. I believe now that my perception was probably influenced by the sentence intonation pattern. What I recorded as a secondary stress in words like $\dot{c}ew\partial\theta amx$ may have been falling pitch, which seems to be characteristic of the last stressed syllable of a phrase, rather than heightened intensity. But I prefer not to try to determine which is "right" and "correct," so in the illustrating sentences I have left the stresses as I recorded them.

I have not attempted an analysis of the sentence intonation pattern or patterns. This needs to be done.

1.4. PHONOTACTICS

Obstruents (excluding the glottals) commonly follow one another in sequences of up to four. A sequence of five is possible, as in /tx "stx" as 'al' 'just standing in shock.' There seem to be no restrictions on the kinds of sequences of obstruents that can occur. Plosives in sequences are rearticulated. Sequences of /ss/ are not uncommon and are clearly audible, usually as a lengthened [s·].

Resonants occur only adjacent to vowels. An initial resonant must be followed by a vowel and a final resonant preceded by one. Medially, sequences of resonant-obstruent, resonant-resonant, and obstruent-resonant occur.

The laryngeals are more restricted. /?/ occurs only adjacent to a vowel. Within the words, it does not occur following an obstruent other than /s/. It can (and commonly does) occur following /s/, when /s/ is a prefix. It can occur before any vowel but does not occur in final position following a schwa. (Its occurrence with resonants has been discussed in §1.2.) /h/ occurs only before vowels. It can follow a resonant or one of the spirants /s/, /ł/, or /x^w/ at a morpheme boundary, but does not occur following other obstruents. It can appear between an unstressed and a stressed vowel. It does not appear between a stressed and an unstressed vowel. (See §1.5.1 and §1.5.8.)

1.5. MORPHOPHONEMICS

Under this heading, I will discuss alternations that occur fairly commonly. They will also be identified, however, in the appropriate sections on the morphology, as will any alternations of limited occurrence.

1.5.1. Losses in Rapid Speech

In other than slow, deliberate speech, there may be optional loss of some instances of schwa, glottal stop, glottalization of resonant, and /h/.

An unstressed schwa following an initial nasal resonant may be lost. It there is a vowel preceding, the nasal may be heard as part of that syllable, as in /tə nəmén/ ~ /tən mén/ 'my father' (ta 'article,' na- 'my,' mén 'father'), /tə məstáyəx''/ ~ /təmstáyəx''/ the person' (ta 'article,' mastáyax'' 'person').

Occasionally /n = / with no preceding vowel appears as a syllabic [n] (written simply /n/), as in /x = n = /x = n

Both medial and final glottal stop may be lost in rapid or casual speech. A glottal stop between a stressed full vowel and an unstressed vowel (which is likely to be assimilated to the preceding one, as discussed below) can be lost, leaving a long vowel but one with a falling pitch contour that seems distinct from the long vowels identified in 1.2.2, such as $/ma^2 q w/ \sim /ma^2 q w/ \sim /ma^$

A glottal stop after an unstressed final (and so necessarily full) vowel may be lost, in which case the vowel may appear as schwa, as in /méqe?/ ~ /méqe/ ~ /méqe/ 'snow.'

The glottalization of resonants after unstressed vowels in final position is often inaudible, as in /léləm/ ~ /léləm/ 'house.' In rapid speech, it may become inaudible elsewhere after unstressed vowels, as in /sməné·m/ ~ /sməné·m/ 'descendants.'

An /h/ before a stressed vowel may be, but is not invariably, lost when preceded by a spirant, as in /wəłhí θ / ~ /wəłí θ / 'be already a long time,' composed of /wəł-/ 'already,' /hí θ / 'last long'; /shá·ỷ/ ~ /sá·ỷ/ 'finished,' the resultative of /háy/ 'stop, finish.' On the other hand, between a spirant and an unstressed vowel, /h/ is probably never heard, as in /šx "əlí/ 'life,' from /šx "/ 'oblique nominalizer' (here 'means of'), /həlí/ 'be alive.'

1.5.2. Assimilation of Schwa

An unstressed schwa may take on the quality of a full vowel adjacent to it or separated from it by a glottal stop. In the Musqueam of CC and JP, this is usual within the word when the stressed vowel is first. In such cases, the schwa will echo the first vowel, as in 'large bird' and 'black bear' given in §1.4.1. For CC and JP, $/má^{2}aq^{w}/$ and $/spé^{2}e\theta/$ are usual and those with the glottal stop lost are rapid or casual forms. For AG, $/má^{2}aq^{w}/$ and $/spé^{2}e\theta/$ are usual. Within the

root, progressive assimilation of schwa to a following full vowel may be less regular. For CC and JP, /ste[?]é/ may be more usual than /stə[?]é/ 'like,' but I recorded / \check{x} ^wə[?]ít/ 'wedge' in the one form only.¹²

Across morpheme boundaries, the assimilation of schwa to a full vowel is also variable, as in /ni⁹ax^w/ ~/niəx^w/ ~/ni·x^w/ 'that you did' (/ni⁹/ ~/ni/ 'auxiliary,' /-ax^w/ 'you (subordinate subject),' /stə⁹é 'ə tθé⁹/ ~ /ste⁹é 'e tθé⁹/ ~ /steé·tθe⁹/ 'like that' (/stə⁹é/ 'like,' /⁹ə/ 'oblique particle,' /tθé⁹/ 'that'), /tə⁹í/ ~ /ti⁹í/ 'this' (/tə/ 'article,' /⁹í/ 'be here'). The more assimilated forms are heard in more rapid or more informal speech.

There are also instances of a sequence of two schwas realized as $/\epsilon/$. This happens when a word-final schwa is followed by a schwa in a particle (with or without an intervening glottal stop). The result is a stressed full vowel, as in /silé-ł/ 'deceased grandparent,' composed of /sílə/ 'grandparent,' /-əł/ 'past'; /?əw $\epsilon/$ 'isn't? didn't? etc.,' composed of /? δ wə/ 'not,' /?ə/ 'interrogative.' These particular forms and a few others with /-əł/ 'past' seem to be usual if not obligatory. This realization of the two schwas as $/\epsilon/$ suggests the possibility that the final vowel in words like /sílə/ 'grandparent' is basically (or historically) an /e/ reduced to /ə/ because it is unstressed.

There are also a few roots that seem to consist of just a consonant and a full vowel. When such a root is followed by a suffix beginning with an unstressed schwa, the result is necessarily a long vowel, as in $/p\acute{a}\cdott/$ 'blow on it,' from $/p\acute{a}/$ 'get hit by the wind,' /-at/ 'transitive'; $/q^w\acute{e}\cdott/$ 'make a hole in it,' from $/q^w\acute{e}/$ 'get through,' /-at/ 'transitive.'

One root of this shape, θi / 'big,' assimilates to the initial vowel of some suffixes, as in $\theta i \cdot q^w$ / 'big-headed,' from $-aq^w$ / 'head'; $s\theta i \cdot wtx^w$ / 'big house ("smokehouse," winter dance house),' from $-ewtx^w$ / 'house.'

1.5.3. Metathesis of Non-Labial Resonant and Schwa

There are a number of words or forms of words in which there is phonetically a somewhat lengthened vowel followed by a resonant other than /m/ or /w/ that may also be somewhat lengthened or syllabic, as in $[sm \notin nt] \sim [sm \notin nt] =$ /sm $\notin nt$ / 'rock, mountain'; $[\dot{q}\dot{a}\cdot yt] \sim [\dot{q}\dot{a}\cdot yt] \sim [\dot{q}\dot{a}\cdot yt] = /\dot{q}\dot{a}\cdot yt$ / 'kill him.' For such words, there is evidence for a basic shape //CAR \Rightarrow T// (C is any consonant,

¹² Looking at variation such as between [má^γəq^w] ~ [má^γaq^w] ~ [máq^w] or [stə^γé ^γə tθé] ~ [ste^γ·tθé^γ], I decided not to overphonemicize and therefore I have written variants as I heard them. For one reason, dialect and idiolect differences are such that what is an occasional casual form for one speaker may be the usual form for another. For another reason, it may be useful to preserve stylistic differences for the same speaker. In the examples, therefore, I have generally written the assimilated schwas as I heard them in the first line and what I believe to be the basic form in the second line. It might be useful to have some diacritical mark to indicate a schwa that has taken on the value of an adjacent full vowel, as in /má^γâq^w/, /tâ^γí/, and /stâ^γé ^γâ tθé^γ/, but I have not attempted this.

A a full vowel [i.e., i/, e/, a/, or u/], R a resonant, and T an obstruent), indicating a shift in position of the resonant and schwa and the assimilation of the schwa. Perhaps historically the resonant became syllabic before the preceding vowel was affected.

The clearest evidence for this is in the forms of the verb roots of the shape /CAR/ with the suffix /-ət/ transitive, as in /Åé·lt/ 'stick it on, apply it,' from /Åél/ 'get stuck'; / \dot{q} á·yt/ 'kill it,' from / \dot{q} áy/ 'die' (see §7.2.1). The process is also seen in the subordinate passive paradigm; the person markers /-él-/ 'I' and /-ál-/ 'we, you (plural)' and the subordinate passive suffix /-ət/ are realized as /-é·lt/ and /-á·lt/ (see §10.2.6).

This process no doubt also accounts for the long vowels in nouns of the shape /CÁ·RT/, as in /smé·nt/ 'rock, mountain,' /k̃wá·nť/ 'porpoise,' /?é·yt/ 'lingcod.' In the case of /smé·nt/, the morphology shows it; the diminutive /smémnət/ implies a base //sménət// (see §8.4.1). There is also comparative evidence in Northern Straits *sŋénət* 'rock,' *k̃wánət*' 'porpoise,' *?éyət* 'lingcod.'

Although phonetically the resonant may share the lengthening in words like /smé·nt/, there seems to be no ambiguity in phonemicizing them this way, but it might be as defensible to write /smén·t/ or /sménnt/. The lengthening must be marked in some way, however, for it is not simply an automatic phonetic phenomenon in words of the shape /CART/; cf. [mémt] /mémt/ 'pass out food.' Nor is the metathesis automatic in words of the shape /CÁRəT/; cf. /téyəq/ 'get angry,'/sméləkw/ 'warrior's hide shirt.'

1.5.4. Glottalized Resonant Disengagement

When a root of the shape /CA[?] takes the suffix /-ət/ transitive, the resulting form has the shape /CA[?] aRt/; cf. /wíl/ 'appear,' /wí[?] alt/ 'make it appear.' The expectable form would be */wílat/ phonetically *[wí[?]lat], and so it appears that the resonant and schwa have switched positions (as in §1.5.3) but with the glottal stop protecting the schwa from assimilation. This process is also seen in the progressive forms of verbs with initial /n/, as in /néčt/ 'change it,' /né[?] anct/ (from //nénact//) 'be changing it.' The term "glottalized resonant disengagement" is, of course, defensible only if one accepts [?R] as / \mathring{R} /.

1.5.5. Coalescence of Glottal Stop and Resonant

Occasionally a sequence /V⁹VR/ (V is any vowel) undergoes an optional change in which the vowels coalesce and the glottal stop appears as glottalization of the resonant. This can happen to the possessive /⁹ən-/ 'your' after an article; for example, /tə ⁹ən-/ can be realized as /tə ⁹ən-/ ~ /te·n/ ~ /təⁿ/.

1.5.6. Loss of the First of Two Identical Resonants

There are words in which long vowels are followed by final resonants (some glottalized, some not) that can be phonetically longer than usual, as in $[\dot{q}^{w}(\cdot n)] \sim [\dot{q}^{w}(\cdot n)]$ phonemicized $/\dot{q}^{w}(\cdot n)'$ (ear,' and $[sp\dot{a}\cdot l] \sim [sp\dot{a}\cdot l]$ phonemicized

/spá·l/ 'raven.' For these we can infer a basic shape //CÝR ∂R // in which the two resonants are identical, except possibly for glottalization. In some instances, it appears that the stressed vowel is a full vowel, in others a schwa. The process //CÝR ∂R // \rightarrow /Cá·R/ (or /CV·RR/?), like that identified in §1.5.3, may have, historically, involved metatheses of the first resonant and the schwa (with or without a stage in which the resonant and schwa became a syllabic resonant) rather than involving a simple loss of the medial resonant.

One instance of this process appears in the passive paradigm where /-ám-/'you (passive)' and /-əm/ 'intransitive' are realized as $/-á\cdotm/$ (see §10.2.6). Another clear case is $/\check{x}^{w}i\cdot m/$ 'flow fast,' composed of $/\check{x}^{w'} \cdot am/$ 'fast' and /-am/'intransitive.' The source of the full vowel /i/ is not clear, but since schwa does not occur long, a lengthened vowel would have to be one full vowel or another.

Similar bases for $/\dot{q}^{w}$ í \dot{n}' (ear' and $/sp\dot{a}\cdot\dot{l}/$ 'raven' are suggested by two diminutive forms. For 'ear,' JP offered a diminutive, saying that although he had never heard it, it would be $/\dot{q}^{w}\dot{q}^{w}n\vartheta\dot{n}/$ 'little ear.' This form implies a base $//\dot{q}^{w}\dot{n}\vartheta\dot{n}/$. A diminutive plural given by CC, /sp ϑ l ϑ l) 'lots of little crows,' implies a singular diminutive (not recorded) */sp ϑ l), which in turn implies a base //sp ϑ l).

Comparative evidence from within Halkomelem supports the case for this process; the Nanaimo dialect has medial resonants in several of these words where Musqueam and other dialects have long vowels (see Elmendorf and Suttles 1960, 32).

This process may also occur in instances of /l/ followed by /ł/, as in /spé-łx̆ən/, probably //spáləłx̆ən// 'prairies,' the plural of /spáłx̆ən/ 'prairie' (see §8.3.4).

1.5.7. Emergent /θ/

Historically, Halkomelem $/\theta$ / is the reflex of an earlier *c. The lexicon seems pretty consistently to reflect this. In two places in the morphology, however, there is a $/\theta$ / that is probably the result of a coalescence of *t and *s. In the transitive verb paradigm, $/\theta$ / appears as the realization of /-t/ transitive and a firstand second-person object marker that I identify as //S// (see §10.2.5). In the possessive paradigm in JP's and AG's speech (but not in CC's and DK's), $/? \partial \theta - /$ appears as the realization of an allomorph of the second-person possessive that I identify as $//? \partial T - //$ and an initial /s/ in the noun or nominalization (see §10.2.4).¹³

1.5.8. Intrusive /h/

There are a few roots that appear alone as CA, as in $/\theta i/$ 'big,' $/q^{w} e'/$ 'pass through,' /pa'/ 'get blown on.' When such a root is followed by a suffix beginning

¹³ My model for this use of morphophonemic symbols was William H. Jacobsen's treatment of Washo morphophonemics in his course in the language at the University of Nevada.

with a stressed vowel, an /h/ appears, as in / θ =hín=s/ 'barrel-chested' (< /-ín=s/ 'chest'), /qw=híl=m/ 'go through' (< /qwé/ 'pass through,' /-íl=m/ 'move toward'), /sp=héls/ 'wind' (< /s-/ 'nominalizer,' /pá/ 'get blown on,' /-éls/ 'activity'). It may be that these roots have an underlying CAC shape (a very common one) in which the final /h/ is not realized after the stressed vowel (a position in which it cannot occur); for example, // θ (h// is realized as / θ (/ 'big.' Or it may be that /h/ is simply automatic when an unstressed vowel is followed by a stressed vowel. (For possible CAh roots, see §7.2.4.)

1.5.9. Alteration of /n/ and /l/

A few suffixes with initial /n/ have forms with initial /l/ when added to a root or stem ending in /l/. The transitive suffix $/-n \Rightarrow x^w/$ 'limited control' appears as /-ləx w/ in /łėdelləx w/ 'know it' and /cellex w/ 'catch up with him' (cf. /celt/ 'follow him'). But this may vary with speakers; JP said /tállax w/ but CC said /tálnəx^w/ for 'understand it, learn about it' (cf. /tál/ 'be understood, be settled'). The suffix /-namet/, the reflexive of /-nex^w/, appears as /-lámet/ in /k^w'é·llàmət/ 'manage to get a hideout' (< /k^wé·l/ 'hide'; cf. /k^wélx/ 'hide him'). The applicative /-nəs/ 'goal' appears as /-ləs/ in /?éyələs/ 'leave him' (cf. /?éyəl/ 'get out of the way'). And the lexical suffix /-nəc/ appears as /-ləc/ in /słállac/ 'bum' (from /łal-/ 'behind'), /s?állac/ 'stump, base of standing tree' (from /?əl-/ 'dummy root'), and /šx^wÅə́pəlləc/ 'tail' (cf. /Åə́p/ 'deep.' /Åpíl/ 'go downward'), and as /-allac/ (probably composed of a connective /-al-/ and $/-n \Rightarrow c/$) in $/x^w \Rightarrow d^w \Rightarrow 11 \Rightarrow ct/$ 'cut its tail off' (from $/t \Rightarrow d^w/$ 'get cut off'). There is also a form /-ələc/ in /x "?ə́pələcəm/ 'wipe one's bum.' The lexical suffix for 'tooth' has the forms /-nís/, /-nəs/, and /-áləs/, the last in /θiθáləs/ 'big teeth,' /Åəlqtələs/ 'long teeth,' etc. The Cowichan form corresponding to /-ələs/ is -əlnəs (Hukari and Peter 1995, 318), suggesting that the Musqueam /-nəs/ became assimilated to the connective /-al-/.

1.5.10. Alternation of /c/ and /k^w/ and of /x/ and /x^w/

In the progressives and resultatives of a few verbs with initial /c/ or /x/ followed by /a/, the /c/ is reduplicated as /k^w/ and the /x/ as /x^w/. This is seen in /cám/ 'go/come inland' and its progressive /cák^wəm/ 'be going/coming inland,' in /xák^w/ 'get bathed' and its progressive /xáx^wk^wəm/ 'be getting bathed,' and in /x^wxápəθət/ 'sit cross-legged' (from the root \sqrt{xap}) and the resultative /sxáx^wəp' 'seated cross-legged' (see also §7.2.1.1). This irregularity is probably the result of historical change. The process must have been:

*kúm > *kúkum > *kúk^wum > *čúk^wəm > cák^wəm

First, the root was reduplicated by a grammatical rule. Then, the second */k/ was labialized under the influence of the preceding /u/. Next, the initial */k/ was palatalized to become */č/, and perhaps at this point the second vowel was

reduced to /a/. In this process, the second */k/ was protected from palatalization by labialization. Finally, the */u/ became /a/, and the */č/ became /c/.

There is one apparent root with /c/ that reduplicates as $/k^w/$ in which the following vowel is not /a/. This is the $/c_{9}-/ \sim /c_{e}-/$ that appears in several interrogative words (see §17.7-17.13), as in $/x^wc\acute{e}l/$ 'go where' and its progressive $/x^wc\acute{e}k^w$ əl/ 'be going where.' This irregularity must be the product of a more complex history.

1.5.11. Vowel Gradation

Alternation of a full vowel, schwa, and zero, depending on the type of root or stem, type of suffix, and placement of stress, occurs widely in the morphology and will be described there, where relevant.

1.5.12. Vowel Mutation

The joining of some suffixes to some stems results in a change in the quality of the stressed vowel, from one full vowel to another, in the stem, or, rarely, in the suffix. Three kinds of mutation of this sort have been discovered. Only the first occurs commonly.

(1) Stem /e/ to /a/. There are several suffixes that have this umlaut effect, causing an /e/ in the stem to appear as /a/. These include both grammatical and lexical suffixes. For some the effect seems quite regular, but for others only a few mutated forms have been recorded. These suffixes are listed below with a few examples showing the mutation. The first three are grammatical suffixes that seem to have this effect quite regularly.

- <i>θət ~ -θát</i>	'oneself'
ṫá?θət	'test oneself' (cf. $\dot{t}\dot{e}^{\gamma}t$ 'test him, try it')
páθəθət	'spread out' (cf. <i>péθət</i> 'spread it out')
<i>k</i> ʷłáθәt	'capsize' (cf. <i>k^wtét</i> 'spill it, tip it over')
θáỷθət	'be fixing oneself' (cf. $\theta \delta y \theta \partial t$ 'fix oneself,' $\theta \delta y t$ 'fix it,' $\theta \delta y t$
	'be fixing it')
náčθət	'become different' (< $n\acute{e}c$ 'differ, be different')
si?áṁθət	'become rich' (< <i>si²ém</i> ' 'rich, upper-class person')
-təl ~ -tál	'each other'
ťəťá təľ	'keep testing each other' (cf. $\dot{t}\dot{e}^{2}t$ 'test him')
nəċnáċtəĺ	'differ from one another' (< <i>néč</i> 'differ')
<i>ċłmántə</i> ľ	'half-sibling with the same father' (< <i>ct</i> - 'fellow-, co-,' <i>mén</i>
	'father')
-a?ł	'attributive'
snáča?ł	'different, somebody else's' (< s- 'nominalizer,' néč 'differ')
si?áṁa?ł	'wealthy person's' (< <i>si²ém</i> ' rich, upper-class person')
syəwána?ł	'old-time, ancestral' (cf. syawén' 'ancestor,' yawén' 'before')

There are two combinations of grammatical suffixes that have this umlauting effect, at least in some words. These are the passive endings $/-n \Rightarrow m/$ (the passive of $/-n \Rightarrow x^w/$ 'limited control transitive') and $/-st \Rightarrow m/$ (the passive of $/-st \Rightarrow x^w/$ 'causative'). We see this effect in the following:

Xán əm	'be brought back as a bride' (cf. $\dot{\lambda} \dot{e}^2 t$ 'go and get him, invite him')
həyá?stəm	'be taken away' (< $h \partial y \dot{e}^{2} st \partial x^{w}$ 'take him away,' < $h \dot{\partial} y e^{2}$ 'leave')
nə ⁹ áməstəm	'be taken away' (< $n\partial^2 \acute{e}me\partial st\partial x^w$ 'take him away,' causative of
	ném 'go')
xtáməstəm	'be swum away with' (< xtémostox ^w 'swim away with it,' xtém
	'swim')

Three lexical suffixes have this effect. Examples with the first are numerous, with the second fewer, and with the third only the one given.

-ás ~ - <i>əs</i>	'face, round object'
x ^w Žáqtəs	'long-faced' (< Žéqt 'long')
?isáləs	'two dollars' (< [?] isélə ~ yəsélə 'two')
pá [?] aθəs	'bear mask (a type of <i>sxwayxwey</i>)'
	$(\leq sp e^{2}e\theta$ 'black bear')
$aq^w \sim -\partial q^w$	'head'
Žáqtaq ^w	'high (of a mountain)' (< Žéqt 'long')
Žəláqtaq"	'long-heads (i.e., the Kwakiutl with
	Quatsino-style head-lengthening)'
<i>yesállaq</i> ^w (probably <i>yəsálaq</i> ^w)	'two-headed' (in a translation of a
	Snohomish song) (JP) (< yəsélə 'two')
-əłcə	'water'
$ heta \hat{a}^{2}t \hat{a}^{2}c \hat{a}$	'dark tide (i.e., a low tide on a moonless night)' ($\langle \theta \ell^2 t$ 'dark')

(2) Stem /a/ to /e/. This has been recorded in one word only, /pé·lť^θe[?]/ 'buzzard (turkey vulture),' which appears to be composed of the root of /spá·l/ 'raven' and the suffix /-it^θe[?]/ 'clothing, blanket.'

(3) Suffix /e/ to /a/. This has been recorded in one word only, /sqੈəq́əxán/ 'partner,' which is composed of /sqੈəq́a?/ 'accompanying' and the suffix /-xən ~ -xén/ 'foot.'

These vowel mutations are no doubt the product of the assimilation of one vowel to that of an adjacent syllable at an earlier stage in the history of the language. In the first and second types, the vowel mutation is comparable to the umlauting effect of a suffix on the stem in Germanic languages, while in the third it is comparable to the effect of the stem on the suffix in the vowel harmony of Uralic and Altaic languages.

1.6. SPECIAL FORMS OF SPEECH

I am aware of some distinctive features that are heard in songs, baby talk, and the speech of a few animal characters in stories. I do not have many examples, however, and so there may be other features I have not recorded.

In songs, to judge by a very few examples, vowels may be lengthened and to some extent changed in quality. In a song composed for a $s\check{x}^w \acute{a}y\check{x}^w \partial y$ performance, the phrase $t\partial n\partial m\acute{e}m\partial i$ 'my dear father' appears as $te \ neme m\partial i$. In this song, the compound prefix $\check{s}x^{w-}$ 'oblique nominalizer' appears as $s\partial x^{w-}$. I do not know whether this is an older pronunciation of the prefix or an instance of a schwa being introduced simply to separate the consonants.

Adults speaking to babies, as mentioned in §1.1.1, will substitute the rare unrounded velars for unrounded uvulars, as in $k\dot{a}^2$ for $q\dot{a}^2$ 'water' and in /ké⁹ek/ for /qé⁹eq/ 'younger sibling' used affectionately. AC's much older half-brother addressed him this way throughout life. The /k/ also substitutes for /q/ in /skánək/ for /sq̃ánəq/ 'dear.' A Tsawwassen woman is said to have been called this as a pet name by her family all her life. I have recorded /k/ in only one word, /skíkənək/, the baby-talk form of /sq̃íqənəq/, the diminutive of /sq̃ánəq/ 'dear.' A Musqueam man is said to have been called this as his ordinary name, and a younger sister might be called this.

In stories told by James Point, the characters Raven and Mink both begin words, generally the initial words in a clauses, with \check{x} -. He also said that an initial \check{x} - was used in talking to dogs, adding, "A trained dog is supposed to talk that way; it will show it's teeth and laugh ' \check{x} ^{...}, \check{x} ^{...}, \check{x} ^{...}." A niece talked to her dog Daisy that way, saying " \check{x} stém $k^{w}ee\theta \check{X}i$? ('What do you want?')," and when offering food, " \check{x} $\check{X}\dot{a}$ wəls $\check{c}x^{w}$ yəwen ('Bark first')."

The very few examples I have of spells (verbal formulae with inherent power) suggest that while they contain esoteric vocabulary, there is nothing distinctive in their phonology.

2 Synopsis of Morphology

2.1. WORDS

A word may consist of a root standing alone and unaltered, or of a root altered by one or more processes of internal modification or accompanied by one or more affixes or both. We can distinguish several classes of words. Verbs (§7) and nouns (§8) constitute two large open sets, adjectives (§9) a smaller open set, and adverbs (§18) a still smaller one. There are also several small closed sets: personal words (§14.2.7), possessive words (§14.2.8), demonstratives (§15.2), interrogative words (§17), numerals (§19), and exclamatory words (§20). Four (lexical) verbs also serve as auxiliary verbs (§3.2.1), and several verbs also serve as prepositions (§3.1).

Except for a few adverbs, all words can function as predicate heads and so there is no basis in this function for distinguishing verbs, nouns, and adjectives. Nevertheless, there are other bases for distinguishing these classes and not doing so would, I believe, needlessly mystify the language, making it seem more exotic than it really is. The most obvious differences are these: verbs have progressive forms and (with a few exceptions) cannot take possessive affixes, nouns do not have progressive forms and do take possessive affixes, while adjectives neither have progressive forms nor take possessive affixes. Verb roots and noun roots also have different canonical shapes.

Compounding seems rare or non-existent. For possible examples, see 12.3.3 and 13.3, and for noun modifiers of nouns, see 4.1.3.¹

2.2. ROOTS AND THEIR INTERNAL MODIFICATION

The great majority of verb roots have the shapes CAC, C₂C, and C₂CC (C is any consonant and A is any full vowel), while noun roots most commonly have

¹ A few recent coinages that are constructed from borrowed elements or are loan translations may perhaps be identified as compounds, such as *lamatú tíntan* 'sheep bell' (CJ *lamatú* 'sheep,' *tíntan* 'bell') (see §4.1.3), *pípa téla* 'paper money' (CJ *pípa* 'paper,' *téla* 'money, dollar'), *číkman téla* 'coins' (CJ *číkman* 'metal, iron'), *kalísmas θqét* 'Christmas tree' (*θqét* 'tree').

the shape CVCVC (V is any vowel), but show a greater variety than verb roots. Adjective roots are more like verb roots in that the most common shapes are C = C and CAC, but there is some variation.

We can identify roots as verbal, nominal, and adjectival, but we must note that there is a prefix that nominalizes verbs and adjectives, there are several prefixes that make verbs of nouns, and there are several ways of making adjectivelike words out of nouns.

Processes of internal modification of the root include reduplication (of initial CV and CVC), infixation, shift in stress and vowel grade, and glottalization of resonants. The first two affect the root only; the last also affects suffixes. These processes express progressive and other aspects, plural, and diminutive. Overall, there is no one-to-one relationship between process of internal modification of the root and grammatical form. Roots of different shapes can undergo different processes to produce grammatically identical forms. It appears that most forms can be predicted from the shape of the root, although there are also many irregular forms.

Verb roots can be internally modified for progressive, durative, iterativedispositional, and resultative aspects, for the plural, for the diminutive, and for various combinations of these. Nearly all verbs have progressive forms. The verb root is identifiable as perfective, as opposed to progressive, aspect (compare the perfectives $p e t^{\theta}$ sew, $s \neq q$ split, tear, and $k^{w} \neq t$ spill, flow with the progressive $p e p a t^{\theta}$ be sewing, 'sasáq' be splitting, be tearing,' and $k^{w} a k^{w} a t$ 'be flowing'). The choice between perfective and progressive forms is obligatory. A number of verbs also have a durative aspect, which can appear in both perfective and progressive forms (compare $dik^w \partial t$ 'bite it' and $did\partial k^w \partial t$ 'be biting it' with $d\partial dk^w et$ 'hold it in the teeth' and $di dk^w et$ 'be holding it in the teeth'). A number also have an iterative-dispositional aspect (as $s \neq d \neq s \neq d$ 'easy to split,' $k^{w} \delta t k^{w} \delta t$ 'cranky, easily tipped over,' $s d \delta k^{w} d \delta k^{w}$ 'prone to bite'), which for a few roots appears in a progressive as well as a perfective form. Most verbs have resultative forms (as $sp e \vec{p} a \vec{t}^{\theta}$ 'sewed,' $ss as i \vec{q}$ 'spit, torn,' and $s \vec{k}^{w} a \vec{k}^{w} i \vec{t}$ 'spilled, capsized'), which are adjective-like and do not have the perfective-progressive distinction. The plural can be marked, optionally, in nearly all of these forms, that is, there can be plural perfective, plural progressive, plural durative, plural iterative-dispositional, and plural resultative forms. The diminutive and diminutive plural are also optionally marked, but only in the progressive and resultative aspects. The internal morphology of the verb is presented in §7.

Noun roots can be internally modified for the plural, the diminutive, and the diminutive plural (compare *céləx* 'hand,' *cəlcéləx* 'hands,' *cécləx* 'little hand,' *cəlécləx* 'lots of little hands'). A few have what seem to be resultative forms. As nouns, they to not have progressive forms, but a noun may be made into a verb by means of a verbalizing affix and then have a progressive form. The internal morphology of the noun is described in §8.

Adjective roots, like noun roots, can be internally modified for plural, diminutive, and diminutive plural. They have progressive forms only if made into verbs through an affix. Their internal morphology is described in §9.

A small set of roots that do not undergo any internal modification can be identified as adverbs. These and a number of other words and phrases that perform adverbial functions are described in §18.

A few of the demonstrative words, personal words, interrogative words, and numerals also undergo internal modification.

2.3. AFFIXES

Strictly speaking, there are prefixes, suffixes, and infixes. However, I have chosen to describe the infixes, along with reduplication and other processes, under the heading "internal morphology of the root," as described above. This leaves prefixes and suffixes to be accounted for under "affixes." In theory, affixes should also be divisible into inflectional or derivational, depending on their involvement in paradigms, and grammatical or lexical, depending on their meaning. In reality, however, a number of Halkomelem affixes mix these categories. For descriptive purposes, I have simply divided the affixes into nonpersonal affixes (§11-13) and personal affixes. The latter are included in the person system (§14).

The non-personal affixes are described under four headings: suffixes of the voice system, aspectual and modal affixes, derivational affixes, and lexical suffixes.

Suffixes of the voice system (see §10) include transitive, intransitive, causative, permissive, applicative, reflexive, reciprocal, and subordinate passive.

Most verb roots are semantically inactive or patient-oriented (i.e., they have glosses like 'get hit,' 'get buried,' or 'get washed,' there being no roots with glosses 'hit,' 'bury,' or 'wash'), but there are a few that are semantically active or agent-oriented (e.g., 'look,' 'seek'). All are grammatically intransitive. They take subjects only.

With the suffixes of the voice system relations are different. A verb composed of an inactive root and an intransitive suffix is still grammatically intransitive but is semantically active (agent-oriented) and can take an oblique object. A verb composed of either an inactive or active root and transitive suffix is grammatically transitive and can take an object. The transitive suffix also provides the necessary base for an object or passive person suffix. The applicatives change the valence of the verb root, allowing what would have been in an oblique relationship to it to be its grammatical object.

The two most commonly used transitive suffixes, as well as their reflexive counterparts, distinguish actions performed with limited control or done accidentally from those performed with full control or purposefully.

The aspectual and modal affixes (§11) include aspectual prefixes and modal suffixes. The aspectual prefixes (§11.1) are a small set of elements with aspectual or adverbial meaning that can precede a predicate head and that express mainly temporal distinctions such as 'already,' 'still,' 'habitually,' but also 'become' and 'somewhat.' One of these, $w\partial_{2}$, tentatively identified as 'established,' occurs very often and in a variety of ways that still present problems in analysis. (Further analysis may also identify some or all of these as particles rather than prefixes.)

The modal suffixes (§11.2) are a still smaller set of elements that can follow suffixes of the voice system. The only two recorded indicate desire or intention and search or arrangement.

The derivational affixes (see §12) include some that have purely grammatical meaning, simply converting a word from one class to another; some that combine grammatical and lexical meaning; and others that have purely lexical meaning. On the basis of form, function, and meaning, we can distinguish a small set of purely grammatical affixes, a small set of verbalizing affixes, and a small set of lexical prefixes.

The purely grammatical affixes (§12.1) include a nominalizing prefix, a prefix that appears in resultative forms of the verb, a prefix that seems to show an oblique relationship between a verb root and lexical suffix, a compound of the nominalizer and the last, and a suffix that forms adjective-like words. Of these, the prefix *s*- 'nominalizer' deserves special mention here. It appears as an essential part of a number of nouns and appears prefixed to verbs and adjectives, which can thereupon take possessives. But it does not function simply as a component of nouns. It plays a role in the formation of one type of relative clause (§4.1.1.2) and in the formation of nominalized clauses (§4.3). The compound prefix δx^{w} - 'oblique nominalizer' plays parallel roles. In their roles as clause nominalizers, *s*- and δx^{w} - may also be identified as particles.

The verbalizing affixes (§12.2) combine this function with lexical meanings, as *c*- 'get, make, do, go to,' *t*- 'partake of,' tx^{w} - 'buy,' and $-\partial l \sim -il$ 'move toward, become,' $-\dot{a}\cdot t$ 'travel by,' $-i\cdot m$ 'die from.' The lexical prefixes (§12.4) include $t \partial m$ - 'time of,' $-t \partial n$ 'originating from,' and $\dot{c}t$ - 'fellow.'

Logically the lexical suffixes should be included among derivational affixes, but they form such a large class that I have given them a separate section (§13). They are nominal in meaning, referring to body parts (as $-\partial s \sim -as$ 'head,' $-aq^w$ 'head,' $-c\partial s$ 'hand'), common artifacts (as $-\partial wtx^w \sim -ewtx^w$ 'house,' $-w\partial t$ 'canoe'), natural phenomena (as $-\partial t p$ 'plant, tree,' $-\partial n\partial p$ 'ground'), and a variety of other things. They are often joined to the root with a connective element, which may give the suffix a narrower meaning. A lexical suffix can be related to a verb root as object, locus, or instrument; to an adjective root as noun head; or to noun root as noun possessor or as noun head of noun modifier. Words formed with lexical suffixes can be verbs, adjectives, or nouns.

2.4. THE PERSON SYSTEM

The personal affixes are only a part of the personal system, which is presented as a whole in §14. The language distinguishes first, second, and third persons in singular and plural. (There are neither dual forms nor inclusive/exclusive distinctions.) These persons are expressed in two sets of words, in particles of two sorts, and in affixes of six types.

For the first and second persons, particles and suffixes distinguish subjects and objects and, among subjects, non-passive from passive and main-clause from dependent-clause subjects. For third person, in main clauses a suffix marks transitive subjects while intransitive subjects and objects are unmarked, but in dependent clauses both transitive and intransitive subjects are marked.² There is also a set of possessive affixes; these are prefixes for first and second person singular, suffixes for first-person plural and third person, and a combination or prefix and suffix for second-person plural. A particle marks plurality for third-person subjects, objects, and possessors.

The two sets of words are personal words and possessive words. The personal words (§14.2.7) can serve as predicates with the sense 'It is I, I am the one who,' etc., and as oblique nominal adjuncts comparable to English personal pronouns as objects of prepositions. The possessive words (§14.2.8) can serve as predicates with the sense 'It is mine,' etc.

2.5. THE ORDERING OF AFFIXES

Derivational prefixes and suffixes form an inner layer around the root, while inflectional affixes form an outer layer. Among derivational affixes, those with lexical meaning stand closer to the root than those with purely grammatical meaning. Among inflectional affixes, those of the voice and person systems stand closer than the aspectual prefixes and modal suffixes.

Preceding the root, the verbalizing and miscellaneous lexical prefixes (derivational with lexical meaning) stand closest to it. If a word formed with a lexical prefix is a noun, a first-person singular or second-person possessive prefix (inflectional) can precede the lexical prefix, as in:

(a) nəċłx wálməx w

nə-ċł-x "ə́lməx " my-fellow-people 'my fellow villagers, my fellow Indians'

If a verb formed with a verbalizing prefix is nominalized, the nominalizer (derivational with purely grammatical meaning), symbolized NOM, precedes the verbalizer and it can be preceded by a possessive, as in:

² Thus as Gerdts (1981) has pointed out, the language is split in its case system, making a nominative-accusative distinction for first and second persons but an ergative-absolutive distinction for third person in main clauses.

(b) nəsclá?θən
 nə-s-c-lá?θən
 my-NOM-make-plate
 'something to serve as my plate'

If a word with a possessive prefix is a predicate, it can take an aspectual prefix, as in:

(c) x^wə²əθwé²ələp.
 x^wə-²əT-s-wé²-ələp
 become-your-NOM-own-your (pl)
 'It has become yours.'

Following the root, the lexical suffixes (derivational) appear first (with or without connective elements). A noun formed with a lexical suffix can have a possessive suffix following it, as in:

```
(d) x "əlməx "éwtx "ct
x "álməx "- éwtx "- ct
Indian-house-our
'our "smokehouse""
```

A verb formed with a lexical suffix can take suffixes of the voice system followed by suffixes of the personal system, as in:

(e) x^wq^wénəctəs.

x^w-q^wé-nəc-t-əs inward-penetrate-bottom-TR-3TR '[She] punches holes in the bottom of it.'

Passive forms are exceptional in that a suffix of the voice system stands on either side of the personal suffix, as in:

```
    (f) x<sup>w</sup>t<sup>θ</sup>íq<sup>w</sup>əsnèləm.
    x<sup>w</sup>-t<sup>θ</sup>íq<sup>w</sup>-əs-n-el-əm
    inward-get.hit-face-TR-I.PASSIVE-INTRANS
    'I was (accidentally) hit on the face.'
```

2.6. AMBIVALENT AFFIXES

Five morphemes, the possessives na- 'my' and 2an- 'your,' the aspectual prefixes wa-2 'established' and wat- 'already,' and the *s*- 'clause nominalizer' (see below) deserve a special comment. I have identified the first four as prefixes, and in the second line of the examples I treat all five as such. However, phonetically they may, but do not necessarily, become attached to preceding morphemes. Where I recorded this, it appears in the first line of the examples.

A possessive may become attached to a preceding article, as in (a).

```
(a) tən mén
tə nə-mén
ART my-father
'my father'
```

One of the aspectual prefixes may become attached to a preceding clause nominalizer, as in (b).

```
(b) səw néms.
```

s-wə-ném-s NOM-EST-go-3POS 'Then he went.'

The clause nominalizer may become attached to a preceding article, as in (c).

```
(c) k<sup>w</sup>s néms
k<sup>w</sup>ə s-ném-s
ART NOM-go-3POS
'when he went'
```

For more complex clusters, see §4.3.4.

2.7. PARTICLES

What distinguishes a particle from an affix is the fact that it is the partner not of a single root or stem, as an affix is, but of a larger construction such as a predicate, a nominal adjunct, a clause, or a sentence. Particles are also movable and separable from the heads of the constructions they relate to, while affixes firmly adhere to their roots or stems. With few exceptions, the particles are unstressed. Phonetically most seem separate from adjacent words (and are written so), but a few seem phonetically like affixes (and are written so). Particles can also form clusters that seem to be phonetic units.

On the basis of the constructions they relate to, we can distinguish particles that relate to predicates and those that relate to nominal adjuncts.

The particles that relate to predicates include clause-initial particles, secondposition particles, and sentence-final tags.

There are four clause-initial particles. These introduce two kinds of embedded clauses. Two are the clause subordinaters, $w\partial_{7}$ 'if, when, that' and $?\partial t$ -'whenever, whatever, that.' These can precede the first word in a clause, marking it as a subordinate clause. If the subordinate clause is non-passive, its subject is expressed by a subordinate subject suffix, if passive by a subordinate passive form. (See §4.2, "Subordinate Clauses.") The other two clause-initial particles are *s*- 'nominalizer' and δx^{w} - 'oblique nominalizer' in their roles as clause nominalizers. Either of these can precede the first word in a clause and serve to nominalize the whole clause. If the nominalized clause is non-passive, its subject is expressed by a possessive; if it is passive, the subject is expressed by a subordinate passive form. (See §4.3, "Nominalized Clauses.") There are about twenty second-position predicate particles. Some of these are introduced in §3.1.2. For details, see §16. Usually these appear after the first word in the predicate. Four distinguish first- and second-person singular and plural subjects in non-passive main clauses. Another, somewhat freer in its possible positions, is the third-person plural marker. These five are dealt with primarily in §14, "The Person System." Of the non-personal second-position predicate particles, two mark tense, past and future. Most of the others are modal, distinguishing questions, requests, exclamations, quality of information, and so on. These and the sentence-final tags are described in §16.

The nominal particles include the articles and the oblique case marker. The articles include (1) a set of six members (with some variants) identifiable by the intersection of two distinctions: gender (feminine versus non-feminine) and position (present and visible, nearby but invisible, remote or hypothetical); and (2) a single article marking a proper noun in the oblique case. The articles serve to introduce nominal adjuncts, and they enter into the formation of several sets of demonstratives that can also introduce nominal adjuncts or stand as nominal adjuncts themselves. The articles, as well as the demonstratives, are described in §15.

The oblique particle 9 serves as a case marker introducing oblique nominal adjuncts (see §3.4) and one kind of noun possessor (§3.8.1).

Finally, there is one conjunction, ${}^{2}\partial y' \sim i$ 'and, but,' which links both coordinate nominal adjuncts (§3.8.4) and coordinate clauses (§5).

3 Syntax 1: Simple Sentences

3.1. THE MINIMAL SENTENCE

A sentence consists minimally of a predicate. A predicate may consist of a single word.

- (a) θí. big 'It's big.'
- (b) ⁹é·nθə. be.I 'It's me.'
- (c) ném. go 'Go!'

Such predicates are probably not very common, except perhaps as commands, as in the last instance. More commonly, this word, now identifiable as the predicate head, is followed by one or more second-position predicate particles (see §3.1.2 below).

- (d) ném łə. go PER 'Go!'
 (e) ném ċə. go QUOT 'He is said to be going.'
 (f) ném cən ce?.
 - go I FUT 'I'll go.'

- (g) ném ?ə čx w k wə. go ROG you then 'Will you go then?'
- (h) k^wáq^w čx^w ce?.
 be.hit you FUT
 'You'll get hit.'
- (i) cák^w də. far EMPH 'It's far!'
- (j) spé²eθ ²ə. black.bear ROG 'Is it a bear?'
- (k) spé⁹eθ ṁ́ə.
 black.bear CERT
 'It's (certainly) a bear.'
- (l) spé²eθ ċa.
 black.bear QUOT
 'It's said to be a bear.'
- (m)spé⁹eθ yəx^w m.
 black.bear INF CERT
 'It must be a bear (even though we cannot see it).'
- (n) swóỷqe? cən. male I 'I am a man.'
- (o) stém ce? k^wə.what FUT then'What will it be (then)?'
- (p) c-yówon ce? mo.
 VERB-sing.possessed FUT CERT
 'He's going to sing his syówon [possessing song].'
- (q) čéwət cən ce?. čéw-ət cən ce? help-TRANSITIVE I FUT 'I'll help him/her/them.'
- (r) čéwəθəs ce?.
 čéw-ət-S-əs ce?.
 help-TR-me-3TR FUT
 'He will help me.'

- (s) k^wəcnéləm ce? m.
 k^wec-n-él-əm ce? m.
 see-tr-I(PAS)-INTR FUT CERT
 'I will be seen. They (somebody) will see me.'
- (t) k^wóxnəct čx^w me.
 k^wix-nəc-t čx^w me.
 name-base-TR you CERT
 'Name a price.'
- (u) nównoct ło do.
 nów-noc-t ło do.
 insert-base-TR PER EMPH
 'Pay him!'

(Note that in (s) and (t) the full vowels of $k^w \acute{e}c$ 'look' and $k^w \acute{i}x$ 'name' are reduced to schwa when the roots are followed by the suffixes *-n-* 'transitive' and $n \partial c$ 'base.')

3.1.1. Predicate Heads

Predicate heads can be bare roots like θi 'big,' $c a k^w$ 'far,' $k^w a q^w$ 'get hit,' derived forms like $s p e^2 e \theta$ 'black bear,' $c y \delta w a \eta$ 'sing a possessing song,' inflected forms like c e w a t 'help him/her/them,' $c e w a \theta a s$ 'he/she helps me,' and $k^w a c n e l a m$ 'I am seen,' and forms with both derivational and inflectional affixes like $k^w \delta x n a c t$ and $n \delta w n a c t$ in the sentences above. (These affixes will be identified later.)

Predicate heads can also be words definable morphologically as verbs ($n\acute{em}$ 'go'), adjectives ($\theta\acute{t}$ 'big'), nouns ($sw\acute{s}\acute{y}qe^2$ 'man'), or members of the closed sets of personal words ($2\acute{e}\cdot n\theta \partial$ 'be-I, me') and interrogative words ($st\acute{em}$ 'what'). In fact, perhaps there are only a few adverbs that cannot serve as predicate heads.

While members of these several classes of words can all serve as predicate heads, it is nevertheless useful to distinguish verbal, adjectival, and nominal predicates. In expanded predicates, members of these classes enter into somewhat different sorts of relations with other elements. And because of analytical problems that adjectival and nominal predicates pose, it seems best to postpone consideration of them until we have considered expanded verbal predicates and their nominal adjuncts. But first let us look at the second component of most minimal sentences.

3.1.2. The Second-Position Predicate Particles

The second-position predicate particles form a closed set of twenty or so members. Many of them can co-occur and do so in a fixed order. A fuller discussion, with examples, appears in §16. Here we need only note those that occur most commonly. The second-position predicate particles include four person markers

 $c \partial n$ 'I'ct 'we' $\check{c}x^w$ 'you (singular)' $ce \cdot p$ 'you (plural)'

which mark first- and second-person subjects of non-passive (intransitive and active transitive) predicates in main clauses. In the absence of any of these or the imperative particle, the subject of a non-passive main clause is understood to be third person. They also include

⁹é·*ltən* 'third person plural' (3PL)

which optionally marks plurality for a third-person subject (or object or possessor). The second-position predicate particles also include two tense markers

 $-\partial t$ 'past' ce^{2} 'future' (FUT)

and a number of modal particles, marking questions

?∂ 'interrogative' (ROG)

requests and instructions

 $te \sim t \partial$ 'imperative' (PER)

degree of certainty and quality of information

$\vec{m}e \sim \vec{m}\partial$ 'certainty' (CERT)	$y \partial x^w$ 'inference' (INF)
$\dot{c}\partial$ 'hearsay, quotative' (QUOT)	<i>ċtwa</i> [?] 'speculative' (SPEC)

relationship to old information or assumptions

 $k^{w}\partial$ 'then' $\dot{q}\partial$ 'emphatic' (EMPH) $\partial\partial l$ 'though, but, adversative' (ADV) ' $a\dot{l}$ 'just, only'

and others, many of them difficult to gloss.

As examples in the following sections will illustrate, the second-position predicate particles, except for 2al 'just' (see §16.2.8) and 2ettan 'third-person plural' (see §14.2.9), always follow the first word in a predicate. When a predicate is expanded by the addition of one or more words before the head, the particles are shifted in position so as to follow the first word.

3.2. EXPANDED VERBAL PREDICATES

A verbal predicate may be expanded by the addition of one or two auxiliary verbs before the head and/or one or more adverbs either before or after the head. We also find verbs with verb complements and compound verbs.

3.2.1. Auxiliary Verbs

Two pairs of verbs function as auxiliaries, a locative pair, ${}^{2}i$ 'be here' and $ni^{2} \sim ni \sim n\partial$ 'be there,' and a directional pair, ${}^{2}mi' \sim mi$ 'come' and $n\acute{m}$ 'go.' I will

refer to these as the locative auxiliaries and the directional auxiliaries. They may also function as predicate heads and as prepositional verbs.

The auxiliaries serve as pegs on which to hang subject (or, in nominalizations, possessive) person markers and non-personal particles. They also serve to set the predicate within a spatial context.

The choice between ${}^{2}i$ and ni^{2} depends on the location of the speaker relative to whatever the predicate refers to. The auxiliary ${}^{2}i$ 'be here' locates the phenomenon or event (whether real or hypothetical, present, past, or future) near the speaker at the time of utterance. The auxiliary ni^{2} 'be there' locates it somewhere else. Consider the following sentences with ${}^{2}i$.

(a) [?]i cən cécəwət.

?i cən čéčəŵ-ət
AUX I be.helping-TR
'I am helping him.'

- (b) ⁹i cən técəl. AUX I arrive.here 'I arrived here.'
- (c) [?]i técəl. AUX arrive.here 'He's here. He has arrived here.'
- (d) ?i ł yox w con mo ?í?tot.
 ?i oł yox w con mo ?í?tot
 AUX-past INF I CERT be.sleeping
 'I must have been asleep.'

(e) ⁹i cən cnáx wəł.
⁹i cən c-náx wəł
AUX I get-canoe
⁶I have a canoe.⁷

In each of these, the i shows that the speaker is where he/she is helping or has arrived, where someone else has arrived, and so on. In the following, with ni^{2} , the speaker is not where the event has happened, is happening, or may happen.

(f) ni cən cewət.

ni? cən cew-ət AUX I help-TR 'I helped him.'

- (g) ni cən tə́pəłt. (JP)
 - ni? cən tə́pəl-t AUX I be.stretching.on.a.frame-TR 'I'm stretching it [a hide] on a frame.'

- (h) ni⁹ títələm.
 ni⁹ títələm
 AUX be.singing-INTR
 'There is someone singing.'
- (i) ni ?ə čx^w ce? q^wiq^wél?
 AUX ROG you FUT make.a.speech
 'Will you be speaking?'
- (j) ni? k^wəcnéləm.
 ni? k^wec-n-él-əm
 AUX look-trt-lpas-intrt
 'I am seen. They see me.'
- (k) ni⁹əł k^wəcnéləm.
 ni⁹-əł k^wéc-n-él-əm
 AUX-past look-TR-PAS-INTR
 'I was seen. They saw me.'
- (1) ni [?] yax ^w ce[?] waném.
 ni[?] [?] yax ^w ce[?] waném
 AUX ROG INF FUT ESTABLISHED-go
 [°] I wonder if he will go.[°]

The locative auxiliaries may indirectly convey information about time. Very often, when the auxiliary ni^2 appears with a perfective form of the verb, the reference is to a past event, as in $ni^2 \ con \ cewat$ 'I helped him.' But this is not because ni^2 is a past marker. It is simply because the event is remote in space from the speaker, as are, in the examples just given, my hide stretching on a frame somewhere, somebody singing somewhere, the speech that you may give, and my being seen by other people. Similarly, the auxiliary 2i often appears with a progressive form and reference to the present, as in $2i \ con \ cec^2 awat$ 'I am helping him.' But again, this is simply because the speaker is present at this ongoing event. In $2i \ con \ tecol$ 'I arrived here,' the reference is to a past event, but the speaker is still present where the event occurred and so uses 2i rather than ni^2 . The locative auxiliaries clearly refer to space and not to time, though perhaps more often than not the 'then' is 'there' and the 'now' is 'here.' The only true tense markers, however, are the particles -at 'past' and ce^2 'future.'

The directional auxiliaries $^{2}\partial mi \sim mi$ 'come' and $n\acute{em}$ 'go' identify actions as involving motion toward or away from the speaker.

(m) mi łe ?éx we?.
 come PER give.food
 'Give me some [of something to eat].'

(n) mi ła k watx wilam.
 come PER enter
 'Come inside.'

(o) ném łə mé⁹x.
 ném łə mé⁹-x
 go PER come.off-TR
 'Take it away.'

These auxiliaries can also have aspect-like senses, $2\partial mi$ meaning 'become' and $n\acute{e}m'$ 'be going to (do something).'

(p) ⁹ əmi x wəθí.
 ⁹ əmi x wə-θí
 come become-big
 'It became big.'

(q) ném cən ⁹íməx. go I walk 'I'm going to walk.'

If auxiliaries appear in sequence, the locational precedes the directional, i^{i} nearly always appearing before $i^{2}\partial m^{i}$ and ni^{2} before $n\acute{m}$.

(r) ⁹i mi łəłí·l. be.here come be.moving.shoreward 'He's coming in [shoreward].'

- (s) ⁹i wəłmi təwyəxáyk dət.
 i wəł-⁹əmi təw-yə-xáyk dət
 be.here already-come somewhat-along-be.getting.cold
 'It's been getting a little colder.'
- (t) ni nem k wənk wən. be.there go be.captured 'He was taken captive.'
- (u) ni nem cpé⁹eθ.
 ni⁹ nem c-pé⁹əθ
 be.there go get-black.bear
 'He has gone hunting bear.'
- (v) ni nem ťák^w ?é·łtən.
 be.there go return.home 3PL
 'They went home.'

The locative auxiliaries can also occur with their own lexical counterparts.

```
(w)?i ?í. (AG)
AUX be.here
'He's here.'
```

```
(x) ni ní?. (AG)
AUX be.there
'He's there.'
```

There is also a set of demonstrative auxiliaries, which combine elements identifiable with the locative auxiliaries and elements of the demonstrative system. They are described in §15.

3.2.2. Adverbs

An adverb is a word that can appear within the predicate as a modifier, specifying frequency, intensity, or some other quality (see §18). Most adverbs can appear initially in the predicate, as do $y\dot{a}\theta$ 'always,' $\dot{X}\dot{e}$ 'again, also,' $t\dot{a}\dot{q}$ 'usually,' and $q\dot{e}^2is$ 'recently' in (a) to (e).

- (a) wəyáθ [?]ə čx^w nəw yə²é·ỷ. (JP) wə-yáθ [?]ə čx^w ni[?] wə-yə-[?]é·ỷ EST-always ROG you AUX EST-along-be.continuing 'Do you always keep going?'
- (b) Xe cən nəwł Xté. (CC) Xé cən ni⁹ wəł-Xté⁹ again I AUX already-do 'I did it again.'
- (c) łód cən wəmi técəl. (DK)
 łód cən wəmi técəl
 usually I EST-AUX(come) arrive.here
 'I generally get here.'
- (d) łódoł xérm. (JP)
 łódoł xérm
 usually-past be.crying
 'She used to be crying.'
- (e) qé⁹is cən ni k^wécnəx^w. (JP) qé⁹is cən ni⁹ k^wéc-nəx^w recently I AUX look-TR 'I saw him just now.'

Some adverbs can appear between the auxiliary and the predicate head, as $\lambda \hat{e}$ 'again, also' and $m \delta k^{w}$ 'all' in (f) and (g).

- (f) ni [?]ə [?]àw ném[?]. (CC) ni[?] [?]ə [?]dé wə-ném AUX ROG also EST-go 'Did he go too?'
- (g) ni cən wəmák^w pánət. (CC)
 ni⁹ cən wə-mák^w pán-ət
 AUX I EST-all get.buried-TR
 'I buried them all.'

And some adverbs can appear after the predicate head, as $q \partial \dot{l} \dot{e}t$ 'again' and $m \partial \vec{k}^{w}$ 'all' in (h) and (i).

- (h) ni nem x wəné nt qəlét. (JP 12)
 ni? nem x wə-né?ənt qəlét
 AUX AUX(go) become-be.getting.night again
 'It became evening again.'
- (i) mi čx^w x^w ax^wét mák^w. (JP) mi čx^w x^w ax^wét mák^w AUX(come) you be.descending-TR all 'Bring them all down.'

3.2.3. Verb Complements

Several verbs often appear with verb complements. Such verbs include $i\dot{a}^2\theta at$ 'try,' $\theta \dot{a}t$ 'intend, try,' $i\dot{\theta}(x \otimes m)$ 'pity, be so kind as to, please,' $i\dot{\theta}x \otimes imat$ 'pity him, be so kind as to ... for him,' $\dot{e}\dot{y}$ 'continue,' $h\dot{e}\dot{y}$ 'go ahead,' $h\dot{a}y$ 'finish,' as in (a) to (e).

- (a) ťá⁹θət łe c^{*}λ⁵m. (CC 1) té⁹-θət łe c^{*}λ⁵m test-self PER jump 'Try to jump.'
- (b) θόt łák^w ... (JP 4) try fly
 'He tried to fly ... [but]'
- (c) t^θíx **əm łe céwət.
 t^θíx **əm łe céw-ət
 please PER help-TR
 'Please help him.'
- (d) ⁹i cən ⁹é·ỷ k^wúk^w ... (CC 12) AUX I be.continuing cook 'I continued to cook ...'

(e) héỷ čx w k v x x v y čx v y ćm. (CC)
 go.ahead you then be.story.telling
 'Go ahead then and tell a story.'

The verb háy 'finish' with $w\partial t$ - 'already' can have the sense of an English perfect tense, as in (f) and (g).

- (f) niwəł háy cyówən. (CC) ni? wəł-háy cyówən AUX already-finish sing.possessing.song
 'He has finished singing. He has already sung.'
- (g) ni?əł > čx^w wəłáy k^wścnəx^w (CC) ni?-əł ?> čx^w wəł-háy k^wéc-nəx^w AUX-past ROG you already-finish look-TR 'Have you ever seen him?'

Verbs of motion can be followed by verb complements, as in (h) and (i).

- (h) ² omí θoł ² ółton. (JP)
 come ADV eat
 'Come and eat now [regardless of other matters].'
- (i) ném cən kwə tákw ?éyələsàmə. (JP 3) ném cən kwə tákw ?éyəl-ləs-àmə go(AUX) I then return.home leave-TR-you 'Now I can go home and leave you.'

3.2.4. Hortative Forms

The auxiliary ${}^{?i}$ or the verb $h\acute{e}y$ 'go ahead' followed by the particle $l\acute{e}$ 'imperative,' with the stress on the particle (possibly a unique case of such a stress pattern), has the sense of a first-person plural imperative: 'Let's go!' JP and AG both gave (a), while CC gave (b).

(a) ²*i i é*. 'Let's go!'
(b) *he ý i é*. 'Let's go!'

AG also used a sequence of two particles, le 'imperative' and e 'plural imperative.'

(c) [?]*i té* [?]*e*[.]. 'Let's go!'

CC used heỷ lé followed by a verb complement, as in (d).

(d) heỷ lé ťáx^w. (CC)

go.ahead PER move.down.toward.shore 'Let's go down toward the shore.'
In JP's usage, however, $he\dot{y} t\dot{e}$ was followed by what seems to be a separate predicate in (e).

(e) héỷ łe ném ct.
 go.ahead PER go we
 'Let's go' or 'Get ready ... we'll go.'

AG's usage in (f) looks like CC's use of heý té.

(f) ?i łé ťíləm. AUX PER sing 'Let's sing.'

However, in (g) it appears that 'sing' is a separate predicate in that the 'plural imperative' follows it. (See also §16.2.9. and §16.2.11.)

(g) ⁹i łé ťíləm ⁹e · AUX PER sing PLPER 'Let's sing.'

3.3. TYPES OF VERBAL PREDICATES

On the basis of suffixes of the voice system (see §10) borne by the predicate head, we can distinguish grammatically three types of verbal predicates: intransitive, active transitive, and passive.

3.3.1. Intransitive Predicates

A predicate in which the head bears no transitive suffix can be identified as grammatically intransitive. Examples of intransitive predicates taken from among the sentences given in the preceding sections are:

- (a) ⁹i técəl. AUX arrive.here 'He has arrived here.'
- (b) ném cən ⁹íməx. go I walk 'I'm going to walk.'
- (c) k^wáq^w čx^w ce?.
 be.hit you FUT
 'You'll get hit.'
- (d) ni nem k ** ank ** án. be.there go be.captured 'He was taken captive.'

(e) mi łe ⁹éx^we⁹.
 come PER give.food
 'Give me some (of something to eat).'

In (e) 'me' is implied by the directional auxiliary *mi* 'come.'

(f) ⁹i cən c-náx^wəł. AUX I get-canoe 'I have a canoe.'

(g) ni nem cpé⁹eθ.
 ni⁹ nem c-pé⁹əθ
 be.there go get-black.bear
 'He has gone bear hunting.'

Some of these sentences are translated with English transitive verbs in the active or passive voice, but this has no bearing on the grammatical status of the Halkomelem sentences. It is simply a reflection of the fact that many Halkomelem roots, such as $k^w a q^w$ 'get hit' and $k^w a n$ 'get taken,' are non-active or patient-oriented, while others, including ${}^2 e x^w e^2$ 'give food,' are active or agent-oriented, and of the fact that the prefix c- 'get, make, do' produces an active form.

There are also four suffixes that produce intransitive verbs, the most commonly used of which are $-\partial m$ 'intransitive' and -els 'activity' (see §10.2).

3.3.2. Active Transitive Predicates

A predicate in which the verb bears one of the transitive suffixes -t, $-n\partial x^{w}$ (~ -n-), -x, etc. (for a full account, see §10.1) may be identified as grammatically transitive, and when this suffix is not followed by the suffix $-\partial m$ 'intransitive,' it is active.

The transitive suffix may be followed by one of the first- and second-person object markers, -S 'me, you,' -Samx ~ -amx 'me,' -Sam ∂ ~ -am ∂ 'you,' -alx" 'us,' and -al ∂ 'you plural.' (The transitive suffix -t and the object marker -S are realized as θ). In the absence of one of these, the object is third person. A transitive form may also bear the third-person transitive subject (ergative) marker - ∂s . (For a fuller account of the person markers, see §14.)

Among the sentences given in the preceding sections, the following are transitive:

```
    (a) ni cən céwət.
    ni? cən céw-ət
    AUX I help-TR
    'I helped him.'
```

- (b) céwəθ cən ce?.
 cew-ət-S cən ce?
 help-τR-me I FUT
 'I'll help you.'
- (c) čéwəθəs ce?.
 čéw-ət-S-əs ce?.
 help-tr-me-tr FUT
 'He will help me.'
- (d) qé⁹is cən ni k^wécnəx^w.
 qé⁹is cən ni⁹ k^wéc-nəx^w
 recently I AUX look-TR
 'I saw him just now.'
- (e) nem ła mé?x.
 nem ła mé?-x
 go PER be.off-TR
 'Take it away.'

3.3.3. Passive Predicates

A predicate in which the verb bears a transitive suffix followed by the suffix $-\partial m$ 'intransitive,' with or without an intervening passive person marker, may be identified as grammatically passive. If there is no passive person marker between the transitive and intransitive suffixes, the subject is third person. The passive person markers are $-Sel - \sim -el - (1, -Sam - \sim -am - 'you,' and -al - 'we, you plural.' The second-person <math>-Sam$ and the $-\partial m$ 'intransitive' are realized as -Sam. (Passive forms are not grammatically accompanied by the subject particles $c\partial n$ 'I,' cx^w 'you,' etc.)

Examples of passive predicates are:

- (a) ni čéw-ət-əm. AUX help-TR-INTR 'He was helped.'
- (b) ^γi cewaθelam.

⁹i čéw-ət-Sel-əm AUX help-TR-I-INTR ⁶I was helped.⁹

(c) ccéwəθà·m ce?

céw-ət-Sam-əm ce? help-TR-you-INTR FUT 'He (she, they) will help you. You will be helped.'

```
(d) ni? ċćw-ət-àl-əm.
AUX help-TR-we-INTR
'We were helped. You (pl.) were helped.'
```

(e) ni⁹ k^wəcná·m.

```
ni? k<sup>w</sup>ec-n-am-əm
AUX see-TR-you-INTR
'He (she, they) sees (see) you. You are seen.'
```

(f) k^wəcnéləm ce[?] m. k^wec-n-el-əm ce[?] m. see-tr-I-INTR FUT CERT 'I'll be seen.'

Speakers' translations 'He will help you' and 'He sees you' reflect the fact that active forms cannot have second-person objects with third-person subjects. Often, however, speakers will translate all of these forms with the active voice and 'they' as the subject.

3.4. NOMINAL ADJUNCTS

In a simple sentence, the predicate may be followed by one or more nominal adjuncts functioning as subject, object, locus, instrument, and so on.

The presence of a nominal adjunct is signalled by one of the deictic elements that constitute the demonstrative system (§15), an article or a demonstrative. These particles and words distinguish gender (non-feminine or feminine) and position (present and visible, near and invisible, or remote or non-existent).

A nominal adjunct consists minimally of a demonstrative standing alone and functioning as a third-person pronoun, such as $k^w\theta e^{2}$ 'he, that one (near, invisible),' $\theta \partial w \dot{X} \dot{a}$ 'she (present, visible),' and so on. Or, a nominal adjunct may consist of a demonstrative or an article followed by a word functioning as the nominal adjunct head, as in $t\partial sp e^{2}e\theta$ 'the/a (non-feminine, present, visible) black bear,' $\theta \partial n\partial m \dot{a} \dot{n} \partial$ 'my (feminine, present, visible) child,' that is, 'my daughter here.' (It is also possible for nominalized clauses and relative clauses to serve as nominal adjuncts, in what are, by that fact, complex sentences, described in §4.)

Nominal adjuncts relate to verbal predicates either directly or obliquely. A direct nominal adjunct is unmarked, while an oblique nominal adjunct may (but see directly below) be preceded by the oblique case marker ⁹a. Direct nominal adjuncts serve as subjects and (direct) objects, while oblique nominal adjuncts serve as loci and goals, instruments, oblique objects of intransitives, agents of passives, and possessors.

The role of oblique also differs from that of direct nominal adjunct in that it can be filled by a personal word, ${}^{2}\acute{e}:n\theta \partial$ 'me,' $n\delta w \partial$ 'you,' and so on (see

§14.2.7). Furthermore, before personal words and proper nouns serving as oblique nominal adjuncts, instead of one of the articles that appear elsewhere we find the oblique article \vec{X} (§15.1.4), which conveys no information about gender or position.

The difference between direct and oblique nominal adjuncts is not always apparent. In the Musqueam dialect as spoken by CC and JP, the particle ?a is optional and in some contexts is usually omitted. In this respect, as in several others, Musqueam stands between Cowichan, in which the particle is obligatory, and Upriver, in which it seems not to occur. In the Musqueam of DK and AG, which is influenced by Cowichan, the particle ?a appears with all oblique nominal adjuncts. Despite the absence of the particle for some speakers, however, the fact that an adjunct is oblique can be established in two ways. First, if a proper noun or personal word is substituted, the article \vec{x} will appear. CC and JP used this article pretty consistently even when omitting the particle. Second, as described later (§4.1.1.2), if the oblique is "extracted" for a relative clause, the verb in the relative clause bears the prefix *s*- (for an oblique object) or $\vec{s}x^{w}$ -(for a locus, goal, or instrument).

Let us now look at nominal adjuncts in relation to the three types of verbal predicates identified in §3.3.

3.4.1. With Intransitive Predicates

An intransitive verbal predicate may have a single direct nominal adjunct, which is its subject, as in (a) to (c).

- (a) ni mé? tə spít⁹ (DK) AUX come.off ART dye 'The dye ran.'
- (b) ?i wə?í łə nəmáňə. (CC)
 ?i wə-?í lə nə-máňə
 AUX EST-be.here ART my-child
 'My daughter is here [somewhere not visible].'
- (c) ni łák^w ⁹óżqəl tə sq^wéləx. (CC) AUX fly exit ART bird 'The bird flew out.'

Intransitive verbal predicates can have oblique nominal adjuncts in several roles. A predicate that expresses location or motion can have an oblique nominal adjunct as its locus or goal, as in (d) to (h).

(d) ⁹i cən ⁹í ⁹ə tən'á. AUX I be.here OBL this 'I am here.'

- (e) ni ní? ə təná·nə. AUX be.there OBL that.over.there 'He/she is over there.'
- (f) ném cən ce? tə spółxən. (CC) ném cən ce? [?ə] tə spółxən go I FUT [OBL] ART pasture/prairie 'I'm going to the pasture.'
- (g) mi čx^w təná səqin. (JP) ?əmí čx^w [?ə] təná səqin come you [OBL] this near.end 'Come to this end of the house.'
- (h) ni čé? > t> smé·nt. (AG)
 ni čé? ?> t> smé·nt.
 AUX land/perch OBL ART rock
 'It landed on the rocks.'

When a subject and a locus or goal both appear, the order of the two adjuncts may be either way. Compare (i) and (j):

- (i) ni ní? X sť^θáməs k^włə nəmɨn. (CC)
 ni ní? [?ə] X sť^θáməs k^włə nə-mɨnə
 AUX be.there [OBL] ART Victoria[Songhees] ART my-child
 'My daughter is in Victoria.'
- (j) ni scocé? to nowáč ?o to púk. (CC)
 ni? scocé? to no-wáč ?o to púk [púk^w]
 AUX atop ART my-watch OBL ART book
 'My watch is on top of the book.'

(Constructions like these, composed of intransitive verbs with oblique nominal adjuncts as loci or goals, also commonly serve as verbal adjuncts, comparable to English prepositional phrases. See §3.5.)

A predicate that refers to some action that may have a patient but is nonactive (patient-oriented) may have an oblique nominal adjunct as instrument, as in (k) and (l).

- (k) ni cən k^wáq^w ?ə k^wθə k^wq^wámən. (AG)
 AUX I get.hit OBL ART ax
 'I got hit by an ax.'
- (l) ni cən łáčcəs ² k wθa łéčtən. (AG)
 AUX I get.hand.cut OBL ART knife
 ⁴ I got cut in the hand with a knife.²

An intransitive predicate that is active (agent-oriented) can have an oblique object. Active intransitive verbs that participate in this construction include some bare roots, some forms with c- 'get, make, do,' some with $-\partial m$ 'intransitive,' and some with $-\ell ls$ 'activity.' These affixes all seem to produce agent-oriented verbs. Examples with bare roots are (m), (n), and (o).

(m) $\dot{m}i$ łe $\dot{\gamma}exwe$? $\Rightarrow \vec{k}^w q \dot{a}$?. (CC) mi lə [?]éx^we[?] [?]ə κ̈́» aá? AUX(come) PER give.food OBL ART water 'Give me some water.' (n) ném lə sówq kwə s?áltən. (CC) ném lə sówq [?ə] kwə s?óltən go PER seek OBL ART food 'Go look for food.' (o) ni ?ə θət ?ə tθé?. (DK) 22 θát a tθé? ni AUX ROG say OBL that 'Did he say that?' Examples with affixes are (p) to (s). (p) ni cən ck^wán k^w náca?. (DK) cən c-k^wán [?ə] ќ∾ n áca? ni get-get.taken [OBL] ART one AUX I 'I was able to get one.' (q) ném cən xté⁹em k^w snétəłqən. (CC) cən xté?-əm [?ə] kw s-nét-əl-qən ném AUX(go) I make-INTR [OBL] ART NOM-night-past-meal 'I'm going to make breakfast.' (r) ni ^γ axí·ls ^γ a k^wθa q^włéy. (AG) ni? ⁹ σxəl-éls ⁹ k^wθə q^włéy AUX paddle-ACT OBL ART log 'They towed the logs.' (s) ni me ném toméls ^γo k^wθo snóx^wołs. (CC) ni mэ ném ťem-éls ?ə k^wθə snáx^wəł-s AUX CERT AUX(go) get.hit-ACT OBL ART canoe-3POS 'He's gone to hew his canoe.'

It is also possible for an oblique nominal adjunct to appear as an instrumental following a predicate head with the suffix *-éls* 'activity,' as in (t). (t) łċćls ?ə tə łċćtən. (AG) łiċ-ċls ?ə tə łċċtən cut-ACT OBL ART knife 'cut with a knife'

But it may not be possible, AG believed, to have an *-éls* form with two oblique adjuncts, one the oblique object and the other the instrumental.

3.4.2. With Active Transitive Predicates

An active transitive predicate may have one or two direct nominal adjuncts.

If the subject is a first or second person, a direct nominal adjunct may appear as the object, as in (a) and (b).

(a) cewat can ce? tawka.

céw-ət cən ce? təwằá help-TR I FUT he/him 'I will help him.'

(b) ni ?ə čx^w k^wəcnəx^w k^wθe? swəyde?. ni ?ə čx^w k^wec-nəx^w k^wθe? swəyde? AUX ROG you look-TR that man 'Did you see that man?'

If the object is a first person, a direct nominal adjunct may appear as the subject, as in (c).

(c) céwəθàmxəs ce? tθe? swójqe?.
 céw-ət-Samx-əs ce? tθe? swójqe?
 help-tR-me-3TR FUT that man
 'That man will help me.'

Active transitive forms with second-person objects and third-person subjects, however, do not occur. Instead we find passive forms.

Unless another condition is met, if a predicate has a third-person subject and a third-person object, a single direct nominal adjunct will be interpreted as the object. Thus (d) must have the first interpretation rather than the second (starred), improbable though it may be.

(d) qíqok^wətəs tə nəsq^wəméy. (DK) qíqok^w-ət-əs tə nə-sq^wəméy be.biting-TR-3TR ART my-dog
'He is biting my dog.' (**'My dog is biting him.')

The other condition is the appearance of $t\partial w \dot{\lambda} a$ and other demonstratives formed with $\dot{\lambda} \dot{a}$ 'be third person.' After a predicate head with the suffix $-\partial s$ 'third-person transitive subject,' $t\partial w \dot{\lambda} a$ or one of its counterparts must be interpreted as subject, as in (e). (e) k^wónətəs təwka swíwləs. (JP 27) k^wón-ət-əs təwka swíwləs get.taken-TR-3TR that young.man 'The young man took it.'

An active transitive predicate may also have both object and subject nominal adjuncts. When this occurs, the normal order seems to be predicate, subject, object (VSO), as in (f).

(f) ni qá·ytəs θə pús θə kwétən. (JP)
 ni? qáy-ət-əs θə pús θə kwétən
 AUX die-TR-3TR ART cat ART mouse
 'The cat killed the mouse.'

But the order predicate, object, subject (VOS) is also possible, as in (g).

(g) k^wék^wəyətəs tə mémənəs t0e swojqe?. (JP)
 k^wék^wəy-ət-əs tə mémənə-s t0e? swojqe?
 be.training-TR-3TR ART children-3POS that man
 'That man is training his sons [preparing them for vision quests].'

In (g), perhaps the sense requires that the father be the subject. But any possible ambiguity can be avoided by the use of $t \partial w \dot{X} a$, which can mark the subject only, as in (h).

(h) ni qá·ytəs tə smóyəθ təwka swóyqe?. (DK)
 ni? qáy-ət-əs tə smóyəθ təwka swóyqe?
 AUX die-TR-3TR ART deer that man
 'The man killed the deer.'

The sentences offered by me,

ni ἀá·ytəs tə swə́y̓qe? təwʌ̈́a smə́yəθ. ni ἀá·ytəs təwʌ̃a smə́yəθ tə swə́y̓qe?.

would both, DK insisted, have to mean 'The deer killed the man.'

This subject-marking function of $t \partial w \dot{\lambda} a$ (or a counterpart) allows for freedom in the ordering of subject and object nominal adjuncts. In texts dictated by JP, there are a few complex sentences suggesting that he used $t \partial w \dot{\lambda} a$ or a counterpart to postpone a subject with dependent elements until after the object.

There is one further constraint on transitive predicates with two third-person participants. A proper name cannot assume the role of subject. It must be the agent in a passive construction.

An active transitive predicate can have an oblique nominal adjunct serving as an instrumental, as in (i), (j), and (k).

- (i) ni cən lám i ?ə kθə smé·nt. (CC)
 ni cən lám i -ət
 ?ə kwθə smé·nt
 AUX I get.hit(by something thrown)-TR OBL ART rock
 'I hit him with a rock.'
- (j) lámət tə smént. (CC)
 lám-ət [?ə] tə smént
 get.hit-TR OBL ART rock
 'Hit him with a rock.'
- (k) łícot 'o to łécton. (AG) łícot 'o to łécton cut-TR OBL ART knife 'Cut it with a knife.'

We also find oblique nominal adjuncts playing other roles when they appear following active transitive predicate heads bearing the suffix $-st ax^w$ 'causative' (see §10.3.2) or one of the applicative suffixes (see §10.4.)

A causative formed from an intransitive verb of motion may have an oblique nominal adjunct for its goal or recipient, as in (l) and (m).

(1) místəx^w čx^w X ²é·nθə. (JP) ²əṁí-stəx^w čx^w [²ə] X ²é·nθə come-CAUS you OBL ART me ³Bring it to me.²
(m)nə²éməstəx^w lə təwXa. (CC) néṁ-stəx^w lə [²ə] təwXa go-CAUS PER OBL him ³Take it to him³

Probably for most verbs of this sort, however, a goal or recipient is expressed in a verbal complement (prepositional phrase) (see §3.5).

A causative formed from an agent-oriented verb may have an oblique nominal adjunct for what is the patient in the action performed by what is its grammatical object (the causee), as in (n) and (o).

 (n) k^wécstəx^w tə John ⁹ə k^wθən pút. (AG) k^wéc-stəx^w tə John ⁹ə k^wθə ⁹ən-pút look-CAUS ART John OBL ART your-boat 'Show John your boat.'

 (o) ni cən ⁹ öłtənəstəx^w k^wθə sẳ í ẳ qəł ⁹ > k^wθə sqéwθ. (AG) ni⁹ cən ⁹ öłtən-stəx^w k^wθə sẳ í ẳ qəł ⁹ > k^wθə sqéwθ AUX I eat-CAUS ART child OBL ART potato 'I fed the child potatoes.' These are more literally 'Cause John to look at your boat' and 'I caused the child to eat potatoes.'

The applicatives $-st \partial x_2^{w_2}$ 'comitative,' -lc- 'benefactive,' and $-\partial s$ 'recipient' allow someone playing a role other than that of patient to become the grammatical object, at which point the original patient must be expressed by an oblique nominal adjunct, as in (p), (q), and (r).

- (p) Xté⁹stəx^w ⁹ə ti⁹í. (AG) Xté⁹-stəx^w ⁹ə tə⁹í do-COM OBL this 'Do this to it.'
- (q) θéyəłcθàmx łe [?ə] kwə táxwac. (CC)
 θéy-łc-t-Samx ła [?a] kwa táxwac
 be.created-BEN-TR-me PER OBL ART bow
 'Make me a bow.'
- (r) ?áx ** 380 àm x čx ** ?* k ** łéctan. (AG)
 ?éx ** as -t Sam x čx ** ?* k ** a łéctan
 give-REC-TR-me you OBL ART knife
 'Give me a knife.'

The last three sentences might be rendered with grammatically more parallel forms as 'Affect it with this,' 'Supply me with a bow,' and 'Present me with a knife.' (For a full analysis of Halkomelem causative and applicative or "redirective" forms, see Gerdts 1981.)

3.4.2. With Passive Predicates

A passive predicate can have only one direct nominal adjunct, which is its subject, as in (a), (b), and (c).

- (a) Žéxətəm tə x^wmöθk^wəÿəm. (CC)
 Žéxən-t-əm tə x^wmöθk^wəÿəm invite-tR-INTR ART Musqueam
 'The Musqueam were invited.'
- (b) X^{*}á·ytəm tə sté⁹ex^{*}əł. (JP 12) X^{*}áy·ət-əm tə sté⁹ex^{*}əł die.off-tR-INTR ART children 'The children were all killed.'
- (c) k^włétəm tə snáx^wəłs. (JP 12) k^wəł-ét-əm tə snáx^wəł-s spill-TR-INTR ART canoe-3POS 'Their canoe was tipped over.'

The agent of a passive predicate is an oblique nominal adjunct, as in (d) to (h).

- (d) módotom ?o to sklotopam. (JP 6)
 módotom ?o to s-klotopam
 swallow-tr-intr OBL ART NOM-fierce
 'They were swallowed by the monster.'
- (e) ni čéwətəm tə čéňəm. (CC)
 ni čéw-ət-əm [?ə] tə čéňəm
 AUX help-TR-INTR OBL ART Shaker
 'He was helped by the Shakers.'
- (f) k^wócnà·m ce? təwẳa. (CC) k^wóc-n-am-əm ce? [?ə] təwẳa look-TR-you(PAS)-INTR FUT OBL him 'He will see you.' (lit. 'You will be seen by him.')
- (g) čéwətəm ce? X Tom təwka. (CC)
 céw-ət-əm ce? [?ə] X Tom təwka
 help-tr-intr fut obl ART Tom he
 'Tom will help him.' (lit. 'He will be helped by Tom.')
- (h) ni⁹əł čéwəθàləm X Tom. (JP) ni⁹-əł čéw-ət-Sal-əm [⁹ə] X Tom AUX-past help-TR-I(PAS)-INTR OBL ART Tom 'Tom helped me.' (lit. 'I was helped by Tom.')

These last three sentences are passive with no active counterparts, since the rules do not permit the active voice with a second-person object and a third-person subject, or with a proper name as subject.

There is yet another constraint on passive constructions. The role of agent cannot be filled by a personal word; for example, the active sentences $ni \ c \partial n$ $\dot{q}\dot{a}\cdot yt$ 'I killed him' and $ni \ \dot{c}x^w \ \dot{q}\dot{a}\cdot yt$ 'You killed him' have no passive counterparts. You cannot say ** $ni \ \dot{q}\dot{a}\cdot yt\partial m$? $\partial \ \dot{X}$? $\dot{e}\cdot n\partial \partial$ or ** $ni \ \dot{q}\dot{a}\cdot yt\partial m$? $\partial \ \dot{X}$ $n\dot{a}w\partial$.

An instrumental can also appear with a passive predicate as an oblique nominal adjunct, as in (i) and (j).

- (i) ni? čéwəθèləm ?ə k^wθə sṫíləm. (CC)
 ni? čéw-ət-Sel-əm ?ə k^wθə s-ṫíləm
 AUX help-TR-I(PAS)-INTR OBL ART NOM-sing
 'I got helped with a song.'
- (j) lálaməθà·m tə yəỷq^wí·n. (CC) lálam-at-Sam-am [?a] ta yaq^wí·n be.getting.hit-TR-you(PAS)-INTR OBL ART light
 'They [the Shakers] are throwing the light [or heat, of their candles] on you.'

We also find, with passives of causatives, oblique nominal adjuncts serving the same purposes that they do with active causatives, as in (k) and (l).

- (k) ... i wəłnə?áməstəm θə déməy tə stáləw. (CC 19)
 ?əy wəł-ném-stəx -əm θə déməy [?ə] tə stáləw and already-go-CAUS-INTR ART girl OBL ART river
 ... and the girl was taken to the river.'
- (l) ?í·ŵəstəm tə syəŵín, tə sxəléls. (CC 19)
 ?í·ŵ-əs-t-əm [?ə] tə syəŵín tə sxəléls be understanding-REC-TR-INTR OBL ART spell ART painting 'She was being taught the spells and the painting.'

A sentence may contain an intransitive predicate with a stated or implied subject followed by a passive form with a different subject where we might expect a transitive and an object. An example is (m).

(m) mi θ ot cewatam tawka. (JP 22)
 mi θot cewatam tawka.
 come try help-tr-intre he
 'They came intending to help him.'

Perhaps because $t \partial w \dot{\lambda} a$ must be a subject, the verb must be passive. For other examples, see §4.3.4, (c) and (d).

3.5. VERBAL ADJUNCTS

A predicate head of any of the three types described in §3.3, with or without nominal adjuncts related as described in §3.4, may be followed by another verb, secondary to the head (and often unstressed), introducing an oblique nominal adjunct. These secondary verbs specify location, direction of motion, and a few notions of other sorts expressed by English prepositions. They serve to relate their oblique nominal adjuncts to the predicate head in ways other than those described in §3.4. Because these secondary verbs have functions like those of English prepositions, they may be conveniently called "prepositional verbs." (Galloway [1977, 483] has called the phrase composed of such a verb and its nominal adjunct a "prepositional phrase.")

Prepositional verbs do not constitute a separate class of words used for this purpose only; most, if not all, can appear as predicate heads. They include the locative verbs ?i 'be here' and ni^2 'be there,' the directional verbs ?ami 'come' and nem' 'go' (all four of which also appear as auxiliaries), derivatives of the last two, x^wami 'come to' and x^wnem' 'go to,' and several other simple verbs, such as tecal 'arrive here,' tas 'arrive there,' x^wte^2 'head toward, than (in comparative constructions),' tali (~ tali?) 'be from,' yatale (~ $yata^2e$) 'be going along.' They include a number of resultative forms (described in §7.7), such as states 'near,' sqaqa' 'accompanying, with,' sta^2e' 'resembling, like,' and so on,

and some more complex forms, such as $y \ge st a l w \ge t a l w = t a l w \ge t a l w = t$

Sentences (a) and (b) illustrate \hat{i} 'be here' used as auxiliary, predicate head, and prepositional verb.

- (a) ?i ?i ?ə kwən słódəwəł. (CC)
 ?i ?i ?ə kwə nə-słódəwəł
 AUX be.here OBL ART my-back
 'He is behind me.'
- (b) ?i ?ám t ?i k ** n słałálwał. (CC)
 ?i ?ám t ?i [?a] k ** a na-słałálwał
 AUX be.sitting be.here OBL ART my-rear.side
 'He is sitting behind me (as in a canoe).'

In (a), the stressed 2i is the predicate head. In (b), the predicate head is 2im d is the sitting' and the 2i following is a prepositional verb, required to locate the oblique nominal adjunct.

The locatives i and *ni* are usually translatable as 'at,' in,' on,' or 'into,' as in (c) to (e).

- (c) nəwł łódelləx wəs ?i kwi šxwqwélewəns. (JP 10)
 ni? wəł-łódelləx w-əs ?i [?ə] kwi šxwqwélewən-s
 AUX already-know-TR-3TR be.here OBL that mind-3POS
 'He already knew in his mind ...'
- (d) łédot ło ni 'o to lotém. (CC) łédot ło ni' 'o to lotém lie-TR PER be.there OBL ART table 'Lay it on the table.'
- (e) ni cən łálqt tə scéxt ni ?ə tə qá?. (CC)
 ni? cən łálq-t tə scéxt ni? ?ə tə qá?
 AUX I immerse-TR ART stick be.there OBL ART water 'I dipped the stick into the water.'

The sequence of locative verb and demonstrative, literally 'at this' or 'at that,' is translated 'here' or 'there' (see §15.2.4.2), as in (f) and (g).

- (f) ?ámət łə ?i ?ə tə?í. (CC)
 ?ámət łə ?i ?ə tə?í
 sit PER be.here OBL this
 'Sit here.'
- (g) wəyáθ cən wək wék wəcnəx w təwka ni ?ə tθé?. (CC)
 wə-yáθ cən wə-k wék wəc-nəx w təwka ni? ?ə tθé?
 EST-always I EST-be.looking-TR him be.there OBL that 'I always see him there.'

Sometimes ni^2 can be translated freely with an instrumental 'by' or 'with' or with a benefactive 'for,' as illustrated in (h) to (k), although perhaps there is still a basic locative meaning.

(h) k^wánət čx^w ni tə d^wənd^wí·ns. (JP) čx^w ni? k™án-at [?ə] tə å^wənå^wí∙'n-s get.taken-TR you be.there [OBL] ART ears-3POS 'Grab him by the ears.' (i) ni ce·p wəkwácnəxw ni te·n qáləm ?i təna swéyəl. (JP 26) ce·p wə- \dot{k} ^wéc-nəx^w ni? [?ə] tə ?ən-qáləm ?i AUX VOU(PL) EST-see-TR be.there OBL ART your-eye be.here [?ə] təna swéyəl OBL this dav 'You have seen it with your [own] eyes today.' (j) sčéčaň niw łádallax was ni? ta célaxs. (JP 32) scécon ni? wo-łódol-lox - os ni? [?ə] tə céləx-s really AUX EST-know-TR-3TR be.there OBL ART hand-3POS 'He [the harpooner] really knew [detected] it with [in] his hands.' (k) ni cən yá·yəs ni ?ə x nówə. (CC) ?a ż ni cən vá·vəs ni? náwa AUX I be working be there OBL ART vou 'I am working for you.'

The directional verbs $2\partial m i$ 'come' and $n \ell m$ 'go' can appear following predicate heads that express motion and can have goals or indirect objects, $2\partial m i$ with the sense 'to (toward the speaker)' and $n \ell m$ with the sense 'to (away from the speaker),' illustrated in (1) to (q).

- (l) wélx ła ta pá·l ?ami ?a x ?é·nθa. (CC)
 wél-x ła ta pá·l ?ami ?a x ?é·nθa throw-TR PER ART ball come OBL ART me 'Throw the ball to me.'
- (m) wélx cən ce? tə pá·l nem ?ə X nówə. (CC)
 wél-x cən ce? tə pá·l nem ?ə X nówə throw-TR I FUT ART ball go OBL ART you
 'I'll throw the ball to you.'
- (n) wélx čx^w nem ² τθé?. (AG) wél-x čx^w nem ² τθé? throw-TR you go OBL that 'Throw it to that person.'

- (o) cć[?]t čx^w nem tə *TV*. (JP) cć[?]-t čx^w nem [[?]ə] tə *TV* alight-TR you go OBL ART TV 'Set it on the television set.'
- (p) ?i θžátəm ?əmi Ž ?é·nθə. (CC)
 ?i θəž-ət-əm ?əmi [?ə] Ž ?é·nθə
 AUX push-tR-INTR come OBL ART me
 'They pushed it against me.' (lit. 'It was pushed against me.')
- (q) qwəsqwəstəm nem tə qá?. (JP 12)
 qwəsqwəs-t-əm nem [?ə] tə qá?
 submerge(PL)-TR-INTR go OBL ART water
 'They were thrown into the water.'

If the predicate head is in the progressive aspect, the prepositional verb may also be, as in (r).

(r) ... i c'oc'é?təm hən'əm k wə cícoł... (JP 16)
?əy' c'oc'é?-t-əm ho'n'əm [?ə] k wə cícoł and be.alighting-TR-INTR be.going OBL ART high '... and they were being set up on something high ...'

The prepositional verbs $x^w \partial \vec{m}i$ and $x^w ne\vec{m}$, formed with the prefix x^{w-2} 'move toward' (see §12.2.6), can be used in the same way as ' $\partial \vec{m}i$ and $ne\vec{m}$ to introduce nominal adjuncts as goals or indirect objects of predicate heads referring to motion. However, they are more commonly used to introduce nominals as indirect objects of predicate heads referring to speech or thought and as beneficiaries of other activities, as in (s) to (u).

- (s) wəyáθ cən wəx^wq^wilq^wəlíwən x^wnem³ ?ə X nówə. (CC) wə-yáθ cən wə-x^wq^wilq^wəlíwən x^wnem³ ?ə X nówə EST-always I EST-be.thinking go.to OBL ART you 'I am always thinking about you.'
- (t) ni čx^w θəq dəł xé⁹em x^wnem k^w cícəł si⁹em ... (CC) ni⁹ čx^w θəq dał xé⁹am x^wnem [⁹a] k^wa cícał si⁹em AUX you pray go.to OBL ART above lord 'You pray to God ...'
- (u) ni ct yá ý s x nem ? a X łwálap. (CC)
 ni? ct yá ý s x nem ? a X łwálap
 AUX we be working go.to OBL ART you(PL)
 'We are working for you.'

Examples of uses of other prepositional verbs may be seen in (v) to (y).

- (v) ni ct ném x «cák teli? > tə léləmct. (CC 5)
 ni? ct ném x «cák teli? ? + tə léləm-ct
 AUX we go go.far from OBL ART house-our
 'We went waaay far from our house ...'
- (w) ni x wənéć x té? X łwśləp ... (JP 26)
 ni? x wənéć x té? [?ə] X łwśləp
 AUX become-different head.toward OBL ART you(PL)
 'She is now different from you ...'
- (x) mi čx^w ?³émət stətés ^x ?⁶·nθə. (JP 9)
 ?³mí čx^w ?³mət stətés [?³] ^x ?⁶·nθə come you sit near OBL ART me
 'Come sit next to me.'
- (y) ni cən tíləm sqaqa kwə nəšx vaqwa?. (CC)
 ni? cən tíləm sqaqa? [?ə] kwə nə-šx vaqwa?
 AUX I sing accompanying OBL ART my-sibling
 'I sang with my brother.'

See also the comparative use of adjectives in §3.7.1.

3.6. ADVERBIAL ADJUNCTS

A predicate may be accompanied by an article or demonstrative followed by an adverb or adjective. Such constructions may be obliquely related to the predicate, although in the examples recorded from Musqueam speakers there is no oblique particle. These adverbial adjuncts usually appear last in the clause.

- (a) mɨ ⁹ɨλɨqəl kɨwə xɨwəm. (JP 24) mi ⁹ɨλɨqəl kɨwə xɨwəm AUX(come) exit ART fast 'He came out right away.'
- (b) xté²stəx^w łe təna x^wóm. (CC) xté²-stəx^w łe təna x^wóm do-COM PER this fast 'Do it right away.'
- (c) nə dépətəs kwə yéyə. (DK)
 ni? dép-ət-əs kwə yéyə [yéye?]
 AUX tie-TR-3TR ART bound.together
 'He wrapped it up tightly.'
- (d) ?éİmi? čx* k*əw qé?is ?al. (JP 3)
 ?éİmi? čx* k*ə wə-qé?is ?al
 be.waiting you ART EST-short.time just
 'Wait a little while.'

- (e) k^w on at čx^w k^wθé? taňa x^wané?ent. (CC 17) k^w on at čx^w k^wθé? taňa x^wané?ant be.taken-TR you that.one this evening 'Take him tonight.'
- (f) ?áx ** əs θ k ** ə ? = x ** í ň. (DK) ?áx ** əs-t-S k ** ə ? = x ** í ň give-TR-me ART little 'Give me a little.'
- (g) ťá?θət čx^w náqəm k^wə ³⁄_λáp. (DK) ťé?-θət čx^w náqəm k^wə ³⁄_λáp test-self you dive ART deep 'Try to dive deep.'
- (h) nəw θáytəm tə məstáyəx^w k^wəw x^wəná?əł ?al ... (JP11) ni wə-θáy-t-əm tə məstáyəx^w k^wə wə-x^wəná?-əł ?al AUX EST-come.into.being-TR-INTR ART people ART EST-first-past just 'People were made in the beginning ... [but] ...'
- (i) θόyt θο hόỷq^w k^wο θí. (JP 2)
 θόy-t
 θο hόỷq^w
 k^wο θí
 come.into.being-TR ART be.burning ART big
 'Make a big fire.'

In this last sentence, θt 'big' might be seen as an object complement, the sentence being literally 'Make the fire big' rather than 'Make the fire greatly.' I do not have enough data to justify setting up such a category, however.

These adverbial adjuncts are perhaps simply phrases substituting for the simple adverbs identified in §3.2.2. Giving them separate status here is based on the presence of the article or demonstrative, which makes them resemble nominal adjuncts. We might also note that in their not having an oblique particle to introduce them, these adverbial adjuncts resemble nominalized clauses used adverbially (see §4.3.3).

3.7. ADJECTIVAL AND NOMINAL PREDICATES

As seen in §3.1.1, adjectives and nouns can appear as predicate heads. Like verbal predicates, adjectival and nominal predicates can be expanded with other elements and can have nominal adjuncts. As predicate heads, however, adjectives and nouns differ from verbs in several respects. Morphologically, adjectives and nouns do not have distinctions of aspect. Syntactically, they differ from verbs in their use of auxiliaries, adjectives appearing with auxiliaries less often and nouns in certain constructions only. Adjectives and nouns can also appear together in nominal predicates expanded with adjectives. (Nominal predicates also differ from both verbal and adjectival predicates in their use in pseudo-cleft sentences, described in §4.1.5.1.1).

3.7.1. Adjectival Predicates

Adjectives may appear as predicate heads accompanied by particles only, as in (a) and (b).

- (a) k^wámk^wəm cən. (JP)
 strong I
 'I'm strong.'
- (b) k^wámk^wəm[?] [?] čx^w. (AG) strong ROG you 'Are you strong?'

They can be preceded by auxiliaries, as in (c) to (f), but they have been recorded less commonly so than have verbs.

- (c) [?]í [?]ə čx^w wə[?]ðý [?]al (often [?]i·čx^wəw[?]ðýal)
 [?]í [?]ə čx^w wə⁻?ðý [?]al
 AUX ROG you EST-good(well) just
 'How are you?' (a common greeting)
- (d) ⁹i cən wə⁹əy ⁹al. (CC)
 ⁹í cən wə-⁹əy ⁹al
 AUX I EST-well just
 ⁴I am well.' (the usual response)
- (e) ⁹i cən ⁹uk^wámk^wəm. (AG) AUX I EST-strong
 ⁶I am strong.⁷ (not as emphatic as k^wámk^wəm cən [AG])
- (f) ni ⁹uk^wámk^wəm. (AG)
 AUX EST-strong
 'He [absent] is strong all right.'

Adjectival predicate heads can be preceded by adverbs acting as intensifiers, as in (g) to (i).

- (g) wəỉán wəqôx.
 EST-too/very EST-many/much
 'There are too many. They are too many. It is too much.'
- (i) wəňán cən wəžáł ἀáἀəỷ. (CC 21)
 EST-very I EST-hurt sick
 'I am really sick.'
 An adjectival predicate may have a noun subject, as in (j) and (k).

- (j) płét tə pték wəm. (CC) thick ART brake.fern
 'The brake ferns are thick.'
- (k) X1? tə sy>wən. (CC) difficult ART song 'The song is hard.'

The subject of an adjectival predicate can be compared with something else by means of $x^w n\acute{em}$ 'go to,' $x^w a\acute{mt}$ 'come to,' or $\check{x}^w t\acute{e}^2$ 'head for, move toward,' used as prepositional verbs (see §3.5) with the sense 'than,' as illustrated in (l) and (m).

(1) \vec{k}^{w} á \vec{m} \vec{k}^{w} ə \vec{m} x w ə \vec{m} i \cdot $\vec{\lambda}$ 9 é \cdot n θ ə. (JP) k^wámk^wəm x^wəmi [?ə] x ?é•nθ⊋ come.to OBL strong ART me 'He's stronger than I.' (m)cák^w k^wámk^wəm x^wnem ?ə ż náwə. (CC) far strong go.to OBL ART you 'He's far stronger than you.'

The focusing word háy 'specifically' (usually with the particle ${}^{2}al$ 'just') preceding the adjective give a superlative sense, as in (n).

 (n) háy ?al k^wám k^wəm x^wté· k^wəw mök^w. (JP) háy ?al k^wám k^wəm x^wté ?ə k^wə wə-mök^w especially just strong head.for OBL ART EST-all 'He's the strongest of all. He is stronger than anybody.'

3.7.2. Nominal Predicates

As shown in §3.1, nominal predicate heads can appear alone, followed by particles and adverbs.

- (a) təmžáýž. (CC 20) təm-žáýž time.of-cold 'It was winter.'
- (b) swóỷqe? cən. (JP) man I 'I am a man.'
- (c) səwwáýqe? ct. men we 'We are men.'

- (d) x ">lməx" ?ə čx". (CC) Indian ROG you 'Are you an Indian?'
- (e) shínhar ín shínhar ín shínhar (JP)
 shínhar ín s
- (f) swóżqe? čx^w Åe. (JP) man you also 'You're a man too.'
- (g) wəqá'əł cə 'al mök'. (JP 1)
 wə-qá'-əł cə 'al mök'.
 EST-water-past QUOT just all
 'It is said that it was all just water.'

There may be a nominal adjunct serving as the subject of the sentence, as in (h) and (j).

- (h) Żcésəł ċə tθé?. (JP 1)
 Żcés-əł ċə tθé?
 island-past QUOT that
 'That [Point Roberts] is said to have been an island.'
- (i) syəŵín k^wθe? sk^wíx. (JP 22) spell.word that name
 'That name is a spell word.'

Auxiliaries have not been recorded with nominal predicates in simple sentences.

3.7.3. Nominal Predicates with Adjectival Modifiers

A nominal predicate head can be preceded by a modifying adjective (or numeral), the noun and its modifier standing in a relationship like that of a verb head and a preceding adverb. As always, the predicate particles follow the first word.

- (a) ⁹óỷ čx^w swóỷqe⁹. (AG) good you man
 'You're a good man.'
- (b) k^wámk^wəm s⁷áləlyəs. (JP 25) k^wámk^wəm s⁷áləlyə-s strong dream-3POS
 'His dream was strong. He had a strong dream.' (lit. 'It was his strong dream.')

In (b), the absence of an article before $s^2 \delta l y \partial s$ 'his dream' makes the whole phrase a predicate.

(c) i wənéč θəł sq^wəléx'. (JP 25)
 ?əỷ wə-néč θəł sq^wəléx and/but EST-different ADV bird
 'But it was a different bird.'

The adjective can be preceded by an intensifying adverb, as in (d).

(d) scécon ce·p wo?éloý sowwójqe?. (JP 12) scécon ce·p wo?éloý sowwójqe? really you(PL) EST-good(PL) men 'You are really good men.'

There may be a nominal adjunct as the subject, as in (e) and (f).

- (e) ⁹ śý swáýqe⁹ ta John. (AG)
 good man ART John
 'John is a good man.'
- (f) k^wámk^wəm scénəm θə nətén. (CC 21) strong Shaker ART my-mother 'My mother was a strong Shaker.'

If the modifier of the noun is a quantifying word, the sentence can have the existential sense of the English 'There is' or 'There are,' as in (g) to (j).

- (g) qáš ce? spédam. (CC) many FUT flower
 'There will be lots of flowers [the tree being covered with buds].'
- (h) sčéčaň yax^w ce? waqáž spédam. (CC) really INF FUT EST-many flower
 'There will really be lots of flowers.'
- (i) yəłédacəs caracterization canoe
 five.conveyances QUOT canoe
 'They say there were five canoes.'
- (j) qážał ča mastáyax^w ni a k^wθé?. (JP 25) many-past QUOT person be.there OBL that
 'It is said there were many people there.'

In the last four examples, because there is no article preceding the noun, the adjective, particles, and noun together must constitute the predicate. In (j), the $ni \ ^2 \partial \ k^{w} \partial \hat{e}^2$ is a verbal adjunct or prepositional phrase modifying the predicate. In (k), there is a subject.

 (k) qôx spk^wôm to lotém. (CC) much dust ART table
 'The table is dusty.'

I am uncertain how to interpret this. It appears to be literally 'The table is much dust.' But is it 'The table has much dust,' or is there an understood prepositional verb and oblique particle making it 'There is much dust on the table'?

3.8. EXPANDED NOMINAL ADJUNCTS IN SIMPLE SENTENCES

Noun heads of nominal adjuncts can take possessive affixes and have noun possessors (§3.8.1), they can be followed by a few of the predicate particles (§3.8.2), and they can have various adjectival modifiers, including noun premodifiers (§3.8.3). Nominal adjuncts can also be expanded by coordination and possibly by apposition.

3.8.1. Possession

The possessive affixes are $n\partial$ - 'my,' $\partial \partial n$ - ~ $\partial \partial T$ - 'your,' -*ct* 'our,' ∂n - ~ ∂T - ... - $\partial l\partial p$ 'your (pl.),' -*s* 'his, her, its,' and -*s* ($\partial e^{-lt}\partial n$) 'their' (see §14.2.4).

Noun possessors can be related to noun heads in one of two ways. In the first, the noun head takes the possessive -s and is followed by the possessor with its own article, as in (a) to (d).

- (a) tə s²íləyəθəns tə stáləw (CC 11)
 tə s²íləyəθən-s tə stáləw
 ART bank-3POS ART river
 'the bank of the river'
- (b) tə słíq^ws tə scé·łtən (CC 13) ART flesh-3POS ART salmon 'the flesh of the salmon'
- (c) tə sq^wéls θə nətén (CC 11)
 tə s-q^wél-s θə nə-tén
 ART NOM-speak-3POS ART my-mother
 'my mother's words'
- (d) k **θə sné?emsk **θə šx **né?m (CC 21)ARTshaman's.vision.power-3POSART'the "power" of the Indian doctor'shaman

A demonstrative can also be a possessor, as in (e).

(e) tə si⁹éms təwkáləm (JP 27) ART leader-3POS those 'the leader of them' In the second way, the noun head, without the possessive *-s*, is followed by an optional oblique particle 2σ , the oblique article \vec{X} , and the possessor noun. With proper nouns as possessors, this type of construction is usual, as in (f) and (g).

(f) θə tén ?ə X xá?xi^θəltən (CC 11) ART mother OBL ART xá?xi^θəltən 'the mother of xá?xi^θəltən [Peter Pierre, the Katzie shaman]'
(g) tə sxxəyem X t^θək téyəč (JP 3) tə sxxəyem [?ə] X t^θək téyəč ART story OBL ART t^θək téyəč 'the story of t^θək téyəč [the name of a magical dog]'

This construction has also been recorded for a few common nouns (see \$15.1.4).

3.8.2. Particles

A few of the second-position predicate particles can also appear following the noun heads of nominal adjuncts, as $-\partial t$ 'past' and ce^2 'future,' in (a) and (b).

- (a) k^wθə nəménəł (CC) k^wθə nə-mén-əł ART my-father-past 'my late father'
- (b) k^wθə nəpúk ce⁹ (CC) k^wθə nə-púk ce⁹ ART my-book FUT 'my book-to-be'

A few also appear with adjectives modifying noun heads (see below). In contrast, the whole range of second-position predicate particles can appear in nominal predicates (see §3.7.2).

3.8.3. Adjectival Modifiers

These include adjectives, numerals, a few adverbs, adjective-like forms of several kinds, and one or more verb forms. It seems that all of these always precede the noun head except for numerals, which sometimes follow.

3.8.3.1. Adjectives

Adjectives are simple and complex (see §4.3). Simple adjectives are roots such as θi 'big,' $\check{x} \acute{e} \dot{w} s$ 'new,' $\dot{p} \acute{a} \dot{q}$ 'white,' and $q \acute{a} l$ 'bad,' and their plural and diminutive forms, as in (a) to (d).

(a) k^wθ θ θ q^włé y (JP)
 ART big log
 'a big log'

- (b) tə xéws pték wəm (CC 14) ART new brake.fern.frond 'new ferns'
- (c) qôx kwôo pôch di nosqwomqwoméyalł (JP)
 many ART white(PL) my-dogs-young
 'I have lots of white puppies.' (lit. 'My white puppies are numerous.')
- (d) x^wləme[?]íwəθəs k^wəủa qʻəl nəstáləs. (JP)
 x^w-ləmé[?]-íwən-t-S-əs k^wəủa qʻəl nə-stáləs
 LOC-kick-rear-TR-me-3TR that bad my-spouse
 'My ugly husband kicked me in the bum.'

Complex adjectives are formed from adjective roots and lexical suffixes, as in (e).

(e) tə θəθíŵs má²aq^w (JP 23)
 tə θíθə-íŵs má²əq^w
 ART big(PL)-body large.bird
 'the larger [kinds of] birds [waterfowl]'

The adjective may be followed by some of the predicate particles. In (f), it is followed by two.

(f) k^wθ³ θí·ł čtwa? m³st³y³x^w (JP 14)
 ART big-past SPEC person
 'that great person, as he must have been'

The adjective can be preceded by an intensifying adverb, as in (g) and (h).

- (g) k^wθ scéc sň w θí q^wtá y θ sn (JP 23)
 ART really EST-big sturgeon
 'a really big sturgeon'
- (h) k^wθə háy ?al Åéqt s?ánəm (JP 23)
 ART specifically just long shaft
 'the longest [harpoon] shaft'

An adjective may precede a possessed noun, as in (i).

 (i) k^wθə qál ⁹əθsxəménələp (JP 2) k^wθə qál ⁹əT-s-xəmén-ələp ART bad your-NOM-enemy-your(PL) 'your bad enemy'

(It is not clear why JP pronounced the initial s- in sxamén 'enemy.')

Sequences of adjectives, either in nominal adjuncts or in predicates, hardly ever occur in the texts. In (j), there is a rare instance of a sequence of two adjectives.

(j) θ t łák^w t w k a q t θ f spá l ... (JP 4)
 try fly that bad big raven
 'That bad big raven [the bigger and greedier of two] tried to fly ...'

(AG said the order 'bad big' in this sentence sounds normal, but I have no basis for any classification of adjectives by the order in which they can appear. Perhaps the phrase was 'that bad Big Raven,' Big Raven being a proper name.)

3.8.3.2. Numerals

Numerals, both simple and complex (i.e., composed of a numeral root and a lexical suffix; see §19), usually precede the nouns they modify, as in (a), (b), and (c).

- (a) ném čə ?á·ł tə ťxámələ stənténəý. (JP 14)
 AUX(go) QUOT embark ART six-person women
 'Six women embarked, it is said.'
- (b) tə isélə məlləs (CC 8) ART two racoon 'two racoons'
- (c) tə nɨcəs smé·nt (JP 6)
 tə nɨca?-əs smé·nt
 ART one-round.object rock
 'one rock/another rock'

If there is an adjective, it will follow the numeral when the numeral precedes the noun, as in (d) and (e).

- (d) wəθə⁹ít cə k^wə qáynəx^wəs tə nöca? ⁹əx^wín smöyəθ. (JP 5) wə-θə⁹ít cə k^wə qáy-nəx^w-əs tə nöca? ⁹əx^wín smöyəθ EST-truly QUOT then die-TR-3TR ART one little deer 'He really killed one little deer then, they say.'
- (e) ... ni? kwánnaxwas kwθa yéýsala ?amámaň sté?exwał. (JP 2) ni? kwán-naxwas kwθa yéýsala ?amámaň sté?axwał AUX get.taken-TR-3TR ART two-persons little(PL) children '... she got two small children.'

But the numeral may also follow the noun, as in (f) and (g). It is not clear what difference, if any, in meaning this makes.

 (f) wəθə²ít k^wánnəx^wəs k^wθə sqaməl isəlélwe²s. (JP 12) wə-θə²ít k^wán-nəx^w-əs k^wθə sqaməl yəsélə-əlwe²s EST-truly get.taken-TR-3TR ART paddle two-paddle 'They truly did get two paddles.' (g) čáləsəm təwka. wəłk wacnax was tə snáx wəł θéməx wəł. (JP 12)
 čál-əs-əm təwka wəł-kwec-nəx and some termination of the second

One possible interpretation is that these numerals that follow their heads are relative clauses (see \$4.1), in which case (f) would be more literally 'They truly did get paddles, which were two (paddles).' See also comment on example (a) in \$3.8.5.)

3.8.3.3. Adverbs

A few words that usually appear as adverbs (see §18) can also appear as noun modifiers. These include $m\delta \vec{k}^{w}$ 'all,' $\vec{\lambda} \vec{e}$ 'also,' $x^{w}\delta n \vec{a}$ 'first,' and $\vec{a}t$ 'earlier, long ago,' among others.

The word $m \delta k^{w}$ 'all' (§18.4.40) is exceptional in that it can appear either before or after the article. Compare (a) with (b) and (c):

- (a) mớk v tə sté?ex vəł (JP 2) all ART children 'all the children'
- (b) təw mák^w sté?ex^wəł (JP 2)
 ART EST-all children
 'all the children'
- (c) təw mök^w stém smöyəθ (JP 3)
 tə wə-mök^w stém smöyəθ
 ART EST-all what deer
 'any kind of [game] animal'

An adverb can precede an adjective, as in (d).

(d) kwaw mákw 'amáman sté'exwał (JP 2) kwa wa-mákw 'amáman sté'axwał ART EST-all little(PL) children 'all the little children'

For the use of wa- 'established' in (b), (c), and (d), see §3.8.3.4. below. Other adverbs appear following the article, as in (e), (f), and (g).

(e) tə yáysələ Xaw swá wilas (JP 12)
 tə yáysələ Xe wa-swá wilas
 ART two.persons also EST-boys
 'two other boys'

- (f) k^wθə wəłátəł məstáyəx^w (JP 1) k^wθə wə-łát-əł məstáyəx^w ART EST-earlier-past person 'the people of long ago'
- (g) k^wθə x^wəňá?ał x^wənítəm (AG) k^wθə x^wəňá?-əł x^wənítəm ART first-past White.person 'the first White people'

(The prefix wa_{2} 'established' appears here in ways that are not, at this stage of analysis, predictable.)

3.8.3.4. Adjective-like Forms

Nouns, adverbs, and interrogative words can be given an adjectival meaning and used attributively (as well as predicatively) by the prefix wa_{2} 'established' and (usually) the particle 2al' 'just.' (Enclosing a verb with wa_{2} ... 2al' gives the meaning 'just, only,' but enclosing a noun, adverb, or interrogative word gives only an adjectival sense.) These forms appear before the noun head, as in (a) to (e).

(a) təw sθéθəyən ⁹al pték^wəm (CC 14) tə pték ^wəm ART EST-blood iust bracken.frond 'the bloody ferns' (b) k^wθəw x^wə́lməx^w ?al məstə́yəx^w (JP 2) k™θə wə-x "álməx" ?aĺ məstáyəx^w ART EST-Indian just person 'the Indian people' (c) k^wəw ^{*}Aəx^wlá⁹as ⁹al məstəyəx^w (JP) k^wə wə-xəx^wlá⁹as ⁹al məstávəx^w ART EST-no.matter just person 'any common people' (d) kwaw talanaca ?al mastayaxw (JP 3) k^wə wə−təl-?ónəcə ⁹al məstáyəx^w ART EST-from-where just person 'people from anywhere' (e) təw məkw val tələnəcə məstəvəx (JP 13) wə-mók^w ⁹al təl-?>nəcə tə məstáyəx ART EST-all just from-where person 'people from everywhere'

Some of these forms can precede adjectives, as in (f) and (h).

- (f) k^wθəw mök^w ?al θiθíŵs má?aq^w (JP 23) k^wθə wə-mök^w ?al θíθə-íŵs má?əq^w ART EST-all just big(PL)-body bird(large) 'all bigger [kinds of] waterfowl'
- (h) k^wəw stém ⁹al ⁹əx^wín t^θít^θk^w (JP 25) k^wə wə-stém ⁹al ⁹əx^wín t^θít^θk^w ART EST-what just little little.bug 'some kind of little bug'

Adjective-like words are also produced from nouns with the prefix *s*- 'nominalizer'(or 'resultative'?) and the suffix $-a^{2}t$ 'attributive formant' (see §12.1.5), as in (i) and (j).

- (i) tə šx^wlməx^wa?ł tóməx^w
 tə s-x^wólməx^w-a?ł tóməx^w
 ART NOM-Indian-ATT land
 'Indian land'
- (j) tə šx^wÅéləma?ł smənmé·nt
 tə s-x^wÅéləm-a?ł smənmé·nt
 NOM ART-Clallam-ATT mountains
 'the Olympic Mountains'

3.8.3.5. Verb Forms as Modifiers

Resultative forms (see §7.7) can precede nouns as adjective-like modifiers, as in (a), (b), and (c).

- (a) k^wθə st^θət^θíx^w sqálle?ł (CC 12) k^wθə st^θət^θíx^w sqálle?ł ART washed diaper 'clean diapers'
- (b) k^wθə syé^xd nəpút (AG) k^wθə syé^xd nə-pút ART painted my-boat 'my painted boat'
- (c) tə səyhá·yəł ċtwa? léləm (JP 2)
 tə səyhá·y-əł ċtwa? léləm ART finished(PL)-past SPEC house 'the houses built then'

In the texts, occasionally another verb form appears before a noun, as in (d).

(d) tə xé·m sxítxqəł (JP 2) ART be.crying child 'the crying child'

I can suggest that this progressive verb form, $\check{x}\acute{e}\cdot n$, may be a relative clause, which may precede a noun it modifies (see §4.1.2, [h] and [i]). However, this suggestion does not relieve me of the obligation to explain why $\Im x^{w} in$ in (e)

(e) k^wθə [?] əx^wín s^{*}λ i^{*}λ qəł (JP 2)
 'the little child'

is not a relative clause, which at this point I cannot do.

3.8.3.6. Noun Premodifiers

Nouns may precede nouns as their modifiers. They will be treated in §4.1.3 in connection with nominal relative clauses.

3.8.4. Coordination

Nominal adjuncts can be linked with the conjunction ${}^{2}\partial y' \sim i$ 'and,' as in (a) to (e).

- (a) Žélxəs tə táməł i tə ste?é k snás. (JP 22)
 Žél-x-əs tə táməł ?əỷ tə stə?é ?ə k snás apply-tR-3TR ART red.ochre and ART like OBL ART oil
 'He applied red ochre and something like oil.'
- (b) k^wónətəs tə sỷólx^wəm i tə t^θélə i tə mósən. (JP 22) k^wón-ət-əs tə sỷólx^wəm 'əỷ tə t^θélə 'əỷ tə mósən get.taken-TR-3TR art lung and ART heart and ART gall 'He took the lungs and the heart and the gall.'
- (c) tə [?]éšx^w, tə k^wá·n, əy tə xés (JP 22) ART seal ART porpoise and ART sea.lion 'seals, porpoises, and sea lions'
- (d) ni ²ónəcə k^w léləm X sməlów i sqəlócəməs. (CC 1)
 ni ²ónəcə k^wə léləm [²ə] X sməlów ²əy sqəlócəməs
 AUX where ART house OBL ART s. and s.
 'Where is the house of sməlów and sqəlócəməs?'
- (e) θόt k^wsə nətén i tən cłilé?em. (CC 21)
 θόt k^wsə nə-tén ?əỷ tə nə-cłilé?em say ART my-mother and ART my-stepparent '... said my mother and my stepfather.'

Nouns cannot be linked with first- and second-person subjects and objects in the same way they are linked with other nouns. Instead, the appropriate plural person marker appears in its proper place and the conjunction and noun appear as a nominal adjunct, as in (f) and (g).

(f) ^γi ct mi háye^γ ^γi k^wθ^a namén. (AG) ?i ct hýve? ?i Μi k^wθə nə-mén AUX we AUX(come) leave and ART my-father 'My father and I left to come here.' (g) ni k^wəcnálx^wəs łə słéni^γ ^γi k^wθə nəsqé^γəq. or ni kwəcnálxwəs ?i kwbə nəsqé?əq lə sléni?. (AG) ni k^wéc-n-álx^w-əs lə sléni? ?i k^wθ_θ n_θ-sqé⁹_θq my-junior.sibling/cousin AUX look-TR-us-3TR ART woman and ART

In an oblique adjunct, a personal word can be coordinated with a noun, and the personal word is plural, as in (h).

'The woman saw my younger brother and me.'

3.8.5. Apposition

It seems that a nominal adjunct can also be amplified by an appositive word or phrase. In the texts, indisputable examples are rare, since other interpretations are often possible. What may be instances of appositive phrases are seen in (a) and (b).

- (a) k^wónətəs tə qóləm, isélə qóləm. (JP 22) k^wón-ət-əs tə qóləm yəsélə qóləm get.taken-TR-3TR ART eye two eye 'He took the eyes, two eyes.'
- (b) ŽéŽələtəs kwə təwqáx sqwél, sŽéləqəm sqwél. (JP 22) ŽéŽəl-ət-əs kwə təw-qáx sqwél sŽéləqəm sqwél be.applying-TR-3TR ART somewhat-many word dangerous word 'He was applying words, powerful words, to it.'

However, it is possible that the final phrases in (a) and (b) are in fact separate sentences: 'They were two eyes'; 'They were powerful words.' Or they could be construed as relative clauses (see §4.1): 'He took the eyes, which were two eyes'; 'He was applying words, which were powerful words.' If this latter is correct, these would be examples of non-restrictive relative clauses, which seem rare or non-existent. In (a), 'two eyes' seems parallel to the complex numerals in sentences (f) and (g) in §3.8.3.2 above. There are also instances of proper nouns that seem to be standing in apposition to the nouns that precede them, as in (c), (d), and (e).

(c) k^wθ^a sxaméns yáq^wałte?x (JP 14) k^wθə sxəmén-s yáq^wəłte⁹x ART enemy-3POS Lekwiltok 'their enemies, the Yuculta [Lekwiltok]' (d) tə mənas tə cicəl si em Jesus máňa-s cícəł si?ém Jesus ta tə ART child-3POS ART high lord Jesus 'the son of God. Jesus' (e) wə⁹íməx ⁹al tə si⁹ém xé·ls. (JP 18) ?aĺ wə-?íməx tə si?ém **ž**é∙ĺs EST-be.walking just transformer ART lord

'The Lord Transformer was just walking.'

Here, too, it is possible that these are all nominal relative clauses and more literally translated 'their enemies, who were the Yuculta,' 'the son of God, who was Jesus,' and 'the lord [honoured person], who was xééls.' Again, they would be non-restrictive relative clauses.

4 Syntax 2: Complex Sentences

A complex sentence consists of a main clause and one or more embedded clauses es termed relative clauses, subordinate clauses, and nominalized clauses. It may be useful to identify each of these clause types here in a preliminary way. In the following examples, the embedded clauses are enclosed in square brackets.

(1) A relative clause is an embedded clause that can modify a nominal adjunct head, such as 'whom I saw' in (a).

(a) tə swóýqe? [ni·n k^wécnəx^w]
tə swóýqe? ni?-ən k^wéc-nəx^w
ART man AUX-I look-TR
'the man [whom I saw]'

(2) A subordinate clause is an embedded clause that can stand as a conditional or complement clause in relation to a main clause, such as 'if you go' in (b).

 (b) k^wócnà·m ce? [wəniəx^w ném]. k^wéc-n-am-əm ce? wə-ni?-əx^w ném look-tr-you(PAS)-INTR FUT if-AUX-you go 'You'll be seen [if you go].'

(3) A nominalized clause is an embedded clause that can serve as a nominal adjunct, such as 'my going' in (c).

```
    (c) sk<sup>w</sup>éy k<sup>w</sup>ə [nəsném].
    sk<sup>w</sup>éy k<sup>w</sup>ə nə-s-ném
impossible ART my-NOM-go
'I can't go.' (lit. '[My going] is impossible.')
```

Each of these clause types has other forms and uses, which will be described in the following sections.

4.1. RELATIVE CLAUSES

The term "relative clause" is one that I will use for embedded clauses resulting from "extraction." These correspond to English attributive clauses ("relative clauses" in the narrowest sense) such as 'who saw me' and 'whom I saw' in 'the man who saw me' and 'the man whom I saw,' and to clauses that appear in English "cleft" and "pseudo-cleft" sentences (Quirk and Greenbaum 1973, 414-17) such as 'who saw me' in 'It was John who saw me' (cleft) and 'John was the one who saw me' (pseudo-cleft).¹ Halkomelem relative clauses (in this broad sense) appear as attributive clauses; as subjects in sentences whose predicates are nouns, personal words, and interrogative words; and even as predicates. When we compare a relative clause with the corresponding simple clause, we see that one of the nominals of the simple clause is missing in the relative clause, having been "extracted" so that it may appear, in the larger context of a complex sentence, as the relative clause head. Thus, when we compare the simple clause 'I saw a bear' and the relative clause 'that I saw,' we see that 'bear' is missing from the relative clause. It has been "extracted" so that it may stand, for example, as the subject in the complex sentence 'The bear that I saw ate our lunch.' In this complex sentence, 'bear' is now modified by (is the head of) the relative clause 'that I saw.' In this way, a relative clause singles out a participant in one event or condition (expressed by one clause) in order to identify that participant with or in another event or condition (expressed by another clause), as in 'the bear that I saw with whatever it was that ate our lunch.' Relative clauses in their various uses in Halkomelem all seem to have this function. In this section, relative clauses are enclosed in square brackets.

Relative clauses often appear modifying noun heads (the extracted nouns), as in (a).

(a) tə swóýqe? [ni·n k^wácnəx^w]
tə swóýqe? ni?-ən k^wéc-nəx^w
ART man AUX-I look-TR
'the man (that, whom) I saw'

However, they can also simply follow an article or demonstrative expressing the gender, position, and number of the extracted noun, as in (b).

(b) tə [ni⋅n k^w ścn əx^w] tə ni[?]- ən k^w śc-n əx^w ART AUX-I look-TR 'the one (that, whom) I saw'

¹ Gerdts (1981, 59-84), discussing extraction, confines the term "relative clause" to attributive clauses and uses the phrase "embedded clauses of extraction" for what I am calling "relative clauses."

Following Gerdts (1981, 62), I will identify this last construction as a relative clause with an "eclipsed" head. The examples that follow in the next section are mainly relative clauses with eclipsed heads.

Depending on the type of predicate embedded, relative clauses can be identified as verbal, nominal, and, possibly, adjectival. We will first consider verbal relative clauses.

4.1.1. Forms of Verbal Relative Clauses

Verbal relative clauses vary in form depending on what is extracted and from what type of clause. One major difference separates relative clauses with extracted subjects and objects from relative clauses with extracted oblique nominal adjuncts. In the first set the predicate head is a finite form of the verb, while in the second set the predicate head is nominalized.

4.1.1.1. With Extracted Subjects and Objects

If what is extracted is the subject of an intransitive or passive clause, the relative clause will have the same form as the corresponding simple clause. Compare the following simple-clause predicates, (a) and (b) with the corresponding relative clauses with eclipsed heads, (a') and (b').

(a') k ^w θə [ni ἀáy]
ART AUX die
'the one who died'
(b´) kʷθə [ni ἀá·ytəm]
k™θə ni qáy-ət-əm
ART AUX die-tr-intr
'the one who was killed

However, if either the subject or the object of a transitive clause is extracted, the relative clause will be different, in all but one case, from the corresponding simple clause. This is because the presence or absence of subject markers must show whether it is the subject or object that has been extracted.

If the object is extracted, the relative clause will bear one of the subordinate subject suffixes (-*e*·*n* ~ - ∂n 'I,' - ∂x^{w} 'you [singular],' - ∂t 'we,' -*e*·*p* ~ - ∂p 'you (plural),' - ∂s 'he, she, it, they'; see §14.2.3). Compare (c) to (f) with (c') to (f'):

(c) čéw-ət cən ce?.	(c') $k^w \theta a$ [cew-at-en ce?]
help-tr I fut	ART help-tr-I FUT
'I'll help him.'	'the one (whom) I'll help'
(d) ni cən céwət.	(d') k ^w θə [ni·n ċéwət]
ni cən céw-ət	k™θə ni ⁹ -ən cew-ət
AUX I help-tr	art aux-I help-tr
'I helped him.'	'the one (whom) I helped'

(e') k ^w θə [cew-ət-əx ^w ce?]
ART help-tr-you FUT
'the one (whom) you'll help'
(f´) k ^w θə [ni céw-ət-əs]
ART AUX help-tr-3sub
'the one (whom) he helped'

In (f) and (f'), the forms of the simple predicate and the corresponding relative clause are identical in appearance. However, the third-person subject markers are probably not the same, that in the simple predicate being the transitive subject marker and that in the relative clause being the subordinate subject marker.

If a third-person subject of a transitive is extracted, the relative clause will have no third-person subject marker. Compare (g) to (i) with (g') to (i'):

(g) céwəθəs ce?	(g´) kʷθə [ċéwəθàmx ce?]
čéw-ət-S(amx)-əs ce?	k™0ə céw-ət-Samx ce?
help-tr-me-3tr FUT	ART help-TR-me FUT
'He will help me.'	'the one who will help me'
(h) čéwəθà·m ce?	(h´) k ^w θə [ccéwəθàmə ce?]
céw-ət-Sam-əm ce?	k ^w θə ccéw-ət-Samə ce?
help-tr-you(pas)-INTR FUT	ART help-tr-you fut
'He will help you.'	'the one who will help you'
(i) ni čéwətəs.	(i') k ^w θə [ni čéwət]
ni? cew-ət-əs	k ^w θə ni? cew-ət
AUX help-tr-3tr	ART AUX help-tr
'He helped him.'	'the one who helped him'

In (h), the form $\dot{c} \epsilon w \partial a m$ in the simple clause is passive, literally 'You are helped,' there being no form in the active paradigm with a second-person object and a third-person subject. However, the corresponding relative clause is active.

In (i'), the absence of the subject marker in the relative clause is what distinguishes it from 'the one he helped,' (f') above.

4.1.1.2. With Extracted Oblique Adjuncts

Relative clauses of this type differ from those of the last type in that in these the predicate head is nominalized. There are also differences within this type depending on whether the extracted nominal adjunct is an oblique object or a locus, goal, or instrument, and on the voice of the predicate.

If what is extracted is an oblique object, the predicate head takes the prefix s- 'nominalizer' (see §12.1.1) and, if the predicate head is intransitive or active transitive, it takes a possessive affix (see §14.2.4) as a subject marker. Compare
the following simple clauses (a) and (b) with the corresponding relative clauses (a') and (b').

- (a) ni cən ?á·m ?ə tə qá?. (AG)
 ni cən ?á-əm ?ə tə qá?
 AUX I call-INTR OBL ART water
 'I called for water.'
- (a') $k^{w}\theta \Rightarrow [n \Rightarrow s^{\gamma} \hat{a} \cdot m] q\hat{a}^{\gamma} (AG)$ $k^{w}\theta \Rightarrow n \Rightarrow s^{\gamma} \hat{a} \cdot \Rightarrow m q\hat{a}^{\gamma}$ ART my-NOM-call-INTR water 'the water I called for'
- (b) ni cən ⁹əxímt ⁹ə k^wθə ⁹əpánəs. (AG) ni⁹ cən ⁹əxím-t ⁹ə k^wθə ⁹ápən-əs AUX I borrow-TR OBL ART ten-face 'I lent him ten dollars.'
- (b') k^wθə [ni nəs?əxímt] (AG) k^wθə ni? nə-s-?əxím-t ART AUX my-NOM-borrow-TR 'what I lent him'

If the predicate is passive, the predicate head in the relative clause is a nominalized passive with the passive person marker (see \$14.2.6) and subordinate passive suffix (see \$10.8). Compare (c) and (c'):

- (c) ni ⁹áx ^wəstəm ⁹ə k^wθə s⁹ðłtən.
 ni⁹ ⁹éx ^w-əs-t-əm ⁹ə k^wθə s⁹ðłtən
 AUX give-REC-TR-INTR OBL ART food
 ⁶ 'They were given food.'
- (c') k^wθə s² sitən [nem s² áx^wəsti·t] k^wθə s² sitən nem s² éx^w-əs-t-əy-ət ART food AUX(go) NOM-give-REC-TR-3SUBPAS-SUBPAS 'the food they were given'

If the extracted oblique nominal adjunct is a locus, goal, or instrument, the predicate head in the relative clause takes the compound prefix δx^{w} - 'oblique nominalizer' (see §12.1.4). This compound prefix is composed of *s*- 'nominalizer' and x^{w} - 'oblique.' It may be roughly translated 'place of, time of, means of, reason for,' and so on. The predicate head also has either a possessive (if intransitive or active transitive) or a subordinate passive person marker (if passive) as a subject marker. It may be preceded by an auxiliary linking it to the noun the clause modifies. Compare (d) to (g) with (d') to (g'):

- (d) ni ní? ?
 ?
 i ní? ?
 i k^w
 spółž
 x
 aux be.there OBL ART prairie
 'He is at the pasture.'
- (d') k^wθə spółžən [ni šx^wní?s] k^wθə spółžən ni? šx^w-ní?-s ART prairie AUX OBLNOM-be.there-3POS
 'the pasture where he is' (lit. 'the pasture that is his place of being there, his locus')
- (e) ni cən ném [?] γ k^wθ xáca?.
 ni? cən ném [?] γ k^wθ xáca?
 AUX I go OBL ART lake
 'I went to the lake.'
- (e') k^wθ × áca? [ni n × x^wn ém]
 k^wθ × áca? ni? n šx^w n ém
 ART lake AUX my-OBLNOM-go
 'the lake where I went' (lit. 'the lake that was my place of going, my goal')
- (f) ni lámɨθəs ?ə tə smé·nt.
 ni lámɨ-ət-S-əs ?ə tə smé·nt
 AUX get.hit-TR-me-3TR OBL ART rock
 'He hit me with a rock.'
- (f') tə smé·nt [ni šx^wláməθəs]
 tə smé·nt ni? šx^w-lám-ət-S-əs
 ART rock AUX OBLNOM-get.hit-TR-me-3SUB
 'the rock he hit me with' (lit. 'the rock that was his means of hitting me')
- (g) ni lám→θèləm '> tə smé·nt.
 ni lám→ət-Sel-əm '> tə smé·nt
 AUX get.hit-TR-I(PAS)-INTR OBL ART rock
 'I was hit with a rock. Somebody hit me with a rock.'
- (g') tə smé·nt [ni šx^wláməθè·lt]
 tə smé·nt ni? šx^w-lám-ət-Sel-ət
 ART rock AUX OBLNOM-get.hit-TR-I(PAS)-SUBPAS
 'the rock I was hit with' (lit. 'the rock that was the means by which I was hit')

4.1.1.3. With Extracted Possessors

If a possessor is extracted, the possessed noun retains the possessive affix, and the relative clause has the form appropriate to the grammatical status of the noun possessed. For example, if the noun possessed is the subject of an intransitive clause, the relative clause will have the same form as the corresponding simple clause. Compare (a) and (a'):

- (a) ni mé?k^wł k^wθə stiqíws k^wθé? swóỷqe?. (AG) ni? mé?k^wł k^wθə stiqíw-s k^wθé? swóỷqe? AUX get.hurt ART horse-3POS that man 'That man's horse got hurt.'
- (a') kwθə (swáyqe?) [ni mé?kwł kwθə stiqíws] (AG) kwθə swáyqe? ni? mé?kwł kwθə stiqíw-s ART man AUX get.hurt ART horse-3POS 'the man whose horse got hurt'

(In considering (a'), AG preferred to omit the noun head, saying that it seemed redundant with the article indicating a male possessor.)

Gerdts (1988, 73-74) has determined that in the Cowichan dialect there are restrictions on what possessors can be extracted. My data are not extensive enough to corroborate this.

4.1.2. Uses of Verbal Relative Clauses

Relative clauses with eclipsed heads can serve as nominal adjuncts. In (a), the relative clause is the subject.

(a) nə hɨye? kwθə [ni-ł céwəθàmx]. (JP) ni hɨye? kwθə ni?-əł céw-ət-Samx AUX leave ART AUX-past help-TR-me
'The one who helped me is gone.'

In (b), it is the object.

(b) ni? cən kwácnəxw kwθə [?i·ł céwəθàmx]. (CC) ni? cən kwéc-nəxw kwθə ?i-əł céw-ət-Samx AUX I look-TR ART AUX-past help-TR-me 'I saw the one who helped me.'

In (c), the relative clause is the possessor of the noun that precedes it.

```
(c) niw x<sup>w</sup>ənhəlí tə t<sup>θ</sup>éle's tθe? ['i·p mi łé·ləstəx<sup>w</sup>]. (JP 22) ni wə-x<sup>w</sup>ən-həlí tə t<sup>θ</sup>éle'-s tθe? 'i-əp mi AUX EST-still-live ART heart-3POS that AUX-you AUX é·l-əstəx<sup>w</sup> move.shoreward-CAUS
'The heart of that which you have brought ashore is still alive.'
```

The examples given above are of relative clauses consisting of predicates only, but relative clauses may also contain nominal adjuncts, as in (d) to (g).

- (d) ni? cən kwácnəxw kwθə [ni? céwət təwka]. (CC) ni? cən kwéc-nəxw kwθə ni? céw-ət təwka AUX I look-TR ART AUX help-TR him 'I saw the one who helped him.'
- (e) ni cən kwácnəxw kwθə [nem ce? céwətəs kwəná-łtən]. (JP)
 ni cən kwác-nəxw kwθə nem ce? céw-ət-əs kwəná-łtən
 AUX I look-TR ART AUX(go) FUT help-TR-3SUB they
 'I saw the one they will help.'
- (f) ... θόt tə [q^wólstəx^w k^wə nəcłilé?əm]. (CC)
 θόt tə q^wél-stəx^w k^wə nə-cłilé?əm
 say ART speak-COM ART my-step.parent
 ... said the one who scolded my late-stepfather.'
- (g) nəw xá?łəmətəm yəθéləy [ni? nem yəsqəqa? ə təwka]. (CC 21) ni wə-xá?łəm-ət-əm [?ə] yəθéləy ni? nem AUX EST-be.watching-TR-INTR OBL those AUX AUX(go) yə-sqaqa? ?ə təwka along-accompanying OBL him 'He was looked after by those who had gone with him.'

In (d), $t \partial w \dot{\lambda} a$ is a third-person object, a role it cannot play in a main clause with a third-person subject indicated by $-\partial s$ suffixed to the verb. Here, in a relative clause, the $-\partial s$ is absent, and there is no such restriction.

Relative clauses can precede their noun heads, as in (h) and (i).

- (h) k^wθə [łáġəlləx^wəx^w] wəłθίθə (CC 10) k^wθə łáġəl-ləx^w-əx^w wəłθίθə ART know-TR-you ritualist 'a ritualist that you know'
- (i) k^wθe? [ni x^wəs?³lyəs] cí·tməx^w (JP 28) k^wθe? ni x^wə-s³³lyə-s cí·tməx^w that AUX become-vision-3POS great.horned.owl 'that great-horned owl that became his vision'

More often, however, it seems that the relative clause follows its noun head, as in (j), (k), and (l).

(j) $k^w \theta \vartheta \vec{k}^w f^2 x^w [y \vartheta h \vartheta \dot{y} q^w ce^2]$ (CC 10) $k^w \theta \vartheta \vec{k}^w f^2 x^w y \vartheta h \vartheta \dot{y} q^w ce^2$ ART pitch along-be.burning FUT 'the pitch that will be burning'

- (k) tə scéxt [ni⁹əł s⁹áx^wəsts] (JP 2)
 tə scéxt ni⁹-əł s-⁹éx^w-əs-t-s
 ART stick AUX-past NOM-give-REC-TR-3POS
 'the sticks he had given them'
- (l) tə sxélməx "cəs [há?k "əxəsəł k "sə nəscáməq"] (CC 19) tə sxélməx "cəs há?k "-əx-əs-əł k "sə ART ritualist's.rattle be.getting.used-TR-3SUB-past ART nə-scáməq" my-great.grandparent
 'the ritualist's rattle that my great-grandmother used'

4.1.3. Nominal Relative Clauses

Nominal predicates, too, can be embedded as relative clauses.² In the texts they have the same form we would expect in corresponding simple clauses. This lack of distinguishing features makes for problems in interpretation, however. First, there are clear examples of nominal relative clauses appearing after their noun heads, as in (a) to (d).

- (a) tə sqwəmqwəméy [stəlqáyə] (JP 3)
 ART dogs wolves
 'the dogs that were wolves'
- (b) k^wθəw mák^w s²áłtən [sxé²xe²s]. (CC 17) k^wθə wə-mák^w s⁻⁹áłtən s-xé²xe²-s ART EST-all NOM-eat NOM-holy-3POS 'all the foods that were his taboos'
- (c) k^wθə spəlk^wít^θa? [wəṫátəł ċtwa? sí·?èm] (JP 16) k^wθə spəlk^wít^θa? wə-ṫát-əł ċtwa? sí·?èm ART dead.person EST-anciently-past SPEC upper-class.people
 'the dead that were leaders long ago'
- (d) tə skíti qəł [[swe?s θə sá?səq^wt] mɨnɨə] (JP 1)
 tə skít qəł s-wé?-s θə sá?səq^wt mɨnɨə
 ART child NOM-own-3POS ART junior child
 'the boy who was the son of the younger sister' (lit. 'the male child who was the son who was the property of the female junior sibling/cousin')

In (d), there is a relative clause (doubly bracketed) within a relative clause. The relative clause $sw e^{2s} \theta s s a^{2s} a q^{w}t$ 'who is the property of (belongs to) the

² If the distinction between verb and noun that I have made is artificial, then every word following an article might be identified as a relative clause, $t \partial sw \partial y q e^{2}$ being 'the one who is a man' rather than 'the man,' and so on.

junior' modifies $m \delta n \delta$ 'who is the child' (the kin term), which modifies $s \tilde{\lambda} t \tilde{\lambda} q \delta t$ 'child' (the age status term). The outer relative clause follows its head, while the inner one precedes its head.

It may be that some noun modifiers that stand before the head are nominal relative clauses. This interpretation seems possible in (e), (f), and (g).

- (e) tə słənłénəý məstáyəx^w (JP 25)
 ART women person
 'the women' (lit. 'the persons who are women'?)
- (f) k^wθə scaléx^wəm məstáyəx^w (CC 17)
 ART old-dancer person
 'the old dancer' (lit. 'the person who is an old-dancer'?)
- (g) tə x*iəlməx məstiyəx (JP 28)
 ART Indian person
 'the Indian people' (lit. 'the persons who are Indians'?)

Compare (g) with (h):

(h) k^wθəw x^wəlməx^w ?al məstəyəx^w
 k^wθə wə-x^wəlməx^w ?al məstəyəx^w
 ART EST-Indian just person
 'the Indian people'

In (h), the combination $w \partial - 2al$ seems to make an adjective-like word of $x^{w} \partial l m \partial x^{w}$ 'Indian.' Here 'Indian' is clearly not a nominal relative clause.

On the other hand, it might be simpler just to identify these modifiers in (e), (f), and (g) as nouns used adjectivally and label them "noun premodifiers."

There are in fact noun premodifiers that cannot be identified as embedded nominal predicates because they relate obliquely (rather than equatively) to the head. They may not be common, but one example is $\delta x^w n \partial w it$ 'far-side' in (i),

(i) tə šx^wnəwíł məstáyəx^w (JP 22)
 ART far.side person
 'the far-side people' (i.e., the people from across the Gulf)

which may be a reduction of (j), which occurred later in the same story.

 (j) yəθéləý [təlí tə šx^wnəwíł] məstáyəx^w (JP 22) yəθéləý təlí [?ə] tə šx^wnəwíł məstáyəx^w those be.from OBL ART far.side person 'those people who are from the far side'

Here the modifier is a verbal relative clause.

Another example of an obliquely related noun premodifier is $l \partial m \partial t \hat{u}$ 'sheep' in (k).

(k) tə ləmətú tíntən nínča (CC 21) ART sheep bell one(DIM) 'a small sheep-bell'

Here, as in (i), it is not possible to interpret either the first or the second noun as a relative clause modifying the other. Neither 'a sheep that is a bell' nor 'a bell that is a sheep' makes sense. But this case may be marginal. The words lomoti 'sheep' and tinton 'bell' are both from Chinook Jargon, and they form here what is probably a loan translation of the English compound *sheep-bell*. So this use of lomoti as a noun premodifier with the oblique sense 'for sheep' may be atypical for Halkomelem. (I suspect that in a more conservative style, one might use the s- a^2t formation and say *slomotional site times the second state to the second

The fact that nouns can stand as modifiers both before and after the nouns they modify means that, at this stage of analysis, it is hard to know how to interpret some juxtaposed nouns. CC interpreted her own

(l) tə s^{*}xéləqəm stəlqáyə ART fierce(PL) wolf(PL)

as 'the fierce wolves,' but DK (her daughter) suggested 'the fierce ones who were wolves.'

JP translated his

(m)təná·nə Żcés smé·nt (JP 15) that.over.there island mountain/rock

as 'that high island,' but AG translated it 'that island that is a mountain.' Later in the story, JP made it $k^{w}\theta a sm ent \lambda c es$, which AG translated either 'rock island' or 'mountainous island.' But again it could be 'that island that is a mountain' or 'that mountain that is an island.'

4.1.4. Adjectival Relative Clauses

Adjectives precede nouns as modifiers in nominal adjuncts (as described in §3.8.3) and in predicates (as described in §3.7.3). They have not been recorded as modifiers following their noun heads. Whether adjective modifiers can be identified as embedded predicates and therefore relative clauses is an open question.

However, adjectives can also appear in adjuncts without nouns, as in (a).

(a) ?áməsθàmx čx^w ?ə k^wθə [?óý]. (AG)
?ém-əs-t-Samx čx^w ?ə k^wθə ?óý
give-REC-TR-me you OBL ART good
'Give me the good one. Give me the one that is good.'

These may be identified as relative clauses with eclipsed heads.

Adjectives also appear in pseudo-cleft sentences (see below), as in (b).

```
    (b) <sup>?</sup>é·nθə [k<sup>w</sup>ámk<sup>w</sup>əm].
    be.I strong
    'I am the one who is strong.'
```

These may be identified as embedded predicates, and so relative clauses in the sense used here.

4.1.5. Pseudo-Cleft and Cleft Sentences

These are sentences that are constructed so as to give prominence or focus to some element (the focal element), such as a noun, pronoun, or interrogative. Both of these sentence types involve the splitting of a simple clause into a focal element and an embedded clause from which the focal element has been extracted.

In a pseudo-cleft sentence, the focal element is the predicate and the embedded clause is either the subject or a predicate complement, as in (a).

```
(a) nówə [ni kwócnəxwèn].
nówə ni? kwéc-nəxw-èn
be.you AUX look-TR-I
'You are the one I saw.'
```

Here $n \delta w \delta$ 'be you' is the predicate and $ni k^w \delta c n \delta x^w e n$ 'whom I saw' is a predicate complement.

In a cleft sentence, the focal element is the subject and the embedded clause is a predicate complement, as in (b).

 (b) wənáy tə swéyəl [wəθə?ít xé?xə]. (JP 21) wənáy tə swéyəl wə-θə?ít xé?xe? be.only ART daylight EST-truly holy 'It is only the Daylight that is truly holy.'

Here $w \partial n \dot{a} y$ 'be only' is the predicate, $t \partial s w \dot{e} y \partial l$ 'the daylight' is the subject, and $w \partial \theta \partial^2 it \dot{x} \dot{e}^2 \dot{x} \partial$ '(that) is truly holy' is a predicate complement.

4.1.5.1. Pseudo-Cleft Sentences

These can have nouns, personal words, and interrogative words as predicates. We will consider them here according to type of predicate.

4.1.5.1.1. With Nominal Predicates

These give prominence to nouns that would be heads of nominal adjuncts in simple sentences. A sentence of this type consists of a noun (or noun with modifying adjective) standing as the predicate, followed by a relative clause formed by the extraction of this noun.

The relative clause may be preceded by an article (as a relative clause with eclipsed head) and serve as the subject of the sentence, as in (a) and (b).

- (a) $s^{2}i \cdot 4q = y^{2}$ $k^{w} + 2 = [ni ni^{2} = 2 + k^{w} + 6e^{2}]$. (JP 13) two-headed.serpent ART AUX be there OBL that 'What is there is a two-headed serpent.'
- (b) sqwəméy kwθə [nəsẳí?]. (JP 2) sqwəméy kwθə nə-s-c-ẳí? dog ART my-NOM-make-value 'What I want is a dog.'

In (a), the predicate is $s^{2i} \cdot lq \partial y'$ be a two-headed serpent' and the subject is $k^{w}l\partial ni ni^{2} \cdot \partial k^{w}\theta d^{2}$ 'that (feminine absent) which is there at that (place).' In (b), the predicate is $sq^{w}\partial m d y'$ be a dog' and the subject is $k^{w}\partial \partial n\partial s \dot{x}i$ 'that which I want,' which is a relative clause with nominalized head because $c \dot{x}i^{2}$ 'want' takes an oblique object. (For this interpretation of $s \dot{x}i^{2}$, see §12.2.1 under [g].)

Sentences of this type with a locative subject, like that of (a), often, though not always, have the existential sense of English 'there is, there are,' as in (c), (d), and (e).

- (c) swóżqe? to [ní? o toní?]. (CC)
 man ART be.there OBL there
 'There is a man over there.' (lit. 'The [male] one over there is a man.')
- (d) słén θθ [ni? θ tení?] (CC)
 woman ART be.there OBL there
 'There is a woman over there.' (lit. 'The [female] one over there is a woman.')
- (e) snáx "əł k "θə [?i ?í ?ə k "ə cécəw]. (CC) canoe ART AUX be.here OBL ART shore
 'There is a canoe on the beach here.' (lit. 'The [thing] here on the beach is a canoe.')

Sentences with this existential sense can also have the relative clause following the predicate directly as a modifier or complement rather than as the subject.

 (f) qáž słanłénaý [nem kwánatam]. (JP 12) qáž słanłénaý nem kwán-at-am many women AUX(go) be.taken-TR-INTR 'There were many women who were captured.'

Sentences like (f) may appear to be simply examples of subject-verb word order, but they are better interpreted as pseudo-cleft sentences.

Sentences with this existential sense can also have ni^{2} 'be there' or 2i 'be here' preceding the noun. The ni^{2} can have the sense 'some,' as in (g) and (h).

- (g) ni? məstáyəx^w k^wθə [məhák^wnəs k^wθə siyéyəsəł niwł nem [?]ík^w].
 ni? məstáyəx^w k^wθə mə-hék^w-nəs k^wθə
 be.there person ART come-remember-GOAL ART syəyéyə-s-əł ni wəł-nem [?]ík^w
 friends-3POS-past AUX already-AUX(go) be.lost
 'Some people came to remember their friends [or relatives] who had passed away.'
- (h) ?i ?ə qáž swíwə [ni ?ə Ž qícəỷ?]. (CC)
 ?i ?ə qáž swíwə ni ?ə Ž qícəỷ
 be.here ROG many eulachon be.there OBL ART Katzie
 'Are there lots of hooligans [eulachons] at Katzie?'

In all of the examples given so far, the noun predicate is an extracted subject. In (i), (j), and (k), they are extracted objects.

- (i) qáž θəlá^γθən [⁹i⁹łtánəstx^wè·n]. (CC) qáž θəlá^γθən ⁹i⁹łtán-əstəx^w-è·n many mouth(DIM.PL) be.eating-CAUS-I
 'I have lots of little mouths to feed.' (lit. 'There are many little mouths that I am feeding.')
- (k) ləyá·m tə słíwayał ['i·p há'kwax]. (CC 21)
 ləyá·m tə s-líwayał 'i-əp há'kw-əx
 devil ART NOM-worship AUX-you(PL) being.used-TR
 'The religion that you are practising is of the Devil.'

In (l) and (m), the noun predicate is an extracted possessor.

- (l) ni? məstáyəx^w [?áy šx^wq^wéləwəns]. (JP 18) ni? məstáyəx^w ?áy šx^wq^wéləwən-s be.there person good feeling-3POS 'Some people were glad.' (lit. 'There were people whose feelings were good.')
- (m) ná·nča? słén >ý [q^w>ós č> t> m>in²>s]. (JP 20)
 ná·nča? słén >ý q^w>ós č> t> m>in²-s
 one.person woman submerge QUOT ART child-3POS
 'There was a woman whose son was drowned.'

In (n), it is an oblique object and so the predicate head in the relative clause bears s- 'nominalizer.'

 (n) sqéwθ k^wθə [ni nəs?əłtánəstəx^w tə sẳíẳqəł]. (AG) sqéwθ k^wθə ni nə-s-?əłtán-əstəx^w tə sẳíẳqəł potato ART AUX my-NOM-eat-CAUS ART child 'It was potatoes that I fed the child.'

And in (o) it is an extracted locus and so the predicate head in the relative clause bears δx^{w} - 'oblique nominalizer.'

- (o) néčewtx^w če lélem te [šx^wni?s [k^ws čł?immáttels te spá·l i te d^wəlítəq]]. (JP 7) lélam ta nécaw-tx^w ca šx^w-ni[?]-s k ^w one-house OUOT house ART OBLNOM-AUX-3POS ART s-cl-?immát-təl-s tə spá·ĺ °əv tə NOM-together-sit(PL DUR?)-each.other-3POS ART raven and ART ἀ^wəlítəq. seagull
 - "Raven and Seagull were living together in one house." (lit. "Their place, when Raven and Seagull were living together, is said to have been one house.")

(The construction beginning with $k^{w}s$ - in (o) is a nominalized clause of a type described in §4.3.2 [7]).

The pseudo-cleft sentence with a nominal predicate is unusual in that the predicate can begin with a demonstrative. Demonstratives, of course, normally introduce adjuncts. In (p), the noun head is preceded by a demonstrative and a particle.

(p) θθ^γín θ k^w θ nθ^γím θ [ni^γé·p k^wék^w θcət]. (JP 26)
 θθ^γín θ k^w θ nθ^{-γ}ím θ ni^γ-é·p k^wék^w θc-θt
 this then my-grandchild AUX-you(PL) be.looking-TR
 'Now it's this granddaughter of mine that you're looking at.'

In (q), the predicate has the same structure but the relative clause is a nominal relative clause.

(q) k^wθe[?] k^w smák^w [st^θók^w ca ?al, mák^w stém st^θók^w] ... (JP 17) k^wθe[?] k^wa smák^w st^θók^w ca ?al, mák^w stém st^θók^w that then ball worm QUOT just all what worm 'that ball now, they say it was just worms [and/or bugs], all kinds of worms ...' (lit. 'It is then that ball that is said to be of worms ...')

4.1.5.1.2. With Pronominal Predicates

A sentence of this type consists of a personal word standing as the predicate head followed by a relative clause. The personal words are $2\delta n \partial \partial \sim 2\delta n \partial \partial 'I$, $n\delta w \partial 'you$,' $tn i m \partial t$ 'we,' $tw \delta l \partial p$ 'you plural,' $\dot{X}\dot{a}$ 'he, she, it,' and $\dot{X}\dot{a} l \partial m$ 'they' (see §14.2.7). The person that is expressed by the predicate head is extracted from the relative clause. The relative clause can be either a predicate complement, following the predicate head directly, or the subject, following an article. Compare the following simple sentences (left) with the corresponding pseudo-cleft sentences (right).

- (a) ni cən ném. AUX I go 'I went.'
- (b) ni ném. ni ném. AUX go 'He went.'
- (c) k^wámk^wəm cən.
 strong I
 'I am strong.'
- (d) nə-méňə yəθéləý.
 my-children those
 'They're my children.'
- (e) ?i ?əmií ?ə X ?é·θə.
 ?i ?əmií ?ə X ?é·θə.
 AUX come OBL ART me
 'He came to me.'
- (f) ni cən k^wácnàmə.
 ni? cən k^wec-n-àmə
 AUX I look-TR-you
 'I saw you.'
- (g) ni kwścnàmxəs. ni? kwec-n-àmx-əs AUX look-TR-me-3TR 'He saw me.'
- (h) ni c'a k^wácnà·m.
 ni[?] c'a k^wéc-n-ám-am
 AUX QUOT look-TR-you-INTR
 'He saw you. You were seen.'

- (a') [?]é·nθə [ni ném].
 be.I AUX go
 'I am the one who went.'
- (b´) Ẋ́á [ni·ł ném].
 Ẋ́á ni?-əł ném
 BE3P AUX-past go
 'He's the one who went.'
- (c') ²é·nθə [k^wámk^wəm].
 be.I strong
 'I am the one who is strong.'
- (d') Žáləm [nə-méməňə]. be.they my-children 'THEY are my children.'
- (e') ?é·nθə [?i šx "əṁís].
 ?é·nθə ?i šx "-?əṁí-s.
 be.I AUX OBLNOM-come-3POS
 'I am the one he came to.'
- (f') nówo [ni kwócnoxwèn].
 nówo ni? kwec-noxw-èn.
 be.you AUX look-tr-I
 'You are the one I saw.'
- (g') ²é·nθə [ni·ł kwścnəxwəs].
 ?é·nθə ni²-əł kwec-nəxw-əs be.I AUX-past look-TR-3SUB
 'I am the one he saw.'
- (h´) nówo co [ni kwocnoxwos].
 nówo co ni? kwocnoxwos].
 be.you QUOT AUX look-TR-3SUB
 You are the one he saw.'

Reportedly, speakers vary in their treatment of complement clauses in sentences of this type. For (g') and (h'), AG gave (i) and (j) as forms he had heard used but judged less correct.

- (i) ⁹é·nθə ni k^wéc-n-amxəs.
 ⁹é·nθə ni k^wéc-n-amx-əs be.I AUX look-TR-me-3TR 'I am the one he saw.'
- (j) nówo ni k^wócnà·m.
 nówo ni k^wéc-n-am-om
 be.you AUX look-TR-you-INTR
 'You are the one he saw.'

In these, the speaker used simple predicates instead of relative clauses. I do not believe CC or JP used such forms.

4.1.5.1.3. With Interrogative Predicates

Some interrogative words such as *wét* 'who,' *stém* 'what,' and so on (see \$17) can be used in questions in which the interrogative word is the predicate and the subject a relative clause, usually with eclipsed head, as in (a), (b), and (c).

- (a) wét k^w ö k^w ö ná tan [yömí]. (CC) wét k^w ö k^w ö ná tan yö ?ömí who then those long-come 'Who are those coming then?'
- (b) stém [?]a[?]a tə [niəx^w k^wənét]. (JP) stém [?]a[?]a tə ni-əx^w k^wən-é-t what ROG ART AUX-you be.taken-DUR-TR 'What is that thing you have?'
- (c) scék »əl k »ə k »ə [ni?əx » łédət]. (CC) scék »əl k »ə k »ə ni?-əx » łéd-ət how.much then ART AUX-you lie-TR 'How much did you put down (as down payment)?'

Other examples appear in §17.

4.1.5.2. Cleft Sentences

In a sentence of this type, the predicate is $\lambda \dot{a}$ 'be third person' (BE3P) or one of a small number of other words that can link a subject, which is the focal element, and a complement, which is a relative clause formed by the extraction of that subject. For example:

(a) Xa tə tám [ni·n cécəwət]. (JP)
 Xa tə tám ni?-ən cécəw-ət
 BE3P ART Tom AUX-I be.helping-TR
 'It's Tom that I am helping.'

Here $\dot{\lambda}a$ is the predicate, 'Tom' the subject, and 'that [whom] I am helping' a relative clause from which Tom has been extracted.³

Other examples of this type of sentence follow in (b) to (i).

- (b) X a tθé? [ném ce?]. (JP)
 BE3P that(MP) go FUT
 'He is the one who will go.'
- (c) Ža tə tám [ni čéwəθàmx]. (JP)
 Ža tə tám ni? čéčəw-ət-Sàmx
 BE3P ART Tom AUX be.helping-TR-me
 'Tom is the one who is helping me.'
- (d) Xa ce? m t t e? swóy qe? [céwə t am a]. (JP)
 Xa ce? m t t e? swóy qe? céw-at-Sàma
 BE3P FUT CERT that man help-TR-you
 'It's that man who will help you.'
- (e) Xa tə?ínə [nəsXí?]. (CC)
 Xa tə?ínə nə-s-c-Xí?
 BE3P this my-NOM-make-valued
 'This is the one I want.'
- (f) X tθe? smé·nt [ni? šx^wlámaθs]. (CC)
 X tθe? smé·nt ni? šx^wlámaðs]. (CC)
 BE3P that rock AUX OBLNOM-get.hit-TR-me-3POS 'That's the rock he hit me with.'
- (g) Žá m o to?í smént [ni? šx "lámoðe·lt]. (CC)
 Žá m o to?í smént ni? šx "-lámot-ot-Se·l-t
 BE3P CERT this rock AUX OBLNOM-get.hit-TR-I-SUBPAS 'This is the rock I was hit with.'
- (h) Xa k^wθe[?] swéyəl [ni·n šx^wcéwət]. (CC)
 Xa k^wθe[?] swéyəl ni[?] nə-šx^w-céw-ət
 BE3P that day AUX my-OBLNOM-help-TR
 'That was the day when I helped him.'

³ The relative clause following the subject is a complement rather than a modifier of the subject. Example (a) is not 'The Tom whom I am helping is the one,' which would make sense only if there were several Toms. Nor is it 'Tom, whom I am helping, is the one,' since there do not seem to be non-restrictive relative clauses.

(i) Ža ?ə ti?í [ni ?əšx^wk^wáq^wət]. Ža m̊ə.
Ža ?ə ti?í ni? ?ə-šx^w-k^wáq^w-ət Ža m̊ə
BE3P ROG this AUX your-OBLNOM-get.hit-TR BE3P CERT
'Is this what you hit him with? Yes, it is.'

In these examples, the relative clause follows the subject. However, it is also possible for the subject to follow the relative clause, as in (j) and (k). It is not clear what difference, if any, this makes in the sentence.

- (j) Ža [ni·ł čéwətè·n (~ ni·nəł čéwət)] tə tám. (JP)
 Ža ni?-əł čéw-ət-è·n (~ ni?-ən-əł čéw-ət) tə tám BE3P AUX-past help-TR-I (AUX-I-past help-TR) ART Tom 'Tom is the one I helped.'
- (k) Ža ni·ł čéwəθamx tθé?. (JP)
 Ža ni?-əł čéw-ət-Samx tθé?
 BE3P AUX-past help-TR-me that(MP)
 'He is the one who helped me.'

In (l), we have a triply complex sentence, containing three nesting relative clauses.

- (l) k^wθe[?] məstáyəx^w [Åa [ni há[?]k^wəx t^θə [k^wík^wəxət t^θə q^wíq^wəməs]]].
 (JP 28)

'It's that person who is the one that uses what we call the "qwíqwməs" [hair headdress].'

It is a pseudo-cleft sentence in which the predicate $(k^w \theta e^2 m \partial s t \dot{\delta} y \partial x^w)$ is extracted from a cleft sentence (of the form $\dot{\lambda}a \ k^w \theta e^2 \ m \partial s t \dot{\delta} y \partial x^w$ ni $h \dot{\delta}^2 k^w \partial x \ t \partial \dots$ 'It is that person who ...'), in which the object $(t^\theta \partial \ k^w i k^w x \partial t \dots)$ is a relative clause with eclipsed head.

There are a few other words that can function instead of $\dot{\lambda}a$ as predicates of cleft sentences or in predicates of pseudo-cleft sentences. These include $h\dot{a}y$ 'specifically,' $won\dot{a}y$ 'only,' $tx^w\dot{a}y$ 'only remaining,' and perhaps a few more, as in (m) to (p).

- (m)i háy θə náňača? [ni tás ?a x díčaý]. (CC 11) and specifically ART one.person AUX reach OBL ART Katzie
 'But the other one went to Katzie.'
- (n) wənáy tə ləplít [nəšx^wnə?ás]. (CC 13) wənáy tə ləplít nə-šx^w-nə?-ás be.only ART priest my-OBLNOM-be.there-face 'The priest was the only one I looked to.'

ART headdress

(o) tx wáy tə łníməł [ni həwhówdw]. (JP 22) tx wáy tə łníməł ni həwhówdw be.only.remaining ART be.we AUX be.drifting(PL) 'We were the only ones left drifting around.'

An adverb ($\S18.2$) can appear within the relative clause in a cleft sentence, as in (p).

(p) Ža wəyáθ wəšx "ón öm tə Safeway. (CC)
 Ža wə-yáθ wə-šx "-h ón ôm [?ə] tə Safeway
 BE3P EST-always EST-OBLNOM-be.going OBL ART Safeway
 'Safeway is where I always go.'

The following seem to be elliptical versions of this kind of sentence, with the $\dot{\chi}a$ omitted in all three and the subject also omitted in the second and third:

- (q) wəłíθ łádallax *é·n k*θe?. (JP)
 wəł-híθ łádal-lax*-é·n k*θe?.
 already-last.long know-tR-I that
 'He's somebody I've known for a long time.'
- (r) wəlíθ 'al sqálqələθəns. (CC 16)
 wəl-híθ 'al s-qálqələθən-s already-last.long just NOM-dream-3POS
 'It's what he was dreaming about for a long time.'
- (s) tx wáy qwáqwəlstamx. (DK) tx wáy qwáqwəl-st-amx be.only.remaining be.speaking-COM-me 'He's the only one who still speaks to me.'

In the last example, the absence of a third-person transitive *-as* makes it clear that the second word is a relative clause.

4.2. SUBORDINATE CLAUSES

A subordinate clause is one that stands apart from the main clause of a sentence but in a relationship subordinate to or dependent on the main clause, such as "when he comes" in "I'll see him when he comes" or "that he goes" in "I'll see that he goes."

A subordinate clause is produced by prefixing one of the two subordinating particles, $w_{\partial -_i}$ 'if, when, that' and ' ∂t - 'whenever, whatever,' to the first word in the predicate and replacing its coordinate subject person marker with a sub-ordinate subject person marker.

The subordinate subject person markers in intransitive or transitive active predicates are $-e \cdot n \sim -\partial n$ 'I,' $-\partial x^{w}$ 'you,' $-\partial t$ 'we,' $-e \cdot p \sim -\partial p$ 'you (plural),' and $-\partial s$ 'third person' (§14.2.3). In passive predicates, the first- and second-person

passive person markers, $-Sel - \sim -el - 1$, $-Sam - \sim -am - 'you$, -al - 'we, you (plural), and $-\partial y$ - 'third person,' are followed by $-\partial t$ 'subordinate passive.' The third-person marker in the subordinate passive is $-\partial y$ - followed by $-\partial t$, usually realized as -it (see §14.2.6 and §10.8).

Compare the following main-clause forms (on the left) with the corresponding subordinate-clause forms (on the right):

(a)	ném cən.	(a′)	wənémè∙n	
	ném cən		wə-néṁ-è∙n	
	go I		if-go-I	
	ʻI go.'		'if/when/that I go'	
(b)	k̃™əcnéləm.	(b′)	wək ^w əcné·lt	
	k [™] ec-n-él-∍m		wə-k ^w ec-n-él-t	
	look-tr-I-intr		if-look-tr-I-subpas	
	'I am seen.'		'if I am seen'	
(c)	ni ⁹ ítət.	(c′)	⁹ əłniəs ⁹ ítət	
	ni ⁹ ⁹ ítət		⁹ əł-ni ⁹ -əs	?ítət
	AUX sleep		whenever-AUX-3SUB	sleep
	'He slept.'		'whenever he slept'	-

Subordinate clauses are used as conditional clauses, indirect commands, indirect questions, and complements of certain kinds of main clauses of temporal and interrogative meaning.

4.2.1. Conditional Clauses

Subordinate clauses with $w\partial$ - and with $?\partial l$ - are both commonly used as conditional clauses. The difference between them is that a subordinate clause with $w\partial$ - refers to a hypothetical future event, while one with $?\partial l$ - refers to a real ongoing activity or series of repeated events. Compare (a) with (b) and (c) with (d):

- (a) k^w ścn əx^w čx^w ce? wə?əmii s técəl. (CC) k^w ec-nəx^w čx^w ce? wə-?əmii-əs técəl look-TR you FUT when-AUX(come)-3SUB arrive.here 'You'll see him when he comes.'
- (b) ni ?ə čx* k*ék*əcnəx* ?əłmis técəl. (CC)
 ni ?> čx* k*ék*əcnəx* ?əł-?əmi-əs técəl
 AUX ROG you be.looking-TR whenever-AUX(come)-3SUB arrive.here
 'Do you see him, when he comes?'
- (c) x^wcél čx^w ce? k^wə wəniəx^w ce? höye? təli? ə təna. (CC)
 x^wcél čx^w ce? k^wə wə-ni?-əx^w ce? höye? təli? ?ə təna
 go.where you FUT then when-AUX-you FUT leave from OBL this
 'Where are you going when you leave here?'

(d) ni [?]əlyámətəs [?]əłniəs [?]ítət. (JP 28) ni [?]əlyə-mət-əs [?]əł-ni[?]-əs [?]ítət AUX have.vision-CON-3TR whenever-AUX-3SUB sleep 'He dreams about it whenever he goes to sleep.'

Time in the future may be expressed by a subordinate clause with a conditional sense, as in (e) and (f).

- (e) k^wəcnámə cən ce? wəwéyələs. (CC) k^wec-n-ámə cən ce? wə-wéyəl-əs see-TR-you I FUT when-become.day-3SUB
 'I'll see you tomorrow.' (lit. 'I'll see you when [another] day comes.')
- (f) ?əṁí čx^w wəx^wəné·ntəs. (CC 21)
 ?əṁí čx^w wə-x^wə-né·nt-əs
 come you when-become-becoming.night-3SUB
 'Come this evening.' (lit. 'Come when it becomes evening.')

Subordinate clauses with a conditional sense may appear after ²dy 'good,' as in (g).

(g) 'ôý ctwa' qo wonémen tá'olt. (JP 3)
'ôý ctwa' qo woném-en tól-t
good SPEC EMPH if-go-I be.understood-TR
'I think it would be better if I went to find out!'

Nominalized clauses occur more often in this context, however (§4.3.3). The conditional sense also seems present in the following sentences with 2 *d*-.

- (h) stə?é čx^w ?əłq^wəlstámə?è·n. stə?é čx^w ?əł-q^wel-st-ámə-?è·n be.like you whenever-speak-com-you-I 'You do whatever I tell you.' (lit. 'You comply whenever I speak to you.')
 (i) hí?, háy łe ?əłpəlqélsəx^w. (JP, quoting grandson to flirting girl)
- hí?, háy łe ?əł-pɔlq́-éls-əx^w
 hey stop PER whenever-glimpse-ACT-you
 'Hey, stop making googoo eyes at me.' (lit. 'Hey, stop whenever you take sidelong glances.')

Subordinate clauses usually follow main clauses, as in the preceding examples, but this is not invariably so, as (j) and (k) show.

(j) wəmi'?è·n xwəntécəl ?i ?ə čxw ce? wə?í. (CC) wə-mií-?e·n xwəntécəl ?i ?ə čxw ce? wə?í if-come-I still-arrive.here AUX ROG you FUT EST-be.here 'If I come early, will you be here?' (k) wəni?əs ce? k*ə wəłxák*, Ża ce? k*θə wəłθíθə [θáyt k*θə scécext [šx*xíxəq́a?q*əm ce?]]. (CC 17)
wə-ni?-əs ce? k*ə wəł-xák* Ża ce? k*θə wəłθíθə when-AUX-3SUB FUT then already-bathe BE3P FUT ART ritualist θáy-t k*θə scécext šx*-xíxəq́-á?q*-əm-s ce?
be.made-TR ART little.stickOBLNOM-be.scratching-head-INTR-3POS FUT 'When he [the new dancer] has finished bathing, then it is the ritualist who will fix the little stick that he will be scratching his head with.'

Sentence (k) also includes two nesting relative clauses, the second (in inner square brackets) modifying a word in the first.

4.2.2. Indirect Commands

Subordinate clauses with wa- occur with this function following main clauses with verbs referring to instructing, requesting, and so on.

- (a) csét cən ce? wənéməs. (CC)
 cəs-ét cən ce? wə-ném-əs
 tell-TR I FUT that-go-3SUB
 'I'll tell him to go.'
- (b) ni cseθámxəs wənémèn. (CC) ni? cəs-et-Sámx-əs wə-ném-èn AUX tell-TR-me-3TR that-go-I 'He made me go.'
- (c) x > t > s k wθ> dákt> w> t él>q > t > n. (JP)
 x > t > s k wθ> dákt> w> t él>q > t > t > n
 be.telling-com-me-3TR ART doctor that-stick.out.tongue-I
 'The doctor told me to stick out my tongue.'
- (d) ni cən nem yá t łə Edna wə té c dàm xəs kw swétə. (JP) ni? cən nem yáh-ət łə Edna wə té c-t-tam x-əs AUX I AUX(go) alert-TR ART Edna that-make-BEN-TR-me-3SUB [?ə] kw swétə OBL ART sweater
 'I went and ordered a sweater from Edna '

4.2.3. Indirect Questions

Subordinate clauses with wa- occur with this function when they contain interrogative words (§17) and follow main clauses with verbs referring to knowing, discovering, and so on.

- (a) ni cən Åəw mélqt wəni?əs ?ə́nəcə θə tíntən. (CC 21)
 ni cən Åe wə-mélq-t wə-ni?-əs ?ə́nəcə θə tíntən
 AUX I also EST-forget-TR that-be.there-3SUB where ART bell
 'I too had forgotten where the bell was.'
- (b) nəwł łódallax was i kwi šx wqwelawans wani s ctámat kwθa siyeyasał.
 (JP 10)
 - ni?wəł-łódjal-lax *-əs?i[?ə] k*i šx *q*élawən-sAUX already-know-TR-3TRbe.hereOBLthatmind-3POSwə-ni-əsctámətk*θəsiyéyə-s-əłthat-AUX-3SUBdo.what.with.oneselfARTfriends-3POS-past'He already knew in his mind what had happened to his late friends.'

4.2.4. Complements of Main Clauses Referring to the Future

Subordinate clauses with *wə*- appear after main clauses referring to future time, as in (a).

(a) wéyəl ce? wədéwəθamèn. (CC)
wéyəl ce? wə-déw-ət-Samə-èn
become.day FUT that-pay-TR-you-I
'I'll pay you tomorrow.' (lit. 'It will be tomorrow that I will pay you.')

Here the subordinate clause is not, properly speaking, a conditional clause, since my paying you is not a condition for the sun rising tomorrow. Compare with (b), given earlier in §4.2.1 (e).

(b) k^wəcnámə cən ce? wəwéyələs.
 k^wec-n-ámə cən ce? wə-wéyəl-əs see-TR-you I FUT when-become.day-3SUB 'I'll see you tomorrow.'

Here 'tomorrow' is a conditional clause and a condition for my seeing you.

Nor can the subordinate clause in (a) be identified as 'when I pay you' in the sense of a relative clause 'on which I pay you.' Compare with (c), given in "Cleft Sentences" (§4.1.5.2).

 (c) X a k^wθe[?] swéyəl ni·n šx^wcéwət.
 X a k^wθe[?] swéyəl ni-ən šx^w-céw-ət be3P that day AUX-I OBLNOM-help-TR 'That was the day when/on which I helped him.'

Here $ni \cdot n$ ($ni \ na$ -) $\delta x^w \dot{c} \dot{e} w a t$ 'on which I helped him' is a proper relative clause identifying $sw \dot{e} y a l$ 'day' as an oblique nominal adjunct in the corresponding main clause, with the sense 'I helped him on that day.'

Presumably, $w \acute{e} y \imath l c e^{2}$ in (a) is not followed by a true relative clause because is not a noun (like $sw\acute{e} y \imath l$) but a verb. Perhaps, then, a subordinate clause provides the only way of relating a future event to a verb relating to future time.

In "when" questions about the future, we find the same construction, as in (d). (See also §17.5.)

(d) təmtém ce? kwə wəniəxw háye?. (CC) təmtém ce? kwə wə-ni?-əxw háye? when FUT then that-AUX-you leave 'When are you leaving?'

4.2.5. Interrogative Complements with ?>ł-

Following an interrogative, the prefix $2\partial t$ and the suffix $-\partial s$ 'third-person subordinate subject' can be affixed to a noun to give the sense 'kind of a,' as in (a) to (e).

- (a) stém ?əłsq^wəméyyəs tθe?. (JP) stém ?əł-sq^wəméyy-əs tθe? what that-be.dog-3SUB that
 'What kind of a dog is it?' (lit. 'What that it may be a dog is that?')
- (b) stém ?əłsqwəmd rəmey skwa ?iəxw kwənét. (JP)
 stém ?əł-sqwəmd rəmey skwa ?i-əxw kwən-é-t
 what that-dogs-3SUB ART AUX-you be.held-DUR-TR
 'What kind of dogs do you have?'
- (c) scék vəl ?al ?əłməstáyəx vəs k vθé?. (JP) scék vəl ?al ?əł-məstáyəx v-əs k vθé? how just that-person-3suB that 'What kind of a person is that guy?'
- (d) stém kwə ?əłskwəléxəs kwə ni ?əšxwkwəléxt. (CC) stém kwə ?əł-s-kwəléx-əs kwə ni? ?ə-šxw-kwəléx-t what then that-NOM-shoot-3SUB ART AUX your-OBLNOM-shoot-TR 'What kind of a gun did you shoot him with?' (lit. 'What kind of a gun was the one you shot him with?')
- (e) Že ?> k^w> w>néć ?>łsq^w>méý>s. (CC)
 Že ?> k^w> w>-néć ?>ł-sq^w>méý->s
 also ROG then EST-different that-dog-3SUB
 'Was it another kind of dog?'

This use of $2\partial t \cdots \partial s$ seems to be restricted to contexts like those in the examples given, and the use is difficult to interpret. The relevant facts are these: There is no separate word for 'kind,' although there are lexical suffixes with the

meaning 'kind' and these appear in words for 'how many kinds,' 'one kind,' and so on. The word *stém* 'what,' however, appears not to combine with lexical suffixes (except -*c* 'tribe'). And so, again (as with a main clause referring to future time), the subordinate clause may serve a purpose not open to other constructions. The prefix may be $2\partial t$ - rather than $w\partial$ - because $2\partial t$ - refers to a continuing state while $w\partial$ - refers to a future or unreal state.

Unlike the construction with $w\partial$ - and $?\partial t$ - identified in the preceding sections (§4.2.1 to 4.2.4.), those described in this section seem to be expansions of the main predicate rather separate subordinate clauses. The nominal adjuncts in the examples given in this section seem to be subjects of the whole predicates that precede them rather than subjects simply of the subordinate clauses, that is, the immediate constituents of (a) are $st\acute{em}$ $?\partial tsq^w \partial m\acute{e}\dot{y}\partial s$ and $t\theta e^2$, rather than $st\acute{em}$ and $?\partial tsq^w \partial m\acute{e}\dot{y}\partial s$ the?

4.3. NOMINALIZED CLAUSES

A nominalized clause is formed by prefixing *s*- 'nominalizer' (or δx^{w} - 'oblique nominalizer' in one subtype) to the first word of a predicate. If the predicate is intransitive or active transitive, its subject must be marked by a possessive affix instead of a main-clause subject marker. Compare the following predicates (left column) with the corresponding nominalizations (right column):

(a)	ném cən.	(a´) nəsném
	ném cən	nə-s-ném
	go I	my-NOM-go
	ʻI goʻ	'my going'
(b)	k ^w ə́cnəx ^w əs.	(b') sk wácn a x ws
	k ^w ec-nəx ^w -əs	s-k ^w ec-nəx ^w -s
	look-tr-3tr	NOM-look-tr-3pos
	'he sees it'	'his seeing it'
(c)	ni cən k vəcnəx v.	(c´) nəsni k [*] *ə́cnəx*
	ni cən kwec-nəxw	nə-s-ni? kwec-nəxw
	AUX I look-tr	my-nom-aux look-tr
	'I saw it'	'my having seen it'

Passive predicates can be nominalized in their subordinate forms only and do not require possessives, since their subjects are already indicated by their passive affixes. Compare (d) and (e) with (d') and (e'):

(d) k ^w əcnéləm	(d') skwəcné·lt
k ^w ec-n-él-əm	s-k ^w ec-n-él-ət
look-tr-I-intr	NOM-look-tr-I-subpas
'I am seen'	'my being seen'

(e) k ^w ácnəm	(e´) skwəcní•t
ḱ ^w eс-n-әт	s-k ^w ec-n-áy-ət
look-tr-intr	NOM-look-tr-3pas-subpas
'he is seen'	'his being seen'

Nominalized clauses can appear within predicates, and they can appear as direct nominal adjuncts, as oblique nominal adjuncts, and as narrative sentences.

4.3.1. Within Predicates

Nominalized clauses can appear following the adverbial words $y \acute{e}t$ 'just now,' $w \partial n \acute{y}$ 'only,' $m \acute{z} \acute{k}^w$ 'all, every,' and perhaps a few others (see §18), as in (a) to (e).

- (a) yéł m
 nəstécəl. (CC)
 yéł m
 nə-s-técəl
 just.now CERT my-NOM-arrive.here
 'I just got here.'
- (b) yéł ma stécals. (CC)
 yéł ma s-técal-s
 just.now CERT NOM-arrive.here-3POS
 'He just got here.'
- (c) yéł m
 n posi técpl. (CC)
 yéł m
 n posovani necesi técpl
 just.now CERT my-NOM-AUX arrive.here
 'I just got here.'
- (d) yéł m
 sos técol. (CC) yéł m
 s-?i-s técol just.now CERT NOM-AUX-3POS arrive.here 'He just got here.'
- (e) yéł m s s ném. (CC)
 yéł m s ni?-s ném
 just.now CERT NOM-AUX-3POS go
 'He just went.'

In (d) and (e), the auxiliary appears as a. In (c), it is probably 2i 'be here,' consistent with *técal* 'arrive here.' But in (d) it is probably ni^2 'be there,' consistent with $n\acute{em}$ 'go.' (JP often used *sis* in seemingly parallel constructions.)

(f) wənáy sq^wéls təwÅa šx^wné?em ... (CC) wənáy s-q^wél-s təwÅa šx^wné?em be.only NOM-speak-3POS that shaman 'That shaman only said ...' (g) môk^w nôsném ?ímôx ... môk^w nô-s-ném ?ímôx every my-NOM-go walk 'Whenever I go about ...'

A construction of this type with $m \delta \vec{k}^w$ 'all, every' has the sense of a conditional clause, but grammatically it is the first of a pair of coordinate clauses (see §5.2.3).

The personal word $\vec{\lambda}\vec{a}$ 'be third person' (see §14.2.7) can be followed by a nominalization with $w\partial t$ - 'already,' the construction having the sense 'be now about to, be now starting,' as in (h) to (k).

- (h) Xan səwł tíləm. (CC)
 Xa nə-s-wəł-tíləm
 BE3P my-NOM-already-sing
 'I'm going to start to sing.'
- (i) Ža mai sawł tilams. (CC)
 Ža mai sawł tilams. (CC)
 BE3P CERT NOM-already-sing-3POS
 'He's going to start to sing.'
- (j) Ža m
 ^{*}a v
 ^{*}əsəwł žté[?]stəx^w. (CC)
 Ža m
 ^{*}a v
 ^{*}a-s-wəł-žté[?]-stəx^w
 BE3P CERT your-NOM-already-do-COM
 ^{*}Do it now.^{*}
- (k) Ža mon sowł nóm. (CC)
 Ža monowa no-s-woł-nóm
 BE3P CERT my-NOM-already-go
 'I'm going now.' (lit. 'It is my already going.')

AG gave his equivalent of (k) as $\dot{\lambda}a \ \dot{m}a \ nasu \ walnem, with the sequence \ walnem, walnem, with the sequence \ walnem, waln$

The word $\dot{X}a$ can also introduce a nominalized clause beginning with *s*-'nominalizer' followed by wa_{2} 'established,' commonly, but not always, realized as saw, but these nominalized clause also commonly occur without any preceding $\dot{X}a$. Because of this, I treat them as a separate type. See §4.3.4.

4.3.2. As Nominal Adjuncts

Nominalized clauses can appear as nominal adjuncts in a number of ways.

(1) As subjects. They can appear as subjects of intransitive verbs, as in (a).

(a) xóq k^wθən sťk^wéwəł. (JP)
xóq k^wθə nə-s-ťək^w-éwəł.
be.finished ART my-NOM-get.stuck-vessel
'I'm all finished with my "corking" [caulking of a boat].' (lit. 'Finished is my caulking.')

They may be subjects of verbs that are adverbial in sense, referring to manner, as in (b).

(b) yəxáxxəxdət tə sxwənxénəms. (JP 10) yə-xáxxəxdət tə sxwənxénəm.
yə-xáxxəxdət tə s-xwənxénəm-s along-turning.from.side.to.side-self ART NOM-run-3POS
'He ran zigzagging along.' (lit. 'His running was turning itself from side to side.')

Or referring to time (perhaps past only), as in (c) and (d).

- (c) híθ k^wən s⁹ím ax.
 - híθ k^wə nə-s-⁹íməx last.long ART my-NOM-be.walking 'I walked for a long time.'
- (d) xăa²áθən swéyəl k^wə nəsni² sk^wtéx^w.
 xăa²áθən swéyəl k^wə nə-s-ni² sk^wtéx^w four day ART my-NOM-AUX inside
 ^cI was in there for four days.²

Nominalized clauses also very often appear as subjects with predicates that are modal in sense (see §18.3), as in (e), (f), and (g).

(e) ⁹ 'θý k^ws ἀa·nθətct. (CC 21)
⁹ 'θý k^wθ s-ἀen-θθθt-ct good ART NOM-return-self-our
'We'd better return.' (lit. 'Our returning would be good.')

Because the article $k^w \partial \sim k^w$ implies remote or non-existent status (see §14.1), an even more literal version of (e) might be 'Be good our hypothetical returning.'

(f) nəsẳí? kwə nəsném.
nə-s-c-ẳí? kwə nə-s-ném
my-NOM-make-valuable ART my-NOM-go
'I want to go.' (lit. 'My hypothetical going is what I want.')

In (f), $n \partial s \dot{\lambda} t^{\gamma}$ is probably a relative clause (see §12.2.1 [g]) rather than a nominalized clause. (g) skwéy kwa nastílam.

sk^wéy k^wə nə-s-tíləm impossible ART my-NOM-sing 'I can't sing.' (lit. 'Impossible is my singing.')

(2) As objects. A nominalized clause can appear as the object of a transitive verb, as in (h).

(h) ?é·nθə ce? xqát tə ?əssqálcəp. (CC 12)
?é·nθə ce? xəq-ət tə ?ə-s-səq-əlcəp
be.I FUT finish-TR ART your-NOM-split-firewood
'I will finish splitting the wood for you myself.' (lit. 'I will be the one who will finish your wood-splitting.')

(3) As oblique objects. In sentences like (i) to (l), the nominalized clauses probably relate to the predicates as oblique objects, although the oblique particle does not appear.

- (i) sí səy cən k vən sném.
 sí səy cən k və nə-s-ném
 fear I ART my-NOM-go
 'I'm afraid to go.' (lit. 'I am afraid of my [hypothetical] going.')
- (j) ni tətá?θət k^ws ?íməxs.
 ni tətá?-θət k^wə s-?íməx-s
 AUX be.testing-self ART NOM-walk-3POS
 'He is trying to walk.' (lit. 'He's testing himself in his walking.')
- (k) ⁹íc'al ct k^ws némct.
 ⁹íc'al ct k^wa s-ném-ct
 uncomplying we ART NOM-go-our
 'We refuse to go.' (lit. 'We are uncomplying in our going.')
- (1) ni⁹əł cən mə šx^wte⁹é·wən k^wə nəsném.
 ni⁹-əł cən mə šx^wte⁹é·wən k^wə nə-s-ném
 AUX-past I CERT conceive ART my-NOM-go
 'I've been intending to go.' (lit. 'I have certainly conceived of my going.')

In all these sentences, the nominalization is translated most naturally by an English infinitive. In some sentences, however, such as (l), the nominalized clause might also be translated by an English subordinate clause with "may," or in an older style with a subjunctive form, identifying the event as hypothetical: 'I have certainly conceived that I may go.'

(4) As indirect quotations, as in (m).

```
(m)ni ma xáťa k<sup>w</sup>a sni?s ném.
    ni? ma
                  х́э́tэ
                               kwa s-ni?-s
                                                       ném
    AUX CERT be.saying ART NOM-AUX-3POS go
    'He said that he went '
  (5) As indicators of purpose, as in (n).
(n) <sup>9</sup>i cən cláysəns k<sup>w</sup>ə nəshá<sup>9</sup>k<sup>w</sup>əx tə stíwəyəł scénəm. (CC 22)
          cən c-lávsəns kwə nə-s-há?kw-əx
    ?i
                                                                tə
                                                                      s-tíwəvəł
                 get-licence ART
                                      my-NOM-be.using-TR ART NOM-worship
    AUX I
       s-cénəm
       NOM-shake
    'I have a licence to use the Shaker religion.' (lit. 'I have a licence for my
       using the religion that is the Shaker.')
  (6) As indicators of cause, as in (o) and (p).
(o) <sup>9</sup>i cən x<sup>w9</sup>ítcsəsmət tθe<sup>9</sup> swáyge<sup>9</sup> k<sup>w</sup>s<sup>9</sup>áys k<sup>w</sup>θe<sup>9</sup> léləms. (JP)
           cən x<sup>w</sup>?ítcsəs-mət tθe? swáyqe? k<sup>w</sup>ə s-?áy-s
    ?i
    AUX I
                 envious-con
                                    that
                                            man
                                                        ART NOM-good-3POS
       k<sup>w</sup>θe? léləm-s
                house-3POS
       that
    'I am envious of that man because he has a good house.' (lit. 'I envy that
       man his house's being good.')
(p) scécon ctow hílok w kwict kwocnámo, háy čxw do kwe?es?i mi
       k<sup>w</sup>écətàlx<sup>w</sup>. (CC)
    scécon ct wo-hílok<sup>w</sup> k<sup>w</sup>
                                        s-ni-ct
                                                         k<sup>™</sup>ec-n-ám⊋
                                                                           háv
    really
              we EST-happy ART NOM-AUX-we see-TR-you
                                                                           finish
                       k<sup>w</sup>ə ?es-?i
                                          mi
                                                   k<sup>w</sup>éc-ət-àlx<sup>w</sup>
       čx w
              ģ
       vou
              EMPH ART
                             your-AUX come
                                                   see-TR-US
    'We were really happy to see you. Thank you for coming to see us.'
(But cf. nominalized clauses with \delta x^{w}- in §4.3.3.)
  (7) Adverbial adjuncts referring to past time. These correspond to English
adverbial clauses introduced by 'when' in the sense of 'at the time (in the past)
that' (in contrast to 'when' in the sense of 'if'; cf. w_{\partial-1} in §4.2.1 and ha^2 in
§5.2.3). They can appear both following and preceding the predicate (the main
clause), as in (q) and (r).
(q) ni cən me kwácnəx kwθəwλa kwsni?s ní? ə kwθé?. (CC)
                        k<sup>w</sup>ác-nax<sup>w</sup> k<sup>w</sup>θawλa k<sup>w</sup>a s-ni?-s
          cən me
                                                                             ní?
    ni
```

AUX I CERT look-TR him ART NOM-AUX-3POS be.there ? γ k^wθέ? OBL that

'I saw him there.' (lit. 'I saw him at his being there.')

(r) k^wsmis técəl k^wθe[?] məstə́vəx^w, ni [?]ə čx^w k^wə́cnəx^w. kwa s-mi-s técəl k^wθe? məstáyəx^w ni ART NOM-AUX(come)-3POS arrive.here that person AUX čxw kwec-nəxw ?a look-tr ROG vou 'When that person got here, did you see him?'

4.3.3. Clauses Nominalized with δx^{w} -

When used to form a nominalized clause, the compound prefix $\check{s}x^{w}$ - 'oblique nominalizer' (composed of *s*- 'nominalizer' and x^{w} - 'oblique relater') has the sense of 'reason why,' and it identifies the clause it nominalizes as the effect of some other event or condition.

These clauses can appear in predicates, usually following $\dot{X}\dot{a}$ 'be third person,' as in (a).

(a) Ža šx^wk^wənéts tə sq^wəmq^wəméy stəlqáyə. (JP 3)
Ža šx^w-k^wən-é-t-s tə sq^wəmq^wəméy stəlqáyə
BE3P OBLNOM-be.held-DUR-TR-3POS ART dogs wolves
'That's why he kept the dogs that were wolves.'

From its context, it is clear that (a) is a complete sentence. However, in (b) and (c), the status of the clause beginning with $\dot{X}a$ is not clear.

(b) ni? híləm Xa šx*ni?s yák**əm. (JP)
ni? híl-əm Xa šx*-ni?-s yák**-əm
AUX fall-INTR BE3P OBLNOM-AUX-3POS break-INTR
'It fell, and that's why it broke.' 'It fell, which is why it broke.'

(c) słénoj Ża šx^wżóżos tθé?. (JP) słénoj Ża šx^w-żóżo-s [?o] tθé? woman BE3P OBLNOM-be.doing-3POS OBL that 'She's a woman, and that's why she does that.' 'She's a woman, which is why she does that.'

Are the clauses beginning with $\dot{\lambda}a$ separate sentences or are they, as in the alternate English versions, relative clauses with the preceding predicates as their heads?

A clause nominalized with δx^{w} - can be introduced by the oblique particle 2∂ , as in (d).

```
(d) ni<sup>?</sup> ct wədáynəx<sup>w</sup> k<sup>w</sup>θə niəł wečmətàlx<sup>w</sup> <sup>?</sup>ə šx<sup>w?</sup>ict mi Åəwnámət.
        (JP 12)
                                                                 wéčmən-t-àĺx<sup>w</sup>
    ni?
                    wə-dáy-nəx<sup>w</sup>
                                         k™θә
                                                   ni-əł
            ct
            we
                    EST-die-TR
                                         ART
                                                   AUX-past
                                                                 watch-TR-us
    AUX
              šx<sup>w</sup>-?i-ct
                                      ?əmi
        22
                                                      Xew-namat
        OBL OBLNOM-AUX-our AUX(come) escape-self
```

'We succeeded in killing those who were guarding us, and that's how we got away.'

A clause nominalized with δx^{w} - can also appear as a subject, as in (e).

```
(e) X1? k*ən šx*?əmi. (JP 3)
X1? k*ə nə-šx*-?əmi
important ART my-OBLNOM-come
'I've come for an important reason.' (lit. 'The reason why I have come is important.')
```

4.3.4. Nominalized Narrative Sentences

Nominalized clauses with $w\partial_{2}$ 'established' appear frequently in narratives, either following $\dot{X}\dot{a}$ 'be third person' as predicate complements or, more often, simply standing alone as complete sentences. The $\dot{X}\dot{a}$ followed by the nominalized clause may be translated 'It is (was) then ...' or 'That is when ...' With no $\dot{X}\dot{a}$, the nominalized clause is generally translated with an initial 'then' or 'so then.'

In these nominalized clauses, the w_{∂_2} precedes the predicate head or, when there is one, the directional auxiliary ($2\partial m i$ 'come,' n e m 'go'), and it is preceded by the *s*- 'nominalizer. It can appear as w_{∂} - prefixed to the verb or as $-\partial w$ suffixed to the *s*-. The *s*- 'nominalizer' and w_{∂} - usually, but not always, appear as $s_{\partial w}$ (phonetically [s ∂w] ~ [siw] ~ [su] (in AG's speech usually *su*?). The verb may have an intransitive or a transitive suffix and, if the latter, an object pronoun suffix. The whole must have a possessive, either at the beginning (if first- or second-person singular) or at the end (if first- or second-person plural or third person).

If there is no locative auxiliary (^{2}i or ni^{2}), the order of elements for an intransitive or active transitive predicate is:

 $(\mathring{A}a)$ nə-, or ?əT- s- wə- VERB (-t, -nəx^w) -ct, -ələp, or -s (BE3P) 1, or 2POS NOM- EST- VERB STEM (-TR) 1PL, 2PL, or 3POS

Examples are shown in (a) to (d).

- (a) nəsəw mi cómət. (CC 5) nə-s-wə-mi cóm-ət my-NOM-EST-AUX(come) pack[carry on the back]-TR
 'Then I packed it up.'
- (b) ⁹∂θw∂łžíl∂x. (JP 22)
 ⁹∂T-s-w∂-łžíl∂x
 your-NOM-EST-stand
 'Then you stand up.'

- (c) səw néms. (CC) s-wə-ném-s NOM-EST-go-3POS 'Then he went.'
- (d) səw kwócnəxws. (CC) s-wə-kwec-nəxw-s NOM-EST-look-TR-3POS 'Then he saw it.'

For a passive predicate (with no locative auxiliary), the order is:

(Ăá)	S-	wə-	VERB	-t or -n-	-Sel-, -Sam-, etc.	-ət
(BE3P)	NOM	EST	VERB STEM	TR	PAS	SUBPAS

An example is shown in (e).

(e) səw yáθəstì·t (CC 12)
 s-wə-yáθ-əs-t-əy-ət
 NOM-EST-tell-RECIP-TR-3SUBPAS-SUBPAS
 'Then they were told.'

If there is a locative auxiliary, the order of elements for an intransitive or active transitive predicate is:

(Žá) nə-,?əT- s- ?i, ni? -ct, -ələp, -s wə- VERB (BE3P) POS NOM AUX POS EST VERB

Examples are shown in (f), (g), and (h).

- (f) nəsnəw ném ?áxqəl. (CC 12)
 nə-s-ni? wə-ném ?áxqəl
 my-NOM-AUX EST-AUX(go) exit
 'Then I went out.'
- (g) səct wəháye?. (CC 8) s-ni?-ct wə-háye?
 NOM-AUX-our EST-leave 'Then we left ...'
- (h) səsəw k^wálləxtəs. (CC 8) s-ni?-s wə-k^wálləx-t-əs NOM-AUX-3POS EST-shoot-TR-3TR 'Then he shot it.'

From forms like *nəsnəw* and from the requirements of syntax, we may infer the presence of an auxiliary (*ni*² or ²*i*) in forms like *səctəw* ~ *səct wə*- and *səsəw* ~ *səs wə*- (other examples of which will appear below). In JP's texts, the third-person forms are usually *sisəw* ~ *sis wə*-. For a passive predicate with a locative auxiliary, the nominalized auxiliary takes a third-person possessive regardless of the person of the passive person marker suffixed to the head, and the order is:

(Åá) s- ⁹i or ni⁹ -s wə- VERB STEM -t, -n- -Sel-, -Sam-, etc.
(BE3P) NOM AUX 3POS EST- VERB STEM TR PAS
-əm
INTR

Examples are shown in (i), (j), and (k).

- (i) səsəw nə?éməstéləm tə qàqəyéwtx^w. (CC 12)
 s-ni?-s wə-nə?ém-əst-él-əm [?ə] tə qàqəy-éwtx^w
 NOM-AUX-3POS EST-go-CAUS-I-INTR OBL ART sick-house
 'Then I was taken to hospital.'
- (j) səsəw céwətəm k^wsə nətén. (CC 21) s-ni?-s wə-céw-ət-əm k^wsə nə-tén NOM-AUX-3POS EST-help-TR-INTR ART(FR) my-mother 'Then my late mother was helped.'
- (k) səsəw mi Xanəm tə yöysələ qeləməy. (CC 11)
 s-?i-s wə-mi Xa-n-əm tə yöysələ
 NOM-AUX-3POS EST-AUX(come) fetch-TR-INTR ART two.persons qeləməy
 young.women
 'So the two young women were brought here as brides.'

In (k), the locative auxiliary concealed in $s \rightarrow s \rightarrow w$ must be 2i 'be here' rather than ni^2 'be there,' as in (i) and (j). This is required by the mi 'come' that follows it.

It is not at all clear what determines the choice of a form with an auxiliary over a form without one. In narratives, *səw* and *səsəw* (or *sisəw*) occur in consecutive sentences with, as yet, no discernible pattern.

There is another problem. With a third-person subject, there are inconsistencies in the verb forms that follow. In discussing $s \partial w$ and $s \partial s \partial w$, CC insisted that the forms implied by the formulas given above are "correct": $s \partial w$ should be followed by a verb with a possessive -s or a subordinate passive ending with $-\partial t$ (as in the left column below), while $s \partial s \partial w$ should be followed by the form appearing in a main clause, the intransitive with zero, the transitive active with $-\partial s$, and the passive with $-\partial m$ (as in the right column).

(l)	səw néms.	(l')	səsəw ném.	
	s-wə-ném-s		s-ni-s	wə-ném
	NOM-EST-go-3pos		NOM-AUX-3POS	EST-go
	'Then he went.'		'Then he went.'	

(m´)səsəw k ^w ácnəx ^w əs.
s-ni-s wə-k ^w ec-nəx ^w -əs
NOM-AUX-3POS EST-see-TR-3SUB
'Then he saw it.'
(n´) səsəw k ^w ánnəm.
s-ni-s wə-k™án-n-əm
NOM-AUX-3POS EST-be.taken-TR-INTR
'Then he is grabbed.'

These forms are consistent with the identification of the initial *s*- of *s*_{*aw*} and *s*_{*as*_{*aw*} as *s*- 'nominalizer' and the second -*s*- of *s*_{*as*_{*aw*} as *s*- 'third-person possessive,' and CC usually followed these rules. However, in the texts she dictated, I occasionally recorded forms like (o) and (p) with *s*_{*aw*} but no -*s* 'third-person possessive' suffixed to the verb.}}

- (o) səw q^wél tə Gabe. (CC 18) s-wə-q^wél[-s] tə Gabe NOM-EST-speak[-3POS] ART Gabe 'Then Gabe spoke.'
- (p) səw csét θə stáləss k^ws ... (CC 12)
 s-wə-cəs-ét[-s]
 θə stáləs-s
 k^w s NOM-EST-tell-TR[-3POS]
 ART spouse-3POS ART NOM 'Then he told his wife to ...'

In (o), where the verb is intransitive, it is possible that CC said sasaw and I misheard it as saw, or that she contracted sasaw to saw. In (p), however, where the verb is active transitive, had the form been sasaw, the verb should have been csétas with the -as 'third-person transitive subject.' Thus is seems more likely that she said saw and then omitted the possessive. In texts dictated by JP, sentences with saw and sisaw follow CC's rules part of the time, but fairly often saw is followed by a main-clause form of the verb. In these instances, perhaps he did contract sisaw to $saw.^4$

⁴ This inconsistency occurs in other dialects as well. In Cowichan texts that I have recorded, $s \partial w$ (su^2) is usually followed by a possessed form, but sometimes it is followed by a main-clause form (cf. Hukari 1982). In Chilliwack, main-clause forms seem to be usual. Galloway (1993, 363) identifies su (glossed 'so, then') as a conjunction, without identifying the *s*- as the nominalizer. He gives a number of sentences with third-person subjects showing su followed by main-clause verbs. When in one example (p. 367) the suffix -*s* does appear (as it should by CC's rules), he suggests that it may be an error. More likely it was the product of a speaker who treated the *s*- as nominalizer, which it clearly is with first- and second-person subjects (see Galloway's examples at the bottom of his p. 363). For most Upriver speakers and perhaps for some

Another inconsistency appears in the placement of the predicate particles. An initial $\vec{X}\vec{a}$ can be followed by one or more particles (*ce*² 'future,' $y \partial x^{w}$ 'inferential,' etc.). In fact, speakers may use $\vec{X}\vec{a}$ to introduce nominalized clauses just to have a peg to hang particles on. In a few instances, however, CC used a particle in a sentence without $\vec{X}\vec{a}$, as in (q) and (r).

- (q) səs yəx^w wəné^{...}m tə ⁹itətéwtx^w. (CC 21) s-ni⁹-s yəx^w wə-ném [⁹ə] tə ⁹itət-éwtx^w NOM-AUX-3POS INF EST-go OBL ART sleep-house 'And then he must have gone to a hotel.'
- (r) səsəw yəx" wək"ánəθəs ⁹é-İtən. (CC 12)
s-ni⁹-s wə-yəx" wə-k"án-ət-S-əs ⁹é-İtən
NOM-AUX-3POS EST-INF EST-be.taken-TR-me-3TR 3PL
'Then they must have taken me.'

In the first example, the $y \ge x^w$ is placed where one might expect it, following the auxiliary and preceding the $w \ge 2^-$, which is prefixed to the verb. In the second, however, it follows $s \ge 3^-$, which contains a $w \ge 2^-$, and is followed by another $w \ge 2^-$. This suggests again that there is a tendency to disengage the cluster of elements that includes the *s*- 'nominalizer' from the rest of the sentence.

This cluster of elements can also include the adverb $\lambda \hat{e}$ 'again, also,' as in (s) and (t).

- (s) səsវəw ²ə́k^wəstəm. (CC 14) s-ni[?]-s វ̇ e wə-²ə́k^wəs-t-əm NOM-AUX-3POS also EST-hang.up-TR-INTR 'Then they were hung up too.'
- (t) nəsikaw 'amət. (CC 21)
 nə-s-ni' ke wə-'amət
 my-NOM-AUX also EST-get.up
 'Then I got up too.'

The prefix $w\partial t$ - 'already' can also appear, either in place of $w\partial_{2}$ or perhaps incorporating it (cf. §11.1.2), as in (u) and (v).

elsewhere, however, Galloway's view may be correct: when there is a third-person subject, $s \partial w$ (or su or su?) may be simply a conjunction with no government of the following verb.

Dialect differences in this respect seem to go back to the beginning of the twentieth century. In Hill-Tout's Kwantlen texts (1903, 430-34), it appears that "ē-tlās-wä" ($^{2}oy' \dot{\lambda}a' swo$ -) is usually followed by a verb with a possessive, while in his Chehalis and Scowlits texts (1904, 336-37, 368-74) it appears that "tla-sõ" ($\dot{\lambda}a sow$) and "sõ" (sow) are usually followed by a main-clause form.

(u) səwł xqóts tə sk^wícs. səw łípəts. (CC 16)
s-[wə-?] wəł-xəq-ót-s tə s-k^wíc-s.
NOM-[EST-?] already-get.finished-TR-3POS ART NOM-butcher-3POS
s-wə-líp-ət-s
NOM-EST-cut.up-TR-3POS
'Then he finished his butchering. And then he cut it up.'

```
    (v) Ža səsŽəwł híqəθət. (CC 13)
    Ža s-ni?-s Že [wə-?]wəł-híq-əθət
    BE3P NOM-AUX-3POS again [EST-?] already-launch-self
    'Then they shoved off again.'
```

At this point, it is not clear just what $w\partial t$ -adds to these sentences. It may simply help establish a temporal relationship between the sentence in which it appears and the following sentence or sentences. A freer translation of the first example above might be 'When he had finished butchering, he cut up the meat.' But not all instances of $sw\partial t$ - lend themselves to this kind of interpretation. Besides, pairs of nominalized narrative sentences with $s\partial w$ can also be interpreted in this fashion.

Although nominalized narrative sentences usually appear in narratives after the action has started, it is also possible to begin a narrative with such a sentence. An account of how CC's parents became Shakers begins thus:

(w) səsəw x »əネəlé	él məstáyəx* k*s	sə nətén.	səsəw x ^w əsqə	olé?ł. (CC 2	1)
s-ni?-s	wə-x™ə-Ճəlél		məstáyəx ^w	k ^w sə	
NOM-AUX-3POS	EST-become-mide	dle-aged	person	ART	
nə-tén.	s-ni ⁹ -s	wə-x ^w ə-	sqəlé?ł		
my-mother	NOM-AUX-3POS	EST-beco	me-sickly		
'My late mothe	r became a middle	e-aged per	rson. She was s	sickly.'	

A more literal translation of this would be: 'It was my late mother's having become a middle-aged person. It was her having become sickly.' A freer translation might be: 'When my mother got to be middle-aged, she became sickly.' However, there is nothing that explicitly marks these two nominalizations as standing in the relationship of conditional clause and main clause.

It is not usual to begin a narrative in this fashion. This last is the only example in my collection of Musqueam texts, but there are other examples in Cowichan texts. More often, nominalized narrative sentences appear after the action has started. Those introduced by $\dot{X}\dot{a}$ seem especially to imply that some precondition has been met. A few examples follow.

In an account of a personal experience, the narrator's husband asks her to buy shotgun shells adding:

(x) Ža ce? nəsəw nem páyətəł kw má?aqw wəwéyələs. (CC 12)
Ža ce? nə-s-wə-nem páyətəł [?ə] kw
BE3P FUT my-NOM-EST-AUX(go) hunt.waterfowl [OBL] ART má?əqw wə-wéyəl-əs
larger.bird when-be.day-3SUB
'Then I'll go duck hunting tomorrow.'

Later, she is chopping wood when her child distracts her:

(y) Xa yəx^w nəsnəw sé⁹t tə k^wq^wámən. (CC 12) Xa yəx^w nə-s-ni⁹ wə-sé⁹-t tə k^wq^wámən BE3P INF my-NOM-AUX EST-rise-TR ART axe 'It must have been then I raised the axe.'

In an account of a deer drive, boys come howling like wolves and -

(z) Ža ce? səw yətátəsəts kwə smént ?é·łtən wəmi·s wəłwélatəs kwəə smáyaə. (CC 7)
Ža ce? s-wə-yə-tátəs-at-s kwə smént ?é·łtən BE3P FUT NOM-EST-along-be.getting.hit-TR-3POS ART rock 3PL wə-mi-əs wəłwéł-at-as kwəə smáyaə when-come-3SUB chase(PL)-TR-3SUB ART deer
'Then they'll be striking rocks together when they come chasing the deer.'

In a war story, a number of people are captured by enemies who came from different places in the north.

```
(aa) λa sisəw x<sup>w</sup>əsθəlθəléq təwkaləm nem x<sup>w</sup>əsk<sup>w</sup>áyəθ. (JP 11)
λa s-ni<sup>2</sup>-s wə-x<sup>w</sup>ə-sθəlθəléq təwkaləm
BE3P NOM-AUX-3POS EST-become-separated(PL.RES) those nem x<sup>w</sup>ə-sk<sup>w</sup>áyəθ
AUX(go) become-slave
'Therefore those taken as slaves became separated.'
```

However, two boys escape and, after paddling hard for a night and a day, they stop to eat.

(bb) Ža yəx^w cə sis wənóq^w təwŽáləm k^wis wəłhá·yθən. (JP 12)
Ža yəx^w cə s-ni?-s wə-nóq^w təwŽáləm k^w
BE3P INF QUOT NOM-AUX-3POS EST-fall.asleep those ART s-ni?-s wəł-háy-əyəθən NOM-AUX-3POS already-finish-mouth
'Then they must have fallen asleep when they had finished eating.'

Sentences consisting of the nominalized clause alone are more common than those with $\dot{\lambda}\dot{a}$. In some narratives, they are by far the most frequent type of sentence. For example, in an account of a hunting expedition, the whole of which

is given in §23.3, of the first twelve sentences, ten are nominalized clauses. Only the first and third (together with two quoted within the fourth) are not. Analysis of narrative structure and style may show that sentences of this type play a role not discernible at present. Such an analysis might also answer some of the other questions raised in this section.

Sentences (cc) and (dd) are examples of intransitives with one subject followed by passives with a different subject, as seen in §3.4.2 (m).

- (cc) səsəw ném tə ni ném k^w cécəw k^w écətəm tə stáləw. (CC 7) s-ni?-s wə-ném tə ni? ném [[?]ə] k^wə cécəw NOM-AUX-3POS EST-go ART AUX go OBL ART shore k[™]éc-ət-əm tə stálow look-tr-intr art river
 - 'Then someone went to the shore to look at the river.' (lit. 'Then went the one who went to the shore was looked at the river.')
- (dd) səw ⁹áməts tə čəčí⁹qən nem ánəstəm təwka x^wk^wələ. (JP 9) s-wə-⁹ámət-s tə čəčí⁹qən nem tánəs-t-əm təwka NOM-EST-sit-3POS ART mink AUX sit.beside-TR-INTR that x^wk^wələ scouring.rush
 - 'So Mink sat down next to Scouring Rush.' (lit. 'So sat Mink was sat beside that Scouring Rush.')
5 Syntax 3: Compound Sentences

A compound sentence consists of two or more predicates joined by the conjunction ${}^{2} \vartheta \dot{y} \sim i$. It is useful to distinguish two broad classes of compound sentences: coordinate and qualifying.

5.1. COORDINATE COMPOUND SENTENCES

In these, the predicates are semantically coordinate and the conjunction is translatable as 'and,' 'but,' or 'and at that time.' A coordinate compound sentence may consist of a series of verbs with the same subject, joined by conjunction, as in (a).

(a) k^wónətəs i cómətəs i tóx^wstx^wəs. (JP 7) k^wón-ət-əs ⁹əý cóm-ət-əs ⁹əý tax^w-stəx^w-əs be.taken-TR-3TR and pack-TR-3TR and move.to.shore-CAUS-3TR 'He took it and put it on his back and brought it down.'

Such a sentence may differ from a simple sentence with a verb complement (cf. $\S3.2.3$) only in the presence of the conjunction.

A coordinate compound sentence can consist of two clauses with different subjects, joined with $2\partial y$ in the sense 'and,' as in (b).

(b) stadwíqan ta smént ?ay Xaw statídw ta stálaw. (CC 7) s-tadw-íqan ta smént ?ay Xe wa-s-tatídw RES-be.cut.off-slope ART mountain and also EST-RES-be.cut.off ta stálaw ART river
'The mountain face is sheer and so is the river['s bank].'

If the two clauses are contradictory, the conjunction may be translated 'but,' as in (c) and (d).

- (c) ⁹5ỷ ti⁹ín i qál ta náča?. (CC) good this but bad ART one
 'This one is good but the other is no good.'
- (d) wənan wəłhiθ kwsəs skwekwəlt θə tintən i ni čxw nem kwənət. (CC 21)wə-nán wəł-híθ k ^w s-ni?-s sk wék walt θə already-last.long EST-verv ART NOM-AUX-3POS hidden(RES) ART tíntən ?∂ỷ ni čxw nem k^wán-ət bell but go be.taken-TR AUX you 'The bell was hidden for a long time but you went and got it.'

A simple sentence can begin with the conjunction, as in (e). What precedes this sentence seems to require interpreting it as an independent sentence.

(e) ² o ³ o

In joining two clauses, the conjunction ${}^{2}\partial y$ can also relate them in time, what is expressed in the first clause preceding what is expressed in the second. English may require a subordinating conjunction, as in (f), (g), and (h).

(f)	x ^{w9} áṁət	cən c	e ⁹ i xo	qót cə	n. (CC	2)				
	x ^w -?ám̀ə	t	cən	ce?	°∂ỷ	xəq-э́t		cən		
	LOC-be.si	tting	Ι	FUT	and	be.finis	hed-tr	Ι		
	'When I	get hoi	ne, I'll	finish	it.' (lit	. 'I will	be home	, and I v	vill fin	ish it.')
(g)	scécon n k ^w ow	i wəł mók*	k ^w átəx [*] wét. (J	™əm k P)	к™θә х	*əlməx	^w éwtx ^w	i ni k ^w á	ocnəx'	"əs
	sčéčən	ni	wəł-ł	™átəx	k™-əm	1	κ ^w θə x	wəlməx	w-éwt	X ^w
	really	AUX	alread	ly-be.t	olazing	-INTR A	art Ii	ndian-ho	use	
	?∂ỷ	ni	kwec-1	iəx ^w -a	əs k	°ə wə	-mźk'	wét		
	and	AUX	look-TI	R-3TR	AI	RT EST	-all	who		
	'The smo	okehou	se was	going	up in f	lames w	hen ever	ybody s	aw it.'	
(h)	wəẳá ?a	l stíwa	əyəłs k	^w ən š	x ^w wél	əy i nəv	w mók*	⁹ al dáy	. (CC	21)
	wə-Żá	?aĺ	s-ťíw	'əyəł-	S	Å™ə	nə-šx v	wéləy	?∍ỷ	ni
	EST-BE3P	just	NOM-	worsh	ip-3PO	S ART	my-par	ents	and	AUX

wə-mə́k^w ⁹al dáy

EST-all just die

'It was the only religion of my parents until they both died.'

5.2. QUALIFYING COMPOUND SENTENCES

These are compound sentences in which the first clause qualifies the second, standing semantically in an adverbial relationship to it. The logic of these

sentences seems to be a simple extension of that of the last three examples in 5.1. We may distinguish several types depending on how the first clause qualifies the second.

5.2.1. With Quantifying First Clauses

In these the first clause qualifies the second by indicating the amount of time or number of things required, as in (a) to (f).

- (a) ²isél swéyəl i nə tás ²ə X s²ámənə. (CC 21)
 ²isél swéyəl ²əy ni² tás ²ə X s²ámənə two days and AUX arrive.there OBL ART Duncan ⁴After two days they arrived at Duncan.²
- (b) ⁹isáləs ⁹al iqélč i ni t^θôx k^wθə nəq^wiéýxən. (JP)
 ⁹isélə-əs ⁹al iqélč ⁹əý ni t^θôx k^wθə nə-q^wiéý-xən two-face just moon and AUX wear.out ART my-log-foot
 ⁶My shoes wore out in just two months.⁷
- (c) xăa'áθən lépàt i lść tθe? šx «qwólstəň. (JP)
 xə'áθən lépàt 'əỷ lść tθe? šx «-qwóls-təň four cup and fill that OBLNOM-boil-INSTR 'It takes four cups to fill that boiling-pot.'
- (d) tx^wk^wín má²aq^w i ²ówk^w (DK) tx^w-k^wín má²əq^w ²əỷ ²ówk^w remaining-how.many larger.bird and run.out 'How many more ducks [are there to cook] before they're gone?'
- (e) cəx "lé? i dáynəx "əs tə smáyə0. (JP 7) cəx "lé? ?əỷ dáy-nəx "-əs tə smáyə0 once.in.a.while and die-TR-3TR ART deer 'Once in a while he killed a deer.'
- (f) ?ówə scék wələs i ke cən wəqwəlstámə. (CC)
 ?ówə scék wəl-əs ?əỷ ke cən wə-qwel-st-ámə not how-3SUB and again I EST-speak-COM-you 'I'll never speak to you again.'

For other examples of this construction, see §18. For the negative construction, see §6.

5.2.2. With Interrogative First Clauses

Two interrogative words (see \$17) can stand as first-clause predicates in sentences of this type. These are *tomtém* 'when' and *nocím* 'why,' as in (a) and (b).

- (a) təmtém ce? i ném čx^w. (DK) when FUT and go you
 'When are you going?'
- (b) nəcím k^we i ném cən. (CC) why then and go I 'Why should I go?'

5.2.3. With Conditional First Clauses

In these, the first clause begins with ha^{2} 'if, when' or $m\delta \vec{k}^{w}$ 'whenever' (lit. 'all'), two words that produce what are in effect conditional clauses. As noted in §4.3.1, $m\delta \vec{k}^{w}$ is followed by a nominalized clause. See also §18.4.18 and 18.4.40.

- qəlét (a) ha? čx^w ce? **x**té i χźł čx w ce?. (CC) if again do and vou FUT hurt you FUT 'If you do it again, you'll get hurt.' [a threat]
- ?aĺ (b) há? čxw ṁе wə-?í i háv cən ce? if vou CERT EST-be.here iust and specifically I FUT wə-ném. (CC) EST-go 'Even if you stay here, I will go.'
- (c) môk^w nôsném ?ímôx ?i wôk^wôcnôx^w côn. (JP) môk^w nô-s-ném ?ímôx ?ôỷ wô-k^wec-nôx^w côn whenever my-NOM-AUX(go) walk and EST-look-TR I 'Whenever I go about, I see him.'
- (d) mák vał nasném x cál θat 'iw k vácnax can. (JP) mák vał na-s-ném x cál-θat 'ay wa-k can can whenever-past my-NOM-AUX(go) go.where-self and EST-look-TR I 'Whenever I went somewhere, I saw him.'

5.2.4. With Modal First Clauses

There are at least three words with a modal sense (see §18.1) that can serve as a predicate head in the first clause. These are $\check{x}^w \delta m$ 'can' (also 'fast'), $c \partial l \ell l$ 'nearly,' and $x^w \ell l \partial q$ 'nearly.' In sentences of this type, the subject of both predicates is the same. A particle marking a first- or second-person subject appears in the first clause and is not repeated in the second.

(a) X^wóm cən i X ow ném. (CC)
 X^wóm cən [?]oğ Xe wo-ném
 can I and also EST-go
 'I can go too.'

(b)	x̃™э́m can[quic 'You're	ckly] liable	čx ^w you to get	i and hurt.'	mé ⁹ k ^w ł. (CC) get.injured	
(c)	x̃™óm can 'Can yo	?ə ROG u make	čx ^w you e string	^າ ວỷ and g figure	x ^{w?} əwélcəs. (J make.string.fig es?'	P) ures
(d)	x̃™́э́m can[quic 'He gets	ckly] s sore [i and takes o	wéť ⁰ . take.c offence	(JP) offence] easily.'	
(e)	cəlél nearly 'I almos	cən I st fell a	i and sleep.'	ni AUX	n áq ^w . (JP) fall.asleep	
 (f) えっwł colél i mi pohéls. (JP) えを woł-colél ?oỷ mi pah-éls again already-nearly and AUX(come) be.blown-act 'The wind will soon be blowing again.' 						
A noun subject, on the other hand, follows the second predicate.						

(g) x *éləq cə i Xx** təm tə xéls ni ?ə tə ... (JP 10)
x *éləq cə ?əỷ Xəx*- təm tə xéls ni
nearly QUOT and defeat-tr-intr ART Transformer be there
?ə tə
OBL ART
'The Transformer was almost beaten at ...'

In the placement of the subject particles and the nominal adjunct, these sentences resemble simple sentences with initial auxiliaries or adverbs. However, the conjunction suggests that they are a special type of compound sentence.

6 Syntax 4: Negation

The basic negative word is $2\delta w \delta$ 'not' (as the most common negator of a predicate), 'no' (as an interjection or answer), 'refuse' (see §6.1.6). Two other negative words, probably derivatives of $2\delta w \delta$, are also used to negate predicates. These are $x^w \delta w \delta'$ 'not yet' and $2\delta w \delta t e^2$ 'none, absent.' (Other words with negative meanings include $sk^w \delta y'$ 'impossible' (§18.4.43) and $sqi2q\delta l'$ 'unable' (§18.4.45).

6.1. NEGATION WITH ?ówə

An affirmative predicate of most types is most commonly negated by embedding it in a matrix clause of which the head is $2\delta w \partial$ 'not.' The embedded clause may be a subordinate clause or a nominalized clause.

6.1.1. Subordinate Clauses with Intransitive and Active Transitive Predicates

The negation with a subordinate clause seems to be the basic (unmarked) form, except that there is one hole in the paradigm that is filled by a negation with a nominalized clause.

If the embedded clause being negated is an intransitive or active transitive subordinate clause, the negative word is followed by a main-clause subject marker (one of the particles $c \partial n$ 'I,' $\check{c} x^w$ 'you,' ct 'we,' $ce \cdot p$ 'you (pl.),' or zero for a third-person subject), while the embedded affirmative, unless it is passive, appears with the corresponding subordinate subject suffix (- $e \cdot n \sim -\partial n$ 'I,' $-\partial x^w$ 'you,' $-\partial t$ 'we,' $-e \cdot p \sim -\partial p$ 'you (pl.),' or $-\partial s$ 'third person').

The negative paradigm for an intransitive predicate without an auxiliary is as shown in (a) to (f).

(a) ⁹ówə cən ném-èn.
 not I go-I
 'I do/will not go.'

- (b) ⁹ Św čx^w ném-əx^w not you go-you 'You do/will not go.'
- (c) ⁹ówə ct ném-ət. not we go-we 'We do/will not go.'
- (d) ⁹ Śwə ce · p ném-è·p.
 not you(PL) go-you(PL)
 'You folks do/will not go.'
- (e) ⁹ówə ném-əs.
 not go-3SUB
 'He/she/it does/will not go.'
- (f) ⁹ówə ném-əs (⁹é-łtən) not go-3SUB 3PL 'They do not go.'

(I am identifying the *némen*, *némax*^w, and so on as subordinate clauses, even though there is no subordinating prefix $w\partial_{-1}$ or $2\partial_{-1}$. Alternatively, one might identify $2\partial_{-1}w\partial_{-1}a$ as an adverb of a unique type that requires a subordinate person marker suffixed to the predicate head. I think, however, that the interpretation presented here will turn out to be the most parsimonious.)

If there is an auxiliary in the embedded affirmative, it will take the subordinate subject suffix, as in (g) and (h).

- (g) ⁹ówə cən ni·n ném.
 ⁹ówə cən ni⁹-ən ném not I AUX-I go
 ⁶I did not go.⁷
- (h) ?ówo con ?i·n yá·ýos.
 ?ówo con ?i·on yá·ýos
 not I AUX-I be.working
 'I am not working.'

If the embedded affirmative is an active transitive predicate, the negation is formed as above for first- and second-person subjects, as in (i) to (m).

 (i) ?ôwô čx^w k^we mélqtôx^w.
 ?ôwô čx^w k^we mélq-t-ôx^w not you then forget-TR-you 'Don't forget it.'

- (j) ?ówô čx^w ží?že?ax^w. (JP)
 ?ówô čx^w ží?že?-ax^w
 not you ashamed-you
 'Don't be shy.'
- (k) ?əwé· cən xté?ełcθame·n kw táxwac. (CC)
 ?əwə ?ə cən xté?-əłc-t-Samə-ən kw táxwac
 not ROG I make-BEN-TR-you-I ART bow
 'Shall I make you a bow?' (lit. 'Shall I not make you a bow?')
- (l) ?ówo čx^w ni?ox^w k^wócnàmx.
 ?ówo čx^w ni?ox^w k^wec-n-àmx not you AUX-you look-tr-me 'You didn't see me.'
- (m)?'> w> c>n ni·n c'éwət.
 ?> w> c>n ni?->n c'éw->t
 not I AUX-I help-TR
 'I didn't help him.'

However, if the embedded affirmative is an active transitive predicate with a third-person subject and no auxiliary, it must be negated as a nominalized clause, as in (n) and (o).

- (n) ?ówə k^ws k^wócnəx^ws.
 ?ówə k^w s-k^wec-nəx^w-s
 not ART NOM-look-TR-3POS
 'He doesn't see him.'
- (o) ?\u00f3w\u00f3 m\u00f3 k\u00eds s k\u00eds s k\u00eds chan xs ?\u00ed-t\u00ed-t\u00ed t\u00ed n
 ?\u00e3w\u00e3 m\u00e3 k\u00ed s k\u00ed wec-n-\u00e3mx-s ?\u00ed-t\u00ed-t\u00e3n
 not CERT ART NOM-look-TR-me-3POS 3PL
 'They do not see me.'

A form like $**? \delta w \partial k \delta c n \partial m x \partial s$ is not possible (CC). On the other hand, if there is an auxiliary, an active transitive predicate with a third-person subject can appear as a subordinate clause, as in (p).

(p) ?ówə ni?əs kwócnàmxəs. (CC)
 ?ówə ni?-əs kwec-n-àmx-əs
 not AUX-3SUB look-TR-me-3TR
 'He didn't see me.'

Here, the first -as is the third-person subordinate subject marker while the second -as is the third-person transitive subject marker. The auxiliary provides a base for the first, and the head provides a base for the second. Without the auxiliary, the head would have to have either two suffixes with the form -as or else

one *-os* serving both functions. Switching to a nominalization in such instances avoids this dilemma.

6.1.2. Subordinate Clauses with Passive Predicates

When the embedded predicate is passive, the negative (with the exceptions noted below) has a third-person subject. This holds regardless of the subject of the embedded passive. The $2 \delta w \partial$ takes no person marker, and if there is no auxiliary, the predicate head takes one of the passive person markers *-el-* 'I,' *-am-*'you,' *-al-* 'we, you (pl.),' or *-\partialy-* 'third-person subordinate passive' (see §14.2.6), and the subordinate passive suffix *-at*, as in (a) and (b).

- (a) ?'awa ce? me k^wacné·lt. (CC)
 ?'awa ce? ma k^wec-n-él-at
 not FUT CERT look-TR-I-SUBPAS
 'I will not be seen.'
- (b) ?ówə ce? me k^wócnì·t. (CC)
 ?ówə ce? me k^wec-n-əy-ət
 not FUT CERT look-TR-3SUBPAS-SUBPAS
 'He will not be seen.'

If there is an auxiliary, it takes the third-person subordinate subject marker $-\partial s$, while the predicate head has a coordinate passive form, that is, the stem takes the passive suffixes for first or second person or zero for third person and the suffix $-\partial m$, as in (c) to (f).

- (c) ?ówə ni⋅s kwácnèləm. (CC, JP)
 ?ówə ni-əs kwec-n-èl-əm
 not AUX-3SUB look-TR-I-INTR
 'I am not seen.'
- (d) ?\u00f3we-l ni?\u00f3s kw\u00e3cn\u00e2lam. (JP)
 ?\u00f3we-\u00e3l ni?\u00e3s kw\u00e3c-n\u00e2l\u00e3m
 ?\u00e3we-\u00e3l ni?\u00e3s kw\u00e3c-n\u00e3l\u00e3m
 not-past AUX-3SUB look-TR-I-INTR
 `I was not seen.'
- (e) ?ówə ni·s yək wócnèləm. (JP)
 ?ówə ni-əs yə-k wec-n-èl-əm
 not AUX-3SUB along-look-TR-I-INTR
 'I wasn't seen by anyone.' (lit. 'I was not seen all along.')
- (f) ?ówə ni·s k̃wə́cnà·m. (JP) ?ówə ni-əs k̃wec-n-am-əm not AUX-3SUB look-TR-you-INTR 'You are not seen.'

Imperative forms are exceptional in that second-person subject markers may appear, as in (g) and (h).

- (g) ⁹ówə ce·p yək^wócnà·lt. (CC)
 - ?ówə ce·p yə-k^wóc-n-àl-ət
 not you(PL) along-look-TR-you(PL)-SUBPAS
 'Don't let yourselves be seen.'
- (h) ?ówo čx^w yok^wócnàmot. (JP)
 ?ówo čx^w yo-k^wec-n-àm-ot
 not you along-look-tr-you-subpas
 'Avoid being seen.'

(After giving [h], JP denied the possibility of $**^2 \delta w \partial c \partial n \ k^w \delta c n \dot{c} lt$ for 'I am not seen.')

Such imperative forms may be influenced by or be simply errors for forms with -s 'permissive' (§10.7), like (i) and (j).

- (i) ?ówəs čx^w də k^wócnàmət. (CC)
 ?ówə-s čx^w də k^wec-n-àm-ət
 not-PERM you EMPH look-TR-you-SUBPAS
 'Don't let yourself be seen.'
- (j) ?ówəs čx^w k^wócnàmət. (AG)
 ?ówə-s čx^w k^wec-n-àm-ət
 not-PERM you look-TR-you-SUBPAS
 'Don't be seen.'

In (i) and (j), the suffix -s 'permissive' has produced a new verb, ²*ówəs* 'let it not be,' which presumably requires 'you' as a subject.

6.1.3. Subordinate Clauses with Non-Verbal Predicates

Adjectival, nominal, and other non-verbal predicates are negated in the manner of intransitive (and most transitive) verbal predicates, a coordinate subject person marker following the negative word and the corresponding subordinate subject person marker suffixed to the first word of the embedded affirmative.

Examples of negated adjectival predicates are shown in (a) and (b).

(b) ⁹ əwéł θí⁹ əs. (AG)
 ⁹ śwə-əł θí-⁹ əs
 not-past big-3SUB
 ⁶ It was not big.²

(See §14.2.3 for AG's treatment of the subordinate subject marker.) Examples of negated nominal predicates are shown in (c) to (g).

- (c) ⁹ówə cən sắíẳqəłèn. (JP)
 ⁹ówə cən sẳíẳqəł-èn not I child-I
 ⁶ I am not a child.'
- (d) ?ówe-ł cən sẳíẳqəłè-n. (JP)
 ?ówe-əł cən sẳíẳqəł-è-n
 not-past I child-I
 'I was not a child.'
- (e) ⁹ śwə sẳíẳ qəłəs θe?. (JP)
 ⁹ śwə sẳíẳ qəł-əs θe?
 not child-3SUB that(FP)
 'She is not a child.'
- (f) ?ówo ct sté?ex vołot. (JP)
 ?ówo ct sté?ex voł-ot
 not we children-we
 'We are not children.'
- (g) ni čx^w wəłćísəm. wəł²ówə čx^w ni²əx^w sẳíẳqəł. (JP)
 ni čx^w wəł-ćísəm wəł-²ówə čx^w ni²-əx^w sẳíẳqəł
 AUX you already-grow already-not you AUX-you child
 'You have grown up. You are not a child any more.'

In (h), the negated predicate consists of an adjective modifying a noun.

(h) ⁹ówə pépłətəs smént i wəłłéłqət θəł. (JP 23)
⁹ówə pépłətəs smént ⁹əỷ wəł-łéłqət θəł not thick(PL)-3SUB rock but already-wide(PL) ADV
⁶ 'They were not thick rocks but they were wide.' (freely, 'They were rocks that were wide but not thick.')

In (i) and (j), the negated predicates each consist of a noun modified by an adjective modified by an intensifier.

- (i) ⁹ όwə nán s θí snáx wəł. (JP)
 ⁹ ówə nán s θí snáx wəł not very-3SUB big canoe
 ⁶ It's not a very big canoe.⁷
- (j) ?ówə cow nános woqóx xwolmoxw. (JP 21)
 ?ówo cow wo-nános wo-qóx xwolmoxw
 not QUOT EST-very-3SUB EST-many people
 'They were not very many people, they say.'

In (k) and (l), the negated predicates each consist of a noun modified by a possessive word (see §14.2.8).

- (k) ?ówo noswé?os nosnóx woł. (CC)
 ?ówo no-s-wé?-os no-snóx woł
 not my-NOM-own-3SUB my-canoe
 'It's not my canoe.'
- (l) ?ówə s?á·łəs snóx wəł. (CC)
 ?ówə s?á·ł-əs snóx wəł
 not our-3SUB canoe
 'It's not our canoe.'

In (m), a possessive word and a personal word (§14.2.7) are negated predicates.

(m)?ówo me swé?sos k*xémon. ?ówo me Xá·s toli? k*0o cwé? k*0o k^wxémən. (CC) ?áwa me s-wé?-s-əs k^wix-émən ?áwa ṁе not CERT NOM-OWN-3POS-3SUB name-residue not CERT Xá-28 tali? k»θa c-wé? [?ə] k «θa kwix-émaň name-residue BE3P-3SUB from ART do-own OBL ART 'That name doesn't belong to him. He doesn't come from the ones who own the right to use that name.' (more lit. 'He doesn't have the right to use the name. He is not one who is from those who own the right to use the name.')

6.1.4. Nominalized Clauses

Negations with $2\delta w \partial$ and nominalized clauses (see §4.3) have been recorded less commonly than those with subordinate clauses, except for the third-person forms required by the subordinate-clause paradigm, as explained above. These forms may be more emphatic than those with subordinate clauses; they can be, but are not always, translated with 'never.' Examples are shown in (a) to (d).

- (a) ⁹ówə kwən skwəcnəxw. (JP)
 - ?ówə k̃™ə nə-s-k̃™ec-nəx™

```
not art my-nom-look-tr
```

'I never see him.' (lit. 'Not my [hypothetical] seeing him.')

(b) ⁹ówə c'ə k^ws⁹ónəx^wθəts tə sxté téqtəqmən. (JP 24)

 9 ówə cə kwə s- 9 ónəxw- θ ət-s tə s- x té téqtəqmən not QUOT ART NOM-stop-self-3POS ART RES-said Constant-Farter 'The one called Constant-Farter never kept still, they say.'

- (c) ?ówə k^wsəw ?óỷs ?al ?əłni⋅s dədá?təl. (JP 22)
 ?ówə k^wə s-wə-?óỷ-s ?al ?əł-ni⋅-əs
 not ART NOM-EST-good-3POS just whenever-AUX-3SUB dədá?-təl
 be.meeting-each.other
 'It never went well when they met.'
- (d) háye θəwẳa łéw. ⁹əwk^ws x^wcák^ws i ẳəwł px^wéls. (JP 22) háye θəwẳa łéw. ⁹əwə k^wə s-x^w-cák^w-s ⁹əỷ ẳe leave that flee not ART NOM-toward-far-3POS and again wəł-pax^w-éls already-blow-ACT
 - 'Away went that thing that was running away. It didn't get far when it blew again.'

There is considerable variation in how $2\delta w \partial is$ realized phonetically in such constructions; $2\delta w \partial \vec{k}^w \dots$ can be $[2\delta w \partial \vec{k}^w] \sim [2\delta \cdot \vec{k}^w]$ and $2\delta w \partial \vec{k}^w \dots$ can be $[2\delta w \partial \vec{k}^w] \sim [2\delta \cdot \vec{k}^w] \sim [2\delta \cdot \vec{k}^w]$ or lose its stress and be $[2\delta w \partial \vec{k}^w] \sim [2\delta \cdot \vec{k}^w]$. I have indicated above only the loss of the second vowel with loss of stress.

6.1.5. With Relative Clauses

A few examples have been recorded of ${}^{2}\delta w \partial$ followed by a relative clause introduced by *l* or $k^{w}l$, which are probably forms of the articles $l\partial$ and $k^{w}l\partial$ (§15.1) and may be translated, if at all, as 'anyone who, anything that.' This construction is used for negative statements of general validity, often translated with 'never.'

- (a) [?]ówə čx^w ł ni dáy tə nówə. (JP 9) ?áwa ni? čx ^w ł dáy tə ກລ໌ພລ not vou ART AUX die ART you 'You yourself don't die [so why should I worry].' (lit. 'You are not anyone who dies, you yourself.' Cf. ⁹5wə čx^w niəx^w dáy 'You do not die.')
- (b) ... ?əỷ ?ówə ce·p k^wł k^wócnəx^w. (JP 26)
 ?əỷ ?ówə ce·p k^wł k^wec-nəx^w
 and not you ART look-TR
 '... but you never saw it.' (lit. 'You are not anyone who saw it.')

AG rephrased (b), shifting the second-person plural subject to the relative clause in (c).

(c) ?ówə k^wł ni·ləp k^wécnəx^w. (AG)
?ówə k^wł ni-ələp k^wec-nəx^w
not ART AUX-you(PL) look-TR
'You never saw it.' (lit. 'It is not anything that you saw.')

In a speech given at a winter dance upriver in a big house that had not been used for the purpose for a long time, JP said:

(d) təna léləm šx^w²íct wəlíθ k^ws²ówəs ł ni yóg^w. (JP 5) k ^w wəł-híθ təna léləm šx^w-⁹í-ct this house OBLNOM-be.here-our already-last.long ART s-?áwa-s ł tágw ni NOM-not-3POS ART AUX burn

'This house where we are, it was a long time without fire.' (perhaps lit. 'It is this house where we are that has long been not anything that has burned.')

And, describing the Coast Mountains, he used (e).

(e) tə '> wə k^wł t⁰ w méqe' ni tə smént. (JP)
tə '> wə k^wł t⁰ w méqe' ni ['>] tə smént
ART not ART disappear snow be there OBL ART mountain
'the place where snow never melts on the mountains' (lit. 'the snow on the mountain that is not anything that melts')

6.1.6. Ellipsis with the Negative

The negative $2\delta w a$ can appear without a negated predicate, as in (a), (b), and (c).

- (a) ?əỷ ?ówə tə nəclíle?em. (CC 21)
 ?əỷ ?ówə tə nə-clíle?em
 but not ART my-stepparent
 'But my stepfather refused.'
- (b) ⁹ówə cən. (JP, AG) not I 'I won't.'

Although in some contexts $\delta w \partial$ may be translated 'refuse,' it does not function as a verb. For example, one cannot say ***ni* con $\delta w \partial$ for 'I refused' (AG), so these are best seen as cases of ellipsis.

6.2. NEGATION WITH x^w∂ŵé

The negative word $x^{w}\partial w \dot{e}$ 'not yet' is probably derived from $2\partial w \partial$ 'not,' but its formation is not transparent. Its use with subordinate clauses is parallel to that of $2\partial w \partial$.

- (a) x^wowć con ni n k^wócnox^w. (JP) x^wowć con ni-on k^wec-nox^w not.yet I AUX-I look-TR 'I haven't seen him yet.'
- (b) x^wəŵé ni·s k^wúk^wtəm. (JP) x^wəŵé ni-əs k^wúk^w-t-əm not.yet AUX-3SUB cook-TR-inTR 'It's not yet cooked.'
- (c) x^wəwé [?]e [?]i·s ġdámqən k^wθe [?]esqéwθ. (CC) x^wəwé [?]ə [?]i-əs ġdámqən k^wθə [?]ə-sqéwθ not.yet ROG AUX-3SUB bloom ART your-potato 'Have your potatoes flowered yet?'
- (d) x^wəŵé ⁹i·s jół, wəx^wənšx^wá·x^wt⁰. (CC)
 x^wəŵé ⁹i-əs jół wə-x^wən-šx^wá·x^wt⁰
 not.yet AUX-3SUB come.to EST-still-crazy/'wild''
 'He's not yet grown up [lit. 'come to his senses']: he's still foolish.'

In (e), the phrase $x^w \partial w \hat{e} c \partial^2 a \hat{l}$ is used adverbially as the first of two coordinate clauses.

(e) x ** owé co ?al i x ** oswóżqe? con. (JP)
 x ** owé ce? ?al ?o y x ** o-swóżqe? con
 not.yet FUT just and become-man I
 'Soon I will be a man.'

6.3. NEGATION WITH ?ówəte?

The negative $2\delta w \delta t e^2$ 'absent, non-existent, none' is probably identifiable as a compound word or phrase composed of $2\delta w \delta$ 'not' and te^2 perhaps 'present, existent' (compare the article $t\partial$). The position of the predicate particles with $2\delta w \delta t e^2$ suggests an uncertain status as a word. The interrogative 2δ follows the $2\delta w \delta$, giving $2\delta w e^2 \delta t e^2 \sim 2\delta w e^2 t e^2$ 'are there none?' while the past particle $-\delta t$ can follow either the $2\delta w \delta$, the te^2 , or both, giving $2\delta w e^2 t t e^2$, $2\delta w \delta t e^2 t$, or $2\delta w e^2 t t e^2 t$ 'there were none.'

Followed by a nominal adjunct, $2\delta w \partial t e^2$ has the sense 'be absent,' as in (a) and (b).

(a) ?ówəte? łə nəwáč, ni yəx^w k^wónətəm. (CC)
?ówəte? łə nə-wáč ni yəx^w k^wón-ət-əm
absent ART my-watch AUX INF be.taken-TR-INTR
'My watch is gone. It must have been taken.'

 (b) ?ówəte? k "θə nəsq"əméy. (AG)
 ?ówšəte? k "θə nə-sq"əméy absent ART my-dog
 'My dog is not around.'

Compare (a) and (b) with (c) and (d), in which the negative is $2 \delta w \partial$.

- (c) ?όw ?i·s ?í k^wθ a nasq^waméý. (AG)
 ?ów ?i-as ?í k^wθ a na-sq^waméý
 not AUX-3SUB be.here ART my-dog
 'My dog is not here.'
- (d) ?ówo co ni·s ní? k "θo nosq "oméy. (AG)
 ?ówo co ni·os ní? k "θo no-sq "oméy" not QUOT AUX-3SUB be.there ART my-dog
 'I'm told my dog is not there.'

If it is a person who is absent, the oblique case is used, as in (e) and (f).

- (e) ?ówəte? ċə X q^wəlítəq. (JP 7)
 ?ówəte? ċə [?ə] X q^wəlítəq
 absent QUOT OBL ART seagull
 'Seagull, it is said, was not there.'
- (f) ?śwəte? čθəł ?ə X Arnold. (AG)
 ?śwəte? čə θəł ?ə X A
 absent QUOT ADV OBL ART A
 'But they say Arnold was not present.'

The oblique case here seems to imply that the subject of such sentences is an implicit "the place" and that the predicate $2\delta wate^2$ is more literally glossed 'be without, be lacking, be devoid of,' the first sentence being more literally 'It [the house] was, it is said, devoid of Seagull.'

Followed by a noun standing as a predicate complement (with no article or demonstrative), $2\delta w \partial t e^2$ has the sense 'be none, be non-existent,' providing the negative counterparts of existential nominal predicates (see §3.7.3 and §4.1.5.1.1) and predicates with *c*- 'have' (see §12.2.1), as in (g) to (k).

- (g) ?śwəte? strí·tkà: (CC 12) none streetcar 'There was no streetcar.'
- (h) ?śwè·te? síl, ?íməθ. (CC 12)
 ?śwə ?e te? síl, ?íməθ
 not ROG existent cloth grandchild
 'Do you have any cloth, grandchild?' (lit. 'Is there no cloth, grandchild?')

(i) ?ówəte? nəsqwəméy. (CC, AG)
 ?ówəte? nə-sqwəméy
 none my-dog
 'I have no dog.'

(Compare [i] with the absent-dog sentences [c] and [d] above.)

- (j) ?əwé·te? ?ənkəpú. (AG)
 ?əwə ?ə te? ?ən-kəpú
 not ROG existent your-coat
 'Don't you have a coat?'
- (k) ?ówəte? nə-kəpú. (AG) absent my-coat 'I have no coat.'

Sentence (k) does not preclude my having a coat at home. I'm simply now coatless and outside in the cold. But compare (l):

- (l) ?'swate? k w@a na-kapú. (AG) absent ART my-coat
 'My coat is not around.'
- This implies that I definitely own a coat but can't lay my hands on it. In (m), the $2\delta w \partial t e^2$ has two 'past' particles.
- (m)?əwé·łtè·ł x wənítəm ?ə k wəha wəłhíθəł. (AG)
 ?əwə-əł te?-əł x wənítəm ?ə k wəha wəł-híθ-əł
 not-past existent-past White.person OBL that already-last.long-past
 'There were no White people long ago.'

The predicate complement may be a noun derived from a verbal root or a nominalization.

(n) ²ówəte² sníws. (JP)
²ówəte² s-níw-s
none NOM-advise-3POS
⁶ They had no advice.' (i.e., 'They are without proper upbringing.')

```
(o) ?\u00f3w\u00e3te? s\u00ex\u00e1?\u00e3 s^\u00ex\u00e3 s^\u00ex\u00e3 s^\u00ex\u00e3 s^\u00e3 

(p) ?ówəte? nəsłódallax<sup>w</sup>. (CC 12)
 ?ówəte? na-s-łódal-lax<sup>w</sup>
 none my-NOM-know-TR
 'I didn't know it.'

- (q) ?ówəte? eθck\*óyxdət. (JP)
   ?ówəte? ?oT-s-c-k\*óyx-dət
   none your-NOM-do-move-self
   'There's nothing you can do about it.'
- (r) x "cél čx". ?ówəte? nəšx "ném. (AG) x "cél čx". ?ówəte? nə-šx"-ném go.where you none my-OBLNOM-go 'Where are you going? Nowhere.' (lit. 'I have no destination.')

The sequence  $2\delta w \partial t e^2 k^w l$  (~  $\partial l$ ) followed by a relative clause has the sense 'There is not anyone who ..., Nobody ..., Nothing ...,' as in (s) to (w).

- (s) ?ówəte? k<sup>w</sup>ł ?ítət. (JP 25) absent ART sleep
  'Nobody slept.' (lit. 'Absent was one who slept.')
- (t) ?ówəte? k<sup>w</sup>ł k<sup>w</sup>óc-nəx<sup>w</sup>. (JP 28) absent ART see-TR 'Nobody sees them.'
- (u) ?'swate? k<sup>w</sup>ł mi wíl. (JP) absent ART AUX(come) appear 'Nothing showed up.'
- (v) ?'swate? ał (~ k<sup>w</sup>ł) ném. (CC) absent ART go
   'Nobody ever goes.'
- (w) '\u00f3w\u00edte e' k w\u00ed k w\u00edcn ex w\u00edcn e v\u00edcn v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn e v\u00edcn

AG also used the interrogative form of this construction to ask a question, as in (x).

(x) ?əwé·te? k<sup>w</sup>ł x<sup>w</sup>ə?í k<sup>w</sup>θə nəmén. (AG)
?əwə ?ə te? k<sup>w</sup>ł x<sup>w</sup>ə?í k<sup>w</sup>θə nə-mén not ROG existent ART become-be.here ART my-father
'Did my father get here?'

See also the interrogative words wét 'who' (\$17.1) and stém 'what' (\$17.3) for other uses of  $2\delta wate^2$ .

# 6.4. NEGATED PREDICATES IN COMPLEX AND COMPOUND SENTENCES

Predicates negated with  $2\delta w\partial$  can appear in relative clauses, subordinate clauses, and nominalized clauses.

#### 6.4.1. In Relative Clauses

Negated predicates can appear in relative clauses, as in (a) to (d).

- (a) k<sup>w</sup>θ <sup>2</sup> <sup>3</sup>w <sup>3</sup> ni <sup>5</sup> s ném (JP) k<sup>w</sup>θ <sup>2</sup> <sup>3</sup>w <sup>3</sup> ni <sup>-</sup> <sup>3</sup>s ném ART not AUX-3SUB go <sup>6</sup> the one who didn't go<sup>2</sup>
- (b) k<sup>w</sup>θ ná n cả? m stáy s x<sup>w</sup> ? śwa ni · s céwałtan (JP) k<sup>w</sup>θ a ná · n cả? m stáy s x<sup>w</sup> ? śwa ni - s céw - ałtan ART one.person person not AUX-3SUB help-people 'the one person who didn't help'
- (c) k<sup>w</sup>θə ?ówə ce? céwətì·t (JP) k<sup>w</sup>θə ?ówə ce? céw-ət-əy-ət ART not FUT help-TR-3SUBPAS-SUBPAS 'the one who won't get helped'
- (d) k<sup>w</sup>θ ?<sup>3</sup>w ni·s d<sup>3</sup>yn m (JP) k<sup>w</sup>θ ?<sup>3</sup>w ni-s d<sup>4</sup>y-n-m ART not AUX-3SUB die-TR-INTR 'the one who was not killed'

#### 6.4.2. In Subordinate Clauses

The  $2\delta w\partial$  can take  $w\partial_{2}$  'if, then,' which requires that it as well as the negated verb have a suffixed subordinate subject person marker, as in (a) and (b).

- (a) wə?əwén némèn ?əỷ sk<sup>w</sup>éy k<sup>w</sup>ə nəsk<sup>w</sup>ácnəx<sup>w</sup>. (JP)
  wə-?áwə-en ném-en ?əỷ sk<sup>w</sup>éy k<sup>w</sup>ə nə-s-k<sup>w</sup>ec-nəx<sup>w</sup>
  if-not-I go-I and impossible ART my-NOM-see-TR
  'If I don't go, I won't see him.'
- (b) wəwés mis x və?ámət k ve ?eeiyéyə i yeł ... (JP 22)
  wə-?ówə-əs ?əmi-əs x və-?ámət k ve if-not-3SUB AUX(come)-3SUB become-be.sitting(at.home) ART ?eT-səyéyə ?əy yeł your-friends and thereafter 'If your friends do not come home, then ...'

#### 6.4.2.1. As Conditional Clauses

In (a) and (b) above, the presence of the conjunction  $2\partial y$  and shows these to be a kind of compound sentence, even though the first clause has the form of a subordinate clause.

What are no doubt compound sentences can also be constructed with  $ha^2$  'if, when' introducing the first of a pair of coordinate clauses, as in (c), (d), and (e).

- (c) ha? cən ?ówə némè:n ?əỷ sk<sup>w</sup>éy k<sup>w</sup>ən sk<sup>w</sup>ócnəx<sup>w</sup>. (JP) ha? cən ?ówə ném-è:n ?əỷ sk<sup>w</sup>éy k<sup>w</sup>ə nə-s-k<sup>w</sup>ec-nəx<sup>w</sup> if I not go-I and impossible ART my-NOM-look-TR 'If I don't go, I won't see him.'
- (d) hát cən '> wə ni n (~ ni n) ném' '> y' '> wét cən ni n (~ ni n) k wə cnəx w. (JP) hát -> to cən '> wə ni '> -> ném' '> y' '> wə -> to cən ni '> -> n k we c-nəx wif-past I not AUX-I go and not-past I AUX-I look-TR 'If I hadn't gone, I wouldn't have seen him.'
- (e) ha? ce·p ce? ?>> háye·p i ném cən qíq jatàla. (CC 21) ha? ce·p ce? ?>> háy-e·p ?>> y ném cən if you(PL) FUT not stop-you(PL) and AUX(go) I qíq jat-àla get.bound-TR-you(PL)
  'If you don't stop, I'll arrest you.'

See §5.2.3 for conditional first clauses like (c) to (e) in the affirmative.

# 6.4.2.2. Expressing Necessity

The word  $sk^w ey$  'impossible' as a main clause predicate followed by an 'if not' clause expresses necessity, as in (a) and (b).

- (a) sk<sup>w</sup>éy m³ w³?³wé·n ném³e·n. (CC) sk<sup>w</sup>éy m³ w³-?³w³-e·n ném³-e·n impossible CERT if-not-I go-I
  'I must go. I have to go. I ought to go.' (lit. 'It is impossible that I do not go.')

See also  $s\vec{k}^w \acute{e} y$  with nominalized negatives in §6.4.3.

# 6.4.2.3. In the Sense 'Or'

An 'if not' clause is also used in the sense 'or,' as in (a) and (b).

(a) səw x<sup>w</sup>ənínsəs k<sup>w</sup>θə stém ?al ?óỷ xáxce? wəwe·s stá?tələwes. (CC 17) wə-x<sup>w</sup>ə-ní-nəs-əs k™θə stém ?aĺ s-[ni-s] NOM-[AUX-3POS] EST-become-be.there-GOAL-3TR ART what just ?áỷ žážce? wə-?śwə-əs stá?tələw-es good little.lake if-not-3SUB creek-3SUB 'Then they reached some good little lake or creek.'

- (b) wənáy k<sup>w</sup>θə sqéqs iwe·s ẳa·s k<sup>w</sup> sxéyəłs iwe·s ẳa·s θə stáləss. (CC 17) ?əv wə-?ówə-əs Åa-əs sqéq-s wənáv k™θә younger.sibling-3POS and if-not-3SUB be.only ART BE3P-3SUB k <sup>w</sup> sxévəł-s ?∍ỷ wə-?áwə-əs Åa-əs θə older.sibling-3POS if-not-3SUB and BE3P-3SUB ART ART stáləs-s spouse-3POS
  - 'It is only his younger sibling or his older sibling or his wife [who can give permission].' (lit. 'It is only his younger sibling and if it is not, the one who is his older sibling and if it is not, the one who is his spouse.'

I have supposed that there is a wa- 'if' in the cluster *iwe*'s 'and if not,' although I did not record it.

# 6.4.3. In Nominalized Clauses

Negated predicates can appear within nominalized clauses, as in (a) to (f).

(a)  ${}^{9}i$  cən tət ${}^{i}a{}^{9}\theta$ ət  ${}^{k}w$ ən s ${}^{9}$ ówə  ${}^{k}w$ óy ${}^{x}\theta$ ət ${}^{w}e$ ·n. (JP) ?i can ťaťá?-θat k™ə nə-s-?áwə k<sup>w</sup>áyx-θət<sup>w</sup>-e⋅n be.testing-self AUX I ART my-NOM-not move-self-I 'I'm trying not to move.' (b) niŵ tətá?θət k<sup>w</sup>s?śwəs k<sup>w</sup>śyxθətəs tθe? (JP) ni? wə-tətá?-θət k <sup>w</sup> s-?áwa-s k<sup>w</sup>áy<sub>x</sub>-θət-əs  $t\theta e^{2}$ AUX EST-be.testing-self ART NOM-not-3POS move-self-3SUB that 'He's trying not to move.' (c) skwéy kws?ówos dáynoxwos. (JP 23) sk<sup>w</sup>év κ̈́» s-?áwə-s dáv-nəx<sup>w</sup>-əs impossible ART NOM-not-3POS die-TR-3TR 'They never failed to kill it.' (d) ni cən tálnəx<sup>w</sup> k<sup>w</sup>s<sup>2</sup>áwəs xésəs θə yəžátəstalx<sup>w</sup> θe<sup>2</sup>. (JP 22) ni? cən tál-nəx<sup>w</sup> s-?áwə-s k w xés-əs A2 AUX I be.understood-TR ART NOM-not-3POS sea.lion-3SUB ART yə-xətə-st-alx<sup>w</sup>  $\theta e^{2}$ along-be.doing-com-us that 'I found out that this thing that was doing that to us was not a sea lion.' (e) ha  $? \epsilon \cdot n \theta = i s k^{w} \epsilon y k^{w} = n s^{2} \delta w = t d^{w} a t \epsilon \cdot n k^{w} = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m w = n s c t i x = m s c t i x = m w = n s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i x = m s c t i$ ha  $?\dot{e}\cdot n\theta \Rightarrow ?\dot{\varphi} \cdot s\dot{k}^{w} \dot{e}y$   $\dot{k}^{w} \Rightarrow n\varphi \cdot s \cdot ?\dot{\varphi} w \Rightarrow t \Rightarrow \dot{q}^{w} \cdot \dot{a}t \cdot \dot{e}\cdot n$   $k^{w} \Rightarrow$ if and impossible ART my-NOM-not be.cut-TR-I be me ART nə-sctíxəm wə-ni-əs my-harpoon.line if-AUX-3SUB 'If it were me, I'd soon [have to] cut my line if it ...'

 (f) ni ct nem k<sup>w</sup>téx<sup>w</sup>t tə sté<sup>9</sup>ex<sup>w</sup>əł nem tə sk<sup>w</sup>uléwtx<sup>w</sup>, Xa swə<sup>9</sup>>wəs ni·s k<sup>w</sup>>>> nəx<sup>w</sup>>> tə sy>>>> k<sup>w</sup>θə sy>wénsəł. (JP)

k<sup>w</sup>téx<sup>w</sup>-t tə ni ct nem sté<sup>9</sup>ex<sup>w</sup>əł nem [<sup>9</sup>ə] tə AUX we AUX(go) enter-TR ART children go [OBL] ART sk<sup>w</sup>ul-éwtx<sup>w</sup> Ža k » źn-nəx »-əs s-wə-?áwə-s ni-əs school-house BE3P NOM-EST-not-3POS AUX-3POS be.taken-TR-3TR s-yá0əs-s k<sup>w</sup>θә yəwén-s-əł tə before-3POS-past ART NOM-transmit-3POS ART

'We sent the kids to school, and so they did not get the traditions of their ancestors.'

The negative  $x^w \partial w \dot{e}$  'not yet' in a nominalized clause, literally 'being not yet' or 'when not yet,' is translated freely as 'before,' as in (g) and (h).

| (g) | хas х                                                                               | wəwes tecələs tə       | žóťəstəm žé∙ls.  | (JP 15) | )                  |       |  |
|-----|-------------------------------------------------------------------------------------|------------------------|------------------|---------|--------------------|-------|--|
|     | х́а                                                                                 | s-x <sup>w</sup> əwé-s | técəl-əs         | tə      | x́ət́ə−st−əm       | xé∙ĺs |  |
|     | BE3P                                                                                | NOM-not.yet-3POS       | arrive.here-3SUB | ART     | be.saying-COM-INTR | xé∙ĺs |  |
|     | 'That was before the one called xé is came.'                                        |                        |                  |         |                    |       |  |
| (h) | n) … k <sup>w</sup> s x <sup>w</sup> əẁés mi∙s wíl≀tə syáq <sup>w</sup> əm. (JP 21) |                        |                  |         |                    |       |  |

| ()                          | k <sup>w</sup> | s-x <sup>w</sup> əŵé-s | mi-əs     | wíl    | tə  | syáq <sup>w</sup> əm |
|-----------------------------|----------------|------------------------|-----------|--------|-----|----------------------|
|                             | ART            | NOM-not.yet-3POS       | come-3SUB | appear | ART | sun                  |
| ' before the sun comes up.' |                |                        |           |        |     |                      |

# **7** Morphology of the Root 1: The Verb

#### 7.1. THE VERB ROOT

The basic shapes of many verb roots may be seen when they appear bare of affixes, as in  $\vec{ak^w}$  'return home,'  $q^w \vec{at}$  'wash ashore,' or  $\vec{k^w} \vec{aq^w}$  'get hit (by something wielded with a clubbing motion),' as in (a) to c).

- (a) ném cən ták<sup>w</sup>.
   AUX(go) I go.home
   'I'm going home.'
- (b) ni? q<sup>w</sup>5ł. AUX wash.ashore 'It washed ashore.'
- (c) k<sup>w</sup>áq<sup>w</sup> čx<sup>w</sup> ce?.
   get.hit you FUT
   'You'll get hit.'

But other verb roots occur (or, at any rate, have been recorded) only with affixes and so their basic shapes must be inferred. And since the presence of an affix may modify the shape of a root and do so differently depending on type of root, process of root expansion, and type of affix, we find that inferring the basic shapes of these roots is not simply a matter of lopping off the affixes but of working out the paradigms and identifying them through analogy. Inferred roots will be written with the root sign ( $\sqrt{}$ ) preceding them.

#### 7.1.1. Canonical Shapes

Most verb roots are di-consonantal. They consist of a consonant, a full vowel (/e/, /a/, or /i/), and a second consonant ( $C_1AC_2$ ), or else a consonant, a schwa, and a second consonant ( $C_1 
abla C_2$ ). There are also a number of tri-consonantal roots that consist of a consonant, a schwa, a second, and a third consonant ( $C_1 
abla C_2 C_3$ ), and there seem to be a few of other shapes ( $C_1AC_2 
abla C_3$ ,  $C_1AC_2AC_3$ ,

 $C_1C_2AC_3$ , etc.). However, some of what appear to be tri-consonantal roots may, upon further analysis, turn out to be di-consonantal roots with petrified affixes. Several roots that appear to have the shape CV (V is any vowel) are probably CVh and thus di-consonantal. There is, however, at least one single-consonant root, the *c*- ~ *c*- that appears in several interrogative words (§17.6 to 17.13) and that may be identical to the verbalizing prefix *c*- 'do, get, make' (§12.2.1).

Limitations on the consonants that can occur in these root types are few. As elsewhere in the language, glottalized resonants do not occur initially. Otherwise, there seems to be no restriction on what consonant can occur as the first. Likewise, as /h/ and /?/ do not occur between schwa and zero, CəC roots cannot have /h/ or /?/ as a second consonant. Otherwise, there seems to be no restriction on what consonant can occur as the second in a di-consonantal root. I have not yet worked out all of the forms of tri-consonantal root, and so cannot yet say what restrictions may occur in them.

Roots of different shapes follow different paradigms. Not only do CAC, C<sub>2</sub>C, and C<sub>2</sub>CC roots behave differently but it also makes a difference whether the first, second, or third consonant is an obstruent (T), a resonant (R), or a laryngeal (H). Accordingly, it is useful to distinguish CAC roots as belonging to subtypes TAT, TAR, TAH, RAT, RAR, RAH, and HAH, to distinguish C<sub>2</sub>C roots as belonging to subtypes T<sub>2</sub>T, T<sub>2</sub>R, R<sub>2</sub>T, R<sub>2</sub>R, H<sub>2</sub>T, and H<sub>2</sub>R, and so on.

#### 7.1.2. Inherent Voice

All verb roots are grammatically intransitive in the sense that they cannot take grammatical objects without first being provided with one of the transitivizing suffixes. In their relationship with grammatical subjects, however, they fall into two classes, non-active and active. Most roots fall into the first of these classes.

(1) Non-active root. These are inherently intransitive or passive in the sense that their grammatical subjects are either acting by themselves or are undergoing or experiencing the action, process, or whatever may be denoted.

Examples of this class are  $t\dot{a}k^{w}$  'return home,'  $t\dot{a}k^{w}$  'fly,'  $k^{w}t^{2}$  'climb, ascend,'  $c\dot{e}^{2}$  'land atop,'  $q^{w}\delta s$  'enter the water,'  $q^{w}\delta t$  'wash ashore,'  $\dot{p}\delta t$  'become conscious,'  $\dot{q}ay$  'die,'  $s\delta \dot{q}$  'split, tear,'  $p\delta n$  'get buried,'  $t^{\theta}\delta \tilde{x}^{w}$  'get washed,'  $k^{w}aq^{w}$  'get hit.'

If the non-active root names an action that involves an agent and it appears bare of affixes, the grammatical subject is the patient, as in (a).

(a) k<sup>w</sup>áq<sup>w</sup> čx<sup>w</sup> ce?
 get.hit you FUT
 'You'll get hit.'

But when a non-active root is transitivized, there is a shift in case relations. Now the grammatical subject is the agent, as in (b). (b) ni? ?> čx<sup>w</sup> k<sup>w</sup>áq<sup>w</sup>-ət
 AUX ROG you get.hit-TR
 'Did you hit it?'

Not all non-active roots can take the transitivizer *-t* (*-\partial t*, *-\dot{\partial} t*, *-\dot{a} t*), but compare the following:

| $\dot{c}\dot{e}^{2}$ 'land atop'                     | $\dot{c}\dot{e}^{2}t$ 'put it atop'             |
|------------------------------------------------------|-------------------------------------------------|
| $q^{w} \delta s$ 'enter the water'                   | $q^{w}s \delta t$ 'put it into the water'       |
| $\dot{p}\dot{\delta t}$ 'become conscious'           | $\vec{p}t\delta t$ 'make him conscious'         |
| <i>q́áy</i> 'die'                                    | <i>ḋá·yt</i> 'kill him'                         |
| <i>sáq</i> 'split, tear'                             | sqét 'split it, tear it'                        |
| $\dot{t}^{\theta} \delta \check{x}^{w}$ 'get washed' | $t^{\theta} \check{x}^{w} \acute{a}t$ 'wash it' |

(See also §10.1, "Transitive Suffixes.")

(2) Active roots. These are inherently active in the sense that the grammatical subject is the agent in the activity denoted. This relatively small group includes  $p e t^{\theta}$  'sew,'  $k^{w} e c$  'look,'  $k^{w} c c$  'cut open, butcher,' q e n 'steal,'  $q^{w} e l$  'speak,'  $y e k^{w}$  'hire,' h e y 'make a canoe,'  $s \delta w q$  'seek,'  $x t e^{2}$  'do.' When an active root is transitivized, there is no shift in case relations: whether the root is bare of affixes or has a transitivizer, the grammatical subject is still the agent. Compare (c) and (d):

| (c) <sup>9</sup> i<br>be.here(AUX)<br>'I am butchering.'    | cən<br>I | k <sup>w</sup> ík <sup>w</sup> əc.<br>be.cutting.open       |
|-------------------------------------------------------------|----------|-------------------------------------------------------------|
| (d) <sup>9</sup> i<br>be.here(AUX)<br>'I am butchering it.' | cən<br>I | k <sup>w</sup> ík <sup>w</sup> əċ-ət.<br>be.cutting.open-tr |
| Compare also:                                               |          |                                                             |
|                                                             |          |                                                             |

| pét <sup>e</sup> 'sew'     | <i>pét<sup>e</sup>at</i> 'sew it'                           |
|----------------------------|-------------------------------------------------------------|
| $k^{w} \acute{e} c$ 'look' | $\vec{k}^{w} \acute{e} c \overrightarrow{a} t$ 'look at it' |
| qéň 'steal'                | <i>qé<sup>9</sup>ant</i> 'steal it'                         |
| sówý 'seek'                | <i>sáwqt</i> 'look for it'                                  |

#### 7.1.3. Internal Modification of the Root

Verb roots undergo a number of processes of internal modification to produce a number of different grammatical forms. Processes of internal modification include reduplication (both CV and CVC), infixing, shift in stress and position of vowel, shift from full vowel to schwa and from schwa to full vowel, vowel lengthening, and glottalization of resonants. The grammatical forms produced by these processes include progressive, plural, diminutive, durative, iterative-dispositional, and resultative forms. Verb roots also undergo internal modification to form stems for various affixes.

There is no one-to-one relationship between process of internal modification and grammatical form. Differently shaped roots may undergo different processes to produce grammatically identical forms. However, most forms can be predicated from the shape of the root, although there are also some irregular forms.

The following sections will deal with the principal forms of the verb and the processes that produce them.

#### 7.2. PROGRESSIVE ASPECT

Nearly all verbs appear in both a simple, unmarked perfective form and a progressive form<sup>1</sup> marked variously, as described below. This opposition of perfective and progressive appears in bare roots functioning as verbs and in verbs composed of roots and various kinds of affixes. It also appears in (or intersects with) plural, durative, and dispositional forms. Diminutive forms, however, occur in the progressive aspect only. Resultatives stand apart in not making this distinction.

Halkomelem progressive forms are often equatable with English progressives, for example, compare i t i a a m 'be singing' (progressive) with i t a m 'sing' (perfective) in the following:

| ni títələm.      | ni ťíləm.  | íləm ce?.       |
|------------------|------------|-----------------|
| 'He is singing.' | 'He sang.' | 'He will sing.' |

But Halkomelem progressives may also be used for habitual action where English would have a simple present or 'used to' for the past, as in (a) to (e).

| (a) | wəyá0 cən     | wəkʷé    | k̓»əcnəxʷ ni ?ə tθé | Р.       |     |      |
|-----|---------------|----------|---------------------|----------|-----|------|
|     | wə-yáθ        | cən      | wə-kwékwəc-nəxw     | ni       | ?ə  | tθé? |
|     | EST-always    | Ι        | EST-be.seeing-TR    | be.there | OBL | that |
|     | 'I always see | e him tl | here.'              |          |     |      |

(b) təli<sup>9</sup> k<sup>w</sup>əns mi k<sup>w</sup>>nnəx<sup>w</sup> k<sup>w</sup>ən šqiləs, Xan səw k<sup>w</sup>ek<sup>w</sup>əcət tə xəθi·lə miməłə. (CC 20)

| təli?                                                              | Ќ <sup>w</sup> ə | nə-s-mi     | k ™án∙ | -nəx <sup>w</sup> | k <sup>w</sup> ə | nə-šqíləs  | Х́а         |
|--------------------------------------------------------------------|------------------|-------------|--------|-------------------|------------------|------------|-------------|
| from                                                               | ART              | my-NOM-come | be.tak | en-TR             | ART              | my-sense   | BE3P        |
| nə                                                                 | -s-wə-           | -kʷékʷəc-ət | tə     | <b>x</b> əθí∙     | lə               | míməłə     |             |
| my-NOM-EST-be.seeing-TR                                            |                  |             | ART    | four.p            | erson            | be.dancing | .possesssed |
| 'From when I first came to my senses, I used to see four dancers.' |                  |             |        |                   |                  |            |             |

<sup>1</sup> The form I am calling "progressive" is the "actual" of Thompson and Thompson (1971), the "continuative" of Galloway (1977) and my own earlier usage, and the "imperfective" of Hukari (1978). I have adopted the terms used by Comrie (1976) for comparable aspects.

- (c) k̇<sup>w</sup>ák̇<sup>w</sup>əc-θət čx<sup>w</sup>.
   be.looking-self you
   'Look after yourself.'
- (d) wə-<sup>2</sup>é<sup>2</sup>x̄əθ čx<sup>w</sup> <sup>2</sup>al.
   EST-be.lying you just
   'Just stay in bed.'
- (e) łód ?ə dídək wəls tən sq wəméy. hé?e. dídək wəls. (DK)
   łód ?ə dídək w-əls tə ?ən-sq wəméy hé?e dídək w-əls
   habitually ROG be.biting-ACT ART your-dog yes be.biting-ACT
   'Does your dog bite? Yes. He bites.'

In these sentences, the Halkomelem for 'see,' 'look after oneself,' 'lie down,' and 'bite' are all progressive forms.

Also, the Halkomelem progressive is used with some verbs to denote conditions for which English would use adjectives, as in  $\vec{\lambda} e \vec{\lambda} \partial t \partial \vec{m}$  'salty, taste salty' (a progressive form; cf.  $\vec{\lambda} e t \partial m q \hat{a}^2$  'salt water'),  $t^{\theta} e t^{\theta} \partial x^w \partial m$  'blue' (seemingly a progressive form, although no perfective has been recorded).

The formation of progressive forms depends on the form of the verb root.

### 7.2.1. Progressives of CAC Roots

These are roots consisting of an obstruent or resonant, a full vowel, and an obstruent or resonant (i.e., TAT, RAT, TAR, and RAR roots, C being any consonant but /h/ or /?/). They behave in the same way (with minor variations and exceptional types noted below) and so may be identified as constituting a single class, the largest in the language.

# 7.2.1.1. Progressives of Bare CAC Roots

The bare roots of this type form their progressives by CV reduplication with the stress on the first vowel and the second appearing as schwa ( $C_1 A C_2 \rightarrow C_1 A C_1 \circ C_2$ ), as in the following TAT roots with their progressives:

| $\dot{p}\dot{e}\dot{t}^{\theta}$ 'sew'  | $\dot{p}\dot{e}\dot{p}\dot{a}\dot{t}^{\theta}$ 'be sewing'     |
|-----------------------------------------|----------------------------------------------------------------|
| $t^{\theta} eq$ 'go bald'               | $t^{\theta} \dot{e} t^{\theta} \partial q$ 'be going bald'     |
| $k^{w} ec$ 'look'                       | <i>k<sup>w</sup>ék<sup>w</sup>əc</i> 'be looking'              |
| <i>iák</i> <sup>w</sup> 'go/come home'  | $i di di di k^w$ 'be going/coming home'                        |
| <i>tás</i> 'fish with a net'            | <i>tátəs</i> 'be fishing with a net'                           |
| <i>łák</i> <sup>w</sup> 'fly'           | <i>łáłak</i> <sup>w</sup> 'be flying'                          |
| $k^w a q^w$ 'get hit'                   | $\vec{k}^{w}\vec{a}\vec{k}^{w}\partial q^{w}$ 'be getting hit' |
| <i>tiq</i> <sup>w</sup> 'bump'          | $tit \partial \dot{q}^w$ 'be bumping'                          |
| $\vec{k}^{w} \vec{i} \vec{c}$ 'butcher' | $\vec{k}^{w} \vec{k}^{w} \vec{a} \vec{c}$ 'be butchering'      |
| $\check{x}^{w}iq^{w}$ 'rub, slip'       | $\check{x}^{w} i \check{x}^{w} \partial q^{w}$ 'be rubbing'    |

Non-initial resonants are generally (or always) glottalized in progressive forms. (Where I have not recorded such a resonant as glottalized, as in 'be

disappearing' below, I may simply have missed the glottalization, since it can be quite light.) Examples of RAT roots with initial /y/ and /m/ and their progressives follow:

| <i>yé ý ak</i> <sup>w</sup> 'be hiring'                   |
|-----------------------------------------------------------|
| yéýəq' 'be toppling'                                      |
| $y \acute{a} \acute{y} \partial \breve{x}^w$ 'be thawing' |
| yí ýðq 'be snowing'                                       |
| mámož 'be getting storm-bound'                            |
|                                                           |

I have no examples with initial /1/.

RAT roots with initial /n/ form their progressives with a reordering or metathesis of elements following the stressed vowel; instead of the expectable  $*/-\dot{n}$ -/, we find /-2 = n-/. I have no example of a bare root of this type, but those with affixes follow:

| néčt 'change it'        | <i>né<sup>9</sup>ənct</i> 'be changing it' |
|-------------------------|--------------------------------------------|
| <i>násθət</i> 'get fat' | ná <sup>?</sup> ansθat 'be getting fat'    |

Examples of TAR roots with their progressives are:

| ,<br><i>tén</i> 'disappear'         | <i>tétən</i> 'be disappearing'                                  |
|-------------------------------------|-----------------------------------------------------------------|
| $d^{\theta} em$ 'be out (the tide)' | $i^{\theta} \acute{e} i^{\theta} i^{\phi} i^{\phi}$ 'be ebbing' |
| <i>téw</i> 'flee'                   | tétaw 'be fleeing'                                              |
| qéň 'steal'                         | qéqəň 'be stealing'                                             |
| <i>q́áy</i> 'die'                   | <i>qáqəy</i> 'be sick'                                          |
| <i>Χ̈́ίw</i> 'flee'                 | <i>ẳí ẳ ә</i> w 'be fleeing'                                    |
| $\dot{q}^{w}$ ím 'disembark'        | $\dot{q}^{w} \dot{q}^{w} \partial m$ 'be disembarking'          |
|                                     |                                                                 |

Note that the final resonant of the root may be glottalized, as in 'steal.' Examples of RAR roots and their progressives are:

| wíľ | 'show up'               | wíwoł 'be showing up'  |       |
|-----|-------------------------|------------------------|-------|
| wéỷ | 'be exposed/discovered' | wéway 'be getting expo | osed' |

There are two small sets of exceptional CAC roots that form their progressives in other ways.

(1) A few roots (but not all) with initial /c/ and /x/ have /k<sup>w</sup>/ and /x<sup>w</sup>/, respectively, in their progressives, as in:

| <i>cám</i> 'go/come inland'            | cákwam 'be going/coming inland'                          |
|----------------------------------------|----------------------------------------------------------|
| $x \acute{a} \acute{k}^w$ 'get bathed' | $x \acute{a} x^w \partial \vec{k}^w$ 'be getting bathed' |
| $x \neq x \neq x$ 'get an erection'    | $x a x^w \partial \vec{\lambda}$ 'have an erection'      |

Other roots of this set include ce- 'do what,' ce- 'borrow,' cel- 'follow,' xap- 'sit cross-legged,' and xal- 'drill,' which have been recorded only with affixes. They will be discussed in connection with those affixes. For an explanation of this irregularity, see §1.5.10. (2) A small set of roots with initial labialized obstruents and the vowel /e/ have /a/ in their progressives. They are:

| $\vec{k}^{w} \vec{e} \vec{y}$ 'get hungry' | $k^w \dot{a} k^w \partial \dot{y}$ 'be feeling hungry' |
|--------------------------------------------|--------------------------------------------------------|
| $q^{w}\acute{e}l$ 'speak'                  | $q^{w} \dot{a} q^{w} \partial \vec{l}$ 'be speaking'   |

#### 7.2.1.2. Progressives of CAC Roots with Affixes

CAC roots take the form  $-\partial t$  of -t 'transitive.' With the exceptions noted below, they form their progressives with the same type of reduplication seen with the bare root ( $C_1 \dot{A} C_2 \partial t \rightarrow C_1 \dot{A} C_1 \partial C_2 \partial t$ ), as in the following progressives of TAT roots.

| $p \acute{t}^{\theta} a t$ 'sew it' | <i>pépət<sup>e</sup>ət</i> 'be sewing it'                            |  |
|-------------------------------------|----------------------------------------------------------------------|--|
| $k^w a q^w \partial t$ 'strike him' | <i>k<sup>w</sup>ák<sup>w</sup>əq<sup>w</sup>ət</i> 'be striking him' |  |
| <i>tí q̇̃ wət</i> 'bump him'        | <i>títəq</i> wət 'be bumping him'                                    |  |
| <i>qé pət</i> 'tie it'              | <i>qéqapat</i> 'be tying it'                                         |  |

Roots of the first three of these have been recorded bare of affixes and are given in §7.2.1.1. The bare root of the fourth,  $\dot{q}\dot{e}\,\dot{p}\partial t$  'tie it,' has not been recorded as such, but its form with *-t* 'transitive' and its resultative form  $s\dot{q}\dot{e}\dot{q}\partial\dot{p}$  'knot' support its identification as a CAC root. Similarly, some of the roots identified below as CAC roots have not been recorded bare of affixes, and their identification is tentative.

RAT roots with -t 'transitive,' except for those with initial /n/, form their progressives in the same way.

| <i>wéť<sup>e</sup>st</i> 'pry it'   | <i>wéwət<sup>i</sup>ət</i> 'be prying it'       |
|-------------------------------------|-------------------------------------------------|
| <i>yák<sup>w</sup>ət</i> 'smash it' | yáý váť vet vet vet vet vet vet vet vet vet vet |
| $mis \partial t$ 'sort them out'    | <i>mímðasðt</i> 'be sorting them out'           |

RAT roots with initial /n/ form their progressives as described above, with the expectable  $*/\dot{n}$  reordered as /2n/, as in:

```
níłət 'decide it'
```

ní?ənłət 'be deciding it'

TAR and RAR roots in combining with *-t* 'transitive' fall into two classes: those with labial resonants as their second consonant and those with non-labial resonants as their second consonant.

Those with /m/, /w/, /m/, or /w/ as the second consonant follow the usual formula, as in:

| qámət 'bend it'             | qáqəmet 'be bending it'             |
|-----------------------------|-------------------------------------|
| <i>cewət</i> 'help him'     | <i>cécowot</i> 'be helping him'     |
| <i>lámət</i> 'throw at him' | <i>lálðmðt</i> 'be throwing at him' |
| níwət 'advise him'          | níňawat 'be advising him'           |

Those with /n/, /l/, and /y/ or /n/, /l/, and /y/ as the second consonant show in their perfective forms the kind of reordering or metathesis already noted in the progressives of roots with initial /n/. If the second consonant is /n/, /l/, or /y/, instead of the expectable resonant followed by schwa, we find the vowel of the root lengthened and followed by the resonant, which may itself be phonetically slightly lengthened. For example, the root  $\vec{\lambda} \epsilon l$  'get stuck, be detained' and -t 'tr,' expectably \*/ $\vec{\lambda} \epsilon l t$ /, is realized as / $\vec{\lambda} \epsilon \cdot l t$ / 'stick it on' (phonetically  $[\vec{\lambda} \epsilon \cdot l t] \sim [\vec{\lambda} \epsilon \cdot l \cdot t])$  and the root  $\vec{q} a y$  'die' and -t 'tr,' expectably \*/ $\vec{q} a y t$ /, is realized as / $\vec{q} \cdot y t$ / 'kill him' (phonetically [ $\vec{q} \cdot y t$ ] ~ [ $\vec{q} \epsilon \cdot i t$ ]). Other examples are:

| <i>pé·lt</i> 'skim it'                           | <i>pépələt</i> 'be skimming it'                                  |
|--------------------------------------------------|------------------------------------------------------------------|
| $\dot{t}\dot{e}\cdot lt$ 'cut it along the back' | $t \dot{e} t \dot{a} \dot{l} a t$ 'be cutting it along the back' |
| <i>Žé·lt</i> 'stick it on'                       | <i>ẳéẳələt</i> 'be sticking it on [him?]'                        |
| <i>cé·lt</i> 'chase him'                         | <i>cék<sup>w</sup>ələt</i> 'be chasing him'                      |
| <i>k</i> <sup>w</sup> é·yt '''train'' him'       | <i>k</i> <sup>w</sup> ék <sup>w</sup> əyət 'be ''training'' him' |

(The verb glossed "train" him' means prepare him for a vision quest.)

If the second consonant is  $/\dot{n}/, /\dot{l}/, \text{ or }/\dot{y}/, \text{ instead of the expectable glottal$ ized resonant followed by schwa, we find the sequence glottal stop (phonetically possibly lost), schwa (phonetically possibly echoing the stressed vowel), $and the resonant. For example, the root <math>\dot{c}\dot{e}\dot{n}$  'be straight' and -t 'transitive,' expectably \*/centl, are realized as /centl (phonetically [centl] ~ [centl]). Other examples are:

| <i>tá?əlt</i> 'learn it'                            | <i>tátələt</i> 'be learning it'  |
|-----------------------------------------------------|----------------------------------|
| <i>čí?əlt</i> 'seize it'                            | <i>čí čəl ət</i> 'be seizing it' |
| <i>má</i> <sup>9</sup> <i>əyt</i> 'aim it'          | <i>mámðyðt</i> 'be aiming it'    |
| <i>wí<sup>9</sup>alt</i> 'show it, make it visible' | wíwałat 'be showing it, etc.'    |
| wé·yt 'publicize it'                                | wéwayat 'be publicizing it'      |

Two CAC roots (so far noted) have irregular forms with -t 'transitive.' One, *néc*' 'be different,' is exceptional in that the suffix appears as -t rather than  $-\partial t$ .

| néct | 'change it' | né?anct | 'be changing if |
|------|-------------|---------|-----------------|
| neci | enunge n    | ne onei | be enanging it  |

The other, composed of the prefix  $x^{w}$ - 'locative' (here 'within'?) and  $q^{w} \acute{e} l$  'speak,' is exceptional in that it forms its progressive on the pattern of a TAH root (see below).

| $x^w q^w \acute{e} \cdot lt$ 'sing out the words | $x^w q^w \partial q^w \acute{e} lt$ 'be singing out the words' |
|--------------------------------------------------|----------------------------------------------------------------|
| (of a syáwan)'                                   | mus-forms                                                      |

With -x 'transitive,' CAC roots form their progressives with the type of reduplication undergone by a bare root and with no linking schwa, as in:

| <i>wélx</i> 'throw it'         | $w \dot{e} \dot{w} \dot{e} \dot{l} x$ 'be throwing it' |
|--------------------------------|--------------------------------------------------------|
| <i>télx</i> 'follow it'        | <i>tétəlx</i> 'be following it'                        |
| $k^{w} \epsilon l x$ 'hide it' | $k^{w} e k^{w} \partial l x$ 'be hiding it'            |

With  $-\partial m$  'intransitive,' CAC roots form their progressives regularly with the type of reduplication undergone by a bare root, as in:

| <i>iícəm</i> 'swim (on the surface)' | <i>it i zcəm</i> 'be swimming, etc.'                 |
|--------------------------------------|------------------------------------------------------|
| <i>čísəm</i> 'grow'                  | <i>čí čəsəm</i> 'be growing'                         |
| <i>méqam 'jump</i> (as a fish)'      | mémadam 'be jumping (as a fish)'                     |
| <i>yáq<sup>w</sup>əm</i> 'sweat'     | yá ý aque yá y a v y y y y y y y y y y y y y y y y y |
| <i>píləm</i> 'overflow'              | <i>pí palam</i> 'be overflowing'                     |
| <i>tiləm</i> 'sing'                  | <i>títələm</i> 'be singing'                          |
| x <sup>w</sup> áyəm 'sell'           | xwáxwəyəm 'be selling'                               |
| qéwəm 'rest'                         | qéqəwəm 'be resting'                                 |
| <i>žínom</i> 'growl'                 | <i>xí x ən əm</i> 'be growling'                      |

There is a small set of words that are or appear to be composed of TAR roots plus  $-\partial m$  'intransitive' or  $-\partial l$  'move toward,' in which the final resonant of the root is lost, leaving a long vowel; for example, //tém $-\partial m//$  'call out' is realized as  $t\acute{e}\cdot m$  and //k<sup>w</sup>él $-\partial l//$  'hide' is realized as  $k^w e \cdot l$ . The underlying forms can be inferred from the transitives  $t\acute{e}m\partial t$  'call him' and  $k^w\acute{e}lx$  'hide it.' These words form their progressives by CV reduplication with the stress on the second vowel, following the pattern of TAH roots (see below), but with that vowel raised to /i·/, as with some T $\partial R$  roots with  $-\partial m$  (see below). Members of this set discovered so far are:

| <i>té</i> · <i>m</i> 'call out'                             | <i>tətí·m</i> 'be calling out'                             |
|-------------------------------------------------------------|------------------------------------------------------------|
| (cf. témət 'call him'                                       | <i>tətí mət</i> 'be calling him')                          |
| $k^{w}e \cdot l$ 'hide'                                     | $k^{w} \partial k^{w} i \cdot \dot{l}$ 'be hiding'         |
| (cf. $k^{w} \ell x$ 'hide it'                               | $k^{w} e k^{w} \partial l x$ 'be hiding it')               |
| <i>łé</i> · <i>l</i> 'come/go shoreward'                    | $l \partial l \hat{l} \hat{l}$ 'be coming/going shoreward' |
| (cf. <i>télx</i> 'move it back from the centre of the fire' | $l \ell l a l x$ 'be moving, etc.')                        |
| <i>tá</i> · <i>l</i> 'come/go seaward'                      | $t \partial t \hat{l}$ 'be coming/going seaward'           |

In the first example,  $t = t i \vec{m} dt$ , the progressive of the transitive seems to be based on the progressive of the form with  $-\partial m$ . In the third example, there is no direct evidence for a root tal-, but the evidence for  $k^{w} \ell l$ - and  $t \ell l$ - implied by  $k^{w} \ell l x$  and  $t \ell l x$  suggests it. (AG gave  $t \partial t \hat{a} \cdot \hat{l}$  as 'be going/coming seaward,' but CC gave  $t \partial t \hat{a} \cdot \hat{l}$  as 'being out to sea,' a possible durative form.)

With  $-n\partial x^w \sim -l\partial x^w$  'limited control,'  $-st\partial x^{w_1}$  'causative,' and  $-st\partial x^{w_2}$  'comitative,' CAC roots appear in the perfective with the vowel weakened to schwa although retaining the stress, but in the progressive the full vowel reappears (C<sub>1</sub> $\delta$ C<sub>2</sub> $- \rightarrow$  C<sub>1</sub> $\Delta$ C<sub>1</sub> $-\partial$ C<sub>2</sub>-), as in:

| <i>pótnəx<sup>w</sup></i> 'recognize him'            | <i>pí pətnəx</i> <sup>w</sup> 'be recognizing him'         |
|------------------------------------------------------|------------------------------------------------------------|
| $p \delta t s t \partial x^w$ 'have him identify it' | <i>pí pətstəx</i> <sup>w</sup> 'be having him identify it' |
| (cf. <i>cpít</i> 'recognize')                        |                                                            |

| $k^w \delta c n \partial x^w$ 'see it'                                   | $k^{w} \epsilon k^{w} a c n a x^{w}$ 'be seeing it'                                            |
|--------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|
| $\vec{k}^{w} \delta cst \partial x^{w}$ 'show it to him'                 | $k^{w} \epsilon k^{w} a cst a x^{w}$ 'be showing it to him'                                    |
| (cf. $\vec{k}^w \acute{e}c$ 'see'                                        | $k^{w} \acute{e} c \partial t$ 'look at it')                                                   |
| $t \neq \hat{q}^w n \neq x^w$ 'bump him (accidentally)'                  | <i>títəq</i> <sup>w</sup> <i>nəx</i> <sup>w</sup> 'be bumping him (accidentally)'              |
| (cf. <i>tí q<sup>w</sup>ət</i> 'bump him')                               |                                                                                                |
| $t^{\theta} \dot{q}^{\psi} n \partial x^{\psi}$ 'hit him (accidentally)' | $i\hat{t}\hat{t}\hat{t}\hat{\theta}\hat{q}^{w}n\partial x^{w}$ 'be hitting him (accidentally)' |
| (cf. $t^{\theta} i q^{W} \partial t$ 'punch him')                        |                                                                                                |
| $k^{w} \delta q^{w} n \partial x^{w}$ 'hit him (accidentally)'           | $k^{w} a k^{w} a q^{w} n \partial x^{w}$ 'be hitting him (accidentally)'                       |
| (cf. $k^w á q^w \partial t$ 'hit him, as with a club')                   |                                                                                                |
| $\dot{c}\delta wn\delta x^w$ 'manage to help him'                        | <i>cécownox</i> <sup>w</sup> 'be managing to help him'                                         |
| (cf. <i>čéwət</i> 'help him')                                            |                                                                                                |
| <i>cólləx</i> <sup>w</sup> 'overtake him'                                | <i>cék<sup>w</sup>əlləx<sup>w</sup></i> 'be overtaking him'                                    |
| (cf. <i>cé·lt</i> 'follow him')                                          |                                                                                                |
| $q^{w} \partial l st \partial x^{w}$ 'speak to him'                      | $q^{w} \hat{a} q^{w} \hat{s} \hat{l} s t \hat{s} x^{w}$ 'be speaking to him'                   |
| (cf. <i>q<sup>w</sup>él</i> 'speak')                                     |                                                                                                |

With *éls* 'activity,' CAC roots appear in the perfective with loss of stress and the vowel reduced to zero where both consonants are obstruents, but again in the progressive CAC roots appear in their full form  $(C_1C_2- \rightarrow C_1AC_1 \rightarrow C_2)$ , as in:

| $p \vec{k}^{w} \vec{e} ls$ 'warm something'                  | $p \acute{e} p \imath \acute{k}^w \imath \acute{l} s$ 'be warming s.t.' |
|--------------------------------------------------------------|-------------------------------------------------------------------------|
| (cf. <i>pékwət</i> 'warm it up')                             |                                                                         |
| $\dot{t}^{\theta}\dot{q}^{w}\dot{e}ls$ 'hit (with the fist)' | $i^{\theta}ii^{\theta}q^{w}ais$ 'be hitting'                            |
| (cf. $\dot{t}^{\theta} \dot{q}^{w} \partial t$ 'punch him')  |                                                                         |
| <i>łk̇̃wéls</i> 'hook s.t., hook fish'                       | <i>títək<sup>w</sup>əls</i> 'be hooking fish'                           |
| (cf. <i>łík<sup>w</sup>ət</i> 'hook it, gaff it')            |                                                                         |
| $k^w q^w \ell ls$ 'hit, go to bat'                           | <i>k៉<sup>w</sup>ák៉<sup>w</sup>əq<sup>w</sup>əls</i> 'be hitting'      |
| (cf. $k^w \acute{a} q^w \partial t$ 'hit him, club him')     |                                                                         |

Or if one or both consonants are resonants, the vowel is reduced to schwa  $(C_1 \ni C_2 - \rightarrow C_1 \acute{A} C_1 \ni C_2 -)$ , as in:

| <i>pəwéls</i> 'freeze s.t.'             | <i>pí pəwəls</i> 'be freezing s.t.'       |
|-----------------------------------------|-------------------------------------------|
| (cf. <i>píŵət</i> 'freeze it')          |                                           |
| <i>nətéls</i> 'give orders'             | ní <sup>?</sup> anłals 'be giving orders' |
| (cf. <i>nílət</i> 'set a time, decide') |                                           |
| <i>ləméls</i> 'throw s.t.'              | <i>lál amals</i> 'be throwing s.t.'       |
| (cf. <i>lámðt</i> 'throw it at him')    |                                           |

With lexical suffixes, CAC roots appear in forms similar to those seen with grammatical suffixes. The form of the root usually depends on whether the suffix is unstressed or stressed and whether it begins with a vowel or with a consonant.

Before an unstressed suffix that begins with a vowel, a CAC root appears as CÁC- in the perfective and likewise in its full form in the progressive (cf. CAC roots with *-om* 'intransitive'), as in:

 $x^wládq^wast$  'slap his face' $x^wlálaq^wast$  'be slapping his face'(cf.  $láq^wast$  'slap him,' -as 'face,'  $x^w$ - 'locative')qlsacan 'make a net'qlqasacan 'be making a net'(cf. qlsat 'make it fast with a line,' -acan 'surface') $llcarq^wam$  'get a haircut' $lllacarq^wam$  'get a haircut' $llacarq^wam$  'be getting a haircut'(cf. llcat 'cut it,' - a'qw 'head,' -am 'intransitive') $\lambda elawst$  'hitch him (a horse) up' $\lambda elawst$  'be hitching him up'(cf.  $\lambda e^{lt}$  'stick it on,' -aws 'body,' -t 'transitive')

There are exceptions, such as

 $x^{wl}\delta m \partial st$  'throw it in his face'  $x^{w}[h]\delta m \partial st$  'be throwing in his face' (cf.  $l\delta m \partial t$  'throw it and hit him,' - $\partial s$  'face')

in which the root is reduced to  $R \Rightarrow R$ - and then inflected like a root of that type (see §7.2.8).

Before an unstressed lexical suffix beginning with a consonant, a CAC root appears as C $\diamond$ C- in the perfective but with the full vowel in the progressive (like CAC roots with  $-n \partial x^{w}$ ).

k wáx w csəm 'knock on a door'  $k^w \acute{a}k^w ax^w csəm$  'be knocking on the door' (cf.  $k^w \acute{a}x^w at$  'knock on it,' -cəs 'hand,' - $\partial m$  'intransitive')  $x^w t\acute{a}lqat$  'answer him'  $x^w t\acute{e}talqat$  'be answering him' (cf.  $t\acute{e}l$  'copy, follow suit,' -qan 'voice,' -t 'transitive')  $p\acute{a}k^w n act$  'put it on the fire to heat'  $\check{s}x^w p\acute{e}pak^w n ac$  's.t. heating' (cf.  $p\acute{e}k^w at$  'warm it,' -nac 'base, butt,' -t 'transitive')

In the last example,  $\delta x^w p \ell p \partial k^w n \partial c$  is derived from the progressive with  $\delta x^w$ -'oblique nominalizer.'

Before a stressed lexical suffix beginning with a vowel, a CAC root appears in the perfective as CC- (if both consonants are obstruents) or C $\Rightarrow$ C- (if one or both are resonants). Before a stressed lexical suffix beginning with a consonant, a CAC root appears in the perfective as C $\Rightarrow$ C-. In the progressives of such verbs, the full vowel of the root may or may not appear and may or may not bear a secondary stress.

nowéyoł 'advise a child'
nonowéýoł 'be advising a child'
(cf. níwoł 'advise him,' -éyoł 'child')
kwocxénom 'keep a lookout'
kwokwocxénom 'be keeping a lookout'
(cf. kwéc 'look,' -éxon 'side,' -om 'intransitive')

# 7.2.1.3. Progressives of CÁCVC Roots

There are a number of what appear to be roots consisting of a consonant, a full vowel, and a second consonant, followed by a schwa and a third consonant (CÁCəC) and that form their progressives as CAC roots do, by CV reduplication with the stress on the first vowel. These include:

| <i>téyəq</i> 'get angry'                           | <i>tétayaq</i> 'be getting angry                              |
|----------------------------------------------------|---------------------------------------------------------------|
| $i \acute{a} \acute{l} a \check{x}^w$ 'chase away' | $i \acute{a} t i \acute{a} l a \check{x}^w$ 'be chasing away' |
| <i>žíləž</i> 'make war'                            | <i>ží žələž</i> 'be making war'                               |
| <i>síwəl</i> 'feel, be aware'                      | sísawał 'be feeling'                                          |
| (possibly with - <i>əl</i> 'move toward')          |                                                               |
| cłéwan 'make mats'                                 | cłéławań 'be making mats'                                     |
| (< c- 'make, have,' and the root of sta            | <i>éwən</i> 'sleeping mat')                                   |
| $k^{w}(t)$ 'live with in-laws'                     | $k^{w}ik^{w}\partial t\partial w$ 'be living with in-laws'    |
| (cf. skwitow 'co-parent-in-law, second             | -degree affine')                                              |
| yá·ys //yáyəs// ʻwork'                             | yá·ýyəs //yáýəyəs// 'be working'                              |

There are at least two roots that might be grouped with the last set, except that the third consonant is a glottal stop, which must be preceded by a full vowel (CÁCAH). This pattern suggests that the full vowel is an underlying schwa.

| <i>cmék<sup>w</sup>e</i> ? 'attend a funeral'         | cméňək <sup>w</sup> e? | 'be attending as funeral' |
|-------------------------------------------------------|------------------------|---------------------------|
| (< c- 'make, have,' cf. $m \acute{e} k^w e^{2t}$ 'con | ndole them')           |                           |
| <i>míłe</i> <sup>2</sup> 'dance possessed'            | mí <i>mðite?</i> 'be   | e dancing possessed'2     |

With *-t* 'transitive,' presumed CÁCəC and CÁCAH roots appear with similar progressive forms, as in:

| <i>ťálðxwðt</i> 'chase him away'      | $\dot{t}\dot{a}\dot{t}\partial l\partial \dot{x}^{w}\partial t$ 'be chasing him away' |
|---------------------------------------|---------------------------------------------------------------------------------------|
| $m\acute{e}k^w e^{2t}$ 'condole them' | <i>mémək</i> <sup>w</sup> e <sup>?</sup> t 'be condoling them'                        |
| síwəlt 'make him aware'               | sísəwəlt 'be letting him know'                                                        |

The schwa before the *-t* in 'chase him away' is expectable if  $t \dot{a} l \partial x^w$  is a root. Its absence in  $m \dot{e} k^w e^2 t$  is also expectable for a root ending with a glottal stop.

<sup>2</sup> As Boas (1897, 645) noted, míte<sup>2</sup>, meaning 'dance possessed by one's syáwan in the winter dance,' is probably a loan from Kwakwala. Cf. Lincoln and Rath (1980), root 261, Kwakiutl miλa 'teasing,' also the Kwakwala name of a series of dances.

Its absence in *siwalt*, however, suggests that the root is *siw-* and that the *-al* is a suffix.

There is also a set of words that are composed of stems of the shape CÁC $\Rightarrow$ C-(in which the full vowel is either /e/ or /a/) and the suffixes  $\neg m$  'intransitive' or  $\neg a$  'move toward' and that form their progressives in quite a different way. Some of these stems are clearly identifiable as expansions of C $\Rightarrow$ CT roots and so this set will be described under that class (see §7.2.10).

#### 7.2.2. Progressives of HAC Roots

Roots consisting of /h/ or /?/, a full vowel, and an obstruent or resonant (HAT and HAR roots) seem to form a single class with HAT and HAR subclasses.

Bare HAT roots form their progressives by inserting a glottal stop after the vowel (HÁT  $\rightarrow$  HÁ<sup>9</sup>T), as in:

| <i>hék</i> <sup>w</sup> 'remember'         | $h \hat{e}^{2} \vec{k}^{w}$ 'be remembering'                |
|--------------------------------------------|-------------------------------------------------------------|
| <i>hák</i> <sup>w</sup> 'get used'         | $h \hat{a}^2 k^w$ 'be getting used'                         |
| $2i\vec{k}^{w}$ 'be lost'                  | $2i^{2}k^{w}$ 'be getting lost'                             |
| ${}^{?}\acute{a}\vec{k}^{w}$ 'get hung up' | ${}^{?}\acute{a}{}^{?}\acute{k}{}^{w}$ 'be getting hung up' |

Bare HAR roots, however, form their progressives by lengthening the vowel and glottalizing the resonant ( $H\dot{A}R \rightarrow H\dot{A} \cdot \vec{R}$ ), as in:

| háy 'stop'                    | $h\dot{a}\dot{y}$ 'be stopping, be coming to a stop' |
|-------------------------------|------------------------------------------------------|
| $h\acute{e}$ ý 'make a canoe' | $h\acute{e}\dot{y}$ 'be making a canoe'              |

A progressive form like  $h\hat{e}^{\gamma}\vec{k}^{w}$  'be remembering' may be the product of CV reduplication with  $/\gamma/$  the realization of /h/ glottalized (as resonants are in progressive forms) and with the second vowel (expectably a schwa) deleted, that is,  $h\hat{e}\vec{k}^{w}$  is reduplicated as  $//he\hat{h}e\vec{k}^{w}// \rightarrow //h\hat{e}^{\gamma}\vec{k}^{w}// \rightarrow h\hat{e}^{\gamma}\vec{k}^{w}$ . A progressive form like  $h\hat{a}\cdot\dot{y}$  'be stopping' may be produced in the same way but with the schwa retained because of the final resonant and the glottal stop deleted, that is,  $h\dot{a}y$  is reduplicated as  $//ha\hat{h}a\dot{y}// \rightarrow //h\hat{a}\gamma\dot{y}// \rightarrow //h\hat{a}\gamma\dot{y}// \rightarrow h\hat{a}\dot{\gamma}\dot{y}$ .

With -t 'transitive,' HAT roots form their progressives like the bare roots.

| $hik^{w} \partial t$ 'give a war cry' | $hi^{2}k^{w} \partial t$ 'be giving a war cry'      |
|---------------------------------------|-----------------------------------------------------|
| <i>híqət</i> 'put it under'           | <i>hí?qət</i> 'be putting it under'                 |
| <i>háq<sup>w</sup>ət</i> 'smell it'   | $h\dot{a}^{\gamma}q^{w}\partial t$ 'be smelling it' |
| <i><sup>9</sup>é pot</i> 'wipe it'    | <i><sup>?</sup>é<sup>?</sup>p̀ət</i> 'be wiping it' |
| <i><sup>?</sup>áťət</i> 'stretch it'  | ${}^{?}\dot{a}{}^{?}\dot{t}at$ 'be stretching it'   |
| $2ik^{w} \partial t$ 'throw it away'  | $2i^{2}k^{w} \partial t$ 'be throwing it away'      |

With *-t* 'transitive,' HAR roots form the perfective according to a pattern already seen in TAR and RAR roots; if the final resonant is a non-labial, the vowel is lengthened and there is no schwa after the resonant. The progressive is formed by inserting a glottal stop, as in:

| <i>hí·lt</i> 'roll it over, push it off'   | hí?əlt 'be rolling it over, etc.'                              |
|--------------------------------------------|----------------------------------------------------------------|
| <sup>9</sup> á·nt 'agree with his request' | <sup>2</sup> á <sup>2</sup> ant 'be agreeing with his request' |

Following arguments used earlier, we might suppose that for the first set above a perfective is the product of  $//hil = t// \rightarrow //hi = t// \rightarrow hi't$  and the progressive of  $//hih = 1 = t// \rightarrow //hi' = 1 = t// \rightarrow hi't$ .

With *-om* 'intransitive' and other unstressed suffixes with initial vowels, HAT roots also follow the pattern of the bare root.

| $h\acute{e} t\dot{\theta}_{\partial m}$ 'breathe'  | <i>hé<sup>?</sup>ť&amp;m</i> 'be breathing'                                    |
|----------------------------------------------------|--------------------------------------------------------------------------------|
| hésəm 'sneeze'                                     | <i>hé<sup>s</sup>səm</i> 'be sneezing'                                         |
| háq <sup>w</sup> əm 'stink'                        | há <sup>?</sup> q <sup>w</sup> ðm 'be stinking'                                |
| $2it \partial m$ 'dress oneself'                   | $2i^{2}t^{\theta} \partial m$ 'be dressing oneself'                            |
| <i><sup>γ</sup>ί x̃ əm</i> 'borrow money'          | $2i^{2} \dot{x} \partial m$ 'be borrowing money'                               |
| $x^{w?} \dot{a} \dot{p} a s a m$ 'wipe one's face' | $x^{w^2} \dot{a}^2 \dot{p} \partial s \partial \dot{m}$ 'be wiping one's face' |
| (cf. <sup>?</sup> é pot 'wipe it,' -os 'face')     |                                                                                |

With *-om* 'intransitive' and other vowel-initial suffixes, HAR roots form their progressives in two ways:

(1) by simply glottalizing the resonant, as in

| híləm | 'roll, fall off' | hiləm | 'be rolling, b | be falling |
|-------|------------------|-------|----------------|------------|
|       | ,                |       | L //           |            |

As I recorded it, the progressive here is phonetically [hi?ləm], which is possibly the realization of //hí?ləm//, which is what the rules ought to produce.

(2) by lengthening the vowel and glottalizing the resonant, as in

| °í wəst            | 'show him, instruct him' | ?í∙ŵəst | 'be showing him, etc.' |
|--------------------|--------------------------|---------|------------------------|
| <sup>9</sup> áməst | 'give it to him'         | ?á∙ṁəst | 'be giving it to him'  |

The roots of these words are  $\sqrt{2}em$  'give' and  $\sqrt{2}tw$  'show,' and both are formed with *-as* 'recipient' (§10.4.3). Here too the progressives may be the realizations of  $//2i^9w$  ast// and  $//2a^9m$  ast//.

One seemingly anomalous verb may belong here:

?é?əm 'give'

?é·mam 'be giving'

off'

This verb is probably composed of  $\sqrt{2em}$  'give' and  $-\partial m$  'intransitive.' The underlying forms may be //?éməm// 'give' and //?é?məm// 'be giving,' but if so, the replacement of the medial resonant by a glottal stop is unique.

With  $-n \partial x^w$  'limited control' and other unstressed suffixes beginning with a consonant, an HAC root appears as H $\Rightarrow$ C- in the perfective and as with vowel-initial suffixes in the progressive, as in:

| $h \neq q^w n \neq x^w$ 'smell it'     | $h a^{?} q^{w} n \partial x^{w}$ 'be smelling it'               |
|----------------------------------------|-----------------------------------------------------------------|
| <i>?ák̈wnəx<sup>w</sup></i> 'lose it'  | $\gamma t^{\gamma} \vec{k}^{w} n \partial x^{w}$ 'be losing it' |
| <i>háynəx</i> <sup>w</sup> 'finish it' | <i>há</i> ·ý <i>nəx</i> <sup>w</sup> 'be finishing it'          |
$x^{w?\delta}x^{w}m\partial t$  'covet it'  $x^{w?\delta}x^{w}m\partial t$  'be coveting it'  $x^{w?\delta}wc\partial st$  'instruct him with the hands'  $x^{w?\ell}wc\partial st$  'be instructing, etc.'

With suffixes beginning with a stressed vowel, a HAC root appears as unstressed H $_{PC}$ - in the perfective but with the full vowel in the progressive, as in:

 ${}^{?}\partial \check{x}\dot{a}\cdot y\partial (n\partial m)$  'shave the beard'  ${}^{?}(\check{x}\partial y\partial \theta (n\partial m)$  'be shaving the beard' (<  ${}^{?}(\check{x}\partial t)$  'scrape it,'  $-\dot{a}y\partial \partial n$  'mouth,'  $-\partial m$  'intransitive')

#### 7.2.2.1. Progressives of HAC(C) > C Roots

There are a few words that may be roots of this shape or may be HAC roots with suffixes that are not now identifiable. These include:

| <sup>?</sup> <i>ítət</i> 'sleep'               | <sup>?</sup> <i>í</i> <sup>?</sup> <i>tit</i> 'be sleeping'    |
|------------------------------------------------|----------------------------------------------------------------|
| $e^{i} \delta \delta \theta$ 'lie down'        | ${}^{?}\acute{e}{}^{?}\check{x}\partial\theta$ 'be lying down' |
| ?íməx 'walk'                                   | ?ímax 'be walking'                                             |
| <sup>?</sup> <i>imən</i> 'sing an incantation' | <sup>?</sup> íman 'be singing an incantation'                  |
| <i>?íləqət</i> 'buy it'                        | <i>?íl̊əqət '</i> be buying it'                                |
| <i>?áləxət</i> 'go get it, fetch it'           | ?álaxat 'be fetching it'                                       |
| <sup>?</sup> élməy 'wait'                      | <i><sup>?</sup>éĺmə</i> ý 'be waiting'                         |

#### 7.2.3. Progressives of TAH Roots

Roots consisting of an obstruent, a full vowel, and /?/ or (possibly) /h/ form their progressives by CV reduplication with the stress on the second vowel and the first appearing as schwa (TÁH  $\rightarrow$  T $_{9}$ TÁH).

Roots with final glottal stop present no problems, and their behaviour will be described first. The pattern of reduplication is seen in a few bare roots of the shape TA?.

| $\vec{k}^{w} i^{\gamma}$ 'climb'       | $k^{w} \partial k^{w} i^{\gamma}$ 'be climbing'                             |
|----------------------------------------|-----------------------------------------------------------------------------|
| $\dot{X}\dot{a}^{2}$ 'calm down'       | $\dot{X} \partial \dot{X} \dot{a}^{2}$ 'be calming down'                    |
| sé <sup>9</sup> 'rise'                 | səsé <sup>2</sup> 'be rising'                                               |
| $\dot{c}\dot{e}^{\gamma}$ 'land atop'  | <i>ċəċé<sup>9</sup></i> 'be landing atop'                                   |
| $\check{x}i^{2}$ 'appear, be revealed' | $\check{x} \partial \check{x} i^{\gamma}$ 'be appearing, be getting caught' |

TA? roots take the -t form of the -t 'transitive' and form their progressives as when bare.

| $s e^{2}t$ 'raise it'                           | $s \ge s e^{2t}$ 'be raising it'                                          |
|-------------------------------------------------|---------------------------------------------------------------------------|
| $\dot{t}\dot{e}^{2}t$ 'try it, test it'         | $\dot{t}\partial\dot{t}\dot{e}^{\gamma}t$ 'be trying it, be testing it'   |
| $\dot{t}^{\theta} \dot{e}^{\gamma} t$ 'chew it' | $t^{\theta} \partial t^{\theta} \dot{e}^{2} t$ 'be chewing it'            |
| $\dot{X}\dot{e}^{2}t$ 'fetch it, invite him'    | $\dot{X} \partial \dot{X} \dot{e}^{2}t$ 'be fetching it, be inviting him' |
| $\dot{c}\dot{e}^{\gamma}t$ 'set it on'          | $\dot{c}\partial\dot{c}\dot{e}^{2}t$ 'be setting it on'                   |
| $k^{w} e^{2t}$ 'let it go, drop it'             | $k^{w} \partial k^{w} \dot{e}^{2} t$ 'be letting it go, be dropping it'   |

| $x^{w} e^{2t}$ 'lessen it, lighten it'              | $x^{w} \partial x^{w} \dot{e}^{2}t$ 'be lessening it, be lightening it'                |
|-----------------------------------------------------|----------------------------------------------------------------------------------------|
| $\dot{q}^{w} \dot{e}^{2} t$ 'sweep it'              | $\dot{q}^{w}\partial\dot{q}^{w}\dot{e}^{\gamma}t$ 'be sweeping it'                     |
| $\dot{\lambda}\dot{a}^{2}t$ 'comfort him, calm him' | $\dot{X} \partial \dot{X} \dot{a}^{2}t$ 'be comforting him, etc.'                      |
| $c\dot{a}^{2}t$ 'pry it apart'                      | $c \partial c \hat{a}^{2} t$ 'be prying it apart'                                      |
| $\dot{q}\dot{a}^{2}t$ 'put it together'             | $\dot{q}$ $\dot{q}$ $\dot{a}$ $\dot{q}$ $\dot{a}$ $\dot{c}$ to be putting it together' |
| $\check{x}i^{2}t$ 'reveal it, create it'            | $\check{x} \partial \check{x} i^{2} t$ 'be revealing it, be creating it'               |

With  $-\partial m$  'intransitive,' the glottal stop of a TA? root may be lost, leaving a long vowel, as in:

```
\dot{t}^{\theta} \dot{e}^{\cdot} m \sim \dot{t}^{\theta} \dot{e}^{2} \partial m 'chew' \dot{t}^{\theta} \partial \dot{t}^{\theta} \dot{e}^{\cdot} \dot{m} 'be chewing'
(cf. \dot{t}^{\theta} \dot{e}^{2} t 'chew it,' \dot{t}^{\theta} \partial \dot{t}^{\theta} \dot{e}^{2} t 'be chewing it')
```

With  $-n \partial x^w$  'limited control' and other consonant-initial unstressed suffixes, a TA? root retains the full vowel in the perfective. The suffix  $-n \partial x^w$  may appear as  $-n x^w$ .

 $s\acute{n}ax^{w} \sim s\acute{n}x^{w}$  'manage to lift it'  $sas\acute{n}x^{w}$  'be managing to lift it'  $\check{x}\acute{n}ax^{w}$  'catch him in the act'  $\check{x}a\check{x}\acute{n}x^{w}$  'be catching him in the act'  $i\acute{e}^{2}xat$  'try out a new dancer'  $ia\acute{t}\acute{e}^{2}xat$  'be trying out a new dancer'  $(< i\acute{e}^{2}t$  'try it,' -xan 'foot,' -t 'transitive')

With a suffix with an initial stressed vowel, a TA? root appears as Ta? in the perfective but is TaTÁ? in the progressive, as in:

 $\dot{\lambda}\partial^2 \dot{e} y\partial t$  'comfort a child'  $\dot{\lambda}\partial \dot{\lambda}\dot{a}\dot{y}\partial t$  'be comforting a child' (cf.  $\dot{\lambda}\dot{a}^2 t$  'comfort him,' - $\dot{e} y\partial t$  'child')

With stressed consonant-initial suffixes, a TA<sup>9</sup> root appears as Tə in the perfective and variably in the progressive, to judge from the following:

| <i>Žəłéləm</i> 'go after food'                                                           | <i>Žé?tələm</i> 'be going after food' (JP)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| <i>Žəłnénəm</i> 'go after sea food'                                                      | $\vec{\lambda}$ $\vec{\lambda}$ $\vec{\lambda}$ $\vec{a}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ $\vec{n}$ |
| (cf. $\dot{\chi} \dot{e}^{\gamma} t$ 'fetch it,' - $l \dot{e} l \sim -l \dot{e} n$ 'food | l,' - <i>əm</i> 'intransitive')                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |

The following verb, evidently formed from the root of  $\vec{\lambda} e^{2t}$  'fetch him' and -  $x \partial n$  'foot,' seems to be irregular. My sources differed in its forms.

| $\vec{X} \dot{e} \cdot x \partial n \sim \vec{X} \partial^2 \dot{e} x \partial n$ 'invite' | <i>ŽiŽéxən</i> 'be inviting' (JP)                |
|--------------------------------------------------------------------------------------------|--------------------------------------------------|
| <i>Žéxən</i> 'invite'                                                                      | $\vec{X}i\vec{X}ex\partial n$ 'be inviting' (CC) |
| <i>Žéxəň</i> 'invite'                                                                      | <i>ẳiẳéxəň</i> 'be inviting' (AG)                |

There are a few possible TAh roots. Two have the shape TA when bare of suffixes.

| pá                                   | 'get hit by the wind' | pəpá                             | 'be getting hit by the wind' (AG) |
|--------------------------------------|-----------------------|----------------------------------|-----------------------------------|
| $q^{\scriptscriptstyle w} \acute{e}$ | 'get through'         | q <sup>w</sup> əq <sup>w</sup> é | ' 'be getting through'            |

With *-t* 'transitive,' a long vowel appears in the perfective and a glottal stop may appear in the progressive.

| <i>pá·t</i> 'blow on it'                | pəpá?ət 'be blowing on it'                                                                                              |
|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------|
| $x^w p \acute{a} t$ 'inflate it'        | $x^w p \partial p \dot{a}^2 \partial t$ 'be inflating it'                                                               |
| $q^{w} \acute{e} t$ 'make a hole in it' | $q^{w} \partial q^{w} \dot{e}^{\gamma} \partial t \sim q^{w} \partial q^{w} \dot{e}^{\cdot} t$ 'be making a hole in it' |

With stressed vowel-initial suffixes, these roots appear as Təh or Tə in the perfective:

From forms like *spahéls* 'wind' and  $q^{w}ah(lam)$  'pass through,' we might suppose that these roots are //pah// and  $//q^{w}eh//$  and that with *-t* 'transitive' we have:

```
pá·t //páhət// pəpá[?]ət //pəpáhət//
```

Another possible TAh root appears in

 $\vec{p}\vec{a}\cdot\vec{m}$  'swell, rise'  $\vec{p}\vec{a}\cdot\vec{m}$  'be swelling, be rising'

for which the resultative is *sp̀ahám* 'swollen, risen,' from which we might suppose the root to be //p̀ah//.

However, a problem with this interpretation is posed by the forms

 $x^{w} \acute{e}t$  'lower it, starve him'  $x^{w} \partial x^{w} \acute{e}t$  'be lowering it, be starving him'

in which the vowels are short. Yet in the derivative  $x^w \acute{e} l c a t$  'throw down a gift for him (from the scaffold in front of the house at a potlatch)' (< -lc- 'benefactive,' -t 'transitive') the vowel is long. And in the derivative  $x^w a h i l a ws$  'half starved' (< il 'move toward,' -aws 'body') there is an /h/. These derivatives would suggest a root //x we h//. But if the root has that form, then why is the vowel of the forms with -t 'transitive' short?

Another possible TAh root, with a differently formed progressive appears in

$$\check{x}\acute{e}$$
'm 'cry'  $\check{x}\acute{e}$ 'm 'be crying'

Cf.  $\check{x} \partial \check{x} \dot{e} \dot{m}$  'several are crying.' For similarly formed progressives, see TəR roots with  $-\partial m$  below. From a derivative  $\check{x} \partial h \dot{a} m \partial l \partial l c \partial$  (CC) (also  $\check{x} \dot{a} m \partial l \partial l c \partial$  [JP]) 'a giant frog that cries like a baby and bestows wealth' (suffix,  $-\delta l \partial l c \partial \sim -\dot{a} \cdot l c \partial$  'giant'), we may infer a root  $\sqrt{\check{x}} ah$ .

### 7.2.3.1. Progressives of TAHT Roots

There are a few roots consisting of an obstruent, a full vowel, /?/ or /h/, and another obstruent. These form their progressives like TAH roots, by CV reduplication with the stress on the second vowel and the first appearing as schwa.

| $l e^{2} \check{x}t$ 'put it on a dish, put s.t. under it' | $t \partial t \hat{e}^2 \check{x} t$ 'be putting it, etc.'                        |
|------------------------------------------------------------|-----------------------------------------------------------------------------------|
| $k^{w}i^{2}x^{w}t$ 'pitch it, seal it with pitch'          | $k^w \partial k^w i^2 x^w t$ 'be pitching it, etc.'                               |
| (cf. $k^w i^2 x^w$ 'pitch' [noun])                         |                                                                                   |
| $\check{x}^{w} \acute{a} \cdot \acute{t} t$ 'brace it'     | $\check{x}^{w} \partial \check{x}^{w} \acute{a} \cdot \acute{t}t$ 'be bracing it' |
| sí <sup>?</sup> ž <i>əm</i> 'duck, dodge'                  | səsí?xəm 'keep ducking'                                                           |

#### 7.2.4. Progressives of RAH Roots

Roots consisting of a resonant, a full vowel, and /?/ or (presumably) /h/ form their progressives by CV reduplication but with the initial resonant replaced by /h/, the first vowel weakened to schwa though (usually) stressed, the following resonant glottalized, and the second vowel variable in quality, stress, and whether followed by /?/ (RÁH  $\rightarrow$  hớrÂH ~ hớrð).

The pattern is seen in two bare roots of the shape RA?:

| $m e^{2}$ 'come off, run (as dye)'          | háma 'be coming off' (AG)                            |
|---------------------------------------------|------------------------------------------------------|
| <i>lé</i> <sup>2</sup> 'get hit just right' | $h \delta l \delta$ 'be getting hit just right' (AG) |

With -t 'transitive,' RA? roots (like TA? roots) appear without a linking schwa:

| lé <sup>?</sup> t' | make it secure' (CC) | hálət          | 'be making it secu | ıre'               |
|--------------------|----------------------|----------------|--------------------|--------------------|
| yá?t               | 'back it up'         | h <i>á ýət</i> | (JP) həỷá?t (DK)   | 'be backing it up' |

They show the same pattern with the transitivizer -x, as in:

| $m e^{2} x$ 'take it off' | $h \dot{\sigma} \dot{m} \dot{e}^2 x$ (CC) $h \dot{\sigma} \dot{m} \dot{\sigma} x$ (JP, AG) 'be taking it off' |
|---------------------------|---------------------------------------------------------------------------------------------------------------|
| $l e^{2} x$ 'put it away' | háláx (CC) hálax (JP, AG) 'be putting it away                                                                 |

The word  $n\acute{em}$  'go' appears, from its progressive form, to contain a root of this class; it is probably composed of  $n\acute{l}$  'be there' and  $-\partial m$  'intransitive.'

ném 'go' hánam 'be going'

Its causative form also supports this inference.

```
n\partial^2 \acute{e}m\partial st\partial x^w 'take it, recite it' h\acute{a}n\partial m\partial st\partial x^w 'be taking it, etc.'
```

With  $-n \partial x^{w}$  'limited control,' a RA? root behaves like a TA? root,  $-n \partial x^{w}$  appearing as  $-n x^{w}$ :

*lénx<sup>w</sup>* 'manage to put it away' (no progressive recorded).

With -éls, it is reduced to Rə?-, as in:

 $l\partial^2 \dot{e}ls$  'put s.t. away'  $h\partial \dot{l}\dot{e}ls$  'be putting away'

('Put something away' is used to mean to give something as an investment, expecting a return.)

With an unstressed vowel-initial suffix, a RA? root retains its stress but evidently may undergo a shift in vowel while the glottal stop is lost and the vowel is lengthened:

 $mi \cdot \partial q^w t$  'cut his head off' $h \partial mi \cdot \partial q^w t$  'be cutting his head off' (CC) $me \partial q \partial t$  'cut his head off' (JP)(no progressive recorded)(cf.  $me^2 x$  'remove it,'  $-q^w$  'head,' -t 'transitive')

There is one possible RAh root (so far discovered), seen in the following forms with -t 'transitive' and  $-\partial m$  'intransitive.'

 $y \dot{a} \cdot t$  'warn him, order from him' $h \partial \dot{y} \dot{a} \cdot t$  'be warning him, etc.' $y \dot{a} \cdot m$  'order (as a sweater from a knitter)' $h \partial \dot{y} \dot{a} \cdot \vec{m}$  'be ordering'

These suggest a root of the shape //yah//, meaning something like 'alert,' with -t 'alert him' and with  $-\partial m$  'alert another for one's own sake.'

# 7.2.4.1. Progressives of RA<sup>2</sup>CC and RA<sup>2</sup> Roots

There are a few words that either have these shapes or are composed of RA? roots with unidentifiable suffixes. They form their progressives like RA? roots.

| $m e^{2k} k^{w} l$ 'get hurt'                        | hámak <sup>w</sup> ł (AG), hamé <sup>2</sup> k <sup>w</sup> ł (CC) 'be getting hurt' |
|------------------------------------------------------|--------------------------------------------------------------------------------------|
| <i>yé<sup>9</sup>ət</i> [yé <sup>9</sup> et] 'vomit' | háýət (JP), həýé <sup>2</sup> t (DK) 'be vomiting'                                   |
| <i>yé<sup>9</sup>ətt</i> 'vomit on him'              | háýatt 'be vomiting on him' (AG)                                                     |

# 7.2.5. Progressives of HAH Roots

The only example so far discovered of a root that seems to be of this shape appears with -t and  $-\partial m$  and has a long vowel in the perfective and a glottal stop followed by schwa in the progressive.

| ?á·t 'call him'         | á?ət 'be calling him |
|-------------------------|----------------------|
| <sup>?</sup> á·m 'call' | ?á·m 'be calling'    |

# 7.2.6. Progressives of T<sub>∂</sub>T Roots

Bare roots of the shape obstruent-schwa-obstruent form their progressives by CV reduplication with the stress on the second vowel  $(T_1 
abla T_2 \rightarrow T_1 
abla T_1 
bla T_2)$ , as in:

| $p \delta q^w$ 'go broke'                                  | <i>pəpáq<sup>w</sup></i> 'be going broke'                              |
|------------------------------------------------------------|------------------------------------------------------------------------|
| $\dot{p}\dot{\delta t}$ 'come to, sober up'                | $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{a}$ $\dot{b}$ $\dot{b}$ coming to' |
| tás 'arrive there'                                         | tətás 'be arriving there'                                              |
| $\dot{t}\dot{a}k^{w}$ 'get stuck in mud'                   | $t \partial t \partial k^w$ 'be getting stuck in mud'                  |
| <i>ťáq</i> <sup>w</sup> 'break'                            | $i \partial i \partial q^w$ 'be breaking'                              |
| $\theta \delta \vec{k}^{w}$ 'tighten up, get straight'     | $\theta \partial \theta \delta \vec{k}^{w}$ 'be tightening up, etc.'   |
| $\dot{t}^{\theta} \dot{\delta} \check{x}^{w}$ 'get washed' | $t^{\theta} \partial t^{\theta} \partial x^{w}$ 'be getting washed'    |
| $s \dot{a} \dot{q}$ 'get torn, get split'                  | $s \partial s \partial \dot{q}$ 'be getting torn, etc.'                |

| $\dot{c}\dot{a}\dot{q}^{w}$ 'get pierced'           | $\dot{c}\partial\dot{c}\dot{d}^{w}$ 'be getting pierced'         |
|-----------------------------------------------------|------------------------------------------------------------------|
| $q^{w} \delta s$ 'go under water'                   | $q^w \partial q^w \delta s$ 'be going under water'               |
| $q^{w} \delta t$ 'wash ashore'                      | $q^{w} \partial q^{w} \partial t$ 'be washing ashore'            |
| $\vec{k}^{w} \delta \vec{t}$ 'spill over, tip over' | $\vec{k}^{w} \partial \vec{k}^{w} \partial t$ 'be spilling over' |

With *-t* 'transitive,' T<sub>2</sub>T roots appear as TT followed by either a full vowel, */e/* or */a/*, or a schwa in the perfective, and their progressives are formed by inserting a full vowel, */e/* or */a/* between the two consonants. We may thus distinguish two subclasses:

(1) With the formula  $TT\dot{A} \rightarrow T\dot{A}T$ , as in:

| <i>sģét</i> 'split it'                        | <i>séqt</i> 'be splitting it'                              |
|-----------------------------------------------|------------------------------------------------------------|
| <i>k<sup>w</sup>łét</i> 'spill it'            | <i>k<sup>w</sup>éłt</i> 'be spilling it'                   |
| <i>,<br/>tk<sup>w</sup>át</i> 'caulk it'      | <i>ťák<sup>w</sup>t</i> 'be caulking it'                   |
| <i>id<sup>w</sup>át</i> 'cut it off'          | $i \dot{t} \dot{a} \dot{q}^{w} t$ 'be cutting it off'      |
| <i>ť<sup>ē</sup>ž<sup>w</sup>át</i> 'wash it' | $\dot{t}^{\theta} \dot{a} \check{x}^{*} t$ 'be washing it' |

The vowels here seem predictable; if the second consonant is rounded, the vowel is /a/; if not, it is /e/.

(2) With the formula TT $\rightarrow T AT -$ , as in:

| $\dot{p}t\delta t$ 'bring him to'            | <i>pélt</i> 'be bringing him to'                          |
|----------------------------------------------|-----------------------------------------------------------|
| tsát 'approach it'                           | tést 'be approaching it'                                  |
| $\theta \vec{k}^{w} \delta t$ 'stretch it'   | $\theta \acute{e} \vec{k}^{w} t$ 'be stretching it'       |
| $\theta \dot{q} \delta t$ 'spear it'         | $\theta \dot{e} \dot{q} t$ 'be spearing it'               |
| $\theta \check{x} \acute{a} t$ 'push him'    | $\theta \dot{e} \dot{x} t$ 'be pushing it'                |
| $\vec{X}\vec{q}^{w}\delta t$ 'get him ready' | $\dot{\chi} \dot{a} \dot{q}^{w} t$ 'be getting him ready' |
| $\dot{c}\dot{q}^{w}\delta t$ 'pierce it'     | $\dot{c}\dot{a}\dot{q}^{w}t$ 'be piercing it'             |

In this set, it appears that only if the second consonant is a rounded uvular, the vowel in the progressive is /a/.

*csét* 'tell him to do so' *cásət* 'be telling him to do so'

The pattern is like that of some of the roots with -m 'intransitive' given below.

There are a few presumptive T 
area T roots that I have recorded with *-t* 'transitive' but never bare. These include:

*iqát* 'insult him (by reference to ancestry)' *iéqt* 'be insulting him ...' *iqét* 'peg/nail it together temporarily' *iéqt* 'be pegging it ...'

(I have recorded no  $*\dot{t}\dot{a}q$  or  $*t\dot{a}q$ .)

With  $-\partial m$  'intransitive,' T $\partial$ T roots fall into five subclasses, four with the stress falling on the suffix in the perfective with variations in the stressed vowel (and so resembling the subclasses with -t 'transitive') and a fifth formed by reduplication (and so resembling the progressive of the bare root).

(1) With a full vowel in both perfective and progressive (TTÁ-  $\rightarrow$  TÁT-), as in:

 $\vec{k}^w x \acute{e}m$  'count'  $\vec{k}^w \acute{e} x image m$  'be counting' (cf.  $\vec{k}^w x \acute{e}t$  'count them,'  $\vec{k}^w \acute{e} xt$  'be counting them')

(2) With a full vowel in the perfective and a schwa in the progressive (TTÁ- $\rightarrow$  TáTə-), as in:

| $\theta x^{w} \acute{a} m$ 'bleed'                      | $\theta \delta x^{w} \delta m$ 'be bleeding' |
|---------------------------------------------------------|----------------------------------------------|
| (cf. $\theta \delta x^w$ 'fade away')                   |                                              |
| <i>čtém</i> 'creep'                                     | <i>čátəm</i> 'be creeping'                   |
| (no $*\dot{c}\delta t$ or $*\dot{c}t\dot{e}t$ recorded) |                                              |
| <i>xtém</i> 'swim underwater'                           | xátam' 'be swimming underwater'              |
| (no $*x \delta t$ or $*x t \delta t$ recorded)          |                                              |

(3) With a schwa in the perfective and full vowel in the progressive (TT $3 \rightarrow TATa$ -), as in:

| (he duinnine) |
|---------------|
| be dripping   |
|               |

(4) With a schwa in both perfective and progressive (TT $5 \rightarrow T\delta T \overline{P}$ -), as in:

| $pk^w \delta m$ 'splash, billow out'                                       | pákwam 'be splashing, etc.'                                  |
|----------------------------------------------------------------------------|--------------------------------------------------------------|
| (cf. $p\vec{k}^w \delta t$ 'scatter it,' $p\vec{e}\vec{k}^w t$ 'be scatter | ering it')                                                   |
| $q \lambda i m$ 'come loose'                                               | $q \delta \tilde{\lambda} \partial m$ 'be coming loose' (AG) |
| (no $*q\delta\lambda$ recorded)                                            |                                                              |

(5) With reduplication, i.e.,  $T \Rightarrow T \Rightarrow T \Rightarrow T \Rightarrow T \Rightarrow T \Rightarrow T \Rightarrow$ 

*tóťam* 'rake herring' *tołóťam* 'be raking herring' (cf. *tŕtt* 'hit it with a swiping motion,' *téťt* 'be hitting it away')

With  $-n\partial x^w$  'limited control,' a T $\partial$ T root appears in the perfective as unstressed T $\partial$ T- while the suffix is stressed as  $-n ex^w$ . However, in the progressive the root appears stressed as T $\Delta$ T and the suffix unstressed as  $-n\partial x^w$ , as in:

| $t \rightarrow sn e^{x^w}$ 'manage to get close to it'                  | $t \acute{e} s n \partial x^w$ 'be managing to, etc.'         |
|-------------------------------------------------------------------------|---------------------------------------------------------------|
| (cf. <i>tsát</i> 'approach it'                                          | <i>tést</i> 'be approaching it')                              |
| $\theta \partial \check{x} n \acute{e} x^w$ 'accidentally push him'     | $\theta \acute{e} \check{x} n \partial x^w$ 'be accidentally' |
| (cf. $\theta \check{x} \delta t$ 'push him away'                        | $\theta \acute{e} \check{x}$ 'be pushing him away')           |
| $\dot{c}\partial \dot{q}^w n \dot{e} x^w$ 'manage to hit it (shooting)' | $\dot{c}\dot{a}\dot{q}^w n \partial x^w$ 'be managing to'     |
| (cf. $\dot{c}\dot{q}^{w}\dot{a}t$ 'pierce it, shoot it'                 | $\dot{c}\dot{a}\dot{q}^{w}t$ 'be piercing it, etc.')          |
| $i \partial q^w n e x^w$ 'manage to cut it off'                         | $i \acute{a} \dot{q}^w n \partial x^w$ 'be managing'          |
| (cf. $i \dot{q}^w \dot{a} t$ 'cut it off, cut it loose'                 | $i \dot{a} \dot{q}^{w} t$ 'be cutting it off, etc.')          |

The verb  $i \partial q^w n ex^w$  also means 'catch him (at s.t.)' or 'catch it (an event).'

With  $-\acute{els}$ , a T $\Rightarrow$ T root appears as it does with -t, as TT- in the perfective and (usually) with a full vowel in the progressive, as in:

| $\dot{X}k^{w} \acute{e}ls$ 'pinch'                                 | <i>Žák<sup>w</sup>əls</i> 'be pinching'                                  |
|--------------------------------------------------------------------|--------------------------------------------------------------------------|
| (cf. $\vec{\lambda}k^{w} \acute{a}t$ 'grab it between the fingers' | $\dot{\lambda} \dot{a} k^{w} t$ 'be grabbing it')                        |
| $\dot{q}p\dot{e}ls$ 'take up a collection'                         | $\dot{q}\dot{e}p\dot{s}$ 'be taking up a collection'                     |
| (cf. $\dot{q}p\delta t$ 'put it together'                          | $\dot{q}\dot{e}pt$ 'be putting it together')                             |
| <i>k</i> <sup>w</sup> <i>téls</i> 'pour'                           | $k^{w} \epsilon t \delta l s$ 'be pouring' (AG)                          |
|                                                                    | (but cf. $\vec{k}^{w} \delta t \delta \vec{l} s$ 'be flowing fast' [JP]) |
| (cf. <i>k<sup>w</sup>tét</i> 'spill it'                            | $k^{w} \epsilon t t$ 'be spilling it')                                   |

With lexical suffixes, a TablaT root always appears unstressed in the perfective, as TT- when followed by a suffix beginning with a vowel and as TablaT- when followed by one beginning with a consonant. (Because the suffix always takes the stress, we find that TablaT roots take the stressed forms of suffixes that appear unstressed when following roots with full vowels.) In the progressive, the stress usually shifts to the root, which then appears as TablaT-, but in some words the suffix keeps the stress and the root appears as TablaT- (as in 'wash one's feet' below).

```
t^{\theta} \check{x}^{w} \acute{a} s \partial m 'wash one's face'
 t^{\theta} \delta \tilde{x}^{w} \delta s \delta m 'be washing one's face'
(< -\dot{a}s \sim -\partial s 'face,' \dot{t}^{\theta}\dot{\delta}\dot{x}^{w} 'get washed,' -\partial m 'intransitive,' cf. \dot{t}^{\theta}\dot{x}^{w}\dot{a}t 'wash it,'
 t^{\theta} \dot{a} \dot{x}^{w} t 'be washing it')
t^{\theta} \check{x}^{w} \acute{e} cs \partial m 'wash one's hands'
 t^{\theta} \delta \tilde{x}^{w} \delta c s \delta m 'be washing one's hands'
(< -\acute{e}c\partial s \sim c\partial s 'hand')
t^{\theta} \partial x^{w} x \ell n \partial m 'wash one's feet'
 t^{\theta} \partial x^{w} x \dot{e} \dot{n} \partial m 'be washing ...' (AG)
(< -x \acute{e}n \sim -x in `foot')
t^{\theta} \check{x}^{w} \acute{a} l w \partial t \partial m 'wash clothes'
 t^{\theta} \delta \tilde{x}^{w} \delta l w \delta t \delta m 'be washing clothes'
(< -\delta l w \partial t 'clothing')
sģálcap 'split firewood'
 ságalcap 'be splitting firewood'
(< -álcəp 'fire, firewood,' sáq 'get split'; cf. sqét 'split it,' séqt 'be splitting it')
tk^wéwət 'caulk a canoe'
 i \delta k^{w} \partial w \partial t 'be caulking a canoe'
(< -\acute{e}w\partial t \sim -w\partial t 'vessel,' \dot{t}\partial k^{w} 'get stuck in mud'; cf. \dot{t}k^{w}\dot{a}t 'caulk it,' \dot{t}\dot{a}k^{w}t 'be
 caulking it')
```

There are a few cases recorded where the root of the progressive has the full vowel that appears in the progressive with *-t*.

 $x^{w} l aqn \acute{e} cam `tow' x^{w} l \acute{e} qn a cam `be towing' (JP)$ (< -néc `butt, base, rump'; cf. lqét `peg it together,' léqt `be pegging it together')
t<sup>2</sup> x<sup>w</sup> é<sup>2</sup> t' wash diapers' (CC, AG) t<sup>2</sup> d x<sup>w</sup> a<sup>2</sup> t (CC), t<sup>2</sup> d x<sup>w</sup> a<sup>2</sup> t (AG) `be
washing diapers'

(< -é?ł 'bedding')

In the last example, the  $-a^{2}t$  in CC's progressive may be a case of vowel harmony.

#### 7.2.6.1. Progressives of TaTaC Roots

The few roots of this shape form their progressive as T<sub>∂</sub>T roots do, by CV reduplication with the stress on the second schwa.

| <i>ťá pəl</i> 'play cards'                                                        | <i>tətá pəl '</i> be playing cards'                                               |
|-----------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|
| $\dot{q}\dot{a}\dot{t}^{\theta}\dot{a}\dot{x}$ 'defecate'                         | $\dot{q}_{\rho}\dot{q}\dot{\delta}\dot{t}^{\theta}_{\rho}\dot{x}$ 'be defecating' |
| $\vec{X} \neq \vec{k}^w \neq \vec{n}$ 'go out (as a fire)'                        | <i>ŹəŹákwən 'be</i> going out'                                                    |
| $\vec{X} \delta \vec{k}^{w} \delta \vec{n}t$ 'extinguish it (a fire or a hatred)' | <i>ŽəŽákwant</i> 'be extinguishing it'                                            |

AG gave these last four forms with  $\dot{n}$ ; I recorded *n* from JP.

The following seems to belong here, again suggesting that the full vowel followed by the glottal stop is an underlying schwa.

 $s \neq \tilde{x}^w a^2$  'urinate'  $s \Rightarrow s \neq \tilde{x}^w a^2$  'be urinating'

### 7.2.6.2. Progressives of TTVC Roots

| $itaq^w$ 'get twisted'                           | $i \delta t \partial q^w$ 'be getting twisted' |
|--------------------------------------------------|------------------------------------------------|
| $ctáq^w$ 'go through'                            | $c \delta t \partial q^w$ 'be going through'   |
| $\vec{\lambda}qit$ 'give birth'                  | <i>ẳáqəł '</i> be giving birth'                |
| $\check{x}\check{t}\check{e}\check{k}^w$ 'carve' | <i>žáťak</i> <sup>w</sup> 'be carving'         |

With -*t* these follow the same pattern.

| <i>ťłáq<sup>w</sup>t</i> 'pry it off'     | <i>ťáłaq</i> <sup>w</sup> t 'be prying it off'      |
|-------------------------------------------|-----------------------------------------------------|
| <i>cłáq<sup>w</sup>t</i> 'put it through' | $c \delta i \partial q^w t$ 'be putting it through' |

There is at least one possible root of the shape CC<sub>2</sub>C:

 $k^{w}t \delta x^{w}$  'enter' (no progressive recorded).

With -t, however, a full vowel appears in the progressive:

 $k^{w}t \acute{e}x^{w}t$  'bring/take it in'  $k^{w} \acute{o}t \partial x^{w}t$  'be bringing/taking it in'

There are other words, recorded only with transitivizers, that follow this pattern. From the examples just given, it appears that the bare roots of these may be either CCAC or CC<sub>2</sub>C in shape.

| <i>tpáłt</i> 'stretch it on a frame (as a hide)'     | <i>tá pəłt</i> 'be stretching it'                       |
|------------------------------------------------------|---------------------------------------------------------|
| $\theta q \dot{a} t t$ 'honour him by giving wealth' | $\theta \delta q \partial t$ 'be honouring him'         |
| $\theta q e n x$ 'stand it up in the ground'         | $\theta \delta q \partial n x$ 'be standing it up' (AG) |

Some or all of these may possibly be T<sub>P</sub>T roots with petrified suffixes.

#### 7.2.7. Progressives of T<sub>∂</sub>R Roots

Roots of this shape form their progressives by CV reduplication with the stress on the second schwa (as T $\Rightarrow$ T roots do) and with the resonant glottalized (T $\Rightarrow$ R)  $\Rightarrow$  T $\Rightarrow$ T $\Rightarrow$ T $\Rightarrow$ R, as in:

| pán 'get buried'                    | <i>pəpáň</i> 'be getting buried'                                    |
|-------------------------------------|---------------------------------------------------------------------|
| $\dot{q}^{w} \delta l$ 'get cooked' | $\dot{q}^{w}\partial\dot{q}^{w}\partial\dot{l}$ 'be getting cooked' |
| $k^{w} \delta n$ 'become possessed' | $k^{w} \partial k^{w} \partial n$ 'be going into a possessed state' |
| $x^{w} \delta y$ 'wake up'          | $x^{w} \partial x^{w} \partial y'$ 'be waking up'                   |

With -t 'transitive,' T<sub>2</sub>R roots fall into two subclasses, with nearly all of them in the first subclass.

(1) With their progressives formed by reduplication following the pattern of the bare roots (and unlike T $\Rightarrow$ T roots, which do not reduplicate with *-t*, as in:

| <i>pánət</i> 'bury it'                                | pəpánat 'be burying it'                                                      |
|-------------------------------------------------------|------------------------------------------------------------------------------|
| $\dot{q}^{w} \delta l \delta t$ 'roast it'            | <i>q<sup>w</sup>əq<sup>w</sup>ələt</i> 'be roasting it'                      |
| <i>k<sup>w</sup>ánət</i> 'take it'                    | <i>k<sup>w</sup>ək<sup>w</sup>ə́nət</i> 'be taking it'                       |
| <i>łəmət</i> 'pick it/them'                           | <i>tətámat</i> 'be picking it/them'                                          |
| $\dot{q}^{w} \dot{\sigma} m \partial t$ 'pull it out' | $\vec{q}^{w} \partial \vec{q}^{w} \partial m \partial t$ 'be pulling it out' |
| $(\langle \dot{q}^{w} \delta m$ 'shed, moult')        |                                                                              |

(2) With their progressives formed by simply glottalizing the resonant, without reduplication.

| <i>łálət</i> 'bail it (as a canoe)' | <i>lálat</i> 'be bailing it' |
|-------------------------------------|------------------------------|
| qówət 'warm it'                     | qáwat 'be warming it' (AG)   |

With *-əm* 'intransitive,' T*>*R roots form their progressives in three ways: (1) by reduplication (with the usual glottalization of resonants)

| <i>páňəm</i> 'plant'         | pəpánam 'be planting'            |
|------------------------------|----------------------------------|
| <i>iśyəm</i> 'stick, adhere' | <i>ť ať á y am</i> 'be sticking' |

(2) by strengthening the vowel (with the usual glottalization)

| čánəm                  | 'shake (as a Shaker)'     | ćéńəṁ | 'be shaking' |
|------------------------|---------------------------|-------|--------------|
| <i>ģ</i> źw <i>ə</i> m | 'howl (as a dog or wolf)' | ģéwəm | 'be howling' |

(3) by simply glottalizing the final resonant

 $\vec{q}^{w}\delta l \delta m$  'bake'  $\vec{q}^{w}\delta l \delta m$  'be baking' (AG)

There are also at least two T $\partial$ R roots that, in the perfective, combine with  $\partial m$  with a loss of the resonant and the appearance of a long vowel /i·/. Their progressives are formed in the way just illustrated, by simply glottalizing the final /m/.

| <i>łí·m</i> //łśm-əm// 'pick' | <i>łí·m</i> //łóm-əm// 'be picking' |
|-------------------------------|-------------------------------------|
| (cf. <i>łámat</i> 'pick them' | <i>łəłámat</i> 'be picking them')   |

 $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  /// $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot m$  // $t\hat{q}\cdot$ 

With  $-n \partial x^w$  'limited control,' a T $\partial$ R root is stressed in the perfective and the suffix is unstressed (unlike T $\partial$ T roots, which appear unstressed and with  $-n \dot{e} x^w$ ). The progressive (in the one example available) is formed by strengthening the vowel and glottalizing the resonant.

 $k^{w} \dot{a} n n \partial x^{w}$  (get it'  $k^{w} \dot{a} \dot{n} n \partial x^{w}$  (JP),  $k^{w} \dot{e} \dot{n} n \partial x^{w}$  (AG) 'be getting it'

From the few examples available, it seems that with  $-\acute{els}$  TəR roots behave variably.

| <i>taméls</i> 'chop'               | <i>t atá mals</i> 'be chopping'                                                |
|------------------------------------|--------------------------------------------------------------------------------|
| (cf. ťámət 'chop it'               | <i>tətámət</i> 'be chopping it')                                               |
| <i>xəléls</i> 'write'              | <i>x̄əxéyəls</i> (CC), <i>x̄əxíls</i> [x̄əxí·ls?] (JP) 'be writing'            |
| (cf. <i>žálət</i> 'paint a design' | <i>x̄əxíələt</i> 'be painting a design')                                       |
| $k^{w}i$ · $ls$ 'grab onto s.t.'   | $k^{w} \partial k^{w} i \cdot ls \sim k^{w} i \cdot ls$ [?] (AG) 'be grabbing' |

With lexical suffixes, T $\partial$ R roots appear in the perfective stressed with unstressed suffixes and unstressed with stressed suffixes, but it is unclear when a suffix will be stressed or unstressed; cf. forms with  $-i\dot{w}s \sim -\partial\dot{w}s$  'body, bird' below. The progressives are formed in ways seen with -t and  $-\partial m$ :

(1) by reduplication

pánəst 'throw dirt in his face' pəpánəst 'be throwing ...' (AG)
(cf. pánət 'bury it,' pəpánət 'be burying it,' -əs 'face')
táləst 'splash water in his face' tətáləst 'be splashing ...' (JP)
(cf. tálət 'bail it,' tálət 'be bailing it,' -əs 'face')

(2) by glottalizing the resonant

 $\vec{q}^{w}\delta m\partial \vec{w}s$  'pluck a bird'  $\vec{q}^{w}\delta m\partial \vec{w}s$  'be plucking a bird' (JP) (cf.  $\vec{q}^{w}\delta m\partial t$  'pull it out,'  $\vec{q}^{w}\partial \vec{q}^{w}\delta m\partial t$  'be pulling it out,'  $-i\vec{w}s \sim -\partial \vec{w}s$  'body, bird')

(1) and (2) both ways

*təmí wst* 'chop its trunk' *tətáməwst* ~ *táməwst* 'be chopping ...' (AG) (cf. *támət* 'chop it' *tətámət* 'be chopping it') *təmí wst* 'knock its branches off' *tətáməwst* - *táməwst* 'be knocking ...' (cf. *támət* 'pick them' *tətámət* 'be picking them')

(3) by strengthening the vowel and glottalizing the resonant of the root

 $\check{x}$  "anxénam 'run' $\check{x}$  "anxanam' 'be running'(possibly  $\check{x}$ "ám 'fast,' -xén ~ -xan 'foot,' -am 'intr.,' i.e., "fast-foot it") $iawi^{\theta} e^{2}am$  'disrobe' $ie\dot{w}i^{\theta}e^{\cdot}m$  'be disrobing'(<  $ia\dot{w}it^{\theta}a^{2}$  'naked')

This last is composed of the suffix  $-t\hat{t}^{\theta}e^{\gamma}$  'robe' and a root possibly identifiable as  $t\hat{e}w$  'escape, be cured,' and  $-\partial m$  'intransitive.' However, a CAC root like  $t\hat{e}w$  ought to be reduplicated in the progressive, so if this root is indeed  $t\hat{e}w$ , it is being treated here like a T $\partial R$  root.

There is one exceptional T $\partial$ R root,  $\partial \partial y$  'come into existence, get made,' which with *-t*, *-\partial m*, and various lexical suffixes, behaves like a C $\partial$ CT root (see §7.2.10), strengthening the vowel and glottalizing the resonant, as in:

 $\theta \delta yt$  'make it, fix it' $\theta \delta yt$  'be making it, be fixing it' $\theta \delta yam$  'make bread' $\theta \delta yam$  'be making bread' $\theta \delta yacan$  'rewrap the dead' $\theta \delta yacan$  'be rewrapping for reburial' $(< -\delta can \sim -i can$  'surface,' -am 'intr.')

# 7.2.7.1. Progressives of T a R a C Roots

The few roots of this shape form their progressives, as  $T \Im R$  roots do, in two ways:

(1) by CV reduplication with the stress on the second schwa, as in:

 $x^{w} \delta l \delta k^{w} t$  'wrap it up'  $x^{w} \delta x^{w} \delta l \delta k^{w} t$  'be wrapping it up'

(2) by glottalizing the medial resonant, as in:

*łáməx*<sup>w</sup> 'rain'

*łómox*<sup>w</sup> 'be raining'

(This may be composed of a TəR root and a suffix; cf. *stámłam* 'dew,' *stém* 'drenched with rain'.)

| <i>čáləx</i> <sup>w</sup> 'go/come upriver' | $\dot{c}\dot{\delta}\dot{l}\partial x^{w}$ 'be going/coming upriver' (AG)       |
|---------------------------------------------|---------------------------------------------------------------------------------|
| <i>ģáwətt</i> 'drum for him'                | <i>q́áwðtt</i> 'be drumming for him'                                            |
| (cf. <i>q́áwət</i> 'drum [noun]')           |                                                                                 |
| <i>táməxt</i> 'braid it'                    | <i>támaxt</i> 'be braiding it'                                                  |
| (but cf. <i>tomxoné?t</i> 'braid her hair   | r,' $timx \partial n e^{2t}$ 'be braiding her hair,' < $-\partial n e^{2} \sim$ |
| -ənə 'ear')                                 |                                                                                 |
| <i>ģáwətəm</i> 'beat a drum'                | $\dot{q}\dot{a}\dot{w}atam$ 'be beating a drum'                                 |

# 7.2.8. Progressives of R<sub>∂</sub>C Roots

Roots of the shapes R $\sigma$ T and R $\sigma$ R are like RAH roots (see §7.2.4) in forming their progressives by a variety of CV reduplication in which the initial resonant is replaced by /h/.

Bare roots of the shape R<sub>2</sub>T follow the formula R<sub>2</sub>T  $\rightarrow$  h<sub>2</sub>R<sup>T</sup>, as in:

| $m \dot{q}$ 'get full, eat one's fill' | hámq 'be getting full'                                  |
|----------------------------------------|---------------------------------------------------------|
| $n \delta q^w$ 'fall asleep'           | $h \dot{a} \dot{n} q^w$ 'be falling asleep'             |
| lás 'cave in, slide (as land)'         | háls 'be falling in, etc.'                              |
| <i>lák</i> <sup>w</sup> 'break'        | hálk <sup>w</sup> 'be breaking'                         |
| $y \delta q^w$ 'burn'                  | $h \delta \dot{y} q^w$ 'be burning, fire'               |
| $w \dot{q}^{w}$ 'drift away'           | $h \dot{\sigma} \dot{w} \dot{q}^{w}$ 'be drifting away' |

Bare roots of the shape R<sub>2</sub>R follow the formula R<sub>2</sub>R  $\rightarrow$  h<sub>2</sub>R<sub>2</sub>R, as in:

*lám* 'be folded, be doubled over' *hálam* 'be getting folded' (AG)

As they appear with *-t* 'transitive,' R $\Rightarrow$ C roots fall into four subclasses, varying in the placement of the stress in the perfective and in the presence of a schwa or full vowel after the two consonants of the root (cf. the subclasses of T $\Rightarrow$ T roots in §7.2.6):

(1) with the stress on the schwa of the root and with a schwa following (R $\Rightarrow$ C $\Rightarrow$ - $\rightarrow$ h $\Rightarrow$ RC $\Rightarrow$ -), as in:

| <i>málat</i> 'dip it ceremoniously'    | hámlət 'be dipping it'                                      |
|----------------------------------------|-------------------------------------------------------------|
| <i>máqat</i> 'swallow it'              | hámqat 'be swallowing it'                                   |
| <i>lácðt</i> 'fill it'                 | hálcat 'be filling it'                                      |
| $y \neq \tilde{x}^w \neq t$ 'untie it' | $h \dot{a} \dot{y} \ddot{x}^{w} \partial t$ 'be untying it' |

Two roots that take transitivizer /-x/ seem to belong here.

| $n \delta \dot{p} \delta x$ 'eat it' | <i>hán pàx</i> 'be eating it' |
|--------------------------------------|-------------------------------|
| <i>nówəx</i> 'insert it'             | háňwax 'be inserting it'      |

(2) with the stress on the schwa of the root and no schwa following (R $\Rightarrow$ C-).

| yáq <sup>w</sup> t | 'burn it' | háỷq <sup>w</sup> t | 'be burning it |
|--------------------|-----------|---------------------|----------------|
| * *                |           |                     | 0              |

(3) with the stress on a schwa following the root in the perfective but with no schwa following the root in the progressive (R $\partial$ C $\rightarrow$   $\rightarrow$   $h\partial$  RC-), as in:

| <i>mətát</i> 'bend it'                               | hámít 'be bending it'                                    |
|------------------------------------------------------|----------------------------------------------------------|
| $l \partial \check{x} \partial t$ 'enlarge the mesh' | $h \neq l \tilde{x}t$ 'be enlarging the mesh (of a net)' |
| <i>məsát</i> 'fold it up'                            | <i>hámst</i> 'be folding it up'                          |

(4) with a stressed full vowel following the root in the perfective and a schwa there in the progressive (R $\partial$ CÁ  $\rightarrow$  H $\partial$ RC $\partial$ ). I have found only one example:

```
lək^wát 'break it'
```

hálkwat 'be breaking it'

With  $-\partial m$  'intransitive,' R $\partial$ C roots may fall into the subclasses seen with -t (cf. T $\partial$ T roots, in §7.2.6), but only one example has been recorded. It would belong to the first subclass.

```
náqam 'dive'
```

hángam' 'be diving'

With lexical suffixes, R<sub>∂</sub>T roots form their progressives with the formula seen in the bare root but with the stress varying with the suffix, as in:

 $m \delta k^{w} \partial l c \partial p$  'gather firewood'  $h \partial m k^{w} \delta l c \partial p$  'be gathering ...' (cf.  $m \delta k^{w} \partial t$  'find it, pick it up,'  $-\delta l c \partial p$  'fire, fuel')  $m \partial t \dot{e} q \partial t$  'bend it down (as a branch)' $h \partial m \dot{t} \partial q \partial t$  'be bending ...'(cf.  $m \partial t \dot{d} t$  'bend it,'  $h \partial m \dot{t} t$  'be bending it,'  $-q \partial n$  [-éqən ?] 'head, end of long object') $y \dot{d} \partial \partial s t$  'tell him' $h \partial y \partial \partial s t$  'be telling him'(cf.  $s y \partial \theta$  'tradition, heritage,'  $s y \partial \partial s s$  'tradition, teaching,'  $-\partial s$  'recipient') $w \partial q^w i l \partial m$  'move downstream' $h \partial w \dot{q}^w \partial l \partial m$  'be moving downstream'(cf.  $w \partial q^w$  'drift with the current,'  $h \partial w d^w$  'be drifting,' -il 'move toward,'  $-\partial m$  'intransitive')

There are a few exceptional R $\partial$ R roots that, with *-t* and *-\partial m*, form their progressives like C $\partial$ CT and a few T $\partial$ R roots, by strengthening the vowel and glottalizing the resonant:

| yáwt 'praise him' | yáŵθət 'be bragging' |
|-------------------|----------------------|
| náyəm 'laugh'     | néý am 'be laughing' |

#### 7.2.9. Progressives of H<sub>∂</sub>C Roots

There is no recorded example of a root of the shape  $h \Rightarrow C$  and only one of the shape  $^{9}\Rightarrow C$ . It forms its progressive by what may be identified as reduplication with the stress on the first vowel, which is strengthened to a full vowel.

 $2\delta q^w$  'lose one's hair'  $2\delta^2 a q^w$  'be losing one's hair' (JP)

It has at least one derivative:

 $^{2}\delta q^{w}\theta \partial t$  'shed hair'  $^{2}\delta \cdot q^{w}\theta \partial t$  'be shedding hair' (JP)

(AG was not familiar with the word  ${}^{2}\delta q^{w}$  but suggested that the progressive ought to be  ${}^{2}\delta q^{w}$ . The next set of words would also suggest this.)

#### 7.2.9.1. Progressives of H arrow C(C)V(C) Roots

There are a few words that appear to be roots of these shapes, with initial glottal stop. They form their progressives by strengthening the schwa to a full vowel, not now predictable, followed by a glottal stop or glottalization of a resonant.

| <i>?áxəl</i> 'paddle'        | $2\hat{i}^2x\hat{j}$ 'be paddling'                         |
|------------------------------|------------------------------------------------------------|
| ?ámət 'sit down, sit up'     | <sup>2</sup> ámat 'be sitting down, be sitting up'         |
| <i>?álya</i> 'have a vision' | <i>?ályə</i> 'be having a vision'                          |
| ?áłtən 'eat'                 | <sup>?</sup> <i>i</i> <sup>?</sup> <i>ttən</i> 'be eating' |

This last word looks as though it might be composed of an HAC root \*?*il* reduced to Hác- with a CVC suffix (*-tan* 'instrument'?).

There is one word with an initial /h/ that may belong here:

| háye <sup>2</sup> (AG) 'go away, leave' | háýa 'be going away, etc.' (AG) |
|-----------------------------------------|---------------------------------|
| háye <sup>?</sup> (JP) 'go away, leave' |                                 |

(As recorded from JP, this should belong in another set. Unfortunately, I did not record a progressive from JP.)

#### 7.2.10. Progressives of C<sub>2</sub>CT Roots

There are a number of roots (or apparent roots) of this shape, with the first consonant of any class (including glottals), the second either an obstruent or a resonant (and, if a resonant, either unglottalized or glottalized), and the third an obstruent. (Further analysis may reveal that some or all of these are really stems composed of CV roots and petrified suffixes or stem formatives, but for now we must consider them roots.) They form their progressives by strengthening the vowel, shifting from schwa to /e/ or /a/ (CáCT  $\rightarrow$  CÁCT). Generally, the vowel in the progressive is /e/ unless the third consonant is rounded, making it /a/, but there are exceptions, such as  $\dot{c}\dot{e}\dot{y}x^wt$  'be drying it,'  $q^w\dot{a}\dot{l}st$  'be boiling it.' If the second consonant is an unglottalized resonant, it is glottalized in the progressive, unless (it seems) the third consonant is a glottalized obstruent. The following are bare roots with their progressives:

| sáwý 'seek'                                               | séwd 'be seeking'                                                                           |    |
|-----------------------------------------------------------|---------------------------------------------------------------------------------------------|----|
| $2 \delta w \vec{k}^w$ 'get used up'                      | $2\acute{e}\vec{w}\vec{k}^{w}$ (AG), $2\acute{a}\vec{w}\vec{k}^{w}$ (DK) 'be getting used u | p' |
| $\tilde{X} \delta l \tilde{x}$ 'spark, throw out a spark' | $\vec{x} \cdot \vec{e} \cdot \vec{x}$ 'be sparking' (AG)                                    |    |

With *-t* 'transitive,' CoCT roots fall into two subclasses, the far greater number being in the first:

(1) with no schwa preceding the -t

| $p \delta s k^w t$ 'call him names'                                 | pásk <sup>wt</sup> 'be calling him names'                                 |
|---------------------------------------------------------------------|---------------------------------------------------------------------------|
| $\dot{p}\delta y \dot{\chi} t$ 'squeeze it'                         | $\dot{p}\dot{e}y\dot{x}t$ 'be squeezing it'                               |
| $m \delta t q^{w} t$ 'dip it'                                       | <i>mátq</i> <sup>w</sup> t 'be dipping it'                                |
| sáyít 'tickle him'                                                  | séyťt 'be tickling him'                                                   |
| <i>sáwqt</i> 'look for it'                                          | séw <i>qt</i> 'be looking for it'                                         |
| $t \delta t \tilde{x} t$ 'shake it back and forth'                  | <i>lét xt</i> 'be shaking it back and forth'                              |
| <i>łəlqt</i> 'dip it'                                               | <i>lélqt</i> 'be dipping it'                                              |
| $i \neq l q^w t$ 'snatch it off him'                                | $i \dot{d} \vec{q}^{w} t$ 'be snatching it off him'                       |
| $\tilde{X} \neq p \tilde{X} t$ 'spread it, scatter it'              | $\dot{\chi} \dot{e} p \dot{x} t$ 'be spreading it, be scattering it'      |
| $\dot{x} \dot{y} \dot{q} t$ 'press on it'                           | $\vec{x} \acute{e} y \acute{q} t$ 'be pressing on it'                     |
| $\hat{X} \neq \hat{X} \hat{X} \hat{X} \hat{X}$ 'throw sparks on it' | $\vec{\lambda} \vec{e} \vec{l} \vec{x} t$ 'be throwing sparks on it' (AG) |
| $\dot{c}\delta yx^{w}t$ 'dry it'                                    | $\dot{c}\dot{e}\dot{y}x^wt$ 'be drying it'                                |
| $k^{w} \delta y \check{x} t$ 'stir it'                              | $k^{w} \acute{e} \acute{y} \check{x} t$ 'be stirring it'                  |
| $x^{w} \delta l  \dot{p} t$ 'fan him'                               | $x^{w} \acute{e} l \acute{p} t$ 'be fanning him'                          |
| $\check{x}\check{\partial}l\check{c}t$ 'twist it around'            | <i>xélct</i> 'be twisting it around'                                      |
| $q^w \delta \vec{l} st$ 'boil it (as an egg)'                       | $q^{w} \dot{a} \dot{l} st$ 'be boiling it'                                |
| <i>hánq<sup>w</sup>t</i> 'warm it up'                               | <i>háňq<sup>w</sup>t</i> 'be warming it up'                               |
| $h \delta y q^w t$ 'recruit him'                                    | $h \dot{q} \dot{y} q^w t$ 'be recruiting him'                             |
| $2 \delta w \vec{k}^w t$ 'use it up'                                | $e^{2}\acute{e}w\dot{k}^{w}t$ 'be using it up' (AG)                       |
|                                                                     |                                                                           |

(2) with a schwa preceding the -t. I have found only one example:

 $\dot{c}\delta lx\delta t$  'delouse him'  $\dot{c}\delta lx\delta t$  'be delousing him (a child)'

With  $-\partial m$  'intransitive' and  $-\partial l$  'move toward,' some but perhaps not all C $\partial$ CT roots appear in the perfective as CÁC $\partial$ T- (the Á being the /e/ or /a/ that appears in the progressive of the *-t* form), and these form their progressives in ways varying slightly depending on the type of consonant in the second position.

(1) If the second consonant is an obstruent, the stressed full vowel is lengthened (for CC though not for AG) and the schwa is dropped (CÁTəT-  $\rightarrow$  CÁ(·)TT-):

 $m \acute{a}t a q^w a m$  'bubble up' $m \acute{a} t q^w a m$  (CC),  $m \acute{a}t q^w a \acute{m}$  (AG) 'be bubbling up'(cf.  $m \acute{a}t q^w t$  'dip it' $m \acute{a}t q^w t$  'be dipping it')

(2) If the second consonant is a resonant, this is glottalized in the progressive and the schwa following it is dropped (CÁR<sub>2</sub>T-  $\rightarrow$  CÁ<sup> $\alpha$ </sup>T-), as in:

| <i>k<sup>w</sup>eyəžəm</i> 'move'                            | k <sup>w</sup> éýx <i>am</i> 'be moving'                                              |
|--------------------------------------------------------------|---------------------------------------------------------------------------------------|
| (cf. $k^w \delta y \check{x}t$ 'move it'                     | $k^{w} \acute{e} \acute{y} \check{x} t$ 'be moving it')                               |
| <i>Řéləxəm</i> 'spark'                                       | <i>Žél žəm</i> 'be sparking' (AG)                                                     |
| (cf. <i>Žálž</i> 'spark'                                     | $\vec{\lambda} \vec{e} \vec{l} \vec{x}$ 'be sparking')                                |
| <i>hánəq<sup>w</sup>əm</i> 'get warm'                        | hánq <sup>w</sup> əm 'be getting warm'                                                |
| (cf. <i>hánq<sup>w</sup>t</i> 'warm it up'                   | <i>hánq<sup>w</sup>t</i> 'be warming it up')                                          |
| $q^w \acute{a} m \partial \check{x}^w \partial l$ 'get thin' | $q^{w} \acute{a} \vec{m} \partial \check{x}^{w} \partial \acute{l}$ 'be getting thin' |
| (cf. $cq^{w} \delta m \check{x}^{w}$ 'thin, skinny')         |                                                                                       |

The last progressive form, 'be getting thin,' seems irregular in not losing its first schwa.

There are words that are composed of stems of these shapes with these suffixes and forming their progressives in these ways. Their roots have not yet been recorded bare of suffixes, but possibly they too are of the shape C<sub>2</sub>CT. These include:

| $pálax^w am$ 'steam (from heat, as a pot)'                                        | $p \acute{a} l x^w \partial m$ 'be steaming'                        |
|-----------------------------------------------------------------------------------|---------------------------------------------------------------------|
| (AG denied ** pálxwt, but cf. spálaxwam 's                                        | steam')                                                             |
| $\dot{p}\dot{a}l\partial\dot{k}^{w}\partial m$ 'boil up (as water from paddling)' | <i>pálkwəm 'be boiling up'</i>                                      |
| łátəq <sup>w</sup> əm 'boil'                                                      | <i>łátq<sup>w</sup>əm</i> ' 'be boiling' (AG)                       |
| <i>łáťaq</i> wam 'snore'                                                          | <i>táťq</i> <sup>w</sup> <i>əm</i> 'be snoring' (CC)                |
| $l\acute{a}\acute{t} aq^w \partial l$ 'get soft (as fish in the sun)'             | $l\dot{a}\dot{t}\dot{q}^{w}\partial l$ 'be getting soft' (CC)       |
| <i>k<sup>w</sup>átəx<sup>w</sup>əm</i> 'burst into flame'                         | $k^{w} \dot{a} \cdot t \partial x^{w} \partial m$ 'be blazing' (JP) |
| x <sup>w</sup> ánək <sup>w</sup> əm 'make a blowing noise,                        | x <sup>w</sup> áňk <sup>w</sup> əm 'be making a blowing             |
| breathe heavily'                                                                  | noise, etc.'                                                        |
|                                                                                   |                                                                     |

With other suffixes, CoCT roots show the forms seen with *-t*, except that in the perfective the stress may be on either the root or the suffix, while in the progressive it seems to be always on the root.

c'əýx<sup>w</sup>éls 'dry fish' c'éýx<sup>w</sup>t 'be drying it')
xálc'əst 'wring its neck' xálc'əst 'be wringing its neck'
(cf. xálc' t'twist it,' xélc't 'be twisting it,' -əs 'face, front end')
səwqánəq 'look for a person' séwqanəq 'be looking for a person'
(cf. səwq' 'seek,' séwq' 'be seeking,' -ánəq 'another person')
təlqác səm 'put one's hands into water' télqəc səm 'be putting ...'
(cf. təlqt 'dip it, launch it,' télqt 'be dipping it,' -cəs ~ -écəs ~ -ácəs 'hand,' -əm 'intransitive')

#### 7.2.11. Unique Progressives

Three roots form their progressives in wholly irregular ways:

(1)  $\check{x}t\acute{e}^{2}$  'do,'  $\check{x}\acute{a}t\acute{a}$  'be doing' (also 'be saying'), in which we see a unique replacement of the unglottalized /t/ in the perfective by a glottalized /t/ in the progressive. This root appears with -am as

 $\check{x}t\acute{e}\cdot am$  'prepare (as food or tea)'  $\check{x} \cdot a\acute{t}\acute{e}\cdot m$  'be preparing'

(2)  $\theta \delta t$  'say,'  $\tilde{x} \delta t \delta$  'be saying' (also 'be doing'), in which there is a (so far) unique example of the progressive of one verb also serving as the progressive of another.

(3)  $k^w \delta ll \delta x$  'shoot,'  $2\hat{e}^2 k^w \delta ll \delta x$  'be shooting,' in which there is a (so far) unique prefixed progressive marker.

#### 7.3. PLURALS

Many, but evidently not all, verbs have plural forms in the perfective and/or progressive aspect. (This section deals with plurals of these forms only. Plurals of duratives and resultatives are discussed under those headings.)

Depending on the transitivity of the verb, plural forms can indicate plurality in the subject, object, event, or result. The plural of an intransitive form can indicate a plural subject; thus, the plural of  $n\acute{em}$  'go,'  $n\acute{el}a\acute{m}$ , can only mean 'several go.' However, the plural of  $i\acute{a}\acute{q}^w$  'break or snap (as a rope or stick),'  $i\acute{a}\acute{q}^w i\acute{a}\acute{q}^w$ , can mean 'several break' or 'one gets broken up into lengths,' and the plural of  $i\acute{a}\acute{q}^w$  'get pierced,' can mean 'get pierced all over' or 'get pierced many times (as by a thistle).' The plural of a transitive form can indicate a plural object as well as a plural subject or event; thus, the plural of  $i\acute{e}\acute{t}\acute{q}^w at$  'hit him with the fist,'  $i\acute{e}\acute{a}i\dot{a}\acute{q}^w at$ , can mean 'one hits several,' 'several hit one,' or 'one hits one several times.' JP glossed  $laml\acute{a}me^2t$ , the plural of  $lam\acute{e}^2t$  'kick him,' as 'kick a lot' – 'one kicks several,' 'several kick one,' or 'one kicks the hell out of one'! Plural forms may therefore, at least with some verbs, serve to produce a kind of intensive or iterative aspect.

Plural verb forms are optional. Plurality of subject or object need not be indicated in any way. Plurality of a third person is often indicated by the particle  ${}^{2}\acute{e}ttan$ , which may obviate the need for indicating it with the verb. Plurality of events can be indicated by adverbs such as  $cax^{w}l\acute{e}$  occasionally,'  $y\acute{a}\theta$  'always,' and so on. Plurals of verbs are only moderately common in my texts, and for many verbs I have been unable to elicit plural forms, especially plural progressives.

Plurals, like progressives, are formed differently according to the shape of the root.

### 7.3.1. CAC Plural Perfectives

CAC roots, bare and with *-t* and *-om*, regularly make plurals of their perfectives with an infixed *-lo*- (which occasionally appears as *-ol*-) or *-lo*- (which occasionally appears as *-ol*-) or *-lo*- (which occasionally appears as *-ol*-) and, if the vowel is */i/*, a lowering to */e/* or, in the presence of a rounded back velar, to */a/*. The variations in the infix are not predictable at this time. Here are examples of perfectives followed by their plurals:

| <i>cám</i> 'go/come up from the shore'                | <i>cáləm</i> 'several go/come up'                                                            |
|-------------------------------------------------------|----------------------------------------------------------------------------------------------|
| <i>q́áy</i> 'die'                                     | <i>qáləy</i> 'several die'                                                                   |
| $\dot{t}ax^w$ 'go/come down to the shore'             | $i d \hat{l} \partial x^w$ 'several go/come down'                                            |
| <i>łák</i> <sup>w</sup> 'fly'                         | <i>tálok</i> <sup>w</sup> 'several fly'                                                      |
| $\dot{q}^{w}$ ím 'disembark'                          | $\dot{q}^{w}$ áləm 'several disembark'                                                       |
| $i^{\theta}i\dot{q}\partial t$ 'step on it'           | $t^{\theta} \acute{e} l \partial \acute{q} \partial t$ 'step on several, several step on it' |
| $\dot{t}^{\theta} \dot{q}^{w} \partial t$ 'punch him' | $t^{\theta} \dot{a} \dot{l} \dot{a} \dot{q}^{w} \partial t$ 'punch several, etc.'            |
| $\theta(\dot{c}\partial t)$ 'stab him'                | $\theta \acute{e} \acute{l} \acute{o} \acute{c} \acute{o} t$ 'stab several, etc.'            |
| <i>łí pət</i> 'cut it up'                             | <i>téləpət</i> 'cut it into pieces, etc.'                                                    |
| <i>tícət</i> 'cut it across'                          | <i>té<sup>9</sup>alčat</i> 'cut it into slices, etc.'                                        |
| <i>čísəm</i> 'grow'                                   | <i>čéləsəm ~ čé·lsəm</i> 'several grow'                                                      |
| <i>xéqam</i> 'open the mouth'                         | <i>xéləqəm</i> 'open the mouths'                                                             |
| $x^{w}$ ísət 'shake it up'                            | $x^{w} e^{2} \partial l s \partial t$ 'shake them up, etc.'                                  |
| $\dot{q}(\vec{k}^{w} \partial t \text{ 'bite him'})$  | $\dot{q}\dot{e}\dot{l}\partial\vec{k}^{w}t$ 'bite several, bite in several                   |
|                                                       | places, several bite him'                                                                    |

(The plural of 'bite' was recorded without a schwa before the *-t*. I cannot explain this apparent irregularity.)

The few irregular plural perfectives will be identified along with irregular plural progressives below.

# 7.3.1.1. CAC Plural Progressives

CAC roots, like other types, generally base their plural progressives on the shape of the root as reflected in the simple (non-plural) progressive. As long as

the shape of the root is clearly reflected by CV reduplication in the simple progressive, the plural progressive is formed by CVC reduplication of the root with the stress on the first vowel and the second appearing as schwa, as in:

| <i>k<sup>w</sup>áq<sup>w</sup></i> 'get hit'     | k̃ <sup>w</sup> ák̃ <sup>w</sup> əq <sup>w</sup> (prog.)                    |
|--------------------------------------------------|-----------------------------------------------------------------------------|
| <i>k<sup>w</sup>áləq<sup>w</sup></i> (pl. perf.) | k̃ <sup>w</sup> áq <sup>w</sup> k̃ <sup>w</sup> əq <sup>w</sup> (pl. prog.) |
| <i>k̈̀«áqʷət '</i> hit/club him'                 | k <sup>w</sup> ák <sup>w</sup> ∂q <sup>w</sup> ∂t (prog.)                   |
| <i>k̈«álə qʷət (</i> pl. perf.)                  | k <sup>w</sup> áq <sup>w</sup> k <sup>w</sup> ∂q <sup>w</sup> t (pl. prog.) |
| <i>ťáx</i> <sup>w</sup> 'go/come down'           | <i>iáiəx</i> <sup>w</sup> (prog.)                                           |
| <i>ťálax<sup>w</sup></i> (pl. perf.)             | <i>iáx<sup>w</sup>iəx<sup>w</sup></i> (pl .prog.)                           |
| <i>łák</i> <sup>w</sup> 'fly'                    | łáł∂k <sup>w</sup> (prog.)                                                  |
| <i>łálok</i> <sup>w</sup> (pl. perf.)            | łákwł∂k <sup>w</sup> (pl. prog.)                                            |
| <i>ťícəm</i> 'swim'                              | <i>itiəcəm</i> (prog.)                                                      |
| <i>ťéləcəm</i> (pl. perf.)                       | <i>itciəcəm</i> (pl. prog.)                                                 |

In the following, the progressive is irregular but the plural progressive seems to follow the pattern of the previous set:

| $q^{w} \acute{e} l$ 'speak'                  | $q^w \acute{a}^? q^w  ightarrow l$ | (prog.)     |
|----------------------------------------------|------------------------------------|-------------|
| $q^{w} \acute{e}^{2} \partial l$ (pl. perf.) | qʷál̊qʷəl̊                         | (pl. prog.) |

In the next two examples, the shape of the root is obscured by its simple (non-plural) progressive form, and so the plural progressive is formed by reduplication of the initial CVC of this simple progressive form, and the stress may be on the second vowel, as in:

| <i>cám</i> 'go/come back up' | cák <sup>w</sup> əm (prog.)                      |
|------------------------------|--------------------------------------------------|
| <i>cáləm</i> (pl. perf.)     | cək <sup>w</sup> cák <sup>w</sup> əm (pl. prog.) |
| ném 'go'                     | háňam (prog.)                                    |
| néləm (pl. perf.)            | hanháňam (pl. prog.)                             |

The irregularities of these simple progressives have been dealt with earlier (in \$7.2.1).

Two CAC roots are irregular in the formation of their plural perfectives by CVC reduplication, and in one the irregularity is taken further in the plural progressive.

| iíiələm (prog.)<br>ií·liələm (pl. prog.)                                                            |
|-----------------------------------------------------------------------------------------------------|
| $k^{w} \partial k^{w} i \cdot l$ (prog.)                                                            |
| (no pl. prog. recorded)                                                                             |
| $k^{w} \acute{e} k^{w} \partial l x \sim k^{w} \acute{e}^{\gamma} k^{w} \partial \dot{l} x$ (prog.) |
| (no pl. prog. recorded)                                                                             |
|                                                                                                     |

Finally, JP gave an irregular plural progressive in the following set:

| <i>lí pət</i> 'cut it up (a carcass)' | <i>títəpət</i> 'be cutting it up'          |
|---------------------------------------|--------------------------------------------|
| <i>téləpət</i> 'cut it into pieces'   | <i>télapat</i> 'be cutting it into pieces' |

Here it appears that the plural progressive is formed from the plural perfective following a pattern in which a medial resonant is simply glottalized, as if *télapat* were a simple (non-plural) perfective. AG, on the other hand, glottalized the resonant in the perfective and so could not produce such a contrast.

#### 7.3.2. HAC and HAC<sub>a</sub>C plurals

In the few examples recorded, HAC and HAC<sub>2</sub>C roots form their plural perfectives as CAC roots do but differ in their plural progressives in the placement of the stress and appearance of glottal stops:

| háy 'stop'                                                     | há·ý (prog.)                                                     |
|----------------------------------------------------------------|------------------------------------------------------------------|
| háləy [háli] (AG) (pl. perf.)                                  | (no pl. prog. recorded)                                          |
| ?ík <sup>w</sup> 'be lost, die' ?élak <sup>w</sup> (pl. perf.) | $2\ell^2 \vec{k}^w$ 'be getting lost'<br>(no pl. prog. recorded) |
| <sup>?</sup> <i>ítət</i> 'go to bed, sleep'                    | <sup>?</sup> í <sup>?</sup> tət (prog.)                          |
| ? <i>a</i> ?əltət (pl. perf.)                                  | <sup>?</sup> ət <sup>?</sup> í <sup>?</sup> tət (pl. prog.)      |
| <sup>?</sup> í <i>məx</i> 'walk'                               | ²ím̀əx //²í²m̀əx//? (prog.)                                      |
| ? <i>əṁíməx</i> (pl. perf.)                                    | ²əm̀í·m̀əx //²əm²í²m̀əx//? (pl. prog.)                           |

In the last two examples, the stress in the plural progressive is on the second vowel rather than on the first. The plural progressive of  ${}^{2}(tat)$  'sleep' could more easily be identified as the product of CVC reduplication if it were  $**{}^{2}at{}^{2}(tat)$  or based on the progressive form  $**{}^{2}a{}^{2}(tat)$ , but the latter is phonologically impossible. The recorded form may be a kind of compromise, as may be the plural progressive of  ${}^{2}(tat)$  'sleep' is clearly not produced by a rounded uvular, as in a number of CAC roots, and so its presence implies, I believe, that the second vowel was once \*u, that is, 'sleep' was  $*{}^{2}itut$  (cf. Lushootseed  ${}^{2}(tut)$  'sleep').

One HAR root resembles the irregular TAR root iilom 'sing' in forming its plural perfective by CVC reduplication:

| <i>híləm</i> 'fall, roll' | híləm (prog.)             |
|---------------------------|---------------------------|
| həlhíləm (pl. perf.)      | hilhílam (DK) (pl. prog.) |

#### 7.3.3. CAH Plurals

I have not recorded any plural forms of TAH or HAH roots. From the following examples, RAH roots appear to form plural perfectives by CV reduplication with the stress on the second vowel:

| $m e^{2x}$ 'remove it'                                                     | hóm e <sup>2</sup> x (CC), hóm əx (AG) (prog.) |
|----------------------------------------------------------------------------|------------------------------------------------|
| $m \partial m e^{2x}$ (pl. perf.)                                          | həmhóm əx (AG) (pl. prog.)                     |
| $l\acute{e}^{2}x \sim l\acute{\iota}^{2}x$ 'put it away'                   | hálex (prog.)                                  |
| $l\partial l\acute{e}^{2}x \sim l\partial l\acute{\iota}^{2}x$ (pl. perf.) | həlháləx (AG) (pl. prog.)                      |

### 7.3.4. C<sub>2</sub>C Plural Perfectives

C $\partial$ C roots seem to follow more than one pattern and, as they do in forming progressives, behave differently as bare roots and when transitivized with *-t*. As bare roots, C $\partial$ C roots seem to form their plural perfectives in two ways:

(1) by reduplication of the first consonant and strengthening of the vowel to /e/ or /a/, a schwa being present only if one of the consonants is a resonant  $(C_1 \diamond C_2 \rightarrow C_1 \land C_1(\diamond) C_2)$ , as in:

| $n \delta q^w$ 'fall asleep'                          | <i>ná·nq<sup>w</sup></i> //nánəq <sup>w</sup> // 'several fall asleep' |
|-------------------------------------------------------|------------------------------------------------------------------------|
| $k^{w} \delta n$ 'become possessed'                   | $k^{w} \acute{e} k^{w} i$ 'several become possessed'                   |
| $x^{w} \delta y$ 'wake up'                            | $x^w \dot{a} x^w \partial y$ 'several wake up'                         |
| <i>tás</i> 'arrive there'                             | téts (CC) 'several arrive there'                                       |
| $w \partial t \partial \tilde{x}$ 'stand up suddenly' | $w \partial t \partial t \dot{x}$ 'several stand up suddenly' (AG)     |
| $(< w \partial$ - 'suddenly')                         |                                                                        |

AG glossed  $t \acute{e}ts$  as 'arrive there in a group,' identifying the simple progressive,  $t \partial t \acute{o}s$ , as 'arrive there one by one.' Perhaps one person cannot "be arriving" in Halkomelem, one being either there or not. I have recorded no plural progressive for this word.

(2) by CVC reduplication with stress on the second schwa ( $C_1 \Rightarrow C_2 \rightarrow C_1 \Rightarrow C_2 C_1 \Rightarrow C_2$ ), as in:

| $\dot{c}\dot{j}\dot{q}^{w}$ 'get pierced' | $\dot{c}$ $\partial \dot{q}^{w}$ $\dot{c}$ $\partial \dot{q}^{w}$ 'get pierced all over' |
|-------------------------------------------|------------------------------------------------------------------------------------------|
| <i>lák</i> <sup>w</sup> 'get broken'      | <i>lək<sup>w</sup>lák<sup>w</sup></i> 'several get broken'                               |
| pán 'get buried'                          | pənpán 'several get buried'                                                              |

I have recorded only a few examples of each of these two types and must acknowledge the possibility that they are not alternatives of the same form but are actually two different forms. The only possible evidence I have for this is what may be two plurals of  $k^{w} \delta n$ . As a bare root,  $k^{w} \delta n$  is usually translated 'get started,' meaning 'go into a trance-like state as a winter dancer does,' and I have glossed it 'become possessed.' I have assumed that it is the root of  $k^{w} \delta n \delta t$  'take it,'  $k^{w} \delta n n \delta x^{w}$  'get it,' and so on, and it does literally mean 'be taken.' A plural formed by the first rule identified above,  $k^{w} \delta k^{w} \delta n$ , was given by AG and glossed 'they get started.' But there is another word,  $k^{w} \delta n k^{w} \delta n$  'be captured' ('be taken by several'?), which seems to be a plural of  $k^{w} \delta n$  formed by the second rule. Alternatively, we might consider  $k^{w} \delta n$  'become possessed' and the root  $k^{w} \delta n \delta t$  'take it' as homophones that form their plurals by different rules. With -t (also -x and  $-\partial m$ ), CoC roots form their plural perfectives by the second of the two patterns seen with the bare root, as in:

| $\theta \vec{k}^{w} \delta t$ 'pull it taut'    | $\theta \partial \vec{k}^{w} \theta \partial \vec{k}^{w} t$ 'pull several taut'    |
|-------------------------------------------------|------------------------------------------------------------------------------------|
| $t^{\theta} \check{x}^{w} \acute{a}t$ 'wash it' | $t^{\theta} \delta \check{x}^{"} t^{\theta} \delta \check{x}^{"} t$ 'wash several' |
| <i>pánət</i> 'bury it'                          | <i>pənpánət</i> 'bury several'                                                     |
| $n \delta \dot{p} \delta x$ 'eat it'            | <i>nəpnápax</i> 'eat several, chew it up'                                          |
| náqəm 'dive'                                    | nəqnáqəm 'several dive'                                                            |

#### 7.3.4.1. CoC Plural Progressives

Like the simple progressives, the plural progressives of C<sub>2</sub>C roots are formed differently depending on whether the initial consonant is an obstruent or a resonant.

T $\ni$ T (and T $\ni$ R?) roots form the plural progressives of bare roots by infixing -*al*- between the first consonant and the first schwa of the simple progressives (C<sub>1</sub> $\ni$ C<sub>1</sub> $\Rightarrow$ C<sub>2</sub> $\rightarrow$  C<sub>1</sub> $\ni$ l $\Rightarrow$ C<sub>1</sub> $\Rightarrow$ C<sub>2</sub>), as in:

| $i a i a q^w$ 'be breaking' | $i \partial \partial i \partial q^w$ 'several are breaking' |
|-----------------------------|-------------------------------------------------------------|
| sasáq 'be cracking'         | sələsáq 'several are cracking'                              |

T<sub>2</sub>T roots form the plural progressives of their forms with *-t* by CVC reduplication of the form that appears in the simple progressive with the reduction of one (usually the second) vowel to schwa ( $C_1 A C_2 - t \rightarrow C_1 A C_2 C_1 = C_2 - t$ ). The pattern is seen in the following sets:

| $i \dot{a} \dot{q}^{w}$ 'break, snap'                                   | <i>ťəťáq</i> <sup>w</sup> (prog.)                                  |
|-------------------------------------------------------------------------|--------------------------------------------------------------------|
| $\dot{t} \partial \dot{q}^{w} \dot{t} \partial \dot{q}^{w}$ (pl. perf.) | <i>ťələťáq</i> <sup>w</sup> (pl. prog.)                            |
| $\dot{t}\dot{q}^{w}\dot{a}t$ 'cut it off'                               | ťáq <sup>w</sup> t (prog.)                                         |
| <i>ťəq<sup>w</sup>ťáq<sup>w</sup>ət</i> (pl. perf.)                     | $\dot{t}\dot{a}\dot{q}^{w}\dot{t}\partial\dot{q}^{w}t$ (pl. prog.) |
| sáq' 'crack, tear, split'                                               | səsáq (prog.)                                                      |
| sésq (pl. perf.)                                                        | sələsə́q (pl. prog.)                                               |
| $s \dot{q} \dot{e} t$ 'split it, tear it'                               | séqt (prog.)                                                       |
| səqsáqət (pl. perf.)                                                    | séqsəqt (pl. prog.)                                                |

The plural perfective of the first set might be translated 'cut it up into lengths,' and of the second 'split it up,' 'tear it into shreds,' or 'tear several.'

R $\circ$ C roots form their plural progressives of both bare roots and transitive forms by reduplication of the initial h $\circ$  $\mathring{R}$  of the simple progressive with the first resonant unglottalized (at least in the forms recorded) and the stress on the second schwa, as in the following sets:

| $n \delta q^w$ 'fall asleep'          | háňq <sup>w</sup> (prog.)        |  |
|---------------------------------------|----------------------------------|--|
| ná·nq <sup>w</sup> (pl. perf.)        | hənhánq <sup>w</sup> (pl. prog.) |  |
| $n \partial w \partial x$ 'insert it' | hánwax (prog.)                   |  |
| nəwnáwax (pl. perf.)                  | hənhánwəx (pl. prog.)            |  |

| <i>náqəm</i> 'dive'                                 | <i>háňqəm</i> (prog.)        |
|-----------------------------------------------------|------------------------------|
| <i>nəqnáqəm</i> (pl. perf.)                         | <i>hənháňqəm</i> (pl. prog.) |
| $w \delta \dot{q}^w$ 'drift away'                   | həẁq'" (prog.)               |
| $w \delta \dot{q}^w w \delta \dot{q}^w$ (pl. perf.) | həwhóẁq'" (pl. prog.)        |

AG also gave a form  $w \dot{a} \dot{q}^w w \partial \dot{q}^w$  'be idling along under power,' which appears to be formed as a transitive would be.

There is one R<sub>2</sub>R root that is irregular in every form:

| náyəm 'laugh'         | néýəm (prog.)         |
|-----------------------|-----------------------|
| hənní yəm (pl. perf.) | nəynéysim (pl. prog.) |

#### 7.3.5. C<sub>2</sub>CT Plurals

CəCT roots form their plurals by reduplication of the CəC- or CAC- of the simple (non-plural) form, as in:

| <i>łślqt</i> 'dip it, immerse it' | <i>łélqt</i> (prog.)        |
|-----------------------------------|-----------------------------|
| <i>łəlłálqt</i> (pl. perf.)       | <i>łélłəlqt</i> (pl. prog.) |

As we have seen earlier (§7.2.7),  $\theta \delta yt$  'fix it,' although it is a T $\partial$ R root plus -*t*, behaves like a C $\partial$ CT root. The following set further illustrates this:

| $\theta \delta yt$ 'fix it'                                                                    | $\theta \acute{e} \acute{y} t$ (prog.)                                                                     |
|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------|
| $\theta \partial y \theta \partial \dot{y} t [\theta i \theta \partial \dot{y} t]$ (pl. perf.) | $\theta \dot{e} \dot{y} \theta \dot{v} \dot{y} t [\theta \dot{e} \dot{y} \theta i^{\gamma} t]$ (pl. prog.) |

A few words composed of CACC stems that may be expansions of C $\Rightarrow$ CT roots and the suffix - am have plurals formed by CVC reduplication, such as:

| <i>páləkwəm</i> 'boil'                                                                         | <i>pálkwəm</i> (prog.)      |
|------------------------------------------------------------------------------------------------|-----------------------------|
| <i>pəlpáləkwəm</i> 'sev. boil' (pl. perf.)                                                     | (no pl. prog. recorded)     |
| <i>Žéləžəm</i> 'spark'                                                                         | <i>Žéĺžəm</i> 'be sparking' |
| $\vec{X}$ $\partial l \vec{X} \dot{e} l \partial \tilde{x} \partial m$ 'spark repeatedly' (CC) | (no pl. prog. recorded)     |

# 7.3.6. H<sub>2</sub>C(C)V(C) Plurals

The few words of these shapes do not follow a consistent pattern. One resembles CAC roots in forming its plural perfective with a full vowel and infixed *-l-*, and it resembles specifically HAC<sub>2</sub>C words in having an "extra" glottal stop in the plural:

| <i>²áxəl</i> 'paddle'                 | $2\hat{i}^2x\hat{j}\hat{l}$ (prog.)                         |
|---------------------------------------|-------------------------------------------------------------|
| <i><sup>?</sup>éləxəl</i> (pl. perf.) | <sup>?</sup> ax <sup>?</sup> í <sup>?</sup> xal (pl. prog.) |

Another forms its plural perfective by CVC reduplication of the simple perfective (cf.  $\dot{t}(l \partial m \text{ 'sing,' above})$  and its plural progressive by CVC reduplication of its simple progressive:

| <sup>2</sup> ómət 'sit down, get up'   | ?ám̀ət (prog.)                   |
|----------------------------------------|----------------------------------|
| <sup>?</sup> <i>ðmðmðt</i> (pl. perf.) | <sup>9</sup> əmá·mət (pl. prog.) |

Another forms its plural perfective in a unique way and its plural progressive by CVC reduplication of the simple progressive but without the "extra" glottal stop:

| <i>?áłtən</i> 'eat'                           | ?í?łtən (prog.)       |
|-----------------------------------------------|-----------------------|
| <sup>9</sup> é <sup>9</sup> əłtən (pl. perf.) | ?əł?íłtən (pl. prog.) |

And here may belong the following:

| háye? (JP), hýye? (AG) 'leave'                              | háýa (AG) (prog.)        |
|-------------------------------------------------------------|--------------------------|
| $h \partial l \partial y \dot{e}^{\gamma}$ (AG) (pl. perf.) | həyháýə (AG) (pl. prog.) |

# 7.4. DIMINUTIVES

Verbs may have diminutive and diminutive plural forms in the progressive aspect. There are no diminutive perfective forms. Diminutive forms may indicate that the subject or the object is small, as in:

| <i>ťəťí<sup>9</sup>ťələm</i> .<br>(dim. of <i>ťíləm</i> 'sing')                                                                                                           | 'A little thing (as a child) is singing.'                                              |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------|
| ?əli <sup>?</sup> í?łtən.<br>(dim. pl. of ?ə́łtən 'eat')                                                                                                                  | 'A lot of little ones are eating.'                                                     |
| $\dot{t}^{\theta} \partial l i \dot{t}^{\theta} \dot{x}^{w} \partial s \partial m.$<br>(dim. pl. of $x^{w} \dot{t}^{\theta} \dot{x}^{w} \dot{a} s \partial m$ 'wash one's | 'They are washing their little faces.' face')                                          |
| <i>čəlé<sup>9</sup>čəwəłtən.</i><br>(dim. pl. of <i>čéwəłtən</i> 'help')                                                                                                  | 'The little ones are helping.'                                                         |
| <sup>?</sup> i cən cəlé <sup>?</sup> cəwət.<br>(dim. pl. of céwət 'help him')                                                                                             | 'I am helping the little one.'                                                         |
| ni cən təlitést $k^w \theta$ ə məlí <sup>2</sup> məq <sup>w</sup> .<br>(dim. pl. of tsət 'approach him' and                                                               | 'I was approaching the little ducks.' dim. pl. of $m\dot{a}^{2}\partial q^{w}$ 'duck') |

Diminutives are formed by CV reduplication of simple progressive forms. If the simple progressive is already formed by reduplication of the root, the diminutive form will thus be doubly reduplicated, and in it the first consonant of the root will appear three times. For the different types of roots, there are differences in placement of stress and quality of vowel. A glottal stop after the stressed vowel seems usual unless it is the final vowel.

For diminutives of TAT and TAR roots, the progressive  $C_1 \dot{A} C_1 \partial C_2$  is reduplicated as  $C_1 \partial C_1 \dot{A}^2 C_1 \partial C_2$ , as in:

| $\dot{p}\dot{e}\dot{t}^{	heta}$ 'sew' | $\vec{p}\vec{e}\vec{p}\vec{a}\vec{t}^{\theta}$ (prog.) | $\vec{p}$ ə $\vec{p}$ ė $^{2}\vec{p}$ ə $\vec{t}^{\theta}$ (dim.) |
|---------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------|
| <i>céwət</i> 'help him'               | cécowot (prog.)                                        | <i>ċəċé<sup>?</sup>ċəwət</i> (dim.)                               |

| <i>icəm</i> 'swim'  | <i>iítəcəm</i> (prog.) | <i>tətí?təcəm</i> (dim.) |
|---------------------|------------------------|--------------------------|
| <i>tíləm</i> 'sing' | <i>títələm</i> (prog.) | tətí?tələm (dim.)        |

For diminutives of other roots and perhaps of derivatives of all roots, there seems to be a general rule reduplicating the first consonant of the progressive with the vowel /i/, as in:

| tsát 'approach it'                                  | tést (prog.)                                 | titést (dim.)                                          |
|-----------------------------------------------------|----------------------------------------------|--------------------------------------------------------|
| <i>ł x í l ə x</i> 'stand'                          | ł <i>žílax</i> (prog.)                       | <i>tiłžíl̇́əx</i> (dim.)                               |
| $\dot{q}^{w} \partial y \hat{l} \partial x$ 'dance' | <i>q</i> ́ <sup>w</sup> əyə́ləx (prog.)      | <i>ἀ<sup>w</sup>iἀ<sup>w</sup>əyə́ləx</i> (dim.)       |
| <i>pé·ls</i> 'blow'                                 | pəpé·ls (prog.)                              | pippé·ls (dim.)                                        |
| $q^w \delta l q^w \delta l$ 'tell'                  | q <sup>w</sup> í ·ĺq <sup>w</sup> əĺ (prog.) | $q^{w}i^{\gamma}q^{w}i^{\cdot}lq^{w}\partial l$ (dim.) |
| həwáləm 'play'                                      | hiwáləm (prog.)                              | hihiŵáĺəm (dim.)                                       |
| <sup>9</sup> <i>ítət</i> 'sleep'                    | <sup>9</sup> í <sup>9</sup> tət (prog.)      | ?i?í?tət (dim.)                                        |
| <i>?áłtən</i> 'eat'                                 | <sup>9</sup> í <sup>9</sup> łtən (prog.)     | ?i?í?łtən (dim.)                                       |

There is only one root with an initial resonant for which I have recorded a diminutive glossed as such:

```
y\dot{a}\cdot ys 'work' y\dot{a}\cdot \dot{y}s (prog.) hiy\dot{a}\dot{y}s (dim.)
```

Here perhaps the initial resonant has been converted into an h, as in the progressives of roots with initial resonant followed by schwa.

JP gave two forms that he glossed as progressives of  $m \delta k^w \partial l c \partial p$  'gather firewood,'  $h \partial m k^w \delta l c \partial p$  and  $h i m k^w \delta l c \partial p$  'be gathering firewood.' Perhaps the second is a diminutive, 'be gathering scraps of firewood.'

#### 7.4.1. Diminutive Plurals

Diminutive forms are made plural with an *-l*- infix. For diminutives formed from TAT and TAR roots, where the first vowel is a schwa, the infix is *-li-*, as in:

| ṕəṕé?ṕəť <sup>θ</sup> | 'be sewing (dim.)'  | <i>pəli pé^pət</i> <sup>0</sup> (dim. pl.) |
|-----------------------|---------------------|--------------------------------------------|
| ťəťí?ťəl`əm           | 'be singing (dim.)' | <i>təlití<sup>9</sup>tələm</i> (dim. pl.)  |

Or the infix is simply -l- replacing the second consonant, as in:

| ċəċé <sup>?</sup> ċəwət | 'be helping him (dim.)' | ċəlé <sup>?</sup> ċəwət | (dim. pl.) |
|-------------------------|-------------------------|-------------------------|------------|
| <i>tətí?təcəm</i>       | 'be swimming (dim.)'    | ťəlí?təcəm              | (dim. pl.) |

For diminutives formed from roots of other types, where the first vowel is i, the infix is  $-\partial l$ , as in:

| <i>titést</i> 'be approaching it (dim.)'                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | təlitést (dim. pl.)           |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------|
| $iit \tilde{x}i \tilde{l} \sigma x$ 'be standing (dim.)'                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | <i>təlítžíləx</i> (dim. pl.)  |
| <sup>?</sup> <i>i</i> <sup>?</sup> <i>t</i> | ? <i>əli?í?tət</i> (dim. pl.) |
| <sup>?</sup> <i>i</i> <sup>?</sup> <i>i</i> <sup>?</sup> <i>i</i> <sup>t</sup> <i>i</i> <sup>n</sup> 'be eating (dim.)'                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ?əli?í?łtən (dim. pl.)        |

A few diminutive plural forms have specific nominal meanings. The diminutive plural of  $\dot{c}isam$  'grow' is  $\dot{c}ali^2salam$  'little ones are growing,' also 'grove of young trees.' Two are used as names for arthropods,  $calic\dot{c}\lambdaam$  'sand fleas (amphipods)' (from  $c\dot{\lambda}am$  'jump,'  $c\dot{c}\lambdaam$  'be jumping') and  $\dot{q}ali\dot{q}awatam$ 'clicking grasshoppers' (from  $\dot{q}awatam$  'beat a drum'). An introduced plant, 'ground ivy (Glecoma hederacea),' is  $\dot{t}^{\theta}al\dot{c}^{2}\dot{t}^{\theta}ax^{w}am$ , the diminutive of  $\dot{t}^{\theta}\dot{c}\dot{t}^{\theta}ax^{w}am$  'blue' (showing, incidentally, that 'blue' is a progressive form, since only progressives have diminutives).

#### 7.5. THE DURATIVE ASPECT

A number of verbs have a durative aspect, and there are a few verbs whose durative forms occur more commonly than other forms. Many verbs, however, appear not to have durative forms.<sup>3</sup>

The durative aspect indicates that an action is prolonged or a position held, as illustrated by the examples below. The durative aspect intersects with the progressive aspect; many, though not all, duratives appear in both perfective and progressive forms.

The durative is marked by the appearance of a stressed full vowel as the final vowel of the stem. Depending on the type of root, there may also be CV reduplication.

TAT, TAR, and T $\partial$ T roots reduplicate the first consonant of the root with  $/\partial/$  appearing as the intervening vowel in the perfective and /i/ in the progressive. Compare the following sets of non-durative (first line) and durative (second line) forms in the perfective (left column) and progressive (right column):

| $\vec{k}^{w} \acute{e} c \partial t$ 'look at it'                                | $\vec{k}^{w}\vec{e}\vec{k}^{w}acat$ 'be looking at it'      |
|----------------------------------------------------------------------------------|-------------------------------------------------------------|
| $\vec{k}^{w} \partial \vec{k}^{w} c \acute{e} t$ 'watch it, keep your eye on it' | $\vec{k}^{w}i\vec{k}^{w}c\acute{e}t$ 'be watching it, etc.' |
| <i>ἀίἐʷət</i> 'bite it'                                                          | $\dot{q}t\dot{q}a\dot{k}^{w}at$ 'be biting it'              |
| <i>ἀəἀἐʷét</i> 'hold it in your teeth'                                           | $\dot{q}i\dot{q}k^{w}\acute{e}t$ 'keep holding it'          |
| <i>ť<sup>θ</sup>í ἀj ≥t</i> 'step on it'                                         | $i^{\theta}ii^{\theta}a\dot{q}at$ 'be stepping on it'       |
| <i>ť<sup>θ</sup>ð ť<sup>θ</sup>ά ℓ</i> 'hold it down with your body'             | $i^{\theta}ii^{\theta}a\dot{q}at$ 'keep holding it down'    |
| <i>k<sup>w</sup>í xət</i> 'name him'                                             | <i>k<sup>w</sup>ík<sup>w</sup>əxət</i> 'be naming him'      |
| <i>k<sup>w</sup>ək<sup>w</sup>xít</i> 'call him by nickname'                     | <i>k<sup>w</sup>ik<sup>w</sup>xít</i> 'be calling him by'   |
| <i>θžát</i> 'push it away'                                                       | $\theta \dot{e} \dot{x} t$ 'be pushing it away'             |
| <i>θəθžét</i> 'hold it away, hold it back'                                       | $\theta i \theta \dot{x} \dot{e} t$ 'keep holding it'       |
| $q^{w} \acute{e} l$ 'speak'                                                      | $q^{w} \dot{a} q^{w} \dot{a} \dot{l}$ 'be speaking'         |
| $q^{w} \acute{e} q^{w} \acute{e} l$ 'give a speech'                              | $q^{w} i q^{w} \dot{e} \dot{l}$ 'be giving a speech'        |

<sup>3</sup> Forms that Leslie (1979) identifies as "augmentative" (p. 44) and "stative" (p. 47) belong here, I believe.

The first five of the six sets of examples just given might suggest that the durative is marked by a suffix  $-\acute{e} \sim -\acute{t}$  that follows the second consonant of the root and is followed by the transitivizer -t. However, the last set seems to show that the durative is formed by a process that occurs within the word regardless of morpheme boundaries. (The form  $p \circ p \acute{e} k^w$  'afloat' may also belong here, but see §7.7.6).

There are TAT and T<sub>2</sub>T roots that seem to have sets of forms that are not as complete as those just given. In the following example, the expectable progressive is missing and forms that appear to be perfective and progressive duratives (in the second line) are glossed as simple progressive.

| <i>k<sup>w</sup>écəm</i> 'scream'                    | (* <i>k<sup>w</sup>ék<sup>w</sup>əcəm</i> , not recorded) |
|------------------------------------------------------|-----------------------------------------------------------|
| <i>k<sup>w</sup>ək<sup>w</sup>cém</i> 'be screaming' | kwikwcém 'be screaming'                                   |

In the next two examples, there seems to be only one durative form.

| <i>xéqəm</i> 'open the mouth'                           | <i>xéxəqəm</i> 'be opening the mouth'                     |
|---------------------------------------------------------|-----------------------------------------------------------|
| (* <i>xəxqém</i> , not recorded)                        | <i>xixqem</i> 'have your mouth open'                      |
| $\dot{q}^w x \dot{a} t$ 'pet it'                        | $\dot{q}^{w} \dot{e} xt$ 'be petting it'                  |
| (* $\dot{q}^w \partial \dot{q}^w x i t$ , not recorded) | $\dot{q}^{w} i \dot{q}^{w} x i t$ 'make up to it/him/her' |

The object of the last form can be an animal or a human member of the opposite sex.

TAH roots with *-t* 'transitivize' form their duratives (as TAT roots with *-t* do) by placing the stressed vowel between the final consonant of the root (? in the examples below) and the *-t*. (In the non-durative transitives of TAT roots, this slot is occupied by schwa, but in TAH counterparts by zero; cf.  $k^w \acute{e} c a t$  'look at it,'  $k^w e^2 t$  'drop it.') Reduplication appears in the progressive only. Compare the following non-durative (first line) and durative (second line) forms. Both perfective and progressive duratives were given by DK; AG had only the perfective durative, which he glossed as 'be leaving it alone.'

| $k^{w} e^{\gamma} t$ 'drop it, let it go'      | $k^{w} \partial k^{w} \dot{e}^{\gamma} t$ 'be dropping it, letting it go' |
|------------------------------------------------|---------------------------------------------------------------------------|
| $k^{w}\partial^{2}\acute{e}t$ 'leave it alone' | $k^{w} \partial k^{w} \partial^{2} \acute{e}t$ 'be leaving it alone'      |

The duratives in the following set were given by DK only:

| $\dot{t}^{\theta} \dot{e}^{\gamma} t$ 'chew it'                        | $t^{\theta} \delta t^{\theta} \delta^{2} t$ 'be chewing it'                                      |
|------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|
| $\dot{t}^{\theta} \partial^{2} \dot{\epsilon} t$ 'continually chew it' | $t^{\theta} \dot{t}^{\theta} \dot{t}^{\theta} \dot{s}^{2} \dot{e} t$ 'be continually chewing it' |

The perfective durative in the following was given by DK; AG did not have it.

| $s e^{2}t$ 'raise it, lift it'      | $s \ge s e^{\gamma} t$ 'be raising it, be lifting it' |
|-------------------------------------|-------------------------------------------------------|
| sə <sup>2</sup> ét 'keep it lifted' |                                                       |

However, the following pair appear to be durative and may be derived from  $s\dot{e}^{2}t$  'raise it' in the sense, as DK suggested, of 'keep his/her hopes up.'

səsə?ét 'flirt with him/her'

sisə?ét 'be flirting with him/her'

Both DK and AG gave these forms.

TəR roots with *-t* form their duratives with a stressed full vowel between the second consonant and the *-t* but without reduplication and apparently without separate perfective and progressive forms, as in the following examples:

k\*ánət 'take it' k\*ək\*ánət 'be taking it' k\*əné?t (JP), k\*ənét (AG) 'hold it, be holding it' k\*ánàm 'get some, get' k\*ək\*ánàm 'be getting some' k\*əném' 'have some, have'

cámat 'take it in your mouth'
càcámat 'be taking it in your mouth'
càmé<sup>2</sup>t (JP), càmét (AG) 'hold it in your mouth, be keeping it in ...'
càmé<sup>2</sup>m' 'be having something in your mouth'
cámát (JP), cámat (AG) 'pack it, load it on your back'
cacámat 'be loading it up ...'
camé<sup>2</sup>t (JP), camét (AG) 'carry it on your back'
cámam 'pack something'
cacámam 'be packing something'
camé<sup>2</sup>m' 'be carrying something'

One C<sub>2</sub>CT root (noted so far) has what seems to be a durative form:

| <i>łślqt</i> 'dip it, immerse it' | <i>lélqt</i> 'be dipping it'  |
|-----------------------------------|-------------------------------|
| <i>lálqìt</i> 'soak it'           | <i>télqit</i> 'be soaking it' |

The following T $\Rightarrow$ R $\Rightarrow$ T root has a durative with the last vowel stressed and a full vowel. DK gave a form with CVC reduplication:

 $\dot{x} \delta n \partial \dot{q}$  'open the eyes' (no prog. recorded)  $\dot{x} \partial n \dot{x} \partial n \dot{q}$  'keep the eyes open'

JP and AG gave the same simple form but differed in the progressive and durative:

*žónoqt* 'open them (the eyes)' *žožánoqt* (AG) 'be opening them from time to time' *žonáqt* (JP), *žonéqt* (AG) 'keep the eyes open'

Another verb with a probable durative form is:

 $2\dot{a}\cdot\dot{t}$  'go/come aboard' (no prog. recorded)  $2\partial\dot{l}\dot{a}^{2}a\dot{t}$  'aboard' A derivative  ${}^{2}\partial l \dot{\partial} l e^{2}ct$  'load it up' (<  $-\partial l e^{2}c$  'vessel') implies an /l/ in the root and suggests an underlying form, // ${}^{2}al\partial l$ //, for  ${}^{2}\dot{a}\cdot l$  (cf. //spal $\partial l$ // for  $sp\dot{a}\cdot l$  'raven').

What appear to be durative progressive forms occur as terms for recreational activities:

xixk<sup>w</sup>ám' 'be in swimming' (cf. xák<sup>w</sup>əm 'bathe,' xáx<sup>w</sup>ək<sup>w</sup>əm 'be bathing')
k<sup>w</sup>ik<sup>w</sup>əlíl 'be playing hide-and-seek' (cf. k<sup>w</sup>e·l 'hide,' k<sup>w</sup>ək<sup>w</sup>í·l 'be hiding')
cic Xím 'be having a jumping contest' (cf. cXám 'jump,' céXəm 'be jumping')
xix pám' 'be whistling for amusement' (cf. xápəm 'whistle,' xáxəpəs 'be whistling at him')

Durative stems (or what appear to be durative stems) also take the reflexive and reciprocal suffixes  $-\theta \partial t$  'oneself' and  $-t\partial l$  'each other,' as in:

 $qiq\partial w \dot{a}\theta \partial t$  'be sun-bathing' (cf.  $q\dot{a}w\partial t$  'warm it,'  $q\dot{a}w\partial \theta \partial t$  'warm oneself')  $\theta \partial \theta \dot{x} (t\partial l$  'push each other' (cf.  $\theta \dot{x} \dot{a}t$  'push him away,'  $\theta \partial \theta \dot{x} \dot{e}t$  'keep it pushed away')

 $mim\vec{\lambda}(t)$  'be paying each other back, be getting even' (cf.  $m\vec{\lambda}\vec{\lambda}$  'repay him')  $\vec{q} \cdot \vec{q} \cdot \vec{x}$  (insult each other' (cf.  $\vec{q} \cdot \vec{x} \cdot \vec{t}$  'insult him,'  $c\vec{q} \cdot \vec{x}$  'black')

 $xix\dot{\lambda}i\theta \partial t$  'be bragging about one's (a male's) sexual experiences' (lit. 'be

keeping oneself in a state of tumescence, ' <  $x \dot{a} \dot{\lambda}$  'get an erection')

There are a few words that I suspect may be durative forms but cannot positively identify as such because I have not yet recorded their roots in other forms:

| <i>čtém</i> 'hear'                                                                                                                  | <i>čí čtém</i> 'be hearing'                                                                              |
|-------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------|
| <i>žłém</i> 'watch'                                                                                                                 | <i>žižłém</i> 'be watching'                                                                              |
| <i>ləhél</i> 'play slahal'                                                                                                          | <i>líləhel</i> 'be playing slahal'                                                                       |
| $m \partial y \hat{a}^{2}t$ 'come down in price'                                                                                    | həmyá?t 'be coming down in price'                                                                        |
| $\check{x}^w \partial \check{x}^w \partial^2 \acute{e}t$ 'imitate him, do as he does'                                               | $\check{x}^{w}i\check{x}^{w}\partial^{2}\acute{e}t$ 'be imitating him'                                   |
| <i>x̃wəx̃wə?éqət</i> 'imitate his speech,<br>repeat his words'                                                                      | $\check{x}^{w}i\check{x}^{w}\partial^{2}\acute{e}q\partial n$ 'be repeating'                             |
| $\dot{q}^{w} \partial \dot{\bar{q}}^{w} \partial n \dot{e} m$ 'have one's voice change<br>(as a boy at puberty)'                    | $\vec{q}^{w} i \vec{q}^{w} \partial n \hat{e} m$ 'be having'<br>(< $\vec{q}^{w} i \cdot \vec{n}$ 'ear'?) |
| <i>ž</i> <sup>w</sup> <i>əyém</i> 'tell a story' (JP), <i>ž</i> <sup>w</sup> <i>iž</i> <sup>w</sup> <i>əyém</i> 'tell stories' (CC) | 'be telling a story' (JP), <i>x</i> <sup>w</sup> ∂ <i>x</i> <sup>w</sup> ∂ <i>y</i> ém                   |
| $\dot{c}i\dot{c}k^{w}\dot{e}^{\gamma}t$ 'have the hiccups' (progressiv                                                              | ve?)                                                                                                     |

 $t^{\theta}it^{\theta}tit$  'be barely hanging on with the teeth or fingers' (DK) (progressive?)

#### 7.5.1. Plural Duratives

Durative forms may be pluralized and diminutivized. As with non-duratives, different root types follow different patterns in forming plurals. Duratives formed from TAT, TAH, and probably some other root types are made plural

with an /1/ infix, as in the following examples (recorded duratives on the first line, plural duratives on the second):

 $k^{w} \partial k^{w} c \acute{em}$  'be screaming'  $k^{w} i k^{w} c \acute{em}$  'be screaming continuously'  $k^{w} \partial l \partial k^{w} c \acute{em}$  'several are screaming'

(no form recorded)  $xix\dot{q}\acute{e}m$  'have your mouth open'  $x\partial x^w\acute{e}m$  'several have their mouths open'

 $\dot{c}\partial^2 \dot{e}t$  'leave it sitting' (no form recorded)  $\dot{c}\partial \partial \dot{c}\dot{e}t$  'leave them sitting'

(The root of the first example is  $\sqrt{k^wec}$ . AG gave  $wak^wec$  'scream suddenly.' JP, DK, and AG all identified  $k^wecam$  as 'scream' and  $k^wak^wcem$  as 'be screaming,' but the latter resembles a durative rather than a progressive. AG said there might be a  $k^weckwacam$ , the expected progressive of  $k^wecam$ , but he had not heard it.)

Duratives formed from T<sub>9</sub>R roots are made plural by CVC reduplication, as in:

 $k^{w} \partial n \dot{e} t$  'have it, hold it, be holding it'  $k^{w} \partial n k^{w} \partial n \dot{e} t$  'hold several, several hold it'

*c'əmét* 'hold it in your mouth, be holding it in your mouth' *c'əmc'əmét* 'several are holding them in their mouths' (DK) (probably *c'əmc'əmét*)

# 7.5.2. Diminutive Duratives

Diminutives of duratives, to judge from the few examples I have, are formed like diminutives of non-durative forms, with CV reduplication with an /i/ vowel and with an /l/ infix for plural, as in:

| <ul> <li>kwanét 'have it, be holding it' (dur.)</li> <li>kwankwanét 'have several, several have it' (dur. pl.)</li> </ul> | k <sup>wi</sup> ?k <sup>w</sup> ∂nét 'have it' (dim. dur.)<br>k <sup>w</sup> ∂lik <sup>w</sup> ∂nét 'have several'<br>(dim. dur. pl.) |
|---------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|
| comét 'hold it in one's mouth' (dur.) coméomét 'several are holding it in their mouths' (dur. pl.)                        | <pre>(no dim. dur. recorded) colicomét 'the little ones are holding   it in their mouths' (DK) (dim. dur.   pl.)</pre>                |

Because the diminutive occurs only in the progressive aspect, the existence of diminutives of these forms shows that they do function as progressives.

# 7.6. THE DISPOSITIONAL-ITERATIVE ASPECT

A number of roots have a form that has two closely related senses. When used without suffixes, it indicates that there is an inclination or propensity toward the action of quality expressed by the root; when used with suffixes, it indicates that the action occurs repeatedly. For these forms with the first sense, the term

"dispositional" or "potential' may be useful, while for the second sense the term "iterative' seems appropriate.<sup>4</sup>

The dispositional-iterative form is produced by CVC reduplication with both vowels schwa (regardless of the vowel of the root) and the stress on the first (C<sub>1</sub>AC<sub>2</sub>, etc.  $\rightarrow$  C<sub>1</sub> $\Rightarrow$ C<sub>2</sub>C<sub>1</sub> $\Rightarrow$ C<sub>2</sub>). For example, from *qéň* 'steal' comes *qáňqaň* 'thief, likely to steal'; from  $k^w \partial t$  'spill, capsize' comes  $k^w \partial t k^w \partial t$  'cranky, likely to capsize.'

These forms, it should be noted, may differ from some plural forms of some roots only in where the stress falls. For example, compare  $s \neq \dot{q} s \neq \dot{q} \cdot \dot{q} s$  to split, easy to tear' and  $s \neq \dot{q} s \neq \dot{q} \cdot \dot{q} s$  to tear' into shreds, tear several,' both from  $s \neq \dot{q} \cdot \dot{q} \cdot \dot{q}$  to reacked.'

Words consisting of the bare root reduplicated in this fashion can have the meaning (someone/something) disposed/inclined/likely to do whatever the root means. Often the first gloss given is an English noun or noun phrase, but adjectives also appear, and verbs with the sense of habitual action, as in the following (a warning about an unwelcome visitor):

(a) syá·čx<sup>w</sup> ?ə θəwẳa. wəyáθ wəck<sup>w</sup>ék<sup>w</sup>əc. ck<sup>w</sup>əck<sup>w</sup>əc. Ճəw qə́nqən. syá·čx<sup>w</sup> ?ə θəwẳa wə-yáθ wə-c-k<sup>w</sup>ék<sup>w</sup>əc be.alerted you OBL DEM(FP) EST-always EST-get-be.looking c-k<sup>w</sup>əck<sup>w</sup>əc Ճe wə-qə́nqən get-look(DISP) also EST-steal(DISP)
'Watch out for her. She's always peeking. She's a snooper. She also steals.'

water out for her. She's arways peeking, she's a shooper. She also stears.

Here  $c\vec{k}^{w}\vec{e}\vec{k}^{w}ac$  'be peeking' (from  $c\vec{k}^{w}\vec{e}c$  'catch sight of something') is in the progressive aspect, while  $c\vec{k}^{w}\vec{a}c\vec{k}^{w}ac$  'snooper, snoopy' and  $q\vec{a}nqan$  'thief, likely to steal' are dispositional-iterative forms.

Other such forms with similar meanings include:

 $q^{w} \delta l q^{w} \delta l$  'talkative, outspoken' ( $\langle q^{w} \ell l$  'speak,' but cf.  $q^{w} \delta l q^{w} \delta l$  'tell')  $w\delta t^{\theta} w_{\delta} t^{\theta}$  'easily offended' ( $\langle w\ell t^{\theta}$  'become offended')  $n\delta tn_{\delta} t$  'willing, co-operative' (cf.  $n\ell t_{\delta} t$  'decide it, set a time for it')  $t\delta m t_{\delta} m$  'cautious, wary' (cf.  $t\delta m_{\delta} \theta_{\delta} t$  'be wary, be suspicious')  $n\delta cn_{\delta} c$  'changeable' ( $\langle n\ell c$  'different,' cf.  $n\ell c t$  'change it')  $q\delta x q_{\delta} x$  'slippery' (cf.  $q\ell x_{\delta} t$  'make it slide,'  $qt x_{\delta} \theta \ell t_{\delta} m$  'I slipped,' lit. 'I was slid')  $k^{w} \delta x k^{w} \delta x$  'lascivious, someone who always wants sex' ( $\langle k^{w} \ell x \rangle$  'copulate')  $t\delta k^{w} t_{\delta} k^{w} \delta n$  'catching, contagious' (cf.  $k^{w} \delta n\ell t_{\delta} m$  'transfer,'  $k^{w} \delta nw \ell t$  'transfer it, as from one canoe to another')

máťmať 'limber' (cf. maťát 'bend it')

<sup>4</sup> Earlier, I called this simply "dispositional." It is the "potential" of Leslie (1979, 42) and the "generalizing aspect" of Thompson and Thompson's description (1971, 279) of Clallam.

A few such dispositional forms with similar meanings seem to require a prefix – *s*- 'nominalizer,'  $\delta x^{w}$ - 'oblique nominalizer,' or  $x^{w}s$ - 'no $x^{w}s$ - 'habitual':

 $\delta x^w \delta k^w x^w \partial k^w$  'drunkard' (cf.  $x^w \delta k^w \partial t$  'polish it' and its resultative  $\delta x^w \delta x^w \partial k^w$  'polished, drunk'; thus 'someone who is inclined to get "polished"))

 $\delta x^w n \delta y n \delta y'$  'someone who laughs at the least little thing' (cf.  $n \delta y \delta m$  'laugh')  $x^w s^2 \delta n \delta n t$  'obedient' (<  $2 \delta \cdot n t$  'obey, agree')

Another word that may belong here is  $xs^2 \delta t \partial t \partial t$  'shrew' (<  $2t \partial t$  'sleep,' lit. 'inclined to sleep' or possibly 'cause sleep,' probably because the animal is said to be found sleeping beside the trail and to be useful as a charm under a cradle to make a child sleep (CC). (The initial xs- is unique in my corpus. The x- may be CC's variant form of  $\delta x^w$ -. See §12.1.4.)

Other such forms seem to have specialized meanings that are not easily linked to the preceding ones, such as:

*tox*<sup>w</sup>*tox*<sup>w</sup> 'move one's residence down to the shore' (< *táx*<sup>w</sup> 'go/come down ...') *cómcom* 'move one's residence up away from the shore' (< *cám* 'go/come up ...')

cák<sup>w</sup>cak<sup>w</sup> 'low tide' (< cák<sup>w</sup> 'far') sxámxam' 'low tide, low water' (< xém 'go aground')</pre>

cówcow 'out on the water, the offing' (cf. cécow 'shore'?)

 $q^{w}\delta \dot{l}q^{w}\delta \dot{l}$  'tell, relate' (<  $q^{w}\ell i$  'speak')

?álalya 'dream' (< ?álya 'have a vision')</pre>

The forms  $q^w \delta l q^w \delta l$  'talkative' and  $q^w \delta l q^w \delta l$  'tell' are the only pair contrasting dispositional and possibly iterative meaning.

Dispositional-iterative forms are also used as stems for various suffixes, generally with an iterative meaning, as in:

- $\dot{t}\delta k^{w}\dot{t}\partial k^{w}\partial l$  'cross back and forth' (as in  $\dot{t}\delta k^{w}\dot{t}\partial k^{w}\partial l \delta x^{w}[h]\delta \dot{y}q^{w}\partial l\partial t$  'ferry,' lit. 'steamer that crosses back and forth,' <  $\dot{t}\delta k^{w}\partial l$  'cross over,' cf.  $\dot{t}\delta k^{w}\partial l$  'be crossing over,' prog.)
- $y \partial \check{x} \partial \check{x} \partial \check{x} \partial \check{x} \partial \partial t$  'be zigzagging along' (cf.  $\check{x} \partial \check{x} \partial t$  'cross him, contradict him,'  $\check{x} \partial i \partial \partial t$  'turn aside, swing around')
- $t^{\theta_{\beta}}\dot{p}t^{\theta_{\beta}}\dot{p}n_{\partial x^{w}}$  'keep opening and shutting the eyes' (cf.  $t^{\theta_{\beta}}\dot{p}n_{\partial x^{w}}$  'shut one's eyes,'  $w_{\partial t}^{\theta_{\ell}}\dot{t}t^{\theta_{\beta}}\dot{p}n_{\partial x^{w}}$  č $x^{w}$  'al 'keep your eyes shut,' prog.)
- xáqxaqam 'Sansum Narrows,' lit. 'keeps opening and shutting its mouth,' in Symplegades fashion in a myth (cf. xéqam 'open one's mouth,' xixqém 'have one's mouth open,' dur. prog.)
- k<sup>w</sup>óck<sup>w</sup>ocnox<sup>w</sup> 'see him from time to time' (cf. k<sup>w</sup>ócnox<sup>w</sup> 'see him,' k<sup>w</sup>ék<sup>w</sup>ocnox<sup>w</sup> 'be seeing him')
- cówcowot 'help him from time to time' (cf. céwot 'help him,' cécowot 'be helping him')

- *ċəwċəwítəl* 'help each other on repeated occasions' (cf. *ċəċəwítəl* 'help each other,' dur.)
- *k*<sup>*w*</sup>*ðck*<sup>*w*</sup>*ðc∂m* 'scream from time to time' (cf. *k*<sup>*w*</sup>*éc∂m* 'scream,' *k*<sup>*w*</sup>*∂k*<sup>*w*</sup>*cém* 'be screaming continuously,' dur.)
- $t \delta \vec{k}^w t \delta \vec{k}^w x \delta n$  'be tripping repeatedly' (<  $t \delta \vec{k}^w x \delta n$  'trip,' <  $t t \vec{k}^w$  'get snagged,'  $x \delta n$  'foot')
- $q^{w} \partial l q^{w} \partial l \dot{a} \cdot y \partial \partial n$  'outspoken' (JP), 'blabbermouth (CC) (< - $\dot{a} \cdot y \partial \partial n$  'mouth')

# 7.6.1. Progressives of Dispositional-Iterative Forms

For most of these dispositional-iterative forms, a separate progressive or a perfective/progressive distinction would seem unnecessary. Nevertheless, it seems that a progressive dispositional-iterative form does exist, although perhaps only for those that are specialized in meaning such that they have no iterative force. One such is  $q^w \delta l q^w \delta l$  'tell, report';  $q^w t l q^w \delta l$  'be telling, be reporting.' This progressive dispositional form occurs fairly often nominalized in the sense of 'what X has been telling' or, freely translated, 'according to X,' as in  $sq^w t l q^w \delta l s$  $k^w \theta \delta sy \delta y \delta l \delta x^w a t$  'according to the old people.'

The only other progressive dispositional forms recorded are:

- *mí tmətθət* 'be getting limber' (< *mátmətθət* 'get limber,' < *mátmət* 'limber'; cf. *mətát* 'bend it') (AG)
- $\vec{k}^{wi} \cdot c\vec{k}^{w} \rightarrow c$  'be watching and waiting' (which AG gave but could not give a perfective for, but cf.  $c\vec{k}^{w} \rightarrow c$  'snooper')

Thus, it appears that the formula for producing the progressive dispositional from the perfective dispositional may be  $C_1 \Rightarrow C_2 C_1 \Rightarrow C_2 \rightarrow C_1 i \cdot C_2 C_1 \Rightarrow C_2$ .

# 7.6.2. Plurals of Dispositional-Iterative Forms

Dispositional forms can be made plural with an *-l-* infix. There seems to be variation in vowel and stress. Compare the following singular (left column) and plural (right column) forms:

| qánqan 'thief'                                                            | nax <sup>w</sup> sqalánqaň (DK), qélaňqaň (AG) 'thieves'                                                        |
|---------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|
| $q^{w} \delta \dot{l} q^{w} \delta \dot{l}$ 'tell'                        | $q^{w} \partial l \partial l q^{w} \partial l$ 'several tell' (AG)                                              |
| $q^{w} \hat{\iota} \cdot \hat{l} q^{w} \hat{\imath} \hat{l}$ 'be telling' | $q^{w} \partial l i \cdot l q^{w} \partial l$ 'several are telling' (AG)                                        |
| $w \dot{i}^{\theta} w \dot{i}^{\theta}$ 'easily offended'                 | $w \acute{e} \vec{l} \imath \acute{t}^{\theta} w \imath \acute{t}^{\theta}$ 'several easily offended ones' (AG) |
| <i>máťmať</i> 'limber, springy'                                           | <i>mél atmat</i> 'several limber ones' (AG)                                                                     |
| <i>ąó žąož</i> 'slippery'                                                 | <i>qéləxqəx</i> 'several slippery ones' (AG)                                                                    |

# 7.6.3. Diminutives of Dispositional-Iterative Forms

The only diminutives I have recorded for dispositional forms are of  $q^{wi} \cdot l q^{w} \partial l$ 'be telling' as given by AG. They are formed as are other diminutives of verbs. The singular diminutive is  $q^{wi} \cdot l q^{w} \partial l$  'the little guy is telling' and the plural is  $q^{w} \partial l i^{2} q^{wi} \cdot l q^{w} \partial l$  'the little guys are telling.'

#### 7.7. RESULTATIVE FORMS

Many verbs, though not all, have resultative forms.<sup>5</sup> These denote the result, product, or finished state of the activity denoted by the base form of the verb. Resultative forms can often be translated with English past participles or adjectives, for example,  $s\dot{p}\dot{e}\dot{p}\dot{a}\dot{t}^{\theta}$  sewed' (cf.  $\dot{p}\dot{e}\dot{t}^{\theta}$  sew,'  $\dot{p}\dot{e}\dot{t}^{\theta}\dot{a}t$  'sew it'),  $sn\dot{n}\dot{a}\dot{w}$  'advised' (cf.  $n\dot{w}\dot{a}t$  'advise him'),  $sh\dot{a}\dot{v}$  'finished, ready' (cf.  $h\dot{a}y$  'finish'),  $stat\dot{e}s$  'near' (cf.  $t\dot{a}s$  'arrive there,'  $ts\dot{a}t$  'approach it'),  $s\theta\partial\theta\dot{e}\dot{k}^{w}$  'straight' (cf.  $\theta\dot{k}\ddot{w}\dot{a}t$  'pull it'),  $s\dot{t}^{\theta}\ddot{x}\ddot{w}\dot{e}cas$  'with washed hands' (cf.  $\dot{t}^{\theta}\ddot{x}\ddot{w}\dot{e}csam$  'wash one's hands'),  $s\ddot{x}xan\dot{a}p$  'plowed' (cf.  $\dot{x}\dot{x}an\partial p$  'plow a field').

The word  $s\dot{\rho}\dot{e}\dot{\rho}\dot{a}\dot{t}^{\theta}$  is also used in  $s\dot{\rho}\dot{e}\dot{\rho}\dot{a}\dot{t}^{\theta}m\dot{a}\dot{s}in$  'sewing machine' and  $s\dot{\rho}\dot{e}\dot{\rho}\dot{a}\dot{t}^{\theta}\dot{x}^{w}ilam$  'sewing thread,' but this does not seem to be a usual usage of resultative forms; these may be simply translations of the English terms.

Resultatives are ordinarily formed by prefixing  $s_{-2}$  'resultative' (§12.1.2) to an internally modified form of the root or root with suffixes. The type of internal modification varies with the type of root; the resultatives of CAC and CaCC roots are modified as they are for progressives, but the resultatives of CaC roots differ from their progressives in having a full second vowel instead of a schwa. (Details appear below.)

The suffixes that may appear in resultatives include  $-\partial m$  'intransitive,'  $-\partial l$ 'move toward,' and a number (as yet undetermined) of lexical suffixes. The transitive suffixes -t,  $-n\partial x^w$ , and -x, and the reflexive and reciprocal suffixes, do not appear in resultative forms. On the other hand, resultatives may be transitivized with the suffix  $-st\partial x^w_l$  'causative' (see §9.3).

Resultatives do not share the perfective-progressive distinction found elsewhere in the aspect system, but they do have plural, diminutive, and diminutive plural forms.

A few forms that appear to be resultatives do not take the *s*- 'resultative' prefix. There are also a few resultatives that appear with a suffixed *-t* 'stative' (\$12.1.6).

#### 7.7.1. Resultatives of CAC and CaCC Roots

Resultatives of roots of these shapes are regularly produced by prefixing s'resultative' to a form that is identical with the progressive of the bare root (or
with what we may infer that form would be from the progressive of the transitive). Compare the following sets:

| PERFECTIVE                             | PROGRESSIVE                                                | RESULTATIVE                                    |
|----------------------------------------|------------------------------------------------------------|------------------------------------------------|
| $\dot{p}\dot{e}\dot{t}^{\theta}$ 'sew' | $\vec{p}\vec{e}\vec{p}\vec{a}\vec{t}^{\theta}$ 'be sewing' | <i>spépət<sup>e</sup></i> 'sewed'              |
| <i>yák<sup>w</sup>ət</i> 'smash it'    | yáý akwat 'be smashing it'                                 | syáýak <sup>w</sup> 'smashed'                  |
| <i>q<sup>w</sup>ím</i> 'disembark'     | $\dot{q}^{w} \dot{q}^{w} \partial m$ 'be disembarking'     | <i>sq<sup>w</sup>íq<sup>w</sup>əm</i> 'ashore' |

<sup>5</sup> I have adopted the term used by Leslie (1979). These are the "participles" of Galloway (1977) and they correspond to the "stative" forms in the Thompsons' description (1971) of Clallam.

| níwət 'advise him'                        | <i>ní ňəwət</i> 'be advising him'                                | sníňow 'advised'                                      |
|-------------------------------------------|------------------------------------------------------------------|-------------------------------------------------------|
| <i>cé<sup>9</sup>ant</i> 'straighten it'  | <i>céconot</i> 'be straightening it'                             | sčéčaň 'straight'                                     |
| <i>xák<sup>w</sup>ət</i> 'bathe him'      | $x \acute{a} x^w \partial \vec{k}^w \partial t$ 'be bathing him' | <i>sxáx<sup>w</sup>ək<sup>w</sup></i> 'bathed'        |
| <i>híqət</i> 'put it under'               | $hi^{2}q  arrow t$ 'be putting it under'                         | $si^2q$ 'underneath'                                  |
| háy 'finish'                              | $h \dot{a} \cdot \dot{y}$ 'be finishing'                         | shá·ý 'finished, ready'                               |
| ?áť 'stretch'                             | ${}^{2}\dot{a}{}^{2}\dot{t}$ 'be stretching'                     | $s^2 \dot{a}^2 \dot{t}$ 'stretched out'               |
| $\dot{t}\dot{e}^{2}t$ 'try it'            | $i \partial i e^{\gamma} t$ 'be trying it'                       | sť at é? 'tried'                                      |
| sé <sup>?</sup> 'rise'                    | səsé? 'be rising'                                                | ssəsé? 'raised'                                       |
| <i>čé<sup>?</sup></i> 'land atop'         | <i>ċəċé</i> <sup>?</sup> 'be landing atop'                       | scoce? 'atop'                                         |
| <i>x<sup>w</sup>ét</i> 'lower it'         | $x^{w} \partial x^{w} \dot{e} t$ 'be lowering it'                | $\check{s}x^w \partial x^w \acute{e}$ 'down, lowered' |
| <i>x<sup>w</sup>pá·t</i> 'inflate it'     | $x^{w}p \partial p \dot{a}^{2} \partial t$ 'be inflating it'     | <i>šx<sup>w</sup>pəpá</i> 'inflated'                  |
| $m e^{i x}$ 'remove it'                   | $h \delta \dot{m} e^2 x$ 'be removing it'                        | səme? 'off'                                           |
| <i>mátq<sup>w</sup>t</i> 'immerse it'     | $m \acute{a} t q^w t$ 'be immersing it'                          | <i>smátq</i> <sup>w</sup> 'immersed' (AG)             |
| <i>tálqt</i> 'dip it'                     | <i>télqt</i> 'be dipping it'                                     | <i>stélq</i> 'in the water'                           |
| $\dot{c}\dot{\delta}\dot{y}x^wt$ 'dry it' | $\dot{c}\dot{e}\dot{y}x^wt$ 'be drying it'                       | $s\dot{c}\dot{e}\dot{y}x^{w}$ 'dry, dried'            |
| • •                                       |                                                                  | · · · · · · · · · · · · · · · · · · ·                 |

#### 7.7.2. Resultatives of C<sub>2</sub>C Roots

Resultatives of roots of this shape are produced by prefixing *s*- 'resultative' to a form that resembles the progressive of the bare root (not the form with *t*- 'transitivizer') except that there is a full vowel between the reduplicated first consonant and the second consonant. Which full vowel will appear is not (at this stage of analysis) predictable. Compare the following sets exemplifying T<sub>∂</sub>T, T<sub>∂</sub>R, and R<sub>∂</sub>T roots:

| PERFECTIVE<br>tás 'arrive there'<br>tsát 'approach it'                                                        | PROGRESSIVE<br><i>tətás</i> 'be arriving there'<br><i>tést</i> 'be approaching it'                                                | RESULTATIVE<br>stətés 'near'                                             |
|---------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|
| <i>θák<sup>w</sup></i> 'get taut'<br><i>θk<sup>w</sup>ət</i> 'pull it'                                        | $\theta \partial \theta \partial \vec{k}^w$ 'be getting taut'<br>$\theta \vec{e} \vec{k}^w t$ 'be pulling it'                     | s0ə0ék <sup>w</sup> 'straight'                                           |
| <i>q<sup>w</sup>ðs</i> 'sink'<br><i>q<sup>w</sup>sðt</i> 'sink it'                                            | $q^{w} \partial q^{w} \delta s$ 'be sinking'<br>$q^{w} \delta s t$ 'be sinking it'                                                | <i>sq<sup>w</sup>əq<sup>w</sup>is</i> 'sunk'                             |
| $\dot{t}\partial \dot{q}^{w}$ 'break, snap'                                                                   | $i \partial i \partial q^w$ 'be breaking'                                                                                         | <i>statiq</i> <sup>w</sup> 'cut off, cut up, shear'                      |
| <i>ṫdʷát</i> 'cut it off'<br><i>sód</i> 'split, tear'<br><i>sdét</i> 'split it'                               | $\dot{t}\dot{a}\dot{q}^wt$ 'be cutting it off'<br>$s as \dot{s}\dot{q}$ 'be splitting'<br>$s \dot{e} \dot{q} t$ 'be splitting it' | ssəsíq 'split, torn'                                                     |
| $\dot{t}^{\theta} \dot{a} \check{x}^{w}$ 'get washed'<br>$\dot{t}^{\theta} \check{x}^{w} \dot{a} t$ 'wash it' | $i^{\theta}\partial i^{\theta}\partial x^{w}$ 'be getting washed' $i^{\theta}\partial x^{w}t$ 'be washing it'                     | $s\dot{t}^{\theta}\partial\dot{t}^{\theta}\dot{t}\check{x}^{w}$ 'washed' |
| <i>čáq<sup>w</sup></i> 'get pierced'<br><i>čq<sup>w</sup>át</i> 'pierce it'                                   | $\dot{c}\partial\dot{c}\dot{d}\dot{q}^{w}$ 'be getting pierced'<br>$\dot{c}\dot{a}\dot{q}^{w}t$ 'be piercing it'                  | <i>sčəčáų</i> <sup>w</sup> 'pierced, shot'                               |

| p                                                                                                 | $p  a p \delta n$ 'be getting buried'<br>$p a p \delta n \delta t$ 'be burying it'<br>$\dot{q}^w \partial \dot{q}^w \delta l$ 'be getting cooked'                         | spəpíň 'buried' sq̃ <sup>w</sup> əq̃ <sup>w</sup> íĺ 'cooked, ripe, well taught'               |
|---------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|
| <i>q̃<sup>w</sup>ə́lət</i> 'roast it'                                                             | $\dot{q}^{w} \partial \dot{q}^{w} \partial \dot{l} \partial t$ 'be roasting it'                                                                                           |                                                                                                |
| <i>k<sup>w</sup>ón</i> 'become<br>possessed'<br><i>k<sup>w</sup>ónət</i> 'take it'                | <ul> <li>k<sup>w</sup> <i>a</i>k<sup>w</sup> <i>á ň</i> 'be going into<br/>a trance'</li> <li>k<sup>w</sup> <i>a</i>k<sup>w</sup> <i>á ň at</i> 'be taking it'</li> </ul> | <i>sk<sup>w</sup>ək<sup>w</sup>ín</i> 'possessed,<br>in a trance'                              |
| <i>lámət</i> 'pick them'<br><i>qáwət</i> 'warm it'                                                | <i>ləlámət</i> 'be picking them'<br><i>qáwət</i> 'be warming it'                                                                                                          | <pre>stəłim 'picked' sqəqiw' 'heated beside the fire, exposed to the heat' (AG)</pre>          |
| <i>x<sup>w</sup>óy</i> 'wake up'                                                                  | <i>x<sup>w</sup>əx<sup>w</sup>ə́y</i> 'be waking up'                                                                                                                      |                                                                                                |
| $\theta \delta yt$ 'fix it'                                                                       | $\theta \dot{e} \dot{y} t$ 'be fixing it'                                                                                                                                 | <i>sθəθόỷ</i> ~ //s-θəθíỷ//?<br>'right'                                                        |
| məq' 'get full'<br>náq <sup>w</sup> 'fall asleep'<br>lák <sup>w</sup> 'break'<br>ná pax' 'eat it' | <ul> <li>hómq' 'be getting full'</li> <li>hónq<sup>w</sup> 'be falling asleep'</li> <li>hólk<sup>w</sup> 'be breaking'</li> <li>hónpox 'be eating it'</li> </ul>          | səmiq 'full'<br>səniq <sup>w</sup> 'asleep'<br>səlik <sup>w</sup> 'broken'<br>sənip 'eaten on' |

# 7.7.3. Resultatives of Other Tri-Consonantal Roots

A few other words that appear to be tri-consonantal roots of shapes other than C<sub>2</sub>CC form their resultatives in a way that does not reflect the progressive form. Their resultatives have a stressed full vowel between the second and third consonant. Compare the following sets:

| PERFECTIVE                                                | PROGRESSIVE                                                                       | RESULTATIVE                                                        |
|-----------------------------------------------------------|-----------------------------------------------------------------------------------|--------------------------------------------------------------------|
| <i>łé<sup>2</sup> x̃t</i> 'put it on a dish, serve it'    | $t \partial t e^{2} \tilde{x} t$ 'be putting it on<br>a dish, be serving it'      | <i>sła?éž</i> 'dished out, laid out<br>on a mat, served'           |
| $k^w i^2 x \partial t$ 'pitch it up'                      | $k^{w} \partial k^{w} i^{2} x^{w} t$ 'be pitching<br>it up'                       | $sk^w \partial^2 i x^w$ 'pitched up'                               |
| <i>x˜wá·ťt</i> 'brace it'                                 | $\check{x}^{w} \partial \check{x}^{w} \dot{a} \dot{t} t$ 'be bracing it'          | <i>s x̃<sup>w</sup> ə<sup>2</sup>át</i> 'braced'                   |
| $\check{x}\acute{e}\cdot\acute{t}^{\theta}t$ 'measure it' | $\check{x} \partial \check{x} \dot{e} \dot{t}^{\theta} \dot{t}$ 'be measuring it' | $s\check{x}\vartheta\acute{e}\check{t}^{\theta}$ 'marked, evident' |
| <i>tpáłt</i> 'stretch it'                                 | <i>tápəłt</i> 'be stretching it'                                                  | st páł 'stretched (as hide)'                                       |
| $itaq^w$ 'get twisted'                                    | $\dot{t}\dot{\delta t}\partial \dot{q}^w$ 'be getting twisted'                    | $s\dot{t}t\dot{a}\dot{q}^{w}$ 'aching from a twist' (AG)           |
| $k^{w}t\delta x^{w}$ 'enter'                              | (prog. not recorded)                                                              | <i>sk<sup>w</sup>téx<sup>w</sup></i> 'inside'                      |
| <i>x<sup>w</sup>álək<sup>w</sup>t</i> 'wrap it<br>up'     | $x^{w} \partial x^{w} \partial l \partial k^{w} t$ 'be<br>wrapping'               | <i>šx<sup>w</sup>əlák</i> <sup>w</sup> 'wrapped up'                |
# 7.7.4. Resultatives of Irregular Roots

Perhaps the only truly irregular roots are  $\theta \delta t$  'say' and  $\check{x}t\dot{e}$  'do,' which have the same progressive form,  $\check{x}\delta\dot{t}\delta$  'be saying, be doing,' and the same resultative,  $s\check{x}\dot{t}\dot{e} \sim s\check{x}\delta\dot{t}\dot{e}$ ? 'said that way, so-called, done that way, the same, prepared.'

# 7.7.5. Resultatives with Suffixes

There are a few words formed with the suffixes  $-\partial m$  'intransitive' and  $-\partial l$  'move toward' that have resultatives that incorporate the suffix. These resultatives resemble those of the set identified above as "other tri-consonantal roots" in that they have a stressed full vowel before the final consonant, in this case that of the suffix.

| PERFECTIVE                                                | PROGRESSIVE                                               | RESULTATIVE                                                            |
|-----------------------------------------------------------|-----------------------------------------------------------|------------------------------------------------------------------------|
| <i>táyam</i> 'stick'                                      | <i>títəyəm</i> 'be sticking'                              | sťəyám 'stuck'                                                         |
| <i>ģáləṁ</i> 'camp'                                       | (no prog. recorded)                                       | sqalím 'camped'                                                        |
| (no perf. recorded)                                       | qəqəməl 'be rising'                                       | sqəmíl 'high (the tide)'                                               |
| $q \vec{\lambda} \delta m$ 'bounce off,<br>be jarred off' | <i>qá Žəm</i> 'be bouncing off,<br>be getting jarred off' | sqŽím 'off' (AG)                                                       |
| náqəm 'dive'                                              | hánqam 'be diving'                                        | <i>sənqém</i> 'submerged,<br>under' (AG)                               |
| <i>pá·m</i> 'swell, rise'                                 | <i>pppá:m</i> 'be swelling'                               | s pohám 'swollen' (AG)                                                 |
| <i>yá·m</i> 'order'                                       | həyá·m 'be ordering'                                      | səyá·m 'ordered' (AG)                                                  |
| $ti^{?}$ become lonely'                                   | (no prog. recorded)                                       | sť a <sup>9</sup> íl' 'lonely'                                         |
| <pre>'éyəl 'get out of the<br/>way, leave'</pre>          | ?é·yəl 'be leaving'                                       | s <sup>2</sup> əyél (CC, JP), s <sup>2</sup> eyíl<br>(AG) 'away, gone' |
| híləm 'roll over,                                         | hílam 'be rolling over,                                   | <i>səhíl</i> 'fallen off, fallen                                       |
| fall off'                                                 | be falling off'                                           | down'                                                                  |
| <i>hí·lt</i> 'roll it over,                               | <i>hí?əİt</i> 'be rolling it                              |                                                                        |
| push it off'                                              | over'                                                     |                                                                        |

Lexical suffixes are also incorporated into resultatives, which vary in form according to type of root and type of suffix. The data are too little for a full account, but it seems that the only resultatives with suffixes in which the roots have the shapes that appear in their plain (non-suffixed) resultatives are those formed from TAT and C<sub>2</sub>CT roots and unstressed suffixes. In other resultatives with suffixes the roots appear in simpler shapes. Also, it seems that a few suffixes have forms that appear only in resultatives, such as  $-\partial n dp$  (otherwise  $-\partial n \partial p$ ) 'ground.'

The following examples illustrate the variations in type of root and suffix so far recorded. (There are so many gaps in the data, such as unrecorded perfective and progressive forms, that it seems better to simply list the resultatives and add whatever relevant data there are in parentheses rather than to tabulate as in previous sections.) (1) TAT roots with stress on the root:

- *stítacaq<sup>w</sup>* 'with one's hair cut short (as traditional for mourners)' (< *-aq<sup>w</sup>* 'head,' res. of *tícaq<sup>w</sup>am* 'get one's hair cut'; cf. *tícat* 'cut it,' *stítac*' 'cut')
- $\delta[x^w]titacas$  'with bow hauled up on the shore' (< -as 'face, forward end'; cf. ticat 'haul it up onto the shore,' stitac 'hauled up')
- $\delta[x^w]tit\partial cn\partial c$  'with stern hauled up on the shore' (<  $-n\partial c$  'rump, bottom, stern')
  - (2) TAT roots with stress on the suffix:
- *słoćnáp* 'disked, harrowed' (< -*ónop* 'ground,' res. of *łćónop* 'disk or harrow a field'; cf. *łćot* 'cut it') (AG)
- $\delta x^{w} \partial \partial t \dot{c} d \partial x^{w} \partial \partial t$  'kerfed and bent' (<  $-\dot{e} \partial x^{w} \partial \partial t$  'tongue,' res. of  $x^{w} \dot{t} \dot{c} \dot{e} \partial x^{w} \partial \partial t$ 'kerf it [wood for a bent-wood box]'; cf.  $t \dot{t} \dot{c} \partial t$  'cut it') (JP)

(3) HAT roots with stress on the root:

- $\delta x^{w?} \delta^2 \dot{p} \partial s$  'with wiped face' (<  $-\partial s$  'face,' res. of  $x^{w?} a \dot{p} \partial s \partial m$  'wipe one's face'; cf.  $\partial e \dot{p} \partial t$  'wipe it') (AG)
  - (4) TAH roots with stress on the suffix:
- $\delta x^w q^w \partial n \acute{e}c$  'with a hole in the bottom' (<  $-n \partial c$  'bottom,' res. of  $x^w q^w \acute{e}^2 n \partial ct$ 'make a hole in the bottom of it'; cf.  $q^w \acute{e} \cdot t$  'make a hole in it,'  $s q^w \partial q^w \acute{e}$  'with a hole in it')
- $\delta x^w \partial n \dot{e} c$  'sitting on the floor' (<  $-n \partial c$  'bottom'; cf.  $x^w \dot{e} t$  'lower it,'  $\delta x^w \partial x^w \dot{e}$  'lowered, down')
  - (5) T<sub>2</sub>T, T<sub>2</sub>R, and R<sub>2</sub>T roots seem all to have the stress on the suffix:
- $st^{\partial} \check{x}^{w} \acute{e} c \rightarrow s$  'with clean hands' (<  $-c \rightarrow s$  'hand,' res. of  $t^{\partial} \check{x}^{w} \acute{e} c \rightarrow m$  'wash one's hands'; cf.  $t^{\partial} \check{x}^{w} \acute{a} t$  'wash it,'  $st^{\partial} \partial t^{\partial} t \check{x}^{w}$  'washed')
- $st^{\theta}\check{x}^{w} \partial x \acute{en}$  'with clean feet' (< -*x* $\partial n$  'foot, leg,' res. of  $t^{\theta}\check{x}^{w} \partial x \acute{en} \partial m$  'wash one's feet') (AG)
- $st\dot{q}^w \acute{e}c \partial s$  'missing a finger' (<  $-c \partial s$  'hand'; cf.  $t\dot{q}^w \acute{a}t$  'cut it off,'  $st\dot{d}t\dot{q}^w$  'cut off, shear')

 $st \dot{q}^w x \acute{e}n$  'missing a leg' (< -x i an 'foot, leg') (JP)

- $s \vec{\lambda} x^{wi} c \partial n$  'covered' (<  $-\partial c \partial n$  'surface,' res. of  $\vec{\lambda} x^{wi} c t$  'cover it'; cf.  $\vec{\lambda} x^{w} a t$  'cover him,'  $s \vec{\lambda} \partial \vec{\lambda} e x^{w}$  'covered')
- s*xənap* 'plowed' (AG) (< -ənəp 'ground,' res. of *xə́nəp* 'plow'; cf. *xét* 'open it')
- spaňás 'with earth thrown in one's face, besprinkled ritually with ochre, feathers, etc.' (< -as 'face,' res. of pánast 'throw earth in his face, besprinkle him with ochre, etc.'; cf. pánat 'bury it,' spapín' 'buried')</p>
- $s\dot{q}^{w}\partial m\partial n\dot{a}p$  'weeded' (<  $-\partial n\partial p$  'ground,' res. of  $\dot{q}^{w}\partial m\partial n\partial p$  'pull up weeds'; cf.  $\dot{q}^{w}\partial m\partial t$  'pluck it')

sèýqás 'sharpened' (< ->s 'face,' res. of yáq>st 'sharpen it, grind it, file it'; cf. háýq>st 'be sharpening it'; cf. yáq>t 'grind it up')

- (6) C<sub>2</sub>CT roots with stress on the root:
- stélqəlwe<sup>2</sup>s 'with one's paddle in the water' (<-álwe<sup>2</sup>s 'paddle,' res. of təlqálwe<sup>2</sup>səm 'hold one's paddle in the water'; cf. tálqt 'dip it,' stélq 'in the water')
- sxálcaq<sup>w</sup> 'silly, crazy' (< -aq<sup>w</sup> 'head'; cf. xálct 'twist it, rotate it,' sxalxélc' 'twisted [plural]')

 $s \check{x} \acute{a} l \acute{c} \partial y \partial \theta \partial n$  'crooked-jawed (salmon)' (< - $\acute{a} y \partial \theta \partial n$  'mouth')

#### 7.7.6. Anomalous Resultative Forms

At least two roots have resultatives formed in the expected way but with the addition of the suffix -t 'stative.' Compare the following:

| PERFECTIVE                        | PROGRESSIVE                                        | RESULTATIVE                                  |
|-----------------------------------|----------------------------------------------------|----------------------------------------------|
| <i>k<sup>w</sup>élx</i> 'hide it' | $k^{w} e k^{w} \partial l x$ 'be hiding it'        | sk <sup>w</sup> ék <sup>w</sup> əlt 'hidden' |
| <i>k<sup>w</sup>é·l</i> 'hide'    | $k^{w} \partial k^{w} i \cdot \dot{l}$ 'be hiding' |                                              |
| <i>pé·lt</i> 'skim it'            | pépələt 'be skimming it'                           | spépəlt 'skimmed'                            |

There are a few forms that seem to be resultatives without the *s*-prefix.

| $\dot{p}\delta k^w$ 'rise to the | $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ $\dot{p}$ | <i>popék</i> <sup>w</sup> 'floating, afloat' |
|----------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------|
| surface'                         | surface' (AG)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | (CC, JP, AG)                                 |
|                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | s <i>pəpek</i> "afloat' (CC,                 |
|                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | DK)                                          |

CC gave both  $\dot{p}\partial\dot{p}\dot{e}k^w$  and  $s\dot{p}\partial\dot{p}\dot{e}k^w$  'floating, afloat.' AG denied the form with *s*- but gave a causative,  $\dot{p}\partial\dot{p}\dot{e}k^wst\partial x^w$  'keep it afloat,' which supports the identification of  $\dot{p}\partial\dot{p}\dot{e}k^w$  as a resultative.

| <i>Žák<sup>w</sup>ən 'go out</i> | <b>X</b> a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x  a  x | 'be going out' | <i>Řék<sup>w</sup>ən</i> 'out, |
|----------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------|--------------------------------|
| (as a fire)'                     |                                                                                                                                                                                                                                                                                                                                                   |                | extinguished'                  |
| <i>Žákwant</i> 'put it out'      | Х́әҲ́ә́к <sup>w</sup> әńt                                                                                                                                                                                                                                                                                                                         | 'be putting    |                                |
|                                  | it out'                                                                                                                                                                                                                                                                                                                                           |                |                                |

The form  $\lambda \hat{e} k^w \partial n$ , with its stress on the first vowel, does not resemble either a resultative or a durative, but semantically it seems to fit the resultative slot.

The following verb has two possible resultative forms:

| θqénx  | 'stand it up | <i>θáqnəx</i> | 'be standing it up' | <i>sθqén</i> 'held upright'                  |
|--------|--------------|---------------|---------------------|----------------------------------------------|
| in the | ground'      |               |                     | $\theta q \acute{e}t$ '1. be spitted upright |
|        |              |               |                     | by a fire; 2. tree, mast'                    |

 $s \ge lk^w a'^2 a y \theta \ge n$  'with a broken mouth' (<  $-a' y \ge \theta \ge n$  'mouth,' res. of  $l \ge k^w a' \cdot y^w \ge n$ 'get one's mouth broken'; cf.  $l \le k^w$  'get broken, break,'  $h \ge lk^w$  'be breaking,'  $l \ge k^w a'$  t'break it,'  $h \ge lk^w \ge t$  'be breaking it,'  $s \ge l(k^w)$  'broken')

The first,  $s\theta q en$ , is more like what one might expect; the second,  $\theta q et$ , is clearly based on the same root, but hard to explain.

# 7.7.7. Plurals of Resultatives

In general, plural forms of the resultative resemble plural progressives as simple (non-plural) resultatives resemble simple progressives. (Or so it seems, but there are great gaps in the data; plural resultatives are somewhat easier to elicit than plural progressives, but for many verbs I have neither form.)

Resultatives formed from TAT and TAR roots are made plural by CVC reduplication of the root. Compare the following perfective, resultative, and plural resultative forms:

| PERFECTIVE                          | RESULTATIVE                                 | PLURAL RESULTATIVE                                                  |
|-------------------------------------|---------------------------------------------|---------------------------------------------------------------------|
| $\dot{q}^{w}$ ím 'disembark'        | sqwíqwəm 'ashore'                           | <i>sq<sup>w</sup>ímq<sup>w</sup>əm</i> 'several ashore'             |
| <i>łíkwət</i> 'hook it'             | stítak <sup>w</sup> 'hooked'                | <i>słík<sup>w</sup>łək<sup>w</sup></i> 'several hooked'             |
| <i>qík<sup>w</sup></i> 'get bitten' | <i>sqíq́ək</i> <sup>w</sup> 'bitten (in one | $s\dot{q}i\dot{k}^{w}\dot{q}\partial\dot{k}^{w}$ 'bitten up, bitten |
|                                     | place)'                                     | in several places'                                                  |
| <i>pá</i> ·yt 'bend it'             | spápoý 'bent'                               | spáýpəý 'winding'                                                   |
| <i>łéqət</i> 'lay it down'          | stétəq 'lying'                              | stédtad 'several lying'                                             |
| tícət 'haul it up'                  | stítać 'hauled up'                          | stíčtač 'several hauled up'                                         |

Resultatives formed from RAH and R<sub>2</sub>C roots seem to be made plural in two ways:

(1) by reduplication of the schwa and resonant following the /s-/, probably identifiable as CVC reduplication with loss of the /h/, as in:

| $m e^{2}x$ 'remove it' | səmé? 'off'                    | səməme? 'several off'    |
|------------------------|--------------------------------|--------------------------|
|                        | //s-həmé?//                    | //s-həmhəmé?//           |
| <i>ná pox</i> 'eat it' | $s \ge n i \dot{p}$ 'eaten on' | sənanip 'eaten up'       |
| (2) by infixing /-     | əl-/ after the /s-/ (or afte   | er the lost /h/), as in: |

```
n \delta q^w 'fall asleep' s \partial n (q^w) 'sound asleep' s \partial l \partial n (q^w) 'several sound asleep'
```

The next example could be formed by either of these methods:

| lé?x | 'put it away' | səlé? 'put away' | sələlé? 'several put away' |
|------|---------------|------------------|----------------------------|
|------|---------------|------------------|----------------------------|

Resultatives formed from T 
i T and T 
i R roots are made plural by infixing /-i-/ after the first root consonant of the simple resultative, as in:

| i <sup>0</sup> óž× | 'get washed'        | $s\dot{t}^{	heta}\partial\dot{t}^{	heta}\dot{t}^{	ilde{w}}$ 'v | washed'      | $s\dot{t}^{	heta}\partial l\partial\dot{t}^{	heta}\dot{t}^{	imes}$ | 'several washed' |
|--------------------|---------------------|----------------------------------------------------------------|--------------|--------------------------------------------------------------------|------------------|
| q <sup>w</sup> ás  | 'go into the water' | <i>sq<sup>w</sup>əq<sup>w</sup>ís</i> 'i                       | n the water' | sqʷələqʷís                                                         | 'several in the  |
|                    |                     |                                                                |              | water'                                                             |                  |
| pźn                | 'get buried'        | spəpín 'bu                                                     | iried'       | spələpín '                                                         | several buried'  |

A resultative formed of a T<sub>P</sub>R root with *-om* is made plural by CVC reduplication:

*dóləm* 'camp out' *sqalím* 'camped out' *sqalqalím* 'several camped out'

Resultatives formed from TAH roots are made plural by infixing /-əl-/ after the first root consonant of the simple (non-plural) resultative, as in:

| $i^{\theta} e^{\gamma} t$ 'chew it'   | $s\dot{t}^{\theta}\partial\dot{t}^{\theta}\dot{e}^{\gamma}$ 'chewed' | $s\dot{t}^{\theta}\partial l\partial \dot{t}^{\theta}\dot{e}^{\gamma}$ 'several |
|---------------------------------------|----------------------------------------------------------------------|---------------------------------------------------------------------------------|
|                                       |                                                                      | chewed up'                                                                      |
| $q^{w} \dot{e} t$ 'make a hole in it' | sq <sup>w</sup> aq <sup>w</sup> é 'hole'                             | $sq^w \partial l \partial q^w \dot{e}$ 'full of holes'                          |

Resultatives formed from HAC roots are made plural by full reduplication of the root as it appears in the simple resultative with the stress on the second vowel:

 $h\dot{a}^{2}k^{w}$  'be used'  $sh\dot{a}^{2}k^{w}$  'used'  $sh\partial k^{w}h\dot{a}^{2}k^{w}$  'several used'

# 7.7.8. Diminutives of Resultatives

The diminutive form of a resultative indicates small size in the person or thing referred to or small amount in the attribute denoted. Diminutives of resultatives are formed by further CV reduplication with stress and vowel quality depending on type of root.

Resultatives of CVC roots form their diminutives with the formula  $s-C_1AC_1 \Rightarrow C_2 \rightarrow s-C_1 \Rightarrow C_1A(?)C_1 \Rightarrow C_2$ , as in:

| spí pow 'frozen' | spəpí <sup>2</sup> pəw 'poor little frozen thing' (CC)      |
|------------------|-------------------------------------------------------------|
| sqáqay 'dead'    | sqaqaqay 'little dead thing' (DK)                           |
| spápoý 'bent'    | spəpá <sup>2</sup> pəý 'slightly bent, slightly drunk' (AG) |

Resultatives of C<sub>2</sub>C, RAH, and (presumably) TAH roots (those for which the first vowel is schwa) form their diminutives with the formula  $s-C_1 \Rightarrow C_1 AC_2 \rightarrow s-C_1 C_1(2)C_1 AC_2$ , as in:

| spəpín' 'buried'                                                         | <i>spìppín</i> 'little buried thing' (CC)                                                             |
|--------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|
| $s\dot{t}^{\theta}\partial\dot{t}^{\theta}\dot{t}\check{x}^{w}$ 'washed' | $s\dot{t}^{\theta}\dot{t}\dot{t}^{\theta}\dot{t}^{\phi}\dot{t}^{\chi_{w}}$ 'little washed thing' (CC) |
| $s\theta \partial \theta \hat{k}^w$ 'straight, taut'                     | $s\theta i\theta \partial \theta \dot{e} \vec{k}^{w}$ 'a bit tight' (AG)                              |
| səmé? //s-həmé?// 'off, removed'                                         | sì·mé <sup>?</sup> //s-hìhəmé?// 'small thing                                                         |
|                                                                          | removed' (CC)                                                                                         |

# 7.7.9. Diminutive Plurals of Resultatives

Diminutive forms of C<sub>2</sub>C and RAH resultatives are made plural, like the resultatives themselves, by infixing /-al-/ after the first root consonant, as in:

| $s\dot{t}^{\theta}\dot{t}^{\theta}\dot{t}^{\chi}$ 'washed'                          | $s\dot{t}^{\theta}\dot{\iota}\dot{t}^{\theta}\dot{t}^{\theta}\dot{t}^{\chi}$ (dim.)  |
|-------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|
| $\dot{t}^{\theta} \partial l \partial \dot{t}^{\theta} \dot{t} \check{x}^{w}$ (pl.) | $s\dot{t}^{\theta}\partial l\dot{t}^{\theta}\dot{t}^{\theta}\dot{t}^{xw}$ (dim. pl.) |
| spəpín' 'buried'                                                                    | spìppín (dim.)                                                                       |

| spələpín (pl.)            | spəlippín (dim. pl.) |
|---------------------------|----------------------|
| səmé? 'off'               | sì·mé? (dim.)        |
| səməme <sup>9</sup> (pl.) | səli:mé? (dim. pl.)  |

I have no examples of diminutive plurals of resultatives of CAC roots.

# 7.8. PARADIGMS

I have attempted to bring together some of the data presented in the preceding sections to show full paradigms of a few representative verbs in the form of flowcharts (Diagrams 1 to 3). In the diagrams, forms appearing in square brackets are inferred from analogous forms in the same paradigm or in paradigms of roots of the same shape, but these are unattested. The blocks enclose symbols for rules for modifying the shape of the root. These are:

- 2 move stress to second vowel
- C reduplicate as C<sub>2</sub>C
- CV reduplicate as CVCə
- CV, 2 reduplicate as CəCý
- CVC reduplicate as CvCCoC
- Preduce full vowel to schwa
- A raise schwa to full vowel or reinstate full vowel
- 0 reduce vowel to zero
- ? infix glottal stop after (full) vowel
- -i- infix -i-
- -il- infix -il-
- -l- infix -*əl* or -*lə*-
- -l- infix -əl- or -l-
- s- prefix *s* 'resultative'
- C > 1 change reduplicated consonant to *-l*-
- k glottalize resonant
- Ci? reduplicate with i?



Diagram 1.  $\dot{q}i\vec{k}^{w}$  'get bitten'



| PERFECTIVE                                                        | PROGRESSIVE                                                                                     | RESULTATIVE |        |
|-------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|-------------|--------|
| $q^{w} \acute{e}l \longrightarrow CV \vec{R}$<br>'speak'          | → qʷáqʷəỉ<br>'be speaking'                                                                      | [none]      | SIM    |
| · i- CVC<br>↓<br>q <sup>w</sup> é <sup>2</sup> əl<br>'sev. speak' | q*álq*əl<br>'sev. are speaking'                                                                 |             | PL     |
|                                                                   | CV, 2, ə<br>↓<br>djwəqwá?qwəl<br>'little one is speaking                                        | <u>}</u>    | DIM    |
|                                                                   | V<br>C > 1<br>V<br>q <sup>w</sup> əlá <sup>9</sup> q <sup>w</sup> əl<br>'little ones are speaki | ng'         | DIM PI |

# Diagram 2. $q^{w} \acute{e}l$ 'speak'







# Diagram 3. $i^{\theta} \delta \check{x}^{w}$ 'get washed'



# 7.8.1. Problematic Paradigms

There are a few verbs that pose special problems in that they have forms that cannot yet be clearly identified as to aspect.

A root  $\sqrt{l\partial x}$  'stand up' has perfective forms with the prefix  $w\partial$ - 'suddenly':  $w\partial t \partial x$  'stand up suddenly' (AG);  $w\partial t \partial t \dot{x}$  'several stand up suddenly' (AG). This root without a suffix has no known progressive forms.

However, a derivative with  $-il\partial x$  (no gloss assignable, possibly composed of -il 'move toward' and an element -x, cf.  $2im\partial x$  'walk,'  $\dot{q}^w \partial y i \partial x$  'dance,'  $-x \partial n$  'foot') appears in two or more aspects. Its forms are:

- 1 *lžíləx* 'stand up' (JP, AG, DK)
- 2 *təltxíləx* (JP), *təlxíləx* (AG, DK) 'several stand up (in unison)'
- 3  $t \check{x} i \dot{l} \partial x$  'be standing' (JP, AG, DK)
- 4 *łəxłəxíləx* 'several are standing' (JP, AG, DK)
- 5  $l \neq x \neq l \neq x$  'be in the act of standing up, stand up from time to time' (AG)
- 6  $l \partial l \dot{x} i \dot{l} \partial x$  'they are standing around' (DK)
- 7  $lil \tilde{x} llo x$  'the little guy is standing up' (AG)
- 8  $l = lit \tilde{x} \tilde{l} a x$  'the little guys are standing up' (AG)

The first of these is the simple perfective and the usual word one elicits for 'stand.' Form 2 is clearly the plural perfective. Forms 3 and 4 were identified as progressive and plural progressive, although they do not exactly follow any pattern. Then AG gave form 5, which looks like a good progressive on the model  $t \partial \hat{m}(wst)$  'chop its trunk,'  $t \partial \hat{m} \partial \hat{w} st$  (~  $t \partial t \partial \hat{m} \partial \hat{w} st$ ) 'be chopping its trunk.' (Note also  $d^w \partial y (l \partial x)$  'dance,'  $d^w \partial y \partial l \partial x$  'be dancing,' although the stress in the progressive is still on the second vowel.) But if form 5 is the progressive, what then are forms 3 and 4? They may be, or once have been, duratives. Form 6 looks like a durative used for recreational activities, but it also looks like form 7, which is a diminutive (a diminutive of a durative?) with the usual diminutive /i/ appearing as / $\partial$ /. Form 8 is readily identifiable as the plural of form 7.

# **8** Morphology of the Root 2: The Noun

Nouns contrast with verbs in that they may take the possessive affixes and are not (or not usually) inflected for the progressive aspect, while verbs (with a few exceptions) must be nominalized to take the possessives, and they are (with a few exceptions) inflected for the progressive aspect. Like verbs, nouns may have plural and diminutive forms, and a few nouns have what seem to be resultative forms. Noun roots differ also from verb roots in their usual shapes. (For verbs functioning as nouns, see §8.9).

# 8.1. TYPES

In composition, nouns fall into several types, depending on whether they are simple roots or roots with suffixes, the type of root and suffix, and whether they have the prefix *s*-nominalizer. These types are:

(1) Simple bare nouns. These are noun roots that are free forms, such as:

| tén 'mother'                   | máňa 'child'                                   |
|--------------------------------|------------------------------------------------|
| céləx 'hand'                   | <i>qáləm</i> 'eye'                             |
| <i>téləw</i> 'arm'             | ${}^{?}\acute{e}\check{s}x^{w}$ 'harbour seal' |
| léləm 'house'                  | <i>iámən</i> 'wall'                            |
| $q\dot{a}^{\gamma}$ 'water'    | <i>xét</i> 'path'                              |
| <i>táž<sup>w</sup>ac</i> 'bow' | <i>Řcés</i> 'island'                           |

(2) *Simple s- nouns*. These are formed from what appear to be bound noun roots or roots with noun-like meanings, as in:

| $st^{\theta} \acute{a} \acute{m}$ 'bone' | sžáňa 'foot'   |
|------------------------------------------|----------------|
| <i>swóýqe</i> <sup>2</sup> 'male, man'   | sqéwθ 'potato' |

| smént 'rock, mountain' | syáł 'wood'   |
|------------------------|---------------|
| scátž 'halibut'        | smáyaθ 'deer' |
| stálow 'river'         | syícom 'sand' |

These roots have not been recorded apart from these words and their derivatives. The shapes of roots of this and the preceding type are discussed below.

(3) *Derived s- nouns*. These are formed from verb roots, verb roots with grammatical affixes, or (rarely) adjectives with the prefix *s*- 'nominalizer,' as in:

```
snét 'night' (< nét 'be night')

spíw' 'ice' (cf. píw \partial t 'freeze it')

st \partial m \partial x^w 'rain' (< t \partial m \partial x^w 'rain')

sn ds 'fat' (< nds 'fat, fatty')

sy d \cdot ys 'work' (< y d \cdot ys 'work')

sk^w i x 'name' (cf. k^w i x \partial t 'name him')

s^2 \partial t \partial t \partial t 'food' (< 2 \partial t \partial t \partial t 'eat')

s^2 \partial t \partial t \partial t 'clothing' (< 2 \partial t \partial t \partial t 'get dressed')

sp \partial h d ls 'wind' (< p d 'get blown,' - els 'activity')
```

The semantic relationship between these nouns and the verbs they are derived from is discussed in §12.1.1.

(4)  $\delta x^{w}$ - *nouns*. These are formed from verbs or adjectives with the compound prefix  $\delta x^{w}$ - 'oblique nominalizer (place of, time of, means of, reason for),' as in:

šx<sup>w</sup>?éxöθ 'bed' (< ?éxöθ 'lie down')</li>
šx<sup>w</sup>∂lí 'life' (< h∂lí 'alive')</li>
šx<sup>w</sup>k<sup>w</sup>ámk<sup>w</sup>∂m' 'strength, power' (< k<sup>w</sup>ámk<sup>w</sup>∂m' 'strong')

See also §12.1.4.

(5) Nouns with lexical suffixes. There are a great variety of these; see §13.4.3.

# 8.2. SHAPES OF NOUN ROOTS

Noun roots differ from verb roots in their most common shapes and in having a greater variety of shapes. The most common shape of the noun root is CVCVC, while that of the verb root is CVC. Of a set of 273 noun roots pulled from my lexical file, just over half have the shape CVCVC, and if we add those of the shapes CV·RC and CV·C, which may have an underlying CVCVC shape, the total is nearly 60 percent. The next most common shapes are CVC (nearly 10 percent), CVCC (nearly 10 percent), CVCV, CVCCVC, and CCVC. There may be a few others.

Some, or even many, of the longer forms may, of course, consist of old roots and affixes no longer identifiable as such. The number of apparent CVCVC roots ending in  $-\partial m$  and  $-\partial n$  suggest that in some instances these endings may be suffixes, perhaps  $-\partial m$  'intransitive' and possibly  $-\partial n$  as a variant of  $-t\partial n$ 'instrument.' But there are also a number of apparent CVCVC roots ending in  $-e^2$ ,  $-a^2$ ,  $-\partial n$ ,  $-\partial l$ ,  $-\partial y$ , and  $-\partial w$ , suggesting that there may once have been some noun-forming process involving resonants or glottalized resonants at work on CVC roots.

The following are some examples of noun roots (or presumable noun roots), grouped by shape. (I am not counting the nominalizing prefix s-.)

With the shape CVC are:

| $sp\delta x^w$ 'tripe'                            | <i>mén</i> 'father'                          |
|---------------------------------------------------|----------------------------------------------|
| smák <sup>w</sup> 'ball'                          | tén 'mother'                                 |
| <i>sť<sup>0</sup>ám</i> 'bone'                    | tín 'hummingbird'                            |
| $s\dot{t}^{\theta}\dot{s}\dot{k}^{w}$ 'bug, worm' | séy 'wool, fur'                              |
| xés 'sea lion'                                    | <i>xét</i> 'path, door'                      |
| qá <sup>?</sup> 'water'                           | <i>qéq</i> 'baby'                            |
| <i>qám</i> 'kelp'                                 | $sq^{\psi}\delta t^{\theta}$ 'willow grouse' |
| syáł 'wood'                                       | <i>swém</i> 'horse clam'                     |
| <i>tét</i> 'fathom' (i.e., the length of the      | outstretched arms; cf. təłét 'unfold it')    |

With the shape CVCV are:

| <i>móňə</i> 'child' (kin term)  | <i>mélə</i> 'bait'                                              |
|---------------------------------|-----------------------------------------------------------------|
| sť <sup>θ</sup> ána 'sculpin'   | $\dot{t}^{\theta} \dot{a} y \partial$ 'fur seal'                |
| sása 'widgeon'                  | síla 'grandparent'                                              |
| <i>sk¤ám̀ə</i> 'ratfish'        | <i>láq<sup>w</sup>ð</i> 'quiver, handbag'                       |
| <i>kʷál̓ə</i> 'belly'           | $\dot{q}\dot{a}\dot{w}\dot{a}$ 'new dancer's staff'             |
| <i>sžáňa</i> 'foot, leg'        | $\check{x}^{w} i \check{x}^{w} \partial$ 'large red sea urchin' |
| $q^w \delta le$ 'hemlock bough' | swíwa 'eulachon' ('hooligan')                                   |
| <i>həmá</i> 'pigeon'            | $2\acute{e}\check{x}\partial$ 'Canada goose'                    |

With the shape CVCC are:

| smátž <sup>w</sup> 'sculpin (sp.?)'                  | <i>mé<sup>9</sup>t</i> 'lining of abdominal cavity of sturgeon' |
|------------------------------------------------------|-----------------------------------------------------------------|
| <i>mślq</i> <sup>w</sup> 'uvula'                     | <i>témx</i> <sup>w</sup> 'gooseberry'                           |
| $\dot{t}\dot{a}^{\gamma}x^{w}$ 'grand fir'           | $\theta i k^{w} t$ 'sea cucumber'                               |
| $\dot{t}^{\theta}iqt$ 'flicker'                      | $t^{\theta} \delta l \dot{c}$ 'scale (of fish or reptile)'      |
| séyd 'clay'                                          | s <i>žámk</i> <sup>w</sup> 'fish-roe cheese'                    |
| scátž 'halibut'                                      | sćéxt 'stick of wood'                                           |
| <i>kʷátq</i> 'frog'                                  | $k^{w}\delta l\dot{c}$ 'dried herring'                          |
| $k^{w}i^{p}x^{w}$ 'pitch'                            | <i>qélq</i> 'rose hip'                                          |
| $sq\acute{e}w\theta$ 'potato'                        | $qi\dot{w}\dot{x}$ 'steelhead'                                  |
| héwt 'rat'                                           | $e^{2}\acute{e}\check{s}x^{w}$ 'harbour seal'                   |
| <pre>?éýž 'crab'</pre>                               | ?éwkw 'possessions'                                             |
| $\dot{q}^{w}\dot{e}\dot{x}t$ 'root plot, garden, orc | hard'                                                           |

With the shape CCVC are:

| $iq^{w} \delta m$ 'thimbleberry'                   | słćém 'weasel'                             |
|----------------------------------------------------|--------------------------------------------|
| $\hat{\lambda} p \acute{e}t$ 'storage bag'         | $\vec{\lambda} s i p$ 'licorice fern root' |
| $\hat{X} l \acute{e} \acute{y}$ 'shovelnose canoe' | <i>Ăcés</i> 'island'                       |
| $k^{w}sic$ 'cut-throat trout'                      | <i>sk<sup>w</sup>qéq</i> 'robin'           |
| s <i>qtán</i> 'bow of canoe'                       |                                            |

With the shape CVCVC are:

 $sp e^{2} \partial \theta$  'black bear'  $m \delta \theta \partial t$  'Indian hemp' másən 'gall' smélak<sup>w</sup> 'hide shirt' *méčən* 'black haw berry'  $m \hat{a}^2 \partial q^w$  'larger bird'  $sm \neq \dot{q}^w a^\gamma$  'great blue heron' támas 'sea otter, velvet'  $t \hat{e}^{\gamma} \partial t$  'harpoon head' stálow 'river' *táman* 'wall' *téləw* 'arm, wing' *téqe*? 'salal berry' *θámən* 'eyebrow'  $s\theta \dot{a}q \dot{a} \dot{y}$  'sockeye'  $st^{\theta}ic \partial m$  'hazelnut' sé?əq 'brake-fern root' słéwać 'herring' stéwan 'sleeping mat' láwaž 'rib' scáła? 'leaf' *xéwəq* 'carrot'  $k^{w} \delta x^{w} \partial \theta$  'coho'  $k^{w} \hat{a} \hat{l} \partial x^{w}$  'chum salmon'  $k^w a x^w a^\gamma$  'box' sqáma? 'breast, milk'  $sq ela \tilde{x}$  'digging stick' sqəléw 'beaver' *démə*ý 'adolescent girl' *žáca*<sup>2</sup> 'lake' sžáýəs 'head' sqwəméy 'dog'  $s\dot{q}^{w} \partial l \dot{e} x^{w}$  'small bird'

 $s \dot{p} i s \partial c$  'conifer cone'  $m \delta t^{\theta} \delta t$  'pus' smálać 'horsefly' mécan 'testicle' *méqe*<sup>?</sup> 'snow (on the ground)' máq<sup>w</sup>əm 'bog'  $sm \delta y \partial \theta$  'deer, animal, meat' táməł 'red ochre' stálas 'spouse' tá xwac 'bow' stíwan 'living sibling's child' *táča*<sup>2</sup> 'skewer for herrings' *stíwad*<sup>w</sup> 'fuller's earth'  $\theta \hat{a} \theta \hat{a} n$  'mouth'  $t^{\theta} \ell l e^{\gamma}$  'heart' sák<sup>w</sup>əm 'whole bark' sá x<sup>w</sup> l 'grass' sténaý 'woman' (but cf. tánat 'weave it')  $sléma\theta$  'fish club' célax 'hand' sxəmén 'enemy, rival' k<sup>w</sup>ásən 'star' kwétań 'mouse' kwálaw 'skin'  $s\vec{k}^{w}\delta y\partial \theta$  'slave' qálam 'eye'  $q \delta l \delta \tilde{x}$  'salmon eggs' s*ģáməl* 'paddle' *àəảí*? 'gut' *xécət* 'fireweed'  $q^{w}\partial^{2} \hat{a} p$  'crabapple' qwánas 'whale' syícom 'sand'

| <i>wénəm</i> 'orphan'<br><sup>2</sup> ánəŵ 'tallow'<br>s <sup>2</sup> áx <sup>w</sup> a <sup>2</sup> 'butter clam'                                                                                                                                                                                                                                                                | <ul> <li>?íməθ 'grandchild'</li> <li>s²ánəm 'harpoon shaft'</li> <li>?éləx 'sibling, cousin of opposite sex'</li> </ul>                                                                                                 |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| With the shape CV·C are:                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                         |
| <i>s pá·l</i> <sup>'</sup> 'raven'<br><i>slí·m</i> 'sandhill crane'<br><i>q<sup>w</sup>í·n</i> 'ear'<br><i>hú·n</i> 'pink (humpback) salmon'                                                                                                                                                                                                                                      | wí·l 'tule'<br>mí·l 'blue grouse'<br>k <sup>w</sup> é·c 'dogfish'                                                                                                                                                       |
| Some, but not all, of these have an un<br>With the shape CV·RC are:                                                                                                                                                                                                                                                                                                               | derlying shape //CVRVC// (see §1.5.6).                                                                                                                                                                                  |
| <i>k̇̀«á·nť</i> 'porpoise'<br>?é·yt 'lingcod'                                                                                                                                                                                                                                                                                                                                     | smé·nt 'stone, mountain'                                                                                                                                                                                                |
| These probably have an underlying sh<br>With the shape CVCCVC are:                                                                                                                                                                                                                                                                                                                | hape //CVRəC// (see §1.5.3).                                                                                                                                                                                            |
| spáł x <sup>w</sup> əm 'lung'<br>mál x <sup>w</sup> əl 'Indian plum'<br>st <sup>θ</sup> e <sup>9</sup> qən 'cattail'<br>łéwqəm 'mussel'<br>ćá <sup>9</sup> k <sup>w</sup> a <sup>9</sup> 'skunk cabbage'<br>xá <sup>9</sup> qən 'marten'<br>swíwləs 'adolescent boy'<br>scínk <sup>w</sup> a <sup>9</sup> 'lightning snake'<br>sk <sup>w</sup> álwəs 'co-parent-in-law, blood kir | spéł žən 'prairie, pasture'<br>méxcən 'louse'<br>sésqəc 'sasquatch'<br>lá?θən 'dish'<br>k <sup>w</sup> əmləx <sup>w</sup> 'root'<br>swəyqe? 'male, man'<br>s?əlqəy 'snake'<br>qi?ta? 'swing'<br>n's spouse's blood kin' |
| With the shape CCVCC are:                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                         |
| $iq\acute{e}l\acute{c}$ 'moon'<br>$s\theta\check{x}\acute{a}^{2}s$ 'Dolly Varden trout'<br>With the share CVCVCV are:                                                                                                                                                                                                                                                             | <i>sẩxé·k<sup>w</sup></i> 'silverweed'                                                                                                                                                                                  |
| máž <sup>w</sup> əyə 'navel'<br>sžəmálə 'small sea snail'                                                                                                                                                                                                                                                                                                                         | <i>smətəli</i> 'beaver-tooth dice game'<br><i>t<sup>ê</sup>áməkwa</i> 'kelp greenling (tommycod)'                                                                                                                       |
| With the shape CVCVCVC are:                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                         |
| $páxələq^w$ 'yellow cedar'<br>$láməq^wa^2$ 'sole'<br>$\dot{q}^walítəq$ 'seagull'<br>$yáx^wale^2$ 'eagle'                                                                                                                                                                                                                                                                          | sála <sup>2</sup> ac 'house mat'<br>qəyiî <sup>2</sup> əc 'elk'<br>syətəwa <sup>2</sup> 'wealth'                                                                                                                        |

There are a few presently unanalyzable nouns of other shapes with four or more consonants, but it seems likely that these and a number of those above will eventually prove to be composed of roots and suffixes.

There are also a few nouns showing CVC reduplication with stress in the first vowel. Some of these have the *s*- prefix but most do not. A disproportionate number are names of animals and plants. Several may be imitative of sounds.

It is also possible that some of these are verb forms. The pattern  $C_1 AC_2 C_1 = C_2$ is that of the plural progressive of the verb, as in  $lak^w lak^w$  'several are flying,' from  $lak^w$  'fly.' Two or three of the words of this shape listed below may have this source, but three of the list definitely do not, since they are loans from Chinook Jargon (marked CJ). The pattern  $C_1 = C_2 = C_2$  is that of the dispositional form of the verb, which may be glossed as a noun, as in qanqan 'steal habitually, thief,' from qen 'steal.' I have omitted identifiable dispositional forms from the list, but it may include others not identified.

With the shape  $C_1 A C_2 C_1 = C_2$  are:

músməs 'cow, bull' (CJ)  $t^{\theta}(x^wt^{\theta}ax^w$  'fishhawk'  $sq^{wt}nq^{w}an$  'bead' xápxap 'an unidentified kind of bird, said to whistle' (cf. xápas 'whistle')  $s\check{x}^wey\check{x}^way$  'a cleansing ritual performed by masked dancers' tintan 'bell, time o'clock' (CJ)  $t^{\theta}tkt^{\theta}ak$  'wagon, buggy' (CJ)

With the shape  $C_1 \circ C_2 C_1 \circ C_2$  are:

| <i>táq<sup>w</sup>təq<sup>w</sup></i> 'red snapper'                      | $\dot{t}^{\theta} \delta x^{w} \dot{t}^{\theta} \partial x^{w}$ 'stinging nettle' |
|--------------------------------------------------------------------------|-----------------------------------------------------------------------------------|
| $\dot{\tilde{X}} \neq \tilde{X} \neq \tilde{X} \neq \tilde{X} $ 'oyster' | sq́ə́łq́əł 'muskrat'                                                              |
| <i>ģážģaž</i> 'tin'                                                      | <i>žámxám</i> 'flowering stalk of the horsetail'                                  |
| <i>x	wə́nxwən</i> 'bullroarer'                                           | ?áňaň //?áň?aň// 'magpie'                                                         |
| a <sup>w</sup> áma <sup>w</sup> am 'bittern' (probably this              | bird, said to be named for its call)                                              |

# 8.3. PLURALS

Nouns referring to human beings all have plural forms, and the use of these forms seems to be obligatory. Nouns referring to body parts, a few animals, and some of the most important artifacts and other inanimate things have plurals that are commonly used. But plurals do not occur for all nouns referring to animals and inanimate things, and the use of most such plurals is optional. When asked for plural forms of names of animals, JP hesitated on several, gave forms that he said were probably correct, and added that they were seldom used. On others, he immediately said that there were no plurals. It seems also that the plural forms that are optional are more likely to be used for a group of the referent, while plural forms used regularly mean simply more than one. The plural of a few words means 'kinds of'; for example,  $má^2 aq^w$  'larger bird' (often glossed 'duck') has a plural  $mamá^2aq^w$  'different kinds of larger birds, mixed ducks' (cf. English 'fish, fishes'). Plurality can, of course, also be indicated by the verb.

As with verbs, plurality in nouns is marked by internal modification of the root in a variety of ways. Plurals may be marked by: (1) an infixed *-l-*, with or without glottalization, and with the stress either preceding of following the infix, (2) CV reduplication, with the stress on either the first or the second vowel, (3) CVC reduplication, with the stress on either the first or the second vowel, or (4) a change in the vowel alone.

To some extent the type of plural marking in nouns is predictable from the shape of the root, but to a lesser extent than with the verbs. To some extent, too, the relationship between root shape and plural marking in the noun parallels that relationship in the verb. There are noun plural forms, however, that have no counterpart among verb plurals, although they may resemble other forms of the verb. I will point to similarities and differences in the presentation below.

There are questions that are not yet answerable. If verbs and nouns have different rules for forming plurals (as the data suggest), which rule applies to a noun derived from a verb root? If roots of different shapes follow different rules, when a full vowel appears as schwa in the presence of a lexical suffix that has this effect, is the plural based on the surface shape with schwa or on the underlying shape with the full vowel?

In the following presentation of plural forms, I have sorted by type of plural marker, merging nouns with roots of different shapes when they have similarly formed plurals.

#### 8.3.1. With Infixed -l-

These vary in where the stress falls and whether the *-l*- is glottalized. The types are:

(1) With the stress on the first vowel, the infix following it. Nearly all of the nouns that have plurals of this type have roots of the shape CAC-, that is, they have an initial consonant, a full vowel, and a second consonant, and whether there is more is not relevant. (Verbs with CAC roots also usually have plurals of this type and of similar subtypes.) There are four subtypes.

(1a) With the infix appearing as  $-l_{\partial}$ - unglottalized (CÁC-  $\rightarrow$  CÁl<sub> $\partial$ </sub>C-)

| $st^{\theta}\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!\!$ | <i>sť<sup>6</sup>áləm</i> 'bones'                                           |
|---------------------------------------------------------------------------------------|-----------------------------------------------------------------------------|
| sťépəý 'snag'                                                                         | sťéləpəý 'snags'                                                            |
| sčéxt 'stick'                                                                         | <i>sčéləxt</i> 'sticks'                                                     |
| $sq\acute{e}w\theta$ 'potato'                                                         | $sq\acute{e}l \partial w\theta$ 'potatoes'                                  |
| <i>šx<sup>w?</sup>áq<sup>w</sup>a</i> ? 'sibling, cousin'                             | <i>šx<sup>w?</sup>áləq'</i> <sup>w</sup> a <sup>?</sup> 'siblings, cousins' |

If the root vowel is *i*, the plural has *e*.

| <i>ť<sup>θ</sup>í kť<sup>θ</sup>ə k</i> 'wagon'                  | <i>ť<sup>e</sup>éləkť<sup>e</sup>ək</i> 'wagons'                    |
|------------------------------------------------------------------|---------------------------------------------------------------------|
| <i>šx<sup>w</sup>əmník<sup>w</sup></i> 'living parent's sibling' | <i>šx<sup>w</sup>əmnélək<sup>w</sup></i> 'living parent's siblings' |

(1b) With the infix as  $-l_{\partial}$  (CÁC-  $\rightarrow$  CÁl<sub> $\partial$ </sub>C). There seems to be nothing that distinguishes the roots of this set from the last.

| $\theta \hat{a} \theta \partial n$ 'mouth' | <i>θál əθən</i> 'mouths'                         |
|--------------------------------------------|--------------------------------------------------|
| $k^w a x^w a^2$ 'box'                      | <i>k៉<sup>w</sup>álðx<sup>w</sup>a</i> ? 'boxes' |
| <i>qéməy</i> 'teenage girl'                | <i>qélamay</i> 'teenage girls'                   |
| <i>sk<sup>w</sup>í x</i> 'name'            | sk <sup>w</sup> élax 'names' (AG)                |
| łéctən 'knife'                             | <i>télocton</i> 'knives'                         |

The last example is a derivative of a verb root; cf.  $lic \partial t$  'cut it' and its plural  $le^{2}\partial lc \partial t$  'cut it into slices.'

(1c) With  $-\dot{l}\partial$ - reordered as  $-\partial\partial l$ - (CÁC-  $\rightarrow$  CÁ $\partial\partial l$ C-. The schwa is often assimilated to the stressed vowel and sometimes, for some speakers, the glottal stop is then lost, resulting in a long vowel. In all of the roots that have plurals with this reordering, the second consonant may be identified as a coronal obstruent, but not all roots of this shape have plurals of this form; cf.  $\theta \dot{a} \theta \partial n$  'mouth' in set (1b) above.

| sčáła <sup>2</sup> 'leaf'                     | <i>sčá?alła?</i> 'leaves'                                                             |
|-----------------------------------------------|---------------------------------------------------------------------------------------|
| <i>xét</i> 'path, door'                       | <i>xé<sup>9</sup>elt</i> 'paths, doors'                                               |
| k <sup>w</sup> ásən 'star'                    | k <sup>w</sup> á <sup>9</sup> alsən 'stars'                                           |
| <i>žáca</i> <sup>2</sup> 'lake'               | <i>žá<sup>?</sup>əlca</i> ? 'lakes'                                                   |
| s <sup>2</sup> áθəs 'face'                    | $s\dot{a}^{2}al\theta \partial s \sim s\dot{a} \cdot l\theta \partial s$ 'faces' (JP) |
| yásaq <sup>w</sup> 'hat'                      | <i>yá?alsaq</i> <sup>w</sup> 'hats'                                                   |
| mécən 'testicle'                              | <i>mé<sup>9</sup>əlcən</i> 'testicles'                                                |
| <i>pút</i> 'boat'                             | <i>pú<sup>2</sup>əlt</i> 'boats'                                                      |
| snéčałtx <sup>w</sup> 'affine'                | sné <sup>2</sup> alčałtx <sup>w</sup> 'affines'                                       |
| <i>nécəwtx</i> <sup>w</sup> 'house next door' | <i>né·lċ́əwtx<sup>w</sup></i> 'houses next door'                                      |
| s²áθnəc 'bay'                                 | s <sup>2</sup> é·lθnəc 'bays'                                                         |

This last may belong here in spite of the vowel change. The root is probably  $\sqrt{2}e\theta$ - of unknown meaning (see §13.3), appearing with schwa before *-nəc* 'bottom.' (Cf. CAC verbs with *-nəx*<sup>w</sup> where the vowel appears as schwa in the perfective but in its full form in the progressive.)

(1d) With -*l*- following the vowel and no schwa (CVC-  $\rightarrow$  CV1C-). Only the following have been noted, one with a full vowel and three with unstressed schwas:

| qéqələ 'baby' (CC)                                                          |                                                                   |
|-----------------------------------------------------------------------------|-------------------------------------------------------------------|
| sqéqələ 'boy baby' (JP)                                                     | sqélqələ 'babies' (JP)                                            |
| <i>məstáyəx</i> <sup>w</sup> 'person'                                       | <i>məlstá yəx<sup>w</sup></i> 'persons'                           |
| stəqáye <sup>2</sup> 'wolf'                                                 | <i>stəlqáye</i> ? 'wolves'                                        |
| <i>šx<sup>w</sup>mək<sup>w</sup>ə</i> <sup>2</sup> élə 'grave, family plot' | <i>šx<sup>w</sup>məlk<sup>w</sup>ə<sup>9</sup>élə</i> 'graveyard' |

This last example may have a full vowel in the root; cf.  $cm\acute{k}^we^2$  'participate in a funeral.'

(2) With the stress in the second vowel, following the infix. These are also mostly CAC roots but there seem to be fewer with plurals of this type. (There

are no verb roots noted with plurals of this type, but verb and noun diminutives commonly have this plural.)

(2a) With  $-\partial l$ - unglottalized (CÝC-  $\rightarrow$  C $\partial$ lÝC-)

| <i>pítxən</i> 'lizard, salamander'                                                    | pəlítxən 'lizards, salamanders'                                                             |
|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|
| músməs 'cow, bull'                                                                    | məlúsməs 'cows, cattle'                                                                     |
| stíqìw 'horse'                                                                        | stəlíqìw 'horses'                                                                           |
| sqé <sup>?</sup> eq ~ sqé·q 'junior sibling,<br>cousin'                               | sqəlé·q 'junior siblings, cousins' (JP)                                                     |
| <i>máqsən</i> 'nose'                                                                  | <i>məláqsən</i> 'noses'                                                                     |
| (2b) With $-\partial \dot{l} - (C\dot{V}C - \rightarrow C\partial \dot{l}\dot{V}C -)$ |                                                                                             |
| <i>sqé<sup>?</sup>əq</i> 'junior sibling, cousin'<br><i>s`Åéləqəm</i> 'fierce being'  | sqəlé <sup>2</sup> əq 'junior siblings, cousins' (CC)<br>s <i>žəléləqəm</i> 'fierce beings' |

(2c) With  $-\partial l \partial$ , neither vowel stressed  $(C_1 \partial C_2 \acute{V} - \rightarrow C_1 \partial l \partial C_2 \acute{V} -)$ . Nouns that have plurals of this form have an unstressed schwa as the first vowel, but not all nouns of this shape have plurals of this form; cf. (1d) above. (Verb forms that have this shape, e.g., some duratives and some resultatives, also have plurals of this type.)

| $i^{\theta} \partial m e \cdot n$ 'arrow [bone pointed?]' | <i>ť<sup>e</sup>ələmé•n</i> 'arrows' |
|-----------------------------------------------------------|--------------------------------------|
| sxəmén 'enemy, rival'                                     | sxələmén 'enemies, rivals'           |
| <i>kəpú</i> 'coat'                                        | kələpú 'coats'                       |
| k <sup>w</sup> əšú 'pig'                                  | k <sup>w</sup> ələšú 'pigs'          |

The first of these probably has the root of  $st^{\partial}a\vec{m}$  'bone' and the plural might be seen as a reduction of  $st^{\partial}ala\vec{m}$  'bones'; the others have roots of the shapes seen. The last two are loans from Chinook Jargon, suggesting that this plural formation is productive for words of this shape.

#### 8.3.2. With CV Reduplication

These vary in stress and quality of the stressed vowel.

(1) With stress on the first vowel  $(C_1 V C_2 \rightarrow C_1 V C_1 \neg C_2 -)$ . There are only a few nouns with plurals of this type. (There are no verbs recorded with plurals formed this way, but it is a common way of forming progressives.)

| $s\dot{t}^{\theta}\delta\dot{k}^{w}$ 'worm, bug'    | <i>sť<sup>6</sup>šť<sup>6</sup>3k</i> <sup>w</sup> 'worms, bugs' |
|-----------------------------------------------------|------------------------------------------------------------------|
| $q^{w} \delta l \delta$ 'bough (esp. hemlock)' (JP) | <i>q<sup>w</sup>áq<sup>w</sup>ələ</i> 'boughs' (JP)              |
| $q^{w} \delta q^{w} \delta l \delta$ 'hemlock' (CC) |                                                                  |
| <i>sq<sup>w</sup>əléx</i> 'small bird'              | <i>sq<sup>w</sup>ðq<sup>w</sup>ðlðx</i> 'small birds'            |

(2) With stress on the second vowel  $(C_1 \acute{V}C_2 \rightarrow C_1 \Im C_1 \acute{V}C_2 \rightarrow)$ . There are a number of nouns with plurals of this type, falling into several subtypes. In nearly all instances, the second consonant is a resonant or a pharyngeal. (Among

verb plurals, there is a parallel in two roots with the shape RAH, but this is the common form of reduplication producing progressives of TAH roots.)

(2a) With no change in vowel. Most have the shape CAH.

| $spé^{\gamma}e\theta$ 'black bear'                                     | $sp = p e^{2} e \theta$ 'black bears'                                                |
|------------------------------------------------------------------------|--------------------------------------------------------------------------------------|
| <i>má</i> <sup>?</sup> <i>aq</i> <sup>w</sup> '''duck'' (larger bird)' | <i>məmá<sup>2</sup>aq</i> <sup>w</sup> 'several kinds of "ducks"                     |
| $st^{\theta} e^{\gamma} q  a n$ 'cattail'                              | $st^{\theta} \partial t^{\theta} \partial q \partial n$ 'place where only cattails   |
|                                                                        | grow'                                                                                |
| <i>lá<sup>?</sup>θən</i> 'dish'                                        | <i>ləlá<sup>2</sup>θən</i> 'dishes'                                                  |
| $l \acute{a} \cdot k$ 'log (cut for a mill)'                           | <i>ləlá</i> ·k 'logs'                                                                |
| $q\dot{a}^{2}$ 'water'                                                 | <i>təmqəqá</i> ? 'freshet time'                                                      |
| <i>ká</i> 'car'                                                        | <i>kəká</i> · 'cars' (JP) (DK gave <i>kəlá</i> · but                                 |
|                                                                        | JP rejected it)                                                                      |
| stíwən 'living sibling's child'                                        | stətíwən 'living sibling's children'                                                 |
| $s\vec{k}^{w}\delta y\partial\theta$ 'slave'                           | $s\vec{k}^{w}\partial\vec{k}^{w}\dot{e}y\partial\theta$ 'slaves' ("but seldom used") |
|                                                                        | – JP)                                                                                |

(2b) With an initial resonant appearing initially as *h* and medially doubled, the vowel unchanged ( $RVC- \rightarrow h \Rightarrow RRVC-$ ). As usual, following the prefix *s*-, the *h* is lost. (A similar appearance of R as *h* is seen in the progressives of verbs of the shapes RAH and R $\Rightarrow$ C. See also [2e] below.)

| léləm 'house'                             | həlléləm 'houses'                                                                                       |
|-------------------------------------------|---------------------------------------------------------------------------------------------------------|
| <i>syálðax<sup>w</sup>a?</i> 'old person' | səyyáləx <sup>w</sup> a <sup>2</sup> [sìyá <sup>2</sup> lax <sup>w</sup> a <sup>2</sup> ] 'old persons' |
| <i>swóýqe</i> ? 'male, man'               | <i>səwwáÿqe</i> ²[suwə̈ÿqe? ~ suw∙ə̈ÿqe?]                                                               |
|                                           | 'males, men'                                                                                            |

(2c) With a vowel change from  $\partial$  to *e*. Only one noun is recorded with a plural of this type:  $m \partial n \partial$  'child, offspring,'  $m \partial n \partial n \partial$  'children, offspring.' (There is a parallel in the plurals of some verbs of the shape T $\partial$ T.)

(2d) With a vowel change from  $\partial$  to  $i(C_1 \partial C_2 \rightarrow C_1 \partial C_1 (C_2 -)$ . Nearly all have the shape  $T \partial R$ - (obstruent-schwa-resonant). (No verbs are known to have plurals of this type, but verb roots with schwa may have resultatives with similar forms.)

| $t \delta m \partial x^w$ 'land, earth'      | <i>tətíməx</i> <sup>w</sup> 'countries, pieces of land'                                                   |
|----------------------------------------------|-----------------------------------------------------------------------------------------------------------|
| tánəqsən 'mallard'                           | tətínəqsən 'mallards'                                                                                     |
| <i>k<sup>w</sup>ámləx<sup>w</sup></i> 'root' | $k^{w} \partial k^{w} i m l \partial x^{w}$ 'bunch of roots'                                              |
| <i>skwálwas</i> 'co-parent-in-law, blood     | $s\vec{k}^{w}\partial\vec{k}^{w}ilw\partial s$ (once recorded                                             |
| relative's affine'                           | <i>sk<sup>w</sup>əlk<sup>w</sup>ilwəs</i> ) 'co-parents-in-law,                                           |
|                                              | secondary affines'                                                                                        |
| qáləm 'eye'                                  | qəqíləm 'eyes'                                                                                            |
| sq'áməl 'paddle'                             | sq <sup>2</sup> q <sup>2</sup> q <sup>2</sup> m <sup>2</sup> l 'paddles'                                  |
| sžáňa 'foot, leg'                            | sžažíňa 'feet, legs'                                                                                      |
| <i>q<sup>w</sup>əqtén</i> 'shoulder'         | $q^{w} \partial q^{w} i \dot{q} t \partial n$ 'shoulders'                                                 |
| $\theta q \acute{e}t$ 'tree'                 | $\theta \partial \theta i q \partial t$ (JP), $\theta \partial q \theta \delta q \partial t$ (CC) 'trees' |

The last two examples have roots of shapes different from those preceding, and are probably not simple nouns;  $q^w \partial \dot{q} t \dot{e} n$  may be composed of a root plus the suffix  $-t\partial n \sim -t\dot{e}n$  'instrument,' and  $\theta q \dot{e} t$  may be a verb form (see §7.7.6).

(2e) With an initial resonant appearing as h initially in the plural and with a change of the vowel from  $\vartheta$  to i (R $\vartheta$ C-  $\rightarrow$  h $\vartheta$ RiC-). Again, following s- the h is lost. (There is one wholly irregular verb,  $n\vartheta y \vartheta m$  'laugh,' that has the plural resembling this type.)

| yá x <sup>w</sup> ələ 'eagle'                           | həyí x̆ <sup>w</sup> lə 'eagles'                |
|---------------------------------------------------------|-------------------------------------------------|
| $sm \delta y \partial \theta$ 'deer, animal, meat'      | $s \partial m i y \partial \theta$ 'deer (pl.)' |
| $sm \acute{q}^w a^2$ 'great blue heron'                 | $s \ge m i \dot{q}^w a^2$ 'great blue herons'   |
| snáx <sup>w</sup> ať 'canoe'                            | səní x <sup>w</sup> əł 'canoes'                 |
| swáď <sup>w</sup> a?ł (JP), swáwď <sup>w</sup> a?ł (CC) | səwíqwa?ł 'mountain-goat wool                   |
| 'mountain-goat wool blanket'                            | blankets' (AG)                                  |
| syáwan 'possessing song'                                | səýíwən [si?yíwən] 'different kinds             |
|                                                         | of possessing songs'                            |

(3) In a word beginning with two consonants, reduplication of the first separated by an unstressed schwa ( $C_1C_2\acute{V} \rightarrow C_1 \neg C_1C_2\acute{V}$ ). Only one example has been recorded:  $t\acute{c}\acute{a}lm \neg x^w$  'wild animal,'  $t \neg t\acute{c}\acute{a}lm \neg x^w$  'all kinds of wild animals' (CC).

#### 8.3.3. With CVC Reduplication

These vary with stress and vowel quality.

(1) With stress on the first vowel  $(C_1 V C_2 - \rightarrow C_1 V C_2 C_1 \neg C_2 -)$ . Only one example has been recorded:  $s p \dot{\delta} q$  'white patch' (on the skin; cf.  $p \dot{\delta} q$  'white'),  $s p \dot{\delta} q p \dot{\delta} q$  'white patches.' (Verbs of the shape CAC have plural progressives of the shape  $C_1 \Delta C_2 C_1 \neg C_2$  and dispositional-iteratives of the shape  $C_1 \dot{\delta} C_2 C_1 \neg C_2$ .)

(2) With the stress in the second vowel  $(C_1VC_2 \rightarrow C_1 \neg C_2C_1 \lor C_2 \rightarrow C_1 \neg C_2C_1 \lor C_2 \rightarrow C_1 \neg C_2C_1 \lor C_2 \rightarrow C_1 \neg C_2C_1 \lor C_2 \rightarrow C_1 \neg C_2C_1 \lor C_2 \rightarrow C_1 \neg C_2C_1 \lor C_2 \rightarrow C_1 \neg C_2C_1 \lor C_2 \rightarrow C_1 \neg C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_2 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_1 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C_2 \lor C_2C$ 

| stáməx 'warrior'               | stəmtáməx 'warriors'                                       |
|--------------------------------|------------------------------------------------------------|
| <i>stáləs</i> 'spouse'         | <i>stəltáləs</i> 'spouses'                                 |
| stálow 'river'                 | stəltáləw 'rivers'                                         |
| <i>téləw</i> 'arm, wing'       | <i>ťalťélaw</i> 'arms, wings'                              |
| <i>iámən</i> 'wall (of house)' | <i>ťəmťámən</i> 'walls'                                    |
| síla 'grandparent, etc.'       | səlsílə 'grandparents, etc.'                               |
| $\theta aman$ 'eyebrow'        | $\theta \partial m \theta \dot{a} m \partial n$ 'eyebrows' |
| sténoý 'female, woman'         | stanténaý 'females, women'                                 |
| <i>Xímən</i> 'tendon'          | <i>ŽəmŽímən</i> 'tendons'                                  |
| <i>céləx</i> 'hand'            | <i>cəlcéləx</i> 'hands'                                    |
|                                |                                                            |

| <i>šx<sup>w</sup>áməcən</i> 'fisher (the animal)' | <i>šx<sup>w</sup>əmx<sup>w</sup>áməcən</i> 'fishers' |
|---------------------------------------------------|------------------------------------------------------|
| <i>žéləw</i> 'spoon'                              | <i>xəlxéləŵ</i> 'spoons'                             |
| wénəm 'orphan'                                    | wənwénəm 'orphans'                                   |
| $\gamma i m \partial \theta$ 'grandchild, etc.'   | $^{?}$ <i>àmíma</i> $\theta$ 'grandchildren, etc.'   |
| <i><sup>?</sup>éləx</i> 'sibling of opposite sex' | <i>°əléləx</i> 'siblings of opposite sex'            |

The following nouns with long vowels have underlying CAR shapes and have plurals of this type with the long vowel retained in the plural.

| spá·l //spáləl// 'raven' | >             | spəlpá·l' | ravens'             |                 |
|--------------------------|---------------|-----------|---------------------|-----------------|
| smé·nt //sménət// 'sto   | ne, mountain' | smənmé•n  | t 'stones, mountain | ıs <sup>:</sup> |

The following probably have underlying C $\Rightarrow$ R $\Rightarrow$ R shapes (see §1.5.6), but their plurals are of the same type.

| $\dot{q}^{w}i\cdot\dot{n}$ 'ear' | <i>ą́wəną́wí·n</i> 'ears'                                                 |
|----------------------------------|---------------------------------------------------------------------------|
| <i>wí</i> ·ĺ 'tule'              | $x^w w \partial l w i \cdot \dot{l} \partial l p$ 'place where only tules |
|                                  | grow'                                                                     |

(3) With stress on a later vowel

| $sp \partial lq^{w} it^{\theta} e^{\gamma}$ 'dead person' | <i>s pəl pəlq<sup>w</sup>ít<sup>9</sup>e<sup>?</sup></i> 'dead persons' |
|-----------------------------------------------------------|-------------------------------------------------------------------------|
| <i>səmáýe</i> ? 'bee, wasp'                               | səmsəmáyə?elə 'bees' or wasps' nest'                                    |
| sq <sup>w</sup> əméý 'dog'                                | sq <sup>w</sup> əmq <sup>w</sup> əméý 'dogs'                            |

#### 8.3.4. With Vowel Change Only

There are a few nouns with plurals of this type. They are not necessarily the products of the same process and are better examined one by one.

(1) The plural of syéye (also recorded  $syéye^2$ ) 'friend, relative' (from  $yéye \sim yéye^2$  'tightly bound, secure,' that is, 'someone to whom one is bound') is phonetically [sìyéye], which is probably /syəyéye/, formed by CV reduplication.

(2) The plural of  $[si^{\circ}\acute{em}]$  'rich person, leader, lord, lady, Sir, Ma'am' is  $[si^{\circ}\acute{em}]$ . These are probably /səẏ́ėḿ/ and /səyəў́ėḿ/, the plural again formed by CV reduplication. The word looks as though it might be the resultative form of an unrecorded root \* yəm or \* ye<sup>2</sup> with the suffix -əm. Compare /səẏ́eḿ/ and the plural /səyəẏ́eḿ/ with səḿe<sup>2</sup> 'removed' and its plural səməḿe<sup>2</sup>, the resultatives of the root of mé<sup>2</sup>x 'remove it' (see §7.7.2). However, the word exists as a cognate or loan in several other languages in the region, namely Squamish  $si^{2}ám^{2}$ , Northern Straits  $si^{2}\acute{em}$ , Lushootseed  $si^{2}\acute{ab}$ . Kuipers (1967, 374) suggests that the Squamish  $si^{2}ám^{2}$  is from  $hi^{2}ám^{2}$  'return home' from the root  $\sqrt{ham^{2}}$ ,  $ham^{2}$  'be covered, come home' (cognate of Musqueam 'amit 'come'), inviting comparison with Latin dominus. Bates et al. (1994, 14) identify the Lushootseed  $si^{2}\acute{ab}$  as a nominalization of ' $i^{2}\acute{ab}$  glossed 'wealth' and as "an endearing term of address to a little boy," but they give no example of the usage of ' $i^{2}\acute{ab}$  with the meaning 'wealth,' their evidence for this gloss evidently being

simply a verb  $2i^{2}abil$  'become wealthy' (cf. Musqueam  $s \partial y \dot{a} m \theta \partial t$  'get rich, become a rich person').

(3) The plural of  $sp \neq l \times an$  'prairie, pasture' is  $sp \neq l \times an$  'prairies, pastures.' It seems possible that the long vowel in the plural is the result of the loss of an infix -l-, the underlying form being //sp $\neq$ l $\neq$ k $\neq$ n//. If so, the pattern does not exactly match any of those recorded.

(4) The plural of swiwlas 'teen-age boy, young man' is swawlas. The singular may be in origin the diminutive of a form \*swalas, compare Lushootseed swalus 'young man of noble parentage' (Bates et al. 1994, 247). Following the pattern of snawal 'canoe' and sninxwal 'little canoe' (see §8.4.2. below), we get swiwlas. The plural is formed as in snanaxwal 'little canoes,' being //swawalas// with the usual metathesis of the resonant and schwa and the two adjacent schwas producing a long full vowel.

(5) The plural of *scowtét* 'man's sister's husband, man or woman's child's spouse' is *sciwotét*. I cannot explain this apparent irregularity.

# 8.3.5. Suppletion

There is one instance of this, in which the plural has another plural form:  $s\vec{\lambda}(\vec{\lambda}qat)$  'child,'  $st\acute{e}^{2}ex^{w}at$  'children,'  $staw(x^{w}at)$  'bunch of children, many children.' The form of the singular suggests that it is the diminutive of an unrecorded  $*s\vec{\lambda}aqat$  (see §8.4.2.), probably from the root of  $\vec{\lambda}aqit$  'give birth.' The plural is clearly from a different root. The third form is hard to explain. It may be a unique example of a different kind of plural. The terms denote 'child' and 'children' referring to life status, not to kinship (see §21.1.1).

# **8.4. DIMINUTIVES**

Like plurals, diminutive forms are not available for all nouns. Those for which diminutives have been recorded include terms for persons, body parts, animals, artifacts, and natural features, but the list is not long. Again, as with plurality, a diminutive sense can also be expressed in the verb.

It seems likely that speakers vary in their use of diminutive forms. CC and DK produced them in elicitation more freely than JP and, I believe, used them more often. This may reflect a sex difference; women may use diminutives more than men do. Diminutive forms not only express the sense of smallness but can also convey a sense of endearment. They are used when speaking about and to children. A little girl may be called *tétən* 'little mother' (DK's father called her this) and a little boy *mémən* 'little father.' Persons feeling close use diminutives to one another. In response to a diminutive used to her, DK used the diminutive of *swétə* 'sweater':

(a) <sup>9</sup>i cən wəł-yə-hɨmɨx θə nə-swéwtə. (DK)
 AUX I already-along-be.removing ART my-sweater(DIM)
 'I am already taking off my little sweater.'

And they are used in speaking of little animals. Someone, DK remembered, after looking into a nest of baby mice, said:

(b) Žéwdom to qoqoláms i sosé'stx vos to molímqsons. (DK)
Žéwdom to qoqolám s 'oy' sosé'-stox vos to be.glittering ART eye(DIM.PL)-3POS and be.rising-CAUS-3TR ART molímqson-s nose(DIM.PL)-3POS
'Their little eyes were glittering and they were lifting up their little noses.'

Probably English with 'little eyes' and 'little noses' does not convey the feeling of the Halkomelem diminutives

When CC could not produce a diminutive for snake, she suggested it was because snakes are not "cute." Anyway, one can refer to the young of an animal simply by suffixing  $-a \cdot lt$  'young' to the uninflected form of the name. Occasionally this suffix occurs with diminutive forms.

Diminutives can also belittle. In relating a bit of gossip, a speaker referred, a bit sarcastically, to a married woman's secret lover as her  $st a^2 t l a s$  'little husband.'

Used in speaking of oneself, diminutives can express humility – real or required by etiquette. About 1917, Herman Guerin was instructed to begin a speech at a potlatch-like event with (c).

(c) q<sup>w</sup>əl-st-ámə cən ce? təňa <sup>?</sup>é·nθə sisləwa? <sup>?</sup>əθk<sup>w</sup>ək<sup>w</sup>əyəθ. (HG) q<sup>w</sup>əl-st-ámə cən ce? təňa <sup>?</sup>é·nθə sisləwa? speak-com-you(PL) I FUT this I poor(DIM) <sup>?</sup>əT-sk<sup>w</sup>ək<sup>w</sup>əyəθ your-slave(DIM)
'I, your poor little slave, will speak to you.'

Diminutives are formed in far fewer ways than are plurals. They are nearly all produced by reduplication of the first consonant. A schwa in the root generally becomes i in the diminutive, a repeated resonant is often glottalized, and a glottal stop may follow a full vowel.

# 8.4.1. With C<sub>1</sub> Reduplication and No Vowel Change

The nouns that follow this rule all have a full vowel in the root or base form. All but a few of this shape follow this rule.

(1) With no glottal stop ( $C_1 A C_2 \rightarrow C_1 A C_1 C_2$ -). A resonant may be glottalized in the diminutive.

| <i>púpt</i> 'little boat'        |
|----------------------------------|
| <i>s pú pəň</i> 'little spoon'   |
| <i>púps</i> 'little cat, kitten' |
| <i>mémən</i> 'little father'     |
|                                  |

| <i>smé·nt</i> //sménət// 'stone'      | smémnət 'little stone'                                            |
|---------------------------------------|-------------------------------------------------------------------|
| <i>máq<sup>w</sup>əm</i> 'bog'        | <i>mámq<sup>w</sup>əm</i> 'little bog'                            |
| <i>tén</i> 'mother'                   | tétəň 'little mother'                                             |
| <i>sť<sup>0</sup>ám</i> 'bone'        | <i>sť<sup>0</sup>áť<sup>0</sup>əm</i> 'little bone'               |
| <i>céləx</i> 'hand'                   | <i>cécləx</i> 'little hand'                                       |
| <i>sk<sup>w</sup>áye</i> ? 'squirrel' | <i>sk<sup>w</sup>ák<sup>w</sup>ye</i> ? 'little squirrel'         |
| <i>k<sup>w</sup>éťaň</i> 'mouse'      | kwékwťań 'little mouse'                                           |
| <i>kʷáxʷa</i> ? 'box'                 | <i>k៉<sup>w</sup>ák៉<sup>w</sup>x<sup>w</sup>a</i> ? 'little box' |
| swéta 'sweater'                       | swéwta 'little sweater'                                           |
|                                       |                                                                   |

(2) With a glottal stop after the vowel ( $C_1A \rightarrow C_1A^{\gamma}C_1$ )

| <i>sčáła</i> ? 'leaf'                                     | <i>sčá?čła</i> ? 'little leaf'                                                            |
|-----------------------------------------------------------|-------------------------------------------------------------------------------------------|
| <i>xét</i> 'path, road, door'                             | $x \dot{e}^2 x t$ 'little path'                                                           |
| <i>žáca</i> <sup>2</sup> 'lake'                           | $\check{x}\check{a}^{2}\check{x}ca^{2}$ 'little lake'                                     |
| $\dot{t}^{\theta} i k \dot{t}^{\theta} i k$ 'wagon, etc.' | $\dot{t}^{\theta} i^{\gamma} \dot{t}^{\theta} i k \dot{t}^{\theta} i k$ 'baby buggy' (JP) |
| <i>stálow</i> 'river'                                     | stá?tləw (JP), státələw (CC) 'creek'                                                      |

In the following, after the glottal stop, the expected sequence of two identical consonants is reduced to one:  $\theta \hat{a} \theta \partial n$  'mouth,'  $\theta \hat{a}^{2} \theta \partial n$  'little mouth.'

# 8.4.2. With C<sub>1</sub> Reduplication and Vowel Change

(1) Without glottalization ( $C_1 \mathfrak{p}$ -,  $C_1 \to C_1 i C_1$ -)

| <i>páwaý</i> 'flounder'                     | <i>pí paway</i> 'little flounder'           |
|---------------------------------------------|---------------------------------------------|
| <i>máqsən</i> 'nose'                        | <i>mímqsən</i> 'little nose'                |
| smá q <sup>w</sup> a? 'heron'               | <i>smímq<sup>w</sup>a</i> ? 'little heron'  |
| smáyaθ 'deer'                               | $smimy \partial \theta$ 'fawn, little deer' |
| <i>táməx<sup>w</sup></i> 'earth, land'      | $titm \partial x^w$ 'little piece of land'  |
| tánaqsan 'mallard'                          | <i>títnəqsən</i> 'little mallard'           |
| $t \delta x^{w} \theta \partial t$ 'tongue' | $titx^w \theta \partial t$ 'little tongue'  |
| snáx <sup>w</sup> ał 'canoe'                | <i>sníňx<sup>w</sup>ət</i> 'little canoe'   |
| ləmətú 'sheep'                              | <i>líİmətú</i> 'little sheep'               |

In the next two examples, a final  $-n\partial$  in the simple form becomes  $-ne^{2}$  in the diminutive, perhaps reflecting a historically earlier form of the word; cf. *NStr ŋdnd*' child').

| mź'nə | 'child (kin term)' | mí mne?    | 'little child' |
|-------|--------------------|------------|----------------|
| sžáňə | 'foot'             | s xí x ne? | 'little foot'  |

The following have a shift in stress from the second vowel in the simple form to the first in the diminutive:

| sq <sup>w</sup> əméý 'dog'   | <i>sq<sup>w</sup>íq<sup>w</sup>me</i> ŷ 'little dog, puppy' |
|------------------------------|-------------------------------------------------------------|
| $\theta q \acute{e}t$ 'tree' | $\theta i \theta q \partial t$ 'little tree'                |

| <i>Ăcés</i> 'island'       | <i>XíXcəs</i> 'little island'                  |
|----------------------------|------------------------------------------------|
| sť <sup>e</sup> ģám 'drop' | $st^{\theta}tt^{\theta}qam$ 'little drip' (AG) |

(2) With a glottal stop after the i  $(C_1 \rightarrow - C_1 i^{\gamma} C_1)$ 

| <i>sθáqaý</i> 'sockeye'               | $s\theta i^{\gamma}\theta q \partial y$ 'little sockeye'                      |
|---------------------------------------|-------------------------------------------------------------------------------|
| $st^{\theta}\delta k^{w}$ 'bug, worm' | $s\dot{t}^{\theta}i^{2}\dot{t}^{\theta}\dot{k}^{w}$ 'little bug, little worm' |
| <i>čséý</i> 'Douglas-fir'             | <i>ċí<sup>?</sup>ċsəý</i> 'little Douglas-fir' (CC)                           |

# 8.4.3. With CV Reduplication

There are a few words in which  $C_1 A$ - is reduplicated  $C_1 A C_1 a$ -:

| sčéxt 'stick'            | sčéčaxt 'little stick'                                      |
|--------------------------|-------------------------------------------------------------|
| sílo 'grandparent, etc.' | sísələ 'grandpa, grandma (address)'                         |
| yásaq <sup>w</sup> 'hat' | yá·ysaq <sup>w</sup> //yáyəsaq <sup>w</sup> // 'little hat' |

There are a few words in which an unstressed schwa is reduplicated along with the first consonant:

| səmáýe? 'bee, wasp'                                        | səsəmáýe? 'little bee, little wasp'                             |
|------------------------------------------------------------|-----------------------------------------------------------------|
| $sp \partial lq^{w} i i^{\theta} e^{\gamma}$ 'dead person' | $sp  app a p a l a q^{w} i t^{\theta} e^{\gamma}$ 'screech owl' |
| $x^{w} \partial n(t \partial m)$ 'White person'            | x <sup>w</sup> əx <sup>w</sup> ənítəm 'White child'             |

There are a few words in which a stressed schwa is replaced by i or  $i^2$ :

| <i>kvála</i> w 'skin'                                  | <i>k៉<sup>w</sup>ík៉<sup>w</sup>ələw</i> 'little skin'                                                                       |
|--------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|
| <i>sk<sup>w</sup>áyəθ</i> 'slave'                      | $s\vec{k}^{w}\vec{i}\vec{k}^{w}\partial y\partial\theta$ (JP), $s\vec{k}^{w}\partial k^{w}\partial y\partial\theta$ (AG, HG) |
|                                                        | 'little slave, small pet animal'                                                                                             |
| <i>x<sup>w</sup>ə́lməx<sup>w</sup></i> 'Indian, tribe' | $x^{wi}x^{w}\partial lm\partial x^{w}$ 'little Indian'                                                                       |

There are also a few diminutives formed in unique ways. It will be better to consider these after the diminutive plurals.

# 8.5. DIMINUTIVE PLURALS

Diminutives of nouns are pluralized either with an infixed *-l-* or with a vowel change.

# 8.5.1. With Infixed -l-

Most nouns with diminutives of type 1 (those with an original full vowel) and some nouns of type 2 (those with schwa replaced by *i*) have diminutive plurals with  $-\partial l$ - preceding the vowel:  $C_1 \dot{A} C_1 - \rightarrow C_1 \partial 1 \dot{A} C_1$ . I will list base forms (left column), as well as diminutives (middle column) and diminutive plurals (right column), but give glosses for the uninflected forms only.

| pús 'cat'           | púps    | pəlúps    |
|---------------------|---------|-----------|
| smé·nt 'rock'       | smémnət | sməlémnət |
| <i>céləx</i> 'hand' | cécləx  | cəlécləx  |

| sk <sup>w</sup> áye? 'squirrel' | sk <sup>w</sup> ák <sup>w</sup> ye? | sk¤əlák <sup>w</sup> ye? |
|---------------------------------|-------------------------------------|--------------------------|
| páwaý 'flounder'                | pí powoý                            | pəli pəwəy               |
| <i>máqsən</i> 'nose'            | mímqsən                             | məlímqsən                |
| táməx <sup>w</sup> 'land'       | títməx <sup>w</sup>                 | təlítməx <sup>w</sup>    |

In the next, English provided the base form:

In the next, no diminutive singular has been recorded:

| sk <sup>w</sup> łé <sup>·</sup> y (JP), sk <sup>w</sup> łéy (CC) | <br>sk <sup>w</sup> əlík <sup>w</sup> łey 'small |
|------------------------------------------------------------------|--------------------------------------------------|
| 'little-neck clam'                                               | little-neck clams'                               |

In the next, the *-l*- of the plural is glottalized:

| <i>k</i> <sup>w</sup> éť ə'n | 'mouse' | <i>k</i> <sup>w</sup> ék <sup>w</sup> ťə'n | 'little mouse' | <i>k</i> <sup>w</sup> əlék <sup>w</sup> tən | 'little mice' |
|------------------------------|---------|--------------------------------------------|----------------|---------------------------------------------|---------------|
|                              |         |                                            |                |                                             |               |

If a glottal stop appears in the diminutive, it also appears in the diminutive plural:

| scáła? 'leaf'     | sčá?čła? (dim.)      | <i>scəlá<sup>2</sup>cła<sup>2</sup></i> (dim. pl.) |
|-------------------|----------------------|----------------------------------------------------|
| <i>xét</i> 'road' | $x e^{2} x t$ (dim.) | xəlé <sup>2</sup> xł (dim. pl.)                    |

In the next three, a glottal stop appears in the plural although there is none in the singular:

| scéxt 'stick'                | sčéčəxt (dim.)                        | scalé <sup>2</sup> caxt (dim. pl.)     |
|------------------------------|---------------------------------------|----------------------------------------|
| $\theta q \acute{e}t$ 'tree' | $\theta i \theta q \partial t$ (dim.) | <i>θəlí<sup>9</sup>θqət</i> (dim. pl.) |
| <i>téwət</i> 'shell'         | <i>tétəwət</i> (dim.)                 | <i>ťəlé<sup>2</sup>ťwət</i> (dim. pl.) |

#### 8.5.2. With Vowel Change

Some nouns that have schwa in the base form replaced by *i* in the diminutive have a diminutive plural in which the *i* is replaced by schwa. There may also be another schwa before  $C_2$ . The rule is  $C_1 i C_1 C_2 \rightarrow C_1 i C_1 (2) C_2$ .

| <i>Žcés</i> 'island'                    | <i>ĂíĂcəs</i> 'islet'                                | <i>ĂóĂəcəc</i> 'islets'                             |
|-----------------------------------------|------------------------------------------------------|-----------------------------------------------------|
| sq <sup>w</sup> əméý 'dog'              | <i>sq<sup>w</sup>íq<sup>w</sup>mə</i> ý 'little dog' | sq <sup>w</sup> əq <sup>w</sup> əme ýa·lł 'puppies' |
| <i>q<sup>w</sup>łéý</i> 'driftwood log' |                                                      | $q^{w} \delta q^{w} t e \dot{y}$ 'scraps of         |
|                                         |                                                      | driftwood'                                          |

If  $C_1$  is a resonant, it may be doubled after the stress:

 $sm \delta y \partial \theta$  'deer'  $sm i m y \partial \theta$  (dim.)  $sm \delta m m \partial y \partial \theta$  (dim. pl.)  $sn \delta x^{w} \partial t$  'canoe'  $sn i n x^{w} \partial t$  (dim.)  $sn \delta n n \partial x^{w} \partial t$  (dim. pl.)

#### **8.6. PROBLEMATIC DIMINUTIVES**

There are a dozen or so nouns that have diminutive or diminutive plural forms that are irregular enough to call for special comment and, in several instances, to suggest a reinterpretation of the base form. I list these here followed by their plural, diminutive, and diminutive plural forms, with a dash where no form has been recorded.

(1)  $st^{\theta} \dot{a} \dot{m}$  'bone,'  $st^{\theta} \dot{a} l \partial \dot{m}$  (pl.),  $st^{\theta} \dot{a} t^{\theta} \partial \dot{m}$  (dim.),  $t^{\theta} \dot{a} t^{\theta} \partial l \dot{a} \dot{m}$  (CC) (dim. pl.). The diminutive plural may be a diminutivized plural rather than the usual pluralized diminutive. The final full vowel and missing *s*- are not explainable.

(2)  $q \delta l \partial m$  'eye,'  $q \partial q \ell l \partial m$  (pl.), —,  $q \partial q \partial l \delta m$  (DK) (dim. pl.). The diminutive plural is similar to (1) in the final vowel.

(3)  $m\dot{a}^2 \partial q^w$  'duck,'  $m\partial m\dot{a}^2 \partial q$  (pl.),  $m\dot{m}\partial q^w$  (dim.),  $m\partial t\dot{m}\partial q^w$  (dim. pl.). The diminutive suggests that the base may be  $//m\dot{a}^2 \partial q^w//$ . The diminutive plural is regularly formed from the diminutive.

(4) *léləm* 'house,' *həlléləm* (pl.), *líləm* (dim.), *hállèm* (CC), *háləllèm* (JP) (dim. pl.). The plural implies a base //lələləm//, but the diminutive implies a base //ləm//. The diminutive plural could have the same base as the plural and the stress and/or vowel pattern of (1) and (2).

(5)  $s\vec{\lambda}\acute{e}l aqam$  'fierce thing,'  $s\vec{\lambda}a\dot{l}\acute{e}l aqam$  (pl.),  $s\vec{\lambda}i'\vec{\lambda}alaqam$  (dim.),  $s\vec{\lambda}a\vec{\lambda}\acute{e}laqam$  (dim. pl.). CC identified the Chilliwack as  $s\vec{\lambda}e'\acute{e}laqam$ , probably //s $\vec{\lambda}a'\acute{e}laqam$ //. Such a base would account for the Musqueam diminutive and diminutive plural forms.

(6)  $s^2 \dot{a} \theta as$  'face,'  $s^2 \dot{a}^2 \partial l \theta as$  (pl.),  $s^2 \partial \theta \dot{a}s$  (JP),  $s^2 \dot{a}^2 \theta as$  (DK) (dim.). The diminutive form given by JP seems to be the product of nothing but a shift in stress. It has no parallel, unless it is in (7). A diminutive plural was not recorded.

(7)  $sp\dot{a}d$  'raven,'  $spalp\dot{a}d$  (pl.),  $spal\dot{a}d$  (dim.) 'crow,'  $spal\dot{a}pla\dot{d}$  (dim. pl.) 'crows.' The long vowel in 'raven' is probably the result of the loss of the first of two identical resonants (see §1.5.6). This is nicely reflected in the diminutive plural;  $spal\dot{a}pla\dot{d}$  implies a diminutive  $*sp\dot{a}pla\dot{d}$ , which in turn implies a base  $*sp\dot{a}la\dot{d}$ , which is what, with the loss of the medial resonant, became  $sp\dot{a}d\dot{d}$ . The word  $spal\dot{a}d$  'crow,' however, does not fit into this sequence. As a diminutive of 'raven,' if it is, it must be the product of the same rule that produced JP's diminutive of 'face' (6).

(8)  $\dot{q}^{w}\dot{t}\cdot\dot{n}$  'ear,'  $\dot{q}^{w}\partial n\dot{q}^{w}\dot{t}\cdot\dot{n}$  (pl.),  $\dot{q}^{w}\dot{i}\dot{q}^{w}\dot{t}\cdot\dot{n}$  (CC),  $\dot{q}^{w}\dot{t}\dot{q}^{w}n\partial\dot{n}$  (JP) (dim.),  $\dot{q}^{w}\partial l\dot{q}^{w}\dot{t}\cdot\dot{n}$  (CC) (dim. pl.). CC's diminutive is the product of CV reduplication but with the stress on the second vowel, which is not usual. JP gave his diminutive during elicitation, saying he had never heard the word used but it would be  $\dot{q}^{w}\dot{t}\dot{q}^{w}n\partial\dot{n}$ . This form, if produced by the usual rule, implies a base  $*\dot{q}^{w}\delta n\partial\dot{n}$ , which is in fact 'ear' in the Nanaimo dialect and no doubt what became, through loss of the medial resonant, the Musqueam  $\dot{q}^{w}\dot{t}\cdot\dot{n}$ .

(9) spáłžan 'prairie,' spéłžan (pl.), spépłžan (dim.), spalépłžan (dim. pl.). As indicated under plurals, the long vowel in the plural may be the product of the loss of an infixed *-l*. On the other hand, the diminutive form implies a short full vowel in the base.

# **8.7. RESULTATIVE FORMS**

There are several words formed from nouns with a prefix *s*- and meaning 'possessing' or having a meaning similar to that of an English past participle used adjectivally; for example, from  $t \ell l \partial$  'money' we have  $st \ell t \partial l \partial$  'having money, monied.' Some, though not all, of these resemble the resultative forms of verbs and so I am tentatively identifying them as resultative forms of nouns and the prefix as  $s_{-2}$  'resultative.'

A small set (of four so far) are formed from nouns that have not undergone any internal modification. The nouns appear below on the left and their resultative forms on the right.

| méxcon 'louse'                              | sméxcon 'lousy'                                         |
|---------------------------------------------|---------------------------------------------------------|
| <i>k៉<sup>w</sup>ál</i> a 'belly'           | sk <sup>w</sup> ála 'pot-bellied'                       |
| <i>qánəx<sup>w</sup></i> 'gullet'           | sqánax <sup>w</sup> 'greedy'                            |
| $i^{\theta} \dot{y} st \partial n$ 'antler' | $st^{\theta} \delta yst \partial n$ 'buck with antlers' |

These might, without too great a stretch, be glossed 'loused,' 'bellied,' 'gulleted,' and 'antlered.' The first of these has a diminutive, sm(mxcon) 'lousy little thing,' from m(mxcon) 'little louse.'

In the others recorded, the noun undergoes internal changes (reduplication, stress shift, etc.) resembling those we see in the resultative forms of verbs (see §7.7).

| <i>smélək</i> <sup>w</sup> 'warrior's hide shirt' | smémolok <sup>w</sup> 'wearing a warrior's shirt'                           |
|---------------------------------------------------|-----------------------------------------------------------------------------|
| <i>méqe</i> ? 'fallen snow'                       | <i>sméməqe</i> <sup>9</sup> 'all snowy'                                     |
| <i>télə</i> 'money'                               | stétələ 'having money' (DK)                                                 |
| <i>łéčtən</i> 'knife'                             | słéłactan 'armed with a knife' (AG)                                         |
| <i>Žímən</i> 'tendon'                             | s Ží Žəmən 'powerful, muscular'                                             |
| $k^w a x^w a^2$ 'box'                             | $s\vec{k}^{w}\vec{a}x^{w}\vec{k}^{w}ax^{w}a^{2}$ 'boxed (pl.)' <sup>1</sup> |
| $q\dot{a}^{2}$ 'water'                            | sqəqá? 'watery, thin (as soup)'                                             |
| <i>táməx<sup>w</sup></i> 'earth'                  | $st \partial m i x^w$ 'dirty (as clothes)'                                  |
| máňa 'child'                                      | səmné? 'having a child, one who has a child'                                |
| snáx <sup>w</sup> ať 'canoe'                      | $s \partial n x^w \partial t$ 'one who has a canoe'                         |
| $sk^w \delta y \partial \theta$ 'slave'           | $s\vec{k}^{w}\partial y\hat{a}\theta$ 'one who has a slave'                 |

There is also a jokingly produced hybrid form: skikf 'skiff,' skikft 'skiffed, having a skiff' (DK). Here the final -t may be -t 'stative' or may be the English -ed.

The following have been recorded as verbs only, the first formed with  $x^w \partial$ -'become' and the second with  $-st \partial x^{w_i}$  'causative,' both affixes that are commonly used with resultatives of verbs:

<sup>1</sup> So recorded, the second vowel may be a schwa echoing the others.

| $k^{w}i^{2}x^{w}$ 'pitch' | $x^{w} \partial s \vec{k}^{w} \partial^{2} i x^{w}$ 'become pitched up' |
|---------------------------|-------------------------------------------------------------------------|
| <i>ləsék</i> 'bag, sack'  | <i>sliləsékstəx</i> <sup>w</sup> 'have it bagged'                       |

There are perhaps semantic limits only on what nouns can have resultative forms. Not enough examples have yet been collected to determine whether they are formed in ways altogether parallel to the formation of resultatives of verbs.

#### 8.8. THE INFLECTION OF VERBS DERIVED FROM NOUNS

Nouns may be used to produce verbs by means of several prefixes, among them c- 'get, make, do,' t- 'partake,'  $tx^{w}$ - 'buy.' These verbalized nouns may have progressive forms, as in:

- (a) <sup>9</sup>i cən tx<sup>w</sup>-łé<sup>9</sup>łəċtən. (AG)
  AUX I buy-knife(PROG)
  'I am buying a knife.' (cf. *łéċtən* 'knife')
- (b) ?i cən tx<sup>w</sup>-sé?pləl. (AG)
  AUX I buy-bread(PROG)
  'I am buying bread.' (cf. səplíl 'bread')
- (c) ni tx<sup>w</sup>-kəká· k<sup>w</sup>θə John. (AG)
   AUX buy-car(PROG) ART John
   'John is buying a car.' (cf. ká· 'car')

Again, however, not enough examples have yet been collected to determine whether these progressives are formed in ways altogether parallel to the formation of progressives of verbs, or whether there are fuller paradigms.

# 8.9. VERBS THAT ACT AS NOUNS

There are some verb forms that can be glossed as nouns. As indicated in §7.4.1,  $colice\dot{A}om$ , the diminutive plural of  $c\dot{e}\dot{A}om$  'be jumping,' literally, 'little ones are jumping' means 'sand fleas'; in §7.6,  $q\dot{o}\dot{n}qo\dot{n}$ , the dispositional-iterative of  $q\dot{e}\dot{n}$  'steal,' means both 'likely to steal' and 'thief'; and in §7.7.6, the possible resultative  $\theta q\dot{e}t$  means both 'standing upright (as a spit by the fire)' and 'tree.' Of course, any verb form can be preceded by an article and serve as a relative clause, so perhaps that is what these forms are when functioning as nouns.

In general, the capacity to take the possessive affixes seems to distinguish nouns from verbs. However, there are verb forms that function as nouns and do take possessives, such as  $h\delta \dot{y}q^w$  'be burning (from  $y\delta q^w$  'burn'), fire';  $h\delta \dot{l}k^w$  'be breaking (from  $l\delta k^w$  'break'), pocketknife'; and  $a'\dot{t}$  'stretch, slingshot.' I cannot say how many such verb forms there are.

# **9** Morphology of the Root 3: Adjectives and Adjective-like Words

Adjectives proper constitute a small class of words that are distinct from both verbs and nouns. Morphologically, adjectives are like verbs and unlike nouns in that they do not take possessives directly; they must first be nominalized with s- 'nominalizer.' They are like nouns and unlike verbs, however, in not having progressive forms. Like both verbs and nouns, they have plural and diminutive forms. However, verbs have diminutives of progressive forms only, while nouns and adjectives have diminutives based on otherwise uninflected forms. Syntactically, adjectives differ from verbs in that they can appear before nouns in predicates (see §3.7.3). Numerals form a class of adjective but will be dealt with in §19.

Adjectives proper are simple and complex. Simple adjectives are mainly bare roots (with or without internal modification for plurality or diminutiveness), while complex adjectives are formed from adjective roots and lexical suffixes.

As indicated in §3.8.3.4, in addition to adjectives proper there are a number of words that can be described as adjective-like.

#### 9.1. SHAPES OF ADJECTIVE ROOTS

Adjective roots appear in the common shapes of verb roots, that is, CVC and CVRC, but also in a few others. The examples that follow include most of the adjective roots so identified.

With the shape CV there is only  $\theta i$  'big' (possibly // $\theta ih$ // and so CVC; see \$1.5.8).

With the shape CVC are the following:

| $\dot{p}\dot{a}\dot{q}$ 'white'   | $m \delta q^w$ 'thick (around, as a rope)'                  |
|-----------------------------------|-------------------------------------------------------------|
| nás 'fat'                         | $t^{\theta} \delta c$ 'stiff'                               |
| $i \delta q^w$ 'wet'              | $\dot{\chi} \delta p$ 'deep'                                |
| $c \hat{a} k^w$ 'far'             | $\dot{X}\dot{\delta}\dot{X}^{w}$ 'hard, solid, difficult'   |
| <i>qál</i> 'bad'                  | $\vec{\chi} t^{2}$ 'expensive, important, difficult'        |
| $q \delta \check{x}$ 'many, much' | <i>ž</i> <sup>w</sup> <i>óm</i> 'fast'                      |
| ²áý 'good'                        | $\check{x}i\check{t}^{\theta}$ 'not fully cooked, not ripe' |

Three words denoting dimensions seem to each consist of a CVC root and a suffix *-t* 'stative' ( $\S$ 12.1.6):

 $\dot{\chi} \dot{e} q t$  'long'  $l \dot{q} \dot{e} t$  'wide' (cf.  $l \dot{e} \dot{q} \partial t$  'lay it down')  $p l \dot{e} t$  'thick (as cloth, bushes, etc.)'

Four words denoting colours each consist of a CVC root and the prefix c-(§12.2.1):

 $ck^{wim}$  'red' $cx^{wik'w}$  'grey'cqix 'black' $cq^{way}$  'green, yellow, pale'

And three others may consist of CVC roots reduplicated:

*pé<sup>°</sup>pe<sup>°</sup>* 'fuzzy, woolly' *yéýe<sup>°</sup>* (//yé<sup>°</sup>ye<sup>°</sup>//) 'secure, tight, connected'

 $\check{x}\acute{e}^{?}\check{x}e^{?}$  'holy, forbidden, extraordinary'

With the shape CVCC are:

 $\theta \hat{e}^{\gamma} t$  'dark'  $\check{x} \hat{e} \hat{w} \hat{s}$  'new'

With the shape CCVC is tsas 'poor (especially in possessions), pitiful, suffering.' With CVCV is  $\tilde{x}ela$  'scarce.' With CVCVC are:

 $\dot{x}\dot{a}\dot{v}\dot{\vec{x}}$  'cold'

| <i>xəlák</i> <sup>w</sup> 'round' | x <sup>w</sup> átəs 'heavy'                     |
|-----------------------------------|-------------------------------------------------|
| <i>hílək</i> <sup>w</sup> 'happy' | <sup>?</sup> <i>ax</i> <sup>w</sup> ín 'little' |
| səléwe <sup>9</sup> 'pitiful'     | <i>mál al 'soft, spongy'</i>                    |
| <i>líləq</i> 'easy, cheap'        | cícəł 'high' (√cəł)                             |
| míman 'little' (cf. mána 'child') |                                                 |
|                                   |                                                 |

Several have an initial s-, which may be  $s_{-2}$  'resultative':

| <i>sžís</i> 'somewhat dark skinned' | smét <sup>e</sup> an 'proud' |
|-------------------------------------|------------------------------|
| stámk <sup>w</sup> 'blunt'          | scowét 'smart'               |

A few are of shapes that suggest roots that have undergone internal modification. Four, each denoting the lower end of a variable quality, look as though they are formed of CVC roots with CV reduplication and, in three of the four, an infixed glottal stop. The pattern is not identifiable with any appearing elsewhere. These are:

 $t\partial^2 t \cdot tx$  'narrow (as something flat)  $q^w \partial^2 t q^w s, q^w \partial^2 t \cdot y \partial q^w s$  'thin (as a rope or stick)'  $\dot{c} \partial \dot{c} t \cdot \dot{x}$  'short' (cf.  $\dot{c} t \cdot \dot{x}$  'how long!'; see §20.2)  $x^w \partial^2 \dot{a} x^w e^2$  'light (not heavy)' (cf.  $x^w \dot{e}^2 t$  'lighten it, make it less')  $k^w \dot{a} k^w \partial s$  (JP, AG),  $k^w \dot{a}^2 k^w \partial s$  (CC) 'hot (in temperature, spicy, passionate)' (<  $k^w \dot{e} s$  'get burnt' with a vowel change; cf.  $k^w \dot{e} k^w \partial s$  'be getting burnt' [AG])
Formed by CVC reduplication is kwámkwam' strong.'

# 9.2. PLURALS

Plural forms of adjectives are used in predicates, such as the plural of  $p \delta q$  'white' in (a).

(a) wəm śk<sup>w</sup> pe pd. (CC)
 wə-m śk<sup>w</sup> pe pd
 EST-all white(PL)
 'They are all white.'

And they are used to modify plural nouns, as in (b).

(b) <sup>?</sup>éləy məlstáyəx<sup>w</sup>. <sup>?</sup>áwəte<sup>?</sup> šxsk<sup>w</sup>éys k<sup>w</sup>eθ nem sqaqamat. (JP) šx<sup>w</sup>-sk<sup>w</sup>év-s ?éləv məlstávəx<sup>w</sup>. ?áwəte? k »a good(PL) person(PL) none OBLNOM-impossible-3POS ART <sup>?</sup>əT-s-nem sqəqamət your-NOM-go being.joined 'They are good people. There is no reason why you can't mix with them.'

In (b), both  $2\delta y'$  good' and  $m \partial st \delta y \partial x^w$  'person' are pluralized. However, a plural adjective may be enough to show plurality; in (c), the adjectives are plural forms but the noun is singular.

(c) Ží<sup>?</sup>s k<sup>w</sup>s mís <sup>2</sup>éləy məstə́yəx<sup>w</sup>, k<sup>w</sup>áləmk<sup>w</sup>əm məstə́yəx<sup>w</sup>. (JP 1)
[s-c-]Ží<sup>?-s</sup> k<sup>w</sup> s-mí-s <sup>2</sup>éləy məstə́yəx<sup>w</sup>
[NOM-get-]value-3POS ART NOM-come-3POS good(PL) person k<sup>w</sup>áləmk<sup>w</sup>əm məstə́yəx<sup>w</sup>
strong(PL) person
'They wanted them to become good people, strong people.'

Plurals are marked in adjectives in ways similar to those marking plurals in verbs and nouns, with an infixed *-l-*, CV reduplication, or CVC reduplication.

Most roots with full vowels form plurals with infixed *-l*-. The stress is on the vowel preceding the *-l*-, and there is no glottalization in the following (the plural form follows the singular):

| <i>žéws</i> 'new'                              | <i>xélə</i> ws                  |
|------------------------------------------------|---------------------------------|
| <i>Žéqt</i> 'long'                             | Žéləqt                          |
| <i>k<sup>w</sup>ámk<sup>w</sup>əm</i> 'strong' | <i>k</i> ¤áləmk <sup>w</sup> əm |
| The -l- is glottalized in:                     |                                 |
| nás 'fat'                                      | náləs                           |
| The -l- and glottal stop are met               | athesized in:                   |
| $sm\acute{t}^{\theta}an$ 'proud'               | smé?əlť <del>0</del> ən         |

The stress follows the -*l*- in:

 $\check{x}\acute{e}^{?}\check{x}e^{?}$  'holy'  $\check{x}\imath\acute{e}^{?}\check{x}e^{?}$ 

One adjective with stressed schwa has its plural formed with -l- and a full vowel in place of the schwa:

 $2\dot{j}\dot{j}$  'good'  $2\dot{\ell}l\dot{j}\dot{j}$ 

Several of the C<sub>2</sub>C shaped adjectives have CV reduplication and a full vowel in place of schwa or zero:

| <i>páq</i> 'white'          | <i>pé p</i> q |
|-----------------------------|---------------|
| <i>qál</i> 'bad'            | qéqəl         |
| x <sup>w</sup> átəs 'heavy' | xwáxwtəs      |
| płét 'thick'                | pé pł ət      |
| <i>lqét</i> 'wide'          | łéłqət        |

One adjective with a full vowel has its plural formed with CV reduplication:

 $\theta i$  'big'

```
θίθə
```

A few with initial C<sub>2</sub>C- and CC- have plurals formed with CVC reduplication:

| səléwe <sup>9</sup> 'poor'        | səlsəléwe?            |
|-----------------------------------|-----------------------|
| <i>xəlák</i> <sup>w</sup> 'round' | xəlxəlák <sup>w</sup> |
| tsás 'poor'                       | təstəsás              |

The following seems irregular:

<sup>?</sup>əmámən

# 9.3. DIMINUTIVES

The few examples of diminutives of adjectives that have been recorded resemble diminutives of nouns.

Adjectives with full vowels show reduplication of the first consonant with a glottal stop after the vowel or glottalization of a resonant.

| <i>žéws</i> 'new'            | $\check{x}\acute{e}^{2}\check{x}$ $\partial\dot{w}s$ 'new little (one)' |
|------------------------------|-------------------------------------------------------------------------|
| nás 'fat'                    | náns 'fat little (one)'                                                 |
| smét <sup>e</sup> ən 'proud' | <i>smémt<sup>e</sup>an</i> 'proud little (one)'                         |

Those with initial C<sub>2</sub> or CC show reduplication of the first consonant, /i/ replacing /<sub>2</sub>/, and possibly a glottal stop following the vowel.

| $\dot{p}\dot{a}\dot{q}$ 'white'   | $\vec{p}\vec{i}\vec{p}\vec{q} \sim \vec{p}\vec{i}^{2}\vec{p}\vec{q}$ 'white little (one)' |
|-----------------------------------|-------------------------------------------------------------------------------------------|
| <i>xəlák</i> <sup>w</sup> 'round' | <i>xixəlák</i> <sup>w</sup> 'round little (one)'                                          |
| <i>x<sup>w</sup>átəs</i> 'heavy'  | $x^{w}i^{2}x^{w}t\partial s$ 'heavy little (one)'                                         |
| <i>tsás</i> 'poor'                | <i>títsəs</i> 'poor little (one)                                                          |
| səléwe? 'pitiful'                 | síslawe? 'pitiful little'                                                                 |

There is one instance of triplication of the initial consonant:  $\theta i$  'big,'  $\theta i \theta_{\partial} \theta i$ 'a little bigger.'

# 9.4. DIMINUTIVE PLURALS

Diminutive plurals are formed either from the diminutive by an infixed *-l-* or from the base form by CV reduplication and schwa replacing a full vowel.

| base                          | plural                 | diminutive                 | diminutive plural                                                                                                             |
|-------------------------------|------------------------|----------------------------|-------------------------------------------------------------------------------------------------------------------------------|
| <i>páq</i> 'white'            | ,<br>pé pở             | pí pở ~ pí <sup>9</sup> pở | pəlí pở ~ pəlí? pở ~ pó pəð                                                                                                   |
| <i>tsás</i> 'poor'            | təstəsás               | títsəs                     | təlítsəs                                                                                                                      |
| nás 'fat'                     | náləs                  | náns                       | náňnəs                                                                                                                        |
| θí 'big'                      | θίθə                   | θίθəθί                     | $\theta \partial li \theta \partial \theta i \sim \theta \partial l i^2 \theta \partial \theta i$<br>'become a little bigger' |
| səléwe <sup>9</sup> 'pitiful' | səlsəléwe <sup>9</sup> | sísləwe?                   | səlísləwe?                                                                                                                    |

# 9.5. VERBS FORMED FROM ADJECTIVE ROOTS

Adjective roots can take the prefix  $x^{w_{\partial}}$  'become' or the reflexive suffix  $-\theta_{\partial t} \sim -\theta \delta t$  with the sense 'become' (see §10.5.1), as in:

 $x^{w} \partial \theta i$  'get big' (<  $\theta i$  'big')  $x \partial^{w} \partial \theta i \partial \partial$  'get big, get bigger'  $n \dot{a} s \theta \partial t$  'get fat' (<  $n \dot{a} s$  'fat')  $^{2} \partial y \dot{\theta} \dot{a} t$  'get better' (<  $^{2} \dot{s} \dot{y}$  'good')  $\dot{k}^{w'} a \dot{m} \dot{k}^{w} \partial m \partial t$  'get stronger' (<  $\dot{k}^{w} \dot{a} \dot{m} \dot{k}^{w} \partial \dot{m}$  'strong')

Adjectives can also take the suffix  $-st \partial x^w$  'causative' (see §10.3), as in:

 $\vec{k}^{w} \vec{a} \vec{m} \vec{k}^{w} \vec{o} \vec{m} \vec{s} \vec{t} \vec{a} \vec{x}^{w}$  'strengthen' (<  $\vec{k}^{w} \vec{a} \vec{m} \vec{k}^{w} \vec{o} \vec{m}$  'strong')  $\vec{x} \vec{e}^{2} \vec{x} \vec{o} \vec{s} \vec{t} \vec{a} \vec{x}^{w}$  'believe to be sacred' (<  $\vec{x} \vec{e}^{2} \vec{x} \vec{e}^{2}$  'holy')

# 9.6. COMPLEX ADJECTIVES

These are adjective-like words composed of adjective roots and lexical suffixes, in which the root denotes some quality and the suffix a class of phenomena. They can often be literally translated with an English compound ending in *-ed*, as in:

| COMPOUND                                                        | ROOT                                           | SUFFIX                                 |
|-----------------------------------------------------------------|------------------------------------------------|----------------------------------------|
| $\theta$ íqsən 'big-nosed'                                      | $\theta i$ 'big'                               | - <i>əqsən</i> 'nose'                  |
| $^{\gamma} \partial \dot{y} \dot{a} \theta$ 'sharp'             | ?áý 'good'                                     | $-a^{2}\theta$ 'edge(lip, mouth)'      |
| $q \partial l \hat{a}^{2} \theta$ 'dull'                        | <i>qál</i> 'bad'                               | $-a^{2}\theta$ 'edge (lip, mouth)'     |
| $t^{\theta}c i t^{\theta}e^{\gamma}$ 'stiff-textured'           | $\dot{t}^{\theta} \delta c$ 'stiff'            | $-i t^{\theta} e^{\gamma}$ 'blanket'   |
| $x^{w}\theta i q \partial n$ 'loud'                             | $\theta i$ 'big'                               | -qən 'throat'                          |
| $p \partial t \hat{a} \cdot y \theta \partial n$ 'thick-lipped' | <i>płát</i> 'thick'                            | $-a \cdot y \theta \partial n$ 'mouth' |
| <i>lcíws</i> 'tired'                                            | <i>tác</i> 'become possessed, out of control'? | -iws 'body'                            |

See §13 for other examples.

Complex adjectives may have plural forms. Compare (a) with (b) and (c) with (d):

- (a) θí x > n t > st<sup>2</sup>θ' q > m. (AG)
   θí x > n t > st<sup>2</sup>θ' q > m
   big-foot ART drop
   'There's a big drop.'
- (b) θίθəxən tə słáməx<sup>w</sup>. (AG)
   θίθə-xən tə słáməx<sup>w</sup>
   big(PL)-foot ART rain
   'The rain is in big drops.'
- (c) x<sup>w</sup>λ aqtíwan tθe? θqét. (JP) x<sup>w</sup>-λ eqt-iwan tθé? θqét inside-long-rump that tree 'That tree is tall.'
- (d) x<sup>w</sup>λ alqtíwan ta θaqθáqat. (JP)
   tall(PL) ART trees
   'The trees are tall.'

Of course, adjective roots with lexical suffixes also form words that are clearly nouns, such as  $p\dot{q}\delta lq \partial n$  'mountain goat' ( $< p\dot{\delta}\dot{q}$  'white,'  $-\partial lq\partial n$  'animal fur, hair').

# 9.7. ADJECTIVE-LIKE WORDS

As indicated in §3.8.3.4, adjective-like words and phrases are produced by enclosing a noun, adverb, or interrogative word with  $w \rightarrow \dots al$  or a noun with  $s - \dots -a^2 l$ .

Resultative forms of verbs (§7.7) and (in the few cases where they exist) of nouns (§ 8.7) can also modify nouns. A few examples are:

```
stətés 'near' (< tás 'arrive there'; cf. tsát 'approach')

s\theta \partial \theta \acute{e} \vec{k}^{w} 'straight' (< \theta \acute{a} \acute{k}^{w} 'get taut'; cf. \theta \acute{k}^{w} \acute{a} t 'pull')

st\acute{e}taw 'bright' (< \sqrt{tew}; cf. téwəl 'light up')

stəm(ix^{w} 'dirty' (< táməx^{w} 'earth')

sqəq\acute{a}^{2} 'thin, watery (as soup)' (< q\acute{a}^{2} 'water')

sinx^{w}\acute{e}t 'having a canoe' (< sn\acute{a}x^{w}\partial t 'canoe')

sta^{2}\acute{t}l 'lonely' (< \acute{t}i^{2}\partial l 'become lonely')

s\acute{a}lq^{w} 'lonely' (< ?)

s\acute{a}llas 'half drunk' (< ?)
```

Other adjective-like words with initial s- are:

s<sup>2</sup>ómot 'lazy' (< <sup>2</sup>ómot 'sit') sqónox<sup>w</sup> 'greedy' (< qónox<sup>w</sup> 'gullet') sk<sup>w</sup>ólo 'pot-bellied (< k<sup>w</sup>ólo 'belly')

Two other verb forms can have adjectival meanings. They appear in predicates but I have not recorded them as modifiers. These are:

(1) Dispositional-iterative forms of verbs (§7.6), such as:

```
m \delta t m \delta t 'limber' (\sqrt{m \delta t}; cf. m \delta t \delta t 'bend')
k^w \delta t k^w \delta t 'cranky (likely to capsize)' (\langle k^w \delta t 'spill')
q^w \delta l q^w \delta t 'talkative' (\langle q^w \delta t 'speak')
```

(2) Perfective and/or progressive forms with  $-\partial m$  'intransitive' (§10.2.1), such as:

| <i>détəm</i> 'sweet'                                                                                        | níq <sup>w</sup> əm 'soft, flexible'                                                              |
|-------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------|
| $t^{\theta} e^{t^{\theta}} a^{w} a^{w} a^{w}$ 'blue'                                                        | <i>títəx<sup>w</sup>əm</i> 'slimy'                                                                |
| séžəm 'sour'                                                                                                | $\vec{X}$ é $\vec{X}$ ə $\vec{i}$ → $\vec{m}$ 'salty' (< $\sqrt{\acute{e}}$ i → $\vec{m}$ 'salt') |
| $i^{\theta} \dot{q}^{w} \partial m, i^{\theta} \dot{a} i^{\theta} \partial \dot{q}^{w} \partial m$ 'rotten' |                                                                                                   |

For more examples of these two forms, see §7.6. and §10.2.1.

A few other adjective-like words are formed with  $-\partial m \sim -\partial m \sim -\partial m$  suffixed to adjectives or verbs, such as:

```
?əỳảm 'strong' (< ?áỳ 'good')
qəlám (JP), qəqəlém (CC) 'weak' (< qál 'bad')
x^w?áỳəm 'clear (water)' (< ?áỳ 'good')
x^wqáləm 'murky (water)' (< qál 'bad')
x^wí·m (//x̃wəm-əm// 'fast (water)' (< x̃wə́m 'fast')
qá?əm 'watery' (as potatoes, cf. sqəqá? 'watery, as of soup, < qá? 'water')
?í?tətəm 'sleepy' (< ?í?tət 'be sleeping')</p>
```

# **10** Non-Personal Affixes 1: Voice

A number of verb suffixes show how the participants in an event relate to it. These include the transitive, intransitive, causative, applicative, reflexive, reciprocal, permissive, and subordinate passive suffixes.

#### **10.1. TRANSITIVE SUFFIXES**

The suffixes -t,  $-n\partial x^w$ , and -x mark forms as simple transitive, that is, they indicate that both a subject and an object are involved in the event, they provide the necessary base for the object person markers and the third-person transitive subject marker, and they make it possible for a verb to have an object as a direct adjunct. The transitivizers -t,  $-n\partial x^w$ , and  $-st\partial x^{w_1}$  are productive, but -x occurs in a only small number of words. Semantically, the differences among them involve the categories of control and perhaps others.

A transitivizer can be suffixed to a verb root (which may, depending on the transitivizer, take a different shape from the one it assumes when appearing bare of affixes) or to a stem composed of a verb root and one or more lexical or grammatical affixes (other than the person markers and a few others that normally follow transitivizers). A transitivizer can appear in any aspect of the verb but the dispositional-iterative and resultative. (See §7 for the forms of roots with these suffixes and §14.2.5 for the object person markers.)

#### 10.1.1. -t 'transitive' (TR)

This is the general or unmarked transitive suffix. It appears as -(V)t except when followed by -S 'first/second-person singular object,' with which it coalesces as  $(V)\theta$ . The presence or type of vowel depends on the form of the root or stem to which the -t is suffixed.

#### 10.1.2. $-n \partial x^w$ 'limited control transitive' (LCTR)

This indicates that the subject is acting unintentionally, with difficulty, or with the possibility of failure.<sup>1</sup> It appears as  $-n \partial x^w$  (or  $-n \dot{e} x^w$  with T $\partial$ T roots) when not followed by a first- or second-person object marker, otherwise as -n-. Allomorphs  $-l \partial x^w \sim -l$ - appear with a few roots ending in  $/1/.^2$ 

Differences in meaning between -t and  $-n\partial x^w$  may be seen in the following sets of roots, -t form and  $-n\partial x^w$  form:

| $\dot{k}^{w}\acute{e}c$ 'see'                 | $k^{w} ec \partial t$ 'look at it'                 | <i>kwácnaxw</i> 'see it'                                |
|-----------------------------------------------|----------------------------------------------------|---------------------------------------------------------|
| k <sup>w</sup> ón 'get taken'                 | k <sup>w</sup> ánət 'take it'                      | <i>k<sup>w</sup>ánnəx<sup>w</sup></i> 'get it'          |
| ${}^{\gamma}ik^{w}$ 'be lost'                 | ${}^{?}ik^{w}\partial t$ 'throw it away'           | ?ák <sup>w</sup> n∂x <sup>w</sup> 'lose it'             |
| <i>tí q</i> <sup>w</sup> 'bump'               | $t i \dot{q}^{w} \partial t$ 'bump it (on          | <i>táď<sup>w</sup>nəx<sup>w</sup></i> 'bump him         |
|                                               | purpose)'                                          | (accidentally)'                                         |
| $i \neq j q^w$ 'break, be cut                 | $i \dot{q}^{w} \dot{a} t$ 'cut it off, cut it      | <i>iáq<sup>w</sup>nəx<sup>w</sup></i> 'accidentally cut |
| off, be cut in two'                           | in two'                                            | it off, manage to cut it off'                           |
| $\dot{c}\dot{a}\dot{q}^{w}$ 'get pierced, get | <i>čq<sup>w</sup>át</i> 'pierce it, poke           | $\dot{c}\dot{a}\dot{q}^wn\partial x^w$ 'manage to shoot |
| shot'                                         | it, shoot it'                                      | it'                                                     |
| $k^w \acute{a} q^w$ 'get hit'                 | <i>k<sup>w</sup>áq<sup>w</sup>ət</i> 'hit it, club | $k^w \delta q^w n \partial x^w$ 'hit it accidental-     |
|                                               | him'                                               | ly, hit it (as a baseball)'                             |

As these sets show, a  $-n\partial x^w$  form may be glossed like the corresponding -t form but with the addition of 'accidentally' or 'manage to' (i.e., 'do after some effort') or because of the semantics of English, it may be glossed with a different word. In each case, the  $-n\partial x^w$  form indicates that the agent is not fully in control of the situation or is playing a less active part in it, 'seeing' rather than 'looking at,' 'getting' rather than 'taking,' 'losing' rather than 'throwing away.' The form  $k^w \delta q^w n \partial x^w$  is 'hit it' when 'it' is a baseball, AG explained, "because there is a chance of missing it."

The allomorph  $-l \partial x^w$  occurs in:

táqallax<sup>w</sup> 'know it, understand it' (root otherwise unrecorded)
cállax<sup>w</sup> 'catch up with him' (cf. cé·lt 'follow him')
tállax<sup>w</sup> (JP), tálnax<sup>w</sup> (CC) 'understand it, learn about it' (cf. tál 'be understood, be settled'; tá²alt 'learn it')

#### **10.1.3.** -*x* 'transitive' (TR)

This seems to be similar to *-t* in meaning and function. It occurs in fewer than twenty words (so far recorded), most of them expressing movement. It carries no implication of limited control. Leslie (1979) noted that his Cowichan

<sup>1</sup> Thompson (1979) discusses control as a major category in Salishan. Arnold Guerin pointed out the "possibility of failure" as a component in the meaning of limited control.

<sup>2</sup> Galloway (1977) identifies the  $-\partial x^w$  of  $-n\partial x^w$  and  $-st\partial x^w$  as a third-person object marker following -n and -st, the counterpart of  $-\emptyset$  following -t and -x, but it seems simpler to treat  $-n\partial x^w$  and  $-st\partial x^w$  as simply allomorphs of -n and -st.

consultants shifted or corrected from  $-\check{s}$  (the Cowichan equivalent of the Musqueam -x) to -t, and he therefore supposed that  $-\check{s}$  is simply an allomorph of -t. At Musqueam, CC and JP did not do this, and I am inclined to give -x separate status. But the evidence is slight. I have only two possible cases of -x contrasting with -t:

(1) CC gave  $l\hat{e}^2 x$  as 'put it away' and  $l\hat{e}^2 t$  as 'make it secure'; however, JP regarded  $l\hat{e}^2 t$  as the Cowichan equivalent of his  $l\hat{i}^2 x$  (and CC's  $l\hat{e}^2 x$ ) 'put it away.'

(2) In Text 22, JP first used  $\lambda e lt$  'apply it' when a ritualist applies powerful words to the sea lion he has carved of cedar, while later he used  $\lambda e lx$  'apply it' when the ritualist applies ochre, oil, and the eyes of a real sea lion. If *-x* has any meaning other than that of *-t*, it may be to indicate physical movement.

This suffix appears in the following:

- $m\hat{e}^2x$  'take it off, remove it' (<  $m\hat{e}^2$  'come off, run (as a dye)'; cf.  $x^w m\hat{e}^2x$  'open it')
- *tél x* (JP), *x*<sup>w</sup>*tél x* (CC) 'follow it (as a trail), notice it' (cf. *tél* (JP), *tél* (CC) 'copy, follow suit')
- *téyqx* 'move it (as something from one room to another)' (cf. *téyqəl* 'move, change residence')
- nówox 'put it in, insert it' (cf. soníw' 'inside')

*télx* 'move it back from the centre of the fire, take it off the stove' (cf. *té·l* 'move shoreward,' *təlíləm* 'move back from the fire, move back into the crowd away from the centre of the big house')

 $\vec{X}p(ix \text{ 'sink it'} (< \vec{X}p(i) \text{ 'go under'}; cf. \vec{X}p(i) \text{ sink it'})$ 

 $k^{w}\acute{e}lx$  'hide him' (cf.  $k^{w}\acute{e}\cdot l //k^{w}\acute{e}l - \mathfrak{l}/l$  'hide')

- *yəléwx* 'pass him' (< *yəléw* 'past'; cf. *yəléwəl* 'get past')
- *wélx* 'throw it' (cf. *wé·ls* //wél-éls// 'throw wealth at a potlatch, have a "scramble"")
- $h\dot{a}^{2}k^{w}\partial x$  'use it' (<  $h\dot{a}^{2}k^{w}$  'be used')
- hiwx 'bring it out into the centre of the house, bring him forward' (cf. hiwal 'move toward the fire')
- x<sup>w</sup>iwx //x<sup>w</sup>hiwx// 'move it to the centre of the fire' (cf. x<sup>w</sup>iwəl 'move upstream,' <sup>?</sup> > hiŵ 'upstream')
- $2i\dot{t}^{\theta}\partial x$  'put it on, wear it' (cf.  $2i\dot{t}^{\theta}\partial m$  'get dressed,' lexical suffix  $-i\dot{t}^{\theta}e^{2}$  'blanket, robe')
- $\theta q \acute{e} n x$  'stand it upright (as a mast)' (<  $s \theta q \acute{e} n$  'be held upright'; cf.  $\theta q \acute{e} t$ 1. 'be standing upright as a spit by a fire,' 2. 'tree')

The progressive of the passive of the last appears in (a).

- (a) yəθəqnəxəm tə swé<sup>9</sup>s tə číkmən. (JP)
  - yə-θəqén-x-əm tə s-wé?-s tə číkmən along-be.held.upright(PROG)-TR-INTR ART NOM-own-3POS ART iron 'The telephone poles are being stood up [set in their holes by the telephone company].'

## 10.2. INTRANSITIVE SUFFIXES

These are  $-\partial m$  'intransitive,'  $-\acute{els}$  'activity,'  $-\acute{\partial n\partial q}$  'someone,' and  $-\acute{\partial t}\partial n$  'someone.'' Verbs with these suffixes are grammatically intransitive, that is, they cannot take the first- and second-person object suffixes or the third-person transitive subject suffix, nor can they have an object as a direct adjunct. However, they are active or agent-oriented, and many with  $-\partial m$  and perhaps all with the other three suffixes are transitive in sense. Some of these with  $-\partial m$  and  $-\acute{els}$ can have objects in oblique phrases (see §3.4).

The suffixes  $-\partial m$  and  $-\dot{e}ls$  generally do not occur with the same root, but there are at least two exceptions. These will illustrate the differences between root, transitive, and these two intransitive forms.

| ROOT                         | TRANSITIVE                         | INTRANSITIVE                   | ACTIVITY                             |
|------------------------------|------------------------------------|--------------------------------|--------------------------------------|
| pán 'get buried'             | <i>pánət</i> 'bury it'             | <i>pánəm</i> 'plant'           | pənéls 'bury something'              |
| $k^{w} \delta n$ 'get taken' | <i>k<sup>w</sup>ánət</i> 'take it' | <i>k<sup>w</sup>ánəm</i> 'get' | <i>k<sup>w</sup>í·ls</i> 'grab hold' |

The suffixes  $-\delta n \partial q$  and  $-\delta l t \partial n$  have each been recorded with only a small number or roots, and neither suffix with roots that take  $-\delta l s$ .

# 10.2.1. ->m 'intransitive' (INTR)

Verbs bearing this suffix are subject-centred, that is, they indicate that the condition exists in the subject, or the action is performed by the subject, or that the action has consequences for the subject. This suffix appears as  $-\delta m$  or  $-\delta m$  with T $\partial$ T roots in the perfective (see §7.2.6), as  $-\delta m$  in the durative (see §7.5) aspect, and otherwise as  $-\delta m$ . It occurs in several sorts of words:

(1) Adjective-like verbs formed from non-active roots (many occurring only with this suffix) that denote conditions or qualities. Several of these evidently usually appear in the progressive aspect and a few (like 'blue') seem to appear only in the progressive. Words of this sort include:

détam, dédatam '(taste) sweet' (cf. sdédatam 'something sweetened')
iétam, iétatam 'sour, bitter' (cf. itiamoat 'become sour,' ialé'iatam 'sorrel, Rumex acetosella L.')
kwékwatam 'taste brackish'
séxam 'taste not sweet, sour'
žétam 'salt, salty, salt salmon,' žéžatam 'taste salty'
iététagam 'be smelling of sweat' (CC)
méxam, mémaxam 'smell like burning rags' (CC)
xwéxwasam 'smell of urine'
séyam 'be sore, sting, taste strong' (cf. sáyamoat 'ache')
títaxam 'slimy' (cf. stíxam 'slime')
tétacam' smell faintly of burning rushes or ferns' (CC)

<sup>3</sup> These last two might better go with the lexical suffixes (§13).

x<sup>w</sup>éx<sup>w</sup>əqəm' 'smell or taste of rotten meat or fish' t<sup>i</sup>θéyəm 'sticky, tacky, get stuck (as a cross-cut saw in a log)' niq<sup>w</sup>əm 'soft, flexible' t<sup>i</sup>θátəm 'get cold (the body),' t<sup>i</sup>θá't<sup>i</sup>θətəm' 'be feeling cold' (cf. st<sup>i</sup>θátəm 'cold [the illness], t<sup>i</sup>θét 'cool off [the day],' t<sup>i</sup>θét<sup>i</sup>θət 'cool, shady, soothing, nourishing') t<sup>i</sup>θét<sup>i</sup>θax<sup>w</sup>əm' 'blue' šx<sup>w</sup>k<sup>w</sup>ítəm 'noisy'

(2) Intransitive verbs formed from non-active roots that name processes or actions that are inherently intransitive. In their relations with forms with other affixes, they fall into several subclasses:

(2.1) Those that have no (known) transitive counterparts with -t 'transitive.' Most of these also seem to have bound roots.

| <i>pá·m</i> 'swell, rise'                          | qáwəm 'howl'                                                      |
|----------------------------------------------------|-------------------------------------------------------------------|
| <i>čísəm</i> 'grow'                                | <i>téyəm</i> 'stick, adhere'                                      |
| qéwəm 'rest'                                       | $q \dot{\lambda} \delta m$ 'bounce off, shake loose'              |
| $h\acute{e}t\acute{\theta}_{\partial m}$ 'breathe' | <i>čánəm</i> 'shake (as a Shaker)'                                |
| hésəm 'sneeze'                                     | <i>čtém</i> 'creep'                                               |
| <i>iáq<sup>w</sup>əm</i> 'cough'                   | xtém 'swim under water (as a fish)'                               |
| $\theta x^{w} \acute{a} m$ 'bleed'                 | <i>méqəm</i> 'jump (as a fish)'                                   |
| $\check{x}\acute{e}m$ 'cry, weep'                  | $p\dot{e}\dot{q}am$ 'bloom' (cf. $\dot{p}\dot{a}\dot{q}$ 'white') |

(2.2) Those with transitive counterparts with -t contrasting with  $-\partial m$  that are causative in meaning. Compare the intransitive (left column) and its transitive counterpart (right column):

| <i>píləm</i> 'overflow'                       | <i>ṕí·lt</i> 'let it overflow'             |
|-----------------------------------------------|--------------------------------------------|
| <i>híləm</i> 'roll, fall off'                 | <i>hí·lt</i> 'roll it, push it off'        |
| $p \vec{k}^{w} \delta m$ 'splash, billow out' | $p\vec{k}^{w}\delta t$ 'scatter it around' |
| <i>k<sup>w</sup>éyəžəm</i> 'move'             | $k^w \delta y \check{x} t$ 'move it'       |
| <i>hánəq<sup>w</sup>əm</i> 'get warm'         | <i>hánq<sup>w</sup>t</i> 'warm it up'      |
| <i>mátəq<sup>w</sup>əm</i> 'bubble up'        | $m \delta t q^w t$ 'dip it in water'       |

(2.3) Those that can be transitivized by suffixing *-t* 'transitive' after the *-om* 'intransitive.' The two suffixes form a transitive the object of which is the goal of the action. (Or, alternatively, we might suppose that the root takes a redirective suffix *-mot*. See §10.4.5.) In some of these there is a shift in stress. Compare the intransitives (left column) with their transitive counterparts (right column):

| <i>ticəm</i> 'swim on the surface' | <i>iícəmət</i> 'swim for it'               |
|------------------------------------|--------------------------------------------|
| náqəm 'dive'                       | <i>nəqámət</i> 'dive for it'               |
| sí x̃ <sup>w</sup> ∂m 'wade'       | $s\check{x}^{w}im\partial t$ 'wade for it' |
| <i>číləm</i> 'climb'               | <i>číləmət</i> 'climb onto it'             |
| <i>cŽám</i> 'jump'                 | <i>cẳámәt 'jump for it, jump him'</i>      |

(2.4) Those formed from non-active roots plus -il 'move toward, become.' Among these are (on the right):

| wáqw   | 'drift, | flow | downstream' |
|--------|---------|------|-------------|
| ?áý 'g | good'   |      |             |

wəq<sup>w</sup>(ləm 'move downstream' 'əŷ(ləm 'become clear (the day)'

For others, see the suffix -il (§12.3.1).

(3) Grammatically intransitive verbs formed from roots that are logically transitive, that is, name actions that have patients. Some of these roots are active, such as  $s\delta w \dot{q}$  'seek,' while others (probably the majority) are non-active, such as  $p\delta n$  'get buried.' In either case, the form with  $-\delta m$  is active, such as  $s\delta w \dot{q} \delta m$  'seek' and  $p\delta n \delta m$  'plant.' We may distinguish two subclasses.

(3.1) Those with  $-\partial m$  suffixed to roots alone, that is, without lexical suffixes. Many or perhaps all of these can have objects in oblique phrases. A few do not seem to have transitive counterparts with *-t*. Three such are:  $t^{\partial}t \cdot m$  'pick berries' (cf.  $st^{\partial}t \cdot m$  'berries'),  $st^{\gamma}x \partial m$  'duck, dodge,' and  $xt d^{\gamma} \partial m$  'prepare' ( $\leq xt d^{\gamma}$  'do').

However, most have transitive counterparts with the transitive suffix -t or -x in place of  $-\partial m$ , although the meaning may differ somewhat. Compare the intransitive forms (left column) with their transitive counterparts (right column):

| $i^{\theta} m$ 'chew'                     | $\dot{t}^{\theta} \dot{e}^{\gamma} t$ 'chew [it]' |
|-------------------------------------------|---------------------------------------------------|
| $\vec{k}^w x \acute{e}m$ 'count'          | $k^w x \epsilon t$ 'count [them]'                 |
| <sup>?</sup> á·m 'call'                   | <sup>?</sup> á·t 'call [him]'                     |
| $2it^{\theta}$ (get dressed)              | $2it^{\theta} ax$ 'put [it] on, wear [it]'        |
| <i>pánəm</i> 'plant'                      | pánət 'bury [it, him]'                            |
| k <sup>w</sup> ánəm 'get'                 | k <sup>w</sup> ánət 'get [it], take [it]'         |
| <i>sáləm</i> 'spin wool'                  | sálət 'spin [it]'                                 |
| <i>q<sup>w</sup>áləm</i> 'bake'           | <i>q̃wálət</i> 'roast [it]'                       |
| <i>táťam</i> 'rake herring or eulachon'   | <i>lťót</i> 'flick [it], hit with a swiping       |
|                                           | motion'                                           |
| <i>tí</i> · <i>m</i> 'pick (as fruit)'    | <i>łímət</i> 'pick [them]'                        |
| $\theta \delta y \partial m$ 'bake bread' | θáyt 'fix [it]'                                   |
| náyəm 'laugh'                             | náyt 'laugh at [him]'                             |
| $y \hat{a} \cdot m$ 'place an order'      | <i>yá∙t</i> 'warn [him]'                          |

(3.2) Those in which  $-\partial m$  is suffixed to a stem composed of a root and a lexical suffix related as verb and object. In these words, if the suffix names a body part, the  $-\partial m$  indicates that the action is undergone by the subject. In contrast, the *-t* in a transitive counterpart indicates that the action is undergone by another. Compare:

| $x^{w}t^{\theta}\check{x}^{w}\check{a}s\partial m$ 'wash one's face'  | $x^{w}t^{\theta}\check{x}^{w}\acute{a}st$ 'wash his face' |
|-----------------------------------------------------------------------|-----------------------------------------------------------|
| $\dot{p}\dot{e}\cdot\partial q^{w}\partial m$ 'straighten one's hair' | $\dot{p}\dot{e}\cdot\partial q^wt$ 'stroke his head/hair' |

(4) Verbs formed from nouns. I have not recorded many of these. The process may not be productive.

patənəm 'hoist sail' (< pástən 'sail'; cf. pá·t 'blow on it,' -tən 'instrument')  $\dot{q}\dot{a}watam$  'drum (v.)' (<  $\dot{q}\dot{a}wat$  'drum (n.)'; cf.  $\dot{q}\dot{a}watt$  'drum for him')

One such group of words requires the prefix c- 'get, do,' as in:

*cténəm* 'call (someone) "mother" (< *tén* 'mother') *cménəm* 'call (someone) "father" (< *mén* 'father') *csìlé*:m 'call (someone) "grandparent" (< *síli* 'grandparent')

(5) Co-ordinate passive forms (see §14.2.6)

(6) A few nouns that may simply be nominalized verbs the roots of which have not been recorded, such as  $s\dot{t}^{\theta}c_{\sigma}m$  'nut, hazelnut,'  $s\dot{a}\dot{k}^{w}\sigma m$  'whole bark,'  $s\theta\dot{e}y\sigma m$  'guest,'  $s\dot{t}\dot{e}x\sigma m$  'lower-class person or group of slave descent.'

# 10.2.2. -éls 'activity' (ACT)

This suffix appears as  $-\acute{els}$  in all perfective forms except for a very few (as yet unpredictable) perfectives in which it is  $-\imath ls$ . In progressives it appears as  $-\imath ls$  or, suffixed to stems with lexical suffixes, as  $-i \cdot ls$ . (See §7 for the forms of stems appearing with  $-\acute{els}$ .)

Many verbs with -*éls* are glossed as transitive verbs without objects or with 'something' object. Compare the following activity forms (left column) with their counterparts with the transitivizers -*t* or -*x*:

| $p \partial \theta \delta ls$ 'spread something' | $p \dot{e} \theta \partial t$ 'spread it'                  |
|--------------------------------------------------|------------------------------------------------------------|
| pánəls 'bury something'                          | <i>pánət</i> 'bury it'                                     |
| <i>təméls</i> 'chop'                             | <i>ťámət</i> 'chop it'                                     |
| nawéls 'insert something'                        | náwax 'insert it'                                          |
| <i>ť<sup>e</sup>ďvéls</i> 'hit'                  | $\dot{t}^{\theta} \dot{t} \dot{q}^{*} \partial t$ 'hit it' |
| <i>Åk<sup>w</sup>éls</i> 'pinch'                 | $\dot{X}k^{w} \dot{a}t$ 'pick it up with the fingertips'   |

But often the *-éls* form is given a restricted meaning referring to some habitual or ritual activity, as in:

| <i>pk<sup>w</sup>éls</i> 'smoke fish'                        | <i>pék<sup>w</sup>ət</i> 'warm it up'                |
|--------------------------------------------------------------|------------------------------------------------------|
| $p\dot{\vec{x}}\acute{e}ls$ 'harpoon sturgeon'               | $\vec{p} \in \vec{X} lt$ 'feel for it'               |
| <i>Žəyqéls '</i> knead bread'                                | $\vec{x} \neq y \vec{q} t$ 'press down on it'        |
| $\theta \partial y q^{w} \ell ls$ 'dig a grave'              | $\theta \acute{e} y q^{w} \partial t$ 'dig it'       |
| <i>qpéls</i> 'collect money'                                 | $\dot{q}p \delta t$ 'put them together'              |
| <i>k<sup>w</sup>ət x<sup>w</sup>éls</i> 'bring something in' | k <sup>w</sup> téx <sup>w</sup> t 'bring/take it in' |
| <i><sup>2</sup> axí·ls</i> 'tow something'                   | <sup>2</sup> áxəl 'paddle'                           |

The term  $p \vec{\lambda} \acute{e} ls$  is literally 'feel for something,' as a harpooner must do with his vertically held harpoon, and  $k^w \partial t x^w \acute{e} ls$  refers especially to bringing a picture of

a deceased person to display it in the big house. The long vowel of  $2 a x i \cdot l s$  is the result of the loss the first of a pair of resonants.

The  $-\acute{els}$  form may also have a subject-centred or intransitive meaning comparable to that of  $-\partial m$  with the sense of (3.2) above. Compare these  $-\acute{els}$  forms and the transitives with -t:

| <i>łəžéls</i> 'help yourself (to food)'            | $i e^{2} \check{x}t$ 'put it in a dish'                         |
|----------------------------------------------------|-----------------------------------------------------------------|
| <i>pəhéls</i> ~ <i>péls</i> 'blow (as the wind)'   | <i>pá</i> · <i>t</i> 'blow on it, blow it out'                  |
| <i>pəwels</i> 'freeze (INTR)'                      | <i>píŵət</i> 'freeze it'                                        |
| $p\check{x}^{w}\acute{e}ls$ 'blow (as a whale or a | $p \acute{a} \check{x}^w \partial t$ 'blow on it (as a shaman), |
| shaman through his hands)'                         | expelling chewed medicine on it'                                |

A number of words for recently introduced tools and utensils consist of progressive *-éls* forms, usually but not always with the compound prefix  $\delta x^{w}$ -'means of' (§12.1.4), as in:

 $\delta x^w lil_{\dot{d}c} ds$  'saw' (cf.  $lic_{\partial t}$  'cut it,'  $lil_{\partial c} dt$  'be cutting it')  $\delta x^w c \dot{a} d^w ds$  'fork (table or pitch)' (cf.  $c \dot{d}^w \delta t$  'pierce it,'  $c \dot{a} d^w t$  'be piercing it')  $\delta x^w p \dot{e} p \partial k^w n \partial c \dot{l} s$  'tea kettle' (cf.  $p \delta k^w n \partial c t$  'put a pot on the fire')  $\dot{p} \dot{t} \dot{p} \partial t^\theta \partial s$  'wringer on washing machine' (cf.  $\dot{p} \dot{t}^\theta \partial t$  'squeeze it, wring it out')

#### 10.2.3. - *ánaq* 'someone else, another person'

Verbs formed with this and the next suffix, like verbs formed with  $-\partial m$  and  $-\delta ls$ , are intransitive and active. However, while verbs with  $-\partial m$  may have a reflexive sense and those with  $-\delta ls$  have implied objects that are not human, the implied objects of verbs with  $-\delta n\partial q$  and  $-\delta lt\partial n$  are other persons. Some verbs with  $-\delta n\partial q$  are:

- səwdánəq 'look for a person, engage in a search' (cf. sáwd 'seek,' sáwdəm 'seek,' sáwdt 'look for him')
- *łəwśnəq* 'perform a cure' (cf. *łéwət* 'treat him, heal him')
- $\dot{X}x^{w} \dot{\partial}n \partial q$  'win, win a contest' (cf.  $\dot{X}x^{w} \dot{\partial}t$  'beat him, as in a game')
- *cəłtənánəq* 'rent out (as a landlord), make a loan' (cf. *cáłtən* 'borrow, rent (as a tenant),' *cáłtənt* 'lend it')
- *nəwèyəłánəq* 'advise someone else's child' (cf. *níŵət* 'advise him,' *nəŵéyəł* 'advise a child,' *sníŵ* 'advice')
- $\dot{\lambda} \delta n \partial q$  'potlatch (v.),'  $s \dot{\lambda} \delta n \partial q$  'potlatch (n.)' (cf.  $\dot{\lambda} e^{2t}$  'go get him, pick them up, invite him/them';  $\dot{\lambda} ex \partial n$  'invite,' < -x $\partial n$  'foot,'  $s \dot{\lambda} ex \partial n$  'feast, local gathering.' The term  $s \dot{\lambda} \delta n \partial q$  may literally mean 'a gathering of people invited from elsewhere.')

This suffix can follow  $-\delta s$  'recipient' and -lc- 'benefactive' (see §10.4.3 and 10.4.4 below). An example with  $-\delta s$  is  $2i\dot{w}\partial s\dot{\partial}n\partial q$  'give directions' ( $< 2i\dot{w}\partial s$  'give directions,' composed of  $\sqrt{2i\dot{w}}$ - 'instruct' +  $-\delta s$  'recipient'; cf.  $2i\dot{w}\partial s t$  'show him, guide him'). For examples with -lc-, see §10.4.4.

# 10.2.4. $-\partial t t \partial n \sim -\partial t t i n$ 'someone, another'

This is probably related to the particle  $2\dot{e} dt \partial n$  'third-person plural.' It seems similar or identical in meaning to  $-\delta n \partial q$  'another.' It has been recorded in fewer words.

*céwəltən* 'help, give help' (cf. *céwət* 'help him') *déwəltən* 'pay, make payment' (cf. *déwət* 'pay him') *xíwəltən* 'restrain (as someone else's child from causing a disturbance)' (cf. *xíwət* 'restrain him, stop him')

háyq<sup>w</sup>əłtən 'recruit' (cf. háyq<sup>w</sup>ət 'recruit him')

# 10.3. THE CAUSATIVE SUFFIX $-st \partial x^{w_{I}}$

The suffix  $-st \partial x_{\mu_1}^w$  (causative' (CAUS) indicates that the subject makes the object do, causes the object to be, or puts the object in whatever act, quality, or state that is named by the root or stem. This suffix also serves as or is homophonous with  $-st \partial x_{\mu_2}^w$  (comitative' (see §10.4.1 below). Historically it may be composed of -s 'permissive' (see §10.7 below) plus -t 'transitive.'

This suffix has several allomorphs. When final in a word, it appears as  $-st \ge x^w$ , except when preceded by a T $\ge$ T root, in which case it bears the stress and appears as  $-st \le x^w$ , or when preceded by a suffix ending in a resonant, in which case it appears as  $-st \ge x^w$ . When followed by -ss 'third-person (transitive or subordinate) subject,' the unstressed  $-st \ge x^w$  may be reduced to  $-st x^w$ . When followed by one of the object person markers, it appears as  $-st = x^w$ , which, unlike -t 'transitive,' does not coalesce with the first- and second-person singular object markers.

The suffix  $-st \partial x^{w_i}$  has two senses: (1) 'cause to be' and (2) 'cause to do.' It has the first sense when suffixed to non-active verbs (see §10.3.1 for "active" versus "non-active"), resultative forms of verbs, verbs bearing intransitive suffixes, reflexive and reciprocal suffixes, adjectives, and verbs formed from nouns. It has the second sense when suffixed to active verbs.

## 10.3.1. In the sense 'cause to be'

With many non-active verbs, the transitive suffixes are causative in sense. Compare the following roots (left column) and transitives formed with -t,  $-n\partial x^w$ , and -x:

| $\dot{c}\dot{e}^{2}$ 'land atop'              | $\dot{c}\dot{e}^{\gamma}t$ 'put it on top'            |
|-----------------------------------------------|-------------------------------------------------------|
| $q^{w} \delta s$ 'go into the water'          | $q^{w}s \delta t$ 'put it into the water'             |
| <i>žáł</i> 'hurt'                             | <i>xtát</i> 'hurt him'                                |
| <i>q́áy</i> 'die'                             | $\dot{q}\dot{a}yn\partial x^{w}$ 'manage to kill him' |
| $m e^{2}$ 'come off'                          | $m e^{2} x$ 'take it off, remove it'                  |
| <i>píləm</i> 'overflow'                       | $\dot{p}i \cdot lt$ 'let it overflow'                 |
| $p \vec{k}^{w} \delta m$ 'splash, billow out' | $p\vec{k}^{w}\delta t$ 'scatter it about'             |
| <i>k<sup>w</sup>éyəžəm</i> 'move'             | $k^{w} \delta y x t$ 'move it'                        |
|                                               |                                                       |

These transitives might also be glossed 'cause it to land atop,' 'cause it to go into the water,' 'cause him to hurt,' 'manage to cause him to die,' and so on.

For many non-active verbs, however,  $-stax_{i}^{w}$  has the same effect. Compare the following roots and their causative forms:

| $ni^{9}$ 'be there'                                   | $ni^{9}st \partial x^{w}$ 'have it there'                                              |
|-------------------------------------------------------|----------------------------------------------------------------------------------------|
| <i>ťák</i> <sup>w</sup> 'go home'                     | $i \delta k^w st \partial x^w$ 'take/bring him home'                                   |
| $\dot{t}\dot{a}x^{w}$ 'go/come shoreward'             | $i \delta \vec{k}^w st \partial x^w$ 'take/bring him to the shore'                     |
| cám 'go/come inland'                                  | cómstox <sup>w</sup> 'take/bring him inland'                                           |
| háye? 'go away'                                       | <i>həyé<sup>9</sup>stəx</i> <sup>w</sup> 'take him away'                               |
| <i>yəwé</i> <sup>2</sup> 'go along'                   | yəweeestəx 'take him along'                                                            |
| <sup>?</sup> əmí 'come'                               | <i>?əmístəx</i> <sup>w</sup> 'bring it here'                                           |
| ?íməx 'walk'                                          | ?íməxstəx <sup>w</sup> 'make him walk'                                                 |
| <i>sí</i> <sup>.</sup> <i>?si</i> <sup>?</sup> 'fear' | <i>sí</i> <sup>.</sup> <i>?si</i> <sup>?</sup> <i>stax</i> <sup>w</sup> 'frighten him' |

These might also be glossed 'cause it to be there,' 'cause him to be home,' 'cause him to be down on the shore,' and so on.

So far I have discovered only one instance of a non-active root or stem taking both  $-st \partial x^{w_{i}}$  and one of the transitivizers, apparently with the same meaning:  $\vec{\lambda} p (lx)$  'sink it' and  $\vec{\lambda} p (l \partial s t \partial x^{w})$  'sink it.' These are based on  $\vec{\lambda} p (l)$  'go under' ( $< \vec{\lambda} \delta p$  'deep,' -il 'move toward') and are glossed identically. Something close to a contrasting pair is seen in l e l x 'move it back from the centre of the fire' and  $l e l \partial s t \partial x^{w}$  'move it toward the shore.' But the -x form is based on the root l e l, seen also in  $l \partial l (l \partial m)$  'move back from the fire (INTR),' while the  $-st \partial x^{w}$  form is based on a derivative, probably the root l e l with the suffix  $-\partial l \sim$ -l l 'move toward.'

Very commonly,  $-stax_{i}$  follows resultatives (see §7.7). Compare the following perfectives (left), resultatives (centre), and transitivized resultatives (right):

| há y 'finish'                                        | shá·ý 'finished'                                     | <i>shá</i> ·ý <i>stəx<sup>w</sup></i> 'have it finished'                         |
|------------------------------------------------------|------------------------------------------------------|----------------------------------------------------------------------------------|
| <i>θáyt</i> 'fix it'                                 | sθəθə́y' 'right'                                     | $s\theta \partial \theta \partial \dot{y} st \partial x^w$ 'keep it on course'   |
| $q^{w} \acute{e} t$ 'perforate it'                   | <i>sq<sup>w</sup>əq<sup>w</sup>é</i> 'perforated'    | $sq^w \partial q^w \acute{e}st \partial x^w$ 'leave a hole in it'                |
| <i>łólqt</i> 'dip it'                                | <i>stélq</i> 'in the water'                          | <i>stélqstəx</i> <sup>w</sup> 'keep it in the water'                             |
| wíl 'appear'                                         | swíwał 'visible'                                     | <i>swíwəlstəx<sup>w</sup></i> 'keep it in sight'                                 |
| $\check{x}\acute{e}\check{t}^{\theta}t$ 'measure it' | $s\check{x}e^{\gamma}\acute{e}t^{\theta}$ 'measured, | $s\check{x}\partial^{2}\acute{e}t^{\theta}st\partial x^{w}$ 'blaze (as a trail), |
|                                                      | marked,                                              | designate (as a time), keep                                                      |
|                                                      | distinguished'                                       | track of it'                                                                     |

The suffix  $-st \partial x^{w_1}$  'causative' can follow the intransitive suffix  $-\partial m$ , with a shift in stress to the  $-\partial m$ :

 $^{?}\partial t^{\theta} \hat{e}m \partial st \partial x^{w}$  'dress him' (>  $^{?}tt^{\theta} \partial m$  'get dressed, dress oneself')  $\dot{c}\partial t^{2}\partial c \dot{d}m \partial st \partial x^{w}$  'take it over to the other side' (<  $\dot{c}\dot{a}\cdot\dot{l}\partial c$  'cross over')  $n\partial^{2}\dot{e}m \partial st \partial x^{w}$  'take it there, recite it' (<  $n\dot{e}m$  //ni?- $\partial m$ // 'go')

It can follow -*ánaq* 'someone else' and -*áłtan* 'other':

 $n\partial w \partial y\partial t \partial n \partial q st \partial x^w$  'get him to give advice to someone else's child' (<  $\sqrt{niw}$  'advise,' - $ey\partial t$  'child')

kwənəwsi in aqstəxw 'get him to work contagious magic on others' (< kwənəwsi in aq 'work contagious magic on others,' < kwi in aws 'work contagious magic,' composed of kwi in 'get taken,' -əws 'body') gewəltinəstəxw 'punish him,' lit. 'make him pay'

It can follow the reflexive  $-\theta \partial t$  (§10.5 below):

(a) si<sup>2</sup>ámθətstám tə <sup>2</sup>al. (AG)
 si<sup>2</sup>ém θət-st-ám tə <sup>2</sup>al
 rich.person-self-CAUS-me PER just
 'I hope that I get wealthy.'

It can follow  $-t\partial \hat{l}$  'reciprocal' (§10.6 below). Compare the reciprocal forms (left column) and their causatives (right column):

| $\theta \delta l \partial q t \delta l$ 'separate from one | $\theta \partial l \partial qt \dot{a} l \partial st \partial x^w$ 'separate them, divide |
|------------------------------------------------------------|-------------------------------------------------------------------------------------------|
| another'                                                   | it'                                                                                       |
| sálsaldtal 'even with one another'                         | səlsəlqtálstəx <sup>w</sup> 'divide it equally'                                           |
| $k^{w}i^{2}a^{2}t\partial l$ 'keep away from one           | <i>k<sup>w</sup>itáləstəx<sup>w</sup></i> 'move them apart'                               |
| another'                                                   |                                                                                           |

In the next two examples,  $-st a x^{w_1}$  follows an adverb and a personal word:

 $q \partial \dot{\ell} \dot{\epsilon} t s t \partial x^w$  'do it to him again' (<  $q \partial \dot{\ell} \dot{\epsilon} t$  'again') (AG)  $\dot{\chi} \dot{a} s t \partial x^w$  'make it so, go ahead with it' (<  $\dot{\chi} a$  'be third person') (AC)

Adjectives (left below) can take  $-st \partial x^{w_1}$  to form causatives (right):

| <i>k<sup>w</sup>ámk<sup>w</sup>ðmstðx<sup>w</sup></i> 'strengthen it' |
|-----------------------------------------------------------------------|
| $\check{x}^w \acute{\partial} mst \partial x^w$ 'speed it up'         |
| cícəłstəx <sup>w</sup> 'pile them high'                               |
| $\check{x}\acute{e}^{?}\check{x}est a x^{w}$ 'believe it sacred'      |
|                                                                       |

Verbs formed from nouns can take  $-st \partial x^{w_1}$  to form causatives. Compare the noun (left), noun stem with verbalizing prefix (centre), and causative (right):

| <i>swá ýqe</i> ? 'man' | <i>x<sup>w</sup>əswə́yqe</i> ? 'become a man' | <i>x<sup>w</sup>əswə́yqe<sup>2</sup>stəx<sup>w</sup></i> 'make |
|------------------------|-----------------------------------------------|----------------------------------------------------------------|
|                        |                                               | a man of him'                                                  |

#### 10.3.2. In the sense 'cause to do'

Finally,  $-st \partial x^{w_1}$  can be suffixed to active (but intransitive) verbs to produce causative forms with the sense 'cause to do.' These verbs include some active roots, like  $k^w \acute{e}c$  'see,' and some inactive roots with  $-\partial m$  'intransitive' and  $-\acute{e}ls$  'activity.' With such verbs there is a clear contrast between forms with the transitivizers -t, -x, and  $-n\partial x^w$  and those with  $-st\partial x^w$ . Compare the following active but intransitive, transitive, and causative forms:

| <i>k<sup>w</sup>éc</i> 'see'                                                                      | $k^{w} ec \partial t$ 'look at it'                                                                                                                        | $k^{w} \delta cst \partial x^{w}$ 'show it to him, make him see it'                                                                                                                                                                 |
|---------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| <sup>?</sup> í <i>ť</i> ∂m 'dress'<br><i>ť<sup>®</sup>x<sup>w</sup>éls</i> 'wash'<br>√ <i>pit</i> | $k^{w\delta}cn\partial x^{w}$ 'see it'<br>? $t^{t}\partial x$ 'wear it'<br>$t^{\theta}x^{w} at$ 'wash it'<br>$p\delta tn\partial x^{w}$ 'recognize<br>it' | ${}^{?} \partial t^{\theta} \epsilon m \partial s t \partial x^{w}$ 'dress him, put it on him'<br>$t^{\theta} x^{w} \epsilon l s s t \partial x^{w}$ 'have him wash it'<br>$p \partial t s t \partial x^{w}$ 'make him identify it' |
| <i>qá²qa</i> ² 'drink'<br>√ <i>nəp</i> ' 'eat'<br><i>²áłtən</i> 'eat'                             | <i>qá<sup>2</sup>qət</i> 'drink it'<br><i>ná ṗəx</i> 'eat it'<br>—                                                                                        | <i>qá</i> <sup>2</sup> <i>qəstəx</i> <sup>w</sup> 'give it to him to drink'<br><i>nəṗ́óməstəx</i> <sup>w</sup> 'make him eat it'<br><i>?ółtənstəx</i> <sup>w</sup> 'feed him'                                                       |

In these cases, the forms with *-t*, *-x*, and *-n* $\partial x^w$  are purely transitive with no causative sense, while the forms with *-st* $\partial x^w_1$  are transitive and causative in that their objects are caused to do whatever is expressed by the verb. Or to put it differently, their objects are agents in implied (embedded) clauses. A couple of sentences will illustrate these syntactic relations:

(a) ném k<sup>w</sup> ścstəx<sup>w</sup> tə John ?ə k<sup>w</sup>θə xéws ?ənpút. (AG) ném k<sup>w</sup>ec-stəx<sup>w</sup> John ?a k™θə **žé**ws tə <sup>9</sup>ən-pút go see-CAUS ART John OBL ART new your-boat 'Go show John your new boat.'

In (a), 'John' is the grammatical object of the causativized verb and 'your boat' is in an oblique phrase. But in the implied (embedded) clause, 'John' is subject and 'your boat' is object.

```
(b) t^θx^wélsstəx^w čx^w k^wθəủ ^γíməθ ^γə tə šx^wí^γltən. (AG)
t^θəx^w-éls-stəx^w čx^w k^wθə ^γən ^γíməθ ^γə tə
get.washed-ACT-CAUS you ART your grandchild OBL ART
šx^w^γl^γltən
dishes
'Have your grandson wash the dishes.'
```

In (b), 'your grandson' is the grammatical object of the causativized verb and 'the dishes' is in an oblique phrase, while in the implied clause 'your grand-child' is subject and 'the dishes' is object.<sup>4</sup>

#### 10.4. APPLICATIVES

The applicative or "redirective" suffixes "redirect" the verb so that what would otherwise have to be in an oblique relationship to it can be its grammatical object. They function as the English prefix *be*- might function if used consistently, as in the second of the following two sentences:

<sup>4</sup> I am indebted to Donna Gerdts for my understanding of this, among many features of the language.

- (1) I sprinkled water (object) on him (oblique).
- (2) I besprinkled him (object) with water (oblique).

Or the suffix -t (of "gift" from "give") in the second of the following two:

(1) I gave a book (object) to him (oblique).

(2) I gifted him (object) with a book (oblique).

And as in these examples, the Halkomelem applicatives generally make it possible for a person rather than a thing to occupy the position of object.<sup>5</sup>

There are five applicatives:  $-st \partial x^{w_2}$  'comitative' (COM),  $-n \partial s$  'goal' (GOAL),  $-m \partial t$  'concern' (CON),  $-\partial s$  'recipient' (RECIP), and -tc- 'benefactive' (BEN). Each of these has a range of meanings, expressed by various English prepositions, and the ranges overlap, so that it is difficult to label them in any very useful way. I have tried to give labels according to what seems to be most distinctive about each, but these labels should be interpreted loosely. The label 'comitative,' for example, is a meant to reflect the need for an English 'with' in glossing a number of forms with  $-st \partial x^{w_2}$ , but some forms with this suffix require 'to' and some 'for.' Where  $-st \partial x^{w_2}$  requires 'for,' it overlaps in meaning with -tc-, which, following established practice, I have labelled 'benefactive.' But, as Galloway (1977, 251) says of the Upriver -tc-, it can also be "malefactive." What is listed here as  $-m\partial t$  presents particular difficulties that require further work. All but  $-\partial s$  'recipient' and -tc- 'benefactive' are also transitivizers. The suffix  $-\partial s$  can be followed by a transitivizer but need not be, while -tc- must be followed by a transitivizer.

The grammatical function of the applicatives seems similar to that performed by many of the lexical suffixes (see §13.4.1) in verbs. For example, the lexical suffix  $-aq^w$  'head' in the verb  $mé \partial q^w t$  'decapitate him' ( $< mé^2$  'be removed'; cf.  $me^2x$  'remove it') provides an object incorporated within the verb and so redirects the verb so that the owner of the head can be its grammatical object. Indeed, the redirective  $-\partial s$  'recipient' may be identical with or a development of the lexical suffix  $-\partial s \sim -\dot{a}s$  'face.'

# 10.4.1. -stax<sup>w</sup><sub>2</sub> 'comitative' (COM)

This is identical in its forms with  $-st \partial x^{w_l}$  'causative,' with which it may be historically identical. It expresses relationships between verb stems and grammatical objects of the sorts commonly expressed in English by *with*, *for*, *to*, and perhaps *about*.

The status of  $-st \partial x_{2}^{w}$  as something other than a causative can be seen in the following examples. First, compare roots (left), *-t* forms (centre), and  $-st \partial x_{2}^{w}$  forms (right) with glosses constructed to show systemic relations (in square brackets):

<sup>5</sup> See Gerdts 1988 on the syntactic functions of applicatives.

| <i>há y</i> 'finish' | <i>háyt</i> 'finish it' | <i>háystəx</i> <sup>w</sup> 'get rid of it' |
|----------------------|-------------------------|---------------------------------------------|
| ['be finished']      | ['make it finished']    | ['be finished with it']                     |
| <i>žół</i> 'hurt'    | <i>žłát</i> 'hurt him'  | <i>žáłstəx</i> <sup>w</sup> 'pity him'      |
| ['feel pain']        | ['make him feel pain']  | ['feel pain with/for him']                  |

Because these roots are non-active, the forms with *-t* are causative in sense, and the forms with *-stax<sup>w</sup>*, express a different kind of relationship.

Second, consider the two meanings of an active verb based on the root of  $\dot{q}\dot{e}wat$  'pay him (for labour)' and their forms with  $-stax_{1}^{w}$  'causative' and  $-stax_{2}^{w}$  'comitative':

| <i>qéwəłtən</i> 'pay penance for sins' | <i>qewəltinəstəx<sup>w</sup></i> , 'punish him' ['make  |
|----------------------------------------|---------------------------------------------------------|
|                                        | him pay']                                               |
| <i>qéwəłtən</i> 'pay people'           | <i>qewəltinəstəx</i> <sup>w</sup> 2 'pay his way' ['pay |
|                                        | for him']                                               |

Other forms with  $-st \partial x_2^w$  illustrate the variety of English prepositions required in translation:

| łéw 'run away'                          | <i>téwstəx</i> <sup>w</sup> 'run away with him/her'               |
|-----------------------------------------|-------------------------------------------------------------------|
| <i>xtém</i> 'swim'                      | xtéməstəx <sup>w</sup> 'swim away with him'                       |
| $q^{w} \acute{e}l$ 'speak'              | $q^{w} \delta l s t \delta x^{w}$ 'speak to him'                  |
| $\theta \delta t$ 'say'                 | $\theta \delta t s t \partial x^w$ 'tell him' ['say to him']      |
| $x^{w}\partial^{2}i\cdot nt$ 'say what' | $x^{w}\partial^{2}i\cdot ntst\partial x^{w}$ 'say what to him'    |
| <i>žté</i> <sup>2</sup> 'do'            | $\check{x}t\acute{e}^{2}st \partial x^{w}$ 'do to him'            |
| <i>qəlét</i> 'do again'                 | $q \partial \hat{l} dt st \partial x^w$ 'do to him again'         |
| $\sqrt{c}$ of what                      | $c \delta s t \partial x^w$ 'do what with him'                    |
| <i>swé</i> <sup>2</sup> 'one's own'     | $swé^{2}st \partial x^{w}$ 'keep it for him'                      |
| ?áý 'good'                              | <i>?áýstəx</i> <sup>w</sup> 'like it' ['feel good about it'?]     |
| <i>qál</i> 'bad'                        | $q \delta l s t \partial x^w$ 'dislike it' ['feel bad about it'?] |

These last two might also be interpreted as formed with  $-st \partial x_{i}^{w}$  'causative,' that is, 'make it good' and 'make it bad.' Such ambiguities suggest that  $-st \partial x_{2}^{w}$  'comitative' may have developed out of  $-st \partial x_{i}^{w}$  'causative.'

#### 10.4.2. -nəs 'goal' (GOAL)

This appears in a few words where the object is a goal that might otherwise be expressed by an oblique adjunct. Like -x 'movement transitive,' it carries no implication of limited control. It appears as -ns after a stressed vowel and otherwise as -ns, except for one word so far recorded in which it appears as -ls.

The following are all of the words so far recorded with  $-n \partial s$  (the intransitive bases are on the left and forms transitivized with  $-n \partial s$  are on the right):

| <sup>?</sup> əmí 'come' | <i>?əmíns</i> 'come after him, come for him' |
|-------------------------|----------------------------------------------|
| ném 'go'                | <i>nə<sup>9</sup>émnəs</i> 'go after him'    |

| $x^{w}\partial^{2}i$ 'reach here' | $x^{w}\partial^{2}ins$ 'reach it here'              |
|-----------------------------------|-----------------------------------------------------|
| *x <sup>w</sup> əní?              | $x^{w} \partial n i ns$ 'reach it there'            |
| ?ámət 'sit'                       | <sup>9</sup> ómətnəs 'sit for her (as a suitor)'    |
| hék <sup>w</sup> 'remember'       | hák <sup>w</sup> nas 'remember it'                  |
| <i>mélq</i> 'forget'              | <i>málqnəs</i> 'forget it, have it slip one's mind' |

(The stem  $x^w \partial ni^2$  has not been recorded, but it must be composed of  $x^w \partial$ -'become' and  $ni^2$  'be there' and the counterpart of  $x^w \partial^2 i$ .)

The allomorph *-l* $\partial s$  appears in  $2eyal\partial s$  'leave him.' The root  $\sqrt{2}ey$  is implied by 2eyal 'get out of the way' (with  $-\partial l$  'move toward') and by  $s^2\partial ynec$  'out of the road' (with *-n* $\partial c$  'base, butt'). See §14.2.5 for a paradigm with the object person markers.

# 10.4.3. -*əs* 'recipient' (RECIP)

This indicates the presence of a recipient. As suggested above, it may be historically identical with  $-\partial s \sim -\dot{a}s$  'face, round object, forward end'; it has the same umlauting effect of converting an /e/ in the stem into an /a/. It is not a transitivizer and is therefore probably not relatable to  $-n\partial s$  'directional,' which is. It can occur finally or followed by a -t 'transitivizer,' which allows the recipient to be the grammatical object.

This suffix is not positively identifiable in very many words. Examples of which we can be certain include two with final  $-\partial s$ :

```
²i\dot{w}as 'show, guide' (< \sqrt{2}i\dot{w} 'instruct')
syá\thetaas 'tradition' (< yá\theta 'tell')
```

Examples with -t transitivizer include the following:

```
²i\vec{w} st 'show him how' (<\sqrt{2}\vec{t}\vec{w} 'instruct')

y\delta\theta st 'tell him' (< y\delta\theta 'tell')

²\delta m st 'give to him' (\sqrt{2}em 'give')

²\delta \vec{x}^{w} st 'give to him' (<\sqrt{2}e\vec{x}^{w} 'give')
```

The root  ${}^{?iw}$  'instruct' is implied by  $x^{w?\delta}wc\partial s$  'instruct manually' (with *-cos* 'hand'),  ${}^{?em}$  'give' by  ${}^{?e?}\partial m$  'give,'  ${}^{?e'm}\partial m$  'be giving,'  ${}^{?d}mn\partial ct$  'put money down on it' (with *-noc* 'base'), and  ${}^{?ex}w$  by  $x^{w?\delta}xwm\partial t$  'covet it.'

The difficulties we encounter in trying to identify this suffix may be illustrated by the following two pairs of words:

```
\vec{k}^{wl} \vec{e} t 'spill it' \vec{k}^{wl} \vec{a} s t 'pour water on it'
k^{w} \vec{e}^{2} t 'let it go' k^{w} \vec{a}^{2} a s t 'turn it loose (as a horse), cast him off (as a lover)'
```

In the first of these, it seems reasonable to identify the suffix as 'face' and interpret the word as 'pour on the face/front of it.' In the second, however, it would be stretching the meaning of 'face' a bit far, and it seems more reasonable to identify the suffix as simply indicating a recipient.

#### 10.4.4. -*łc*- 'benefactive' (BEN)

Like *-t* 'transitive,' this may appear with a vowel, as -(V)tc-, the presence or type of vowel depending on the preceding root or stem. In function -tc- relates an active verb and its patient to a "beneficiary," someone who is affected (not necessarily beneficially) by or has an interest in the act. With a benefactive form, a patient must appear in an oblique phrase. If *-t* 'transitive' follows the *-tc*-, the beneficiary is the grammatical object. Compare (a) and (b):

(a) ni cən yáxwət tə xwíləm.
 ni cən yáxwət tə xwíləm
 AUX I free-TR ART rope
 'I untied the rope.'

(b) ni cən yáx vəlcət tə Tom ? ə tə x víləm. <u></u>¥<sup>w</sup>íĺ2m ni? cən váž<sup>w</sup>-əłc-ət tə Tom ?ə tə I free-BEN-TR AUX ART Tom rope OBL ART 'I untied the rope for Tom.'

The suffix -ic- may be followed by -t 'transitive,'  $-\partial m$  'intransitive,' and  $-\partial n\partial q$  'another.' It has not been recorded in word final position or with a transitivizer other than -t. In the following, the transitivizer -x is replaced by -t following -ic-:  $n\partial \dot{p}\partial x$  'eat it,'  $n\partial \dot{p}\partial c\partial t$  'eat it for him.'

It appears that any active verb may have a benefactive counterpart. Some benefactives, however, are interpreted idiomatically. Compare the following:

| <i>k<sup>w</sup>áq<sup>w</sup>ət</i> 'strike it' | <i>k<sup>w</sup>áq<sup>w</sup>əłcət</i> 'strike it for him'                             |
|--------------------------------------------------|-----------------------------------------------------------------------------------------|
| <i>žté<sup>9</sup>əm</i> 'make'                  | <i>xté<sup>2</sup>əłcət</i> 'make it for him'                                           |
| <i>kvécət</i> 'look at it'                       | <i>kwécəłcət</i> 'read it to him'                                                       |
| <i>xálət</i> 'paint it, write it'                | <i>žáləłcət</i> 'write for him/to him'                                                  |
| <i>k<sup>w</sup>ənét</i> 'hold it'               | <i>k<sup>w</sup>ənéłcət</i> 'owe him' (lit. 'hold it for him')                          |
| <i>kʷəỷét</i> 'stop him'                         | <i>k</i> <sup>w</sup> <i>ətéłcət</i> 'refuse something to him, deny him something'      |
| $k^{w}\partial^{2}\acute{e}t$ 'leave it alone'   | $k^{w}\partial^{2}\acute{e}tc\partial t$ 'leave it be for him, respect his right to it' |

A benefactive can be the counterpart of either a transitive or an intransitive active form with  $-\partial m$  'intransitive' or  $-\ell ls$  'activity,' thus:  $k^w t \ell t$  'spill it,'  $k^w t \ell ls$  'pour,'  $k^w t \delta t c \partial t$  'pour it for him, spill it for him.'

Context may show the difference. Compare (c) and (d):

(c) ni cən kwłáłcət <sup>γ</sup>ə kwθə tí.

| ni?    | cən     | k <sup>™</sup> əł-łc-t | $^{9}$ ə | k™θə | tí  |
|--------|---------|------------------------|----------|------|-----|
| AUX    | Ι       | spill-ben-tr           | OBL      | ART  | tea |
| 'I pou | ared te | a for him.'            |          |      |     |

(d) ni cən kwłźłcət <sup>9</sup>ə kwθə tís. (AG)

| ni?     | cən      | k̃™əł-łc-t    | ?ə  | kwӨə | tí-s     |
|---------|----------|---------------|-----|------|----------|
| AUX     | Ι        | spill-ben-tr  | OBL | ART  | tea-3POS |
| 'I spil | lled his | tea for him.' |     |      |          |

Evidently if it is *his* tea, as in (d), it has already been poured, and so it must be spilled.

Followed by  $-\partial m$  'intransitive,' -lc- indicates that the act is performed for someone but without specifying the person, as in  $m\ell^2\partial lc\partial m$  'remove something for someone' (cf.  $m\ell^2 x$  'remove it'),  $k^w \partial n\ell lc\partial m$  'carry something for someone' (cf.  $k^w \partial n\ell t$  'hold it'). But context can indicate for whose benefit the act is performed, as in (e).

(e) <sup>?</sup>i cən x<sup>w</sup>t<sup>0</sup>dáləs. <sup>?</sup>ówe<sup>.</sup> yəx<sup>w</sup> čx<sup>w</sup> té<sup>?</sup>təx<sup>w</sup> mé<sup>?</sup>əłcəm. x<sup>w</sup>-ť<sup>θ</sup>əġ-áləs ?i cən ?∕awə ?ə vəx<sup>w</sup> čx w té?-t-əx\* in-press-eye AUX I not ROG INF you try-TR-you mé?-əłc-əm come off-BEN-INTR 'I got something in my eye. Would you like to try to get it out for me?'

I got something in my eye. Would you like to try to get it out for me?

The only other suffix recorded after -lc- is  $-\partial n \partial q$  'another,' which appears in:

*k*<sup>*w</sup></sup><i>onèłcónoq* 'owe people' (lit. 'hold for people')</sup>

- $\delta x^{w} e^{i} l c \delta n \partial q$  'potlatch gift for a designated person' (cf.  $x^{w} e^{i} l c \partial t$  'throw down a gift,' <  $x^{w} e^{i} l$  'lower it') (JP)
- *syeqəłcánəq* 'replacement for a person (as one slave given for another)' (JP) (cf. <sup>?</sup>*əýéqt* 'exchange it')

## 10.4.5. -mət 'concern' (CON)

This suffix can appear variously  $(-\delta m \partial t \sim -\partial m \partial t \sim -m \partial t \sim -m (t \sim -m e^{2}t)$ . It may be that the first element is in origin simply  $-\partial m$  'intransitive,' with a stressed vowel when in the durative aspect, and the final is -t 'transitive.' If so, however, the combination seems to have come to lead a life of its own. It transitivizes verbs that are (with a few exceptions) non-active. If the verb denotes a physical activity, the grammatical object is usually its goal; if a mental activity, its target and/or cause. English translations can require the prepositions *after*; for; to, at, by, or simply concerning, hence the admittedly tentative label.

The development of a new suffix out of a combination seems traceable. First, as indicated above ( $\S10.2$  [2.2]), several verbs ending in *-om* can take *-t* to produce transitives the objects of which are goals, as in:

```
cíləmət 'climb up it' (< cíləm 'climb up')
nəqémət 'dive for it' (< náqəm 'dive') (AG)
```

There are also a few verbs that do not appear with  $-\partial m$  that take  $-\partial m\partial t$  with the same effect, such as:

- ya·ysámət 'work on it/at it' (< yá·ys //yáyəs// 'work')
- $\dot{q}$ *ətí nəmət* 'go along (as through a village) inviting them' (cf.  $\dot{q}$ *ətá* $\theta$ *ən* 'go along')
- $\dot{q} \partial p s \partial m \partial t$  'get together and go after him' (<  $\dot{q} p \dot{a} s t$  'call them together')

There are verbs with *-om* of less concrete meaning that take *-t*, such as:

- $y \neq x c \neq z$  (sk for it' (<  $y \neq x c \neq z$  (make a request')
- xá·łəmət 'accept him (as suitor), take care of him' (< xá²łəm 'agree, give
  assent')</pre>
- $t^{\theta}x^{w}im\partial t \sim t^{\theta}\delta x^{w}m\partial t$  'pity him, be merciful to him' (<  $t^{\theta}ix^{w}\partial m$  'pity, be merciful')

Other verbs without  $-\partial m$  can take  $-m\partial t \sim -m\acute{e}t \sim -m\acute{e}^{2}t \sim -m\acute{t}t$  with the range of meanings indicated above:

yók<sup>w</sup>mət 'hire him' (< yék<sup>w</sup> 'hire')  $q^w \delta lmət$  'scold him' (<  $q^w \ell l$  'speak') '>lyámət 'dream about him' (< ' $2\delta lya$  'have a vision') hók<sup>w</sup>mət (JP), hók<sup>w</sup>me<sup>2</sup>t (AG, DK) 'come to remember him' (< hék<sup>w</sup> 'remember') tcíwsmət 'get tired of him' (< tcíws 'get tired') síwəlmət 'be aware of him' (AG) (< síwəl 'be aware') cíwəlme<sup>2</sup>t 'get annoyed with him' (AG) (< cíwəl 'get annoyed') Xí?mət 'prefer it, want it' (< Xí? 'valuable') hílək<sup>w</sup>mət 'welcome it' (< hílək<sup>w</sup> 'happy') qxeləmít 'gang up on him' (< qxéləc 'many persons') k<sup>w</sup>i·ck<sup>w</sup>əcmət 'watch something for him' (< k<sup>w</sup>éc 'see') məqmít 'get full of it' (ES/AG) (< máq 'get full') téyəqme<sup>2</sup>t 'get angry with him' (< téyəq' 'get angry')

The last was AG's term, but he accepted a passive  $i i q \delta m \partial t \partial m$ , used by CC, which implies a differently shaped active.

The above is not an exhaustive list. This suffix or combination of suffixes may be productive. Moreover, AG gave a form  $stctwsme^{2}t$  'tired of him,' composed of stctws 'tired,' which appears to be a resultative form, suggesting that, like  $-stax^{w}$  'causative,'  $me^{2}t$  may be suffixed to resultatives.<sup>6</sup>

## 10.5. REFLEXIVES

These are  $-\theta \partial t$ ,  $-n \dot{a} m \partial t$ , and  $-st \partial n \dot{a} m \partial t$ . Undoubtedly they are composed of the transitivizers -t,  $-n \partial x^w$ , and the causative  $-st \partial x^w$ , respectively, plus one or more other suffixes. But they occur more widely than the transitivizers and causative

<sup>6</sup> Gerdts (1999, 85-86) identifies a Cowichan applicative  $-me^2$  (which is followed by -t 'transitive') as a benefactive, the counterpart of -lc-, used with intransitive verbs.

do alone or with the person markers, and are more conveniently treated as separate entities.

## 10.5.1. -θət 'oneself'

This produces a large number of reflexive forms that include the counterparts not only of English *raise oneself* and *shake oneself* but also of many English intransitive forms like *spread out* and *get fat*. It is no doubt composed of *-t* 'transitive' and a suffix //-Sat// 'self.' It appears as  $-\theta at$  following a CoC root in its CoC form, otherwise it appears as  $-\theta at$ . Like some other suffixes with an /a/ vowel, it converts an /e/ in the stem to /a/. It seems to be fully productive with roots that take *-t*, it joins them in the same fashion as *-t*, and progressives are formed by the same rules, all of this giving good evidence that it is nothing other than *-t* plus a //-Sat//. Forms with  $-\theta at$  paralleling forms with *-t* include:

| <i>sé<sup>9</sup>t</i> 'raise it'                   | $s\dot{a}^{2}\theta \partial t$ 'raise oneself'                                                        |
|-----------------------------------------------------|--------------------------------------------------------------------------------------------------------|
| $p \epsilon \theta \partial t$ 'spread it out'      | $p\dot{a}\theta\partial\partial t$ 'spread out (INTR)'                                                 |
| $\tilde{X} \dot{a}^{2} t$ 'comfort him'             | $\dot{\chi}\dot{a}^{2}\Theta \partial t$ 'calm down'                                                   |
| <i>hí·lt</i> 'roll it off'                          | $hi \cdot l\theta \partial t$ 'let oneself fall' (cf. $hi l \partial m$ 'fall off roll')               |
|                                                     | 011, 1011 )                                                                                            |
| <i>tsát</i> 'approach it, bring it close'           | $ts \delta \theta \partial t$ 'approach' (cf. $t \delta s$ 'arrive there')                             |
| $\vec{k}^{w} t \acute{e} t$ 'tip it over, spill it' | <i>k<sup>w</sup>łáθət</i> 'capsize'                                                                    |
| $k^{w} \delta y \check{x} t$ 'move it, shake it'    | $k^{w} \delta y \check{x} \theta \delta t$ 'move, act' (cf. $k^{w} \delta y \delta \check{x} \delta m$ |
|                                                     | 'shake')                                                                                               |

The reflexive  $-\theta \partial t$  also follows roots that do not take or are not prepared to take -t, providing an inchoative meaning, 'get oneself, make oneself, become,' as in:

| nás 'fat'                               | násθət 'get fat'                                                     |
|-----------------------------------------|----------------------------------------------------------------------|
| $\vec{k}^{w} \acute{es}$ 'get scorched' | $k^w \acute{a} s \theta \partial t$ 'become hot (weather)'           |
| $i \delta q^w$ 'wet'                    | $l \partial q^{w} \theta \delta t$ 'get all wet'                     |
| $k^{w} \delta n$ 'become possessed'     | <i>k</i> <sup>w</sup> <i>onθát</i> "get started" (become possessed)' |
| ²áỷ 'good'                              | $^{2}\partial \dot{y}\theta \dot{a}t$ 'get better'                   |
| $\theta i$ 'big'                        | <i>θiθát</i> 'get bigger'                                            |
| si <sup>9</sup> ém 'rich person'        | $si^2 \acute{am} \theta at$ 'become rich'                            |

Here too belong words in which  $-\theta \partial t$  follows  $-\partial m$  'intransitive,' with a shift of stress to that suffix, as in:

| $i^{\theta} \dot{q}^{w} \partial m$ 'be rotten' | $t^{\theta} \dot{q}^{w} \dot{a} m \theta \partial t$ 'become rotten'      |
|-------------------------------------------------|---------------------------------------------------------------------------|
| háq <sup>w</sup> əm 'stink'                     | $h \partial q^w \acute{a} m \theta \partial t$ 'stink' ['become stinky'?] |

And possibly formed with -om are:

| sŽéləqəm 'fierce (be                           | ing)' <i>Χ̈́əlqэ́mθət</i> | 'be exceptionally bad'     |
|------------------------------------------------|---------------------------|----------------------------|
| <i>šx<sup>w</sup>né<sup>9</sup>əm</i> 'shaman' | x <sup>w</sup> ən'əm      | 'train to become a shaman' |

I have recorded one example of the reflexive and inchoative senses in contrast; AG gave  $lq^w \delta \theta \delta t$  'wet one's pants' and  $l\delta q^w \theta \delta t$  'get wet' (<  $l\delta q^w$  'be wet').

#### 10.5.2. -námət 'oneself (limited control)'

This is the reflexive of  $-n \partial x^w$  'transitive (limited control).' It can mean 'do to oneself unintentionally,' 'do to oneself in spite of difficulties,' or 'be able to do (get started doing) in spite of difficulties.' It is no doubt composed of  $-n \partial x^w$  (probably the -na-) plus  $-\partial t$  'self,' perhaps linked with  $-\partial m$  'intransitive.' It appears as  $-l \dot{a} m \partial t$  following a final -l in a stem, otherwise as  $-n \dot{a} m \partial t$ .

That  $-n \dot{a} m \partial t$  is to  $-n \partial x^w$  as  $-\theta \partial t$  is to -t is shown by the following set:

| $k^{w} \delta l \partial xt \sim k^{w} \delta l l \partial xt$ 'shoot him | $k^{w} \delta ll \partial x \theta \partial t$ 'shoot oneself |  |  |
|---------------------------------------------------------------------------|---------------------------------------------------------------|--|--|
| intentionally'                                                            | intentionally' (CC)                                           |  |  |
| <i>k<sup>w</sup>álləxnəx<sup>w</sup></i> 'shoot him                       | k <sup>w</sup> állaxnàmat 'shoot oneself                      |  |  |
| accidentally'                                                             | accidentally' (CC)                                            |  |  |

Like  $-\theta \partial t$ ,  $-nam\partial t$  can be suffixed to both forms that ordinarily take transitive suffixes and forms that do not. In the following,  $-nam\partial t$  is suffixed to a root that usually takes the transitivizer  $-n\partial x^w$ :

 $t \partial \dot{q}^w n \dot{a} m \partial t$  'bump oneself accidentally' (cf.  $t \partial \dot{q}^w n \partial x^w$  'bump him accidentally,'  $t (\dot{q}^w \partial t$  'bump him on purpose')

 $k^w \partial q^w n \dot{a} m \partial t$  'hit/chop oneself accidentally' (cf.  $k^w \dot{a} q^w n \partial x^w$  'hit/chop him accidentally,'  $k^w \dot{a} q^w \partial t$  'hit him on purpose')

- $k^w \partial cn \dot{a}m \partial t$  'see oneself (as when walking past a mirror or in a dream)' (cf.  $k^w \partial cn \partial x^w$  'see him,'  $k^w \partial c \partial t$  'look at him,'  $k^w \partial k^w c \dot{a} s \partial m$  'be looking at one-self in a mirror,'  $\langle k^w \dot{e}c$  'see,'  $-\dot{a}s$  'face,'  $-\partial m$  'intransitive')
- $se^{2n\acute{a}m\partial t}$  'manage to raise oneself (raise oneself but only after difficulty)' (cf.  $s\acute{a}^{2}\theta\partial t$  'raise oneself,'  $s\acute{e}^{2t}$  'raise him, lift it')

In the following,  $-n\acute{a}m \partial t$  is suffixed to intransitive forms, including some with  $-\partial m$  'intransitive' and  $-\acute{e}ls$  'activity.' Here  $-n\acute{a}m \partial t$  can have the sense 'do in spite of oneself' as well as 'do in spite of difficulties,' as in:

náyəmnámət 'get to laughing in spite of oneself' (< náyəm 'laugh')

*téyəqnámət* 'get mad against one's intention' (< *téyəq* 'get angry')

 $n \partial q^w n \acute{a} m \partial t$  'oversleep (i.e., sleep beyond one's intentions), get to sleep in spite of difficulties' (<  $n \acute{a} q^w$  'fall asleep')

 $k^w \partial^2 \partial lsn \acute{a}m \partial t$  'lose one's grip, let go unintentionally' (<  $k^w \partial^2 \acute{e}ls$  'let go'; cf.  $k^w \acute{e}^2 t$  'let it go, drop it')

*iíləmnàmət* 'get to singing' (< *iíləm* 'sing')

?íməxnàmət 'be able to walk' (< ?íməx 'walk')</pre>

 $k^{w'} \acute{e} \cdot ll \grave{a} m \partial t$  'manage to get a hideout' (<  $k^{w} \acute{e} \cdot l$  'hide'; cf.  $k^{w} \acute{e} l x$  'hide him')  $\theta \acute{l} n \grave{a} m \partial t$  'improve, get lucky, recover strength' (<  $\theta \acute{t}$  'big')

 $\partial i y n \dot{a} m \partial t$  'good to watch, good to see' (<  $\partial i \dot{y}$  'good'; cf.  $\partial i \dot{y} \partial i \dot{t}$  'get better)

Perhaps *-námət* in this last implies good in the eyes of others, regardless of the subject's own intentions.

# 10.5.3. -stanámat 'oneself (causative)'

This is the reflexive of  $-st a x^{w_1}$  'causative' and is probably composed of this suffix plus  $-n \acute{am} at$  'oneself (limited control).' Its basic meaning is 'make oneself,' but it may be translated as 'feel' or 'pretend to.' The allomorph  $-st \acute{an} a m at$ , reducible to  $-st \acute{an} m at$ , appears after an unstressed stem, while  $-st an \acute{am} at$ appears after a stem that bears a stress. Some examples are:

yè<sup>2</sup>etstənámət 'make oneself vomit' (cf. yé<sup>2</sup>etstəx<sup>w</sup> 'make him vomit') xəłstánəmət 'feel sorry for oneself' (cf. xáłstəx<sup>w</sup> 'feel sorry for him') təystánmət 'feel cheap' (cf. té<sup>2</sup>it //téyət// 'insult him') dàdəystənámət 'pretend one is sick' (< dádəy' 'sick') x<sup>w</sup>àx<sup>w</sup>ək<sup>w</sup>stənámət 'pretend one is drunk' (< šx<sup>w</sup>áx<sup>w</sup>ək<sup>w</sup> 'drunk')

# 10.6. THE RECIPROCAL $-t\partial \vec{l}$ 'EACH OTHER'

The only reciprocal suffix that has been recorded is  $-t\partial \hat{l}$  'each other,' which appears as  $-t\hat{a}\hat{l}$  in a few words that may be durative or resultative forms and when it is followed by  $-st\partial x^w$  'causative.' The distinction between full and limited control seen in the transitives seems to be lacking in the reciprocal. The suffix  $-t\partial \hat{l}$  may be composed of -t 'transitive' plus an element  $-a\hat{l}$ . However, it differs from -t 'transitive' in that it may appear without the schwa that normally precedes the -t with a CAC root, as in:

 $t \acute{a} s t \vec{a} \vec{l}$  'collide' (<  $t \acute{a} s$  'get hit'; cf.  $t \acute{a} s \vec{a} t$  'hit it')  $t \acute{q}^w t \vec{a} \vec{l}$  'collide' ( $t \acute{q}^w$  'bump'; cf.  $t \acute{q}^w \vec{a} t$  'bump it')

Like  $-\theta \partial t$ , it converts an /e/ in the stem to /a/ (as in a few examples given below). Its reciprocal sense may be illustrated by the following:

*ləqáltəl* 'jibe, be properly aligned' (< *ləqél* 'be on the mark, be just right'; cf. *ləqélt* 'put it right,' *ləqálθət* 'get in line')

*?əýáqtəl* 'exchange with each other' (cf. *?əýéqt* 'exchange it, replace it')

*xəmá·ntəl* 'be enemies, be rivals' (< *xəmén* 'enemy, rival'; cf. *xəmé·nt* 'play against him')

x<sup>w</sup>cámastał 'meet one another' (< x<sup>w</sup>cámas 'meet'; cf. x<sup>w</sup>cámast 'meet him') cẩámatał 'jump and grab one another' (< cẩám 'jump'; cf. cẩámat 'jump at him')

 $t \dot{a} \dot{x} \partial t \partial t$  'distribute game/catch' (cf.  $t \dot{x} \partial \theta \partial t$  'divide, branch out,'  $t \dot{a} \dot{x} \partial s t$  'give him his share')

*téyəmtəl* 'stick together' (< *téyəm* 'stick, adhere')

Some words with -tal have been recorded in what appears to be the progressive only, such as:

- $k^{w} \partial k^{w} \hat{a}^{2} t \partial \hat{l}$  'separate, divorce' (cf.  $k^{w} \hat{e}^{2} t$  'let it go, drop it,'  $k^{w} \partial k^{w} \hat{e}^{2} t$  'be letting it go')
- $\dot{q}\partial\dot{q}\dot{a}^{2}t\partial\dot{l}$  'happen to meet unexpectedly' (cf.  $\dot{q}\dot{a}^{2}t$  'put it together,'  $\dot{q}\dot{a}^{2}\theta\partial t$  'join')
- $\vec{\lambda} = \vec{\lambda} t^2 t \hat{s} \hat{t}^2 t \hat{s} \hat{t}^2 t \hat{s} \hat{t}^2$  'like each other' ( $\langle \vec{\lambda} t^2$ ' dear, important, difficult'; cf.  $s \vec{\lambda} t^2$  'something wanted, liked')

Other words with  $-t\partial \vec{l}$  occur in what appears to be the durative aspect, such as:

 $\theta \partial \theta \check{x} i^{2} t \partial \check{l}$  'push each other' (cf.  $\theta \partial \theta \check{x} \acute{e} t$  'keep pushing (dur.),'  $\theta \check{x} \delta t$  'push it,' <  $\theta \delta \check{x}$  'move under pressure')

*ćəćəwítəl* 'help each other' (cf. *ćéwət* 'help him')

 $\dot{q}\partial\dot{q}\dot{x}\dot{a}t\partial\dot{l}$  'insult each other' (cf.  $\dot{q}\dot{x}\dot{a}t$  'insult him,'  $c\dot{q}\dot{q}\dot{x}$  'black')

*pəptítəl* 'challenge each other' (cf. *ptém* 'ask')

Others are glossed in a way that suggests they are in a progressive durative form, such as:

- *mimŽítəl* 'be paying each other back (as in a feud)' (cf. *máŽət* 'repay him,' *máməŽət* 'be repaying him')
- $q^{wi}q^{w}$   $\hat{m}c$   $\hat{a}st \hat{d}$  'be having their arms around each others necks' (cf.  $q^{w}$   $\hat{m}c$   $\hat{a}st$  'hug him, put one's arms around his neck')

Forms produced by still other internal modification of the root are:

nočnáčtol 'differ from one another' (< néč 'be different')</p>
cáck<sup>w</sup>tol 'be far apart' (< cák<sup>w</sup> 'far')
q<sup>w</sup>ólq<sup>w</sup>oltol 'talk together' (< q<sup>w</sup>ólq<sup>w</sup>ol dispositional of q<sup>w</sup>él 'speak')
sólsolqtal 'even with one another' (< soltq' 'even, equal,' the resultative of loqet 'make it even')</p>

The suffix  $-t\partial \hat{l}$  can be followed by  $-st\partial x^w$  'causative,' as in:

 $\dot{q}\dot{a}^{2}t\dot{a}l\partial st\partial x^{w}$  'put them together' (<  $\dot{q}\dot{a}^{2}t$  'put it together')  $\theta\partial l\partial qt\dot{a}l\partial st\partial x^{w}$  'separate it, divide it' (<  $\theta\partial l\partial qt\partial l$  'separate from one another')  $s\partial ls\partial l\dot{q}t\dot{a}l\partial st\partial x^{w}$  'divide evenly' (<  $s\delta ls\partial l\dot{q}t\dot{a}l$  'even with one another')

## 10.7. THE PERMISSIVE -s

The suffix -s 'permissive' (PERM) occurs in imperative constructions with the sense "let" as in "Let him go," or "Let it be" as in "Let it be you." It may be suffixed to transitive and intransitive verbs and to personal words. Historically, it may be the -s of  $-stax^w$  'causative,' but that suffix is itself a transitivizer and never follows one, while -s 'permissive' has much greater freedom in what it can follow, as illustrated in the following:

(a) k<sup>w</sup>ánəts čx<sup>w</sup>. k<sup>™</sup>án-ət-s čx<sup>w</sup>. be.taken-TR-PERM you 'Let him have it.' (b) híləms łe <sup>9</sup>al. híləm-s ?aĺ łe fall-PERM PER just 'Just let it fall.' (c) Žás le ném. Žá-s łe ném BE3P-PERM PER go 'Let him go.' (d) nówos łe ném. ném łe náwə-s be.you-PERM PER go 'You be the one to go.' (e) táx ws čx w ? ps xté? st px w. táx<sup>w</sup>-s čxw ?ə-s-xté?-stəx\* later-PERM vou your-NOM-do-CAUS 'Do it later.' (f) spépək<sup>w</sup>s čx<sup>w</sup> <sup>9</sup>al. ?aĺ s-pépək<sup>w</sup>-s čxw **RES-warm-PERM** you just 'Leave it warming.' (g) <sup>9</sup>éli<sup>9</sup>s čx<sup>w</sup>. (AG) éləy-s čx<sup>w</sup> good(PL)-PERM you 'Make them good.' (said when ordering shoes) (h) <sup>9</sup>ówəs té<sup>9</sup>s čx<sup>w</sup> k<sup>w</sup>ə wét stálk<sup>w</sup>ł hóỷθəstè<sup>9</sup>ep. (CC 10) ?áwə-s k<sup>™</sup>a wét sťálk<sup>w</sup>ł té?-s čxw not-PERM appear-PERM you ART who non-dancer háydəs-t-è?ep be.telling-TR-you(PL) 'You will not tell any non-dancer.' (lit. 'Let it be none [?5wəte?] the [hypothetical] non-dancer whom you [plural] tell.')

(i) mèsí, nəšx<sup>w</sup>?ád<sup>w</sup>a?. háystəx<sup>w</sup> čx<sup>w</sup> tə tsás ?əššq<sup>w</sup>éləwən. Xás čx<sup>w</sup> k<sup>w</sup>ə cícəł si?ém hé?k<sup>w</sup>ənəsəx<sup>w</sup>. (CC)

nə-šž<sup>w?</sup>ád<sup>w</sup>a? mèsí háv-stəx<sup>w</sup> čxw tə tsás thanks my-sibling stop-CAUS ART pitiful you <sup>9</sup>əs-šx<sup>w</sup>q<sup>w</sup>éləwən Åa-s čx<sup>w</sup> k<sup>™</sup>ə cícəł si?ém vour-feeling BE3P-PERM you ART above lord hé?kw-ənəs-əxw remember-GOAL-vou

'Halleluia, my brother/sister. Get rid of your sad thoughts, and remember God.' (lit. 'Let it be the Lord Above whom you are remembering' – a Shaker counsel)

# 10.8. THE SUBORDINATE PASSIVE - *at*

The suffix  $-\partial t$  'subordinate passive' (SUBPAS) follows the passive personal suffixes in subordinate passive forms (see §14.2.6). It may be identical with -t 'stative' (§12.1.6).

# **11** Non-Personal Affixes 2: Aspectual and Modal Affixes

These are two sets of elements that appear first and last in the predicate head. They are tentatively grouped together here simply because they stand apart from the derivational affixes and the inflectional affixes of the voice and personal systems. Further analysis may suggest a different grouping.

## **11.1. ASPECTUAL PREFIXES**

These are some nine elements that express distinctions mainly of a temporal sort that might be called aspectual. They are:  $wa_{2}$  'established,' wat- 'already,'  $x^{w}an$ - 'still,'  $ya_{1}$  'along,'  $wa_{3}$  'suddenly,'  $ya_{2}$  'first,'  $nax^{w}s$ - 'habitually,'  $x^{w}a$ - 'become,' taw- 'somewhat.' These appear in the predicate and, with the exception of  $wa_{2}$  and wat-, invariably before the head. These two,  $wa_{2}$  and wat-, can also appear before a directional auxiliary, while  $wa_{2}$  can also appear before an adverb. Because the head of a predicate can be any kind of word, these elements may precede  $s_{1}$  '(word) nominalizer' (see §12.1.1) and the possessives na- 'my' and  $an \sim a^{2}aT$ - 'your' or  $s_{2}$  'resultative.'

The two,  $wa_2$  'established' and wal- 'already,' are also particle-like in that they may become attached to a preceding particle, auxiliary, or adverb. Because of these features, they were identified in §2.6 as "ambivalent affixes." They are not particles as defined in §2.5 in that they seem (with the possible exception of  $wa_2$  in some usages) grammatically related simply to the following word rather than to some larger construction. They are also neither mobile, as the predicate particles are, nor separable from their head words as the articles may be separated from their noun heads by various modifiers.

Some of these prefixes can co-occur. Table 11.1 shows in Roman numerals the relative positions in which they have been recorded and in Arabic numerals the order in which they are described below. The suffixes in the same column seem to occupy the same position and not to co-occur.

Table 11.1

| Aspectual Prefixes                           |                                    |                            |                                             |
|----------------------------------------------|------------------------------------|----------------------------|---------------------------------------------|
| Ι                                            | Π                                  | III                        | IV                                          |
| 1. <i>wə</i> - <sub>2</sub><br>'established' | 4. <i>x<sup>w</sup>ən-</i> 'still' | 5. <i>yə-</i> 1<br>'along' | 6. <i>x<sup>w</sup>ə</i> -<br>'become'      |
| 2. wəł-<br>'already'                         |                                    |                            | 7. $w \partial_{3}$<br>'suddenly'           |
| 3. <i>təw-</i> 'somewhat'                    |                                    |                            | 8. <i>y</i> ∂-₂<br>'first'                  |
|                                              |                                    |                            | 9. <i>nəx<sup>w</sup>s-</i><br>'habitually' |

Examples are too few for absolute certainty, but there are instances of  $w_{\partial_2}$ before  $x^w_{\partial n}$ ,  $y_{\partial_1}$ , and  $x^w_{\partial_2}$ ; of  $w_{\partial_1}$  before  $y_{\partial_1}$ ,  $x^w_{\partial_2}$ , and  $w_{\partial_3}$ ; of  $t_{\partial w}$  before  $x^w_{\partial n}$  and  $y_{\partial_2}$ ; and of  $x^w_{\partial n}$  before  $y_{\partial_1}$ . In the materials dictated by CC and JP, I have noted only one instance of  $w_{\partial_2}$  and  $w_{\partial_1}$  co-occurring, with  $w_{\partial_1}$  preceding  $w_{\partial_2}$ . On the other hand, AG used  $2u^2$  (corresponding to  $w_{\partial_2}$ ) before  $w_{\partial_1}$  (see §4.3.1, following example [k]). More work is needed, especially on the various uses and privileges of occurrence of  $w_{\partial_2}$ .

## 11.1.1. wa-2 'established' (EST)

This is used often and in a variety of ways with functions and meanings that are not yet clear. The gloss 'established' is an approximation of one apparent meaning but perhaps misses others. In some constructions, it seems simply to link one element with another. It is possible that two or more homophonous elements have been identified here as  $w\partial_{2}$ .

As used by older people (AC, CC, JP),  $w \partial_{2}$  appears in slower, more deliberate speech as  $w\partial_{2}$  but in more rapid or informal speech, it may become attached to a preceding element (a particle, auxiliary, or adverb) as  $-\partial w$  or -w, as in s - + $w\partial_{2}$  can become  $s\partial w$ ,  $ni^{2} - + w\partial_{2}$  can become  $ni\dot{w}$  or  $n\partial w$ ,  $\dot{\lambda}e + w\partial_{2}$  can become  $\dot{\lambda}\partial w$ . Most younger speakers, however, are said not to use the  $w\partial_{2}$  form but to say  $^{2}\partial w$  or  $^{2}\partial \dot{w}$  ( $^{2}u$  or  $^{2}u^{2}$ ) in the Cowichan manner, as did DK and AG.

 $w_{\partial_2}$  is clearly distinct from the subordinating particle  $w_{\partial_1}$  'if, when, that,' which is always accompanied by a subordinate subject marker and which, in the speech of CC and JP, is invariably  $w_{\partial}$  (see §4.2, "Subordinate Clauses"). It is also distinct from  $w_{\partial_3}$  'suddenly' (see below).

 $w_{\partial_2}$  commonly appears preceding a predicate head that is itself preceded by an auxiliary or some other word, and it usually seems to add so little lexical meaning to the construction that one is tempted to suppose that it simply serves some grammatical function, perhaps to link the predicate head more clearly to whatever precedes it. However, the following pair of sentences contrasting the absence and presence of  $wa_{2}$  (unfortunately, the only such pair available) shows that it can indicate an established or continuing state or an established fact:

- (a) <sup>9</sup>i cən c-náx<sup>w</sup>əł.
   AUX I get-canoe
   'I have a canoe. I have obtained a canoe.'
- (b) ?i cən wə-c-nə́x wəł. AUX I EST-get-canoe 'I own a canoe. I do have a canoe.'

Other examples are not incompatible with this interpretation:

- (c) ?í·ł cən wəcnáx wəł.
   ?í-əł cən wə-c-náx wəł
   AUX-past I EST-get-canoe
   'I had a canoe.'
- (d) <sup>9</sup>i wə-<sup>9</sup>í k<sup>w</sup>θə nə-mɨňə. (CC)
   AUX EST-be.here ART my-child
   'My son is here.'
- (e) ni 'aw 'ay'. niŵ 'ay'. ni' 'a wa-'ay' ni' wa-'ay' AUX ROG EST-good AUX EST-good 'Is it good? It's good.'
- (f) Xa maw híwaqw.
   Xá ma wa-hiwaqw
   BE3P CERT EST-head
   'He's the head.'

However, in the following, the  $wa_{2}$  is prefixed to a verb complement referring to an activity that has not yet begun:

(g) k \*\* an θát can wak \*\* atx \*\* álcap. (CC)
 k \*\* an - θát can wa-k \*\* átax \*\* - álcap
 be.taken-self I EST-be.entering-firewood
 'I'm going to start packing in wood.'

With a few adverbs, such as  $\dot{n}\dot{a}n$  'too, very,'  $y\dot{a}\theta$  'always,'  $m\dot{\sigma}k^{w}$  'all,' and  $\dot{t}\dot{a}t$  'long before,'  $w\partial_{2}$  usually, if not always, appears prefixed to both the adverb and the predicate head, as in:

- (h) wəủán wəqóx. ~ wəủánəw qóx. wə-ủán wə-qóx
   EST-too EST-many
   'It's too much. There are too many.'
- (i) wəyáθ cən wəċćċəwət.
   wə-yáθ cən wə-ċćċəw-ət
   EST-always I EST-be.helping-TR
   'I'm always helping him.'
- (j) wəmák<sup>w</sup> wəpépd.
   wə-mák<sup>w</sup> wə-pépd
   EST-all EST-white(PL)
   'They're all white.'
- (k) wəiát wəšx "né?em. wə-tát wə-šx "né?em EST-earlier EST-shaman 'She had previously been a shaman.'
- (1) wəňán čx<sup>w</sup> wənəsẳí?.
  wə-ňán čx<sup>w</sup> wə-nə-s-c-ẳí?
  EST-very you EST-my-NOM-do-value
  'I really love you. I love you very much.' (lit. 'You are very much what I want.')
- (m)?i ct wəłwəảánəw mi x "cák". (JP 22)
  ?i ct wəł-wə-ảán wə-?əải x "-cák"
  AUX we already-EST-very EST-come move.toward-far
  'We've come too far.'

Following scécon 'really,' wo-2 appears before the head only, as in:

- (n) sčéčəň wə<sup>9</sup>áý. really EST-good 'It's really good.'
- (o) sčéčəň wə-θá·q<sup>w</sup>.
   really EST-big.headed
   'He's really got a big head.'

A prefix  $w\partial$ - commonly occurs with the particle 2al' (just' (see §16.2.8). The combination  $w\partial$ - ... 2al' can have the sense 'just, only' given by 2al' alone, or it can simply identify what it encloses as an adjective-like form. I tentatively identify the prefix in this construction as  $w\partial_2$  but must point out that it can be ordered differently, as in:

(p) xém i ni x vəwəqadqəy ?al.
 xém ?əy ni? x və-wə-qadqəy ?al
 cry and AUX become-EST-sick just
 'It [a baby] cried itself sick.'

Here  $w_{\partial_2}$  follows rather than precedes  $x^w_{\partial}$  'become,' but perhaps the  $x^w_{\partial}$  is prefixed to the whole construction  $w_{\partial q} \dot{a} \dot{q}_{\partial y} \dot{a} \dot{l}$  'sick.' (Other examples of constructions with  $w_{\partial}$  ... ' $a \dot{l}$  appear in §3.8.3.4).

Like any other aspectual prefix,  $wa_{2}$  can be preceded by *s*- 'clause nominalizer.' In (q),  $wa_{2}$  appears before an adverb in a nominalized clause:

(q) nəski? kwə ?əswəmákwələp ?é?əm. (CC)
 nə-s-ki? kwə ?ə-s-wə-mákw-ələp ?é?əm
 my-NOM-want ART your-NOM-EST-all-you(PL) give
 'I want you all to give.'

In (r),  $wa_2$  appears twice, evidently linking the adverb that is the predicate of the main clause and the resultative form that is the predicate of the embedded nominalized clause:

(r) wəyáθ k<sup>w</sup>s wəstə<sup>9</sup>és k<sup>w</sup>ə ...
wə-yáθ k<sup>w</sup> s-wə-stə<sup>9</sup>é-s k<sup>w</sup>ə
EST-always ART NOM-EST-like-3POS ART
'It was always as though ... It seemed that always ...' (JP 20)

 $wa_2$  also appears following *s*- 'clause nominalizer' in nominalized clauses that stand as sentences in narratives (see §4.3.4, "Nominalized Narrative Sentences").

 $w\partial_{2}$  is probably also what appears as  $-\partial w$ - in the demonstrative set  $\theta \partial w \dot{\lambda} \dot{a}$ 'she,'  $t\partial w \dot{\lambda} \dot{a}$  'he,' and so on (see §15.2.2.2).

## 11.1.2. *wəł-* 'already'

This appears commonly before a predicate head but, like  $wa_{-2}$  'established,' it can also appear before a directional auxiliary. Like  $wa_{-2}$  too, it can appear in two different shapes. When attached to some preceding element, it can appear as -awt, but more often it appears as wat. It usually has the sense 'already' or forms what corresponds to an English present perfect tense, as in:

(a) ni? cən wəł-skakidw. (CC)
 AUX I already-bundled
 'I'm already bundled up.'

| (b) ni | wəł-ném              | tánnəc | tə  | łqélċ. (CC) |
|--------|----------------------|--------|-----|-------------|
| AUX    | already-AUX(go)      | set    | ART | moon        |
| ʻTh    | e moon has just gone | down.' |     |             |

It may have the sense 'still,' as in:

(c) Xowoł táwon. (AC) also already-town
'It's more town.' (i.e., 'It's still town all the way out there.')

With a progressive form, it may have the sense 'be starting to,' as in:

(d) ni mɨ wəłyəhɨnɨm. (CC) ni? mɨ wəł-yə-hɨnɨm AUX CERT already-along-be.going 'He's starting to go.'

With a negative, wol-may be translated 'no longer, not anymore,' as in:

(e) ni čx<sup>w</sup> wəłćísəm. wəł?ówə čx<sup>w</sup> ni?əx<sup>w</sup> sźíźgəł. (JP) ni? čx w wəł-císəm. wəł-?śwa čx w ni?-əx<sup>w</sup> sŽíŽqəł AUX vou already-grow already-not vou AUX-you child 'You have grown up. You're not a child anymore.'

What appears to be  $w\partial t$ - occurs in a few words that must be listed in the lexicon:

wəł yé ýsələ 'two animals' (< yé ýsələ 'two persons' < yəsélə 'two')</li>
wəłθίθə 'ritualist,' i.e., a person who uses spells (syəŵín) professionally (possibly < θί 'great')</li>

## 11.1.3. tow- 'like, somewhat, a bit, sort of (like)'

This is possibly composed of a root  $t\hat{e}^{\gamma}$  'resemble, appear,' which can be inferred from  $t\partial^{\gamma}\hat{e}m\partial x$  'resemble,'  $st\partial^{\gamma}\hat{e}$  'like,' and possibly ' $\partial w\partial te^{\gamma}$  'none,' and  $w\partial_{\gamma_{2}}$  'established.' The sense 'like, resembling' is seen in:

- (a) təw-x "ənítəm. (AC) like-White.person'They're like White people.'
- (b) təw<sup>2</sup>í<sup>2</sup>tət <sup>2</sup>al tən höyq<sup>w</sup>. (DK) təw-<sup>2</sup>í<sup>2</sup>tət <sup>2</sup>al tə <sup>2</sup>ən-höyq<sup>w</sup> like-be.sleeping just ART your-be.burning 'Your fire is just like it's sleeping.'
- (c) smótəqsən təw-tóki (AC) snotty like-turkey 'snotty like a turkey'

(d) təwspəlq<sup>w</sup>ít<sup> $\theta$ </sup>e<sup> $\gamma$ </sup> <sup> $\gamma$ </sup>ámət ni <sup> $\gamma$ </sup>ə tə s<sup> $\gamma$ </sup>é<sup> $\lambda$ </sup>q i wəł $\theta$ é<sup> $\gamma$ </sup>t (DK) təw-spəlq<sup>w</sup>ít<sup>θ</sup>e? ?ámət ni ?ə s?éŽq °∂ỷ tə like-owl be.sitting be.there OBL outside and ART wəł-θé?t already-dark 'She's like an owl sitting outside in the dark.' (said by a man of his sulking wife)

This sense is stronger with the addition of  $st\partial^2 e'$  like' or the suffix  $-am\partial x$  'resemble,' and such constructions are said to be more usual, as in (e) and (f).

- (e) təw-x \*\* ənítəm-àməx.
   like-White.person-appear
   'They look like White people.'
- (f) təwste?é· kw ləpláš. (DK)
  təw-ste?é ?ə kw ləpláš
  somewhat-like OBL ART plank
  'It's like a plank.' (a lady says of her own cake comparing it with her friend's)

The sense 'somewhat, a bit' appears to be more common, as in:

- (g) təwcəcł>nəp tətiməx<sup>w</sup> (JP 19)
   təw-cicəł->nəp tətiməx<sup>w</sup>
   somewhat-high-ground lands
   'somewhat higher ground'
- (h) mi ce·p təwłtí. (DK)
   ?əmi ce·p təw-ł-tí
   come you(PL) somewhat-partake-tea
   'Come and have some tea.'
- (i) mi łə təwqíkwəm. (AG)
   mi łə təw-qíkw-əm
   come PER somewhat-bite-INTR
   'Give me a bite.'
- (j) təw-?í čx<sup>w</sup> ?al. (JP) somewhat-be.here you just
   'Stay a while. Stay a while.'
- (k) ?i cən təwkwámkwəmdət kwən sni łtí. (JP)
  ?i cən təw-kwámkwəm-dət kwə nə-s-ni? ł-tí
  AUX I somewhat-strong-self ART my-NOM-AUX partake-tea
  'I became a little stronger when I drank tea.'
(1) təwx wn by a mas łda, si ?ém. təw-x<sup>w</sup>-náyəm-əs łə ģə, si?ém like-inside-laugh-face PER EMPH sir 'Smile once in a while, Mister.' (said by an older relative to a sullen adolescent) (m) səctəw höye? i ct təwx wənyə?é.y yə?əx?íxəl i ... (CC 4) s-ni? wə-háye? ?əỷ ct [ni?] ct NOM-AUX we EST-leave and AUX we təw-x<sup>w</sup>ən-yə-<sup>9</sup>é·ý və-?əx?íxəl ?∂ỷ like-still-along-continue along-be.paddling(PL) and 'Then we left and we sort of continued paddling [leisurely] along when ...'

From the sense 'somewhat' may come the use of *tow*- to form a weak imperative, as in:

(n) təw-xəl-ət le. (DK) like-write-tr PER 'Write it down.'

(Perhaps 'Write it down, eh?' or 'You might write it down,' said to be more polite than a simple  $\check{x}\check{a}lat$  *le* 'Write it down.')

Prefixed to derivatives of  $k^{win}$  'how many' (§17.20), *tow*-gives the sense 'several,' as in:

(o) təwk<sup>w</sup>i·lə sté<sup>?</sup>ex<sup>w</sup>əł təw-k<sup>w</sup>in-ələ sté<sup>?</sup>ex<sup>w</sup>əł somewhat-how.many-person children 'several children'

 (p) k<sup>w</sup>θə təwk<sup>w</sup>ənmàt wəqəl ?al (JP 22) k<sup>w</sup>θə təw-k<sup>w</sup>in-mat wə-qəl ?al ART somewhat-how.many-kind EST-bad just 'several bad things'

# 11.1.4. x<sup>w</sup> an- 'still, soon'

This has been recorded most often with the meaning 'still':

- (a) x wən-yə-?í?tət. (DK) still-along-be.sleeping
   'He's still sleeping.'
- (b) k<sup>w</sup>θə <sup>?</sup> əx<sup>w</sup>ín sੈííλqəł nə x<sup>w</sup>əns<sup>?</sup> éλq (JP 2) k<sup>w</sup>θə <sup>?</sup> əx<sup>w</sup>ín sੈííλqəł ni<sup>?</sup> x<sup>w</sup>ən-s<sup>?</sup> éλq ART little child AUX still-outside 'a little child who was still outside'

- (c) kwəns xwənsxíxqəł
   kwə nə-s-xwən-sxíxqəł
   ART my-NOM-still-child
   'when I was still a child'
- (d) x wənnétəł i wəl?ómət tə T. (CC 6)
  x wən-nétəl ?əỳ wəl-?ómət tə T
  still-morning and already-get.up ART T
  'It was early morning when T got up.'

But in some contexts, perhaps only with verbs of motion, the gloss is 'soon':

(e) sčéčən wə-?ýý wə-x wən-hýye?-əs. (CC 13)
really EST-good if-soon-leave-3SUB
'It would be really good if she left soon.'

The word  $x^{w} \partial nt \acute{e} c \partial l$  (<  $t \acute{e} c \partial l$  'arrive here') appears to be an idiom meaning 'early':

(f) ?əmi cən x wən-técəl.
 come I soon-arrive.here
 'I'll come early.'

#### 11.1.5. ya-, 'along'

This has the sense 'while moving' or 'simultaneously.' It occurs commonly with progressive forms. It can indicate action progressing while the actor is moving or, perhaps, while other events are occurring. Compare (a) and (b):

- (a) <sup>9</sup>i cən yə-hɨme?-x.
   AUX I along-be.removing-TR
   'I'm removing them (while moving along).'
- (b) <sup>?</sup>i cən həm e<sup>?</sup>-x. (CC) AUX I be.removing-TR 'I'm removing them (but standing still).'

In (c), (d), and (e),  $y \partial_{-1}$  is also prefixed to progressive forms:

(c) yə-títələm. (DK) along-be.singing

'He's singing away [as a child might while playing].'

(d) səsəw <sup>9</sup>áx qəl θəw xa yəch áy wən. (CC 11)

s-ni-s wə-? $\delta$ Åqəl  $\theta$ əwÅa yə-c-h $\delta$ ýwən NOM-AUX-3POS EST-go.out she along-do-singing.possessed 'Then she went out singing her possessing song (sy $\delta$ wən).'

yə-há·ý. (AG) (e) ni along-be.stopping AUX 'It's stopping. It is coming to a stop.'

In (f), it is prefixed to a durative form:

(f) ha? čx<sup>w</sup> k<sup>w</sup>ə nem qíqə@amx i nem cən yək<sup>w</sup>ənét tən láysəns. (CC 21) ha? čxw k <sup>w</sup>ə nem qíq-ət-Samx ?∂ỷ ném cən if vou then AUX(go) be.bound-TR-me and AUX(go) I və-k<sup>w</sup>ən-é-t nə-láysəns tə my-license along-be.taken-DUR-TR ART

'If you have me arrested, I'll take along my [Shaker minister's] licence.'

In (g), it is prefixed to a resultative form:

(g) səw höye?  $\theta$ ə skwáye kwí? yəšxwxwəxwdwíwən tə sdáyəs sx pélgən. (JP 19) s-wə-həve?[-s] sk<sup>w</sup>áve? k<sup>™</sup>í? θə NOM-EST-leave-3POS ART squirrel ascend yə-šx <sup>w</sup>x <sup>w</sup>əx <sup>w</sup>d<sup>w</sup>íwən sắpélqən [?ə] sģáyəs tə

along-bound.around.middle OBL ART down feather 'Then away climbed the squirrel, the downy feathers tied around its waist '

In (h), it is prefixed to the nominalized head of a relative clause with an extracted instrument.

(h) 🕺 a mэ ni? və-šx<sup>w</sup>-k<sup>w</sup>ák<sup>w</sup>əq<sup>w</sup>-əls <sup>9</sup>é·łtən. (CC) along-OBLNOM-be.getting.hit-ACT BE3P CERT AUX 3PL 'That's what they were hitting with.'

It is prefixed to a few forms indicating location to produce forms indicating direction of motion. Compare (i) with (j) and (k) with (l):

| (i) | ni?     | sí?q     | tə    | šx <sup>w9</sup> ámət. |
|-----|---------|----------|-------|------------------------|
|     | AUX     | under    | ART   | bed                    |
|     | ʻIt's u | nder the | bed.' |                        |
| (j) | néṁ     | cən      | ce?   | yə-sí <sup>9</sup> q.  |
|     | go      | Ι        | FUT   | along-under            |
|     | ʻI'm g  | going un | der.' | -                      |
| (k) | ni?     | scłálw   | əł.   |                        |
|     | AUX     | above    |       |                        |
|     | (       |          |       |                        |

It's above.

(l) wélx yəscłálwəł tə léləm.

| wél-x     | yə-scłálwəł      | [?ə] | tə  | léləm |
|-----------|------------------|------|-----|-------|
| throw-TR  | along-above      | OBL  | ART | house |
| 'Throw it | over the house.' |      |     |       |

It is also prefixed to some of the numerals to produce forms used in counting canoes, cars, and so on. (For those recorded, see §19.2.)

**11.1.6.**  $y_{\partial_2}$  (~ *i*-) 'first, before doing anything else, above all (first)' This seems to appear with perfective forms only, in contrast to  $y_{\partial_1}$ , which usually appears with progressive forms. Most recorded instances have been in direct or implied commands or requests, as in (a) to (d) but not (e).

| (a) | yə?áłtəı                                                       | n ce•p                                                                       | yéł <sup>9</sup> əsné                                                                     | ém <sup>9</sup> éyə                             | ləsàİx™                                        | . (CC)                               |                                              |
|-----|----------------------------------------------------------------|------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------|------------------------------------------------|--------------------------------------|----------------------------------------------|
|     | yə-?ə́łtə                                                      | on c                                                                         | e∙p yé                                                                                    | eł ?ə                                           | -s-ném                                         | °éy∙                                 | -ləs-àlx <sup>w</sup>                        |
|     | first-eat                                                      | У                                                                            | ou the                                                                                    | en yo                                           | ur                                             | NOM                                  | -go.away-TR-us                               |
|     | 'Eat befo                                                      | ore you                                                                      | leave us.                                                                                 | ,                                               |                                                |                                      |                                              |
| (b) | ?áỷ k™sị<br>?áỷ k<br>good A<br>'We'd be                        | yəłé·lc<br>c <sup>w</sup> s·<br>.RTN<br>etter go                             | t, qəİét h<br>-yə-łé·l-c<br>OM-first-g<br>ashore fir                                      | iθínəm<br>ct,<br>co.shorev<br>rst and a         | . (JP 12)<br>vard-ou<br>gain hay               | )<br>qəlé<br>r agair<br>ve somet     | t həyθínəm<br>finish.eating<br>hing to eat.' |
| (c) | ?áwə ce<br>?áwə<br>not<br>'Don't le                            | •p yək<br>ce∙p<br>you<br>et yours                                            | <sup>w</sup> ácnà∙lt.<br>yə-k <sup>w</sup> ec<br>first-see<br>selves be s                 | (JP)<br>c-n-àl-ə<br>-TR-you-<br>seen.'          | t<br>SUBPAS                                    |                                      |                                              |
| (d) | <sup>?</sup> áwə ce<br><sup>?</sup> áwə<br>not<br>'Don't le    | <ul> <li>p isx €</li> <li>ce · p</li> <li>you</li> <li>et it cate</li> </ul> | nhcə <sup>9</sup> e•p.<br>yə-sxár<br>first-be.j<br>ch you.'                               | (JP 13)<br>acə <sup>?</sup> -e∙p<br>prey-you    |                                                |                                      |                                              |
| (e) | ?ówə ni<br>?ówə<br>not<br>wəł-x<br>alread<br>'He did<br>surpri | ∴s yəq́í<br>ni?-əs<br>AUX-3SI<br>wə-ní?<br>ly-becoi<br>not reac<br>ise.')    | <sup>9</sup> θət i ni<br>yə-<br>UB first<br><sup>9</sup> -nəs<br>me-be.the<br>ct in time, | ? cən w<br>qí?θət<br>-react<br>re-TR<br>and I w | vəłx <sup>w</sup> ən<br>?əỷ<br>and<br>ent up t | íns. (JP)<br>ni?<br>AUX<br>o him.' ( | )<br>cən<br>I<br>i.e., ʻI caught him by      |

#### 11.1.7. $n \partial x^{w}s$ - (~ $x^{w}s$ -) 'habitually' (HAB)

It is possible that this is a compound prefix, but there is no evidence for it at present. It appears prefixed to perfective forms of verbs, as in (a) to (c).

- (a) nəx<sup>w</sup>shəwi<sup>θ</sup>é<sup>γ</sup>θəs. (JP) nəx<sup>w</sup>s-həwi<sup>θ</sup>é<sup>γ</sup>-t-S-əs HAB-tease-TR-me-3TR
  'He has the habit of teasing me.'
- (b) nəx<sup>w</sup>słák<sup>w</sup>xən cən <sup>9</sup>əł<sup>9</sup>íməxèn. (JP) nəx<sup>w</sup>s-łák<sup>w</sup>xən cən <sup>9</sup>əł<sup>9</sup>íməx-èn HAB-trip I whenever-walk-I 'I'm always tripping whenever I walk.'
- (c) nəx «scéwəltən k «əna Tom. (JP) nəx «s-céw-əltən k «əna Tom HAB-help-others that Tom
   'Tom is always helping.'

(This last sentence was JP's comment after I had elicited a set of active and passive forms beginning with "Tom helps George.")

This prefix appears in terms for persons engaged in activities habitually or professionally, as in:

nəx<sup>w</sup>s<sup>2</sup>əhé<sup>·</sup>wə (CC), nəx<sup>w</sup>shé<sup>·</sup>wə (JP) 'deer hunter, inland hunter' (< ?əhé<sup>·</sup>wə, hé<sup>·</sup>wə 'hunt deer')
nəx<sup>w</sup>s pá yətət 'duck hunter' (< pá yətət 'hunt ducks')</li>
nəx<sup>w</sup>sk<sup>w</sup>áy x θət 'good provider' (< k<sup>w</sup>áy x θət 'act, move oneself')
nəx<sup>w</sup>s kánəq 'potlatcher' (< kánəq 'give a potlatch')</li>

However, as the earlier examples show, this is an aspect marker, not simply an agent marker equivalent to the English *-er*.

The form  $x^w s$ - has been recorded in the following words only:

 $x^{wsl} \partial h \dot{e}l$  'good bone-game player' (<  $l \partial h \dot{e}l$  'play the bone game') (JP)  $x^{wsl} \partial \dot{e}t^{\theta}$  ' person who always makes fun of others' (cf.  $l \partial \dot{e}t^{\theta}t$  'make fun of him')  $x^{wsl} \dot{p} \dot{a} \dot{\lambda} \partial m$  '''real'' (confirmed) smoker' (<  $l \dot{p} \dot{a} \dot{\lambda} \partial m$  'smoke tobacco') (DK)

# 11.2. MODAL SUFFIXES

Only two of these have been identified. They differ from the derivational suffixes in that they follow suffixes of the voice system, for example,  $-\partial m$  'intransitive,'  $-n\partial x^{w}$  'transitive limited control,'  $-\partial \partial t$  'reflexive.' They may be followed by the third-person transitive subject marker.

#### 11.2.1. -*ilman* 'want to, intend to, seem about to'

This is suffixed to verbs in both perfective and progressive aspects. It may appear as  $-\delta lm \partial n$  in the progressive.

(a) <sup>9</sup>i cənəwł <sup>9</sup>itətəlmən. (DK)
<sup>9</sup>i cən wəł-<sup>9</sup>itət-əlmən
AUX I already-sleep-want
<sup>6</sup>I want to sleep now.<sup>2</sup>

- (b) ?i cən həmx lə la mən. (CC)
  ?i cən həmx la max la max la be.fainting-want
  'I'm about to faint.'
- (c) k<sup>w</sup>θəw stém <sup>?</sup>al ni qáqaynax<sup>w</sup>almanas (JP 22)
   k<sup>w</sup>θə wə-stém <sup>?</sup>al ni<sup>?</sup> qáqay-nax<sup>w</sup>-almanas
   ART EST-what just AUX be.dying-TR-want-3SUB
   'whatever they wanted to kill'
- (d) Xa sxí?s kws kwannoxwólmons towxa xółe e tθe qól mostóyoxw. (JP 25) xa s-c217-s k w s-k<sup>w</sup>àn-nəx<sup>w</sup>-əlmən-s BE3P NOM-want-3POS ART NOM-being.taken-TR-want-3POS təwẩa х́а́f э ?a tθe aál məstávəx<sup>w</sup> that be.doing OBL that bad person 'That is what that bad person who was doing that intended.'

#### 11.2.2. $-i \cdot t \sim -i \partial t (//-\partial y \partial t//)$ 'arrange to, seek to (arrange)'

The meaning of this suffix is not entirely clear. Examples of its use suggest that it indicates that an activity occurs outside the normal social group through a contractual arrangement. It appears as  $-i^2\partial t$  (//- $\partial y^2\partial t$ //) in the progressive.

- (a) ni? ct ném ?>it=n-í-l. (DK) AUX we go eat-arrange 'We invited ourselves to eat there [at a restaurant].'
- (b) ni? ct ném ?ìtət-í·ł. (DK) AUX we go sleep-arrange 'We went to a hotel.'
- (c) <sup>9</sup>i <sup>9</sup>i<sup>9</sup>i<sup>1</sup>tən-i<sup>9</sup>ə<sup>1</sup>. (AG) AUX be.eating-arrange 'He's eating in a restaurant.'
- (d) ném łe θáyəmθət-ì·ł nem [?ə] yəθé?. (JP) go PER stay.with-arrange go OBL them 'Go and ask permission to stay with them.'

(e) ni? θəθəỳcəm-í?əł.
AUX be.sleeping.together-arrange
'She/he is looking for someone to sleep with [have sex with].'

It also appears in ?  $\partial m \partial t \partial t$  (< ? $\partial m \partial t$  'sit,' used for 'go to the bathroom' in some families).

# **12** Non-Personal Affixes 3: Derivational Affixes

We will consider here all of the derivational affixes except the lexical suffixes, which will be presented in §13. Included here are the derivational affixes with purely grammatical meaning, the verbalizing prefixes, the verbal suffixes, and miscellaneous lexical prefixes.

#### 12.1. PURELY GRAMMATICAL DERIVATIONAL AFFIXES

There are only three, or possibly four, simple prefixes, one compound prefix, and a suffix that belong here.

#### 12.1.1. s-1 'word nominalizer' (NOM)

This prefix, perhaps the most commonly used in the language, appears in three kinds of words:

(1) Simple *s*- nouns (see \$8.1 [2]), that is, nouns with bound roots (reduplicated forms showing *s*- to be a prefix), as in:

sxána 'foot' (cf. sxaxína 'feet')
sqwaméy 'dog' (cf. sqwamqwaméy 'dogs')
smént 'rock, mountain' (cf. smanmént 'rocks, mountains')

(2) Derived *s*-nouns (see \$8.1 [3]), that is, nouns formed from verb roots. These may name the object of the action designated by the verb, as in:

s<sup>2</sup>áltan 'food' (< <sup>2</sup>áltan 'eat,' i.e., that which is eaten) s<sup>2</sup>tí<sup>2</sup>am 'clothing' (< <sup>2</sup>tí<sup>2</sup>am 'get dressed,' i.e., that which is put on) spíw 'ice' (cf. píwat 'freeze it,' i.e., that which is frozen) sxíték<sup>w</sup> 'carving' (< xíték<sup>w</sup> 'carve,' i.e., that which is carved) sk<sup>w</sup>íx 'name' (cf. k<sup>w</sup>íxat 'name it,' i.e., that which is given as a name) sx<sup>w</sup>aýém 'story, myth' (< x<sup>w</sup>aýém 'tell a story, recite a myth')<sup>1</sup> syáθas 'story, account' (cf. yáθast 'tell him')

<sup>1</sup> Some use the word sx\*syém for any narrative, as JP did at times, but he also used it to refer to events involving Xéls and to the world before Xéls came (see §18.4.9, example [m]). I think

Or they may name the activities themselves, as in:

syá·ys 'work' (< yá·ys 'work') sŽónəq 'potlatch' (< Žónəq 'give a potlatch')

In one instance, the noun designates the substance the presence of which is indicated by an adjective: snás 'fat, grease' (< nás 'fat' as person, 'fatty' as food).

(3) Nominalizations of predicate heads in relative clauses with extracted oblique adjuncts (cf. §4.1.1.2), as in:

(a) tə s<sup>2</sup>áłtən [ni·n sxté<sup>2</sup>em] (DK)
tə s<sup>2</sup>áłtən ni<sup>2</sup> nə-s-xté<sup>2</sup>əm
ART food AUX my-NOM-prepare
'the food that I prepared'

In this relative clause (bracketed), the head verb ( $\check{x}t\acute{e}^{2}\partial m$  'prepare') is nominalized because the extracted word that the relative clause modifies ( $s^{2}\delta lt\partial n$ 'food') is the oblique object of this verb in a corresponding simple clause, such as:

 (b) ném θəł šté?em kw s²áłtən. (DK) ném θəł šté?em [?ə] kw s²áłtən go ADV prepare OBL ART food
 'Go now and make dinner.' ('Go anyway and prepare some food.')

(Here the object must be identified as oblique even with the particle  $2\partial$  missing, because the verb is intransitive.)

Nominalizations of this kind seem analogous to at least some of those of the second kind – see (2) above – in their semantic relationship to the verbs from which they are derived; for example,  $n \partial s \tilde{x} t \dot{e}^2 \partial m$  'what I prepared' is to  $\tilde{x} t \dot{e}^2 \partial m$  con 'I prepare' as  $n \partial s^2 \partial t \partial n$  'my food (what I eat)' is to ' $\partial t \partial t \partial n$  con 'I eat.' It seems likely that these two kinds of words are historically related. However, the nominalizations in relative clauses must have possessives, while nouns like  $s^2 \partial t \partial n$  'food' need not, and so nouns like  $s^2 \partial t \partial n$  cannot simply be identified as relative clauses.

It seems useful to distinguish s- 'word nominalizer' and s- 'clause nominalizer,' although they are no doubt historically identical. The latter can be identified as a particle in the sense of an element that relates to a construction larger than a word. (See §2.7.)

# 12.1.2. s-2 'resultative' (RES)

This prefix occurs with nearly all resultative forms of the verb (see §7.7). It also

that in more traditional usage, a story involving animal characters or the work of  $\check{X}\acute{e}ls$  was a  $s\check{x}^*s\check{y}\acute{e}n$  and an account of an historical event or traditional practice was a  $sy\acute{a}\theta s$ .

occurs with some forms of the noun that may be identified as resultative (see \$8.7).

It is distinct from  $s_{-1}$  'nominalizer' in that resultative forms do not take possessives (and so are verbs, not nouns, morphologically defined), while the principal function of  $s_{-}$  'nominalizer' is precisely to convert a verb into a form that may take a possessive, that is, to make a noun (morphologically defined) of it.

The identify in sound of  $s_{-2}$  'resultative' and  $s_{-1}$  'nominalizer' may be a recent development in Halkomelem. In neighbouring Salish languages, the counterpart of Halkomelem  $s_{-2}$  'resultative' has an initial glottal stop and schwa, as in Squamish 'as- (Kuipers 1967, 111).

#### 12.1.3. x<sup>w</sup>-, 'oblique relater' (OBREL)

This is a tentative identification. In a few instances, it appears that  $x^{w_{-1}}$  may relate a root and a lexical suffix as verb and instrument rather than verb and object, as in:

 $x^{w?} \delta w c \partial st$  'show him with the hands' (< ?tw- 'show,' -  $c \partial s$  'hand,' -t 'transitive')  $x^{w} k^{w} \ell^{2} c \partial s$  'drop from the hands' (cf.  $k^{w} \ell^{2} t$  'drop it, let it go')

in contrast to  $k^w \delta n \partial c \partial st$  'take his hand' (cf.  $k^w \delta n \partial t$  'take it'), which has no prefix  $x^w$ - and in which the root and lexical suffix are related as verb and object.

In some of the words where  $x^{w_{-1}}$  seems to have a locative sense, it may also mark an oblique relationship between root and suffix. Or, it may be that this function has developed from the locative sense of  $x^{w_{-3}}$  (cf. §12.4.8).

This interpretation of  $x^{w_{-_i}}$  as an 'oblique relater,' although based on only a few examples, casts  $x^{w_{-_i}}$  here in a role similar to the one it plays in the 'oblique nominalizer'  $\delta x^{w_{-_i}}$ .

#### **12.1.4.** $\delta x^{w}$ - 'oblique word nominalizer' (OBLNOM)

This is a compound of  $s_{-1}$  'word nominalizer' and  $x^{w_{-1}}$  'oblique relater.' The usual Musqueam form is  $\delta x^{w_{-}}$  but both CC and JP occasionally used  $\delta$ -, which is the usual Cowichan form, often correcting afterwards to  $\delta x^{w_{-}}$ . In a few words and nominalized clauses, CC used x- in place of  $\delta x^{w_{-}}$ . I cannot account for this variation. In JP's Musqueam, the sequence  $\partial T$ - 'your' followed by  $\delta x^{w_{-}}$  is realized as  $\partial \theta x^{w_{-}}$ . This compound prefix appears in four kinds of words:

(1) Nominalizations of predicate heads in relative clauses with extracted oblique nominal adjuncts other than objects. The extracted nominal adjuncts are loci, goals, and instruments (see §4.1.1.2) and the  $\delta x^{w}$ - can be glossed 'at which, to which, by which,' as in:

```
(a) Xa m ə ti?í sk vəléx [ni nəšx k vələxt]. (JP)
Xa m ə ti?í sk vəléx ni? nə-šx k vələx-t
BE3P CERT this gun AUX my-OBNOM-shoot-TR
'This gun is the one that I shot it with.'
```

In the relative clause (bracketed), the head verb  $k^w \delta l \delta xt$  'shoot it' is nominalized with  $\delta x^w$ -because the extracted  $sk^w \partial l \delta x$  'gun' is the instrument of a corresponding simple clause:

(b) ni cən k<sup>w</sup>ələx-t ?ə ti?í sk<sup>w</sup>əléx. AUX I shoot-TR OBL this gun 'I shot it with this gun.'

(2) Words that refer to instruments. These may be formed from verbs in the perfective or progressive aspects, without or with suffixes. The prefix has the sense 'means by which' and the root or stem designates the activity, as in:

 $\delta x^{w} \delta \delta \partial \phi'$  (<  $\delta \delta \partial \phi'$  (ie down')  $\delta x^{w}(h)\partial li$  (life' ('the "soul" that leaves the body at death,' <  $h\partial li$  'alive')  $\delta x^{w} \delta \delta \delta \phi'$  (<  $\gamma \delta \delta \delta \phi'$  (<  $\gamma \delta \delta \delta \phi'$  (<  $\gamma \delta \delta \delta \phi'$ )  $\delta x^{w} \delta \delta \delta \phi'$  (<  $\gamma \delta \delta \delta \phi'$ ) be working')  $\delta x^{w} \delta \delta \delta \phi \phi'$  (cf.  $\delta \delta \delta \delta \phi \phi'$ ); cf.  $\delta x \delta \delta t$  'open it up,' - $\delta h \delta \phi'$  (ground')  $\delta x^{w} \delta \delta \delta \phi'$  (cf.  $c \delta c \delta \delta \phi'$  (split it with a froe,' -els 'activity')

A few such words have the suffix *-ton* 'instrument,' which seems redundant:

 $\delta x^{w}k^{w}i^{2}t\partial n$  'ladder' (<  $k^{w}i^{2}$  'ascend')  $\delta x^{w}qa^{2}qa^{2}t\partial n$  'drinking tube' (<  $qa^{2}qa^{2}$  'drink')

Words of this kind seem to be related to those of the previous set (the predicate heads nominalized with  $\check{s}x^{w}$ -) in a way parallel to the relationship between the second and third set of words formed with *s*-(§12.1.1 above).

(3) Terms for persons. These include several kinship terms, which are given in §21. One of these is  $\delta x^w st l \partial \delta$  'grandparent's spouse, spouse's grandparent, grandparent's sibling's spouse' ( $< st l \partial \delta$  'grandparent, grandparent's sibling'). This word and others might suggest that the prefix  $\delta x^w$ - may be simply glossed 'in-law,' but a more accurate gloss might still be 'locus' or 'means,'  $\delta x^w st l \partial \delta$ being literally 'one who is in the place of grandparent' or 'one who functions as grandparent.'

Other terms for persons include:

 $\delta x^w sw \delta^{2}$  'master, owner' (<  $sw \delta^{2}$  'property,' <  $w \delta^{2}$  'own')  $\delta x^w si^{2} \delta m$  'boss, leader' (<  $si^{2} \delta m$  'rich person, upper-class person')  $\delta x^w n \delta^{2} \delta m$  'shaman' (cf.  $sn \delta^{2} \delta m$  'shaman's tutelary, "Indian doctor's power")

In these too, 'locus' or 'means' may be implied. The  $\delta x^w n e^{i\partial m}$  is both the possessor of and is possessed by his or her  $sne^{i\partial m}$ .

(4) Words in which a root and suffix are related obliquely. In words like the following, the  $\delta x^{w}$ - cannot be interpreted as 'by which' or 'at which.' It seems rather that the  $x^{w}$ - may serve to relate the root and suffix obliquely (cf. §12.1.3 above), and the *s*- then nominalizes this construction.

 $\delta x^{w}m \hat{u}sm \partial s \hat{d} \partial c$  'cow manure' (<  $m \hat{u}sm \partial s$  'cow,'  $-\delta \hat{d} \partial c$  'dung')  $\delta x^{w} \hat{c} \hat{k} \partial n \hat{d} \partial c$  'chicken manure' (<  $\tilde{c} \hat{k} \partial n$  'chicken')  $\delta x^{w} \hat{l} \partial m \hat{d} \hat{l} \hat{c}$  'bottle' (<  $l \hat{d} m$  'rum, liquor,'  $-\hat{e} l \partial$  'place for, container for')  $\delta x^{w} \hat{p} \hat{a} t \partial n \hat{e} \hat{l} \hat{a}$  'mast' (<  $p \hat{a} t \partial n$  'sail')

As with *s*- 'nominalizer,' we may distinguish  $\check{s}x^{w}$ - as 'oblique word nominalizer' and as 'oblique clause nominalizer' (see §4.3.3).

# 12.1.5. -*a*?*t* 'attributive' (ATT)

As mentioned in §3.8.3.4, when this is accompanied by  $s_{-1}$  'nominalizer' and sometimes internal change in the stem, it produces attributive or adjectival forms (§9.7).<sup>2</sup> Internal changes include the umlauting of /e/ to /a/ seen with several other suffixes but also a shift in stress. The latter appears in only two Musqueam forms given by JP and three Katzie forms given by SP. While the full form of the suffix is  $-a^2t$ , it is often heard as -at or -at. Some phrases with attributives formed by this combination of prefix and suffix are:

 $\delta x^{w} \delta lm \partial x^{w} a^{2} l t \delta m \partial x^{w}$  'Indian land' (<  $x^{w} \delta lm \partial x^{w}$  'Indian')  $\delta x^{w} \partial n(t \partial m a^{2}t s d^{w} \partial y(t \partial x \text{ 'White-style dance'}) (< x^{w} \partial n(t \partial m \text{ 'White'}))$ *smílka?ł šx<sup>w</sup>ləmélə* 'milk bottle' (< E. 'milk')  $syawana^{2}t \ sx^{w}q^{w}elawan$  'old people's ways of thinking' (< syawen'forebears.'  $< v \partial w \dot{e} \dot{n}$  'before') snáca?t q'''ext 'someone else's "claim" (e.g., wapato pond)' (< néc 'different')  $scilar m am a^{2}aq^{w}$  'birds of the heights (high-flying birds, such as eagles, buzzards, etc.)' (cf. *cícəł* 'above,' JP) słanayá?  $i š x^w q^w a ltan$  'woman's word, women's language' (< s l e n a y'woman,' JP)  $sté^{2}ex^{w}\partial ta^{2}t$   $\delta x^{w}q^{w}\partial tan$  'baby talk' (as using /k/ for /q/,  $< sté^{2}ex^{w}\partial t$ 'children,' JP) scówcował słéwoł 'pilchard' (lit. 'deep-sea herring,' < cówcow 'offing,' CC) scəwcéwał  $\dot{q}^{w}$ əlítəq 'offshore seagull' (a tern?) (JP) scəwcáwəł  $\dot{q}^{w} \dot{a} \dot{x}^{w} \partial q s$  ' $\dot{q}^{w} \dot{a} \dot{x}^{w} \partial q s$  (a type of  $s y \dot{a} w \partial n$ ) from the sea' (SP)  $sk^{w}\dot{\lambda}\dot{a}ma^{i}t$  ( $\dot{x}\dot{a}ca^{2}$ ) 'Coquitlam Lake' ( $\leq k^{w}ik^{w}a\dot{\lambda}am$  'Coquitlam') (JP)  $s\dot{q}^{w}a\cdot n\dot{\chi}il\partial t$  stálow 'Fraser River' (<  $\dot{q}^{w}\dot{a}\cdot n\dot{\chi}\partial t$ ,  $\dot{q}^{w}\dot{a}^{2}a\dot{n}\dot{\chi}\partial t$  'Kwantlen, i.e., river of the Kwantlen,' perhaps referring to the Fraser as it runs through Kwantlen territory only (SP)  $\delta x^{w}m\partial\theta k^{w}\partial\gamma\delta m\partial t$  sy $\delta\theta$  'Musqueam tradition, Musqueam inheritance' (SP) A few attributive forms with  $s - \dots -a^{2}t$  have become established in special senses, such as:

<sup>2</sup> I assume that the prefix s- is  $s_{-j}$  'nominalizer' rather than  $s_{-2}$  'resultative' because the Squamish counterpart *-ut* takes the Squamish nominalizer s-, not '200-, the Squamish counterpart of  $s_{-2}$  (Kuipers 1967, 128).

syíxa?ł 'southern or Puget Sound canoe type' (< yíx 'Puget Sound,' JP) sq<sup>w</sup>x<sup>w</sup>ám∂xa?ł 'Squamish canoe type' (< sq<sup>w</sup>x<sup>w</sup>ám∂x 'Squamish,' JP; DK gave sqx<sup>w</sup>ám∂x)

#### 12.1.6. -t 'stative' (STAT)

This is a rare suffix. It appears in a few adjectives and nouns:

- *łąćt* 'wide' (cf. *lq* 'across, other side,' *łéqəməx* 'flat country,' *łąćcəs* 'five,' *łéqət* 'lay it down')
- *płét* 'thick' (cf. Squamish *pəłc* 'thick-lipped' and comparative data in Kuipers [1967, 248])

*Åeqt* 'long'

 $\theta q \acute{e}t$  'standing upright, tree' (cf.  $\theta q \acute{e}n$  'be held upright,'  $\theta q \acute{e}nx$  'stand it up') táyt 'upstream' (cf. Cowichan táyəl 'go/come upstream')

- $t \delta y w \partial t$  'the "North" (i.e., Johnstone Strait and beyond)' (cf.  $x^w t \delta y \partial w \partial t$  'the "Northern" tribes,' probably related to the last; the northern end of the Strait of Georgia begins "upstream")
- *sžálačt* 'place where strong currents meet' (cf. *sžálač* 'current,' *žálčt* 'turn it')

It also appears in two anomalous resultative forms:

skwékwəlt 'hidden' (cf. kwélx 'hide it,' kwékwəlx 'be hiding it,' kwé'l 'hide,'
kwəkwí'l 'be hiding')

spépəlt 'skimmed' (cf. pélt 'skim it,' pépələt 'be skimming it')

It may also be identical with -t 'subordinate passive' (§10.8).

#### 12.2. VERBALIZING PREFIXES

There are seven prefixes that have lexical meanings and also serve to make verbs of nominal or adjectival roots or stems. They are: *c*- 'get, have, make, do,'  $x^{w}\partial$ - 'become,'  $tx^{w}-_{1}$  'buy,'  $t-_{1}$  'partake,'  $\dot{\lambda}$ - 'go to,' ct- 'die of,' and  $x^{w}-_{2}$  'move toward.' Of these, only *c*- and  $x^{w}\partial$ - require any discussion. For the others I have simply listed a few examples.

#### 12.2.1. c- 'get, have, make, do'

This is a kind of all-purpose verbalizer that converts the root or stem to which it is prefixed into an incorporated object. It is probably what remains of an old root, as suggested by the reduplicated form  $c \partial k^w$ , given below. It may also be the initial element in the first- and second-person subject particles (§14.2.1), perhaps once serving as an auxiliary to which the pronominal forms now serving as subordinate subjects were added.

Prefixed to nouns, *c*- provides the most usual way of expressing 'get' or 'have,' and in this sense it seems fully productive. Alone, it has a perfective sense and is usually translatable as 'get.' With an auxiliary and  $wa_{-2}$  'established,' it has an imperfective sense and is usually 'have,' as in (a) to (c).

- (a) c-wáč cən ce?. get-watch I FUT 'I'm going to get a watch.'
- (b) ?í ?ə čx<sup>w</sup> wə-c-wáč. here ROG you EST-get-watch 'Do you have a watch?'
- (c)?í-łcənwəcwáč.?í-əłcənwə-c-wáčhere-pastIEST-get-watch'I used to have a watch.'

With bound roots or stems requiring s-(s-nouns), c-replaces the s-, as in:

- (d) ?í cən cnáx wəł.
  'I have a canoe. I got a canoe.' (cf. snáx wəł 'canoe')
- (e) <sup>9</sup>í cən x<sup>w</sup>əcq<sup>w</sup>əméy.

'I have become the owner of a dog.' (cf. *sq<sup>w</sup>əméy* 'dog')

The sense of 'make' or 'do' may be restricted to contexts where this interpretation is more likely than 'have' or 'get.' Compare  $cn \delta x^w \partial t$  'have a canoe' in (d) with (f).

(f) scowét k «scnóx »ołs.

scəwet k<sup>w</sup> s-c-náx<sup>w</sup>əł-s clever ART NOM-make-canoe-3POS 'He is a good canoe-maker.' (lit. 'He is clever at his making a canoe.')

Other examples are:

 $cm\acute{a}n\acute{a}$  'have a baby' (<  $m\acute{a}n\acute{a}$  'child, offspring')  $cp\acute{e}^{2}\partial\theta$  'hunt black bear' (<  $sp\acute{e}^{2}\partial\theta$  'black bear')  $ct\acute{a}kt\emph{a}$  'get a doctor' (<  $t\acute{a}kt\emph{a}$  'medical doctor')  $ct\acute{e}tam$  'be doing what' (<  $st\acute{e}m$  'what')  $ct\acute{e}w\emph{a}n$  'make mats' (<  $st\acute{e}w\emph{a}n$  'mat')  $cp\acute{a}t\emph{a}n$  'sail (v.)' (<  $p\acute{a}t\emph{a}n$  'sail [n.]')  $c\acute{q}t\acute{l}\emph{a}$  'make dried fish' (<  $s\acute{q}t\acute{l}\emph{a}$  'winter supplies, especially of dried salmon')  $cy\acute{a}w\emph{a}n$  'sing a winter dance song' (<  $sy\acute{a}w\emph{a}n$  'winter dance song')  $cy\emph{a}w\acute{t}n$  'recite a spell' (<  $syawt\acute{n}$  'spell')

With kinship terms, the prefix c- and the suffix  $-\partial m$  'intransitive' give the sense 'call, address as,' as in:

```
cménəm 'call "father" (< mén 'father')
csìlé·m 'call "grandparent" (< sílə 'grandparent')
```

There are two interrogative verbs formed with c-:

*ctétəm* 'be doing what'

ctámət 'do what, experience what, suffer what'

The first is formed with the root of *stém* 'what.' It seems to appear in the progressive only. The second is probably also formed with the same root and a suffix that causes a change from /e/ to /a/, possibly the final  $-\partial t$  of the reflexive  $-\partial \partial t$ , which has the same effect; the term might more literally be interpreted 'what happens to oneself.'

The verb *ctámət* is perfective as given and has an anomalous progressive:  $cak^wstámat$  'be doing what, etc.' This form is interesting for two reasons. It shows *c*- undergoing reduplication (see §1.5.10 for /c/ and /k<sup>w</sup>/ in reduplicated forms), and thus it must be functioning as a root. And it shows that the underlying form of the perfective is //c-stamət//. This raises the question whether words like  $cn\delta x^w\partial t$  'have a canoe, make a canoe,'  $cp\ell^2\partial\theta$  'hunt black bear,' etc., are formed by prefixing *c*- to  $-n\delta x^w\partial t$ ,  $-p\ell^2\partial\theta$ , etc., or by prefixing *c*- to  $sn\delta x^w\partial t$ ,  $sp\ell^2\partial\theta$ , etc., and then deleting the *s*.

The prefix *c*- may also be identifiable with the initial element in the interrogative words  $c \delta s t \partial x^w$  'do what with it' and  $x^w c \delta l$  'go where.' (See §17.7 and 17.9.)

Verbs formed with c- prefixed to roots of other sorts (verbal, adjectival, and even adverbial) are active but intransitive. It is not clear how productive the prefix is with such roots, but the following have been recorded:

 $c\vec{k}^{w}\acute{e}c$  'get a look, see anything, happen to see' (<  $\vec{k}^{w}\acute{e}c$  'look'; cf.  $\vec{k}^{w}\acute{e}c\partial t$  'look at it,'  $\vec{k}^{w}\acute{o}cn\partial x^{w}$  'see it')

 $ck^{w} \delta n$  'get a hold, get anything' ( $\langle k^{w} \delta n$  'get taken'; cf.  $k^{w} \delta n \delta t$  'take it')  $ch \delta a q^{w}$  'catch a smell of something, get a whiff' (cf.  $h \delta a q^{w} \delta t$  'smell it')  $cm \delta k^{w}$  'find something, salvage something' (cf.  $m \delta k^{w} \delta t$  'find it, salvage it')  $cp \delta t$  'realize who someone is' (cf.  $p \delta tn \delta x^{w}$  'recognize him')  $cq \delta x$  'have much, have lots of something' ( $\langle q \delta x$  'much, many')  $c \lambda t^{2}$  'want something' ( $\langle \lambda t^{2}$  'expensive, important')  $c \lambda t^{2}$  'do something again, resume' ( $\langle \lambda t^{2}$  'again, also')  $cw \delta t^{2}$  'own' ( $\langle \sqrt{w} \delta t^{2}$  'own')

Verbs formed with c- and verbal or adjectival roots present a problem in the way some relative clauses are formed. Because these verbs are active but intransitive, they can have oblique objects, as in (g).

(g) chí? cən ?ə kwə nəšx wsé?t tə nəká: (AG)
c-hí? cən ?ə kwə nə-šx se?-t tə nə-ká: make-valuable I OBL ART my-OBLNOM-rise-TR ART my-car 'I want something to lift my car with.'

In a relative clause formed through the extraction of an oblique object, the predicate head is nominalized with s- (see §4.1.2). However, for two of these verbs with c-, it appears that in relative clauses the c- is deleted. This, at any rate, seems the simplest interpretation of forms like the following:

- (h) nəsmək<sup>w</sup>. (JP)
  nə-s-[c-]mək<sup>w</sup>
  my-NOM-make-find
  'I found it.' (lit. 'It is what I found.')
- (i) nəskí? k<sup>w</sup> səplíl. (AG) nə-s-[c-]kí? k<sup>w</sup> səplíl my-NOM-make-valuable ART bread
  'I want some bread.' (lit. 'Bread is what I want.')
- (j) niw smák<sup>w</sup>s <sup>2</sup>al k<sup>w</sup>θ<sup>3</sup> ni<sup>2</sup> há<sup>2</sup>k<sup>w</sup><sup>3</sup> ax<sup>3</sup>s. (JP) ni<sup>2</sup> w<sup>3</sup>-s-[c-]mák<sup>w</sup>-s <sup>2</sup>al k<sup>w</sup>θ<sup>3</sup> ni<sup>2</sup> há<sup>2</sup>k<sup>w</sup>-<sup>3</sup>x-<sup>3</sup>s
  AUX EST-NOM-make-find-3POS just ART AUX be.using-TR-3SUB 'He just picked up what he is using.' ('What he is using is just what he picked up.')
- (k)  $\lambda a k \circ \theta a t^{\theta} e^{1} a s n a s \lambda i^{?}$ . (JP 22)

 $\dot{\lambda}$ a k<sup>w</sup> $\theta$ ə  $\dot{t}^{\theta}\dot{e}\dot{l}$ ə-s nə-s-[c-] $\dot{\lambda}$ í? BE3P ART heart-3POS my-NOM-make-valuable 'It is its heart that I want.'

On the other hand, verbs formed with c- in the sense 'get' or 'have' prefixed to noun roots can be nominalized with s- without the loss of the c-, as in:

(l) θόγt con ce? to sála?ac sclélomct. (CC)
θόγ-t con ce? to sála?ac s-c-lélom-ct
be.made-TR I FUT ART mat NOM-have-house-our
'I'll fix the mats for our house.' (lit. 'I will fix the mats that will be what we will have as a house.')

(m)ni cən cməkw kwdə tewəł nəscla?dən. (CC) c-mák<sup>w</sup> ni? cən [?ə] k»θə łéwał I make-find clam shell AUX OBL ART nə-s-c-lá?θən my-NOM-have-plate 'I found a clamshell for my plate.' (lit. 'I found a clamshell that will be what I have as a plate.')

Prefixed to verbs with intransitive suffixes, c- gives a causative sense, as in:

```
ck^{w}\delta y\dot{x}\theta \partial t 'cause something' (< k^{w}\delta y\dot{x}\theta \partial t 'act, move oneself')
c\dot{q}\dot{e}\dot{l}\partial m 'convince oneself' (< \dot{q}\dot{e}l 'believe,' -\partial m 'intransitive')
c^{2}\partial y(l\partial m 'make good weather (magically)' (< ^{2}\partial y(l\partial m 'become clear')
```

Several terms for colours occur with c-. These are:

 $ck^{wi}m'$  red' (cf.  $sk^{w}m'$  red,'  $sk^{w}k^{w}m'$  reddish')  $c\dot{q}i\dot{x}$  'black' (cf.  $\dot{q}i\dot{x}\partial y\partial$  'Negro')  $cq^{w}\dot{a}y$  'green' (CC), 'yellow or green' (JP), 'pale' (cf.  $q^{w}\dot{a}y\partial l$  'turn pale')  $cx^{w}i\dot{k}^{w}$  (JP),  $\dot{c}x^{w}i\dot{q}^{w}$  (CC) 'grey'  $cp\dot{a}x^{w}$  'blond' (cf.  $sp\dot{a}x^{w}$  'tripe')

Here *c*- is the counterpart of the separate prefixes for colour terms in closely related Salishan languages, such as na- in Northern Straits. One colour term, paid 'white,' occurs without *c*-. There is a parallel to 'pale/turn pale' above in  $cq^wam\lambda^w$  'thin, skinny' (cf.  $q^wam\lambda^wal$  'become thin, get skinny').

# 12.2.2. $x = \partial - - x = i - become'$

This provides the most common way of expressing 'become, come to be' (but cf. also the reflexive  $-\theta \partial t$  in §10.5.1 and the verbal suffix  $-\partial l \sim -i l$  in §12.3.1). It appears as  $x^{w}\partial$ - in most contexts. In (a) it is prefixed to both an adjective and a verb:

(a) niw nem <sup>?</sup>al x<sup>w</sup>ək<sup>w</sup>ámk<sup>w</sup>əm k<sup>w</sup>snis x<sup>w</sup>ək<sup>w</sup>ənétəs ni <sup>?</sup>ə k<sup>w</sup>i sməstáyəx<sup>w</sup>. (JP 28) wə-nem ?aĺ x »ə- k »á m k »ə m k w ni s-ni-s AUX EST-go iust become-strong NOM-AUX-3POS ART x wə-k wəné-t-əs ?a κ<sup>w</sup>i ni s-məstáyəx<sup>w</sup> become-be.taken(DUR)-TR-3TR be.there OBL DEM NOM-person 'It [his vision] just gets so strong that he comes to have it in his "system.""

In (b), it is prefixed to a resultative form:

```
(b) ni x^wəsəmné?. (DK)
```

ni x<sup>w</sup>ə-s-mə́nə

AUX become-RES-child

'He has a kid.' (lit. 'He has become childed.')

And in (c), (d), and (e), it is prefixed to nominalized forms or nouns:

 (c) x<sup>w</sup>ə<sup>?</sup>əθwé<sup>?</sup>ələp. (JP 17) x<sup>w</sup>ə<sup>-</sup>?əT-s-wé<sup>?</sup>-ələp become-your-NOM-own-2PL 'It has become yours.'

(d) k<sup>w</sup>θe? ni x<sup>w</sup>∂-s?óly∂-s cí·tm∂x<sup>w</sup> (JP 28)
 that AUX become-vision-3POS great.horned.owl
 'that owl that had become his vision'

(e) ni ct łǎíləxstəx w kwsÅas xwəčífct. (AG)
ni ct łǎíləx-stəx kw s-Åa-s xwə-číf-ct
AUX we stand-CAUS ART NOM-BE3P-3POS become-chief-our
'We elected him chief.' (lit. 'We made him stand up so that he was the one who became our chief.')

This suffix appears as  $x^{w}i$ -before personal and interrogative words, as in (f), (g), and (h).

- (f) héỷ łe, x<sup>w</sup>inśwəs neṁ x<sup>w</sup>nəćíwən. (JP 10) héỷ łe x<sup>w</sup>i-nśwə-s neṁ x<sup>w</sup>nəćíwən go.ahead PER become-you-PERM go investigate 'Well, now you go investigate.' ('It's your turn to ...,' lit. 'Let it become you to ...')
- (g) x<sup>w</sup>i<sup>2</sup> i<sup>2</sup> θn θ n em xésəl. (JP 10) x<sup>w</sup>i<sup>-2</sup> inθ n em xésəl become-I go face.danger
   'It's up to me to go face the danger.'
- (h) héỷ x<sup>w</sup>istém (CC) héỷ x<sup>w</sup>i-stém go.ahead become-what 'What's next?'

# 12.2.3. tx<sup>w</sup>-1 'buy'

- (a) ném čx<sup>w</sup> tx<sup>w</sup>səplíl.
   ném čx<sup>w</sup> tx<sup>w</sup>-səplíl
   go you buy-bread
   'Go buy bread.'
- (b) tx "wéč cən ce?.
   tx "-wéč cən ce?
   buy-watch I FUT
   'I'll buy a watch.'
- (c) ni cən tx<sup>w</sup>ká: ni cən tx<sup>w</sup>-ká: AUX I buy-car 'I bought a car.'

# 12.2.4. *t-1* 'partake, experience'

łqá? 'drink water' (< qá? 'water'; cf. qá?qa? 'drink')
łtí 'drink tea' (< tí 'tea')
lčém 'eat "jam" (< čém 'jam, any canned fruit')</pre>

*łčémt* 'make it into ''jam''' (DK) *łpáŹam* 'smoke (v.)' (cf. *spáŹam* 'smoke [n.]') *łk<sup>w</sup>ak<sup>w</sup>í<sup>2</sup>x<sup>w</sup>* 'be chewing gum' (< k<sup>w</sup>í<sup>2</sup>x<sup>w</sup> 'pitch') *łk<sup>w</sup>álk<sup>w</sup>ala* 'be having a bellyache' (< k<sup>w</sup>ála 'belly') *łqálqala* ~ *cłqála* ~ *cłqá<sup>2</sup>qala* 'thirsty' (cf. *qá<sup>2</sup>* 'water') *łx<sup>w</sup>áx<sup>w</sup>asam* 'eat whipped soapberries' (cf. *sx<sup>w</sup>ésam* 'soapberries')

The last may be a progressive in the pattern of  $q^w \acute{e}l$  'speak,'  $q^w \acute{a}q^w \acute{o}l$  'be speaking.'

# 12.2.5. X- 'go to'

 $\dot{X}t\dot{a}w\partial n$  'go to town' (<  $t\dot{a}w\partial n$  'town')  $\dot{X}n\dot{e}c\partial wtx^w$  'visit, call on a neighbour' (<  $n\dot{e}c\partial wtx^w$  'next door, next room')

# 12.2.6. ct- 'die of'

*cłlém* 'die of drink' (< *lém* 'liquor,' < E. 'rum') *cłą* $anx^{w}(lam$  'die of overeating' (<  $qánax^{w}$  'gullet,' -*ílam* 'move toward'; cf. *sqánax<sup>w</sup>* 'greedy') *chł* $^{\theta} á xtan$  'die of poison' (<  $t^{\theta} á xtan$  'poison')

# 12.2.7. $x^{w}$ -2 'move toward'

With stems referring to location,  $x^{w_{-2}}$  seems to function as a verbalizer implying motion toward, as in:

 $x^w c \acute{a} k^w$  'go far' (<  $c \acute{a} k^w$  'far')  $x^w c \acute{a} l \partial q^w$  'go up into the woods' (<  $c \acute{a} l \partial q^w$  'the woods, back country')  $x^w c \acute{c} c \partial l$  'go high' (<  $c \acute{c} c \partial l$  'above')  $x^w c \acute{o} w c \partial w$  'go out from shore' (<  $c \acute{o} w c \partial w$  'the offing')

It may co-occur with the suffix  $-\partial l \sim -il \partial m$  (§12.3.1), which has a similar function, as in:

 $x^{w} \vec{\lambda} p i l$  'go straight down into the ground' ( $\langle \vec{\lambda} p i l$  'go downward')  $x^{w} i w a l$  'move upstream, move toward the middle of the house' ( $\langle h i w a l$ 'move closer to the fire')

 $x^{w}c\acute{e}l$  'go where' ( $<\sqrt{c}$ , cf.  $c\acute{a}stax^{w}$  'do what with it')

# 12.3. VERBAL SUFFIXES

There are four of these. The first is not always verbalizing since the root may itself be verbal.

# 12.3.1. $-\partial l \sim -i l$ 'move toward, turn to, become'

This is very productive and can be followed by transitive and intransitive suffixes.

 $2\hat{\lambda}q\partial l$  'go/come out' (cf.  $s^2\hat{\ell}\hat{\lambda}q$  'outside')

 $^{2}\partial \vec{X}qilt$  'kick out (a sponger, spouse, etc.)' (cf.  $^{2}\partial \vec{X}q\partial l$ )

híwəl 'move toward the fire' (cf. híwx 'bring forward')

x"íwəl 'go/come upstream' (probably //x"-híwəl//, cf. ?əhíw 'upstream')

yəléwəl 'get past' (cf. yəléw 'past, passed,' yəléwx 'pass him')

 $\vec{\lambda} pil$  'move downward, sink, go/come downstairs' (prog.  $\vec{\lambda} \delta p \partial l$ ,  $< \vec{\lambda} \delta p$ 

'deep'; cf.  $\vec{X} i \vec{X} \partial p$  'down')

 $\vec{X}p(lx$  'sink it' (cf.  $\vec{X}p(l)$ , also  $\vec{X}p(l) = tax^w$  'sink it, abandon it')

 $q^{w} \dot{a} y \partial l$  'turn pale' (cf.  $cq^{w} \dot{a} y$  'yellow, green, pale')

*qəlqí·l* 'be ruined' (< *qál* 'bad,' probably //qəlqálə1//)

qəlqí·lt 'ruin it' (cf. qəlqí·l)

téwəl 'get light, dawn' (cf. téwət 'light it, throw a light on it')

The compound suffix -*il* om is -*il* plus -om 'intransitive':

*həqíləm* 'go under (a bed, house, etc.)' (cf. *sí*?*q* 'underneath,' *híqət* 'put it under')

 $w \partial \dot{q}^w i l \partial m$  'go/come downstream' (<  $w \partial \dot{q}^w$  'drift, float with the current')

təlíləm 'move back from the fire, away from the centre of the house' (cf.

*lélx* 'move it back from the fire, move it from the centre of the house')  $x^w ck^w il_{\partial m}$  'go far away' (<  $c \dot{a} k^w$  'far')

 $x\dot{c}il\partial m$  'go into the woods' (cf.  $xi\dot{c}\partial t$  'put it into the woods, thrust it in,'  $sxix\partial c$  'the woods, the bush')

 $k^{w} \partial t x^{w} (l \partial m \text{ 'go/come inside'} (cf. k^{w} t \partial x^{w} \text{ 'get inside'})$ 

 $q^{w} \partial h(l \partial m \text{ 'go through' (cf. } q^{w} \acute{e} \text{ 'get through')})$ 

 $^{2} \partial y i l \partial m$  'become clear (the weather)' (<  $^{2} \partial y$  'good')

 $\dot{\lambda} \partial x^w la^2 as (l \partial m \text{ 'go to nothing, become worthless'} (< \dot{\lambda} \partial x^w la^2 as \text{ 'of no value'})$  $qa^2 q \partial^2 (l \partial m \text{ 'get the drinking habit'} (< q a^2 q a^2 \text{ 'drink'})$ 

# 12.3.2. -*í*·*m* 'die from'

This is possibly composed of some element of the shape  $//-\partial R-//$  and the suffix *-om* 'intransitive.' It has not been recorded in many words. It may require a prefix *c*- or *s*-.

(a) ni<sup>9</sup> ct<sup>θ</sup>ayə<sup>9</sup>í·m. (CC)
 ni<sup>9</sup> c-t<sup>θ</sup>áyə-í·m

AUX get-fur.seal-die.from

'They died hunting fur seals.'

(b) ni<sup>9</sup> skaití·m. (CC)

ni? s-Åqíł-í·m

AUX NOM[?]-give.birth-die.from

'She died in childbirth.'

(c) cǎənǎí·m. (JP 4) c-ǎíləǎ-í·m get[?]-make.war-die.from "fallen in battle"

The last example is something Raven says in a myth. Presumably the /n/ for /l/ is old-fashioned Musqueam.

# 12.3.3. -á·ł 'travel by'

There is a word  $2\dot{a}\cdot t$  'go aboard,' and so words in which  $-\dot{a}\cdot t$  appears as the second element might be considered compounds. However, compounds are rare or non-existent (in the present analysis) and besides, the meaning of  $-\dot{a}\cdot t$  is not identical with that of  $2\dot{a}\cdot t$ . Therefore  $-\dot{a}\cdot t$  is listed here as a suffix. There is also a possible lexical suffix  $-a\cdot t$  'canoe, vehicle' (§13.6, 32), for which I have only two clear attestations. In the sense 'travel by,'  $-a\cdot t$  appears to be fully productive.

- (a) ném ct wékən-à·ł. (CC 7) go we wagon-travel.by 'We'll go by wagon.'
- (b) ni ct nem t<sup>0</sup>fikt<sup>0</sup>ik-à·ł. (JP) AUX we go wagon-travel.by 'We went by wagon.'
- (c) báysikl-à·ł čx<sup>w</sup> (JP)
   bicycle-travel.by you
   'Use your bike.'

In (a), AC used a term for 'wagon' borrowed from English, while in (b), JP used one from Chinook Jargon.

snáxa<sup>w</sup>łà·ł 'go by canoe' (JP)
lilú·tà·ł 'go by train' (JP) (> lilút > 'railroad')

# 12.3.4. - $\dot{a}m\partial x \sim -\dot{a}\cdot m\partial x$ 'look, appear'

The form  $-ám \partial x$  is usual, as in:

scəkwəláməx 'how does it look?' (< scékwəl 'how?')
?əyaməx 'pretty, handsome' (< ?ay' 'good')
qəlaməx 'ugly' (< qal 'bad')
nəcaməx 'look different' (< néc 'different')</pre>

The form  $-\dot{a}\cdot m\partial x$  was recorded in one word:  $t\partial^2 \dot{a}\cdot m\partial x$  'resemble' (<  $te^2$ - 'be like'; cf.  $ste^2\dot{e}$  'like'). But perhaps the long vowel is a product of the root having a form  $te^2e$ . This word is used as in (a), (b), and (c).

| tə?á∙m;                          | эx                                                                                       | [?ə]                                                                                                                                                        | k <sup>™</sup> ə                                                                                                      | məstáyə                                                                                                                                                                                               | x <sup>w</sup> . (J                                                                                                                                                                                       | P)                                                                                                                                                                                                       |                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                 |
|----------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| resembl                          | e                                                                                        | OBL                                                                                                                                                         | ART                                                                                                                   | person                                                                                                                                                                                                |                                                                                                                                                                                                           |                                                                                                                                                                                                          |                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                 |
| 'It look                         | s like                                                                                   | a perso                                                                                                                                                     | n.'                                                                                                                   |                                                                                                                                                                                                       |                                                                                                                                                                                                           |                                                                                                                                                                                                          |                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                 |
| tə <sup>9</sup> á∙ma<br>resembl  | əx<br>le                                                                                 | ?∂<br>ROG                                                                                                                                                   | k <sup>w</sup> ə<br>then                                                                                              | t∍wÅa<br>he                                                                                                                                                                                           | [?ə<br>0bi                                                                                                                                                                                                | ] Ž<br>. AI                                                                                                                                                                                              | ?é<br>rt m                                                                                                                                                                                                                                                 | ∙nθə.<br>e                                                                                                                                                                                                                                                                                                                                      |
| 'Does h                          | e lool                                                                                   | k like m                                                                                                                                                    | ne?'                                                                                                                  |                                                                                                                                                                                                       |                                                                                                                                                                                                           |                                                                                                                                                                                                          |                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                 |
| ni<br><sup>AUX</sup><br>'Do I lo | ?ə<br>ROG<br>ok lił                                                                      | cən<br>I<br>xe him?                                                                                                                                         | k <sup>w</sup> ə<br>then                                                                                              | tə⁰á∙m<br>resemb                                                                                                                                                                                      | əx<br>le                                                                                                                                                                                                  | [?ə]<br>OBL                                                                                                                                                                                              | təwี่ส์<br>him                                                                                                                                                                                                                                             | ι.                                                                                                                                                                                                                                                                                                                                              |
|                                  | tə?á·ma<br>resembl<br>'It look<br>tə?á·ma<br>resembl<br>'Does h<br>ni<br>AUX<br>'Do I lo | tə <sup>9</sup> á·məx<br>resemble<br>'It looks like<br>tə <sup>9</sup> á·məx<br>resemble<br>'Does he lool<br>ni <sup>9</sup> ə<br>AUX ROG<br>'Do I look lik | tə?á·məx[?ə]resembleOBL'It looks like a persotə?á·məx?əresembleROG'Does he look like mni?əAUXROGI'Do I look like him? | tə? $\dot{a} \cdot m$ əx[?ə] $\dot{k}$ "əresembleOBLART'It looks like a person.'tə? $\dot{a} \cdot m$ əx?ək "əresembleROGthen'Does he look like me?'ni?əni?əcənk "əAUXROGIthen'Do I look like him?''' | tə?á·məx[?ə]kwəməstáyəresembleOBLARTperson'It looks like a person.'tə?á·məx?əkwətə?á·məx?əkwətəwằaresembleROGthenhe'Does he look like me?'ni?əcənni?əcənkwətə?á·mAUXROGIthenresemble'Do I look like him?' | tə?á·məx[?ə]kwəməstáyəxw. (JresembleOBLARTperson'It looks like a person.'tə?á·məx?əkwətə?á·məx?əkwətəwÅa[?əresembleROGthenheOBL'Does he look like me?'ni?əcənkwəAUXROGIthenresemble'Do I look like him?' | tə?á·məx[?ə] $\vec{k}$ wəməstáyəx w. (JP)resembleOBLARTperson'It looks like a person.'tə?á·məx?ək wətə?á·məx?ək wətəwÅa[?ə]tə?á·məx?ək wətəwÅa[?ə]'Does he look like me?'ni?əcənk wəni?əcənk wətə?á·məx[?ə]AUXROGIthenresembleOBL'Do I look like him?''III | tə?á·məx [?ə] k <sup>w</sup> ə məstáyəx <sup>w</sup> . (JP)<br>resemble OBL ART person<br>'It looks like a person.'<br>tə?á·məx ?ə k <sup>w</sup> ə təwÅa [?ə] Å ?é<br>resemble ROG then he OBL ART mu<br>'Does he look like me?'<br>ni ?ə cən k <sup>w</sup> ə tə?á·məx [?ə] təwÅa<br>AUX ROG I then resemble OBL him<br>'Do I look like him?' |

#### 12.4. MISCELLANEOUS LEXICAL PREFIXES

There are a few prefixes with lexical meaning and less clear grammatical status than those presented under §12.2. They have not been recorded as co-occurring.

#### 12.4.1. ma- 'come'

This is identifiable with the verb  $2\partial \dot{m}i$  'come' and in some contexts might easily be interpreted as  $2\partial \dot{m}i$  used as an auxiliary. However, this interpretation is shown to be false by (a).

| (a) | ) səw mətécəls tə léləmct. (CC 21) |                   |     |           |  |
|-----|------------------------------------|-------------------|-----|-----------|--|
|     | s-wə-mə-técəl-s                    | [ <sup>9</sup> ə] | tə  | léləm-ct  |  |
|     | NOM-EST-come-arrive.here-3POS      | OBL               | ART | house-our |  |
|     | 'Then she came to our house.'      |                   |     |           |  |

Here the fact that the possessive -s follows  $m \partial t \dot{e} c \partial l$  (rather than  $m \partial$ ) shows that  $m \partial t \dot{a} c \partial l$  is treated here as a single word, and  $m \partial$ - as a prefix. This might also be seen as a case of compounding.

This prefix has been recorded in:

*mətécəl* 'come here' (< *técəl* 'arrive here') *məq́a*·*n* $\theta$ *i* 'come back here' (< *q́a*·*n* $\theta$ *i* 'return') *məq̂*<sup>w</sup>*im* 'come ashore' (< *q̂*<sup>w</sup>*im* 'disembark') *mək*<sup>w</sup>*ini* 'bring it' (< *k*<sup>w</sup>*ini* 'take it') *məták*<sup>w</sup>*stəx*<sup>w</sup> 'bring him home' (< *ták*<sup>w</sup>*stəx*<sup>w</sup> 'take him home')

# 12.4.2. tom-'time of, season for'

t 
amtém 'when' (cf. stém 'what')  $t amlíle^{2}$  'salmonberry season' (<  $líle^{2}$  'salmonberry')  $t am \check{x} \check{a} \check{y} \check{x}$  'winter' (<  $\check{x} \check{a} \check{y} \check{x}$  'cold')

The prefix  $t \partial n - \sim t \partial n$  in the following is probably a variant of the last:

tənmílə ~ təmmílə 'winter dancing season' (< mílə 'dance possessed with
 one's "song" [syśwən]') (CC)</pre>

 $t \partial n q^w \partial t q^w \partial t x \partial n$  'period of continuous fog' (cf.  $s q^w \partial t x \partial n$  'fog,'  $q^w i q^w t i m x \partial n$  'drizzle') (JP)

For others, see §22.2.7.

# 12.4.3. tan- 'from'

 $t \partial n y i x$  'a wet south wind' ( $\langle \sqrt{y} i x$  'Puget Sound'; cf.  $x^w \partial y i x \partial l$  'Lushootseed speakers,'  $\delta x^w \partial y i x \partial l a^2 l$  'southern, of Puget Sound')

təncáləq<sup>w</sup> 'a squally west wind, "bush wind" from over Vancouver Island' (< cáləq<sup>w</sup> 'inland, "the bush")

- tənsXé·m 'a kind of "song" [syśwən] that comes from crying' (< Xé·m 'cry,' s- 'nominalizer')
- *tənsi<sup>2</sup>ém* 'upper-class, from high-class people' (< *si<sup>2</sup>ém* 'rich, upper-class person')

*tənstéxəm* 'of lower-class origin' (< *stéxəm* 'lower-class group')

 $t \partial n s \dot{c} \partial l \dot{e} x^w$  'people from  $s \dot{c} \partial l e x^w$  'p.n., the upper end of the Musqueam village'  $t \partial n s^2 \partial \dot{x} q \dot{a} l w \partial t$  'outsider, someone from outside the family' ( $\leq s^2 \partial \dot{x} q \dot{a} l w \partial t$  'outside')

The prefix *təl*-, a variant of the last, is so far identified in the following words only. It may appear only before a glottal stop.

talanaca 'from where?' (< ?ánaca 'where?')
tali (JP), tali? (CC) 'from' (< ?i 'be here')
talanac 'a rainy southeast wind' (< s?ánac 'bay')
\$x\*talá·wmax" 'where one is from' (< \$x\*\*- 'where/why,' -a\*\*- 'connector,'
-max" 'place, people')</pre>

# 12.4.4. tx<sup>w</sup>-2 '(something) belonging to'

This seems to be used only with wét 'who' and with personal names.

- (a) tx<sup>w</sup>-wét ti<sup>9</sup>í. belonging.to-who this 'Whose is this?'
- (b) tx<sup>w</sup>-John tθe?.
   belonging.to-John that 'That's John's.'
- (c) Xa cos k<sup>w</sup>θo tx<sup>w</sup>-Arnold. (AG)
   BE3P QUOT that belonging.to-Arnold 'It's said to be the one that's Arnold's.'

Compare (c) with (d):

(d) Xa co swe? o X Arnold. (AG)
 BE3P QUOT property OBL ART Arnold
 'It is said to be Arnold's.'

# 12.4.5. tx<sup>w</sup>-<sub>3</sub> 'remaining, only'

 $tx^{w}k^{w}(n \text{ 'how many more?'} (< k^{w}(n \text{ 'how many?'})$  $tx^{w}n\delta ca^{2} \text{ 'one more'} (< n\delta ca^{2} \text{ 'one'})$  $tx^{w}\delta y \text{ 'be only, be an exception, except for' (< h\delta y \text{ 'be stipulated'})}$  $w \delta tx^{w}st^{\theta}\delta m^{2}al \text{ 'nothing but bones, skinny'} (< st^{\theta}\delta m \text{ 'bone,' }w\delta - ... ^{2}al \text{ 'only'})$ 

# 12.4.6. *t*-<sub>2</sub> 'portion'

 $l\theta \delta l a q$  'some, a part' (cf.  $\theta \delta l a q \theta a t$  'separate')  $l s \delta \dot{q}$  'half (portion of meat, a dollar, etc.)' ( $\leq s \delta \dot{q}$  'be split, be torn')  $l t \delta \dot{q}^w$  'half (a stick, bone, etc.)' ( $\leq t \delta \dot{q}^w$  'break')  $l s p \ell l a w$  'members of the Sparrow [surname] family'

# 12.4.7. *čł-* 'fellow, co-'

This can be prefixed to nouns and a variety of verb forms.

*čłx<sup>w</sup>ólməx<sup>w</sup>* 'fellow villagers, fellow Indians' (< *x<sup>w</sup>ólməx<sup>w</sup>* 'village, tribe, Indian')

 $\dot{c}t\dot{x}\dot{e}\cdot m$  'deceased child's spouse's parent, child's deceased spouse's parent' (i.e., 'co-parent-in-law after the death of a linking child,' lit. 'co-weeper,'  $\leq \check{x}\dot{e}\cdot m$  'cry')

 $\dot{c}t^{2}\partial\dot{q}^{w}t^{2}t\partial\dot{l}$  'siblings and cousins' (<  $^{2}\partial\dot{q}^{w}t^{2}t\partial\dot{l}$  'be related as siblings or cousins')

 $\dot{c}lk$  "*íntəl*' opponent in a fight' (< k"*íntəl*' fight')

 $\dot{c}tst\dot{e}^{2}ex^{w}\partial t$  'the other children' (lit. 'co-kids,' <  $st\dot{e}^{2}ex^{w}\partial t$  'children')

 $\dot{c}tw\acute{e}t$  'someone you share with' (lit. 'co-owner,' <  $w\acute{e}$ ' 'own' or  $w\acute{e}t$  'who?')  $\dot{c}tm\acute{a}nt\partial l$  'half-sibling with the same father' (<  $m\acute{e}n$  'father,'  $-t\partial l$  'mutual')  $t\partial si^{2}\acute{e}m$  ni  $\dot{c}tq^{w}dtals$  'the big man he was talking with' (<  $q^{w}iq^{w}dtals$  'be talking together,' -s 'his'; lit. 'the rich man who was his interlocutor') (JP 3)

# 12.4.8. x<sup>w</sup>-, 'inward'

A prefixed  $x^{w}$ - occurs in a large number of words. In many, its meaning and functions are, as yet, quite unclear. The label 'inward' is meant as a catch-all for occurrences of  $x^{w}$ - that have not been segregated under the labels 'move toward' and 'oblique relater.' I am assuming that all occurrences can ultimately be placed in one of these three categories. It may be that there are more than

three homophonous prefixes, or, on the other hand, there may be only one that has developed a wide range of meaning and function.

The notions inward, inhering, possessing seem to be conveyed by  $x^{w}$ - in a number of sorts of words. Some of these are:

(1) Active verbs, such as:

 $x^w c \delta l \partial w t$  'turn it inside out' (<  $c \delta l \partial w t$  'turn it over, as a featherbed')  $x^w \delta x \delta x \delta t$  'cut it open, break it open' (<  $\delta x \delta t$  'open it, as a can,' - $\delta x \partial t$  'side')

(2) With lexical suffixes. The prefix  $x^w$ - occurs often, though not always, with -*íwən* 'behind/inside,' -*qən* 'throat/voice/speech,' and -*łél*- 'food'; it occurs fairly often with -*as* in the sense 'face,' but not 'moon, dollar'; and it occurs occasionally with others.

It is possible that some of these suffixes simply attract or require an accompanying  $x^{w}$ , but it seems also possible to identify this  $x^{w}$ - with a locative 'in' or 'on' or a possessive 'with' or 'having,' although it is not possible to say why it should occur in one word and not in another. Compare the following with  $x^{w}$ -:

```
x^{w}\theta \hat{a} \cdot s 'with a big face'

x^{w}\theta \partial h(w\partial n) 'with a big behind'

x^{w}\theta \hat{i} q\partial n 'loud' (lit. 'big-voiced')

x^{w}\theta \hat{e} \cdot q 'with a big penis'

x^{w}\hat{c} \hat{q}^{w} \hat{i} w\partial n 'get pricked in the behind'

x^{w}ma \hat{\lambda} \hat{t} \hat{e} \hat{l} \partial m 'repay food'
```

with the following without  $x^{w}$ -:

 $\theta \dot{a} \cdot q^w$  'big headed'  $\theta \dot{i} q s \partial n$  'with a big nose'  $\theta \partial h \dot{a} y \partial \theta \partial n$  'with a big mouth'  $\theta \dot{i}^2 \dot{a} l \partial s$  'with big eyes'  $\theta \dot{e} \cdot q$  'with a big penis' (JP)  $\dot{c} \dot{q}^w \dot{a} \cdot l c \partial$  'get pricked in the belly'  $\dot{c} \dot{q}^w (n \partial s$  'get pricked in the chest'  $m a \dot{\lambda} \dot{e} l \dot{l}^\theta e^2$  'repay wealth'

(3) In words concerned with opening and shutting, such as:

 $x^{w}t\delta q$  'shut (intr., as a door does)' (<  $t\delta q$  'be cut off, be taken by surprise')  $x^{w}taq\acute{e}t$  (CC),  $taq\acute{e}t$  (JP) 'shut it (anything vertical, as a door)'  $x^{w}w\acute{e}t^{\partial}t$  'lock it' (<  $w\acute{e}t^{\partial}t$  'pry it')  $x^{w}laklit$  'lock it' (< laklit 'key' [< F. *le clef* via CJ])  $x^{w}m\acute{e}^{2}x$  'unlock it' (<  $m\acute{e}^{2}x$  'take it off, remove it')  $x^{w}y\acute{a}\check{x}^{w}at$  'unlock it' (<  $y\acute{a}\check{x}^{w}at$  'untie it, free it') (4) In words referring to mental states, such as:

 $x^{w}\dot{\lambda}i^{2}$  'stingy with money' ( $<\dot{\lambda}i^{2}$  'expensive, difficult')  $x^{w}l\dot{l}\partial q$  'generous' ( $< l\dot{l}\partial q$  'cheap, easy')  $x^{w}s\dot{\iota}^{2}si^{2}$  'afraid, cowardly' ( $< s\dot{\iota}^{2}si^{2}$  'fear')  $x^{w}q\dot{\sigma}lw\partial t$  'mean, tough person' ( $< q\dot{\sigma}l$  'bad,' -w $\partial t$  'vessel')

(5) In place names.  $x^{w}$ - occurs in a number of place names and in a few words for kinds of places, all formed from the names of plants or animals and often with the suffix *-am* 'intransitive.' The sense of  $x^{w}$ - here seems to be 'where there is/are –' or 'having –.' For example:

x<sup>w</sup>m *i*θk<sup>w</sup>∂y<sup>i</sup>∂m 'p.n., Musqueam' (< m*i*θk<sup>w</sup>∂y' 'a species of plant')
 x<sup>w</sup>q*im∂k<sup>w</sup>∂m* 'p.n., a spring at Musqueam' (< sq*im∂k<sup>w</sup>* 'octopus,' once there in a legend)

 $x^{ws} \dot{a} \cdot \dot{q}^{w}$  'p.n., a stream near Pitt Lake' (<  $s \dot{a} \cdot \dot{q}^{w}$  'cow parsnip')

*x<sup>w</sup>mécənəł p* 'p.n., a site near St. Mungo's Cannery' (< *mécənəł p* 'black haw')

 $x^{w}\dot{c}\dot{a}\dot{l}\dot{a}\dot{k}^{w}\dot{a}^{2}\dot{a}\dot{m}$  'a place where many skunk cabbages ( $\dot{c}\dot{a}^{2}\dot{k}^{w}a^{2}$ ) grow'

 $x^{w}w\partial lwi \cdot l\partial lp$  'a place where nothing but tules  $(wi \cdot l)$  grow'

(6) In words for states of water or weather. These are formed, like those of the last set, with  $-\partial m$ .

 $x^{w^{9}} \delta y_{\partial m}$  'clear (water)' (<  ${}^{2} \delta y$  'good')  $x^{w} q \delta l_{\partial m}$  'murky' (<  $q \delta l$  'bad')  $x^{w} \check{x}^{w} i \cdot m$  'be flowing fast' (<  $\check{x}^{w} \delta m //x^{w} \cdot \check{x}^{w} \delta m \cdot \vartheta m //$  'fast')  $x^{w^{9}} \delta n \partial x^{w} \partial m$  'be slack water' (<  ${}^{2} \delta n \partial x^{w}$  'stop')  $x^{w^{2}} \delta \theta t \dot{e} \cdot n \partial m$  'become cloudy' (cf.  $\check{s} x^{w^{2}} \dot{e} \theta t \partial n$  'cloud')

(7) With the meaning 'vulva.' This appears to be the only instance of a bodypart prefix, the counterpart of the various suffixes referring to body parts. It occurs in:

 $x^{w}\theta i$  'big vulva' (<  $\theta i$  'big')

 $x^{w?}\delta \dot{y}$  'clean vulva' (<  $^{2}\delta \dot{y}$  'good, clean')

 $x^{w}k^{w}$  an ét 'put one's hand on a woman's vulva'

 $\delta x^w q^w in$  'woman's pubic hair' (cf.  $q^w in \partial ws$  'body hair,'  $q^w in eq$  'male pubic hair')

and possibly in  $x^w m \delta \dot{q}^w$  'miscarry' (<  $m \delta \dot{q}^w$  'burst').

In this set of words, does  $x^{w}$ - have a meaning such as 'inward' used euphemistically or is there a missing lexical suffix that was deleted euphemistically?

(8) A prefix  $n \partial x^{w}$ - apparently identical with  $x^{w}$ - 'inward' has been recorded in one word only:  $n \partial x^{w} m \partial \theta k^{w} \partial y \partial m t \dot{w} s$  'a person originally from Musqueam but now living somewhere else' ( $\langle x^{w} m \dot{\partial} \theta k^{w} \partial y \partial m t \rangle$  Musqueam, '-*iws* 'body').

# 12.4.9. wa-3 'suddenly, with a burst'

Perhaps this should be classed with the aspectual prefixes. (For AG and perhaps others for whom  $w_{\partial_2}$  is 2u- or  $2u^2$ - this is still  $w_{\partial_2}$ .) Its use is illustrated in (a) to (e).

- (a) nəwł wəpśk<sup>w</sup> tə spák ams. (CC)
  ni wəł wə-pśk<sup>w</sup> tə
  AUX already suddenly-burst.forth.in.small.particles ART spák am-s
  smoke-3POS
  'His smoke has appeared. His smoke has suddenly burst out.'
- (b) ni wəpɨkw kwθə pá·kw. (AG)
  ni wə-pɨkw kwθə
  AUX suddenly-burst.forth.in.small.particles ART
  pá·kw
  mature.Chinook.salmon
  'The big old spring salmon splashed.'

The root  $\sqrt{p\delta k^w}$ , which appears in the last two examples, has not been recorded as such, but it appears in  $pk^w \delta m$  'billow forth, splash forth,'  $pk^w \delta t$  'scatter it about (fine stuff like down),'  $spk^w \delta m$  'dust,' and  $sp\delta k^w$  'flour.'

- (c) ni cən cód kwsis wəkwtáxw. (DK)
  ni cən cód kw s-ni-s wə-kwtáxw
  AUX I be.surprised ART NOM-AUX-3POS suddenly-enter
  'I was surprised when she walked in [suddenly entered].'
- (d) x "ənsnét i niwəł wəhíq. (AG) x "ən-snét '>əŷ ni wəł-wə-híq still-night and AUX already-suddenly-shove.off
  'It was still night when he shoved off [suddenly launched his canoe].'
- (e) ni wək<sup>w</sup>éc k<sup>w</sup>θə ni k<sup>w</sup>í. (AG)
  ni wə-k<sup>w</sup>éc k<sup>w</sup>θə ni <sup>9</sup> k<sup>w</sup>í
  AUX suddenly-scream ART be.there OBL there
  'There was a sudden scream over there.' (lit. 'An invisible one who is over there suddenly screamed.')

# 12.4.10. $h \partial \dot{n} - \sim h \partial \dot{m} - \sim \partial \partial \dot{m} - \sim h \partial n -$ 'come'

This has been recorded with only a few roots. The variation does not seem to be significant. (CC used  $h \partial n'$ -. JP recognized that some people say  $h \partial n'$ - but he generally said  $h \partial m'$ - or  $2 \partial m'$ -. AG used  $h \partial n$ -.)

həncéw 'come down to the shore' (pl. həncéləw, cf. cécəw 'shore') (also figuratively 'return from the dead')
həncáləqw 'come back to life' (cf. < cécəw 'shore,' -əs 'face'?)</li>
həncáləq<sup>w</sup> 'get up to the house from the shore' (< cáləq<sup>w</sup> 'inland') (AG)
hənàmət 'get home' (< ?ámət 'sit,' x<sup>w</sup>?ámət 'be home') (AG)
hənx<sup>w</sup>é 'get down (as from up in a tree)' (cf. x<sup>w</sup>ét 'lower it') (AG)

# **13** Non-Personal Affixes 4: Lexical Suffixes

It is possible to isolate nearly 150 elements that may be lexical suffixes or connectives plus lexical suffixes. About 100 occur in enough words to be clearly identifiable in meaning. A good many of these are very productive. Others have been recorded in only one or two words each and so are less easily glossed. Some can be identified only as possible lexical suffixes.

#### 13.1. FORMS

The lexical suffixes themselves (disregarding the connectives) have a variety of forms: -C, -VC, -VCC, -VCVC, -VCCC. Glottalized obstruents occur but seem less common in lexical suffixes than in roots.

The lexical suffixes are often joined to the root by a connective element that consists of a vowel  $-/\partial/$ , /e/, or /a/- plus a resonant -/1/, /w/, or /y/, with or without glottalization. With some lexical suffixes, the connective seems to add little or nothing to the meaning; for example,  $-tx^w$ ,  $-\partial tx^w$ ,  $-\partial wtx^w$ , and  $-ewtx^w$  all seem to mean 'house.' But with others the connective appears to give a more restricted meaning, for example,  $-e\check{x}\partial n$  'side, perimeter' and  $-\partial te\check{x}\partial n$  'arm.'

Several lexical suffixes have alternate forms with a final  $-\partial n$ , such as  $-a\theta$ ,  $-a\theta\partial n$ , etc., 'mouth,' etc.; -ic,  $-ic\partial n$ , etc. 'back,' etc.; and  $-\partial qs$ ,  $-\partial qs\partial n$ , etc. 'nose,' etc.

A few have long vowels, which are probably the result of the loss of the first of two identical resonants (see §1.5.6); thus,  $-\acute{e}\cdot n$  'side' may be  $-\partial n$  'ear' plus a final  $-\partial n$ .

Most lexical suffixes (without or with connectives) have unstressed and stressed variants, often with vowels of different quality. Where the stress falls can depend on the nature of the root, the presence of other suffixes, or the inflection of the whole word for plural, progressive, resultative, and so on (see §7). For a few lexical suffixes, a final /ł/ appears as /l/ before *-t* 'transitive,' an alternation not observed elsewhere in the language, and for a few others a final /n/ is lost before *-t* 'transitive.'

These variations have been an obstacle to identifying some lexical suffixes. The problem is further complicated by several cases of complete or partial homophony, where two or three suffixes are homophones in all or some of their variant forms.

The lexical suffixes usually follow bare roots or roots with  $s_{-1}$  'nominalizer.' Instances of a lexical suffix following a grammatical suffix exist but are rare. Instances of one lexical suffix following another are more common (see below). Roots to which lexical suffixes are added appear with different vowel grades, depending in the shape of the root and the type of suffix (see §7). Moreover, with certain suffixes, such as -as 'face,' and  $-aq^w$  'head,' an /e/ vowel in the root will appear as /a/ (see §1.5.12).

Some lexical suffixes, when used to refer to body parts and food, require (varying with the speaker) the prefix  $x^{w}$ -, 'inward' (§12.4.8) before the root.

#### 13.2. MEANINGS

The lexical suffixes that can be clearly identified as to meaning denote body parts (head, hand, foot, etc.), some artifacts (house, canoe, paddle), some natural phenomena (vegetation, water, fish), a few statuses of persons (child, spouse), a few shapes (round object, long object, container) and parts (surface, edge, end, base), and a few other things less easily categorized.

The "somatic" suffixes (Kuipers 1967, 120), that is, those denoting body parts, form the largest and perhaps most complete set. But there seems to be no basis for setting them off from the others or giving them any kind of primacy. Some of the somatic suffixes can also have non-somatic reference, such as -aqsan 'nose, point of land'; -as 'face, round object, moon, dollar'; -qan 'throat, speech, meal, opening facing upward'; -dlas 'eye, net mesh.' In each of these, the somatic meaning may be historically and conceptually primary, but this is conjecture. Other somatic suffixes seem to be formed from suffixes of more general meaning by connectives that restrict to narrower, somatic meaning, as in -dlexan 'arm' from -dexan 'side,'  $-dya\partial an$  'mouth' from  $-d\theta an$  'margin,' and possibly  $-almax^w$  'breast, milk' from  $-max^w$  'land, people.' Therefore it seems better not to prejudge the lexical suffixes as to primary meaning.<sup>1</sup>

A number of lexical suffixes appear commonly with numeral roots as numerical classifiers (see §19.2), but these too do not form an exclusive set. Most of them occur with other roots or stems. Nor does it seem possible to subcategorize the lexical suffixes on the basis of what grammatical affixes they may take. Some have not been recorded with *-t* 'transitive' or *-om* 'intransitive,' but the

<sup>1</sup> Hinkson (1999) has provided strong circumstantial evidence for the primacy of the body-part meanings of three lexical suffixes and their semantic extensions each in several directions.

reason may lie simply in the nature of things rather than in grammar.<sup>2</sup> Accordingly, I have not tried to sort out the lexical suffixes into types but treat them as a single set.

Many lexical suffixes are broad or abstract in meaning and without synonymous words in the lexicon. For example, there is no single word corresponding in meaning to *-mat* 'kind, part'; *-mon* 'residue'; *-os* 'round object, face, moon, dollar'; *-ét* 'times'; *-qon* 'throat, voice, speech, meal, opening facing upward'; *-wot* 'canoe, vessel, container'; or *-ot p* 'plant, tree, vegetation.'

Other lexical suffixes, however, are matched by words with the same or closely similar meanings. Compare the suffixes in the left column with the words on the right:

| $-aq^w$ 'head'                      | sžáýas 'head'          |
|-------------------------------------|------------------------|
| -cəs 'hand, bough'                  | céləx 'hand'           |
| - <i>xən</i> 'foot, leg'            | sžáňa 'foot, leg'      |
| -éwtx <sup>w</sup> 'house'          | léləm 'house'          |
| - <i>əlwe<sup>2</sup>s</i> 'paddle' | <i>sģáməl</i> 'paddle' |

#### **13.3. LEXICAL SUFFIXES AND RELATED WORDS**

There are also a few instances where the lexical suffix and the word with the same meaning are clearly related. In one instance, the word seems to be no more than the suffix with an initial glottal stop and the prefix  $s_{-i}$  'nominalizer':

-*ínəs* 'chest, beach' s<sup>2</sup>*íləs* (CC), s<sup>2</sup>*ínəs* (JP) 'chest'

In another, a verb root consists of the lexical suffix with the addition of an initial glottal stop and the loss of a final glottal stop. Compare:

| $-i\dot{t}^{\theta}e^{\gamma}$ 'robe, clothing' | $2it^{\theta} \partial m$ 'get dressed'              |
|-------------------------------------------------|------------------------------------------------------|
|                                                 | $2it^{\theta} ax$ 'put it on'                        |
|                                                 | s <sup>2</sup> íť <sup>θ</sup> am 'blanket, clothes' |
|                                                 |                                                      |

In a few words, the suffix appears on a single consonant, *m*-, *t*-, *y*-,  $\theta$ -:

| <i>áqsən</i> 'nose, point' | <i>máqsən</i> 'nose'       |
|----------------------------|----------------------------|
| alacan 'testicles'         | mécən 'testicle'           |
| <i>á psəm</i> 'neck, nape' | <i>tápsəm</i> 'neck, nape' |

<sup>2</sup> Leslie (1979, 68-85) separates what I am calling "lexical suffixes" into two groups, "non-inflective suffixes" and "lexical suffixes," asserting that the first group cannot take suffixes of the voice system while the second can. He further divides the second group according to whether they take *-am* only, *-t*, *-tal*, and *-nax*<sup>w</sup> only, or all four. However, I have some counter-examples, which I have tried to include in the data presented, and I believe that it is too soon to make any claims of this sort, for Musqueam at any rate.

| - $\ell l \partial x^{w} \theta \partial l$ 'tongue' | $t \neq x^w \theta \geq t$ 'tongue'        |
|------------------------------------------------------|--------------------------------------------|
| - $m \partial x^w$ 'land, people'                    | támax <sup>w</sup> 'land, earth'           |
| -ánəs 'tooth'                                        | <i>yánəs</i> 'tooth'                       |
| $-a\theta \partial n$ 'margin'                       | $\theta \hat{a} \theta \partial n$ 'mouth' |

I have assumed that the initial consonant of the word was simply a dummy root that serves no purpose other than to provide a base for the suffix. However, Kinkade (1998) has argued that these words,  $m \delta q s \partial n$ , etc., may be the sources of the suffixes. That is, the suffixes were historically produced by lopping off the initial consonant of the free form. A possible example of this process can be seen in  $t \delta m x c \partial n$  'fine-tooth comb,' which may be composed of the root of  $t \delta m \partial t$  'pick it/them' and  $m \delta x c \partial n$  'louse' minus the initial part.

In a few more words, there is a root,  ${}^{2}\theta - {}^{2}e\theta$ - or  ${}^{2}\partial l - {}^{2}el$ -, with or without *s*- 'nominalizer,' that seems to limit the suffix to a somatic or spatial reference.

| - <i>əs</i> 'round object, etc.' | $s^2 \dot{a} \theta \partial s$ 'face'                                  |
|----------------------------------|-------------------------------------------------------------------------|
| <i>-nəc</i> 'butt, bay'          | s <sup>2</sup> áθnəc 'bay'                                              |
| <i>-qən</i> 'throat, etc.'       | $\delta x^{w^2} \delta \theta \partial q \partial n$ 'inside of throat' |
| <i>-qən</i> 'front, etc.'        | ?éθəqən 'front'                                                         |
| - <i>əqsən</i> 'nose, point'     | s <sup>2</sup> álqsən 'point of land'                                   |
| -éxən 'side'                     | <i>?íləxən</i> 'end of house, road, etc.'                               |

In these words, it seems to me that the initial element is more likely a root with some (now unclear) semantic content.

Another possible example of the last type is  $s^{2i} \cdot lq \partial y'$  'two-headed serpent' (which probably has the underlying form  $//s^{2i} l \partial lq \partial y'$ ); cf. Cowichan and Northern Straits  $s^{2i} n \partial lq \partial y'$ ), which seems to be composed of an element  $^{2il}$  of unknown meaning and  $^{2i} \partial lq \partial y'$  'snake.'

The word  $s\check{x}\delta\check{x}\delta$ tnet 'Sunday, week' ( $\langle \check{x}\ell\check{x}e^2\check{x}e^2$  'holy') and a few others are formed with a linking element -*t*-, which may be a reduction of -*a*<sup>2</sup>*t* 'attributive,' and *nét* 'become night.' These terms, which are clearly of modern origin, might be identified as compounds (Gerdts and Hinkson 1996, 3) or as containing a new lexical suffix. I have listed them as containing a lexical suffix, number 34 in §13.6 (see also §22.2.3).

#### **13.4. RELATIONS BETWEEN SUFFIX AND ROOT**

Roots and lexical suffixes can be combined to relate to each other in different ways (roots and suffixes playing different grammatical roles) and to produce words that have different grammatical functions (as verbs, adjectives, or nouns). We can categorize these complex words by their type of root, the grammatical relation between root and suffix, and the grammatical functions that they have.

# 13.4.1. With Verbal Roots

A lexical suffix can be related to a verbal root as patient, locus, or instrument, that is, in a role played in syntax by an adjunct, or as head with the root as modifier.

*13.4.1.1. Root and Suffix Related as Verb and Patient, Locus, or Instrument* This is probably the largest class of words with lexical suffixes. They are all verbs.

Words of this type may occur without any additional suffixes, being then necessarily intransitive:

*táčcəs* 'get one's hand cut, cut one's hand (accidentally)' (< *tíč-* 'cut,' -*cəs* 'hand')

 $l\partial k^{w}\dot{a}\cdot y\partial \partial n$  'get one's mouth broken' (<  $l\partial k^{w}$  'get broken,' - $\dot{a}y\partial \partial n$  'mouth')  $\dot{c}\dot{a}lc\partial s$  'switch sides when paddling' (<  $\dot{c}\dot{a}l$ - 'switch,' - $c\partial s$  'hand')

səwqʻánəp 'seek land to settle on' (< sáwq' 'seek,' -ənəp 'land')

 $tx^{w\delta lc\partial p}$  'bring down firewood' (<  $tax^{w}$  'move down to shore,'  $-\delta lc\partial p$  'fire, firewood')

Words of this type can also appear with suffixes of the voice system (see \$10), except for the reflexive suffixes  $-\theta \partial t$  and so on.

táčcəst 'cut his hand' (< tíć- 'cut,' -cəs 'hand,' -t 'tr.')</li>
táčcəsnəx<sup>w</sup> 'cut him on the hand (accidentally)' (< -nəx<sup>w</sup> 'tr. limited control')
k<sup>w</sup>ək<sup>w</sup>ónəcəstəl 'be shaking hands' (cf. k<sup>w</sup>ónət 'take it,' k<sup>w</sup>ónəcəs 'take his hand,' -təl 'reciprocal')

Following lexical suffixes, a reflexive (or middle voice) sense is conveyed by *-om* 'intransitive.' Compare the following pairs:

 $x^{wt} t^{\theta} \check{x}^{w} \dot{a} st$  'wash his [another's] face' (<  $x^{w}$ - 'inward,'  $t^{\theta} \check{s} \check{x}^{w}$  'get washed,' -*ás* 'face')

 $x^{w}t^{\theta}\check{x}^{w}\acute{a}s\partial m$  'wash one's [own] face'

*tí caq<sup>w</sup>t* 'cut one's hair' (< *tí c*- 'cut,' -*aq<sup>w</sup>* 'head') *tí caq<sup>w</sup>əm* 'get a haircut'

 $\theta \partial y \dot{e}^{2} t t$  'make his bed' (cf.  $\theta \dot{\partial} y t$  'fix it,'  $-\dot{e}^{2} t$  'bedding')  $\theta \partial y \dot{e}^{2} t \partial m$  'make one's bed'

```
x^w?al xté·lt 'go get food for him' (cf. ?ál a x at 'go get it,' -tél- 'food')
x^w?al xtél am 'go get food for oneself'
```

There are also forms with  $-\partial m$  'intransitive' where this suffix does not appear to have a reflexive meaning, as in  $x\partial t\partial m\partial yt\partial m$  'take care of a [anyone's] child' (cf.  $x\dot{a}^{2}t\partial m\partial t$  'look after him, take care of it,'  $-\dot{\delta}y\partial t$  'child').

In most words of this type, the lexical suffix probably relates to the verb as patient. In a word like  $x^{w^{2}} \partial l \tilde{x} t \dot{e}^{2} l t$  'go get food for him,' there seems to be no question that the suffix denotes a patient and also functions as an applicative (see §10.4), allowing the recipient to appear as the grammatical object.

In words containing a suffix referring to a body part, this suffix might be seen as locative, for example, in  $x^w t^\theta \tilde{x}^w \delta t$  as literally 'wash him on the face,'  $l\delta cc \partial s$  as literally 'get cut on the hand.' Gerdts (1981) has argued, however, that these body-part suffixes denote patients and allow their possessors to appear as grammatical objects.

In the following word, it seems that the suffix denotes a patient and allows the instrument to appear as the grammatical object:  $x^w l \partial \vec{m} l \dot{e} l t$  'throw it into the fire' (<  $l \dot{a} \vec{m}$  'get hit by something thrown'; cf.  $l \dot{a} \vec{m} \partial t$  'hit him with something thrown,'  $-l \dot{e} l$ - 'fire').

On the other hand, the suffix pretty clearly denotes a locus rather than a patient in the following:

 $^{2}\partial t\partial n \dot{a}\theta \partial n$  'eat along the way' (<  $^{2}\partial t\partial n$  'eat,'  $-\dot{a}\theta \partial n$  'margin, edge')

 $x^w \dot{q} \partial w \hat{l} t$  'go with him in a canoe' (<  $x^{w}$ - 'inward,'  $\dot{q} \hat{a}^2$  'accompany,' - $w \hat{l} \hat{l} \sim -w \partial \hat{l}$  'canoe')

 $k^{w} \acute{e}^{2} c \partial s$  'drop it from the hand' (cf.  $k^{w} \acute{e}^{2} t$  'drop it, let it go,' -  $c \partial s$  'hand')

And the suffix denotes an instrument in the following examples:

 $\vec{k}^{w} c \dot{a} l \partial s$  'see with one's own eyes' (<  $\vec{k}^{w} \dot{e} c$  'see,' - $\dot{a} l \partial s$  'eye')

 $x^{w?} \delta w c \partial st$  'show him with the hands how something is done' (<  $x^{w}$ 'oblique,' 2iw- 'understand'; cf.  $2iw \partial s$  'show, guide,' - $c \partial s$  'hand')

 $k^{w} \partial x^{w} q s \partial n \partial m$  'peck (as a woodpecker)' (cf.  $k^{w} dx^{w} \partial t$  'knock on it,'  $-\partial q s \partial n$  'nose')

yəpèpəźxénəm' 'be feeling along with the feet (in shallow water looking for cockles)' (< yə- 'along,' péź- 'feel,' -xən 'foot')</p>

#### 13.4.1.2. Root and Suffix Related as Modifier and Head

These are words that function as nouns, the root specifying the kind, by function or source, and the suffix designating the class (cf. §13.3.3.2 below):

?itatéwtx<sup>w</sup> 'hotel' (< ?ítat 'sleep,' -éwtx<sup>w</sup> 'house') qiqéwtx<sup>w</sup> 'jail' (< qíq' 'be bound, be put to jail') téyawał 'racing canoe' (< téy 'race,' -awał 'canoe, vessel') xilaxáwał 'battleship' (< xílax 'make war') łałak<sup>w</sup>ówał 'flying machine, airplane' (< łáłak<sup>w</sup> 'be flying') k<sup>w</sup>ayak<sup>w</sup>ólxan 'trolling line' (< k<sup>w</sup>úyak<sup>w</sup> 'troll,' -alxan 'line')

This interpretation also seems possible for words with three other suffixes:

(1) The large number of words for instruments formed with the suffix  $-t \partial n \sim -t \partial n' \sim -t \partial n$ , such as:

 $\dot{p} \delta \dot{t}^{\theta} t \partial n$  'needle' (<  $\dot{p} \delta \dot{t}^{\theta}$ 'sew')  $t \delta \dot{k}^{w} t \partial n$  'gaff hook' (cf.  $t (\dot{k}^{w} \partial t$  'hook it')  $s \delta \dot{l} s \partial \dot{t} \partial n$  'spindle' (cf.  $s \delta l \partial t$  'spin it')  $\dot{q}^{w} \dot{e}^{2} t \partial n$  'broom' (cf.  $\dot{q}^{w} \dot{e}^{2} t$  'sweep it')  $\dot{\lambda} \partial \dot{x}^{w} t \dot{e} n$  'blanket' (<  $\dot{\lambda} \delta \ddot{x}^{w}$  'get covered')  $\dot{t}^{\theta} \partial \ddot{x}^{w} t \dot{e} n$  'cleansing rite' (<  $\dot{t}^{\theta} \ddot{x}^{w}$  'get washed')

(2) Words formed with the suffix  $-m\partial n \sim -m\partial n$  'instrument' (probably not productive):

 $\vec{k}^w q^w \delta m \partial n$  'ax' (<  $\vec{k}^w \delta q^w$  'get hit with a clubbing motion')  $t \delta t \delta m \partial n$  'herring rake' (cf.  $t t \delta t$  'flick it,'  $t \delta t \delta m$  'catch herring with a rake')

3) Words formed with the suffix *-mən* ~ *-eṁ̀ən* 'residue':

lək<sup>w</sup>émən 'something broken (as the broken end of a stick)' (< lók<sup>w</sup> 'break, get broken') tócmən 'sawdust' (< tíc- 'cut')</pre>

# 13.4.2. With Adjective Roots

It seems that an adjective root and a lexical suffix can be related only as modifier and head, as in:

 $\begin{array}{l} \theta i qs \partial n \text{ 'big-nosed'} (< \theta i \text{ 'big,'} - \partial qs \partial n \text{ 'nose'}) \\ \vec{\lambda} \partial q \partial t \delta p s \partial m \text{ 'long-necked'} (< \vec{\lambda} e q t \text{ 'long,'} - \delta p s \partial m \text{ 'neck'}) \\ \vec{\lambda} \partial q t e l c \text{ 'long-haired'} (< \vec{\lambda} e q t \text{ 'long,'} - e l c \text{ 'hair, line'}) \\ \vec{\lambda} \partial q t e l c \text{ 'long-haired'} (< \vec{\lambda} e q t \text{ 'long,'} - e l c \text{ 'hair, line'}) \\ \vec{\mu} q \delta l q \partial n \text{ 'mountain goat'} (< \vec{p} \delta q \text{ 'white,'} - \partial l q \partial n \text{ 'pelt'}) \\ \vec{x} e^2 \tilde{x} e^2 \delta y \partial t \text{ 'abnormal baby'} (< \tilde{x} e^2 \tilde{x} e^2 \text{ 'holy, forbidden,'} - \partial y \partial t \text{ 'child'}) \end{array}$ 

Some of these words, like the last two above, are nominal in meaning, but most words of this type are adjectival in meaning. For example,  $\theta i q s \partial n$  must be glossed 'big-nosed' or 'having a big nose' rather than simply 'big nose.' Compare (a) and (b):

- (a) θíqsən tθe?. (AG)
   θi-əqsən tθe?
   big-nose that
   'He has a big nose.'
- (b) θí máqsan tθe?. (AG) big nose that 'That's a big nose.'

These words may be used predicatively, as is  $\theta i q s \partial n$  above, or attributively, as in (c) and (d).

- (c) k<sup>w</sup>θə θíqsən sq<sup>w</sup>əméý (AG) k<sup>w</sup>θə θi-əqsən sq<sup>w</sup>əméý ART big-nose dog
   'the big-nosed dog'
- (d) θə kaqtélč qémay (AG)
   θa keqt-élč qémay
   ART long-hair girl
   'the long-haired girl'

Some lexical suffixes can form adjectives that can modify nouns of similar but narrower meaning, as in (e) to (g).

- (e) k<sup>w</sup>θə p'dáls smé·nt (JP, AG) k<sup>w</sup>θə p'áq'-als smé·nt ART white-spherical.object rock 'the white rock'
- (f) tə sθé·łp θqét (AG)
   tə s-θi-əłp θqét
   ART NOM-big-vegetation tree
   'the big tree'
- (g) <sup>9</sup> əx<sup>w</sup>ín o ws čík on (JP)
   <sup>9</sup> əx<sup>w</sup>ín i ws čík on small-body/bird chicken
   'small chicken'

Words formed from numeral roots (see \$19) have the same relationship to the nouns they modify, as in (h) and (i).

- (h) tə nɨcɨqən sítən (JP 4)
   tə nɨcɨa?-əqəň sítən
   ART one-container basket
   'one basket'
- (i) tə ixim-ələ si si si yanı (JP 4)
   ART six-person women
   'six women'

However, it seems that a suffix cannot appear with a noun of closely similar meaning; for example, one cannot say  $**t\partial \vec{\lambda}\partial qt \ell \vec{l} \vec{c} m \ell q \partial n$  'long(-haired) hair.' These restrictions need to be explored.

# 13.4.3. With Noun Roots

A noun root or stem (as a root with s- 'nominalizer') and a lexical suffix can be related in one of two opposite ways, as head and modifier (the root or
stem designating the genus and the suffix the species) or as modifier and head (vice versa).

## 13.4.3.1. Related as Head and Modifier

When a suffix denoting a body part follows a noun root or stem, it can have the sense of a noun possessor or the first member of an English compound, as in:

 $st^{\theta}amaq^{w}$  'skull' (<  $st^{\theta}am$  'bone,'  $-aq^{w}$  'head,' i.e., 'head-bones')  $st^{\theta}amaq^{w}$  'skull' (<  $st^{\theta}am$  'bone,'  $-aq^{w}$  'head,' i.e., 'head-bones')  $q^{w}inaya\thetaan$  'bones of the hand, hand-bones' (< -cas 'hand')  $q^{w}inaya\thetaan$  'beard' (<  $q^{w}in$  'hair of face and body,'  $-aya\thetaan$  'mouth')  $staq^{w}alas$  'gums' (<  $stiq^{w}$  'flesh,' -alas 'tooth')  $snasalmax^{w}$  'butter' (< snas 'fat, grease,'  $-almax^{w}$  'breast, milk,' i.e., 'milk-grease')

## 13.4.3.2. Related as Modifier and Head

In the great majority of words with noun roots or stems, however, including some in which body-part suffixes are used in non-somatic senses, the root or stem is the modifier, specifying the kind, and the suffix is the head, identifying the class.

The root may identify the source or material, as it does in:

 $\dot{t}^{\theta} \hat{t} w \partial q a q^{w}$  'elder-head, i.e., an arrow with a blunt head of elderberry wood' (<  $\dot{t}^{\theta} \hat{t} w \partial q$  'red elderberry,' - $a q^{w}$  'head')

 $\dot{t}\dot{a}^{2}x^{w}c_{2}s$  'grand fir bough' ( $\langle \dot{t}\dot{a}^{2}x^{w}$  'grand fir,' - $c_{2}s$  'hand, twig')  $musmas\delta lmax^{w}$  'cow's milk' ( $\langle musmas$  'cow,' - $\delta lmax^{w}$  'breast, milk')  $sala^{2}ac\dot{e}wtx^{w}$  'mat house' ( $\langle s\dot{a}la^{2}ac$  'house-mat,' - $\dot{e}wtx^{w}$  'house')  $?\dot{e}\dot{s}x^{w}\dot{e}\dot{m}an$  'seal-oil' ( $\langle ?\dot{e}\dot{s}x^{w}$  'seal,' - $\dot{e}\dot{m}an$  'liquid')  $wi\cdot l\dot{e}^{2}l$  'tule mat' ( $\langle wi\cdot l$  'tule,' - $\dot{e}^{2}l$  'bedding')

Or the root may identify the purpose or end to which the head is dedicated:

 $sa\check{x}^{w} \partial l\acute{e}wtx^{w}$  'barn' (<  $sa\check{x}^{w}\partial l$  'grass, hay,' - $\acute{e}wtx^{w}$  'house')  $qew\theta\acute{e}wtx^{w}$  'potato cellar' (<  $sq\acute{e}w\theta$  'potato,' - $\acute{e}wtx^{w}$  'house')  $s\thetae\theta\partial y \partial n\acute{e}l\partial$  'blood vessel' (<  $s\theta\acute{e}\theta\partial y \partial n$  'blood,' - $\acute{e}l\partial$  'place for, container')  $put\acute{e}lwe^{2}s$  'oar' (<  $p\acute{u}t$  'boat,' - $\acute{e}lwe^{2}s$  'paddle')

All the examples given so far in this section are words that function as nouns. However, there may also be words of this type that function as adjectives. The proper way of saying porpoise harpoon, JP said, is (or once was)  $k^* \acute{a} n \acute{t} a t c a$  $s^2 \acute{a} n am$ , lit. 'porpoise ( $k^* \acute{a} \cdot n \acute{t}$ )-water (-a t c a) shaft ( $s^2 \acute{a} n am$ ).' This example, although isolated in the present data, suggests the possibility that some of the lexical suffixes can (or once could) be used to produce modifying forms of nouns. I have not recorded any lexical suffix for 'harpoon' or 'spear,' and I believe that in traditional usage a simple noun could not modify another noun as in English "porpoise harpoon," although some present-day speakers use such constructions. As others have suggested (e.g., T. Thompson in Haeberlin 1974, 220), the capacity to create forms with lexical suffixes is one that diminished early in the decline in fluency in the Coast Salish languages, and I suspect that this is one basis for statements we hear about how in previous generations people spoke "the old language."

## **13.5. COMPOUNDING OF SUFFIXES**

Lexical suffixes can appear following other lexical suffixes, as in:

 $i^{\theta} \check{x} \check{a}^{2} c^{2} pm \partial n$  'residue of a fire' ( $\langle i^{\theta} \check{x}$  'be burnt up,' - $c^{2} p$  'fire,' - $m \partial n$  'residue')  $\partial \check{x} \partial \hat{l} q \partial n \partial \hat{l} w \partial t$  'buckskin shirt' ( $\langle \partial \check{x} \check{x} \partial \hat{l} q \partial n$  'buckskin,' composed of  $\partial i \check{x}$ -

'scrape' and -*álqan* 'hide,' plus -*álwat* 'garment')

- $n \partial w \dot{a} q^w c \partial s t \partial n$  'thimble' (<  $n \partial w$  'enter,'  $-a q^w$  'head,'  $-c \partial s$  'hand,'  $-t \partial n$  'instrument,' lit. 'instrument for inserting the head of the hand')
- *sk*<sup>w</sup>*acàstanálas* 'eyeglasses' (< *sk*<sup>w</sup>*acástan* 'mirror, window,' composed of *s*-'nominalizer,' *k*<sup>w</sup>*éc* 'look,' *-as* 'face,' *-tan* 'instrument,' thus lit. 'instrument for seeing the face,' plus *-álas* 'eye')

Two words, unusual in that lexical suffixes follow grammatical suffixes, are:

- $s^{2} \partial t^{\theta} \dot{a} \dot{m} \partial s$  'shawl' (<  $s^{2} \dot{t}^{t} \dot{\theta} \partial m$  'clothing,' composed of *s* 'nominalizer,' the lexical suffix  $\dot{t}^{t} \dot{\theta}$  'clothing, blanket,' and  $-\partial m$  'intransitive' plus  $-\partial s$  'face')
- *qəqàma*<sup>2</sup>*stéyət* 'be nursing a child' (< *qáma*<sup>2</sup>*stəx*<sup>w</sup> 'nurse it,' composed of the root of *sqáma*<sup>2</sup> 'breast, milk' and *-stəx*<sup>w</sup> 'causative' plus *-əyət* 'child')

## 13.6. THE INVENTORY

What follows is a list of well-identified lexical suffixes, each with illustrative words, followed by a second list of possible lexical suffixes, each with what appears to be evidence for its status. I have not included the suffixes that appear in personal names for reasons discussed at the end of this section.

The suffixes are ordered by initial consonant, without consideration of the vowels. The order is: p,  $\dot{p}$ , m, t,  $\dot{t}$ , n,  $\theta$ ,  $\dot{t}^{\theta}$ , s, t,  $\ddot{\lambda}$ , l, c,  $\dot{c}$ , x,  $k^w$ ,  $\dot{k}^w$ ,  $x^w$ , q,  $\dot{q}$ ,  $\ddot{x}$ ,  $q^w$ ,  $\dot{q}^w$ ,  $\ddot{x}^w$ , y, w, h. Connective elements are ignored; for example,  $-\partial lm \partial x^w$  as a variant of  $-m\partial x^w$  is listed with  $-m\partial x^w$ , and  $-\partial lm\partial x^w$  as a separate suffix follows  $-m\partial x^w$ .

1 -é $p \sim -\partial p$  'base, bottom'? (cf. Haeberlin 1974, 230, 4.28 'bottom, back part') only in:

 $tq\acute{e}p$  'tidal pound' (<  $t\acute{a}q$  'be closed off')  $y\acute{a}q^w ap$  'fell a tree by burning' (<  $y\acute{a}q^w$  'burn')  $sq^w\acute{e}qap$  'shuttlecock cod lure' (< ?)

3 -*ápsəm* ~ -*əpsém*- 'neck, nape'

 $t \delta p s \partial m$  'nape' (<  $t \partial$ - dummy root?)  $\dot{\tilde{\lambda}} \partial q t \delta p s \partial m$  'long-necked' (<  $\dot{\tilde{\lambda}} \ell q t$  'long')  $t \partial m \partial t \delta p s \partial m$  'pileated woodpecker' (<  $t \delta m \partial t$  'red ochre')  $l \partial k^w \delta p s \partial m$  'break one's neck' (<  $l \delta k^w$  'get broken')  $\dot{\tilde{\lambda}} \partial k^w \partial p s \delta m$  'grab him by the neck' (cf.  $\dot{\tilde{\lambda}} k^w \delta t$  'grab it')

4 -*mət* 'appearance'?

5 -mat 'kind, piece, part'

 $q\delta \tilde{x}mat$  'many kinds' ( $< q\delta \tilde{x}$  'many')  $t\delta x^w mat$  'three pieces' ( $< tf x^w$  'three')  $\theta f mat$ ,  $\theta f \theta \partial mat$  'greater part, greater' ( $< \theta f$  'big')  $^2 \sigma x^w f n mat$  'smaller part, smaller portion'

6 -mən ~ - $\delta$ mən ~ - $\partial$ mən ~ -min 'instrument'

 $n \delta l m \partial n$  'plan' (cf.  $n \ell l \partial t$  'set a time')  $k^w q^w \delta m \partial n$  'axe' (cf.  $k^w \delta q^w \partial t$  'hit it, as with a club')  $l \delta \ell \partial m \partial n$  'herring rake' ( $< l \delta \ell \partial m$  'rake herring'; cf.  $l \ell d t$  'flick it')  $k^w \partial c m \ell n$  'deer-hoof rattle' (cf.  $k^w \ell c \partial m$  'scream')

7  $-m \partial n \sim -m i n \sim -m i \cdot n$  'location, position'

yəwénmən 'area in front of house' (< yəwén' 'before, first') 'əncémən 'all parts of the body' (< 'ánəcə 'where') təsmín 'next younger sibling' (< tás 'arrive there') tətésmən 'one closest' (< stətés 'near') sənÅe?mí:n 'next older sibling' (< sánÅe? 'senior sibling/cousin')</p>

8 -mən ~ -émən ~ -mín 'residue' (an extension of 7?)

*i<sup>t</sup>θánman* 'small bones left over after a meal' (< *si<sup>t</sup>θán*' 'bone') *lak* "éman' 'something broken, as the broken end of a stick' (< *lák*" 'get broken') *i<sup>t</sup>θžmín* 'junk, worthless stuff' (< *i<sup>t</sup>θž* š 'get used up'; cf. *i<sup>t</sup>θč* š 'sell cheap, pawn') *s<sup>2</sup>∂lqánman* 'top of tree left after felling and sawing' (< ?)</li> *x*"aš x"tálawèman 'become the track of a river' (< stálaw' river,' JP 13)</li>

9 -éman 'extracted liquid' (an extension of 8?)

?ešx<sup>w</sup>ém'n 'seal oil' (< ?éšx<sup>w</sup> 'harbour seal') k<sup>w</sup>a·niém'n 'porpoise oil' (< k<sup>w</sup>á·ni 'porpoise') sq<sup>w</sup>i·lməx<sup>w</sup>ém'n 'blackberry juice' (< sq<sup>w</sup>i·lməx<sup>w</sup> 'blackberries') ck<sup>w</sup>imém'n 'red liquid' (< ck<sup>w</sup>ím 'red')

10  $-\partial m\partial t^{\theta} \sim -\dot{e}m\partial t^{\theta}$  long object'

 $sk^{w}(namat^{\theta})$  'how many (poles, etc.)?' ( $< k^{w}(n)$  'how many?')  $nac'emat^{\theta}$  'one (long object)' (< nac'a 'one')  $\vec{\lambda}aqtemat^{\theta}$  'tall (person)' ( $< \vec{\lambda}eqt$  'long')  $cac'(\cdot\vec{\lambda}amat^{\theta})$  'short (person)' ( $< cac'(\cdot\vec{\lambda})$  'short')

```
11 -m \partial x \sim -\dot{a} m \partial x \sim -\dot{a} \dot{m} \partial x 'country, person' (cf. 12)
```

*téqəməx* 'flats, flat country (as the Fraser delta)' (cf. *tqét* 'wide') *stáməx* 'warrior' (dummy root *t*-?)

- *qáləməx* 'Bud, Pal' (cf. *qéqələ* 'baby,' also Shuswap *qəlmux*<sup>w</sup> 'person,' Kuipers 1974, 236)
- sqx<sup>w</sup>ámðx</sup> (also recorded sq<sup>w</sup>x<sup>w</sup>ámðx) 'Squamish' (people and country up Howe Sound) (< ?)</p>

12 
$$-m \partial x^w \sim -\acute{e}m \partial x^w \sim -\partial lm \partial x^w \sim \partial wm \partial x^w \sim -a\dot{w}m \partial x^w$$
 'place, people (cluster?)

 $t \neq m \neq x^w$  'earth, land, soil' (< *t*-dummy root?)

 $x^{w} \delta lm \partial x^{w}$  'village, Indian' (< ?)

 $\delta x^{w} l e l e m \partial x^{w}$  'sea hunter-fisher' ( $\delta x^{w}$ - 'oblique nominalizer'; cf. l e l 'go shoreward')

```
s\dot{q}^{w}\dot{t}\cdot\dot{l}m\partial x^{w} 'blackberries' (//s-\dot{q}^{w}\partial l-\partial lm\partial x^{w}//, < \dot{q}^{w}\partial l 'be cooked, ripe')
\partial w\dot{k}^{w}\partial lm\partial x^{w} 'all different kinds of Indians' (< \partial \dot{w}\dot{k}^{w} 'be used up')
```

```
nəċáwməx^w 'one people ("tribe")' (< náċa² 'one')
```

 $n\partial c \partial w m \partial x^w$  'different people' (<  $n \dot{e} \dot{c}$  'different,' the root is different from the last, but the derivatives are homophonous)

 $sk^w x \delta w m \partial x^w$  'place name, name of a country' (<  $sk^w i x$  'name')  $q \check{x} \delta w m \partial x^w$  'many different kinds of people' (<  $q \delta \check{x}$  'many')  $\check{x}^w t \partial \dot{d} \cdot w m \partial x^w$  'where one belongs (native place)' (<  $t \partial \dot{l}$  - 'from')

13 -*álməx*<sup>w</sup> 'breast, milk, spring (of water)' (an extension of 12?)

θiθálməx<sup>w</sup> 'big-breasted' qətálməx<sup>w</sup>əm 'bind one's breasts' (cf. qítət 'bind it') pt<sup>θ</sup>álməx<sup>w</sup> 'milk a cow' (cf. pát<sup>θ</sup>ət 'wring it out') musməsálməx<sup>w</sup> 'cow's milk' (< músməs 'cow') snəsálməx<sup>w</sup> 'butter' (< snás 'fat, oil') sqimək<sup>w</sup>álməx<sup>w</sup> 'p.n., Devilfish Spring' (< sqímək<sup>w</sup> 'octopus') 14 -ama? 'body'

 $\theta \partial h \dot{a} m a^2$  'big-bodied' (<  $\theta i$  'big')  $\dot{k}^w \dot{a} \dot{m} \dot{k}^w \partial \dot{m} \dot{a} m a^2$  'strong-bodied' (<  $\dot{k}^w \dot{a} \dot{m} \dot{k}^w \partial \dot{m}$  'strong')  $q^w \partial^2 i \cdot y \partial q^w s \dot{a} \dot{m} a^2$  'thin person' (<  $q^w \partial^2 i \cdot y \partial q^w s$  'thin' [CC])

15 -*i*·ma?? (perhaps //-əy-əma?//, a connective -*əy*- with 14, -ama? 'body')

<sup>2</sup>óýí·ma<sup>2</sup> 'clean (person, house)' (< <sup>2</sup>óý 'good') qàlí·ma<sup>2</sup> 'dirty (person, house)' (< qól 'bad')</p>

16  $-t \partial n \sim -t \partial n' \sim -t \ell n' \sim -t \ell n'$  'instrument'

 $\dot{p}\dot{a}\dot{t}^{\theta}tan$  'needle' ( $\langle \dot{p}\dot{a}^{\theta}$  'sew')  $\dot{p}\dot{a}\dot{x}tan$  'hide scraper' (cf.  $\dot{r}\dot{x}at$  'scrape it')  $\dot{t}\dot{e}\dot{c}tan$  'knife, fish knife' (cf.  $\dot{t}\dot{c}\dot{c}at$  'cut it')  $\dot{x}ax^{w}ten$  'blanket' ( $\langle \dot{X}\dot{a}x^{w}$  'get covered'; cf.  $\dot{x}x^{w}at$  'cover it')  $\dot{t}^{\theta}\dot{a}\dot{x}^{w}ten$  'cleansing rite' ( $\langle \dot{t}^{\theta}\dot{x}\dot{x}^{w}$  'get washed'; cf.  $\dot{t}^{\theta}\dot{x}wat$  'wash it')  $\dot{q}^{w}e^{2}tan$  'broom' (cf.  $\dot{q}^{w}e^{2}t$  'sweep it')

17  $-t \partial n \sim -t \partial n'$  'person' (an extension of 16?)

control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control control c

18  $-tx^{w} \sim -\delta \dot{w}tx^{w} \sim -\delta wtx^{w} \sim -\delta t\delta x^{w}$  'house'

 $\theta \delta \dot{m} t x^w$  'two houses' (<  $\theta \delta m$ - ~  $\theta \delta m$ - 'two')  $t x^w \delta \dot{w} t x^w$  'three houses' (<  $t \dot{t} x^w$  'three')  $n c \dot{c} \delta w t x^w$  'one house' (<  $n \delta \dot{c} a^2$  'one')  $n \dot{c} \dot{c} \delta w t x^w$  'next door, next room' (<  $n \dot{c} \dot{c}$  'different')  $s \dot{a} l a^2 a c \dot{e} w t x^w$  'mat house' (<  $s \dot{a} l a^2 a c$  'house mat')  $s \theta \dot{e} w t x^w$  'big house (large, old-style house)' (<  $\theta \dot{t}$  'big')  $t \dot{w} \partial y \partial \dot{e} w t x^w$  'church' (<  $t \dot{t} \dot{w} \partial y \partial t$  'worship')  $s i l \dot{e} w t x^w$  'tent' (< s i l 'cloth, cotton, < E. 'sail')  $s c t \dot{e}^2 e l t \partial x^w$  'roof, upper storey' (cf.  $c i c c \partial t$  'above')

18.1 -é·ltəx<sup>w</sup> 'spouse' (an extension of 18, perhaps //-ələltəx<sup>w</sup>//, a double connective)

x<sup>w?</sup> aysé·ltax<sup>w</sup> 'have two wives' (< yaséla ~ <sup>2</sup>iséla 'two') x<sup>w</sup>tx<sup>w</sup>é·ltax<sup>w</sup> 'have three wives' (< tí x<sup>w</sup> 'three') x<sup>w</sup>calé·ltax<sup>w</sup> 'make off with someone's spouse' (cf. cí·lt 'grab something away from him') 18.2 - $\partial t x^w$  (perhaps - $\partial t$ - is a connective plus - $t x^w$  'house')

 $snécolt x^w$  'affines. "opposite friends" (=  $sk^w \delta lw \delta s$ ) (< néc 'other. different') (JP)

19  $-\delta n \partial \sim -\partial n e^{2}$  'ear. side'

 $\vec{\lambda}$  alqtána 'long-eared, deer' (cf.  $\vec{\lambda}$  éqt 'long,'  $\vec{\lambda}$  élaqt 'long pl.')  $\dot{t}^{\theta}\dot{q}^{w}\delta n\delta t$  'hit him on the ear' (cf.  $\dot{t}^{\theta}\dot{t}\dot{q}^{w}\delta t$  'hit, punch')  $q^{w} \dot{e} \cdot n e^{\gamma t}$  'pierce his ears' (cf.  $q^{w} \dot{e} \cdot t$  'make a hole in it')  $x^{w_2} \partial y \partial n \ell^2$  'listen' (<  $x^{w_2}$  'locative,'  $2 \partial y$  'good') łąźnam 'tie up (a boat), drop anchor' (cf. łąćt 'peg together, i.e., hold from the side'?)  $q^{w}s \delta n \partial m$  'drop anchor' (<  $q^{w} \delta s$  'go into the water')  $q^{w}s \neq n \neq t \neq n$  'anchor' (last plus *-t \neq n* 'instrument') *miģáňatan* 'pot-hook' (cf. *mí ģat* 'stick into the ground')  $\delta x^{w} \theta e^{i} y \partial n \partial t \partial n$  'dessert' (cf.  $\theta \partial y t$  'fix it,' i.e., something fixed on the side?) <sup>?</sup>*iməxné*<sup>?</sup>*tən* 'visitor' (< <sup>?</sup>*íməx* 'walk,' -*tən* 'person'?)  $qi \dot{x} \partial n \dot{e}^2 t \partial n$ ,  $q \partial q \dot{i}^2 \dot{x} \partial n \dot{e}^2 t \partial n$  'shadow' (< cf.  $qi \dot{x} \partial t$  'slide it,' -t  $\partial n$  'person'?) 19.1 -é·n (//-ənən//?) 'side, component, tip, feather' (cf. 19)

*stité* n 'upper end (of house or village)' ( $< t \neq yt$  'upstream')  $st \partial n \partial v \dot{e} n$  'female component (as flat leaf of cattail)' (<  $st \dot{e} n \partial y$  'female, woman')

 $w \partial y q e^{2} \dot{e} \cdot n$  'male component (as flowering stalk of cattail)' (<  $w \partial y q e^{2}$ 'male, man')

- $sa^2saq^wte^n$  'most junior sibling/cousin' (<  $sa^2saq^wt$  'junior sibling/ cousin')
- $syalax^wa^2 e'n$  'oldest person' (<  $syalax^wa^2$  'old person')
- $y \partial q^{w} \dot{e} \cdot n \partial m$  'light up (the end of a cigarette)' (<  $y \dot{a} q^{w}$  'burn,'  $\partial m$  'intr.')  $m \partial \dot{q}^w a^2 \dot{e} \cdot n$  'heron feather (used to induce vomiting)' (<  $sm \partial \dot{q}^w a^2$  'great blue heron')  $\dot{t}^{\theta}iqt\dot{e}n$  'flicker feather (worn by a shaman)' ( $\langle \dot{t}^{\theta}iqt$  'flicker')

```
k^{w} \delta y \check{x} \theta t \check{e} n 'style' (< k^{w} \delta y \check{x} \theta \delta t 'act,' lit. 'move oneself')
```

19.2 -*i*·*n* (a variant of 19.1? in this word only?)

 $h \partial \dot{y} q^{w} i \cdot n$  'lantern, lamp, candle, light' (<  $h \partial \dot{y} q^{w}$  'be burning, fire')

20  $-\partial n \sim -\partial n'$  'instrument'? (perhaps in loans from Northern Straits)

 $k^{w} \dot{a} x^{w} \partial n$  'long pole used by  $\dot{q}^{w} \dot{a} \dot{x}^{w} \partial qs$  dancer to strike roof planks' (cf.  $k^{w} \dot{a} x^{w} \partial t$  'knock on it')

 $x \neq m \neq n$  'trawl net (used upriver)' (cf. Northern Straits  $x \neq m \neq n$ )

*pí*<sup>2</sup>*k*<sup>w</sup>*∂n* 'splint for cooking fish' (< ?) *tí*<sup>2</sup>*c∂n* 'deadfall' (< ?) *snáw∂n* 'keepsake' (< ?)

21  $-\delta n \partial p \sim -\partial n \delta p$  (in resultative forms) 'ground'

cəcłánap 'high ground' (< cícəł 'high, above')  $\dot{\vec{X}}pánap$  'low ground' (<  $\dot{\vec{X}}ap$  'low') səwdánap 'look for ground to settle on' (< sáwd' 'seek') dwámanap 'pull up weeds' (cf. dwámat 'pull it out') słažánaptan 'floor' (< s- 'nominalizer,' łaž 'stand,' -tan 'instrument') słažanáp 'floored, floored house' (< s- 'res.,' łaž 'stand') słcanáp 'disked (as a field)' (cf. łícat 'cut it, slice it') ładánap 'daybreak' (<?)

22 -*in* $\partial s \sim -$ *i* $l_{\partial s} \sim -\partial n \partial s$  'chest, beach'

s<sup>2</sup>(nəs (JP), s<sup>2</sup>(ləs (CC) 'chest' (s- 'nominalizer' with suffix?) ləqtínəs 'broad-chested' (< lqet 'wide') xtínəs 'have an aching chest' (< xət 'hurt') həyqwənəs 'have heartburn' (< həyqw 'be burning') t<sup>9</sup>qwinəst 'hit him in the chest' (cf. t<sup>9</sup>tqwat 'hit him') xəqətínəs 'p.n., on Lulu Island above Steveston,' lit. 'long beach' (< xeqt 'long')

23  $-\delta n \partial s \sim -\delta l \partial s \sim -n \delta$  'tooth'

yánas 'tooth' (< y- dummy root?) θiθálas 'big-toothed' (< θίθa 'big pl.') staq<sup>w</sup>élas 'gums' (< stíq<sup>w</sup> 'flesh') sxaxk<sup>w</sup>nís 'something jammed into the teeth' (< s- 'resultative,' xák<sup>w</sup> 'be wedged in')

24  $-n\partial c \sim -l\partial c \sim -n\acute{e}c \sim -\partial ll\partial c$  'butt, tail, base, bay, price, suffix(?)'

sť $^{\partial}$ ámnəc 'hip bones' (< sť $^{\partial}$ ám' 'bone') x"čénacəm 'sit on chair, perch' (< x"- 'locative,' čé? 'alight, be set') x"<sup>2</sup>á polocom 'wipe one's bottom' (cf. ?épət 'wipe it') x"tod" solloct 'cut its tail off' (cf. td"at 'cut it off') pók"nəct 'put it (a pot) on the fire' (cf. pék"ət 'warm it over a fire') yəpənéc 'going with the wind' (< pah 'blow') d"sómnəct 'pull it out by the roots' (cf. d"sómət 'pull it out') tocnəcé·ls 'mow (a field)' (< root of ticət 'cut it' plus -e·ls 'activity') stítəcinəc 'mowed' (s- 'resultative') šx"títəcinəci·ls 'mowing machine' (šx"- 'oblique nominalizer,' -i·ls 'prog. of act.') s<sup>2</sup> *δ*θ*n∂c* 'bay' (< s- 'nominalizer,' <sup>2</sup>*e*θ- restrictive dummy root?) *q̂∂tnéc* 'go around (as the end of a lake)' (cf. *q̂∂táθ∂n* 'go along the edge')
k<sup>w</sup>*δxn∂ct* 'name its price' (cf. k<sup>w</sup>*íx∂t* 'name it')
<sup>2</sup>*δmn∂ct* 'put money down on it' (cf. <sup>2</sup>*e*<sup>2</sup>*∂m* 'give')
x<sup>w</sup>*lén∂yn∂ct* 'make a woman's name of it (a man's name)' (< slén∂y 'woman') (AG)

25  $-\dot{e}\cdot nx^w \sim -\dot{e}\cdot n\partial x^w \sim \dot{e}n\partial x^w$  'fish, food, fish run, season, (bad) weather' (?)

*pk<sup>w</sup>é·nx<sup>w</sup>* 'smoke fish (as salmon)' (cf. *pék<sup>w</sup>ət* 'warm it') *c̄əŷx<sup>w</sup>é·nəx<sup>w</sup>* 'smoke little fish (herring, etc.)' (< *c̄óŷx<sup>w</sup>* 'dry') *qx̆é·nx<sup>w</sup>* 'big run of fish, good year' (< *qóx* 'many') *təmhú·ṅ̀e·nx<sup>w</sup>* 'humpback year' (< *təm-* 'time,' *hú·ṅ* 'humpback [pink]
salmon') *xeṫd<sup>e</sup>énax<sup>w</sup>* 'p.n., Mt. Baker' (cf. *x̆e·ṫ<sup>e</sup>t* 'measure it,' lit. 'measures the fish run or measures the season'?) *spé·nx<sup>w</sup>* 'camas' (//pən-énəx<sup>w</sup>//, < pán 'be buried')</li> *?əŷyé·nəx<sup>w</sup>* 'good year, good run of salmon' (JP) *?əǧyé·nəx<sup>w</sup>* 'bring a storm' (√?, 'an old word" [JP])

25.1 - $\partial l en \partial x^w \sim -l en \partial x^w$  'season'

 $x^{w} \partial s \partial l e n \partial x^{w}$  'a month in the fall when leaves fall' (cf.  $x^{w} i s \partial t$  'shake it')  $h \partial y l e n \partial x^{w}$  'a month in the fall after the last named' (< h a y 'stop, be finished')

```
25.2 -i \cdot nx^w \sim -wi \cdot nx^w 'year'
```

 $\theta \partial m i \cdot n x^w$  'two years' (cf.  $\theta em - \sim \theta \partial m -$  'two')  $\vec{k}^w \partial n \partial w i \cdot n x^w$  'how many years?' ( $\langle \vec{k}^w i n$  'how many?')  $\vec{k}^w x \partial w i \cdot n x^w$  'age (of someone or something)' (cf.  $\vec{k}^w x \ell t$  'count them')

26  $-a\theta \partial n \sim -\partial \theta \sim -a^2\theta \sim -a^2\theta \sim -\partial \partial n \sim -a^2y\theta \partial n (//-ay_2\theta_2n//) \sim -a^2y\theta_2$ (before *-t* 'transitive')  $\sim -\partial y_2\theta (n - \sim -\theta (n \text{ (before } -\partial m \text{ 'intransitive')})$ 'mouth, lip, margin, edge'

θάθən 'mouth' (θ- dummy root?)
möqa?θ 'gift of food to take home after feast' (< möq' 'full, as from eating')</li>
háyθən 'finish eating' (< háy 'finish')</li>
ləkwá·yθən 'get one's mouth broken' (< lókw 'get broken')</li>
qə²á·yθən 'saliva' (< qá² 'water')</li>
θəhá·yθən 'big-mouthed' (< θί 'big')</li>
t<sup>2</sup>θ<sup>4</sup>q<sup>w</sup>á·yθət 'punch him in the mouth' (cf. t<sup>2</sup>θíq<sup>w</sup>ət 'punch him')
qələyəθínəm 'curse' (< qál 'bad')</li>

 $s^{2i}l\theta$  'outlet of lake' (//s-2il-20//, s- 'nominalizer,' 2il-'side'?)  $2\partial \tilde{\lambda} q a^2 \theta$  'go out into the open' (cf.  $2\partial \tilde{\lambda} q \partial l$  'go out')  $2\partial \dot{y} \dot{a}^2 \theta$  'sharp (blade)' (<  $2\partial \dot{y}$  'good')  $q \partial l \dot{a} \cdot \theta$  'dull' (<  $q \partial l$  'bad')  $sn\partial^2 d\theta \partial n$  'far side (of stream, etc.)' (< s- 'nominalizer,'  $ni^2$  'be there')  $\dot{q} \partial t \dot{a} \theta \partial n$  'walk along (a shore, etc.)' (cf.  $\dot{q} \partial t n \dot{e} c$  'walk around')  $\dot{q}_{\partial t}\theta$  in  $\partial m_{\partial t}$  'go along through the village inviting people' (the last plus -*am* 'intransitive' and -*at* 'transitive')  $^{2}\partial t\partial n \dot{\theta}\partial n$  'eat along the way (as berries off the bush)' (<  $^{2}\partial t\partial n$  'eat') 26.1 -  $a v a \theta \sim - \dot{a} \dot{v} a^2 \theta$  'head of weapon'? (connective plus -  $a^2 \theta$  'edge'?)  $q \partial \theta x^{w} \dot{a} y a \theta$  'prong of multi-pronged spear, foreshaft of doubleforeshafted harpoon' (cf.  $q \hat{e} \theta \partial l p$  'ocean spray')  $x p \dot{a} \dot{v} a^2 \theta$  'arrow of hazel shoot with no head' (< x a p- 'whistle'?) 27  $-i\dot{t}^{\theta}e^{\gamma} \sim -\dot{e}\cdot l\dot{t}^{\theta}e^{\gamma}$  'blanket, wealth'  $\dot{t}^{\theta} c i \dot{t}^{\theta} e^{\gamma}$  'stiff (cloth)' (<  $\dot{t}^{\theta} \dot{\sigma} c$  'stiff')  $t \partial w i \dot{t}^{\theta} e^{\gamma}$  'naked' (<  $t \dot{e} \dot{w}$  'escape')  $s\vec{\lambda}p(t^{\theta}e^{\gamma})$  'slip, undergarment' (s- 'nominalizer,'  $\vec{\lambda}\delta p$  'deep')  $m\partial \vec{\lambda} \dot{e} \cdot l \dot{t}^{\theta} e^{\gamma}$  'return wealth' (cf.  $m \dot{a} \vec{\lambda} \partial t$  'return it')  $2it^{\theta}\partial m$  'get dressed' (2- dummy root? plus  $-\partial m$  'intransitive')  $s^{2}it^{\theta} \partial m$  'clothing, blanket' (< s- 'nominalizer' plus last) 28  $-\partial s \sim -\dot{a}s$  'face, round object, moon, dollar, bow of canoe, bank'  $s^2 \hat{a} \theta \partial s$  'face' (s- 'nominalizer,'  ${}^2 \hat{e} \theta$ - restrictive root?)  $\delta x^{w} k^{w} \delta s \delta s$  'with face painted black'' (<  $k^{w} \delta s$  'get scorched')  $st^{\theta} \acute{a}m as$  'cheek bones' ( $< st^{\theta} \acute{a} \acute{m}$  'bone')  $x^{w} \vec{\lambda} \dot{a} q t \partial s$  'long-faced' (<  $\vec{\lambda} \dot{e} q t$  'long')  $x^{w}l\dot{a}dt as$  'with flattened head' (<  $l\dot{a}dt$  'wide')  $\delta x^{w}k^{w}\delta m \vec{\lambda} \delta s$  'with unflattened head (cf.  $sk^{w}\delta m \vec{\lambda}$  'bump')  $y \dot{a} \dot{q} \partial s t$  'sharpen it' (cf.  $y \dot{a} \dot{q} \partial t$  'grind it up')  $s \partial y \dot{q} \dot{a} s$  'sharpened' (res. of last) *ti<sup>2</sup>tás* 'facing upstream' (< *táyt* 'upstream') *<sup>2</sup>isálas* 'two dollars, two months' (< *yaséla* 'two') *lí*·*ms* 'a month name' (< *slí*·*m* 'sandhill crane')  $\check{s}x^{w}t i t \partial c \partial s$  'drawn up with bow on bank' (< s- 'resultative,'  $x^{w}$ - 'inward,'  $\sqrt{\text{ of } t(\dot{c} \partial t)}$  'draw it up with the bow on the bank') cíłas 'steep bank' (cf. cícał 'high') 29 -éł 'times'

 $k^{w} \partial n \ell t$  'how many times?' (<  $k^{w} i n$  'how many?')  $t x^{w} \ell t$  'three times' (<  $t i x^{w}$  'three')  $q \check{x} \ell t$  'often' (<  $q \delta \check{x}$  'many') 30  $-\dot{e}^{\gamma}t$  'bed, bedding'

*wi·lé?t* 'tule mat' (<  $wt \cdot t$  'tule')  $t^{\theta} \tilde{x}^{w} \tilde{e}^{2} t$  'wash diapers' (cf.  $t^{\theta} \tilde{x}^{w} \tilde{a} t$  'wash it')  $\theta \partial y \tilde{e}^{2} t \partial m$  'make one's bed' (cf.  $\theta \delta y t$  'fix it')  $\theta \partial y \tilde{e}^{2} t t$  'make his (another's) bed'

- 31 -e<sup>?</sup>t, -étət, -tət ~ -átət ~ tél- (before -ət 'transitive' and -əm 'intransitive') 'throat, breath, wind, food'
  - ?álłał 'throat' (< ?al-?) maqwáłał 'throat' (< máqw 'thick') iiqwłéłl 'choke him' (cf. tqwát 'cut it off') ?éqe?ł 'get a bone in one's throat' (< ?) ćtćkwe?ł 'have the hiccups' (< ?) tqé?ł 'have one's breath blocked' (cf. xwtáq 'be shut') hawqwéłał 'wind from upriver' (< háwqw 'progressive of wáqw 'drift downstream') xwkámłał 'eat enough' (< kám 'be enough') xwmakłélam 'return food' (cf. mákat 'repay it') xwnakłélam 'return food' (cf. ?émaqt 'take something to him') xwnawnacłélam 'pay back what one has eaten' (< naw- 'insert,' -nac 'price')
- 32  $-a \cdot t$  'canoe, vehicle' (cf.  $-a \cdot t$  'travel by means of' §12.3.3), also  $-\partial l \partial t$ ?

 $m\partial k^w t a t$  'have one's car break down' ( $< m\acute{e}^2 k^w t$  'be injured') (AG)  $s\dot{q}a^2\dot{a}t$  'companion boat' ( $< \dot{q}\dot{a}^2$  'accompany') (AG)  $w\partial \dot{q}^w \dot{a}t$  'lose a canoe by having it drift away' ( $< w\dot{a}\dot{q}^w$  'drift')  $p\partial lq^w \dot{a}t\partial m$  'have the last dance of the season' (lit. 'smash canoes') ( $< p\dot{a}q^w$  'break, go broke')  $?\acute{e}\dot{y}\ddot{x}\dot{a}t$  'western painted turtle' ( $<?\acute{e}\dot{y}\ddot{x}$  'crab'?)  $x^w\dot{a}\dot{y}q^w\partial a\dot{d}t$  'steamship' ( $< x^w$ - 'inside,'  $h\dot{a}\dot{y}q^w$  'be burning, fire')?

33 -*ət p* 'plant, bush, tree, vegetation'

 $sn\acute{a}\acute{c}atp$  'one (tree, etc.)' (<  $n\acute{a}\acute{c}a^{2}$  'one')  $s\theta\acute{e}tp$  'big (tree)' (<  $\theta\acute{t}$  'big')  $\acute{t}q^{w}\acute{a}matp$  'thimbleberry bush' (<  $\acute{t}q^{w}\acute{a}m$  'thimbleberry')  $t\acute{a}\breve{x}^{w}acatp$  'yew tree' (<  $t\acute{a}\breve{x}^{w}ac$  'bow')  $?\acute{e}palsatp$  'apple tree' (<  $?\acute{e}pals$  'apple')

34 -*lnet* ~ -*lné*<sup>2</sup> 'day of the week, week, day' (composed of -*a*<sup>2</sup>*l* 'attributive' and *nét* 'be night'?)

*sžážəłnet* 'Sunday, week' (< *žé<sup>2</sup>že<sup>2</sup>* 'holy') *syəléwəłnet* 'Monday' (< *yəléw* 'come after') łəxwəlnet 'three days' (

35 -łsxé (CC), -łcyé (JP) 'tens'

łəxwəłsxé 'thirty' (< łíxw 'three') ixəməlsxé 'sixty' (< ixám 'six')</pre>

36 -łél- 'fire'

 $x^{w}\theta ayal \ell lt$  'stir up the fire' (cf.  $x^{w}\theta e y \dot{x} x \dot{a} n am$  'shaking the feet')  $x^{w}lam l \ell lt$  'throw it into the fire' (cf.  $l \dot{a} m at$  'hit by throwing')  $\dot{x} x^{w}y a q^{w} \ell \ell l t an$  'dry kindling wood' (<  $y \dot{a} q^{w}$  'burn,' -tan' 'instrument')

37 -*əłcə* 'water'

 $l\check{x}^{w}\delta lc\partial$  'spit (v.)' (cf.  $l\check{x}^{w}\acute{a}t$  'spit it out')  $lq\delta lc\partial lc\partial$  'moonlit low tide' (<  $lq\ell lc$  'moon')  $\theta \delta^{2}t\partial lc\partial$  'low tide on a moonless night' (<  $\theta \ell^{2}t$  'dark')

38  $-\partial l\partial t c \partial \sim -\dot{a} t c \partial \sim -\dot{a} l c t \sim -\dot{e} l c t$  'belly, mind'

xáłalałca 'have a stomach-ache' (< xáł 'hurt')</li>
cd<sup>w</sup>á łca 'get pricked in the belly' (< cád<sup>w</sup> 'get pierced')
sxaldá łca 'have a striped belly' (< sxéld 'stripe')</li>
xáłálcł 'regret, be sorry' (< xáł 'hurt') (CC)</li>
2á yalcł 'feel good, 'have a good mind'' [feel well disposed]' (< ?á y' 'good')</li>
samk<sup>w</sup>élcł 'worried' (< smák<sup>w</sup> 'ball, lump,' lit. 'lump in belly') (JP)

39  $-\partial l\partial t c\partial \sim -\dot{a} \cdot t c\partial$  'giant'

*xámələlcə* (JP), *xəhámələlcə* (CC) 'a giant frog that cries like a baby' (< *xém* 'cry') *k<sup>w</sup>ećá·lce* (CC), *k<sup>w</sup>əćá·lcə* (AG) 'shark' (< *k<sup>w</sup>é·ć* 'dogfish') *t<sup>θ</sup>əxtá·lcə* 'a giant clam' (< *t<sup>θ</sup>óxət* 'gravel'?)

40  $-\acute{el}\partial \sim -\partial l\partial$  'place for, container for' (with  $\check{s}x^w$ - if an internal organ or an artifact)

 $s\theta e \theta ayan éla$  'blood vessel' (<  $s\theta \acute{e} \theta a \acute{y}an$  'blood')  $sa \acute{n}sa \acute{n}a \acute{y}a^{2}\acute{e}la$  'hornet's nest' (<  $sa \acute{n}sa \acute{n}a \acute{y}a$  'hornet')  $\check{s}x "i^{\theta}ela^{2}\acute{e}la$  'solar plexus' (<  $i^{\theta}ele^{2}$  'heart')  $\check{s}x "ha \acute{y}q "\acute{e}la$  'hearth' (<  $h \acute{a} \acute{y}q "$  'burning, fire')  $\check{s}x "lam \acute{e}la$  'bottle' (<  $l\acute{e}m$  'liquor, rum')  $\check{s}x "la\theta \acute{e}rla$  'cupboard' (//šx "-la $\theta$ an -ela//, <  $l\acute{a}\theta$ an 'dish') 41 - $\acute{e}l\partial \sim -\partial l\partial$  'person'

 $l_{\partial x}$  "élə 'three persons' (< lix " 'three') k "i·lə 'how many persons?' (//k "in-ələ//, < k "in 'how many?') sqéqələ 'baby' (cf.  $sqé^{2}eq$  'younger sibling, junior cousin')

42  $-\partial le^2 \sim -\partial la \sim -\partial li \sim \partial yi$  'game, sport'?

 $\theta \delta k^w \partial l e^2$  'tug-of-war' (cf.  $\theta k^w \delta t$  'pull it')  $s c \delta q^w \partial l e^2$  'a ball game like shinny' (< ?)  $\theta \delta q^w \partial l a$  (CC),  $\theta \delta q^w \partial l e$  (JP) 'bet' (cf.  $\theta \delta q^w \partial t$  'report on him'?)  $sm \partial t \partial t$  'a game played by women with dice of beaver teeth' (< ?)  $s \partial k^w \partial y t$  'shuttlecock used in girl's game' (< ?)

43 -*áləs* ~ -*aləs* ~ -*áləs* 'eye, mesh (of net), star, appearance, tendency(?)'

 $\theta i \theta a l as$  'big eyes, big mesh in net, big holes in knitting' (<  $\theta i \theta a$  'big plural')  $x^w m a d^w a l ast$  'poke him in the eye' (<  $m a d^w$  'burst,' -t 'transitive')

 $k^w c a l \partial s$  'see with one's own eyes' (<  $k^w e c$  'see')

k<sup>w</sup>onlcye<sup>2</sup>álos 'how many tens of mesh? (the measure of the depth of a net)' (< k<sup>w</sup>ín 'how many?' -lcye 'times ten')

 $\delta x^w k^w i l \delta l as$  'morning star' (< Cowichan  $k^w \delta y a l$  'become day')  $s k^w i k^w a m \delta l as$  'salmon pink, reddish' (<  $c k^w \ell m$  'red')  $\delta x^w i x^w k^w \delta l as$  'greyish' (<  $c x^w \ell k^w$  'grey')  $cic t \delta l as$  'steep slope' (<  $c \ell c a t$  'above')  $t d^w \delta l as$  'maple dish' (? <  $t \delta d^w$  'be cut off')  $s x k^w \delta l as$  'salmon backbone' (? <  $x \delta k^w$  'be stuck between')

44 -áls 'rock, spherical object, round berry'

t 
anti-dis 'reddish (rock)' (< t anti-dish 'red ochre')  $k^w ant als$  'a kind of hard black rock' (<  $k^w ant$  'porpoise')  $k^w a k^w q^w i^2 als$  'shinny' (<  $k^w a q^w$  'get hit [durative form?]')  $q^w a m c a ds$  'cranberry' (<?)

45 -əls 'weapon'

 $\theta \partial y \partial s \partial w$  'prepare for hunting' (cf.  $\theta \partial yt$  'fix it,'  $-\partial m$  'intransitive'?)  $k^{w}t \cdot ls \partial m$  'pick up something as a weapon' (//k won-ols-om//, cf. k wondt' (take it,'  $-\partial m$  'intransitive')  $k^{w}t \cdot ls \partial m dt$  'pick it up as a weapon' ( $-\partial t$  'transitive')  $k^{w}\partial n dt$  'hold it') ,

46  $-\delta l \partial l \sim -\delta l (l - (before - \partial m 'intransitive') 'crotch, between the legs'$ 

 $\delta x^{w?i} \cdot l_{\partial t}$  'crotch' (// $\delta x^{w-?i} l_{\partial t} = 1 \frac{1}{2} \frac{1}{2} \cdot \frac{1}{2}$ )  $x^{w} sisit \dot{\delta} l_{\partial t} t$  'tickle her between the legs' (cf.  $s \delta y \dot{t} t$  'tickle him/her')  $m \delta s \delta t$  'fold it up') 47 - $\acute{alt} \sim \acute{alt}$  'young'

słənəỳálł 'girl' (< słénəỳ 'woman') swəỳqe²álł 'boy' (< swáỳqe² 'man') swiẁləsálł 'boy' (< swíẁləs 'young man') spe²eθálł 'black-bear cub' (< spé²eθ 'black bear') məmə'nálł 'young (pl.) of an animal' (cf. ?əmámə'n 'little [pl.]')

48 -éləc 'person' (cf. 41)

*qžéləc* 'many people' (< *qáž* 'many') *šx<sup>w</sup>te<sup>2</sup>é·ləc* 'each person, whatever person' (cf. *šx<sup>w</sup>te<sup>2</sup>é* 'just as,' *ste<sup>2</sup>é* 'like')

49  $-\delta l \partial c$  'dung'

*šx<sup>w</sup>musməsáləc* 'cow manure' (< *músməs* 'cow') *šx<sup>w</sup>čikənáləc* 'chicken manure' (< *číkən* 'chicken')

50  $-e\dot{l}e^{2}c \sim -\partial le^{2}c \sim -li^{2}c$  'container'

 $i^{\theta}imele^{2}c$  'berry basket' (cf.  $i^{\theta}im \cdot m [//i^{\theta}im \cdot m //?]$  'pick berries')  $i^{\theta}s \partial le^{2}ct$  'nail it up (as a coffin)' (cf.  $i^{\theta}is \partial t$  'nail it')  $qa^{2}li^{2}c$  'water box' ( $< qa^{2}$  'water')  $ctele^{2}c$  'cover, lid (of box) ( $< \sqrt{c}\partial t$  'above')

50.1 -əlé?c 'bundle'

 $n\partial c\partial l \hat{e}^{2}c$  'one bundle (as of blankets)' (<  $n\partial ca^{2}$  'one')  $\dot{q}\dot{p}\partial l \hat{e}^{2}ct$  'tie them up in a bundle' (cf.  $\dot{q}\dot{e}\dot{p}\partial t$  'tie it up')  $y\partial \tilde{x}^{w}\partial l e^{2}ct$  'untie it (something in a bundle)' (cf.  $y\partial \tilde{x}^{w}\partial t$  'untie it')

52  $-\acute{elc} \sim -\acute{olc}$  'hair, line'

 $\vec{\lambda}$   $\partial qt \acute{e} l \acute{c}$  'long-haired' (<  $\vec{\lambda} \acute{e} qt$  'long')  $c \acute{q} \check{x} \acute{e} l \acute{c}$  'black-haired' (<  $c \acute{q} \acute{t} \check{x}$  'black')  $k^{w} \partial n \acute{e} l \acute{c}$  'hold a harpoon line' (cf.  $k^{w} \partial n \acute{e} t$  'hold it')

53  $-\delta l x \partial n$  'line'

k<sup>w</sup>əyək<sup>w</sup>əlxən (CC), k<sup>w</sup>əýk<sup>w</sup>əlxən (JP) 'trolling line' (cf. k<sup>w</sup>ú·yək<sup>w</sup> [CC], k<sup>w</sup>ówyək<sup>w</sup>əm [JP] 'troll') Xəqtəlxən 'long line' (< Xéqt 'long')</p>

54  $-álk^{wl} \sim -\partial lk^{wl}$  'winter dancer, possessing song'

*žəŵsálk*<sup>w</sup>*t* 'new dancer (this year's initiate)' (< *žéŵs* 'new') *?əmxálk*<sup>w</sup>*t* '''run'' a new dancer' (< *?íməx* 'walk') *stálk*<sup>w</sup>*t* 'non-dancer' (< ?) *x*<sup>w</sup>*áý∂lk*<sup>w</sup>*t* 'finish the season as a new dancer' (< *háy* 'finish') *t∂qáy∂*<sup>2</sup>*álk*<sup>w</sup>*t* 'having a wolf song' (cf. *st∂qáye*? 'wolf')

54.1 -alkwł 'child of'

*x<sup>w</sup>mi*θ*k<sup>w</sup>i*y*imilk<sup>w</sup>i* 'Musqueam child' (< *x<sup>w</sup>mi*θ*k<sup>w</sup>iyim* 'Musqueam') *s pélawalk<sup>w</sup>i* 'Sparrow child' (< surname *Sparrow*)

55 -éləq 'wave'

 $sq^{w}t\ell l aq$  '(moderate) surf' (<  $q^{w}\delta t$  'wash ashore')  $\theta i\theta \ell l aq$  'with big waves' (<  $\theta t \theta a$  'big (pl.)') qt' l aq 'rough, choppy (water)' (//qal-alaq//, <  $q\delta l$  'bad')  $h\delta yalaq$  'wave' (< ?)

56 -éləqəp 'smell, taste, sound' (perhaps 'trace')

scəkwəliləqəp 'how does it smell?' (< scékwəli' how?') 'əyiləqəp 'smell good' (< '>j' good') qilləqəp 'smell bad, taste bitter' (< qil' bad') pəpəti<sup>θ</sup>iniləqəp 'smell of skunk' (< spəpəti<sup>θ</sup>in 'skunk') stéwətiləqəp 'taste like herring' (< stéwəti' 'herring')  $\theta x^{w} eiləqəp$  'die down, fade away (a sound)' (<  $\theta = x^{w}$  'disappear')

57 -eləqən ~ -ələqən 'line, stream?'

sctélaqan 'cork line (of net)' (cf. cícał 'above') s $\vec{x}p$ élaqan 'lead [the metal] line, lower edge of net' (cf.  $\vec{x}$ í $\vec{x}$ ap 'below')  $\vec{x}$ "ámalaqan 'fast leak (in boat or roof) (<  $\vec{x}$ "ám 'fast')

58 -*álqən* 'pelt, fur, hide'

pdálqan 'mountain goat' (< páq' white') tacálqan 'shear wool' (cf. tícat 'cut it') kwaqwálqan 'beat wool' (cf. kwáqwat 'beat it') kwaqwálqatan ''sword'' used for beating wool' (< -tan 'instrument') sXpélqan 'feather' (< Xáp 'deep'?) 'axálqan 'buckskin' (cf. ?íxat 'scrape it')

59 - $\acute{elqt} \sim -\partial lqt$  'catch, game'

 $t \partial w \dot{e} l q t$  'lose a fish, miss a shot' (<  $t \dot{e} w$  'escape')  $q \check{x} \dot{e} l q t$  'have a big catch' (<  $q \dot{a} \check{x}$  'many')  $k'' i n \partial l q t$  'how many caught?' (< k'' i n 'how many?')

60  $-\partial \partial x \partial t^{\theta} \sim -\partial t \partial x \partial t^{\theta}$  'barrier'

 $c\vec{\lambda}\delta m \partial \partial x \partial t^{\theta}$  'jump over a fence' (<  $c\vec{\lambda}\delta m$  'jump')  $\vec{k}^{w}\partial n \partial t^{\partial \theta}$  'pass over a barrier' (<  $\vec{k}^{w}\partial n$ - 'transfer')  $s\partial x^{w}\partial t^{\theta} \partial x \partial t^{\theta}$  'wet the bed' (<  $s\delta x^{w}a^{2}$  'urinate') 61 -*álwət* 'garment'

 $\dot{p}\dot{q}\dot{s}lwat$  'Hudson's Bay Company blanket' ( $\langle \dot{p}\dot{s}\dot{q}'$  white')  $\dot{t}^{\theta}\ddot{x}''\dot{s}lwatam$  'wash one's clothes' (cf.  $\dot{t}^{\theta}\ddot{x}''\dot{s}t$  'wash it')  $\partial \ddot{x}\dot{s}\dot{l}qan\partial lwat$  'buckskin shirt' ( $\langle \partial \ddot{x}\dot{s}lqan$  'buckskin')  $\partial mx''\dot{s}lwat$  'raincoat' ( $\langle l\dot{s}max''$  'rain [v.]'; cf.  $s\dot{t}\dot{s}max''$  'rain [n.]')  $\dot{k}''ix''\dot{s}lwat$  'raincoat' ( $\langle k''i'x'''$  'pitch')  $\ddot{s}x'''\dot{p}i\dot{p}i\dot{t}^{\theta}\partial lwatam$  'wringer (on a washing machine)' (cf.  $\dot{p}i\dot{t}^{\theta}\partial t$  'wring it out')

62 -əlwəs 'heart, side of chest'

<sup>2</sup> $\partial \dot{y} \dot{e} \cdot lw \partial s$  'brave' (< <sup>2</sup> $\partial \dot{y}$  'good') *q* $\partial l \dot{a} lw \partial s$  'cowardly' (< *q* $\partial l$  'bad')  $\dot{t}^{\theta} \dot{q}^{w} \dot{e} lw \partial st$  'punch him on the side' (cf.  $\dot{t}^{\theta} \dot{q}^{w} \partial t$  'punch him')

63  $-\acute{elwe}^2s \sim -\acute{olwe}^2s$  'paddle'

tənəýélwe<sup>2</sup>s 'woman's paddle' (< sténəý 'woman') isəlálwe<sup>2</sup>s 'two paddles' (< yəsélə 'two') ġéwəlwe<sup>2</sup>s 'pay the paddles (i.e., pay those who helped bring food for affines)' (cf. ġéwət 'pay him') təlqálwe<sup>2</sup>səm 'put one's paddle into the water' (cf. təlqt 'dip it') stélqəlwe<sup>2</sup>s 'with unraised paddle (for a quiet approach)' (resultative of last) ġáyəlwe<sup>2</sup>s 'weaken' (< ġáy 'die')</p>

64  $-ic\partial n \sim -\partial c\partial n \sim -ic - \sim -\partial c$  (before -t 'transitive' and  $-\partial m$  'intransitive') 'back, surface, top'

 $\check{x} \partial \check{x} \dot{\rho}(c \partial n \text{ 'chipmunk' (cf. } \check{x} \dot{\ell} \partial t \text{ 'scratch it')}$   $sk^{w} \dot{a} \dot{m} \partial c \partial n \text{ 'hump-backed' (cf. } sk^{w} \partial n \dot{\ell}^{\theta} \text{ 'lump,' } sk^{w} \partial m \dot{\vec{x}} \text{ 'ridge')}$   $sc \dot{t} c \partial n \text{ 'top, surface' (cf. } c \dot{t} c \partial t \text{ 'above')}$   $c \dot{t} c t \text{ 'put it on top' (cf. last)}$   $s \partial h \dot{t} c \partial n \text{ 'this side' (< s- 'nominalizer,' ?t 'be here')}$  $sn \partial^{2} \dot{t} c \partial n \text{ 'that side, other side' (< nt^{2} \text{ 'be there')}}$ 

 $64.1 - e^{2}c \sim -\acute{e}l\partial c \sim -\partial\partial c \sim -lic$  'route across'

 $c \dot{\lambda} \dot{a} \dot{m} e^2 ct$  'jump over it (as a log), skip it' (<  $c \dot{\lambda} \dot{a} \dot{m}$  'jump')  $\dot{t} \dot{q}^w \dot{e}^2 lec$  'take a short cut (as a pass between islands)' (cf.  $\dot{t} \dot{q}^w \dot{a} t$  'cut it off')  $\dot{c} \dot{a} \cdot \dot{l} ac$  'cross over (an island), go over (a hill)' (<  $\dot{c} \dot{e}^2$  'land atop')  $\dot{t} a \dot{q} a \dot{t} c$  'other side (as of a hill)' (cf.  $t \dot{q} \dot{e} n$  'other end') 64.2 - c 'low tide'? (- $\partial c$  'surface'?)

nótcom 'shift from low tides in the daytime to low tides at night (in the fall)' (< nét 'become night')</pre>

wéyəlcəm 'shift from low tides at night to lows in the day (in the spring)' (< wéyəl 'become day')</pre>

64.3 - $\partial wic \sim -\partial w \partial c$  'back' (perhaps a connective and 64,  $-ic \sim -\partial c$  'surface')

*łəqtáwəc ~ łəqtəwíc* 'broad-backed' (< *łqét* 'wide') *xəłəwíc* 'have an aching back' (< *xát* 'hurt') *sxəlxèlqəwíc* 'striped backed' (< *sxélq* 'stripe')

65 - $\partial c$  'hundred'?

*nécowoc* 'one hundred' (< *nécow*- combining form of *nóca*' 'one')  $\theta \acute{e} \vec{m} c$  'two hundred' (<  $\theta em$ - 'two')

```
66 -écən ~ áləcən ~ éləcən 'testicle(s)'
```

*mécən* 'testicle' (< *mə*- dummy root?)  $\theta i \theta \delta l a c an \sim \theta \partial h \delta l a c an$  'having big testicles' (<  $\theta i$  'big')  $x^w l a m e^2 \delta l a c at$  'kick him in the balls' (cf.  $l a m e^2 t$  'kick him')  $x^w p i \lambda \delta l a c t$  'grab him by the balls' (cf.  $p \delta y \lambda t$  'squeeze it')  $x^w k^w a k^w a^2 e l a c an$  'be lying on one's back with legs flexed' (<  $k^w t^2$ 'ascend')

67  $-c \partial p \sim -\partial l c \partial p \sim -\partial l c \partial p$  'fire, firewood'

 $q^{w} \acute{a}^{2} c \partial p$  'spark' (<  $q^{w} \acute{e}$  'get through'?)  $\dot{q}^{w} \acute{a}^{2} i c \partial p$  'soot' (cf.  $\dot{q}^{w} \acute{e} \cdot yt$  'scorch it')  $p \acute{e} \cdot \dot{l} c \partial p$  'blow on a fire' (cf.  $p \acute{a} \cdot t$  'blow on it')  $y \acute{a} q^{w} \partial c \partial p$  'light a fire' (<  $y \acute{a} q^{w}$  'burn')  $k^{w} \partial t x^{w} \acute{d} c \partial p$  'bring in firewood' (cf.  $k^{w} t \acute{e} x^{w} t$  'bring it in')  $s \acute{q} \acute{d} \dot{c} \partial p$  'split firewood' (<  $s \acute{a} \dot{q}$  'get split')

68  $-c \partial s \sim -c s$ - (before  $-\partial m$  'intr.')  $\sim -c i s$  'hand, branch (of tree)'

 $s\dot{t}^{\theta}\dot{a}\dot{n}css$  'bones of the hand' ( $\leq s\dot{t}^{\theta}\dot{a}\dot{m}$  'bone')  $sid^{\theta}\dot{a}\dot{n}css$  'having small hands' ( $\leq samama\dot{n}$  'little [pl.]')  $t\dot{a}\dot{c}css$  (CC),  $t\dot{e}\dot{c}css$  (JP) 'get cut in the hand' (cf.  $t\dot{t}\dot{c}d$  'cut it')  $s\dot{t}\dot{q}^{w}\dot{e}css$  'missing a finger' ( $\leq s$ - 'resultative,'  $t\dot{a}\dot{q}^{w}$  'get cut off')  $k^{w}\dot{a}nacsst$  'take him by the hand, shake his hand' (cf.  $k^{w}\dot{a}nat$  'take it')  $s\dot{e}^{2}csam$  'raise one's hands' (cf.  $s\dot{e}^{2}t$  'raise it')  $sx^{w}tacis$  'way of doing something' ( $\leq sx^{w}$ - 'oblique nominalizer,' root of  $ste^{2}\dot{e}$  'like'?)  $t\dot{a}^{2}x^{w}css$  'grand fir branch' ( $\leq t\dot{a}^{2}x^{w}$  'grand fir')  $pt\dot{a}tcas$  'thick with branches' ( $\leq pt\dot{e}t$  'thick') 68.1 -*álowcos* 'finger' (-*cos* 'hand' with a double connective?)

*Åəlqətál`əwcəs* 'long-fingered' (< *Åéləqt* 'long [pl.]')
sən*Åáləwcəs* 'thumb' (< sán*Åe*<sup>2</sup> 'older sibling, senior cousin')
sa<sup>2</sup>asq<sup>w</sup>tál`əwcəs 'little finger' (< sá<sup>2</sup>səq<sup>w</sup>t 'younger sibling, junior cousin')

69  $-x \partial n \sim -x \partial - -x \partial n \sim -x i n \sim -x i n'$  'foot, leg, brim (of hat)'

 $\dot{c}al\dot{a}l\dot{a}\dot{x}an \text{ 'short-legged' (cf. } \dot{c}a\dot{c}t\dot{x} \text{ 'short')}$   $sn\dot{a}sxan \text{ 'marrow' (< snás 'fat, oil')}$   $q^wt\dot{e}\dot{y}xan \text{ 'shoe' (< q^wt\dot{e}\dot{y} \text{ 'driftwood, log')}}$   $t\dot{e}\dot{c}xan \text{ 'cut one's foot' (cf. } t\dot{t}\dot{c}at \text{ 'cut it')}$   $lak^wx\acute{en} \text{ 'break a leg' (< l\dot{a}k^w \text{ 'get broken')}}$   $s\dot{e}^2xanam \text{ 'raise a leg' (cf. } s\dot{e}^2t \text{ 'raise it')}$   $t\dot{e}^{x}wx\acute{enam} \text{ 'wash one's feet' (cf. } t\dot{e}^{x}wat \text{ 'wash it')}$   $k^wanxat \text{ 'grab him by the foot' (cf. } k^wanat \text{ 'take it')}$   $daxint \text{ 'accompany him' (< da^2 \text{ 'accompany')}}$   $sx^wnaxin \text{ 'feet' (< sx^w - 'oblique nominalizer,' } ni^2 \text{ 'be there')}$   $\theta(xan \text{ 'broad-brimmed (of hat)' (< \theta i \text{ 'big'})}$   $sdadaxin \text{ 'partner' (< da^2 \text{ 'accompany')} (see \S1.5.12 [3])$ 

69.1 - $\dot{a}\dot{l}\partial\dot{w}x\partial n$  'toe' (- $x\partial n$  'foot' with double connective? cf. 68.1)

 $\theta i \theta a \hat{l} \partial w x \partial n$  'big-toed' (<  $\theta i \theta \partial \phi$  'big [pl.]') sən $\hat{\lambda} a \hat{l} \partial w x \partial n$  'big toe' (<  $s \dot{a} n \hat{\lambda} e^{2}$  'older sibling, senior cousin')

```
69.2 - xən 'drop, droplet'
```

 $\theta(\theta \partial x \partial n \text{ 'in big drops'} (< \theta(\theta \partial \text{ 'big [pl.]'})))$  $\check{x}^{*} \acute{a} \acute{c} x \partial n \text{ 'stop raining'} (< ?)$  $q^{*i} q^{*t} \acute{m} x \partial n \text{ 'Scotch mist'} (cf. sq^{*t} x \partial m \text{ 'fog'})$ 

70  $-\vec{k}^w e^2 \sim -\vec{k}^w a^2$  'dead person'?

nówok<sup>w</sup>e<sup>2</sup> (CC), nówok<sup>w</sup>a<sup>2</sup> (JP) 'coffin' (cf. nówox 'insert it') cmék<sup>w</sup>e<sup>2</sup> 'attend a funeral' (< c- 'make, get'; cf. mé<sup>2</sup>x 'remove it'?)

71  $-\partial x^{w} \theta \partial t \sim \acute{e} l \partial x^{w} \theta \partial t$  'tongue'

 $t \delta x^w \theta \partial t$  'tongue' (<  $t \partial$ - dummy root)  $\dot{X} \partial q t \dot{e} l \partial x^w \theta \partial t$  'long-tongued' (<  $\dot{X} \dot{e} q t$  'long')  $s t \dot{c} \dot{e} l \partial x^w \theta \partial t$  'with a cut tongue' (< s- 'resultative,'  $t \dot{c}$ - 'cut')

72  $-x^{w} \partial t$  'canoe' (cf. 90)

 $n\acute{c} \partial x^w \partial t$  'one canoe' (<  $n\acute{c} a^2$  'one')  $\theta\acute{e}m\partial x^w \partial t$  'two canoes' (<  $\theta\acute{e}m$ - 'two')  $sn\acute{z}x^w \partial t$  'canoe (gen.), vehicle' (< ?) 73  $-eq \sim -\partial q \sim -q$  'penis, projection, rear'

 $x^{w}\theta \dot{e} q \sim \theta \dot{e} q$  'having a big penis' (<  $\theta i$  'big')  $q^{w}i \dot{n} e q$  'man's pubic hair' (<  $q^{w}i n$  'hair on face and body')  $k^{w} \partial n \dot{e} q t$  'grab him by the penis' (cf.  $k^{w} \partial n \partial t$  'take it')  $p \partial n \dot{e} q$  'slow match (cedar bark burning in clamshell' (<  $p \partial n$  'be buried')  $y \dot{a} q^{w} \partial q$  'light it (as a wick)' (<  $y \partial q^{w}$  'burn')  $s^{2}il\dot{e}^{2}eq$  'stern (of boat) (< ?)  $c\dot{e}^{1}lq\partial m$  'follow' (perhaps //cəl-əlq-əm// from *cel*- 'follow' or  $c\dot{e}\cdot lt$ //cél-ət// 'follow him')  $\ddot{\lambda} c\dot{e} lq\partial m$  'sneak after ducks' (cf.  $\dot{\lambda} i \dot{c} \partial t$  'sneak up on him')

73.1 - $\partial y e^{2}q \sim a y e^{2}q \sim -\dot{a} y \partial q$  'projection'

 $x^{w^{2}}\partial y^{2} \delta d y^{2} q$  (CC),  $x^{w^{2}} i s \delta l d y^{2} q$  'double-barrelled gun' (<  $y \delta s \delta d v^{2}$  'two')  $x^{w} n \delta c d y^{2} q$  'single-barrelled gun' (<  $n \delta c d a^{2}$  'one')  $\delta x^{w} t \delta x \delta d y^{2} q$  'prong' (cf.  $st(t \delta x t)$  'spread out')  $t x \delta t \delta y^{2} q t$  'spread its legs (as a ladder)' (cf. last)

74  $-q \partial n \sim -q i n$  'head, end, bow of canoe'

*méqən* 'hair' (< *m*- of *máqsən* ?)  $q^{w}éqən$  'bud out' (<  $q^{w}é$  'get through')  $s\dot{p}\dot{q}\dot{a}mqan$  'form a flower' (<  $s\dot{p}\dot{e}\dot{q}am$  'flower,'  $\dot{p}\dot{s}\dot{q}$  'white, bright')  $sm\dot{s}t^{\theta}qan$  'brain' (<  $mat^{\theta}$ - 'soft'? cf.  $mat^{\theta}il\thetaat$  'become soft')  $t\dot{s}\dot{q}qan$  'head of a long river' (cf.  $t\dot{q}\dot{e}$ 'n 'other end')  $ta\dot{q}qin$  'other end of the house (as last)'  $xixa\dot{q}qinam$  'be scratching one's head' (cf.  $xi\dot{q}at$  'scratch it')  $x^{w}awqint$  'push it ahead' (<  $x^{w}$ - 'move toward,' hiw 'upstream, toward fire')

 $\dot{c} \partial l q i n \partial m$  'turn one's canoe around, turn one's back' (<  $\dot{c} \partial l$ - 'switch')

<sup>?</sup>é $\theta$ əqən 'front' (< <sup>?</sup>e $\theta$ - restrictive root) sctiqən 'top of a hill' (cf. cícət 'above') s $\vec{x}$ píqən 'bottom of a hill' (cf.  $\vec{x}$ í $\vec{x}$ əp 'below')  $\vec{x}$ əpqénəm 'descend (a mountain), go downhill' (cf.  $\vec{x}$ í $\vec{x}$ əp 'below')  $\vec{q}^{w}$ áləqən 'shore of bare rock' (cf.  $\vec{q}^{w}$ í<sup>?</sup>əlt 'uncover it')  $\vec{k}^{w}$ í<sup>?</sup>qən 'climb (a mountain), go uphill' (<  $\vec{k}^{w}$ í<sup>?</sup> ascend') təqqént 'put something down in front of him' (cf. téqət 'lay it down')

74.2 - $q \partial n \sim -q \partial$ - (before -*t* 'transitive') ~ -q i n- (before - $\partial m$  'intransitive') 'throat, voice, speech, meal, opening facing upward' (an extension of 75?)

 $x^{w}t^{\theta}\delta t_{q}\delta t$  'quench his thirst' (cf.  $t^{\theta}\delta t_{\partial m}$  'cold,'  $t^{\theta}\delta t^{\theta}\delta t$  'refreshing')  $t^{\theta}\delta x^{w}q(n\delta m)$  'wash one's throat (have a beer)' (cf.  $t^{\theta}x^{w}\delta t$  'wash it')  $x^{w}\theta i q \partial n$  'loud' (<  $\theta i$  'big')  $x^{w}m \delta \theta k^{w}\partial y \partial m q \partial n$  'Musqueam speech'  $q \partial w i c \partial n q \partial n$  'Cowichan speech' (<  $q \partial w i c \partial n$  'Cowichan')  $k \ln j a j q \partial n s t \partial x^{w}$  'speak English to him' (<  $k \ln j a j$  'Englishman,' < CJ)<sup>3</sup>  $\delta x^{w}n \ell t \partial t q \partial n$  'breakfast' (<  $\delta x^{w}$ - 'oblique nominalizer,'  $n \ell t \partial t$  'morning')  $x^{w}p \delta n \partial q \partial t$  'fill it up (e.g., a hole in the ground)' (<  $p \delta n$  'get buried')

74.3 - $\partial q \partial n \sim -\partial q \partial n'$  'container' (an extension of 74.2?)

```
\check{x}\partial\theta (n\partial q\partial n' \text{ four (containers)'} (< \check{x}a^2 \acute{a} \partial \partial n' \text{ four'})
l\dot{q}\acute{e}cs\partial q\partial n' \text{ five (containers)'} (< l\dot{q}\acute{e}c\partial s' \text{ five'})
\theta (\partial \partial q\partial n' \text{ big (containers)'} (< \theta (\partial s' \text{ big [pl.]'})
```

75 -qs ~  $\partial qs$  ~ -qs $\partial n$  ~ - $\partial qs\partial n$  ~ qsin 'nose, snout, point (of land)'

*máqsan* 'nose' (< *ma*- dummy root?)  $q^w áyqs$  'a large gull with a yellow beak' (cf.  $cq^w áy$  'yellow, green') sxáqaqs 'nose ornament' (< xáq 'be finished'?)  $\theta iqsan$  'big-nosed' (<  $\theta i$  'big')  $st^{\theta} \delta \dot{m}qsan$  'bones of the nose' (<  $st^{\theta} \dot{a} \dot{m}$  'bone') tácqst 'cut off its snout' (< tic- 'cut')  $k^w \delta qsanam$  'turn one's nose up' (<  $k^w t^2$  'ascend'?)  $s^2 \delta lqsan$  'point (of land), Point Grey' (< 2al- restrictive root?)  $\dot{x}aqt\delta qsan$  'long nose, long point' (<  $\dot{x}eqt$  'long') yaqsin (CC),  $yaqs\delta nam$  (JP) 'go against the wind' (<?)

76  $-\acute{e} \check{x} \partial n \sim -\partial \check{x} \partial n \sim -\check{x} \acute{e} n$  (before  $-\partial m$  'intransitive') 'arm, side, branch, perimeter'

se·léžanam 'raise one's arms' (cf. sé<sup>2</sup>t 'raise it')
pay payéžan 'have arms akimbo' (cf. pá·yt 'bend it'?)
<sup>2</sup>(lažan 'far end (of a house or road)' (< ?íl-?)</li>
šx<sup>w</sup>lak<sup>w</sup>éžan 'corner (in house' (< lák<sup>w</sup> 'be broken,' lit. 'where the edge is broken' [DK])
táytažan 'p.n., the upper end of the row of houses at scaléx<sup>w</sup>' (< táyt 'upstream')</li>
na<sup>2</sup>éžan 'reach an end' (< ní<sup>2</sup> 'be there')
lálqažat 'dip it (a blanket or garment) partly into the water' (cf. lálqt 'dip it')
maiéžat 'bend a branch down' (cf. maiát 'bend it')
k<sup>w</sup>ac žénam 'keep a lookout, keep watch' (< k<sup>w</sup>éc 'look')
sqaléžan 'stockade, fence' (cf. qálam 'camp')<sup>4</sup>

<sup>3</sup> When JP was a boy, older monolingual persons used this word in asking the young to interpret.

<sup>4</sup> This is also a Chinook Jargon word, but it must be Salish in origin.

76.1 -*əléxən* 'arm' (probably 76 with -*əl*- connective)

*Åəlqətəléžən* 'long-armed' (< *Åéləqt* 'long [pl.]') st<sup>θ</sup>əməléžən 'arm bone' (< st<sup>θ</sup>ám' 'bone') sé·léžənəm 'raise one's arms' (cf. sé·t 'raise it') qəpəléžət 'tie it (as a kerchief) to his arm (as he dances)' (cf. qépət 'tie it up')

```
77 -q^w \sim -\partial q^w \sim \partial q^w \sim aq^w \sim -a^2 q^w \sim -i^2 q^w \sim -\partial l \partial q^w \sim -awaq^w 'head'
```

k<sup>w</sup>áq<sup>w</sup>q<sup>w</sup> 'bladderwrack' (< k<sup>w</sup>áq<sup>w</sup> 'get hit,' because the dry bladders burst with a pop when struck?)
i<sup>θ</sup>pí<sup>?</sup>q<sup>w</sup> 'get one's head filthy' (< i<sup>θ</sup>∂p' 'get filthy')
?<sub>∂x<sup>w</sup>ínaq<sup>w</sup></sub> 'small-headed' (< ?<sub>∂x<sup>w</sup>ín</sub> 'small')
x<sub>∂łaq<sup>w</sup></sub> 'have a headache' (< x<sub>∂l</sub> 'hurt')
tíca<sup>2</sup>q<sup>w</sup>∂m' 'get one's hair cut' (cf. tíc∂t 'cut it')
x<sup>w</sup>i<sup>θ</sup>q<sup>w</sup>∂l∂q<sup>w</sup>t' 'hit him on the head' (cf. i<sup>θ</sup>tid<sup>w</sup>∂t' punch him')
híwaq<sup>w</sup> 'headman, leader' (< hiw' 'upstream, toward the fire')</li>
yásaq<sup>w</sup> 'hat' (<?)</li>
w<sub>∂ł</sub>x<sup>w</sup>áwaq<sup>w</sup> 'three (animals)' (< w<sub>∂</sub>- '''established" aspectual prefix,'
tíx<sup>w</sup> 'three')
Xáqtaq<sup>w</sup> 'high (mountain)' (< Xéqt 'long')</li>

```
77.1 -áwðqw 'hat'
```

łəmxwáwəqw 'rain-hat' (< łóməxw 'rain') kwəmləxwáwəqw 'spruce-root basketry hat' (< kwómləxw 'root') kwixwáwəqw 'rain-hat of waterproof fabric' (< kwí?xw 'pitch, gum')</pre>

```
78 -\acute{ey} \sim -\partial \dot{y} 'plant, tree, wood'
```

 $q^{w}t\acute{e}$ ý 'driftwood, log' (<  $q^{w}\acute{a}t$  'wash ashore')  $pq^{w}\acute{e}$ ý 'rotten wood' (<  $p\acute{a}q^{w}$  'get broken up')  $\acute{c}s\acute{e}$ ý 'Douglas-fir wood' (<  $\acute{c}$ ís- 'grow'?)  $s\acute{e}k^{w} \partial m \partial y$  'birch tree' (<  $s\acute{a}k^{w} \partial m$  'whole bark')

79  $-\partial y$  'someone who does (agent)'?

sžəžé·ý 'child that cries a lot' (cf. žé·m 'cry')
sténəỷ 'woman' (< tən- 'weave'?)
qéməỷ 'young woman, teenaged girl' (< ?)</pre>

80  $-\partial y' \sim -ay'$  'fish'?

 $s\theta \dot{a}qa\dot{y}$  'sockeye salmon' (< ?)  $st^{\theta}\dot{a}q^{w}\dot{a}\dot{y}$  'chinook ("spring") salmon, salmon (gen.)' (cf. next)  $st^{\theta}q^{w}\dot{a}\dot{y}$  'little fish' (cf.  $t^{\theta}\dot{a}^{2}aq^{w}$  'stickleback')  $w\dot{e}ca\dot{y}$  'perch' (< ?)  $\dot{x}\dot{q}^{w}\dot{a}\dot{y}$  'salmon milt'(?) (< ?) 81  $-\partial ye \sim -\partial \dot{y}e \sim -\dot{a}\dot{y}e$  'dear one' (? in forms of address), 'small creature'(?)

?íməye 'grandchild (address)' (< ?íməθ 'grandchild') níkwəye 'uncle, aunt (address)' (cf. šxwəmníkw 'living parent's sibling') səmáye 'bee, hornet' (< ?) qwínəye 'clown dancer wearing bearskin appearing with sxwáyxwəy ' (< qwín 'hair on face and body') qíxəye 'Negro' (cf. cqíx 'black')

82 -éýə $n \sim \delta l y \partial n$  'net, trap'

 $q^wse\dot{y}\partial n$  'set a gill net' (<  $q^w\delta s$  'go into the water')  $\dot{k}^wte\dot{y}\partial n$  'set a net parallel to the shore' (cf.  $\dot{k}^w\dot{a}t\partial st$  'spread it [a net]')  $\theta\dot{e}k^w\partial \dot{y}\partial n$  'be pulling a net' (cf.  $\theta\dot{k}^w\delta t$  'pull on it')  $\dot{\lambda}\delta y\dot{q}e\dot{y}\partial n$  'be caught in a trap' (cf.  $\dot{\lambda}\delta y\dot{q}t$  'press down on it')

83  $-\partial y \dot{a} s \sim -\partial \dot{y} \dot{a} s \sim -\partial y \partial s$  'circular figure,' 'come full circle'?

 $q \partial q \partial y \delta s$  'barrel' (cf.  $q \partial q$  'taut,'  $q \partial q \delta t$  'make it taut,'  $q(q \partial t$  'bind it')  $k^w \partial m \partial x^w \partial y \delta s$  'watertight coiled basket' ( $\langle k^w \delta m \partial x^w$  'root')  $tq^w i^2 \delta s$  'tight around the middle' ( $\langle t \delta q^w$  'tight')  $h \partial n c \delta w \partial y \partial s \sim 2 \partial m c \delta w \partial y \partial s$  'come back to life, return after thought dead' ( $\langle h \partial n c \delta w \sim 2 \partial m c \delta w$  'come down to the shore')

84 - $\acute{e}y\partial t \sim \acute{o}y\partial t \sim -e\dot{y}\dot{t}$ - (before - $\partial m$  'intransitive') 'child, people, ceremony'

 $\dot{\lambda} \partial^2 \dot{e} y \partial t$  'stop a baby's crying' (cf.  $\dot{\lambda} \dot{a}^2 t$  'comfort him')  $n \partial w \dot{e} y \partial t$  'advise a child' (cf.  $n \dot{t} \dot{w} \partial t$  'advise him')  $x \partial t \dot{a} m e \dot{y} t \partial m$  'take care of a baby' (cf.  $x \dot{a}^2 t \partial m \partial t$  'take care of it')  $\dot{x} e^2 \dot{x} e^2 \dot{a} y \partial t$  'abnormal baby' ( $\langle \dot{x} \dot{e}^2 \dot{x} \partial t \rangle$  'holy, taboo')  $\dot{c} \dot{a} y \partial t$  'thank a crowd' (cf.  $\dot{c} \dot{t} t \sim \dot{c} \dot{t} \partial t$  'thank him')  $\theta \partial \theta \dot{x}^w \dot{e} y \partial t$  'girl's puberty ceremony ( $\langle \theta \partial \theta \partial \dot{x}^w$  'first menstruation')

85 -*iwən* ~ -*iwə*- (before -*t* 'transitive') 'inside, middle, waist, behind, rump, trunk (of tree)'

- $x^{w} \partial q^{w} \acute{a}m \partial w \partial n$  "smelly behind" (word coined by Coquitlam William for 'automobile') (<  $h\acute{a}q^{w}\partial m$  'smell')  $x^{w} \acute{\lambda} \partial qti(w\partial n$  'tall (tree)' (<  $\acute{\lambda}\acute{e}qt$  'long')
- 86  $-\dot{a}\dot{w}\partial\theta \sim -\partial w\partial\theta$  'kind, ingredient'

təwx<sup>w</sup>k<sup>w</sup>(nəwəθ 'about how many kinds?' (< təw- 'somewhat,' k<sup>w</sup>(n 'how many?')
x<sup>w</sup>nəċáŵəθ 'one kind, one ingredient' (< náċa<sup>2</sup> 'one')
x<sup>w</sup>tx<sup>w</sup>áŵəθ 'three kinds' (< tíx<sup>w</sup> 'three')
qžáŵəθ 'many kinds' (< qáž 'many')</li>

*płtws* 'regain consciousness' (<  $p \delta t$  'come to, become conscious')  $s \partial w \dot{q} t \dot{w} s$  'search for a lost person' (<  $s \delta \dot{w} \dot{q}$  'seek')  $q^w t n \partial w s$  'body hair' (<  $q^w t n$  'hair other than head hair')  $\dot{\lambda} \ell l \partial w s t$  'harness it' (<  $\dot{\lambda} \ell l$  'get stuck, be constrained')  $m \partial^2 t w s t$  'unharness it' (cf.  $m \ell^2 x$  'remove it')  $x^w \partial k^w t \dot{w} s \partial m$  'straighten oneself up' (cf.  $\partial k^w \delta t$  'stretch it')  $n \partial t \dot{w} s$  'one (bird)' (<  $n \delta t c a^2$  'one')  $\dot{q}^w \delta m \partial w s$  'pluck it (a bird)' (cf.  $\dot{q}^w \delta m a t$  'pull it out')  $k^w s t$  'singe it (a bird)' (<  $k^w \ell s$  'get scorched')  $\check{x} \delta t a l \partial w s$  (CC),  $\check{x} t \ell l \partial w s$  (JP) 'have a pain' (<  $\check{x} \delta t$  'hurt')  $^2 \partial w k^w t l \partial w s$  'become exhausted' (<  $^2 \partial w k^w$  'be all gone, used up')

88 -éwəs 'figure, back, trunk (of body)'

<sup>?</sup> $\partial \dot{y} \dot{e} w \partial s$  'having a good figure' (< <sup>?</sup> $\dot{s} \dot{y}$  'good')  $\dot{\vec{x}} \partial q t \dot{e} w \partial s$  'having a long back (trunk)' (<  $\dot{\vec{x}} \dot{e} q t$  'long')  $s \theta \dot{k}^w \dot{e} \dot{w} \partial s$  'of straight bearing' (cf.  $\theta \dot{k}^w \dot{\sigma} t$  'pull it,'  $s \theta \partial \theta \dot{e} \dot{k}^w$  'straight')  $s t q^w \dot{e} \dot{w} \partial s$  'too tight around the waist' (<  $t \dot{\sigma} q^w$  'tight')

89 -wət~ -wít ~ -ewət ~ -awət -wíl (before -t 'transitive' and -əm 'intransitive') ~ -wəl- (before -tən 'instrument') 'canoe, vessel, mind'(?)

táyewəl 'racing canoe' (< téy 'race')  $tk^{w}éwəl$  'caulk a canoe' (<  $ták^{w}$  'stuck in mud') xiləxiawəl 'battleship' (< xiləx 'make war')  $x^{w}k^{w} = nwilt$  'transfer it from one craft to another' (<  $k^{w} = n$ - 'transfer')  $t^{\theta} = x^{w} wilt$  'wash dishes' (cf.  $t^{\theta} x^{w} at$  'wash it')  $x^{w}t^{\theta} ax^{w} wilt = n$  'dish pan' (< -tən 'instrument')  $x^{w}q i wəl$  'mean, tough person' (< qil 'bad')  $x^{w}qaxwit$  'tricky' (< qax 'many') 90  $-wit \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\dot{alw$ 

 $\dot{\vec{x}}k^w \partial w \hat{i}lt$  'grab him by the sides' (cf.  $\dot{\vec{x}}k^w \hat{a}t$  'grab it')  $\dot{\vec{x}}x^w n \partial w \hat{i}t$  'other side of the Gulf' (<  $n\hat{i}^2$  'be there')  $sn\partial^2 \dot{\vec{a}} w \partial t$  'other side (of house, log, etc.)' (<  $n\hat{i}^2$  'be there')  $sh\hat{\vec{a}} w \partial t$  'this side (of house, etc.)' (<  $\hat{i}t$  'be here')  $sct\hat{\vec{a}} w \partial t$  'upper side, top' (cf.  $c\hat{i}c\partial t$  'above')  $ct\hat{\vec{a}} w \partial t$  'upper side, top' (cf. last)  $s\vec{x}p\hat{\vec{a}} w \partial t$  'under side' (cf.  $\vec{x}t\vec{x}\partial p$  'below')  $\vec{x}p\hat{\vec{a}} w \hat{\vec{i}} dm$  'move under (cf. last)'

91  $-we^{2}t \sim \partial we^{2}t$  'back'

 $st \delta \dot{q} w e^{2}t$  'back (of human body)' (cf.  $t \dot{q} \dot{e} \cdot n$  'other end')  $l \delta k^{w} \delta w e^{2}t$  'break one's back' ( $\leq l \delta k^{w}$  'get broken')  $t^{\theta} \dot{q}^{w} \delta \dot{w} \dot{e} tt$  'punch him on the back' (cf.  $t^{\theta} \dot{t} \dot{q}^{w} \delta t$  'punch him')

## **13.7. POSSIBLE LEXICAL SUFFIXES**

The following appear in too few words to be safely identified as lexical suffixes.

1 -*émət* 'stored food' (only one example)

?əwkwémət 'run out of food' (AG) (< ?ówkw 'be used up'), 'lose one's breath (DK)

2 -*tín* variant of 16, -*tən* 'instrument'?

 $\delta x^{w}q \partial lt \hat{n}$  'hate, bad feelings, bad health' (DK), 'disliked one' (AG) (<  $q \delta l$  'bad')

3 *-ť* ?

*témətəm* 'get drenched with rain' (cf. *táməx*<sup>w</sup> 'rain,' *stámtəm* 'dew')

4 -*ťén* ?

 $\check{x} \partial \check{y} s \check{t} \acute{e} \check{n}$  'nit' (<  $s \check{x} \partial \check{y} \partial s$  'head'?)

5 ->nt 'day'?

 $\theta \delta m \partial nt$  'two (days)' (<  $\theta \partial m$ - ~  $\theta \delta m$ - 'two')

6 -*ánəm* ?

```
syəlánəm 'year' (cf. syáləxwa? 'old person,' yəléw 'past')
```

7  $i^{\theta}$ ?

sk<sup>w</sup>ámi<sup>θ</sup> 'lump (as on body)' (cf. sk<sup>w</sup>ámacan 'hump-backed,' sk<sup>w</sup>ám<sup>X</sup> 'bulge, ridge') sk<sup>w</sup> ∂mit<sup>θ</sup>á·yθ∂n 'chin' (last plus -á·yθ∂n 'mouth,' i.e., 'mouth-bump') sk<sup>w</sup>∂mit<sup>θ</sup>∂léx∂n 'elbow' (same plus -∂léx∂n 'arm') titét<sup>θ</sup>t 'make fun of him because of accident or bad luck' (cf. té<sup>2</sup>it 'insult him') h∂wit<sup>θ</sup>é<sup>2</sup>t 'tease him' (cf. h∂wál∂m 'play')

8 -sím 'berry'?

*dəle·dəsím* 'twinberry (berry of *Lonicera involucrata*)' (< Cowichan *dəlé·də* 'crow'; cf. *dəlé·dəl p* 'twinberry bush')

9  $-\partial t q \partial y$  'snake'

```
²ółqoż 'snake'
s²í·łqoż (Cowichan, Northern Straits s²ínołqoż) 'two-headed serpent' (< ?)</p>
```

10 -əł x e ?

 $\theta \partial q \partial t \dot{x} \dot{e}$  'kneel' (CC) (<  $\theta \delta \dot{q}$  'be speared'?)  $\theta \partial q \partial t \ddot{x} \dot{e}^{2} \partial m$  'pray' (CC)  $\dot{t}^{3} \dot{q} \partial t \ddot{x} \dot{e}$  'kneel' (JP) (cf.  $\dot{t}^{4} \dot{q} \partial t$  'step on it')

11 - オ?

sk<sup>w</sup>óm<sup>3</sup>/<sub>d</sub> 'protuberance on the body, bulge' (cf. sk<sup>w</sup>óm<sup>d</sup> 'lump on the body'; cf. 7 above)
šx<sup>w</sup>k<sup>w</sup>óm<sup>3</sup>/<sub>d</sub> 'p.n., at Musqueam, glossed 'ridge'
šx<sup>w</sup>k<sup>w</sup>óm<sup>3</sup>/<sub>d</sub> s 'unflattened head' (last with -*os* 'face')

```
12 -\partial l \partial 'standing object'?
```

```
x^wk^wól^a 'scouring rush' (cf. x^wák^wðt 'polish it')
xóla (AC), sxól^a (JP) 'penis' (cf. xéwðt 'vulva,' with -éwðt 'vessel'? xé²t 'hold it in front, hold it on one's lap')
```

```
13 -íləx ? (cf. 19 below)
```

```
l\check{x}il\partial x 'stand' (cf. w\partial l\partial \check{x} 'stand up suddenly')
\dot{q}^{w}\partial yil\partial x 'dance' (< ?)
```

14 -é·ləq?

*qiqəŵé·ləq* 'menstrual pad (of shredded cedar bark)' (< *qíqəŵ* 'be menstruating')

```
15 -alwəm ~ -əlwəm? (cf. ?\acute{a}lwam 'be left alone')
```

```
x^wnəċálwəm (JP), x^wnəċálwəm (DK) "one family" (perhaps the descendants of one couple, JP believed) (< náċa? 'one')</p>
```

16 -*c* 'origin'?

*témc* 'from what place? from what people?' (cf. *stém* 'what?')

17 -cála? ~ ->čále??

*qətqətcála*<sup>2</sup> (JP), *qətqətəčále*<sup>2</sup> (CC) 'spider' (cf. *qítət* 'wrap something around it, bind it')

18 -*ċ* ~ -*əċ* ?

sálc 'go around' (cf. sálat 'spin it,' sálam 'spin,' sálqt 'swing it around') mímalac 'someone who is always making mistakes' (cf. málmal 'misidentify')

 $y \delta l y \delta l \delta c$  'storm, bad weather' (< ?)

19 -*əx* ? (cf. 13 above)

?ímax 'walk' (< ?ím- suggested by ?amícan 'walk on sand, as in shallow water looking for flounders,' < -ícan 'surface')</pre>

```
20 -k^w?
```

 $\check{x}\acute{e}\check{p}k^{w}\partial wst\partial m$  'have rheumatism' (<  $\check{x}i\check{p}$ - of  $\check{x}i\check{p}\partial t$  'scratch it,'  $\check{x}\check{p}ilt$ 'weaken him'? plus  $-\partial ws$  'body' and third-person passive suffixes; cf.  $\check{x}\acute{e}\check{p}q\partial wst\partial m$  'have rheumatism, have bones aching')

 $\dot{q}\partial \vec{p}k^w \acute{a}\cdot y \partial \partial \partial \dot{a}\cdot m$  'your teeth are chattering' (<  $\sqrt{\text{ of } \dot{q}\dot{e}\dot{p}\partial t}$  'tie it up,'  $\dot{q}\dot{e}l\,\vec{p}t\partial m$  'have a cramp' plus  $-\dot{a}\cdot y \partial \partial n$  'mouth' and second-person passive suffixes)

 $\dot{q}\delta m k^{w} \theta \delta t$  'shrink oneself' (cf.  $\dot{q}\delta m \delta s t$  'make room for him,'  $-\theta \delta t$  'self')  $\dot{\chi}\delta m k^{w} t$  'crush a louse between the teeth' (< ?)

21  $-\partial x^w$ ?

łámax<sup>w</sup> 'rain' (cf. słámłam 'dew,' słé·mł 'drenched with rain') lamáx<sup>w</sup> 'make the noise of something dropped, go plunk' (< lám 'get hit by something thrown'?)

22 - $\acute{e}x^w$  'times'?

 $n\partial \dot{c}ex^w$  'once' (<  $n\partial \dot{c}a^2$  'one'; cf.  $\partial \partial me'$  'twice,'  $lx^w ell$  'three times,' and other forms with -ell)

23  $-\partial x^{w}a^{2}$ ?

syálax<sup>w</sup>a? 'old person' (cf. syaléw' 'past')

24 -q?

?émaqt 'take for him, return something borrowed to him' (cf. ?é?am 'give')  $x^{w^2}\partial mql \dot{e}lt$  'take food to him' (<  ${}^{?}em$ - of  ${}^{?}e^{?}\partial m$  'give,' - $l\dot{e}l$ - 'food')  ${}^{?}ey \partial qt$  'put it to one side' (cf.  ${}^{?}ey \partial l$  'go to one side,' - $\partial l$  'move toward')  $m\dot{e}lq$  'forget' (cf.  $m\dot{a}lm\partial l$  'be confused, misidentify')  $\check{x}e\dot{p}q\partial wst\partial m$  'have rheumatism, have bones aching' (cf.  $\check{x}e\dot{p}k^{w}\partial wst\partial m$ 'have rheumatism,' see 23 above)

25  $-\dot{q}$ ?

 $s\delta l\dot{q}t$  'swing it around (e.g., a sling)' (<  $s\delta l$ - 'spin'; cf.  $s\delta l\dot{c}$  'go around')  $\dot{c}\delta l\dot{q}\delta s$  'sling' (<  $\dot{c}\delta l$ - 'turn,' - $\delta s$  'round object'?)  $y\delta m\dot{q}\theta\delta t$  'rub oneself with cedar boughs' (< ?, - $\theta\delta t$  'self')

26 -*ž* ?

 $x^{w}\theta \partial y \dot{x} \dot{t} \dot{e} lt$  'stir up the fire' (cf.  $\theta \dot{\partial} yt$  'fix it'? - $\dot{t} \dot{e} l$ - 'fire')  $\theta \dot{\partial} y \dot{x} \partial l q \partial n$  'fluff out wool' (cf.  $\theta \dot{\partial} yt$  'fix it'? - $\partial l q \partial n$  'fur')  $\dot{x} \dot{\partial} p \dot{x} t$  'scatter it, sow it' (<  $\dot{x} \partial p$  'down'?)

27  $-\dot{q}^{w}$  'fully, vigorously'?

 $y \delta t \dot{q}^{w} t$  'scour it' (< ?)  $\dot{t} \delta \dot{q}^{w} t$  'pry it off' (< ?)  $\delta \delta l \dot{q}^{w} t$  'snatch it off' (< ?)  $c \delta \dot{t} \dot{q}^{w} t$  'grind it up' (< ?)

```
28 -é 'times'?
```

 $\theta \partial m \acute{e}$  'twice' (<  $\theta \partial m$ - ~  $\theta \acute{e}m$ - 'two')

### **13.8. SUFFIXES IN FORMAL PERSONAL NAMES**

I have not included the suffixes that appear in formal personal names, that is, the names that are formally given at potlatch-like events, as opposed to baby names and nicknames that are acquired informally. A formal name generally consists of an initial element (root or stem) and one of a fairly small number of suffixes that identify the name as masculine or feminine. In some names the initial element seems meaningful, but in most it does not. Several of the suffixes seem identical with Halkomelem lexical suffixes, but others are not. Suffixes in names that have been borne by Musqueam men include  $-lenax^w$ , -alaq, -altan,  $-altx^w$ , and -qinam, and those in names that have been borne by Musqueam women are  $-tana^{-t}$ , -aye, and -alwat.

Names of this sort have been passed down over many generations, across language boundaries, and similar or identical forms appear throughout the Coast Salish social continuum to which the Musqueam belong. It seems that a name tends to become adapted to the phonology of the language in which it is used, though not always so. It would probably not be productive to try to analyze these names in any local context. Coast Salish formal names are best seen as the property of the Coast Salish social network and not of any one people or language. They cannot be treated as simply additional items in the lexicon of any one language. Because of their regional, multilinguistic use, they pose a multilinguistic, perhaps pan-Salish, problem in analysis and historical reconstruction. (For a good discussion of the problem, see Kennedy 2000, 269-77.)

# **14** Person Markers

The Halkomelem person markers, those forms that correspond in meaning to English personal pronouns, include several sets of affixes, one set of particles, and two sets of words. Because these several sets play different roles in the grammar of the language, each set has been introduced in the syntactic context where it appears. It also seems useful, however, to bring them all together here in a single section.

#### 14.1. PERSON AND NUMBER

The person markers distinguish first, second, and third persons in singular and plural number, but the distinctions are not altogether consistently made throughout the system. The first- and second-person object suffixes are identical in one form. In the passive, a single form appears for both first-person and second-person plural. In most sets, the third person has no separate plural form; plurality is marked by adding a pluralizing particle when it is not shown by other means, as by a plural noun or verb form.

It is also possible that number has not always been a simple distinction between one and more than one. James Point stated that the forms I identify as second-person "singular" could be (or were once) used in addressing a married couple, a pair of brothers, or even a family, the "plural" forms being used for a larger or less integrated group. This does not appear to be current or recent usage. Arnold Guerin identified the "singular" forms as strictly singular. There are, however, a few instances (not yet gathered and examined) in material dictated by both Christine Charles and James Point of singular forms used when two or more persons were addressed. Moreover, in a text dictated by Andrew Charles, there is an instance of a first-person singular form used to mean 'we (he and I),' referring to the speaker and his cousin, who was his hunting partner, in speaking to a group of strangers. Thus the question of number seems worth further investigation. There is no dual number or inclusive/exclusive distinction, unless the usage just mentioned could once have provided one. In both CC's and JP's texts, there are instances of second-person singular forms used in the sense of 'one,' as happens in colloquial English, for example, "If you ever went there, they would ..." Since there does not appear to be any other way of saying 'one,' one may assume that this usage may be native, not borrowed.

#### 14.2. FORMS

Table 14.1 lists the person markers, each set in a column. They will be described in the following order: (1) the first- and second-person coordinate subject particles, (2) the subordinate subject suffixes, (3) the third-person transitive subject suffix, (4) the possessive affixes, (5) the object suffixes, (6) the passive suffixes, (7) the personal words, (8) the possessive words, and (9) the third-person plural particle.

| Table  | 14. | 1 |
|--------|-----|---|
| 100000 |     | - |

| The pe | rson m          | arkers                            |                |                               |                     |         |               |                   |                                      |                                                  |
|--------|-----------------|-----------------------------------|----------------|-------------------------------|---------------------|---------|---------------|-------------------|--------------------------------------|--------------------------------------------------|
|        | 1<br>subj       | 2<br>subro<br>subj                | d p            | 4<br>oss                      | 5<br>obj            |         | 6<br>pass     | 7<br>pers<br>word | ŗ                                    | 8<br>poss<br>vord                                |
| 1sg    | cən             | -e·n<br>-ən<br>- <sup>?</sup> e·r | 1              | nə-                           | -S<br>-Samx<br>-amx | ţ       | -Sel-<br>-el- | ?ánθə<br>?é·nθə   | na                                   | pswé <sup>9</sup>                                |
| 2sg    | Čx <sup>w</sup> | - <i>∂x</i> *                     | v 9<br>9       | ən-<br><sup>9</sup> ə-<br>əT- | -S<br>-Samə<br>-amə | ,       | -Sam<br>-am-  | náwə              | ?ə<br>?ə                             | swé <sup>?</sup><br><del>O</del> wé <sup>?</sup> |
| 1pl    | ct              | -ət                               |                | -ct                           | -alx <sup>w</sup>   |         | -al           | łníməł            | s                                    | ?á·ł                                             |
| 2pl    | ce·p            | -e•p                              | , ?<br>?<br>-? | ən-<br>əT-<br>ələp            | -alə                |         | -al-          | łwálap            | <sup>2</sup> әѕн<br><sup>2</sup> әӨv | vé <sup>v</sup> ələp<br>vé <sup>v</sup> ələp     |
|        | 1               | 2                                 | 3              | 4                             | 5                   |         | 6             | 7                 | 8                                    | 9                                                |
|        | subj            | subord<br>subj                    | tr sub         | poss                          | obj                 | p<br>co | ass<br>subrd  | pers<br>word      | poss<br>word                         | pl<br>part                                       |
| 3 sg   | 0               | - <i>∂S</i>                       | - <i>∂S</i>    | -5                            | 0                   | 0       | -əy-          | Żá                | swé <sup>2</sup> s                   | 0                                                |
| 3 pl   | 0               | -ƏS                               | - <i>∂S</i>    | - <i>S</i>                    | 0                   | 0       | -әу-          | Х́аlәт            | swé <sup>9</sup> s                   | ?é∙łtən                                          |

Phonologically the coordinate subject, subordinate subject, and possessive sets can be grouped together, as can the object and passive sets, but the two groups have nothing in common. The coordinate subject set contains some of the same elements that appear in the possessive set. The object and passive sets likewise share several elements. The personal words seem to share some elements with the first group. The possessive words contain the possessive affixes except for one case of suppletion. The third-person plural particle seems unrelated to the other person markers. Some further analysis will be presented below.

Constraints may be summarized as follows: In the active paradigm, a third person cannot play the role of subject with a second person as object. Forms corresponding literally to 'he sees you' cannot occur; instead we find passive forms like 'you are seen.' In the passive, only third persons can be agents. Forms like 'I am seen by you,' 'you are seen by me,' and 'he is seen by me' are not possible. Such relations can be expressed in the active only, as 'you see me,' and so on. The first-person plural possessive suffix cannot form a possessive word in the manner of the other possessive affixes; instead we find a wholly irregular form.

## 14.2.1. The First- and Second-Person Subject Particles

These must be grouped with the second-position predicate particles (see §16.2). As members of that set, they move to follow an auxiliary or adverb. They are:

| cən             | ʻI'           | ct 'we'         |
|-----------------|---------------|-----------------|
| čx <sup>w</sup> | 'you (sing.)' | ce·p 'you (pl.) |

They are composed of an initial element c- (/c/ regularly becomes / $\check{c}$ / before / $x^{w}$ /) and a set of final elements identifiable with final elements of the subordinate subject set (§14.2.3). (This c- is possibly identifiable with the prefix c-'get, have, make, do.' One can conjecture that it was once an auxiliary verb.)

These particles mark first- and second-person subjects in both intransitive and transitive predicates in coordinate constructions (main clauses), as in the following sentences:

| (a) a | ném cən ce <sup>?</sup> . '       | I'll go.'           | (b)  | ném ct ce?. 'W                          | Ve'll go.'           |
|-------|-----------------------------------|---------------------|------|-----------------------------------------|----------------------|
| (c) i | ném čx <sup>w</sup> ce?.          | 'You'll go.'        | (d)  | ném ce <sup>.</sup> p ce <sup>?</sup> . | 'You folks will go.' |
| (e) a | céwət cən ce <sup>9</sup> .       | 'I'll help him/her. | '(f) | céwət ct ce?.                           | We'll help him/her.' |
| (g) ( | <i>čéwət čx<sup>w</sup> ce</i> ?. | 'You'll help        | (h)  | céwət ce·p ce?                          | . 'You folks will    |
|       | him/her.'                         |                     |      | help him/he                             | r.'                  |

With an intransitive predicate head, a third-person subject is marked by zero (i.e., is indicated by an absence of a first- or second-person marker) and plurality is optionally indicated by the particle <sup>?</sup>*é*·*l*tan (see §14.2.9), as in the following:

```
(i) ném ce?. 'He/she/it/they will go.' (j) ném ce? ?é·łtan. 'They will go.'
```

With a transitive predicate head, however, a third-person subject must be marked by the suffix given next.

#### 14.2.2. The Third-Person Transitive Subject Marker

A suffix  $-\partial s$  marks a third-person subject of a transitive verb in a main clause, contrasting with zero marking a third-person subject of an intransitive verb in a main clause. It is identical in form with the suffix  $-\partial s$  that marks a third-person subject in a subordinate clause (see next set).

It always follows the transitivizer (and object person marker, if any). Unlike the first- and second-person subject particles, which move to follow an auxiliary or adverb, this suffix stays with the transitivized verb itself. Compare the following pairs of sentences:

| (a) | céwət cən ce? | 'I will help        | (b) | <i>céwətəs ce</i> ? | 'He/she will help |
|-----|---------------|---------------------|-----|---------------------|-------------------|
|     | him/her.'     |                     |     | him/her.'           |                   |
| (c) | ni cən céwət. | 'I helped him/her.' | (d) | ni ćéwətəs.         | 'He/she helped    |
|     |               |                     |     | him/her.'           |                   |

In the second pair, the particle  $c \ge n$  'I' moves to follow the auxiliary ni, while the suffix  $- \ge s$  'he/she' does not.

Again, plurality of a third person may be indicated with the particle <sup>2</sup>*é*·*lton*, as in:

(e) k<sup>w</sup>⇒c-n-ámx-⇒s ce? ?é·łt∍n.
 see-TR-me-3TR FUT 3PL
 'They will see me.'

#### 14.2.3. The Subordinate Subject Person Markers

These are the suffixes:

| <i>-e·n ~ -∂n ~ -?e·n '</i> I' | <i>-∂t</i> 'we'                                  |
|--------------------------------|--------------------------------------------------|
| $-\partial x^w$ 'you (sing.)'  | $-\acute{e}\cdot p \sim -\partial p$ 'you (pl.)' |
| - <i>əs</i> 'third person'     |                                                  |

The first- and second-person forms have, as indicated above, the same final elements that occur in the corresponding subject particles. The third-person form bears a similar relationship to the third-person possessive suffix *-s*. The second-person plural  $-\hat{e} \cdot p$  has probably developed from the  $-\partial \partial p$  that occurs in the possessive set and in the personal words through the loss of the medial resonant. In AG's speech, it appeared as  $-e^2ep$  in progressive forms.

The allomorphs  $-\partial n$  and  $-\partial p$  appear after the auxiliaries ?i 'be here' and  $ni \sim ni^2$  'be there.' In this environment, the schwas of the whole set may be realized as vowel length; for example, the auxiliary  $ni^2$  with the subordinate subject set yields  $ni \cdot n$ ,  $ni \cdot x^w$ ,  $ni \cdot t$ ,  $ni \cdot p$ , and  $ni \cdot s$ . The first-person allomorph  $?\acute{e} \cdot n$  occurs following a vowel.

These suffixes mark the subject of a verb (either intransitive of transitive) in a subordinate clause. They regularly appear in three types of constructions:

(1) In subordinate clauses introduced by wa- 'if, when, that' or 2at- 'whenever, whatever' (see §4.2), as in:

| wənéme'n 'if I go'                     | wənémət 'if we go'                         |
|----------------------------------------|--------------------------------------------|
| <i>wənéməx<sup>w</sup></i> 'if you go' | <i>wənémè·p</i> 'if you folks go'          |
| wənéməs 'if [third person] goes'       | wənéməs ( <sup>?</sup> éłtən) 'if they go' |

(2) In the negative paradigm (see §6), in what may be seen as subordinate clauses complementing a main clause headed by  $2\delta w\partial$  'not' or  $x^w\partial w\dot{e}$  'not yet,' as in:

- (a) <sup>9</sup>ówə cən ném-è·n. not I go-I 'I don't go.'
- (b) <sup>9</sup>ówə cən ni·n ném. <sup>9</sup>ówə cən ni<sup>9</sup>-ən ném not I there-I go 'I didn't go.'
- (c) <sup>2</sup>ówə ném-əs. not go-3SUB 'He doesn't go.'

For further discussion and examples, see §6.1.1.

(3) In relative clauses with extracted objects (see §4.1.1.1), such as:

(d) k<sup>w</sup>θ o céw ot èn ce?
 k<sup>w</sup>θ o céw ot - èn ce?
 ART help-TR-I FUT
 'the one I will help'

(e) k<sup>w</sup>θə ni·n céwət
 k<sup>w</sup>θə ni-ən céw-ət
 ART AUX-I help-TR
 'the one I helped'

For further discussion and examples, see §4.1, "Relative Clauses."

## 14.2.4. The Possessive Affixes

These are:

```
n\partial - \operatorname{'my'} -ct 'our'

\partial \partial n - \sim \partial \partial T - \operatorname{'your'} \partial \partial n - \sim \partial \partial T - \ldots -\partial \partial \partial p 'your (pl.)'

-s 'his/her/its/their (3POS)'
```

As in other Salishan languages, the possessives are split between prefixes and suffixes. The possessives of the first- and second-person singular are prefixes,

the possessives of the first-person plural and the third person are suffixes, and the second-person plural possessive is a discontinuous form, a combination of the prefix for the singular and a suffix that makes it plural. As in other paradigms, a third person is optionally pluralized with the particle *?é'ttən*.

| nəléləm 'my house'           | <i>léləmct</i> 'our house'                  |
|------------------------------|---------------------------------------------|
| <i>?ənléləm</i> 'your house' | <sup>?</sup> ənléləmələp 'your (pl.) house  |
| léləms 'his/her/its house'   | léləms ( <sup>?</sup> é·łtən) 'their house' |

The first-person plural possessive is anomalous in being identical with the subject particle ct 'we,' which is composed of an element c- that occurs in the other subject particles and a -t that we can identify as 'first-person plural.' As a possessive, -ct is probably a borrowing from the subject particle set.

The second-person prefix appears as  ${}^{2}\partial T$ - before /s/ and otherwise as  ${}^{2}\partial n$ -. For some speakers (e.g., CC and DK), the  ${}^{2}\partial T$ - is realized as  ${}^{2}\partial$ - or  ${}^{2}\partial s$ -, while for others (e.g., JP) the //T// coalesces with the initial /s/ as / $\theta$ /. Compare usages with  $m\dot{a}\dot{n}\dot{a}$  'child' and  $sq^{w}\partial m\dot{e}\dot{y}$  'dog':

| CC | <sup>?</sup> ənmə́nə 'your child' | ?əsqwəméy ~ ?əssqwəméy 'your dog'             |
|----|-----------------------------------|-----------------------------------------------|
| JP | ?ənmáňa 'your child'              | <sup>?</sup> əθq <sup>w</sup> əméý 'your dog' |

This difference is said to go back to dialect differences between the Musqueam settlements of  $m \acute{a} l \partial y$  (on the downstream side) and  $s \acute{c} \partial l \acute{e} x^w$  (on the upstream side), the form with  $/\theta$ / being that of  $s \acute{c} \partial l \acute{e} x^w$ . (Historically, the form with /n/ is probably the older. In the  $m \acute{a} l \partial y$  dialect, the /n/ assimilated to the /s/ or was dropped, while in the  $s \acute{c} \partial l \acute{e} x^w$  dialect it became a stop \*t before /s/, then the \*ts coalesced to become \*c, which regularly became  $/\theta/$ .)

The third-person possessive affixed to a word ending in /s/ remains an /s/, and the resulting /ss/ is generally audible. Some speakers of the Island dialect dissimilate, converting the possessive to  $\theta$ ; the Musqueam speakers for whom I have data did not do this.

Optional morphophonemic processes (see §1.5.1) create other variations. Both of the prefixed possessives tend to coalesce, especially in rapid speech, with a preceding article or word ending in a vowel. With  $n\partial$ - 'my,' the schwa may be dropped, leaving -n, so that  $t\partial n\partial l\ell \partial m$  'my (present visible) house' may become  $t\partial n \, l\ell l\partial m$ . With  $\partial n - 2\partial T$ - 'your,' one of two things can happen. The schwa of the  $2\partial n$ - form may be dropped, leaving the glottal stop and resonant to become a glottalized resonant, so that  $t\partial \, 2\partial n \, l\ell \, \partial m$  'your (present visible) house' may become  $t\partial n \, l\ell \, l\partial m$ . This leaves the glottalization of the resonant to carry the burden of distinguishing 'my' and 'your.' Or, the

<sup>1</sup> Such small distinctions are evidently not unique to Halkomelem. Franz Boas, working with a Pentlatch speaker in 1886, complained, "It took me an hour to distinguish between 'I' and 'you'" (Rohner 1969, 65).

sequence of two schwas produced by an article followed by 'your' may become  $/e^{9}e/or/e^{-}$ , so that  $k^{w}\partial^{2}\partial\theta q^{w}\partial m\dot{e}\dot{y}$  'your (nearby, invisible) dog' may become  $k^{w}e^{\cdot}\theta q^{w}\partial m\dot{e}\dot{y}$ .

The possessives are strictly limited in the kinds of words to which they can be affixed. (It is just these limitations that provide the basis for distinguishing the major classes of words in the language. See  $\S2.1$ .) The possessives can be affixed to nouns and to nominalized non-passive verbs only.

Affixed to nouns, the possessives indicate the person of the possessor, "possession" having the usual sense of relating kin to kin, property to owner, part to whole, and so on, as in: *nomén* 'my father,' *Pontén* 'your mother,' *móňos* 'his/her child,' *nosq*<sup>w</sup>*oméy*' 'my dog,' *lélomct* 'our house,' *Ponpútolop* 'your (pl.) boat,' *nocélox* 'my hand,' *máqsons* 'his/her nose.'

When the possessor is a noun, it follows the word taking the possessive, as in:

- (a) tə θάθən-s tə sẳíẳqəł
   ART mouth-3POS ART child(status)
   'the child's mouth'
- (b) tə mɨnɨ-s θə sɨnke?
   ART child(kin)-3POS ART senior.sibling 'the son of the older sister'
- (c) tə θάθən-s tə mönö-s
   ART mouth-3POS ART child(kin)-3POS
   'her son's mouth'
- (d) tə qłán-s k<sup>w</sup>θə snáx<sup>w</sup>əł
   ART box-3POS ART canoe
   'the bow of the canoe'
- (e) tə šx<sup>w</sup>lək<sup>w</sup>éxən-s tə léləm ART corner-3POS ART house 'the corner of the house'
- (f) k<sup>w</sup> sn ə<sup>2</sup>álwəł-s tə léləm ART far.side-3POS ART house
   'the other side of the house'

When affixed to nominalized non-passive verbs, that is, active verbs with the prefix s- 'nominalizer' (see §12.1.1), Halkomelem possessives (like English possessives with gerunds) mark underlying subjects. Compare the following main-clause forms (left column) with their nominalizations (nominalized clauses):

- (a) ném cən go I 'I go'
- (b) čéwəθàmx čx<sup>w</sup>
   čéw-ət-Samx čx<sup>w</sup>
   help-TR-me you
   'you help me'
- (c) ni? ném AUX go 'he went'
- (d) łíċ-ət-əs cut-TR-3TR 'he cuts it'

- (a') nə-s-ném my-NOM-go 'my going'
- (b') ?əθċćwəθàmx ?əT-s-ċćw-ət-Samx your-NOM-help-TR-me 'your helping me'
- (c') s-ni?-s ném NOM-AUX-3POS go 'his having gone'
- (d') s-łíċ-ət-s NOM-cut-TR-3POS 'his cutting it'

In each of these pairs, the possessive in the nominalized clause corresponds to the subject marker in the main-clause form (see §4.3 for the uses of nominalized clauses).

The possessives cannot be affixed to nominalized passive forms. This rule is consistent with the fact that the subjects of main-clause passives are not marked by the same elements that occur with main-clause non-passives. (See §14.2.6 below.)

Most importantly, the possessives also cannot be affixed to any non-passive verb that has not been nominalized. For example, \*\**noném* 'my go,' \*\**nocéwat* 'my help him,' and so on are not possible forms. (For exceptional cases, see §8.9.)

#### 14.2.5. The Object Person Markers

These are the suffixes:

| $-S \sim -Samx \sim -amx$ 'me' | $-alx^{w}$ 'us'        |
|--------------------------------|------------------------|
| <i>-S ~ -Samə ~ amə '</i> you' | <i>-alə</i> 'you (pl.) |

These forms appear to be composed of identifiable elements, -S and -am-'non-third-person singular,' -al- 'non-third-person plural,' -x and  $-x^{w}$  'first person,' and -a 'second person.' The third person is not marked.

An object person marker can be suffixed to a transitive verb only, that is, to a verb stem that has a transitive suffix (see \$10.1.1).

The element -S occurs regularly with the transitivizer -t and with -t only, with which it coalesces as  $/\theta/$ . It may stand alone to mean 'me' or 'you' or be followed by -amx 'me' or  $-am\vartheta$  'you.' Thus with the root  $\dot{c}\dot{e}w$ - 'help' and -t 'transitive' we find:

| $\dot{c}\dot{e}w\partial\theta \sim \dot{c}\dot{e}w\partial\theta amx$ 'help me'         | <i>ćéwətàl x<sup>w</sup> '</i> help us' |
|------------------------------------------------------------------------------------------|-----------------------------------------|
| $\dot{c}\dot{e}w\partial\theta \sim \dot{c}\dot{e}w\partial\theta am\partial$ 'help you' | <i>céwətàlə</i> 'help you (pl.)'        |
| <i>céwət</i> 'help him/her/them'                                                         |                                         |

Such forms are normally accompanied by subject person markers (see Table 14.2). There is, however, one notable hole in the pattern: the third-person transitive subject marker cannot appear with a second-person object marker. At these points in the paradigm, there is a switch to the passive.

| Object-subject paradigm |                              |                |                                       |                  |                              |  |  |
|-------------------------|------------------------------|----------------|---------------------------------------|------------------|------------------------------|--|--|
|                         |                              |                |                                       |                  | Him/                         |  |  |
| Subj/obj                | Me                           | You            | Us                                    | You (pl.)        | her/them                     |  |  |
| Ι                       |                              | céwəθ(àmə) cən |                                       | céwətàlə cən     | céwət cən                    |  |  |
| You                     | ćéwəθ(àmx) čx <sup>w</sup>   |                | ćéwətàlx <sup>w</sup> čx <sup>w</sup> |                  | <i>čéwət čx</i> <sup>w</sup> |  |  |
| We                      |                              | ćéwәθàmә ct    |                                       | céwətàlə ct      | <i>čéwət ct</i>              |  |  |
| You (pl.)               | с́е́wәθàmx се <sup>.</sup> р |                | ćéwətàİx™ ce∙p                        |                  | čéw∂t ce∙p                   |  |  |
| He                      | čéwəθ(àmx)əs                 | (čéwəθà·m)     |                                       | <i>céwətàləm</i> | ćéwətəs                      |  |  |

For this reason, a form with -S alone, unaccompanied by a differentiating -amx or -ama, is not ambiguous. With a second- or third-person subject marker, the -S has to mean 'me,' and with a first-person subject marker, it has to mean 'you,' as in:

 $\dot{c} \epsilon w \partial \theta \, \dot{c} x^w \, c e^2$ . 'You will help me.'  $\dot{c} \epsilon w \partial \theta \partial s \, c e^2$ . 'He will help me.'  $\dot{c} \epsilon w \partial \theta \partial s \, c e^2$ . 'I will help you.'

For 'He will help you,' we find:

*céwəθà·m.* lit. 'You will be helped.' *céwətàləm ce*<sup>2</sup>. lit. 'You folks will be helped.'

often translated 'They will help you' or 'Somebody will help you.'

With other transitive suffixes, the object set lacks the -S that appears with -t. With the root  $k^{w} \acute{e}c$  'look' and  $-n - \sim -n \partial x^{w}$  'limited control,' we find:

| <i>kwəcnámx</i> 'see me'                            | <i>k<sup>w</sup>əcnálx</i> <sup>w</sup> 'see us' |
|-----------------------------------------------------|--------------------------------------------------|
| <i>kwəcnámə</i> 'see you'                           | <i>kwəcnálə</i> 'see you folks'                  |
| $k^w \delta c n \partial x^w$ 'see him/her/it/them' |                                                  |

With  $y \partial w \dot{e}^{\gamma}$  'go along' and  $-st - -st \partial x^{w}$  'causative,' the paradigm is:

| <i>yəwé<sup>9</sup>stàmx</i> 'take me along'                              | <i>yəwé<sup>9</sup>stàİx</i> <sup>w</sup> 'take us along' |
|---------------------------------------------------------------------------|-----------------------------------------------------------|
| yəwei yəwei you along'                                                    | yəwé <sup>9</sup> stàlə 'take you (pl.) along'            |
| $y \partial w \dot{e}^{2} st \partial x^{w}$ 'take him/her/it/them' along |                                                           |
With the root of  $k^{w} \acute{e} l$  (//k<sup>w</sup>él-əl//) 'hide' and the transitivizer -*x*, the paradigm is:

*k<sup>w</sup>élxàmx* 'hide me' *k<sup>w</sup>élxàmə* 'hide you' *k<sup>w</sup>élx* 'hide him/her/it/them' *k<sup>w</sup>élxàlx<sup>w</sup>* 'hide us' *k<sup>w</sup>élxàlə* 'hide you folks'

And with the root  $h\acute{e}\vec{k}^{w}$  'remember' and the transitivizer *-nəs*, we find:

 $hék^w n \partial s amx$  'remember me' $hék^w n \partial s al x^w$  'remember us' $hék^w n \partial s am\partial$  'remember you' $hék^w n \partial s al \partial$  'remember you folks' $hék^w n \partial s$  'remember him/her/it/them'

In every case, however, a third-person subject and second-person object causes a switch to the passive paradigm.

In the progressive aspect, the resonants in the suffixes -amx,  $-am\partial$ , and  $-al\partial$  may be glottalized, as in:

(a) wəyáθ cən wəċéċəwəθàma.

wə-yáθ cən wə-cécəw-ət-Samə Est-always I Est-help(PROG)-tR-you 'I'm always helping you.'

(Or so it appears in the material dictated by CC and JP, although I must admit there are inconsistencies. On the other hand, AG seemed to use a glottalized -ama 'me' and an unglottalized -ama 'you' consistently in both perfective and progressive aspects.)

### 14.2.6. The Passive Person Markers

These mark the subjects in passive forms. (For "passive," see §3.3, "Types of Verbal Predicates.") They are the following suffixes:

| -Sel- ~ -el- 'first-person singular'     | -al- 'first-person plural, second- |
|------------------------------------------|------------------------------------|
|                                          | person plural'                     |
| -Sam- ~ -am- 'second-person singular'    |                                    |
| - <i>ay</i> - 'third-person subordinate' |                                    |

The first- and second-person forms share some components with their counterparts in the object set. The -S- 'non-third-person singular' and the -al- 'non-third-person plural' are identical. But in the passive set the -am- is confined to the second-person singular while another element, -el-, appears as the first-person singular. The first-person plural and second-person plural are not differentiated. The -ay- 'third-person subordinate passive' does not resemble anything else.

Like the object set, these must preceded by a transitivizer. Unlike the object set, they cannot be word-final; they must be followed by a passive marker, either  $-\partial m$  'intransitive' (§10.2.1) in main clauses or  $-\partial t$  'subordinate passive'

(\$10.8) in subordinate clauses and nominalizations. In main clauses, the third person has no marker, but in subordinate clauses and nominalizations, it is marked by  $-\partial y$ -.

These suffixes combine according to several morphophonemic rules. As with the object set, the element -S- coalesces with -t transitive as  $\theta$  but does not appear with the other transitivizers. Moreover, in the coordinate passive, when the second-person singular is followed by  $-\partial m$ , the medial resonant is lost and the vowel lengthened,  $-am - + -\partial m$  becoming  $-a \cdot m$ . In the subordinate passive, there is the kind of metathesis seen also when certain roots take the -t transitive (see §1.5.3) whereby  $-el - + -\partial t$  becomes  $-e \cdot lt$  and  $-al - + -\partial t$  becomes  $-a \cdot lt$ . Finally, the sequence  $-\partial y - + -\partial t$  usually, though not in very deliberate speech, becomes  $-i \cdot t$ . (The corresponding form in the Cowichan dialect is  $-ew \partial t$ .)

The paradigm based on *cewat* 'help him' in the coordinate passive is:

*céwəθeləm* 'I am helped' *céwəθà·m* 'you are helped' *céwətàləm* 'we are helped, you (pl.) are helped' *céwətəm* 'he/she is helped, they are helped'

In the subordinate passive, it is:

wočéwo0e·lt 'if I am helped'
wočéwo0a·m 'if you are helped'
wočéwo1a·lt 'if we are helped, if you (pl.) are helped'
wočéwo1i·t 'if he/she is helped, if they are helped'

The paradigm based on  $k^{w} \delta c n \partial x^{w}$  'see him' in the coordinate passive is:

 $k^{w} a cn \ell am$  'I am seen'  $k^{w} a cn a m$  'you are seen'  $k^{w} a cn a l am$  'we are seen, you (pl.) are seen'  $k^{w} a cn a m$  'he/she is seen, they are seen'

In the subordinate passive it is:

wok<sup>w</sup>ocné·lt 'if I am seen' wok<sup>w</sup>ocnámot 'if you are seen' wok<sup>w</sup>ocná·lt 'if we are seen, if you (pl.) are seen' wok<sup>w</sup>ocní·t 'if he/she is seen, if they are seen'

The passive paradigm differs from the intransitive and active transitive paradigms in that, ordinarily, neither the coordinate subject particles ( $c \partial n$ ,  $c x^w$ , etc.) nor the subordinate subject suffixes ( $-e \cdot n$ ,  $-\partial x^w$ , etc.) play any part in it. In a form like  $c e^w \partial \theta e^l \partial m c e^2$  'I will be helped,' the subject is sufficiently marked by -Sel-. Forms like \*\* $c e^w \partial \theta e^l \partial m c \partial n c e^2$  do not seem to occur. However, in a negative clause with an auxiliary, the auxiliary will be followed by a thirdperson subordinate subject marker (see §6.1.2). The passive is nominalized with *s*-nominalizer prefixed to the subordinate form, as in:

*scéwəθè·lt* 'my being helped' *scéwəθàmət* 'your being helped' *scéwətà·lt* 'our being helped, your (pl.) being helped' *scéwət*à·*tt* 'his/her being helped, their being helped'

As indicated above, the nominalized passive differs from other nominalizations in that it does not take a possessive, forms like  $**nascewa\thetaelt$  evidently being badly formed. This absence of a possessive in the nominalization is consistent with the absence of a subject particle following the main-clause form of the passive.

#### 14.2.7. The Personal Words

These are words as defined in §2.1. They are:

| 1st person | ? <i>ə́nθə ~ ?é·nθə '</i> I'       | łníməł 'we'               |
|------------|------------------------------------|---------------------------|
| 2nd person | náwa 'you'                         | <i>łwálap</i> 'you folks' |
| 3rd person | $\dot{\chi} \dot{a}$ 'he, she, it' | <i>Žáləm</i> 'they'       |

The plural forms are partly analyzable. The *t*- of the first-person plural and the second-person plural may be *t*- 'portion' ( $\S12.4.6$ ). The  $-\partial l \partial p$  of the second-person plural occurs as a suffix with that meaning in the possessive. The *-w*- of  $lw \partial l \partial p$  is conceivably a reduction of  $n \partial w \partial$ . The third-person plural is probably the singular  $\dot{\lambda} \dot{a}$  plus an *-l*- that appears commonly as a pluralizing infix and a final *-om* that may conceivably be the intransitive suffix.

Like other words (including the possessive words) but unlike the other person markers, the members of this set can function as predicates. They can also, with restrictions not imposed on other words, function as or enter into the formation of adjuncts.

The first-person forms  $2\delta n\theta \partial$  and  $2\epsilon n\theta \partial$  (the latter generally so recorded from JP and CC but recorded  $2\epsilon n\theta \partial$  from AG) both appear as predicates. (AG suggested that the form  $2\epsilon n\theta \partial$  is emphatic, but data from JP and CC are not sufficient to support this. JP thought that the difference was a matter of dialect or idiolect.) Only the form  $2\epsilon n\theta \partial$  (or  $2\epsilon n\theta \partial$ ) appears as an adjunct.

#### 14.2.7.1. As Predicate Heads

Standing alone or with predicate particles, these words usually have the sense of 'It's me,' 'It's you,' and so on, or 'I'm the one,' 'You're the one,' or, in answer to questions, 'I will,' and so on.

Followed by relative clauses, in a type of pseudo-cleft sentence (§4.1.5.1.2), they are usually translated 'I am the one who ...,' 'You are the one who ...,' and so on.

- (a) <sup>9</sup>όnθ cc<sup>9</sup> ném. (JP)
   be.I FUT go
   'I'm the one that will go.'
- (b) <sup>2</sup>é·nθə k<sup>\*</sup>ámk<sup>\*</sup>əm.
  be.I strong
  'I'm the one who's strong.'
- (c) <sup>9</sup> in θ i c-yá·ỳ s.
  be.I AUX get-be.working
  'I'm the one that's doing it.'
- (d) nəwé <sup>γ</sup>e ce<sup>γ</sup> θáyt. <sup>γ</sup>ánθə ce<sup>γ</sup>. (AG) náwə <sup>γ</sup>ə ce<sup>γ</sup> θáy-t <sup>γ</sup>ánθə ce<sup>γ</sup> be.you ROG FUT fix-TR be.I FUT 'Will you be the one who'll fix it? I will.'
- (e) łníməł ce? nem kwónət kwθə s?ítθəms.
   be.we FUT go take-TR ART clothing-3POS 'We will be the ones to go get his clothes.'
- (f) łwólap ce? hóỷq<sup>w</sup>-alcap.
  be.you(PL) FUT be.burning-firewood
  'You people will be the ones to look after the fire.'
- (g) Ža ni ł ném. (JP) Ža ni ²- əł ném BE3P AUX-past go 'He is the one who went.'
- (h) Žáləm nə-méməñə. (CC)
   be.they my-children
   'They are my children.'
- (i) Žáləm ni k wónət k wθə télə. (AG) be.they AUX take-TR ART money
   'They are the ones who took the money.'

The third-person singular  $\lambda a$ , perhaps usually unstressed, can take an adjunct, which is either introduced by or consists of a deictic element, as a grammatical subject. If there is no more than a subject, the  $\lambda a$  serves simply to put it in focus:

(j) X̂a k<sup>w</sup> śná. (AG)
 BE3P this
 'He [just out of sight] is the one.'

(k) えa tə nə-məndə. (CC) BE3P ART my-child 'It's my son.'

However, if there is a subject and a relative clause, the  $\dot{x}a$  then functions as a copula, linking the two as subject and complement in a cleft sentence (§4.1.5.2).

The complement (i.e., the relative clause) may follow the  $\dot{\chi}a$  directly and be followed by the subject (i.e., the adjunct), as in (l) and (m).

(l) Ža ni·ł céwəθàmx tθe?. х́а ni?-əł céw-ət-Sàmx tθé? BE3P AUX-past help-TR-me that 'He is the one who helped me.' (m) Xa nəswé? nəpúk tə?ínə. (CC) х́а na-swé? nə-púk ta?ína BE3P mv-own my-book this 'This is my book.' Or the complement may follow the subject, as in (n) and (o). (n) Åa ma ta?í sqwaméy ni díkwaθ. (CC) díkw-ət-S Х́а mэ tə?í sq<sup>w</sup>əméỷ ni? this dog get.bitten-TR BE3P CERT AUX me 'This is the dog that bit me. It's this dog that bit me.' (o) Xa mo to Andrew cwé? to sqwoméy. (CC) х́а 'nэ tə Andrew c-wé? [?ə] sq<sup>w</sup>əméỷ tə BE3P CERT ART Α get-own OBL ART dog 'It's Andrew who owns the dog.'

# 14.2.7.2. As Adjuncts

The personal words can also function as adjuncts. As an adjunct, a first-person or second-person word is preceded by an article. The third-person counterparts, however, are a set of words formed by a coalescence of the third-person words with articles and the prefix  $wa_{-2}$ . This set is described in §15.2.2.2. The present section will describe the uses as adjuncts of first- and second-person words only.

As direct adjuncts, the first- and second-person words usually relate anaphorically to person markers accompanying the verb, and they perform one or both of two functions that cannot be performed by the person markers. They give emphasis to the person (there being no emphatic forms of the person markers) and/or they provide a peg on which to hang a relative clause.

In (a) and (b) they serve to give emphasis:

(a) <sup>9</sup>ówə čx<sup>w</sup> ł ni dáy tə nówə i ... (JP 9) ?∕áwə čxw ł ni °∂ỷ dáy tə náwə not vou ART die and AUX ART vou 'You yourself don't die and ...' (b) niwł Xám k<sup>w</sup>θə ni θdéls məstə́yəx<sup>w</sup> <sup>9</sup>ə šx<sup>w9</sup>ə́wəct ni·t ném dáθət tə łníməł. (JP 22) ni? wəł-Żám k<sup>w</sup>θa ni? θəq-éls məstáyəx<sup>w</sup> AUX already-be.enough ART AUX pierce-ACT people ?a šx<sup>w</sup>-?śwə-ct ni?-ət ném dá?-0ət tə OBL OBLNOM-not-our accompany-self AUX-we go ART łnímał be we 'There were enough people who had harpooned [it], and so that's why we ourselves didn't go along.' In (c) to (f), the personal words serve as heads of relative clauses: (c) ... k<sup>w</sup>snict wətá<sup>9</sup>tələt <sup>9</sup>al tə łníməł <sup>9</sup>əwə niət ném x<sup>w</sup>əsdədá<sup>9</sup>. (JP 22) s-ni<sup>9</sup>-ct wə-tá<sup>9</sup>təl-ət ?aĺ k w tə łníməł ?śwə NOM-AUX-our EST-be.learning-TR just be we ART ART not ném x wə-sdədá? ni?-ət become-accompanied AUX-we go "... when we who didn't go along were just studying it." (d) tsás ct tə łníməł ni ə k<sup>w</sup>θə ni·n šx<sup>w</sup>ní?. (JP 3) ?a tsás łnímał ni? k »θə ni? ct tə poor we ART be we be.there OBL ART be there nə-šx<sup>w</sup>-ní? my-OBLNOM-be.there 'We are poor, we who are at my place.' (e) <sup>9</sup>ówə ni·s mélqtàləm tə łwóləp təli·  $\vec{k}$ <sup>w</sup>i šx<sup>w</sup>ní<sup>9</sup>ct. (JP 5) ?∕awə ni?-əs mélq-t-àlə-m təli tə łwśləp forget-TR-you-INTR not AUX-3SUB ART be.you from κ<sup>w</sup>i ?a šx<sup>w</sup>-ní?-ct that OBLNOM-be.there-our OBL 'He didn't forget you people from our place.' (f) Xe cən wəłqəlét cí tàlə tə łwóləp nəmémənə. (JP 8) х́е cən wəł-qəlét ćí-ət-àlə tə łwślap I too already-again thank-TR-you art be.you(PL) nə-mémənə my-children 'I'll thank you all again, you who are my children.'

Second-person words can also appear without co-referential person markers when used in a vocative sense as in (g), as perhaps they were also in (e) and (f).

(g) wənéxən <sup>9</sup>al tθé<sup>9</sup> k<sup>w</sup>ən sq<sup>w</sup>él, tə náwə nəswənméyəł. (JP 8) ?aĺ [?ə] tθé? k<sup>™</sup>ə nə-s-q<sup>w</sup>él, wə-néxən tə EST-end iust OBL that ART my-NOM-speak art nə-swənméyəł náwa my-deceased.sibling's.child be vou 'My words just end there, my niece.'

As oblique adjuncts, the personal words also perform a function that cannot be performed by any of the personal affixes or particles, relating to the verbs they follow as oblique objects. As oblique adjuncts, they require the oblique article  $\vec{x}$  but the oblique case marker 2a may (at least in JP's and CC's idiolects) be omitted.

- (h) x<sup>w</sup>əňíwən čx<sup>w</sup> <sup>2</sup> → λ <sup>2</sup> é·nθə. (CC) x<sup>w</sup>əňíwən čx<sup>w</sup> <sup>2</sup> → λ <sup>2</sup> é·nθə remember you OBL ART be.I 'Remember me.'
  (i) tə<sup>2</sup>á·məx <sup>2</sup> → k<sup>w</sup>ə təwka λ <sup>2</sup>é·nθə. (CC)
- tə?á·məx ?ə k<sup>w</sup>ə təwλa [?ə] λ ?é·nθə resemble ROG then he [OBL] ART be.I 'Does he look like me?'

Often the immediately preceding verb has the function of an English preposition, giving a more specific direction to the action expressed by the main verb (see § 3.5).

(j) wélx čx<sup>w</sup> x<sup>w</sup>əmi  $\lambda$  <sup>9</sup>é·n $\theta$ ə. (CC) x<sup>w</sup>-?əmi ż ?é•nθ⊋ wél-x čxw [?ə] throw-TR LOC-come be.I vou OBL ART 'Throw it to me.' (k) wélx cən ce? nem ?ə 式 nówə. (AG) ?ə ż wél-x cən ce? nem náwə throw-TR I FUT go OBL ART be.you 'I'll throw it to you.' (l) təwk<sup>w</sup>í·lə təli X łwśləp (JP 22) tawk<sup>w</sup>í·la təli [?ə] Χ łwśləp several.persons from be.you(PL) OBL. ART 'several of you'

As oblique adjuncts, the personal words of the first and second persons can function as oblique objects only. They cannot function as agents of passives.

Sentences like \*\**ni qá·ytəm ?* $\partial \vec{X}$  ? $\acute{e}\cdot\partial \partial$  'He was killed by me' or \*\**ni qá·ytəm* ? $\partial \vec{X}$  *nówo* 'He was killed by you' are not possible.

## 14.2.8. The Possessive Words

The fact of possession may be expressed by two words:  $sw\acute{e}^{2}$  'one's own' (pl.  $swaw\acute{e}^{2}$ ), a nominalization of  $\sqrt{we^{2}}$  'own,' to which the possessive affixes other than *-ct* 'our' are attached; and  $s^{2}\acute{a}\cdot\acute{t}$  (CC, JP) or  $s^{2}\acute{a}\cdot\acute{t}$  (AG) 'ours,' \*\* $sw\acute{e}^{2}ct$  being rejected.

The paradigm is:

| <i>nəswé</i> <sup>2</sup> 'it's mine'                                 | $s^2 \dot{a} \cdot t \sim s^2 \dot{a} \cdot 2t$ 'it's ours'                                                                       |
|-----------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------|
| $^{?}\partial\theta w e^{?} \sim ^{?}\partial s w e^{?}$ 'it's yours' | $^{?}\partial\theta w \dot{e}^{?}\partial l \partial p \sim ^{?}\partial s w \dot{e}^{?}\partial l \partial p$ 'it's yours (pl.)' |
| swé <sup>9</sup> s 'it's his/hers/its'                                | swé <sup>2</sup> s ( <sup>2</sup> é·łtən) 'it's theirs'                                                                           |

Some anomalies also occur in the way these words are used with nouns possessed. With the first-person singular and the second person, the prefix is repeated on the noun. With the third person, CC and JP did not repeat the suffix, while AG did. With the second-person plural, the suffix is not repeated.

```
nəswé² nəsq^wəméy. 'It's my dog.'
²∂θwé² ²∂θq^wəméy. 'It's your dog.'
swé²s sq^wəméy. (CC, JP) swé²s sq^wəméys. (AG) 'It's his dog.'
s²á·ł sq^wəméy. 'It's our dog.'
²∂θwé²∂ləp ²∂npút. 'It's you guys's boat.' (AG)
```

As elsewhere, the third-person plural particle does not appear if plurality is indicated by other means. Compare (a) and (b):

| (a) | swé?-s     | ?é∙łtən       | léləm-s          | tə         | ởớq. (AG)  |
|-----|------------|---------------|------------------|------------|------------|
|     | own-3POS   | 3PL           | house-3POS       | ART        | white      |
|     | 'The white | house is thei | rs. The white or | ne is thei | ir house.' |
|     |            |               |                  |            |            |

| (b) | swé?-s        | tθéləy     | tə  | léləm. (AG) |
|-----|---------------|------------|-----|-------------|
|     | own-3POS      | those/they | ART | house       |
|     | 'The house is | theirs.'   |     |             |

# 14.2.9. The Third-Person Plural Particle

This is the particle 2ettan, which may be classed with the predicate particles, although it is unusual in its position. It is probably composed of a base, not now identifiable, and the suffix -attan 'others' (see §10.2.4). It indicates plurality in a third-person subject, object (when the subject is other than third person), or possessor, when plurality is not indicated by other means.

When  $\frac{2}{e}$  to pluralizes the subject or object, it appears last in the predicate, as in (a) to (d).

- (a) ni? ném ?é·łtən. AUX go 3PL 'They went.'
- (b) wə-ném-əs ?é-ltən if-go-3SUB 3PL 'if they go'
- (c) k<sup>\*</sup>wocnàmxəs ce? ?é·łtən. k<sup>\*</sup>ec-n-amx-əs ce? ?é·łtən see-TR-me-3TR FUT 3PL
   'They will see me.'
- (d) ném cən ce? cew-ət ?e.ltən. go I FUT help-TR 3PL 'I'm going to help them.'

In a sentence with a third-person subject and a third-person object,  ${}^{?}\acute{e}tt \partial n$  must refer to the subject, as in (e).

(e) ni? k<sup>w</sup>él-x-əs ?é·łtən. AUX hide-TR-3TR 3PL 'They hid it.' (not 'He hid them.')

When  $\frac{\partial e}{\partial t}$  pluralizes a possessor, it follows the possessive person marker, as in (f) and (i).

| (f)                          | х́а  | sìl-éwtx <sup>w</sup> -s | ?é∙łtən | tə  | sála?ac.  |
|------------------------------|------|--------------------------|---------|-----|-----------|
|                              | BE3P | canvas-house-3POS        | 3PL     | ART | house.mat |
| 'The mats were their tents.' |      |                          |         |     |           |

If plurality is indicated by a plural noun, then  $2\acute{e}\cdot ttan$  cannot be used. Compare (g) and (h) and (i) and (j):

| (g) | ni?       | ἀá·y-t-əm       | ?é∙łtən. |                                        |
|-----|-----------|-----------------|----------|----------------------------------------|
|     | AUX       | die-tr-intr     | 3PL      |                                        |
|     | 'They     | were killed.'   |          |                                        |
| (h) | ni        | ἀá·y-t-əm       | tə       | sq <sup>w</sup> əmq <sup>w</sup> əméy. |
|     | AUX       | die-tr-intr     | ART      | dogs                                   |
|     | 'The de   | ogs were killed | l.'      | -                                      |
| (i) | х́а       | léləm-s         | ?é∙łtəı  | 1.                                     |
|     | BE3P      | house-3POS      | 3PL      |                                        |
|     | 'It's the | eir house.'     |          |                                        |

| (j) | х́а     | léləm-s         | tə      | x <sup>w</sup> ələnítəm | təní?. |
|-----|---------|-----------------|---------|-------------------------|--------|
|     | BE3P    | house-3POS      | ART     | white.people            | that   |
|     | 'That o | ne there is the | White p | eople's house.'         |        |

But neither \*\* $ni^{2} \dot{q}\dot{a}\cdot yt\partial m^{2}\dot{e}\cdot t\partial n t\partial sq^{w}\partial m\dot{e}\dot{y}$  nor \*\* $\dot{X}a \ l\acute{e}l\partial ms^{2}\dot{e}\cdot t\partial n t\partial x^{w}\partial l\partial n(t\partial m)$  is possible.

# **15** The Demonstrative System

Under this heading are included the various deictic elements that introduce nominal adjuncts or else stand as nominal adjuncts. These elements are the articles and the demonstratives. The latter include three sets of words that are either fuller forms of the articles or forms compounded of the articles and other elements. Together the articles and demonstratives form a system that pervades the language.

### 15.1. THE ARTICLES

These are the unstressed particles that stand initially in nominal adjuncts ( $\S3.4$ ) and adverbial adjuncts ( $\S3.6$ ). Some examples of usage will be found in sections describing each of these, where the focus is on syntactic functions. Other examples appear below, where the focus is on the features of meaning the articles express.

With one exception, the articles may be seen (as in Table 15.1) as a set structured by the intersection of two genders and three positions or degrees of proximity. The genders are non-feminine (more accurate than "masculine" but under M below) and feminine (F). The positions are present and visible (p); nearby, or somehow accessible, but invisible (n); and remote, out of reach, or non-existent – as deceased persons or hypothetical events (r). The exceptional article appears with oblique adjuncts of a few types only and does not express gender or positions.<sup>1</sup>

<sup>1</sup> It seems possible that historically the articles were formed from separate elements, perhaps *t*-, *s*-, *k*<sup>w</sup>, *t*-, and glottalization. Viewed synchronically, however, the set cannot be easily analyzed into meaningful components.

| The articles               |                                               |                                            |                |
|----------------------------|-----------------------------------------------|--------------------------------------------|----------------|
|                            | Non-feminine (M)                              | Feminine (F)                               | Oblique (obl.) |
| Present and visible (p)    | $t \partial (t^{	heta} \partial)$             | heta  angle                                |                |
| Nearby and invisible (n)   | $k^w 	heta \partial$ , $k^w \partial$ , $k^w$ | łə, k <sup>w</sup> łə, ł, k <sup>w</sup> ł | Ż              |
| Remote or hypothetical (r) | К <sup>w</sup> ə, К <sup>w</sup>              | k <sup>w</sup> sə                          |                |

Table 15.1

The table notes one variation in form that is the result of Cowichan influence at Musqueam. The usual Musqueam non-feminine present article is  $t\partial$ , but JP and DK occasionally used  $t^{\theta}\partial$  (phonetically [ $t^{\theta}\partial$ ]), which is the usual Cowichan form. There are also said to be variations in usage that are the result of Upriver influence (Elmendorf and Suttles 1960, 12).

The glosses indicated by the table, non-feminine present (MP), feminine present (FP), and so on, are no more than approximations. All of the distinctions that may be conveyed by choices of articles have yet to be fully analyzed, as the following account of gender and position will show.

### 15.1.1. Gender

This distinction is relevant when the article is used with a noun or a relative clause; nominalizations and adverbial phrases take only non-feminine forms. The distinction is "natural" rather than "grammatical," that is, whether a Halkomelem speaker uses a non-feminine article or demonstrative with a noun or relative clause depends, as does an English speaker's choice of pronoun, on facts external to the grammar. Halkomelem nouns, like English nouns and unlike Chinook or French nouns, do not fall into grammatical classes of masculine and feminine.

When the reference is to persons, the articles distinguish males and females. Compare:

(a) tə nəmáňə 'my son (here)'

- (b) θ anamána 'my daughter (here)'
- (c) tə ni ném 'the (male) one who went'
- (d) θ ni ném 'the (female) one who went'

This distinction of males and females through the articles (and demonstratives) is probably essential to the functioning of the kinship terms (see §21), which nearly all, like  $m \dot{n} \dot{n} \dot{a}$  'child,' make no sex distinction in themselves. In fact, in the total lexicon there may be no more than seven words that can refer to persons of one sex only. These are:  $sw \dot{s} \dot{y} q e^2$  'man/male,'  $s \dot{t} \dot{e} n \dot{s} \dot{y}$ 'woman/female,'  $sw \dot{t} \dot{w} l \dot{s} s$  'young man,'  $\dot{q} \dot{e} \dot{m} \dot{s} \dot{y}$  'young woman,'  $m \dot{e} n$  'father,' *tén* 'mother,'  $\delta x^{w} \ell \partial x$  'woman's sister-in-law.' The indication of sex through the article is redundant for these seven words only.<sup>2</sup>

But this distinction can be made for singular males and females only. The non-feminine is used for plurals of either sex and for female as well as male persons when referred to in a generic since. Compare:

- (e) tə *txámələ słənłénəy* 'six women, the six women'
- (f)  $\theta \partial s t \dot{e} n \partial y$  'a woman, the woman'
- (g) i ha? ni? tə déməy ... (CC 19) and if be.there ART(MP) girl 'and if there was any girl who ...'
- (h) θə ἀéməỷ 'a girl, the girl'

The non-feminine article is also used with personal words regardless of the sex of the referent:

(i) Ža k<sup>w</sup> ládjellex<sup>w</sup>e<sup>n</sup> te<sup>γ</sup>é<sup>n</sup>θe, Mrs. Andrew Charles. (CC 12)
Ža k<sup>w</sup> ládjel-lex<sup>w</sup>-e<sup>n</sup> te<sup>γ</sup>é<sup>n</sup>θe
BE3P then know-TR-I ART(MP) I
'That is what I, Mrs. Andrew Charles, know.'

With words referring to animals, in some elicited sentences the article seem to refer to sex, but in narratives it seems that non-feminine forms are used for larger animals and feminine forms for smaller ones. However, the distinction between the generic and the specific may also determine choices; in an account of sturgeon harpooning, JP used non-feminine articles for sturgeon in discussing general principles but switched to feminine forms when describing a specific kind of situation.

With words for inanimate things, it also seems that feminine forms are preferred for smaller things, but there may well be other considerations that I have not yet detected.

#### 15.1.2. Position

When the reference is to tangible things, usually the difference between present and near forms seems to be a matter of proximity and visibility. Compare (a), (b), and (c):

<sup>2</sup> The root of  $\delta x^{w^2} \ell l \partial x$  'woman's sister-in-law' is  $2\ell l \partial x$  'sibling of the opposite sex,' i.e., a man's sister or a woman's brother. For this word, too, perhaps the gender of the article is redundant because the sex of the referent may be inferred from the sex of the speaker.

(a) ni <sup>9</sup>źnącą łą nawáč. (CC) ni? ?ánaca nə-wáč łə be.there where mv-watch ART(FN) 'Where is my watch?' (b) <sup>9</sup>í wə<sup>9</sup>í tə nəsłəq<sup>w</sup>ə<sup>9</sup>élə łə nəwáč. (CC) ?í wə-?í [?ə] tə nə-słəq<sup>w</sup>ə<sup>9</sup>élə łə be here EST-be here my-pocket OBL ART(MP) ART(FN) nə-wáč my-watch 'My watch is here in my pocket (on me).' (c) now ní? ło nosłoq<sup>w</sup>o?élo ło nowáč. (CC) ni? wə-ní? [?ə] łə nə-słəq<sup>w</sup>ə<sup>9</sup>élə be there EST-be there OBL my-pocket ART(FN) nə-wáč łə ART(FN) mv-watch 'My watch is there in my pocket.'

In the situations given in all three of these sentences, my watch is out of sight and so marked by a nearby but invisible form. In (b), my pocket is in the jacket I am wearing and so my pocket is marked by a present and visible form. And in (c), my pocket is in my jacket somewhere else and so my pocket marked by a nearby but invisible form. My watch is consistently marked feminine, perhaps because it is small. But my pocket is non-feminine in the second sentence and feminine in the third. I do not know the reason for this change in gender.

Position is of course also indicated by the verbs 2i 'be here' and  $ni^2$  'be there,' which locate the speaker. This is seen in the previous sentences and in (d), (e), and (f).

(d) ni? sk<sup>w</sup>téx<sup>w</sup> tə léləm k<sup>w</sup>θə sq<sup>w</sup>əméy. (CC) ni? sk <sup>w</sup>téx <sup>w</sup> [?ə] léləm sq<sup>w</sup>əméỷ tə k»Өә be,there inside OBL ART(MP) house ART(MN) dog 'The dog is in the house.' (e) ni<sup>9</sup> nem k<sup>w</sup>ətx<sup>w</sup>íləm k<sup>w</sup>θə sq<sup>w</sup>əméy. (CC) nem k<sup>w</sup>ətx<sup>w</sup>íləm k<sup>w</sup>θə sq<sup>w</sup>əméỷ ni AUX(be.there) enter ART(MN) dog go 'The dog went in.' (f) <sup>9</sup>i mi <sup>9</sup>áxqəl tə sqwəméy. (CC) ?i ?əmí ?áẳqəl sq<sup>w</sup>əméỷ tə AUX(be.here) exit come ART(MP) dog 'The dog came out.'

In the situations given in all three of these sentences, the speaker is outside the house. This is indicated by the choice of ?i or ni? in all three sentences as well as by the choice of ?omi or nem in (e) and (f). (In the English, the choice of *come* or *go* would establish the position of the speaker in [e] and [f], but *is* in [d] would not.) In (d) the dog is invisible, being inside the house while the speaker is outside. In (e), he is invisible, having gone in. But in (f), he is visible, having come out. The house is, of course, visible.

Although the choice of present or near forms of the article is usually determined by actual position, as in the examples just given, it is also possible to use a near form to indicate former absence. For example, upon recovering a lost pocketknife, one might – even while holding it – say:

(g) Xa m v k v l n n h j k v. (CC)
Xa m k v l n n - h j k v
BE3P CERT ART(FN) my-pocketknife
'It's my (lost) pocketknife.'

On the other hand, in storytelling the narrator usually uses the present forms for persons and things that are "present" in the narrative, but there may be switching to convey other information, such as the deceased status of a character (see below).

The near forms have variants. The non-feminine near forms are  $k^w \theta \partial_i k^w \partial_i$ and  $k^w$ . When the reference is to persons and things,  $k^w \theta \partial_i$  is more usual but  $k^w \partial_i$ occasionally occurs. With nominalized clauses used as nominal adjuncts (§4.3.2), the form is more often  $k^w \partial_i$  or  $k^w$ .

The feminine near forms are  $l \partial$  and  $k^w l \partial$ . The first occurs more commonly. The second may imply greater distance or inaccessibility. Compare (h) and (i):

| (h) | n i?<br>be.there<br>'My dat | s?éẳq<br>e outside<br>ughter is ou | łə<br>e ART(FN)<br>ıtside.' | nə-mə'n<br>my-child  | ə. (CC)           |                |
|-----|-----------------------------|------------------------------------|-----------------------------|----------------------|-------------------|----------------|
| (i) | ni                          | n i <sup>9</sup>                   | Ź                           | sť <sup>θ</sup> áməs | k <sup>w</sup> łə | nə-mə́nə. (CC) |
|     | AUX                         | be.there                           | ART(OBL)                    | Victoria             | ART(FN)           | my-child       |

CC said that in the first sentence there would be no choice in the article, but in the second there is a choice between  $l_{\partial}$  and  $k^{w}l_{\partial}$ . If she had been speaking of a son, she said, in both sentences the form would be  $k^{w}\theta_{\partial}$ .

The remote forms,  $k^w \partial$  and  $k^w s \partial$ , are used with persons to show they are deceased, as in:

 $\vec{k}^{w}\partial n\partial s t \vec{l}\partial$  'my late grandfather/great uncle'  $k^{w}s\partial n\partial s t \vec{l}\partial$  'my late grandmother/great aunt'

'My daughter is in Victoria.'

However, deceased status may also be indicated by using the near forms while adding the particle  $-\partial t$  'past':

 $k^{w}\theta \partial n \partial s \hat{l} e \cdot l$  'my late grandfather/great uncle'  $l \partial n \partial s \hat{l} e \cdot l$  'my late grandmother/great aunt'

The feminine remote form  $k^{w}s\partial$  seems to be used for persons only. The non-feminine  $k^{w}\partial$  or  $k^{w}$ , however, has several other uses.

In reference to location in space, t a and  $k^w a$  contrast in distinguishing simply what is visible to the speaker and what is invisible, around in back, around the corner, and so on, as in (j) to (n).

(j) ni tə nə<sup>γ</sup>éθəqən ni? [?ə] nə-?é0əqən tə be there OBL my-front ART(MP) 'in front of me (in a canoe)' (lit. 'there at my front') (k) ni k<sup>w</sup>ə nəsłi<sup>9</sup>á<sup>9</sup>aq<sup>w</sup>t ni? [?ə] k<sup>™</sup>ə nə-słi?á?agwt be there OBL ART(MR) my-rear 'behind me (in a canoe)' (lit. 'there at my rear')

(l) ném cən tə spółxən.

| ném    | cən        | [?ə]      | tə             | spółžən         |
|--------|------------|-----------|----------------|-----------------|
| go     | Ι          | OBL       | ART(MP)        | prairie/pasture |
| ʻI'm g | oing to tl | ne pastur | e (not far and | l visible).'    |

(m) ném cən kwə spółxən.

CC explained that if the pasture is known to the listener, instead of (m), one might say (n).

(n) ném cən k<sup>w</sup>θə spółxən.

ném cən [?ə] k<sup>w</sup>θə spółxən go I OBL ART(MN) prairie/pasture 'I'm going to the pasture (out of sight).'

The possibility that  $k^w \theta \partial$  and  $k^w \partial$  differ in their involvement of the listener in other contexts deserves further consideration.

In reference to time, the contrast is between the present and the past. Compare the following, which appear as adverbial adjuncts: today'
k<sup>\*</sup> coléqoł 'yesterday'
k<sup>\*</sup>oňa łowołné? 'the day before yesterday'

(The demonstratives  $t \partial n a$  and  $k^w \partial n a$  are formed with  $t \partial$  and  $k^w \partial$ , as discussed below.)

The article  $\vec{k}^{w} \partial$  is also used for unknown and/or hypothetical things, as in (o), (p), and (q).

- (o) nəsží? kwə páy. (CC) nə-s-c-ží? kwə páy my-NOM-do-value ART(MR) pie 'I want some pie.'
- (p) tə?á·məx k<sup>w</sup>ə swóyqe?. (CC)
  tə?á·məx [?ə] k<sup>w</sup>ə swóyqe?
  resemble OBL ART(MR) male
  'It looks like a man.'

| (q) | ?áỷ k™  | s némct sáv    | vật kử pśweż    | tə cáửcəử. | (AC 1)                |                |
|-----|---------|----------------|-----------------|------------|-----------------------|----------------|
|     | ?áỷ     | k <sup>w</sup> | s-ném-ct        | sáwd-t     | <b>k</b> <sup>w</sup> | <b>p</b> ́áwaỷ |
|     | good    | ART(MN)        | NOM-go-our      | seek-TR    | ART(MR)               | flounder       |
|     | [?ə]    | tə             | cáwcəw          |            |                       |                |
|     | OBL     | ART(MP)        | offing          |            |                       |                |
|     | 'We'd b | etter go loo   | k for flounders | offshore.' |                       |                |

The article  $k^{w} \partial$  also appears with nominalizations referring to hypothetical events, as in (r).

(r) nəskí? kwə nəsném.
nə-s-c-kí? kwə nə-s-ném
my-NOM-do-value ART(MR) my-NOM-go
'I want to go.' (lit. 'What I want is my hypothetical going.')

(See also §4.3, "Nominalized Clauses.")

Still another use of  $k^w \partial$  is to identify words as quoted terms presumably unfamiliar to the listener, as in (s), (t), and (u).

- (s) k<sup>w</sup>ík<sup>w</sup>əxtəm k<sup>w</sup>ə méθəł. (JP 1)
   is.being.named ART(MR) méθəł
   'It is called méθəł [possibly Indian hemp, Apocynum cannabinum].'
- (t) tə  $k^{w}ik^{w}$ əx təm  $k^{w}$ ə sésqec. (JP 2) ART(MP) is.being.named ART(MR) sésqec 'that which is called a "sasquatch"

 (u) səw ni<sup>9</sup>s θə žátəstəm swe<sup>9</sup>s stəltaləs k<sup>w</sup>ík<sup>w</sup>əxtəs k<sup>w</sup>ə x<sup>w</sup>əlməx<sup>w</sup> k<sup>w</sup>ə qəlqəlít. (JP 2)

s-wə-ni<sup>?</sup>-s \*áfastam θə swe?-s are.said.to.be NOM-EST-be there-3POS ART(FP) property-3POS stəltaləs[-s ?] x \*álməx \* k Ȓk »əxt-əs k »ə spouses-3POS be.naming-3SUB ART(MN) Indian k<sup>™</sup>ə qəlqəlíł ART(MR) aəlaəlíł 'Then there were the ones said to be their wives, whom the Indians called

qəlqəlíł.'

In (u), JP used the FP form before  $\check{x}\check{d}istam$  'said to be,' gender presumably taking precedence over the plural 'wives'; he used the MN form before 'people'; and he used the MR  $k^{w}a$  before the name qalqalit (the Basket Ogress), perhaps to mark it as a quoted form. In the narrative that followed, however, he used feminine forms when he referred to her.

From AG, I recorded:

(v) k<sup>w</sup>ək<sup>w</sup>xít ct ce<sup>9</sup> ək<sup>w</sup> θəhá·yθən.
 k<sup>w</sup>ək<sup>w</sup>xi-t ct ce<sup>9</sup> ?ə k<sup>w</sup>ə θəhá:yθən name(DUR)-TR we FUT OBL ART(MR) big.mouth
 'We shall call [keep calling] him "Bigmouth."

Here the oblique particle  ${}^{2}\partial$ , which CC and JP often omitted, shows that these quoted-term phrases ( $\vec{k} * \partial m \dot{\partial} \theta \partial t$ , etc.) are syntactically oblique adjuncts, syntactically like adverbial adjuncts.

However, when a quoted term is presumably familiar to the listener, the article may be  $t_{\partial}$  (for examples, see §17.15).

In adverbial adjuncts (see §3.6), the article  $k^{w}\partial$  appears before adjectives, as in (w).

(w) m<sup>3</sup> <sup>9</sup>ážqəl k<sup>w</sup>ə x<sup>w</sup>ám. (JP 24)

?əmí ?óẳqəl [?ə] kwə xwám
come exit OBL ART(MR) fast
'He came out right away.'

So far, in nearly all of the examples given to illustrate how the articles distinguish position, they have preceded nouns. The same distinctions are made when the articles precede relative clauses, as in (x), (y), and (z).

(x) stém k<sup>w</sup>ə tə k<sup>w</sup>ənétəx<sup>w</sup>.

| stém  | k <sup>w</sup> ə | tə        | k "ən-é-t-əx " |
|-------|------------------|-----------|----------------|
| what  | then             | ART(MP)   | get-DUR-TR-you |
| 'What | are you          | holding?' |                |

- (y) ni cən kwócnəx kwoə ni·nəł céwət. (JP)
  ni? cən kwóc-nəx kwoə ni?-ən-əł céw-ət AUX I see-TR ART(MN) AUX-I-past help-TR 'I saw the one I had helped.'
- (z) wét k<sup>w</sup> a ni céc awaθàma. (JP)
   wét k<sup>w</sup> a ni? céc awat-Sama
   who ART(MR) AUX be.helping-TR-you
   'Who helps you (when you are away at work)?'

In (x), what is being held is visible; in (y), the one who had been helped is not visible; and in (z), the speaker is not familiar with the situation and the helper is unknown.

What appear to be the FN forms t and  $k^w t$ , with variants  $2 \delta t$  (CC, DK) and  $k^w \delta t$  (DK), can appear after  $2 \delta w \delta$  'not' (§6.1.4) and  $2 \delta w \delta t e^2$  'none' (§6.3) followed directly by a relative clause or followed by the interrogative words  $w \delta t$  'who' (§17.1) and  $s t \delta m$  'what' (§17.3) followed by a relative clause, as in (aa) and (bb).

- (aa) <sup>9</sup>ówəte<sup>9</sup> k<sup>w</sup>ł <sup>9</sup>ítət. (JP 25)
  none ART sleep
  'Nobody slept. There was no one who slept.'
- (bb) <sup>?</sup>ówəte<sup>?</sup> k<sup>w</sup>ł wét k<sup>\*</sup>wóc-nəx<sup>w</sup>-e·n. (AG) none ART who look-TR-I 'I haven't seen anybody.'

There also seems to be a predicate particle  $k^{wl}$  (§16.2.16), perhaps related to this usage of the article in a way not yet clear.

# 15.1.3. The Contrast with English

Some of the examples given above may suggest that there is a tendency for the Halkomelem  $t_{\partial}$  and  $\theta_{\partial}$  to correspond to the English definite article *the* and for the Halkomelem  $k^{w_{\partial}}$  to correspond to the English indefinite article a/an (with a singular count noun), *some* (with a mass noun), or zero (with a plural). But there is really no parallel. In fact, any Halkomelem article can appear in a context where it must be translated as indefinite in English, or in a context where it must be translated as definite. Consider the articles in (a), (b), and (c).

(a) səctəw kwácnəx tə čičí?qan. (CC 8)
 s-ni?-ct wə-kwác-nəx tə čičí?qan
 NOM-AUX-our EST-see-TR ART(MP) mink
 'Then we saw a mink.'

(b) k<sup>w</sup>án-ət-əs k<sup>w</sup>í<sup>9</sup>x<sup>w</sup>. (JP 2) tə ART(MP) pitch get-TR-3TR 'She took (some) pitch.' (c) ném ćə ?á·ł řžám-ala słənłénəy. (JP 4) tə OUOT embark ART(MP) six-person go women

'Six women embarked (it is said).'

In these sentences, the article  $t\partial$  must be translated as indefinite in English because, in the narratives where the sentences appear, the mink, pitch, and six women have not been mentioned before. But they appear all "present" in the viewpoint of the narrator and so must take the article  $t\partial$  in Halkomelem.

The reverse is illustrated by (d) and (e).

- (e) stém k<sup>w</sup>ə sk<sup>w</sup>íxs ti<sup>?</sup>í. (CC) stém k<sup>w</sup>ə sk<sup>w</sup>íx-s tə<sup>?</sup>í what ART(MR) name-3POS this 'What do you call this? What is the name of this?'

The conditional clause (d) is from a statement made by Scouring Rush to Mink warning him of the dangers of standing beside him in the water. The log has not been previously mentioned in the narrative and so the article must be indefinite in English. The log is real though not yet visible to Scouring Rush and so takes the Halkomelem  $k^w \theta \partial$  non-feminine near article. The more literal English translation of (e) requires the definite article, presumably because English speakers take the existence of a name as given even though they do not yet know it, while the Halkomelem sentence requires the  $k^w \partial$  precisely because the name is not yet part of the real world of, or accessible to, the speaker.

However, the demonstratives do seem to be used for persons or things already known and so may function as definite articles. (See §15.2.3.)

# 15.1.4. The Oblique Article

The article  $\vec{X}$ , which is neutral as to gender and position, has the following uses:

(1) It is obligatory before proper names in oblique adjuncts (see §3.4), as in (a), (b), and (c).

(a) čéwətəm ce? え Tom tə Jack. (JP)

| ćéwətəm     | ce?      | [?ə]        | Ż            | Tom       | tə       | Jack |
|-------------|----------|-------------|--------------|-----------|----------|------|
| be.helped   | FUT      | OBL         | ART(OBL)     | Tom       | ART(MP)  | Jack |
| 'Tom will h | elp Jacl | k.' (lit. ' | Jack will be | helped by | y Tom.') |      |

(b) <sup>?</sup>i təlí<sup>?</sup> əž Seattle. (CC) ż ?i talí? 22 Seattle be here from OBL ART(OBL) Seattle 'He came from Seattle' (c) swé? m a X Andrew ta sq<sup>w</sup>améy. (CC) swé? mэ [?ə] ź sq<sup>w</sup>əméỷ Andrew tə property CERT OBL Andrew ART(MP) dog ART(OBL) 'The dog belongs to Andrew.' (lit. 'The dog is the property of Andrew.') (2) It is obligatory before personal words standing as oblique adjuncts, as in (d), (e), and (f). (d) wélx čx<sup>w</sup> x<sup>w</sup>əmi  $\lambda$  (~  $^{2}$ ə $\lambda$ )  $^{2}$ é·n $\theta$ ə. (JP) x w-?əmí ż ?é•nθ∋ wél-x čxw 22 throw-TR toward-come OBL ART(OBL) be me vou 'Throw it to me.' (e) ni cən yá·yəs ni <sup>9</sup>əx nówə. (CC) vá·ỷəs ni? ź ni? cən ?a náwa AUX I be.working be.there OBL ART(OBL) be.vou 'I am working for you.' (f) təwkwí·lə təlí x łwóləp. (JP 22) wə-k<sup>w</sup>ín-ələ [?ə] ź tə talí? łwśləp ART(MP) EST-how.many-person from OBL ART(OBL) you(pl.) 'several of you' (3) It provides an optional way of relating a possessed noun to its noun possessor, as in (g) and (h).

- (g) k<sup>w</sup> όl ο w λ spé?eθ (JP)
  hide ART(OBL) black.bear
  'bear hide, hide of a bear' (cf. k<sup>w</sup>θο k<sup>w</sup> ól o ws to spé?eθ 'a/the bear's hide' [AG])
- (h) słéwən X pipá⋅m (CC, JP)
   sleeping.mat ART(OBL) frog
   'broad-leaf plantain' (lit. 'frog mat')

Such forms are not common, so it is not clear whether there are any restrictions on the kind of noun that may be used in them.

(4) The article  $\vec{x}$  has also been recorded in a few other sentences that may not reflect general usage. JP gave (i) and (j), but AG would have to use a possessive, saying (k).

- (i) c-qáš Ř ?éwkw. (JP)
   get-much ART(OBL) wealth
   'He's got lots of everything. He is well off.' (lit. 'He has much of wealth.')
- (j) c-qáš Ř télə tθé?. (JP)
   get-much ART(OBL) money that.one(MP)
   'He has lots of money.'

CC gave (1), but JP rejected this. Compare with his (m):

(l) niwł słóm ź ?ápən i kwə nóca?. (CC) ni? wəł-słóm Χ ?ápən °∂ỷ be there already-wearing.off ART(OBL) ten and k<sup>™</sup>a náča? ART(MR) one 'It's half past eleven.' (m) táx<sup>w</sup> wiléw nóca? (JP) táž<sup>w</sup> n áda? wə-yəléw be.dead.centre EST-past one

'half past one'

The oblique article is restricted to these uses. Unlike the other articles, it cannot introduce nominalizations or relative clauses.

# **15.2. THE DEMONSTRATIVES**

These express the same categories of gender and proximity as the articles, and also express definiteness. Further, they have plural forms and, in one set, diminutives. Like the articles, most of the demonstratives can stand before nouns as determiners. Unlike the articles, all of the demonstratives can also stand alone as adjuncts with the functions of demonstrative pronouns, third-person pronouns, and locatives.<sup>3</sup>

There are several sets of demonstratives. The total number of forms recorded is close to forty, and there may be a few not yet recorded. Because members

<sup>3</sup> The distinction between third-person and demonstrative pronouns and between pronouns and locatives are, of course, distinctions of English. Whether a Halkomelem form such as  $t\theta e^{2}$  is translated with the third-person pronoun 'he' or 'it' or with the demonstrative pronoun 'that' is probably a matter of how strong the deictic function of  $t\theta e^{2}$  is felt to be and whether the referent is a person or a thing, things being more comfortably called 'that' than people. And when  $t\theta e^{2}$  is translated 'there,' it is simply because English requires it in that context.

of the different sets have parallel functions, it seems best to present them all first and then discuss their functions.

The glosses given in the tables below are intended to show the range of functions and meanings, but the list is not exhaustive. Many of the non-feminine forms may appear as determiners with non-human plurals and so might also be glossed 'these' or 'those.'

#### 15.2.1. The Simple Demonstratives

The forms recorded are given in Table 15.2.

| Table | 15.2 |
|-------|------|
|-------|------|

| The simple demonstratives  |                                                                          |                                              |                                                    |  |  |
|----------------------------|--------------------------------------------------------------------------|----------------------------------------------|----------------------------------------------------|--|--|
|                            | Non-feminine (M)                                                         | Feminine (F)                                 | Plural (pl.                                        |  |  |
| Present and visible (p)    | $t\theta \hat{e}^{9}$ 'that, he, him, it, there'                         | $\theta e^{2}$ 'that, she, her'              | yəθé <sup>?</sup> , yəθéləỷ<br>'those, they, them' |  |  |
| Nearby and invisible (n)   | <i>k</i> <sup>w</sup> θ <i>é</i> <sup>2</sup> 'that, he, him, it, there' | <i>k<sup>w</sup>łé</i> ? 'that,<br>she, her' | <i>k<sup>w</sup>θéləỷ</i> 'those, they, them'      |  |  |
| Remote or hypothetical (r) | $\vec{k}^{wi}$ 'that, there'                                             |                                              |                                                    |  |  |
|                            |                                                                          |                                              |                                                    |  |  |

The relationship of these forms with the articles is very close. Four of the five singular forms ( $\theta e^{2}$ ,  $k^{w}\theta e^{2}$ ,  $k^{w}te^{2}$ , and  $k^{w}t$ ) may be no more than stressed or fuller forms of the corresponding articles ( $\theta a$ ,  $k^{w}\theta a$ ,  $k^{w}ta$ , and  $k^{w}a$ ). Data from other dialects are useful in interpreting the other forms. The Cowichan non-feminine present article is  $t^{\theta}a$  [ $t^{\theta}a$ ], which seems closer to the Musqueam  $t\theta e^{2}$  than the Musqueam article ta. Chilliwack has a human plural article ya (Galloway 1993, 390), which seems identifiable with the Musqueam plural element ya. The Musqueam plurals  $ya\theta e^{2}ay^{2}$  and  $k^{w}\theta e^{2}ay^{2}$  are clearly formed with the plural infix -la-, but the Cowichan singular demonstratives  $t^{\theta}ey$ ,  $\theta ey$ , and so on seem to provide better bases than the Musqueam singulars.

## 15.2.2. The Compound Demonstratives

These are words each composed of an article and a root, possibly with other elements. They are probably all relative clauses that have become more or less fossilized. We may distinguish, by the type of root, locative and third-person demonstratives.

## 15.2.2.1. The Locative Demonstratives

These are formed with 2i 'be here' and  $ni^2$  'be there,' which occur constantly as locative and auxiliary verbs, a third element  $-n\dot{a}$  'be here' (or possibly simply 'exist'), otherwise unattested, and perhaps a fourth element  $-n\dot{a}$  (or  $-n\dot{a}$ ) 'be

yonder,' only tentatively suggested. In some forms there is also a final  $-n\partial$ , which cannot be identified, although it seem possible that it might be an unstressed  $-n\partial a$  'exist.'

This set of demonstratives seems to have no forms corresponding to the articles  $k^{w}\theta a$  and la. It distinguishes only two positions, perhaps simply visible and invisible. There are plurals formed with  $-\dot{n}a$ , but no others. Forms recorded so far are given in Table 15.3.

| The locative demonstratives   |                                                                                                                                                              |                                                                   |                                                                  |  |
|-------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------|------------------------------------------------------------------|--|
|                               | Non-feminine (M)                                                                                                                                             | Feminine (F)                                                      | Plural (pl.                                                      |  |
| Present and visible (p)       | <i>te<sup>?</sup>í</i> 'this, here,<br>this way'<br><i>tə<sup>?</sup>ínə</i> 'this'<br><i>təní?</i> 'there, that way'<br><i>təníňə</i> 'that'                | <i>θə<sup>9</sup>í 't</i> his'<br><i>θə<sup>9</sup>ínə</i> 'this' | <i>təǹá ltən</i> these,<br>they, them'                           |  |
|                               | <i>təna</i> 'this, he, him'                                                                                                                                  | $\theta$ əna 'this, she, her'                                     | yəna' 'these, they,<br>them'<br>yəna'ttən 'they,<br>them, those' |  |
|                               | <i>tənˈá·nə</i> 'that over<br>there, over there'<br><i>təni?[ə]tənˈá·nə</i><br>'that way over there'                                                         |                                                                   |                                                                  |  |
| Remote or<br>hypothetical (r) | <ul> <li>k<sup>w</sup>∂<sup>2</sup>i 'around that way'</li> <li>k<sup>w</sup>∂ná 'he, him'</li> <li>k<sup>w</sup>∂ná n∂ 'way over</li> <li>there'</li> </ul> | <i>k<sup>w</sup>səna</i> 'she, her'                               | <i>k<sup>w</sup>əǹá łtən</i> 'they,<br>them'                     |  |

Table 15.3

The forms with 2i and  $ni^2$  are transparently relative clauses; for example,  $t\partial^2 i$  (usually pronounced [ti2i]) is no doubt //t $\partial^2 i$ // 'the one who/that is here.' If the final  $-n\partial$  is (as tentatively suggested) a reduction of  $-na^2 i$  (exist,' then  $t\partial^2 i n\partial^2 may$  be 'the one who/that exists here.' The complex form  $t\partial ni^2\partial t\partial na^2 n\partial^2 may$  be something like 'the one/who that is there at that which is way over there.' (See §17.17 for a comparable use of  $t\partial^2 2\partial n\partial c\partial^2$ , lit. 'the one that is where,' for 'which.')

The plural markers are the *y* $\partial$ - of the simple set and, very likely, the third-person plural particle  $?\acute{e}tt\partial n$ . Thus  $\vec{k}^w \partial n \dot{a} tt\partial n$  is probably  $//\vec{k}^w \partial n \dot{a} ?\acute{e}tt\partial n//$ , meaning something like 'the invisible ones who exist.'

#### 15.2.2.2. The Third-Person Demonstratives

These are based on  $\lambda \dot{a}$  'be third person.' They form a set that has members in all three positions. There are also forms with  $-\partial t$  'past' for deceased persons and there are diminutives (for at least the present visible position) formed by reduplication. And there are still other forms containing  $ni^{2}$  'be there.' Attested forms are given in Table 15.4. There may well be others.

| The third-person demonstratives |                              |                        |              |  |  |
|---------------------------------|------------------------------|------------------------|--------------|--|--|
|                                 | Non-feminine                 | Feminine               | Plural       |  |  |
| Present and visible             | təwŻá                        | θəwŻá                  | təwŽáləm     |  |  |
| Nearby and invisible            | k™ӨәwՃá                      | łəwŻá                  | k¤0əw Żáləm  |  |  |
| Remote or hypothetical          | <i>k</i> <sup>w</sup> əw Xá  | k <sup>w</sup> səwŽá   |              |  |  |
| Deceased                        | <i>k</i> <sup>w</sup> əwĂá:t | k <sup>w</sup> səwŽá·ł | k¤0əwŽáləməł |  |  |
| Diminutive present              | təwŽáŽəm                     | θəwŽáŽəm               | təwŻəŻaŻám   |  |  |
| <i>ni</i> <sup>9</sup> present  | tənəwŽá                      | 0ənəw Žá               |              |  |  |

#### Table 15.4

These forms too, like the locative demonstratives, are probably relative clauses in origin,  $t \partial w \dot{\lambda} \dot{a}$  being a contraction of  $//t \partial w \partial_{-2} \dot{\lambda} \dot{a} / /$  'the one who is third person' and so on. In rapid speech, the stress falls equally on both syllables or may appear to fall on the first. The Cowichan counterparts are  $t^{\theta} \partial w n \dot{t} \dot{t}$  and so on, based on  $n \dot{t} \dot{t}$ , the Island counterpart of Mainland  $\dot{X} \dot{a}$ .

The forms  $t \partial n \partial w \dot{\lambda} \dot{a}$  and  $\theta \partial n \partial w \dot{\lambda} \dot{a}$  are probably //tə ni? wə-2  $\dot{\lambda} \dot{a}$ // 'the one who is third person there' or 'the one who is there being the third person' and so on. These forms seem to be rare; they occur in a few sentences given by CC during elicitation but have not shown up in the texts.

#### 15.2.3. Use as Determiners

It appears that nearly all of the demonstratives can stand before nouns as determiners. (There may be restrictions on some of the plural forms.) As determiners, the demonstratives may be translated 'this,' 'that,' 'these,' and 'those,' but for all but the locatives formed with 2i,  $ni^2$ , and  $-na^2$ , this deictic sense is weak. Perhaps a demonstrative is often chosen over an article mainly because it is definite, while the article is not. At any rate, demonstratives seem to appear more often in Halkomelem than in English, and often a Halkomelem demonstrative can be translated as easily with an English definite article. As determiners, the demonstratives are also often unstressed.

The simple demonstratives ( $t\theta \hat{e}^2$ , etc.) generally have the sense of a weak 'that':

- swóyge?. (CC) (a) wét k <sup>w</sup>ə  $t\theta e^{2}$ who then DEM(MN) man 'Who is that man?' (b) syáwín k<sup>w</sup>θe? sk<sup>w</sup>íx. (JP 23) spell.word DEM(MN) name 'That [or 'this' or 'the'] name is a spell word.' (c) Åa k <sup>w</sup>ə k<sup>w</sup>łe? aəlaəlíłəł ctwa?. (JP 2) then Basket.Ogress BE3P DEM(FN) SPEC 'It was that Basket Ogress [now deceased, known from tradition].' (d) ném k wi cən spółxon. (CC)
- (d) nem can k<sup>-1</sup> sparan. (CC) go I DEM(MR) prairie/pasture 'I'm going to that pasture.'

The fact that  $\vec{k} \cdot \vec{i}$  implies that something is simply invisible and not necessarily distant is clear from (e).

(e) <sup>?</sup>i k<sup>w</sup>i s<sup>?</sup>é<sup>x</sup>q k<sup>w</sup>θ<sup>a</sup> n<sup>a</sup>ká<sup>.</sup> (JP)

<sup>9</sup>i [<sup>9</sup> $\partial$ ]  $\vec{k}^w$ i s<sup>9</sup> $\vec{e}$ Åq  $k^w$  $\theta \partial$  n $\partial$ -kábe.here OBL that(MR) outside ART(MN) my-car 'My car is outside.'

The 2i tells us that the car is nearby while the  $k^{wi}$  tells us that the outside is invisible. We are in the house and the car is parked outside, out of sight.

The locatives  $t\partial^2 i$  and  $t\partial^2 in\partial$  contrast with  $t\theta e^2$ , and  $t\theta e^2$  contrasts with  $t\partial ni^2$ , and so on. Compare the following:

tə<sup>2</sup>i sq<sup>w</sup>əméy' 'this dog'
tθé<sup>2</sup> sq<sup>w</sup>əméy' 'that dog'
tə<sup>2</sup>í púk<sup>w</sup> 'this book [in my hand]'
tθé<sup>2</sup> púk<sup>w</sup> 'that book [somewhere near me]'
tənína púk<sup>w</sup> 'that book [over there]'
təni<sup>2</sup>ətəná:nə púk<sup>w</sup> 'that book way over there'
tə<sup>2</sup>ínə téqəməx 'this flat country [here in the Fraser Delta]'
təná:nə scəwaθən 'the outer side over there [i.e., Tsawwassen]'

The forms  $t\partial n\dot{a}$ ,  $\theta \partial n\dot{a}$ , and so on seem to refer to what is simply present rather than right at hand:

təňa swéyəl 'this day, today'
?i ə təňa táməx<sup>w</sup> 'on this earth'
θəňa nəšx<sup>w</sup>əmník<sup>w</sup> 'this aunt of mine' (CC 12)
yəňa yáýsələ 'these two persons'

In (f) and (g), təňá contrasts with təňá nə.

- (f) mí čx<sup>w</sup> (x<sup>w</sup>te? ə) təna səqín. (JP) come you move.toward OBL DEM(MP) near.end
   'Come to this end of the house.'
- (g) ném čx<sup>w</sup> [?ə] təňá·nə snəqíň. (JP)
   go you OBL DEM(MP) far.end
   'Go to that end of the house.'

However, the nouns also contrast; s = qin' near end of a house' contains 2i 'be here,' while sn = qin' far end of a house' contains  $ni^2$  'be there.'

In contrast with  $t\partial n\dot{a}\cdot n\partial$ , which implies distance but visibility,  $k^w \partial n\dot{a}\cdot n\partial$  can refer to something distant and unseen:

(h) ck<sup>w</sup>ik<sup>w</sup>éýəm k<sup>w</sup>θə məlstéyəx<sup>w</sup> ni k<sup>w</sup>əňá·nə cak<sup>w</sup>. (JP)
 be.starving ART(MN) persons be-there DEM(MR) far
 'People are starving in that far-off country.'

The forms  $k^w \partial n \dot{a}$  and  $k^w s \partial n \dot{a}$  indicate invisibility, but when referring to persons do not imply deceased status as the corresponding articles ( $k^w \partial$  and  $k^w s \partial$ ) do.

(i) žáťa k<sup>w</sup>aňa bá·s ... (JP)
 be.saying DEM(MR) boss
 'The ''boss'' says ...'

The forms  $t \partial n \dot{a}$  and  $\theta \partial n \dot{a}$  are unique among the demonstratives in that they can also serve as contemptuous second-person pronouns in vocative phrases, as in (j), (k), and (l).

- (j) 'a. təňa qś. l sqwəméy spá. l. (JP 4)
  aah DEM(MP) baad dog raven
  'Oh, you *baad* dog Raven!' (said by women from whom Raven has stolen salmonberries)
- (k) še?, θəňa ἀćməỷ. (JP)
  hey DEM(FP) girl
  'Hey, you girl!' (said to a young man who has done something effeminate, such as using <sup>2</sup>*iš* instead of *šé*<sup>2</sup> as an interjection)
- (l) təňa qál smimxcaň. (AG)
   DEM(MP) bad lousy(DIM)
   'You lousy little thing!'

The third-person determiners  $(t \partial w \dot{\lambda} a, \text{ etc.})$  when used as determiners have a weak deictic sense; they probably often serve simply to mark definiteness. They are most frequently used with persons:

*təw*Åa swíwləs 'that/the young man' θəwÅa qéməy' 'that/the young woman' As determiners, however, they are not restricted to persons. They are also used for animals:

(m)?ówəte? k wə šx wnéms təwẩa smóyəθ. (CC 7)
 ?ówəte? k wə šx w-ném-s təwẩa smóyəθ
 none ART(MN) OBLNOM-go-3POS DEM(MP) deer
 'The deer has/have no place to go.'

(n) ?ówə k<sup>w</sup>s k<sup>w</sup>óyxöəts θəwảa q<sup>w</sup>tá·yθən. (JP 23)
 ?ówə k<sup>w</sup> s-k<sup>w</sup>óyxö-θət-s θəwảa q<sup>w</sup>tá·yθən not ART(MN) NOM-move-self-3POS DEM(FP) sturgeon
 'The sturgeon doesn't move.'

They seem generally not used for inanimate things, but I have found two exceptions in texts:

| (0) | sẳíẳəc      | tə       | s?áθəs                               | ni           | ə      |                                   |
|-----|-------------|----------|--------------------------------------|--------------|--------|-----------------------------------|
|     | designed    | ART(MP   | ) face                               | be.there     | OBL    |                                   |
|     | təwźa       | sxélr    | məx <sup>w</sup> cəs. (CC 1          | .9)          |        |                                   |
|     | DEM(MP)     | ritual   | ist's.rattle                         |              |        |                                   |
|     | 'A face was | s carved | on that/this [rit                    | ualist's] ra | ttle.' |                                   |
| (p) | wəná∙y      | ćə       | k <sup>w</sup> -s-ni <sup>9</sup> -s | sčé          | ćən    | wə- <sup>9</sup> áwk <sup>w</sup> |
|     | be.only     | QUOT     | ART-NOM-AUX-3                        | spos trul    | у      | EST-be.used.up                    |

'It was only when that poison was truly used up that ...'

t<sup>θ</sup>óxtən. (JP 27)

poison

But these may not be exceptions after all. Both the rattle and the poison may be seen as active entities.

As determiners, the third-person demonstratives seem to have the same anaphoric restriction that they have as demonstrative pronouns. This is discussed below.

The use of plural forms of all sets of demonstratives seems to vary with the kind of noun. If a noun that is plural in form or sense refers to human beings, a demonstrative used as a determiner is almost always a plural form, as in examples above. There are exceptions, however, as in (q), where plurality is not marked.

tow่งใด

DEM(MP)

(q) təwÅa čé·ymən (CC 12)
 DEM(MP) Chinese
 'the Chinese' [farmers renting reserve land]

With animals, the choice of a plural seems to be optional. Non-plural forms may be used, perhaps especially when plurality is marked by other means. Compare the following sentences from a single passage in JP 23.

- (r) ni xém təwẳáləm má?əq<sup>w</sup>.
   AUX come.to.surface DEM(PL) duck
   'Those ducks came to the surface.'
- (s) X a k<sup>w</sup> > ni šx<sup>w</sup>k<sup>w</sup>aq<sup>w</sup> > ts tθe? má? > q<sup>w</sup>.
  X a k<sup>w</sup> > ni šx<sup>w</sup>-k<sup>w</sup>aq<sup>w</sup> > ts tθe? má? > q<sup>w</sup>
  BE3P then AUX OBLNOM-club DEM(MP) duck
  'That's what he used to club the ducks with.'
- (t) sisəw ném <sup>9</sup>al lək<sup>w</sup>lók<sup>w</sup> tə tópsəms təwka má<sup>9</sup>əq<sup>w</sup>. ?al s-ni-s wə-ném lək "lák" tápsəm-s tə NOM-AUX-3POS EST-AUX(go) just break(PL) neck-3POS ART təwila má?əa<sup>w</sup> duck DEM(MP) 'Then the necks of those ducks were just broken.'
- (u) k<sup>w</sup> ank<sup>w</sup> ánatas tθe? má?aq<sup>w</sup>.

| k*ənk*án-ət-əs       | t0e?    | má?əq* |
|----------------------|---------|--------|
| be.taken(PL)-TR-3TR  | DEM(MP) | duck   |
| 'He took the ducks.' |         |        |

The word  $m\dot{a}^{2}\partial q^{w}$  'duck' (really any larger bird) is not plural in form. In (r), plurality is indicated by the demonstrative, in (s) by nothing, in (t) by 'get broken,' and in (u) by 'take.'

Inanimate plurals generally take non-plural forms, as in (v).

(v) k " $\theta$ e? smənmé·nt (JP) DEM(MN) rocks 'the/those rocks'

But again there are exceptions, as in (w), in which plurality is indicated in all three words.

(w) yəθéləỷ θéməx wəł səníx wəł (JP)
 DEM(PL) two.canoe canoes
 'those two canoes'

Like the articles, the demonstratives can introduce relative clauses (see §4.1.2). In this function, they seem to be used very much the same way as when introducing nouns.

## 15.2.4. Use as Adjuncts

The demonstratives can also stand alone as direct or oblique adjuncts, serving as third-person pronouns, demonstrative pronouns, and locatives. There are differences among the three sets of demonstratives in these functions. Demonstratives of the simple and locative sets, when standing as direct adjuncts and as oblique adjuncts in many contexts, can refer to either human or non-human entities and so serve as third-person or demonstrative pronouns, that is, be translated as 'he,' 'she,' 'this,' 'that,' and so on. However, when standing as oblique adjuncts following verbs implying location or motion in a particular direction, demonstratives of the simple and third-person sets can refer to place and serve as locatives, that is, be translated as 'here,' 'there,' and so on. When used as adjuncts, demonstratives of the third-person set always refer to human beings and so must be translated as third-person pronouns. For purposes of exposition, it seems useful to separate pronominal from locative uses.

# 15.2.4.1. Pronominal Uses

In the following sentences, demonstratives of the simple set appear as direct adjuncts and refer to either human or non-human entities.

- (a) wét tθé<sup>?</sup>. (JP) who that(MP) 'Who is that/he?' (b) stém k <sup>w</sup>ə k<sup>w</sup>θé?. (CC) what then that(MN) 'What is that?' (c) yə-mí tθé?. (JP) along-come that(MP) 'He is coming.' (d) Åcés-əł ćə tθé?. (JP 1) island-past OUOT that(MP) 'It [Point Roberts] was an island, they say.' (e) təw-nínadwam
- (e) təw-níňəq̇<sup>w</sup>əm tθé?. (JP) somewhat-soft(PL) that(MP)
   'They are soft.'
- (f) há<sup>9</sup>k<sup>w</sup>-əx łe tθé<sup>9</sup>. (JP)
   be.used-TR PER that(MP)
   'Use that!'
- (g) <sup>9</sup> śwə sẳ íẳ qəł-əs θé?. (JP)
   not child-3suB that(FP)
   'She is not a child.'

ćə k<sup>w</sup>łe<sup>?</sup>. (CC) (h) ni k\*án-aws-t-am be.taken-body-TR-INTR that(FN) AUX OUOT 'She was harmed by sorcery, they say.' (i) səwáyqe? yəθéləỷ. (JP) men those(MP) 'They are men.' ćə ťá k w (i) ném yəθéləỷ. (CC) AUX(go) OUOT go.home those(MP) 'They are going home, they say.' (k) ni ťák \* k<sup>w</sup>θéləỷ. (CC) go.home those(MN) AUX 'They went home.' In (l) to (p), the simple demonstratives are oblique adjuncts. (1) ste<sup>?</sup>é <sup>?</sup>e t $\theta$ e<sup>?</sup> stə?é tθé? ?a like OBL that 'thus, like that' (m)x<sup>w</sup>ən xəté·tθe? x "ən-xəre ?ə tθé? still-be.doing OBL that 'while doing that, in the course of that' х́а šx<sup>w</sup>-žáťa-s (n) słénaý tθé?. (JP) [?ə] woman BE3P OBLNOM-be.doing-3POS OBL that 'She's a woman, that's why she does that.' (o) wél-x čxw nem ?a tθé?. (AG) throw-TR go him vou OBL 'Throw it to him.' (p)  $t = 1i^{2} \cdot 2 \cdot k = 0$  (CC 21) ?a k<sup>w</sup>θé? ?∂ỷ təlí? from OBL DEM(MN) and 'from that [time, event], after that, from then on ...'

In (l) to (n), the  $st\partial^2 \acute{e}$  'be like' and  $\check{x}\dot{\partial}i\dot{a}$  'be doing' are intransitive but can have objects in the oblique case. In (o), *nem* is a verb denoting motion, but it is used here as a preposition; the main verb denotes an action that normally has a human recipient and so the demonstrative refers to a human being. In (p), the demonstrative might also be interpreted as a locative.

The locative demonstratives formed with 2i and  $ni^2$  generally refer to things and not to people.

(a) stém ti<sup>9</sup>í. (AG) what this(MP) 'What's this?' (r) há?k<sup>w</sup>-əx łe tə<sup>9</sup>í. (CC) be.used-TR PER this(MP) 'Use this!' (s)  $n = st^{\theta} i^{\gamma}$ tə<sup>9</sup>ínə. (JP) my-loot this(MP) 'This is my loot.' (t) stém təní?. (CC) k »ə what then that(MP) 'What's that?' Those formed with  $-\dot{n}a$  more often refer to people. (u) tém-c təna. (AG) what-tribe this(MP) 'Where is this man from?' (v) ném co <sup>9</sup>al lítotos θoná to shés to s<sup>9</sup>ít<sup>θ</sup>om ni hak<sup>w</sup>oxos k<sup>w</sup>s yégawałtan. (JP) ném ćə ?aĺ łíł-ət-əs θəná tə distribute-TR-3TR iust she go OUOT ART(MP) s?íť<sup>e</sup>əm shé-s tə ni hak<sup>w</sup>-əx-əs k <sup>w</sup> remainder-3POS blanket ART AUX use-TR-3TR ART s-yé-daw-altan NOM-along-pay-others 'This woman will go around distributing the remainder of the blankets that she used to pay the people.' (w) Ža sé∙nŽe yəna. (JP 29) senior BE3P they 'They are older ones.' (x) Žáləm ćtwa? yəná·łtən. (JP) BE3P(PL) SPEC thev 'They must be the ones.' (v) Ža kwana. (AG) BE3P that(MR) 'He's the one [just out of sight].' (z) <sup>9</sup>i qíqəw k<sup>w</sup>səná. (JP) AUX be.menstruating that(FR) 'She's having her period.'

The last sentence answers the question, "Why isn't your wife helping?" The auxiliary i indicates that she is nearby but the form  $k^w s \partial n \dot{a}$  indicates that she is out of sight.

The locative demonstratives in these examples are all direct adjuncts. In (aa), a locative demonstrative is an oblique adjunct.

(aa) Že nəw ste?é ?al ə tə?ínə. (CC 28)
Že ni? wə-stə?é ?al ?ə tə?ínə also AUX EST-like just OBL this 'It's the same as this.'

When used as either direct or oblique adjuncts, the third-person demonstratives refer to persons only.

(bb) ném ?ə k <sup>w</sup>ə təwÅa. (CC) that(MP) go ROG then 'Is he going?' (cc) səw q<sup>w</sup>éls θəwÅa ... (CC) s-wə-q<sup>w</sup>él-s θ**∋**wÅa NOM-EST-speak-3POS that(FP) 'Then she said ...' (dd) ném ?a k »a tawŻálam go ROG then those(MP) 'Are they going?' (ee) cewat can ce? tawka. (CC) céw-ət cən ce? tawża help-TR I FUT that(MP) 'I'll help him.' (ff) čéwətəm ce<sup>9</sup> Å Tom təwÅa. (CC) céw-ət-əm ce? [?ə] ź Tom təwźa help-TR-INTR FUT OBL ART Tom that(MP) 'Tom will help him.' (lit. 'He will be helped by Tom.') (gg) ni ž™ánxənəm k™θəwÅáləm. i ... (JP) be.running those(MN) and AUX 'They were running when ...' (hh) k<sup>w</sup>ýcna·m ce<sup>9</sup> təwňa. (CC) k<sup>w</sup>ec-n-a·m təwẩa ce? [?ə] see-TR-you FUT OBL that(MP) 'He'll see you.' (lit. 'You will be seen by him.')

- (ii) ni? cən yá·ýəs ni? [?ə] k<sup>w</sup>θəwka. (CC)
   AUX I be.working be.there [OBL] that(MN)
   'I'm working for him.'
- (jj) ste<sup>9</sup>é cən [<sup>9</sup>ə] təwÅa. (CC) be.like I [OBL] that(MP) 'I'm the same as him.'

(kk) sow kwócnoxws kwowła to nóca? smóyoθ ka yowewołotos to stəlqáyə. (CC 16) k้™วพ≯ี่ล์ s-wa-k<sup>w</sup>ác-nax<sup>w</sup>-s х́а n áča? tə smáyəθ NOM-EST-see-TR-3POS DEM(MR) ART(MP) one deer BE3P yə-wéwəł-ət-əs stəlqáyə tə along-be.chasing-TR-3TR wolves ART

'Then he [deceased] saw a deer, which was the one that the wolves were chasing.'

In this last sentence, the narrator (AC) identified the principal character, his deceased junior cousin, with a form that shows deceased status, but for the deer and the wolves, he used present visible forms.

There is an anaphoric restriction, mentioned above, on the use of the thirdperson demonstratives: after *-as* 'third-person transitive subject,'  $taw \dot{x}a$ , and so on can refer to the transitive subject only. The implications of this for syntax are discussed in §3.4.2.

# 15.2.4.2. Locative Uses

Demonstratives of the simple and the locative sets, standing as oblique adjuncts after verbs denoting location or directional motion, serve as locatives.

With 2i 'be here' and  $ni^2$  'be there' used both as main verbs and as prepositions, these demonstratives have the senses 'here' and 'there.'

?é·łtən

3PL

(a) ni čə ní<sup>γ</sup> k<sup>w</sup>θé<sup>γ</sup>. (JP) ćə ni? ní? [?ə] k<sup>w</sup>θé? OUOT be.there DEM(MN) AUX OBL 'It's reportedly here.' (b) səsəw ni<sup>9</sup> ə t $\theta$ é<sup>9</sup> é·łtən. (CC 11) s-ni?-s wə-ní? ?a tθé? NOM-AUX-3POS EST-be.there OBL DEM(MP)

'And so they stayed there.'

(c) wəvá $\theta$  cən wək<sup>w</sup>ék<sup>w</sup>əcnəx<sup>w</sup> təw<sup>\*</sup> ta ni<sup>?</sup> ə t $\theta$ é<sup>?</sup>. (CC) wə-k<sup>w</sup>ék<sup>w</sup>əc-nəx<sup>w</sup> təwิ่ล wə-váθ cən ni? 22 I EST-be.seeing-TR him EST-always be there **OBL** tθé? DEM(MP) 'I always see him there.' (d) ni<sup>?</sup> cən me k<sup>w</sup>ə́cnəx<sup>w</sup> k<sup>w</sup>θəwk̃a k<sup>w</sup>sni<sup>?</sup>s ni<sup>?</sup>ək<sup>w</sup>θé<sup>?</sup>. (CC) mэ k<sup>™</sup>ác-n∍x<sup>™</sup> k™θəwጰa k <sup>w</sup> ni? cən s-ni?-s AUX I CERT see-TR him(MN) ART NOM-AUX-3POS ni? ?2 k<sup>w</sup>θé? be there OBL DEM(MN) 'I saw him there.' (lit. 'I saw him when he was there.') (e) ?í ?a təná tə púk<sup>w</sup>. (CC) be here DEM(MP) book OBL ART(MP) 'The book is here.' (f) <sup>9</sup>ámət łe ?i ?a tə<sup>9</sup>í. (CC) sit PER be.here OBL DEM(MP) 'Sit here ' (g) ha<sup>9</sup> cən dá·yt <sup>9</sup>i·təna [<sup>9</sup>i <sup>9</sup>ə təna] ... (JP 22) há? cən dáy-ət ?i ?a təná if die-tr be.here I OBL DEM(MP) 'If I kill it here '

Also, following  $2\partial m i$  'come' and  $n \epsilon m$  'go' used as prepositions, these demonstratives have the sense 'here' and 'there.'

(h) mí łe mi <sup>9</sup> tə<sup>9</sup>í. (CC) ?əmí łe ?amí °ə tə?í come come DEM(MP) PER OBL 'Come here.' (i) ném łe nem ?ə təní?. (CC) go PER go OBL DEM(MP) 'Go there.'

Following  $\check{x}^w t \acute{e}^{\gamma}$  'go toward,'  $n \acute{asom}$  'look toward,' and perhaps some other verbs, these demonstratives have the sense 'this way,' 'that way,' and so on.

| (j) | ằ™té?       | ct     | ce?  | [?ə]  | tə <sup>9</sup> í. (CC 8) |
|-----|-------------|--------|------|-------|---------------------------|
|     | head.for    | we     | FUT  | [OBL] | this(MP)                  |
|     | 'Let's head | this w | ay.' |       |                           |

(k) x<sup>w</sup>té təní?. (JP) čx w [?ə] head.for that(MP) you [OBL] 'Go that way." (1) ?śwə čx w x<sup>w</sup>té?-∍x<sup>w</sup> [?ə] tθé?. (JP) vou head.for-vou [OBL] that(MP) not 'Don't go that way.' mi x™té? (m)<sup>?</sup>i təná. (JP) ə come head.for OBL this(MP) AUX 'He's coming this way.' k<sup>w</sup>ə<sup>γ</sup>í. (JP) (n) ném čxw x™té? ə head.for that(MR) go vou OBL 'Go around the other way.' čxw [?ə] təní?. (JP) (o) násəm face vou [OBL] that (MP) 'Look that way.'

# **15.3. THE DEMONSTRATIVE AUXILIARIES**

There is a set of four words that play the same role syntactically as the locative auxiliaries (§3.2.1), while calling attention to the existence of an entity or the immediacy of an event and identifying it as to gender. Each of the four has an initial element seemingly identifiable as a locative auxiliary. This is followed by the oblique particle  $2\sigma$  and an article  $t\sigma$  (MP) and  $\theta\sigma$  (FP). The article agrees in gender with whatever the word calls attention to. The demonstrative auxiliaries are shown in Table 15.5.

| Table 15.5                    |                |                      |        |  |
|-------------------------------|----------------|----------------------|--------|--|
| The demonstrative auxiliaries |                |                      |        |  |
|                               | Non-feminine   | Feminine             | Plural |  |
| ?í 'be here'                  | ?í?ətə ~       | <sup>?</sup> í?əθə ~ |        |  |
|                               | ²íətə ~ ²í·tə  | ²í əθə ~ ²í·θə       |        |  |
| <i>ná</i> 'be over there'     | ná?ətə ~ ná•tə | ná?əθə ~ ná·θə       | ná?əỷe |  |

The initial element 2i is certainly 2i be here.' The  $n\dot{a}$  (or  $n\dot{a}^2$ ) is probably a variant of  $n\dot{i}^2$  be there.' The Cowichan forms (cf. Leslie 1979, 103), which are used by AG, show a variant of 2i as well; they are  $2\dot{e}^2\partial t$ ,  $2\dot{e}^2\partial \theta$ ,  $n\dot{a}^2\partial t$ ,  $n\dot{a}^2\partial \theta$ .

Like other auxiliaries, the demonstrative auxiliaries precede the predicate head.
- (a) <sup>?</sup>í<sup>?</sup>ətə přípələm. (CC) be.here(MP) be.overflowing 'It's overflowing.'
- (b) ná<sup>2</sup> θθθ yθmé·mt. (CC)
   be.there(FP) along-be.distributing
   'She's passing something out.'
- (c) <sup>9</sup>iətə Åəwł mi hét<sup>θ</sup>əm. (JP 22)
   <sup>9</sup>i<sup>9</sup>ətə Åə wəł-mi hét<sup>θ</sup>əm be.here again already-AUX(come) breathe 'He's starting to breathe again.'
- (d) q<sup>w</sup>á··q<sup>w</sup>ələž łe, ná<sup>2</sup>ətə yəyéyəq tə θqét. (CC) q<sup>w</sup>áq<sup>w</sup>ələž łe, ná<sup>2</sup>ətə yə-yéyəq tə θqét take.care PER be.there along-be.falling ART tree 'Watch out, the tree is falling!'

In these examples, the gender of the auxiliary agrees with that of the subject, but in (e) (a statement by a male speaker) it agrees with that of the object, which is what attention is directed toward.

(e) <sup>9</sup>i<sup>9</sup>∂∂ c∂n k<sup>w</sup>∂n-é-t θ∂na qól ... (JP 25)
 be.here I be.taken-DUR-TR this bad
 'I'm holding this bad thing that ...'

As with the locative auxiliaries, a demonstrative auxiliary can be followed by a predicate head that is its lexical counterpart, as in (f) and (g).

- (f) <sup>9</sup>í<sup>9</sup>ətə <sup>9</sup>í tə nə-máñə. (CC) be.here be.here ART my-child 'My son is here [visible].'
- (g) <sup>9</sup>i<sup>9</sup>∂θ∂w <sup>9</sup>í θ∂ n∂máň∂. (CC)
  <sup>9</sup>i<sup>9</sup>∂θ∂ w∂-<sup>9</sup>í θ∂ n∂-máň∂
  be.here EST-be.here ART my-child
  <sup>6</sup>My daughter is here [visible].<sup>7</sup>

Compare (h) and (i), in which the gender is shown by the article only.

(h) ?í wə-?í k w∂ə nə-máňa. (CC)
 AUX(be.here) EST-be.here ART my-child
 'My son is here [but not visible].'

(i) <sup>9</sup>í wə-<sup>9</sup>í lə nə-möňə. (CC)
 AUX(be.here) EST-be.here ART my child
 'My daughter is here [but not visible].'

In (j), (k), and (l), the locative is repeated again in the demonstrative following the oblique particle.

| (j) | ná?ətə ní                            | ? ə təní? (             | (təw처a)   | . (CC)                |         |      |
|-----|--------------------------------------|-------------------------|-----------|-----------------------|---------|------|
|     | ná?ətə                               | ní?                     | $^{9}$ ə  | təní?                 | təwŻá   |      |
|     | be.there                             | be.there                | OBL       | that                  | he      |      |
|     | 'There he                            | is. He is th            | ere.'     |                       |         |      |
| (k) | ná?ə0ə ni                            | í? ə təní?.             | (CC)      |                       |         |      |
|     | ná?ə0ə                               | ní?                     | ?ə        | təní <sup>9</sup>     |         |      |
|     | be.there                             | be.there                | OBL       | that                  |         |      |
|     | 'There she                           | is. She is              | there.'   |                       |         |      |
| (l) | <sup>γ</sup> i∙tə <sup>γ</sup> i∙ tə | o <sup>9</sup> ínə təna | qál. (JI  | P 27)                 |         |      |
|     | <sup>9</sup> i <sup>9</sup> ətə      | ?í                      | ?ə        | tə <sup>9</sup> ínə   | təna    | qэ́l |
|     | be.here                              | be.here                 | OBL       | this                  | this    | bad  |
|     | 'Here is th                          | is bad one.             | ,         |                       |         |      |
| ]   | The plural f                         | orm was re              | corded    | only once,            | in (m). |      |
| (m) | )ná <sup>9</sup> əỷe                 | 1                       | nənhán    | q <sup>w</sup> . (CC) |         |      |
|     | be.over.the                          | ere(PL) ł               | be.fallin | g.asleep              |         |      |

'They are falling asleep.'

Presumably the ye in the locative auxiliary is the element ye that appears in other plural forms. I do not know why the y/ is glottalized.

# **16** Predicate Particles and Tags

Under this heading, I have grouped three sets of particles (as defined in §2.7): the clause-initial particles, the second-position predicate particles, and the sentence-final tags.

#### **16.1. CLAUSE-INITIAL PARTICLES**

These are the two clause subordinaters  $w \partial_{\tau_1}$  'if, when, that,' and  $2\partial t$ -' whenever, whatever, that,' and the two clause nominalizers *s*- 'nominalizer' and  $\delta x^w$ -' oblique nominalizer.' For uses of the first two, see §4.2, "Subordinate Clauses"; for uses of the second two, see §4.3, "Nominalized Clauses."

#### **16.2. SECOND-POSITION PREDICATE PARTICLES**

There are more than twenty particles that can appear within the predicate. Some were introduced in §3.1.2. Here we will examine all that I have recorded. Most of them are mobile, appearing after the first word of the predicate whatever that may be. If the only word is the head, the particle will follow it; if the head is preceded by an auxiliary, the particle will follow the auxiliary; if the auxiliary is preceded by an adverb, the particle will follow the adverb. Two of these particles, however, can become more firmly attached to words and override this principle. A few can also appear in nominal adjuncts.

Four of the second-position particles are main-clause subject markers, two mark tense, one is restrictive in sense, while the rest are, broadly speaking, modal. The subject markers are  $c \partial n$  'I,'  $\check{c}x^w$  'you,' ct 'we,' and  $ce \cdot p$  'you plural' (see §14.2.1). The two tense markers are  $-\partial t$  'past' and  $ce^2$  'future.' The particle  $2a\dot{l}$ , often glossed 'just, only,' is restrictive in sense. The particles  $-\partial t$  'past' and ' $2a\dot{l}$  'just' are the ones that seem occasionally to behave like word suffixes. The modal particles distinguish questions, requests, quality of information and degree of certainty, relationship to previous assumptions, emotional tone, and so on.

Semantically, the modal particles present greater problems than perhaps any other feature of the language. Some are very common and their meanings are generally quite clear, and yet one of these can appear in a rare context and suddenly be pretty opaque. Others seem to occur in only a few contexts, and I have too few examples to be sure of their meanings. All of them, of course, occur in sentences with intonation patterns that have not yet been worked out, and some of them no doubt convey feelings that are expressed by English intonation patterns. An ideal description, therefore, would set the Halkomelem particles in their intonational contexts and gloss them with English words and intonation patterns.

Many of the second-position predicate particles can co-occur and do so in a fixed order. When  $2\partial$  'interrogative' and  $y\partial x^w$  'inferential' occur together,  $2\partial$  always precedes  $y\partial x^w$ ; these two always precede the subject markers;  $m\partial$  'certain' always follows the subject markers; and so on. Some, like  $ce^2$  'future' and  $t\partial$  'imperative,' have never been discovered in the same clause and seem to occupy the same position in relation to the others. Table 16.1 lists the second-position predicate particles by the positions they occupy. In the materials available, however, not all appear often enough with others to make the ordering of all of them certain.

|                                              |           |                         |                          |                     |                    |                       |                       |                            |                           |                     |                          | _ |
|----------------------------------------------|-----------|-------------------------|--------------------------|---------------------|--------------------|-----------------------|-----------------------|----------------------------|---------------------------|---------------------|--------------------------|---|
| Order of second-position predicate particles |           |                         |                          |                     |                    |                       |                       |                            |                           |                     |                          |   |
| 1                                            | 2         | 3                       | 4                        | 5                   | 6                  | 7                     | 8                     | 9                          | 10                        | 11                  | 12                       |   |
| - <i>əł</i><br>'past'                        | ²∂<br>ROG | y∂x <sup>w</sup><br>INF | cən<br>'I'               | <i>с</i> ́ә<br>QUOT | <i>ce</i> ?<br>Fut | <i>²aἰ</i><br>'just'  | <i>м</i> э<br>CERT    | k <sup>w</sup> ∂<br>'then' | k <sup>w</sup> ł<br>'yet' | <i>wa</i> ?<br>PRES | <sup>?</sup> а?а<br>ROG! |   |
|                                              |           |                         | čx <sup>w</sup><br>'you' | ° <i>e∙</i><br>PLPR | <i>łә</i><br>PER   | ² <i>ewəł</i><br>REAS | <i>ċtwa</i> ?<br>SPEC | <i>q́ә</i><br>Емрн         |                           |                     |                          |   |
|                                              |           |                         | ct<br>'we'               |                     |                    | yeł<br>EXP            |                       | <i>θəł</i><br>ADV          |                           |                     |                          |   |
|                                              |           |                         | <i>ce·p</i><br>'you'     |                     |                    |                       |                       |                            |                           |                     |                          |   |

Table 16.1

These particles, with the exception of the coordinate subject markers (*c* $\partial n$ ,  $\check{c}x^w$ , etc.), are described below in the order in which they are listed in Table 16.1.

#### 16.2.1. -*əł* 'past'

Phonologically this behaves like a suffix, having no initial glottal stop and always coalescing with a final vowel, and so I write it as such. Grammatically, however, it is a particle in that it (usually) follows inflectional suffixes (but not altogether consistently). It has, it seems, unusual freedom in where it can occur. It can follow words of any type, including auxiliaries, adverbs, personal words, demonstratives, and others that take few if any other particles or suffixes. Following a verb head in a predicate with no auxiliary,  $-\partial t$  indicates past intention or incipient action, as in (a) and (b).

- (a) čéwət-əł cən. (JP, AG, DK)
   čéw-ət-əł cən
   help-TR-past I
   'I was going to help him.'
- (b) čéwətəł ?ə čx<sup>w</sup>.
  céw-ət-əł ?ə čx<sup>w</sup>.
  help-past ROG you
  'Were you going to help him?'

There are only two occurrences, (c) and (d), of this form in texts dictated by CC, and she interpreted both as simple past.

- (c)  $\vec{k}$  wécətəł cən nəcéləx i wəlstətiq tə nəsnəxcəs. (CC 12) nə-céləx wəł-stətíd\* k<sup>w</sup>éc-ət-əł cən ?əv tə see-TR-past I mv-hand and already-cut.through ART nə-snážcəs mv-finger 'I looked at my hand and my finger was already cut through.'
- (d) sá?asəməł wəłkwócnəxwəs ... (CC 16) sé?-as-əm-əł wəł-kwóc-nəxw-əs raise-head-INTR-past already-see-TR-3TR
  'He lifted his head and saw ...'

However, in these contexts these forms could easily be reinterpreted as incipient past; (c) might better be 'When I turned to look at my hand, my finger was already cut through,' and (d) might better be 'He went to raise his head and saw ...' or 'As he raised his head, he saw ...' In other contexts (with auxiliaries, etc.) there seems to be no question that  $-\partial t$  is simply 'past.'

Following auxiliaries, -*əł* produces what is often translated as an English perfect tense.

(e) niəł (~ ni·ł) cən cewət.

- ni-əł cən ccw-ət AUX-past I help-TR 'I had helped him.'
- (f) ni·ł cən wəł(h)áy qwáqwəl kw xwəné?ent ... (JP 8)
  ni-əł cən wəł-háy qwáqwəl kw
  AUX-past I already-finish be.speaking ART xwə-né?ent
  become-being.night

'I have already spoken to you last night ...'

In relative clauses, it produces what is interpreted as a simple past.

(g) k<sup>w</sup>θə ni·ł ccówəθàmx
 k<sup>w</sup>θə ni-əł cców-ət-Sàmx
 ART AUX-past help-TR-me

'the one who helped me'

There seems to be, however, some variation in where it is placed and its relationship with a subordinate subject marker. Compare (h) and (i):

| (h) | k <sup>w</sup> θə | ni∙nəł ċéwət | (JP)    |
|-----|-------------------|--------------|---------|
|     | k <sup>w</sup> θə | ni-ən-əł     | cew-ət  |
|     | ART               | AUX-I-past   | help-TR |
|     | 'the o            | ne I helped' | -       |

(i) tə niəł cèwəté n swáyqe? (AG)
 tə ni-əł cèw-ət-é n swáyqe?
 ART AUX-past help-TR-I man
 'the man I helped'

In (h), the  $-\partial t$  follows the subordinate subject marker  $-\partial n$  (~  $-\dot{e}\cdot n$ ) 'I,' but in (i), the  $-\partial t$  seems to have pushed the subordinate subject marker over to the predicate head.

In relative clauses, the  $-\partial t$  can also follow the predicate head:

(j) k<sup>w</sup>θ<sup>a</sup> ni háy<sup>a</sup>ł sx<sup>a</sup>x<sup>a</sup>łnét (CC) háy-əł sžəžəłnét k<sup>w</sup>θə ni ART AUX finish-past week 'last week' (k) ni cən wəkwácnəxw kwθə ni há?kwəxəsəł kwsθédts tə qwtá yθən. (JP 23) wə-k<sup>w</sup>éc-nəx<sup>w</sup> ni cən k »θə ni há?k<sup>w</sup>-əx-əs-əł AUX I EST-see-TR AUX be.using-TR-3SUB-past ART k w s-θéd-t-s q<sup>w</sup>tá·yθən tə NOM-be.getting.speared-TR-3POS ART ART sturgeon 'I saw what he was using when he harpooned sturgeon.'

The following examples illustrate the freedom  $-\partial t$  has in its co-occurrence with other types of words. Compare (l) and (m), (n) and (o), and (p) and (q):

(l) qəẁicənəł kʷθə nəménəł.
 qəẁicən-əł kʷθə nə-mén-əł
 Cowichan-past ART my-father-past
 'My late father was Cowichan.'

- (m) qəẁícən k<sup>w</sup>θə nəmén.
   Cowichan ART my-father
   'My father is Cowichan.'
- (n) ?í-ł cən wəcnáx wəł.<br/>?í-əł cən wə-c-snáx wəł<br/>AUX-past I EST-get-canoe<br/>'I had a canoe.'
- (o) <sup>9</sup>í cən wəcnə́x <sup>w</sup>əł. <sup>9</sup>í cən wə-c-snə́x <sup>w</sup>əł AUX I EST-get-canoe <sup>6</sup>I have a canoe.<sup>7</sup>
- (p) ?ówè·ł té? nosnóx woł.
   ?ówo-oł té? no-snóx woł not-past appear my-canoe
   'I didn't have a canoe.'
- (q) ?ówəté? nəsnóx wəł.
   ?ówəté? nə-snóx wəł none (not.appear) my-canoe
   'I don't have a canoe.'
- (r) mớk w-əł nə-s-ném ... all-past my-NOM-go 'Whenever I went ...'
- (s) <sup>9</sup>é·nθe·ł nem ...
   <sup>9</sup>é·nθə-əł nem be.I-past go
   <sup>6</sup> 'I was the one who went ...'
- (t) stéməł yəx<sup>w</sup> <sup>2</sup>a<sup>2</sup>a.
  stém-əł yəx<sup>w</sup> <sup>2</sup>a<sup>2</sup>a
  what-past INF ROG!
  'I wonder what it could have been.'
- (u) k<sup>w</sup>ona·ł spónwo (DK)
   k<sup>w</sup>ona·oł spónwo
   that(MR)-past summer[?]
   'last summer'

With possessed nouns, the position of  $-\partial t$  'past' in relation to the possessive seems variable. Compare (v) and (w):

- (v) k<sup>w</sup>θ s-yəwenderi kardıra s-yəwenderi kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri kardıra seri seri kardıra seri seri kard
- (w) k<sup>w</sup>θ s-yəŵén-s-əł (JP)
   ART NOM-before-3POS-past
   'his ancestors'

The  $-\partial t$  'past' is followed by a lexical suffix in at least one word:  $\delta x^w n \dot{e} t \partial t q \partial n$ 'breakfast' (<  $n \dot{e} t \partial t$  'morning,' itself <  $n \dot{e} t$  'night').

## 16.2.2. % 'interrogative' (ROG)

This converts a statement or possible statement into a yes-or-no question. It does not occur with an interrogative word.

- (a) ni ?a 1áč full AUX ROG 'Is it full?' (cf. ni láć. 'It's full.') ?a ?é·łtən. (b) ném go ROG 3PL 'Are they going?' (cf. *ném 'é tran*. 'They're going.') (c)  $spé^{9}e\theta$ ?ə. bear ROG 'Is it a bear?' (cf.  $sp e^{2}e\theta c\partial$ . 'It's said to be a bear.') (d) ni ?ə čx<sup>w</sup> k<sup>w</sup>ácnəx<sup>w</sup>. ni ?a čxw k<sup>w</sup>éc-n-əx<sup>w</sup> ROG see-TR-you AUX you 'Did you see him?' (e) Ža ?a ?ən-háĺk™. your-be.breaking [pocketknife] BE3P ROG 'Is it your pocketknife?' (f) cew-ət-əł ?ə čx<sup>w</sup>.
- (f) cew-ət-əł 'ə cxw. help-TR-past ROG you 'Were you going to help him?'

The interrogative particle tends to assume the quality of a preceding vowel or to coalesce with it; thus  $\dot{\lambda}a \,^2 \partial$  may become  $\dot{\lambda}a \,^2 a$ ,  $ni \,^2 \partial$  may become ni, and so on. Following an unstressed schwa, the interrogative particle may cause the stress to shift to it and may coalesce with it as a long e. Thus //?áwə ?ə// quite regularly becomes  $2\partial w \dot{e} \,^2 e$  or  $2\partial w \dot{e}$ , while //-ámə ?ə// may become  $-am \dot{e} \,^2 e$  or  $-am \dot{e}$ , as in (g) and (h).

- (g) <sup>?</sup>əwé<sup>.</sup> cən ném̀e<sup>.</sup>n. <sup>?</sup>ówə <sup>?</sup>ə cən néṁ-e<sup>.</sup>n not ROG I go-I <sup>°</sup>Can't I go?'
- (h) x<sup>w</sup>əyθàmé <sup>?</sup>e cən.
  x<sup>w</sup>əy-t-Samə <sup>?</sup>ə cən
  wake-TR-you ROG I
  'Shall I wake you?'

#### 16.2.3. y a x<sup>w</sup> 'inferential' (INF)

This indicates that a statement is based on inference. In translations we often find 'must' in the sense of necessary inference (e.g., 'He must have gone') as opposed to obligation (e.g., 'I must go').

- (a) ni yəx<sup>w</sup> wəł-háye?. AUX INF already-leave 'He must have left.'
- (b) ni yəx<sup>w</sup> k<sup>w</sup>ə́n-ət-əm. AUX INF be.taken-TR-INTR 'It must have been taken.'
- (c) ni yəx<sup>w</sup> ct nəq<sup>w</sup>-námət.
   AUX INF we fall.asleep-self
   'We must have fallen asleep.'

The combination  $2^{\circ}$  'interrogative' and  $y \partial x^{w}$  'inferential' is usually translated 'I wonder if ...,' as in (d) and (e).

| (d) | ?i cən   | ı x <sup>w</sup> t <sup>0</sup> q <sup>w</sup> a | áləs. ?əwé ?e yəx* čx* t    | é?təx™ m  | é·łcəm.    | (JP)             |                 |
|-----|----------|--------------------------------------------------|-----------------------------|-----------|------------|------------------|-----------------|
|     | ?i       | cən                                              | xʷ-ṫ <sup>0</sup> əḍʷ-áləs  | ?∍wé      | °e         | yəx <sup>w</sup> | čx <sup>w</sup> |
|     | AUX      | Ι                                                | inward-get.pierced-eye      | not       | ROG        | INF              | you             |
|     | ἰé?-     | t-əx <sup>w</sup>                                | mé <sup>9</sup> -łc-əm.     |           |            |                  |                 |
|     | try-     | TR <b>-</b> you                                  | be.removed-BEN-INTR         |           |            |                  |                 |
|     | 'I got : | somethin                                         | ng in my eye. I wonder if y | you would | l try to g | et it out f      | `or me.'        |

(e) ni <sup>?</sup> ə y ax<sup>w</sup> <sup>?</sup> us θ a b a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s n a s

For other uses of  $y \partial x^w$  with  $\dot{m}\partial$  and other particles, see those below.

## 16.2.4. *с*́*ә* 'quotative' (QUOT)

This indicates that the statement is based on hearsay.

- (a) ném čə təwĂa. go QUOT that(MP) 'He is going, they say.'
- (b) spé<sup>γ</sup>eθ ċə.
  bear QUOT
  'It is said to be a bear. It is supposed to be a bear.'
- (c) sÅéləqəm-əł ċə. (JP 6) dangerous.being-past QUOT
   'It is said that it was a dangerous being.'
- (d) ni cən cə ném. AUX I QUOT go 'I'm supposed to have gone. It is said that I went.'
- (e) ni ċə t<sup>θ</sup>əxtén-t-əm.
   AUX QUOT poison-TR-INTR
   'He was poisoned, according to rumour.'

Compare (e) with (f).

(f) ni  $\dot{m}$ ə  $\dot{t}^{\theta}$ əxtén-t-əm. AUX CERT poison-TR-INTR 'He was certainly poisoned.'

Preceding the particles  $ce^{2}$  'future' and  $\theta \partial t$  'adversative,'  $\dot{c} \partial$  appears as  $\dot{c}$ . See  $ce^{2}$  for an example of  $\dot{c}ce^{2}$  and  $\theta \partial t$  for an example of  $\dot{c}\theta \partial t$ .

## 16.2.5. <sup>9</sup>e<sup>•</sup> 'plural imperative' (PLPER)

This expresses a request or instruction to several. It may, but need not, be followed by  $l \partial$  'imperative-optative.'

- (a) háy <sup>?</sup>e. (AG) stop PLPER 'Cut it out, you people!'
- (b) łí·m ?e·. (AG) pick PLPER 'Pick (you several)!'
- (c) łí·m <sup>9</sup>e· łe, łí·m <sup>9</sup>e· łe. (CC)
   pick PLPER PER pick PLPER PER
   'You kids keep on picking.'
- (d) k<sup>w</sup>óyx̆-θət <sup>2</sup>e· łe. (CC) move-self PLPER PER
   <sup>°</sup>Do something about it, all of you.<sup>°</sup>

(e) ?5łton ?è· łe. (JP)
eat PLPER PER
'Go ahead and eat it, all of you.' (to children at the table)

AG, in contrast to CC and JP, put the <sup>?</sup>e<sup>.</sup> after the *le* when using both in (f).

(f) ?i łé ?e: AUX PLPER PER 'Let's go.'

#### 16.2.6. ce? 'future' (FUT)

This simply indicates future time.

- (a) k<sup>w</sup>áq<sup>w</sup> čx<sup>w</sup> ce?.
   get.hit you FUT
   'You'll get hit.'
- (b) céw-əθ cən ce?. help-you I FUT 'I'll help you.'
- (c) <sup>?</sup>é·nθə ce<sup>?</sup> c<sup>3</sup>m-ət.
  be.I FUT carry.on.back-TR
  'I will pack it. I will be the one who will pack it.'
- (d) ném čx<sup>w</sup> sówdom k<sup>w</sup> hóýq<sup>w</sup>ct ce?.
  ném čx<sup>w</sup> sówdom k<sup>w</sup> hóýq<sup>w</sup>ct ce?
  go you seek-INTR ART be.burning-our FUT
  'Go look for what will be our fire (i.e., firewood).'
- (e) ?əpé·lə ct ce? wəni·t háye?.
  ?əpən-é·lə ct ce? wə-ni-ət háye?
  ten-person we FUT when-AUX-we leave
  'There will be ten of us when we go.'
- (f) wə-nét-əs ce? ... when-be.night-3SUB FUT 'When it is night ...'
- (g) nə-s-wé? ce? nə-sẳpíwən. my-NOM-own FUT my-shirt 'It will be my shirt.'
- (h) ni <sup>?</sup> y y x<sup>w</sup> ce<sup>?</sup> w y ném.
  ni <sup>?</sup> y y x<sup>w</sup> ce<sup>?</sup> w y ném
  AUX ROG INF FUT EST-go
  'I wonder if he will go.'

- (i) k<sup>w</sup>ónnəx<sup>w</sup> yəx<sup>w</sup> cən ce? k<sup>w</sup> télə. (JP) k<sup>w</sup>ón-nəx<sup>w</sup> yəx<sup>w</sup> cən ce? k<sup>w</sup> télə take-TR INF I FUT ART money 'I must be going to get some money.' (because my hand itches)
- (j) scécản yax ce? waqáž spédam. (CC) scécan yax ce? wa-qáž spédam truly INF FUT EST-many flower
   'There should really be lots of flowers.' (the tree being covered with buds)

(k) ném ct <sup>\*</sup>λé<sup>γ</sup>t k<sup>w</sup>θ<sup>3</sup> m<sup>3</sup>stéy<sup>3</sup>x<sup>w</sup> ni k<sup>w</sup>θ<sup>3</sup> n<sup>3</sup>ca<sup>γ</sup> ce<sup>γ</sup> swéy<sup>3</sup>l. (JP 16) λé?-t ném ct k<sup>w</sup>θə məstévəx<sup>w</sup> ni [?ə] we gather-TR person be there go ART OBL ce? k<sup>w</sup>θə n śća? swéyəl FUT dav ART one

'We are going to call the people for a coming day.' (for a potlatch)

 (l) ném cce?. (AG) ném co ce?
 go QUOT FUT
 'They say he'll go.'

The sequence  $y \partial x^w$  'inferential' and  $ce^2$  'future' also seems to have a special meaning in the formal speeches of the winter dance house, as in (m) and (n).

(m) nówo yox w ce? xá?łomot to hóygw, wonémos ce? dwoyílox to ž∍wsálk™ł. xá?łəm-ət háỷqw náwa vəx<sup>w</sup> ce? tə be.you INF agree-TR ART be.burning FUT žeŵs-álk™ł. ce? d<sup>w</sup>əyíləx wə-ném-əs tə when-AUX(go)-3SUB FUT dance ART new-dancer 'You will (honour us to) be the one to look after the fire, when the new dancer dances '

(n) <sup>9</sup>é·nθə yəx<sup>w</sup> ce<sup>9</sup>.
be.I INF FUT
'I am (honoured to be) the one.' (said in reply to the preceding)

The particle  $ce^{\gamma}$  is joined with  $wa_{-2}$  in (o).

(o) ha? cew səwóyqe? i wəlé·llàmət ni kwəw ?ónəcə ?al. (JP 22) wə-səwáyqe? ?∍ỷ k<sup>™</sup>ə ha? wə-łé·l-nàmət ni ce? if EST-men and EST-land-self FUT be at ART ?aĺ wə-?śnəcə EST-where iust

'If they are men, they will be able to land somewhere.'

#### 16.2.7. *b* ~ *le* 'imperative-optative' (PER)

This expresses a request or a wish. A request or command can consist of the verb alone or with a second-person subject particle or (if made to several) with  ${}^{2}e$  'plural imperative,' as in (a) to (d).

| (a) | néṁ.    |              | (b) | néṁ     | čx <sup>w</sup> . |
|-----|---------|--------------|-----|---------|-------------------|
|     | go      |              |     | go      | you               |
|     | 'Go!'   |              |     | (You)   | go!'              |
| (c) | néṁ     | ce·p.        | (d) | néṁ     | <sup>9</sup> e.   |
|     | go      | you(PL)      |     | go      | PLPER             |
|     | '(You p | people) go!' |     | '(You p | eople) go!'       |

But a request may also be expressed with  $l_{\partial}$ , as in (e) to (l).

- (e) ném łə. (f) ném <sup>2</sup>e· łe. go PER go PLPER PER 'Go!'
  (g) héỷ łə ném. (h) <sup>2</sup>əmí ła čéwaθàmx.
- (g) héỷ là ném. go.ahead PER go 'Let's go! Go ahead and go!'
- (i) tá?θət ła cxiám.
   te?-θat ła cxiám
   test-self PER jump
   'Try to jump.'
- (k) t<sup>2</sup><sup>6</sup>fx<sup>w</sup>əm 1ə cew-ət. pity PER help-TR 'Please help him.'
- 'Come and help me.'
  (j) ?áx "əst lə təwẩa. ?ex "-əs-t lə təwẩa

łə

PER

céw-ət-Sàmx

help-TR-me

- give-RECIP-TR PER that(MP) 'Give it to him.'
- (l) nówə-s łə. be.you-PERM PER 'You be the one.'

?omí

come

The particle  $l \partial$  may combine with  $k^w \partial$  'consequential' as  $lk^w \partial$ :

(m)<sup>9</sup>ámət łk<sup>w</sup>ə.

<sup>9</sup>ómət lə kwə
sit PER then
'Sit down then. Try to sit down. Sit down and let's see how it goes.'

It may also combine with  $\dot{q}\partial$  'emphatic' as  $\dot{t}\dot{q}\partial$ :

(n) ?ômət łda.
?ômət la da
sit PER EMPH
'Sit down, won't you! For Pete's sake, sit down!'

The combination of  $l_{\partial}$  and  ${}^{2}a\dot{l}$  'restrictive' expresses hope. The usual translation is 'I hope (that) ...,' but a more literal translation might be 'may it be that ...' Examples are (o) to (u).

| (0) | mí                  |                        | łə               | ?aĺ                        | x <sup>∞</sup> ə- <sup>9</sup> í. |                           |
|-----|---------------------|------------------------|------------------|----------------------------|-----------------------------------|---------------------------|
|     | AUX(co<br>'I hope   | me)<br>he co           | PER<br>mes.'     | just                       | become-be.h                       | ere                       |
| (p) | mí<br>come          | čx <sup>w</sup><br>vou | łə<br>PER        | ?aĺ<br>iust                | x™ə- <sup>9</sup> í.<br>become-be | here                      |
|     | 'I hope             | you c                  | ome.'            | J                          |                                   |                           |
| (q) | ni?                 | łə                     | ?aĺ              | х́té <sup>9</sup> -s       | stəx <sup>w</sup> .               |                           |
|     | AUX<br>'I hope      | PER<br>he do           | just<br>es it.'  | do-co                      | М                                 |                           |
| (r) | ni?                 | čxw                    | łə               | ?ai≀                       | žté?-stəx™.                       |                           |
|     | AUX<br>'I hope      | you<br>you d           | PER<br>o it.'    | just                       | do-сом                            |                           |
| (s) | ni <sup>9</sup> cər | n łə 🤈                 | al kʷác          | nəx <sup>w</sup> .         |                                   |                           |
|     | ni?                 | cən                    | łə               | ?aĺ                        | k <sup>™</sup> ec-n∍x             | W                         |
|     | AUX<br>'I hope      | I<br>I see             | PER<br>it.'      | Just                       | see-TR                            |                           |
| (t) | wə-θəʻ              | Pít 🛛                  | łə               | ?aİ                        | técəl.                            |                           |
|     | EST-true<br>'I hope | e<br>it's tri          | PER<br>ue that l | just<br>ne come            | arrive.here<br>s.'                |                           |
| (u) | ṁí                  |                        | łə               | ?aİ                        | wə-θə?ít                          | técəl.                    |
|     | AUX(co<br>'I hope   | me)<br>he rea          | PER<br>ally com  | just<br>nes.'              | EST-true                          | arrive.here               |
| T   | The co-o            | ccurre                 | ence of a        | $t_{\partial}$ and $^{2}a$ | il may, howey                     | ver, be accidental, as in |

(v) wə-?í lə ?al. EST-be.here PER just 'Stay here!'

in which  $l \partial$  makes a command of  $w \partial^2 i \partial^2 a \dot{l}$  'just be here.'

This particle appears stressed as  $l\acute{e}$  following ?i or  $he\acute{y}$  with the sense 'Let's' (see §3.2.4).

# 16.2.8. ?al 'just'

This often serves to limit in some way the word it follows. Often, though not always, this word is preceded by  $w_{\partial_2}$  'established' (§11.1.1). Often a translation is not needed, but if one is needed, it is usually 'just' or 'only.' The

particle is unusual in that it commonly follows the predicate head rather than simply the first word in the predicate.

When the predicate head stands alone,  ${}^{2}al$  appears in the order indicated by its position in Table 16.1, namely, following  $ce^{2}$  'future' or  $l_{\partial}$  'imperative' and preceding  $\dot{m}_{\partial}$  'certain' or  $k^{w}_{\partial}$  'then,' as in (a), (b), and (c).

- (a) dólom ce? ?al ?é·łtən. camp FUT just 3PL 'They'll just stay (camp) there.' (b)  $ni^{?}$  ce<sup>?</sup> ?al t $\theta e^{?}$  k<sup>w</sup>scev x<sup>w</sup>s. ní? ce? ?aĺ tθé? s-čéỷ x<sup>w</sup>-s k w NOM-dry-3POS AUX ROG just that/there ART 'It will stay there to get dry.'
- (c) k<sup>w</sup>ék<sup>w</sup> ⇒c ce ·p ?al k<sup>w</sup> ⇒.
   be.looking you(PL) just then 'Just watch then.'

Normally the al follows an auxiliary as in (c); however, a predicate head enclosed by the combination  $wa_2 \dots al$  can be preceded by an auxiliary, as in (d).

| (d) | ni cən                                  | wətítəq | ™nəx <sup>w</sup> ?al. |      |  |  |  |  |
|-----|-----------------------------------------|---------|------------------------|------|--|--|--|--|
|     | ni                                      | cən     | wə-títəq॑ʷ-nəx ʷ       | °aĺ. |  |  |  |  |
|     | AUX                                     | Ι       | EST-be.bumping-TR      | just |  |  |  |  |
|     | 'I just kept accidentally bumping him.' |         |                        |      |  |  |  |  |

Moreover, as indicated elsewhere (§3.8.3),  $wa_2 \dots al$  can enclose what are simply adjective-like forms, in which this combination of prefix and particle seems to have no lexical meaning at all.

- (e) <sup>\*</sup>λa há<sup>?</sup>k<sup>w</sup> ni k<sup>w</sup>θəw sxéxəm <sup>?</sup>al qá<sup>?</sup>. (JP 23) Хa há?k\* ni [?ə] k<sup>w</sup>θə wə-sxéxəm ?aĺ aá? be used BE3P be at EST-shallow OBL ART just water 'That was the one used in shallow water.'
- (f) ?i ?ə čx<sup>w</sup> wə-?śỷ ?al.
  AUX ROG you EST-good just
  'How are you?' (lit. 'Are you well?')
- (g) k<sup>w</sup>əw X<sup>\*</sup>əx<sup>w</sup>lá<sup>2</sup>as <sup>2</sup>al məst<sup>5</sup>yəx<sup>w</sup> k<sup>w</sup>ə wə-X<sup>\*</sup>əx<sup>w</sup>lá<sup>2</sup>as <sup>2</sup>al məst<sup>5</sup>yəx<sup>w</sup> ART EST-of.no.account just person 'any common people'

With  $h \dot{a} y$  'be singled out,'  ${}^{2}a \dot{l}$  (without the  $w \partial_{-2}$ ) expresses the superlative, as in (h), (i), and (j).

(h) Xa ckwə tə háy <sup>9</sup>al cəcí X ... (JP 1) ?al ċəċí∙ネ х́а ćə k »ə tə háv BE3P OUOT then specifically iust short ART 'It was said to be the shortest ...' (i) Χ̃a háy <sup>γ</sup>al θí təna sq<sup>w</sup>əméy (x<sup>w</sup>ném <sup>γ</sup>ə tə cłsq<sup>w</sup>əmq<sup>w</sup>əméys). (AG) х́а háy ?aĺ θí təna sq<sup>w</sup>əméy x<sup>w</sup>ném ?a specifically iust BE3P big this dog to OBL tə ċł-sqwəmqwəméy-s co-dogs-3POS ART 'This dog is the biggest (of the dogs).' (lit. 'among his fellow dogs') (j) <sup>?</sup>iəł cən cká· <sup>?</sup>ə k<sup>w</sup>θə háy <sup>?</sup>al qól. (AG) ?al ?i-əł cən c-ká∙ ?ə k»θa háy qál AUX-past I get-car OBL ART specifically just bad 'I had a car that was the worst ' With a negative,  ${}^{2}al$  can have the force of 'at all,' as in (k). (k) <sup>?</sup>ówə cən <sup>?</sup>al k<sup>w</sup>ócnəx<sup>w</sup>è·n. ?áwa cən ?aĺ k<sup>w</sup>éc-n∍x<sup>w</sup>-e·n I not iust see-TR-I 'I won't see him at all.' With interrogative words,  ${}^{2}a\dot{l}$  (usually with  $wa_{2}$ ) forms their indefinite counterparts, as in (1) to (q). (1) k<sup>w</sup>aw stém <sup>9</sup>al k<sup>™</sup>ə ?aĺ wə-stém EST-what just ART 'something' (m) k<sup>w</sup> aw wét <sup>9</sup>al k<sup>™</sup>ə ?aĺ wa-wét ART EST-who just 'somebody' (n) k<sup>w</sup>əw k<sup>w</sup>ín <sup>9</sup>al k<sup>™</sup>ə wə-k<sup>w</sup>ín ?aĺ ART EST-how.many just 'a few'

(o) ?əłnéməs ?áləxətəs k<sup>w</sup>θəw stém ?al
 ?əł-ném-əs ?áləx-ət-əs k<sup>w</sup>θə wə-stém ?al
 whenever-go-3SUB get.food-TR-3TR ART EST-what just
 'whenever they went out after anything'

- (p) k<sup>w</sup>θəw mók<sup>w</sup> ?al stèm k<sup>w</sup>θə wə-mók<sup>w</sup> ?al stèm ART EST-all just what 'anything' (cf. mók<sup>w</sup> stèm 'everything')
- (q) <sup>?</sup> anca k<sup>w</sup>an słédat. <sup>?</sup> anca <sup>?</sup> al. (DK)
   <sup>?</sup> anca k<sup>w</sup>a na-s-łéd-at. <sup>?</sup> anca <sup>?</sup> al where ART my-NOM-lie-TR where just 'Where shall I lay this? Anywhere.'

## 16.2.9. ?ewəł 'reassuring' (REAS)

This is unusual in that it can bear a stress. It expresses the feeling that the hearer should already have the information.

(a) ni ?ewəł k<sup>w</sup>ánətəm ?ə<sup>\*</sup> John. (AG)
ni ?ewəł k<sup>w</sup>ón-ət-əm ?ə <sup>\*</sup> John
AUX REAS take-TR-INTR OBL ART John
'John took it (lit. "It was taken by John"), as I've already mentioned (or "as you should know").'

It is also used as a polite way of making a request, as in (b) to (e).

- (b) ?áłtan ?e· ?éwał, si·?ém. (JP)
  eat PLPER REAS Gentlefolk
  'Go ahead and enjoy your meal.'
  'Just go ahead and eat.' (in a plaintive tone) (AG's reading of JP's sentence)
- (c) hèỷ <sup>?</sup>éwəł. go.ahead REAS 'Goodbye.'
- (d) <sup>2</sup>əswá<sup>2</sup>θətənələp <sup>2</sup>éwəł, si<sup>2</sup>ém. (CC)
   <sup>2</sup>ə-s-wá<sup>2</sup>θətən-ələp <sup>2</sup>éwəł si<sup>2</sup>ém
   your-NOM-sing.along.with.dancer-your(PL) REAS honoured.ones
   'With your help, ladies and gentlemen.'

(CC used these words to thank the people when her husband danced in the winter dance.)

(e) ném łe <sup>?</sup>éwəł. (AG) go PER REAS 'Go ahead then.'

## 16.2.10. *yet* 'expectable' (EXP)

This expresses the feelings of familiarity, affection, or exasperation conveyed by *again, as usual, that old* ..., and so on.

- (a) <sup>9</sup>i čx<sup>w</sup> wəctámət. ni[·] čx<sup>w</sup> ýeł cmədmədəm. (JP 4) ?i čxw wa-ctámat ?a čxw **v**eł ni EST-do.what.with.oneself AUX AUX vou ROG you EXP cməqməqəm over.eat 'What's the matter with you? Have you overeaten as usual?' (b) ni cən yeł k wəšúθət. ni cən **v**eł k™əšú-θət EXT pig-self AUX T 'I made a pig of myself as usual.' (AG) (c)  $\dot{\lambda}a$  yeł  $\dot{k}^{w}\dot{a}\dot{k}^{w}\dot{a}q^{w}n\dot{a}c\dot{i}$ . (CC 3) Żа k<sup>w</sup>àk<sup>w</sup>əq<sup>w</sup>-nəc-í·ls **v**eł BE3P EXP be.striking-butt-ACT 'It's those Little Choppers [forest dwarves] again.' (d) Åa **v**eł Mary. (DK) EXP Mary BE3P
  - 'It's that old Mary. It's Mary again.'
- (e) mi čx<sup>w</sup> veł hé?k<sup>w</sup> a k<sup>w</sup>ań cłwét. (JP) k<sup>™</sup>ə ?ən-cł-wét mi čxw **v**eł hé?k\* ə AUX(come) vou EXP remember ART your-fellow-who OBL 'Remember as usual your fellow being (the other poor guy like yourself).'

## 16.2.11. *m*<sup>*i*</sup> (sometimes *m*<sup>*e*</sup>) 'certain' (CERT)

This indicates that there is no doubt about the statement.

- (a) spé<sup>?</sup>eθ ṁ<sub>.</sub>
  bear CERT
  'It's a bear.' (certainly, observably, cf. spé<sup>?</sup>eθ ċ<sub>.</sub> ċ<sub>.</sub> 'It's said to be a bear'; spé<sup>?</sup>eθ ċ<sub>.</sub> ċ<sub>.</sub> wa<sup>?</sup> 'It might be a bear')
- (b) wəłθí?θə mɨ θəwka.
   ritualist CERT she
   'She is a ritualist.'
- (c) Xa m → k<sup>w</sup>θ n → h → lk<sup>w</sup>.
   BE3P CERT ART my-be.breaking
   'It is my pocketknife.' (that he has in his possession, as I noticed just now)
- (d)  $\dot{\lambda}a$   $\dot{m}$ ə  $\dot{\gamma}\epsilon$   $n\theta$ ə sqəl  $\dot{a}$  cəməs. BE3P CERT I sqəl  $\dot{a}$  cəməs  $\dot{\gamma}$ I am Sqəl  $\dot{a}$  cəməs.

- (e) Xa mö nösöwiném.
   Xa mö nö-s-wöi-ném
   BE3P CERT my-NOM-already-go
   'I am going now.'
- (f) ni čx<sup>w</sup> m
   <sup>w</sup> céwəθ.
   ni čx<sup>w</sup> m
   <sup>w</sup> céw-ət-S
   AUX you CERT help-TR-me
   'You did help me.'
- (g) páq<sup>w</sup> cən ce? m̀ə. go.broke I FUT CERT 'I'll go broke.'
- (h) nəsẳí? mɨə.
   nə-s-c-ẳí? mɨə
   my-NOM-make-valuable CERT
   'I want it.'
- (i) <sup>9</sup>ówəte<sup>9</sup> mə łə nəwáč, ni yəx<sup>w</sup> sqén. ?áwate? mэ łə nə-wáč ni yəx <sup>w</sup> s-aén be none CERT mv-watch ART AUX NOM-steal INF 'My watch is not there; it must have been stolen.' (lit. 'something stolen')

This particle can, and often does, occur with  $y \partial x^w$  'inferential' to add to the certainty of the inference.

- (j) spé<sup>2</sup>eθ yəx<sup>w</sup> mö.
   bear INF CERT
   'It must be a bear.'
- (k) spé<sup>9</sup>eθ-əł yəx<sup>w</sup> m.
   bear-past INF CERT
   'It must have been a bear.'
- (1) ?i·ł yəx<sup>w</sup> cən mə ?í?tət.
  ?i-əł yəx<sup>w</sup> cən mə ?í?tət
  AUX-past INF I CERT be.sleeping
  'I must have been asleep.'

It can also co-occur with  $\dot{c}\partial$  'quotative,' as in the following, where it presumably indicates conviction in what someone else has reported. (m)  $^{9}a$  ni cə məw  $\theta$ ə<sup>9</sup>*it*, wə $\theta$ ə<sup>9</sup>*it* cə ni mé<sup>9</sup>k<sup>w</sup>ł. (JP) 22. ni ćə mə wa- $\theta$ a?ít wa- $\theta$ a?ít ćə ni oh AUX OUOT CERT EST-frue EST-true OUOT AUX mé?k<sup>w</sup>ł be.injured 'Oh, it's true; he really got hurt.'

(n) <sup>9</sup>uθə<sup>9</sup>ít cɨ mɨ. (AG, prompted by the last sentence) wə-θə<sup>9</sup>ít cɨ mɨ.
EST-true QUOT CERT
'It's reported that it's true.'

In a phrase introduced by the article  $\vec{k}^{w}$  'indefinite,'  $\vec{m}\partial$  has the effect of positively affirming an identity.

- (o) wənan wəskeləqəm, kwəw stəlqayə mə. (JP 3)
   wənan wəskeləqəm kwə wəstəlqayə mə
   EST-very EST-fierce ART EST-wolves CERT
   'They were too fierce of course, they were wolves.'
- (p) k<sup>w</sup>əw spé<sup>γ</sup>eθ m<sub>2</sub>.

k<sup>w</sup> = w=-spé?eθ m=
ART EST-bear CERT
'Of course, it's a bear.' (which caused something such as a dog being mauled) (JP)
'What can you expect of a bear?' (as when one has broken into a cooler) (CC)

## 16.2.12. *ctwa*? 'speculative' (SPEC)

This indicates that the statement is based on supposition. It is weaker than  $y \partial x^w$  'inferential' in sentences like the following:

- (a) ni čtwa? wəłk wónətəm.
  - ni čtwa? wəł-kwón-ət-əm
    AUX SPEC already-be.taken-TR-INTR
    'It may have been taken by now.' (cf. ni yəx<sup>w</sup> k<sup>w</sup>ónətəm. 'It must have been taken.')
- (b)  $spé^{\gamma}e\theta$  ctwa<sup> $\gamma$ </sup>

bear SPEC

'It might be a bear.' (cf.  $sp\acute{e}^{2}e\theta y \partial x^{w} \dot{m}\partial$ . 'It must be a bear.')

(c) Xáləm čtwa? yəná-ltən.
 be.they SPEC those
 'These must [may] be the ones.'

It occurs often in phrases referring to points in time or the passage of time to indicate estimation, as in (d), (e), and (f).

| (d) | ni                                            | ċtwa?      | wəł-tàx'    | N        | swéyəl.    |     |  |  |  |
|-----|-----------------------------------------------|------------|-------------|----------|------------|-----|--|--|--|
|     | AUX                                           | SPEC       | already-r   | ight.on  | day        |     |  |  |  |
|     | 'It mus                                       | t be noon  | by now.'    |          |            |     |  |  |  |
| (e) | ni                                            | ċtwa?      | wəł-xə?a    | íθən     | tíntən     | i   |  |  |  |
|     | AUX                                           | SPEC       | already-f   | our      | clock      | and |  |  |  |
|     | 'It mus                                       | t have bee | en four o'o | clock wł | nen'       |     |  |  |  |
| (f) | ni                                            | ċtwa?      | isélə       | tə       | nét-əł.    |     |  |  |  |
|     | AUX                                           | SPEC       | two         | ART      | night-past |     |  |  |  |
|     | 'It must have been about two in the morning.' |            |             |          |            |     |  |  |  |

It also occurs in narratives in attributive phrases, evidently to indicate that the attribution is based on the evidence of tradition rather than observation, as in (g).

(g) swé?s sk<sup>w</sup>óỷxθət k<sup>w</sup>θə θí·ł ċtwa? məstéyəx<sup>w</sup>, stáməx, qìyəplénəx<sup>w</sup>.
 (JP 14)

| s-wé <sup>9</sup> -s | s-k*áýx-6 | Əət    | k™θә                | θí-ə·ł   | ćtwa? |
|----------------------|-----------|--------|---------------------|----------|-------|
| NOM-own-3POS         | NOM-move  | e-self | ART                 | big-past | SPEC  |
| məstéyəx™            | stáməx    | qìyə   | plénəx <sup>w</sup> |          |       |
| person               | warrior   | Capi   | lano                |          |       |
|                      |           |        |                     |          |       |

'It was the work of that great person that was, the warrior, Capilano.' (or perhaps 'the great person he must have been')

#### 16.2.13. k = a - k = e 'then'

This is a weak 'then' in the sense of 'therefore, consequently, so,' and so on. It occurs very commonly in questions but also appears in instructions and in simple statements. In questions especially its effect is so weak that it may be better omitted in a free translation.

- (a) stém k<sup>w</sup> τθe?.
   what then that
   'What is that (then)? (So) what is that?'
- (b) stém ce? k<sup>w</sup>e. what FUT then 'What will it be (then)?'
- (c) stém k v t k v nétax v. stém k v t k v né-t-ax v what then ART hold-TR-you 'What is it you are holding?'

- (d) wét k<sup>w</sup> tθe? swýge?.
   who then that man
   'Who is that man?'
- (e) wét kwə kwə ?i Żénəq.
   who then ART AUX be.potlatching
   'Who is it (then) who is potlatching?'
- (f) ctámət cən ce? k<sup>w</sup>e. do.what I FUT then 'What shall I do (then)?'
- (g) ném čx<sup>w</sup> k<sup>w</sup>ə. go you then 'Are you going (then)?'
- (h) ?ômət čx<sup>w</sup> k<sup>w</sup>ə, si?ém, ?əłtən.
   sit you then sir eat
   'Sit down (then), sir, and eat.'
- (i) héỷ čx<sup>w</sup> k<sup>w</sup>ə. go.ahead you then 'Go ahead (then).'
- (j) Ždéý k<sup>w</sup>ə.
   how.far then
   'My, it's far!'
- (k) ni? ctwa? k<sup>w</sup> ž ž β²άθ syslán syslán i ...
   AUX SPEC then four years and 'It was about four years later when ...' (CC 20)
- (l) x " >ná c>n c> ?al k "> x "X > yk ">sθám > i w>łc X > mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathematica mathem

| x <sup>w</sup> əná? | cən     | ce''   | 'al            | k <sup>w</sup> ə | x "X əyk "-əs | -t-Sámə              | "əý   |
|---------------------|---------|--------|----------------|------------------|---------------|----------------------|-------|
| first               | Ι       | FUT    | just           | then             | wink-RECIP-T  | rr-you               | and   |
| wəł-cź              | ám-ət   | č      | X <sup>w</sup> | k™θә             | ná?ənċa?      | ť <sup>θ</sup> éc-ət |       |
| already             | -jump-т | rr y   | ou             | ART              | one.person    | pull.hair-           | TR    |
| (11)                | 1 (11   | × (1 ) | т · ·          | i .              | • 4           | 1                    | 1 1 . |

The moment (then) that I wink at you, you jump the one and grab him by the hair.'

(m) ném k wə.

'Now go.' (said in a low voice to someone impatient to leave, at a suitable moment, as when a speaker has finished his speech [AG])

#### 16.2.14. *q*́*ә* 'emphatic' (ЕМРН)

This gives emphasis to a statement and may also have an adversative quality.

- (a) cá∵k<sup>w</sup> ἀə. far EMPH 'It's *fa---r!*'
- (b) spé<sup>2</sup> >θ d' =.
   bear EMPH
   'It's a bear!'
- (c) <sup>9</sup>i cən qə t<sup>θ</sup>át<sup>θ</sup>əłəm, məna.
   AUX I EMPH cold child
   'I'm *cold*, child.' (though you may be warm, so don't open the door) (CC's sentence, AG's interpretation)
- (d) <sup>9</sup> όwə də xésəs tθe<sup>9</sup>. (JP 22)
  <sup>9</sup> ówə də xés-əs tθe<sup>9</sup>
  not EMPH sea.lion-3SUB that
  <sup>6</sup> That isn't a sea-lion!' (in spite of what you thought)
- (e) <sup>9</sup>óỷ ctwa<sup>9</sup> də wənémèn tá<sup>9</sup>əlt.
  <sup>9</sup>óỳ ctwa<sup>9</sup> də wəném-èn tá<sup>9</sup>əl-t good SPEC EMPH EST-go-I learn-TR
  <sup>6</sup>I guess it would be better if I went to find out!<sup>9</sup>
- (f) háy čx<sup>w</sup> ἀə. stop you EMPH 'Thank you.'
- (g) ná·w d;ə. hey EMPH 'Well, hello.' (an acknowledging greeting)

#### **16.2.15.** *θ ə t* 'adversative' (ADV)

This contrasts the statement or instruction with another existing condition and is translatable as 'however,' 'nevertheless,' 'but,' 'still,' 'even so,' 'instead,' 'regardless,' and so on.

(a) <sup>γ</sup>ówə cən mə <sup>γ</sup>i·n təw<sup>γ</sup>óỷ, <sup>γ</sup>i cən təwdádəỷ i wəném cən θəł sk<sup>w</sup>úl. ?áwa cən ṁә <sup>?</sup>i-ən təw-?áv ?icən somewhat-good I I not CERT AUX-I AUX təw-dádəy ?∍ỷ wə-ném cən skwúl θəł somewhat-sick and EST-go Ι ADV go.to.school 'I don't feel well; I'm a little sick, but I'm still going to school.'

(b) š[x<sup>w</sup>]te<sup>γ</sup>é·wən k<sup>w</sup>sni<sup>γ</sup>s <sup>γ</sup>í<sup>γ</sup>tət k<sup>w</sup>sni<sup>γ</sup>s <sup>γ</sup>í<sup>γ</sup>tən, i ni θəł qé<sup>γ</sup>əntəm Å stəlqáyə. (JP 7) š[x<sup>w</sup>]te<sup>9</sup>é·wən s-ni?-s ?í?tət k w k <sup>w</sup> think be.sleeping ART NOM-AUX-3POS ART ?í?łtən ?∂ỷ θəł qé?ən-t-əm s-ni?-s ni NOM-AUX-3POS be.eating and AUX ADV steal-TR-INTR [?ə] Χ stəlqáyə OBL wolves ART 'He thought that while he was sleeping he had eaten, but instead he had been robbed by the wolves.' (c) <sup>γ</sup>i čθəł wəłsźələźák stəm tə šx dí-lənəs. (JP 27) wəł-sźələźák -st-əm ?i ć2 θəł tə AUX OUOT ADV already-grabbed.with.fingers-CAUS-INTR ART šx vəlí·lənə-s cheeks-3POS 'But he already had their fingers digging into his cheeks.' (d) <sup>γ</sup>i yáyəd<sup>w</sup>əm k<sup>w</sup>θə John <sup>γ</sup>i <sup>γ</sup>i cən θəł t<sup>θ</sup>át<sup>θ</sup>əłəm təna <sup>γ</sup>é·nθə. (AG) °∂ỷ ?i yáỷədٍwəm k™θə John ?i cən θəł be.sweating ART John and I AUX AUX ADV ť<sup>0</sup>áť<sup>0</sup>əłəm təna ?é·nθə. cold this be I 'John was sweating, but I was cold myself.' (e) Xa cow skelogom i womosteyox<sup>w</sup> θoł. (JP 6) х́а ćə wə-skéləqəm ?∍ử wə-məstéyəx<sup>w</sup> θəł BE3P OUOT EST-dangerous.being and EST-person ADV 'He was said to be a dangerous being but still human.' (f) ni čθəł ném. (AG) ni ćə θəł ném AUX OUOT ADV go 'He did go, though, I am told.' (g) ném cən ce? θəł. (AG) Ι FUT ADV go 'I want to go anyway.' (h) ni ma θəł ném sqé<sup>9</sup>eq-s. (AG) ə k<sup>w</sup>θə younger.brother-3POS AUX CERT ADV go ROG ART 'But he did go to his younger brother.' (perhaps following 'He did not go to the funeral ...') (i) ni ctwa? θəł ném. (AG) SPEC AUX ADV go 'I imagine he did go, though.'

In instructions,  $\theta \partial l$  has an urging sense what may be conveyed by 'now, right away,' in contrast to whatever else one might be doing.

- (j) ném θəł.
   go ADV
   'Go ahead now!' (said to someone who has stopped)
- (k) <sup>γ</sup>źnəx<sup>w</sup> θəł.

```
stop ADV
```

'Now stop!' (said, in a stronger tone, to someone who has ignored earlier requests to<sup>2</sup>ónəx<sup>w</sup> 'stop')

- (l) <sup>2</sup> ∂mí θ∂ł <sup>2</sup> ∂łt ∂n.
   come ADV eat
   'Come and eat now!' (instead of standing there)
- (m)?ôłtən θəł.
   eat ADV
   'Go ahead and eat now!' (instead of sitting there)

#### 16.2.16. *k<sup>w</sup>t* 'by now, by then'(?)

This has been recorded in a few sentences following  $hi\theta$  'be a long time.'

- (a) híθ k<sup>w</sup>ł. (AG)
   last.long yet
   'What a long time it has been!'
- (b) hí $\theta$  k<sup>w</sup>ł yeł sis x<sup>w</sup>k<sup>w</sup>ón yə $\theta$ éləy. (JP) hí $\theta$  k<sup>w</sup>ł yeł s-ni-s x<sup>w</sup>k<sup>w</sup>ón yə $\theta$ éləy last.long yet after.which NOM-AUX-3POS lose one.person 'It was a long while before they lost one of their number.'

This predicate particle  $k^w l$  is distinct in function from the  $k^w l$  that can appear after  ${}^{2}\delta w {}^{2}$  or  ${}^{2}\delta w {}^{2}e^{2}$  or one' and be followed by a relative clause (§6.1.3, 6.3). That  $k^w l$  is probably an article (§15.1). However, in sentences with  ${}^{2}\delta w {}^{2}e^{2}k^w l$  ..., it is easy to infer a meaning 'yet, up to now,' as in (c).

(c) ?ówəte? k<sup>w</sup>ł x<sup>w</sup>ə?ínsamx. (AG)
?ówəte? k<sup>w</sup>ł x<sup>w</sup>ə-?í-nəs-amx
none ART become-be.here-GOAL-me
'Nobody has gotten to me [yet].' (lit. 'The one who has reached me here is none.')

Such an ascription of meaning may possibly have led to the use of the article as a predicate particle.

## 16.2.17. wa? 'presumptive' (PRES)

This seems to indicate that the statement is presumed by the speaker or someone else to be true, while also indicating doubt or surprise. Not many examples of usage have been recorded, however, and so this is one of the more elusive particles. It may be a component in the particle  $\dot{c}twa^2$  'speculative' (§16.2.12) and possibly in the adverbs  $2iw\dot{a}wa$  'maybe' (§18.4.34) and  $w\dot{a}\dot{c}a$  'probably' (§18.4.35), all of which also express some qualification about the truth of a statement.

In (a) and (b),  $wa^{2}$  seems to have an inferential sense.

- (a) híθ-əł cən k<sup>w</sup>ł wa? <sup>2</sup>í<sup>2</sup>tət. (JP 10)
   last.long-past I yet PRES be.sleeping
   'I must have been sleeping a long time.'
- (b) híθ-əł cən wa?. (AG) last.long-past I PRES
   'I must have been there a long time.'

It appears with the prefix  $t \partial w$ - 'somewhat,' the combination having the sense 'seem like,' as in (c) to (f).

- (c) təw-məŻél cən ce? wa?. (CC) like-faint I FUT PRES 'I feel like I'm going to faint.'
- (d) təw-?i wa? dádəy. (CC)
   like-AUX PRES sick
   'He looks sick.'
- (e) stém, stém yəx<sup>w</sup> tθé?. təwcí·tməx<sup>w</sup> wa?. (DK)
   stém stém yəx<sup>w</sup> tθé? təw-cí·tməx<sup>w</sup> wa?
   what what INF that like-great.horned.owl PRES
   'What can that be? It looks like an owl.'
- (f) təw-spé?eθ wa?. (AG)
  like-bear PREs
  'It appears to be a bear. Say, that looks like a bear.'

The combination  $k^{w} \partial w a^{2}$  has the sense 'one assumes, contrary to fact,' as in (g) and (h).

(g) ni<sup>9</sup>əł cən kwə wa<sup>9</sup> xwəy kwənəs<sup>9</sup>i həndəx. (AG) ni?-əł cən k <sup>w</sup>ə wa? x<sup>w</sup>áy k <sup>w</sup>ə nə-s-?i AUX-past I then PRES awaken ART my-NOM-AUX hốnp-əx be.eating-TR

'You'd think I would have awakened while I was eating.'

(h) ném kwə wa? Xxwət. (AG) ném kwə wa? Xəxw-ət go then PRES be.defeated-TR 'He thinks he'll go and beat him (but we don't believe it).'

In a question,  $k^{w} \partial w a^{2}$  can convey a feeling of outrage, as in (i).

(i) wét k<sup>w</sup>ə wa<sup>9</sup> wəx<sup>w</sup>páspəsk<sup>w</sup>əstəx<sup>w</sup> k<sup>w</sup>e·θyéyə θəna xəte<sup>9</sup>etθe<sup>2</sup>. (JP 15) wa? wə-x<sup>w</sup>-páspəsk<sup>w</sup>-əs-t-əx<sup>w</sup> k<sup>™</sup>ə wét k »ə who then that-inside-be.mocking-face-TR-vou PRES ART <sup>9</sup> əT-s-yéyə θəna **x**əfə ?a  $t\theta e^{2}$ vour-NOM-tightly.bound this be.doing OBL that 'Who the hell are you to ridicule the face of your friend, you doing that?!'

The combination  $k^w \partial w a^2$  also occurs with exclamatory words (see §20.2), as in (j).

(j) X̃íc k<sup>w</sup>∂ wa? t∂ X̃<sup>w</sup>íl∂m-s. (AG) how.long then PRES ART rope-3POS
 'What a long rope he has!'

## 16.2.18. <sup>9</sup>á<sup>9</sup>a 'emphatic interrogative' (ROG!)

This is used with interrogative words to emphasize the question. (These words do not require any particle and cannot take  $2\sigma$  'interrogative.')

| (a) | stém    | ?á?a       | ti <sup>9</sup> í.                         |   |
|-----|---------|------------|--------------------------------------------|---|
|     | what    | ROG!       | this                                       |   |
|     | 'What's | this?!' (c | f. stém ti <sup>?</sup> í. 'What's this?') | i |

- (b) stém ?á?a tə niəx<sup>w</sup> k<sup>w</sup>ənét. stém ?á?a tə ni-əx<sup>w</sup> k<sup>w</sup>əné-t what ROG! ART AUX-you hold-TR 'What's that thing you have?'
- (c) x \* cél čx \* ?á?a. go.where you ROG! 'Where are you going?'

Like  $2\partial$  'interrogative,' it is also used with  $y \partial x^w$  'inferential' to express curiosity, but with an interrogative word.

(d) stém yəx<sup>w</sup> <sup>9</sup>a<sup>9</sup>a tθe<sup>9</sup>.
 what INF ROG! that
 'I wonder what that is.'

- (e) cástax<sup>w</sup> sq<sup>w</sup>əmq<sup>w</sup>əméy-s k<sup>w</sup>θə vəx<sup>w</sup> čx<sup>w</sup> ?a?a k<sup>w</sup>θə do.what.with ROG! dogs-3POS INF you ART ART aál məstéyəx<sup>w</sup>. (JP 3) bad person
  - 'I wonder what's best to do (lit. "what you can do") about the dogs of that bad person.'

Following a vowel,  ${}^{2}\dot{a}{}^{2}a$  appears as  $\cdot \dot{l}\partial -$  for example, //ni  ${}^{2}\dot{a}{}^{2}a$ // becomes  $ni\cdot \dot{l}\partial$ , as in (f).

(f) ní·la ctámat łeň <sup>9</sup>ímaθ.

ní <sup>9</sup>a<sup>9</sup>a ctámət łə <sup>9</sup>ən-<sup>9</sup>íməθ AUX ROG! do.what.with.self ART you-grandchild 'What happened to your granddaughter?'

The particle  ${}^{2}a{}^{2}a$  is also used in sentences without interrogative words to form confirming questions, as in (g), (h), and (i).

(g) ni·la kwánatas. (AG)

ni <sup>?</sup>a<sup>?</sup>a k<sup>w</sup>ón-ət-əs AUX ROG! be.taken-TR-3TR 'Oh, so he did take it, did he?'

(h) 🕺 a·lɔ. (DK)

Åa <sup>?</sup>a<sup>?</sup>a BE3P ROG! 'Oh, so that's why, is it?'

(i) <sup>?</sup> əwé·la ni?as ném k<sup>w</sup>θa John.
 <sup>?</sup> əwa <sup>?</sup>a<sup>?</sup>a ni<sup>?</sup>-as ném k<sup>w</sup>θa John not ROG! AUX-3SUB go ART John 'Oh, didn't John go?'

The particle  $2\hat{a}a$  also appears with exclamatory words (§20.2.5), as in (j) to (m).

(j) líləq <sup>?</sup>à<sup>?</sup>a. (JP 7) easy ROG!

'It's tough!' (lit. something like 'Easy? Ha!' said by Seagull to Raven in a myth)

(k) x<sup>w</sup>əwém <sup>9</sup>a<sup>9</sup>a tə <sup>9</sup>ənléləmələp. (AG) x<sup>w</sup>əwém <sup>9</sup>a<sup>9</sup>a tə <sup>9</sup>ən-léləmələp how.big ROG! ART your-house-2PL 'What a big house you people have!'

- (l) sắćẳəl cən ?a?a. (AG) stuck I ROG! 'What a fast runner I am!'
- (m)<sup>?</sup>á<sup>.</sup>?à<sup>.</sup>, ni yəx<sup>w</sup> <sup>?</sup>a<sup>?</sup>a. (JP 7) aah AUX INF ROG! 'Oh, I guess that's so.' (Raven to Seagull)

#### **16.3. SENTENCE-FINAL TAGS**

These occur as final elements in sentences. Two have been recorded.

#### 16.3.1. ? *a* ~ ? *e* 'question tag'

This is added to a statement to seek confirmation.

| (a) cák <sup>w</sup> | k™θә     | Darrington        | <sup>9</sup> а. | cá∿k™ | <b>q๋</b> ә. (AG) |
|----------------------|----------|-------------------|-----------------|-------|-------------------|
| far                  | ART      | Darrington        | eh?             | far   | EMPH              |
| 'So Da               | rrington | is a long way, eh | ? Sure, j       | far!' |                   |

#### 16.3.2. yé 'taunting echo'

This expresses a thought supposedly held by another before his downfall.

- (a) swóyqe? con yé. (AG) man I eh?
  'I'm a man!' or perhaps 'I'm a man? Ha!' (said to someone after he has been beaten in a contest of strength)
- (b) šx<sup>w</sup>óx<sup>w</sup>a? cən yé. (AG) light I eh?

'So I'm a fast runner, eh?' (lit. 'I'm light, eh?' said to the once-proud loser of a footrace)

AG suggested that this may be identical to the response  $y \dot{e}$  used by listeners to a myth recital to encourage the narrator to continue, the common meaning being 'so you said.'

# **17** Interrogative Words

These are words that ask questions eliciting information. In this context, some correspond to the English interrogatives 'who,' 'what,' 'when,' 'where,' and so on, while others correspond to verbal phrases such as 'do what,' 'say what,' 'go where,' and so on. Interrogative words can also appear in embedded questions and in indefinites such as 'somebody,' 'anywhere,' and so on.

In simple questions, interrogative words function as predicate heads, for example,  $w\acute{e}t$  'who' in (a), in which the subject is a relative clause.

(a) wét kwə kwə <sup>9</sup>i Åé·nəq. (CC)
wét kwə kwə <sup>9</sup>i Åé·nəq
who then ART AUX be.potlatching
'Who is potlatching?' (lit. 'Who then is the one who is potlatching?')

An embedded question, such as 'who took it' in 'I know who took it,' is expressed by a subordinate clause with the interrogative word as its predicate head, as in (b).

 (b) łəġəlləx<sup>w</sup> cən <sup>2</sup>uwétəs k<sup>w</sup>θə ni k<sup>w</sup>ánət. (AG) łáġəl-ləx<sup>w</sup> cən wə-wét-əs k<sup>w</sup>θə ni<sup>2</sup> k<sup>w</sup>án-ət know-tr I that-who-3SUB ART AUX get.taken-tr <sup>4</sup> I know who took it.<sup>2</sup>

When interrogative words appear in adjuncts, they have an indefinite sense, as  $w\acute{e}t$  'who' in (c).

(c) wək "án ətəs k wét ... (CC 17)
 wə-k "án - ət-əs k v ə wét if-get.taken-TR-3SUB ART who 'if he takes anybody ...'

Usually they appear with other elements giving the sense 'any,' 'every,' and 'no.' Examples are given under the individual words below.

It seems that an interrogative word can also appear as a predicate head and have an indefinite sense if it bears a third-person subordinate subject suffix, such as  $st\acute{em}$  'what' in (d).

 (d) stéməs k<sup>w</sup>θə ni n k<sup>w</sup>ácnəx<sup>w</sup>. (CC) stém-əs k<sup>w</sup>θə ni?-ən k<sup>w</sup>éc-nəx<sup>w</sup> what-3SUB ART AUX-I look-TR
 'I saw something.'

It is not clear why there is a subordinate subject marker in what appears to be a main clause. Perhaps a sentence like (d) is an ellipsis of something like 'I don't know what it was I saw.'

As indicated in the section on particles (§16.2.2), interrogative words cannot be followed by the simple interrogative particle 2a, but they may be followed by  $2a^2a$  'emphatic interrogative' (§16.2.18). They are also often followed by the particle  $k^wa$  'then' (§16.2.13), often left untranslated.

The interrogative words are listed in Table 17.1 after their roots. They are described in the following sections in the order shown in the table. (Forms with lexical suffixes are omitted here but will be included in the later discussion.)

| Interrogative words                          |                                               |                                  |  |
|----------------------------------------------|-----------------------------------------------|----------------------------------|--|
| Root                                         | Word                                          | Gloss                            |  |
| wet 'who'                                    | 1. <i>wét</i>                                 | 'who?'                           |  |
|                                              | 2. $tx^w w \acute{e}t$                        | 'whose?'                         |  |
| tem 'what'                                   | 3. stém                                       | 'what?'                          |  |
|                                              | 4. témc                                       | 'what people?'                   |  |
|                                              | 5. təmtém                                     | 'when?'                          |  |
|                                              | 6. ctétəm                                     | 'be doing what?'                 |  |
| <i>c</i> ∂- 'do what'                        | 7. cástəx <sup>w</sup>                        | 'do what with it? about it?'     |  |
|                                              | 8. ctámət                                     | 'do what with oneself?'          |  |
|                                              | 9. x <sup>w</sup> cél                         | 'go where?'                      |  |
|                                              | 10. x <sup>w</sup> cálθət                     | 'go where?'                      |  |
|                                              | 11. x <sup>w</sup> cálstəx <sup>w</sup>       | 'put it where?'                  |  |
|                                              | 12. scék <sup>w</sup> əl                      | 'how? how much?'                 |  |
|                                              | 13. scək <sup>w</sup> əlím                    | 'how? how constructed/intended?' |  |
| <sup>9</sup> <i>i</i> • <i>nt</i> 'say what' | 14. $x^{w?i}$ ·nt                             | 'say what?'                      |  |
|                                              | 15. $sx^w \partial^2 i \cdot nt$              | 'what meaning?'                  |  |
|                                              | 16. $x^w \partial^2 i \cdot nst \partial x^w$ | 'say what to him?'               |  |
| <sup>9</sup> ánəcə 'where'                   | 17. <sup>9</sup> ánəcə                        | 'where?'                         |  |
|                                              | 18. <sup>9</sup> ánəcəstəx <sup>w</sup>       | 'put it where?'                  |  |
| nəcím 'why'                                  | 19. nəcím                                     | 'why?'                           |  |
| $k^{w}$ <i>in</i> 'how many'                 | 20. k <sup>w</sup> ín                         | 'how many?'                      |  |

Table 17.1

## 17.1. *wét* (CC, JP, AG), *təwét* (CC, DK) 'WHO?' *wé<sup>9</sup>əlt* 'WHO (PLURAL)?' (AG)

The form  $t \partial w \acute{e}t$ , which CC used occasionally and DK regularly, is the usual Upriver form (Galloway 1977, 345).<sup>1</sup> A plural was recorded from AG only. Examples of usage follow.

- (a) wét tθé?. (JP)
   who that
   'Who is that guy?'
- (b) wét k<sup>w</sup> tθe? swáy qe?. (CC) who then that male 'Who is that man?'
- (c) wét ce? k<sup>w</sup> → mi céw → dàm →. (JP)
  wét ce? k<sup>w</sup> → mi céw → t-Sam →
  who FUT ART AUX(come) help-TR-you
  'Who is going to help you?' (lit. 'The one who comes to help you will be who?')
- (d) wét kwała yańi. (CC)
   wét kwała yała yała yała
   who then that along-come
   'Who is coming? Who is it then that's coming?'

In the next three sentences, the subject is plural. CC and DK expressed plurality with demonstratives, AG with a plural 'who.'

| (e) | ) wét kʷə k̊ʷənํá·łtən yəmํí. (CC) |                  |                     |                  |  |  |  |
|-----|------------------------------------|------------------|---------------------|------------------|--|--|--|
|     | wét                                | k <sup>w</sup> ə | <b>k</b> ™ən≀á•ltən | yə-?əmí          |  |  |  |
|     | who                                | then             | those               | along-come       |  |  |  |
|     | 'Who i                             | s comin          | g? Who are tho      | se coming then?' |  |  |  |

- (f) təwét yəθéləỷ. (DK)
   who they
   'Who are they?'
- (g) <sup>3</sup>⁄<sub>λ</sub>a wé<sup>9</sup> lt k<sup>w</sup>θ <sup>9</sup>i yə-<sup>9</sup>í·max. (AG)
   BE3P who(PL) ART AUX along-be.walking
   'Who are those walking (along)?'

The interrogative *wét* 'who' is used in asking for personal names. (For names of things, *stém* 'what' is used.)

<sup>1</sup> That the *t*- of *twét* is a prefix rather than the article *t*<sub>2</sub> is clearly indicated by such Upriver forms as *wttwét*<sub>2</sub> s'who is it' and  $k^w$  *twét*<sub>2</sub> s'someone' (Galloway 1977, 349, 373).

 (h) wét k<sup>w</sup>e <sup>2</sup>eθk<sup>w</sup>íx. Jimmy. (JP) wét k<sup>w</sup>a <sup>2</sup>eT-sk<sup>w</sup>íx who ART your-name 'What is your name?' 'Jimmy.'

As illustrated earlier, wét as a nominal adjunct has the sense 'somebody.' With the limiting  $w_{\partial_2} \dots 2al$ , it forms  $k^w \partial w$  wét 2al 'anybody.'

(i) ni čx<sup>w</sup>  $\vec{k}$ <sup>w</sup> $\dot{\gamma}$ cn $\dot{\gamma}$ x<sup>w</sup>  $\vec{k}$ <sup>w</sup> $\dot{\gamma}$ w wét  $\gamma$ al. (CC) ni? čx <sup>w</sup> k<sup>w</sup>éc-nəx<sup>w</sup> k<sup>™</sup>ə ?aĺ [?ə] wa-wét ROG look-TR AUX vou ART EST-who just 'Did you see anybody?'

A phrase composed of  $m \delta \vec{k}^w$  'all' (with or without  $w \partial_{2} \dots 2 \vec{al}$ ) and wét has the sense 'everybody' or (with a negative) 'anybody.'

| (j) | ném čx              | <b>P</b> 3) |               |                  |                     |     |  |  |  |
|-----|---------------------|-------------|---------------|------------------|---------------------|-----|--|--|--|
|     | ném čx <sup>w</sup> |             | yэ́Ө-əs-t     | k <sup>™</sup> ə | wə-mák <sup>w</sup> | wét |  |  |  |
|     | go                  | you         | tell-recip-tr | ART              | EST-all             | who |  |  |  |
|     | 'Go tell everybody' |             |               |                  |                     |     |  |  |  |
|     |                     |             |               |                  |                     |     |  |  |  |

(k) ?ówə sźísəs kwəw mókw ?al wét kws néms ... (JP 3) ?áwa s-cŽí-s-əs k<sup>™</sup>∂ wə-mák<sup>w</sup> ?aĺ wét k »ə not NOM-want-3POS-3SUB ART EST-all who iust ART s-ném-s NOM-go-3POS 'He didn't want anybody to go ...'

Following  $2\delta w \partial t e^2$  'none,' which usually stands as a predicate, wét has the sense 'anybody.' Or, the sequence can be rendered 'nobody.' The wét may appear as a nominal adjunct, introduced by an article indicating absence and followed by a relative clause, as in (l) to (n).

(1) ?'swəte? c'ə k<sup>w</sup>ł wét ném. (JP 3) none QUOT ART who go
'Nobody ever went, they say.' (lit. 'Was absent reportedly any who that went.')

(m)?ówəte? kw wét łóqol-ləx kweə swéyol ... (JP 2)
 none ART who know-TR ART day
 'Nobody knew the day ...'

(n) ?ówəte? k<sup>w</sup>ł wét k<sup>w</sup>ócnəx<sup>w</sup>e<sup>n</sup>. (AG)
?ówəte? k<sup>w</sup>ł wét k<sup>w</sup>éc<sup>-</sup>nəx<sup>w</sup>-e<sup>n</sup>
none ART who look-TR-I
'I haven't seen anybody.' (lit. 'There no who whom I see.')

Or *wét* may be preceded by no article, in which case it is presumably a part of the predicate, as in (o) and (p).

- (o) ?ówəte? cə wét ném. (DK)
   ?ówəte? ce? wét ném
   none FUT who go
   'Nobody will go.'
- (p) təwét. <sup>9</sup>ówəte<sup>9</sup> wét. (DK) who none who 'Who is it?' 'Nobody.'

It should be added that  $2\delta w \partial t e^2$  alone can have the sense 'nobody,' more literally 'there is none who,' as in (q).

(q) ?ówəte? k<sup>w</sup>ł k<sup>w</sup>əcnəx<sup>w</sup>. (JP 28)
 ?ówəte? k<sup>w</sup>ł k<sup>w</sup>ec-nəx<sup>w</sup>
 none ART look-TR
 'Nobody sees them.'

See also <sup>?</sup>ówəte<sup>?</sup> (§6.3).

## 17.2. tx<sup>w</sup>wét (AG), təwét (JP) 'WHOSE?'

The form  $tx^w w\acute{et}$  is clearly  $w\acute{et}$  'who' with the prefix  $tx^{w}$ -2 'belonging to' (see §12.4.4), which occurs otherwise only with personal names. The ta- of  $taw\acute{et}$  is not identifiable. (I recorded  $taw\acute{et}$  from JP only a few times, and unfortunately I cannot find any form for 'whose' in material from CC. I would be inclined to suppose that I might have misheard JP's form except that Hill-Tout (1902, 420) gives the Kwantlen forms as wet 'who' and too-wet 'whose.')

(a) tx<sup>w</sup>-wét ti?í. (AG) belonging.to-who 'Whose is this?'

## 17.3. stém 'WHAT?' stéləm 'WHAT (PLURAL)?'

This is evidently composed of a root *tem* 'what' and *s*- 'nominalizer.' The plural has been recorded from AG only. Example of usage are (a) to (f).

- (a) stém <sup>?</sup>a<sup>?</sup>a. (CC) what <sup>ROG!</sup> 'What is it?'
- (b) stém ce? k<sup>w</sup>e. (CC) what FUT then 'What will it be (then)?'

- (c) stém k<sup>w</sup>θé?. (CC) k "ə what then that 'What's that?' (d) stém <sup>9</sup>a<sup>9</sup>a tə niəx<sup>w</sup> k<sup>w</sup>ənét. (JP) ?a?a ni?-əx<sup>w</sup> k »əné-t stém tə what ROG hold-TR AUX-you ART 'What is that thing you have?' (e) stém k<sup>w</sup><sup>2</sup> sk<sup>w</sup>íxs ti<sup>9</sup>í. (CC)
- stém k<sup>w</sup>ə sk<sup>w</sup>íx-s tə?í what ART name-3POS this 'What do you call this? What is the name of this?'
- (f) stéləm tə ni yəxəlxélqəm. (AG) stéləm tə ni? yə-xəlxélqəm what(PL) ART AUX along-be.briefly.appearing(PL)
  'What are those things we are glimpsing?' (lit. 'What [pl.] are those that are briefly appearing?')

The word *stém* takes the variant  $x^{wi}$ - of the prefix generally appearing as  $x^{w}$ -'become,' as in (g).

(g) héỷ x<sup>w</sup>i-stém. (CC) go.ahead become-what 'What's next?' (lit. 'Go ahead become what?')

The notion 'what kind of a ... ?' is expressed in two ways. One is with a conditional clause (see §4.2.1) introduced by  $2\partial t$ -generally 'when' or 'whenever,' as in (h).

(h) stém <sup>2</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - sq<sup>w</sup> <sup>3</sup> - s

The other is simply with the article  $k^{w}\partial$ , which indicates that something is unknown, as in (i).

(i) stém k<sup>w</sup> ləplá·š ti<sup>9</sup>í. (DK)
 stém k<sup>w</sup>ə ləpláš tə<sup>9</sup>í
 what ART board this
 'What kind of board is this?'

To serve in an embedded question,  $st\acute{em}$  appears as the predicate of a subordinate clause, as in (j), (k), and (l). (i)  $\vec{k}^{w}$ ácstálx<sup>w</sup> čx<sup>w</sup> wəstéməs  $\vec{k}^{w}$ ə <sup>?</sup>ə $\theta$ x<sup>w</sup>?ávələp. (JP 1) k<sup>™</sup>éc-st-alx<sup>™</sup> čxw wa-stém-as k<sup>w</sup>⊃ look-CAUS-us that-what-3SUB vou ART ?əT-šx<sup>w</sup>-?áý-ələp your-OBLNOM-good-your(PL) 'Show us what you're good for.' (lit. 'Make us see what is the reason for your being good.') (k) x̃<sup>w</sup>óm k<sup>w</sup>stólləx<sup>w</sup>s wəstéməs k<sup>w</sup>θə swé<sup>2</sup>s s<sup>2</sup>ólyə k<sup>w</sup>θe<sup>2</sup> məstóyəx<sup>w</sup>. (JP 28) хॅ<sup>∞</sup>э́т k <sup>w</sup>ə s-tál-lax w-s wa-stém-as k<sup>w</sup>θa

fast ART NOM-be.understood-TR-3POS that-what-3SUB ART swé?-s s?ályə k<sup>w</sup>θe? məstáyəx<sup>w</sup> own-3POS vision that person 'They can immediately tell what the vision of that person is.'

 (l) yöθəsθ <sup>2</sup>ustéməs k<sup>w</sup>ə ni dá·yt. (AG) yöθ-əs-t-S wə-stém-əs k<sup>w</sup>ə ni<sup>2</sup> dáy-ət tell-RECIP-TR-me that-what-3SUB ART AUX die-TR 'Tell me what killed him.'

In (m) (and for a similar one, see §20.3.7 [a]), there is no main verb indicating an embedded question, but one seems to be implied.

(m)?ômət łə wəstéməs k<sup>w</sup>e ?eθλí?. (JP 3)
?ômət łə wə-stém-əs k<sup>w</sup>ə ?əT-s-cλí?
sit PER that-what-3SUB ART your-NOM-want
'Sit down [and tell me] what it is that you want.'

Indefinites with *stém* are analogous to those formed with *wét* 'who,' as in (n), (o), and (p).

- (n) ha mí k<sup>w</sup>θə stém ... (JP 9) ha<sup>9</sup> <sup>9</sup>əmí k<sup>w</sup>θə stém if come ART what 'If anything comes ...'
- (o) ni čx<sup>w</sup> k<sup>w</sup>ýcnyx<sup>w</sup> k<sup>w</sup>yw stém <sup>9</sup>al. (CC) ni? čx<sup>w</sup> k<sup>w</sup>éc-nəx<sup>w</sup> k<sup>w</sup>∍ ?aĺ [?ə] wə-stém look-tr AUX ROG vou ART EST-what just 'Did you see anything?'
- (p) stém k<sup>w</sup>əs<sup>2</sup>x<sup>2</sup>i<sup>2</sup>, stém <sup>2</sup>a<sup>1</sup> (DK) stém k<sup>w</sup>ə <sup>2</sup>ə-s-c<sup>2</sup>x<sup>2</sup>i<sup>2</sup> stém <sup>2</sup>a<sup>1</sup> what ART your-NOM-want what just 'What do you want [e.g., of food laid out]? Anything.'
As a modifier,  $(w \partial -)st \acute{em} ?a \vec{l}$  has the sense 'any kind of' or 'every kind of,' as in (q).

(q) k<sup>w</sup>θə stém <sup>9</sup>al sm ə́yəθ (JP 1)
 ART what just deer
 'any kind of [game] animal [lit. deer]'

The phrase  $(w \partial -)m \partial \vec{k}^w st \, \epsilon m$  is 'everything' or, as a modifier, 'every kind of,' as in (r) and (s).

(r) qáx ca kwow mákw stém, smáyao, spé?eo, i kwaw stém ?al s?áłtən ... (JP 3) ćə k™θə wə-mák<sup>w</sup> stém. aźž smáyəθ, deer EST-all what plentiful OUOT ART 2a1 кw spé<sup>γ</sup>eθ. ?∍ỷ wə-stém s?áłtan bear and ART EST-what iust food

'They say there was everything - deer, bear, and all kinds of food ...'

(s) təw mɨkw stém smáyəθ (JP 3)
 tə wə-mɨkw stém smáyəθ
 ART EST-all what deer
 'deer or anything, any kind of game'

The phrase  $2\delta w \partial t e^2 st \dot{e}m$  means 'nothing,' as in (t) to (x).

- (t) i wəł?ówəte? stém (JP 2)
   ?əỷ wəł-?ówəte? stém and already-none what 'but there was already nothing there'
- (u) tx<sup>w</sup>-?ówəte? cə stém ... (JP 7) remaining-none QUOT none 'There was nothing left, it is said ...'
- (v) ?\u00f3w\u00e3te? k<sup>w</sup> st\u00e3m ni k<sup>w</sup>\u00e3cn\u00e3x \u00e3v\u00e3 ?\u00e3cH\u00e4tan. (JP 27)
  ?\u00e3w\u00e3te? k<sup>w</sup> st\u00e3m ni? k<sup>w</sup>\u00e3c-n\u00e3x \u00e3v-\u00e3s ?\u00e3cH\u00e4tan
  none ART what AUX see-TR-3SUB 3PL
  'They didn't see anything.' (lit. 'None was the something that they saw.')
- (w) ?ówəte? kw stém s-mé?kwł-s ... (JP 25)
  none ART what NOM-get.hurt-3POS
  'There wasn't any injury ...' (lit. 'None [was] his any hurt.')
- (x) stém k wə k wəné-t-əx w. ?ówəte? stém. (DK)
   what ART hold-TR-you(SUB) none what
   'What have you got?' 'Nothing.'

The interrogative *stém* may be the basis for a word *stámone* ~ *stámone* ~ *thing*, thingy' (something you can't put a name to), as in (y), (z), and (aa).

- <sup>γ</sup>əwé· čx<sup>w</sup> niəx<sup>w</sup> k<sup>w</sup>ə́cnəx<sup>w</sup> k<sup>w</sup>θə nəstámənə. (DK) (y) ?awa ?a čx<sup>w</sup> ni-əx " k<sup>™</sup>ác-nax™ k»θə nə-stámənə. not ROG vou AUX-vou see-TR ART my-thingy 'Have you seen my thingy around?' mi łdwa to nioxw kwonét stámone. (JP) (z) wa? k<sup>w</sup>əné-t mi łə ģθ tə ni-əx<sup>w</sup> come PRES[?] hold-TR PER EMPH ART AUX-you stáməne thing 'Bring me that thing you've got.'
- (aa) tén łe tə 'əθstáməne. (JP)
  tén łe tə 'əT s-stáməne
  disappear PER ART your NOM-thing
  'Hide your thing.' (lit. 'Disappear your thing,' said by mother to little boy with his fly undone)

# 17.4. témc (PL. téləmc) 'WHAT PEOPLE'

This is composed of the root *tém* 'what' and a suffix -c not otherwise attested.

- (a) témc čx<sup>w</sup>. (CC, JP, AG) what.people you
  'Where are you from? What reserve are you from? What tribe are you from? What nationality are you?'
- (b) témc yəx<sup>w</sup> <sup>2</sup>a<sup>2</sup>a tθé<sup>2</sup>. (JP) what.people INF ROG! that 'I wonder what nationality he is!'
- (c) téləmc təňá·łtən. (AG) what.people(PL) they 'Where are they from?'
- (d) k<sup>w</sup> əw m ák<sup>w</sup> ?al témc (JP 6)
   k<sup>w</sup> wə-m ák<sup>w</sup> ?al témc
   ART EST-all just what.people
   'any tribe'

# 17.5. təmtém 'WHEN?'

This comes from  $t \partial m$ - 'time of' and  $t \acute{e}m$  'what.' To ask a question, it can appear as a main-clause predicate with the questioned action expressed in a coordinate clause linked by  $\partial a \dot{y} \sim i$  'and,' as in (a) and (b).

(a) təmtém ce? i ném čx<sup>w</sup>. (DK) təmtém ce? ?əỷ ném čx<sup>w</sup> when FUT and go you 'When are you going?' (lit. 'When will it be and you go?')
(b) təmtém i ni čx<sup>w</sup> tás. (CC)

| təmtém    | ?∍ỷ     | ni?     | čxw | tás          |
|-----------|---------|---------|-----|--------------|
| when      | and     | AUX     | you | arrive.there |
| 'When did | you get | there?' | -   |              |

This construction can be used for questions about both future and past. Alternatively, to ask a question about the future, *təmtém* can be followed by a subordinate clause expressing the questioned action, as in (c) and (d).

| (c) | təmtém    | ce?                 | k <sup>w</sup> ə | wə-ném-è∙n. (C0                     | C)          |
|-----|-----------|---------------------|------------------|-------------------------------------|-------------|
|     | when      | FUT                 | then             | if/that-go-I                        |             |
|     | 'When sha | ll I go?'           | (lit. 'Wl        | nen will it be then the             | nat I go?') |
| (d) | təmtém c  | e? k <sup>w</sup> ə | wəniəx           | w hóye?. (CC)                       |             |
|     | təmtém    | ce?                 | k <sup>w</sup> ə | wə-ni <sup>9</sup> -əx <sup>w</sup> | háye?       |
|     | when      | FUT                 | then             | if/that-AUX-you                     | leave       |
|     | 'When are | you lea             | ving?'           |                                     |             |

Or, to ask a question about the past,  $t \ge mt \le m$  can alternatively be linked by  $\dot{X}a$  'be third person' to a nominalization expressing the questioned action, as in (e) and (f).

| (e) | х́а   | k <sup>w</sup> | təmtém         | <sup>9</sup> ə-s- <sup>9</sup> i | k <sup>w</sup> ác-n-àmx. (CC) |
|-----|-------|----------------|----------------|----------------------------------|-------------------------------|
|     | BE3P  | ART            | when           | your-NOM-AUX                     | see-TR-me                     |
|     | 'When | did you        | see me?' (lit. | 'Your having seen                | me is when?')                 |

| (f) | х́а   | k <sup>w</sup> | təmtém     | <sup>9</sup> ə-s- <sup>9</sup> i | técəl. (CC) |
|-----|-------|----------------|------------|----------------------------------|-------------|
|     | BE3P  | ART            | when       | your-NOM-AUX                     | arrive.here |
|     | 'When | did you        | get here?' |                                  |             |

In a subordinate construction, *təmtém* can serve as the 'when' of an embedded question, as in (g).

| (g) | ?áwəté?   | k™ łóq́əli            | ləx <sup>w</sup> v | vətəmté            | məs ce <sup>9</sup> i Åəw mi | técəl. (J | P 2) |
|-----|-----------|-----------------------|--------------------|--------------------|------------------------------|-----------|------|
|     | ?áwəté?   | <b>k</b> <sup>w</sup> | łáģə               | l-ləx <sup>w</sup> | wə-təmtém-əs                 | ce?       | °∍ỷ  |
|     | none      | ART                   | knov               | v-TR               | if/that-when-3SUB            | FUT       | and  |
|     | х́е       | wə-?a                 | mi                 | técəl              |                              |           |      |
|     | again     | EST-CC                | me                 | arrive.h           | nere                         |           |      |
|     | 'No one k | new whe               | n she              | would co           | ome again.'                  |           |      |

Or it can have an indefinite meaning, as in (h).

(h) ... i ném cən ce? θəł wənśwnəct wətəmtéməs ?al. (JP) ?∍ỷ ném cən ce? θəł wə-náŵnəc-t and go I FUT ADV EST-pav-TR ?aĺ wə-təmtém-əs if-when-3SUB iust '... but I'll still go and pay him sometime.'

The combination of a negative and tamtém means 'never,' as in (i).

(i) sqíqas ce·p k<sup>w</sup>sté<sup>9</sup>etθe<sup>9</sup>. <sup>9</sup>ówa tamtémas i qalét ce·p yáx<sup>w</sup>aθat. (JP)  $t\theta e^{\gamma}$ sģíģəs sté?e ?a ?áwa ce·p k <sup>w</sup>ə like knot vou(PL) ART OBL that not tamtém-as ?∍ỷ aəlét ce·p váx<sup>w</sup>-əθət when-3SUB and again you(PL) be.united-self 'You are knotted together. Never undo yourselves again.' (said to a couple at their wedding)

In (j), the *təmtém* is nominalized.

| (j) | ?áw∍ k  | x <sup>w</sup> s təm | téms i <sup>9</sup> ax¤əstəs | (JP 1) |                                       |
|-----|---------|----------------------|------------------------------|--------|---------------------------------------|
|     | ?áwə    | k w                  | s-təmtém-s                   | °∂ỷ    | <sup>?</sup> ex <sup>w</sup> -əs-t-əs |
|     | not     | ART                  | NOM-when-3POS                | and    | give-recip-tr-3tr                     |
|     | 'They 1 | never ga             | ve them'                     |        |                                       |

More commonly, however, 'never' is ?ówo scék volos (see §17.12 below).

## 17.6. ctétəm 'BE DOING WHAT?'

This is probably composed of the verbalizing prefix c- 'get, do, make' and *tém* 'what' reduplicated for the progressive aspect. (A perfective \*ctém has not been recorded.)

- (a) ctétəm k wə. (CC) be.doing.what then 'What is he doing?'
- (b) ctétəm čx<sup>w</sup>. (CC) be.doing.what you 'What are you doing?'
- (c) <sup>9</sup>i cən k<sup>w</sup>ə ctétəm. (CC) AUX(here) I then be.doing.what 'What am I doing?' (as in a game – 'guess what?')

# 17.7. cástax" 'DO WHAT WITH IT? DO WHAT ABOUT IT?'

This is evidently composed of a root  $c\delta$ - (reduplicated  $c\delta k^w \partial$ - for the progressive aspect; see §1.5.10) 'do what?' and the suffix  $-st \partial x^w_2$  'comitative' (see §10.4.1).

- (a) côstax<sup>w</sup> čx<sup>w</sup> k<sup>w</sup>e. (CC)
   do.what.with.it you then
   'What are you going to do with it?'
- (b) ?i čx<sup>w</sup> k<sup>w</sup>e cék<sup>w</sup>əstəx<sup>w</sup>. (CC) AUX you then be.doing.what.with.it 'What are you doing with it?'
- (c) cóstəx <sup>w</sup> yəx <sup>w</sup> čx <sup>w</sup> <sup>9</sup>a<sup>9</sup>a k<sup>w</sup>θə sq<sup>w</sup>əmq<sup>w</sup>əméys k<sup>w</sup>θə gól məstáyəx<sup>w</sup>. (JP 3) cástax » čx<sup>w</sup> ?a?a k <sup>w</sup>θə vəx<sup>w</sup> do what about them ROG INF vou ART sq<sup>w</sup>əmq<sup>w</sup>əméy-s k<sup>w</sup>θə aál məstéyəxw dogs-3POS ART bad person

'I wonder what you/we can do about the dogs of that bad person.'

(d) cék<sup>w</sup>stəx<sup>w</sup> čx<sup>w</sup> ce<sup>γ</sup> ti k<sup>w</sup>ənétəx<sup>w</sup> wənéməx<sup>w</sup> tás k<sup>w</sup>e·θyəx<sup>w</sup>ánəm. (JP) cék «stəx » čx<sup>w</sup> ce? k "əné-t-əx" wə-ném-əx<sup>w</sup> ti do.what vou FUT this hold-TR-vou when-vou tás <sup>9</sup>əT-s-yə-x<sup>w</sup>-hə́nəḿ [?ə] k "ə arive.there OBL ART vour-NOM-along-move.toward-be.going 'What are you going to do with that you're taking along when you get to where you're going?'

#### 17.8. ctámət 'DO WHAT WITH ONESELF?'

This word is usually translated 'do what?' or 'have what happen to one?' but the progressive,  $c \partial k^w st \acute{am} \partial t$ , suggests that the perfective has the underlying form //c-stámət//, and so this may be simply the reflexive of  $c \acute{a} st \partial x^w$  'do what with/about it?' – being composed of the root  $c \partial$ - 'do what' and a suffix -stám $\partial t$ 'with oneself, about oneself' (cf. -stanám $\partial t$  ~ -stán $\partial m \partial t$ , §10.5.3), meaning 'do what with oneself?'

- (a) ctámət čx<sup>w</sup>. (JP) do.what.with.oneself you 'What do you want to do?'
- (b) cək<sup>w</sup>stámət čx<sup>w</sup>. (JP) be.doing.what.with.oneself you 'What are you doing?'
- (c) i čx<sup>w</sup> cək<sup>w</sup>stámət. (CC)

?ičx wcək wstámətAUXyoube.doing.what.with.oneself'What are you up to?'

- (d) ctámət cən ce? kwe. (CC) do.what.with.oneself I FUT then 'What shall I do? What am I going to do now?'
- (e) ?i čx<sup>w</sup> ctámət. (CC)
  AUX you do.what.with.oneself
  'What's the matter with you?' (lit. 'What have you done with yourself?')
- (f) ni·lə ctámət łeň ?íməθ. (CC)
   ni? ?a?a(?) ctámət łə ?ən-?íməθ
   AUX(there) ROG! do.what.with.oneself ART your-grandchild
   'What's the matter with your granddaughter?'

In a subordinate clause, *ctámət* can serve as the predicate of an embedded question.

(g) nəwł ł<br/>ádjəlləx "<br/>əs i k̈"i šx "q"éləwəns wəni<br/>·s ctámət k" $\theta$ ə

| siyé  | ýəsəł. (JP 10) |                    |      |          |                  |      |
|-------|----------------|--------------------|------|----------|------------------|------|
| ni?   | wəł-łádal-la   | x <sup>w</sup> -əs | °i   | [?ə]     | κ <sup>w</sup> i |      |
| AUX   | already-know-  | -TR-3TR            | here | OBL      | DEM              |      |
| šx™c  | q™éləwən-s     | wə-ni?             | '-əs | ctámət   |                  | k™θə |
| mine  | 1-3POS         | if-AUX-3           | 3SUB | do.what. | with.oneself     | ART  |
| siyé  | ýə-s-əł        |                    |      |          |                  |      |
| frien | ds-3POS-past   |                    |      |          |                  |      |
|       |                |                    |      |          |                  |      |

'He knew in his mind what had happened to his [deceased] friends.'

In (h), stámət has the indefinite sense of 'do anything.'

```
(h) skwéy kws ctáməts. (JP 1)
```

sk<sup>w</sup>éy k<sup>w</sup> s-ctámət-s impossible ART NOM-do.what.with.oneself-3POS 'They couldn't do anything.' (lit. 'Their doing what with themselves was impossible.')

# 17.9. x<sup>w</sup>cél 'GO WHERE?'

This is probably composed of  $x^{w}$ - 'motion toward,' the root  $c \rightarrow$  'do what,' and  $-\partial l$  'move toward.' AG gave the progressive  $x^{w}c \acute{e}k^{w}\partial l$ .

- (a) x<sup>w</sup>cél čx<sup>w</sup>. <sup>9</sup>ówəte<sup>9</sup> nə-šx<sup>w</sup>-ném. (AG)
  go.where you none my-OBLNOM-go
  'Where are you going?' 'Nowhere.' (lit. 'None is my destination.')
- (b) x<sup>w</sup>cél čx<sup>w</sup> ce? k<sup>w</sup>ə. (CC) go.where you FUT then 'Where will you go (then)?'

(c) ni yəx "cék "əl. (AG)

ni? yə-x<sup>w</sup>cék<sup>w</sup>əl

AUX along-be.going.where

'Which way (on what road) is he going?' (lit. 'Where is he going along?')

#### 17.10. *x<sup>w</sup>cδl*θ*δt* 'GO WHERE?'

This is composed of  $x^w c \ell l$  'go where' and  $-\theta \partial t$  'oneself.' The progressive is  $x^w c \delta k^w \partial l \partial \partial t$ . AG gave a perfective plural,  $x^w c \partial c \ell l \partial \partial t$ , and a progressive plural,  $x^w c \partial k^w c \delta k^w \partial \partial t$ .

- (a) x<sup>w</sup>cólθət čx<sup>w</sup>. (JP)
   go.where you
   'Where are you going?'
- (b) yə-x<sup>w</sup>cák<sup>w</sup>əlθət čx<sup>w</sup>. (JP) along-be.going.where you
   'Where are you going?' (said to someone seen walking and stopping)

Like other interrogative words,  $x^w c \delta l \theta \delta t$  can also have an indefinite sense, as in (c).

 (c) mák wał nasném x wcálθat <sup>9</sup>iw k wácnax w can. (JP) mák wał na-s-ném x wcálθat <sup>9</sup>aý wa-k wác-nax w can all-past my-NOM-go go.where and EST-see-TR I 'Whenever I went somewhere, I saw him.'

#### 17.11. x<sup>w</sup>célstəx<sup>w</sup> 'PUT WHERE?'

This is composed of  $x^{w}c\ell i$  'go where' and  $-stax^{w_{i}}$  'causative.' The only instance I have of this word is the following reported exchange between a husband and wife:

| (a) | ni ə č | éx™ łáďa               | olləx <sup>w</sup> w  | əni∙n | x <sup>w</sup> célst | əx <sup>w</sup> k <sup>w</sup> θ | ə nəq <sup>w</sup> ło | éyxən. <sup>9</sup> i cən           |
|-----|--------|------------------------|-----------------------|-------|----------------------|----------------------------------|-----------------------|-------------------------------------|
|     | ċəń    | né <sup>9</sup> t. (JP | )                     |       |                      |                                  |                       |                                     |
|     | ni?    | °ə                     | čx <sup>w</sup>       | łádal | l-ləx <sup>w</sup>   | wə-ni                            | ?-ən                  | x <sup>w</sup> célstəx <sup>w</sup> |
|     | AUX    | ROG                    | you                   | know  | /-TR                 | if/that-                         | -AUX-I                | put.where                           |
|     | k»θ    | ə na                   | ə-q <sup>w</sup> łéyx | ən.   | ?i                   | cən                              | ċəṁé                  | ?-t                                 |
|     | ART    | m                      | y-shoe                |       | AUX                  | Ι                                | be.ho                 | lding.in.mouth-TR                   |
| 4   | Do you | ı know v               | where I p             | ut my | shoes?'              | 'I'm hole                        | ding ther             | n in my mouth.'                     |

# 17.12. *scék\*əl* 'HOW? HOW MUCH? WHO?'

This is probably the resultative form of  $x^w c \ell l$  'go where,' with a literal meaning something like 'where gone.'

- (a) <sup>9</sup>i scék<sup>w</sup>əl k<sup>w</sup>θən mén. (AG)
  <sup>9</sup>i scék<sup>w</sup>əl k<sup>w</sup>θə <sup>9</sup>ən-mén
  AUX how ART your-father
  <sup>6</sup>How is your father?'
  (b) ni k<sup>w</sup>e scék<sup>w</sup>əl (CC)
- ni? k<sup>w</sup>e scék<sup>w</sup>əl AUX then how 'How was it?'
- (c) wə-scék<sup>w</sup>əl<sup>?</sup>al<sup>\*</sup> tə swéyəl. (JP) EST-how just ART day 'How is the weather?'
- (d) yə-scék wəl čx w. (CC) along-how you
  'How (by what means) are you going?'
- (e) yə-scék wəl cən ce? kwə nə-s-ném. (CC) along-how I FUT ART my-NOM-go
   'How (by what means) shall I go?'
- (f) scék »əl k »s cák »s tθe?. (JP)
  scék »əl k » s-cák »-s tθe?
  how ART NOM-far-3POS that
  'How far is that?' (lit. 'How much is that's being far?')
- (g) scék wəl k wə k wə ni?-əx k łéd-ət. (CC)
  how.much then ART AUX-you lay.down-TR
  'How much did you put down [as down payment]?' (lit. 'How much then is what you laid down?')

In certain, perhaps very limited contexts, *scék<sup>w</sup>al* can mean 'who' or 'what.'

- (h) Xa k<sup>w</sup>∂ scék<sup>w</sup>∂l. (CC)
   BE3P then who
   'Who is he?'
- (i) Xa k<sup>w</sup> scék<sup>w</sup> sl tθe?. (AG)
  BE3P then who that
  'Who is that?' ("Seems to be the same as wét tθe?," according to AG)

The interrogative meaning is absent in (j).

 (j) môk<sup>w</sup> scék<sup>w</sup>ol (JP) all how.much 'everything' The negative  $2\delta w \partial sc \delta k^w \partial l \partial s$  is the usual form for 'never' (but see §17.5,  $t \partial m t \delta m$  above). A literal interpretation might be 'it has not gone anywhere' or 'it has not reached any amount.' The form usually appears as the first of a pair of coordinate clauses. For examples, see §18.4.7.

In a subordinate clause,  $sc\acute{e}k^w \partial l$  can serve as the predicate of an embedded question.

(k) ha?  $^{2}$   $^{2}$   $\theta$   $^{3}$   $^{2}$   $^{2}$   $\theta$   $^{3}$   $^{2}$   $^{2}$   $\theta$   $^{3}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^{2}$   $^$ ?aT-s-c-X1? ?aT-s-k<sup>w</sup>ác-nax<sup>w</sup> ha? k <sup>w</sup>ə if your-NOM-make-value ART vour-NOM-see-TR wə-scék »əl-əs k w s-θí-s that-how-3SUB NOM-big-3POS ART 'If you want to see how big it is ...'

The word  $sc\acute{e}k^{w}\partial l$  also appears, with the stressed vowel appearing as schwa, as a stem with two lexical suffixes,  $-\acute{e}l\partial q\partial p$  'smell, taste' and  $-\acute{a}m\partial x$  'appear.'

```
(l) scək »əléləqəp. (JP)'How does it smell?'
```

| (m) scək wəláməx. 1                                             | ni ə ck <sup>"</sup> | vím əy r | ni ə t <sup>ə</sup> et <sup>ə</sup> əx | »əm. (JP) | )   |          |
|-----------------------------------------------------------------|----------------------|----------|----------------------------------------|-----------|-----|----------|
| scək <sup>w</sup> əláməx                                        | ni?                  | ?ə       | ck <sup>w</sup> ím                     | ?әу       | ni? | $^{9}$ ə |
| how.appear<br>t <sup>e</sup> et <sup>e</sup> əx <sup>w</sup> əm | AUX                  | ROG      | red                                    | or        | AUX | ROG      |
| blue                                                            |                      |          |                                        |           |     |          |

'What colour is it? (lit. "How does it look?") Is it red or is it blue?'

#### 17.13. scək"əlím 'HOW? HOW CONSTRUCTED? HOW INTENDED?'

This is probably composed of  $sc\acute{e}k * \partial l$  'how' and  $-\partial m$  'intransitive,' with the stress shift to the last syllable because of its being a resultative form. (Cf.  $c\partial l\acute{t}m$  'even,' §18.4.39?) (CC and AG gave the word as  $sc\partial k * \partial l\acute{t}m$ , JP as  $c\partial k * \partial l\acute{t}m$ .)

- (a) ni kwe scəkwəlím. (CC)
   ni? kwe scəkwəlím
   AUX then how
   'How was it?'
- (b) cək wəlím ce? kwe θyá ý as xwte ək wθe θyé ya ni nem xwəsqíqəd. (JP) cək <sup>w</sup>əlím ce? Å<sup>w</sup>∂ <sup>9</sup>əT-s-yá·ýəs x<sup>w</sup>te? ?a how intended FUT ART vour-NOM-be.working toward OBL <sup>9</sup>əT-syéỷə k »θə ni? nem x<sup>w</sup>ə-sqíqəq your-friend become-bound ART AUX go 'What are you going to do about your friend who was taken to jail?'

 (c) scək »əlím k wθə s-q slq vəl. (AG) how.intended ART NOM-tell
 'How do you make out that message?' (lit. 'How intended is the telling?')

#### 17.14. x<sup>w</sup>ə<sup>9</sup>í·nt 'SAY WHAT?'

The progressive is  $x^w \partial^2 i^2 \partial nt$ , the plural perfective  $x^w \partial nt (AG)$ , and the plural progressive  $x^w \partial nt^2 \partial nt$  (AG). These forms suggest that the word is composed of a prefix  $x^w \partial$ - and a root with the underlying form //?inət//.

- (a) ni x<sup>w</sup>ə<sup>9</sup>í·nt. (AG) ni<sup>9</sup> x<sup>w</sup>ə<sup>9</sup>í·nt AUX say.what 'What did he say?'
- (b) ?i x<sup>w</sup>ə?í?ənt. (AG) AUX be.saying.what 'What is he saying?'
- (c) x<sup>w</sup>∂<sup>2</sup>í<sup>2</sup>∂nt k<sup>w</sup>∂ t∂ n∂wÅa. (CC) be saying what then ART that.one
   'What is he [a person not known] talking about?'
- (d) x<sup>w</sup>ə<sup>2</sup>í<sup>2</sup>ənt k<sup>w</sup>e k<sup>w</sup>ə<sup>2</sup>əswe<sup>2</sup> <sup>2</sup>əšx<sup>w</sup>q<sup>w</sup>éləwən. (CC) x<sup>w</sup>ə<sup>2</sup>í<sup>2</sup>ənt k<sup>w</sup>e k<sup>w</sup>ə<sup>2</sup>ən-swe<sup>2</sup> <sup>2</sup>ən-šx<sup>w</sup>q<sup>w</sup>éləwən be.saying.what then ART your-own your-mind 'What do you think about it?' (lit. 'What is saying then your own mind?')
- (e) x<sup>w</sup>ə<sup>2</sup>i<sup>2</sup>ənt <sup>2</sup>a· θə šx<sup>w</sup>-q<sup>w</sup>él-s tə be.saying.what ROG! ART OBLNOM-speak-3POS ART syáwən-s. (JP 1) song-3POS
  'What are the words to his song?' (lit. 'What do the words of his song say?')

## 17.15. sx<sup>w</sup> ∂<sup>9</sup> í·nt 'WHAT MEANING?'

This is composed of *s*- 'resultative'(?) (recorded as *s*-, not  $\check{s}$ -) and  $x^{w}\partial^{2}i\cdot nt$  'say what.'

(a) sx<sup>w</sup>ə<sup>9</sup>í·nt k »ə tə k<sup>w</sup>ł. (AG) what.meaning then ART k <sup>w</sup>ł 'What does [the particle] k<sup>w</sup> mean?' (b) sx<sup>w</sup>ə<sup>9</sup>í·nt-s k <sup>w</sup>ə tə  $k^{w}$ ł. (AG) what.meaning-3POS then k <sup>w</sup>ł ART

'What does he mean by  $k^{w}t$ ?'

(c) sx<sup>w</sup>ə<sup>?</sup>í·nts <sup>?</sup>a<sup>?</sup>a tə x<sup>w</sup>ənítəm tə watch. (JP) mean.what ROG! ART White.person ART watch 'What do the White people mean by the word *watch*?'

# 17.16. $x^{w}\partial^{2}i \cdot ntst\partial x^{w}$ 'SAY WHAT TO HIM?' (AG), 'SAY WHAT TO HIM ABOUT SOMETHING?' (JP)

This is composed of  $x^w \partial^2 t \cdot nt$  'say what' and  $-st \partial x^w_2$  'comitative.'

- (a) x<sup>w</sup>ə<sup>9</sup>í·ntstəx<sup>w</sup> čx<sup>w</sup> ce<sup>9</sup>. (AG)
   say.what.to.him you FUT
   'What will you say to him?'
- (b) ?i čx<sup>w</sup> x<sup>w</sup> ?í?əntstamx. (AG)
  ?i čx<sup>w</sup> x<sup>w</sup> ?í?ənt-st-amx
  AUX you be.saying-com-me
  'What are you saying to me?'
- (c) ?śwəte? kw xwə?intstalə ?əθyəwenələp. (JP 8)
   ?śwəte? kw xwə?int-stalə ?əT-syəwen-ələp
   none ART say.what-to.you(PL) your-before-2PL
   'Nobody can say anything to you about your tradition.' (lit. 'your past')

#### 17.17. ?ónaca (CC, JP), ?ónca (AG, DK) 'WHERE?'

This does not seem related to other interrogatives. The form  $2\delta nc\partial$  is probably Cowichan.

- (a) nə <sup>9</sup>ánəcə lə nəwáč. (CC) ni? <sup>9</sup>ánəcə lə nə-wáč AUX where ART my-watch 'Where is my watch?'
- (b) ni <sup>2</sup>όnəcə k<sup>w</sup>θə <sup>2</sup>əθxéýəł. (JP 1) ni<sup>2</sup> <sup>2</sup>ónəcə k<sup>w</sup>θə <sup>2</sup>əT-s-xéýəł AUX where ART your-NOM-senior.sibling/cousin 'Where is your senior cousin?'
- ?ə (c) ni čxw ?ánaca. ?i ?í təná. (CC) cən vou where Ι be.here this AUX AUX OBL 'Where are you?' 'I am here.'

In a subordinate clause,  $2 \delta n \partial c \partial$  serves as the predicate of an embedded question, as in (d).

| (d) ni cə                                                      | n Żəw 1                                                   | mélqt                               | wəni?əs ?                                                                    | śnəcə                              | θə tíntə                               | n. (CC                                                | 21)                                         |                                  |
|----------------------------------------------------------------|-----------------------------------------------------------|-------------------------------------|------------------------------------------------------------------------------|------------------------------------|----------------------------------------|-------------------------------------------------------|---------------------------------------------|----------------------------------|
| ni?                                                            | cən<br>I                                                  | х́е<br>too                          | wə-mélq                                                                      | l-t<br>t TD                        | wə-ni <sup>9</sup> -                   | ·ƏS<br>hara as                                        | UD                                          | <sup>9</sup> ánəcə<br>whore      |
| AUX<br>θə<br>ART<br>'I too                                     | tín<br>bell<br>had fors                                   | too<br>tən<br>gotten v              | where the b                                                                  | ell was                            | .'                                     | nere-38                                               | UВ                                          | where                            |
| The wo                                                         | rd mák                                                    | <sup>v</sup> 'all' se               | erves to for                                                                 | m <i>mál</i>                       | <sup>w ?</sup> ánəcə                   | 'every                                                | where                                       | ,' as in (e).                    |
| (e) mák <sup>w</sup><br>all<br>'Every                          | ?án<br>whe<br>where t                                     | əcə<br>ere<br>here wa               | wə-qá <sup>?</sup><br>EST-water<br>as only wat                               | a<br>ju<br>ter.'                   | İ. (JP 1)<br>ıst                       |                                                       |                                             |                                  |
| The par                                                        | ticle <sup>9</sup> al                                     | 'just' f                            | orms 'anyv                                                                   | where,'                            | as in (f).                             |                                                       |                                             |                                  |
| (f) <sup>?</sup> án cə<br><sup>?</sup> án cə<br>where<br>'When | k <sup>w</sup> ən s<br>k <sup>w</sup><br>AR<br>re shall l | słédot.<br>o n<br>r m<br>layito     | ?əncə ?al.<br>ə-s-łéq-ət<br>ıy-nom-lie-<br>down?' 'Ar                        | (DK)<br>TR<br>nywher               | <sup>9</sup> əncə<br>where<br>e.'      | <sup>9</sup> aỉ<br>just                               |                                             |                                  |
| The pre                                                        | fix <i>təl-</i>                                           | from'                               | forms <i>tələ́i</i>                                                          | 1 <i>əcə</i> 'c                    | ome from                               | where                                                 | ?' as i                                     | n (g).                           |
| (g) tələna<br>from.v<br>'When                                  | oco<br>vhere<br>e are yo                                  | čx <sup>w</sup> .<br>you<br>ou from | təlí?<br>be.from<br>? I'm from                                               | cən<br>I<br>Vanco                  | [?ə]<br>OBL<br>uver.'                  | Х́<br>ART                                             | Vanc<br>Vanc                                | ouver. (CC)<br>ouver             |
| The pre                                                        | fix x <sup>w</sup> ə-                                     | 'becon                              | ne' forms x                                                                  | :"ə <sup>2</sup> ánə               | <i>cə</i> 'get to                      | where                                                 | ?' as i                                     | n (h).                           |
| (h) <sup>9</sup> áwə<br><sup>9</sup> áwə<br>not<br>'He di      | ni∙s<br>ni?-a<br>AUX-<br>dn't seer                        | st<br>əs<br>3SUB<br>m to get        | te <sup>9</sup> é k <sup>w</sup><br>stə <sup>9</sup> é<br>like<br>tanywhere. | əw :<br>[?ə]<br>OBL<br>.' (lit. '] | x "ə?ə́nəc<br>k "ə<br>ART<br>t was not | xə. (JP 1<br>wə-x <sup>w</sup><br>EST-get<br>like his | 9)<br>ə <sup>9</sup> ánə<br>to.wh<br>gettir | cə<br>lere<br>lg anywhere.')     |
| An artic<br>and sema<br>(see §15.2                             | cle and<br>ntically<br>2.2.1).                            | ? <i>ánəcə</i><br>the into          | form the interrogative of                                                    | nterrog<br>counter                 | ative 'wh<br>rpart of th               | nich?' v<br>he loca                                   | vhich<br>tive d                             | is structurally<br>emonstratives |
| (i) ⊼a<br>BE3P<br>'Whic                                        | k <sup>™</sup> ə-?<br>ART-w<br>h one is                   | án əcə.<br>here<br>it?'             | (JP)                                                                         |                                    |                                        |                                                       |                                             |                                  |

(j) Xa k<sup>w</sup>∂ t∂-<sup>9</sup>∂n∂c∂ púk n∂-swé?. (CC)
 BE3P then ART-where book my-own
 'Which book is mine?'

In (i),  $k^w \partial^2 \delta n \partial c \partial$  is used pronominally, and in (j),  $t \partial^2 \delta n \partial c \partial$  is used adjectivally. In (i),  $k^w \partial$  indicates that the choices are not in view of the speaker, while in (j), the  $t \partial$  indicates that the choice is among books that are present and visible.

#### 17.18. <sup>9</sup>ánacastax<sup>w</sup> 'PUT IT WHERE?'

This is composed of  $? \delta n \partial c \partial$  'where?' and  $-st \partial x^{w_1}$  'causative.'

(a) <sup>?</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a)</sup> <sup>(a</sup>

In (b),  ${}^{?} \partial n \partial c \partial s t \partial x^{w}$  appears in the passive voice in a dependent clause:

(b) sówdt ct ce? wonis ?ónocostom kws ni?s skwékwolt. (JP 12) sáwd-t ct ce? wə-ni?-əs ?ánacastam k <sup>w</sup> seek-TR we FUT if/that-AUX-3SUB be.put.where ART s-ni?-s sk wék walt hidden NOM-AUX-3POS 'We'll look for where they were put when they were hidden.'

#### 17.19. n acím 'WHY?'

For a question 'why do ..., why did ...,' and so on, the interrogative word is followed by a verb with the oblique nominalizer  $\check{s}x^{w}$ - (CC sometimes used *x*- in this context) and a possessive.

- (a) nəcím k<sup>w</sup>e <sup>9</sup>əšx<sup>w</sup>ném. (CC) nəcím k<sup>w</sup>e <sup>9</sup>ə-šx-ném why then your-OBLNOM-go 'Why are you going?'
- (b) nəcím kwe ?əxni?ələp ném. (CC)
   nəcím kwe ?ə-x-ni-?ələp ném
   why then your-OBLNOM-AUX-2PLPOS go
   'Why did you folks go?'
- (c) nəcím kwe xəs ném. (CC)
   nəcím kwə šxw-ni?-s ném
   why then OBLNOM-AUX-3POS go
   'Why did he go?'

In (c), *x* $\partial s$  is probably a reduction of  $\delta x^w ni^2 s$ . However,  $n\partial c i m$  may also be followed by a coordinate clause, giving the sense 'why should ...,' as in (d).

(d) nəcím kwe i ném cən. (CC)
 nəcím kwe ?əỷ ném cən why then and go I
 'Why should I go?'

#### 17.20. *k<sup>w</sup>ín* 'HOW MANY?'

In its usual sense,  $k^{w}in$  stands as a predicate with whatever is asked about standing as its subject.

 (a) k<sup>w</sup>ín k<sup>w</sup> téłs θə <sup>?</sup>əswáltən. (JP) k<sup>w</sup>ín k<sup>w</sup> téł-s θə <sup>?</sup>əT-swáltən how.many ART fathom-3POS ART your-net 'How many fathoms is your net?'

The word k''(n) may also stand as the first of a pair of coordinate clauses and mean 'what time?' – that is, 'how many (hours)?' (Cf. *tomtém* 'when' above.) Presumably this is a modern usage.

| (b) k <sup>w</sup> ín ce <sup>9</sup> i | ném čy  | « <sup>w</sup> . (DK) |     |     |
|-----------------------------------------|---------|-----------------------|-----|-----|
| <b>k</b> <sup>w</sup> ín                | ce?     | °∍ỷ                   | ném | čxw |
| how.many                                | FUT     | and                   | go  | you |
| 'What time                              | are you | going?'               |     |     |

In some other contexts, for example, as a noun modifier,  $k^{win}$  can mean 'a few, several, a number of,' as in (c) and (d).

| (c) | k™θә      | təw-k <sup>w</sup> ín | swáwləs (JP 14) |
|-----|-----------|-----------------------|-----------------|
|     | ART       | somewhat-how.many     | young.men       |
|     | 'a few yo | oung men'             |                 |
|     | -         |                       |                 |

(d) i  $\ddot{k}$  k<sup>w</sup>ín swéyəl i  $\ddot{\lambda}$ ə nəwł ném t $\theta$ e?. (JP 1) k<sup>™</sup>ín wəł-ném the? ?i swévəl ?∍ỷ х́е ni? how.manv dav and already-go he AUX again AUX 'After several days, he went again.'

The word  $k^w in$  also takes a number of lexical suffixes, many of which also appear with numerals. (For examples, see §19.2.) For canoes, instead of a lexical suffix, there is the prefix ya- 'along' and the root is reduplicated:  $yak^w ik^w an$  'how many canoes/conveyances?'

Most, if not all, of these interrogatives can also appear in a context where they mean 'several,' as in (e) to (i).

- (e) təw-yək<sup>w</sup>ík<sup>w</sup>ən (JP) somewhat-how.many.canoes 'several canoes'
- (f) yək<sup>w</sup>ík<sup>w</sup>ənəs ctwa<sup>?</sup> snəx<sup>w</sup>əł. (JP) yə-k<sup>w</sup>ík<sup>w</sup>ən-əs ctwa<sup>?</sup> snəx<sup>w</sup>əł along-how.many-3SUB SPEC canoe
   'There were I don't know how many canoes.'

- (g) təw-kwənéł (JP) somewhat-how.many.times 'several times'
- (h) təw-kwí·lə somewhat-how.many.people'several people'

The last can be pluralized, as in (i).

(i) təw-k<sup>w</sup>əlk<sup>w</sup>í·lə (JP 16)
 somewhat-how.many.people(PL)
 'several groups of people'

# **18** Adverbs and Adverbial and Modal Words and Phrases

Under this heading are included a number of words and phrases that can qualify a predicate in ways that may be termed adverbial or modal. Some of these words and phrases appear in but one kind of grammatical relationship to a predicate, while others can appear in several kinds. Some have no function to perform other than adverbial or modal, while others may also function as adjectives or verbs. Nevertheless, there seems to be a core of words (disregarding now the phrases) here that cannot be placed anywhere else. Most of these undergo no internal modification, that is, they are not inflected for aspect, number, and so on, and they might be seen to form a class on that basis. On the other hand, a few that seem to belong here on the basis of function are themselves inflected forms.

Adverbial and modal functions have already been identified in different contexts. Here, §18.1 reviews these functions, §18.2 considers adverbs as a category, §18.3 reviews the grammatical relations of what are called here adverbial and modal predicates, and §18.4 catalogues the words and phrases that have been recorded serving as adverbs and as adverbial and modal predicates and gives examples of usage.

#### **18.1. ADVERBIAL AND MODAL FUNCTIONS**

Adverbial functions are performed by modifiers standing in various relationships to the predicates or clauses they modify. Adverbs (see §3.2.2) stand within the predicate in ways reviewed in §18.2. Verbal adjuncts (analogous to English prepositional phrases – see §3.5) and adverbial adjuncts (§3.6) stand outside the predicate but within the clause. Subordinate clauses used as conditional clauses (§4.2.1) and nominalized clauses used adverbially (§4.3.2) stand outside the clause. In all of these, what is modified is a main clause predicate or predicate head, and the adverbial modifier is grammatically dependent on or subordinate to it. There is still another way, however, in which an adverbial modifier can relate to the predicate it modifies. It can stand as a predicate and main clause itself while the modified predicate (or clause) relates to it as a subordinate clause or a coordinate clause. Adverbial modifiers of this sort may be said to function as "adverbial predicates," and the words and phrases that can function so are called here "adverbial words and phrases." Examples of adverbial predicates appear in several places in §4 and §5. The kinds of grammatical relations they have with predicates are reviewed in §18.3.

Functions termed "modal" (admittedly a vague concept) are performed by some of the predicate particles (§16.2) and by the modal suffixes (§11.2). There are also a few words that can stand in the same grammatical relationships to predicates (or clauses) as do the adverbial words and that might be termed "modal words" serving as "modal predicates."

Actually, there is not a very clear distinction between the "adverbial" and the "modal" in this context, it being a matter of semantics rather than grammar. In English, grammar seems to make a clear distinction: "I cannot hear him" has a modal auxiliary, while "I barely hear him" has an adverb. In Halkomelem, however, "cannot" and "barely" are both expressed by adverbial/modal predicates with embedded qualified clauses.

Some of these adverbial and/or modal words can also function as subordinating conjunctions. Examples are given in §18.4.

#### 18.2. ADVERBS

Adverbs are identified in §3.2.2 as words that can stand within a predicate as modifiers. They can appear in various positions.

In verbal predicates, they commonly appear initially, as with  $y\dot{a}\theta$  'always' and  $q\dot{e}^{\gamma}is$  'recently' in (a) and (b).

| (a) | <ul> <li>a) wə-yáθ-əł</li> <li>EST-always-past</li> <li>'I was always sick.'</li> </ul> |     | cən<br>I | wə-ἀáἀəỷ. (DK)<br>Est-sick |  |
|-----|-----------------------------------------------------------------------------------------|-----|----------|----------------------------|--|
| (b) | qé <sup>9</sup> is                                                                      | cən | ni       | kʷác-nəxʷ. (JP)            |  |
|     | recently                                                                                | I   | AUX      | look-tr                    |  |

Some can also appear between the auxiliary and the predicate head, as with  $m \delta k^{w}$  'all' in (c).

(c) ni cən wə-mə́k<sup>w</sup> pə́n-ət. (CC) AUX I EST-all get.buried-TR 'I buried them all.'

'I saw him just now.'

Or following the head, as with  $\lambda e$  'also' in (d).

(d) sis wəqaynəx wəs ke ?é-ltən. (JP 12)
s-ni?-s wə-qay-nəx w-əs ke ?é-ltən NOM-AUX-3POS EST-die-TR-3TR also 3PL
'Then they killed him too.'

In adjectival predicates (§3.7.1) and in nominal predicates with adjectival modifiers (§3.7.3), an intensifying adverb can precede the adjective, as with  $\dot{n}\dot{a}n$  'very' and  $s\dot{c}\dot{e}\dot{c}\dot{a}\dot{n}$  'really' in (e) and (f).

- (e) wəủán əw k<sup>w</sup>ám k<sup>w</sup>əm θəủa spəhéls. (JP 24)
   wə-ủán wə-k<sup>w</sup>ám k<sup>w</sup>əm θəủa spəhéls
   EST-very EST-strong this wind
   'This wind is very strong.'
- (f) scécon yox<sup>w</sup> ce? wo-qóx spédom.
   really INF FUT EST-many flower
   'There will really be lots of flowers.'

In nominal predicates,  $\dot{\lambda}e$  'also' can appear either before or after the head, as in (g) and (h).

(g) Xowoł táwon. (AC)
Xowoł-táwon
also already-town
'It's still town [way out there too].'

(h) wəxpéy ke tə ni šx qpéqəts. (JP 27)
 wə-xpéy ke tə sx qəp-éqən-t-s
 EST-cedar also ART OBLNOM-stick-throat-TR-3POS
 'What he covered it with was also cedar.'

The words most often recorded functioning as adverbs specify frequency, relative time, intensity, specificity, and similar qualities. There are fewer than twenty of these. They include:  $y\dot{a}\theta$  'always,'  $t\dot{a}\dot{q}$  'usually,'  $\dot{a}t$  'earlier, long ago,'  $q\dot{e}^{2}is$  'recently,'  $\dot{\lambda}\dot{e}$  'again, also,'  $q\dot{a}\dot{l}\dot{e}t$  'again, also,'  $h\dot{a}^{2}$  'if, when,'  $x^{w}\dot{a}n\dot{a}^{2}$ 'first,'  $ya\dot{w}\dot{e}n$  'before,'  $li^{2}\dot{a}^{2}aq^{w}t$  'next,'  $\dot{n}\dot{a}n$  'very, too,'  $s\dot{c}\dot{e}\dot{c}a\dot{n}$  'really,'  $\theta a^{2}it$ 'truly,'  $t\dot{a}\ddot{x}^{w}$  'precisely,'  $h\dot{a}y$  'specifically,'  $wan\dot{a}y$  'only,'  $m\ddot{a}\ddot{k}^{w}$  'all.'

Words that function as adverbs, however, do not constitute a clearly definable class. It seems that there are very few words that serve as adverbs only, having no other functions. Only  $h\dot{a}^2$  'if, when,'  $l\dot{a}\dot{q}$  'usually,'  $\dot{n}\dot{a}n$  'very, too,' and  $s\dot{c}\dot{e}\dot{c}\dot{a}n$  'really' have been recorded as adverbs but nowhere else. Most words that commonly serve as adverbs may also appear elsewhere, as noun modifiers in nominal adjuncts (§3.8.3.3), such as  $\dot{\lambda}\dot{e}$  'also' and  $m\dot{a}k^w$  'all'; as heads of adverbial adjuncts (§3.6), such as  $q\dot{e}^2is$  'recently'; or as predicate heads, such as  $y\dot{a}\theta$  'always' and  $t\dot{a}\dot{x}^w$  'precisely, adjust.' (Examples of usage of all of these are found in §18.4.)

Moreover, at least some words that function primarily as adjectives (see §9) can also occasionally serve as adverbs, such as  $\theta i$  'big,'  $c \dot{a} k^w$  'far,' and  $\check{x} \dot{a} t$  'painful, hurt' in (i), (j), and (k).

(i) scécon con niw θí q<sup>w</sup>íx<sup>w</sup> ... (JP 24) scécan can ni? wə-θí q<sup>w</sup>íž<sup>w</sup> really T EST-big miss AUX 'I really made a big mistake ...' (lit. 'I really greatly missed ...') cá∿k™ x wə-?áý. (CC 21) (i) ni cən AUX I far become-good 'I am far better.' (k) žáł dádəý. hurt/sore sick 'He was very sick.'

These adjectives used as adverbs are not catalogued in §18.4.

There is still another construction in which the predicate contains a modifier that may be identified as an adverb. In this construction (see \$4.3.1), the predicate head is a nominalized clause that is preceded by the modifier, as with *yét* 'just now' in (1).

| (l) | yéł          | мэ      | s-ném-s. (CC) |
|-----|--------------|---------|---------------|
|     | just.now     | CERT    | NOM-go-3POS   |
|     | 'He's just g | going.' |               |

Only four words have been recorded functioning as modifiers in this construction:  $m \delta \vec{k}^w$  'all' (with the sense 'whenever'),  $w \partial n \delta y$  'only,'  $c \partial l \ell l \partial a l$  'barely,' and  $y \ell t$  'just now, after which.' Of these,  $m \delta \vec{k}^w$  and  $w \partial n \delta y$  appear as adverbs in the usual fashion, while  $c \partial l \ell l (\partial a l)$  and  $y \ell t$  have not been so recorded.

#### **18.3. ADVERBIAL AND MODAL PREDICATES**

These are words and phrases that stand as main-clause predicates followed by clauses that they in some way qualify. There are three kinds of grammatical relationships that can exist between an adverbial or modal predicate and the qualified clause:

(1) The qualified clause appears as a nominalized clause standing as a nominal adjunct (identified in §4.3.2), as in (a) and (b).

(a) sk<sup>w</sup>éy k<sup>w</sup>ən sném. (JP)
 sk<sup>w</sup>éy k<sup>w</sup>ə nə-s-ném
 impossible ART my-NOM-go
 'I can't go.'

(b) híθ k<sup>w</sup>əns<sup>9</sup>íṁ̀əx. (CC)

hí $\theta$  k<sup>w</sup> $\Rightarrow$  n $\Rightarrow$ -s-?ím $\Rightarrow$ x last.long ART my-NOM-be.walking 'I was walking a long time.'

In (a), the nominal adjunct is probably a subject, the sentence being more literally 'My going is impossible.' In (b), the nominal adjunct may be a subject ('My walking took a long time'), or it may be a nominalized clause used adverbially ('It took a long time when I was walking').

(2) The qualified clause stands as a subordinate clause (see §4.2), as in (c) and (d).

| (c) | sk wéy ma wa             | ə <sup>9</sup> áwè∙n ( | (~ wəwé∙n) némè | •n. (CC) | )     |          |
|-----|--------------------------|------------------------|-----------------|----------|-------|----------|
|     | sk <sup>w</sup> éy       | мэ                     | wə-?śwə-e·n     | nér      | n-e∙n |          |
|     | impossible               | CERT                   | that-not-I      | go-I     |       |          |
|     | 'I must go.'             |                        |                 | -        |       |          |
| (d) | wəłqžéł <sup>9</sup> əłr | némè∙n k               | wi scáməs. (CC) |          |       |          |
|     | wəł-qžéł                 |                        | ?əł-néṁ-e∙n     | [?ə]     | k™i   | scáməs   |
|     | already-many             | .times                 | that-go-I       | OBL      | that  | Songhees |
|     | 'I have often            | gone to V              | /ictoria.'      |          |       |          |

In both of these the subordinate clauses are probably better identified as complements rather than conditional clauses (lit. 'It is impossible that I not go' rather than 'It will be impossible if I do not go,' etc.).

(3) The qualified clause stands as the second of a pair of coordinate clauses (see §5.2), as in (e) and (f).

| (e) | ċəx™lé? ċə i  | ném háy  | e <sup>9</sup> tə c | wé <sup>9</sup> tə lo | éləm. (JP | 12) |                  |      |
|-----|---------------|----------|---------------------|-----------------------|-----------|-----|------------------|------|
|     | ċəx™lé?       | ćə       | °∂ỷ                 | ném                   | háye?     | tə  | cwé <sup>9</sup> | [?ə] |
|     | sometimes     | QUOT     | and                 | go                    | leave     | ART | own              | OBL  |
|     | tə            | léləm    |                     |                       |           |     |                  |      |
|     | ART           | house    |                     |                       |           |     |                  |      |
|     | 'Somotimos ti | ha aumar | of the h            |                       | t onvorv' |     |                  |      |

'Sometimes the owner of the house went away.'

| (f) | ž™́э́m   | °ə       | čx <sup>w</sup> | ?∍ỷ    | x <sup>w</sup> ?əwélcəs. (JP) |
|-----|----------|----------|-----------------|--------|-------------------------------|
|     | fast     | ROG      | you             | and    | make.string.figures           |
|     | 'Can you | u make s | tring figu      | ires?' |                               |

In both of these sentences, the  ${}^{2}o\dot{y} \sim i$ , which is 'and' in most contexts and occasionally 'but,' might be better translated 'when.' The sentences might be better translated 'There were said to be occasions when the owner of the house left,' and 'Are you fast when you make string figures?'

The words that have been recorded as adverbial predicates include a few that have also been recorded as adverbs ( $y\dot{a}\theta$  'always,'  $y\partial\dot{w}\dot{e}n$  'first,'  $h\dot{a}y$  'specifically,'  $w\partial n\dot{a}y$  'only'). But most that have been recorded in one function have

not been recorded in the other. A few of these adverbial words have been recorded as simple predicates ( $hi\theta$  'last long,'  $n\acute{e}c$  'be different'). Some of the adverbial and modal words have been recorded with more than one kind of qualified clause, while others have been recorded with one kind only.

As with adverbs, the function of adverbial predicates can be assumed by words that usually serve as adjectives, such as  $\vec{k}^w \vec{a} \vec{m} \vec{k}^w \vec{a} \vec{m}$  'strong' and ' $\vec{a} y \vec{a} m$  'slow' in (g) and (h).

- (g) k<sup>w</sup>ámk<sup>w</sup>əm k<sup>w</sup>s k<sup>w</sup>áyəxəms. (JP 28)
   k<sup>w</sup>ámk<sup>w</sup>əm k<sup>w</sup>ə s-k<sup>w</sup>áyəxəm-s
   strong ART NOM-be.shaking-3POS
   'They shook strongly.'
- (h) wə?áyəm ?al k\*sni?s ném yəlíləqstx\*\*əs. (JP 23)
  wə-?áyəm ?al k\*\*ə s-ni?-s ném
  EST-slow just ART NOM-AUX-3POS AUX(go)
  yə-líləq-stəx\*\*-əs
  along-easy-CAUS-3TR
  'Slowly he eased it [the harpoon line] out.'

Here (g) may be more literally rendered either 'Their shaking was strong' or 'They were strong when they shook,' and (h) seems best as 'He was slow when he went easing it out.'

Some of the words considered here (háy 'specifically,' wanáy 'only,'  $tx^wáy$  'only remaining') enter into a kind of construction in which they qualify the subject rather than the predicate and so cannot be considered adverbs. In these cleft sentences (§4.1.5.2), the subject is either a noun followed by a relative clause or simply a relative clause, as in (i) and (j).

| (i) | wənáy       | ?aĺ      | tə       | k™áx™a?        | ni        | k <sup>w</sup> ən-é-t-əs. (JP 3) |
|-----|-------------|----------|----------|----------------|-----------|----------------------------------|
|     | be.only     | just     | ART      | box            | AUX       | get.taken-DUR-TR-3SUB            |
|     | 'It was on  | ly a box | that he  | held.'         |           |                                  |
| (j) | wənáy       | ?aĺ      | mi-əx    | K <sup>w</sup> | tál-ləx™. | (JP 28)                          |
|     | be.only     | just     | AUX(C    | ome)-you       | be.unders | stood-tr                         |
|     | 'That's all | you ca   | me to kn | low.'          |           |                                  |

In (j), the *wonáy* might be taken for an adverb, but the fact that what follows is a relative clause suggests that (j) is a cleft sentence, even though the relative clause is not preceded by an article. This suggests the possibility that sentences in which these predicates take first- and second-person subject particles are also cleft sentences. Compare (k) and (l):

| (k) | tx wáy qwáqwəlsta: | m x. (DK)                |
|-----|--------------------|--------------------------|
|     | tx™áy              | q*áq*əl-stəx*-amx        |
|     | only.remaining     | be.speaking-COM-me       |
|     | 'He's the only one | who still speaks to me.' |

| (l) | tx™áy               | ct     | q <sup>w</sup> áq <sup>w</sup> əl-stəx <sup>w</sup> | k <sup>w</sup> ə | ł (DK)         |
|-----|---------------------|--------|-----------------------------------------------------|------------------|----------------|
|     | only.remaining      | we     | be.speaking-COM                                     | ART              | portion        |
|     | 'We are the only of | nes wh | o still speak to the _                              | s.' (su          | rname deleted) |

In (k), the status of  $q^w \dot{a} q^w \partial lstam x$  as a relative clause is indisputable (there being no  $-\partial s$  'third-person transitive subject' required by a main-clause verb), and so the status of the whole as a cleft sentence is indisputable. In (l), with other than a third person subject,  $q^w \dot{a} q^w \partial lst \partial x^w$  has a form appropriate for either a main-clause or relative-clause predicate head. The presence of ct 'we' seems to identify  $tx^w \dot{a} y$  as an adverb and  $q^w \dot{a} q^w \partial lst \partial x^w$  as predicate head. However, DK's translation shows that she saw the second sentence as constructed like the first.

#### 18.4. CATALOGUE

This section lists forty-seven words and phrases that have been recorded serving as adverbs, adverbial predicates, or modal predicates, and gives examples of the functions in which they have been recorded. The list is certainly not exhaustive. These words and phrases are ordered by semantic categories. Numbers 1-8 express frequency, 9-23 express duration and relative time, 24-35 are intensifiers and quantifiers, 36-40 are restricters and quantifiers, and 41-47 are more purely modal terms. They are described in the order shown in Table 18.1.

| Adverbs and adverbial and mo | dal wor | ds and phrases                                     |  |
|------------------------------|---------|----------------------------------------------------|--|
| Category                     | Terms   |                                                    |  |
| Terms expressing frequency   | 1       | $y \dot{a} \theta$ 'always'                        |  |
|                              | 2       | táď 'usually'                                      |  |
|                              | 3       | $\dot{c} \partial x^{w} l \dot{e}^{2}$ 'sometimes' |  |
|                              | 4       | <i>néč</i> 'other times'                           |  |
|                              | 5       | $n \partial \dot{c} e x^w$ 'once,' etc.            |  |
|                              | 6       | <i>qžét</i> 'often'                                |  |
|                              | 7       | ?áwa scék <sup>w</sup> alas 'never'                |  |
|                              | 8       | mák <sup>w</sup> swéyal 'every day,' etc.          |  |

►

| Table I | 18 |
|---------|----|
|---------|----|

| 18 |
|----|
|    |

| Category                                       | Tern                                                                                  | 15                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|------------------------------------------------|---------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Terms expressing duration<br>and relative time | 9<br>10<br>11<br>12<br>13<br>14<br>15<br>16<br>17<br>18<br>19<br>20<br>21<br>22<br>23 | $h\ell\theta$ 'for a long time<br>$n\delta ca^2 sw \ell y al$ 'for one day,' etc.<br>tat 'earlier'<br>$q\ell^2 is$ 'recently'<br>$y\ell\ell$ 'just now'<br>$w\ell y al$ 'tomorrow,' etc.<br>$\lambda \ell$ 'again, also'<br>$qal \ell t$ 'again, also'<br>ha2' 'if, when'<br>$m\delta k^w$ 'whenever'<br>$x^w aha^2$ 'first'<br>$ya \ell \ell t$ 'after'<br>$ta \chi^w$ 'later'<br>$x^w an \chi_a e^2 e(t\theta e^2)$ 'meanwhile'                                                                          |
| Intensifiers and quantifiers                   | 24<br>25<br>26<br>27<br>28<br>29<br>30<br>31<br>32<br>33<br>34<br>35                  | $\dot{n}\dot{a}n$ 'very, too'<br>$s\dot{c}\dot{c}\dot{c}\dot{n}$ 'really'<br>$\partial \partial^{2}\dot{t}$ 'truly'<br>$t\dot{a}\ddot{x}^{w}$ 'precisely'<br>$c\partial l\dot{\ell}$ 'nearly'<br>$c\partial l\dot{\ell}$ 'al 'barely'<br>$x^{w}\dot{\ell}\partial q$ 'nearly'<br>$(w\partial)q\dot{e}q\partial l '2d$ 'barely'<br>$snl^{2}\partial nl$ 'enough'<br>$\ddot{X}_{\partial x}^{w}l\dot{a}^{2}as$ 'no matter'<br>$^{2}iw\dot{a}\dot{w}\partial$ 'maybe'<br>$w\dot{a}\dot{l}\partial$ 'probably' |
| Restricters and quantifiers                    | 36<br>37<br>38<br>39<br>40                                                            | háy 'specifically'<br>wənáy 'only'<br>tx <sup>w</sup> áy 'only remaining'<br>cəlím' 'even'<br>mók <sup>w</sup> 'all'                                                                                                                                                                                                                                                                                                                                                                                       |
| Modal words                                    | 41<br>42<br>43<br>44<br>45<br>46<br>47                                                | $ \tilde{x}^{*}\delta m$ 'fast, can'<br>$s\tilde{X}\delta\tilde{X}\delta m$ 'enough, can'<br>$sk^{*}e'y$ 'impossible'<br>scowet 'capable'<br>$sqi^{2}q\delta$ 'unable'<br>$^{2}\delta\dot{y}$ 'better'<br>$s\tilde{X}i^{2}$ 'want                                                                                                                                                                                                                                                                          |

# 18.4.1. $y\dot{a}\theta$ 'always, frequently'

This has been recorded as an adverb only. It may precede the predicate head, as in (a).

(a) wə-yáθ-əł cən wə-qáqay. (DK)
 EST-always-past I EST-sick
 'I was always sick.'

Or it may follow the head (but with a different meaning), as in (b).

- (b) wə-qaq y 24
   (b) wə-qaq y 24
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   (c) wə-yaq (DK)
   <l
- Or it may be followed by a nominalized clause, as in (c).

| (c) yáθ k <sup>w</sup> ə | nəsniŵ           | yəłə <sup>9</sup> é. (JP) |                         |
|--------------------------|------------------|---------------------------|-------------------------|
| yáθ                      | k <sup>™</sup> ə | nə-s-ni?                  | wə-yə-łə <sup>9</sup> é |
| always                   | ART              | my-NOM-AUX                | EST-along-following     |
| 'I always                | s go the s       |                           |                         |

In the question (d) and the answer (e), the order changes.

- (d) wəyáθ <sup>?</sup> ö čx<sup>w</sup> ni w yəh ś nɨ m. (JP)
  wə-yáθ <sup>?</sup> ö čx<sup>w</sup> ni <sup>?</sup> wə-yə-h ś nɨ m.
  EST-always ROG you AUX EST-along-be.going 'Do you always go that way (or there)?'
- (e) ni cən wəyáθ wəyhônôm. (JP)
   ni? cən wə-yáθ wə-yə-hônôm
   AUX I EST-always EST-along-be.going
   'Yes, I always do.'

# 18.4.2. *łóq* 'usually, generally, habitually'

This has been recorded as an adverb only. It appears initially, as in (a) and (b).

(a) łód <sup>9</sup>ə dídək wəls tən sqwəméy. (DK) łáď ?a dídak<sup>w</sup>−als tə <sup>9</sup>ən-sq<sup>w</sup>əméy usually ROG be.biting-ACT your dog ART 'Does your dog bite?' (b) łóġ ni náčagan lísek k<sup>w</sup>í<sup>?</sup>x<sup>w</sup>. (CC 5) tə usually one.container sack AUX ART pitch 'It usually takes one sack of pitch.' (lit. 'The pitch is usually one sack.')

(For other examples, see §3.2.2. There is a related noun,  $l\dot{e}\dot{q}$  'fashion, custom,' as in  $l\partial l\dot{e}\dot{q}s$  to most  $\partial y_{\partial x^{w}}$  'the people's ways.')

#### 18.4.3. $\dot{c} \partial x^{w} l \dot{e}^{9}$ 'sometimes, there are times when'

This is probably more than a simple root, but its composition is unclear. It has been recorded in one construction only, as the first in a pair of coordinate clauses, as in (a) and (b).

- (a) cox "lé? i ném con to néc šx "imélo. (CC) cax <sup>w</sup>lé? ?∂ỷ ném [?ə] néċ šx <sup>w</sup>iméla cən tə different sometimes and OBL go T ART store 'Sometimes I go to other stores.'
- (b) c→x \*\*lé? i d→yn → x \*\*→ s t→ sm→y→θ. (JP7)
   c→x \*\*lé? ?→y d₄y-n→x \*\*→ s t→ sm→y→θ
   sometimes and die-TR-3TR ART deer
   'Once in a while he killed a deer.'

#### 18.4.4. néč 'at other times, sometimes'

The basic meaning of  $n\acute{c}$  is 'differ, be different, be strange.' It has inflected forms,  $n\acute{c}\acute{c}t$  'change it,'  $n\acute{a}\acute{c}\theta \imath t$  'change (oneself),' and so on, and many forms with derivational suffixes. It can appear as a noun modifier with the sense 'different' or 'other,' as in (a) in §18.4.3. Followed by a nominalized clause introduced by  $k^w$ , it has the meaning 'at other times,' as in (a) below.

(a) néč k<sup>w</sup>s néms ž<sup>w</sup>ənxénəm nem tə šnəwíł i ... (JP 1) néċ k \* ə s-ném-s x<sup>w</sup>ənxénəm nem [?ə] other times ART NOM-AUX(NOM)-3POS run go OBL šnawíł ?∂ỷ tə far side and ART 'At other times they would run to the other side and ...'

Repeated in two clauses, it may have the senses 'at one time' and 'at another time,' as in (b).

(b) néč kws nóča? swéyəl i mi técəl xwə?ámət i néč kws ni?s isélə swéyəl i mi xwə?ámət ?é:ltən (JP 12)

| ~                      |                  |                    | · · · · · · · · · · · · · · · · · · · |                   |      |       |                  |
|------------------------|------------------|--------------------|---------------------------------------|-------------------|------|-------|------------------|
| néč                    | k <sup>w</sup> ə | s-n ść             | a?[-s]                                | swé               | yəl  | °∍ỷ   | mi               |
| sometimes              | ART              | NOM-C              | one-3POS                              | day               |      | and   | come             |
| técəl                  | Х <sup>w</sup> ə | -?áṁət             |                                       | °∂ỷ               | néċ  |       | k <sup>w</sup> ə |
| arrive.here            | beco             | ome-be.s           | seated                                | and               | some | times | ART              |
| s-ni?-s                |                  | <sup>9</sup> isélə | swéyə                                 | l <sup>?</sup> əỷ | n    | 'ni   |                  |
| NOM-AUX-31             | POS              | two                | day                                   | and               | c    | ome   |                  |
| x <sup>w</sup> ə-?ámət |                  | ?é∙łt              | ən                                    |                   |      |       |                  |
| become-be.             | seated           | 3PL                |                                       |                   |      |       |                  |

'Sometimes they came home after one day, and other times they came home after two days.'

#### 18.4.5. noćéx<sup>w</sup> 'once, one time' and so on

This word (<  $n\dot{a}\dot{c}a^2$  'one'),  $\theta\partial m\dot{e}$  'twice' (<  $\theta em$ - 'two'),  $lx^w\dot{e}l$  'three times' (<  $lix^w$  'three'),  $l\partial w\dot{k}^w\partial n\dot{e}l$  'a number of times' (<  $k^win$  'how many'), and probably other words referring to numbers of times (see §19.2 [2]) can occur followed by nominalized clauses introduced with  $k^w$ , as in (a) and (b).

(a) nočéx<sup>w</sup> čo k<sup>w</sup>smis tóq to sxixé<sup>9</sup>ł. (JP 12) n əcéx <sup>w</sup> ćə k "ə s-mi-s táq get.blocked once OUOT ART NOM-AUX(come)-3POS sxixé?ł tə Sechelt ART 'Once the Sechelt were taken by surprise.'

| (b) θəmé k | θəmé k <sup>w</sup> s hánəməst <sup>w</sup> əs tə syáwəns. (JP) |                       |          |           |  |  |
|------------|-----------------------------------------------------------------|-----------------------|----------|-----------|--|--|
| θəmé       | k <sup>w</sup> ə                                                | tə                    | syáwən-s |           |  |  |
| twice      | ART                                                             | NOM-be.going-CAUS-3TR | ART      | song-3POS |  |  |
| 'Twice l   | he sang h                                                       | is song.'             |          |           |  |  |

They can also occur as the first of two coordinate clauses, as in (c) and (d). In (c),  $n\partial \dot{c} ex^w$  is translated 'once in a while,' but such an interpretation is not likely for  $\partial \partial m \dot{e}$  and so on.

(c) nəcex i ni təwk vənnəx vəs k vəw stem val svəttəns. (JP 12) nəčéx <sup>w</sup> ?∍ỷ ni? təw-k<sup>w</sup>án-nəx<sup>w</sup>-əs k<sup>w</sup>⊃ once and somewhat-get.taken-TR-3TR AUX ART ?aĺ s?ə́łtən-s wə-stém EST-what food-3POS iust 'Once in a while they found something to eat.'

 (d) łx "éł i ni sẳáẳəm. (JP 21) łíx "-éł ?əỷ ni? sẳáẳəm three-times and AUX enough 'Three times would be enough.'

In (e),  $n\partial \dot{c} ex^w$  'once' is the subject of the first clause, followed by a nominalized clause in which the first word is  $\dot{\chi} e$  'again,' which is followed by another nominalized clause.

(e) ni čkwə təs kwθə nəcexw kwskesəw mi tecəl təwkaləm sənixwəł. (JP 14) ni? ćə k "ə tás k<sup>w</sup>θə nəcéx <sup>w</sup> k »ə arrive.there AUX OUOT then ART one.time ART s-že-s [s-]wə-mi técəl tawkalam [NOM-]EST-AUX(come) arrive.here those NOM-again-3POS səníx <sup>w</sup>əł canoes

'There came (then, it is said) another time when those canoes came again.'

#### 18.4.6. qxéł 'often, many times'

Composed of  $q \delta \dot{x}$  'many' and  $-\dot{e}t$  'times,' this can occur with a conditional clause introduced with  $^{2}\partial t$ - 'whenever,' as in (a).

(a) wəłqxéł ?əłnémen k<sup>w</sup>i scáməs. (CC)
wəł-qxéł ?əł-ném-en [?ə] k<sup>w</sup>i scáməs already-many.times whenever-go-I OBL that Songhees
'I have often been to Victoria.' (lit. 'It is many times that I go to Victoria.')

#### 18.4.7. ?*ówə scék* »*ðləs* 'never'

Composed of  $2\delta w \partial$  'not' and  $sc \dot{e}k w \partial \dot{l}$  'how, how much' (lit. 'not to any extent'), this can appear as the first of a pair of coordinate clauses, as in (a).

(a) <sup>9</sup>ówə scék <sup>w</sup>ələs i <sup>1</sup>⁄<sub>e</sub> cən wəq<sup>w</sup>əlstàmə. (CC)
 <sup>9</sup>ówə scék <sup>w</sup>əl-əs <sup>9</sup>əỷ <sup>1</sup>⁄<sub>e</sub> cən wə-q<sup>w</sup>el-st-amə not how.much-3SUB and again I EST-speak-COM-you <sup>1</sup>I'll never speak to you again.<sup>2</sup>

#### 18.4.8. $m \delta \vec{k}^w s w \delta y \delta l$ 'every day' and so on

This term (composed of  $m \delta k^w$  'all' and  $sw \epsilon y \delta l$  'day'),  $m \delta k^w n \epsilon t \delta l$  'every morning,' and similar terms can occur followed by a nominalized clause introduced by  $k^w$ , as in (a), (b), and (c).

- (a) mók<sup>w</sup> swéyəl k<sup>w</sup>s wənéms k<sup>w</sup>áyxθət tə stáləss. (JP 20) mákw swéyəl k <sup>w</sup>ə s-wə-ném-s k<sup>w</sup>éỷ<sub>x</sub>-θət tə all dav NOM-EST-go-3POS be.moving-self ART ART stáləs-s spouse-3POS 'Her husband went out every day.'
- (b) ni  $t\theta e^{\gamma} k^{w}s \check{x} \acute{e} \dot{m}s m \acute{s} \check{k}^{w} s w \acute{e} v \imath l$ . (JP 20) ni? ?a k "ə s- xé·m-s mákw tθe? swévəl be there OBL that ART NOM-be.crying-3POS all day 'There she cried every day.'
- (c) mák<sup>w</sup> nétał k<sup>w</sup>e <sup>?</sup>e $\theta$ yá·ýas k<sup>w</sup> ste<sup>?</sup>é <sup>?</sup>e t $\theta$ e<sup>?</sup>. (JP 21) mákw nétəł k <sup>w</sup>ə ?əT-s-vá·vəs [?ə] k <sup>w</sup>ə your-NOM-be.working all morning ART OBL ART sta?é ?e tθé? like OBL that 'Every morning you do the same thing.'

In (d), however, 'every day' appears in an adverbial phrase following the predicate.

(d) <sup>?</sup>i<sup>·</sup>ł ct spápətən k<sup>w</sup>ə mə́k<sup>w</sup> swéyəl. (DK) mákw ?i-əł ct spápətən [?ə] k<sup>w</sup>∂ swévəl AUX-past we be.sailing OBL ART all dav 'We were sailing constantly every day.'

#### 18.4.9. $hi\theta$ 'last long, be a long time'

This can appear as a predicate head, with an auxiliary as in (a) and with an adverb as in (b).

- (a) nem təw-híθ. (JP 3) AUX(go) somewhat-last.long 'Some time passed.'
- (b) wə-ảán ở twa? wəł-híð. (JP 15) EST-very SPEC already-last.long 'It must have been very long ago.'

It can precede the predicate head as an adverb where the head is a verb, as in (c), and where the head is a noun, as in (d).

- (c) híθəł cən k<sup>w</sup>ł wa<sup>9</sup> <sup>9</sup>í<sup>9</sup>tət. (JP 10) híθ-əł cən k<sup>w</sup>ł wa<sup>9</sup> <sup>9</sup>í<sup>9</sup>tət last.long-past I by.then PRES be.sleeping 'I must have been sleeping a long time.'
- (d) wəłíθ să<sup>w</sup>əỷém. (JP 15)
   wəł-híθ să<sup>w</sup>əỳém already-last.long story
   'It's an old story.'

It can appear as a predicate head followed by a nominalized verb, as in (e), or by a nominalized clause, as in (f) and (g).

- (e) wəłí@ ?al sqálqələ@əns. (CC 16)
   wəł-hí@ ?al s-qálqələ@ən-s already-last.long just NOM-dream-3POS
   'For a long time he had been dreaming.'
- (f) wəłíθ kwəns?í ə təủa. (JP)
   wəł-híθ kwə nə-s-?í ?ə təủa
   already-last.long ART my-NOM-be.here OBL this
   'I've been here a long time.' (lit. 'My being here has lasted long.')

(g) <sup>2</sup> śwə híθəs k<sup>w</sup>e <sup>2</sup>eθ<sup>2</sup> éİməỷ. (JP 3)
<sup>2</sup> śwə híθ-əs k<sup>w</sup>ə <sup>2</sup> əT-s-<sup>2</sup> éİməỷ
not last.long-3SUB ART your-NOM-be.waiting
<sup>2</sup> You won't have long to wait.' (lit. 'Your waiting will not be long.')

It can be followed by a subordinate clause, as in (h).

(h) wəlíθ łáġəlləx wen k wei (JP)
 wəl-híθ łáġəl-ləx wei (kwei (P))
 already-last.long know-tR-I that
 'I've known him for a long time.'

It can appear as a predicate in a pair of coordinate clauses, as in (i) and (j).

- (i) híθ i yeł sis təwkwámkwəmdət. (JP 24)
   híθ <sup>?</sup>əỷ yéł s-ni<sup>?</sup>-s təw-kwámkwəmdət
   last.long and now NOM-AUX-3POS somewhat-strong-self
   'It took some time before he became strong.'
- (j) <sup>9</sup>ówə ni·s yəhíθ i ni x<sup>w</sup>ə<sup>9</sup>ámət təs tə léləms. (JP) ?∕awə ni?-əs və-híθ ?∍ỷ ni? not AUX-3SUB along-last.long and AUX x \*\*-?ámet lélam-s tás [?ə] tə become-be.sitting arrive.there OBL house-3POS ART 'It didn't take long for him to get home to his house.'

It can also appear in an adverbial adjunct (§3.6), as in (k).

(k) sk<sup>w</sup>éý k<sup>w</sup>sňes wə<sup>2</sup>ánəx<sup>w</sup> k<sup>w</sup>ə híθ. (JP 22) sk<sup>w</sup>éý k<sup>w</sup>ə s-ňe-s wə-<sup>2</sup>ánəx<sup>w</sup> k<sup>w</sup>ə impossible ART NOM-be.again-3POS EST-stop ART híθ last.long
'They could not stop again for long.'

In (l), the adverbial adjunct appears to stand as a sentence by itself. Compare (m), which is all one clause.

| <ol> <li>k<sup>w</sup>əwłíθəł ctwa?. wəłk<sup>w</sup>ín ctwa? k<sup>w</sup>s nécowors méqe?. (JP 1</li> </ol> |                  |                        |       |                       |       |  |  |
|---------------------------------------------------------------------------------------------------------------|------------------|------------------------|-------|-----------------------|-------|--|--|
|                                                                                                               | <b>k</b> ™ə      | wəł-híθ-əł             | ċtwa? | wəł-k <sup>w</sup> íń | ċtwa? |  |  |
|                                                                                                               | ART              | already-last.long-past | SPEC  | already-how.many      | SPEC  |  |  |
|                                                                                                               | k <sup>w</sup> ə | s-néčəwəc-s            | méqe? |                       |       |  |  |
|                                                                                                               | ART              | NOM-hundred-3POS       | snow  |                       |       |  |  |
|                                                                                                               |                  |                        |       |                       | -     |  |  |

'It must have been long ago. It must have been hundreds of years ago.'

(m)k<sup>w</sup>əwłíθəł ctwa? k<sup>w</sup>sx<sup>w</sup>ənsx<sup>w</sup>əyemsəł cə ?al k<sup>w</sup>əw mək<sup>w</sup> stem ?i <sup>?</sup>∂ tən táməx<sup>w</sup> ... (JP 6) k̃™⊋ wəł-hí0-əł ctwa? k <sup>w</sup>ə s-x ">n-s x "> y ć m-s->ł already-last.long-past ART SPEC NOM-still-myth-3POS-past ART 2a1 k<sup>™</sup>ə wa-mak<sup>w</sup> 9 ?ə təna ćə stem OUOT iust ART EST-all what be here OBL this támaxw earth 'Long ago, when everything was still myth on this earth ...' (i.e., before the coming of  $\check{X}\acute{e}\cdot \dot{l}s$ )

#### 18.4.10. nóća? swéyəl 'one day' and so on

This and other measures of time can be followed by nominalized clauses, as in (a) to (d).

(a) nóčos łgélč k<sup>w</sup>sos ní<sup>9</sup> k<sup>w</sup>so notén. (CC 21) náčas łaélċ k <sup>w</sup>sə s-ni?-s ní? k<sup>w</sup>sə one face moon NOM-AUX-3POS be there ART ART nə-tén my-mother 'My late mother was there for one month.' (lit. 'My late mother's being there was one month' or 'It was one month that my late mother was there.') (b) xă²áθən nétəł xă²áθən x wəné?ent k ws xəxələts. (CC 10) xa?áθən xa?áθən nétəł x<sup>w</sup>ané?ent k <sup>w</sup>ə four morning four evening ART s-žəžəl-t-s NOM-be.painting-TR-3POS 'She paints him four mornings and four evenings.' (c) wəłkwín syəlánəm kwə nəs?ówe·ł cyówən. (CC) wəł-k™ín k™ə syəlánəm nə-s-?áwə-əł already-how.many vear my-NOM-not-past ART [ł] c-yáwan do-possessing.song ART 'For a number of years I have not sung.' (d) niwəł <sup>9</sup>ápən syəlánəm k<sup>w</sup>ənsni·ł k<sup>w</sup>ə́cnəx<sup>w</sup>. (DK) ni? syəlánəm wəł-?ápən nə-s-ni?-əł k <sup>w</sup>ə already-ten AUX years ART my-NOM-AUX-past k<sup>™</sup>éc-nex™ look-TR 'It was ten years ago that I saw him.'

They can serve as the first of a pair of coordinate clauses, as in (e).

(e) isélə ?al swéyəl i mi x \*ə?ám t. (CC 13)
 yəsélə ?al swéyəl ?əỷ mi x \*ə-?ám t
 two just day and AUX(come) become-be.sitting 'In two days they were home.'

And they can appear in verbal adjuncts, as in (f).

(f) səw x<sup>w</sup>q<sup>w</sup>əlq<sup>w</sup>éləwən tə si<sup>2</sup>ém i təna nəča<sup>2</sup> swéyəl. (JP 22) s-wə-x<sup>w</sup>q<sup>w</sup>əlq<sup>w</sup>éləwən[-s] si?ém ?i tə [?ə] təna NOM-EST-be.thinking-3POS ART leader be here OBL this n áca? swéyəl one dav

'Then one day the headman was thinking.' (lit. 'Then the headman was thinking in this one day,' the 'in' being expressed with *?i* 'be here')

(For other examples, see §5.2.1, [a] and [b]. For measures of time, see §22.2.)

#### 18.4.11. *ťát* 'long before, from long ago, earlier'

This usually appears with  $wa_{2}$  'established.' It can appear before the predicate head, as in (a) and (b).

| (a) | wətat wəh                            | á?k¤əx  | əs tə x <sup>w</sup> n                   | náθk™əỷəm   | k™θ∋ st    | e <sup>γ</sup> é <sup>γ</sup> e tθe <sup>γ</sup> . (JP) |
|-----|--------------------------------------|---------|------------------------------------------|-------------|------------|---------------------------------------------------------|
|     | wə-łát                               |         | wə-há <sup>9</sup> k <sup>w</sup> -əx-əs |             | tə         | x <sup>w</sup> m ó0 k <sup>w</sup> o'y om               |
|     | EST-long.before                      |         | EST-get.used-TR-3TR                      |             | ART        | Musqueam                                                |
|     | k <sup>w</sup> θə stə <sup>γ</sup> é |         | $^{9}$                                   | tθé?        |            |                                                         |
|     | ART                                  | like    | OBL                                      | that        |            |                                                         |
|     | The Muser                            | looma h | ave haan                                 | uning comot | aina lilea | that from long ago                                      |

'The Musqueams have been using something like that from long ago.'

| (b) | wətát ?ə  | čx <sup>w</sup> wəh | lék™nəs | k <sup>w</sup> θə s | ste?é ?e tθe?. (JP)      |      |       |
|-----|-----------|---------------------|---------|---------------------|--------------------------|------|-------|
|     | wə-łát    |                     | $^{9}$  | čx <sup>w</sup>     | wə-hék <sup>w</sup> -nəs | k™θә | stə?é |
|     | EST-long. | before              | ROG     | you                 | EST-remember-TR          | ART  | like  |
|     | ?ə        | tθé?                |         |                     |                          |      |       |
|     | OBL       | that                |         |                     |                          |      |       |
|     | 'Do vou   | emember             | anythin | a like t            | hat from before?         |      |       |

'Do you remember anything like that from before?

In (c), it appears to modify a noun.

.

(c) k \*θ = w = ·tát - əłm = stáy = x \* (JP 1)ARTEST-long.before-pastpeople'the old-time people'

Alternatively, one might identify the phrase as a relative clause, 'the ones who were people long before,' 'people' being a predicate head.

# 18.4.12. $q \dot{e}^{2} i s$ (probably // $q \dot{e} \dot{y} \dot{a} s$ //) 'recently, just a short time ago, formerly'

This can appear before the auxiliary, as in (a) and (b), and in (c), which is a relative clause.

- (a) gé<sup>9</sup>is cən ni təlləx<sup>w</sup> ti<sup>9</sup>i. (JP) tál-ləx <sup>w</sup> ti?í qéỷəs cən ni? be understood-TR recently this I AUX 'I just found that out now.' (b) qé?is nə x wək wənátəl ?é.łtən. (JP) aévəs ni? x <sup>w</sup>ə-k <sup>w</sup>ən-é-təl ?é·łtən recently AUX become-get.taken-DUR-RECIP 3PL 'They recently got together.' (c) tə qe<sup>9</sup>is <sup>9</sup>i·n k<sup>w</sup>ə́nnəx<sup>w</sup> sq<sup>w</sup>əméy (JP)
- tə qeyəs <sup>9</sup>i-ən k<sup>w</sup>ənrəx<sup>w</sup> sq<sup>w</sup>ənrey (JF) tə qeyəs <sup>9</sup>i-ən k<sup>w</sup>ən-nəx<sup>w</sup> sq<sup>w</sup>əmey ART recently AUX-I get.taken-TR dog 'the dog I've just acquired'

The phrase  $t \partial n a q d^{2} is$ , translated 'recently,' may follow the predicate head, as in (d).

(d) nəwł ste<sup>γ</sup>é k<sup>w</sup>əw x<sup>w</sup>csí·m <sup>γ</sup>al təna qé<sup>γ</sup>is i ni θəł wəxáca<sup>γ</sup>. (JP 13) wał-sta?é [?ə] k<sup>™</sup>a wə-x<sup>w</sup>-cis-í·m ni? AUX already-like OBL ART EST-inward(?)-grow-die.from ?aĺ [?ə] təna aévəs ?∂ỷ ni? θəł wə-xáca? just OBL this short.time.ago and EST-lake AUX ADV 'It's become sort of grown-in recently, but there's still a lake there.'

The phrase  $w \partial q \dot{e}^{2} is 2 a \dot{l}$  for a short time' can appear as a predicate in a simple sentence, as in (e).

(e) wəqé<sup>9</sup>is cən ce<sup>9</sup> <sup>9</sup>al. (JP)
 wə-qéyəs cən ce<sup>9</sup> <sup>9</sup>al
 EST-recently I FUT just
 'I'll be right back. I'll be only a moment.'

Or it can appear as an adverbial phrase following a predicate, as in (f) and (g).

(f) <sup>9</sup>élməý čx<sup>w</sup> k<sup>w</sup>əw qe<sup>9</sup>is <sup>9</sup>al. (JP)
 <sup>9</sup>élməý čx<sup>w</sup> k<sup>w</sup>ə wə-qeýəs <sup>9</sup>al
 wait you ART EST-recently just
 <sup>6</sup>Wait a little while.<sup>2</sup>

(g)  $x^{w}$  and  $k^{w}$  and  $ta^{\gamma}$  all ta  $\gamma$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^{\gamma}$  and  $ta^$ x<sup>w</sup>ən-k<sup>w</sup>ən-é-t čx w 2aĺ tə <sup>9</sup>əT-sq<sup>w</sup>əmq<sup>w</sup>əméy still-be.taken-DUR-TR iust vour-dogs vou ART wə-qéyəs ?aĺ EST-recently iust 'Just hold your dogs for a moment.'

But a great difference in meaning is seen in (h), where the  $w\partial t$ - 'already' implies past and the particle  $\dot{c}twa^2$  may imply that the statement is based on tradition.

(h) wəłqé<sup>9</sup>is <sup>9</sup>ə ctwa<sup>9</sup> (JP 13)
 wəł-qéýəs <sup>9</sup>ə ctwa<sup>9</sup>
 already-formerly ROG SPEC
 'It must have been long ago.'

#### 18.4.13. yét 'only now, only then'

This is followed by a nominalized verb. As the head of a single clause, it has the sense 'just now, only now,' as in (a). (For other examples see §4.3.1 [a] to [e].)

(a) wəyéł sis técəl. (CC) wə-yéł s-<sup>9</sup>i-s técəl EST-only.now NOM-AUX-3POS arrive.here 'He's just arrived.'

As the head of the second of a pair of coordinate clauses, it is often translated 'before' with the sense 'after which, only then,' as in (b) and (c).

(b) təwhíθ ċə i yéł sis ném. (JP)
 təw-híθ ċə 'əỷ yéł s-ni'-s ném somewhat-last.long QUOT and only.then NOM-AUX-3POS go 'It took a while before he went.'

In (c), the first *yét* appears to have the first sense and the second *yét* the second sense.

(c) yéł s?ict stəté?e. táx<sup>w</sup> snét ?i yéł səs hənłélt k<sup>w</sup>θə x<sup>w</sup>óýq<sup>w</sup>ələł. (AG) stəté?e s-<sup>9</sup>i-ct táxw ?i yéł snét precisely only.now NOM.AUX.our stay(PL) night and hən-łél-t s-?i-s k»θə véł only.then NOM-AUX-3POS come-move.shoreward-STAT ART x ȇý q »alał steamer

'There we were waiting. It was midnight before the ship came in.'

#### 18.4.14. wéyəl 'tomorrow' and so on

Two words, wéyəl 'be day' and nét 'be night,' are the basis of several terms for times of day and night, such as swéyəl 'day' (also 'daylight, sky'), təna wéyəl 'today,' yəwéwəyəl 'dawn' (lit. 'becoming daylight'), nétəł 'morning' (lit. 'night-past'), x<sup>w</sup>əné<sup>2</sup>ənt 'evening' (lit. 'becoming night'), and so on. These are considered in §22.2.1. Here I will simply review the roles of these words in syntax.

As illustrated in 4.2.4 (a), *wéyəl ce*<sup>2</sup> (lit. 'it will be day') with the sense 'tomorrow' can appear as a main clause referring to the future with the qualified predicate as an accompanying subordinate clause. This is seen also in (a).

| (a) | wéyəl         | ce?     | wə-łák*·   | -əx <sup>w</sup> | wə-ni    | i?-əs    | ce?     | neṁ      |
|-----|---------------|---------|------------|------------------|----------|----------|---------|----------|
|     | become.day    | fut     | may-go.h   | iome-you         | when-    | AUX-38   | SUB FUT | go       |
|     | ?áwk™         | tə      | spíŵ       | ni               | ?ə       | tə       | stáləŵ  | . (CC 7) |
|     | be.exhausted  | ART     | ice        | be.there         | OBL      | ART      | river   |          |
|     | 'Tomorrow you | will go | o home, if | the ice in       | the rive | r is all | gone.'  |          |

As illustrated in §4.2.4 (b), *wəwéyələs* (lit. 'when it is day') with the sense 'tomorrow' can appear as a subordinate clause with the qualified predicate as the main clause. This is seen also in (b).

(b) ?óý k\*sowł némct x\*íwol woweyolos. (CC 1)
 ?óý k\*o s-woł-ném-ct x\*íwol wo-weyol-os good ART NOM-already-go-our move.upstream when-day-3SUB
 'We'd better start going upriver tomorrow.'

The temporal modifier can also be the first of two coordinate clauses, as in (c) and (d).

- (c) ni cowł yowewoyol i yeł sis łé l ?é łton. (JP 14) wəł-yə-wéwəyəl ni? ćə ?∂ỷ yeł already-first-becoming.day and only.then AUX OUOT ?é·łtən s-?i-s łé·l NOM-AUX-3POS go.ashore 3PL 'It was early in the morning when they went ashore.'
- (d) x wənnétəł i wəł?ómət tə Dave. (CC 6)
  x wən-nét-əł ?əỷ wəł-?ómət tə Dave still-night-past and already-get.up ART Dave 'Early in the morning Dave got up.'

These words referring to time also appear in adverbial adjuncts, as in (e).

(e) ni-ł cən wəłáy qwáqwəl kw xwənént ... (JP 8)
ni?-əł cən wəł-háy qwáqwəl kwə xwəné?ənt
AUX-past I already-finish be.speaking ART evening
'I have already spoken last night ...'

Or they may follow a prepositional verb (§3.5), as in (f).

(f)  $\lambda a ni^{9}e \cdot p k^{*} \Rightarrow cn \Rightarrow x^{*} i t \Rightarrow na wéyəl t \theta e^{9} ni \cdot n ne^{9} \notin m \Rightarrow st \Rightarrow x^{*}$ . (JP 26) х́а ni<sup>?</sup>-e·p k<sup>™</sup>ác-nəx<sup>™</sup> ?i təna wévəl AUX-vou(PL) look-TR be here this become.day BE3P  $t\theta e^{2}$ ni?-ən ném-əstəx<sup>w</sup> AUX-I that go-CAUS 'What I've talked about is what you have seen today.'

#### 18.4.15. Xe (CC, JP), Xe<sup>9</sup> (AG) 'again, also, too'

This tends to join with  $wa_2$  'established' as  $\lambda aw$  and with wat 'already' as  $\lambda awt$ . It can appear initially before a predicate head, as in (a) ...

(a) Žəw si·?ém. (JP 16)
Žə wə-si·?ém
also EST-honoured.persons
'They were also important people.'

... before an auxiliary, as in (b), (c), and (d) ...

- (b) Χ c cən ce? niw ste?é. (JP)
   Χ cən ce? ni? wə-stə?é
   also I FUT AUX EST-like
   'I'll be the same.'
- (c) ni <sup>?</sup>ə Ẋ̀əw ném. (CC) ni<sup>?</sup> <sup>?</sup>ə Ẋ̀ə wə-ném. AUX ROG also EST-go 'Did he go too?'
- (d) səw ?ówəs Åe·s nəw k<sup>w</sup>ócnəx<sup>w</sup>. (JP2) s-wə-?ówə-s Åe-əs ni? wə-k<sup>w</sup>óc-nəx<sup>w</sup> NOM-EST-not-3POS again-3SUB AUX EST-look-TR 'Then they did not see them again.'

... after an auxiliary, as in (e) ...

(e) <sup>?</sup>i ni <sup>\*</sup>⁄<sub>λ</sub> awł qalét qá<sup>?</sup>qa<sup>?</sup>. (CC 21)
<sup>?</sup>oỷ ni<sup>?</sup> <sup>\*</sup>⁄<sub>λ</sub> e wał-qalét qá<sup>?</sup>qa<sup>?</sup>
and AUX again already-do.again drink
<sup>°</sup>And he would drink again.<sup>°</sup>

... or after the predicate head, as in (f).

(f) səsəw kwəllextəs Åe. (CC 3) s-ni?-s wə-kwəllex-t-əs Åe NOM-AUX-3POS EST-shoot-TR-3TR also 'Then he shot them too.' **18.4.16.** *qəlét* 'do again, again,' or with a negative '(no) more, (no) longer' This can appear as a predicate head, as in (a) and (b) ...

(a) qəlét ?ə čx<sup>w</sup>. (JP)
 do.again ROG you
 'Will you have more?'

... before the head, as in (b) and (c) ...

- (b) qəlét nə<sup>γ</sup>éməstx wəs kwθə syówəns. (JP) qəlét ném-əstəx w-əs kwθə syówən-s again go-CAUS-3TR ART possessing.song-3POS 'Again he sang his song.'
- (c) nem čx<sup>w</sup> qəlét x<sup>w</sup>ə-<sup>9</sup>əx<sup>w</sup>ín, nə-sqéxə. (JP 3)
   AUX(go) you again become-small my-dog
   'Become small again, my dog.' (not the usual word for 'dog' in this magical command)

... or after the head, as in (d) and (e).

- (d) səw xák<sup>w</sup>əms qəlét. (CC 10) s-wə-xák<sup>w</sup>-əm-s qəlét NOM-EST-get.bathed-INTR-3POS again 'Then he bathes again.'
- (e) ?ówo con ?i·n dádoỷ qolét. (CC)
   ?ówo con ?i-on dádoỷ qolét
   not I AUX-I sick longer
   'I am no longer sick.'

It can appear together with  $\vec{\lambda} \vec{e}$  'again,' as in (f).

(f) k<sup>w</sup>səwł <sup>2</sup>ík<sup>w</sup> tə  $\check{x}\acute{a}^{?}\check{x}\acute{t}^{\theta}$ əltən, <sup>2</sup>ówəte<sup>?</sup>  $\check{x}$  gəlét s<sup>\*</sup>xéləgəm šx<sup>w</sup>né<sup>?</sup>em. (CC 11) k "a s-wəł-?íkw tə žá<sup>γ</sup>žť<sup>θ</sup>àltən ż ?áwate? NOM-already-be.lost Pierre be not ART ART ART qəlét sžéləqəm šx<sup>w</sup>né<sup>9</sup>em again powerful shaman 'After Pierre died, there was no longer any powerful shaman.'

# 18.4.17. ha? 'if, when'

This appears exclusively as the first word in a predicate that is usually a conditional clause in a compound sentence (see  $\S5.2.3$ ), as in (a), (b), and (c).

(a) ha? cən ce? ? syqt qəlét i x vədádəy. (CC 20) ha? cən ce? ?áyq-t qəlét ?ąγ́ x<sup>w</sup>ə-dádəy if be.out.of.way-TR become-sick I FUT again and 'If I put it [his song] aside again, he'll become sick.'
(b) ha? ce·p ce? ?ówo háye·p i ném con gídotalo. (CC 21) ha? ce·p ce? ?áwa háy-e·p ?∍ỷ ném cən if vou not stop-vou and AUX(go) I FUT qíq-t-àlə be.bound-TR-you 'If you don't stop, I'll have you arrested.' (lit. 'I'm going to arrest you')

(c) ha·ł cən <sup>9</sup>ówə ni·n ném əy <sup>9</sup>ówe·ł cən ni·n k<sup>w</sup>ócnəx<sup>w</sup>. (JP) ni?-ən ha?-əł cən ?∕awə ném ?∍ỷ ?áwə-əł cən if-past I not AUX-I go and not-past I ni?-ən k<sup>™</sup>ác-n∍x<sup>™</sup> AUX-I look-TR 'If I had not gone, I would not have seen him.'

In (d), the conditional clause is followed by what could stand as a compound sentence.

(d) ha? cən ném i cəx "lé? i k "ácnəx" cən. (CC) ném ?∂ỷ ċəx™lé? ?av≀ ha? cən k<sup>™</sup>ác-n∍x<sup>™</sup> cən when I go and usuallv and look-tr I 'When I go, I generally see him.'

See also §5.2.3 (a) and (b).

#### 18.4.18. $m \delta \vec{k}^w$ 'whenever'

This is the gloss that can be given  $m \delta \vec{k}^w$  'all' when it appears first in a predicate, followed by a nominalized clause, the whole construction standing as the first of two coordinate clauses, as in (a).

(a) môk nosném îímox i wok vôcnox (JP)
môk no-s-ném îímox ?oý wo-k vôc-nox all my-NOM-go walk and EST-look-TR
Whenever I go about, I see him.'

A more literal translation of this sentence might be 'It is my every going walking when I see him.' (See also §5.2.3 [d], and for other uses of  $m \delta \vec{k}^w$ , see §18.4.40.)

#### 18.4.19. xwoňá? 'first, when first, as soon as'

When followed by al' just' in a single clause, this has the sense 'first,' as in (a) and (b).

(a)  $x^* = n\dot{a} \cdot l \dot{c}twa^{\gamma} \cdot a\dot{l} k^* ik^* = xt = m \dot{k}^* T.$  (JP 23) ćtwa? ?aĺ k<sup>w</sup>ík<sup>w</sup>əx-t-əm k<sup>™</sup>a x<sup>w</sup>əná?-əł [?ə] Т being.named-TR-INTR first-past SPEC just OBL ART Т 'He was the first to be named T'

(b) tə smént xwəảá ?al ni wélxəs (JP 6)
 tə smént xwəảá? ?al ni? wél-x-əs
 ART rock first just AUX throw-TR-3SUB
 'the rock he threw first, the first rock he threw'

Without 2al, in the first of two coordinate clauses, it has the sense 'when first,' as in (c) and (d).

- (c) x<sup>w</sup>əňá<sup>?</sup> cən <sup>?</sup>i técəl i ni<sup>?</sup> cən wəłk<sup>w</sup>ácnəx<sup>w</sup> k<sup>w</sup>θəwÅa. (CC) x "əná? cən ?i técəl ?∂ỷ ni? cən first T AUX arrive here and T AUX wəł-k<sup>w</sup>éc-nəx<sup>w</sup> k<sup>w</sup>θaw<sup>\*</sup>λa already-look-TR him 'When I first came here, I saw him.'
- (d) x<sup>w</sup>əňá? mi téwəl i wəł?émət tə swá·ŵləs tə nem ce? yə?émməx. (CC 6) x<sup>w</sup>əná? mi téwəl ?əỷ wəł-?śmət tə first AUX(come) become.light already-get.up and ART swá∙ŵləs nem ce? yə-?ómməx tə bovs ART AUX(20) FUT along-be.hunting 'When it first became light, the boys, the one who would be going hunting, got up.'

With  ${}^{2}al$ , in a simple clause followed by a nominalized clause introduced by *sowl*, it has the sense 'as soon as, the moment that,' as in (e).

(e)  $x^{w}$ ənda? cə ?al  $x^{w}$ (sə $\theta$ ət, səwł ma·l c(səm. (JP 3) x "əná? ćə ?aĺ x<sup>w</sup>ís-əθət. s-wəł-mi ?aĺ first OUOT iust shake-self NOM-already-AUX(come) iust cís-əm grow-INTR 'The moment it shook itself, it began to grow.'

It also occurs with the sense 'when first' within nominalized clauses introduced with  $k^{w}\partial$  (cf. §4.3.2 [7]), as in (f) and (g).

(f) k<sup>w</sup>sx<sup>w</sup>əná?səł ctwa? técəl tə x<sup>w</sup>ənítəm, ?ówe·ł te? stíqìw. (CC) k »ə s-x »əná?-s-əł ćtwa? técəl x <sup>w</sup>ənítəm. tə arrive here NOM-first-3POS-past SPEC ART ART European ?áwe-əł te? stíqìw horse not-past appear 'When the White people first came, there were no horses.'

 (g) k<sup>w</sup>əθni<sup>9</sup> x<sup>w</sup>əňá <sup>9</sup>al k<sup>w</sup>əcnəx<sup>w</sup> ... (JP 3) k<sup>w</sup>ə <sup>9</sup>əT-s-ni<sup>9</sup> x<sup>w</sup>əňá<sup>9</sup> <sup>9</sup>al k<sup>w</sup>əc-nəx<sup>w</sup> ART your-NOM-AUX first just look-TR 'When you first saw it ...' It can also appear alone as a noun modifier, as in (h).

 (h) k<sup>w</sup>θθ x<sup>w</sup>θňá?ał x<sup>w</sup>θnítəm (AG) k<sup>w</sup>θθ x<sup>w</sup>θňá?-θł x<sup>w</sup>θnítəm ART first-past European 'the first White people'

#### 

This has the meaning 'before, ahead, go first' in both spatial and temporal senses, the second no doubt stemming from the first. It can appear as a predicate head in a simple sentence, as in (a), where the sense is both, and (b), where it might be either.

- (a) ném łe yəwén. (CC) go PER first 'Go ahead.' (or 'Go first.')
- (b) ni? yəŵéň. (CC)
  AUX go.before
  'He's ahead.' (or 'He went first.')

It can precede a nominalized clause, as in (c), where it has a temporal sense.

| (c) | ) yəẁén c'ə kʷs mis p'ə́kʷ kʷθə sítənsəł cəmé?təs. (JP 16)       |      |                  |                    |                 |                   |  |
|-----|------------------------------------------------------------------|------|------------------|--------------------|-----------------|-------------------|--|
|     | yəwen cə kwə                                                     |      | k <sup>w</sup> ə | s-mi-s             | ởák™            | k <sup>w</sup> θə |  |
|     | first                                                            | QUOT | ART              | NOM-AUX(come)-3POS | come.to.surface | ART               |  |
|     | sítən-s-əł cəm-é <sup>9</sup> -t-əs                              |      |                  |                    |                 |                   |  |
|     | basket-3POS-past carry.on.back-DUR-TR-3SUB                       |      |                  |                    |                 |                   |  |
|     | 'First the basket she carried on her back comes to the surface.' |      |                  |                    |                 |                   |  |

It may also appear within a clause before the head, as in (d), where the context, a myth, indicated that the sense is temporal.

(d) ?śwəte? k<sup>w</sup>θə ni·ł yəwéń neṁ csétəm. (JP 10)
?śwəte? k<sup>w</sup>θə ni?-əł yəwéň neṁ csé-t-əm absent ART AUX-past before AUX(go) commission-TR-INTR
'There was no sign of the one who had been sent before.' (lit. 'The one who had been sent before was none.')

In (e), it bears the suffix  $-\partial t$  'past,' which makes the reference clearly temporal.

(e) k<sup>w</sup>łə yəwén di di kwən atəls tθe? (JP)
 k<sup>w</sup>łə yəwén di di kwən di di kwən e-təl s
 tθe?
 ART earlier-past fellow-get.taken-DUR-RECIP-3POS that
 'the former wife of that person'

There is also a nominalization, *syowéň* 'ancestors, heritage.' (See also §22.1.8.4.)

#### 18.4.21. *łi?á?aq\*t* (probably //łəỷá?əq\*t//) 'after'

This has the meaning 'after, later, behind, next' in both temporal and spatial senses. It is possibly a resultative form composed of  $\sqrt{te\dot{y}}$  'degrade' (cf.  $t\dot{e}^{2}it$  //téy³t// 'put him down,' teystánmat 'feel cheap'),  $-aq^{w}$  'head,' and -t 'stative.' (See also §22.8.1.4.) It can appear as the predicate head in a simple sentence, as in (a).

(a) ni łi?á?aq<sup>w</sup>t. (JP)
ni? łi?á?aq<sup>w</sup>t
AUX go.last
'He's behind [as one walking].'

Or it can appear before the predicate head either before an auxiliary, as in (b), where the predicate is a relative clause, or after an auxiliary, as in (c).

(b) k<sup>w</sup>θə smé·nt ni łi²á·q<sup>w</sup>t wélxəs (JP 6) k<sup>w</sup>θə smé·nt ni? łi²á²aq<sup>w</sup>t wél-x-əs ART rock AUX next throw-TR-3SUB 'the second rock that he threw, the rock that he threw next'

(c) k<sup>w</sup>θ<sub>θ</sub> łi<sup>2</sup>á<sup>2</sup>aq<sup>w</sup>t ni yá·ý<sub>θ</sub>s k<sup>w</sup><sub>θ</sub> sté<sup>. 2</sup>e tθe<sup>2</sup> s<sup>\*</sup>λ<sup>2</sup>án<sub>θ</sub>q (JP 16) k<sup>w</sup>θə łi?á?aq<sup>w</sup>t ni? yá·ỷəs [?ə] ?a k »a sta?é ART last AUX be,working OBL ART like OBL t0e? sẩánəq potlatch that 'the last one who did such a thing as a potlatch'

# 18.4.22. *tá*x<sup>w</sup> (CC, JP, DK), *tá*x<sup>w</sup> (AG) 'occur later, later, follow (an example)'

This (cf. the possible homophone in §18.4.27) can appear as the predicate of a simple sentence, as in (a) and (b).

- (a) táx<sup>w</sup> ce?. (CC, DK)
   occur.later FUT
   'It will occur later.'
- (b) niw táx<sup>w</sup> ?al o to sqodá?s. (DK)
   ni? wo-táx<sup>w</sup> ?al ?o to sqodá?-s
   AUX EST-follow.example just OBL ART accompanying-3POS
   'He was only following his companions.'

Or it can precede a predicate head, as in (c).

(c) táx<sup>w</sup> čx<sup>w</sup> ?al xté?-stəx<sup>w</sup>. (CC) later you just do-COM 'Do it later.'

# 18.4.23. $x^{w} \partial n \partial \tilde{x} \partial \tilde{t} \partial^{2} e t \theta e^{2}$ 'meanwhile, in the course of that, after some time, finally'

This is in origin  $x^w \partial n \check{x} \partial t \partial e^{2}$  'be still doing that.' However, the shift in stress and optional omission of  $t\partial e^{2}$  'that' with retention of the oblique particle (hanging there without an "object") suggests that the literal meaning gets lost. It has been rendered 'all of a sudden' where suddenness was not implied by the context (the literal meaning of the English phrase having perhaps also been lost) or 'finally.'

- (a) x<sup>w</sup>ən ẍət́é<sup>9</sup>etθe iwəł q<sup>w</sup>él tə sá<sup>9</sup>səq<sup>w</sup>t ... (AC 1) x <sup>w</sup>ən- x źł ə ?a t0e? ?∂ỷ wəł-a<sup>w</sup>él sá?səg<sup>w</sup>t tə still-be.doing that and already-speak OBL junior ART 'All of a sudden the junior cousin spoke ...'
- (b) x<sup>w</sup>ən xəté<sup>γ</sup>e θə sténəy i wəłk<sup>w</sup>ə́cnəx<sup>w</sup>əs tə s<sup>γ</sup>ə́lyəs ... (CC 11) x "ən-xətə ?a θə słénəỷ <sup>9</sup>əỷ wəł-k<sup>w</sup>éc-nəx<sup>w</sup>-əs tə still-be.doing already-look-TR-3TR OBL ART woman and ART s?ályə-s vision-3POS 'Finally the woman saw her vision ...' (lit. 'The woman was still doing it

#### when she saw her vision.')

#### 18.4.24. nán 'too, very, very much'

This has been recorded as an adverb only, and nearly always with  $wa_{2}$  'established' preceding both it and the word it modifies (see also §3.7.1 [g] and [i]).

| (a) | a) wənˈánəw k̃ "ámˈk̃ "əmˈ θən'a spəhéls. (JP 24)<br>wə-n'án wə-k̃ "ámˈk̃ "əm' θən'a spəhéls |                    |           |                   |           |                                |  |  |
|-----|----------------------------------------------------------------------------------------------|--------------------|-----------|-------------------|-----------|--------------------------------|--|--|
|     | EST-very                                                                                     | ES ES              | T-strong  |                   | this      | wind                           |  |  |
|     | 'This w                                                                                      | ind is v           | very stro | ıg.'              |           |                                |  |  |
| (b) | ?áwə                                                                                         | čxw                | wə-nái    | 1-əx <sup>w</sup> | nem       | x <sup>w</sup> -cícəł. (JP 24) |  |  |
| . / | not                                                                                          | vou                | EST-too   | -vou              | AUX(20    | ) move.toward-high             |  |  |
|     | 'Don't                                                                                       | $\frac{1}{20}$ too | high.'    | J                 |           | )                              |  |  |
|     |                                                                                              |                    | ,         |                   |           |                                |  |  |
| (c) | wənán                                                                                        | čx <sup>w</sup> w  | ənəsží?   | (CC)              |           |                                |  |  |
|     | wə-nán                                                                                       | l                  | čxw       | wə-n              | ə-s-c-Xí  | [?                             |  |  |
|     | EST-very.much you EST-my-NOM-do-value                                                        |                    |           |                   |           |                                |  |  |
|     | 'I love y                                                                                    | you ver            | y much.   | ' (lit. 'Y        | ou are ve | ery much what I want.')        |  |  |
| (d) | wəłk™á                                                                                       | k™əv č             | Həł wər   | hán. (JP          | 9)        |                                |  |  |
| ()  | wəł-kw                                                                                       | ák™əv              |           | ćə                | θəł       | wə-nán                         |  |  |
|     | already-                                                                                     | be hun             | gering    | OUOT              | ADV       | EST-verv                       |  |  |
|     | anouay                                                                                       | 0.0.11m            | B         | 2001              |           | 201                            |  |  |

- (e) nánow sk<sup>w</sup>éy. (DK) nán wo-sk<sup>w</sup>éy very EST-impossible 'It's completely impossible.'
- (f) skéləqəm, wənanəw kwónkwən tə sxi?íws. (JP) skéləqəm wə-nan wə-kwónkwən tə sxi?íws fierce EST-very EST-contagious ART smallpox 'Smallpox is fiercely catching.'

#### 18.4.25. scécoń 'really, certainly'

This is a resultative form, literally 'straightened' (cf.  $\dot{c}\acute{e}n$  'lie straight,'  $\dot{c}\acute{e}\cdot nt$  'straighten it'). It usually occurs initially, as in (a) to (d).

- (a) sčéčəň wə-cák<sup>w</sup>. (CC) really EST-far 'It was really far.'
- (b) sčéčaň ce·p wa-?élaý sawwáýqe?. (JP 12) really you EST-good(PL) men
   'You're really good men.'
- (c) scécon wo-łéw. (JP 12) really EST-flee 'They were really escaping.'
- (d) scécon-oł wo-cowét k<sup>w</sup>θo no-mén-oł. (JP 23) really-past EST-smart ART my-father-past 'My father was really smart.'
- (e) scécon niw léxol to swéyol <sup>γ</sup>o kθo sx<sup>w</sup>ox<sup>w</sup>á<sup>γ</sup>as ni xónod. (AG) scécon ni wə-léžəl swéyəl °ə kθa tə really sky AUX EST-open.to.light ART OBL ART sx<sup>w</sup>əx<sup>w</sup>á<sup>9</sup>as ni xánəď thunder open.eyes AUX 'The sky really lit up when the thunder(bird) opened its eyes.'

It can occur following an auxiliary, as in (f).

(f) ha·ł qól swéyəl i ni·ł scéccon woqáy. (JP 17) ha?-əł qól swéyəl ?əỷ ni?-əł scéccon wo-qáy if-past bad day and AUX-past really EST-die 'If it had been a bad day, he would surely have died.'

#### 18.4.26. *θ »<sup>?</sup>ít* 'truly'

This is perhaps from  $\theta \delta y$  'come into existence' ('become real' ?) with the suffix *-t* 'stative' that appears in some adjectives. It occurs initially or after an auxiliary, and perhaps always with  $w \partial_{2}$  'established'.

- (a) wə-θə<sup>9</sup>ít k<sup>w</sup>án-nəx<sup>w</sup>-əs k<sup>w</sup>θə sq<sup>2</sup>áməl <sup>9</sup>isəlálwe<sup>9</sup>s. (JP 12) EST-truly get.taken-TR-3TR ART paddle two.paddles 'They truly did get two paddles.'
- (b) wə-θə?ít yəx<sup>w</sup> sλéləqəm. (JP)
   EST-truly INF dangerous.being
   'It truly must be a monster.'
- (c) niw θθ<sup>2</sup>ít čáwnax<sup>w</sup>. (JP 27) ni<sup>9</sup> wθθ<sup>2</sup>ít čew-nax<sup>w</sup> AUX EST-truly help-TR
   'It really helped him.'

It is used adjectivally in (d).

(d) ni cow scécoń wosté.  $\theta^{\gamma}$ ek wow  $\theta^{\gamma}$ it xés. (JP 22) ni? ćə wə-scécən wa-sta?é ?a k<sup>™</sup>a wa- $\theta$ a?ít EST-really AUX OUOT EST-like EST-real OBL ART xés sea lion 'It really looked like a real sea lion.'

# 18.4.27. $t \dot{a} \ddot{x}^w \sim t \dot{a} x^w$ (CC, JP), $t \dot{a} \ddot{x}^w$ (DK), $t \dot{a} x^w$ (AG) 'adjust, exactly, just'

This variation in recorded form is unusual. A seemingly related resultative form,  $st \partial t i \tilde{x}^w$  'true,' implies a base form  $t \partial \tilde{x}^w$ . For a possible homophone see §18.4.22. This appears as a predicate head in (a).

 (a) táx<sup>w</sup> čx<sup>w</sup> ce?. (CC) adjust you FUT 'You'll adjust.'

(This sentence was actually rendered 'You'll get used to him' and identified as something said to a newly married daughter.)

Used adverbially, it can precede an auxiliary, as in (b) and (c).

(b) táx v yəx cə ?al niw x vələqéł. (JP 22)
táx yəx cə ?al ni? wə-x və-ləqéł
exactly INF QUOT just AUX EST-become-be.accurate
'It must have just hit the mark.'

(c) tá<sup>..</sup> x<sup>w</sup> nəw ném nét i nə ct tás. (CC 3) táž<sup>w</sup> ni? wə-ném nét ?∂ỷ ni? ct EST-AUX(20) become.night exactly AUX and AUX we tás arrive there

'It had juuust become night when we got there.'

#### 18.4.28. cəlél 'nearly, about to'

This appears as the predicate of the first of a pair of coordinate clauses, as in (a) and (b).

- (a) cəlél čx<sup>w</sup> i ni nóq<sup>w</sup>. (JP)
   cəlél čx<sup>w</sup> ?əỷ ni? nóq<sup>w</sup>
   nearly you and AUX fall.asleep
   'You nearly fell asleep.'
- (b) cəlél i pédam k θa naq aprovention and bloom k θa na-q aprovention and bloom are my-apple-plant 'My apple tree is about to bloom.'

In (a) the pronoun subject appears in the first clause, but in (b) the noun subject appears in the second clause.

#### 18.4.29. cəlél <sup>9</sup>al 'barely'

This appears with nominalized clauses introduced with  $k^w$ , as in (a).

(a) cəlél ?al k<sup>w</sup>smis čičłémətəm tə sqəlícəməs. (JP 1)
cəlél ?al k<sup>w</sup>ə s-miss čičłém-ət-əm tə barely just ART NOM-AUX(come)-3POS hear-TR-INTR ART sqəlícəməs
sqəlícəməs
`Sqəlícəməs could barely be heard.'

-

#### 18.4.30. *x<sup>w</sup>éləq* 'nearly'

This can appear as the predicate head in the first clause of a compound sentence, as in (a).

(a) ni cən wəθə<sup>2</sup>ít nem x<sup>w</sup>éləq i x<sup>w</sup>əníns k<sup>w</sup>θə nin šx<sup>w</sup>dáy ... (JP 27) ni? wə-θə?ít nem cən x <sup>w</sup>éləq ?∍ỷ AUX Ι EST-trulv AUX(go) nearly and x<sup>w</sup>a-ní?-nas k™θə ni? nə-šx<sup>w</sup>-dáy become-be.there-GOAL my-OBLNOM-die ART AUX 'I truly nearly came to my death ...'

(For another example see §5.2.4 [g].)

There is an iterative form seen in (b), from a myth, said by Hemlock in a storm.

(b) <sup>9</sup>i cən wəłx "ślx" ələq i yéd. (JP 24)
<sup>9</sup>i cən wəł-x "ślx" ələq <sup>9</sup>əğ yéd AUX I already-getting.near and fall 'I'm almost falling now.'

#### 18.4.31. wəqedəl ?al 'barely'

This appears to be based on a root  $\dot{q}\dot{e}l$ , not identifiable. It can stand before a nominalized predicate head, as in (a).

(a) wədédəl cal sqwənáməts təwka qèyəplénəx vəy tə sdá?s ná?ənca?. (JP 14) wə-dédəl ?aĺ s-g<sup>w</sup>e-namət-s tawźa ćə EST-barely NOM-penetrate-manage-3POS that OUOT just sģá?-s ná?ənča? qèyəplénəx<sup>w</sup> ?∍ỷ tə Capilano and ART companion-3POS one.person 'Capilano and one companion barely got through.'

I have no other examples.

#### 18.4.32. sní?onł 'enough, just enough, permitting'

This is a resultative form (cf. nitot 'set a time for it, establish a plan for it'). It can occur as a simple predicate head, as in (a).

 (a) wá·lə ni·s wəsní?ənł. (JP 3) wá·lə ni?-əs wə-sní?ənł perhaps AUX-3SUB EST-enough 'I guess it's enough [payment].'

It can also introduce a nominalized clause with  $k^w$ , as in (b).

(b) niw sq<sup>w</sup>ələq<sup>w</sup>éstəm sní<sup>γ</sup>ənł <sup>γ</sup>al k<sup>w</sup>smis <sup>γ</sup>éλqəl k<sup>w</sup>θə słék<sup>w</sup>əms ?é·łtən. (JP 1) wə-sq<sup>w</sup>ələq<sup>w</sup>é-st-əm sní?ənł ?aĺ k <sup>w</sup>ə ni? AUX EST-hole.punched-CAUS-INTR permitting just ART ?éÅaəl s-mi-s k™θə słék<sup>w</sup>əm-s NOM-AUX(come)-3POS be.coming.out ART breath-3POS ?é·łtən 3PL 'They [two large boxes] had holes made in them just enough so their [the cousins'] breath could come out.'

# 18.4.33. $\dot{\lambda} \partial x^w l \dot{a}^2 a s$ ( ${}^2a \dot{l}$ ) 'worthless, of no account, no matter that, even if'

This can be used as a noun modifier with the sense 'common' with the implication of 'not much worth,' as in (a).

 (a) K<sup>w</sup>əw X<sup>\*</sup>əx<sup>w</sup>lá<sup>2</sup>as <sup>2</sup>al məst<sup>5</sup>yəx<sup>w</sup> (JP 14) K<sup>w</sup>ə wə-X<sup>\*</sup>əx<sup>w</sup>lá<sup>2</sup>as <sup>2</sup>al məst<sup>5</sup>yəx<sup>w</sup> ART EST-no.matter just person <sup>6</sup>any common people<sup>2</sup>

In this sense, it is the base of  $\hat{X} \partial x^w la^2 as i l \partial m$  'go to nothing, become worthless.'

It can appear as a predicate with a subordinate clause, the whole construction standing as the first of two coordinate clauses, as in (b) and (c).

(b) Åəx "lá?as ?al wə?í.s ?é.łtən i wəném cən. (CC) ?al Åəx™lá?as wa-?í-as ?é·łt∍n ?∂ử wə-ném if-be.here-3SUB iust no.matter 3PL and EST-go cən I 'Even if they stay, I'll go.'

Or it can appear as a predicate with a nominalized clause as a nominal adjunct. In (d), no conjunction was recorded, so perhaps what follows is a second sentence, although JP translated it as one.

(d) Åəx<sup>w</sup>lá<sup>2</sup>as k<sup>w</sup>smis wəθé<sup>2</sup>t snét, mi<sup>..</sup>?é·ỷ x<sup>w</sup>ə<sup>2</sup>ínsəs tə ni<sup>2</sup>əł šx<sup>w</sup>ni<sup>2</sup>s k<sup>w</sup>sni<sup>9</sup>s nem k<sup>w</sup>ənk<sup>w</sup>ən. (JP 12) ึ่∧əx <sup>w</sup>lá?as k "ə s-mi-s wa-θé?t snét ART NOM-AUX(come)-3POS no.matter EST-dark night mi və-?é·v x<sup>w</sup>ə-<sup>9</sup>í-nəs-əs tə become-be.here-GOAL-3TR along-be.continuing AUX(come) ART s-ni?-s ni?-əł šx<sup>w</sup>-ni<sup>?</sup>-s nem k "ə OBLNOM-be.there-3POS AUX-past ART NOM-AUX-3POS AUX(go) k<sup>w</sup>ənk<sup>w</sup>ə́n be.captured

'Even if it became a dark night, they kept going toward the place where they had been taken captive.' (perhaps more literally, 'No matter that it became a dark night. On they came toward what had been the place where they were taken captive.')

# 18.4.34. <sup>?</sup>*iwáw*́ə (CC), <sup>?</sup>*iwáw*́ə<sup>?</sup> ~ <sup>?</sup>*iwów*́wə (JP), <sup>?</sup>*iwáw*́a<sup>?</sup> (AG) 'maybe, perhaps'

(The first form listed above for JP may have been an error for  $2iw\dot{a}wa^2$ .) This term occurs initially only, as in (a) to (d).

- (a)  $^{9}$ iwáwa spé $^{9}$ e $\theta$  kwa ni kwłét. (CC) k̃™⊃ ?iwáwa spé<sup>9</sup>e $\theta$ ni? k<sup>™</sup>əł-ét black.bear Spill-TR perhaps ART AUX 'It might be a bear that spilled it.' (b) <sup>?</sup>iwáŵə<sup>?</sup> ćáď čx<sup>w</sup>. (JP) perhaps get.shot vou 'You might get shot.'
- (c) <sup>9</sup>iwáwa<sup>9</sup> ni ném. (AG) <sup>9</sup>iwáwa<sup>9</sup> ni<sup>9</sup> ném perhaps AUX go 'He may have gone.'
- (d) <sup>9</sup>iwóww. (JP)'Maybe. Possibly [but who knows?]'

After giving (d), JP added that some people say (e).

(e) <sup>9</sup>iwówwo me. perhaps certain

Presumably it means something like 'I believe it's possible.'

This term looks as if it may be a contraction of elements. In (a), it might be interpreted as being composed of the auxiliary 2i, the particle  $wa^2$  'presumptive' (§16.2.17), and wa- 'established.' This interpretation is hard to reconcile with the other examples, however.

**18.4.35.**  $w\dot{a}\dot{l}a$  (CC, JP, AG),  $w\dot{a}\cdot\dot{l}a$  (DK) 'probably, presumably, I guess' This functions in two different ways. When it stands initially, the predicate head has a subordinate subject suffix, as in (a).

(a) wála dísacan-e⋅n ce?. (JP)
 probably make.net-I FUT
 'I guess I'd better make a net.'

But when it appears after an auxiliary, there is no such suffix. The meaning appears to be identical. Compare (b) and (c), (d) and (e), and (f) and (g):

(b) wála spé?eθas kwa ni kwłét. (CC)
wála spé?eθas kwa ni? kwałet
probably black.bear-3SUB ART AUX spill-TR
'It was probably a bear that tipped it over.'

- (c) spé?eθ wál kwa ni kwłét. (AG)
  spé?eθ wál kwa ni? kwał-ét
  black.bear probably ART AUX spill-TR
  'I guess it was a bear that tipped it over.'
- (d) wálə ni?-əs-əł ném. (AG) probably AUX-3SUB-past go 'I guess he had gone.'
- (e) ni wála ném. (AG)
  ni? wála ném
  AUX probably go
  'I guess he must have gone.'
- (f) wál > ?i·n >i ?i?tət. (AG) wál > ?i-ən-əł ?í?tət probably AUX-I-past be.sleeping 'I guess I must have been sleeping.'
- (g) ?i·ł cən wálə ?í?tət. (AG)
  ?i-əł cən wálə ?í?tət
  AUX-past I probably be.sleeping
  'I guess I must have been sleeping.'

It can also appear after the predicate head, as in the relative clause at the end of (h).

| (h) | cəmໍe໌·mໍ cໍə tə sítən θəwໍλa qẻmໍəỷ tθí·mໍ wálə. (JP 15) |        |           |         |          |            |             |    |
|-----|-----------------------------------------------------------|--------|-----------|---------|----------|------------|-------------|----|
|     | cəṁé·ṁ                                                    |        | ćə        | [?ə]    | tə       | sítən      | θəwÅa       |    |
|     | be.carrying.on.one's                                      | .back  | QUOT      | OBL     | ART      | basket     | that        |    |
|     | ģéməỷ                                                     | ť⁰í∙ṁ  |           | v       | válə     |            |             |    |
|     | young.woman                                               | be.pic | king.bei  | rries p | resum    | ably       |             |    |
|     | 'That girl who was p                                      | oresum | hably pic | king be | erries c | arried a b | asket on he | er |
|     | back.'                                                    |        |           |         |          |            |             |    |

Perhaps walowide a has two different sources. The walowide a hat appears initially may contain  $2\delta wo$  'not,' which requires a subordinate subject suffix on the predicate head. It may be  $//2 \approx 2a^2a//$ , though this can appear as  $2\delta we'lo$  (see §16.2.18 [i]). The walowide a hat appears after an auxiliary or the predicate head may be composed of  $wa^2$  'presumptive' (§16.2.17) and  $2a^2a$  'emphatic interrogative.'

#### 18.4.36. háy 'specifically, uniquely, as for, more'

This is homophonous but probably not identical with háy 'stop, finish.' It has several uses. As an adverb, it seems to select the subject for emphasis or focus, as in (a).

(a) ha  $\check{c}x^w$  me wə<sup> $\gamma$ </sup>í <sup> $\gamma$ </sup>al i háy cən ce<sup> $\gamma$ </sup> wəném. (CC) ?al ha? čxw mə wə-?í ?∂ỷ háv cən if you EST-be.here just CERT and specifically I ce? wə-ném EST-go FUT 'Even if you stay here, I will go.'

Often  $h \dot{a} y$  appears in pairs indicating a contrast between two subjects, as in (b).

(b) háy wəłmímələ tə x<sup>w</sup>láməy i háy ?áwə tə qəwícən. (DK) háv wəł mímala tə x <sup>w</sup>lóm o v ?∍ỷ specifically alreadv be.winter.dancing ART Lummi and qəẁícən háy ?áwə tə specifically not ART Cowichan 'The Lummies are already dancing but the Cowichans are not.'

It can be a predicate head with a noun subject that is the subject of a following clause, as in (c).

(c) i háy θə sá<sup>2</sup>səq<sup>w</sup>t, k<sup>w</sup>ənətəs tə tewəł, səw ... (JP 1) , téwəł ?∍ỷ θə sá?səq<sup>w</sup>t k<sup>w</sup>án-ət-əs háy tə and specifically ART iunior get.taken-TR-3TR ART shell s-wa-NOM-EST-'But as for the younger, she took a shell and ...'

In (c), the second clause is a main clause, as indicated by the transitive subject marker -as. In other sentences, the second clause may be a relative clause, in which case we have a kind of cleft sentence (see §4.1.5.2). A clear example of this is (d).

(d) háy θə són ke? łélq-t tə səqé·n ... (AC 1) specifically ART senior dip-TR ART feather 'It was the older who dipped a feather ...'

In (e),  $h \dot{a} y$  precedes a noun head, which is followed by a subject that is a relative clause. Here it functions like  $\dot{\chi} \dot{a}$  in similar cleft sentences (§4.1.5.2).

(e) háy ce? s?áx<sup>w</sup>a? k<sup>w</sup>ə ném-ət c'óýx<sup>w</sup>-t. (CC 1) specifically FUT butter.clam ART go-we dry-TR
'It will be butter clams that we will dry [in contrast to others who will dry salmon].'

Preceding an adjective (as indicated in §3.7.1), *háy* gives a comparative (or superlative) sense, as in (f) and (g).

- (f) háy nə xwám kwsəs mi wədwiləm. (CC 1) háy ni? **ằ**™́γm k »ə s-?i-s Μi more AUX fast ART NOM-AUX-3POS AUX(come) wád<sup>w</sup>-iləm drift-move.toward 'It was faster when they came downstream.'
- (g) k<sup>w</sup>θe? háy ?al c'əcí·λ s?inəm (JP 23)
   that more just short shaft
   'that shortest [harpoon] shaft'

#### 18.4.37. wənáy 'only'

In this word, the wa-may be wa-2 'established,' but this is the only form recorded. It can appear as an adverb, as in (a), where the predicate head appears to be the final word.

(a) wənáy ct ce? ?al ?ślwəm. (DK) only we FUT just left.home
 'We'll be the only ones left at home.'

It has not often been recorded as such, however. It more often appears as a predicate head. In this function,  $w \partial n \dot{a} y$  can be simply followed by a noun subject, as in (b) and (c).

- (b) wənáy ?al k<sup>w</sup>θəw x<sup>w</sup>əlməx<sup>w</sup> ?al məstəyəx<sup>w</sup>. (JP 2) wənáy ?al k<sup>w</sup>θə wə-x<sup>w</sup>əlməx<sup>w</sup> ?al məstəyəx<sup>w</sup> only just ART EST-Indian just person 'There were only Indian people.'
- (c) wənáy ti<sup>?</sup>í nəsqaməl. (JP)
  wənáy tə<sup>?</sup>í nə-sqaməl
  only this my-paddle
  'This is the only paddle I have.' (lit. 'It is only this one that is my paddle.')

The noun subject can be followed by a relative clause, making a kind of cleft sentence (see §4.1.5.2), as in (d).

(d) wənáy tə swéyəl wə-θə?ít xéxe?. (JP 21) only ART daylight EST-truly holy
'Only the Daylight is truly holy.' (lit. 'It is only the Daylight that is truly holy.')

In (e), the subject is not a noun but a relative clause with a noun head, which is modified by another relative clause, to which a subordinate clause is attached.

- (e) wənáy ?al k wθe? ni šx wq wáltans ni? ax w cicłémat waniax w ?ímax ni <sup>9</sup>ə tə smé·nt. (JP) wənáv ?aĺ k™θe? šx<sup>w</sup>-a<sup>w</sup>él-tən-s ni? ni?-əx<sup>w</sup> OBLNOM-speak-INSTR-3POS only iust that AUX-you AUX ?ímax ni? ciclém-ət wə-ni?-əx<sup>w</sup> ?a tə hear-TR when-AUX-vou be.walking be.there OBL ART smé∙nt mountain
  - 'It's only that which is its voice that you hear when you are walking in the mountains.'

The word  $w \partial n \dot{a} y$  can also appear as a predicate head followed by a nominalized clause, as in (f).

(f) wəná. y mə nəsni? təlnəx k k sis wəlx vəx vətəs nəceləx. (CC 12) tál-nəx<sup>w</sup> k "ə s-?i-s wənáv mэ na-s-ni? understand-TR only CERT mv-NOM-AUX ART NOM-AUX-3POS wəł-x »ə-x »átəs [tə] nə-céləx already-become-numb ART mv-hand 'I realized it only when my hand became numb.'

This construction appears a few times in CC's texts.

More commonly,  $w \partial n \dot{a} y$  appears as a predicate followed by a nominalized clause introduced with  $k^w$ , as in (g).

| (g) | wənáy k   | <sup>w</sup> səw i | mict <sup>9</sup> al łéw. (JP 3) |      |      |
|-----|-----------|--------------------|----------------------------------|------|------|
|     | wənáy     | k <sup>w</sup> ə   | s-wə-mi-ct                       | ?aĺ  | łéŵ  |
|     | only      | ART                | NOM-EST-AUX(come)-our            | just | flee |
|     | 'The only | thing              | we can do is run away.'          |      |      |

This construction may serve as a kind of conditional clause in relation to a following coordinate clause, as in (h).

(h) wənáy k<sup>w</sup>s<sup>2</sup>á·θàmət i yéł <sup>2</sup>əθném. (JP 3)
wənáy k<sup>w</sup>ə s-<sup>2</sup>á-ət-Sam-ət <sup>2</sup>əỷ yéł
only ART NOM-call-TR-you(PAS)-SUBPAS and now
<sup>2</sup>əT-s-ném
your-NOM-go
'Only if you were invited, then you might go.' (lit. something like 'Your

being invited is the only thing and then [or after which] it is your going.')

In (i), no conjunction was recorded and so grammatically there are two sentences, yet in sense the first sentence (or clause) is subordinate to the second.

(i) wənáy cə kwsnis scécen wə? wkw təwha tê xtən. xwc alewtəs ? e tən. (JP 27) scécan wa-?áwkw wənáv ćə k »ə s-ni?-s EST-get.used.up only OUOT ART NOM-AUX-3POS really i<sup>0</sup>ážtən ?é·łt∍n tawźa x<sup>w</sup>-cálaw-t-as that poison inward-turn.over-TR-3TR 3PL

'Only when that poison was really used up, they turned it [the poison bag] inside out.'

The word  $w \partial n \dot{a} y$  can also appear as a predicate followed by a conditional clause, this whole standing as the first of two coordinate clauses (cf.  $\dot{\lambda} \partial x^w l \dot{a}^2 as$ , §18.4.33), as in (j).

(j) wənáy wəniəx<sup>w</sup> k<sup>w</sup>əném k<sup>w</sup>əw stém ?al s?5ltən ?əθ?éx<sup>w</sup>e?t i yéł <sup>γ</sup>əθwə<sup>γ</sup>áỷ <sup>γ</sup>al. (JP) wə-ni<sup>9</sup>-əx<sup>w</sup> k<sup>™</sup>ə wənáv k<sup>w</sup>ən-é-m [?ə] wa-stém only if-AUX-vou get.taken-DUR-INTR OBL ART EST-what ?al s?áłtan ?∂T-s-?éx<sup>w</sup>e?-t ?∂ỷ véł food your-NOM-give.food-TR then iust and ?əT-s-wə-?áv ?a1 your-NOM-EST-good just 'Only if you had something to eat to give him, were you safe.'

As a predicate,  $w \partial n \dot{a} y$  can also be followed by a relative clause alone, as in (k) and in the second sentence in (l).

- (k) wənáy ?al ni?-əx<sup>w</sup> tól-ləx<sup>w</sup>. (JP 28) only just AUX-you be.understood-TR
   'That's all you came to know.'
- (l) wənáy  $k^w st^{\theta} = \lambda^w i ltct$  tə ləpát i tə ləlá? $\theta = n$ . wənáy  $\lambda^w ha^2 k^w = xct$ k<sup>w</sup>θe<sup>?</sup> qá<sup>?</sup>. (JP) s-t<sup>θ</sup>əx<sup>w</sup>-wíl-t-ct wənáv k <sup>w</sup>ə ?∍ỷ tə ləpát NOM-get.washed-vessel-TR-our only ART ART cup and ləlá?0ən wənáy šx<sup>w</sup>-há<sup>9</sup>k<sup>w</sup>-əx-ct k<sup>w</sup>θe? tə aá?

ART dishes only OBLNOM-get.used-TR-our that water 'It is only when we wash our cups and dishes. That's the only thing we use that water for.'

#### 18.4.38. tx<sup>w</sup>áy 'only remaining'

This is probably  $tx^{w}$ - 'remaining' and háy 'specifically.' It can appear as an adverb, as in (a).

(a) tx \*\*áy
 cən ?al ?ámət. (DK)
 only.remaining I
 just be.sitting
 'I'm the only one at home.'

It can appear as a predicate head with a noun subject, as in (b).

(b) tx wáy ?al tə šx wq 5? əl əq w. (JP 4) only.remaining just ART juice 'Only the juice was left.'

Or it can appear with a pronoun (or demonstrative) subject followed by a relative clause, as in (c).

(c) wə-tx<sup>w</sup>áy ce<sup>?</sup> ?al təwka yə-?í?xəl. (JP 9) EST-only.remaining FUT just he along-be.paddling 'It will be only he [the man in the bow] who paddles.'

Or it can be followed by a relative clause only, as in (d).

(d) tx wáy q wáq wəlstamx. (DK)

tx\*áyq\*áq\*əl-stəx\*-amxonly.remainingbe.speaking-COM-me'He's the only one who still speaks to me.'

The word  $tx^w \dot{a}y$  can also appear in a verbal adjunct (see §3.5) as a kind of prepositional verb with the sense 'except for,' as in (e).

(e) scécon niw x<sup>w</sup>ómnoc k<sup>w</sup>θo smálovał x<sup>w</sup>ólmox<sup>w</sup> tx<sup>w</sup>áy k<sup>w</sup>oňa Paddy. (JP) sčéčən ni? wə-x<sup>w</sup>ámnəc k™θə s-máləy-a?ł EST-become.extinct really NOM-Mali-ATT AUX ART x <sup>w</sup>ə́lməx <sup>w</sup> tx<sup>w</sup>áy k<sup>w</sup>əna Paddv people except that Paddv 'The Mali people are all gone except for Paddy.'

As a verbal adjunct, however, this is exceptional in that the noun that follows the  $tx^w \dot{a}y$  is its subject, whereas other prepositional verbs are followed by oblique objects.

#### 18.4.39. cəlím 'even'

This has been recorded in only one construction, as a predicate with a nominal adjunct as subject followed by a coordinate clause, as in (a) and (b).

ni? ťθáď<sup>w</sup>əm-θət prunes cəlím peaches °∂ỷ tə tə rotten-self prunes peaches and AUX ART even ART х́э wə-t<sup>θ</sup>ád<sup>w</sup>əm-θət ni? AUX also EST-rotten-self

'The prunes are rotten. Even the peaches have also become rotten.'

?aĺ cəlím  $t \theta e?$ wa-?áwate? šx "łák "łak"-s ?∍ỷ ช้อ even that EST-none iust connection-3POS and also wə-mi xétq-0ət interfere-self EST-AUX(come) 'Even that guy who has no connection is also interfering.'

#### 18.4.40. *mók*<sup>w</sup> 'all'

This was discussed under §18.4.18, when it had the sense 'whenever.' With the sense 'all,' it is used as an adverb to refer, for some speakers, to the object of a transitive or the subject of an intransitive predicate, but not the subject of a transitive. It can appear before or after the head.

(a) mó<sup>.</sup>·k<sup>w</sup> nówoxos nem to sítons. (JP 22) mók<sup>w</sup> nów-əx-əs nem [?ə] sítan-s tə all basket-3POS enter-TR-3TR go OBL ART 'He put them all into his basket.' (b) ni cən wəmák<sup>w</sup> pánət. (CC) ni? cən wə-mák<sup>w</sup> pán-ət AUX I EST-all get.buried-TR 'I buried them all." (c) mi čx<sup>w</sup> x<sup>w</sup>əx<sup>w</sup>ét mók<sup>w</sup>. (JP) mákw mi čxw x wəx wé-t AUX(come) vou be.descending-TR all 'Bring them all down.' (d) nəw mák<sup>w</sup> spapín. (CC) ni? wə-mǿk<sup>™</sup> spəpín AUX EST-all buried 'They are all buried.' (e) mók<sup>w</sup> wočenom to mi técol toli? X s?ámono. (CC 12) mákw técəl wa-cénam tə mi təli? [?ə] all EST-Shaker ART AUX(come) arrive.here from OBL Ż s?ámənə ART Duncan 'Those who came from Duncan were all Shakers'

In her translation, CC gave one sentence, (f), in which  $m \delta k^w$  refers to a transitive subject.

(f) mák ma naw k k v lx as 'é ltan. (CC)
mák ma ni' wa-k k v l-x-as 'é ltan all certain AUX EST-be.hiding-TR-3TR 3PL
'They are all hiding it.'

To AG, however, this sentence seemed to mean 'They are hiding all of them.' To match CC's translation, he offered (g).

(g) môk<sup>w</sup> ?é·łtən ni k<sup>w</sup>ék<sup>w</sup>olx. (AG) môk<sup>w</sup> ?é·łtən ni? k<sup>w</sup>ék<sup>w</sup>ol-x all 3PL AUX be.hiding-TR 'They are all hiding it.'

This is literally 'It is all of them who are hiding it.' The predicate  $m\delta k^w$  'é tan refers to an unexpressed subject, the absent head of the relative clause  $ni k^w \delta k^w \delta l x$ . This type of cleft sentence seems to provide a way of letting  $m\delta k^w$  refer to what is semantically a transitive subject.

The word  $m\delta \vec{k}^w$  can appear as a noun modifier either following the article (and preceded by  $w\partial_{-2}$ ), as in (h).

```
(h) ném yałéłaqatas taw mák ta sté?ex al. (JP 2)
ném yałéłaqatas taw mák ta sté?ex al. (JP 2)
ném yałéłaqatas taw mák tawa mak ta
```

He went and whispered to all the children.

Or it can appear before the article, as in (i).

(i) səw kwəkwi?xwaləstəs məkw tə ste?exwəł. (JP 2)
s-[ni?-s] wə-kwəkwi?xwaləs-t-əs məkw tə ste?exwəł
NOM-AUX-3POS EST-be.pitching-eye-TR-3TR all ART children
'Then she pitched up the eyes of all of the children.'

However, it can be argued that in this last sentence  $m\delta k^w$  is an adverb and part of the predicate rather than a noun modifier and part of the subject.

Similarly, for the following two sentences, (j) and (k), two interpretations seem possible: either  $m \delta k^w$  modifies a noun predicate head or  $m \delta k^w$  is the predicate (cf. similar sentences with  $h \delta y$  and  $w \partial n \delta y$ ) and the noun is the subject. In either case, the noun is followed by a relative clause that shows that it is an extracted transitive subject.

 (j) mák ta x máθk vajam naw x k vécast yaθé? i mak vatx véls ta šxállas. (CC)
 mák ta x máθk vajam ni? wa x k k vé cas ta vaθé?

| m ək " | tə i  | х "тәөк "әуәт                        | n17  | W9-7  | k*-tk*e-cəs-t          | уәθе" |
|--------|-------|--------------------------------------|------|-------|------------------------|-------|
| all    | ART ] | Musqueam                             | AUX  | EST-C | DBL-caulk-hand-TR      | those |
| ?i     | m̀i   | k <sup>w</sup> tóx <sup>w</sup> -éls | [?ə] | tə    | šx <sup>w</sup> žálləs |       |
| AUX    | come  | e enter-ACT                          | OBL  | ART   | picture                |       |

'All of the Musqueams filled the hands of [gave fifty cents each to] those who brought in the picture.' (lit. 'It was all the Musqueams who ...' or 'All were the Musqueams who ...') (k) səsəw mák<sup>w</sup> tə Catholic məstáyax<sup>w</sup> x<sup>w</sup>əqálstax<sup>w</sup> tə nəšx<sup>w</sup>wélay k<sup>w</sup>s<sup>x</sup>as tə ni sq<sup>w</sup>áq<sup>w</sup>əls tə Father Chirouse. (CC 21) wə-mśk<sup>w</sup> s-ni?-s tə Catholic məstávəx<sup>w</sup> Catholic NOM-AUX-3POS EST-all ART people x<sup>w</sup>ə-q*á*l-stəx<sup>w</sup> nə-šx<sup>w</sup>wéləy k »ə s-2a-s tə become-bad-CAUS ART my-parents ART NOM-BE3P-3POS tə ni? s-q\*áq\*əl-s Father Chirouse tə AUX NOM-be.speaking-3POS ART Father Chirouse ART 'Then all of the Catholic people came to dislike my parents because of what Father Chirouse was saying.' (lit. 'Then it was all the Catholic people who ...' or 'All were the Catholic people who ...')

#### 18.4.41. $\check{x}^{w} \acute{a} m$ 'move fast, be quick, immediately, be able to'

As a predicate with a noun or pronoun subject, this has the sense 'move fast, be quick,' as in (a) and (b).

- (a) x̃<sup>w</sup>ôm k<sup>w</sup>θə dôwət-s. (AG)
   fast ART drum-3POS
   'His drum [beat] is fast.'
- (b) ?i čx<sup>w</sup> wənánəw mi ž<sup>w</sup>óm. (JP)
  ?i čx<sup>w</sup> wə-nán wə-mi ž<sup>w</sup>óm
  AUX you EST-very EST-AUX(come) move.fast
  'You came very fast.'

As a predicate with a nominalized clause as subject, it has the sense 'immediately,' 'can immediately,' or perhaps simply 'can,' as in (c), (d), and (e).

(c) x\*\*ám k\*\*anasném. (CC) x̃™́́э́m k<sup>™</sup>a nə-s-ném move.fast ART mv-NOM-go 'I will go quickly. I can go right away.' (d) x̃<sup>w</sup>óm k<sup>w</sup>smis csétəs k<sup>w</sup>θə sq<sup>w</sup>əmq<sup>w</sup>əméys i wəłmi <sup>γ</sup>əyé<sup>γ</sup>θà·m. (JP 3) x̃™́́э́m k »ə s-mi-s cse-t-əs k»θə fast ART NOM-AUX(come)-3POS send-TR-3TR ART sq<sup>w</sup>əmq<sup>w</sup>əméy-s ?∍ỷ wəł-mi

```
dogs-3POS and already-AUX(come)
```

<sup>9</sup>əyé<sup>9</sup>-t-Sam-əm

attack-TR-you(PAS)-INTR

'He would immediately send his dogs and they would attack you.'

(e) ha? ct ce? xílos i xwóm kws qwénàmotct. (JP 12) ha? ct ce? xíləs ?∂ỷ ž™́э́m k »ə if we FUT succeed and fast ART s-q<sup>w</sup>é-namət-ct NOM-penetrate-manage-our 'If we succeed [in this plan], we can get through.'

As a predicate followed by a coordinate clause (with the same subject), it has the sense 'can, may, likely to,' as in (f) and (g).

- (f) x̃<sup>w</sup>óm cən i k<sup>w</sup>ólləxθàmə. (JP)
   x̃<sup>w</sup>óm cən <sup>9</sup>əỷ k<sup>w</sup>ólləx-t-Samə fast I and shoot-TR-you
   'I could shoot you.'
- (g) xxim čxw i mé?kwł. (JP)
  xxim čxw ?əý mé?kwł
  fast you and get.injured
  You're liable to get hurt.'

(For other examples, see §5.2.4 [c] and [d].) It also appears in adverbial adjuncts, as in (h).

(h) səw nem kwəm kwənətəs tə sxixqəł. (JP 2)
 s-wə-nem[-s] kwə wə-xwəm kwən-ət-əs tə sxixqəł
 NOM-EST-go-3POS ART EST-fast get.taken-TR-3TR ART child
 'Then she went quickly and took the child.'

For other examples, see §3.6.

#### 18.4.42. sŽáŽəm 'enough, ought, can'

This is the resultative of  $\lambda am$  'get enough.' It can mean simply 'enough,' as in (a).

(a) niwł skákom ste?é o tθe?. (JP)
ni? woł-skákom sto?é ?o tθe?
AUX already-enough like OBL that
'It's enough like that. That's plenty now.'

As a predicate followed by a nominalized clause, it has the sense 'ought, had better,' as in (b).

(b) skákom kwan sném. (CC) skákom kwa na-s-ném enough ART my-NOM-go 'I'd better go.' With an auxiliary, however, it has the sense 'can,' as in (c).

(c) niw skákom kwo nosném wowéyolos. (CC)
 ni? wo-skákom kwo no-s-ném wo-wéyolos
 AUX EST-enough ART my-NOM-go when-become.day-3SUB
 'I can go tomorrow.'

#### 18.4.43. $s\vec{k}^{w}\acute{e}y$ 'impossible, unable'

When this is a modal predicate and the qualified clause is affirmative, the qualified clause is nominalized, as in (a), (b), and (c).

- (a) skwéy kwənsném. (JP) sk<sup>w</sup>év k<sup>™</sup>ə nə-s-ném impossible ART mv-NOM-go 'I can't go.' (b) skwéy co kwsnéms kwow tolonoco ?al mostoyoxw. (JP 3) skwév ćə k <sup>w</sup>ə s-ném-s k™a wə-təl-?ánəcə ?al impossible OUOT ART NOM-go-3POS ART EST-from-where just məstáyəx<sup>w</sup> person 'Those from elsewhere didn't dare go.' (c) skwéy ma kwskwácní t. (CC)
- (c) sk ey mə k sk əchit. (CC) sk ey mə k xə s-k ec-n-i-ət impossible CERT ART NOM-look-TR-3SUBPAS-SUBPAS 'He can't be seen.'

The combination  $s\vec{k}^{w}\vec{e}y$  and a negative, literally 'cannot not,' has the sense 'must.' The negative can be a nominalized clause, as in (d).

(d) sk<sup>w</sup>éy k<sup>w</sup>əns?<sup>3</sup>wə némen. (JP)
 sk<sup>w</sup>éy k<sup>w</sup>ə nə-s-?<sup>3</sup>wə némen.
 impossible ART my-NOM-not go-I
 <sup>4</sup>I must go.<sup>2</sup>

Or the negative can be in a subordinate clause, as in (e) and (f).

 (e) sk<sup>w</sup>éy wəwe·s céwəθè·lt. (JP) sk<sup>w</sup>éy wə-<sup>9</sup>ówə-əs céw-ət-Sel-ət impossible if-no-3sub help-tr-I(PAS)-subPAS 'I have to be helped.'

(f) sk<sup>w</sup>éyəł wəwé·səł ni·s céwətəm. (JP)
sk<sup>w</sup>éy-əł wə-?ówə-əs-əł ni?-əs céw-ət-əm impossible-past if-not-3sub-past AUX-3sub help-TR-INTR 'He had to be helped.'

As a predicate with a noun or pronoun subject,  $sk^w ey$  has the sense 'very ill,' as in (g).

(g) <sup>9</sup>i scék<sup>w</sup>əl k<sup>w</sup>θən mén. sk<sup>w</sup>éy. (DK)
<sup>9</sup>i scék<sup>w</sup>əl k<sup>w</sup>θə <sup>9</sup>ən-mén sk<sup>w</sup>éy
AUX how ART your-father very.ill
'How is your father?' 'Very ill.'

#### 18.4.44. scowét 'smart, capable, able'

With a noun or pronoun subject, this is rendered 'smart, clever.' With a nominalized clause following, as a kind of complement or adverbial clause, it is rendered 'capable of' (perhaps only in the sense 'know how to do'), as in (a) and (b).

- (a) scowét k<sup>w</sup>s cnóx<sup>w</sup>ołs. (JP) scowét k<sup>w</sup>o s-c-nóx<sup>w</sup>oł-s capable ART NOM-make-canoe-3POS 'He's a good canoe-maker.' (perhaps lit. 'He is clever when he makes canoes.')
- (b) ni <sup>?</sup> ö čx<sup>w</sup> wəscəwet k<sup>w</sup>əθ x<sup>w</sup>əlməx<sup>w</sup>qən. (JP) ni<sup>?</sup> <sup>?</sup> ö čx<sup>w</sup> wə-scowet k<sup>w</sup>ə <sup>?</sup> öT-s-x<sup>w</sup>əlməx<sup>w</sup>-qə AUX ROG you EST-capable ART your-NOM-Indian-speech 'Can you speak Indian?'

# 18.4.45. $sqi^{2}q\partial \hat{l}$ (CC), $sqiq\partial \hat{l}$ (JP), 'unable, ignorant (of how to do something)'

This seems to be a resultative form but the root is not identifiable.

- (a) sqí?qəl cən kwə nə-s-dəy-t. (CC)
   unable I ART my-NOM-be.created-TR
   'I don't know how to do it.'
- (b) sqíqəl k<sup>w</sup>s cá?əms yəθéləy. (JP) sqíqəl k<sup>w</sup>ə s-cá?-əm-s unable ART NOM-explain.genealogical.connection-INTR-3POS yəθéləý they
  'They don't know how to explain how they're related.'

#### 18.4.46. <sup>9</sup>óý 'better'

As an adjective,  $2\delta y$  is 'good,' but it can appear as a modal predicate, followed by either a nominalized clause adjunct or a subordinate clause and rendered 'it would be good/well if' or 'one had better.'

(a) <sup>γ</sup>όỷ k<sup>w</sup>sťí·tct θǝwÅa ... (CC 11) s-fi-ət-ct Aswin ?áỷ k <sup>w</sup>ə good ART NOM-beg-TR-our her 'We'd better ask her ...' (b)  $^{9}$ óý kws galétet nem lí m kwaw  $^{9}$ axwín  $^{9}$ al. (JP 4) s-qəlét-ct łí∙m k<sup>™</sup>a wə-?əx "ín ?áv≀ k <sup>w</sup>ə nem pick good NOM-again-our ART EST-little ART go ?aĺ just 'We'd better go again and pick [berries] a little.' (c)  $^{9}$ áý ctwa? wə-វănəq-è.n. (JP 16) good SPEC if-potlatch-I 'I think I'd better put up a potlatch.' (d) scécon wo?óy wox wonhóye?es θoňa nošx womník. (CC 12) sčéčəň wə-?áv wə-x<sup>w</sup>ən-hóye?-əs θəna if-still-leave-3SUB this really EST-good nə-šx <sup>w</sup>əmník <sup>w</sup> my-living.parent's.sibling

'It would certainly be good if this aunt of mine were gone.'

#### 18.4.47. s*ží*? 'want, like'

Evidently always bearing a possessive affix, this is probably a nominalization of  $c \vec{\lambda} t^{2}$  'want' and a relative clause with extracted oblique object (see §4.1.1.2 and §12.2.1), that is,  $n \partial s \vec{\lambda} t^{2}$  is literally 'what I want.' It can appear as a predicate with a noun subject, as in (a) and (b).

- (a) nəsẳí? k̃<sup>w</sup>ə páy. (CC)
  nə-s-[c-]ẳí? k̃<sup>w</sup>ə páy
  my-NOM-do-value ART pie
  'I want some pie.' (lit. 'Pie is what I want.')
- (b) nəski? tə?ínə. (CC)
  nə-s-[c-]kí? tə?ínə
  my-NOM-do-value this
  'I want this one.' (lit. 'This one is what I want.')

Or it can appear in the usual position of a relative clause in a cleft sentence, as in (c).

```
(c) Xa ti²ínə nəsXí?. (CC)
Xa tə²ínə nə-s-[c-]Xí?
BE3P this my-NOM-do-value
'This is the one I want.' (lit. 'It is this one that is what I want.')
```

In (d), it is the predicate head with an adverb taking a subject particle and no noun subject.

(d) wəňán čx<sup>w</sup> wənəsẳí?. (CC) wə-ňán čx<sup>w</sup> wə-nə-s-[c-]ẳí? EST-very.much you EST-my-NOM-do-value 'I love you very much.' (lit. 'You are very much what I want.')

It can appear as a predicate with a nominalized clause as subject, as in (e), (f), and (g).

- (e) nəsẳí? kwə nəsném. (JP)
  nə-s-[c-]ẳí? kwə nə-s-ném
  my-NOM-do-value ART my-NOM-go
  'I want to go.' (lit. 'My going is what I want.')
- (f) ?ówə nəsẳí·s kwə nəsném. (JP)
  ?ówə nə-s-[c-]ẳí?-əs kwə nə-s-ném not my-NOM-do-value-3SUB ART my-NOM-go
  'I don't want to go.' (lit. 'My going is not what I want.')
- (g) <sup>2</sup> θ Å(<sup>2</sup>) θ Å(<sup>w</sup> θ) s c śc w θ à m θ. (JP)
  <sup>2</sup> θ T-s-[c-]Å(<sup>2</sup>) <sup>2</sup> θ Å(<sup>w</sup> θ) n θ-s-c śc w-θt-Sam θ. your-NOM-do-value ROG ART my-NOM-help-TR-you
  <sup>2</sup> 'Do you want me to help you?' (lit. 'Is my helping you what you want?')

### **19** Numerals

Halkomelem is like other Native languages of northwestern North America in having a decimal system of numerals capable of generating numbers into the thousands.

#### 19.1. SIMPLE FORMS

The numerals from one to ten are:

| <i>náča</i> <sup>2</sup> 'one'                                                                       | <i>ťžóm</i> 'six'                                   |
|------------------------------------------------------------------------------------------------------|-----------------------------------------------------|
| <sup>?</sup> isélə ~ yəsélə 'two'                                                                    | $\dot{t}^{\theta} \dot{a}^{\gamma} k^{w} s$ 'seven' |
| <i>tíx</i> <sup>w</sup> 'three'                                                                      | tqéce? 'eight'                                      |
| $\check{x}\partial^{2}\acute{a}\theta\partial n \sim \check{x}a^{2}\acute{a}\theta\partial n$ 'four' | tú·x <sup>w</sup> 'nine'                            |
| <i>łąćcəs</i> 'five'                                                                                 | <sup>2</sup> ápən ten'                              |

The first two numerals have bound alternate forms,  $n\acute{e}caw$ - 'one-,' which appears in one or two words, and  $\theta\acute{e}m$ - ~  $\thetaam$ - 'two-,' which appears in six or seven words.

A traditional way of reckoning is to count on the fingers, starting with the little finger of the left hand as 'one,' proceeding to the left thumb ('five'), crossing to the right thumb ('six'), and finishing with the right little finger ('ten'). Because the direction is consistent, each finger is identified with a number and so the outstretched hands may become a device for visualizing numerical relations. It is my impression, although I do not have the data to support it, that this practice also made it possible to signal numbers easily and consistently with the hands.

The linkage of numbers and hands makes etymological speculation about three of these numerals inviting (cf. Elmendorf 1962; Galloway 1977, 418-20). 'Five' seems to consist of the root of  $l\dot{q}\acute{e}t$  'wide' and the suffix *-cəs* 'hand,' probably referring to the display of a spread-out hand. 'Six' may contain a root  $\dot{t}a\dot{x}$  'bifurcate' (cf.  $\dot{t}\dot{x}\dot{a}\theta at$  'branch out'), perhaps referring to the shift to the right hand. 'Eight' may have the root  $t\dot{a}q$  'be closed in, be surrounded,' with a suffix not otherwise attested in Halkomelem but possibly identifiable with

'hand' in several other Coast Salish languages. (The Cowichan dialect has  $t\hat{e}^2c\partial s$  'eight,' possibly with a root  $t\hat{e}^2$  'appear' and  $-c\partial s$  'hand.') Elmendorf (1962, 8) suggests that the Musqueam (and Upriver) word is a loan from Lushootseed. Hess (1976, 468) identifies the Lushootseed  $tq\dot{a}\dot{c}i$  'eight' as "lit. 'closed hand.''' Possibly the source of this word was a gesture in which the right middle finger ("eight") met the thumb in a circle.

The numerals from eleven to nineteen run:  ${}^{2}áp \partial n {}^{2}i t \partial n \partial c a^{2} \sim {}^{2}áp \partial n {}^{2}i k^{w}$  $n \partial c a^{2}$  'eleven' (lit. 'one and ten') and so on, the article varying with the position of the referent.

'Twenty' is  $\dot{c}k^w \delta x$ . The word is unique in the system and not transparent in Halkomelem; however, the Clallam word is  $n\partial \dot{c}x^w k^w \delta s$  'twenty,' lit. 'one count' (Thompson and Thompson 1971, 270), and the Halkomelem word almost certainly has the same etymology (from  $n\delta \dot{c}a^2$  'one' and  $k^w \delta x$  'count'), referring to the twenty digits of the human body.

Multiples of ten from 'thirty' to 'ninety' are formed from unstressed variants of the numerals three to nine with the suffix  $-\partial t sx \dot{e} \sim -t sx \dot{e}$  (CC) or  $-\partial t cy \dot{e}$  (JP) 'times ten.' The forms given by CC are:

| łəx <sup>w</sup> əłsxé 'thirty'          | $i^{\theta} \partial k^{w} s \partial t s x e$ 'seventy' |  |  |
|------------------------------------------|----------------------------------------------------------|--|--|
| <i>žəθənłsxé</i> 'forty'                 | təqəcəłsxé 'eighty'                                      |  |  |
| łəqacsałsxé 'fifty'<br>łxamałsxé 'sixty' | <i>tu</i> · <i>x</i> <sup>w</sup> <i>əlsxé</i> 'ninety'  |  |  |
| TT 1 1 1                                 |                                                          |  |  |

The hundreds are:

| <i>néčawac</i> 'one hundred'                      | $lix^w$ néčowoć 'three hundred,' etc. |
|---------------------------------------------------|---------------------------------------|
| $\theta \dot{e} \dot{m} \partial c$ 'two hundred' | ?ápən nécəwəc 'one thousand'          |

'One hundred' and 'two hundred' are formed with the bound forms of 'one' and 'two' and a suffix  $-\partial c$  (- $\partial c$ ?), which cannot be identified.

These are the simple forms of the numerals. They are used both substantivally and attributively in reference to many kinds of things. In the following sentence from a text (CC 14) on fishing, they refer to fish and days.

| (a) səw cxánce?s ?iwáw | a? ?ápən ?iv                    | váwa? čl | ć™áx ?iwáw | va? łģécəs | tə |
|------------------------|---------------------------------|----------|------------|------------|----|
| náča? swéyəl. (CC      | 14)                             |          |            |            |    |
| s-wə-c-xənce?-s        | <sup>9</sup> iwáwa <sup>9</sup> | ?ápən    | ?iwáwa?    | ċk̇̀»́áx   |    |
| NOM-EST-get-prey-3POS  | maybe                           | ten      | maybe      | twenty     |    |
| ?iwáwa? łģécəs         | tə náč                          | a? swé   | yəl        |            |    |
| maybe five             | ART one                         | day      |            |            |    |
| (                      |                                 |          |            | <i>~</i> · |    |

'Then he made a catch, maybe ten, maybe twenty, maybe five in one day.'

And in one from a text on hunting, the numeral refers to animals.

(b) səsəw kwácnəxwəs tə ?isélə məlləs. (CC 8)
 s-ni-s wə-kwec-nəxw-əs tə ?isélə məlləs
 NOM-AUX-3POS EST-see-TR-3TR ART two racoon
 'Then he saw two racoons.'

There are also, however, several sets of complex forms that are used for particular classes of things.

#### 19.2. COMPOUND FORMS

These are mostly composed each of a numeral stem and a lexical suffix, but there are a few formed in other ways. Perhaps all of the lexical suffixes that appear in these forms can also occur with  $k^w i n$  'how many' (§17.20) and nearly all with  $q \delta \tilde{x}$  'many.' A few suffixes may be limited to use with numerals and these two roots, and so might be identified as "numerical classifiers," but most have no such limitation. A few of these sets of complex forms have membership running to ten and beyond, but others are limited to fewer members. The following may not be a complete list.

(1) Persons. 'One person' and 'two persons' are reduplicated forms of 'one' and 'two,' while the rest are formed with the suffix  $-\acute{el}\partial \sim -\partial l\partial$  'person.'

| náňače? (CC) 'one [person]'                       | <i>tqecə<sup>2</sup>élə</i> 'eight'                                   |
|---------------------------------------------------|-----------------------------------------------------------------------|
| yáýsala 'two [persons]'                           | <i>tu</i> · <i>x</i> <sup>w</sup> <i>élə</i> 'nine'                   |
| <i>łəx<sup>w</sup>élə</i> 'three [etc.]'          | ? <i>əpé·lə</i> 'ten'                                                 |
| $\check{x}\partial\theta i\cdot l\partial$ 'four' | <sup>9</sup> ápən i k <sup>w</sup> nánoče <sup>9</sup> 'eleven,' etc. |
| <i>łąecséla</i> 'five'                            | $d\vec{k}^{w}x \ell l a$ 'twenty'                                     |
| <i>txámələ</i> 'six'                              | $\vec{k}^{w}i\cdot l \partial$ 'how many?'                            |
| <i>ť<sup>θ</sup>ak<sup>w</sup>sélə</i> 'seven'    |                                                                       |

For JP, 'one person' was  $n\dot{a}\dot{n}\dot{a}\dot{c}a^2$  or  $n\dot{a}\cdot n\dot{c}a^2$ . These are used to refer to any human beings, as in (a) and (b).

- (a) žəθí·lə ce? k<sup>w</sup>θə sq<sup>\*</sup>áq<sup>\*</sup>əł. (CC 7) žə?áθən-ələ ce? k<sup>w</sup>θə sq<sup>\*</sup>áq<sup>\*</sup>əł four-person FUT ART ready 'There will be four [men] lying in wait.'
  (b) ném cə ?á·ł tə txám-ələ słənłénəỷ. (JP 4)
- go QUOT embark ART six-person women 'Six women are said to have embarked.'

For multiples of ten above twenty, the simple numerals are used, such as  $t \partial x^w \partial t s x \acute{e} m \partial s t \acute{e} y \partial x^w$  'thirty people.' 'One person' and 'two persons' are further reduplicated in  $n \partial c n \acute{a} \cdot n \acute{c} a^2$  'one by one' and  $y \partial s y \partial s \partial a$  'two by two.'

(2) Times. The numerals for 'times' are irregular for 'once' and 'twice' but others recorded are formed with the suffix -*ét*. They are:  $n\partial c ex^{w}$  'once,'  $\theta \partial m e$ 

'twice,'  $lx^{w\ell l}$  'three times,'  $\dot{x}\partial \partial n\ell l$  'four times,'  $l\dot{q}acs\ell l$  'five times,'  $\dot{t}\dot{x}am\ell l$  'six times,'  $\dot{t}^{\theta}ak^{w}s\ell l$  'seven times,'  $apan\ell l$  'ten times,'  $\dot{k}^{w}an\ell l$  'how many times?' 'Eight times' and 'nine times' were thought to be irregular but not remembered. (For use, see §18.4.3.)

(3) Round objects. These are formed with the suffix  $-\delta s \sim -\delta s$  'face, round object' (which produces a shift from  $\dot{e}$  to  $\dot{a}$  in a root). They are used for counting months (moons) and dollars. Those recorded are:  $n\delta\dot{c}\delta s$  'one,'  $2is\dot{a}l\delta s$  'two,'  $tixw\delta s$  'three,'  $\dot{x}\delta\theta in\delta s$  'four,'  $2\delta\rho\dot{a}n\delta s$  'ten,'  $\dot{k}win\delta s$  'how many (months, dollars)?' It is likely that there are more forms in the set and that they can be used for other round objects.

(4) Canoes and vehicles. Forms recorded are:  $n\acute{e}caxwat$  'one canoe,'  $\theta\acute{e}maxwat$  'two canoes,' yatitaxw 'three canoes,'  $ya\check{x}\acute{a}^2a\theta an$  'four canoes,'  $yatad\acute{e}cas$  (CC), ' $it\acute{a}dacas$  (JP) 'five canoes.' 'One canoe' and 'two canoes' have the bound forms of 'one' and 'two' with a suffix that may be a variant of -wat 'vessel' (perhaps also appearing in  $sn\acute{a}xwat$  'canoe,' the root of which is unidentifiable). The three other members of this set have the prefix ya- '(moving) along' and expanded forms of the roots, possibly progressive forms.

(5) Houses. Only three numerals have been recorded:  $n\acute{c}owtx^w$  'one house,'  $\theta\acute{e}mtx^w$  'two houses,'  $tx^w\acute{o}wtx^w$  'three houses,' plus  $k^winowtx^w$  'how many houses?' The suffix is  $-\acute{e}wtx^w \sim -owtx^w \sim -tx^w$  'house.'

(6) Containers. Numbers from 'one' through 'six' have been recorded:  $n\dot{a}\dot{c}\partial q\partial\dot{n}$ , 'iséləqəh, tíx"əqəh, xə $\theta$ ínəqəh, tģécsəqəh, txəməqəh. The suffix may be the same as -qən 'throat.' These are used for baskets, cups, and so on, usually meaning 'containerful.'

(7) Birds. The suffix is  $-i\dot{w}s \sim -i\dot{w}s$  'body.' These may also be used for counting rabbits and other small game. A full set from 'one' to 'ten' and 'twenty' have been recorded:  $n\partial \dot{c}(\dot{w}s, \ ^2is\partial i\dot{w}s, \ tx^{w}i\dot{w}s, \ \dot{x}\partial\theta ini\dot{w}s, \ t\dot{q}\acute{e}c\partial\dot{w}s$  (probably  $t\dot{q}\acute{e}cs\partial\dot{w}s$ ),  $t\dot{x}\acute{a}m\partial\dot{w}s, \ t\dot{\theta}ak^{w}si\dot{w}s, \ tq\acute{e}ce^{2}i\dot{w}s, \ tu\cdot x^{w}i\dot{w}s, \ ^2\partial p\acute{e}n\partial\dot{w}s, \ \dot{c}k^{w}xi\dot{w}s,$ plus  $k^{w}(n\partial\dot{w}s$  'how many?'

(8) Trees. 'One' to 'five' have been recorded:  $sn\dot{c}\partial tp$ ,  $s^{2}is\dot{c}^{2}etp$ ,  $stix v\partial tp$ ,  $s\ddot{x}\partial\theta t tp$ ,  $st\dot{q}\dot{e}cs\partial tp$ . The suffix  $-\partial tp$  'vegetation' is very productive. I cannot account for the prefix *s*-, presumably 'nominalizer.'

(9) Parts or kinds. The suffix *-mat* seems to have these several meanings; cf.  $n\dot{a}\dot{c}a^{2}mat$  'one piece, whole,' *?isélamat* 'two pieces,'  $t\dot{a}x^{w}mat$  'three pieces,'  $k^{w}anmat$  'how many kinds,'  $q\dot{a}xmat$  'many kinds' are the only numerals recorded.

(10) Kinds. JP gave this set as 'kinds,' but I am uncertain of the suffix and the gloss. He gave only 'one' to 'three':  $x^w n \partial c \dot{a} w \partial \theta$ ,  $x^w i s \partial l \dot{a} w \partial \theta$ , and  $x^w t x^w \dot{a} w \partial \theta$ , and  $k^w i n \partial w \partial \theta$  'how many kinds of animals?'

(11) Years. I have recorded only 'one' to 'three':  $n\partial cawi \cdot nx^w$ ,  $\partial \partial mi \cdot nx^w$ ,  $tx^w \partial winx^w$ . The suffix is  $-\partial wi \cdot nx^w$  except with the bound form of 'two'; cf.  $k^w \partial n\partial wi \cdot nx^w$  'how many years.'

(12) Paddles. Only one form has been recorded:  $2is\partial l\acute{e}lwe^2s$  'two paddles.' The suffix is  $-\partial lwe^2s$  'paddle.' There are probably more in the set.

(13) Bundles. Only one form has been recorded:  $n\partial c\partial l \ell^2 c$  'one bundle.' The suffix is  $-\partial l \ell^2 c$  'bundle.' There are probably more in the set.

(14) Net mesh. Four forms have been recorded:  $\dot{c}k^w \partial x \dot{a} \partial s$  'twenty mesh,'  $t \partial x^w \partial t c y \dot{e}^2 \dot{a} \partial s$  'thirty mesh,'  $\ddot{x} \partial \theta \partial n t c y \dot{e}^2 \dot{a} \partial s$  'forty mesh,' and  $k^w \partial n t c y e^2 \dot{a} \partial s$  'how many tens of mesh?' The suffix is  $-\dot{a} \partial s$  'eye.' These are used in giving the depth of a gill net, the length being given in fathoms.

(15) Days. Days are usually counted with the simple numerals, but these two forms are also used, perhaps usually in counting ahead:  $\theta \rightarrow m \rightarrow nt$  'two days,' meaning also 'day after tomorrow,'  $t \rightarrow x^w \rightarrow t n \acute{e}t$  'three days.' For  $-\partial t n \acute{e}t$ , see §22.2.3.

(16) Long objects. Only  $n\partial \dot{c} \ell m \partial \dot{t}^{\theta}$  'one' has been recorded. The suffix is  $-\ell m \partial \dot{t}^{\theta} \sim -\partial m \partial \dot{t}^{\theta}$  which appears in  $s \dot{k}^{w} (n \partial m \partial \dot{t}^{\theta}$  'how many long things (poles, pilings, toothpicks, hairs),'  $\dot{\lambda} \partial q \ell \ell m \partial \dot{t}^{\theta}$  'tall (as a person),' and so on.

See Gerdts et al. 2002 for a discussion of numeral classifiers in Halkomelem, and Shaw et al. 2002 for greatly expanded lists in Musqueam.

#### 19.3. ORDINALS

These are formed from compound numerals with s- 'nominalizer' and -s 'thirdperson possessive,' as in (a) to (d).

- (a) tə́s kwəə sxəə(iləs ... (CC 17) tə́s kwəə s-xə?aən-ələ-s arrive.there ART NOM-four-person-3POS until the fourth person ...'
- (b) X a k<sup>w</sup>θ s š s θ s n é ł s ce? ... (CC 17)
  X a k<sup>w</sup>θ s s x s s a h s c e l s s c e l s s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l s c e l
- (c) Xa s-yôýsələ-s. (JP 10) BE3P NOM-two.persons-3POS 'That was the second person.'

(d) təs tə səpé·ləs i tə ná?ənċa? (JP 10)
 təs tə s-?ápən-ələ-s ?əy tə ná?ənċa?
 arrive.there ART NOM-ten-person-3POS and ART one.person
 'until the eleventh (person)'

The use of *s*- 'nominalizer' to form ordinals has a parallel in English: *true* – *truth; four* – *fourth*. The -*s* 'third-person possessive' may be interpreted as 'of them,' that is, of whatever set the ordinal belongs in.

It may be that these forms can be used only substantivally and not attributively, cf. (e), in which the simple numeral modifies 'morning' but with the meaning 'fourth.'

 (e) Xa tə xa'aθən nét-əł. (CC 17)
 BE3P ART four night-past 'It's the fourth morning.'

Some of the names of the days of the week appear to be ordinals (see  $\S22.2.3$ ).

# **20** Exclamations and Interjections

#### 20.1. EXCLAMATIONS

Exclamatory sentences seem to have no distinctive features of syntax. They probably do have a distinctive intonation (not yet analyzed), but otherwise they may be marked simply by particles. (See  $\dot{q}\partial$ ,  $k^wt$ ,  $wa^2$ , and  $^2\dot{a}^2a$ , §16.1.14, 16.1.17, and 16.1.18.)

Some exclamations are notable, however, for their use of words of a special class, here termed exclamatory-interrogative adjectives. A few of these seem to have literal meanings opposite those intended.

Examples of words used for their opposite meaning, in a kind of sarcasm, are the following with  $l \ell l a q$  'easy' and  $s \vec{\lambda} \ell^2 \vec{\lambda} a l$ , seemingly identifiable with  $s \vec{\lambda} \ell \vec{\lambda} a l$  'stuck (in one place)' or a diminutive of it.

- (a) <sup>9</sup>a<sup>•</sup>, 1íləq <sup>9</sup>a<sup>9</sup>a. (JP 7)
  oh easy ROG!
  'Oh, it's hard!' (lit. more like 'Easy? Ha!')
- (b) sắć?ắəỉ cən ?a?a. (AG) stuck I ROG! 'What a fast runner I am!'

#### 20.2. EXCLAMATORY-INTERROGATIVE ADJECTIVES

These are words that identify qualities of an undetermined degree. They appear in exclamatory sentences with the sense 'how \_\_\_\_!' They (or some of them) can also have the sense 'however \_\_\_\_,' '\_\_\_ to whatever extent,' or 'nobody knows how \_\_\_\_.' Seven words of this sort have been noted.

#### 20.2.1. Ž*ąć ý* 'how far! however far, for an unknown distance'

(a) Åqéý k<sup>w</sup>ə. (CC, JP)
 how.far then
 'My, that's far! Gee, that's far!'

- (b) <sup>\*</sup>⁄<sub>A</sub> <sup>d</sup> <sup>e</sup> <sup>d</sup> <sup>e</sup> <sup>d</sup> <sup>w</sup> <sup>a</sup>. (JP 22) how.far ROG SPEC then 'You don't know how far.'
- (c) Ždéý k<sup>w</sup> ava? ta ?iwał mi šx<sup>w</sup>?ímaxct. (AG)
  Ždéý k<sup>w</sup> ava? ta ?i wał-mi šx<sup>w</sup>-?ímax-ct how.far then PRESUM ART AUX already-come OBLNOM-walk-our 'What a distance we have walked!' (lit. 'How far must be then the place to which we have already walked!')
- (d) ?a; šx<sup>w</sup>á·x<sup>w</sup>ť<sup>0</sup> čx<sup>w</sup>. x<sup>w</sup>əźdéy ?e čx<sup>w</sup> ctwa? wənéməx<sup>w</sup> ?á·ł tə niəx<sup>w</sup> θ<sub>9</sub>vt. (JP) ?a. šx<sup>w</sup>á·x<sup>w</sup>t<sup>θ</sup> čx<sup>w</sup>. x<sup>w</sup>əŻdéy ?a čx<sup>w</sup> ctwa? wə-ném-əx<sup>w</sup> you how.far aw crazv SPEC if-go-you ROG you ?á·ł [?ə] tə ni?-əx<sup>w</sup> θáv-t travel.in OBL ART AUX-vou be.made-TR 'Aw, you're crazy. How far do you think you'll go with that thing you've

'Aw, you're crazy. How far do you think you'll go with that thing you've made?'

#### 20.2.2. *léq* 'how fast! however fast, at an unknown speed'

- (a) léq k "ə. (CC) how.fast then 'My, what speed!'
- (b) ?énəcè, léq k wə tə x wəy qwələł. (JP) oh.my! how.fast then ART steamship
  'My, the steamer's not slow! Wow, how fast the steamer is!'
- (c) léq <sup>?</sup>ə. (JP) how.fast ROG 'It wasn't slow!'
- (d) ?a:, léq ?ə čx<sup>w</sup> ce?. (JP) aw how.fast ROG you FUT
  'Aw, you haven't got that speed!' (said to someone boasting that he'll run a certain distance in a certain time)
- (e) léq <sup>9</sup>e k<sup>w</sup>is nem xtáməstàləm. (JP 22)
  léq <sup>9</sup> k<sup>w</sup> s-ni<sup>9</sup>-s ném
  how.fast ROG ART NOM-AUX-3POS AUX(go)
  xtém-stəx<sup>w</sup>-al-əm
  swim-CAUS-we-INTR
  'Nobody knows how fast we were taken away.' (lit. 'It was an unknown

speed when we were swum away with.')

#### 20.2.3. tə<sup>9</sup>éň 'how wonderful! what a ! how powerful!'

- (a) tə<sup>9</sup>éň k<sup>w</sup>ə tə s<sup>\*</sup>λéləqəm. (JP) how.wonderful then ART fierce.being
   'What a monster!'
- (b) tə?én k włą k »ə sx ">x "á?as i k<sup>w</sup>évžθət how wonderful thunder be.moving.self then ART AUX k̃™əna x<sup>w</sup>əné·nt. (JP) evening that 'Boy, that was some thunder last night!'
- (c) tə?éň k wə tə snáx wəł-s. (CC) how.wonderful then ART canoe-3POS
   'What a wonderful canoe he has!'

#### 20.2.4. wéwə 'how wonderful!' (synonymous with §20.2.3?)

 (a) wéwə k<sup>w</sup>ə tə s⊀éləqəm. (JP) how.wonderful then ART monster
 'What a monster!'

#### 20.2.5. x<sup>w</sup> *wém* 'how big!'

(a) x \*\* wém ?a?a tə ?ən-léləm \*\* ələp. (AG) how.big ROG! ART your-house-your(PL)
 'What a big house you guys have!'

#### 20.2.6. Xíć 'how long!'

Compare  $\dot{c}\partial\dot{c}i\cdot\dot{z}$  'short,' of which this may be the root. Perhaps this is another instance of a word used for the opposite meaning.

(a) X1C k<sup>w</sup>∂ wa? t∂ X<sup>w</sup>1∂m-s. (AG) how.long then PRES ART rope-3POS
 'What a long rope he has!'

#### 20.2.7. *cižəmət<sup>9</sup>* 'how tall!'

Compare  $\dot{c}\partial\dot{c}\dot{t}\dot{\vec{x}}\partial m\partial\dot{t}^{\theta}$  'short in stature,'  $< \dot{c}\partial\dot{c}\dot{t}\dot{\vec{x}}$  'short,'  $-\partial m\partial\dot{t}^{\theta}$  'long object.'

(a) cíλ = mət<sup>θ</sup> k wə tθe? swəy qe?. (JP) how.tall then that man
 'Boy, that fellow is tall!'

#### 20.3. INTERJECTIONS

An interjection can appear either alone as a complete utterance or before a sentence but without any syntactic relationship to it. Interjections express affirmation and denial, some emotional quality, or simply some acknowledgment of a second person. They include a few sounds that do not appear elsewhere. The following is probably not a complete list. The ordering is arbitrary.

20.3.1. hé?e 'ves' ġíġək <sup>w</sup>əls <sup>9</sup>ən-sq<sup>w</sup>əméy hé?e łáď ?ə ta habitually be.biting.something ROG ART vour-dog ves dídəkwəls be.biting.something 'Does your dog bite?' 'Yes, he bites.'

#### 20.3.2. ?ówə 'no'

Identical with  $2\delta w \partial$  'not,' which does have a syntactic role (§6.1).

(a) <sup>9</sup>ówə, <sup>1</sup>⁄<sub>A</sub> a nə-swé<sup>9</sup> nə-méñə. (JP 1)
 no BE3P my-own my-child
 'No, it's my child.'

#### 20.3.3. ná 'what, yes' (as in answer to hearing one's name called)

## 20.3.4. $n \dot{a} \vec{w} \sim n \dot{a} \cdot \vec{w} \sim n \dot{a} \cdot \vec{w}$ (in answer to one's name), 'hey' (to get someone's attention)

In the second sense, this is used especially with a spouse or a close friend, and may take the particle  $\dot{q}_{\partial}$  'emphatic.' In (a), AG explained, it shows that the speaker knows that the person addressed is there.

(a) ná·w dó si?ém.'Hello!' 'Well, hello!'

To a spouse it may imply affection, as in (b).

(b) ném cən wəłyá·ys, naw. (AG) 'I'm going to work now, dear.'

#### 20.3.5. $n i^{9} \sim n i$ 'hey there, excuse me'

This is probably identical with  $ni^{2}$  'be there.'

(a) ni?. si?ém. ?śwe· čx\* neməx\* té?t ... (JP 19) ni? si?ém ?áwa ṫé?-t ?a čx<sup>w</sup> nem-ax\* hey.there AUX(go)-you sir not ROG you try-TR 'Say, sir, wouldn't you try ...'

#### 20.3.6. <sup>9</sup>*i* 'here' (to attract attention)

This is probably ?*i* 'be here.'

| (a) ?í                           | ۶í   | ?í,  | °∕awə | čx <sup>w</sup> | k »ək »ə́n-ət-əx ». (JP 19) |  |  |
|----------------------------------|------|------|-------|-----------------|-----------------------------|--|--|
| here                             | here | here | not   | you             | being.taken-TR-you          |  |  |
| 'Hey, hey, hey! Don't touch it.' |      |      |       |                 |                             |  |  |

#### 20.3.7. $y \neq y \neq \cdot$ well, all right'

This is said in urging someone to speak, and is also used by the audience in a myth telling to urge the teller to continue. (Cf.  $y\dot{e}$  'taunting echo,' §16.3.2.)

- (a) yè, qálamax, stémas kweθ xwamí. (JP 9)
  yè, qálamax, [wa-] stémas kwa ?aT-šxw-amí
  all.right pal that what-3SUB ART your-OBLNOM-come
  'All right, pal, tell me what you've come for.'
- (b) ye, <sup>9</sup>əyí·t ce·p kwə <sup>9</sup>əθqwəlqwəl ni·p cəlstamət. (JP 22) ye, <sup>9</sup>e·y·í-t ce·p kwə <sup>9</sup>əT-s-qwəlqwəl all.right continue-DUR-TR you(PL) ART your-NOM-tell ni<sup>9</sup>-əp cəlstamət AUX-you(PL) happen.to(PL)
  'Well, proceed with your story of whatever happened to you.'

#### 20.3.8. <sup>9</sup>á· 'ah, oh, aw, well'

This often seems to have no function other than to begin a statement. It is often followed by a term of address (such as  $si^2\acute{em}$  'sir, ma'am,' or a kinship term). In narratives with dialogue, 'a may serve primarily to indicate to the audience that a new speaker is speaking.

- (a) ?a·, ni cən wəsẳáẳəm ... (JP 9)
   ?a· ni? cən wə-sẳáẳəm ah AUX I EST-enough
   'Well, I'm ready ...'
- (b) ?a· si?ém, wi?é·ý čx<sup>w</sup> ?al. (JP 3)
  ?a· si?ém wə-yə-?é·ý čx<sup>w</sup> ?al ah sir EST-along-continue you just 'Well, sir, just continue.'
- (c) <sup>?</sup>a· <sup>?</sup>ówö čx<sup>w</sup> xöte·x<sup>w</sup> tθé?. (JP 4)
  <sup>?</sup>a· <sup>?</sup>ówö čx<sup>w</sup> xötö-öx<sup>w</sup> [?ö] tθé?
  ah not you be.saying-you OBL that
  'Oh, don't say that.'

#### 20.3.9. <sup>9</sup>0·, <sup>9</sup>5· 'oh'

These are perhaps only variants of the last.
(a) <sup>?</sup>o· móňə, <sup>?</sup>i cən x<sup>w</sup>əčéňəm. ni cən cá<sup>..</sup>k<sup>w</sup> x<sup>w</sup>ə<sup>?</sup>óỷ. (CC 21) ?i x <sup>w</sup>ə-scénəm 20. mána cən ni con cákw child T become-Shaker I oh AUX AUX far x<sup>w</sup>ə-?śż become-good

'Oh, child, I have become a Shaker. I have become far better.'

(b) ɔ· sċċċən ce·p wə-?éləỷ səwáỳqe?. (JP 12)
 oh really you(PL) EST-good(PL) men
 'Oh, you're really good men.'

#### 20.3.10. <sup>9</sup>áà<sup>9</sup>, <sup>9</sup>á<sup>9</sup>à 'ahhhh! aha!'

- (a)  ${}^{2}\dot{a}\dot{a}$  naətə wə $\theta$ ə ${}^{2}\dot{t}$  x "əsəl $\dot{t}$ k" tə  $\theta$ á $\theta$ əns tə s $\dot{t}$ éləqəm. (JP 6) ?áà? ?ə wə-θə?ít x \* ə-səlík \* na tə to. ahhh be there OBL ART EST-truly become-broken ART θáθən-s sẳéləqəm tə mouth-3POS ART fierce.being 'Ahhh! There the mouth of the monster is truly broken.'
- (b)  $?\dot{a}?\dot{a}$ , səw  $\theta\dot{a}t$  tə sp $\dot{a}\cdot\dot{l}$ , ni? yəx<sup>w</sup> a?a ... (JP 7) s-wə-θát[-s] spá·ĺ. ?á?à∙ tə ni? a?a vəx<sup>w</sup> ROG! ahhh NOM-EST-sav-3POS ART raven be there INF "Aha", said Raven, "It must be so ..."

#### 20.3.11. °5 'hmm'

This is perhaps a more casual  $2\dot{a}$ .

səmémt<sup>0</sup>ən. (a) **ž**?**õ** θávəlsəm cən ce? i nem prepare.weapons hmm proud(DIM) I go FUT and cən ... (JP 7) T

'Hmmm, proud little thing, I'll prepare and I'll go ...'

Raven here is speaking to Seagull. Raven, Mink, and some other characters prefix  $\check{x}$ - to some words.

#### 20.3.12. ?óš 'well!'

This expresses disapproval. It is used mainly by women. If a woman sees a girl working with male motions (pulling toward herself rather than pushing away), she might say:

(a) ?áš, šté·wən ?ə čx<sup>w</sup> k<sup>w</sup>əθswáýqe?. (JP)
?áš šx<sup>w</sup>te?éwən ?ə čx<sup>w</sup> k<sup>w</sup>ə ?əT-s-swáýqe?
well think ROG you ART your-NOM-man
'Well, do you think you're a man?!'

# 20.3.13. šé? 'well!'

Like the last, but used by both sexes. If a man heard a boy say  $2\delta \dot{s}$ , he might say:

(a) šé?, θəňa ἀéməỷ. (JP)
 well this young.woman
 'Hey, are you a girl?!'

**20.3.14.**  $\check{s} \acute{a}$ ,  $\check{s} \acute{a}$ ? 'pshaw! oh, heck, etc.' (DK) Perhaps the same as the last.

# 20.3.15. <sup>9</sup>écənè 'oh my!' (JP)

An expression of surprise.

# 20.3.16. <sup>9</sup>énene 'oh my!' (AC)

The same as the last?

# 20.3.17. yá-? 'my!'

An expression of surprise, as in (a).

(a) <sup>9</sup>a<sup>9</sup> yá<sup>...9</sup> (CC)'Oh my!' (as when seeing big fish)

# 20.3.18. yu · 'wow!' [?] (JP)

An expression of surprise and perhaps appreciation or condemnation.

# **21** Kinship Terms

In every language, terms for relatives constitute a small, semantically structured group of words that are of cultural interest because they reflect features of the social structure of the speakers. In Musqueam and other Halkomelem dialects, the simple terms for relatives constitute an especially small set, only two of which, *mén* 'father' and *tén* 'mother,' have precise English equivalents, there being no simple Musqueam equivalent of 'brother,' 'sister,' 'grandfather,' 'aunt,' 'cousin,' and so on. This lack of correspondence is the result of differences in semantic principles, which are probably related to social structure, and of differences in grammar.

Differences of the first kind will be discussed with each group of terms below. Grammar is relevant in that it has allowed for some of what is probably historically a great reduction in the number of Proto-Salish terms.<sup>1</sup> Most English terms (and terms in several other Salishan languages) distinguish the sex of the referent – for example, a 'brother' is invariably male and a 'sister' female; of the bare terms in Musqueam, only those corresponding to 'father' and 'mother' make such a distinction. For all of the Musqueam terms, the article or demonstrative identifies the sex of the referent, as well as other facts ignored by English. On the other hand, several Musqueam terms make distinctions, such as whether the linking relative is living or dead or whether the speaker ("ego") is male or female, that English terms ignore.

We can distinguish terms for consanguineal ("blood") relatives from those for affinal relatives ('in-laws"). In Musqueam the two sets are structured differently. We can further distinguish three sets of each of the following: (1) terms of personal reference, (2) terms of address, and (3) terms of impersonal reference.

<sup>1</sup> See Elmendorf 1961 on the reduction of Proto-Salish kinship terms, and my comments in Suttles 1965.

Terms of personal reference appear with possessive affixes and in grammatical contexts that require articles. Terms of address are the terms of personal reference without any prefixes other than possessives; they may or may not take possessive affixes, and they appear without articles. Terms of impersonal reference appear in grammatical contexts with articles, but like the verb forms that they probably are, they cannot take possessive affixes. Three of the most common of these bear no relationship to the terms of personal reference; the rest are derivatives of the terms of personal reference. The terms of personal reference and address are described in §21.1 and §21.2, and those of impersonal reference in §21.3.

# 21.1. CONSANGUINEAL RELATIVES

I will first present the terms of personal reference and address for parents, children, siblings, and cousins, then the terms of impersonal reference for these, and finally all sets for other relatives.

# 21.1.1. Parents and Children

The basic terms are:

mén 'father' (pl.?, dim. méməň, address mé?, méməň)
tén 'mother' (pl.?, dim. tétəň, address té?, tétəň)
máňə 'child' (pl. méməňə, dim. mímnə, address máňə)

The terms of address for father and mother, especially the diminutives, which have the sense of 'dear little father' and 'dear little mother,' may be used in talking to a little boy or little girl. The term  $m\acute{e}n∂$  is 'child' in the sense of 'son, daughter, offspring,' not in the sense of 'baby'  $(sq\acute{e}q∂l∂)$  or 'child'  $(s\dot{x}i\dot{x}q∂l)$  as a life-status term. An adult is still the  $m\acute{a}n∂$  of his or her parents. However, there is a related or derived form, '2mi̇́am∂n' little.'

When used in reference, the basic terms must have possessives (my, your, etc.):  $n \partial m \acute{e}n$  'my father,'  $\partial n t \acute{e}n$  'your mother,'  $m \partial n \partial s$  'his child, her child, their child.' And they must also be preceded by articles ( $t\partial$ ,  $\partial \partial$ , etc.) or demonstratives:  $t\partial n \partial m \acute{e}n$  'my father (who is here now),'  $\partial \partial n \partial t \acute{e}n$  'my mother (who is here now),'  $\partial \partial n \partial m \acute{o}n \dot{\partial} \dot{n}$  'my son (who is here now).' Thus the articles indicate whether the person referred to is male or female and whether present and visible, somewhere not visible, or deceased.

If a relative is deceased, this must be indicated either though the article or through the particle  $-\partial t$  'past.' A deceased parent is referred to as  $k^w \partial n \partial m \acute{e} n$  or  $k^w \partial \partial n \partial m \acute{e} n \partial t$  'my late father,'  $k^w s \partial n \partial t \acute{e} n$  or  $t \partial n \partial t \acute{e} n \partial t$  'my late mother.'

There is also a word  $\delta x^w w \delta^2$  'parent' (from  $w \delta^2$  'own,' meaning 'someone to whom one belongs.' In its plural form, this term is probably more commonly used with a wider meaning than simply parent:  $\delta x^w w \delta \partial y$  'parents, uncles and

aunts, those to whom one belongs,' that is, relatives of an older generation. Because it is plural, it takes the masculine (non-feminine) article:  $t a n a \delta x^w w \ell a y$  'my parents (who are here now).' This term is used in addressing parents or uncles and aunts in formal speeches, as in (a).

(a) tə łwślap našx wélay, yaθastála can k θa našx qwélawan.
ta łwślap na-šx wélay yaθas-t-ála can k θa
ART you(PL) my-to.whom.belong tell-TR-you(PL) I ART na-šx qwélawan my-thought
'My parents (or my uncles), I will tell you my feelings.'

A deceased parent may be referred to with the term  $\delta x^w t \delta l e$  (lit. 'origin'?), as in:  $k^w \partial n \partial \delta x^w t \delta l e$  'my late father,'  $k^w s \partial n \partial \delta x^w t \delta l e$  'my late mother.'

The term for 'child' has a derivative, sman e'm 'children and other junior relatives.' It is used in addressing a group of younger relatives that might include children, nephews, and nieces, and grandchildren. These are the relatives whom one might take responsibility for and lecture to.

#### 21.1.2. Siblings and Cousins

Halkomelem does not have separate words for siblings and cousins. The terms used for siblings can also be used for relatives in one's own generation as distant as fourth cousins, persons who share a great-great-great-grandparent. In theory, at least, a marriage between a couple related to that degree was still incestuous.

There are four basic terms:

- $sx^2 \acute{e}y \partial t$  'older sibling, senior cousin' (pl.  $sx^2 \partial x^2 \acute{e}y \partial t$ , address  $2\acute{e}y \partial t$ , possibly related to the suffix  $-\acute{e}y \partial t$  'child')
- $sq\acute{e}^{2} \partial q$  ( $sq\acute{e}^{2} eq$ ,  $sq\acute{e} \cdot q$ ) 'younger sibling, junior cousin' (pl.  $sq\partial l\acute{e} \cdot q$ , address  $q\acute{e}^{2} \partial q$ ,  $q\acute{e} \cdot q$ , possibly related to  $q\acute{e}q\partial l\partial$ , identified as 'baby' by CC and as 'boy baby' by JP, and  $q\acute{e}q$ , identified as Cowichan for 'baby' by CC and as 'girl baby' by JP)
- 2élax 'sibling or cousin of the opposite sex' (that is, sister or female cousin of a male, brother or male cousin of a female) (pl. 2alélax, address not recorded)
- $\delta x^{w^2} \dot{a} \dot{q}^w a^2$  'sibling or cousin (older or younger of either sex)' (pl.  $\delta x^{w^2} \dot{a} l \partial \dot{q}^w a^2$ , also used in address, reciprocal  $\partial \partial \dot{q}^w i^2 t \partial \dot{l}$ ,  $\partial \dot{q}^w i^2 t \partial \dot{l}$ ,  $y \partial \dot{q}^w i^2 t \partial \dot{l}$  'related as siblings or cousins')<sup>2</sup>

The first two terms distinguish siblings on the basis of their relative age – older brothers and sisters from younger brothers and sisters. Your older brother is your  $sx^2\acute{e}y\partial t$ , and your younger brother is your  $sq\acute{e}^2\partial q$ . When these terms

<sup>2</sup> SP gave a possibly related Katzie term,  $s \delta \dot{q}^w$ , which he identified as a lineage going back to one's great-great-great-great-great-great-great to term.

are used for cousins, they distinguish senior from junior lines of descent, that is, your father or mother's older brother or sister's child is your  $sx^2éy\partial t$ , even if that cousin is younger than you, and your parent's younger sibling's child is your  $sq\hat{e}^2\partial q$ , even if that cousin is older than you. This principle is used to maintain seniority over the generations. If you have a distant cousin whose father or mother was a cousin of your father or mother, and you know that your father or mother called his or her father or mother  ${}^2\acute{e}y\partial t$ , then you must call that distant cousin  ${}^2\acute{e}y\partial t$  no matter what the age difference. This principle is consistent with primogeniture, the transmission of names and other property to the eldest child, which may have been generally followed in Aboriginal society.<sup>3</sup>

Again, when used in reference, these terms must have possessives and articles. And like 'child,' the article shows whether the person you are referring to is a female or a male.

 $\theta a \ nas x^2 e y a t' my$  older sister/senior female cousin (who is here)'  $k^w \theta a \ nas q e q t' my$  younger brother/junior male cousin (who is not here)'  $ta \ na^2 e t a x t' my$  brother/male cousin (who is here) (a woman speaking)'  $ta \ na^2 e t a x t' my$  sister/female cousin (who is absent) (a man speaking)'

Thus, while the bare terms do not distinguish brothers (or male cousins) from sisters (or female cousins), the terms in a grammatical context do.

The plural of the term  $\xi x^{w_2} \dot{a} \dot{q}^w a^2$  is used in speeches addressed to groups identified, at least for purposes of establishing solidarity, as relatives, as in (a).

(a) <sup>9</sup>á· si·<sup>9</sup>ém nə-šx<sup>w</sup><sup>9</sup>áləq<sup>w</sup>a<sup>9</sup>
 o honoured.ones my-siblings/cousins
 O my honoured brothers and sisters.<sup>2</sup>

The root of  $\delta x^{w^2} \dot{a} \dot{q}^w a^2$  is probably the basis for  $s^2 \dot{a} \dot{q}^w \partial^2 \delta y \partial t$  'youngest child of a pregnant mother, next older sibling of an unborn child,' that is, the child who is about to be replaced as youngest by the one in the womb.<sup>4</sup>

Other terms for siblings will be given later, in §21.3, "Terms of Impersonal Reference."

# 21.1.3. Uncles and Aunts, Nephews and Nieces

As one might expect from the terms for siblings and cousins, the terms for parents' siblings include their cousins, and the terms for siblings' children include

<sup>3</sup> However, this principle is seen in kinship terminologies of peoples in, among other regions, the Great Basin, for whom property was not especially important. It is characteristic of Spier's "Yuman" type of kinship terminology (Spier 1925, 75-76).

<sup>4</sup> This term appeared in a text some time after I had elicited what I had supposed to be a full set of kinship terms in part by the traditional genealogical method, asking, "What do you call your father's brother Sam's wife Jane," and so on. As I could not possibly have thought of such a term, it shows the value of texts in filling out what might seem to be well-known areas of the lexicon.

cousins' children. The Musqueam terms also differ from English in that there is one set of terms while the linking relative is alive and another if the linking relative has died. Thus we have the following four terms:

šx<sup>w</sup>əmník<sup>w</sup> (CC), šx<sup>w</sup>əmník<sup>w</sup> (JP) 'parent's sibling or cousin during the lifetime of the parent' (pl. šx<sup>w</sup>əmnélək<sup>w</sup>, šxəmnélək<sup>w</sup>, address ník<sup>w</sup> or ník<sup>w</sup>iye)

- *stíwən* 'sibling or cousin's child during the lifetime of the sibling or cousin' (pl. *stətíwən*, dim. *stítwən*, dim. pl. *stələtətíwən*, address *tíwən*, *títwən*)
- $\delta x^w m \partial \theta \partial y \dot{e} t$  'parent's sibling or cousin after the death of the parent' (pl.  $\delta x^w m \partial l m \partial l \theta \partial y \dot{e} t$ , address  $m \partial \theta \partial y \dot{e} t$ )
- swanméyt, swanméyat (< wénam 'orphan,' -éyat 'child') 'sibling or cousin's child after the death of the sibling or cousin' (pl. not recorded, address wanméyt, wanméyt)

Thus while my mother was alive I would have referred to her brother (my uncle) as  $n\partial sx^w \partial mnik^w$ , and he would have referred to me (his nephew) as  $n\partial stiw\partial n$ . But after my mother was gone, I would have referred to him as  $n\partial sx^w m\partial \partial y dt$ , and he would have referred to me as  $n\partial sw \partial mdy dt$ . Similarly, while my brother was alive, I would have referred to his daughter (my niece) as  $n\partial stiw\partial n$ , but after his death, I would have had to call her  $n\partial sw \partial mdy dt$ . Presumably, when people change from the first pair of terms to the second, they are reminded of the link between them and their responsibility for keeping the kinship tie.

In the system of classification used by anthropologists (e.g., Murdock 1949), Halkomelem kinship terms for relatives in parents' generations are "lineal," that is they distinguish between parents and parent's siblings but do not distinguish mother's siblings from father's siblings. In this, Halkomelem and other Coast Salish terms are like English and unlike those of Interior Salish languages, which are "bifurcate collateral," that is, they distinguish parents, father's siblings, and mother's siblings, and they are unlike those of the Northern Coast, which are "bifurcate merging," that is, they have a single term for father and father's brother and a single term for mother and mother's sister. Lineal and bifurcate collateral terminologies are consistent with bilateral descent, while a bifurcate merging terminology is consistent with unilineal descent, the principle basic to the system of matrilineal clans of the Northern Coast.

#### 21.1.4. Grandparents, Grandchildren, and Beyond

Just as the terms for siblings also refer to other relatives in one's own generation, that is, to cousins, so too do the terms for grandparents, grandchildren, great-grandparents, and so on refer to others of these relatives' generations. The basic terms are:

 $s(l_{\partial} \text{ 'grandfather, grandmother, great uncle, great aunt, 'that is, 'relative in the second ascending generation' (pl. <math>s_{\partial} l_{\delta}(l_{\partial})$ , address  $s(l_{\partial})$ )

- <sup>2</sup>*iməθ* 'grandson, granddaughter, grandnephew, grandniece,' that is, 'relative in the second descending generation' (pl. <sup>2</sup>*imiməθ*, address <sup>2</sup>*iməθ*, <sup>2</sup>*iməyə*)
- scámaq<sup>w</sup> 'great-grandparent, great-grandparent's sibling or cousin, greatgrandchild, sibling or cousin's great grandchild,' that is, 'relative of the third ascending or descending generation' (pl. scálamaq<sup>w</sup>, dim. scá<sup>2</sup>camaq<sup>w</sup>, dim. pl. scalá<sup>2</sup>camaq<sup>w</sup>, address cámaq<sup>w</sup>)
- <sup>2</sup>5k<sup>w</sup>∂y∂q<sup>w</sup>(CC), <sup>2</sup>5k<sup>w</sup>∂yàq<sup>w</sup>(JP) 'great-great-grandparent, etc.,' that is, 'relative of the fourth ascending or descending generation'
- $t^{\theta}\delta \dot{p}\delta y\delta q^{w}$  (CC),  $t^{\theta}\delta \dot{p}\delta y\delta q^{w}$  (JP) 'great-great-great-grandparent, etc.,' that is, 'relative of the fifth ascending or descending generation' (also refers to the cap at the stem of a strawberry or salmonberry)

The diminutive  $s\dot{c}\dot{a}^{2}\dot{c}amaq^{w}$  might be used, JP said, by a great-grandparent addressing a great-grandchild. I did not record terms of address for relatives in the fourth and fifth ascending or descending generations. Probably one does not ordinarily meet a lineal ancestor or descendant of such status, but collaterals could be contemporaries.

# 21.1.5. Relatives and Kin Groups

Kinship was and is of great importance in Coast Salish society. The most general term referring to kinship is  $sy\acute{y}a$  (<  $y\acute{y}a$  'secure, tightly tied') 'friend, relative' (pl.  $siy\acute{y}a$  //syəyéýe//). This term perhaps literally means 'bond.' In the old days, friendship without kinship may have been uncommon. When non-relatives wanted to establish close ties, they soon negotiated marriages and thus established kinship. And so the meaning 'relative' is probably more basic than 'friend.' The term has a reciprocal  $y\acute{a}'itai //y\acute{a}y\acute{a}tai//$  'be related.' JP said that one might address a "distant sila" as  $y\acute{e}ya$  as a sign of respect. In the big house, speeches commonly begin:  $?\acute{a}\cdot si\cdot?\acute{e}m$   $nasiy\acute{e}ya$  'O my honoured friends/ relatives.'

There is a noun,  $\dot{c}\dot{a}^{\gamma}$  'genealogical connection,' and a verb derivative,  $\dot{c}\dot{a}^{\gamma}\partial m$  'explain a relationship,' which JP remembered hearing his mother use in the following:

```
(a) ni ² čx^w łźdellex^w k^w čá² ct x^wté· yeθéley Åew nesiyéye. (JP)
 ?ə
 čxw
 łáġəl-ləx™
 cá?-ct
 **té
 ni
 k w
 ?a
 vou
 know-TR
 connection-our
 toward OBL
 AUX ROG
 ART
 yəθéləy Že
 wə-nə-siyéyə
 those
 also
 EST-my-relations
```

'Do you know how we're related to those ones who are my relatives?'

(b) qí<sup>9</sup>qəl k<sup>w</sup>scá<sup>9</sup>əms yəθéləy.

| qí?qəl                                                | k <sup>w</sup> | s-čá?əm-s                  | yəθéləy |  |  |
|-------------------------------------------------------|----------------|----------------------------|---------|--|--|
| ignorant                                              | ART            | NOM-explain.genealogy-3POS | those   |  |  |
| 'They don't know how to explain how they're related.' |                |                            |         |  |  |

Ancestors are referred to as:

 $sy \partial w \dot{e} \dot{n}$  (<  $y \partial w \dot{e} \dot{n}$  'before') (lit. 'what was before,' also 'heritage')  $k^w \partial \partial \dot{m} \dot{i} t y \partial^2 \dot{e} \dot{y} eqt$  'those whom we are replacing' (CC 1)

Descendants are referred to by several terms:

 $st^{\theta} \dot{a}q^{w} \partial m$  'offspring, descendants' ('suckling'?; cf.  $t^{\theta} \dot{a}q^{w} \partial t$  'suck it')  $s^{2} \partial y \dot{e}q$  'descendant' (lit. 'replacement') as in (c)

(c) Xawłmi s<sup>9</sup>əyeqs k<sup>w</sup>θə x<sup>w</sup>əncə́lwəməł. (JP)

 $\dot{\chi}$ a wəł-mi s<sup>?</sup>əỷéq-s k<sup>w</sup>θə x<sup>w</sup>ənċślwəm-əł BE3P already-come replacement-3POS ART one.family-past 'They are descendants of one family.'

The only term elicited for 'family' is  $x^w n \partial c \dot{a} l w \partial m \sim x^w n \partial c \dot{a} l w \partial m \sim x^w \partial n \dot{c} \dot{a} l w \partial m (< n \dot{a} \dot{c} a^2 \text{ 'one'} + ?)$  'one family.' JP believed that this term refers to the descendants of one couple, now not necessarily the residents of one house.

(d) <sup>9</sup>ónθə wəłmi st<sup>0</sup>áq<sup>w</sup>əms k<sup>w</sup>θə x<sup>w</sup>ənciólwəməł (~ k<sup>w</sup>θə x<sup>w</sup>ənciólwəməł məstáyəx<sup>w</sup>). (JP) sť<sup>θ</sup>ág<sup>w</sup>əm-s ?ánθə wəł-mi x »ən cəl wəm-əł k <sup>w</sup>θə be.I already-come descendant-3POS one.family-past ART x <sup>w</sup>əncʻəlwəm-əł  $(\sim k^{w}\theta a)$ məstáyəx<sup>w</sup>) ART one.family-past person

'I am the descendant of that one family (~ the people of that one family).'

#### 21.2. AFFINAL TERMS

Under this heading I will deal with terms for spouses and relationships resulting from marriage, as well as terms for in-laws.

#### 21.2.1. Spouses

There is a single term for spouse: *stálos* 'spouse, husband, wife' (pl. *stoltálos*, dim. *stá?tolos*, *státolos*).

The diminutive may be used affectionately, as in  $\theta a$  nas $\dot{q}$ ánaq nastátalas 'my dear little wife,' or derisively to refer to a married woman's lover. A spouse may be addressed simply with  $n\dot{a}\dot{w}$  'hey.' There is a reciprocal form, *titalástal* 'related as husband and wife' or simply 'married couple.'

Because in the old days a man could have more than one wife, there is a term  $sx\dot{a}\dot{y}a$  'co-wife' (pl.  $sxax\dot{a}\dot{y}e$ ).<sup>5</sup> This term has a reciprocal form,  $xax^wayi^2ta\dot{l}$  (JP) or  $xax^way\dot{a}^2ta\dot{l}$  (CC) 'be co-wives, be wives of the same man.' Something

<sup>5</sup> Somewhere I have seen the term  $sx \dot{a} \dot{y} \partial$  identified with  $x \dot{a} y e^2$  'maggot.' These words are not the same, and the resemblance is probably of no significance.

of the relations between women married to the same man is suggested by the belief that while eating either swiwa, the Fraser River eulachon ("hooligan"), or  $s^2 \dot{a} y a n a x^w$ , the smaller variety that runs in the Squamish River, you should not mention the other, "because they are co-wives."

Although a woman could not have two husbands, there is also a term,  $sx \partial y \acute{e} seq$ , which refers to another man with whom one shares (or has shared?) a woman, as a husband and a wife's lover (perhaps a former husband), not a situation that men would openly agree to. The term may contain the root of  $y \partial s \acute{e} l \partial$  'two' and the suffix *-eq* 'penis.' Some people regard it as obscene.

There are terms for having multiple wives:

```
x^{w?} \partial y s \acute{e} l t \partial x^{w} 'have two wives' (< y \partial s \acute{e} l \partial 'two')
x^{wl} x^{w\acute{e}} l t \partial x^{w} 'have three wives' (< l i x^{w} 'three')
```

A person who has lost a spouse is a  $sye^{2}tan$  'widow, widower.' In the old days, a person who has lost a spouse was encouraged or even obliged to marry a close relative of the deceased, following practices technically called "the levirate" and "the sororate."

#### 21.2.2. Primary Affines

There are four terms that refer to the relatives called in English 'mother-in-law,' 'father-in-law,' 'sister-in-law,' 'brother-in-law,' 'daughter-in-law,' and 'son-in-law.' However, while the English terms are simply based on the consanguineal terms used by the spouse, the Musqueam terms follow a different pattern. They are:

 $\delta k^{w}(t \partial w')$  ( $\delta x^{w}$ - 'oblique nominalizer'?  $k^{w}(t \partial w')$  'to live with one's in-laws') 'father-in-law, mother-in-law, wife's brother' (pl.  $\delta k^{w} e' \partial w')$ 

scowtéł 'son-in-law, daughter-in-law, man's sister's husband' (pl. sciwotéł)

- smétəx<sup>w</sup>tən 'sibling-in-law of the opposite sex,' that is, '(man's) brother's wife, (woman's) sister's husband, wife's sister, husband's brother' (pl. smé<sup>2</sup>əltəx<sup>w</sup>tən, address métəx<sup>w</sup>tən)
- $\delta x^{w^2} \ell \partial x$  (<  $\delta x^{w}$  'oblique nominalizer,'  $\ell \ell \partial x$  'sibling/cousin of opposite sex,' i.e., 'one who has a woman's brother') 'sister-in-law (woman speaking),'i.e., 'husband's sister, brother's wife (woman speaking)' (pl.  $\delta x^{w^2} \partial \ell \partial x$ )

Musqueam terms for blood relatives, like English terms, keep the generations apart. Except for  $2\ell l a x$  'sibling/cousin of the opposite sex,' they mean the same thing when used by either men or women. For males but not for females, however, the Musqueam terms for in-laws follow an overriding principle.

For a man, the term used for his wife's parents applies also to a man in his own generation – his wife's brother, while the term used for his child's spouse applies also to a man of his own generation – his sister's husband. The principle here is probably one that reflects the nature of marriage in aboriginal society: a man (with his family) was responsible for his wife to her parents and to her brothers (his  $\delta \vec{k}^{w} \acute{e} t \partial \vec{w}$ ), while the men responsible to him for their wives were the husbands of his daughters and his sisters (his *sciwatét*). His male  $\delta \vec{k}^{w} \acute{e} t \partial \vec{w}$  are men from whom he received his wife, while his male *sciwatét* are men who received their wives from him or his family.

For a woman, the terms for in-laws keep the generations separate. A woman's  $\delta \vec{k}'' \cdot \delta \vec{w}$  are simply her husband's parents, and her *sciwot*  $\delta \vec{k}$  are simply her children's spouses. She refers to both her husband's brother and her sister's husband as *smétax*  $\vec{v} \cdot \delta n$ , and they refer to her by the same term. She refers to both her husband's sister and her brother's wife as  $\delta x'' \cdot \delta n$ , and they refer to her by the same term.

The term that is used reciprocally between the sexes,  $sm\acute{e}tax^wtan$  'sibling-inlaw of the opposite sex,' is used by persons that might marry in the event of the death of a spouse. If one of a married couple dies, the four terms in-law are replaced by a single term:  $\acute{c}\acute{e}.\acute{y}e$  'deceased spouse's relative, spouse of deceased relative' (pl.  $\acute{c}a\dot{y}\acute{e}.\acute{y}e$ , impers.  $\acute{c}a\acute{c}\acute{e}\acute{y}\acute{e}$ ).

There is a verb derived from this term:  $\dot{c}\dot{e}\cdot\dot{y}em$  'marry ones  $\dot{c}\dot{e}\cdot\dot{y}e$ '. Such a marriage was common in the old days. If a husband or wife died, the family of the deceased might provide a close relative of the deceased as a new spouse. This provided for children to the marriage and kept up the alliance between families.

#### 21.2.3. Secondary Affines

There is another in-law term for which there is no equivalent in English (although some other European languages have one):  $s\vec{k}^w\delta lwas$  'co-parent-inlaw,' that is, 'child's spouse's parent (or other close relative)' (pl.  $s\vec{k}^w\delta l\vec{k}^w lwas$ [CC],  $s\vec{k}^w\delta k^w lwas$  [JP]; reciprocal  $\vec{k}^w t \cdot lwastal$  [CC],  $s\vec{k}^w \delta lwastal$  [JP] 'be allied through the marriage of children').

This is a very important term. Marriages were alliances between families, arranged by and maintained by the people who became co-parents-in-law. Another term for this kind of affine is  $snéc_{\partial t}tx^w$  (< néc 'different' +  $-\partial tx^w$ ?; cf.  $-tx^w$  'house,'  $-eltx^w$  'spouse') 'opposite friend' (JP's gloss) (pl.  $sné^2 \partial lc \partial t^w$ ) (JP).

After the death of one of a married couple, the co-parents-in-law became  $dt \times dt + dt$  'deceased child's spouse's parent, child's deceased spouse's parent, 'lit. 'co-weeper' (pl.  $dt \times dt \times dt + dt$ ).

And there is still another term for which there is no English equivalent:  $sn\partial c\partial ll\partial q \sim s\partial nc\partial ll\partial q (< n\partial ca^2 \text{ one' or } ncc' \text{ different'} + ?)$  'spouse's sibling's spouse,' that is, 'husband's brother's wife, wife's sister's husband, etc.' (pl. not recorded, address  $n\partial c\partial ll\partial q$ ).

There are also a few terms that correspond to persons sometimes called "relatives by marriage" in English:  $xcep\theta$  'uncle by marriage (aunt's husband), aunt by marriage (uncle's wife)'  $\dot{s}x^wsil\partial$  'step-grandparent, great uncle/aunt by marriage'  $\dot{s}x^w?im\partial\theta$  'grandchild's spouse'

# 21.2.4. Step, Half, and Adopted Relatives

A few terms refer to relationships coming from second and plural marriages:

- *snəċáwəyət* (< *náċa*<sup>2</sup> 'one,' *-eyet* 'child') 'half-sibling' (reciprocal *hinċəwyáłtəť* 'related as half-siblings'). This term, JP thought, was perhaps restricted to half-siblings born of the same mother.
- *cłmántəl* (< *cł* 'co-,' *mén* 'father,' *-təl* 'reciprocal') 'half-sibling sharing the same father' (JP). I did not elicit a *tén* 'mother' counterpart.

clilé<sup>9</sup>em 'step-parent'

- *sqé*·*qstənàmət* 'step-sibling' (< *sq*<sup>?</sup>*əq* 'younger sibling, etc.' + causative reflexive suffix [see §10.5.3], lit. 'pretend younger sibling')
- $sk^w ami?t am$  (CC),  $k^w ami?t am$  (AG) (<  $\sqrt{}$  of  $k^w amat$  'raise,' -ayat 'child') 'adopted child'

scimné<sup>9</sup>em (< ?) 'adopted child'

# 21.3. TERMS OF IMPERSONAL REFERENCE

In narratives, terms other than those given above are often used for relatives of the characters, identifying them in relation to one another. The two such terms most commonly used are:

 $s \delta n \dot{Z} e^{2}$  'older sibling, cousin senior in descent' (pl.  $s \dot{e} \cdot n \dot{Z} e^{2}$ , dim. not recorded)  $s \delta s a q^{w} t$  'younger sibling, cousin junior in descent' (pl. not recorded, dim.  $s \partial^{2} \delta \cdot s a q^{w} t$ )

These terms cannot take possessives, and so they cannot be used for one's own relatives or another's. You cannot say  $**nasán \dot{A}e^2$  or  $**nasá^2saq^wt$ . And because they cannot refer to one's own relatives, they cannot be used in talking to a relative. To call out to your older brother, you have to say  $2\dot{a} \cdot 2\dot{e}yat$ . You cannot call out  $**2\dot{a} \cdot sán \dot{A}e^2$ . For these reasons, I have called them (Suttles 1965) "terms of impersonal reference." In narratives these terms follow articles, as  $ta sán \dot{A}e^2$  'the older (of a pair of brothers), the senior (of a pair of cousins)' and  $ta sásaq^wt$  'the younger (of a pair of brothers), the junior (of a pair of cousins), 'referring to persons who have already been identified in the story.

They can, however, be used as modifiers of terms of personal reference, as in (a) and (b).

(a) k<sup>w</sup>θ s s n k<sup>w</sup>θ s s n k<sup>w</sup>θ s s n k<sup>w</sup>θ s s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n k<sup>w</sup>θ s n

(b) Xa sə<sup>9</sup>á·səq<sup>w</sup>t nə-sx<sup>9</sup>éyəł BE3P junior(DIM) my-older.sibling 'It's my youngest older brother.'

There are also derivatives that can take possessives, such as:

- (c) k<sup>w</sup>θ∂ n∂s∂nŽe<sup>2</sup>mí·n (CC)
   'my next older brother'
- (d) nəsənŽé<sup>2</sup>tən (CC)
  'my older sister' (pl. sé·nŽe<sup>2</sup>tən [JP])

Two derivatives refer to groups:

 $sa^2saq^wt\dot{e}n$  'younger group' (used for the first run of chum salmon) (JP)  $san\dot{\lambda}e^2\dot{e}n$  'older group' (used for the second run of chum salmon) (JP)

And these can have plural forms, as in (e).

(e) Xaw mək<sup>w</sup> səlá<sup>2</sup>səq<sup>w</sup>te<sup>n</sup>. (JP)
 'They are the younger ones.' (referring to three younger siblings)

In not taking possessive affixes, these terms are like verbs rather than nouns. In fact, they may be verb forms. The term  $s \delta n \tilde{\lambda} e^2$  looks like a resultative form and  $s \delta^2 s \delta q^w t$  like a progressive form, but I cannot identify the roots.

For 'parent' and 'child' the terms of impersonal reference are:

```
cí²cət 'parent' (pl. cəlí²cət) (CC), syí²əl 'parent' (pl. siyí²əl) (JP)
səmné² 'parent, one who has a child' (CC)
mímne² 'child' (pl. məlé·mne²) (CC)
```

The term  $ci^2c\partial t$  may be historically related to Clallam  $c\partial t$  'father,' and if it is a progressive form, it may mean something like 'one who is parenting.' This term was given by CC; JP believed it was Cowichan and said that the Musqueam word was  $syi^2\partial l$ . The term  $s\partial m e^2$  is a resultative meaning 'childed.' The term of impersonal reference for 'child' is also derived from  $m\partial n\partial$  'child,' the term of personal reference.

Again, the sibling terms can be used with this term, as in  $k^w \theta \partial s \delta n \dot{k} e^2 m \dot{m} m e^2$  'the oldest child (who is male).' The plural appears in  $t \partial l i^2 \partial k^w \theta \partial m \partial \dot{e} \dot{m} n e^2$  'one of the children.'

It was CC who first led me to an understanding of the difference between these terms and the set that appears with possessives. After we had established a basis for identifying them, I asked her for more, and she produced a full set, which I give below in the right column, following the personal set (with glosses abbreviated in standard anthropological fashion).

| <sup>9</sup> éləx                  | 'sb of op sex' | ?ál∂x                                 |
|------------------------------------|----------------|---------------------------------------|
| šx <sup>w</sup> əṁník <sup>w</sup> | ʻpa sb'        | š x <sup>w</sup> əmnának <sup>w</sup> |

| stíwən                  | 'sb ch'                  | státəwən                                                                         |
|-------------------------|--------------------------|----------------------------------------------------------------------------------|
| šx <sup>w</sup> məθəỷéł | 'dec pa sb'              | šx <sup>w</sup> əməl <sup>†</sup> əθéyəlł                                        |
| swənméyəł               | 'dec sb ch'              | swáwənməył                                                                       |
| sílə                    | ʻgr pa'                  | sásələ                                                                           |
| ?ímə <del>0</del>       | 'gr ch'                  | ?śm̀əO                                                                           |
| scáməqw                 | 'gt gr pa'               | scəcáməq <sup>w</sup>                                                            |
| škwíław                 | 'sp pa, wi br'           | šk <sup>w</sup> ák <sup>w</sup> əłəw, pl. šk <sup>w</sup> álək <sup>w</sup> əłəw |
| scəwtéł                 | 'ch sp, si hu'           | scəwítəlł                                                                        |
| smétəx <sup>w</sup> tən | 'hu br, wi sis, etc.'    | səmtéx <sup>w</sup> tən                                                          |
| šx <sup>w?</sup> éləx   | 'hu sis, br wi'          | <i>šx<sup>w?</sup>áléx</i> (probably <i>šx<sup>w?</sup>álax</i> )                |
| ċé·yə                   | 'dec sp rel, dec rel sp' | caceyé, pl. calaceyé                                                             |
| sk <sup>w</sup> álwəs   | 'ch sp pa'               | pl. sk <sup>w</sup> ək <sup>w</sup> əlwés                                        |
| snəčálləq               | 'sp sb sp'               | səncəleq                                                                         |

Clearly, these terms of impersonal reference are all derived from those of personal reference, but the question remains as to how. In the impersonal set, there is usually reduplication of the first consonant, a full first vowel is reduced to schwa, the stress may fall on this vowel or on a later one, possibly a stressed full vowel, and resonants may be glottalized. In these respects, these forms resemble one or another aspect of the verb, but it is not possible to say unequivocally that they are progressive, durative, or what. One of the problems for analysis is that several of the personal set are composed of roots and affixes that cannot be identified. Most of these terms of impersonal reference have not appeared in the texts I have recorded, and, unfortunately, I did not try to elicit them from other speakers. They remain a problem for further analysis and comparative work.

# **22** Space and Time

Distinctions involving space and time are required by the grammar in nearly every utterance in ways that have been described at various places in the preceding sections. Here I will simply summarize these and then describe the treatment of time and space in the lexicon.

#### 22.1. SPACE

Grammar often requires the expression of location or direction in relation to the speaker or to the focus of interest in a narrative. Many predicates require auxiliaries, and thus the speaker must choose between ?i 'be here' and ni? 'be there,' which identify an event as appearing or occurring near the speaker or away from the speaker, or between ?ami ~ mi 'come' and nem' 'go,' which identify any movement as toward the speaker or away from the speaker (see §3.2.1). Every nominal adjunct must be introduced by an article or demonstrative or consist of a demonstrative that locates it in relation to the speaker or the focus of interest by indicating whether it is present and visible, nearby but invisible, or remote, non-existent, or hypothetical (see §15.1.2). The locative demonstratives (§15.2.2.1) and demonstrative auxiliaries (§15.3), formed with ?i 'be here' (with a variant ?e) and ni? 'be there' (and a variant na?), distinguish a 'yonder' position as well as a 'here' and a 'there.'

In the lexicon, direction and location are also expressed by about fifty roots and their derivatives. Many of these roots are verbs that can stand alone (without affixes) to express motion or, less often, position. Many have progressive and resultative forms and can take transitive, causative, reflexive, and reciprocal suffixes. The resultative forms usually express location. There are also some roots that can only be inferred from complex forms and cannot be classified.

Most of these roots take one or more affixes that have a spatial meaning. The most common of these are the suffixes  $-\partial l \sim -il$ ,  $-il\partial m$  'move toward' (§12.3.1),  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{al}w\partial l$ ,  $-\dot{a$ 

26), and the prefixes  $x^{w_{\partial}}$  'become' (§12.2.2),  $y_{\partial}$  'along' (§11.1.5.),  $x^{w_{-2}}$  'move toward' (§12.2.7), and *h\_{\partial}n\_-*, *h\_{\partial}m\_-*, '*come'* (§12.4.10). To economize I will not gloss these affixes in the following three sections.

We will first consider terms referring to location or direction in relation to features of the environment and to the house, fire, and canoe, before considering terms of general reference.

#### 22.1.1. The Expression of Movement and Location in Relation to Water

The direction of movement and location on land or water are commonly indicated with words that refer to the shore and the flow of water. In many contexts, these words are the counterparts of cardinal directional terms in English. The basic terms are illustrated in the following diagram.



These terms<sup>1</sup> can be categorized as follows:

|   | DIRECTION                                             | LOCATION                                            |
|---|-------------------------------------------------------|-----------------------------------------------------|
| 1 | On land in relation to the shore                      |                                                     |
|   | <i>cám</i> 'go/come up inland'                        | $c\acute{a}l \partial q^w$ 'inland, up in the bush' |
|   | $\dot{t}\dot{a}x^{w}$ 'go/come down toward the shore' | <i>cécəw</i> 'down on the shore'                    |

<sup>1</sup> See Hess 1979a for comparable terms in Northern Straits and Lushootseed.

| 2 | On water in relation to the shore           |                                     |  |  |  |
|---|---------------------------------------------|-------------------------------------|--|--|--|
|   | <i>tá</i> · <i>l</i> 'go/come out seaward'  | <i>cówcow</i> 'out at sea, out from |  |  |  |
|   | 5                                           | shore, out on the tide flats'       |  |  |  |
|   | <i>lé·l</i> 'go/come in shoreward, ashore'  |                                     |  |  |  |
| 3 | On land or water in relation to the flow    | of water                            |  |  |  |
|   | <i>x<sup>w</sup>íwəl</i> 'go/come upstream' | <i>?əhíw</i> 'upstream'             |  |  |  |
|   | <i>wəqwiləm</i> 'go/come downstream'        | <i>ł ał ás</i> 'downstream'         |  |  |  |

The terms  $x^{wi}(wal)$  and  $wad^{wi}(lam)$  are used for movement both on the river and on land along the river. On Burrard Inlet, to go toward the head is  $x^{wi}(wal)$  'go upstream,' but to go toward open water is  $2\delta \dot{X}qa\theta$  'go out.' Out on Georgia Strait ("the Gulf"), according to JP, you can only specify where you are going or use the terms  $t\delta ywat$  'north' or yix 'south.' (See §22.1.1.4.) The terms  $t\delta \cdot l$ and  $t\delta \cdot l$  are used for movement on both the river and the salt water.

The directional terms are verbs, as in (a) and (b).

- (a) ni m³ yacak wcák wam. (CC)'They are going up from the shore.'
- (b) ?i wəłmi tátəx w. (JP)'He's coming down to the shore.'

The locational terms were used with articles as nouns by CC and JP, as in (c) and (d).

- (c) ni ni? [?ə] kwə cécəw. (CC)
  'He's down on the beach.'
- (d) <sup>9</sup>óý k<sup>w</sup>snemct sówqt k<sup>w</sup> pówoý [<sup>9</sup>o] to céwcow. (AC 1)
   'Let's go look for flounders out from shore.'

AG, however, used some of them as predicate heads and had plural forms, as in (e) and (f).

- (e) ni cécəŵ. (AG) 'He's down.'
- (f) ni cələwcəw. (AG) 'They are out offshore.'

Inflected and derived forms are given in the next three sections.

#### 22.1.1.1. Inland versus Shoreward

The directional terms  $c\acute{a}m$  'go/come inland' and  $t\acute{a}x^w$  'go/come toward the shore' are simple verb roots with progressive, causative, and other forms;  $c\acute{a}m$  has a progressive  $c\acute{a}k^w \partial m$ , plural  $c\acute{a}l\partial m$ , progressive plural  $c\partial k^w c\acute{a}k^w \partial m$ , and causative  $c\acute{a}mst \partial x^w$  'take/bring it up from the shore,' and  $t\acute{a}x^w$  has a progressive

 $t \dot{a} t \partial x^w$ , plural  $t \dot{a} l \partial x^w$ , progressive plural  $t \dot{a} x^w t \partial x^w$ , and causative  $t \partial x^w s t \partial x^w$ 'take/bring it down to the shore.' What appear to be forms in the dispositionaliterative aspect (§7.6) are  $c \partial m c \partial m$  'move one's residence inland' and  $t \partial x^w t \partial x^w$ 'move one's residence down to the shore.'

The root of the locational term  $c\dot{a}l\partial q^w$  is not identifiable;  $c\dot{e}c\partial w$  is from the root cew- 'shore.' These two also have verbal derivatives,  $h\partial nc\dot{a}l\partial q^w$  'get up to the house from the beach' (AG) and  $h\partial nc\dot{e}w \sim h\partial mc\dot{e}w \sim 2\partial mc\dot{e}w$  'get down to the shore' (JP). The latter has a transitive with  $-n\partial x^w$ , implied by a recorded passive  $2\partial mc\dot{e}wn\partial m$  'be brought down from the bush (as a deer),' a causative implied by a recorded passive  $h\partial mc\dot{e}wst\partial m$  'be brought down from the bush (as a deer),' a derivative (as a dead animal or the body of someone who had been lost),' and a derivative  $2\partial mc\dot{a}w\partial y\partial s$  'come back to life' (with  $-\partial y\partial s$  'circular figure').

As this last word implies, movement to and from inland can also refer euphemistically or metaphorically to movement to and from the grave, as in (a) and (b).

(a) niwł ném cómstom. (CC)

'They have taken the body to the graveyard.' (lit. 'He was taken inland.')

(b) ni <sup>9</sup>əmcáwəyəs kwə ni-ł <sup>9</sup>íkw. (JP)

'The one who was lost has (unexpectedly) returned.'

Reference to the bush (the woods) is more specific in  $x\dot{c}tlam$  'enter the woods' (cf.  $x\dot{c}at$  'put it in the bush,' sxixac 'the bush,' the root of which may mean simply 'thrust into'; cf. also  $x^w\dot{c}at$  'put it in the middle,'  $\check{s}x^wax^w\dot{c}c$  'with-in, among').

The two locational terms are the bases of the contrasting pair  $scalq^w alwal$ 'uphill side' and scaw alwal 'shore side.' The root *cew*- also appears in scaw elac 'out on the river bank' (also a place name at Musqueam) and scaw alaon 'Tsawwassen' (lit. 'shore margin,'  $< -a\theta an$  'mouth'), and in other derivatives given below.

# 22.1.1.2. Offshore versus Shoreward

The directional terms  $t\dot{a}\cdot l$  'go/come offshore/seaward' and  $t\dot{e}\cdot l$  'go/come shoreward' are composed each of a root (probably  $\sqrt{ta}$ - 'out'? and  $\sqrt{tel}$ - 'back, behind') and the suffix  $-\partial l$  'move toward.' The progressives are  $t\partial t\dot{a}\cdot l$  (CC) or  $t\partial t\dot{r}\cdot l$  (JP) and  $t\partial t\dot{r}\cdot l$ . (AG also used  $t\partial t\dot{a}\cdot l$ , but JP believed this was the Cowichan progressive.) Causatives are  $t\dot{a}\cdot l\partial st\partial x^w$  'take/bring it out from shore' and  $t\dot{e}\cdot l\partial st\partial x^w$  'take/bring it ashore.' (The root  $t\dot{e}l$ - also has a transitive form  $t\dot{e}lx$  – see §22.1.4.)

The locational term  $c\dot{\delta w}c\dot{\delta w}$  'out on the water, out on the tide flats' looks like a form of *cew*- 'shore' in the iterative aspect, but it is not clear why this might be so. This term also has a verbal form,  $x^w \partial c \dot{\delta w} ce \dot{w}$  'get out in the water,' as in (a). (a) ni čx<sup>w</sup> wəňánəw x<sup>w</sup>əčáwcəw. (JP)
 'You got out too far.'

An adjectival form appears in *scewcéwał*  $\dot{q}^{w}$ *alítaq* "offshore seagull" (possibly the long-tailed jaeger) (JP). There is a derivative *ciŵcawás* 'be looking out to sea' with *-as* 'face.'

A term  $t\acute{e}tt$  'ashore' (landed but still aboard the vessel as opposed to  $d^{w}tm$  'disembark') was given as the locational counterpart of  $t\acute{e}t$  'go/come ashore.' It appears to be  $t\acute{e}t$  plus -t 'stative'; the resultative is  $st\acute{e}talt$  'ashore, shore' (JP),  $st\acute{e}talt$  (AG). Derivatives are  $stalt\acute{a}twat$  'shore side (as the side of an island facing the mainland)' (JP) and  $sx^wt\acute{e}tlmax^w$  'sea hunter and fisher, expert at getting fish and sea mammals' (JP).

#### 22.1.1.3. Upstream versus Downstream

The directional term  $x^{wiwal}$  'go/come upstream' (progressive  $x^{wiwal}$ ) is composed of the prefix  $x^{w}$ - 'move toward,' the root  $\sqrt{hiw}$  'ahead,' and the suffix -al 'move toward,' while  $wad^{wilam}$  'go/come downstream' (progressive  $h\dot{sw}d^{walam}$ ) is composed of the root  $w\dot{sd}d^{w}$  'drift with the current' and the compound suffix -ilam 'move toward.' There is another, evidently less frequently used term,  $c\dot{sl}ax^{w}$  'go/come upstream' (progressive  $c\dot{sl}ax^{w}$ ), which appears in the names  $s\dot{c}al\dot{e}x^{w}$ , that of the upper part of the Musqueam village, and  $s\dot{c}alx^{wig}q^{w}$  'Chilliwack.'

The locational  ${}^{2}\partial h\dot{w}$  is from  $\sqrt{hiw}$  'ahead,' perhaps a resultative form without the usual *s*- 'resultative,' and  $t\partial t\dot{a}s$  is possibly from a root  $\sqrt{te}$ - or  $\sqrt{ti}$ -'back, behind.' (For other derivatives of *hiw*, see §22.1.8.4.)

The Cowichan counterpart of  ${}^{?}ahiw$  is tayt (< tay- 'upstream' and -t' 'stative'; cf. tayal 'go/come upstream'). The root  $\sqrt{tay}$  appears in several Musqueam terms, contrasting with wadw 'drift with the current,' as in staytalwat[sti?tawał] 'upstream side' versus sawdwatwat 'downstream side,' used for something outside other than a house and taytas [ti?tas] 'facing upstream' versus hawdwas 'facing downstream.'

#### 22.1.1.4. The Regional Axis

The terms  $t \delta y w \partial t$  and y i x are often identified as 'north' and 'south,' but actually refer to the two ends of the great inland sea that the Halkomelem language straddles,  $t \delta y w \partial t$  to the northern end of the Strait of Georgia and y i x to Puget Sound.

The term  $t \delta y w \partial t$  may be derived from the  $\sqrt{t} \partial y$ - of  $t \delta y t$  'upstream' and  $t \delta y \partial l$ 'go/come upstream,' the northern end of the Strait of Georgia being perceived as the upstream end of the waterway leading out the Strait of Juan de Fuca to the ocean. The term may also refer to a westerly direction, as when JP said,  $sk^{w}tk^{w}\partial m$  to  $t\delta yw\partial t$  'It's kind of red in the west,' and its derivative  $st\delta yw\partial t$ refers to a wind that varies from place to place (see §22.1.7). The related term  $x^{wt} \delta y w \partial l$ , usually glossed 'Northern Indians,' refers to the peoples from the northern end of the Strait of Georgia on northward.

The term yix (//yáx//?) may be cognate with the laš of Lushootseed  $(dx^wlašúcid)$  the language of Puget Sound. A derivative,  $x^wyixal$ , is said to refer to the people "from around La Conner south," that is to the speakers of Lushootseed, although sometimes the Samish, who spoke Northern Straits, are included. The adjectival form  $syixa^2t$  refers to the canoe type favoured by Puget Sound peoples.<sup>2</sup>

These terms also appear in two names for winds, given in §22.1.5. (For a more extended discussion, see Suttles 1987c.)

#### 22.1.2. In Relation to the House and Fire

The traditional house was a large wooden structure, rectangular in form and with a single-pitched (shed) roof, standing parallel to the shore. Until the latter half of the nineteenth century, it consisted of a permanent framework of posts and beams covered with removable roof and wall planks, which could be laid across canoes to make "rafts" and be used at other sites. A smaller house may have had an entrance at one end and another on the side facing away from the water; a larger house may also have had one or more on the side facing the water. In winter the walls were lined with mats, which could also be removed for summer use over poles as temporary structures. Inside the house, a bed platform ( $l\acute{e}lwe^2s$ , probably from  $\sqrt{le^2}$  'put away,'  $-elwe^2s$  'paddle') wide enough for two adults and a baby extended around the walls, and partitions extended partway out from it to divide the house into family sections. Each family had a fire within its section, but for ceremonial occasions these were combined into a single fire in the centre of the house.

After milled lumber became available in the late nineteenth century, the walls and roofs became permanent features of the house and two-pitched (gable) roofs were usual, but these houses were generally built with the same orientation. Some of these structures continued to be used for winter dances long after they had been abandoned as residences, until the middle of the twentieth century, when they were replaced by more "modern" structures. The term for any structure lived in is *lélam* 'house.' The term for a large old-style house, especially one used for the winter dance or other ceremonial activities, is  $s\theta \acute{e} \cdot wt x^w$ 'big house' ( $< \theta i$  'big,'  $-awtx^w$  '-house').<sup>3</sup>

<sup>2</sup> A Clallam, Sam Ulmer, identified nax yáš as the people of Whidbey Island, at Coupeville, that is, the Skagit proper. On the other hand, Gibbs (1863, vi) wrote that the Skagits called the Lummi "Nūkh-lésh."

<sup>3</sup> At the time that I was doing most of my research, the English term for this structure was either "big-house" or "smoke-house." In recent years, the term "longhouse" seems to have widely replaced these terms, but I do not remember ever hearing "longhouse" used by Native people before the 1960s or '70s. I believe it came into general use about the same time as "elder," which I also never heard until the 1960s or '70s.

Terms expressing direction and location in relation to the house are illustrated in the following diagram.



Direction in relation to the house is expressed with derivatives of the roots referring to the shore and the flow of water, with the suffixes  $-\dot{as} \sim -\partial s$  '-face, -front' and  $-\dot{e}\dot{x}\partial n$  '-upper arm, side.'

- *xcawás* (CC) (< *x*-, probably  $\check{s}x^{w}$  'oblique nominalizer,' *caw* 'shore') 'front (water side) of house' (lit. 'facing the shore')
- *xcəlqwéxən* (CC), *šxwcəlqwéxən* (JP) (< *cáləqw* 'inland') 'back (inland side) of house'

 $t \partial y t \dot{e} \check{x} \partial n$  [ti<sup>?</sup>té $\check{x} \partial n$ ],  $\check{s} x^w t \partial \dot{y} t \dot{e} \check{x} \partial n$  (<  $t \partial y t$  'upstream') 'upstream side (of house)'  $h \partial w \dot{q}^w \dot{e} \check{x} \partial n$ ,  $\check{s} x^w \partial w \dot{q}^w \dot{e} \check{x} \partial n$  (<  $h \partial w \dot{q}^w$  progressive of  $w \partial \dot{q}^w$  'go with the current') 'downstream side (of house)'

(I did not record 'front of the house' from JP but assume he would have used an initial  $\delta x^{w}$ - instead of CC's *x*-.) These terms are used as in (a) and (b).

- (a) ni ni? ə kwə xcəwás. (CC)
   'He is in front of the house.'
- (b) ni ni? ə kwə xcəlqwéxən. (CC)
   'He is in back of the house.'

They can be used to identify the corners of the house, as in (c) and (d).

- (c) ni ?ə kwe šxwəwdwéxən scəlqwéxən šxwłəkwéxəns kwe léləm. (AC)
   'He's at the lower back corner of the house.'
- (d) ni ni? > k<sup>w</sup>θ> štitéxэns k<sup>w</sup>θ> lél>m ni? > k<sup>w</sup>θ> šl>k<sup>w</sup>éx>n. (AC)
   'It's in the corner of the upstream side of the house.'

In (d), AC used  $\delta t \partial k^w \dot{e} \dot{x} \partial n$  for an inside corner, but JP distinguished  $\delta x^w l \partial k^w \dot{e} \dot{x} \partial n$  'corner of the outside of a house' from  $\delta x^w \dot{e} \dot{e} q \partial n$  'corner inside the house,' as in (e) and (f).

- (e) ni ní tə šx vcéqəň.'It's way in the corner.'
- (f) nistəx " čx " tə šx "céqəň. (JP)
  'Put it away in the corner.'

Within the house, direction in relation to the centre is expressed with some of the same terms or with terms derived from the same roots as those used in relation to water.

The term  $x^w i w \partial i$  'move upstream' also means 'move out into the centre of the house, toward the fire.' Other derivatives of  $\sqrt{hiw}$  (see §22.1.8.4) with related meanings are hiwx or  $x^w i w x$  (-x 'transitive') 'take/bring him/her forward (a person to the centre of the house in front of people), bring it out (as food stored in the back),'  $x^w i w x$  'move it over into the centre of the fire, move it onto the flame (as something being cooked),'  $2\partial h i w s t \partial x^w$  'place it at the edge of the fire, put it upriver,'  $x^w \partial x^w i w$  'front (as of the bed platform), side toward the fire,'  $x^w \partial x^w i w s t \partial x^w$  'keep it close to the fire,'  $x^w \partial x^w i w \partial q \partial n$  'front (of a person, that is, nearer the fire),' as in (a) and (b).

- (a) <sup>?</sup>i·ł <sup>?</sup>amət <sup>?</sup>i <sup>?</sup>ə tən x<sup>w</sup>əx<sup>w</sup>íwəqən. (CC)
  'He was sitting in front of me.'
- (b) ni x vəx víwəqən k s ?əmá·mət[s]. (JP) 'They're sitting in front of the bed platform.'

The term  $t\dot{a}\cdot l$  'move out onto the water, seaward' has a causative  $t\dot{a}\cdot l \rightarrow st \rightarrow x^{w}$ , which can mean 'move it out (as a table into the middle of the room).'

The root of  $l \notin l$  'move shoreward' appears in  $l = l \ell l m$  'move back (from the centre of the house toward the bed platform, into the crowd, away from the fire, as at a big event),' and  $l \notin l x$  'take/bring back (from the centre of the house, from the fire, as to seat a dancer),'  $l = l \ell l$  'back, toward the wall,' as in (c) and (d).

(c) ni łəłél. (JP)

'He is in the back.'

(d) ni łələłél k<sup>w</sup>is <sup>?</sup>əṁá·mət. (JP) 'They are sitting in the back.'

One can also use  $t \partial t a \cdot \hat{l} w \partial t$  'behind, on the side away from the fire,' as in (e) and (f).

(e) <sup>9</sup>i <sup>9</sup>ámət <sup>9</sup>i k<sup>w</sup>ən łəłá·lwəł. (CC)
'He's sitting behind me (in the big house).'

(f) nin łəłálwəł θən stáləs.

'My wife is behind me (nearer the wall in the bed).'

Terms for location relative to the speaker are derivatives of ?i 'be here' and ni? 'be there' (see §22.1.8.1), such as saqin' this end of the house' and snaqin' 'that end of the house,'  $s^2aha^2alcap$  or  $saha^2lcap$  'this side of the fire' and  $sna^2a^2alcap$  'that side of the fire,' sahalwat 'this side' and snahalwat 'that side.' The last is illustrated in (g).

(g) ném cən ce? kwə snə?álwəłs tə léləm.

'I'm going to the other side of the house.'

One can also use *loqqín* 'other end of the house.'

Going around inside the house is  $x^w \dot{q} \dot{a} q s \partial m$  'make a complete circuit,' a term also used for completing a full year. For the ceremonial circuit, see §22.1.3.

Coming or going in and out and inside and outside the house are expressed by the general terms  $k^w t \delta x^w$  and  $\sqrt{2\delta X}q$ - and their derivatives given in §22.1.8.5, and leaving and returning by those given in §22.1.8.3. There are two terms relating more specifically to the house:  $t \delta k^w$  'go/come home,' which has a causative  $t \delta k^w s t \delta x^w$  'take it home.' The term  $2\delta m \delta t$  'sit down (if standing), get up (if lying), be at home' has the derivational forms  $\delta x^w \delta m \delta t$  'house site, bed' and  $h \delta m \delta t$  'get home.'

Usually the houses in a village stood in a row along the shore. On a stream, the upriver end of the row was called  $stayt\acute{e}n$  'upper end' or  $t\acute{a}yta\check{x}an$  'upper side,' and the lower end was called the  $sa\dot{w}\dot{q}^w\acute{e}n$  'lower end' or  $w\acute{a}\dot{q}^wa\check{x}an$  'lower side.' The houses at the ends of the row at  $s\acute{c}al\acute{e}x^w$  at Musqueam were called  $t\acute{a}yta\check{x}an$  and  $w\acute{a}\dot{q}^wa\check{x}an$ . On the salt water, as at Tsawwassen, the ends of the village were  $stayt\acute{e}n$  'north end' and  $syi^2x\acute{e}n$  'south end.'

#### 22.1.3. The Ceremonial Circuit

As elsewhere in this region, the ceremonial circuit is counterclockwise. Participants in the *smíte*<sup>2</sup>, the winter dance, move around the house in a counterclockwise direction. The dancers in a  $s\check{x}^w \acute{a} y\check{x}^w \partial y$  performance circle counterclockwise. Participants in a Shaker service proceed around the church to shake each others' hands and pass before the prayer table in a counterclockwise direction. Even a lady serving tea to her friends, CC explained, should proceed around the table in a counterclockwise direction. To move clockwise is  $qi\dot{q}\partial\theta\partial t$  'bind oneself' ( $\langle qi\dot{q}$  'get bound'; cf.  $qi\dot{q}\partial t$  'bind it') and to move counterclockwise is  $y\dot{x}^w\partial\theta\partial t$  'free oneself' ( $\langle y\dot{x}^w$  'get free' (cf.  $y\dot{x}^w\partial t$  'free it, untie it'; cf. also  $y\dot{x}^w$  'thaw'). Thus, if you move clockwise you are tying yourself up, but if you move counterclockwise you are freeing yourself. The winter dancers moving counterclockwise around the house are "unwinding."

It seems likely that these reflexive terms are extensions in meaning arising from the practice of binding something, such as an arrow shaft, with a clockwise motion, requiring that it must be freed with a counterclockwise motion. (I am assuming that this was the ancient practice.) Once the meaning was extended from the action of binding and unbinding to the direction of the circuit, the implications of the terms would likely have reinforced the practice.

In the big house, there is another reason for moving in a counterclockwise direction, especially when you are entering for the first time and greeting people. If you turn to the right as you enter, you have the people seated along the wall at your right (the honoured side) and you can (in the modern tradition) shake their hands more easily.

The ceremonial circuit in this region contrasts with that of much of the rest of Native North America, where it is clockwise. This direction is often identified as "sun-wise" or "with the sun," because the sun is seen, especially at higher latitudes (in the Northern Hemisphere), as moving in a clockwise direction. (Indeed, it must be the movement of the shadow on a sundial that makes clockwise "clockwise.") The direction in this region, whatever its origin, is consistent with the importance of the house as the theatre of ceremonial activity, and with the absence of any special concern for the cardinal directions.<sup>4</sup>

#### 22.1.4. In Relation to the Canoe

A few words seem to apply specifically to movement and location in relation to and within a canoe, or in recent times perhaps to any conveyance.



<sup>4</sup> This discussion is taken from Suttles 1987c.

Terms for movement in relation to a canoe are:

(1)  $\dot{a}\cdot t$  (probably //?aləł//) 'embark, go/come aboard,' the durative (?) of which is  ${}^{?}\partial l \dot{a}{}^{?}\partial t$  'be aboard.' Compare (a) and (b):

- (a) ném cən ?á·ł. 'I'm going aboard.'
- (b) <sup>?</sup>i cən <sup>?</sup>əlá<sup>?</sup>ał. 'I am aboard.'

This has the causative  ${}^{2}\dot{a} \cdot tstax^{w}$  'put him aboard' and derivatives  ${}^{2}\partial t\dot{a}le^{2}cam$  'load aboard' and  ${}^{2}\partial t\dot{a}le^{2}ct$  'load it aboard' (<  ${}^{-}\partial le^{2}c$  'bundle'). (Compare the suffix  ${}^{-}a\cdot t$  'travel by,' §12.3.3.)

(2)  $\dot{q}^{w}im$  'disembark, go/come ashore,' which has the transitive  $\dot{q}^{w}im\partial t$  'let him off, take it out of the water,' the resultative  $s\dot{q}^{w}i\dot{q}^{w}\partial m$  'ashore, disembarked,' and the derivative  $\dot{q}^{w}im\partial le^{2}ct$  'unload it.'

Terms for position in a canoe are:  $\dot{q}t\dot{a}n$ ,  $s\dot{q}t\dot{a}n$  'bow,' ' $tle\cdot q$  'stern,' and ' $\delta nw\partial t$  'centre, amidships.' Example of use are:

- (c) ném čx<sup>w</sup> ?íle·q. (JP)'Go to the stern.'
- (d) nem čx<sup>w</sup> tə ?ʻánwəł. (JP) 'Go sit in the middle.'
- (e) ni shá·y k<sup>w</sup>θə šx<sup>w</sup>ní?s k<sup>w</sup>θe? hóỷq<sup>w</sup> ni ?ə tθe scécon wəsqłáns tə snóx<sup>w</sup>əł. (JP 23)

'A place was prepared for that fire in the very bow of the canoe.'

 (f) k<sup>w</sup>óyxöt tə šxqaqawał ni k<sup>w</sup>θə <sup>9</sup>íle·q, <sup>9</sup>óxəl scécan wətíməθət <sup>9</sup>óxəl. (JP 23)

'The partner in the stern acts and paddles with all his might.'

Relative positions fore and aft in a canoe are expressed by  ${}^{2}\acute{\theta}\partial q\partial n$  'forward,'  $sli^{2}\acute{a}q^{w}t$  or  $li^{2}\acute{a}^{2}aq^{w}tm\partial n$  'aft, behind.'

The two sides of a canoe are  $s^2 \partial y \dot{a} l w \partial t$  [s<sup>2</sup>i<sup>2</sup>álwə<sup>1</sup>] 'starboard side, right side' (<  $^2\delta y$  'good,' - $\dot{a} l w \partial t$  'side,' and  $s \dot{z} \partial k w \partial t$  'port side, left side' (<  $t^{\partial f} k w a^2$  'left').

The terms for movement by canoe are:

- ?áxəl 'paddle (v.; cf. sq́áməl 'paddle,' n.), row (a skiff),' also 'take food to one's affines' (probably composed of a root √?∂x- and the suffix -∂l 'move toward'), s?áxəl 'gift of food for affines,' ?∂xí·ls //?∂xəléls// 'tow,' ?∂xíləstəm 'be paddled (used of either a canoe or a person being taken somewhere ).'
- $w\dot{a}\dot{q}^{w}$  'drift with the current.' This is the root of  $w\dot{a}\dot{q}^{w}ilam$  'go/come down-stream' and other terms given in the previous sections.

*x<sup>w</sup>əwqínt* 'go ahead with it' (< *x<sup>w</sup>- hiw-* 'ahead,' *-qin* 'head, forward end,' *-t* 'transitive')

 $y\dot{a}^{2}t$  'back it up' (<  $\sqrt{ya^{2}}$  'reverse?' -*t* 'transitive')

- $\dot{x}\dot{a}\dot{\lambda}\partial\theta\partial t$  'turn aside, swing around' ( $\langle \sqrt{\dot{x}e\dot{\lambda}}$  'be at right angles'; cf.  $\dot{x}\dot{e}\dot{\lambda}\partial t$ 'cross him, contradict him,'  $\dot{x}x^w\dot{x}\dot{e}\dot{\lambda}$  'weir,'  $\dot{x}\dot{\lambda}\partial lw\partial t\partial n$  'canoe thwart' [CC]; cf.  $l\dot{x}\partial lw\partial t\partial n$  'canoe thwart' [JP])
- yəpənéc 'go with the wind' (< yə- 'along,' pənéc 'fair wind,' composed of pá 'get blown on' and -nəc ~ -néc 'base, butt')

yəqsín 'go against the wind' (< yə- 'along,' -qsən 'nose'?)

 $\dot{x} \partial \dot{x} \dot{A} \partial n \dot{e}^{2} e$  'go with a wind from the side' (<  $\dot{x} e \dot{X}$  'be at right angles,' - $\partial n \dot{e} c$  'ear, side')

The terms for 'move upstream' and 'move downstream' apply equally to movement on water and movement on land. There are very likely several, perhaps many, other terms relating to movement in a canoe and of a canoe that I have not noted.

# 22.1.5. Directions and Winds

While there are no simple terms for the cardinal directions, there are several terms identified as 'north wind,' 'west wind,' and so on. However, these glosses are inaccurate or incomplete. Direction is only one component of meaning; others are season, temperature, and precipitation. Nine winds were identified by either AC or JP or both:

(1) *stáywat* (< *táywat* 'north'), a wind that blows mainly in summer, from the west to northwest, and brings good weather. Both AC and JP identified it as 'west wind.' At the Musqueam village, it comes from over Point Grey, according to AC, from up the gulf (the Strait of Georgia), according to JP. Upriver, where some of the Musqueam used to go for a supply of fish (see §23.1), it provided ideal conditions for drying the fish; it is said that it blows upriver every day in the daytime and is pretty strong after 10:00 AM, lasting through the day.

(2)  $h \partial w \dot{q}^w \dot{e} t \partial t$ , a wind that blows downstream at night during the fishing season on the Fraser River. It is said to be caused by the flow of the river.

(3)  $sc \delta s$ , a wind identified by JP as a south-to-southwest wind and described as swinging around past Victoria and coming up through the gulf. The name means or implies that it swings around, he said.

(4)  $t \partial n c a d \partial q^w$  (ton- 'from,'  $c a d \partial q^w$  'inland, the "bush""), a southwest wind that blows from over Vancouver Island and may bring squalls. It is said to be called a "bush wind" by some whites on Vancouver Island.

(5)  $t\partial \dot{d}\partial \theta \partial c (t\partial l - from, \dot{\partial} \partial \theta \partial c bay')$ , a southeast or south wind that brings rain, sometimes for weeks at a time. At the Musqueam village, it comes from the southeast, "from where the sun rises," according to JP, from over San Juan Island (due south), according to AC. (The very destructive Typhoon Frieda of October 1962 was identified as this wind.) (6)  $t \partial n y \partial x$  (ton- 'from,'  $y \partial x$  "south," Puget Sound'), a south wind that sometimes but not always brings showers. Both AC and JP identified it as coming over Point Roberts (south southeast).

(7)  $t \partial ny \dot{a}t m \partial \ddot{x}$ , identified by JP as a warm and wet wind that blows from the south in the winter and fights the cold. It is sent by  $y \dot{a}t m \partial \ddot{x}$ , "a big, black, ugly man at the south end" who appears in a myth JP told. CC did not know the name. (See Suttles 1987c for more on the name.)

(8)  $s\acute{a}tac$ , a cold winter wind from the north to northeast. It blows only in winter and brings a cold snap with ice and snow. AC called the wind  $s\acute{a}satac$ , but JP called it  $s\acute{a}tac$  and used  $tams\acute{a}satac$  for a period of continuing cold weather.

(9)  $sq^w \delta \check{x}^w \partial m \partial x$  (<  $sq^w \check{x}^w \delta m \partial x$  'Squamish'), a north wind that may blow in the summer. JP said that a wind from the same direction as  $s\delta t \partial c$  in summer would not be called by this name because it would not be cold, but AG said a  $sq^w \delta \check{x}^w \partial m \partial x$  is also cold.

#### 22.1.6. Measurement of Space

Perhaps only artifacts were measured in earlier times. The *tét* 'fathom,' the space that can be covered by the outstretched arms, was the unit for measuring the length of a line or canoe or the length and breadth of a house. The *táłcəs* 'span' (< *tét* 'fathom,' -*cəs* 'hand'), the length covered by the outstretched hand, was the unit for measuring pieces of blanket and other smaller things.

The  $s \check{x} \dot{a} \dot{n} \dot{a}$  'foot' was also used by JP to describe the length of a harpoon shaft, among other things. He identified this unit as the distance between the knuckles of the little fingers when the two hands form fists with the thumbs out and touching at their tips. The length measured by one hand held this way is a  $ls\dot{a}\dot{q}s\check{x}\dot{a}\dot{n}\dot{a}$  'half foot' or  $l\dot{a}\dot{q}ws\check{x}\dot{a}\dot{n}\dot{a}$  'cut portion of foot.' It is possible, if not likely, that this is a calque of the English measure.

The distance between two places could be expressed by the time taken to travel there.

#### 22.1.7. Place Names

Musqueam place names do not differ in pattern from those of other peoples in this region. Typically there are names for every minor feature of the landscape that people know intimately. Often there are no names for some of the major features, such as large islands and bodies of salt water, probably because people's movements were directed toward specific places for specific purposes. But there are names for distant mountains and mountain ranges, perhaps because these were used in setting courses on longer trips or used to mark seasonal changes in the path of the sun.

Generally, most place names are analyzable, but there are always a few that defy analysis. These may be either old worn-down names in the local language or else names adopted from another language. One common kind of analyzable

name is descriptive of the place. Some are simple terms such as  $2\delta l qsan$  'point' for Point Grey and  $2\delta\theta nac$  'bay' for Burrardview. Some are composed of the name of some plant or animal found there with the prefix  $x^{w}$ - and the suffix  $-\partial m$ . The name  $x^w m \delta \theta k^w a \dot{y} am$ , that of the Musqueam village, is an example, being composed of  $m \delta \theta k^w a \dot{y}$ , the name of a plant, and the prefix and suffix. Another common type is a term that reflects some event in the remote past, especially one involving  $\dot{x} \dot{e} \dot{l}s$ , the Transformer, such as  $\dot{q} \dot{e} \dot{w} am$  'howling,' the name of a rock near Point Grey said to have been given this name because a dog there was going to bite the Transformer.

Appendix 2 contains a list of names of places and peoples.

# 22.1.8. General Terms

These are terms referring to direction and location without regard to specific features of the environment. They are ordered by concepts that are admittedly based on English and not necessarily on Halkomelem. While the roots listed here are general in usage, some of their derivatives refer to features of the environment.

# 22.1.8.1. The Concepts "Here" and "There" and "To" and "From"

The two pairs of verbs used as auxiliaries can also function as simple predicates, and each is the base of several other forms. The verbs ?i 'be here' and ni?'be there' are the bases of terms that contrast the here and the there, as in xwa?i'get here' versus xwani? 'get there'; xwa?inx' bring here,' xwa?ins 'reach it (here)' versus xwanins 'reach it (there)'; sahilwal 'this side' versus sna?ilwal'that side (of house, log, etc.)'; sahican 'this side' versus sna?ian 'that side, the other side'; ?ahis 'look this way, turn this way' versus na?is 'look toward,' na?asam 'go toward.' Others based on ni? include na?isan 'reach an end,' sna?iaban 'far side (of a road, creek, etc.),' sna?ilwal 'other side (of a log, etc.),' and sné·lic 'other side (as of a hill).'

The  ${}^{?}i$  and  $ni{}^{?}$  with the prefix  $\xi x^{w}$ - 'oblique nominalizer' and a possessive form what are essentially relative clauses (see §4.1.1.2) that express 'place where,' as in (a) to (d).

- (a) Xa nəšx<sup>w</sup>ní? k<sup>w</sup>ə nəsyá·ýəs. (AG)
   'That's where I work.'
- (b) Xa šx<sup>w</sup>ní<sup>2</sup>səł k<sup>w</sup>s yá·ýəss. (AG)
   'That's where he used to work.'
- (c) Xa šx<sup>w</sup>ní?s k<sup>w</sup>s k<sup>w</sup>ánnəx<sup>w</sup>s k<sup>w</sup>θə θíθə q<sup>w</sup>tá·yθən. (JP 23)
   'That's where they used to get the biggest sturgeons.'
- (d) təna léləm 'i šx<sup>w</sup>íct (JP 5)
   'this house where we are'

The verb 2i with the prefix tal- 'from' (see §12.4.3) forms tali (also recorded  $tali^2$  and tali from CC) 'be from,' as in (e).

(e) tə łwálap talí kwi šxwní?ct (JP 5)

'You people from down where we come from' (lit. 'You [pl.] who are from that [remote] place where we are,' said by speaker addressing fellow Musqueam in a speech made upriver)

The second pair of auxiliaries,  $? \partial mi$  'come' and  $n \acute{em}$  'go,' are derived from the first pair;  $? \partial mi$  is composed of ?i and the prefix  $? \partial m$ - 'come,' and  $n \acute{em}$  is composed of ni? and the suffix  $-\partial m$  'intransitive.' As a command,  $? \partial mi$  has an alternative form,  $? \partial m\acute{e}$  'come!' Both have causative forms,  $? \partial mi st \partial x^w$  'bring it' and  $n \partial^2 \acute{em} \partial s t \partial x^w$  'take it.' (The form of both the stem and the suffix of the latter indicate that  $n\acute{em}$  is not a simple root [see §10.3], as do the transitive  $n \partial^2 \acute{em} n \partial s$  'go after him' and a reflexive form,  $n \partial^2 \partial mn \acute{am} d$  'get to a destination.' The progressive  $h \acute{an} \partial m$  'be going' suggests the underlying form  $//n \acute{o}? \partial m//.)$ 

Motion toward is also expressed by  $\check{x}^w t \acute{e}^2$  'head for, go toward.' Its causative  $\check{x}^w t \acute{e}^2 s t \partial x^w$  seems to mean simply 'take it away.'

A pair of verbs, *técəl* 'arrive here, reach here' and *tás* 'arrive there, reach there,' indicate the end point of motion. The first, *técəl*, has not been recorded with transitive or causative suffixes, but *tás* has inflected forms with the sense 'near' (see §22.1.8.2).

Five of these verbs, 2i,  $ni^2$ ,  $2\partial mi$ ,  $n\acute{em}$ , and  $\breve{x}^wt\acute{e}^2$  and the derivative  $t\partial li$  are used as prepositional verbs with the senses 'in,' 'at,' 'to,' 'toward,' and 'from.' This use of  $t\partial li$  is seen in (f).

(f) mók<sup>w</sup> woscénom to mi técol tolí λ s<sup>9</sup>ámono. (CC 6)

'The ones who had come from Duncan were all Shakers.'

For examples of the others, see §3.5.

Hereafter in this section, terms are simply listed under semantic categories. The word or root (preceded by a bullet) is followed by inflected and derived forms and their glosses and, for most, by examples of usage. The symbol  $\sqrt{precedes}$  morphemes that I have recorded with affixes only, but some or all of these may be independent words not yet recorded.

#### 22.1.8.2. The Concepts "Near" and "Far"

- tás 'arrive there.' tsát 'approach it,' tasnéx<sup>w</sup> 'get close to it,' tsáθat 'approach,' statés 'near.'
  - (a) ni<sup>9</sup> stətés. (CC) 'He's close.'
  - (b) ném cən stətés. (CC) 'I'm going close.'

- $\sqrt{cim}$  'approach.'  $cim \partial l$  'get close,'  $cim \partial \theta \partial t$  'go near,'  $cim \partial t$  'put it close.'
- $\sqrt{pti}$  'beside.' s pti 'along side,' pti 'dest 'go alongside,' s pti 'deside each other.'
  - (c) ni<sup>?</sup> spłíd. (CC) 'He's alongside.'
  - (d) ni ct wəpłiqtəl 'al tə łníməł. (JP 22) 'We came alongside one another.'
  - (e) ni<sup>9</sup> ct <sup>9</sup>ámət spłíqtəl. (CC) 'We sat together.'
- $\sqrt{t}$  *i*  $\delta n$  'sit beside.' *t*  $\delta n \partial s t$  'sit beside him/her,' *st*  $\delta n \partial s$  'sitting beside,' *t*  $\delta n \delta s t \partial l$  'sit together' (cf. also  $\delta x^w t \delta n \partial x \partial n$  'neighbour').
  - (f) <sup>?</sup>i tən stə́nəs. (CC) 'He's sitting beside me.'
- cák<sup>w</sup> 'far.' x<sup>w</sup>cák<sup>w</sup> 'go far,' x<sup>w</sup>ck<sup>w</sup>íləm 'go far away,' x<sup>w</sup>ck<sup>w</sup>íləməstəx<sup>w</sup> 'take someone far away,' cáck<sup>w</sup>təl 'be far apart,' yəłcák<sup>w</sup> 'far behind (while moving)' (< yə- 'along,' and possibly l-2 'portion'). (For others see §22.2.5, "The Tide.")</li>
  - (g) scécon wocák<sup>w</sup>. (CC)
    - 'It's really far.'
  - (h) scék "əl k "s cák "s. (CC) 'How far is it?'
  - (i) ni<sup>9</sup> cən cák<sup>w</sup> x<sup>w</sup>ə<sup>9</sup>áý. (CC)
    'I'm much better.' (lit. 'I've become far better.')
  - (j) ni yəłcák<sup>w</sup> tə péstən 'ak Australia II. (AG)
     'The Americans were far behind the Australia II.'

# 22.1.8.3. The Concepts "Leaving," "Staying Away," and "Returning"

- √<sup>2</sup>ey- (?) 'away.' <sup>2</sup>éyəl 'come/go away,' s<sup>2</sup>éyəl 'be away,' <sup>2</sup>éyələs 'leave him/her/it.'
  - (a) <sup>9</sup>éyəl təli tə šx<sup>w</sup>k<sup>w</sup>cástən. (JP)
     'Come away from the window.'
- $\sqrt{k^w e \dot{y}}$  'stay away.'  $k^w \dot{a} \dot{y} \partial \partial t$  [k<sup>w</sup>á'i $\theta \partial t$ ] 'move away, stay away,'  $sk^w \partial \dot{y} \dot{e}$ ' 'stay away.'
  - (b) sk<sup>w</sup>əyé<sup>?</sup> čx<sup>w</sup> <sup>?</sup>al. (JP) 'Keep away.'
- $\sqrt{x \partial n}$  (?).  $x \partial n x \acute{en} \dot{x}$  out of the way, isolated' (this appears to be an *s*-less resultative),  $x \partial n x \acute{enst} \partial x^{w}$  'put it out of reach.'
  - (c) nem ct ce? k<sup>w</sup> xənxén. (JP)
    'We'll go out of their way (e.g., to talk privately).'

- *háye* (also recorded *háye*?) 'go away, leave,' *hayé*?*stax*<sup>w</sup> 'take away.'
  - (d) ném cən háye?. (JP) 'I'm going away.'
  - (e) ni yəhəyə.
    - 'He's leaving.'
  - (f) ni həliyé?. 'They went away.'
  - (g) ni yəhəyháya k wθə x wələnítəm. (AG)'The White people are going away.'
- hé ŵ a 'away, not at home, be hunting, absent.' <sup>2</sup> ahé ŵ a 'hunt deer,' naxwahé ŵ a 'land hunter' (cf. šxwłć lemaxw 'sea hunter'), sawí 'gone, away,' háwè nxw 'poor year (for salmon)' [absent season], opposite of qxé nxw 'good year' (> qá x 'many').
  - (h) ni? hé·ŵə. (CC) 'He's away.'
  - (i) ni səwí. (JP)'He's gone for a while. He'll be away for a while.'
- $\sqrt{\dot{q}en}$  'return.'  $\dot{q}\dot{a}nn\partial x^w$  'get it back,'  $\dot{q}\dot{a}nst\partial x^w$  'return it,'  $\dot{q}\dot{a}\cdot n\theta\partial t$  'return.'
  - (j) <sup>9</sup>ówə ct ce<sup>9</sup> wəňánət nem x<sup>w</sup>cák i mi ct ἀá·nθət. (JP 22)
     'We won't go far and we'll come back.'
  - (k) məqá·nθət tə s?ɨtən ni?əł nem s?éx we?tct təwkaləm. (CC)
     'The food that was our gift to them has returned.'
  - (l) ném čx<sup>w</sup> qá·nθət nem k<sup>w</sup>θe <sup>9</sup>əθx<sup>w</sup>?ámət. (JP)
     'Go back to bed.'

# 22.1.8.4. The Concepts "Before," "Behind," and "Pass"

- yəŵéň 'before, ahead, first (in both spatial and temporal senses).' yəŵánθət 'move ahead,' yəŵéňmən 'front.' (See also §18.4.20.)
- $\sqrt{hiw}$  'ahead (in space only?).'  $hiwaq^w$  'headman, leader' ( $-aq^w$  'head'),  $x^w \partial wqin\partial m$  'move up (as toward the pillow, to a child in bed)' ( $x^w$ -,  $-q\partial n$  'head, front end,'  $-\partial m$  'intransitive'),  $x^w \partial wqint$  'move it ahead,'  $x^w awqont$  'be moving it ahead.' For other derivatives, see §22.1.1.3 and §22.1.2.
- *loqet* 'hit the mark, get it just right.' *loqett* 'put it in place,' *loqett* 'get in the way, get on the right bearing,' *loqettol* 'jibe,' xwoloqet 'move in front, come into place,' *loqetmon* 'just opposite, directly in front (in line with something).'
- *yəléw*, *yəléw* 'pass (in space or time).' *yəléwx*, *yəléwx* 'pass him,' *yəléwəl* 'get past.'
  - (a) ni<sup>9</sup> yəlew. (CC) 'It's passed.'

- (b) ném cən yəléwəl. (CC)'I'm going by, I'm going to pass.'
- √*tel* 'back' (see §22.1.1 and §22.1.2).
- *ti<sup>2</sup>á<sup>2</sup>aq<sup>w</sup>t* 'behind, after, next, last (in both spatial and temporal senses).' *sti<sup>2</sup>á<sup>2</sup>aq<sup>w</sup>t* 'behind,' *ti<sup>2</sup>á<sup>2</sup>aq<sup>w</sup>tmən* 'behind.' (See also §18.4.21.)
  - (c) <sup>9</sup>i yəli<sup>9</sup>á<sup>9</sup>aq<sup>w</sup>t <sup>9</sup>əλ <sup>9</sup>é·nθə. (CC)
     'He's behind me (going along).'
  - (d) <sup>9</sup>i <sup>9</sup>ámət <sup>9</sup>i k<sup>w</sup>ən <sup>1</sup>i<sup>9</sup>á<sup>9</sup>aq<sup>w</sup>t. (CC)
    'He's sitting behind me (outside or in a canoe).'
  - (e) <sup>9</sup>i <sup>9</sup>i k<sup>w</sup>ən łə<sup>9</sup>á<sup>9</sup>aq<sup>w</sup>tmən k<sup>w</sup>sis <sup>9</sup>i <sup>9</sup>ámət.
    'He's sitting behind me (in a canoe).'
  - (f) ni co towyołcák w yołi?á?oqwt to na?onca?. (JP 14)
     'There was one who was kind of far behind the other.'

# 22.1.8.5. The Concepts "In," "Out," "Middle," "Between," and "Among"

- kwtóxw 'get inside, enter abruptly.' kwtéxwt 'take/bring it in,' skwtéxw 'inside (anything),' kwotxwílom 'go/come in.'
- $\sqrt{2\partial \dot{X}q}$  'move out.'  $2\partial \dot{X}q\partial l$  'go/come out,'  $2\partial \dot{X}q \dot{l}t$  'eject, throw out (as a sponger or a spouse),'  $2\partial \dot{X}q \dot{a}\theta$  'go out into open water from an inlet,'  $s^2 \dot{e} \dot{X}q$  'outside,'  $s^2 \partial \dot{X}q \dot{a} \dot{l} w \partial t$  'outside (as outside of the family).'
- $\sqrt{n\partial w}$  'enter.'  $n\partial w\partial x$  'insert it,'  $s\partial n w$  'inside,'  $\partial n w (n\partial s$  'middle,' etc. (This is the root of the Cowichan  $n\partial w i l\partial m$  'enter,' for which Musqueam has  $k^w \partial t x^w (l\partial m)$ , but Musqueam has both resultatives,  $s\partial n w$  and  $sk^w t e x^w$ , for 'inside.')
- $\sqrt{x^w}\partial c'$  'be within.'  $x^w c' \partial \partial d'$  'go among, go between,'  $x^w c' \partial t$  'insert it,'  $\check{x}^w (x^w \partial c'$  'within, among,'  $\check{x}^w c' \dot{w} \partial n$  'slipped in between' (-*iw* $\partial n$  'middle').
  - (a) ni? šx<sup>w</sup>əx<sup>w</sup>íć. (CC)
     'It's in between.'
  - (b) ni nem x̄<sup>w</sup>ənxénəm tə pús x<sup>w</sup>cáθət tə lətém i tə x<sup>w</sup>cénəctən. 'The cat ran between the table and the chair.'
- $\sqrt{x\partial c}$  'be within.'  $xic\partial t$  'put it in the woods,'  $xcil\partial m$  'enter the woods,'  $sxix\partial c$  'in the woods.'
- cłáq<sup>w</sup> 'pass through.' cłáq<sup>w</sup>θət 'go through,' cłáq<sup>w</sup>t 'poke it through,' scłáq<sup>w</sup> 'through.'
  - (c) niwəł cłáq<sup>w</sup>. (CC)
    - 'He is through.'
  - (d) n
     x
     x
     enxén
     m t
     pús cłáq
     v
     θ
     t t
     ifl
     ifle
     cat ran through the house.'

- √?eýe? 'middle?' šx<sup>w</sup>?éýe? or šx<sup>w</sup>?óye? 'middle,' šx<sup>w</sup>?éýe?stom 'be placed in the middle.'
  - (e) ni<sup>?</sup> šx<sup>w?</sup>éýe. (CC) 'It's in the middle.'

#### 22.1.8.6. The Concepts "Along," "By Way of," "Across," and "Around"

- $\sqrt{\dot{q}at}$  'go following a natural feature.'  $\dot{q}atn\acute{e}c$  'go around (as a lake or bay),'  $\dot{q}at\acute{a}\thetaan$  'go along the bank,'  $\dot{q}\acute{a}tast$  'go back and forth in front of it,'  $\dot{q}at\theta(namat)$  'go along through the village inviting people,'  $\dot{q}t\acute{e}wat$  'go across (a small bridge),'  $\dot{s}x^w\dot{q}\acute{a}tawat$  'foot bridge,'  $ya\dot{q}\acute{a}taqan$  'stay (be staying?) at the same level (on a hillside).'
  - (a) ném cən ἀətnéc tə xáca? (s²áθnəc). (CC)
     'I'm going around the lake (bay).'
- \$\sqrt{l\delta^2\ell}{e}\$ 'go by way of.' \$y\u00e9l\u00e9^2\ell\$ (JP), \$y\u00e9l\u00e9l\u00e9 (CC) 'go the same way, go by way of.'
  - (b) nem yələlé 'ə tə cécəw. (CC)'Go along the beach.'
  - (c) nem cə ce? nə́pəc yələlé x Andy. (DK) 'I'll send it with Andy.'
- $\sqrt{t}\acute{e}k^{w}$ -,  $t\acute{e}k^{w}\partial l$  'go across (a road, stream, the gulf).'
- √*l*∂*q*<sup>2</sup> · broad, opposite side, straight ahead.' *lqqt* · wide,' *lqqcos* · five' (< -*cos* · hand,' i.e., 'the spread-out hand'), *leqqmox* · flats, flat country (as the Fraser delta),' *lqqin* · other end (of a line),' *loqqlic* · other side (as of a peninsula),' *loqqqin* · other end of a house' (JP), *loqqqn* · head of a big river like the Fraser, people living way upriver as around Kamloops,' *lqqxon* · one house' [?], *lqqin* · one side of the mouth' (CC), *sloqqwol* · back (of body), back (location),' *loqqnc* · rump,' *x<sup>w</sup>lolqos* · go straight ahead.'

(e) wəx<sup>w</sup>łźłdos čx<sup>w</sup> <sup>9</sup>al. (JP)
 'Just go straight ahead.'

- sálč 'go around.' sálčstax<sup>w</sup> 'surrounded,' sálčalwał 'around,' salčalwált 'go around.'
  - (f) ném cən sólč. (CC)

'I'm going to go around.'

- (g) no x˜vonxénom to sqvoméy solč to lélom. (CC)
   'The dog ran around the house.'
- (h) ni? sélč tə qá?. (CC)'There's water all around.'

# 22.1.8.7. The Concepts "Turn," "Left," and "Right"

- $\sqrt{\dot{x}a\dot{x}}$  'be at right angles.'  $\dot{x}\dot{e}\dot{x}\dot{a}t$  'cross him, contradict him,'  $\dot{x}\dot{a}\dot{x}\partial\partial at$  'turn aside, swing around (as when travelling in a canoe),'  $y\partial\dot{x}\dot{a}\dot{x}\dot{x}\partial\partial at$  'be zigzagging (as while running),'  $\dot{s}x^w\dot{x}\dot{e}\dot{x}$  'weir (built across a stream),'  $\dot{x}\dot{x}\dot{a}lw\partial tan$  'thwart in canoe' (< -wd 'canoe,' -ton 'instrument'),  $\ddot{x}\partial \dot{x}\dot{x}\partial n\dot{e}e$ 'go with the wind from the side.'
- $\sqrt{cal}$  'turn, switch, reverse.' calq(n am 'turn a canoe around,' calasam 'turn the head, turn and look back,' calqat 'get him to turn around,' calq(nt 'make it turn around,' calcalq(nam 'topple over,' calcas 'switch sides when paddling,' calaws 'switch sides when felling a tree or a spring-board (- $tws \sim aws$ 'body').
- $t^{\theta}tk^{w}a$  'left.'  $st^{\theta}\partial k^{w}\partial^{2}tws$  (CC),  $t^{\theta}k^{w}\partial^{2}tws$  (JP) 'left hand, left shoulder, left side of body,'  $st^{\theta}k^{w}\partial^{2}twst$  'left side, left (direction).'
  - (a) násəm čx<sup>w</sup> tə  $\theta t^{\theta} k^{w}$ ə<sup>?</sup>álwəł. (JP)
  - Look to your left.'
    (b) ném čx<sup>w</sup> te<sup>9</sup>e st<sup>θ</sup>ík<sup>w</sup>ə. (CC)
    - 'Go to the left.'
- ?∂ýíŵs ~ s?∂ýíŵs 'right hand, right (direction).' (< ?áý 'good,' -iŵs 'body').</li>
  (c) ném čx<sup>w</sup> te?e s?∂ýíŵs. (CC)
  'Go to your right.'

# 22.1.8.8. The Concepts "Up" and "Down," and "Above" and "Below"

- \$\sqrt{cit-`above.' cicət`above, high,' scticən`top, surface,' ctict`put it on top,' sctidwət`above.'
  - (a) ni<sup>?</sup> cicəł. 'He's up above.'
  - (b) sq<sup>w</sup>əq<sup>w</sup>é ni<sup>?</sup> tə cícəł 'perforated at the top'
- $\vec{X} \delta p$  'deep.'  $\vec{X} p \ell l$  'go down, sink,'  $\vec{X} p \ell l x$  'sink it,'  $s \vec{X} p \ell q \partial n$  'bottom of a hill,'  $\vec{X} \partial p q \ell n \partial m$  'descend a mountain, go downhill,'  $s \vec{X} p \ell a l w \partial l$  'underside, underneath,'  $\vec{X} p a l w \ell \partial m$  'go under.'
  - (c) ni? **Åí**Åəp.

'He's down below (as someone in a hole). It's low (a ship in the water).'

- (d) ni? skpálwəł kwθə pípə ?ə kwθə púkw.
   'There is paper under the book.'
- (e) ni x<sup>w</sup>ənxénəm tə pús yəs<sup>3</sup>pálwəł tə léləm.
   'The cat ran under the house.'

The terms  $cic\partial t$  'above' and  $\vec{\lambda}i\vec{\lambda}\partial p$  'below' are also used attributively, as in  $cic\partial t sm\acute{e}nt$  'high mountain,'  $cic\partial t si^{2}\acute{e}m$  'God' (lit. 'Lord Above,' probably in origin a calque of the Chinook Jargon  $s\vec{\lambda}xali tayi$ ), and  $\vec{\lambda}i\vec{\lambda}\partial p h\vec{\partial}yq^{w}$  'Hell' (lit. 'fire below'). There is also a true attributive form  $scila^{2}t$  'of the heights,'

as in  $scilar t m m a^2 a q^w$  'birds of the heights (as vultures, ravens, golden eagles).'

- $k^w i^2$  'ascend.'  $k^w i^2 st \partial x^w$  'raise it,'  $k^w i^2 q \partial n$  'climb (a mountain), go uphill,'  $k^w \partial^2 as$  'look upward,'  $k^w \partial q s \partial n \partial m$  'turn up one's nose, sniff' (<  $-\partial q s \partial n$  'nose,'  $-\partial m$  'intransitive'),  $s x^w k^w i^2 t \partial n$  'ladder' (<  $s x^w$  'oblique nominalizer,'  $-t \partial n$  'instrument').
  - (f) ni? cən k<sup>w</sup>í?.'I climbed up.'
  - (g) <sup>?</sup>i cən k<sup>w</sup>ək<sup>w</sup>i<sup>?</sup>.'I'm climbing.'
- *ċe*<sup>2</sup> 'land on top.' *ċé*<sup>2</sup>*t* 'put it up on something,' *ċá*·*lwət* 'go over,' *ċá*·*ləc* 'go over (an island, a hill),' x<sup>w</sup>*ċén∂c∂m* 'sit on a chair,' *ċ*(*l∂m* 'climb up, mount (a horse),' *sċeċć*<sup>2</sup> 'on top, mounted (as on a horse), put up on something.' (h) ném cən ċá·*l*wəł. (CC)

'I'm going over (a log, canoe, etc.), to the far side (of a car, etc.).'

- (i) ném cən cá·ləc. (CC)
  'I'm going over (a hill, etc.).'
- $\sqrt{se^2}$  'raise.'  $se^{2t}$  'lift it up,'  $se^{2\theta \partial t}$  'raise oneself.'
- $\sqrt{x^w e}$  'lower.'  $x^w \acute{e}t$  'lower it,'  $x^w \acute{a}^2 \theta a t$  'go down' (lit. 'lower oneself'),  $\check{s} x^w a x^w \acute{e}$  'down below, on the ground,'  $\check{s} x^w a n \acute{e}c$  'seated on the floor,' and  $\check{s} x^w \acute{e} n a c$  'the drop-off (the edge of a shallow shelf, where deep water begins).'
- $\sqrt{hiq}$  'move under.'  $hiq \partial t$  'put it under, shove it out into the water,'  $si^2q$  'underneath.'
  - (j) ni<sup>o</sup>əł me sí<sup>o</sup>q tə lətém łə pús.'The cat was under the table.'
  - (k) ni x<sup>w</sup>ənxénəm tə pús yəsí?q tə léləm.'The cat ran under the house.'

#### 22.1.8.9. The Concept "Everywhere"

- $t \dot{e} \cdot t \dot{\theta}$  everywhere' (cf. Cowichan  $t \dot{e} t \dot{\theta} t$  'scatter it').  $t \dot{a} t \dot{\theta} \partial t$  'scatter (intr.),'  $t \dot{e} t \dot{a} t \dot{\theta}$  'go from one place to another,'  $s t \dot{e} t \dot{a} t \dot{\theta}$  'spread out, all over, everywhere.'
  - (a) ni<sup>?</sup> cən té t<sup>θ</sup> k<sup>w</sup>ən sówq <sup>?</sup>i <sup>?</sup>ówəte<sup>?</sup>. (CC)
     'I looked everywhere and there isn't anything.'
- *?əsé p* 'complete, completely (spatially), be finished (temporally).' *?əsó pnəx<sup>w</sup>* 'complete it (as something being studied),' *s?é?sə p* 'all over.'
  - (b) yôm quôt čx<sup>w</sup> ?əsép te ?eθləx<sup>w</sup>íws. (JP 21)
     'Rub yourself all over your body.'
  - (c) i ni <sup>?</sup>əsép tə să<sup>w</sup>əýém. (JP 1)'And the story is finished.'

- (d) s<sup>9</sup>é<sup>9</sup>səp təwǎa ni θáynəx »əs xpéy (JP 22)
   'all over that cedar he had made'
- $m \neq k^{w} \geq n \neq c \neq c \neq c \neq s = 17.17$ ).

# 22.2. TIME

The time of an event relative to the present is expressed by the particles  $-\partial t$  'past' and  $ce^2$  'future' (see §15). The use of  $-\partial t$  to express the past is not obligatory, but the use of  $ce^2$  for the future may be. Duration, frequency, and iteration are expressed by internal modifications of the verb root that distinguish perfective, progressive, durative, and other aspects.

The auxiliaries i 'be here' and  $ni^{2}$  'be there' may appear to refer to time, but, as explained in §3.2.1, this is only because 'the here' is more often 'now' and 'the there' more often 'then.'

The demonstrative system expresses spatial relations, but in some contexts may imply temporal relations. What is present spatially may be present temporally, as in *tona wéyol* 'today,' and what is remote may be past, as in  $k^w$  coléqol 'yesterday' (see §15.1.2).

Verbs of coming and going can be used in referring to a point in time. In an account of his family history, JP said:

 (a) ne<sup>9</sup>éməstəx<sup>w</sup> cən k<sup>w</sup>e <sup>9</sup> k<sup>w</sup>θə yəwénəł šx<sup>w</sup>ənÅas k<sup>w</sup>θən scáməq<sup>w</sup>əł. (JP 23)

ném-stəx<sup>w</sup> k<sup>w</sup>e ?ə k<sup>w</sup>θə yəwendə k cən go-CAUS I then OBL ART before-past šx<sup>w</sup>-x<sup>w</sup>ən-Ža-s nə-scámoqw-oł k»θə OBLNOM-still-BE3P-3POS ART my-great.grandfather-past 'I'll go back to the time of my late great-grandfather.' (lit. 'I'll take it to

```
the past when my late great-grandfather still was.')
```

And later he said:

(b) mal técəl tás k wθə swé?s swéyəl k wθə nəswé? nəmén. (JP 23)
?əmi ?al técəl tás k wθə swé?-s swéyəl k wθə come just arrive.here arrive.there ART own-3POS day ART nə-swé?-əł nə-mén. my-own-past my-father
'We come now to the time of my own father.' (lit. 'It arrives here to the day of my own late father.')

The compound prefix  $\delta x^{w}$ - 'oblique nominalizer' (§4.3.3, §12.1.4) with  $t\delta s$  'arrive there, reach' can refer to a point in space, as in (c).
- (c) ni tás k<sup>w</sup>θə šx<sup>w</sup>tás i nə x<sup>w</sup>əné<sup>9</sup>ent. (CC 1)
  - lit. 'They arrived where they arrived and it became night.' (translated freely by DK as 'They went as far as they could go by the time night came.')

Or it can refer to a point in time, as in (d).

(d) ... 'θłni·s tós k wθθ ni šx wtóss. (JP 17)
 whenever-AUX-3POS arrive.there ART AUX OBLNOM-arrive.there-3POS
 '... whenever there came a time for it'

That time itself "comes" is implied by (e).

(e) təňa ce? yəmi sxáxəłnet 'this coming week'

Adverbs are part of the lexicon, but have been described in their grammatical context in §18. There are some twenty-three adverbs and adverbial phrases that express frequency, duration, and relative time (see §18.4.1 to 18.4.23). See also  $y\partial w\dot{e} \dot{n}$  'before' and  $ti^2 \dot{a}^2 aq^w t$  'after' (in §22.1.8.4) and  $2\partial s\dot{e} \dot{p}$  'complete' (in §22.1.8.9).

### 22.2.1. Day and Night

The 'day' as a period of time is swéyəl (< wéyəl 'be day'), also 'daylight, sky.' The daylight is distinct from the 'sun,'  $syáq^w am$  (<  $yáq^w$  'burn'?). The daylight comes in the morning before the sun, which is, of course, not visible every day. The daylight was regarded as a powerful force, perhaps a deity. In a text dictated by JP, a grandfather instructs his grandson to face the direction from which the sun rises and before it does to ask the daylight to grant his wishes. The grandfather says:

(a) wənáy tə swéyəl wəθə<sup>γ</sup>ít xéxə. (JP21)
 'Only the Daylight is truly holy.'

The term *swéyəl* can also be translated 'weather,' as in (b).

(b) wəscék vəl 'al tə swéyəl. qəlqələm tə swéyəl. (JP)
'How is the weather?' 'It's changing for the worse.'

(Possibly related to this sense is *cwəywéyəl* 'get caught in a storm.')

The night as a period of time is  $sn\acute{e}t$  (<  $n\acute{e}t$  'be night'). The two roots  $w\acute{e}yel$  and  $n\acute{e}t$  are the basis for terms for periods of the day and night:

yəwéwəyəl 'dawn' (lit. 'becoming daylight') táx<sup>w</sup> swéyəl 'noon' (< táx<sup>w</sup> 'adjust, fit, be just so') yəléw táx<sup>w</sup> swéyəl 'after noon' nétəł 'morning' (lit. 'night-past') x<sup>w</sup>əné<sup>2</sup>ənt 'evening' (lit. 'becoming night') wəlhíθ snét 'late at night'

Other terms for the break of day are  $l\dot{q}\delta n\partial p$  'daybreak,' referring to the coming light, not to the sun (etymology unclear; JP said that  $-\partial n\partial p$  was not the suffix for 'ground'), and  $t\dot{e}w\partial l$  'dawn, get light,' recorded in  $x^wna^2$   $\dot{m}i$   $t\dot{e}w\partial l$ 'when daylight came' (cf.  $st\dot{e}t\partial w$  'bright,'  $t\dot{e}w\partial t\partial m$  'be lit, have light thrown on').

Present, future, and past days are expressed differently. 'Today' is  $t \partial n a w \dot{e} y \partial l$ , lit. 'this day.' 'Tomorrow' is either  $w \partial w \dot{e} y \partial l \partial s$  (lit. 'when it becomes day'), a conditional clause, or  $w \dot{e} y \partial l c e^2$  ('day will break'), a main clause accompanied by a conditional clause (see §4.2.4 for examples). 'Yesterday' is  $k^w c \partial l \dot{e} q \partial l$  (composed of the article  $k^w$  implying remote or absent status, a root not now identifiable, and, probably, the particle  $-\partial l$  'past'). Other such terms are:  $k^w \partial n a l \dot{\partial} w \partial l n e^2$  'the day before yesterday' ( $< k^w \partial n a$  'that absent,' possibly  $l \dot{e} w$  'escape' +  $-\partial l n e^2$  ?),  $k^w y \partial w \dot{e} n \partial l$  snét 'the night before last,'  $\theta \dot{a} m \partial n t$  'day after tomorrow,' lit. 'two days,' i.e., 'in two days' ( $< \theta \partial m - \sim \theta e m$ - 'two' +  $-\partial n t$  ?).

Days are usually counted with the simple numerals, but the following two forms are also used, perhaps usually in counting ahead:  $\theta \neq m \neq nt$  'two days,' meaning also 'day after tomorrow,' and  $t \neq x^w \neq nt$  'three days.' For - p t nt see §22.2.3, "The Week."

The rising of the sun or moon is expressed by  $k^w \dot{a} \dot{n} \partial s$ ,  $k^w \dot{a} \dot{n} \partial s \partial m$  (>  $k^w \dot{e} n$  'go over,' - $\partial s$  '-face') 'rise, come into view,' as in (c), (d), and (e).

- (c) wək wán əs tə syáq wəm. (AG)'The sun has risen over the hills.'
- (d) wəyeł smis k<sup>w</sup>ánəsəm tə łqélċ. (AC)
   'The moon is just coming up.'
- (e) mi yək wá ?k wənəsəm tə syá q wəm. (CC)
   'The sun is rising.'

A term, wəłqoyilt 'be up' (etymology unclear) was also used by JP in (f).

```
(f) ⁹ómət ⁹é·ł də, wəłdəyílt.
```

'Get up. The sun is up (said by a mother to children).'

The setting of the sun or moon is expressed by  $\dot{t}^{\theta} \delta \check{x}^{w}$  'get used up, worn out, burnt up,' as in (g) and (h) ...

- (g) ném yət<sup>0</sup>ət<sup>0</sup>áx<sup>w</sup> tə syáq<sup>w</sup>əm. (CC)
   'The sun is setting.'
- (h) ni wəłt<sup>0</sup> jăx<sup>w</sup> tə łqélč. (JP)
   'The moon has set.'

- ... by *tonnoc* (< *ten* 'disappear,' -*noc* '-base') as in (i) ...
- (i) niwəł nem tánnəc tə łqélč. (AC)'The moon has just gone down.'
- ... and by  $k^w en$  'go over,' as in (j).
- (j) ni k<sup>w</sup>én tə syáq<sup>w</sup>əm. (AG)'The sun set.'

If the moon rises later than expected, it is said that:

(k) niwəł łək̃<sup>w</sup>xən tə łqélċ. (AC)'The moon got tripped.'

Answering a question on eclipses, AC gave (l) and (m).

- (l) ni<sup>9</sup>yeł [ni<sup>9</sup>əł] wəłtatəl tə łqélc i tə syaq<sup>w</sup>əm.
  'The moon and the sun have gone behind each other.' (a solar eclipse)
- (m)x<sup>w</sup>ciθət əỷeł tə timəx<sup>w</sup> (ə) tə syáq<sup>w</sup>əm i tə łqélć.
  'The earth comes between the sun and the moon.' (a lunar eclipse)

These sentences must reflect a recent understanding of the phenomena.

#### 22.2.2. Time by the Clock

For several generations, speakers of Halkomelem have reckoned the time of day by hours with clocks. The terms for 'hour' and 'minutes' are  $2\dot{a}w\partial$  and  $m\dot{a}n\partial c$  from English. The term for 'clock, o'clock' and 'bell' is  $tint\partial n$  from Chinook Jargon. Examples of usage are:

- (a) scék<sup>w</sup>əl k<sup>w</sup>ə Å™ə <sup>9</sup>əsqéqəwəłtən. náčas °∂v k<sup>w</sup>∂ how then ART your-be.paying-people. one-face and ART k\*∍ łsźd néča? <sup>?</sup>áwə. (CC) half ART one hour 'How much are you paying?' 'A dollar and a half an hour.'
- (b) tx<sup>w</sup>?ápən mónəc i ?ápən i k<sup>w</sup>ə nóča?. (CC) remaining-ten minutes and ten and ART one 'It's ten minutes to eleven.' (lit. 'There are ten minutes more and it will be eleven.')
- (c) niwł słóm  $\vec{\lambda}$  ?ápən i  $\vec{k}$  wə nóča?. (CC) AUX already-worn.off OBL ten and ART one 'It's half past eleven.'
- (d) táx<sup>w</sup> wiléw nóća?. (JP) táx<sup>w</sup> wo-yoléw nóća? just.at EST-pass one 'It's half past one.'

(e) tá·x<sup>w</sup> wəłsód tíntən k<sup>w</sup>s yəlews xa?áθən tə x<sup>w</sup>əneont. (AC)
 'It was just half past four in the afternoon.'

### 22.2.3. The Week

The week was introduced along with Christianity in the nineteenth century, and so the names of the days of the week must have been devised at that time. These names are:

 $s\check{x}\check{a}\check{x}int$  'Sunday,' lit. 'holy day,'  $<\check{x}\check{e}^{?}\check{x}e$  'holy'  $sya\check{l}\acute{e}wainet$  'Monday,' lit. 'after day,'  $< ya\check{l}\acute{e}w$  'past'  $s\theta\acute{a}mants$  'Tuesday,' lit. 'second day' stixws 'Wednesday,' lit. 'second day'  $s\check{x}a^{?}\acute{a}\thetaans$  'Thursday,' lit. 'fourth'  $st\check{q}\acute{e}cas[s?]$  'Friday,' lit. 'fifth'  $t\check{q}w\acute{a}tam$  'Saturday,' lit. 'be cut off'

The word  $s\check{x}\check{a}\check{x}inter$  is also used for 'week,' as in  $tinas s\check{x}\check{a}\check{x}inter$  'this week,'  $k^w\theta i$  in  $h\check{a}yi$   $s\check{x}\check{a}\check{x}inter$  'last week' (lit. 'the week that has finished'),  $tinas ce^2 yint s\check{x}\check{a}\check{x}inter$  'this coming week,'  $n\check{a}\check{c}a^2 s\check{x}\check{a}\check{x}inter$  'one week,'  $is\acute{e}i$   $s\check{x}\check{a}\check{x}inter$  'two weeks.'

The ending *-ətnet* also appears in *tásətnet* 'week' (< *tás* 'arrive there'?), *náčastnet* 'one week' (presumably a contraction of *náča*? *sžážatnet*), *sk*<sup>w</sup>*ántnet* 'what day of the week is it?' (< *k*<sup>w</sup>*ín* 'how many'), and *tax*<sup>w</sup>*ətnet* 'three days' (< *tíx*<sup>w</sup> 'three'). Thus it appears that we may have here a new lexical suffix with the possible meanings 'week,' 'day of the week,' and 'day.' A similar ending appears in *k*<sup>w</sup>*oňa towotne*? 'the day before yesterday' (< *téw* 'flee'?). (See also §13.3.)

### 22.2.4. The Month

The month, that is, the lunation or lunar month, is lqélc 'moon,' reckoned as beginning with the new moon,  $\check{x}\acute{ews} lq\acute{elc}$ . Phases of the moon were given by AC as follows:

- (a) wəłxéws θə łí?łqəlć.'The moon is new.'
- (b) natəwł yəqəqác tə łqélć.'The moon is in the first quarter.' (lit. 'The moon is swelling.')
- (c) wəcəlél i x<sup>w</sup>əłsóq tə łqélć.
  'It's a quarter moon.'
- (d) ni? wəłx wəłséd tə łqélć.'It's a half moon.'

- (e) niwəł yəléwəłsód [yəléw wəłséd ?] tə łqélć.'The moon is three-quarters full.'
- (f) wəłqśc tə łqélć. 'The moon is full.'

Months may be identified as  $t \partial n a \, lq \ell l c$  'this month,' and  $t \partial n i w \partial t \, \ell^{\theta} \delta \check{x} \, lq \ell l c$  'last month.' JP could give no further such terms.

Months are counted with the suffix  $-\partial s$  'face, moon, dollar,' as in  $2is \dot{a}l \partial s$  'two months, two dollars' ( $\leq y \partial s \dot{e}l \partial$  'two'). A pregnant woman might be asked (g).

```
(g) ni čx^w wəłk^wínəs.
```

'How many moons have you gone?'

And the answer might be (h).

```
(h) 'i cən wəłxəθínəs.'I have four moons.' (i.e., 'I am four months along.')
```

The mother of a baby might be asked (i).

(i) niwəł kwínəs łqélċ təna skíkqəł.
 'How old is the child?'

And the answer might be (j).

(j) niwəł <sup>9</sup>isáləs łqélć. 'Two months.'

For the names of the months, see §22.2.8.

### 22.2.5. The Tide

It is important to know the state of the tide and to predict the tide for shellfish gathering, several kinds of fishing and hunting, and travel on water, both on the sea and on the river. Pacific tides have diurnal inequality; each day, full highs and lows alternate with half tides. As everywhere, the full highs and lows vary through the lunar cycle, the highest and lowest coming with the new moon and the full moon. In addition to the monthly cycle, there is an annual cycle. The timing of the highest highs and lowest lows moves through the year, the lowest tides coming during the middle of the night in winter and the middle of the day in summer, the lowest of all being near the winter and summer solstices.

There is no term corresponding precisely to English 'tide,' but  $q\dot{a}^2$  'water' is often so translated, as is the suffix  $-\partial c \dot{a}$  'water.' Terms for tidal conditions are  $sq\dot{a}m\partial l$  'flooding tide,'  $q\dot{a}q\partial m\partial l$  'be flooding,'  $s\dot{t}^{\theta}\dot{e}m$  'ebbing tide,'  $t^{\theta}\dot{e}t^{\theta}\partial m$  'be ebbing,'  $s\partial l\partial t\dot{c}$  'high tide' (<  $l\dot{a}\dot{c}$  'full'),  $l\dot{a}\dot{c}l\partial \dot{c}$  'be just high water,'  $sx\dot{a}mx\partial m$  'low tide,'  $x^w\dot{c}\dot{a}k^w$  'long run-out (of tide),'  $c\dot{a}k^wc\partial k^w$  'low water,'  $ck^w\dot{a}l\partial s$  'place that has lots of sandbars and/or mudflats,'  $x^w\dot{A}\partial n\partial x^w\partial m$  'be slack water (that is, at the turn of the tide).' The period between the new and full moon is a

period of half tides called  $k^{wi}k^{w}ec \partial m$ , possibly from  $k^{w}ec$  'see,' because this is a time when the water in the river is clear and fish can see nets.

The relevance of moonlight to activity on the shore or just offshore is reflected in the terms  $lq \delta lc \delta lc \delta$  'moonlit low tide' ( $\leq lq \ell lc' moon$ ,'  $-\delta lc \delta$  '-water') and  $\theta \delta' l \delta lc \delta$  'low tide on a moonless night' ( $\leq \theta \ell^2 t$  'dark').

The annual cycle is reflected in the terms  $n \delta t c \partial m$  'shift from low tides in the daytime to low tides at night (in the fall)' (<  $n \delta t$  'become night,' -c '-back, -surface,' - $\partial m$  'intr.') and  $w \delta y \partial c \partial m$  'shift from low tides at night to low tides in the day (in the spring)' (<  $w \delta y \partial l$  'become day').

Tidal conditions are important not only on the saltwater shore but also on the river. The Fraser rises and falls with the tide. When the river is low, the flood tide is said to produce slack water as far up as Mission, about 80 kilometres (50 miles) from its mouth and at about the boundary between the Downriver and Upriver dialects of Halkomelem. Pitt Lake in Katzie country has a tide. Under optimal conditions, as in December, when the river is lowest and the tides highest, it is said that taking a canoe from Musqueam, you could catch the up-bound tide and on one flood tide make it to the Katzie village at Port Hammond. During the summer freshet, however, the water slacks to New Westminster only.

### 22.2.6. The Year

The word for 'year' is syalanam (cf.  $syalaxwa^2$  'old person,' yalew 'past,' suffix not identifiable). It appears in tana syalanam 'this year.' Another word, spanwa (spanwe??) ('another year'?), appears in  $waspanwa^2as$   $ce^2$ ~  $waspanwe^2es$   $ce^2$  'next year' (JP),  $k^w spanwa$  'last year' (JP), and  $k^wanad$ spanwa 'last summer' (DK).

The suffix  $-i \cdot nx^w \sim -wi \cdot nx^w$  'year' is suffixed to numerals to count years and forms a few other terms:  $n\partial cawi nx^w$  'one year,'  $\partial \partial mi \cdot nx^w$  'two years' ( $< \partial em - \partial \partial m$ - 'two'),  $k^w \partial n\partial wi \cdot nx^w$  'how many years?' ( $< k^w i n$  'how many?'),  $k^w n\partial wi \cdot nx^w \sim k^w n\partial w \cdot i nx^w \partial t$  'year before last' (the article  $k^w$ ,  $ni^2$  'be there'),  $k^w x \partial w i \cdot nx^w$  'age (of someone or something)' (cf.  $k^w x \partial t$  'count them').

Years may also be counted as  $méqe^2$  'snow (on the ground),' as in the response (b).

- (a) ni čx<sup>w</sup> wəłk<sup>w</sup>ənəwi·nəx<sup>w</sup>.'How old are you?'
- (b) ni cən wəłxəθənłčé méqe?.'I am forty years old.' (lit. 'forty winters')

To the question asked about a child, the answer might be (c).

(c) ni məwł nəċáŵinx<sup>w</sup>.'One year.'

Or  $x^w \dot{q} \dot{a} q s \partial n$  'make a circuit' may be used, as in (d).

- (d) ni məwł tás k<sup>w</sup>θə šx<sup>w</sup>qáqsəns. (JP)
  - 'He has reached the day he was born, he is exactly one year old.' (lit. 'He has reached his complete circuit.')

The age of an object can be expressed the same way as that of a person.

(e) ni<sup>9</sup>əs ctwa<sup>9</sup> wəłk<sup>w</sup>ənəwi<sup>•</sup>nəx<sup>w</sup> tə léləm. (JP)
<sup>•</sup>I wonder how old the house is.<sup>°</sup>

Years can also be counted with the ordinary word for 'year,' as in (f).

- (f) niwł łąćcəs syiəlánəm kwənsni šxwceléxwəm.
  - 'I have been a dancer for five years.' (lit. 'It has been five years that I have been a dancer.')

### 22.2.7. The Seasons

A division of the year into four seasons may not have been Aboriginal. Both CC and JP gave  $t \partial m k^w a \dot{l} \partial k^w \partial s$  'summer' ( $t \partial m$ - 'season'; cf.  $k^w a^2 k^w \partial s$  'hot') and  $t \partial m \check{x} \partial \dot{y} \dot{X}$  'winter' ( $\langle \check{x} \partial \dot{y} \dot{X}$  'cold'), and for 'this summer' JP gave  $t \partial n \dot{a} s k^w a \dot{k}^w \partial s$  (s- 'nominalizer,'  $k^w a \dot{k}^w \partial s$  'hot'). For 'spring' CC gave a descriptive sentence, (a).

(a) <sup>9</sup>əmí yək<sup>w</sup>ék<sup>w</sup>ələs.

'It's becoming warmer.'

But JP gave  $\dot{q}^{w}\dot{l}\delta s$  'spring' (lit. 'uncover'; cf.  $\dot{q}^{w}\dot{l}^{2}\delta lt$  'uncover it,'  $-\delta s$  'face') or  $t\delta m\dot{q}^{w}\dot{l}\delta s$  'spring.'

For fall CC gave  $haylénax^w$ , which JP said was the name of a month; he also said, "There is no real word," but you can refer to  $x^{wisalénax^w}$ , which is the name of a moon, or you can say yalqánap (< yeq 'topple over' + -anap 'ground'), which means that "the grass and other things go down."

Periods of time are distinguished by means of a prefix or one of two suffixes. The prefix is  $t \partial m - \sim t \partial n - \sim t \partial n$  'season for, time of,' which appears in the words for 'summer' and 'winter' given above and in the following:

*təmlíle*? 'salmonberry season' (< *líle*? 'salmonberry')

 $t \partial m \theta \partial q \partial y$  'sockeye season' (cf.  $s \theta \partial q \partial y$  'sockeye')

 $t \partial m \vec{k}^w \dot{a} \vec{l} \partial x^w$  'dog [chum] salmon season' (<  $\vec{k}^w \dot{a} \vec{l} \partial x^w$  'chum salmon')

*təmsásətəc* 'a period of continuing cold weather' (< *sásətəc* 'cold northeast wind')

 $t \partial m q \partial q \dot{a}^{2}$  'the freshet (when the Fraser rises in the spring)' (<  $q \dot{a}^{2}$  'water')  $t \partial m \dot{t}^{\theta} \dot{t}^{m}$  'berry-picking time' (<  $\dot{t}^{\theta} \dot{t}^{m}$  'pick berries'; cf.  $s \dot{t}^{\theta} \dot{t}^{m}$  'berries')  $t \partial m h \dot{a} p s$  'hop-picking time' (< E. hops)

tənmilə ~ təmmilə 'winter dancing season' (< milə ~ mile<sup>9</sup> 'dance with one's syáwən') *tənqwətqwətxən* 'period of continuous fog' (cf. *sqwətxən* 'fog,' *qwiqwtimxən* 'drizzle')

*wəłtəm?íyəs, təm?i?íyəs, təm?əẏ́əẏ́əs* (< *?íyəs* 'enjoy oneself') in early spring when the weather improves (JP)

The two suffixes are probably related. One is  $-\partial l en \partial x^w \sim -l e n \partial x^w$  'season,' which appears in:

 $x^{w} \partial s \partial l \dot{e} n \partial x^{w}$  'a month in the fall when leaves fall' (cf.  $x^{w} \dot{i} s \partial t$  'shake it')  $h \partial y l \dot{e} n \partial x^{w}$  'the following month in the fall' (<  $h \dot{a} y$  'stop, be finished')

The other is  $-\dot{e}\cdot nx^w \sim -\dot{e}\cdot n\partial x^w \sim \dot{e}n\partial x^w$  'fish, fish run, season,' and possibly 'food,' which appears in:

 $p\vec{k}^{w}\vec{e}\cdot nx^{w}$  'smoke fish (as salmon)' (cf.  $p\vec{e}\vec{k}^{w}\partial t$  'warm it')

 $\dot{c}\partial\dot{y}x^{w}\dot{e}\cdot n\partial x^{w}$  'smoke little fish (herring, etc.)' (<  $\dot{c}\partial\dot{y}x^{w}$  'dry')

 $q \check{x} \acute{e} \cdot n x^w$  'big run of fish, good year' (<  $q \acute{a} \check{x}$  'many')

 $t \partial mh \dot{u} \cdot \dot{n} e \cdot n x^w$  'humpback year' (<  $h \dot{u} \cdot \dot{n}$  'humpback [pink] salmon,' which run in alternate years)

 $\check{x}e\dot{t}^{\theta}\acute{e}nax^{w}$  'Mt. Baker' (lit. 'measures the fish run' or 'measures the season'? cf.  $\check{x}\acute{e}\cdot\dot{t}^{\theta}t$  'measure it')

## 22.2.8. The Calendar

In earlier times the moons were named, but neither of my principal sources, AC (with CC) and JP, could give very many names for the moons. JP gave the names of seven moons; AC remembered only four, three the same as JP's, for a total of eight names. These are, of course, names of lunar months, which cannot correspond to the names of months in the Gregorian (solar) calendar, hence my sources identified them as coming "about" when a Gregorian month came. The names were:

x<sup>w</sup>əsəlénəx<sup>w</sup> (CC), x<sup>w</sup>isəlénəx<sup>w</sup> (JP) (cf. x<sup>w</sup>ísət 'shake it,' referring to the wind blowing leaves off the trees) around October
həylénəx<sup>w</sup> (< háy 'stop, be finished') around November (JP)</li>
xcəlwéstən (CC) (probably xcəlwé<sup>2</sup>stən 'put paddles away,' < √ of xícət 'put it in the bush' + -əlwe<sup>2</sup>s 'paddle') about December

pənéq (CC, JP) (probably pənéq 'slow match') about January<sup>5</sup>
mímən lqélč ('small moon') (JP) about February
wółžəs "when you begin to hear the frogs" (JP) about March

<sup>5</sup> JP believed that the name means 'covered,' from  $p \delta n$  'be buried,' referring to snow on the ground, and he added that the name was avoided by women. Probably the suffix has been identified as that for 'penis.' However, Lummi, Katzie, and Cowichan sources all identified this moon name as the term for 'slow match,' a device for carrying fire in a canoe, a coil of twisted cedar bark rope lit at one end, which was kept in a horse-clam shell.

*li*·*ms* (cf. *sli*·*m* 'sandhill crane') when the sandhill cranes come to the cranberry bogs, about May (JP), *li*·*ms* 'September'? (AC/CC) (They came for the "crane-berries," JP said.)

*Žéqt łqélč* ('long moon') "about June or July" (JP)

In earlier times there may have been more names for moons. However, there may not have been complete agreement on which names to use for each moon, different families naming them differently according to their activities, and there may have been names for no more than ten moons.

The problem with naming moons after natural events, like the croaking of frogs or the migration of cranes, is that these events are determined by the solar year. The only natural events determined by the phases of the moon are the tides. The solar year is about ten days longer than twelve lunar months, and if we had names for twelve moons and counted them continuously, after three years it would appear that the frogs and cranes were both a month late. In a few more years, the names of the moons would no longer make any sense at all.

One way of adjusting a lunar calendar to the solar year is to insert a thirteenth month every four years or so, as the Jewish and Chinese calendars do. Another, probably more common way is to begin counting moons with some event that is determined by the solar year, count up to, say, ten moons, stop counting, and then begin again when the event that you started with comes again. In this way a "Sandhill Crane Moon" will correspond to the coming of the sandhill cranes.

Counting moons, leaving a gap, and beginning the count again with the observation of something determined by the solar year is just what was done by people upriver from the Musqueam. According to Diamond Jenness (1955, 7-9), the Katzie, who spoke a Downriver Halkomelem dialect very close to that of the Musqueam, counted ten months beginning with the arrival of the sockeve in August of the Gregorian calendar, leaving a period covering June and July with two names but not regarded as part of the count. The Chehalis, an Upriver Halkomelem-speaking people, as reported by Charles Hill-Tout (1904, 334-35), began a count of moons with the chinook salmon spawning in October, counted ten moons, and stopped counting in July, leaving a period called by a term said to refer to the coming together of the ends of the year. Neither Jenness nor Hill-Tout gave any reason for this uncounted period. But James Teit (1900, 339), describing the practice of the Ntlakapmux (Thompson), is very clear about it. The Ntlakapmux began a moon count with the rutting of the deer or some other animal in the fall, counted eleven months, and left a period as "the rest of the year." "This indefinite period of unnamed months," Teit explains, "enabled the Indians to bring the lunar and solar years into harmony."

On Vancouver Island, lists of twelve and thirteen names of lunar months have been recorded from Cowichan speakers among others (Donckele 1882; Jenness 1934; 1955, 87; Suttles Cowichan notes). These lists may simply have been the result of attempts to make the old moon names fit the Gregorian calendar, or they may reflect a lunar calendar that was adjusted twice during the solar year, perhaps at the summer and winter solstices. Jenness (1934) reports that the Cowichan observed the winter solstice.

The calendar used by the Musqueam in earlier times must have been of this sort, although it is not clear what observation or observations may have been made. Timekeepers could have observed some biological event or some astronomical event. According to JP, there was a "weatherman" who watched progression of the sun along the horizon at sunrise and observed the solstices when the sun appeared to stand still or waver in the point at which it rises. The name given Mount Baker,  $xet^{\theta}enax^{w}$  'measure fish' or 'measure season,' seems to imply its use as a marker of the seasons. People also observed changes in the stars and noted in particular the rising of the constellation called  $st^{\theta}ana$  'bullhead,' probably the Pleiades, just before dawn.<sup>6</sup>

JP made two statements about the beginning of the year. On one occasion, he said that the year begins with  $\dot{q}^w t \dot{l} \sigma s$  ('spring'), right before  $w \delta t \dot{x} \sigma s$ . On another occasion, shortly after the first, he said that  $x^w \dot{c} \delta l \sigma \sigma n$  'the turn of the year,' that is, the beginning of the new year, was in the fall, explaining that "the word really means the turn from old to new" and adding that "the next is  $\dot{q}^w t \dot{l} \sigma s$ ." His translation of the first term implies that the root is  $\dot{c} \delta l$ - 'turn round,' but it seems possible that it could be  $x^w \sigma \dot{c}$ - 'insert,' which could imply a gap in the moon count. His two statements suggest that there may have been two times during the year when a moon count was started, but these do not correspond to the solstices. We can only conclude that at this time we really do not know how the old calendar worked.

<sup>6</sup> The fish is probably the cabezon, a large species of sculpin. The constellation is often called "Little Dipper" in English.

# **23** Sample Texts

# 23.1. GETTING WINTER SUPPLIES (CC 1)

| k <sup>w</sup> θə šx <sup>w</sup> te <sup>9</sup> és k <sup>w</sup> θə syəẁén'əłct k <sup>w</sup> səw<br>k <sup>w</sup> θə šx <sup>w</sup> -tə <sup>9</sup> é-s k <sup>w</sup> θə s-yəẁén'-əł-ct k<br>ART OBLNOM-be.like-3POS ART NOM-go.bef<br>3POS<br>It is the way of our ancestors when they v | ł néṁs<br><sup>w</sup> s-wəł-néṁ-s<br>ore-past-our ART NOM-already-go<br>vent                                                                             | - |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|---|
| ?áləžətəs kwdə sqíləs ?i kwdə cáyxw s?á<br>?áləž-ət-əs kwdə s-qílə-s ?əy kwdə cáyx<br>provide-tr-3tr ART NOM-supply-3POS and AR<br>to get their winter supplies of dried salmon                                                                                                                    | x̃*a <sup>9</sup> s.<br>* s <sup>9</sup> áx̃*a <sup>9</sup> -s<br>ατ dry butter.clam-3POS<br>and their dried butter clams.                                | 2 |
| k <sup>w</sup> áyžθət k <sup>w</sup> θə syəẁén'əlct.<br>k <sup>w</sup> áyž-θət k <sup>w</sup> θə s-yəẁén'-əl-ct<br>move-self ART NOM-go.before-past-our<br>Our ancestors acted.                                                                                                                    | səw q <sup>w</sup> álq <sup>w</sup> əltəls,<br>s-wə- q <sup>w</sup> álq <sup>w</sup> əl-təl-s<br>NOM-EST-tell-RECIP-3POS<br>They told each other,         | 3 |
| "'?óỷ k <sup>w</sup> səwł némct x <sup>w</sup> íwəl wəwéyələs<br>?óỷ k <sup>w</sup> s-wəł-ném-ct x <sup>w</sup> íwəl wə-wéyəl-<br>good ART NOM-already-go-our move.upstrea<br>"We'd better start going upriver tomorrow                                                                            | əs<br>m when-become.day-3SUB                                                                                                                              | 4 |
| k <sup>w</sup> səwł némct cqílə."<br>k <sup>w</sup> s-wəł-ném-ct c-qílə<br>ART NOM-already-go-our get-supply<br>to get started drying salmon."                                                                                                                                                     |                                                                                                                                                           | 5 |
| səw q <sup>w</sup> éls tə łθáləq,<br>s-wə-q <sup>w</sup> él-s tə ł-θáləq<br>NOM-EST-speak-3POS ART part-divide<br>Then some said,                                                                                                                                                                  | " <sup>?</sup> óỷ. hôye <sup>?</sup> ct wowéyolos.<br>'ôỷ hôye <sup>?</sup> ct wo-wéyolos.<br>good leave we when-day-3sub<br>"Good. We'll leave tomorrow. | 6 |

| Xa ce? k*ə səwł Xd*ótct ?al k*0ə hák*əXa ce? k*ə s-wəł-Xd*-ót-ct ?al k*0ə hálBE3P FUT then NOM-already-prepare-TR-our jiNow then we'll just start preparing what we                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | xət."<br>k <sup>w</sup> -əx-ət<br>ust ART use-TR-we<br>will use."                                                                                        | 7  |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|----|
| wéyəl, səwł ?á·łs ?al ?é·łtən.<br>wéyəl s-wəł-?á·ł-s ?al ?é·łtən<br>day NOM-already-embark-3POS just<br>3PL<br>In the morning they just started to<br>get into their canoes.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | səw q <sup>w</sup> éls tə łθáləq,<br>s-wə-q <sup>w</sup> él-s tə ł-θáləq<br>NOM-EST-speak-3POS ART<br>PART-divide<br>Then some said,                     | 8  |
| "tás ce? ? a x dwá n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n x a n | háy ce <sup>9</sup> łníməł<br>háy ce <sup>9</sup> łníməł<br>specify FUT we                                                                               | 9  |
| <ul> <li>At Kwannen we win separate.</li> <li>x*té? e k*θe šx*śk*əwəł ném ct<br/>ce? əẩ smád*əc.</li> <li>x*té? ?ə k*θə šx*-x*śk*-əwəł ném<br/>ct ce? ?ə ẩ smád*əc.</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | háy ce?<br>háy ce?                                                                                                                                       | 10 |
| dragging-canoe go we FUT OBL ART<br>Point.Roberts<br>who will head for the portage and go<br>to Point Roberts.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | It will be                                                                                                                                               |    |
| s?áx̃ <sup>w</sup> a? k̃ <sup>w</sup> ə némət cʻəýx <sup>w</sup> t.<br>s?áx̃ <sup>w</sup> a? k̃ <sup>w</sup> ə ném-ət cʻəýx <sup>w</sup> -t.<br>butter.clam ART go-we dry-TR<br>butter clams that we will go and dry.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | háy ce <sup>9</sup> k <sup>w</sup> ə tə łwóləp<br>háy ce <sup>9</sup> k <sup>w</sup> ə tə łwóləp<br>specify FUT then ART you(PL)<br>It will be you folks | 11 |
| wəyə <sup>9</sup> é·ỷ <sup>9</sup> al ném c'əỷx <sup>w</sup> éls k <sup>w</sup> šx <sup>w</sup> łíc'əs.'<br>wə-yə- <sup>9</sup> é·ỷ <sup>9</sup> al ném c'əỷx <sup>w</sup> -éls k <sup>w</sup> s-x <sup>w</sup> -łí<br>EST-along-continue just go dry-ACT ART NOM<br>who will continue on to go and dry scored f                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | ,<br>'c-əs<br>1-within-cut-face<br>'ish.''                                                                                                               | 12 |
| səw hóye?s təwẳáləm x™íwəl.<br>s-wə-hóye?-s təwẳáləm x™íwəl<br>NOM-EST-leave-3POS they move.upstream<br>Then away they went upstream.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | ni tás k <sup>w</sup> θə<br>ni tás [?ə] k <sup>w</sup> θə<br>AUX arrive.there [?ə] ART<br>They got to a certain place (as                                | 13 |

| <ul> <li>šx "tás i nə x "né?ent.</li> <li>šx "-tás ?əỷ ni? x "ə-né?ənt</li> <li>OBLNOM-arrive.there and AUX becor being.night</li> </ul>                                                                                        | ne-                                                                  | səw q <sup>w</sup> éls <sup>9</sup> é·łtən,<br>s-wə-q <sup>w</sup> él-s <sup>9</sup> éłtən,<br>NOM-EST-speak-3POS 3PL                                                                                   | 14       |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------|
| far as they could) by the time it bec<br>night.                                                                                                                                                                                 | came                                                                 | Then they said,                                                                                                                                                                                         |          |
| " <sup>9</sup> ə́y k <sup>w</sup> s qʻələm̀ct."<br><sup>9</sup> ə́y k <sup>w</sup> s-qʻələm̀-ct<br>good ART NOM- camp-our<br>"We had better camp."                                                                              |                                                                      | səw łé·ls <sup>9</sup> é·łtən.<br>s-wə-łé·l-s <sup>9</sup> é·łtən<br>NOM-EST-move.ashore-3POS 3PI<br>Then they landed.                                                                                  | 15       |
| səw θόyti t tə sála <sup>9</sup> ac.<br>s-wə-θόy-t-əy-ət tə sála <sup>9</sup> ac<br>NOM-EST-be.created-TR-3PAS-SUBPAS<br>ART house.mat                                                                                          |                                                                      | Xa sílewtx <sup>w</sup> s<br>Xa síl-ewtx <sup>w</sup> -s<br>BE3P canvas-house-3POS                                                                                                                      | 16       |
| Then they put up the house-mats.                                                                                                                                                                                                |                                                                      | The house-mats were                                                                                                                                                                                     |          |
| <sup>9</sup> é·łtən tə sála <sup>9</sup> ac.<br><sup>9</sup> é·łtən tə sála <sup>9</sup> ac<br>3PL ART house.mat<br>their tents.                                                                                                | səw há·<br>s-wə-há<br>NOM-EST<br>becor<br>Then the                   | yðəns tə x <sup>w</sup> əné <sup>9</sup> entqən.<br>íy-əðən-s tə x <sup>w</sup> ə-né <sup>9</sup> ent-qər<br><sup>2</sup> -finish-mouth-3POS ART<br>ne-being.night-throat<br>ey finished eating supper. | 17<br>1  |
| səw <sup>9</sup> a <sup>9</sup> əltət.<br>s-[ni <sup>9</sup> -s] wə- <sup>9</sup> a <sup>9</sup> əltət.<br>NOM-AUX-3POS-EST-sleep(PL)<br>Then they slept.                                                                       | x <sup>w</sup> ənné<br>x <sup>w</sup> ən-ne<br>still-nigh<br>When it | təł ?i wəł?i?ímət.<br>ét-əł ?i wəł-?i?ímət<br>nt-past and already-get.up(PL)<br>was still dark they got up.                                                                                             | 18       |
| səw yáq <sup>w</sup> əlcəps tə səwáyye?.<br>s-wə-yáq <sup>w</sup> -əlcəp-s tə səwáyye?<br>NOM-EST burn-firewood-3POS ART<br>men<br>Then the men lit fires.                                                                      | səw :<br>s-wə<br>NOM-<br>WO<br>Then                                  | xté <sup>9</sup> ems tə słənłénəý tə<br>-xté <sup>9</sup> -əm-s tə słənłénəý tə<br>EST-make-INTR-3POS ART<br>omen ART<br>the women made                                                                 | 19       |
| <ul> <li>šx "nétəłqən <sup>9</sup>i yə<sup>9</sup>é·ý tə səwáýc</li> <li>šx "-nét-əł-qən <sup>9</sup>əy yə-<sup>9</sup>é·ý tə sə</li> <li>OBLNOM-night-past-throat and along</li> <li>be.removing-TR-INTR ART house.</li> </ul> | qe <sup>?</sup> hóm<br>owóýqe?<br>g-be.cont<br>mat                   | əxəm tə sála?ac.<br>həm-əx-əm tə sála?ac<br>inuing ART men                                                                                                                                              | 20       |
|                                                                                                                                                                                                                                 | oving the                                                            |                                                                                                                                                                                                         | <b>.</b> |
| səw ste'le tθê' 'lətən 'leitən<br>s-[ni?-s] wə-stə'le 'lə tθê' 'lətər<br>'leitən                                                                                                                                                | 1                                                                    | səw há·yθən.<br>s-wə-háy-əθən[-s]                                                                                                                                                                       | 21       |
| NOM-AUX-3POS EST-like OBL that eat                                                                                                                                                                                              | 3PL                                                                  | NOM-EST-finish-mouth-3POS                                                                                                                                                                               |          |

And thus they ate.

NOM-EST-finish-mouth-3POS And they finished eating.

| səw <sup>9</sup> əlłəle <sup>9</sup> cəm.<br>s-[ni <sup>9</sup> -s] wə- <sup>9</sup> á·ł-əle <sup>9</sup> c-əm<br>NOM-AUX-3POS EST-aboard-<br>container-INTR<br>And they loaded up.                                                                                             | <ul> <li>Åa səsÅəwł híqəθət</li> <li>Åa s-[ni?]-s Åe wəł-híq-əθət</li> <li>BE3P NOM-AUX-3POS<br/>again already-go.under-self</li> <li>Then they shoved off again</li> </ul>                                                                                                    | 22       |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------|
| wəyəxăi ?al tθé?<br>wə-yə-xăi ?a.l tθé?<br>EST-along-be.doing just<br>that<br>and just doing that                                                                                                                                                                               | <sup>9</sup> i nəw tös <sup>9</sup> al q <sup>w</sup> á·n <sup>*</sup> Xöl. <sup>9</sup> əy ni <sup>9</sup> wə-tös <sup>9</sup> al [ <sup>9</sup> ö X <sup>*</sup> ] q <sup>w</sup> á·n <sup>*</sup> Xöl and AUX EST-arrive.there just OBL ART Kwantlen they reached Kwantlen. | 23       |
| səsəw θələqtəl yəθe? nem tx «á<br>s-[ni?]-s wə-θələq-təl yəθe? ne<br>NOM-AUX-3POS EST-separate-RECIP<br>Then the ones who were going to                                                                                                                                         | ·ł.<br>em łx <sup>w</sup> á·ł<br>those go dig.clams<br>dig clams separated                                                                                                                                                                                                     | 24       |
| x "ənsté?e ?i wəłtə́s ?əʎ Hope<br>x "ən-sté?e ?əy wəł-tə́s ?ə ʎ Ho<br>still-stay and already-reach OBL AR<br>The ones who were going to make                                                                                                                                    | yəθé <sup>9</sup> ni neṁ čx <sup>w</sup> łíċəs<br>ppe yəθé <sup>9</sup> ni neṁ c-x <sup>w</sup> -łíċ-əs<br>T <i>Hope</i> they AUX go do-within-cut-face<br>scored fish continued on to Hope.                                                                                   | 25       |
| Xa səsəwł cqílə ?al k™snis ce<br>Xa s-[ni?]-s wəł-c-qílə ?al k<br>BE3P NOM-AUX-3POS already-do-s<br>Right away they started to get th                                                                                                                                           | ? čáýx <sup>w</sup><br>v s-ni?-s ce? čáýx <sup>w</sup><br>supply just ART NOM-AUX-3POS FUT dry<br>neir winter supply so it would become dry                                                                                                                                    | 26<br>y. |
| i yeł s <sup>9</sup> əmis wəd <sup>w</sup> íləm yəlá <sup>9</sup> ał<br><sup>9</sup> əy yeł s- <sup>9</sup> əmi-s wəd <sup>w</sup> íləm yəl<br>and after.which NOM-come-3POS m<br>And then they came downstream l                                                               | əle <sup>9</sup> c tə sqíləs.<br>á <sup>9</sup> əlále <sup>9</sup> c tə s-qíləs<br>ove.downstream loaded ART NOM-supply<br>oaded with their dried fish.                                                                                                                        | 27       |
| háy nə xॅ <sup>w</sup> ə́m k <sup>w</sup> səsmi wəd <sup>w</sup> il<br>háy ni? xॅ <sup>w</sup> ə́m k <sup>w</sup> -s-[?i]-s ?ən<br>specify AUX fast ART NOM-AUX-3<br>Coming downstream was faster.                                                                              | əm.<br>ii wəqʾʷíləm<br>POS come downstream.                                                                                                                                                                                                                                    | 28       |
| <sup>9</sup> isélə <sup>9</sup> al swéyəl <sup>9</sup> i mi x <sup>w</sup> ə <sup>9</sup> ám<br><sup>9</sup> isélə <sup>9</sup> al swéyəl <sup>9</sup> i mi x <sup>w</sup> ə- <sup>9</sup> ár<br>two just day and come become-be<br>In just two days they were home.            | ət.<br>nət<br>.sitting                                                                                                                                                                                                                                                         | 29       |
| <sup>?</sup> i háy k <sup>w</sup> θə ni <sup>?</sup> tətx <sup>w</sup> á <sup>?</sup> ał nəw<br><sup>?</sup> i háy k <sup>w</sup> θə ni <sup>?</sup> tətx <sup>w</sup> á <sup>?</sup> ał ni <sup>?</sup> v<br>and specify ART AUX be.clamming<br>But those who were clamming we | x™ənhé∙wə.<br>və-x™ənhé∙wə<br>AUX EST-still-be.absent<br>re still away.                                                                                                                                                                                                        | 30       |

| háy <sup>9</sup> áyəm tə s <sup>9</sup> áxॅ <sup>w</sup> a <sup>9</sup> k <sup>w</sup> s c <sup>3</sup><br>háy <sup>9</sup> áyəm tə s <sup>9</sup> áxॅ <sup>w</sup> a                                                       | óỷ x™s.                                                                                                                    | qôž syá·ys.<br>qôž s-yá·ys                                                                         | 31 |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|----|
| s-čýx <sup>w</sup> -s<br>specify slow ART butter.clam A<br>drv-3POS                                                                                                                                                         | RT NOM-                                                                                                                    | much NOM-work                                                                                      |    |
| Butter clams are slower when                                                                                                                                                                                                | they dry.                                                                                                                  | It's a lot of work.                                                                                |    |
| səsəw x <sup>w</sup> əl?áṁət yəθé? ni<br>s-[ni?]-s wə-x <sup>w</sup> ə-?áṁət ya<br>NOM-AUX-3POS EST-become-h<br>When those who had been g                                                                                   | i·ł cἀíἀələ,<br>əθé? ni?-əł c<br>nome they AU2<br>etting winter                                                            | z-qíqələ<br>x-past get-supply<br>supplies got home,                                                | 32 |
| səsəw ?əỷáqtəl tə s?əttəns ?<br>s-[ni?]-s wə-?eỷeq-təl tə s-<br>NOM-AUX-3POS EST-exchange-e:<br>supply<br>they exchanged their foods, the                                                                                   | Pé·łtən, tə sq<br>Páłtən-s Pé·ł<br>ach.other ART<br>e dried salmoi                                                         | ílə<br>tən tə s-qilə<br>NOM-eat-3POS 3PL ART NOM-<br>1                                             | 33 |
| i tə s?áẍwa?.<br>?əỷ tə s?áẍwa?<br>and ART butter.clam<br>and the butter-clams.                                                                                                                                             | Xa šx™te ésa<br>Xa šx™-te?é<br>BE3P OBLNOM<br>That is the w                                                                | əł k™θə ṁi∙t<br>-s-əł k™θə ṁi-ə∙t<br>-be.like-3POS-past ART AUX-we<br>ay it was with those whom we | 34 |
| yə <sup>9</sup> é·ýeqt k <sup>w</sup> shəýdəstalx <sup>w</sup> s k<br>yə- <sup>9</sup> é·ýeq-t k <sup>w</sup> s-həýdəs-t-a<br>along-exchanging-TR ART NOM-<br>OBLNOM-belong(PL)-past-our<br>are replacing as our parents we | x <sup>w</sup> θə šx <sup>w</sup> əwé<br>İx <sup>w</sup> -s k <sup>w</sup> θə š<br>-be.telling-τR-<br>r<br>ere telling us. | ləyəłct<br>x <sup>w</sup> -wéləy-əł-ct<br>·us-3POS ART                                             | 35 |

This text was dictated by Christine Charles on 23 November 1960 and reviewed with Della Kew in November 1972.

### Line

1-2. This is a statement of what is going to follow, perhaps a kind of title.

1.  $\sqrt{t\partial^2 \acute{e}}$  'be like' appears in  $st\partial^2 \acute{e}$  (usually pronounced  $ste^2 \acute{e}$ ), meaning 'be like, resemble, be the same as,' and in  $\check{s}x^w t\partial^2 \acute{e}$  'way, manner,' lit. 'that which is like.' Compare  $st\acute{e}^2 e$  'remain, stay put, continue as before,' as in line 25.

2.  ${}^{2}al \partial \check{x} \partial t$  'get it, prepare it, provide it' usually refers to food.  $\sqrt{d}t \hat{l} \partial$  has not been recorded without an affix. The noun  $s\dot{d}t\hat{l}\partial$ , often glossed 'winter supplies,' refers to any kind of preserved food but especially dried or smoked salmon. The verb  $c\dot{d}t\hat{l}\partial$  means 'preserve food,' especially 'dry salmon.' The  $s^{2}\dot{a}\check{x}^{*}a^{2}$  is the butter clam (*Saxidomus giganteus*).

3.  $k^{w} \delta y \dot{x} \theta \partial t$ , lit. 'move oneself,' is often used to begin a narrative or a new segment of one.  $sy \partial w \dot{e} \dot{n}$ , lit. 'what has gone before,' can mean 'ancestors' or simply 'heritage.'

9.  $\vec{q}^{w} \acute{a} \cdot \vec{n} \vec{\lambda} \cdot \vec{a}$  'Kwantlen' is the name of the people at or near the mouth of the Salmon River, near the second site of Fort Langley. They have often been called "the Langley people."

10. The Musqueam who were going to dig clams went up the Fraser as far as Kwantlen, went up the Salmon River, portaged across the prairie to the Nicomekl, and down that stream to Boundary Bay. Although  $smá\dot{q}^w \partial c$  is the name for Point Roberts, CC initially translated it here as "Boundary Bay." This must have been a well-known route. In 1824, the first Hudson's Bay Company expedition to the Fraser was guided through it by Snohomish and Skagit people from Puget Sound. The term for 'portage' is probably  $\check{s}x^w$ - $x^w \partial \check{k}^w \partial w \partial t$ , but it is given in the first line as recorded.

12.  $\delta x^w lt cost was made by removing the backbone and scoring the flesh so that it could open up and dry more readily. Mrs. Charles described the catching and processing of salmon after dictating this text. For a description and photographs, see Wilson Duff's$ *The Upper Stalo Indians*, Anthropology in British Columbia, Memoirs 1, BC Provincial Museum, Victoria, 1952, 63-66.

16.  $sála^2ac$  'house-mat' refers to the larger mat used for lining the walls of the plank house or for making temporary shelters at camping sites, as here. These were made of tules  $(wi \cdot l)$  or of cattails  $(st^{\theta} e^2 q an)$ . The st e wan 'bed-mat' was a smaller cattail mat used for sleeping or simply sitting on. Both were made by sewing with a long matting needle and a mat creaser. sil 'canvas' is from the English *sail*.

22.  ${}^{2}\partial lt \delta le^{2}c \partial m$  'load up' probably consists of  ${}^{2}\dot{a}t$  'board (a vessel),' which probably has the underlying form //  ${}^{2}al \partial t$  //, the suffix  $-\partial le^{2}c$  'container,' and  $-\partial m$  'intransitive.' The resultative of  ${}^{2}\dot{a}t$  is  ${}^{2}\partial l\dot{a}^{2}\partial t$  'aboard,' for which CC gave an unusual plural,  $y\partial l\dot{a}^{2}\partial t$ . The form in line 27,  $y\partial l\dot{a}^{2}\partial t\delta le^{2}c$  'loaded,' appears to be based on this form.

32. Recorded  $x^{w} \partial l^2 a m \partial t$ , but more likely  $x^{w} \partial^2 a \dot{m} \partial t$  with  $x^{w} \partial -$  'become' or  $x^{w} \partial n^2 a \dot{m} \partial t$  with  $x^{w} \partial n$ - 'already.'

34-35. 'Those whom we are replacing' is another respectful term for the ancestors.

### 23.2. THE DEER DRIVE (CC 2)

| ném ct x <sup>w</sup> á <sup>2</sup> aθ ném <sup>2</sup> v <i>Å Halfway.</i><br>ném ct x <sup>w</sup> á <sup>2</sup> aθ ném <sup>2</sup> v <i>Å Halfway</i><br>go we deer.drive go OBL ART Halfway<br>We are going on a deer drive at Halfway. |                   | <sup>9</sup> əpé·lə ct ce <sup>9</sup><br><sup>9</sup> əpən-elə ct ce <sup>9</sup><br>ten-persons we FUT<br>There will be ten of us | 1 |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------|-------------------------------------------------------------------------------------------------------------------------------------|---|
| wəní·t háye?.                                                                                                                                                                                                                                  | ném ct wékənà·ł.  | sdəlím ct ce?.                                                                                                                      | 2 |
| wə-ní?-ət háye?                                                                                                                                                                                                                                | ném ct wékən-à·ł  | sdəlím ct ce?                                                                                                                       |   |
| EST-AUX-we leave                                                                                                                                                                                                                               | go we wagon-by    | camped.out we FUT                                                                                                                   |   |
| when we leave                                                                                                                                                                                                                                  | We'll go by wagon | We'll be camped out                                                                                                                 |   |

| x̃əθí·lə ce? kʷθə sqႆʷáqႆʷəł.<br>x̃ə?áθən-elə ce? kʷθə sqႆʷáqႆʷəł.<br>four-persons FUT ART lying.in.wait<br>Four will be lying in wait.                                                                                                | えá ce?<br>えá ce?<br>BE3P FUT<br>They will be the ones who will                                                                         | 3  |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|----|
| k*áləxt k*0ə smáyə0 wəmírs ce? v<br>k*áləx-t k*0ə smáyə0 wəmírəs c<br>shoot-TR ART deer when-come-3SUB FU<br>shoot the deer when they start coming                                                                                     | wəłcłáq <sup>w</sup> θət.<br>e <sup>9</sup> wəł-cłáq <sup>w</sup> -θət<br>л already-pass.through-self<br>; through.                    | 4  |
| <ul> <li>i Xá ce? k<sup>w</sup> to sowá·wlosà·lł ném</li> <li>?oý Xá ce? k<sup>w</sup> to sowá·wlos-à·lł n</li> <li>and BE3P FUT then ART youths-young g</li> <li>deer.drive</li> <li>And then it will be the boys who go v</li> </ul> | nəsqʻá? kwəns ném xwá?að<br>ém nə-sqʻá? kwə nə-s-ném xwá?að<br>go my-company ART my-NOM-go<br>vith me when I go to drive the deer,     | 5  |
| yəqéwəm ce? ?éłtən ste?é ?ə k <sup>w</sup> st<br>yə-qéwəm ce? ?éłtən stə?é ?ə k <sup>w</sup> st<br>along-be.howling FUT 3PL like OBL ART<br>howling like wolves as they go along.                                                      | rəqéye<br>təqéye<br>r wolf                                                                                                             | 6  |
| Xá ce? səw yətátəsəts k <sup>w</sup> ə smént<br>Xá ce? s-wə-yə-tátəs-ət-s k <sup>w</sup> ə smé<br>BE3P FUT NOM-EST-along-be.hitting.tog<br>They'll be the ones striking stones tog                                                     | ?é·łtən<br>§·nt ?é·łtən<br>gether-TR-3POS ART stone 3PL<br>gether                                                                      | 7  |
| wəmí s wəłwéłətəs k və smáyəð.<br>wəmí s wəłwéł-ət-əs k və smáy<br>when-come-3SUB be.chasing-TR-3SUB<br>when they come chasing the deer.                                                                                               | αθ<br>ART deer                                                                                                                         | 8  |
| stḋ <sup>w</sup> íqən tə smé∙nt i Åəw stətíd <sup>w</sup><br>stḋ <sup>w</sup> íqən tə smé∙nt <sup>9</sup> əỷ Åe wə-st<br>steep ART mountain and also EST-sh<br>The mountainside is steep and too t                                     | ′ tə stáləw.<br>ətíqೆ™ tə stáləw<br>eer ART river<br>he river bank is sheer.                                                           | 9  |
| sk <sup>w</sup> éy k <sup>w</sup> ə k <sup>w</sup> snéms hí·lθət.<br>sk <sup>w</sup> éy k <sup>w</sup> ə s-ném-s hí·l-θət<br>impossible then ART NOM-go-3POS<br>roll-self<br>They can't let themselves go.                             | wəỉán wəẳáp wəỉán wəcícəł.<br>wə-ỉán wə-ẳáp wə-ỉán wə-cícəł<br>EST-too EST-deep EST-too EST-high<br>It is too far down and too far up. | 10 |
| ?ówəte? kwə šxwnéms təwka smáy<br>?ówəte? kwə šxw-ném-s təwka smá<br>none ART OBLNOM-go-3POS that deer<br>The deer have no place to go.                                                                                                | ຈ <del>0</del> .<br>ອັງຈ <del>0</del>                                                                                                  | 11 |

| nəw ném ?al xॅwən<br>ni? wə-ném ?al xॅv<br>AUX EST-go only run<br>The deer can only ru                                                                          | xénəm təwẩa smáyəθ<br>*ənxénəm təwẩa smáyəθ<br>n that deer<br>un                                                                                                                                                                      | 12 |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----|
| ném ?ə k <sup>w</sup> θéləy n<br>ném ?ə k <sup>w</sup> θéləy n<br>go OBL those AUX al<br>to those who are alro                                                  | əwł sdʷádʷəł.<br>i? wəł-sdʷádʷəł<br>ready-lying.in.wait<br>eady lying in wait.                                                                                                                                                        | 13 |
| səsəw həlk <sup>w</sup> ə́ləxtə<br>s-[ni <sup>?</sup> ] -s est-həlk<br>NOM-AUX-3POS EST-s<br>And so they shoot th                                               | os ?é·łtən.<br>∵*áləx-t-əs ?é·łtən<br>hoot(PL)-TR-3TR 3PL<br>hem.                                                                                                                                                                     | 14 |
| k <sup>w</sup> səsəwł há·y k<br>k <sup>w</sup> s-[ni <sup>?</sup> ]-s w<br>wə-k <sup>w</sup> íc̀-ət-əs<br>ART NOM-AUX-3P<br>EST-cut.open-TH<br>When they have f | "ə se <sup>9</sup> k "áləx səsəw k <sup>*</sup> "ícətəs<br>əł-há·y k <sup>*</sup> ə s-he <sup>9</sup> k "áləx[-s] s-[ni <sup>9</sup> ]-s<br>os already-finish ART NOM-shoot-3POS NOM-AUX-3POS<br>R-3TR<br>inished shooting, they draw | 15 |
| tə səmíyəθ.<br>tə səmíyəθ.<br>ART deer(PL)<br>the deer.                                                                                                         | səsəwł łípətəm. səsəw ṫáxətəl<br>s-[ni?]-s wəł-łíp-ət-əm s-[ni?]-s wə-ṫáx-ətəl<br>NOM-AUX-3POS already-quarter-TR-INTR NOM-AUX-3POS<br>EST-divide-RECIP<br>Then when they are quartered, they divide them up.                         | 16 |
| ?é·łtən.<br>?é·łtən<br>3PL                                                                                                                                      | i Žá səsəwł hílək <sup>w</sup> ?al ?é·łtən.<br>?əỷ Žá s-[ni?]-s wəł-hílək <sup>w</sup> ?al ?é·łtən.<br>and BE3P NOM-AUX-3POS already-get.ready just 3PL<br>And that's when they just get ready.                                       | 17 |
| k <sup>w</sup> səwł há∙y k <sup>w</sup> słáł<br>k <sup>w</sup> s-wəł-há∙y[-s] l<br>ART NOM-already-fin<br>When they have fini                                   | əxətəl tə sxəncəs,<br>k™ s-tátəx-ətəl tə sxəncə-s<br>nish-[3POS] ART NOM-divide-RECIP ART game-3POS<br>ished their division of the game,                                                                                              | 18 |
| k*səwł snéms ťák<br>k* s-wəł-[0]-ném-<br>ART NOM-already-NO<br>then they go home.                                                                               | <sup>»</sup> .<br>∙s ťák <sup>™</sup><br>pM-go-3POS return.home                                                                                                                                                                       | 19 |

This text was narrated by Andrew Charles Sr. on the Musqueam Reserve on 26 October 1960, with Mrs. Charles repeating more slowly to allow me to write it down. The translation is based on work with CC. Although we had used a

tape recorder on two or three previous occasions, this was our first attempt at writing down a dictated narrative. I had asked AC if he would give me an account of some activity he participated in when he was young. He said he would think about it, and the next day he gave the first thirteen lines. I believe this is a thought-out composition. He added lines 14 to 19 after I requested information on the distribution of the game.

Although AC was Musqueam, he had lived as a boy on one of the Cowichan reserves near Duncan. He was fourteen when he went with "the bunch" on a deer drive. They had rifles then, but at an earlier time they would have used bows and arrows. "Halfway" is on the Cowichan River about halfway between "Duncan's" and Cowichan Lake. That was the only place where they got deer by driving them. They had tried other places but these were too open and so they couldn't get any deer. Above the pass at Halfway was a place that was all alders and grass, where the deer fed. The boys chased the deer from there into the pass.

AC once watched the men who waited space themselves out. They took places on both sides of the trail, spaced out alternately so that they were not shooting at each other. The first man usually hit the most; the last man might not have much chance. They might get five or six at one time. If only a few persons were driving the deer, they could not get many, because the deer would hide. Deer are smart. They prefer the open to shelter under the trees – even in the rain. It is said that they don't like the heavy drops from the trees because they soak through.

The division of the deer was even, but if, say, five persons came from the same house, they might not quarter their carcass but take it home whole.

### Line

2.  $s\dot{q}alim$  'encamped, camped out,' resultative of  $\dot{q}alam$  'camp.' The root appears in  $\dot{q}a\dot{l}man$  'campsite,'  $\dot{q}alac\theta at$  'go under shelter,'  $x^w\dot{q}alec$  'shelter (as for a fire to protect it from the wind),' and probably  $\dot{q}alaxan$  'stockade, fence' and  $s\dot{q}alexan$  'stockaded, fenced in.'

3.  $s\dot{q}^{w}\dot{a}\dot{q}^{w}\partial t$  'lying in wait, be ready,' resultative of  $\dot{q}\dot{a}t\partial t$  'lie in wait for him/them, get ready for it (as an attack).'

4. The word for 'shoot' or 'spurt out' was recorded as  $k^w \delta l \partial x$ ,  $k^w \delta l l \delta x$ ,  $k^w \delta l l \delta x$ ,  $k^w \delta l l \delta x$ ,  $k^w \delta l l \delta x$ . It has an irregular progressive, recorded  $2 \delta^2 k^w \delta l \delta x$ ,  $2 \epsilon^2 k^w \delta l \delta x$ , and possibly  $h \epsilon^2 k^w \delta l \delta x$  'be shooting, be spurting,' which appears in line 14. The transitives are  $k^w \delta l \delta x t \sim k^w \delta l \delta x t$  'shoot him (deliberately)' and  $k^w \delta l \delta x t$  or ' $2 \delta k^w \delta l \delta x t$ ' shoot several,' which appears in line 13. A noun derivative,  $s k^w \delta l \delta x t$  (plural  $s \delta l k^w \delta l \delta x$ ) means 'arrow' or 'gun.'

5. In writing 'and' as *i*, I am following DK (see §1, note 9).  $s \partial w \dot{a} \cdot w \partial a s$  'young men' is the plural, as given by CC, of  $sw(\dot{w}las$  'young man.' (JP gave the plural as  $sw\dot{a} \cdot \dot{w}las$ .) The suffix  $-a \cdot lt$  'young' added to the name for an animal produces

the term for the young of that species. Added to young man or  $dem \partial y$  'young woman,' it produces 'boy' or 'girl.' A 'boy' is between 10 and 20 years old. sda' is a nominalization of da' 'accompany.'

6.  $\dot{q}\acute{e}\vec{w}\partial m$  (also recorded  $\dot{q}\acute{e}\vec{w}\partial \vec{m}$ ) 'be howling,' progressive of  $\dot{q}\acute{o}w\partial m$ 'howl.'  $st\partial^2\acute{e}$  'like, the same as, accordingly,' is probably the resultative of a root  $\sqrt{te^2}$  'be alike(?), appear(?).' Cf.  $t\partial^2\acute{a}\cdot m\partial x$  'resemble,'  $st\partial^2\acute{m}$  'manner, way,'  $\check{s}x^wt\partial cis$  'way (of doing something),'  $\check{s}x^wte^2\acute{e}\cdot\vec{w}\partial n$  'think,'  $2\dot{\delta}w\partial te^2$ 'none.'

7.  $t\acute{e}tasat$  'be hitting together,' progressive of  $t\acute{a}sat$  'hit together,' <  $t\acute{a}s$  (also recorded [t<sup>h</sup>ás]) 'get hit, get bumped.'

8. *wəłwétət* 'be driving, chasing away,' progressive of *wétət* 'drive away,' root not recorded.

9.  $st\dot{q}^{w}iq\partial n$  'steep, sheer (in slope of land)' <  $t\partial \dot{q}^{w}$  'be cut off, be broken off,' -*iqon* 'front, slope.'  $st\partial t\dot{q}^{w}$  'cut up, cut off, cut straight, sheer,' resultative of  $t\partial \dot{q}^{w}$  'be cut off.'

14-15. See note for line 4.

16.  $s \partial m i y \partial \theta$  'deer,' plural of  $s m \delta y \partial \theta$  'deer.'  $t \delta x \partial t \partial t$  'divide, distribute (a catch or game),'  $\leq t \delta x$ ,  $-t \partial t$  'reciprocal'; cf.  $t \delta x \partial s t$  'share with him,'  $t \delta x \delta \partial t$  'branch out (as a trail or stream).'

18. *ťáťažatal* 'be dividing,' progressive of *ťážatal*. *sxáňca* 'game (of hunter), catch (of fisherman), prey (of predator)'; cf. *cxáňca* 'take as game, catch, prey.'

19. It seems to me that the second nominalizer should not be there.

### 23.3. NIGHT HUNTING IN THE GULF ISLANDS (CC 3)

| ni? neṁ hóyq <sup>w</sup> θəs k <sup>w</sup> neṁ hólx́əwe?.<br>ni? neṁ hóyq <sup>w</sup> -t-S-əs k <sup>w</sup> ə neṁ hólx́əwe? <sup>3</sup><br>AUX AUX(go) recruit-TR-me-3TR ART AUX(go) be.night.hunting<br>Someone came to get me to go night hunting with him. | 1 |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---|
| səct wəháye? ték wəl nem ?e ネ sqθáqa ł.<br>s-ni?-ct wə-háye? ték wəl nem ?ə 木 sqθáqa lł<br>NOM-AUX-our EST-leave cross go OBL ART Active.Pass<br>So we left and crossed over to "Plumper's Pass" [Active Pass].                                                    | 2 |
| tá··X <sup>w</sup> nəw nem nét i nə ct tás.<br>táX <sup>w</sup> ni? wə-nem nét ?əỷ ni? ct tás<br>exactly AUX EST-AUX(go) be.night and AUX we arrive.there<br>It had just become night when we got there.                                                           | 3 |
| səct wəkʷú·kʷ tə šxʷné?entqənct.<br>s-ni?-ct wə-kʷú·kʷ[-t] tə šxʷ-né?ənt-qən-ct<br>NOM-AUX-our EST-cook-TR ART OBLNOM-evening-meal-our<br>Then we cooked our supper.                                                                                               | 4 |

| səw q <sup>w</sup> él tə gíb, "ž <sup>w</sup> té? c<br>s-wə-q <sup>w</sup> él[-s] tə gíb ž <sup>w</sup> té<br>NOM-EST-speak-3pos ART G<br>FUT this<br>Then Gabe said, "We'll go | et ce? tə?í.<br>5? ct ce? ta<br>babe head.fo<br>this way [p                                 | o?í<br>or we<br>pointing south]                                                                   | Xໍa wəyáθəł<br>Xໍa wə-yáθ-əł<br>BE3P EST-always-past<br>. It's where I always     | 5        |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|----------|
| nəšx <sup>w</sup> ə́ǹəm̀."<br>nə-šx <sup>w</sup> -hə́ǹəm̀<br>my-OBLNOM-be.going<br>used to go."                                                                                 | səctəw k <sup>w</sup><br>s-[ni?]ct v<br>NOM-AUX-w<br>Then we sa                             | ácnəx <sup>w</sup> tə čič<br>və-k <sup>w</sup> ec-nəx <sup>w</sup><br>ve EST-look-TR<br>w a mink. | í <sup>9</sup> ἀən.<br>tə čičí <sup>9</sup> ἀən<br>ART mink                       | 6        |
| səsəw k*əlləx*təs.<br>s-[ni?]-s wə-k*əlləx*-t-<br>NOM-AUX-3POS EST-shoot-TI<br>Then he [Gabriel] shot it.                                                                       | əs<br>r-3tr                                                                                 | ?á. səct λəw<br>?a. s-[ni?]-c<br>well-NOM-AUX<br>already-lea<br>Well, then we                     | ł hóye?.<br>ct Że woł-hóye?<br>c-our again<br>ve<br>went off again.               | 7        |
| səsəw k <sup>w</sup> ə́cnəx <sup>w</sup> əs tə isé<br>s-[ni <sup>?</sup> ]-s wə-k <sup>w</sup> ec-nəx <sup>w</sup> -<br>NOM-[AUX]-3POS EST-look-1<br>Then he saw two racoons.   | lə məlləs.<br>əs tə yəse<br>IR-3TR ART t                                                    | lə məlləs<br>wo raccoon                                                                           | səsəw<br>s-[ni <sup>9</sup> ]-s wə-<br>NOM-AUX-3POS EST-<br>So he                 | 8        |
| k™álləxtəs Åe.<br>k™álləx-t-əs Åe<br>shoot-TR-3TR also<br>shot them too.                                                                                                        |                                                                                             | səctəw mók <sup>w</sup><br>s-[ni?]-ct wə<br>NOM-AUX-our<br>Then we all w                          | ′ ?əlá.ł<br>p-mák <sup>w</sup> ?əlá.ł<br>EST-all aboard<br>rere aboard [the canoe | 9<br>:]. |
| səctəw həye?, i ct təwx<br>s-ni?-ct wə-həye? ?i ct<br>NOM-AUX-our EST-leave AU<br>Then we left and we were                                                                      | ″ənyə <sup>9</sup> é∙ỷ<br>təw-x™ən-<br>x we somev<br>still leisurel                         | yə- <sup>9</sup> é·ỷ<br>what-still-along<br>ly                                                    | g-be.continuing                                                                   | 10       |
| yə?əx?í?xəl i ct wəłċłéń<br>yə-?əx?í?xəl ?i ct wəł-ċ<br>along-be.paddling(PL) AUX<br>paddling along when we ho                                                                  | nət k <sup>w</sup> ə <sup>9</sup> i<br>dém-ət k <sup>w</sup> a<br>we already<br>eard someth | síqəs<br>ə <sup>9</sup> i yə-síqəs<br>-hear-tr Art A<br>ning whistling.                           | UX along-whistle                                                                  | 11       |
| (k <sup>w</sup> əsəwł háy k <sup>w</sup> səs síqəs<br>k <sup>w</sup> ə s-[ni <sup>9</sup> ]-s wəł-háy k <sup>v</sup><br>ART NOM-AUX-3POS already<br>(When the whistling stoppo  | s<br>"ə s-[ni?]-<br>-stop ART N<br>ed                                                       | s síqəs<br>OM-AUX-3POS v                                                                          | vhistle                                                                           | 12       |
| k <sup>w</sup> səctəwł ċłéṁət k <sup>w</sup> səs<br>k <sup>w</sup> ə s-[ni?]-ct wəł-ċłéń<br>ART NOM-AUX-our already-l<br>we heard something chopp                               | k <sup>w</sup> q <sup>w</sup> éls<br>'n-ət k <sup>w</sup> ə s-<br>hear-tr Art<br>ing        | [ni?]-s kwaqw<br>NOM-AUX-3POS                                                                     | z-éls<br>5 get.struck-ACT                                                         | 13       |

| "px <sup>w</sup> ·· px <sup>w</sup> ·· X́····")                                                                                                                                                                           | səw q <sup>w</sup> éls gíb,<br>s-wə-q <sup>w</sup> él-s [tə] gíb<br>NOM-EST-speak-3POS ART Gabe                                                                | 14 |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------|----|
| "pow! pow! swooooosh.")                                                                                                                                                                                                   | Then Gabe said,                                                                                                                                                |    |
| "Xa ýeł k"àk"əq"nəcí·ls.<br>Xa ýeł k"ák"əq"-nəc-í·ls<br>BE3P EXP be.getting.struck-base-ACT<br>"It's those Little Choppers.                                                                                               | sk <sup>w</sup> éy k <sup>w</sup> s cxáncact<br>sk <sup>w</sup> éy k <sup>w</sup> a s-c-xánca-ct<br>impossible ART NOM-get-game-our<br>We can't get anything.  | 15 |
| ni ct wəłcłém at k <sup>w</sup> sis yéw d ta léy<br>ni? ct wał-cłém-at k <sup>w</sup> a s-ni?-s yé<br>AUX we already-hear-TR ART NOM-AUX<br>We just heard a big Douglas-fir fall.                                         | əłp.<br>d tə lé·ỷəłp<br>z-3POS fall ART mature.Douglas.fir                                                                                                     | 16 |
| °ởỷ k <sup>™</sup> s ἀá∙nθətct.<br>°ởỷ k <sup>™</sup> ə s-ἀén-əθət-ct<br>good ART NOM-return-self-our<br>We'd better go back.                                                                                             | ném ct ce? tə ?əx «ín ẳíẳcəs."<br>ném ct ce? [?ə] tə ?əx «ín ẳíẳcəs<br>go we FUT OBL ART little islet<br>We'll go to the little island."                       | 17 |
| səctəw tás.<br>s-[ni?]-ct wə-tás<br>NOM-AUX-our EST-arrive.there<br>Then we got there.                                                                                                                                    | səctəw dváləm.<br>s-[ni?]-ct wə-dváləm<br>NOM-AUX-our EST-disembark(PL)<br>Then we [Gabe and I] got off.                                                       | 18 |
| <ul> <li>tx way Henry Louie ?əlá?ał</li> <li>tə snóx wəł.</li> <li>tx wáy H.L. ?əlá?əł [?ə] tə snóx wəł</li> <li>only H.L. be.aboard OBL ART canoe</li> <li>Only Henry Louie stayed aboard</li> <li>the canoe.</li> </ul> | səw k <sup>w</sup> ə́yxॅθət tə Gabriel.<br>s-wə-k <sup>w</sup> ə́yx̆-θət[-s] tə G.<br>NOM-EST-move-self-3POS ART G<br>Then Gabriel started off.                | 19 |
| səw kwácnəxwəs tə smáyəθ.<br>s-[ni?-s] wə-kwec-nəxw-əs<br>tə smáyəθ<br>NOM-AUX-3POS EST-look-TR-3TR<br>ART deer                                                                                                           | səw kʷə́lləxtəs.<br>s-[ni?-s] wə-kʷə́lləx-t-əs<br>NOM-AUX-[3POS] EST-shoot-tr-3tr                                                                              | 20 |
| And he saw a deer.                                                                                                                                                                                                        | And he shot it.                                                                                                                                                |    |
| səsəw hí··ləm tə smə́yəθ<br>s-[ni?-s] wə-híl-əm tə smə́yəθ<br>NOM-AUX-3POS EST-roll-INTR ART<br>deer                                                                                                                      | tá…ž <sup>w</sup> wəx <sup>w</sup> ə <sup>?</sup> əĺá?ał tə<br>táž <sup>w</sup> wə-x <sup>w</sup> ə- <sup>?</sup> əĺá?əł [?ə]<br>exactly EST-become-aboard OBL | 21 |
| Then the deer <i>rolled</i> down                                                                                                                                                                                          | right into                                                                                                                                                     |    |

23

| tə snáx <sup>w</sup> əł | tá··ێ* wəx *ək *ək *cá··stəm tə smə́yəθ.            | 22 |
|-------------------------|-----------------------------------------------------|----|
| tə snáx*əł              | táx̆ wə-x ̆ə-k̆ ék̆ ̆əc-ás-t-əm tə sməyəθ           |    |
| ART canoe               | exactly EST-become-be.looking-face-TR-INTR ART deer |    |
| the canoe,              | so he was looking <i>right</i> into its face.       |    |

```
⁹ówəte⁹ słódallax^ws wani⁹as ctámat ti⁹í x^wa⁹alá⁹ał.
```

```
?ówəte? s-łódol-lox w-s wo-ni?-os ctámot to?í x wo-?olá?oł
```

none NOM-know-TR-3POS that-AUX-3SUB do.what.with.self this becomeaboard

He didn't know what happened to this thing that came aboard.

This account was told by Andrew Charles, speaking fairly fast, and relayed at dictation speed by Christine Charles on 2 November 1960. The incident happened when AC was a young man and living at Cowichan. The three of them, Gabriel Joe, Henry Louie, and Andrew, went by canoe from the Cowichan River out around Saltspring Island to "Plumper's Pass." Only Gabriel had a gun. In earlier times, hunters used pitch flares, but Gabriel had a "pit lamp," a miner's lamp worn on the forehead.

#### Line

1. The Halkomelem is literally 'went and recruited me,' but 'came and recruited me' seems more normal English. Perhaps in English the reference point of 'come' and 'go' moves with the speaker, while in Halkomelem it is the focus of the speech event, so that if I am no longer where I was recruited, they 'went' and recruited me.

 $h\delta l \dot{x} \partial w e^2$ , progressive of  $l\partial \dot{x} i \dot{w} \partial$  'hunt with a torch' (or in recent times with a miner's pit lamp, hence  $h\partial l \dot{x} \partial w \dot{e}^2$  is often rendered 'be going pit-lamping.'

3.  $t \dot{a} \dot{x}^{w}$  'exactly,' so recorded from CC; recorded  $t \dot{a} \ddot{x}^{w}$  from DK and  $t \dot{a} x^{w}$  from AG.

4.  $k^{w} \hat{u} k^{w}$  'cook,' from English. The *-t* 'transitive' was not recorded but was probably there; grammar requires either *-t* or  $2^{\circ}$  'oblique.'  $ne^{2} \partial nt$ , progressive of  $n\acute{e}t$  'be night.'

6.  $h \delta n \delta n \delta n$ , progressive of  $n \delta n \delta n$  'go.'  $k^w \delta c n \delta x^w$  'see it,' <  $k^w \delta c$  'see.' The full vowel of the root appears as schwa before  $-n \delta x^w$  'limited control transitive.'

9. <sup>?</sup>*əĺá*·*l* 'aboard,' probably a durative of <sup>?</sup>*á*·*l* (//<sup>?</sup>a1əł//?) 'go aboard.'

10-11. There are three instances here of i or  ${}^{?}i$  in a context where, because the place is elsewhere and the time is past, I would expect ni?. Perhaps the  ${}^{?}i$  makes the action more immediate.

10.  $?\acute{e}\cdot \dot{y}$  'be continuing,' progressive of  $?\acute{e}y$  'continue.'

11.  $\frac{\partial x}{\partial t}$ , plural progressive of  $\frac{\partial x}{\partial l}$  'paddle (v.).'

12-14. The material in parentheses was added after the rest of the text was dictated. This was at my request that the story include the sounds that AC had used in an earlier discussion. The  $px^w$  is sharp with gradually decreasing volume. CC remarked on its echoing quality. The  $\check{x}^{\cdots}$  rises in volume and then gradually falls and fades like the sound of a tree falling and echoing.

14. An article seems required, Gabe being the subject.

15. The particle  $\dot{yet}$  'expectable' perhaps gives the feeling of 'It's those pesky Little Choppers again!'  $\vec{k}^w \partial \vec{k}^w \partial q^w n \partial c t' \dot{t}s$  is literally 'be striking at the base of something with a clubbing or chopping motion.' 'Little Choppers' was CC's term. They are forest dwarves who fell trees. (For a discussion of these and similar beings, see Suttles 1987, 73-99.)

16.  $l\acute{e}\cdot \acute{y} \partial l p$ , from the root of  $sl\acute{e}\cdot \acute{y}$  'Douglas-fir bark' (thick bark suitable for fuel) and  $-\partial l p$  'plant,' and hence 'mature Douglas-fir tree.'

17. XíXcas 'islet,' diminutive of Xcés 'island.'

18.  $\dot{q}^{w} \acute{a} l \partial m$ , plural of  $\dot{q}^{w} \acute{i} m$  'disembark.'

19-20. According to CC's stated preference, the verb following the first two instances of  $s \partial w$  ought to be followed by -s 'third-person possessive,' but because the third instance is followed by  $-\partial s$  'third-person transitive subject,' it ought to be  $s \partial s \partial w$ . Apparent errors of this sort may have been mine in the course of taking dictation.

22.  $k^w \partial k^w c a st \partial m$  is a passive form, 'is being looked at in the face.' The deer is the subject of this passive. The agent must be Henry Louie, who was in the canoe.

1

3

## 23.4. MINK, BEAVER, AND SCOURING RUSH (JP 9)

ném cả ta čačí? dạn Xnécawtx<sup>w</sup> nem ta lélams ta sqalew. ném cả ta čačí? dạn Xnécawtx<sup>w</sup> nem [?a] ta lélam-s ta sqalew. go QUOT ART mink visit to [OBL] ART house-3POS ART beaver It is told that Mink made a visit to the house of Beaver.

wəx<sup>w</sup>k<sup>w</sup> ´ələ cə <sup>9</sup>al tə cícəsəm ni tə s<sup>9</sup>əx qalwəls tə leləm x sqəlew.
wə-x<sup>w</sup>k<sup>w</sup> ´ələ cə <sup>9</sup>al tə cícəsəm ni<sup>9</sup> [<sup>9</sup>ə] tə s<sup>9</sup>əx qalwəl-s tə leləm
[<sup>9</sup>ə] x sqəlew.

EST-scouring.rush QUOT just ART be.growing at [OBL] ART outside-3POS ART house [OBL] ART beaver

It was all scouring rushes, the growth outside of the house of Beaver.

ném cảw k<sup>w</sup>táx<sup>w</sup> tawẩa čačí<sup>9</sup>dan ta lélam ẩ sqaléw. ném cả wa-k<sup>w</sup>táx<sup>w</sup> tawẩa čačí<sup>9</sup>dan [?a] ta lélam [?a] ẩ sqaléw. go QUOT suddenly-enter that mink OBL ART house OBL ART beaver Mink went into the house of Beaver.

| <sup>9</sup> i c'ə scənedi devi scalara sqəledi. <sup>9</sup> i c'ə scənedi devi scalara sqəledi. AUX QUOT reclined ART beaver Beaver was reclining.                                                                                                                                    | yəwkwákwəy cə təwxa cəci?qən.<br>yə-wə-kwákwəy cə təwxa cəci?qən.<br>along-EST-hungry QUOT that mink<br>Mink was hungry.                                                                                                                                       | 4  |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----|
| <ul> <li>səw q*éls tə sqəléw, "yé, qáləm</li> <li>s-wə-q*él-s tə sqəléw yé qáləm</li> <li>NOM-EST-speak-3POS ART beaver we</li> <li>your-OBLNOM-come</li> <li>Then Beaver spoke, "Well, Pal, wh</li> </ul>                                                                              | əx, stéməs k <sup>w</sup> eθ x <sup>w</sup> əṁí."<br>əx stém-əs k <sup>w</sup> ə <sup>9</sup> əT-šx <sup>w</sup> - <sup>9</sup> əṁí<br>ll fellow what-3sub ART<br>at might be the reason for your coming?                                                      | 5  |
| <ul> <li>"?á.," səw q<sup>w</sup>éls tə čəčí?q'ən, "</li> <li>?á. s-wə-q<sup>w</sup>él-s tə čəčí?q'ən x.</li> <li>Ah NOM-EST-speak-3POS ART minl</li> <li>what</li> <li>"Ah " said Mink " my khhh-ches</li> </ul>                                                                       | x̂·wəłx̃łínəs cən. x̃?ówəte? stém<br>wəł-x̃əł-ínəs cən x̃?ówəte? stém<br>k kkhh-already-hurt-chest I kkhh-none<br>t hurts I kkhh-hayen't any                                                                                                                   | 6  |
| nəs <sup>9</sup> ə́łtən. XXa kwə wə <sup>9</sup> i·xw təwc<br>nə-s- <sup>9</sup> ə́łtən XXa kwə wə- <sup>9</sup> i-əxw<br>my-NOM-EAT kkhh-BE3P then if-AUX<br>khhh-food. The khh-chance that yo                                                                                         | cáílə k <sup>w</sup> stém ?al<br>təw-c-áílə k <sup>w</sup> stém ?al<br>-you somewhat-get-store ART what just<br>u might have something stored away                                                                                                             | 7  |
| xnəšx <sup>w</sup> i wələmi k <sup>w</sup> ətx <sup>w</sup> iləm." x-nə-šx <sup>w</sup> - <sup>9</sup> i wəl- <sup>9</sup> əmi k <sup>w</sup> ətxilən<br>kkhh-my-OBLNOM-AUX already-com<br>is khh-why I came in."                                                                       | n<br>ne enter                                                                                                                                                                                                                                                  | 8  |
| "?á·," θớt cờ tờ sqəlcw, "?əθx<br>?á· θớt cờ tờ sqəlcw ?əT-šx "-<br>Aw say QUOT ART beaver your-OF<br>"Aw," said Beaver, "How could y                                                                                                                                                   | " <sup>?</sup> əwé <sup>?</sup> e k <sup>w</sup> ə <sup>?</sup> áyè <sup>?</sup> təx <sup>w</sup><br><sup>?</sup> əwə <sup>?</sup> ə k <sup>w</sup> ə <sup>?</sup> áyè <sup>?</sup> -t-əx <sup>w</sup><br>BLNOM-not ROG then devour-TR-you<br>you begin to eat | 9  |
| tən swe? nəs?áłtən. kwəw syáł 7<br>tə nə-s-we? nə-s-?áłtən kwə wə<br>ART my-NOM-own my-NOM-eat ART<br>NOM-supply<br>my food? After all, my own supp                                                                                                                                     | 'al m̀ə tən swé? sq́ílə.<br>-syáł ?al m̀ə tə nə-s-wé? s-q́ílə.<br>EST-wood just CERT ART my-NOM-own<br>Ilies are just wood.                                                                                                                                    | 10 |
| sk <sup>w</sup> éy k <sup>w</sup> ə0dák <sup>w</sup> ənəx <sup>w</sup> <sup>9</sup> iwáwə lá<br>sk <sup>w</sup> éy k <sup>w</sup> ə <sup>9</sup> əT-s-dək <sup>w</sup> -nəx <sup>w</sup> <sup>9</sup> iwa<br>impossible ART your-NOM-bite-TR li<br>You couldn't bite it without your te | ok «lək » ?al ten yənəs."<br>áwə lək «lək » ?al tə ?ən-yənəs<br>kely break(PL) just ART your-tooth<br>eth very likely breaking."                                                                                                                               | 11 |
| "X?á. si?ém, ?á.," θ>t tə čəčí?<br>X -?á. si?ém ?á. θ>t tə čəčí?q<br>k*[~ k*i] s?éXq<br>khhh-ah Sir ah say ART mink NOM<br>"Kkhh-yes, Sir, yes," said Mink.                                                                                                                             | ἀǝἦ. səw ?áᢜqǝl. tǝs kʷs?éᢜq.<br>ǝἦ s-wə?áᢜqǝl[-s] tǝs [?ǝ]<br><i>ι</i> -EST-exit[-3POS] reach OBL ART outside<br>And he went out. When he got outside,                                                                                                        | 12 |
|                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                |    |

səw x<sup>w</sup>ixa tə x<sup>w</sup>k<sup>w</sup>ələ. q<sup>w</sup>əlstx<sup>w</sup>əs, hək<sup>w</sup>nəsəs təwxa cəči<sup>9</sup>qən 13 s-wə-x<sup>w</sup>i-<sup>\*</sup>⁄<sub>1</sub>a tə x<sup>w</sup>k<sup>w</sup>ələ q<sup>w</sup>el-stəx<sup>w</sup>-əs hek<sup>w</sup>-nəs-əs təw<sup>\*</sup>⁄<sub>2</sub>a čəčí<sup>?</sup>qən NOM-EST-become-BE3P ART scouring.rush speak-COM-3TR remember-GOAL-3TR that mink it became Scouring Rush's turn. Mink spoke to him. He remembered k<sup>w</sup>s wəyáθs wəsłélq tə x<sup>w</sup>k<sup>w</sup>álə. səw łdíltəs 14  $k^w$  s-wə-yá $\theta$ -s wə-słélq tə x<sup>w</sup>k<sup>w</sup>álə s-[ni-s] wə-łdíl-t-əs ART NOM-EST-always-3POS EST-immersed ART scouring.rush NOM-AUX-3POS EST-ask-TR-3TR that Scouring Rush was always under water, so he asked him, "sk wéy <sup>γ</sup>ə wətítənàstələt. Xán səw cécəwəθàmə 15 skwéy ?ə wə-titənàs-təl-ət Xá nə-s-wə-cécəw-ət-Samə impossible ROG if-be.sitting.beside-each.other-we BE3P my-NOM-EST-be helping-TR-you "Couldn't we sit beside one another? Then I will be helping you wəmi·s kw st<sup>θ</sup>qwáy." 16 wə-mi-əs kw stoqwáv if-come-3SUB ART small fish when any little fish come." wəłk wák wəy con wanan. Xa sx wx stas toe? taw Xa cací? dan. 17 wəł-kwákwəy cə bəł wə-nán xa sxw-xátə-s tbé? təwka cəcí?dən already-hungry QUOT though EST-very BE3P OBLNOM-be.saying-3POS that that mink But Mink was very hungry. That's why he said that. səw q<sup>w</sup>éls tə x<sup>w</sup>k<sup>w</sup>ələ, 18 s-wə-q<sup>w</sup>él-s tə x<sup>w</sup>k<sup>w</sup>ələ NOM-EST-speak-3POS ART scouring.rush Then Scouring-rush spoke, "x̃ə<sup>9</sup>mm, <sup>9</sup>əθx <sup>w</sup>əwé <sup>9</sup>ə k<sup>w</sup>ə x័əte·x<sup>w</sup> tən swe<sup>9</sup> nəsk<sup>w</sup>ə́yxθət. 19 kkhh-hmm your-OBLNOM-not ROG then be-doing-you ART my-NOM-own my-NOM-move-self "Khhh-hmm, how could you do as I do? ha? mi kweb stém ?i waxwkwáyebasam can ?al. 20ha? mi kweð stém ? vy wo-xwkwáy dosom con ?al if come what and EST-be.shaking.head I just When something comes, I just shake my head.

| γanaca cx * k*a wami's k*θa θi q*ley. γánaca čx* k*a wami's k*θa θí q*léy where you then if-come-3SUB ART big log Where would you be if a big log comes?                                                                                                                                                        | 21 |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----|
| <sup>d</sup> <sup>0</sup> íq <sup>2</sup> əθà·m, né·m yəxálcdət. sk <sup>w</sup> éy k <sup>w</sup> ed həlí." <sup>d</sup> <sup>0</sup> íq <sup>2</sup> ət-Sà·m ném yə-xálc-dət sk <sup>w</sup> éy k <sup>w</sup> ə əT-s-həlí. be.stepped.on-tr-you(PAS) AUX along-be.turning-self imposssible ART your-NOM-live | 22 |
| You'll get stepped on. It will go rolling along. You couldn't live."                                                                                                                                                                                                                                            |    |
| "X·?õ" sow dot to čočí?qon, "?ówo čx" łni qáy to nówo,<br>X·?n. s-wo-dot[-s] to čočí?qon ?ówo čx" ł ni qáy to nówo<br>kkhh-hmm NOM-EST-say-3POS ART mink not you yet AUX die ART you<br>"Kkhh hmm," said Mink, "You haven't died yet,                                                                           | 23 |
| ?i ?é·ሰθə ctwa? kʷáṁkʷəṁ xঁʷte· አႆ nówə."<br>?əỷ ?é·ሰθə ctwa? kʷáṁkʷəṁ xঁʷte ?ə አႆ nówə<br>and I spec strong toward OBL ART you<br>and I must be stronger than you."                                                                                                                                            | 24 |
| "?a.," səw θəts cə tə x.«k.»ələ, "ha? nem yəlew k.»θə q.»ley<br>?a. s-wə-θət-s cə tə x.«k.»ələ ha? nem yəlew k.»θə q.»ley<br>ah NOM-EST-say-3POS QUOT ART scouring.rush if go pass ART log<br>"Well," said Scouring Rush. "If a log passes                                                                      | 25 |
| k™smis ť <sup>0</sup> íq́ə0amxəs ?i Że? cən wə?ámət ?al,<br>k™ s-mi-s ť <sup>0</sup> íqํ-ət-Samx-əs ?əỷ Že? cən wə-?ámət ?al<br>ART NOM-AUX-3POS step.on-TR-me-3TR and again I EST-sit.up just<br>when it has rolled over me, I just get up again and                                                           | 26 |
| nəsẩəw x <sup>w</sup> k <sup>w</sup> əydəsəm ?al."<br>nə-s-ẩe? wə-x <sup>w</sup> k <sup>w</sup> əydəsəm ?al<br>my-NOM-again EST-shake.head just<br>then I just shake my head again."                                                                                                                            | 27 |
| "ž?əॖ.,"ðát cð tð čðčí?qðn, "?áý ctwa? kwðn swðté?t."<br>ž?nn. ðát cð tð čðčí?qðn ?áý ctwa? kwð nð-s-wð-té?-t<br>kkhh-hmm say quot art mink good spec art my-nom-est-try-tr<br>"Kkhh-hmm," said Mink, "Perhaps I'd better try it."                                                                              | 28 |
| "héỷ θəł ?éwəł," səw θəts cə tə x <sup>w</sup> k <sup>w</sup> əldə.<br>héỷ θəł ?éwəł s-wə-θət-s cə tə x <sup>w</sup> k <sup>w</sup> əldə<br>proceed though REAS NOM-EST-say-3POS QUOT ART scouring.rush<br>"Go ahead then," said Scouring Rush.                                                                 | 29 |

```
"mi čx^w ?ámat statés Χ ?é·nθa."
 30
mi čx^w ⁹ómət stətés [⁹ə] Å ⁹é·nθə
come you sit near [OBL] ART me
"Come sit near me"
 sow 'omots to čočí'don nem tonostom towka x k volo.
 31
 swo-'omot-s to čočí'don nem tonos-t-om towka x k vólo
 NOM-EST-Sit-3POS ART mink AUX sit.beside-TR-INTR that scouring.rush
 So Mink sat down next to Scouring Rush.
?ówo co ni·s yohíθ ?iwoł mi wíl to qwłey. θí qwłey. yožá·lcθot.
 32
?ówo co ni-os yo-híθ ?oỷ woł-mi wíl to qwłéy θí qwłéy yo-xálc-θot
not QUOT AUX-3SUB along last.long and already-come appear ART log big log
 along-be.turning-self
It wasn't long when a log appeared, a big log, rolling along.
 "syá" čx kwe," sow dóts to xwkwólo. "i towł yomi to?od xomén."
 33
 syá· čx^w k^wə s-wə-θət-s tə x^wk^wələ ⁹i⁹ətə wəł-yə-mi tə ⁹əT-sxəmén
 be-warned you then NOM-EST-say-3POS ART scouring.rush here already-along-
 come your-enemy
 "Be careful," said Scouring Rush. "Here comes your enemy."
 "X?ó", ?ó"," səw θáts ca ta čačí?qan.
 34
 "X?ó" ?ó" s-wə-θát-s ca ta čačí?daň
 kkhh-oh oh NOM-EST-say-3POS QUOT ART mink
 "Kkhh-oh, oh," said Mink.
 wəłmí tə qwłey, t^{\theta}ídətəm ⁹é·łtən, nem yəlew. ⁹ómət tə xwkwólə
 35
 wəł-mí tə qwłéy t⁰íd-ət-əm ?é·ltən nem yəléw ?émət tə xwkwála
 already-come ART log step.on-TR-INTR 3PL go pass sit.up ART scouring.rush
 When the log came, they got rolled over, and it went past. Scouring Rush
 sat up,
x^wk^wəyθəsəm ste⁹etəwyáθ wəsk^wáyžθəts,
 36
x^wk^wəyθəsəm ste⁹é [⁹ə] tə wə-yáθ wə-s-k^wáyž-θət-s
shaking.the.head like [OBL] ART EST-always EST-move-self-3POS
shaking his head as he always did,
cáləsəm sówqtəs k^wθə stónəsəł čəčí?qan.
 37
cáləsəm sówq-t-əs k^wθə stónəs-əł čəčí?qon
turn.to.look seek-TR-3TR ART sitting.beside-past mink
and he looked back seeking Mink, who had been sitting beside him,
[?]i ni cowł nem yohówdw yox wkwo?íqoń sť⁰oxnís.
 38
'vay ni ca wał-nem ya-háwdw ya-xwkwa'íqań sť
but AUX QUOT already-go along-drifting along-belly.up teeth.showing
But he was gone, drifting along downstream, belly up, teeth showing.
```

ni ?əsép tə sxॅwəyém X čəčí?qən ?i tə sqəléw ?i tə xwk̈wələ. 39 ni ?əsép tə sxॅwəyém [?ə] X čəčí?qən ?əy tə sqəléw ?əy tə xwk̈wələ AUX end ART story [OBL] ART mink and ART beaver and ART scouring.rush So ends the story of Mink, Beaver, and Scouring Rush.

This text was dictated by James Point on 22 March 1963, and reviewed with Arnold Guerin in 1983.

Line

1.  $\dot{\lambda} n\acute{e}c \partial w t x^w$  'visit.' Composed of the prefix  $\dot{\lambda}$ - 'go to,' the root  $n\acute{e}c$  'differ, be different,' and the lexical suffix  $-\partial w t x^w \sim -\acute{e}w t x^w$  'house.'

2.  $x^{w}k^{w}\delta l \partial$  'scouring rush' – the species *Equisetum hiemale*, a relative of the horsetail; it and dogfish skin were used like sandpaper. The word  $x^{w}k^{w}\delta l \partial$  has the same root as  $x^{w}\delta k^{w}\partial t$  'polish it.'

cícosom. Progressive of císom 'grow.'

 $s^2 \delta \dot{\lambda} q \dot{a} l w \partial t$ . Composed of the prefix *s*- 'nominalizer,' the root  $\sqrt{2} e \dot{\lambda} q$  'out,' and the lexical suffix  $-\dot{a} l w \partial t$  'side.' AG pronounced the word  $s^2 \delta \dot{\lambda} q \dot{a} l w e^2 t$ .

3. The phrase  $t \ge l \le l \ge d$  is a q  $\exists \vec{X} = s = q \ge d \le d$  is an alternate way of saying 'the house of Beaver' or perhaps 'the house of the beaver.' However, because the article  $\vec{X}$  must appear before a proper noun or a pronoun in the oblique case, the use of it here suggests that Beaver, Mink, and Scouring Rush are proper nouns, the names of characters.

4. scańewas 'reclining.' Probably a resultative form, composed of *s*- 'resultative,' the root can 'lean against, be propped up,' and the lexical suffix -iws 'body.' JP said that the word implied that Beaver was lying back with his hands clasped behind his neck, looking self-satisfied, we may assume.

 $y_{\partial}$ - 'along.' Generally implies movement or duration; here it may suggest that Mink's hunger was an ongoing condition.

5. *yé* 'well.' This interjection might also be translated 'all right' or 'go ahead.' It is used to urge someone to speak or to continue, as by the audience listening to a story being told.

 $q\delta l \partial m \partial x$  'pal.' Evidently a familiar, perhaps patronizing, form of address. AG thought the root might be  $q\delta l$  'bad,' but it looks like an Interior Salish word for 'person.'

*stéməs*. The *-əs* is 'third-person subordinate subject' (3SUB). The subordinate subject series appears in subordinate clauses. Here there is probably an implied main clause such as 'I wonder ...'

6. In stories, Mink, Raven, and perhaps some other disreputable characters begin words with the sound  $\check{x}$  ...

9.  ${}^{2}\partial\theta x^{w}{}^{2}\partial w \hat{e}{}^{2}e \ k^{w}\partial \ {}^{2}\delta y \hat{e}t \partial x^{w}$  ... This is literally 'Is it then the reason for your not eating my food?' The same construction appears in line 19 with a similar interpretation, 'How could you ...?'

10.  $sw\acute{e}^{2}$  with the *s*- 'nominalizer' is literally 'property';  $cw\acute{e}^{2}$  with the *c*-'verbalizer' is 'to own.' Speakers vary in whether they use the possessive with both the  $sw\acute{e}^{2}$  and the noun that follows. Here JP used the possessive with the following noun the first time and omitted it the second. When AG read the story, he put in a possessive where JP had omitted it.

 $k^{w} \partial w syát ?al model$ . The  $k^{w} \partial s$  implies hypothetical existence, indicating that Beaver is referring to *some* wood, not necessarily any that Mink can see. However, the particle  $m\partial \partial (CERT)$  implies that the statement is based on direct experience, as opposed to hearsay. Beaver of course *knows* that's what his food is. The interpretation 'of course, it is' was given to the same construction in another text.

11.  $l \neq k^w l \neq k^w$  is the plural of  $l \neq k^w$  'break, get broken.'

14. stélq. Resultative of the root talq, seen in talqt 'immerse it, dip it.'

15. titanastal 'be sitting beside each other,' the progressive of tanastal 'sit beside each other,' which is the reciprocal of tanast 'sit beside him/her.'

16. Some people identify  $st^{\theta}q^{w}ay$  as 'trout,' others as any small fish.

17.  $\dot{x}\dot{\partial t}\partial$ . Progressive of both  $\theta \dot{\partial t}$  'say,' as it is here, and  $\dot{x}t\dot{e}^{2}$  'do,' as in line 19.

21.  $q^{w}t\dot{e}\dot{y}$  is usually glossed 'log,' but it is composed of the root  $q^{w}\delta t$  'wash ashore' and  $-\dot{e}\dot{y}$ , an old lexical suffix meaning 'tree' or 'plant,' and so the word probably originally meant specifically a driftwood log.

23. The  $\tilde{a}$  is a nasalized schwa.

26. <sup>2</sup>*ámət* is 'sit down' if one is standing or 'sit up, get up' if one is lying down.

30. statés 'near,' the resultative of tás 'arrive there.'

31. Mink is the subject of  $2\delta m \partial t$  'sit down,' but  $t\delta n \partial s t \partial m$  is the third-person passive form of  $t\delta n \partial s t$  'sit beside,' and its subject is Scouring Rush. This shift to the passive may occur when the previous verb is intransitive. See also line 35. The use of  $t\partial w \dot{\lambda} a$  also indicates that Scouring Rush is the subject.

33. syá 'careful,' lit. 'warned,' the resultative of yát 'warn him/her.'

35.  $t^{\theta}i\dot{q}atam$  is a passive form. Mink and Scouring Rush, indicated by  $2\dot{e}ttan$  'third-person plural' (3PL) are its subject.

38.  $st \delta n \partial s$ , the nominalization of  $t \delta n \partial s$  'sit beside.' The phrase  $k^w \theta \partial s t \delta n \partial s \partial t \delta t \delta t$  might also be translated 'the one who sat beside him, who was Mink.'

The conjunction  ${}^{2}i \sim {}^{2}\partial y$  can mean either 'and' or 'but' depending on context.  $h \dot{\sigma} w \dot{q}^{w}$  is the progressive of  $w \dot{\sigma} \dot{q}^{w}$  'drift downstream.'

 $\check{x}^{w}\check{k}^{w}\partial^{2}iq\partial n$  'belly up' is composed of the prefix  $x^{w}$ , which seems often required when the reference is to a body part, the root  $\check{k}^{w}i^{\gamma}$  'ascend, upward,' and the suffix  $-iq\partial n$  'front.'

 $s\dot{t}^{\theta}\partial \dot{x}n\dot{t}s$  'with teeth showing' is composed of *s*- 'resultative,' a root  $\dot{t}^{\theta}\partial \dot{x}$ , not identifiable, and  $-\dot{a}n\partial s \sim -n\dot{t}s$  'tooth.' There is a verb  $\dot{t}^{\theta}\partial \dot{x}n\dot{t}s\partial m$  'show the teeth, grin.'

# 23.5. THE ORIGIN OF THE NAME "MUSQUEAM" (JP 13)

| <ul> <li>wəłqé?is ?ə ċtwa? tə sq<sup>w</sup>í·lq<sup>w</sup>əls k<sup>w</sup>θə yəwendəł məstayəx<sup>w</sup>.</li> <li>wəł-qeydəs ?ə ċtwa? tə s-q<sup>w</sup>í·lq<sup>w</sup>əl-s k<sup>w</sup>θə yəwendəl məstayəx<sup>w</sup></li> <li>already-formerly ROG SPEC ART NOM-be.telling-3POS ART before-past people</li> <li>It must have been long ago, according to what the people of the past used to tell.</li> </ul>                                                  | 1      |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------|
| ni c'ə kwə ?i kwikwəxtəm kwə xwməmqwe.m. nəwł ste?e<br>ni c'ə ?kwə ?i kwikwəx-t-əm kwə xwməmqwe.m ni? wəł-ste?e<br>AUX QUOT ART AUX be.naming-TR-INTR ART mumquaam AUX already-like<br>There was a place called "Mumquaam." It is somewhat                                                                                                                                                                                                                                 | 2      |
| k̃wəw xʷcႆsí·m ?al təna qé?is i ni θəł wəxăca?.<br>k̃wə wə-xʷ-cႆsí·m ?al [?ə] təna qé?is ?əỷ ni? θəł wə-xăca?<br>ART EST-within-grow just [OBL] this time.past and AUX but EST-lake<br>grown in now, but is still a lake.                                                                                                                                                                                                                                                  | 3      |
| sxéxəm, <sup>9</sup> əx <sup>w</sup> ín, q <sup>w</sup> əmcáls k <sup>w</sup> θə ni cícəsəm ni <sup>9</sup> ə k <sup>w</sup> θe <sup>9</sup> .<br>sxéxəm <sup>9</sup> əx <sup>w</sup> ín q <sup>w</sup> əmcáls k <sup>w</sup> θə ni cícəsəm ni <sup>9</sup> ə k <sup>w</sup> θe <sup>9</sup><br>shallow small cranberry ART AUX be.growing be.there OBL that<br>It is shallow, small, and cranberries are what is growing there.                                           | 4      |
| <sup>?</sup> əwé·'łte·ł ċθəł kw stém ni ?ə kwθe?. wəxáca? ?al, ?əxwín xáca?.<br>?əwə-əł-te?-əł ċə θəł kw stém ni ?ə kwθe? wə-xáca? ?al ?əxwín xáca?<br>not-past-appear-past QUOT but ART what be.there OBL that EST-lake just small<br>lake<br>There was nothing there, though. There was just a lake, a small lake                                                                                                                                                        | 5      |
| Xa k <sup>w</sup> ə xə́təstəm šx <sup>w</sup> əní?s k <sup>w</sup> θe? sX́éləqəməł. s?i·łqəy.<br>Xa k <sup>w</sup> ə xə́tə-st-əm šx <sup>w</sup> -ní?-s k <sup>w</sup> θe? sX́éləqəm-əł. s?i·łqəy.<br>BE3P ART be.saying-CAUS-INTR OBLNOM-be.there-3POS ART fierce-past seelthkey<br>It was said to be where that monster was. It was a seelthkey.                                                                                                                         | 6<br>у |
| wəyáθ wəsq <sup>w</sup> í·lq <sup>w</sup> əl. sk <sup>w</sup> éy cə k <sup>w</sup> s nems tsə́θət k <sup>w</sup> ə <sup>9</sup> íməx.<br>wə-yáθ wə-sq <sup>w</sup> í·lq <sup>w</sup> əl. sk <sup>w</sup> éy cə k <sup>w</sup> s-nem-s təs-θət k <sup>w</sup> ə [yə-] <sup>9</sup> íməx<br>EST-always EST-told impossible QUOT ART NOM-go-3POS reach-self ART<br>[along-]be.walking<br>They were always telling. It was impossible for someone walking to go<br>near there. | 7      |
| x wəna? al mi císəm tə skíkqəł iwəł yə́dəstəm, ya təm.<br>x wəna? al mi císəm tə skíkqəł ?əy wəł-yə́d-əs-t-əm ya təm<br>when.first just come grow ART child and already-tell-RECIP-TR-INTR<br>As soon as a child grew up, he was told about it and warned.                                                                                                                                                                                                                 | 8      |

| "?ówə čx <sup>w</sup> hónəməx <sup>w</sup> k <sup>w</sup> łe xáca? ni k <sup>w</sup> i cáləq <sup>w</sup> . skéləqəm.<br>?ówə čx <sup>w</sup> hónəm-əx <sup>w</sup> k <sup>w</sup> łe xáca? ni [?ə] k <sup>w</sup> i cáləq <sup>w</sup> skéləqəm<br>not you be.going-you that lake in [OBL] that inland fierce.being<br>"Don't be going to that lake back up there. There is a monster.                                                                                                        | 9   |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|
| s <sup>9</sup> í·łqəỷ k <sup>w</sup> łə ni ní <sup>9</sup> ə k <sup>w</sup> θe <sup>9</sup> . ha <sup>9</sup> čx <sup>w</sup> wəṁé·n i neṁ čx <sup>w</sup> ,<br>s <sup>9</sup> í·łqəỷ k <sup>w</sup> łə ni ní <sup>9</sup> <sup>9</sup> ə k <sup>w</sup> θe <sup>9</sup> ha <sup>9</sup> čx <sup>w</sup> wə-ṁé·n <sup>9</sup> əỷ neṁ čx <sup>w</sup> ,<br>seelthkey ART AUX be.there OBL that if you EST-weak and go you<br>It's is a seelthkey that is there. If you're foolish enough to go, | 10  |
| sk <sup>w</sup> éy k <sup>w</sup> əθ <sup>*</sup> λəw mi qá·nθət. Xa cə əθwəném címəl,<br>sk <sup>w</sup> éy k <sup>w</sup> ə 'əT-s-Xe wə-mi qá·nθət Xa cə 'əT-s-wə-ném címəl<br>impossible ART your-NOM-also EST-AUX(come) return-self BE3P QUOT<br>your-NOM-EST-go approach<br>you won't be able to return. They say if you go near.                                                                                                                                                         | 11  |
| wəłxəlxəlcda m. Xa ?ə0niw qáy."<br>wəł-xəlxəlc-t-Sa·m Xa ?əT-s-ni? wə-qáy<br>already twist(PL)-TR-you(PAS) BE3P your-NOM-AUX EST-die<br>you will get twisted up, and you will die."                                                                                                                                                                                                                                                                                                            | 12  |
| Ža šx <sup>w</sup> sk <sup>w</sup> éys k <sup>w</sup> wét ném. wəθə?ít sŽéləqəm.<br>Ža šx <sup>w</sup> -sk <sup>w</sup> éy-s k <sup>w</sup> ə wét ném wə-θə?ít sŽéləqəm<br>вЕЗР ОВLNOM-impossible-зРОЅ АRT who go EST-true fierce<br>That's why no one could go there. It was truly a monster.                                                                                                                                                                                                 | 13  |
| néč čə k <sup>w</sup> s wəłnéms x <sup>w</sup> əné∙nt, cəlél i θətíl,<br>néč čə k <sup>w</sup> s-wəł-ném-s x <sup>w</sup> ə-né?ənt cəlél ?əỷ θət-íl<br>different QUOT ART NOM-already-go-3POS become-being.night nearly<br>and dark-become<br>Sometimes when it was evening nearly dark                                                                                                                                                                                                        | 14  |
| <ul> <li>iwəł cłém t was evening, neury tana,</li> <li>iwəł cłém t čx<sup>w</sup> k<sup>w</sup>θə má?aq<sup>w</sup> wəniəx<sup>w</sup> yə?ím t təwstətés.</li> <li>?əỷ wəł-cłém -ət čx<sup>w</sup> k<sup>w</sup>θə má?əq<sup>w</sup> wə-ni?-əx<sup>w</sup> yə-?ím təw-stətés and already-hear-TR you ART duck if-AUX-you along-be.walking somewhat-near</li> </ul>                                                                                                                             | 15  |
| you neard a duck il you were walking lairiy near.<br>"q <sup>w</sup> ǽ·· q <sup>w</sup> ǽ·· q <sup>w</sup> ǽ··" ste <sup>9</sup> é· k̄ <sup>w</sup> tónəqsən. X̄a cdəł k <sup>w</sup> s wəndáns wəx <sup>w</sup> θíqən.<br>q <sup>w</sup> ǽ·· q <sup>w</sup> ǽ·· ste <sup>9</sup> é <sup>9</sup> ə k̄ <sup>w</sup> tónəqsən X̄a cə dəł k <sup>w</sup> s-wə-nán-s<br>wə-x <sup>w</sup> -θí-qən                                                                                                  | 16  |
| quack quack like OBL ART mallard BE3P QUOT but ART NOM-EST-very-3<br>in-big-throat<br>"Quack! Quack!" It was like a mallard, but it was very loud.                                                                                                                                                                                                                                                                                                                                             | POS |

| ni ἀəw híθ k <sup>w</sup> sni?s ní ?ə k <sup>w</sup> θe? ێáca?əł ἀtwa?, i yeł<br>ni? ἀə wə-híθ k <sup>w</sup> s-ni?-s ní? ?ə k <sup>w</sup> θe? ێáca?-əł ἀtwa? ?əỷ yeł<br>AUX QUOT EST-last.long ART NOM-AUX-3POS be.there OBL ART lake-past<br>SPEC and after.which<br>It was quite a while that it was there in what was then a lake before                                        | 25 |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----|
| sismi, Žəw sq <sup>w</sup> í·lq <sup>w</sup> əls k <sup>w</sup> θə sìyáləx <sup>w</sup> a·ł ċtwa?. mi <sup>…</sup> qəyá?θət,<br>s- <sup>?</sup> i-s mi Že wə-s-q <sup>w</sup> í·lq <sup>w</sup> əl-s k <sup>w</sup> θə syəyáləx <sup>w</sup> a-əł ċtwa? mi <sup>…</sup><br>qəyá?-θət,<br>NOM-AUX-3POS come again EST-NOM-tell-3POS ART old.people-past SPEC<br>come leave.den-self   | 26 |
| it came out, according to what the old people also said. It left its den                                                                                                                                                                                                                                                                                                             |    |
| <ul> <li>mi ťá "x ". wəs xə<sup>9</sup>ét<sup>θ</sup>. təwq á x məst áy ə x " wəs x ə xí<sup>9</sup>st ə x ",</li> <li>mi ťá x wə-s xə<sup>9</sup>ét<sup>θ</sup> təw-q á x məst áy ə x " wə-s x a xí<sup>9</sup>-st ə x "</li> <li>come down.shoreward EST-marked somewhat-many person</li> <li>EST-revealed-CAUS</li> </ul>                                                         | 27 |
| to come down to the shore. It was in plain view. There were quite a few people who had the chance to see it                                                                                                                                                                                                                                                                          |    |
| <ul> <li>k<sup>w</sup>is wəłmi yə?é·ý. wəxi·xłémətəm k<sup>w</sup>θə niwłíθ</li> <li>k<sup>w</sup> s-?i-s wəł-mi yə-?é·ý wə-xi·xłém-ət-əm [?ə] k<sup>w</sup>θə ni? wəł-híθ</li> <li>ART NOM-AUX-3POS already-AUX along-be.continuing EST-be.watching-TR-INTR [OBL] ART AUX aleady-last.long</li> <li>as it came along. It was being watched by people who had known about</li> </ul> | 28 |
| k <sup>w</sup> słádallax <sup>w</sup> as mastáyax <sup>w</sup> .<br>k <sup>w</sup> s-łádal-lax <sup>w</sup> -as mastáyax <sup>w</sup><br>ART NOM-know-TR-3TR person<br>it for a long time.                                                                                                                                                                                           | 29 |
| <ul> <li>"sᢜéləqəm. sᢜéləqəm, łwáləp, k<sup>w</sup>á?əyθət ce·p. ?áwə ce·p i sxañcə?e·p."</li> <li>ᢜéləqəm sᢜéləqəm łwáləp k<sup>w</sup>á?əy-θət ce·p ?áwə ce·p yə-sxañcə?-e·p</li> <li>fierce fierce you.folks keep.away-self you(PL) not you(PL) beforebe.victim you(PL)</li> <li>"It's a monster. A monster, you folks. Move away. Don't let it catch you."</li> </ul>            | 30 |
| <ul> <li>səw ste?é k<sup>w</sup>θe x<sup>w</sup>əlməx<sup>w</sup>əł čtwa?. ne<sup></sup>m θələqtəl tə x<sup>w</sup>əlməx<sup>w</sup>əł čtwa?.</li> <li>s-wə-ste?é k<sup>w</sup>θe x<sup>w</sup>əlməx<sup>w</sup>-əł čtwa? nem θələq-təl tə x<sup>w</sup>əlməx<sup>w</sup>-əł čtwa?</li> <li>NOM-EST-like ART people-past SPEC go divide-each.other ART people-past SPEC</li> </ul>   | 31 |
| So the people remained that way. The people divided                                                                                                                                                                                                                                                                                                                                  |    |

| ste <sup>9</sup> é <sup>9</sup> e k <sup>w</sup> əw cxéłstəm təw <sup>*</sup> xa s <sup>*</sup> xéləqəm k <sup>w</sup> ík <sup>w</sup> əxtəm k <sup>*</sup> wə s <sup>9</sup> í·łqə <sup>*</sup> y.<br>ste <sup>9</sup> é <sup>9</sup> ə k <sup>*</sup> wə wə-c-xéł-st-əm təw <sup>*</sup> xa s <sup>*</sup> xéləqəm k <sup>*</sup> ík <sup>*</sup> wəx-t-əm<br>k <sup>*</sup> wə s <sup>9</sup> í·łqə <sup>*</sup> y<br>like OBL ART EST-make-path-CAUS-INTR that fierce one be paming-TR-INTR | 32            |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------|
| ART seelthkey<br>so as to make a path for that monster that is called the seelthkey.                                                                                                                                                                                                                                                                                                                                                                                                            |               |
| <ul> <li>səw nem yə?é·ý ne…m tə θí stáləw. nem x »əšx »táləwèmən</li> <li>s-wə-nem[-s] yə-?é·ý nem [?ə] tə θí stáləw nem x »ə-šx »-táləw-èmən</li> <li>NOM-EST-go-3POS along-be.continuing go OBL ART big river AUX(go) become-OBLNOM-river-track</li> <li>It continued on to the big river. Where it went became the course</li> </ul>                                                                                                                                                         | 33            |
| <ul> <li>tə ni šx "yəłe?és. mək" stem tə ni t<sup>0</sup>íq təs cíc əsəm, sáx "əl, pték "əm,</li> <li>tə ni šx "-yə-lə?é-s mək" stem tə t<sup>0</sup>íq -ət-əs cíc əsəm sáx "əl pték "əm ART AUX OBLNOM-along-follow-3POS all what ART AUX step.on-TR-3SUB be.growing grass fern</li> <li>of a stream. Everything that it passed over, grass, ferns,</li> </ul>                                                                                                                                 | 34            |
| <ul> <li>k<sup>w</sup>θəw mök<sup>w</sup> ?al stém, nə qáy, k<sup>w</sup>snis wəłnem yəléw k<sup>w</sup>łəwka<br/>skéləqəm.</li> <li>k<sup>w</sup>θə wə-mök<sup>w</sup> ?al stém ni? qáy k<sup>w</sup> s-ni?-s wəł-nem yəléw<br/>k<sup>w</sup>łəwka skéləqəm</li> <li>ART EST-all just what AUX die ART NOM-AUX-3POS already-go pass that<br/>fierce.one</li> <li>everything, died as that monster passed by.</li> </ul>                                                                        | 35            |
| təwqə́x cə mstə́yəx <sup>w</sup> k <sup>w</sup> θə niw scécən wək <sup>w</sup> cáləs.<br>təw-qə́x cə məstə́yəx <sup>w</sup> k <sup>w</sup> θə ni? wə-scécən wə-k <sup>w</sup> ec-áləs<br>somewhat-many QUOT person ART AUX EST-really EST-see-eye<br>There were many people who saw it with their own eyes.                                                                                                                                                                                     | 36            |
| <ul> <li><sup>9</sup>óý sžłéms. iséla ta sžáýass yak "anátal k "snis yaž "ák "Øat.</li> <li><sup>9</sup>óý s-žłém-s yaséla ta sžáýas-s ya-k "anátal k "s-ni?-s ya-ž "ák "-Øa</li> <li>good NOM-watch-3POS two ART head-3POS along-together ART NOM-AUX-3PO</li> <li>along-be.dragging-self</li> <li>They had a good view. It had two heads together as it was dragging itself along.</li> </ul>                                                                                                 | 37<br>ət<br>s |
| stə <sup>9</sup> é ce k <sup>w</sup> θí šx <sup>w</sup> k <sup>w</sup> é <sup>9</sup> em tə s <sup>9</sup> ənx <sup>w</sup> ícəns θəwÅa sÅéləqəm.<br>stə <sup>9</sup> é cə k <sup>w</sup> θí šx <sup>w</sup> k <sup>w</sup> é <sup>9</sup> em tə s <sup>9</sup> ənx <sup>w</sup> ícən-s θəwÅa sÅéləqəm<br>like QUOT ART big storage.basket ART middle-3POS that fierce.one<br>The middle of that monster was like a big storage basket.                                                         | 38            |

| yəpáİx "əm cə. Xa k "ə xətəstəm sx "qənx "eləs,<br>yə-páİx "əm cə Xa k "ə xətə-st-əm sx "qənx "elə-s<br>along-be.steaming QUOT BE3P ART be.saying-CAUS-INTR stomach-3POS<br>It was steaming. That is what is called the stomach,                                                                                                                                                                                                                                                 | 39 |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----|
| šx <sup>w</sup> ni <sup>9</sup> s təwmək <sup>w</sup> stem smóds təwka skeləqəm.<br>šx <sup>w</sup> -ni <sup>9</sup> -s təw-mək <sup>w</sup> stem s-mód-s təwka skeləqəm<br>OBLNOM-AUX-3POS somewhat-all what NOM-swallow-3POS that fierce.one<br>where everything which that monster swallows goes.                                                                                                                                                                             | 40 |
| səw žátəs k¤θəwŽáləməł siyáləx¤a?, Ža šx¤ni?s wənảán<br>s-wə-žátə-s k¤θəwŽáləm-əł syəyáləx¤a? Ža šx¤-ni?-s wə-nán<br>NOM-EST-be.saying-3POS those-past old.persons BE3P OBLNOM-AUX-3POS<br>EST-very<br>Then the old people were saying, that is why this creek is very                                                                                                                                                                                                           | 41 |
| wəspàỷ pəỷ təna stá?tləw. ni šx "yələ?és k "snis nem? 5Åqa?θ<br>wə-spàỳ pəỷ təna stá?tləw ni? šx "-yə-lə?é-s k "s-ni?-s nem<br>?5Åq-a?θ<br>EST-crooked this creek AUX OBLNOM-along-follow-3POS ART NOM-AUX-3POS<br>go.out-mouth<br>crooked. It is the path it followed when it went out,                                                                                                                                                                                         | 42 |
| k <sup>w</sup> θəwÅa sÅéləqəməł.<br>k <sup>w</sup> θəwÅa sÅéləqəm-əł<br>that fierce-past<br>that monster.                                                                                                                                                                                                                                                                                                                                                                        | 43 |
| <ul> <li>i.º. c kwə mi.º. xwcáləwən, səw mi císəm tə sáxwəl ?ə tə ni?əł</li> <li>?i c kwə mi xwcáləwən s-wə-mi[-s] císəm tə sáxwəl ?ə tə ni?-əł</li> <li>AUX QUOT then come next.year NOM-EST-AUX-3POS grow ART plant OBL</li> <li>ART AUX-past</li> <li>When the next year came, then grass grew at what had been the place where</li> </ul>                                                                                                                                    | 44 |
| <ul> <li>šx "yəłə?és k"s nem yə?é.ý k "ØəwÅa sÅéləqəməł čtwa?</li> <li>šx "-yə-łə?é-s k" s-[ni?]-s nem yə-?é.ý k "ØəwÅa sÅéləqəm-əł čtwa?</li> <li>OBLNOM-along-follow-3POS ART NOM-AUX-3POS go along-be.continuing that fierce.one-past SPEC</li> <li>it had passed when it went along, that monster</li> </ul>                                                                                                                                                                 | 45 |
| $k^{v}$ ík <sup>w</sup> əxtəm $k^{v}$ ə s <sup>?</sup> í·łqəỷ. néc sáž <sup>w</sup> əl k <sup>w</sup> θé?. k <sup>w</sup> ík <sup>w</sup> əxətəs<br>k <sup>w</sup> ík <sup>w</sup> əx-t-əm [?ə] $k^{v}$ ə s <sup>?</sup> í·łqəỷ néc sáž <sup>w</sup> əl k <sup>w</sup> θé? k <sup>w</sup> ík <sup>w</sup> əx-ət-əs<br>be.naming-tr-INTR OBL ART seelthkey different grass that be.naming-tR-3TR<br>that was called a seelthkey. That was a different plant. The people called it | 46 |
| tə x <sup>w</sup> əlməx <sup>w</sup> əł ctwa <sup>9</sup> k <sup>w</sup> məθk <sup>w</sup> əỷ. mɨi <sup>w</sup> qəlét yəlánəm, 4<br>mi x <sup>w</sup> cáləyyən                                                                                                                                                                                                                                                                                                                  | 7 |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---|
| tə x"əlməx"-əł ctwa? k" mə́ $\theta$ k"əỷ mi qəlét yəlánəm mi x"cáləwən<br>ART people-past SPEC ART muthkwey come again year come next.year<br>muthkwey. When it was another year, next year,                                                                                                                                                                                                                                                                                   |   |
| niwəł nem tét <sup><math>\theta</math></sup> k <sup>w</sup> is nem císəm təwka sáxəl, mə $\theta$ k <sup>w</sup> əy. 4<br>ni <sup>?</sup> wəł-nem tét <sup><math>\theta</math></sup> k <sup>w</sup> s-ni <sup>?</sup> -s nem císəm təwka sáxəl mə $\theta$ k <sup>w</sup> əy<br>AUX already-go everywhere ART NOM-AUX-3POS go grow that grass muthkwey<br>then that plant muthkwey spread out when it grew.                                                                     | 8 |
| niw pépədəm stə?é ?ə kw xəw néc sáxwəl. ?ówəte? kwə xəve 4<br>ni? wə-pépədəm stə?é ?ə kw xe wə-néc sáxwəl ?ówəte? kwə xe<br>wə-stə?é                                                                                                                                                                                                                                                                                                                                            | 9 |
| AUX EST-be.blooming like OBL ART again EST-different grass none ART again EST-like                                                                                                                                                                                                                                                                                                                                                                                              |   |
| It bloomed like any other plant. But it was not like any other                                                                                                                                                                                                                                                                                                                                                                                                                  |   |
| sáž <sup>w</sup> əl. <sup>9</sup> əwé·ł cə k <sup>w</sup> s t <sup>θ</sup> íqats k <sup>w</sup> θəw tátəł məstáyəx <sup>w</sup> . xé <sup>9</sup> xe <sup>9</sup> stəm. 5 <sup>w</sup> sáž <sup>w</sup> əl <sup>9</sup> əwə-əł cə k <sup>w</sup> s-t <sup>θ</sup> íq-ət-s k <sup>w</sup> θə wə-tát-əł məstáyəx <sup>w</sup> xé <sup>9</sup> xe <sup>9</sup> -st-əm.                                                                                                             | 0 |
| grass not-past QUOT ART NOM-get.stepped.on-3POS ART EST-ancient-past<br>person holy-CAUS-INTR<br>plant. The old people did not step on it. It was thought sacred.                                                                                                                                                                                                                                                                                                               |   |
| swe's sqəlqəlílləts ctwa' ['ə] tə skéləqəmət ctwa', s'ítqəy. 5<br>s-we'-s s-qəlqəlíl-tət-s ctwa' ['ə] tə skéləqəmət ctwa' s'ítqəy.<br>NOM-own-3POS NOM-soil-self-3POS SPEC OBL ART fierce past SPEC seelthkey<br>It was from the droppings of the monster, the seelthkey.                                                                                                                                                                                                       | 1 |
| min swəckánəms ttde? sáx əl ni t k k k v xət k ə mədk ə y,5mi s-wəckánəm-s tsáx əl ni?-ət k k k v x-ət k ə mədk ə y5come NOM-fall-side[?]-3POS that plant AUX-we be.naming ART muthkwey1It is what fell, that plant that we call muthkwey,5                                                                                                                                                                                                                                     | 2 |
| niw x <sup>w</sup> əqáx siétət <sup>0</sup> k <sup>w</sup> s cicəsəms. Xa k <sup>w</sup> ə šx <sup>w</sup> is k <sup>w</sup> (xətəs 5<br>ni <sup>?</sup> wə-x <sup>w</sup> ə-qáx siétət <sup>0</sup> k <sup>w</sup> s-cicəsəm-s Xa k <sup>w</sup> ə šx <sup>w</sup> -ni <sup>?</sup> -s k <sup>w</sup> (x-ət-əs<br>AUX EST- become-spread.out ART NOM-be.growing-3POS BE3P then<br>OBLNOM-AUX-3POS name-TR-3TR<br>which came to be growing plentifully all over. So that is why | 3 |
| k "θəw tátəł x "ślməx" təňa táməx" i šx";5k "θə wə-tát-əł x "ślməx" təňa táməx" ?i šx"-?i-ct [?ə]ART EST-ancient-past people this land be.here OBLNOM-be.here-our OBLthe people of long ago named this place where we are                                                                                                                                                                                                                                                       | 4 |

k<sup>w</sup> a x<sup>w</sup>m áθk<sup>w</sup> aýam. wał yalel <sup>9</sup> a ctwa?. wałhíθ, i can mi skík qał 55
 k<sup>w</sup> a x<sup>w</sup>m áθk<sup>w</sup> aýam wał-yalel <sup>9</sup> a ctwa? wał-híθ<sup>9</sup> i can mi skík qał ART having-muthkwey-INTR already-long.time SPEC already-last.long AUX I come child

56

57

"Musqueam." A long time must have passed. When I was a child

tə <sup>9</sup>é·ňθə i nəwł k<sup>w</sup>ík<sup>w</sup>əxtəm k<sup>w</sup> x<sup>w</sup>mə́θk<sup>w</sup>əyəm təňa. tə <sup>9</sup>é·ňθə <sup>9</sup>əỷ ni<sup>9</sup> wəł-k<sup>w</sup>ík<sup>w</sup>əx-t-əm k<sup>w</sup> x<sup>w</sup>mə́θk<sup>w</sup>əyəm təňa. ART be.I and AUX already-be.naming-TR-INTR ART Musqueam this myself, they had already been calling this "Musqueam."

xáťa k<sup>w</sup>łan síle ł palíq<sup>w</sup>aye.
xáťa k<sup>w</sup>ła na-síla-ał palíq<sup>w</sup>aye.
say ART my-grandparent-past Peliqwiye
So said my grandmother Peliqwiye.

This text was dictated by James Point on 1 April 1963 and checked with Arnold Guerin on 27 August 1983.

### Line

1.  $q^{w}i \cdot l q^{w} \partial l$  'be telling,' progressive of  $q^{w} \partial l q^{w} \partial l$  'tell, relate.'

2. AG said he would have begun the sentence with  $ni \ c^{2} \ ^{2} \ k^{w} \partial \sigma$  rather than JP's  $ni \ c^{2} \ k^{w} \partial$ , but reconsidering this, I think JP's original is consistent with the syntax of the sentence as I understand it: 'There was, it is said, what was called Mumquaam.' An oblique particle  $2\sigma$  would be appropriate if there were a different subject, identified as *at* the place called Mumquaam.

 $x^w m \partial m q^w \dot{e} \cdot m$ , from  $m \dot{a} q^w \partial m$  'bog'; a place name, that of a bog near Chaldecott Park. The name may be based on the diminutive  $m \dot{a} \dot{m} q^w \partial m$  'little bog' with the combination of the prefix  $x^w$ - and the suffix  $-\partial m$ , the underlying form being  $//x^w - m a \dot{m} q^w \partial m - \partial m / /$ , with the medial *m* lost and a shift in stress. This combination of affixes with the meaning 'place where there are' is common in place names over the region.

3.  $x^w \dot{c} s \dot{t} \cdot m$  'grown up,' from  $\dot{c} \dot{t} s \partial m$  'grow,' perhaps with the  $x^w$ - and  $-\partial m$  seen in place names. The underlying form would be  $//x^w$ - $\dot{c} is \partial m - \partial m //$  'place where there are growing things.' I have inserted, in square brackets, an oblique particle that AG would have used.

4. *číčasam* 'be growing,' progressive of *čisam* 'grow.'

6.  $\check{x} \circ \check{t} \circ s t \circ m$  'said to be, so-called.' JP used this term fairly often as a modifier, as he did here.

 $\check{s}\check{x}^{w}\partial ni^{2}$  'place where.' AG indicated that this is pronounced either  $\check{s}x^{w}ni^{2}$  or  $\check{s}x^{w}\partial ni^{2}$ .

 $s \mathring{\lambda} \acute{e} l a q a m$ . This word can be translated in various ways. JP may have suggested 'monster' in this context, but in another narrative he referred to the warrior Capilano as  $s \mathring{\lambda} \acute{e} l a q a m$ , meaning 'fierce' or 'dangerous,' and CC used it in

546

the context  $s\dot{\lambda}\acute{e}laqam \check{s}x^wn\acute{e}^am$ , which I translate 'powerful shaman.' AG commented on how broadly the term is used and how hard to give it a single gloss. He suggested 'fearsome being' in this context and said it could be used for feared things such as a whale at sea or a ghost; a little bug is a  $s\dot{\lambda}i'\dot{\lambda}alaqam$  (the diminutive).

 $s^{2}i \cdot lq_{\partial}y'$  seelthkey.' This is the two-headed serpent known to the speakers of Northern Straits as the  $s^{2}in\partial lq_{\partial}y'$  and to the speakers of Kwakwala as the *sisiul*. The name in Halkomelem and Northern Straits appears to be composed of an unidentifiable element and  $2\partial lq_{\partial}y'$  snake.'

10.  $k^{wl}\partial$ , JP was not consistent in which gender he used with the seelthkey, but often, as here, it was feminine.  $w\partial \dot{m} \dot{e} n$  'foolish, not having good judgment,' probably composed of  $w\partial$ - 'established' and  $\dot{m} \dot{e} n$  'weak.'

12.  $\check{x}\partial l\check{x}\partial l\dot{c}\partial a \cdot m$  'you'll get twisted up,' probably a plural form, that is, 'you'll get twisted in several places'; cf.  $\check{x}\partial l\dot{c}t$  'turn it around, twist it,'  $\check{x}\partial l\dot{c}t$ 'be turning it.' A seelthkey was said to cause a kind of paralysis that contorts the body of its victim.

13.  $\delta x^{w} s k^{w} e' y s$  'cause of its being impossible, reason why it is impossible'

15.  $m\dot{a}^{2} \partial q^{w}$ , usually translated 'duck,' this term really refers to any larger bird; the eagle is a  $m\dot{a}^{2} \partial q^{w}$ .

16. tánəqsən 'mallard,' composed of an unidentifiable root and the suffix -əqsən 'nose, bill.'

17.  $\check{s}x^wc\check{x}\partial \acute{e}t^{\theta}s$  'means by which it is made evident.' The root is that of  $\check{x}\acute{e}t^{\theta}t$  'measure it' and  $s\check{x}\partial \acute{e}t^{\theta}$  'measured, marked, known, evident.' I heard JP say  $\check{s}x^w\check{x}\partial \acute{e}t^{\theta}s$  here and in another context, but AG (on tape) clearly said  $\check{s}x^wc\check{x}\partial \acute{e}t^{\theta}s$ . The verbalizing prefix *c*-seems reasonable.

 $-\partial t \ ctwa^2$ - 'past speculative.' JP used this fairly often when referring to people of the past. If any translation is needed, it would be 'who must have been.'

20. má·ĺmal. Progressive of malmal 'make a mistake, lose one's way.'

27.  $s\check{x}\partial \acute{e}t^{\partial}$ . See 17 above.  $s\check{x}\partial\check{x}i^{2}st\partial x^{w}$  'witness it,' causative of  $s\check{x}\partial\check{x}i^{2}$  'visible, seen'; cf.  $\check{x}i^{2}$  'appear, be revealed,'  $\check{x}i^{2}t$  'reveal, disclose.'

28. xixłém. Progressive of xłém 'watch.'

30. AG corrected what I had written *i*  $sx\delta nc \sigma$  as  $y \partial sc\delta nc \sigma$ , and identified the  $y \partial$ - as 'before,' saying that the  $y \partial$ - could be omitted and in fact the sentence would be better without it.

33. The 'big river' is of course the Fraser. The 'river-track' became the bed of the creek that flows past  $s\hat{c}\partial l\hat{e}x^w$ .

34.  $s\dot{a}\dot{x}^{w}\partial l$  is often translated 'grass,' but it has a wider meaning, since it includes flowering plants that die down in the fall, as seen in what follows.

37. *sxłém* 'sight, view,' nominalization of *xłém* 'watch.' AG pronounced the word *sxłém*.

38.  $\delta x^{w}k^{w}e^{2}\partial m$  'storage basket,' also given as  $sk^{w}e^{2}\partial m$ .

39. pálx » am. Progressive of pálax » am 'steam, come to a boil.'

39.  $\delta x^w q \partial n x^w \ell l \partial$  'stomach.' Composed of  $\delta x^w$ - 'oblique nominalizer,'  $q \partial n \partial x^w$  'throat, gullet,' and  $-\ell l \partial$  'container'; cf.  $sq \partial n \partial x^w$  'greedy,'  $clq \partial n x^w \ell l \partial m$  'die of overeating.'  $sm \partial q ds$  'that which it swallowed,' a relative clause, from  $m \partial q$  'swallow.'

41.  $sy \partial y a d \partial x^w a^\gamma$  [siyá d  $\partial x^w a^\gamma$ ] plural of  $sy a d \partial x^w a^\gamma$  old person.'

42. spáýpaý 'crooked, winding,' plural of spápaý 'bent,' resultative of pá·yt 'bend it.'

42.  $st a^2 t l \partial w$  'creek,' diminutive of  $st a l \partial w$  'river.'  $2\partial A q a^2 \theta$  'go out into open water' (as from Burrard Inlet into the gulf).

44.  $x^{w}\dot{c}\dot{e}l\partial w\partial n$  'turn of the year' was JP's gloss, presumably from  $\sqrt{\dot{c}al}$  'turn'; cf.  $\dot{c}\dot{a}l\partial s\partial m$  'turn the head (to look in another direction),' said to come in the fall.

45.  $k^w s$ . AG said he would have used  $k^w is$  instead. This is short for  $k^w sni^2 s$ . 47.  $m \delta \theta k^w \partial y$ . The identity of this plant, for which Musqueam was named, is not clear. (See Appendix 2, under the name "Musqueam.")

51.  $q \partial l q \partial l (l \theta \partial t \text{ 'soil oneself, defecate'; cf. } q \partial l q \partial l (l t \text{ 'spoil it,' from } q \partial l \text{ 'bad,' used today as a euphemism for } d \partial t \partial t \partial x$ .

52.  $w \partial c \dot{\lambda} a n \partial m$  'fall beside'(?), from  $w \partial c \partial \dot{\lambda}$  'fall' (also recorded  $w \partial c \partial \dot{\lambda}$ , cf. Cowichan  $w \partial \dot{\lambda} \partial c$ ),  $-a n \partial m$  '-side'? (cf.  $-\dot{e} \cdot n$  'side,'  $-\partial n \partial a \sim -\partial n e^2$  'ear').

55.  $x^w m \delta \theta k^w \delta y \delta m$  'Musqueam,' formed with  $x^w$ - and  $-\delta m$ , 'place of  $m \delta \theta k^w \delta y$ .'

57.  $k^w t \partial n \partial s i l e^t$  'my deceased grandmother.' JP used both the feminine remote article  $k^w t \partial$  and  $-\partial t$  'past.' Some older people speaking English nearly always use *poor* to convey the fact of deceased status. Others say *late*. I have generally ignored this in a free translation.

 $\dot{p}\partial l(q^w \partial y \partial$  (also recorded  $p\partial l(q^w \partial y \partial)$ ) was JP's mother's mother, and she was AG's great-great-aunt, the sister of his father's mother's mother.

# Appendix 1 Index of Grammatical Elements and Lexical Affixes

This index lists all prefixes and suffixes, both grammatical and lexical, all particles, and words (and a few phrases) of special classes the functions of which are described in this grammar. Each entry gives the number of the section where the item is identified. In a few cases, other numbers refer to sections describing special uses.

The entries are ordered in a phonological (rather than alphabetical) order. Glottalized resonants are not given separate status but are ordered with the plain resonants. Suffixes with initial vowels are distinguished thereby, but otherwise vowels are ignored in the ordering. The order is as follows:

 $p \ \dot{p} \ m \ t \ \dot{t} \ n \ \theta \ t^{\theta} \ \dot{t}^{\theta} \ s \ S \ \dot{s} \ \dot{t} \ \dot{\lambda} \ l \ c \ \dot{c} \ c \ x \ k^{w} \ \dot{k}^{w} \ x^{w} \ q \ \dot{q} \ \dot{x} \ q^{w} \ \dot{q}^{w} \ \dot{x}^{w} \ y \ w \ h^{\gamma} \ \vartheta \ e \ a \ i$ 

#### т

*mə*- 'come' (< *?əmi* 'come'), lexical prefix, §12.4.1.

 $\vec{m} \rightarrow \vec{m} e$  'certainty' (CERT), predicate particle, §16.2.11.

- *mi* 'come' (< *?əmi* 'come'), directional auxiliary (AUX), §3.2.1; prepositional verb, §3.5.
- -mət ~ -mat ~ -mit ~ -met ~ - $me^{2}t$  'concern' (CON), applicative suffix, §10.4.5.

-mət 'appearance'(?), lexical suffix, §13.6, 4.

-mat 'kind, piece, part,' lexical suffix, §13.6, 5.

-mən ~ -əmən ~ -əmən ~ -mín 'instrument,' lexical suffix, §13.6, 6.

-mən ~ -émən ~ -mín 'residue,' lexical suffix, \$13.6.8 (cf. -émən 'extracted liquid').

 $-m \partial x \sim -\dot{a}m \partial x \sim -\dot{a}m \partial x$  'country, person,' lexical suffix, §13.6.11 (cf.  $-m \partial x^w$ , etc. 'place, people').

*mók*<sup>w</sup> 'all, every,' §18.4.40, §18.4.8; 'whenever,' §18.4.18, §4.3.1.

 $<sup>-</sup>m \partial x^{w} \sim -\acute{e}m \partial x^{w} \sim -\acute{a}lm \partial x^{w} \sim -\acute{a}wm \partial x^{w}$  'place, people, cluster'(?), lexical suffix, §13.6.12 (cf.  $-\acute{a}lm \partial x^{w}$  'breast, etc.').

#### t

- $-t \sim -\partial t \sim -\acute{e}t \sim -\acute{a}t$  'transitive' (TR), §10.1.1.
- *-t* 'stative' (STAT), §12.1.6.
- tə 'a, the (non-feminine, present and visible),' article (ART), §15.1.
- $t \partial m \sim t \partial n \sim t \partial n'$  'time of, season for,' lexical prefix, §12.4.2.
- $\sqrt{t\acute{e}m}$  'what.' See *stém* 'what?' §17.3.
- təmtém 'when?' interrogative word, §17.5.
- témc 'what people?' interrogative word, §17.4.
- $t \partial n \sim t \partial n \sim t \partial m$  'time of, season for,' lexical prefix, §12.4.2.
- tən- ~ təl- 'from,' lexical prefix, §12.4.3.
- $-t\partial n \sim -t\partial n' \sim -te' n'$  'instrument,' lexical suffix, §13.6, 16.
- təní? 'there, that way,' locative demonstrative, non-feminine, visible, §15.2.2.1.
- tənínə 'that,' locative demonstrative, non-feminine, visible, §15.2.2.1.
- təňá 'this, he/him,' locative demonstrative, non-feminine, visible, §15.2.2.1.
- *təná:nə* 'that over there, over there,' locative demonstrative, non-feminine, visible, §15.2.2.1.
- təňá tran 'they/them,' locative demonstrative, plural, visible, §15.2.2.1.
- tənəwŽá 'he/him, that,' third-person demonstrative, §15.2.2.2.
- $t\theta \hat{e}^{2}$  'that, he/him, it, there,' simple demonstrative, non-feminine, visible, §15.2.1.
- təl- ~ tən- 'from,' lexical prefix, §12.4.3.
- təlí?~ təlí 'be from, from,' as prepositional verb, §3.5.
- $-t\partial l \sim t \dot{a} l$  'each other,' reciprocal suffix (RECIP), §10.6.
- $tx^{w}$ -, 'buy,' verbalizing prefix, §12.2.3.
- $tx^{w_{-2}}$  (something) belonging to,' lexical prefix, §12.4.4.
- $tx^{w}$ -3 'remaining, only,' lexical prefix, §12.4.5.
- $-tx^{w} \sim -\partial w tx^{w} \sim -\delta w tx^{w} \sim -\partial l t \partial x^{w}$  'house,' lexical suffix, §13.6, 18.
- $tx^{w} \dot{a}y$  'only remaining, except for,' §18.4.38, §4.1.5.2.
- $tx^w w \acute{e}t \sim t \partial w \acute{e}t$  'whose?' interrogative word, §17.2.
- *tú*·*x*<sup>*w*</sup> 'nine,' §19.
- tqéce? 'eight,' §19.
- $t \dot{a} \ddot{x}^{w} \sim t \dot{a} x^{w}$  'later, occur later, follow (an example),' §18.4.22.
- $t \dot{a} \ddot{x}^{w} \sim t \dot{a} x^{w} \sim t \dot{a} \ddot{x}^{w}$  'adjust, exactly, just,' §18.4.27.
- taw- 'like, somewhat, a bit, sort of (like),' aspectual prefix, §11.1.3.
- *təwét ~ wét* 'who?' §17.1.
- $t \partial w \acute{e}t \sim t x^w w \acute{e}t$  'whose?' §17.2.
- $t \partial w \dot{\lambda} \dot{a}$  'he/him, it, that,' non-feminine, visible third-person pronoun and demonstrative, §15.2.2.2.
- $t \partial w \dot{\lambda} \dot{a} l \partial m$  'they/them, those,' plural third-person pronoun and demonstrative, §15.2.2.2.
- *təwẩá ẩəm* 'that little one,' non-feminine diminutive third-person pronoun and demonstrative, §15.2.2.2.

- *təwǎəǎaǎám* 'those little ones,' plural diminutive third-person pronoun and demonstrative, §15.2.2.2.
- $t\partial^2 t \sim ti^2 t$  'this, here, this way,' locative demonstrative, non-feminine, visible, §15.2.2.1.
- tə<sup>9</sup>ínə 'this,' locative demonstrative, non-feminine, visible, §15.2.2.1.
- *tə*<sup>2</sup>*én* 'how wonderful! what a \_\_\_\_! how powerful!' exclamatory-interrogative adjective, §20.2.3.

## ť

 $-\dot{t}$  (?) possible lexical suffix, §13.7, 3.

*tát* (usually *wətát*) 'long before, from long ago, earlier,' §18.4.11.

*-téň* (?) possible lexical suffix, §13.7, 4.

*ť ž óm* 'six,' §19.

*tá*<sup>2</sup>θət 'try,' §3.2.3.

### n

-n- ~  $-n \partial x^{w}$  'limited control transitive suffix' (LCTR), §10.1.2.

*nə*- 'my,' first-person singular possessive prefix, §14.2.4.

 $ni^{\circ}$  'be there, some,' preceding a noun in a pseudo-cleft sentence, §4.1.5.1.1.

 $ni^{2} \sim ni \sim n\partial$  'there,' locative auxiliary verb, §3.2.1.

 $ni^{9} \sim ni$  'in, at, on, onto by, with, for,' prepositional verb, §3.5.

 $ni^{\circ} \sim ni$  'hey there, excuse me,' interjection, §20.3.5.

ná 'what? yes?' in reply to call, §20.3.3.

 $-\dot{n}a$  'be here'(?), 'exist'(?), follows the articles in a series of locative demonstratives, §15.2.2.1

 $-\dot{n}a$  ~ -na 'be yonder,' in locative demonstratives, §15.2.2.1

ném 'go.'

*ném* ~ *nem* 'go,' directional auxiliary (AUX), §3.2.1; 'to (away from speaker),' prepositional verb, §3.5.

 $-n\acute{am} \partial t \sim -l\acute{am} \partial t$  'oneself (limited control),' reflexive suffix corresponding to  $-n\partial x^{w}$  'limited control transitive,'' §10.5.2.

*ňán* (usually *wəňán*) 'too, very, very much,' §18.4.24.

-nəs 'goal' (GOAL), applicative suffix, §10.4.2.

 $-nis \sim -\partial n\partial s \sim -\partial l\partial s$  'tooth,' lexical suffix, §13.6, 23.

 $-n\partial c \sim -n\dot{e}c \sim -\partial l\partial c$  'butt, tail, base, bay, price,' lexical suffix, §13.6, 24.  $n\partial c\dot{i}m$  'why?' interrogative word, §17.19.

- *néc* 'different, strange, at other times, sometimes, there are different times (when),' §18.4.4.
- $n\partial \dot{c} ex^w$  'once, one time, once in a while,' §18.4.5.

nécow- 'one' (combining form), §19.

*náča*<sup>2</sup> 'one,' §19, §18.4.10.

 $<sup>-</sup>n\partial x^{w} \sim -n\acute{e}x^{w} \sim -n \sim -l\partial x^{w} \sim -l$  'limited control transitive' (LCTR), §10.1.2.  $n\partial x^{w}$ - (identical with  $x^{w}$ - 'inward'?) lexical prefix, §12.4.8.

 $n \partial x^{w}s$ - ~  $x^{w}s$ - 'habitually' (HAB), aspectual prefix, §11.1.8.

náy (usually wənáy) 'only,' §18.4.37.

 $n \dot{a} \dot{w} \sim n \dot{a} \cdot \dot{w} \sim n \dot{a} w$  'yes (in answer to call), hey,' §20.3.4.

nówo 'you,' second-person singular personal word, §14.2.7.

ná<sup>9</sup> *ətə* 'there,' demonstrative auxiliary verb, non-feminine, §15.3.

 $n\acute{a}^{2}\partial\theta\partial$  'there,' demonstrative auxiliary verb, feminine, §15.3.

### θ

- transitive' and -S 'me, you (singular),' §14.2.5.

 $\theta$  'a, the,' article, feminine, present and visible (ART), §15.1.

 $\theta \acute{e}m$ - ~  $\theta am$ - 'two' (combining form), §19.

θəmé 'twice,' §18.4.5, §19.2, 2.

-θam- realization of -t 'transitive' and -Sam- 'you,' second-person passive subject, §14.2.6.

-*θamx* realization of -t 'transitive' and -Samx 'me,' §14.2.5.

-θamə realization of -t 'transitive' and -Samə 'you,' second-person singular object, §14.2.5.

*θit* 'intend, try,' §3.2.3.

 $\theta \delta t$  'say.'

 $-\theta \partial t \sim -\theta \dot{a}t$  'oneself,' reflexive suffix corresponding to -t 'transitive,' §10.5.1.

 $-\theta \partial n \sim -a^2 \theta$  'mouth,' lexical suffix, §13.6, 26.

 $-\theta in \sim -a^2\theta$ , etc. 'mouth, lip, margin, edge,' lexical suffix, §13.6, 26.

*θəňá* 'this, she/her,' locative demonstrative, feminine, visible, §15.2.2.1.

 $\theta \partial n \partial w \dot{X} a$  'she/her, that,' third-person demonstrative, §15.2.2.2.

 $\theta \partial t$  'however, nevertheless, but, still, etc.,' adversative predicate particle (ADV), §16.2.15.

-*θél*-realization of -*t* 'transitive' and -*Sél*- 'I,' first-person singular passive subject, §14.2.6.

 $\theta \partial w \dot{X} \dot{a}$  'she/her, that,' third-person demonstrative, §15.2.2.2.

 $\theta \partial w \dot{\lambda} \dot{a} \dot{\lambda} \partial \dot{m}$  'that little one,' third-person feminine diminutive demonstrative, §15.2.2.2.

 $\theta e^{\gamma}$  'that, she/her,' simple demonstrative, feminine, visible, §15.2.1.

 $\theta \partial^2 i$  'this,' locative demonstrative, feminine, visible, §15.2.2.1.

 $\theta \partial^2 it$  (usually  $w \partial \theta \partial^2 it$ ) 'truly, really,' §18.4.26.

### t<sup>θ</sup>∂

 $t^{\theta}\partial$  'a, the,' article, Cowichan equivalent of  $t\partial$ , occasionally used by some Musqueam speakers, §15.1.

## ťθ

 $-\dot{t}^{\theta}(?)$  possible lexical suffix, §13.7, 7.  $\dot{t}^{\theta}(x^{w}\partial m \, \text{`pity, be so kind as to ..., please ...,' §3.2.3.}$  $\dot{t}^{\theta}x^{w}(m\partial t \, \text{`pity him, be so kind as to ... for him,' §3.2.3.}$  $\dot{t}^{\theta}\dot{a}^{2}k^{w}s \, \text{`seven,' §19.}$ 

S

- *s*- $_1$  'nominalizer' (NOM), derivational prefix, §12.1.1; in nominalized clauses, §4.3; in relative clauses with extracted oblique objects, §4.1.1.2.
- s-2 'resultative' (RES), derivational prefix, §12.1.2. Cf. resultative aspect, §7.7.
- -s<sub>1</sub> 'his, her, its, their,' third-person possessive suffix, §14.2.4.
- -s<sub>2</sub> 'permissive,' §10.7; following ?ówô 'not,' §6.1.1.
- s-... -a?t forming adjective-like words, §3.8.3.4.
- -sím 'berry'(?), possible lexical suffix, §13.7, 8.
- $-st \rightarrow -st \partial x^{w_1}$  'causative' (CAUS), §10.3.
- stém 'what?' interrogative word, §17.3.
- statés 'near,' as prepositional verb, §3.5.
- $-st \partial x^{w_{1}} \sim -st \partial x^{w} \sim -\partial st \partial x^{w} \sim -st$  'causative' (CAUS), §10.3.
- -*st* $\partial x_{2}^{w}$  'comitative'(COM), applicative, §10.4.1.
- -stənámət ~ -stənəmət ~ -stənmət 'oneself,' causative reflexive suffix corresponding to  $-st a x^{w_1}$  causative, §10.5.3.
- $st\partial^2 \acute{e} \sim ste^2 \acute{e}$  'resembling, like,' as prepositional verb, §3.5.
- sní?anł 'enough, just enough, permitting,' §18.4.32.
- $s \rightarrow s \rightarrow w \sim s i s \rightarrow w$  'then he/she/it/they,' composed of  $s_{-1}$  'nominalizer,' the auxiliary  $ni^{2}$  or 2i, the third-person possessive  $-s_{1}$ , and the aspectual prefix  $w \rightarrow 2^{-2}$  'established,' §4.3.4.
- $s \vec{X} i^{2}$  'want, like,' §18.4.47.
- sŽáŽəm 'enough, ought, can,' §18.4.42.
- *sc* compound of *s* 'nominalizer' and the verbalizing prefix *c* 'get, make, etc.,' §12.2.1.
- scék<sup>w</sup>əl<sup>i</sup> 'how? how much? who?' interrogative word, §17.12.
- *scək*<sup>w</sup>*əlím* 'how? how constructed? how intended?' interrogative word, §17.13. *scəwét* 'smart, capable, able,' §18.4.44.
- scécoń 'really, certainly,' §18.4.25.
- skwéy 'impossible, unable,' §18.4.43.
- $sx^{w}\partial^{2}i\cdot nt$  'what meaning?' §17.15.
- $sqi^{2}q\partial l \sim sqiq\partial l$  'unable, ignorant (of how to do something),' §18.4.45.
- sqeqá? 'accompanying, with,' as prepositional verb, §3.5.
- *səw* 'then,' composed of *s* 'nominalizer' and the aspectual prefix  $wa_{2}$  'established,' §4.3.4.
- swé? 'one's own,' basis of possessive words other than 'ours,' §14.2.8.
- $s^2 \dot{a} \cdot t \sim s^2 \dot{a} \cdot 2t$  'ours,' first-person plural possessive word, §14.2.8.

### S

- -S 'me, you (sing.),' first- and second-person singular object suffix following -t 'transitive,' the sequence realized as  $-\theta$ , §14.2.5.
- -Samx ~ -S 'me,' first-person singular object suffix following -t 'transitive,' \$14.2.5.

- -Sam- 'you (sing.),' second-person passive subject suffix following -t 'transitive,' §14.2.6.
- -Sam $a \sim -S$  'you,' second-person singular object suffix following -t 'transitive,' \$14.2.5.
- -*Sél* 'I,' first-person singular passive subject suffix following *-t* 'transitive,' §14.2.6.

#### š

 $\delta x^{w}$ - 'oblique nominalizer' (OBLNOM) ('place where, time when, means of, reason why'), derivational prefix composed of *s*- 'nominalizer' and  $x^{w}$ - 'oblique,' §12.1.4; in relative clauses with extracted oblique nominal adjuncts, §4.1.1.2; in nominalized clauses, §4.3.3.

### ł

- *t*-, 'partake, experience,' verbalizing prefix, §12.2.4.
- $t_{-2}$  'portion,' lexical prefix, §12.4.6.
- $l \rightarrow l e \sim l$  'imperative-optative' (PER), predicate particle, §16.2.7.
- $l \rightarrow k^w l \rightarrow a$ , the, article (ART), feminine, near or accessible but invisible, §15.1.

 $t \sim k^w t \sim {}^2 \partial t$  same as last, introducing relative clauses following negatives, §6.3, (s) to (x).

*łníməł* 'we, us,' first-person plural personal word, §14.2.7.

 $-ln\acute{e}t \sim -ln\acute{e}^{\circ}$  'day of the week, day,' lexical suffix, §13.6.34.

 $-lsxé \sim -lcyé$  'tens,' lexical suffix §13.6, 35.

 $-t\partial t \sim -\partial t\partial t \sim -t \ell t$  'throat, breath, wind, food,' lexical suffix, §13.6, 31.

-tél- 'fire,' lexical suffix, §13.6, 36.

-*tc*- 'benefactive' (BEN), applicative suffix, §10.4.4.

 $-tcyé \sim -tsxé$  'tens,' lexical suffix, §13.6, 35.

 $ti x^w$  'three,' numeral, §19.

 $tx^{w} \acute{e}t$  'three times,' §18.4.5.

łqécəs 'five,' §19.

 $t \dot{a} \dot{q}$  'usually, generally, habitually,' §18.4.2.

*łəwźá* 'she/her, that,' third-person demonstrative, §15.2.2.2.

łwólap 'you (pl.),' second-person plural personal word, §14.2.7.

*li?á·aq<sup>w</sup>t* 'after, next, following, behind, last, go last,' §18.4.21.

### ź

 $\vec{\lambda}$  oblique article, §15.1.4.

 $\vec{\lambda}$ - 'go to,' verbalizing prefix, §12.2.5.

 $-\hat{X}$  (?) possible lexical suffix, §13.7, 11.

 $\dot{X}\dot{a}$  'he, she, it (is),' third-person singular personal word, §14.2.7; in the formation of demonstratives, §15.2.2.2; in nominalized narrative sentences (i.e., with *sow*, *sosow*, etc.), §4.3.4; followed by *s*-*wot*-with the sense 'be about to, be now starting,' §4.3.1; in cleft sentences, §4.1.5.2.

 $\dot{\chi}$ áləm 'they (are),' third-person plural personal word, §14.2.7.

 $\tilde{X}e \sim \tilde{X}e^{2}$  'again, also, too,' §18.4.15.

 $\dot{\chi}$ *ic* 'how long!' exclamatory-interrogative adjective, §20.2.6.

 $\dot{\lambda} \partial x^{w} l \dot{a}^{2} as$  ( $^{2} a \dot{l}$ ) 'worthless, of no account, no matter that, even if,' §18.4.33.

 $\dot{\chi}\dot{q}\dot{e}\dot{y}$  'how far! however far, for an unknown distance,' exclamatory-interrogative adjective, §20.2.1.

#### l

 $-l \sim -l \partial x^{w} (\rightarrow -n \partial x^{w})$  'limited control transitive' (TR), §10.1.2.

-lámət ( $\rightarrow$  -námət) 'oneself (limited control),' §10.5.2.

 $-lén \partial x^w \sim -\partial lén \partial x^w$  'season,' lexical suffix, §13.6, 25.1.

 $-l \partial c (\rightarrow -n \partial c)$  'base, etc.,' lexical suffix, §13.6, 24.

 $-lic \sim -e^{2}c$ , etc. 'route across,' lexical suffix, §13.6, 64.1.

 $-l \partial x^{w} (\rightarrow -n \partial x^{w})$  'limited control transitive' (LCTR), §10.1.2.

*léq* 'how fast! however fast, at an unknown speed,' exclamatory-interrogative adjective, §20.2.2.

 $-li^2c \sim -\partial le^2c \sim -\partial le^2c$  'container,' lexical suffix, §13.6, 50.

#### С

c- 'get, have, make, do,' verbalizing prefix, §12.2.1.

-c 'origin'(?), possible lexical suffix, §13.7, 16.

-c 'low tide,' lexical suffix, §13.6, 64.2.

 $-c\partial p \sim -\partial lc\partial p \sim -\partial lc\partial p$  'fire, firewood,' lexical suffix, §13.6, 67.

*ce*·*p* 'you (pl.),' second-person plural subject particle, §14.2.1.

ct 'we,' first-person plural subject particle, §14.2.1.

-ct 'our,' first-person plural possessive suffix, §14.2.4.

ctétam 'be doing what?' interrogative word, §17.6.

ctámət 'do what with oneself? do what? have what happen to one?' interrogative word, §17.8.

con 'I,' first-person singular subject particle, §14.2.1.

 $-c \rightarrow s \sim -c s \sim -c i s$  'hand, branch (of tree)' lexical suffix, §13.6, 68.

 $c \delta st \partial x^w$  'do what with it? do what about it?' interrogative word, §17.7.

ct- 'die of,' verbalizing prefix, §12.2.6.

- cəlél 'nearly, about to,' §18.4.28.
- *cəlél ?al* 'barely,' §18.4.29.
- *cəlím* 'even,' §18.4.39.

-cala? ~ ->čále? (?) possible lexical suffix, §13.7.17.

*ce*<sup>9</sup> 'future,' predicate particle, §16.2.6.

 $\check{c}x^{w}$  (//c-x<sup>w</sup>//) 'you,' second-person singular subject particle, §14.2.1.

#### ċ

 $-\dot{c} \sim -\partial \dot{c}$  (?) possible lexical suffix, §13.7, 18.

 $\dot{c} \partial \sim \dot{c}$ - 'quotative' (QUOT), predicate particle, §16.2.4.

*ċtwa*<sup>2</sup> 'speculative' (SPEC), predicate particle, §16.2.12.

*ċt*- 'fellow, co-,' lexical prefix, §12.4.7.

 $\dot{c}i\dot{\lambda}\partial m\partial\dot{t}^{\theta}$  'how tall!' exclamatory-interrogative word, §20.2.7.

 $\dot{c}\dot{k}^{w}\delta x$  'twenty,' numeral, §19.

 $\dot{c} \partial x^{w} l \dot{e}^{\gamma}$  'sometimes, there are times when,' §18.4.3.

### x

-*x* 'transitive' (TR), §10.1.3.

 $-x \partial n \sim -x \partial - \sim -x \ell n \sim -x i n$  'foot, leg, brim (of hat),' lexical suffix, §13.6, 69.  $-x \partial n$  'drop, droplet,' lexical suffix, §13.6, 69.2.

### k<sup>w</sup>

 $k^{w}$  o 'then,' predicate particle, §16.2.13.

- $k^{w}\partial \sim k^{w}$  'a, the,' article (ART), non-feminine, near and accessible but invisible (cf.  $k^{w}\partial\partial$ ), usual with nominalizations rather than tangibles, §15.1.
- $k^{W}\theta \partial$  'a, the,' article (ART), non-feminine, near and accessible but invisible, §15.1.
- $k^{w}\theta \hat{e}l\partial \hat{y}$  'those, they/them,' simple plural demonstrative, near, invisible, §15.2.1.
- $k^{w}\theta \partial w \dot{\lambda} \dot{a}$  'he/him, it, that,' third-person demonstrative, non-feminine, near, invisible, §15.2.2.2.
- $k^{w}\theta \partial w \dot{X} \dot{a} l \partial m$  'they/them, those,' third-person plural demonstrative, near, invisible, §15.2.2.2.
- $k^{w}\theta \partial w \dot{\lambda} \dot{a} l \partial m \partial t$  'they/them (deceased),' third-person demonstrative, §15.2.2.2.
- $k^{w}\theta \hat{e}^{\gamma}$  'that, he/him, it, there,' simple demonstrative, non-feminine, near and accessible but invisible, §15.2.1.
- $k^{w}s\partial$  'a, the,' article (ART), feminine, remote, §15.1.
- k<sup>w</sup>səňá 'she/her,' locative demonstrative, feminine, invisible, §15.2.2.1.
- $k^{w}s \partial w \dot{\lambda} \dot{a}$  'she/her,' third-person demonstrative, feminine, remote, §15.2.2.2.
- $k^{w}s \partial w \dot{\vec{x}} \dot{a} \dot{t}$  'she/her (deceased),' third-person demonstrative, §15.2.2.2.

 $k^{w}l$  'by now, by then'(?), predicate particle, §16.2.16.

- $k^{w}l \partial \sim l \partial$  'a, the,' article (ART), feminine, near and accessible but invisible, §15.1. Cf.  $l \partial$ .
- $k^{w}l\dot{e}^{2}$  'that, she/her,' simple demonstrative, feminine, near or accessible but invisible, §15.2.1.

### Ќ<sup>w</sup>

 $\vec{k}^{w} \partial \sim \vec{k}^{w}$  'a, the,' article (ART), non-feminine, remote, §15.1.

 $-\dot{k}^{w}(?)$  possible lexical suffix, §13.7, 20.

 $k^{wi}$  'that, there,' simple demonstrative, non-feminine, remote, §15.2.1.

 $k^{win}$  'how many?' interrogative word, §17.20.

 $k^{w} \partial n \dot{a}$  'he/him,' locative demonstrative, non-feminine, invisible, §15.2.2.1.

 $k^{w} \partial n \dot{a} l t \partial n$  'they/them,' locative demonstrative, plural, invisible, §15.2.2.1.

- *k*<sup>w</sup>*oná*·*no* 'way over there,' locative demonstrative, non-feminine, invisible, §15.2.2.1.
- $\vec{k}^{w} \partial w \vec{\lambda} \vec{a}$  'he/him, that,' third-person demonstrative, non-feminine, remote, §15.2.2.2.
- $k^{w} \partial w \tilde{\lambda} \dot{a} \cdot t$  'he/him (deceased),' third-person demonstrative, §15.2.2.2.
- $-k^{w}e^{\gamma} \sim -k^{w}a^{\gamma}$  'dead person'(?), lexical suffix, §13.6, 70.
- $k^{w}\partial^{2}i$  'around that way,' locative demonstrative, non-feminine, invisible, §15.2.2.1.

### $x^w$

- $x^{w_{-1}}$  'oblique relater' (OBREL), derivational prefix, §12.1.3.
- $x^{w}$ -2 'move toward,' verbalizing prefix, §12.2.7.
- $x^{w}$ -3 'inward, inhering, possessing, vulva,' lexical prefix, §12.4.8.
- $x^{w} \partial \sim x^{w} i$  'become,' aspectual prefix, §12.2.2.
- $x^{w}$  and 'to (toward speaker),' prepositional verb, §3.5; 'than,' §3.7.1.
- $x^{w} \partial n$  'still, soon,' aspectual prefix, §11.1.4.
- x<sup>w</sup>ném' 'to (away from speaker), for,' prepositional verb, §3.5; 'than,' §3.7.1.
- $x^{w} \partial nt \acute{e} c \partial l$  'early'; cf.  $x^{w} \partial n$ -, §11.1.4.
- $x^{w} \partial n \dot{x} \partial t \dot{e}^{2} \partial t \theta e^{2}$  'meanwhile, in the course of that, after some time, finally,' §18.4.23.
- $x^{w} \partial n \dot{a}^{\gamma}$  'first, when first, as soon as,' §18.4.19.
- x<sup>w</sup>əné<sup>9</sup>ənt 'become evening,' §18.4.14. k<sup>w</sup> x<sup>w</sup>əné<sup>9</sup>ənt 'last evening.'
- $x^{w}s$  ~  $n \partial x^{w}s$  'habitually' (HAB), aspectual prefix, §11.1.7.
- $-x^{w}\partial t$  'canoe,' lexical suffix, §13.6, 73.
- *x<sup>w</sup>éləq* 'nearly,' §18.4.30.
- $x^{w}c\acute{e}l$  'go where?' interrogative word, §17.9.
- $x^{w}c\delta l\theta \partial t$  'go where?' interrogative word, §17.10.
- $x^{w} c \ell l st \partial x^{w}$  'put where?' interrogative word, §17.11.
- $x^w \partial w \dot{e}$  'not yet,' §6.2.
- $x^{w} \partial w \acute{e}m$  'how big!' exclamatory-interrogative adjective, §20.2.5.
- $x^{w}\partial^{2}i \cdot nt$  'say what?' interrogative word, §17.14.
- $x^{w}\partial^{2}i \cdot ntst \partial x^{w}$  'say what to him? say what to him about something?' interrogative word, §17.16.

### q

 $-q \sim -\partial lq$  (?) possible lexical suffix, §13.7, 25.

 $-q \partial n \sim -q i n$  'head, end, bow of canoe,' lexical suffix, §13.6, 75.

-qən ~ -qen ~ -íqən 'front, slope,' lexical suffix, §13.6, 75.1.

-qən ~ -qə- ~ -qín 'throat, voice, speech, meal, opening facing upward,' lexical suffix, §13.7, 75.2.

 $-qs \sim -\partial qs \sim -\partial qs \partial n \sim -\partial qs \partial n \sim -qsin$  'nose, snout, point (of land),' lexical suffix, §13.6, 76.

qəlét 'do again, again, (no) more, (no) longer,' §18.4.16.

 $q \check{x} \acute{e} t$  'often, many times,' §18.4.6.

qé<sup>2</sup>is (//qéyəs//?) 'recently, just a short time ago, formerly,' §18.4.12.

## ģ

- $\vec{q}$  (?) possible lexical suffix, §13.7, 25.  $\vec{q}$  emphatic' (EMPH), predicate particle, §16.2.14.

### ň

- $\dot{x}$  (?) possible lexical suffix, §13.7, 26. - $\dot{x}\dot{e}n \sim -\dot{e}\dot{x}\partial n$  'arm, side, branch, perimeter,' lexical suffix §13.6, 77  $\ddot{x}\partial^{2}\dot{a}\theta\partial n \sim \ddot{x}a^{2}\dot{a}\theta\partial n$  'four,' numeral, §19.

### $q^w$

 $-q^{w} \sim -^{2}q^{w} \sim -\partial q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{w} \sim -\partial^{2}q^{$ 

### ģ<sup>w</sup>

 $-\dot{q}^{w}$  'fully, vigorously'(?), possible lexical suffix, §13.7, 27.

### *х*<sup>w</sup>

 $\check{x}^{w} \acute{a}m$  'move fast, be quick, immediately, be able to, §18.4.41.  $\check{x}^{w} t \acute{e}^{2}$  'head toward, toward, 'as prepositional verb, §3.5; 'than,' §3.7.1.

### y

 $y_{\partial_{-1}}$  'along, while moving, simultaneously,' aspectual prefix, §11.1.5.

 $ya_{2}$  'first, before doing anything else, above all,' aspectual prefix, \$11.1.6.

 $y_{\partial_3}$  'plural' in demonstratives, §15.2.1, §15.2.2.

yé 'taunting echo,' sentence-final tag, §16.3.2.

 $y\dot{e} \sim y\dot{e}$  'well, all right,' interjection, §20.3.7.

yəňá 'these, they/them,' locative demonstrative, plural, visible, §15.2.2.1.

yəňáłtən 'they/them,' locative demonstrative, plural, visible, §15.2.2.1.

 $y \partial \theta \hat{e}^{\gamma}$  'those, they/them,' simple demonstrative, plural, present, visible, §15.2.1.

 $y\dot{a}\theta$  (often  $w\partial y\dot{a}\theta$ ) 'always, frequently,' §18.4.1.

 $y \partial s \acute{e} l \partial \sim \partial i s \acute{e} l \partial$  'two,' numeral, §19.

*yét* 'only now, only then,' §18.4.13, §4.3.1.

yeł 'expectable' (EXP), predicate particle, §16.2.10.

 $y \ge x^w$  'inferential' (INF), predicate particle, §16.2.3.

y*əwéň* 'before,' §18.4.20.

#### W

 $w \partial_{-1}$  'if, when, that,' subordinating particle, §4.2.

*wa*-<sub>2</sub> 'established, continuing, now' (EST), aspectual prefix, §11.1.1.

 $w \partial_{2} \dots \partial_{a} l$  forming adjectival phrases, §3.8.3.3.

 $w \partial_{3}$  'suddenly, with a burst,' lexical prefix, §12.4.9.

wét ~ təwét 'who?' interrogative word, §17.1.

 $w \partial t \dot{a} t \rightarrow t \dot{a} t$ .

 $w \partial n \dot{a} n \rightarrow n \dot{a} n.$ 

wənáy 'only, be only,' §18.4.37, §4.1.5.2, §4.3.1.

-wi· $nx^w \sim -i$ · $nx^w$  'year,' lexical suffix, §13.6, 25.2.

 $w \partial \theta \partial^2 it \rightarrow \theta \partial^2 it.$ 

*wəł-* 'already,' aspectual prefix, §11.1.2.

 $-w\partial t \sim -\partial w\partial t \sim -\dot{e}w\partial t \sim -\dot{a}w\partial t \sim -wit \sim -w\partial t \sim -wil -$  'canoe, vessel, mind'(?), lexical suffix, §13.6, 90.

 $-wit \sim -\dot{alw}\partial t \sim -\dot{alw}\partial t \sim -\partial w\partial t \sim -\partial w\partial t - \omega \sim -\partial w\partial t - \omega \sim -\partial w\partial t \sim -\partial w\partial t \sim -\partial w\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t \sim -\partial t$ 

 $wala \sim w\dot{a}\dot{l}a$  'probably, presumably, I guess,' §18.4.35.

 $w \partial q \dot{e}^{\gamma} is \ ^{\gamma} a \dot{l}$  'for a short time,' §18.4.12.

wəqeeqal ?al 'barely,' §18.4.31.

wéyəl 'become day, tomorrow' §18.4.14. təňa wéyəl 'today.'

wéwə 'how wonderful!' exclamatory-interrogative adjective, §20.2.4.

wa<sup>9</sup> 'presumptive (it seems that ...)' (PRES), predicate particle, §16.2.17.

 $-we^{2}t \sim -\partial we^{2}t$  'back,' lexical suffix, §13.6, 92.

### h

həm- ~ hən- ~ pəm- 'come,' lexical prefix, §12.4.10.

 $hi\theta$  'last long, be a long time,' §18.4.9.

héỷ 'go ahead'; héỷ le 'Let's go. Let's ...,' §3.2.4.

háy 'stop, finish'; wəłháy ~ wəłáy 'have already ...,' §3.2.3.

háy 'specifically, uniquely, as for, more,' §18.4.36; in cleft sentences, §4.1.5.2;

háy <sup>2</sup>al ... 'most, -est,' with superlative sense, §3.7.1.

ha? 'if, when,' §18.4.17.

hé?e 'yes,' §20.3.1.

### 9

<sup>9</sup> oblique particle (OBL), §2.7, §3.4, §3.8.1.

<sup>9</sup> 'interrogative' (ROG), predicate particle, §16.2.2.

 $^{?}\partial$ - ( $\rightarrow$   $^{?}\partial n$ -) 'your,' second-person possessive prefix, §14.2.4.

 $^{2}a \sim ^{2}e$  question tag ('eh? isn't it? etc.'), sentence-final tag, §16.3.1.

<sup>9</sup>e· plural imperative (PLPER), predicate particle, §16.2.5.

?í 'be here,' lexical verb.

*i* 'here,' locative auxiliary verb (AUX), §3.2.1; 'in, at,' prepositional verb, §3.5.

?í 'here,' interjection, §20.3.6.

- <sup>2</sup>*i té* 'let's go, let's ...' (cf. *héỷ té*), §3.2.4.
- *<sup>9</sup>á pən* 'ten,' numeral, §19.
- $^{9}\partial m$  ( $\rightarrow h\partial m$ -) 'come,' lexical prefix, §12.4.10.
- *<sup>9</sup>∂mí* 'come,' lexical verb.
- *°omi ~ mi* 'come,' directional auxiliary verb (AUX), §3.2.1; 'to (toward speaker),' prepositional verb, §3.5.
- $^{2}\partial n \sim ^{2}\partial T \sim ^{2}\partial ^{\prime}$ your,' second-person possessive prefix (singular if there is no suffix, plurality being indicated by  $-\partial l \partial p$ ), §14.2.4.
- ${}^{2}\delta n\theta \partial \sim {}^{2}\epsilon' n\theta \partial \sim {}^{2}\epsilon' n\theta \partial {}^{4}$ , first-person singular personal word, §14.2.7; in pseudo-cleft sentences, §4.1.5.1.2.
- $^{?}\delta n \partial c \partial \sim ^{?}\delta n c \partial \circ$  where?' interrogative word, §17.17.
- <sup>2</sup>*inacastax*<sup>w</sup> 'put it where?' interrogative word, §17.18.
- <sup>2</sup>∂θ- realization of <sup>2</sup>∂T- 'your' and s- 'nominalizer,' §1.5.7, §14.2.4.
- *<sup>?</sup>isélə* ~ *yəsélə* 'two,' numeral, §19.
- <sup>9</sup>*ət* 'whenever, whatever, that,' subordinating particle, §4.2.
- ${}^{?}\partial t$  variant of  $t\partial \sim k^{w}t\partial$ ?, which see.
- <sup>9</sup>é·*ltən* third-person plural particle, §14.2.9.
- *<sup>2</sup>al* 'just,' predicate particle, §16.2.8.
- $-\partial q^{w}$ , etc. 'head,' lexical suffix, §13.6, 77
- *<sup>2</sup>óý* 'good, better,' §18.4.46.
- $^{2}\partial y \sim i$  'and,' conjunction linking nominal adjuncts, §3.8.4; 'and, but, and at that time,' conjunction linking clauses in compound sentences, §5.

 $?\acute{e}\cdot\acute{y}$  'continue,' §3.2.3.

- <sup>9</sup>ówə 'not,' §6.1; 'no,' §20.3.2.
- *<sup>?</sup>ówə scék<sup>w</sup>ələs* 'never,' §18.4.7.
- <sup>9</sup>ówəte<sup>9</sup> 'absent, non-existent, none,' §6.3.
- <sup>9</sup>éwəł 'reassuring' (REAS), predicate particle, §16.2.9.
- $^{2}iwawa \sim ^{2}iwawa^{2} \sim ^{2}iwawa \sim ^{2}iwawa^{2}$  'maybe, perhaps,' §18.4.34.
- ${}^{2}a'^{a}$  'emphatic interrogative' (ROG!), predicate particle, §16.2.18.

<sup>2</sup>*i*<sup>2</sup>*ətə* 'here,' demonstrative auxiliary verb, non-feminine, §15.3.

 $2i^{2}\partial\theta$  'here,' demonstrative auxiliary verb, feminine, §15.3.

### ð

 $-\partial p \sim -\dot{e} \cdot p$  'you (pl.),' second-person plural subordinate subject suffix, §14.2.3.

- $-\partial p \sim -\dot{e} p$  'base, bottom'(?), possible lexical suffix, §13.6, 1.
- $-\partial p \sim -\acute{e}p$  'hair'(?), lexical suffix, §13.6, 2.
- -*ápsəm* ~ -*əpsém* 'neck, nape,' lexical suffix, §13.6, 3.
- $-\partial m \sim -\delta m \sim -\delta m$  'intransitive' (INTR), §10.2.1.
- $-\partial m\partial t$  ( $\rightarrow -m\partial t$ ) 'concern' (CON), applicative suffix, §10.4.5.
- $-\delta m \partial n \sim -\delta m \partial n \sim -m \partial n \sim -m (n \text{ instrument,' lexical suffix, §13.6, 6.}$
- $-\partial m\partial t^{\theta} \sim -\partial m\partial t^{\theta}$  'long object,' lexical suffix, §13.6, 10.
- -*ət* 'we,' first-person plural subordinate subject suffix, §14.2.3.
- -*ət* 'subordinate passive,' §10.8.

- $-\partial n \sim -\dot{e} \cdot n$  'I,' first-person singular subordinate subject suffix, §14.2.3.
- $-\partial n \sim -\partial n'$  instrument,' lexical suffix, §13.6, 20.
- $-\partial n\partial \sim -\partial ne^{\gamma}$  'ear, side' lexical suffix, §13.6, 19.
- -ənəp 'ground,' lexical suffix, §13.6, 21.
- -ont 'day'(?), possible lexical suffix, §13.7, 5.
- -*ənəs* ~ -*ínəs* ~ *íləs* 'chest, beach,' lexical suffix, §13.6, 22.
- $-\partial n\partial s \sim -\partial l\partial s \sim -nis$  'tooth,' lexical suffix, §13.6, 23.
- -ənəq 'someone else, another person,' intransitive suffix, §10.2.3.
- $-\partial\theta \sim -a^2\theta \sim -\dot{a}^2\theta \sim -\dot{a}\theta\partial n \sim \theta i n$  'mouth, edge, margin,' lexical suffix, §13.6, 26.
- -*əs* 'he, she, it, they,' third-person transitive subject suffix, §14.2.2.
- -*əs* 'he, she, it, they,' third-person subordinate subject suffix, §14.2.3.
- -*əs* 'recipient' (RECIP), applicative suffix, §10.4.3.
- $-\partial s \sim -\dot{a}s$  'face, round object, moon, dollar, bow of canoe, bank,' lexical suffix, §13.6, 28.
- -əł 'past,' predicate particle, §16.2.1.
- -*ət p* 'plant, bush, tree, vegetation,' lexical suffix,' §13.6, 33.
- $-\partial t \partial n \sim -\partial t i n$  'someone, another,' intransitive suffix, §10.2.4.
- $-\partial t x^w$  lexical suffix, §13.6, 18.2.
- -əłəł ~ -łəł ~ -łél- 'throat, breath, wind, food,' lexical suffix, §13.6, 31.
- -*əłcə* 'water,' lexical suffix, §13.6, 37.
- -əłqəý 'snake,' possible lexical suffix, §13.7, 9.
- -ətxe (?) possible lexical suffix, §13.7, 10.
- $-\partial l \sim -il$  'move toward, turn to, become,' verbal suffix, §12.3.1.
- -ələ ~ -élə 'place for, container for,' lexical suffix, §13.6, 40.
- $-\partial l \partial \sim -\dot{e} l \partial$  'person,' lexical suffix, §13.6, 41.
- -*ələ* 'standing object'(?), possible lexical suffix, §13.7, 12.
- -*ələp* second-person possessive plural suffix, always occurs with  $2\partial n 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial r 2\partial$
- -*álmən* 'want to, intend to, seem about to,' modal suffix, §11.2.1.
- $-\partial lm \partial x^w \sim -m \partial x^w$ , etc. 'place, people, cluster'(?), lexical suffix, §13.6, 12.
- -*álməx<sup>w</sup>* 'breast, milk, spring (of water),' lexical suffix, §13.6, 13.
- $-\partial t \partial x^{w} (\rightarrow -tx^{w})$  'house,' lexical suffix, §13.6, 18.
- $-\partial l e n \partial x^w \sim -l e n \partial x^w$  'season,' lexical suffix, §13.6, 25.1.
- $-\partial ls$  'activity' (ACT), equivalent of -els in the progressive aspect, §10.2.2.
- $-\partial ls \sim -ls$  'weapon,' lexical suffix, §13.6, 45.
- $-\partial l\partial s (\rightarrow -\partial n\partial s)$  'tooth,' lexical suffix, §13.6, 23.
- $-\partial l\partial t \sim -\partial l ll$  'crotch, between the legs,' lexical suffix, §13.6, 46.
- -*ələtcə* ~ -*a*·*tcə* 'belly, mind,' lexical suffix, §13.6, 38.
- $-\partial l\partial tc\partial \sim -a \cdot tc\partial$  'giant,' lexical suffix, §13.6, 39.
- $-\delta ll \partial c \rightarrow -n \partial c$  'butt, tail, base, bay, price,' lexical suffix, §13.6, 24.
- $-\partial l\partial c \rightarrow -e^{2}c$  'route across,' lexical suffix, §13.6, 64.1.
- -ələc 'dung,' lexical suffix, §13.6, 49.

- $-\partial lc\partial p \rightarrow -c\partial p$  'fire, firewood,' lexical suffix, §13.6, 67.
- $-\partial l\partial c\partial n \rightarrow -\dot{e}c\partial n$  'testicle(s),' lexical suffix, §13.6, 66.
- -əlcł 'mind, feelings,' lexical suffix, §13.6, 38.
- $-\partial l x \partial n$  'line,' lexical suffix, §13.6, 53.
- $-\partial lk^{wl} \sim -\dot{a}lk^{wl}$  'winter dancer, possessing song,' lexical suffix, §13.6, 54.
- -*álqən* 'pelt, fur, hide,' lexical suffix, §13.6, 58.
- $-\partial lqt \sim -\acute{e}lqt$  'catch, game,' lexical suffix, §13.6, 59.
- -əléxən 'arm,' lexical suffix, §13.5, 77.1.
- $-\partial \partial x \partial t^{\theta} \sim -\dot{e} \partial x \partial t^{\theta}$  'barrier,' lexical suffix, §13.6, 60.
- $-\partial l y \partial n \sim -\dot{e} \dot{y} \partial n$  'net, trap,' lexical suffix, §13.6, 83.
- -*əlwəm* ~ -*álwəm* (?) possible lexical suffix, §13.7, 15.
- -álwət 'garment,' lexical suffix, §13.6, 61.
- -əlwəs 'heart, side of chest,' lexical suffix, §13.6, 62.
- -ələws ~ -íws 'body, bird,' lexical suffix, §13.6, 88.
- $-\delta l w \partial t \sim -\partial l w \partial l \rightarrow -w i t$  'side,' lexical suffix, §13.6, 91.
- $-\partial le^{2} \sim -\partial li \sim -\partial yi$  'game, sport'(?), lexical suffix, §13.6, 42.
- $-\partial le^{2}c \sim -e\dot{l}e^{2}c \sim -li^{2}c$  'container,' lexical suffix, §13.6, 50.
- -*əlé*<sup>2</sup>*c* 'bundle,' lexical suffix, §13.6, 50.1.
- -*əc* 'hundred,' lexical suffix, §13.6, 65.
- $-\partial c \partial n \sim -\partial c \rightarrow -i c \partial n$  'back, surface, top,' lexical suffix, §13.6, 64.
- -*əcale*<sup>2</sup> ~ -*cala*<sup>2</sup> (?) possible lexical suffix, §13.7, 17.
- $-\partial \dot{c} \sim -\dot{c}$  (?) possible lexical suffix, §13.7, 18.
- $-\partial x$  (?) possible lexical suffix, §13.7, 19.
- $-\partial x^{w}$  'you,' second-person singular subordinate subject suffix, §14.2.3.
- $-\partial x^{w}$  (?) possible lexical suffix, §13.7, 21.
- $-\partial x^{w} \theta \partial t \sim -\acute{e} l \partial x^{w} \theta \partial t$  'tongue,' lexical suffix, §13.6, 72.
- $-\partial x^{w}a^{\gamma}$  (?) possible lexical suffix, §13.7, 23.
- $-\partial q\partial n \sim -iq\partial n$ , etc. 'front, slope,' lexical suffix, §13.6, 75.1.
- $-\partial q \partial n \sim -\partial q \partial n'$  container,' lexical suffix, §13.7, 75.3.
- -*əqs* ~ -*əqsən* ~ -*qs*, etc. 'nose, snout, point (of land),' lexical suffix, §13.6, 76.
- $-\partial \tilde{x} \partial n \sim \dot{e} \tilde{x} \partial n \sim -\dot{x} \dot{e} n$  'arm, side, branch, perimeter,' lexical suffix, §13.6, 77.
- $-\partial q^w \sim -q^w$ , etc. 'head,' lexical suffix, §13.6, 78.
- $-\partial y \sim -i$  'third-person subordinate passive subject suffix,' §14.2.6.
- $-\partial y' \sim -\dot{e}y'$  'plant, tree, wood,' lexical suffix, §13.6, 79.
- $-\partial y$  'someone who does (agent)'(?), lexical suffix, §13.6, 80.
- $-\partial \dot{y} \sim -a \dot{y}$  'fish'(?), lexical suffix, §13.6, 81.
- $-\partial ye \sim -\partial \dot{y}e \sim -\dot{a}\dot{y}e$  'dear one(?), small creature(?), 'lexical suffix, §13.6, 82.
- $-\partial y i \sim \partial l i^{\circ} \sim -\partial l i$  'game, sport'(?), lexical suffix, §13.6, 42.
- $-\partial y\partial t \sim -i \cdot t$  realization of  $-\partial y$  'third-person subordinate passive subject' and  $-\partial t$  'subordinate passive,' §14.2.6.
- -*∂yás* ~ -*∂ýás* ~ -*∂y∂s* 'circular figure,' lexical suffix, §13.6, 84.
- $-\delta y \partial t \sim -\delta y \partial t \sim -\delta y \partial t \sim -\delta y \partial t$  'child, people, ceremony,' lexical suffix, §13.6, 85.
- $-\partial \dot{y}e^{2}q \sim -a\dot{y}e^{2}q \sim -ay\partial q$  'projection,' lexical suffix, §13.6, 74.1.

 $-\partial wm \partial x^w \rightarrow -m \partial x^w$  'place, people, cluster'(?), lexical suffix, §13.6, 12.

- $-\partial wtx^w \rightarrow -tx^w$  'house,' lexical suffix, §13.6, 18.
- -*awan* ~ -*iwan* 'inside, middle, waist, behind, rump, trunk (of tree),' lexical suffix, §13.6, 86.

 $-\partial w \partial \theta \sim -a \dot{w} \partial \theta$  'kind, ingredient,' lexical suffix, §13.6, 87.

 $-\partial \dot{ws} \rightarrow -i\dot{ws}$  'body, bird,' lexical suffix, §13.6, 88.

 $-\partial w\partial t \rightarrow -w\partial t$  'canoe, vessel, mind'(?), lexical suffix, §13.6, 90.

- $-\partial we^{2}t \sim -we^{2}t$  'back,' lexical suffix, §13.6, 92.
- -*əwíc* ~ -*əwəc* 'back,' lexical suffix, §13.6, 64.3.

#### e

- -é 'times'(?), possible lexical suffix, §13.7, 28.
- $-\acute{e}p \sim -\partial p$  'base, bottom'(?), lexical suffix, §13.6, 1.
- $-\acute{e}p \sim -\partial p$  'hair'(?),lexical suffix, §13.6, 2.
- $-\acute{e}\cdot p \sim -\partial p$  'you (pl.),' second-person plural subordinate subject suffix, §14.2.3.
- -émon ~ -mon ~ -min 'residue,' lexical suffix, §13.6, 8.
- -émon 'extracted liquid,' lexical suffix, §13.6, 9.
- -émət<sup> $\theta$ </sup> ~ -əmət<sup> $\theta$ </sup> 'long object,' lexical suffix, §13.6, 10.
- $-\acute{em} a x^w \rightarrow -m a x^w$  'place, people, cluster'(?), lexical suffix, §13.6, 12.
- $-\acute{e}\cdot n \sim -\partial n$  'I,' first-person singular subordinate subject suffix, §14.2.3.
- - $\acute{e}$ ·n 'side, component, tip, feather,' lexical suffix, §13.6, 19.1.
- $-\dot{e}\cdot nx^w \sim -\dot{e}\cdot n\partial x^w \sim -\dot{e}n\partial x^w$  'fish, food, fish run, season, (bad) weather'(?), lexical suffix, §13.6, 25.
- -éł 'times,' lexical suffix, §13.6, 29.
- -él- ~ -Sél- 'I,' first-person singular passive subject suffix, §14.2.6.
- -élə ~-ələ 'place for, container for,' lexical suffix, §13.6, 40.
- $-\acute{el} \sim -\partial l \partial$  'person,' lexical suffix, §13.6, 41.
- -é·lt realization of -él-, first-person singular passive subject suffix, and -*ət* subordinate passive suffix, §14.2.6.
- -é·ltəx<sup>w</sup> 'spouse,' lexical suffix, §13.6, 18.1; §21.2.1.
- -é $l\dot{t}^{\theta}e^{\gamma} \sim -i\dot{t}^{\theta}e^{\gamma}$  'blanket, wealth,' lexical suffix, §13.6, 27.
- -éls 'activity' (ACT), §10.2.2.
- -éləc 'person,' lexical suffix, §13.6, 48.
- $-\acute{e}l ac \sim -e^{2}c$ , etc. 'route across,' lexical suffix, §13.6, 64.1.
- $-\acute{el}\partial c\partial n \sim -\partial \partial c\partial n \sim -\acute{ec}\partial n$  'testicle(s),' lexical suffix, §13.6, 66.
- $-\acute{elc} \sim -\acute{olc}$  'hair, line,' lexical suffix, §13.6, 52.
- -éləq 'wave,' lexical suffix, §13.6, 55.
- -éləqəp 'smell, taste, sound' (perhaps 'trace'), lexical suffix, §13.6, 56.
- -éləqən ~ -éləqən 'line, stream,' lexical suffix, §13.6, 57.
- -élqł ~ -əlqł 'catch, game,' lexical suffix, §13.6, 59.
- $-\acute{e}l\partial \check{x}\partial \acute{t}^{\theta} \sim -\partial l\partial \check{x}\partial \acute{t}^{\theta}$  'barrier,' lexical suffix, §13.6, 60.
- $-\acute{e}l\partial \dot{w}s \rightarrow -i\dot{w}s$  'body, bird,' lexical suffix, §13.6, 88.
- $-\acute{elwe}^{2s} \sim -\partial \acute{lwe}^{2s}$  'paddle,' lexical suffix, §13.6, 63.

- $-\acute{ele}^{2}c \sim -\partial le^{2}c \sim -li^{2}c$  'container,' lexical suffix, §13.6, 50.
- -écən ~ éləcən ~ -áləcən 'testicle(s),' lexical suffix, §13.6, 66.
- $-\acute{e}x^w$  'times'(?), possible lexical suffix, §13.7, 22.
- -eq 'penis, projection, rear,' lexical suffix, §13.6, 74.
- $-\acute{e}\check{x}\partial n \sim -\partial\check{x}\partial n \sim -\check{x}in$  'arm, side, branch, perimeter,' lexical suffix, §13.6, 77.
- $-\acute{e}\dot{y} \sim -\partial \dot{y}$  'plant, tree, wood,' lexical suffix, §13.6, 79.
- $-\acute{e}$ yən ~  $-\emph{o}$ lyən 'net, trap,' lexical suffix, §13.6, 83.
- $-\acute{e}\dot{y}\partial t \sim -\partial y\partial t \sim -\acute{e}\dot{y}t$  'child, people, ceremony,' lexical suffix, §13.6, 85.
- $-\acute{e}wtx^w \rightarrow -tx^w$  'house,' lexical suffix, §13.6, 18.
- -éwəs 'figure, back, trunk (of body),' lexical suffix, §13.6, 89.
- $-\acute{e}w\partial t \rightarrow -w\partial t$  'canoe, vessel, mind'(?), lexical suffix, §13.6, 90.
- $-\dot{e}^{2}t$  'bed, bedding,' lexical suffix, §13.6, 30.
- -e?t 'throat, breath, wind, food,' lexical suffix, §13.6, 31.
- $-e^{2}c \sim -\acute{e}l\partial c \sim -\partial l\partial c \sim -lic$  'route across,' lexical suffix, §13.6, 64.1.

### а

- -am- ~ -Sam- 'you,' second-person singular passive subject suffix, §14.2.6.
- -*a*·*m* realization of -*am* ~ -*Sam* 'you,' second-person singular passive subject suffix, and -*am* 'intransitive,' §14.2.6.
- -amə ~ -Samə 'you,' second-person singular object suffix, §14.2.5.
- -amx ~ -Samx 'me,' first-person singular object suffix, §14.2.5.
- $-ám\partial x \sim -ám\partial x \sim -m\partial x$  'country, person,' lexical suffix, §13.6, 11.
- -á·məx ~ -amx 'look, appear,' verbal suffix, §12.3.4.
- -áma? 'body,' lexical suffix, §13.6, 14.
- -anom (?) possible lexical suffix, §13.7, 6.
- $-\dot{a}\theta \partial n \sim -\partial \theta$ , etc. 'mouth, lip, margin, edge,' lexical suffix, §13.6, 26.
- $-ás \sim -\partial s$  'face, round object, moon, dollar, bow of canoe, bank,' lexical suffix, §13.6, 28.
- -á·t 'travel by,' verbal suffix, §12.3.3.
- -a·tcə ~ -ələtcə 'belly, mind,' lexical suffix, §13.6, 38.
- $-a tc \partial \sim -\partial l \partial t c \partial$  'giant,' lexical suffix, §13.6, 39.
- -al- 'we, you (pl.),' first- and second-person plural passive subject suffix, §14.2.6.
- -alə 'you (pl.),' second-person plural object suffix, §14.2.5.
- -*a*·*lt* realization of -*al*-, first- and second-person plural passive subject, and -*ət*, subordinate passive suffix, §14.2.6.
- $-a\dot{l}x^{w}$  'us,' first-person plural object suffix, §14.2.5.
- $-ál_{\partial s} \sim -\partial l_{\partial s}$  'eye, mesh (of net), star, appearance, tendency'(?), lexical suffix, §13.6, 43.
- -áls 'rock, spherical, object, round berry,' lexical suffix, §13.6, 44.
- -alt ~ -alt 'young,' lexical suffix, \$13.6, 47.
- -álk<sup>w</sup>ł 'winter dancer, possessing song,' lexical suffix, §13.6, 54.

- -álk<sup>w</sup>t 'child of,' lexical suffix, §13.6, 54.1.
- -alwam (?) possible lexical suffix, §13.7, 15.
- -áləŵcəs 'finger,' lexical suffix, §13.6, 68.1.
- -áləwxən 'toe,' lexical suffix, §13.6, 69.1.
- $-aq^{w} \sim -a^{2}q^{w} \sim -q^{w}$ , etc. 'head,' lexical suffix, §13.6, 78.
- $-a\dot{y}e \sim \partial y\partial \sim -\partial \dot{y}\partial$  'dear one (?), small creature (?), 'lexical suffix, §13.6, 82.
- $-aya\theta \sim -a\dot{y}a^{2}\theta$  'head of weapon'(?), lexical suffix, §13.6, 26.1.
- $-\dot{a}\cdot y\theta \partial n \sim -a^2\theta$ , etc. 'mouth, lip, margin, edge,' lexical suffix, §13.6, 26.
- $-a\dot{y}e^{2}q \sim \partial\dot{y}e^{2}q \sim -\dot{a}\dot{y}\partial q$  'projection,' lexical suffix, §13.6, 74.1.
- $-a\dot{w}m\partial x^w \sim -m\partial x^w$  'place, people, cluster'(?), lexical suffix, §13.6, 12.
- $-\dot{a}\dot{w}\partial\theta \sim -\partial w\partial\theta$  'kind, ingredient,' lexical suffix, §13.6, 87.
- $-\dot{a}w\partial t \rightarrow -w\partial t$  'canoe, vessel, mind'(?), lexical suffix, §13.6, 90.
- $-\dot{alwal} \sim -alwil \sim -wil$ , etc. 'side,' lexical suffix, §13.6, 91.
- $-awaq^{w} \rightarrow -q^{w}$  'head,' lexical suffix, §13.6, 78.
- $-\dot{a}\dot{w}aq^{w}$  'hat,' lexical suffix, §13.6, 78.1.
- $-a^{2}\theta \rightarrow -\theta \partial n$  'mouth, lip, margin, edge,' lexical suffix, §13.6, 26.
- -a?t 'attribute formative' (ATT), derivational suffix, §12.1.5.

### i

- -i- ~  $-\partial y$  third-person subordinate passive subject suffix, §14.2.6.
- -*i*·*m* 'die from,' verbalizing suffix, §12.3.2.
- -*i*·*ma*<sup>?</sup> (?) lexical suffix, §13.6, 15.
- *-i*·*t* realization of *-aya*-, third-person subordinate passive subject, and *-at*, subordinate passive suffix, §14.2.6.
- -*i*·*n* (?) lexical suffix, §13.6, 19.2.
- $-in \partial s \sim -il \partial s \sim -\partial n \partial s$  'chest, beach,' lexical suffix, §13.6, 22.
- -*i*· $nx^{w} \sim -wi$ · $nx^{w}$  'year,' lexical suffix, §13.6, 25.2.
- $-i\dot{t}^{\theta}e^{\gamma} \sim -e^{i}\dot{t}^{\theta}e^{\gamma}$  'blanket, wealth,' lexical suffix, §13.6, 27.
- $-i \cdot t \sim -i \partial t$  'arrange to, seek to (arrange),' modal suffix, §11.2.2.
- $-il \sim -\partial l$  'move toward, turn to, become,' verbalizing suffix, §12.3.1.
- -*íləm* compound of -*íl* 'move toward' and -*əm* 'intransitive,' §12.3.1.
- $-il_{\partial s} \sim -in_{\partial s} \sim -\partial n_{\partial s}$  'chest, beach,' lexical suffix, §13.6, 22.
- $-i \cdot ls$  'activity' (ACT), equivalent of -els in the progressive aspect following a lexical suffix, §10.2.2.
- -*íləx* (?) possible lexical suffix, §13.7, 13.
- $-il\partial ws \rightarrow -iws$  'body, bird,' lexical suffix, §13.6, 88.
- $-ic \partial n \sim -\partial c \partial n \sim -ic \sim -\partial c$ 'back, surface, top,' lexical suffix, §13.6, 64.
- $-iq \partial n \sim -\partial q \partial n \sim -q \partial n \sim -q \partial n$  'front, slope,' lexical suffix, §13.6, 75.1.
- -*íwən* ~ -*əwən* ~ *íwə* 'inside, middle, waist, behind, rump, trunk (of tree),' lexical suffix, §13.6, 86.
- $-i\dot{w}s \sim -\partial\dot{w}s \sim -il\partial\dot{w}s \sim -\partial\partial\dot{w}s$  'body, bird,' lexical suffix, §13.6, 88.

# Appendix 2 Names of Places and Peoples

I will list first the names recorded from Musqueam sources for places or features within the territory that the Musqueam people regard as having been their own in the recent past. This territory extends from the north shore of Burrard Inlet south to the main channel of the Fraser River and eastward to where the North Arm separates from the main channel. Some of this territory may be disputed by others. While Musqueam traditions tell that they once had villages on Burrard Inlet at the Capilano River and at Jericho, for several generations Squamish people have been settling there. Kuipers (1969, 32-38) and others have recorded a greater number of Squamish place names on Burrard Inlet than I have recorded Musqueam place names.<sup>1</sup> What I have recorded for the whole territory is probably only a fraction of the names that were once in use, however.

These names were given by Andrew Charles (AC) and Christine Charles (CC) in the late 1950s, by James Point (JP) in the 1960s, by Arnold Guerin (AG) in the early 1980s, and by Ed Sparrow (ES) in 1994. I have also included names recorded by David Rozen (R) in 1979 from an unidentified source or sources, transliterated into the orthography used here. Where there are several versions of a name, I have listed first the one I believe to be most accurate. I will begin with the Musqueam village and its immediate environs and then list names to the north, east, and south.

#### The Musqueam Village

(1)  $x^w m \delta \theta k^w \delta y \delta m$  (<  $m \delta \theta k^w \delta y$  'a rush-like plant') Musqueam, the village and the people.

In the middle of the nineteenth century, Musqueam consisted of two settlements,  $m \dot{a} l \partial \dot{y}$  and  $s \dot{c} \partial l \dot{e} x^w$ , on the north bank of the North Arm, within the present Musqueam Reserve. The settlement of  $m \dot{a} l \partial \dot{y}$  was near the bank of the

<sup>1</sup> See Suttles 1996 for a comparison of Musqueam and Squamish place names around Burrard Inlet.



Map 2 Map of Musqueam Territory

river, below the cemetery. A few hundred metres upriver was the mouth of a stream with two tributaries, one a creek flowing out of the forest to the north, the other a slough flowing from the east. The settlement of  $s\dot{c}al\dot{e}x^w$  was within the conflux of these two streams, a few hundred metres above the mouth of the creek. There may have been only a single large house at  $m\dot{a}la\dot{y}$ , but there may have been as many as a dozen at  $s\dot{c}al\dot{e}x^w$ . By the mid-1890s, the space between the two settlements was partly filled with new houses.

The name "Musqueam" was first recorded, as "Misquiame," on 2 July 1808 by Simon Fraser, who was guided "up a small winding river to a lake to the village." After an hour in the village, Fraser's party found their canoe stranded by an ebbing tide and had to drag it "some distance" to the water. Downriver they saw a second village, but did not land (Lamb 1980, 106). It is likely that the village Fraser visited was  $s\hat{c}al\hat{e}x^w$ . The one he only saw may have been  $m\hat{a}la\hat{y}$ , or it may possibly have been  $\hat{q}\hat{a}la\hat{x}an$ .

The plant for which the village is named has been described variously. The Charleses said that the plant, which they had never seen, was said to have been grasslike, with an edible root. It no longer exists because of grazing cattle. JP said it had sharp thorns, and possibly small violet flowers, and it was now gone because of burning and dyking. However, Kinkade (1986) has shown that comparative linguistic evidence indicates that the name originally designated the blackcap, which, aside from its thorns, does not fit either description.

The Charleses said that the plant underwent cycles of abundance; it could be reduced to small numbers and then increase greatly in a short time, and the Musqueam people were pleased to compare themselves to the plant in this respect. JP told a story of how a two-headed serpent came down out of a small lake at  $x^w m \partial m q^w \dot{e} \cdot m$  and through the village, making the winding bed of the stream and leaving the plant  $m \partial \theta k^w \partial y$ , which was regarded as sacred. (See §23.5 for the text.) During my work with them, however, the Charleses did not mention this story, but they learned from an older person that the name  $x^w m \partial \theta k^w \partial y^\partial m$  properly applies to a site between  $m d l \partial y$  and  $s c \partial l e x^w$  where Mink was playing with a log and got tangled in this grass and got stuck to the log.

Some of the named sites within the village were identified by references to houses that existed in the period when the work was done, from the late 1940s to the early 1970s. I have omitted references to occupants of earlier houses.

(2)  $x^w q i \dot{m} \partial k^w \partial m$  ('octopus place' [AC]),  $sqim\partial k^w \partial lm\partial x^w$  ('octopus breast' [JP]  $\leq sq i \dot{m} \partial k^w$  'octopus, ''devilfish,'''  $-\partial lm\partial x^w$  'breast, milk, spring'). A spring near  $m \dot{a} l \partial \dot{y}$ , where  $\dot{x} \dot{e} \dot{l} s$  destroyed a giant octopus.

(3)  $m \acute{a} l \partial \acute{y}$  (< ?) below the cemetery (AC, CC). From the west end of the occupied area below the cemetery to Cornelius Johnny's house (JP).

(4)  $x^{w} \vec{\lambda} \partial p i c \partial n$  (AC, CC, JP) (>  $\vec{\lambda} \partial p$  'deep, down, below,' -*icon* '-back, -surface'). From the last to Nelson Dan's (AC, CC). Between  $m \dot{a} l \partial \dot{y}$  and  $\dot{t}^{\theta} i \dot{t}^{\theta} \partial \tilde{x} \partial t$  (JP).

(5)  $x^w m \delta \theta k^w \delta y \delta m$  (see [1] above). From Nelson Dan's to Alice Louie's (AC, CC).

(6)  $t^{\theta} t^{t} \theta^{\delta} x \partial t$  ('midden'?  $\leq t^{\theta} \delta x \partial t$  'white beach of mixed shell and pebbles'). Extending from Alice Louie's to the creek (AC, CC). From the site of the new smokehouse along the beach upriver (JP).

(7)  $pi pq^w \partial c \partial n$  ('sandy on high land,'  $< pq^w i c \partial n$  'sand on the beach,'  $'break,' <math>-i c \partial n$  'back, surface'). From George Robert's house to the creek (AC, CC). Two big houses stood here, east of the church (JP).

(8)  $\delta x^w k^w \delta m \dot{\lambda}$  ('ridge,'  $< \sqrt{k^w \partial m}$ , cf.  $\delta x^w k^w \delta m \dot{\lambda} \partial s$  'with unflattened head, with forehead sticking out,'  $s k^w \delta m \dot{t}^{\theta}$  'lump,'  $k^w \delta m \dot{t}^{\partial \sigma}$  'humpbacked'). The

ridge behind Alice Louie's house (AC, CC). There was just one big house here (JP).

(9)  $st \acute{a}^2 t l \vartheta \acute{w}$  ('creek') (AC, CC, JP). The creek that flows through Musqueam, separating  $s \acute{c} \vartheta \acute{e} x^w$  from  $m \acute{a} l \vartheta \acute{y}$  and the other named sites.

(10)  $s\dot{t}\acute{e}lq \partial y$  (AC, CC) (<  $\sqrt{\dot{t}}\partial lq$  'get muddy'?). A stream that flows from the northwest into the creek.

(11)  $sc \partial w \dot{e} \dot{l} \partial c$  (AC, CC) (<  $\sqrt{cew}$  'shore,' - $\dot{e} \dot{l} \partial c$  'route across'). The grassy area along the mouth of the creek, also called  $s p \partial t \dot{x} \partial n$  'meadow, prairie, pasture.'

(12)  $s\dot{c}al\dot{e}x^{w}$  ('upstream,' resultative of  $c\dot{a}lax^{w}$  'go upriver'). The upper village. According to the Charleses, it consisted of two rows of houses, the gap between them filled by another house at the east end, the whole forming a rectangle open at the west end. The creek from the north flowed just to the west of this rectangle. The front row of houses was on a ridge running parallel to the slough. Unless the tide was low, canoes could be brought up the creek and into the slough. The space within the rectangle formed by the houses was a play-ground, in which the stone trophy called *qiscám* stood. JP identified the houses along the front ridge, but was less clear about location of the others. The house sites were sites 13 to 20 below.

(13)  $w \delta \dot{q}^w \delta \dot{x} \partial n$  (AC, CC, JP) ('downstream side,'  $\langle w \delta \dot{q}^w$  'drift with current,' - $\dot{e} \dot{x} \partial n$  'arm, side'). The lower (western) end of the ridge. AC and CC thought that two or three houses had stood here. JP thought that before his time there may have been four houses on this segment of the ridge, one or two of them standing at right angles to it. When he was a boy, there were two houses there. The upper one he identified as  $s \partial m \dot{e} \partial \partial$  (see next).

(14) s<sup>2</sup>aména (AC, CC), saméne (JP) ("seems to mean 'missing one ear,"  $< \sqrt{2}em$ - 'give,' -éna 'ear'). AC and CC identified this as a site above  $w\dot{a}\dot{q}^wa\dot{x}an$ . JP identified it as a part of that segment.

(15) titməlqsən (AC, CC) ('painted end,'  $< t\delta m \partial t$  'red ochre, red paint,'  $-\partial qs \partial n$  '-nose, -point). So called because a beam end was painted to prevent rot (AC). This house was said to have been built at right angles to the ridge and the other houses. It had three posts in front and sloped upward to the rear. It was gone before JP's time. Vincent Stogan's house was there in 1962 (JP).

(16) *q*əlámqən (AC, CC, JP) (<?).

(17)  $t \delta y t \check{x} \partial n$  ('upstream side,'  $\leq t \delta y t$  'upstream,'  $-e \check{x} \partial n$  '-arm, -side'). There were two houses on this segment of the ridge.

(18)  $x^{w}l\dot{a}mlamas$  (AC, CC, JP) (<  $l\dot{a}m$  'get struck by something thrown,' -ss '-face'). The site of a house oriented at right angles to the front row. The ball used on the playing field used to strike the side of the house, hence the name (AC, CC). JP thought that this house may have been on the hillside to the east rather than where the Charleses indicated.

(19)  $scolq^w \acute{a}\theta \partial n$  ('woods-edge' <  $c\acute{a}l \dot{o}q^w$  'inland,'  $-a\theta \partial n$  '-lip, margin'). A row of houses parallel to the first (AC, CC). JP thought this site lay east of the ridge with sites 13 to 17, along the slough, at about a 45° angle to the ridge.

(20)  $k^w \partial k^w q^w \partial l e \check{x} \partial n$  (AC, CC) (<  $k^w \acute{a} q^w$  'get hit by something held in the hand,'  $- \acute{e}\check{x} \partial n$  '-arm, side'). A house at the upper (eastern) end of  $sc \partial l k^w \acute{a} \theta \partial n$ , opposite  $t \acute{a} y t \check{x} \partial n$ .

(21)  $\check{x}\acute{a}c\partial_{\partial}q^{w}(R)$  ('lake at the head,' <  $\check{x}\acute{a}ca^{2}$  'lake,' - $\partial_{\partial}q^{w}$ '-head'). A small lake that is the source of Musqueam Creek, near 25th Avenue and Camosun Street, according to Rozen's source.

(22)  $x^{w}m\partial m q^{w}\dot{e}^{m}$  (AC, JP) ('boggy place,' <  $m\dot{a}q^{w}\partial m$  'bog'). A swamp where the Catholic School is now (AC). A bog that was until recently a small, shallow lake, from which the two-headed serpent came in an account of how Musqueam got its name (JP; see §23.5). This is probably the same as the name Rozen recorded as  $m\partial q^{w}\dot{e}^{m}$ , off Crown Street and 25th Avenue.

### North Shore of North Arm Below the Village

(23)  $h \delta m l \partial s \delta m$  (AG),  $h \delta m l \partial s \delta m$  (CC, JP). A big rock at the west end of the reserve. The name is said to mean 'bend over to drink directly from the water,' but this word was given as  $h \delta m \delta \cdot y s \delta m$ . A person doing this was turned into stone by  $\check{x} \acute{e} \cdot \dot{l} s$ .

(24)  $\delta x^w \delta y t \partial n$  (AG),  $\delta x^w \delta y t \partial n$  (CC),  $\delta x^w \delta y t \partial n$  (JP) ('chamber pot'). A smaller rock next to the last, to whom it had belonged. It was spilled at the time of the transformation (AG).

(25)  $\check{x}\partial(p)\check{x}p\check{a}\check{y}\partial tp$  (R) ('cedar trees'). A place near Point-No-Point.

(26)  $sy \dot{e}^2 t \partial n$  (CC) ('widow, widower'). A rock below the Simon Fraser monument, a widow turned into stone by  $\dot{x} \dot{e} \cdot \dot{l} s$  (AC).

(27)  $\dot{q}i\dot{q}\partial \partial x \partial n$  (CC),  $\dot{q}i\dot{q}\partial x \partial n$  (AG) (dim. of next). A place near Wreck Beach (in its earlier sense, not in its recent extension).

(28)  $\dot{q}\delta l \delta \tilde{x} \partial n$  ('stockade,' probably  $\langle \sqrt{\dot{q}}\partial l$ - 'camp,'  $-\dot{e}\tilde{x}\partial n$  'arm, side'). A place south of Point Grey, where Capilano had a fort (JP).

(29) <sup>2</sup>*álqsən* (AC, AG) (cf. *s*<sup>2</sup>*álqsən* 'point'). Point Grey.

(30)  $\dot{t}^{\theta} \dot{c} \partial l q^{\psi}$  (AC),  $\dot{t}^{\theta} c \partial l q^{\psi}$  (JP),  $\dot{t}^{\theta} \partial c \partial l q^{\psi}$  (R) (possibly from  $\dot{t}^{\theta} c \partial t$  'grab by the hair,' or possibly a plural of  $\dot{t}^{\theta} c t^{2} q^{\psi}$  'hard head'). AC identified the name as that of a rock seen at low tide where people would poke the bottom and sing,  $^{2a} \cdot st \dot{e} y \partial w \partial t^{2a}$  'Oh, West Wind, Oh,' to bring the west wind. JP identified the name as 'Point Grey,' saying that the name implies that the point is stormy, always rough. August Jack, a Squamish source, identified this as a man who had intended to blow away "the great man" (presumably  $\check{x} \dot{e} \cdot \dot{l} s$ ) but was instead transformed into a rock, the biggest on the shore at Point Grey (Matthews 1955, 394).

(31)  $\dot{q}\dot{e}\dot{w}\partial m$  (JP, AG) ('howling'). A rock near Point Grey that was a dog about to bite  $\dot{x}\dot{e}\dot{l}s$ .

(32) iaibtom (JP) ('shivering,' progressive of  $i^{\theta}atom$  'feel cold'). A rock of white granite below the University of British Columbia. According to JP, it was an old lady who was crying here (presumably when  $\dot{x}\dot{e}\cdot\dot{l}s$  came and transformed her). Tim Moody, a Squamish source, identified the name as that of a

little ravine with a spring directly below the BC Tel cable hut (Matthews 1955, 395).

### **Burrard Inlet**

(33)  $p \partial q^w \delta c \partial n$ ,  $p i p q^w \partial c \partial n$  (R) ('sand on the beach,' 'be broken' and*-icon*'back, surface'). A sandbar off Spanish Banks.

(34)  $\delta x^w sicom$  (ES) (so recorded, probably 'sandy place'  $\leq \delta x^w$ -, syicom 'sand'). Spanish Banks.

(35)  $q^w \partial^2 a p \partial t p$ ,  $q^w a^2 a p \partial t p$  (R) ('crabapple tree'). A creek at Spanish Banks.

(36)  ${}^{2} \partial y \dot{a} lm \partial x^{w}$  or  ${}^{2} i \dot{a} lm \partial x^{w}$  (AC, JP, ES)  ${}^{2} i y \dot{e} lm \partial x^{w}$  (AG) (<  ${}^{2} \dot{\partial} y \dot{y}$  'good' and either a connective  ${}^{-al}$ -plus  ${}^{-m} \partial x^{w}$  'earth, people' or  ${}^{-alm \partial x^{w}}$  'breast, milk, spring,' thus either 'good place' or 'good spring water'). Jericho Beach. According to Musqueam tradition, this was the site of a Musqueam village at the time of the warrior Capilano. (The Northern Straits version of the name refers to a place at American Camp on San Juan Island that was a parklike prairie and also the site of a good spring.)

(37)  $\dot{t}^{\theta} \partial m \dot{t}^{\theta} \dot{a} m \partial \dot{l} s$  (AG),  $\dot{t}^{\theta} \partial m \dot{t}^{\theta} \dot{a} m \partial l s$  (JP, R) ('grindstone,' from the root of  $\dot{t}^{\theta} \partial m \partial l s$  'file'). A creek at Bayswater Street.

(38)  $sk^w \dot{a}yus$ ,  $sk^w \dot{a}y \partial \dot{w}s$  (R). Kitsilano Beach. The Squamish is  $sq^w \dot{a}yus$ , possibly 'burnt face,'  $< q^w \dot{a}y$  'burn, scorch' and *-us* 'face,' which might have referred to the hillside after a logging operation there. If the Musqueam is correctly transcribed, it would seem to mean 'disabled body,'  $< sk^w \dot{e}y$  'disabled,' *-ows* 'body.'

(39)  $s \partial n \dot{a} q^w$  (AC, JP, AG, ES, R) (according to information obtained by Donna Gerdts, the resultative of  $n \partial^2 \dot{a} q^w$  'direct the head there'). The site of the Kitsilano Reserve at the entrance to False Creek.

(40)  $2\delta y \partial x \partial n$  (ES) (<  $2\delta y'$  'good,'  $-x\partial n$  '-foot,' hence probably 'good under foot'). English Bay.

(41)  $stit dw a d^{w}$  (R) (dim. of  $st dw a d^{w}$  'fuller's earth'). Second Beach.

(42) *stxílax* (*s*- 'nominalizer,' *txílax* 'be standing'). Siwash Rock.

(43)  $\check{x}\acute{a}^2\check{x}c\vartheta$  (ES),  $\check{x}\acute{a}\check{x}c\vartheta$  (R) (probably  $\check{x}\acute{a}^2\check{x}ca^2$ , dim. of  $\check{x}\acute{a}ca^2$  'lake'). Beaver Creek and Lake.

(44)  $\check{x}^w \acute{a} y \check{x}^w \partial y$  (JP),  $\check{x}^w \acute{a} \dot{y} \check{x}^w \partial \dot{y}$  (ES) (<  $s \check{x}^w \acute{a} y \check{x}^w \partial y$  'masked dance performance'). Lumberman's Arch. The name is from the tradition that an ancestor received the privilege there.

(45) spápayaq (ES) (the final  $\dot{q}$  may be an error for q, cf. Squamish pápiyaq, possibly  $< \sqrt{pay}$  'be bent' and  $-a\dot{y}aq$  'projection,' i.e., 'bent at the end'). Brockton Point.

(46)  $s\vec{k}^{w}\theta\vec{e}^{2}(\mathbf{R})$ . Deadman's Island. The name is probably an error for  $sk^{w}\theta\vec{e}^{2}$  or  $sk^{w}\theta\vec{e}^{2}s$ , the Island Halkomelem for 'island.' The Squamish name for the islet is the cognate  $sk^{w}c\vec{a}^{2}s$  'island.' The Musqueam for 'island' is  $\vec{\lambda}c\vec{e}s$ .

(47)  $\vec{p}\vec{q}\vec{a}\vec{l}s$  (R) ('white rock' <  $\vec{p}\vec{a}\vec{q}$  'white,' - $\vec{a}\vec{l}s$  'rock'). The foot of Granville Street.

(48)  $\dot{q} \partial m \dot{q} \partial m \partial l \partial l p$  (ES, R) ('bigleaf maple trees,' <  $\dot{q} \partial m \partial l$  'paddle'). Hastings Mill.

(49)  $\check{x}a\check{x}ipi(?alp)$  (R) (<  $\check{x}p\acute{e}\check{y}$  'Western red cedar'). Cedar Cove. The Squamish given by Rozen (recorded by Randy Bouchard) is  $\check{x}ap\check{x}ápa\check{y}ay$ , which appears to be the plural of  $\check{x}ápa\check{y}ay$  'young cedar.' The Musqueam form recorded by Rozen is not one I have recorded, but it looks like a possible plural.

(50)  $tomtom(x^wton)$ . CC gave this for the Burrard Reserve, but it is properly the name for Belcarra. The name is the same in Squamish and seems to be from Squamish  $tom(x^w)$  earth, land,' reduplicated, probably for plural, with the suffix -ton 'instrument.' The Halkomelem for 'earth' is  $tomox^w$ . AG identified  $tom(x^wton)$ , the unreduplicated form, as referring to crossed logs filled with earth to prevent erosion. The similar name  $tom(x^wton)$  is the name in the Nooksack language of a Nooksack village near Sumas, Washington, said to have consisted of pit houses. The name suggests this use, but it is not the term for pit house, which is sqom(n) in Nooksack and in Squamish (so given by Hill-Tout 1900, 485), sqom(n) in Halkomelem, also  $sponewtx^w$  in Musqueam. The village at Belcarra is said to have been the earlier home of the Burrardview people. Probably for this reason CC identified the name with the Burrardview Reserve.

(51) solilwota<sup>2</sup>t (ES), solilwota<sup>1</sup> (JP), s<sup>2</sup>olilwota<sup>2</sup>t (AG). Indian Arm and Indian River. This is also said to be the name of the Burrardview people, whose earlier home was at Belcarra and whose salmon stream was Indian River, which was also used by the Musqueam. According to Musqueam and Katzie sources, they formerly spoke Halkomelem and later adopted Squamish. The name is probably derived, by  $R \Rightarrow \rightarrow h \Rightarrow R$ - reduplication and the combination s- 'nominalizer' and -a<sup>2</sup>t 'attributive,' from the name of the group identified by Boas (1887, 132) as the "Lel'elot" (in modern symbols probably *lalilut*), the Squamish "family" on Burrard Inlet. The Squamish name is sol<sup>2</sup>(1<sup>2</sup>utut. The affixes correspond to those of Halkomelem, but since this type of reduplication is usual in Halkomelem but rare if present at all in Squamish, the name is probably Halkomelem in origin.

(52)  ${}^{2}\delta\theta n \partial c$  (ES) (cf.  ${}^{2}\delta\theta n \partial c$  'bay'). Burrardview Reserve. The Squamish is the cognate  ${}^{2}\delta cna\check{c}$  or  ${}^{2}\delta cna\check{c}$  'bay.' CC gave the name of the Burrardview Reserve as  $t \partial m t \partial m (x^w t \partial n, no)$  doubt because this is the name of the earlier village of the Burrard people.

(53)  $\vec{\lambda} \delta t \partial m \delta t q \partial^2$  (R) (<  $\vec{\lambda} \delta t \partial m$  'salt'?). Mackay Creek. Kuipers translates the Squamish name  $\vec{\lambda} a t \partial m \hat{a}^2 \partial l k^w$  as 'salt water,' composed of Squamish  $\vec{\lambda} a t \partial m$  and a suffix containing the formative  $-k^w$  'found in words connected with the sea,' and he adds that the name "is otherwise not quite clear." The suffix in the Musqueam name Rozen recorded is not identifiable, unless it is an error for  $-\partial t c \partial^2 water.'$ 

(54)  $x^{w}m\delta l\partial c\partial \partial n$  (AC),  $x^{w}m\delta l\dot{c}\partial \partial n$  (JP, AG). Capilano River. According to AC, the name is from  $m\delta l\partial c$  'horsefly,' because these were numerous there. AG

believed the root to be  $mal\dot{c}$  'roll,' which appears in  $m\delta l\dot{c}\theta at$  'roll (as salmon when spawning),' and the suffix  $-\theta an$  'mouth, lip, margin.' The name may thus mean 'horseflies at the mouth' or 'where they [presumably salmon] roll at the mouth.' According to Musqueam traditions, this was a Musqueam village site. Simon Pierre of Katzie gave the name of the Capilano River as  $x^w m\delta \theta k^w ay amat$  (probably more accurately transcribed  $x^w m\delta \theta k^w ay ama^2t$ ) meaning '[river] of the Musqueam.'

### Up the North Arm

(55)  $\check{s}x^w \partial m q^w \partial y \dot{e}m$  (AC) (<  $m \acute{a}q^w \partial m$  'bog'?). Celtic.

(56)  $\dot{c} \partial sn\dot{a}^{2}am$  (CC, AG),  $\dot{c} \partial sn\dot{a}^{2}\partial m$  (JP). Marpole. JP knew two stories about the people who lived here, people who spoke the same as the Musqueam. They were wiped out by smallpox.

(57)  $sk^w t \acute{e}x^w q \partial i$  (AG),  $sk^w t \acute{e}x^w q \partial n$  (AC) ('entering the throat,'  $< sk^w t \acute{e}x^w$  'inside,'  $-q \partial n$  'throat'). Wiggins Drift. A place where a creek runs into the Fraser near the site of the bridge to Lulu Island, so called in English from the name of a settler and the fact that fishermen made a drift from there down.

(58)  $\check{s}x^{w}\check{k}^{w}\partial l\check{\lambda}\acute{e}l\partial$  ('coupling place,' <  $\check{k}^{w}\acute{e}\check{\lambda}$  'copulate,' -*elo* '-place for'). A creek that flows into the North Arm, so called because it was used as a lovers' lane.

(59)  $\dot{c}\partial\dot{w}\dot{x}\dot{e}l\partial m\partial t$  (R) (< ?). The east end of Mitchell Island. Rozen identified this name as 'whistling (people),' but cf. (63).

### Islands between the North Arm and South Arm (Richmond)

(60)  $x^w \partial y \dot{e} y \partial t$  (AC),  $x^w i y \dot{e} \dot{y} \dot{t}$  (AG) (< ?). Iona Island.

(61)  $sq^ws \dot{a}\theta \partial n$  (AC, JP) ('sloping into the water, sunken shore,'  $< q^w \dot{a}s$  'enter the water,'  $-\dot{a}\theta \partial n$  'lip, margin'). IR #3, on the northwestern tip of Sea Island, sometimes used for the whole island.

(62) xicomálsom (R) (perhaps xcomálsom, < ?). A place on the western shore of Sea Island.

(63)  $x^w \delta \dot{y} q \partial \partial a$  (AG),  $x^w \delta \dot{y} q \partial \partial a$  (JP),  $x^w \delta \dot{y} q \partial \partial a$  (CC),  $w \delta \dot{y} q \partial \partial a$  (R) (means 'crook of land' [JP], < ?). A place on the southwestern tip of Sea Island, at the mouth of the Middle Arm. It is said that there were once people here who whistled when they talked. They came from the black ducks that whistle (the  $x \delta p \partial \partial \theta$  'American scoter,' <  $\sqrt{xep}$  'whistle'). They did not behave correctly when  $\dot{x} \dot{e} \dot{l} s$  came, and so disappeared (JP).

(64)  $s\dot{p}\delta l\partial k^{w}\partial qs$  (AG),  $sp\delta l\partial k^{w}\partial qs$  (AC, JP) ('boiling point' <  $\dot{p}\delta l\partial k^{w}$  'boil up,'  $-\partial qs$  '-nose, -point'). The northwestern tip of Lulu Island, at the mouth of the Middle Arm, site of the Terra Nova cannery. It was once a Musqueam village site, and continued to be used as a campsite. There were graves there.

(65)  $\partial nx^{w}ic\partial n'(R)$  (? <  $\partial indx^{w}$  'stop,' -*icon* 'back, surface'). On the western shore of Lulu Island, halfway between Terra Nova and Garry Point (R).

(66)  $\dot{q}^{w} \dot{e} y a^{2} \check{x}^{w}$  (JP, AG) (< ?). Garry Point. A summer campsite for some of the Musqueam people. JP's father stayed there and so took "Point" as his surname (JP).

(67)  $\dot{q}^w \delta y a \check{x}^w st \acute{a} l \partial w$  (AC). The main channel at Steveston.

(68)  $q^{w}t\dot{a}\dot{y}\partial m$  (R) (probably  $q^{w}t\dot{e}\dot{y}\partial m$  or  $x^{w}q^{w}t\dot{e}\dot{y}\partial m$ , 'driftwood place,' <  $q^{w}t\dot{e}\dot{y}$  'driftwood log'). Another campsite at the southwestern end of Lulu Island, east of Garry Point.

(69)  $\lambda aqatinas$  (AC, JP, AG) ('long shore,'  $< \lambda eqt$  'long,' -inas '-chest, beach'). On the south shore of Lulu Island, above where Ladner ferry was (AC), Woodward's Landing (AG). This was the great summer village of the Cowichan-speaking people of Vancouver Island.

### Westham Island and South Shore of South Arm

(70)  $^{2} \delta l eq s \partial n$  (JP, R) ('points'?  $< s^{2} \delta l q s \partial n$  'point'; cf.  $^{2} \delta l q s \partial n$  'Point Grey'). The northwestern point of Westham Island (JP), the south end of Westham Island (R).

(71)  $x^{w}ti\hat{c}\partial m$  (AG, R),  $x^{w}tic\partial m$  (JP) ('cut inside out,' <  $x^{w}$ - 'inside,'  $\sqrt{ti\hat{c}}$  'cut'). Canoe Pass. According to AG, the name comes from the practice of cutting rushes to let salmon pass through.

(72)  $s\dot{c}\delta l \partial x^w q \partial n'$  (JP, AG, R) ( $\leq \dot{c}\delta l \partial x^w$  'go upstream,'  $-q \partial n'$  'throat'). Ladner.

(73)  $p \partial t \check{x} \partial n \acute{e}m \partial x$  (JP, AG) ('meadow land'? cf.  $s p \partial t \check{x} \partial n$  'prairie, meadow,' - $\acute{e}m \partial x$  '-country'). An area a little above Ladner. AG translated the name as 'meadow flat.' Rozen recorded  $p \partial t \check{x} \partial n \acute{e} w m \partial x^w$  'meadow land' as the name for the western end of Deas Island and simply  $s p \partial t \check{x} \partial n$  'meadow' for "The Delta Area."

(74)  $x^{w}m\acute{e}\acute{c}\partial n\partial t p$  (JP) ('black haw trees,' <  $m\acute{e}\acute{c}\partial n$  'black haw fruit'). The high land from Sunbury down to Mud Bay.

(75)  $q^{w} \partial q^{w} \partial^{2} a p \partial t p$  (AC) ('crabapple trees'). The site of St. Mungo's Cannery.

(76)  $s \partial w \dot{q}^w \dot{e} q s \partial n$  (R) ('drifting away point,'  $< h \partial w \dot{q}^w$  'be drifting with current,'  $-\partial q s \partial n$  'nose, point'). The southwestern side of Annacis Island.

### At and Above the Forks

(77)  $y \delta l \partial t k^{w} \partial r$ ,  $y \delta l \partial t k^{w} \partial r$  (R) (< ?). The western end of "Tree Island" (Poplar Island?).

(78) *sk<sup>w</sup>téx<sup>w</sup>qən* (R). The eastern end of "Tree Island"; but cf. (57).

(79)  $w \delta \dot{q}^w a \check{x} \partial n s \check{x}^w \dot{a} y m \partial t$  (AC). Usual designation for IR below New Westminster.

(80)  $s \check{x}^w \partial y \acute{em}$  (AC) ('story, myth'). A rock on the western side of the Patullo Bridge.

(81)  $s\check{x}^{w}\acute{e}yamat$  (AG) (<  $s\check{x}^{w}ay\acute{e}m$  'myth'). New Westminster.

(82) qiqéyt (JP, AG), qəyqəyt (AC). South Westminster IR.

### Place Names and Tribal Names

These names for peoples, marked (tr), are also place names in the sense that they may be used to refer to the home territories of the peoples they name.

```
To the east:
```

 $k^{wi}k^{w}\partial \dot{X}\partial m$  Coquitlam (tr),  $sk^{w}\dot{X}\partial ma\cdot i$  ( $\dot{X}\dot{a}xa^{2}$ ) (the lake) belonging to the Coquitlam (JP).

 $\dot{q}i\dot{c}\partial\dot{y}$  ('moss') Katzie (tr).

 $\dot{q}^{w}\dot{a}^{2}a\dot{n}\dot{\lambda}\partial\dot{l}$  'Langley' (Kwantlen) (tr).

 $m \hat{e} \theta x^w \partial \dot{y}$  Matsqui (tr).

 $s \partial m \acute{e} \theta$  Sumas (tr).

 $s\dot{c}\partial lx^{w}\dot{i}^{2}q^{w}$  (<  $\dot{c}\dot{\partial}l\partial x^{w}$  'go upstream,'  $-i^{2}q^{w}$  'head') Chilliwack (tr).

 $s\dot{q} \partial w\dot{q} in (\sqrt{\dot{q}} iw \text{ 'go round a point,' } -q \partial n \sim -q in \text{ 'head'})$  Skulkayn IR.

 $s\dot{q}\dot{a}\dot{w}n\partial c$  (<  $\sqrt{\dot{q}}iw$  'go round a point,'  $-n\partial c$  'butt bay') Scowlitz, 'Harrison Mills' (tr).

scopinos (< cep 'land on,' -inos 'chest, beach') Chehalis (tr).

 $s\theta q^{w} \acute{a}t$  (< ?) Douglas and other Lillooet (tr).

 $x^ws\acute{e}^2eq$  (<  $s\acute{e}^2eq$  'bracken root') Nooksack (tr),  $\check{s}x^ws\acute{e}^2eqa^2l$  smənmént 'Nooksack Mountains,' i.e., the North Cascades.

 $\check{x}e\dot{t}^{\theta}enax^{w}$  (<  $\check{\chi}\check{x}e\dot{t}^{\theta}$  'measure, mark,' -enax<sup>w</sup> 'season, fish') Mount Baker.

 $l \dot{a} \dot{q} a t e^{2}$  (< Klickitat?) Plateau tribes, described by JP as a "belt" of people from "our  $l \dot{a} \dot{q} q a n$ " (the Upper Fraser River) to Yakima; according to AG, the term refers to people with braids, cowboy hats, and teepees (tr).

### To the north:

 $sq^w \check{x}^w \acute{a} \acute{m} \partial x$  (AC),  $sq\check{x}^w \acute{a} \acute{m} \partial x$  (DK) Squamish (tr),  $sq^w \check{x}^w \acute{a} \acute{m} \partial x a^2 t$  (AC, JP) ('Squamish' +  $-a^2 t$  'attributive') Howe Sound, especially the eastern shore, where Musqueam hunters went for mountain goats (also the Squamish-style canoe).

 $\check{x}^{w}t\acute{e}\cdot k^{w}i \ l\partial t\acute{a}lw\partial t-ct \ smanme:nt$  'mountains behind us,' i.e., the Coast Mountains north and east of Vancouver.

sxi<sup>2</sup>xé<sup>2</sup>t (JP) Sechelt (tr).

*čiyátəlič* (JP) Skookumchuck Narrows, in Sechelt country.

sxwə<sup>2</sup>íti Lasquiti Island (AG), Texada Island (JP).

*ċí<sup>2</sup>ċsəm* (dim. of *ċíċsəm* 'be growing') Texada Island (AG).

 $\dot{q}^{w} \delta w m \partial x^{w} s$  (AG) The Comox (tr).

 $y \delta q^w \partial t e^2 \check{x}$  (JP) (< Kwak'wala  $li G^w i t d a^2 \check{x}^w$ , probably via Comox, which has /y/ for /l/) Lekwiltok (the people now at Campbell River and Cape Mudge) (tr).

 $k^{w} \acute{e} k^{w} e^{2t}$  (AG) (< Kwak'wala  $k^{w} \acute{a} g^{y} ut$ ) The Kwakiutl, the people of Queen Charlotte Strait (tr).<sup>2</sup>

<sup>2</sup> A form like k<sup>w</sup>ák<sup>w</sup>∂t appears not only in Halkomelem and I believe in other Salishan languages but also in Heiltsuk, and it evidently refers to more than simply the people who settled at Fort Rupert, whom Boas identified as k<sup>w</sup>ág<sup>y</sup>ut.

 $x^{w}t\delta y awal$  (AC) (<  $\sqrt{t}ay$  'upstream'; cf.  $t\delta yt$  'upstream,'  $t\delta yal$  'go upstream,'  $t\delta y wat$  'west wind,' 'north') The northern peoples (Comox and Cape Mudge northward [AC]).

### To the west:

 $\delta x^w n \partial w (la^2 l sm \partial m m e'nt$  'far-side mountains,' i.e., the mountains on Vancouver Island.

 $x^{w}k^{w}alax^{w}am$  (<  $k^{w}alax^{w}$  'chum salmon') Qualicum.

 $sn \partial w n \partial w \partial s$  (AC) ('looking inward,'  $< \sqrt{n \partial w}$ - 'enter,'  $-\partial s$  '-face') Nanoose (tr).  $sn \partial n \dot{e} y m \partial x^{w}$  (AC) (< ?, probably not  $< n \dot{e} y \partial m$  'laugh') Nanaimo (tr).

 $i \acute{t} \circ i \partial q e^{\gamma}$  (AC) Valdez Island (tr).

 $p \partial n \hat{e} \hat{l} \partial \tilde{x} \partial \hat{t}^{\theta}$  (AC) Kuper Island (tr).

*xcomínos* (AC) Yellow Point ('Chemainus Bay,' the Kulleet Bay of charts) (tr).  $x^{w}c\delta s \delta y$  (AC) Westholm (tr).

 $q \partial wic \partial n$  ('warm the back,'  $< \sqrt{q} \partial w$  'warm,' -*icon* '-back') (AC) Cowichan (tr). This name, or rather its Cowichan form  $q \partial w \partial c \partial n$ , was identified by Cowichan speakers as that of Mt. Tzouhalem (named by the Whites after a famous warrior) because of its bare side facing the sun. Cowichan villages (identified by AC and CC) are:  $s^2 d m \partial n e^2$  Duncan,  $k^w d m \partial c \partial n$  Quamichan,  $x^w \partial l q^w s \ell l \partial$  Koksilah,  $q^w \partial m \partial y \ell q \partial n$  Komiaken,  $\tilde{x}(n \partial p s \partial m$  Kanipsim,  $l \partial m \ell d c$  Clemclemalitz.

s?áməne? žáce Cowichan Lake.

*xsénəc* (CC) Saanich (tr). Villages are: *méləžəł* Malahat, actually a Cowichan-speaking reserve, but it is generally classified as Saanich, *pákwəcən* Cole Bay,  $x^ws\dot{s}\dot{y}\dot{q}\partial m$  Patricia Bay,  $st^{\theta}\acute{e}wtx^w$  'East Camp.'

sť<sup>θ</sup>áməs Victoria (also mətúliə).

*ləkwámən* Victoria people (tr).

sá?ak<sup>w</sup> Sooke (tr).

*məθéləməx*<sup>w</sup> West Coast (tr).

To the south:

scəwáθən (JP) Tsawwassen (tr).

smáq<sup>w</sup> *oc* (JP) Point Roberts.

 $x^{w}l \delta m \partial \dot{y}$  (JP) Lummi (tr).

*swélə* x (JP) Orcas Island.

 $x^w y i x \partial l$  (CC) the people from LaConner south, i.e., the speakers of Lushootseed (tr).  $\check{s} x^w y i x \partial l a^2 l$  smanmént the Cascades east of Puget Sound (JP).

 $\delta x^{w} \lambda e^{l} \partial m$  Clallam (tr),  $\delta x^{w} \lambda e^{l} \partial ma^{2} t$  smonme nt 'Clallam mountains,' i.e., the Olympic Mountains (JP).

# Appendix 3 A History of Work on Halkomelem

#### **Early Vocabularies**

Probably the earliest published example of Halkomelem is a short Cowichan word list, one of a number of Northwest Coast vocabularies collected by W.F. Tolmie, probably in the late 1830s, and appended by John Scouler to "Observations on the Indigenous Tribes of the N.W. Coast of America" (Scouler 1841). The words are poorly recorded and some are wrongly identified, no doubt because of the difficulties Tolmie must have had eliciting them through the recently introduced Chinook Jargon.

From the late 1850s, we have lists in the Cowichan and Chilliwack dialects collected by Charles Wilson and published in what is the earliest ethnographic article on the Halkomelem area (Wilson 1866), and we have lists in the Nanaimo and Tait dialects (about 1,100 words) collected by George Gibbs and published in his "Comparative Vocabularies" (Gibbs et al. 1877, 270-83). From the 1870s, we have word lists collected by Tolmie and G.M. Dawson and included in their Comparative Vocabularies of the Indian Tribes of British Columbia (1884, 38B-49B). These later vocabularies were recorded with somewhat greater accuracy in the identification of the words but without much improvement in transcription. If Tolmie and Dawson heard the difference between glottalized (ejective) and unglottalized plosives or between velars and uvulars, or had sorted out the laterals and the back fricatives, they give no indication of it in their spellings. The same symbol (letter or sequence of letters) may represent several sounds and the same sound may be represented by several symbols, and their words are generously sprinkled with gratuitous hyphens.

In 1882 the Roman Catholic missionary priest Fr. G.C. Donckele collected a vocabulary, now in the British Columbia Provincial Archives (Donckele 1882), of "la langue sauvage Cowichan" to complete a printed form evidently designed for use in French Oceania. It contains some interesting items but it is no better in the way it represents the sounds of the language. Fr. Donckele may

have left other manuscripts on Halkomelem. I was told that he tried preaching in Halkomelem rather than the usual Chinook Jargon but was not very fluent. A Protestant missionary, the Reverend C.M. Tate, is also said to have learned Halkomelem well enough to translate hymns (see Pilling 1893, 67, where Halkomelem appears as "Ankomelum"). I was told that he could indeed communicate in the native language but amused people by mixing dialectal forms. The Reverend Thomas Crosby may also have spoken some Halkomelem (which he wrote "An-ko-me-num"), but the few words given in his book (Crosby 1907) suggest that he too did not make the distinctions that the Native phonological system requires. Anything recorded by these and other nineteenth-century missionaries would certainly be useful in establishing the times and places when particular words were in use, but given the inadequacy of systems of transcription then available, it seems unlikely that they could contribute much to our understanding of Halkomelem grammar.

"Comparative vocabularies" such as those of Gibbs and Tolmie and Dawson did serve to suggest relationships among languages and had allowed Hale and Gallatin to propose a "Salish" or "Selish" family (Powell 1891, 102-5). But they were certainly not adequate for any more refined taxonomy. Even in the 1890s, it seems that the various samples of Salishan all stood coordinate, with dialects of the same language given equal status with separate languages. Powell (1891, 104-5) lists as Salishan sixty-four "principal tribes," of which six (Cowichan, Kwantlen, Nanaimo, Nanoose, Sumas, and Tait) are names for groups of Halkomelem speakers. Pilling (1893) lists as Salishan "languages" five names (Kawichan, Kwantlen, Snanaimuk, Stalo, and Tait) that actually designate groups of Halkomelem speakers.

### Boas

In 1886, just two years after the publication of Tolmie and Dawson's *Comparative Vocabularies*, a profound change occurred. Franz Boas, then a young man, made his first trip to British Columbia, where he met Tolmie, then an old man soon to die. Later Boas wrote home that he had managed in a few weeks to learn more about the Indians than Tolmie had in over fifty years on the Northwest Coast (Rohner 1969, 69). With respect to language, this extravagant statement was probably correct. In recording Native words and in discovering linguistic relations, Boas had moved far ahead.

Boas's earliest transcriptions are certainly not wholly accurate (nor perhaps are anyone's), but his earliest publications (e.g., Boas 1888) show that he was aware of the opposition of glottalized and unglottalized plosives and of velars and uvulars and was using an orthography that permitted him to make most of the distinctions that are phonemic, that is, significant to speakers of the Native languages. His principal faults were that he did not record glottal stops after vowels or distinguish labialized back consonants in all environments. During the next few years he corrected these faults, so that by the time he was producing the reports of the Jesup North Pacific Expedition, around 1900, his transcriptions of Kwakiutl were meticulously accurate phonetically. Unfortunately, however, he did not continue work on the Salishan languages of the Strait of Georgia into that period, and so we have only his earlier transcriptions for Halkomelem and neighbouring languages, far superior to what came before and to much of what was done later but still not altogether accurate.

Boas's classification of the Coast Salish of Georgia Strait was also far superior to any before it. In 1887 he published an article (Boas 1887) in German that accurately distinguished the languages and presented a map that has not been excelled (and has been almost wholly ignored). The taxonomy was modified only slightly when presented again in later works (Boas 1890, 805-6; 1897, 320-21). As a statement of what forms of speech were spoken where, it needs no correction today.

Unfortunately, however, Boas always referred to the languages of the region as "dialects," using that word simply in the sense of "related forms of speech" (Thompson 1973, 992), while in recent decades the word has been used for mutually intelligible forms of speech (e.g., Sapir 1931), forms not mutually intelligible being called "languages." Boas almost certainly knew that his "dialects" such as "Sqxō'mic" (Squamish), "Qau'etcin" (Cowichan, i.e., Halkomelem), and "Lku'ñgEn" (Songhees, i.e., Northern Straits) were not intelligible; that they "diverge unusually strongly from one another" (weichen ausserordentlich stark voneinander ab). Some later anthropologists (e.g., Barnett 1942, 380), however, seem not to have understood Boas's usage and to have supposed that all of the "Coast Salish" speak dialects of a single Coast Salish language.

Boas is also the author of the first published material on Halkomelem grammar, less than three pages on the Nanaimo dialect, presented as one of a number of sketches published in his report to the British Association for the Advancement of Science (Boas 1891, 680-83). Short though it was, it showed that roots undergo internal modification for plural, diminutive, and progressive forms; it identified some of the affixes and auxiliaries expressing tense and mode and some of the pronominal and demonstrative elements; and it noted the categories of gender, presence versus absence, and visibility versus invisibility.

Boas seems to have done no further work with Halkomelem. He left a few pages of manuscript.

### Hill-Tout

Boas was succeeded among the Coast Salish of British Columbia by Charles Hill-Tout, a local scholar who worked in archeology, ethnography, and linguistics. Among his many publications are three (Hill-Tout 1903, 1904, 1907) that present Halkomelem material. These contain: (1) notes on the grammar of the

Chilliwack dialect, with a glossary of about a thousand words (1902, 369-400); (2) notes on the Kwantlen dialect, with a short vocabulary and two texts (1902, 415-41); (3) notes on the Chehalis ("StsEélits) dialect, with one text (1904, 334-39); (4) five texts in the Scowlitz ("Sk aúlits") dialect (1904, 368-76); and (5) two texts in the Cowichan dialect (1907, 366-72). The whole, including translations of the texts, amounts to a total of about seventy-five pages, which was fairly respectable for the time, being more than the shortest sketches that appear in the *Handbook of American Indian Languages* (Boas 1911). It has serious defects, however.

Phonologically, Hill-Tout's work is retrogressive. It suffers from the same faults as Boas's of the 1880s – the failure to record glottal stops after vowels and with resonants and to record labialization of back consonants except before vowels – while it also suffers from Hill-Tout's own failure to distinguish glottalized plosives, which Boas did from his first work. (Hill-Tout's raised commas simply separate consonant clusters, presumably to mark the absence of a vowel where English speakers might expect one.) Moreover, Hill-Tout was often inaccurate in distinguishing dentals from palatals, uvulars from velars, and even back fricatives from stops. Hill-Tout refers to Boas's earlier work and must have been influenced by it, but he met Boas only once and clearly did not get much encouragement from him (Maud 1978, 1:15, 4:9). It seems that Hill-Tout did not fully grasp principles that Boas had discovered, and was unable to move ahead on his own.

Hill-Tout did go beyond Boas in his exploration of Halkomelem grammar, although not in the direction Boas would have gone. Rather than seeking to describe the language in its own terms, Hill-Tout pretty well kept to the categories of traditional grammar. And rather than analyzing forms and identifying sets of morphemes, Hill-Tout simply gave extensive paradigms. He did not account for all of the morphemes in his material; for example, subordinate subject pronominal suffixes occur in his paradigms, but he nowhere identified them and listed them parallel to their coordinate counterparts. Perhaps because of the defects in his transcription he could not have accounted for everything. One verb root, which he does identify, occurs in five different forms – not in reality but in Hill-Tout's inconsistent spelling.

Nevertheless, Hill-Tout's work is useful. It attests to the existence of words and features of grammar at the times when and places where he recorded them. His texts were, I believe, recorded carefully – within the limits of his capacity to transcribe the language. I found it possible, working with James Point, to reconstruct one of Hill-Tout's Kwantlen texts. Mr. Point found only a few places where he might have used a different term or construction.

### **Twentieth-Century Ethnographers**

For half a century after Hill-Tout's work, Halkomelem was almost wholly neglected. Hill-Tout's last publication reporting work on the language appeared in
1907; the next publication reporting work on it was that of Elmendorf and Suttles in 1960.

During this time, a few ethnographers worked in the Halkomelem area – Curtis and his associates around 1912, Barnett and Jenness in the mid-1930s, Marian Smith and her students in the mid-1940s, Lane in the late 1940s, Duff in 1949 and 1950. Most recorded Native terms deemed culturally significant as well as proper names, but the quality of their transcriptions is quite variable. Terms in Curtis (1913) are spelled in a system that is not wholly adequate to the task but that is at least used consistently and (within its limits) fairly accurately. Those appearing in Barnett (1955) are more in the tradition of Tolmie and Dawson. Those appearing in Jenness (1934-35), Smith (1949), and Duff (1952) are somewhat better but still sometimes quite inconsistent. All are, of course, useful in documenting the existence of the terms but are not reliable for any study that requires phonetic accuracy.

## This Study

My own work on Halkomelem grew rather gradually out of the practice of ethnography and the conviction, dating back to student days, that ethnography ought to be based on more than "a smattering of knowledge" of the Native language (Boas 1911, 60). As a student at the University of Washington, I had had some experience with the Northern Straits language working on Northern Straits ethnography for my dissertation (Suttles 1951). I first recorded some Halkomelem in 1952, while teaching at the University of British Columbia, when I undertook the task of re-eliciting from Simon Pierre of Katzie (see Suttles 1955) the Native terms recorded by Diamond Jenness in the 1930s in the course of doing ethnographic work with Simon's father, Peter Pierre – the "Old Pierre" of *The Faith of a Coast Salish Indian* (Jenness 1955). I did not proceed very far linguistically with Simon because, while he was a pleasure to work with on ethnography, his lack of interest in language (to say nothing of his lack of teeth) made linguistic work difficult. There are certainly errors in my transcription of Katzie words.

The opportunity to work with people more suitable for my purposes came in 1957, when Harry Hawthorn suggested that I continue ethnographic work that he had begun with Andrew and Christine Charles at Musqueam, which is very near the university. Mr. Charles was an excellent source of traditional knowledge, and Mrs. Charles proved to be an ideal teacher of her language, patient with my ignorance and yet intolerant of my mistakes.

During 1957 I elicited a fairly extensive vocabulary, reviewing what I had gotten from Simon Pierre, getting Halkomelem equivalents of Northern Straits terms I had recorded still earlier, and adding many new items. I also got some grammatical material and a good deal of ethnography.

During the summer of 1958, William W. Elmendorf was at the university and joined me in working with the Charleses. We re-elicited much of the vocabulary that I had recorded the previous year, expanded on it in some areas, and matched the Musqueam with forms in Cowichan, from Mr. Charles, and Chilliwack, from Mrs. Charles and several people we interviewed briefly in the Chilliwack area. The result of this collaboration was an article (Elmendorf and Suttles 1960) in which we showed something of the phonological and lexical differences among these three dialects and discussed apparent evidence for recent change in the use of the articles. In particular, we showed that the three dialects seem to differ lexically to different degrees in different semantic categories.

I continued to work with the Charleses, for a few days only in 1959 and early 1960, then more frequently during the fall of 1960, when Mrs. Charles dictated nine texts of ethnographic and historical interest. I had hoped to continue this work during my sabbatical year of 1961-62 but Mr. Charles died during the summer of 1961 and Mrs. Charles, understandably, did not want to continue immediately with the work that we had all enjoyed so much together. And so during that year I collected materials, mainly on tape, in the Cowichan and Tait dialects, wrote a brief sketch of Musqueam grammar (essentially the first draft of the present work), and mainly observed and studied the winter ceremonialism, which had been developing and expanding for some years.

During the academic year 1962-63, I worked again briefly with Mrs. Charles, who dictated three more texts. In September 1962, I also began with James Point of Musqueam. Mr. Point was born in 1881. Although older than the Charleses, he was less traditional in his outlook. Nevertheless, he had a good knowledge of myths and, like Mrs. Charles, he was a patient and conscientious teacher. I worked with him often during 1962-63, and by July 1963 he had dictated twenty-seven texts totalling just 300 handwritten pages.

During the summer of 1963, I left the University of British Columbia to teach at the University of Nevada at Reno, where I taught for three years before going, in 1966, to Portland State University. While at Nevada, I was able to spend one short period at Musqueam, in 1965, working on problems in morphology with both Mrs. Charles and Mr. Point and recording one more text from Mr. Point.

After I went to Portland State, my work with Halkomelem had its productive and unproductive periods. During the summer of 1967, I worked again very briefly with Mrs. Charles (for the last time), Mr. Point, and Mrs. Della Kew, the daughter of Andrew and Christine Charles. During the academic year 1968-69, I experimented with teaching Halkomelem to university students. For the class, I ran off ditto copies of several texts and of grammatical notes, which might be seen as the second draft of the present work. During the summer of 1969, one of the students, Bonnie McCay, was able to spend a month at Musqueam eliciting and tape-recording materials from Mr. Point and also several people with whom I had not worked, among them Ed Brown and Steven August. In July 1971, I worked again with Mr. Point, mainly going through my lexical file, rechecking, and filling out paradigms. We worked very hard. He was then 90 but insisted on putting in around six hours a day with only an occasional day off. This was our last extended work together, although I consulted him again briefly in 1973. He died in 1979 at the age of 98 (older according to some). In the fall of 1973, Della Kew visited Portland, and we worked through her mother's texts, checking transcription and translation.

In June 1975, Dr. Barbara Efrat, then Curator of Linguistics at the British Columbia Provincial Museum, invited me to write a reference grammar of Halkomelem for a series of publications that the museum was planning. I accepted the invitation, but a number of other obligations slowed down the work. I did, however, spend several short periods between 1979 and 1983 working with Arnold Guerin, and worked again briefly with Della Kew in 1979. I finally turned in the last portions of the grammar in January 1985. Soon afterward, however, the museum abandoned the planned series, and this version of the grammar was never published. Because I had other work to do, I did nothing further with it for several years.

I had typed the version I submitted to the Provincial Museum (now the Royal British Columbia Museum) on an IBM Selectric typewriter. This version has been consulted by several linguists. When I went back to it, however, I felt that it had to be revised and redone on a computer. I had sent a copy of the typed version to Brent Galloway; he had it entered into a computer, using a font that he had devised for Upriver Halkomelem, and he sent it to me on disks in June 1993. By this time I was using a computer, but in spite of heroic efforts on Brent's part, I never found it possible to use his font. However, the format and the English were there and available, so in the end I simply retyped the phonetic symbols using the Straight font devised by Charles Ulrich. The present version differs from the typed version principally in formatting, but I have added more examples and made several changes in terminology and in the interpretation of the data.

## **Recent Linguistic Work**

Since the mid-1960s, a good deal of work has appeared on Halkomelem, especially on the Upriver and Island dialects. For Upriver we have a vocabulary collected by a non-linguist, Oliver Wells (1965), a master's thesis on phonology by Jimmy Harris (1966), and a doctoral dissertation and a series of articles describing the grammar and a number of semantic domains by Brent Galloway (1971, 1973). For Island Halkomelem we have a master's thesis on Cowichan phonology by Tiiu Kava (1969), a master's thesis on the Cowichan aspect system by M.K. Jones (1976), a doctoral dissertation on Cowichan grammar by Adrian Leslie (1979), a series of articles on grammatical problems by Thomas Hukari beginning in the mid-1970s, a text in Cowichan by Hukari, Ruby Peters, and Ellen White (1977), a Cowichan dictionary by Hukari and Peters (1995), and a doctoral dissertation by Donna B. Gerdts (1981, 1988), author of a series of articles on problems in syntax. Gerdts's master's thesis (1977) was a survey of lexical differences among a number of dialects. Nancy C. Turner and M.A.M. Bell (1971) assembled ethnobotanical data that include Island Halkomelem plant names. David Rozen (1977a, 1977b, 1978a, 1978b) gathered both Island and Downriver place names and Cowichan ethnozoological data and transcribed Catholic prayers and hymns.

For much recent work on Island dialects by Gerdts, Hukari, and Mercedes Hinkson, on Musqueam by Patricia Shaw, Susan Blake, and Jill Campbell, and on Upriver dialects by Galloway and Marina Wiltschko, see the bibliographies in Czaykowska-Higgins and Kinkade 1998 and Mithun 1999.

## Bibliography

- Barnett, Homer G. 1942. The southern extent of totem pole carving. *Pacific Northwest Quarterly* 33: 378-89.
- -. 1955. *The Coast Salish of British Columbia*. University of Oregon Monographic Studies in Anthropology 4. Eugene.
- Bates, Dawn, Thom Hess, and Vi Hilbert. 1994. *Lushootseed Dictionary*. Seattle: University of Washington Press.
- Boas, Franz. 1887. Zur ethnologie Britisch-Kolombiens. *Petermann's Geographische Mitteilungen* 33(5): 129-33.
- -. 1888. The Indians of British Columbia. *Transactions of the Royal Society of Canada for the Year 1888* 6, sec. 2, 47-57.
- -. 1889. Preliminary notes on the Indians of British Columbia. Pp. 236-42 in 58th Report of the British Association for the Advancement of Science for 1888. London.
- -. 1889. Notes on the Snanaimuq. American Anthropologist 2: 321-28.
- -. 1890. First general report on the Indians of British Columbia. Pp. 801-93 in 59th Report of the British Association for the Advancement of Science for 1889. London.
- -. 1891. Second general report on the Indians of British Columbia. Pp. 562-715 in 60th Report of the British Association for the Advancement of Science for 1890. London.
- -. 1894. Indian tribes of the Lower Fraser River. Pp. 454-63 in 64th Report of the British Association for the Advancement of Science for 1895. London.
- -.1897. The social organization and secret societies of the Kwakiutl Indians. Pp. 311-38 in *Report of the US National Museum for 1895*. Washington.
- -, ed. 1911. *Handbook of American Indian Languages, Part 1*. Bureau of American Ethnology Bulletin 40. Washington, DC.
- Boyd, R.T. 1999. *The Coming of the Spirit of Pestilence: Introduced Diseases and Population Decline among Northwest Coast Indians, 1774-1874.* Vancouver: UBC Press/Seattle: University of Washington Press.
- Canada, Department of Indian Affairs and Northern Development, Indian Affairs Branch. 1970. *Linguistic and Cultural Affiliations of Canadian Indian Bands*. Edited and compiled by G.W. Neville. Ottawa: Queen's Printer.
- Comrie, Bernard. 1976. Aspect: An Introduction to the Study of Verbal Aspect and Related Problems. Cambridge: Cambridge University Press.
- Crosby, Thomas. 1907. Among the An-ko-me-nums; or Flathead Tribes of Indians of the Pacific Coast. Toronto: William Briggs.

- Curtis, Edward S. 1907-30. *The North American Indian: Being a Series of Volumes Picturing and Describing the Indians of the United States and the Dominion of Canada.* Edited by Fredrick W. Hodge. 20 vols. Norwood, MA: Plimpton Press. (Reprinted: New York: Johnson Reprint, 1970)
- Czaykowska-Higgins, Ewa, and M. Dale Kinkade, eds. 1998. *Salish Languages and Linguistics: Theoretical and Descriptive Perspectives.* Trends in Linguistics: Studies and Monographs 107. Berlin: Mouton de Gruyer.
- Donckele, Fr. G.C. 1882. Vocabulaire de la langue Sauvage Cowichan, Ile Vancouver. Recuelli par Mr. Gustave Ch. Donckele, Prêtre Missionaire A Cowichan, Ile Vancouver. Colombie Britannique. Le 18 Octobre 1882. Archives of British Columbia F/8/D72.
- Duff, Wilson. 1952. *The Upper Stalo Indians of the Fraser Valley, British Columbia.* Anthropology in British Columbia, Memoirs 1. Victoria: British Columbia Provincial Museum.
- -. 1964. The Indian History of British Columbia. Vol. 1, Impact of the White Man. Anthropology in British Columbia, Memoirs 5. Victoria: British Columbia Provincial Museum.
- Elmendorf, W.W. 1960. *The Structure of Twana Culture*. Research Studies 28(3), Monographic Supplement 2. Pullman: Washington State University. (Reprinted 1992)
- -. 1961. System changes in Salish kinship terminologies. Southwestern Journal of Anthropology 17(4): 165-82.
- -. 1962. Relations of Oregon Salish as evidenced in numerical stems. *Anthropological Linguistics* 4: 1-16.
- Elmendorf, W.W., and W. Suttles. 1960. Pattern and change in Halkomelem Salish dialects. *Anthropological Linguistics* 2: 1-32.
- Galloway, Brent. 1977. A grammar of Chilliwack Halkomelem. PhD dissertation, University of California at Berkeley.
- –. 1993. A Grammar of Upriver Halkomelem. University of California Publications in Linguistics 96. Berkeley: University of California Press.
- Gerdts, Donna B. 1977. A dialect survey of Halkomelem Salish. MA thesis, Department of Linguistics, University of British Columbia.
- -. 1981. Object and absolutive in Halkomelem Salish. PhD dissertation, University of California at San Diego.
- -. 1988. Object and Absolutive in Halkomelem Salish. New York: Garland Publishing.
- -. 1999. Combinatory properties of Halkomelem lexical suffixes. Papers for the 34th International Conference of Salish and Neighboring Languages, Kamloops, BC, Secwepemc Cultural Education Society and Simon Fraser University.
- Gerdts, Donna, and Mercedes Hinkson. 1996. Salish lexical suffixes: A case of decategorialization. Pp. 163-76 in *Proceedings of the Conference on Conceptual Structure, Discourse, and Language*, edited by A. Goldberg. Stanford CA: Center for the Study of Language and Information.
- Gerdts, Donna B., Mercedes Q. Hinkson, and Thomas E. Hukari. 2002. Numeral classifiers in Halkomelem. Pp. 147-73 in *Papers for the 37th International Conference on Salish and Neighbouring Languages*. University of British Columbia Working Papers in Linguistics, edited by Carrie Gillon, Naomi Sawai, and Rachel Wojdak.
- Gibbs, George. 1863. *Alphabetical Vocabularies of the Clallam and Lummi*. New York: Cramoisy Press.
- 1877. Tribes of Western Washington and Northwestern Oregon. Contributions of North American Ethnology 1(2): 157-361. Washington, DC: US Geographical and Geological Survey of the Rocky Mountain Region.

- Gibbs, George, W.F. Tolmie, and G. Mengari. 1877. Comparative vocabularies. *Contributions of North American Ethnology* 1(2): 247-83. Edited by John Wesley Powell. Washington, DC: US Geographical and Geological Survey of the Rocky Mountain Region.
- Gleason, H.A. Jr. 1961. *An Introduction to Descriptive Linguistics*. Revised ed. New York: Holt, Rinehart and Winston.
- Goddard, Pliny E. 1934. *Indians of the Northwest Coast.* 2nd ed. Handbook Series 10. New York: American Museum of Natural History.
- Guerin, Arnold. n.d. *Musqueam Language, Books 1 and 2*. Musqueam: Musqueam Cultural Committee.
- Guerin, Arnold, and Jay Powell. 1975. Hundum?i?num?: Musqueam Language Book 1. Vancouver: Musqueam Band.
- Haeberlin, Herman. 1974. Distribution of the Salish substantival [lexical] suffixes. Edited by M. Terry Thompson. Anthropological Linguistics 16(6): 219-350.
- Harris, Jimmy G. 1966. The phonology of Chilliwack Halkomelem. MA thesis, University of Washington.
- Hess, Thomas M. 1976. Dictionary of Puget Salish. Seattle: University of Washington.
- -. 1979a. A comparison of marine and riverine orientation vocabulary in two Coast Salish languages. *Anthropological Linguistics* 21: 363-78.
- -. 1979b. Central Coast Salish words for *deer*: their wavelike distribution. *International Journal of American Linguistics* 45(1): 5-16.
- Hill-Tout, Charles. 1900. Notes on the Sk-qō'mic of British Columbia, a branch of the Great Salish Stock of North America. Pp. 472-549 (Appendix 2) in 70th Report of the British Association for the Advancement of Science for 1900. London.
- -. 1903. Ethnological studies of Mainland Halkōmē'lem, a division of the Salish of British Columbia. Pp. 355-449 in 72nd Report for the British Association for the Advancement of Science for 1902. London.
- -. 1904. Ethnological report on the Steē'lis and Sk·aúlits tribes of the Halkōme'lem division of the Salish of British Columbia. *Journal of the Anthropological Institute of Great Britain and Ireland* 34: 311-76.
- -. 1907. Report on the ethnology of the south-eastern tribes of Vancouver Island, British Columbia. *Journal of the Royal Anthropological Institute of Great Britain and Ireland* 37: 306-74.
- Hinkson, Mercedes Quesney. 1999. Salishan lexical suffixes: A study in the conceptualization of space. PhD dissertation, Department of Linguistics, Simon Fraser University.
- Hodge, Frederick W., ed. 1907-10. *Handbook of American Indians North of Mexico*. 2 vols. Bureau of American Ethnology 30. Washington, DC.
- Hukari, Thomas E. 1982. Conjunctive /so<sup>?</sup>/ in Cowichan. Pp. 107-17 in Working Papers for the 17th International Conference on Salish and Neighboring Languages, 1-11 August 1982, Portland, OR: Portland State University.
- Hukari, Thomas E., and Ruby Peter, eds. 1995. *The Cowichan Dictionary of the Hul'qumi'num' Dialect of the Cowichan People*. Cowichan Tribes.
- Hukari, Thomas E., Ruby Peter, and Ellen White. 1977. Halkomelem. Pp. 33-68 in *Northwest Coast Texts*, edited by Barry R. Carlson. Native American Text Series 2(3). Chicago: University of Chicago Press.
- Jacobs, Melville. 1936. *Texts in Chinook Jargon*. Publications in Anthropology 7(1): 1-27. Seattle: University of Washington.
- Jenness, Diamond. 1934. [Unpublished Coast Salish fieldnotes]. Manuscript No. 1103.6, Ethnology Division, National Museum of Civilization, Ottawa.

- -. 1934. [Unpublished field notes]. Manuscript in Canadian Ethnology Service Archives, National Museum of Civilization, Ottawa.
- -. 1934-35. The Saanich Indians of Vancouver Island. Manuscript No. VII-G-8M in Canadian Ethnology Service Archives, National Museum of Civilization, Ottawa.
- -. 1955. The Faith of Coast Salish Indian. Anthropology in British Columbia, Memoirs
  3. Victoria: British Columbia Provincial Museum.
- Jones, Michael K. 1978. Morphophonemic properties of the Cowichan actual aspect. MA thesis, Department of Linguistics, University of Victoria.
- Kava, Tiiu. 1969. A phonology of Cowichan. MA thesis, University of Victoria.
- Kennedy, Dorothy I.D. 1993. Looking for the tribe in the wrong places: An examination of the Central Coast Salish social network. MA thesis, University of Victoria.
- -. 2000. Threads to the past: The construction and transformation of kinship in the Coast Salish social network. D.Phil. thesis, Oxford University.
- Kew, J.E. Michael. 1970. Coast Salish ceremonial life: Status and identity in a modern village. PhD dissertation, Department of Anthropology, University of Washington.
- -. 1980. Sculpture and engraving of the Central Coast Salish Indians. Museum of Anthropology Notes 9. Vancouver: University of British Columbia.
- Kinkade, M. Dale. 1986. Blackcaps and Musqueam. Pp. 60-36 in *Papers for the 21st International Conference on Salishan Languages*, University of Washington, Seattle.
- -. 1998. Origins of Salishan lexical suffixes. Pp. 266-91 in Papers for the 33rd International Conference on Salish and Neighboring Languages, University of Washington, Seattle.
- Kuipers, Aert H. 1967. The Squamish Language: Grammar, Texts, Dictionary. The Hague: Mouton.
- -. 1969. The Squamish Language: Grammar, Texts, Dictionary. Part II. The Hague: Mouton.
- Lamb, W. Kaye, ed. 1980. *Simon Fraser: Letters and Journals, 1806-1808.* Toronto: Macmillan of Canada.
- Lane, Barbara S. 1953. A comparative and analytic study of some aspects of Northwest Coast religion. PhD dissertation, Department of Anthropology, University of Washington.
- Leslie, Adrian. 1979. A grammar of the Cowichan dialect of Halkomelem Salish. PhD dissertation in Linguistics, Department of Linguistics, University of Victoria.
- Levine, Robert D., and Freda Cooper. 1976. The suppression of BC languages: Filling in the gaps in the documentary record. *Sound Heritage* 4(3-4): 43-75.
- Lincoln, Neville J., and John C. Rath. 1980. North Wakashan Comparative Root List. National Museum of Man Mercury Series. Canadian Ethnology Service Paper No. 68. Ottawa.
- Maclachlan, Morag, ed. 1998. *The Fort Langley Journals, 1827-30.* Vancouver: UBC Press.
- Matthews, Maj. J.S., comp. 1955. Conversations with Khahtsalano, 1932-1954. Vancouver: Vancouver City Archives.
- Maud, Ralph, ed. 1978. *The Salish People: The Local Contributions of Charles Hill-Tout.* 4 vols. Vancouver: Talonbooks.
- Mithun, Marianne. 1999. Languages of Native North America. Cambridge University Press.
- Mooney, James. 1928. *The Aboriginal Population of America North of Mexico*. Smithsonian Miscellaneous Collections 80(7). Washington, DC: Smithsonian Institution.

Murdock, G.P. 1949. Social Structure. New York: Macmillan.

- Newcomb, Charles F. 1909. *Guide to the Anthropological Collection in the Provincial Museum*. Victoria: Richard Wolfenden.
- Pilling, James C. 1893. *Bibliography of the Salishan Languages*. Bureau of American Ethnology Bulletin 16. Washington, DC: Smithsonian Institution.
- Powell, John Wesley. 1891. Indian language families of America North of Mexico. Pp. 1-142 in *7th Annual Report of the Bureau of American Ethnology for the Years 1885-1886*. Washington.
- Quirk, Randolph, and Sidney Greenbaum. 1973. A Concise Grammar of Contemporary English. New York: Harcourt, Brace, Jovanovich.
- Rohner, Ronald P., ed. 1969. *The Ethnography of Franz Boas. Letters and Diaries of Franz Boas Written on the Northwest Coast from 1886 to 1931.* Chicago: University of Chicago Press.
- Rozen, David, ed. 1977. Hel<u>k</u>'emínem St'íwi7elh: Indian prayers of Southeastern Vancouver Island. Victoria?
- -. 1978. The Ethnozoology of the Cowichan Indian People of British Columbia. Vol. 1, Fish, Beach Foods, and Marine Mammals. n.p.
- 1979. Ethnographic sites in the Greater Vancouver Regional District. Report submitted to the Heritage Conservation Branch, Archaeology Division, Ministry of the Provincial Secretary, Victoria, BC.
- –. 1985. Place-names of the Island Halkomelem Indian people. BA thesis in Anthropology, University of British Columbia.
- Sapir, Edward. 1931. Dialect. Pp. 5123-26 in *Encyclopaedia of the Social Sciences*. New York: Macmillan. Reprinted as pp. 83-88 in *Selected Writings of Edward Sapir in Language, Culture, and Personality*, edited by David G. Mandelbaum. Berkeley and Los Angeles: University of California Press.
- Scouler, John. 1841. Observations on the indigenous tribes of the N.W. Coast of America. *Journal of the Royal Geographical Society of London* 11: 215-51.
- Shaw, Patricia A., Susan J. Blake, and Jill Campbell. 2002. Numerals and lexical suffixes in həndəmínəm. Pp. 293-334 in *Papers for the 37th International Conference on Salish and Neighbouring Languages*. University of British Columbia Working Papers in Linguistics, edited by Carrie Gillon, Naomi Sawai, and Rachel Wojdak.
- Smith, Marian W. 1949. *Indians of the Urban Northwest*. Columbia University Contributions to Anthropology 36. New York: Columbia University Press.
- Spier, Leslie. 1925. *The Distribution of Kinship Systems in North America*. Publications in Anthropology 1(2): 69-88. Seattle: University of Washington.
- Suttles, Wayne Prescott. 1951. Economic life of the Coast Salish of Haro and Rosario Straits. PhD dissertation, Department of Anthropology, University of Washington. (Printed without proper identification in *Coast Salish and Western Washington Indians I*. New York: Garland Publishing)
- -. 1955. Katzie Ethnographic Notes. Anthropology in British Columbia, Memoirs 2. Victoria: British Columbia Provincial Museum. (Reissued 1973)
- 1958. Private knowledge, morality, and social classes among the Coast Salish. American Anthropologist 60: 497-507. (Reprinted in Suttles 1987a)
- -. 1960. Affinal ties, subsistence, and prestige among the Coast Salish. *American Anthropologist* 62: 296-305. (Reprinted in Suttles 1987a)
- -. 1963. The persistence of intervillage ties among the Coast Salish. *Ethnology* 2: 512-25.

- -. 1965. Linguistic means for anthropological ends on the Northwest Coast. *Canadian Journal of Linguistics* 10(2-3): 156-66.
- -. 1968. Coping with abundance: subsistence on the Northwest Coast. Pp. 55-68 in *Man* the Hunter; edited by Richard B. Lee and Irven De Vore. Chicago: Aldine. (Reprinted in Suttles 1987a)
- 1977. The "Coast Salish" of the Georgia-Puget Basin another look. Puget Soundings (April): 22-25.
- 1987a. Coast Salish Essays. Vancouver: Talonbooks/Seattle: University of Washington Press.
- –. 1987b. Cultural diversity within the Coast Salish continuum. Pp. 243-49 in *Ethnicity* and Culture, edited by Reginald Auger, Margaret F. Glass, Scott McEachern, and Peter H. McCartney. Calgary: Archaeological Association, University of Calgary.
- -. 1987c. Four anthropological-linguistic notes and queries. Pp. 185-92 in *Papers from the 22nd International Conference on Salish and Neighboring Languages*, University of Victoria, 13-15 August 1987, edited by John Dunn. Norman, OK: University of Oklahoma, Department of Linguistics.
- -. 1990. Central Coast Salish. Pp. 265-87 in *Handbook of North American Indians*, vol.
  7. Washington, DC: Smithsonian Institution.
- -. 1991. The shed-roof house. Pp. 212-22 in A Time of Gathering: Native Heritage in Washington State, edited by Robin K. Wright. Thomas Burke Memorial Washington State Museum Monograph 7. Seattle: University of Washington Press.
- 1994. Review of A Grammar of Upriver Halkomelem by Brent D. Galloway. Anthropological Linguistics 36: 386-89.
- -. 1996. Linguistic evidence for Burrard Inlet as former Halkomelem territory. Pp. 283-96 in Papers for the 31st International Conference on Salish and Neighbouring Languages, University of British Columbia, 15-17 August 1996.
- 1998. The ethnographic significance of the Fort Langley Journals. Pp. 163-210 in *The* Fort Langley Journals, 1827-1830, edited by Morag Maclachlan. Vancouver: UBC Press.
- Suttles, Wayne, and William W. Elmendorf. 1963. Linguistic evidence for Salish prehistory. Pp. 40-52 in *Symposium on Language and Culture: Proceedings of the 1962 Annual Spring Meeting of the American Ethnological Society*, edited by Viola E. Garfield. Seattle.
- Teit, James A. 1900. *The Thompson Indians of British Columbia*. Edited by Franz Boas. Memoirs of the American Museum of Natural History 2, Anthropology 1(4); Publications of the Jesup North Pacific Expedition 1(4). New York.
- Thompson, Laurence C. 1973. The Northwest. Pp. 979-1045 in *Current Trends in Linguistics*, vol. 10: *Linguistics in North America*, edited by Thomas A. Sebeok. The Hague: Mouton.
- -. 1979. Salishan and the northwest. Pp. 692-765 in *The Languages of Native North America: Historical and Comparative Assessment*, edited by Lyle Campbell and Marianne Mithun. Austin: University of Texas Press.
- Thompson, Laurence C., and M. Dale Kinkade. 1990. Languages. Pp. 30-51 in *Handbook of North American Indians*, vol. 7. Washington, DC: Smithsonian Institution.
- Thompson, Laurence C., and M. Terry Thompson. 1971. Clallam: A preview. Pp. 251-294 in *Studies in American Indian Languages*, edited by Jesse Sawyer. University of California Publications in Linguistics 65. Berkeley: University of California Press.
- Tolmie, William F., and George M. Dawson. 1884. *Comparative Vocabularies of the Indian Tribes of British Columbia, with a Map Illustrating Distribution*. Montreal: Dawson Brothers.

- Turner, Nancy C., and Marcus A.M. Bell. 1971. The Ethnobotany of the Coast Salish Indians of Vancouver Island. *Economic Botany* 25(1): 63-104.
- Wells, Oliver. 1965. A Vocabulary of Native Words in the Halkomelem Language. Chilliwack, BC: Oliver and Casey Wells.
- White, Ellen, and Peter Wilson. 1975. syəθsstən? štən?ni?tən? čłhən?təl?. Pp. 155-71 in *Papers from the 10th International Conference on Salishan Languages*. Lektos, Special Issue. Louisville, KY: University of Louisville.
- Wilson, Charles W. 1866. Report on the Indian tribes inhabiting the country in the vicinity of the 49th parallel of north latitude. *Transactions of the Ethnological Society of London*, n.s., 4: 275-332.

## Index

Adjective-like forms, §3.8.3.4 Adjective-like words, §9.7 Adjectives, §3.8.3.1, §9 complex, §9.6 diminutive plurals of, §9.4 diminutives of, §9.3 exclamatory-interrogative, §20.2 plurals of, §9.2 shapes of the root, §9.1 Adjuncts, adverbial, §3.6 nominal, §3.4 verbal, §3.5 Adverbial adjuncts, §3.6 Adverbial predicates, §18.3 Adverbs, §18, §18.2, §3.2.2, §3.8.3.3 Affixes, §2.3 ordering of, §2.5 Alternation of /n/ and/l/, §1.5.9 of /c/ and /k w/, §1.5.10 of /x/ and x<sup>w</sup>/, §1.5.10 Alveolar consonants, §1.1.1 Animal talk, §1.6 Applicative suffixes, §10.4 Apposition, §3.8.5 Articles, §15.1 Aspect dispositional-iterative, §7.6 durative, §7.5 perfective, §7.2 progressive, §7.2 Aspectual prefixes, §11.1 Auxiliary verbs, §3.2.1

Baby talk, §1.6 Calendar, §22.2.8 Canoe, expression of motion and location in relation to, §22.1.4 Causative suffix, §10.3 Ceremonial circuit, §22.1.3 Clause-initial particles, §2.7 Cleft sentences, §4.1, §4.1.5.2 Coalescence of glottal stop and resonant, §1.5.5 Commands, indirect, §4.2.2 Compound sentences, §5 coordinate, §5.1 qualifying, §5.2 Compounds, §2.1, §12.3.3, §13.3 Conditional clauses, §4.2.1 Conjunction, §2.7 Consonants, §1.1 Coordination, §3.8.4 Day and night, expression of, §22.2.1 Demonstrative auxiliaries, §15.3 Demonstratives, §15.2 Derivational affixes, §12 Diminutive duratives, §7.5.2 Diminutive plurals of adjectives, §9.4 of nouns, §8.5 of resultative forms, §7.7.9 Diminutives of adjectives, §9.3 of dispositional-iterative forms, §7.6.3 of nouns, §8.4 of resultative forms, §7.7.8

of verbs, §7.4 Direction and location, general terms expressing, §22.1.8 Directions and winds, §22.1.5 Dispositional-iterative aspect, §7.6 Durative aspect, §7.5 Ejectives, §1.1.1 Ellipsis with the negative, §6.1.6 Emergent /0/, §1.5.7 Exclamations, §20.1 Exclamatory-interrogative adjectives, §20.2 Fire, expression of motion and location in relation to, §22.1.2 First-person plural imperatives, §3.2.4 Gender, §15.1.1 Glottal stop, §1.1.1 coalescence with resonant, §1.5.5 Glottalization, §1.1.1 Glottalized resonants, §1.1.2 Glottalized resonant disengagement, §1.5.4 Hortative forms, §3.2.4 House, expression of motion and location in relation to, §22.1.2 Inchoatives xwa-, §12.2.2 -əl, -il, §12.3.1 -0ət, §10.5.1 Indirect commands, §4.2.2 Indirect questions, §4.2.3 Interjections, §20.3 Interrogative words, §17 Intransitive suffixes, §10.2 Intrusive /h/, §1.5.8 Kinship terms, §21 Labial consonants, §1.1.1. Labialized uvular consonants, §1.1.1 Labialized velar consonants, §1.1.1 Lateral consonants, §1.1.1 Lexical prefixes, §12.4 Lexical suffixes, §13 Location and direction, general terms expressing, §22.1.8 Locative demonstratives, §15.2.2.1 Metathesis of non-labial resonant and schwa, §1.5.3 Modal predicates, §18.3

Modal suffixes, §11.2 Month, §22.2.4 Morphophonemics, §1.5 Necessity, expression of, §6.4.2.2 Negation, §6 with Pawa, §6.1 with ?ówəte?, §6.3 with x<sup>w</sup>əŵé, §6.2 Nominal adjuncts, §3.4 Nominalized clauses, §4.3 in negated predicates, §6.4.3 in negative constructions, §6.1.1, §6.1.4 with *šx<sup>w</sup>*-, §4.3.3 Nominalized narrative sentences, §4.3.4 Nominalizer s-1, §12.1.1 Noun diminutive plurals, §8.5 Noun diminutives, §8.4 Noun plurals, §8.3 Nouns inflection of, §8 resultative forms, §8.7 shapes of the root, §8.2 types of, §8.1 Numerals, §19, §3.8.3.2 ordinal, §19.3 Numerical classifiers, §19.2 Oblique article, §15.1.4 Oblique nominalizer  $\delta x^{w}$ -, §12.1.4 Oblique particle, §2.7 Obstruents, §1.1.1 'Or,' expression of, §6.4.2.3 Palatal consonants, §1.1.1 Paradigms of verbs, §7.8 Particles clause initial, §16.1 definition of, §2.7 following nouns, §3.8.2 second-position predicate, §3.1.2, §16.2 subordinating ( $w \partial_l$ - and  $? \partial l$ -), §4.2 third-person plural, §14.2.9 Permissive suffix, §10.7 Person markers, §14 subordinate subject, §4.2 Person system, §2.4, §14 Personal words, §14.2.7 Phonology, §1 Phonotactics, §1.4 Pitch, §1.3

Place names, §22.1.7, Appendix 2 Plosives, §1.1.1 Plural diminutive verbs, §7.4.1 Plural duratives, §7.5.1 Plural verbs, §7.3 Plurals of adjectives §9.2 of dispositional-iterative forms, §7.6.2 of nouns, §8.3 of resultative forms, §7.7.7 Position, as expressed by articles, §15.1.2 Possession, §3.8.1 Possessive affixes, §14.2.4 Possessive words, §14.2.8 Predicate heads, §3.1.1 Predicates, §3.1 active transitive, §3.3.2 adjectival, §3.7.1 intransitive, §3.3.1 nominal, §3.7.2 passive, §3.3.3 verbal, §3.3 Prepositions, see Verbal adjuncts, §3.5 Progressive aspect, §7.2 Progressives of dispositional-iterative forms, §7.6.1 Pronouns, see Person markers Pseudo-cleft sentences, §4.1.5.1 Questions, indirect, §4.2.3 Rapid speech, losses in, §1.5.1 Reciprocal suffix, §10.6 Reflexive suffixes, §10.5 Regional axis, §22.1.1.4 Relative clauses, §4.1 in negative constructions, §6.1.5 with negated predicates, §6.4.1 Resonants, §1.1.1, §1.1.2 loss of first of two identical, §1.5.6 Resultative forms of the verb, §7.7 Resultative prefix, §12.1.2 Roots, §2.2 internal modification of, §2.2 Schwa, §1.2 assimilation of, §1.5.2 loss of in rapid speech, §1.5.1 Seasons, §22.2.7 Sentences complex, §4 compound, §5

simple, §3 Space grammatical elements and terms expressing, §22.1 measurement of, §22.1.6 Spirants, §1.1.1 Stress, §1.2, §1.3 Subordinate clauses, §4.2 in negative constructions, §6.1.1, §6.1.2, §6.1.3 referring to the future, §4.2.4 with '>t- meaning 'kind of,' §4.2.6 Subordinate passive suffix, §10.8 Tags, sentence-final, §16.3 Texts, §23 Third-person demonstratives, §15.2.2.2 Tide, §22.2.5 Time by the clock, §22.2.2 expression of in grammar, §22.2 grammatical elements and terms expressing, §22.2 Transitive suffixes, §10.1 Uvular consonants, §1.1.1 Velar consonants, §1.1.1 Verb complements, §3.2.3 Verb forms as modifiers, §3.8.3.5 Verb plurals, §7.3 Verb roots canonical shapes, §7.1.1 inherent voice, §7.1.2 Verbal adjuncts, §3.5 Verbal predicates, types of, §3.3 Verbal suffixes, §12.3 Verbalizing prefixes, §12.2 Verbs derived from nouns, inflection of, §8.8 Verbs formed from adjective roots, §9.5 Verbs that act as nouns, §8.9 Verbs, diminutive forms, §7.4 Verbs, resultative forms, §7.7 Visibility, as expressed by articles, §15.1.2 Voice, §10 inherent in verb roots, §7.1.2 Vowel gradation, §1.5.11 Vowel length, §1.2.1 sources of, §8.6 Vowel mutation, §1.5.12

Vowels full, §1.2 Water, expression of motion and location in relation to, §22.1.1 Week, §22.2.3 Winds, §22.1.5 Word definition of, §2.1 classes of §2.1 Year, §22.2.6 Set in Times New Roman, Scala, Bernalillo, Straight and First Nations New by Artegraphica Design Co. Ltd.Printed and bound in Canada by FriesensCopy editor: Francis ChowCartographer: Cameron Suttles